







### THE POLITICAL HISTORY

#### OF THE

## UNITED STATES OF AMERICA,

#### DURING

# THE GREAT REBELLION,

#### FROM NOVEMBER 6, 1860, TO JULY 4, 1864;

#### INCLUDING

A CLASSIFIED SUMMARY OF THE LEGISLATION OF THE SECOND SESSION OF THE THIRTY-SIXTH CON-GRESS, THE THREE SESSIONS OF THE THIRTY-SEVENTH CONGRESS, THE FIRST SESSION OF THE THIRTY-EIGHTH CONGRESS, WITH THE VOTES THEREON,

AND THE IMPORTANT

EXECUTIVE, JUDICIAL, AND POLITICO-MILITARY FACTS OF THAT EVENTFUL PERIOD;

TOGETHER WITH THE

ORGANIZATION, LEGISLATION, AND GENERAL PROCEEDINGS

OF THE

**REBEL ADMINISTRATION.** 

BY EDWARD MCPHERSON,

OF GETTYSBURG, PENNSYLVANIA,

CLERK OF THE HOUSE OF REPRESENTATIVES OF THE U. S.

WASHINGTON, D. C.: **PUBLISHED** BY PHILP & SOLOMONS. NEW YORK: D. APPLETON & CO. 1864. Entered according to Act of Congress, in the year 1864, by

PHILP & SOLOMONS,

In the Clerk's Office of the District Court of the United States for the District of Columbia.

Stereotyped by MCGILL & WITHEROW, Washington, D. C.

### PREFACE.

THIS volume is intended to be a Record of the Legislation, and the general Political History of the United States, for the last four years—a period of unexampled activity and of singularly deep interest and importance, whether reference be had to the vast material interests involved in the stupendous struggle, or the precedents, principles, and measures which the Convulsion has produced. It is further intended to be a Record rather of those salient facts which embody or illustrate principles, than of those which relate to men or parties, and hence have transient and inferior significance.

So abundant have been the materials, that compression has been a necessity. Selection has been made with the purpose of presenting, fully and fairly, the facts as they are, and the agencies by which they came—viewing all else as subordinate.

The first Ninety pages are devoted to the period of Secession, and contain a narrative of the successive steps in the movement in each State, in chronological order; also, the elaborate justifying papers of the South Carolina Convention, with counter-selections from other authorities; together with a condensation of the various propositions of Adjustment made in or out of Congress and the vote upon each taken in either body, and the various Official Papers of the day tending to show the relations of the parties, the wrongs complained of, and the remedies proposed. Closely examining this Record, it is difficult for a candid person to escape the conviction that Adjustment was hopeless—Revolution being the predetermined purpose of the reckless men who had obtained control of the State machinery of most of the slaveholding States. This conviction will be strengthened by study of what has since transpired.

It will be remembered that the Thirty-Sixth Congress proposed permanently to settle the security of slavery in the slaveholding States by an amendment of the Constitution, which was adopted by a two-thirds vote in each House. And that it completely disposed of the Territorial feature of the difficulties by agreeing upon, and almost unanimously passing, bills organizing Territories covering the entire area owned by the Government. The record of these two important historical facts is given within. They have great significance in establishing the character of the Rebellion.

#### PREFACE.

The copy of the Constitution of the United States is believed to be strictly accurate in text and punctuation, which, it is understood, can be said of only one other copy in print—that in the work known as Hickey's Constitution. The statement of the differences between it and the Rebel Constitution has been made with extreme care. The common index to the two instruments shows, at a glance, wherein they differ, and will be found both interesting and convenient—the whole chapter possessing special value to large classes of persons.

In presenting the facts upon each subject of legislation, the general plan has been: first, to state the result reached, with the final votes; and, then, such proceedings, in the intermediate stages, as are of adequate importance, or necessary to explain the position of Members. This preparation involved constant selection, concerning which there may be differences of opinion—some thinking that too much detail on one subject is given; others, too little of another. In all cases the rule stated, governed. As far as it has been possible to obtain the Rebel legislation on the same or corresponding subjects it has been added, with such of their orders and proclamations as were connected with them. A comparison of the two, and the dates of enactment or issue, will prove of service in dispelling delusions and correcting general misconceptions.

Besides the legislation proper, the volume contains, in a classified form, all the Messages, Proclamations, Orders, Correspondence, and Addresses of the President; the Diplomacy of the Secretary of State; valuable letters and papers from the Secretaries of the Treasury, of War, of the Navy, of the Interior, and from the Postmaster General; Opinions of the Attorney General upon permanent public questions; those of the Orders of Commanding Officers which are within the scope of the work; the Decisions of the Courts; and such other data as properly belong therein—the whole forming a multitudinous mass of facts, to any one of which the classification adopted, and the copious index appended, will, it is hoped, make it easy to refer.

The votes by Yeas and Nays have been carefully compared with the Official Journals of Congress. In preparing these lists, the names of those persons have, for comparison's sake, been *italicized*, who were elected by, or were at the time generally co-operating with, the Democratic party. All others are in roman.

Under "Our Foreign Relations" will be found much of permanent value, as well as of current interest and dispute.

The chapter on the "Conspiracy of Disunion" contains several very interesting documents, chief of which are the extract from U. S. Senator Maclay's journal of 1789, recording, probably, the first threat of disunion uttered in Congress, and upon a subject which remained a matter of complaint in some quarters down to the period of Secession; and the Minutes of the Proceedings of the Police Commissioners of Baltimore in 1861, one of the most flagrant as well as one of the latest outbursts of treason. Other portions of this chapter will richly bear examination. I greatly regret that want of space has required the omission of many other facts, gathered from our political history, tending to reveal the true character of this foul conspiracy against Liberty, this crime against Humanity.

The lists of the organization of the Rebel "Provisional" and "Permanent" Government have been made up from every accessible source, and, though not complete, are more nearly so than any other yet published north of the Potomae, and as nearly so as present facilities afford. They are the result of careful and extensive examination. As a matter of interest, the names of those of the conspirators who were once members of the Congress of the Union have been put in *italic*.

This work was undertaken a few months ago without a realizing sense of the labor it involved. I can scarcely hope to have escaped errors, both of omission and commission, but have striven to make it fair, impartial, and truthful. It deals with the most momentous events of this Century, which will be studied while civil Government exists. I trust that the volume will be of service to those consulting it, and that its general effect will be to help strengthen the purpose of the American people to maintain their Unity, their Freedom, and their Power.

### EDWARD McPHERSON.

August 11, 1864.

Digitized by the Internet Archive in 2011 with funding from The Institute of Museum and Library Services through an Indiana State Library LSTA Grant

### TABLE OF CONTENTS.

Page.

1

2

#### THE ELECTORAL AND POPULAR VOTE FOR. PRESIDENT, IN 1860.....

DEVELOPMENT OF THE SECESSION MOVEMENT-Action of Conventions in South Carolina, Georarkansas, Torda, Louislana, Alabama, Arkansas, Torda, Louislana, Alabama, Arkansas, Torda, North Carolina, Tennesseo, Virginia, Kentucky, and Missouri-Insurreo-tionary Proceedings in Delaware and Maryland. tionary Proceedings in Delaware and Maryland. —Inter-State Commissioners—Organization of a "Southern Congress," and Provisional Gov-ernment—Address of South Carolina to the Slaveholding States, her Declaration of Inde-pendence, and Debates on them—Speech of: Alexander H. Stephens before the Georgia Legis-lature, Nov. 14, 1860—Extracts from Addressee by A. H. Stephens, July, 1859, and Jan., 1861; James H. Hammond, October, 1858; and R. M. T. Hunter, 1860—Extract from the Appeal for Recognition, by Yancey, Rost, and Mann, and Earl Russell's Reply—Seizure and Surrender of Public Property, from November 4, 1860 to March. 4, 1861—Changes in President Buchanar's Cab-inet—Correspondence between President Buch-4, 1861—Changes in President Buchanan's Cab-inet—Correspondence between President Buch-anan and the South Carolina "Commission-ers"—Demand for Surrender of Fort Sunter— Report on the Transfer of Arms to the South in 1859 and 1860—Davis's Bill for the Sale of Government Arms to the States—How the Tel-Government and this to battes—Intrigues for a Pacific, Republic—Mayor Wood's Message Recommend-ing that New York be made a Free City—"Per-sonal Liberty" Laws.

PROCEEDINGS OF THE GOVERNMENT IN RE-LATION TO THE ACTION OF THE INSURREC-TIONARY STATES .....

> Names of the Senators and Representatives of Atorney General Black's Opinion on the Powers of the President End Representatives of the President Buchanan's Last Annual Message-Attorney General Black's Opinion on the Powers of the President-The House Committee of Thirty-Three and their Proposition for Adjust-Thirty-Three and their Proposition for Adjust-ment, together with Abstracts of all other Propo-sitions, and votes thereon—Votes on Resolutions, respecting the "Personal Liberty" Laws, the Union, Major Anderson's Course, Coercion, Non-Interference with Slavery, and on the Bills to, Suppress Insurrection, and to provide for the Collection of Customs—Report of Committee, upon the Danger of the Capital, and Vote upon-Branch's Resolution to withdraw Troons fromupon the Danger of the Capital, and vote upon-Branch's Resolution to with/draw Troops from-the District of Columbia, with Secretary Holt's. Report—Disposition of the Navy, and Vote of Censure upon Secretary Toucey—Propositions, in Congress by Mason, Hunter, Clingman, Craigo, and others—Settlement of the Question of Slaw rown in the Townized. very in the Territories.

THE CONSTITUTION .....

Constitution of the United States-Points of Difference between It and the "Confederate" Constitution, with an Index to both—Speech of Alexander II. Stephens, expounding the "Con-federate" Constitution.

ADMINISTRATION OF ABRAHAM LINCOLN...... 105

President Lincoln's Inaugural Address—Secre-tary Seward and the "Confederate Commis-sioners," with Statements of Judge Campbell, and Thurlow Weed—The President's Reply to, the Virginia Delegation—Commencement of hos-

#### ADMINISTRATION OF ABRAHAM LINCOLN-Continued.

- Thirty-Eighth Congress-Annual Message 1861, War Power? called out-Call for 75,000 Men, and all subsequent Calls arranged in Chronological Order-Mational Legislation on Military Affairs-"Confederate" Legislation and Proclamations and Orders-The Thirty-Seventh Congress-President's Message of July, 1861, December, 1861, and December, 1862-The Thirty-Eighth Congress-Annual Message, 1863 -Ammesty Proclamations, and Circular of the Atterney General-Proclamations concerning: Attorney General-Droclamations concerning: the Blockade, Non-Intercourse with States in Re-bellion, and declaring Boundaries of the Rebellion.
- THE AFRICAN SLAVE TRADE...... 150 The Seward-Lyons Treaty—Vote in the Senate-upon bill to give it effect—Action of the "Confederate" Congress on Slave Trade-Jefferson Davis's Veto thereof—Intercepted Despatch from Judah P. Benjamin to L. Q. C. Lamar.
- ARREST OF CITIZENS, THE WRIT OF HABEAS CORPUS, AND SUPPRESSION OF NEWS-
  - PAPERS ..... Arrest of Members of the Maryland Legislature. and of the Baltimore Police Commissioners-Orders of Gen. McClellan and Secretary Came-ron-John Merryman's Case and Chief Justice. ron-John Merryman's Case and Chief Justice. Taney's Opinion-Attorney General Bate's, Opinion on the President's Power to Arrest and to Suspend the Privilege of the Writ of Habcas Corpus-Views of Horace Binney and Theophilus Parsons-Case of C. L. Vallandigham; Decision of the Supreme Court therein; his Lotter on Retailation; his return to Chio, and Speech at Hamilton-Proclamations of the President Sus-pending the Privilege of the Writ of Habcas Corpus-Indemnification of the President Sus-Case of George W. Jones vs. W. H. Seward-"Confederate" Legislation upon the Suspension of the Writ-Suppressions and Seizures of News-papers, with the Proceedings of the Courts, Congress, and the Post Office Department.
- CONFISCATION AND EMANCIPATION. ..... 195

The Confiscation Bills, and Amendatory Joint, Resolution, and Special Message thereon-Emancipation in the Thirty-Seventh Congress-Proposed Repeal of the Joint Resolution afore-said—Sequestration in the Rebel States—Judi-cial and Military Proceedings under the Confiscation Law-Proclamation thereon-President's Message, March, 1862, recommending Compen-sated Emancipation—Congressional Proceedings thereon-Interview of Border State Congress-men with the President-Emancipation in the District of Columbia-The President's Apthe District of Columbia—The President's Ap-peal to the Border State Congressmen, and their. Reply—Extract from the President's Annual Message, December, 1862—Emancipation in Maryland and Proceedings of the Constitutional Convention thereof—Emancipation Proclama-mations—Votes thereon and Resolutions con-cerning them—Interview between the Chicago-Deputation and the President—Address of the Loyal Governors—Mr. Boutwell's Statement concerning the issue of the Proclamation—Let-ters of Charles Summer and Owen Loyeloy. ters of Charles Sumner and Owen Lovejoy.

vii

Page.

48

91

REPEAL OF THE FUGITIVE SLAVE LAWS, "CON-

TRABANDS," AND KINDRED SUBJECTS...... 234 ABANDS," AND KINDRED SUBJECTS...... Votes on the Passage of the Acts of 1793 and 1850-Repealing Movements in the Thirty-Second, Thirty-Third, Thirty-Seventh, and Thirty-Eighth Congresses-Consus Report rela-ting to the Escape of Fugitive Slaves from 1850 to 1860-The New Article of War-Employment of Slaves in Government Dock-Yards, & C-Re-cognition of Hayti and Liberia-Robert Small-Proposed Removal of the Disqualification of Color in carrying the Mails-Negro Suffrage in the District of Columbia and Montana Territory -Exclusion of Colored Persons from Rai-cars--Exclusion of Colored Persons from Rail-cars-Colored Persons around Natrears-Repeal of Lawa-regulating the Coastwise Slave Trade-Orders and Letters concerning "Contrabands," by Gens. McClellau and Butler, and Secretary Cameron-Fremont's Proclamation of Eman-cipation, and Correspondence with the President therenpon-"Contrabands" in the District of Columbia-Gen. Burnside's Proclamations by Gens. Halleck, Buell, Hooker, McDowell, Don-bleday and others-General Instructions by the President concerning "Contrabands"-Gens Phelps and Butler on Arming Negroes-Pro-posed Congressional Censure of Gen. Halleck's Order No. 3-Prohibition of Slavery in the Ter-ritories-Amendments to the Constitution, pro-posed in the Thirty-Eighth Congress, First Ses-sion-Resolutions on Slavery in the States, in the same Congress-Bureau of Freedmen's Affaira. SLATION, ORDERS, PROCLAMATIONS AND Colored Persons as Witnesses-Repeal of Laws

LEGISLATION, ORDERS, PROCLAMATIONS AND PROPOSITIONS, RELATIVE TO THE WAR,

AND TO "PEACE"...... 261 The Enrollment Acts of 1863 and 1864, with the votes upon all their leading Features and Char-acteristics-Resolutions relative to the Enrollacteristics-Resolutions relative to the Enroll-ment-Orders of the War Department enforcing the Draft of 1862-Gen. McClellan's Recommen-dation of a Draft in 1861-Colored Soldiers and their Pay-Opinion of Attorney General Bates respecting the pay of Rev. S. Harrison, colored Chaplain of the 54th Mass. Regiment-Rules and Orders for the Protection of colored Sol-diers and the President's Speech thereon-Use and Orders for the Protection of colored Sol-diers and the President's Speech thereon-Use of colored men in the "Confederate" Military Service-Negro Enlistment Act of the Tennes-see Rebel Legislature-"Confederate" Legisla-tion upon the Treatment of captured Colored Troops and their Officers-Homesteads for Sol-diers-Unemployed Generals-Resolutions apon the Objects and Prosecution of the War, in the Thirty-Seventh and Thirty-Eighth Congresses-"Peace" Propositions in the Same-Correspond-ence between the President and Fernando Wood -The Niagrara Falls Conference and Correspond--The Niagara Falls Conference and Correspond-once-Peace Propositions in the Rebel Con-gress-Correspondence between Governor Vance and Jefferson Davis-Reported Statement of Davis to Gilmore.

MILITARY ORDERS RESPECTING ELECTIONS., 308

Orders of Gens. McClellan, Dix, and Schenck.— Governor Bradford's Proclamation of 1863, and the President's Letter to the Governor—Orders, in Delaware, Kentucky, and Missouri—Orders, concerning Impressment of Property—Proposed Legislation upon Military Interference in Elections.

The Reconstruction Bill, with the President's Proclamation thereon, and Statement of Sena-tor Wade and Representative Davis-Electoral tor Wade and Representative Davis—Electora Vote of Rebel States—Proposed Commission of Inquiry—Senators from Arkansas,—Process of Reconstruction in Arkansas, Louisiana, and Vir-ginia—Resolutions by sundry Senators and Rep-resentatives concerning the Relations of Rebel States to the Government—Rebel Views of Rebel States, and Legislatures of Rebel States, with Statements of prominent Rebels.

MISCELLANEOUS PAPERS AND SPEECHES OF 

Message of May 29, 1862—Remarks at Union Meeting in Washington City, August 6, 1862— Response to a Screnade, July, 1863—Speech at the Philadelphia Fair, June 16, 1864—Lettera to Horace Greeley, to the Springfield Mass Conven-tion, to Col. A. G. Hodges, of Kentucky, and to the Grant Meeting in New York, June, 1864 1864.

The Trent Affair—Monarchical Intrigues in Cen-tral and South America—Alleged Foreign En-listments—Foreign Modiation, being Letters from Secretary Seward to Governor Hicks and M. Drouyn de l'Huys, and from Lord Lyons to Earl Russell, with his Views on those of New York Democrats respecting Foreign Mediation— The French in Mexico—Congressional Action thereon—The Arguelles Case.

Summary of Financial Legislation from Decem-Summary of Financial Legislation from Decem-ber, 1860, to June 30, 1864—Special War Income Tax, and Votes thereon—The "Legal Tender" Question—Loan Bill of 1864—National Currency Acts—Internal Revenue Acts—Proposed Tax on Slaves—Tariff Acts of 1862 and 1864—Taxea in Insurrectionary Districts—The Public Credit in 1860 and 1861—Statements of Public Debt from June 30, 1860 to June 30, 1864—" Confed-erate" Finances, with their Tax, Funding, and Tithing Acts. Tithing Acts.

- The President's Views on Colonization—Incom-patibility of Civil and Military Office—Fishing Bounties—Acts to Prohibit Polygamy; declaring. Bounties—Acts to Prohibit Polygamy; declaring certain Persons incligible to Ohice; and to pun-ish Conspiracy—Letters of Marque—Enabling Act for Nebraska—Admission of West Virginia —Opinions of Attorney General Batcs on Citizen-ship, and on the pay of Colored Soldiers— —McClellan's Letters recommending a Polit-ical Policy in the Conduct of the War, and favoring Woodward's Election in Pennsylvania —Proposed Censure of President Lincoln and Ex-President Buchanan—Censure of Repre-sentatives Long and Harris. sentatives Long and Harris.
- Threats of Dissolution in the First Congress, 1789—Prophetic Utterances of Jackson, Benton, and Clay—Southern Disunion Congressional Cancus in 1835—Early Hopes of the Rebels— Ex-President Pierce's Letter to Jefferson Davis, Ex-President Pierce's Letter to Jetterson Davis, 1860—The Disunion Programme—Letter of D. L. Yulee, January 7, 1861—Douglas's Last Words—Progress of the Conspiracy in Maryland —Minutes of the Baltimore Police Commission-ers during "the Reign of Terror"—Report to the Baltimore Councils on Expenditure of the 5500 000 recompendent for Orderance Rummers \$500,000 appropriated for Ordnance Purposes-Legislative Action thereon, and other Proceed-ings by the Maryland Legislature of 1861—Sun-dry Rebel Items.
- THE REBEL ADMINISTRATION ...... 400 The Provisional President, Cabinet, and Con-gress, with Memorandum of Changes -- The "Permanent" Administration--The First Con-gress, and Changes therein--The Second Congress\_

NATIONAL POLITICAL CONVENTIONS IN 1864..... ..... 403 The National Union Convention and Letters of Acceptance by President Lincoln and Andrew Johnson-The Cleveland Convention, and the Letters of Acceptance of Fremont and Coch-rane-Col. Cochrane's Address to his regiment, Navaraber 13, 1861 November 13, 1861. INDEX..... 417

.

viii

## RESULT OF THE PRESIDENTIAL ELECTION OF 1860.

New Hampshire5337,519 $441$ $25,881$ $2,11$ Massachusetts134 $106,533$ $22,331$ $34,372$ $5,92$ Rhode Island4 $12,244$ $*7,707$ $7,707$ Connecticut6 $43,792$ $3,291$ $15,522$ $14,66$ Vermont5 $362,646$ $*312,510$ $*312,510$ New York43 $58,324$ $*62,801$ $*26,901$ New Jersey43 $58,324$ $*62,801$ $*178,8^{2}$ Delaware27 $268,030$ $12,776$ $16,765$ $*178,8^{2}$ Delaware3 $3,815$ $3,864$ $1,023$ $7,33$ Maryland1510 $44,990$ $2,701$ $48,53$ North Carolina1510 $42,886$ $11,590$ $51,83$ Kentucky12 $1,364$ $66,058$ $25,651$ $53,14$ Tennessee12 $23,610$ $12,194$ $187,232$ $11,44$ Louisiana13 $13,30,033$ $5,306$ $115,509$ $12,27$ Indiana13 $172,161$ $4,913$ $160,215$ $2,4,70$ Missouri9 $4$ $20,094$ $5,227$ $28,77$ Michigan6 $88,480$ $405$ $65,057$ $88,131,33$ Arkansas6 $88,480$ $405$ $65,057$ $85,111$ Michigan4 $39,173$ $6,817$ $38,516$ $34,37$ Michigan4 $22,069$ $62$ $11,920$ <td< th=""><th></th><th>ELE</th><th>CTOR</th><th>AL VO</th><th>TE.</th><th></th><th></th></td<>		ELE	CTOR	AL VO	TE.				
New Hampshire5337,519 $441$ $25,881$ $2,11$ Massachusetts134 $106,533$ $22,331$ $34,372$ $5,92$ Rhode Island4 $12,244$ $*7,707$ $5,922$ $14,66$ Connecticut6 $43,792$ $3,291$ $15,522$ $14,66$ Vermont5 $362,646$ $*312,510$ $*7,707$ New York $35$ $362,646$ $*312,510$ New Jersey $4$ $3$ $58,324$ $*62,801$ Pennsylvania $27$ $268,030$ $12,776$ $16,765$ Delaware $3$ $3,815$ $3,864$ $1,023$ $7,33$ Maryland $4$ $3$ $58,324$ $*62,801$ Virginia $15$ $10$ $42,986$ $16,290$ $74,33$ North Carolina $15$ $10$ $42,986$ $16,290$ $74,33$ North Carolina $12$ $1,364$ $66,058$ $25,651$ $53,12$ Tennessee $12$ $1,364$ $66,058$ $25,651$ $53,12$ Missisippi $11$ $172,161$ $4,913$ $160,215$ $2,44$ Alabama $13$ $139,033$ $5,306$ $115,509$ $12,21$ Illinois $11$ $9$ $17,028$ $58,372$ $28,801$ $31,33$ Arkansas $6$ $88,480$ $405$ $65,057$ $8,51$ Mississippi $4$ $70,409$ $1,763$ $55,111$ $1,00$ Wisconsin $4$ $22,069$ $62$ $19,20$ $763$ </td <td>States.</td> <td>Lincoln.</td> <td>Bell.</td> <td>Douglas.</td> <td>Brockinridge.</td> <td>Lincoln.</td> <td>Bell.</td> <td>Douglas</td> <td>Breckinridge.</td>	States.	Lincoln.	Bell.	Douglas.	Brockinridge.	Lincoln.	Bell.	Douglas	Breckinridge.
California         4         39,173         6,817         38,516         34,33           Minnesota         4         22,069         62         11,920         74	New Hampshire Massachusetts Rhode Island Connecticut Vermont New York New Jersey Pennsylvania Delaware Maryland Virginia North Carolina South Carolina Georgia Kentucky Tennessee Ohio Louisiana Mississippi Indiana Illinois Alabama Missouri Arkansas Michigan Florida Texas Iowa	5 13 4 6 5 35 4 27 23 13 11 6 4	12		8 10 8 10 6 7 9 4 3	37,519 106,533 12,244 43,792 33,808 362,646 58,324 268,030 3,815 2,294 1,929 1,364 231,610 139,033 172,161 17,028 88,480 70,409	$\begin{array}{r} 441\\ 22,331\\ 3,291\\ 1,969\\ 12,776\\ 3,864\\ 41,760\\ 74,681\\ 44,990\\ No\\ 42,886\\ 66,058\\ 69,274\\ 12,194\\ 20,204\\ 25,040\\ 5,306\\ 4,913\\ 27,875\\ 58,372\\ 20,094\\ 405\\ 55,437\\ *15,438\\ 1,763\\ \end{array}$	$\begin{array}{c} 25,881\\ 34,372\\ *7,707\\ 15,522\\ 6,849\\ *312,510\\ *62,801\\ 16,765\\ 1,023\\ 5,966\\ 16,290\\ 2,701\\ \textbf{popular}\\ 11,590\\ 25,651\\ 11,350\\ 187,232\\ 7,625\\ 3,283\\ 115,509\\ 160,215\\ 13,651\\ 58,801\\ 5,227\\ 65,057\\ 367\\ 55,111\\ \end{array}$	$\begin{array}{c} 6,368\\ 2,112\\ 5,939\\ 14,641\\ 218\\ *178,871\\ 7,337\\ 42,482\\ 74,323\\ 48,539\\ vote.\\ 51,889\\ 53,143\\ 64,709\\ 11,405\\ 22,681\\ 40,797\\ 12,295\\ 2,404\\ 48,831\\ 31,317\\ 28,732\\ 805\\ 8,543\\ 47,548\\ 1,048\\ 888\\ \end{array}$
Total	California Minnesota Oregon	$\begin{array}{c} 4\\ 4\\ 3\end{array}$			79	39,173 22,069 5,270	6,817 62 183	38,516 11,920 3,951	34,334 748 5,006 847,953

Lincoln	over	Douglas	491,295
**	"	Breckinridge	1,018,499
"	**	Bell	1,275,821
Other c	andid	lates over Lincoln	947,289

\* Fusion.

7

### SECESSION MOVEMENT DEVELOPED.

IMMEDIATELY thereupon, and clearly by concert of action previously arranged, various disunion Governors hastily took steps to procure the passage of ordinances of secession by Conventions of their States, artfully using the unsettled excitements of the Presidential canvass to that end.

These proceedings in brief were as follows:

#### SOUTH CAROLINA.

November 6th, 1860. Legislature met to choose Presidential electors, who voted for Breckinridge and Lane for President and Vice President. Gov. WILLIAM H. GIST recommended in his message that in the event of Abraham Lincoln's election to the Presidency, a convention of the people of the State be immediately called to consider and determine for themselves the mode and measure of redress. He expressed the opinion that the only alternative left is the "secession of South Carolina from the Federal Union."

7th. United States officials resigned at Charleston.

10th. U. S. Senators JAMES H. HAMMOND and JAMES CHESNUT, Jr., resigned their seats in the Senate. Convention called to meet Dec. 17th. Delegates to be elected Dec. 6th.

13th. Collection of debts due to citizens of non-slaveholding States stayed. FRANCIS W. PICKENS elected Governor, who appointed a cabinet consisting of A. G. MAGRATH Secretary of State. DAVID F. JAMISON Secretary of War, C. G. MEMMINGER Secretary of Treas-Ury, W. W. HARLLEE P. M. General, ALBERT C. GARLINGTON Secretary of Interior.

17th. Ordinance of Secession adopted unanimously.

21st. Commissioners appointed (BARN-WELL, ADAMS, and ORR) to proceed to Washington to treat for the possession of U.S. Government property within the limits of South Carolina. Commissioners appointed to the other Slaveholding States. Southern Congress proposed.

24th. Representatives in Congress withdrew.

Gov. PICKENS issued a proclamation "an-

nouncing the repeal, Dec. 20th, 1860, by the good people of South Carolina," of the Ordinance of May 23d, 1788, and "the dissolution of the union between the State of South Carolina and other States under the name of the United States of America," and proclaiming to the world "that the State of South Carolina is, as she has a right to be, a separate, sovereign, free and independent State, and, as such, has a right to levy war, conclude peace, negotiate treaties, leagues, or covenants, and to do all acts whatsoever that rightfully appertain to a free and independent State.

Done in the eighty-fifth year of the sovereignty and independence of South Carolina.'

Jan. 3d, 1861. South Carolina Commissioners left Washington.

4th. Convention appointed T. J. Withers, L. M. Keitt, W.W. Boyce, Jas. Chesnut, Jr., R. B. Rhett, Jr., R. W. Barnwell, and C. G. Memminger, delegates to Southern Congress.

5th. Convention adjourned, subject to the call of the Governor.

14th. Legislature declared that any at-tempt to reinforce Fort Sumter would be considered an open act of hostility and a declaration of war. Approved the Governor's action in firing on the Star of the West. Accepted the services of the Catawba Indians.

27th. Received Judge Robertson, Commissioner from Virginia, but rejected the proposition for a conference and co-operative action.\*

\* The resolutions are :

Resolutions are: Resoluted unanimously, That the General Assembly of South Carolina tenders to the Legislature of Virginia their acknowledgment of the friendly motives which inspired the mission entrusted to Hon. Judge Robertson, her Commissioner.

missioner. Resolved unanimously, That candor, which is due to the long-continued sympathy and respect which has sub-sisted hetween Virginia and South Carolina, induces the Assembly to declare with frankness that they do not deem it advisable to initiate negotiations, when they have no desire or intention to promote the ultimate object in view. That object is declared, in the resolution of the Virginia Legislature, to be the procurement of amendments to, or new guarantees in, the Constitution of the United States.

March 26th. Convention met in Charleston.

April 3d. Ratified "Confederate" Constitution—yeas 114, nays 16.

8th. Transferred forts, etc. to "Confederate" government.

#### GEORGIA.

November 8th, 1860. Legislature met pursuant to previous arrangement.

18th. Convention called. Legislature appropriated \$1,000,000 to arm the State.

Dec. 3d. Resolutions adopted in the Legislature proposing a Conference of the South-

ern States at Atlanta, Feb. 20th. January 17th, 1861. Convention met. Received Commissioners from South Carolina and Alabama.

18th. Resolutions declaring it the right and duty of Georgia to secede, adoptedyeas 165, nays 130.

19th. Ordinance of Secession passedyeas 208, nays 89.

21st. Senators and Representatives in Congress withdrew.

24th. Elected Delegates to Southern Congress at Montgomery, Alabama.

28th. Elected Commissioners to other Slaveholding States.

29th. Adopted an address "to the South and the world."

March 7th. Convention reassembled.

16th. Ratified the "Confederate" Consti-

tution—yeas 96, nays 5. 20th. Ordinance passed authorizing the "Confederate" government to occupy, use and possess the forts, navy yards, arsenals, and custom houses within the limits of said State.

April 26th. Governor Brown issued a proclamation ordering the repudiation by the citizens of Georgia of all debts due Northern men.

Resolved unanimously, That the separation of South Carolina from the Federal Union is final, and she has no further interest in the Constitution of the United States; and that the only appropriate negotiations between her and the Federal Government are as to their mutual relations as foreign States.

Lations as foreign States. Resolved unanimously, That this Assembly further owes it to her friendly relations with the State of Virginia to declare that they have no confidence in the Federal Government of the United States; that the most solemu pledges of that government have been disregarded; that under pretence of preserving property, hostile troops have been attempted to be introduced into one of the fortresses of this State concealed in the hold of a vessel of comof this State, concealed in the hold of a vessel of com-merce, with a view to subjugate the people of South Car-olina, and that even since the authorities at Washington have been informed of the present mediation of Virginia,

have been informed of the present mediation of Virginia, a vessel of war has been sent to the South, and troops and munitions of war concentrated on the soil of Virginia. *Resolved unanimously*, That in these circumstances this Assembly, with renewed assurances of cordial re-spect and esteem for the people of Virginia, and high con-sideration for her Commissioner, decline entering into the merginicing request.

sideration for her Commissioner, decline entering into the negotiations proposed. The Charleston Mercury of an earlier date thus alluded to Border State embassies: "Hear them, if you please; treat them with civility; feed them, and drench them in champagne—and let them go! Let us act as if they had never come, as if they had not spoken, as if they did not exist; and let them seek to preserve their Treasmry pap through some more supple agency than ours. The time has gone by when the voice of a Virginia politician, though he coo like a dove, should be heard in the land of a patriotic people."

#### MISSISSIPPI.

November 26th, 1860. Legislature met Nov. 26th, and adjourned Nov. 30th. Election for Convention fixed for Dec. 20th, Convention to meet Jan. 7th. Convention bills and secession resolutions passed unani-Commissioners appointed to other mously. Slaveholding States to secure "their co-operation in effecting measures for their common defence and safety.'

Jan. 7th, 1861. Convention assembled.

9th. Ordinance of Secession passed-yeas 84, nays 15.

In the ordinance the people of the State of Mississippi express their consent to form a federal union with such of the States as have seceded or may secede from the Union of the United States of America, upon the basis of the present Constitution of the United States, except such parts thereof as embrace other portions than such seceding States.

10th. Commissioners from other States Resolutions adopted, recognizing received. South Carolina as sovereign and independent.

Jan. 12th. Representatives in Congress withdrew.

19th. The committee on the Confederacy in the Legislature reported resolutions to provide for a Southern Confederacy, and to establish a provisional government for seceding States and States hereafter seceding. 21st. Senators in Congress withdrew.

March 30th. Ratified "Confederate" Constitution-yeas 78, nays 7.

#### FLORIDA.

November 26th, 1860. Legislature met. Governor M. S. Perry recommended immediate secession.

Dec. 1st. Convention bill passed.

Jan. 3d, 1861. Convention met.

7th. Commissioners from South Carolina and Alabama received and heard.

10th. Ordinance of Secession passedyeas 62, nays 7.

18th. Delegates appointed to Southern Congress at Montgomery.

21st. Senators and Representatives in Congress withdrew.

Feb. 14th. Act passed by the Legislature declaring that after any actual collision between Federal troops and those in the employ of Florida, the act of holding office under the Federal government shall be declared treason, and the person convicted shall suffer death. Transferred control of government property captured, to the "Confederate" government.

#### LOUISIANA.

December 10th, 1860. Legislature met.

11th. Convention called for Jan. 23d. Military bill passed.

12th. Commissioners from Mississippi received and heard. Governor instructed to communicate with Governors of other southern States.

January 23d, 1861. Convention met and | organized. Received and heard Commissioners from South Carolina and Alabama.

25th. Ordinance of Secession passedyeas 113, nays 17. Convention refused to submit the ordinance to the people by a vote of 84 to 45. This was subsequently reconsidered, and the ordinance was submitted. The vote upon it as declared was 20,448 in favor, and 17,296 against.

Feb. 5th. Senators withdrew from Congress, also the Representatives, except John E. Bouligny. State flag adopted. Pilots at the Balize prohibited from bringing over the bar any United States vessels of war.

March 7th. Ordinance adopted in secret \*session transferring to "Confederate" States government \$536,000, being the amount of bullion in the U.S. mint and customs seized by the State.\*

16th. An ordinance voted down, submitting the "Confederate" Constitution to the people-yeas 26, nays 74.

21st. Ratified the "Confederate" Constitution-yeas 101, nays 7. Governor authorized to transfer the arms and property captured from the United States to the "Confederate" Government.

27th. Convention adjourned sine die.

#### ALABAMA.

January 7th, 1861. Convention met.

8th. Received and heard the Commissioner from South Carolina.

11th. Ordinance of Secession passed in secret session—yeas 61, nays 39. Proposition to submit ordinance to the people lost -yeas 47, nays 53.

14th. Legislature met pursuant to previous action.

19th. Delegates elected to the Southern Congress.

21st. Representatives and Senators in Congress withdrew.

26th. Commissioners appointed to treat with United States Government relative to the United States forts, arsenals, etc., within the State.

The Convention requested the people of the States of Delaware, Maryland, Virginia, North Carolina, South Carolina, Florida, Georgia, Mississippi, Louisiana, Texas, Arkansas, Tennessee, Kentucky and Missouri to meet the people of Alabama by their delegates in Convention, February 4th, 1861, at Montgomery, for the purpose of consulting as to the most effectual mode of securing concerted or harmonious action in whatever measures may be deemed most desirable for their common peace and security.-Military Bill passed. Commissioners appointed to other Slaveholding States.

March 4th. Convention re-assembled.

13th. Ratified "Confederate" Constitution,

\* March 14th, 1861. The "Confederate" Congress passed a resolution accepting these funds, with "a high sense of the patriotic liberality of the State of Louisiana."

Transferred control of forts, yeas 87, nays 6. arsenals, etc., to "Confederate" Government

#### ARKANSAS.

January 16th, 1861. Legislature passed Convention Bill. Vote of the people on the Convention was 27,412 for it, and 15,826 against it.

February 18th. Delegates elected.

March 4th. Convention met.

18th. The Ordinance of Secession de feated-yeas 35, nays 39. The Convention effected a compromise by agreeing to submit the question of co-operation or secession to the people on the 1st Monday in August.

May 6th. Passed Secession Ordinanceyeas 69, nays 1. Authorized her delegates to the Provisional Congress, to transfer the arsenal at Little Rock and hospital at Napoleon to the "Confederate" Government.

#### TEXAS.

January 21st, 1861. Legislature met. 28th. People's State Convention met.

29th. Legislature passed a resolution declaring that the Federal Government has no power to coerce a Sovereign State after she has pronounced her separation from the Federal Union.

February 1st. Ordinance of Secession passed in Convention-yeas 166, nays 7. Military Bill passed.

7th. Ordinance passed, forming the foundation of a Southern Confederacy. Delegates to the Southern Congress elected. Also an act passed submitting the Ordinance of Secession to a vote of the people.

23d. Secession Ordinance voted on by the people; adopted by a vote of 34,794 in favor, and 11,235 against it.

March 4th. Convention declared the State out of the Union. Gov. Houston issued a proclamation to that effect.

16th. Convention by a vote of 127 to 4 deposed Gov. Houston, declaring his seat vacant. Gov. Houston issued a proclamation to the people protesting against this action of the Convention.

20th. Legislature confirmed the action of the Convention in deposing Gov. Houston by a vote of 53 to 11. Transferred forts, etc., to "Confederate" Government.

23d. Ratified the "Confederate" Constitution-yeas 68, nays 2.

NORTH CAROLINA. November 20th, 1860. Legislature met. Gov. Ellis recommended that the Legislature invite a conference of the Southern States, or failing in that, send one or more delegates to the neighboring States so as to secure concert of action. He recommended a thorough reorganization of the militia, and the enrolment of all persons between 18 and 45 years, and the organization of a corps of ten thousand men; also, a Convention, to assemble immediately after the proposed consultation with other Southern States shall have terminated.

December 9th. Joint Committee on Federal Relations agreed to report a Convention Bill.

17th. Bill appropriating \$300,000 to arm the State, debated.

18th. Senate passed above bill-yeas 41, nays 3.

20th. Commissioners from Alabama and Mississippi received and heard-the latter,

J. Thompson, by letter. 22d. Senate Bill to arm the State failed to pass the House.

22d. Adjourned till January 7th.

January 8th, 1861. Senate Bill arming the State passed the House, yeas 73, nays 26. 30th. Passed Convention Bill-election to

take place February 28th. No Secession Ordinance to be valid without being ratified by

a majority of the qualified voters of the State. 31st. Elected Thos. L. Clingman United States Senator.

February 13th. Commissioners from Georgia publicly received. 20th. Mr. Hoke elected Adjutant General

of the State. Military Bill passed.

28th. Election of Delegates to Convention took place.

28th. The vote for a Convention was 46,671; against 47,333-majority against a Convention 661.

May 1st. Extra session of the Legislature met at the call of Gov. Ellis. The same day they passed a Convention Bill, ordering the election of delegates on the 15th.

2d. Legislature adjourned.

13th. Election of delegates to the Convention took place.

20th. Convention met at Raleigh.

21st. Ordinance of Secession passed; also the "Confederate" Constitution ratified.

June 5th. Ordinance passed, ceded the arsenal at Fayetteville, and transferred magazines, etc., to the "Confederate" Government.

#### TENNESSEE.

January 6th, 1861. Legislature met.

12th. Passed Convention Bill.

30th. Commissioners to Washington appointed.

February 8th. People voted no Conven-tion: 67,360 to 54,156. May 1st. Legislature passed a joint resolu-tion authorizing the Governor to appoint Commissioners to enter into a military

thon: 67,360 to 24,120.
May 1st. Legislature passed a joint resolution authorizing the Governor to appoint Commissioners to enter into a military league with the authorities of the "Confederate" States.
Tth. Legislature in secret session ratified the league entered into by A. O. W. Totten, Gustavus A. Henry, Washington Barrow, Gustavus A. Henry, Washington Barrow, W. Hilliard, Commissioner for "Confederate" States, stipulating that Tennessee until she became a member of the Confederacy placed the whole military force of the State under the control of the President of the "Confederate" States, and turned over to the "Confederate" States all the public property, naval stores and munitions of war.

Passed the Senate, yeas 14, nays 6, absent and not voting 5; the House, yeas 42, nays 15, absent and not voting, 18.\* Also a Declaration of Independence and Ordinance dissolving the Federal relations between Tennessee and the United States, and an ordinance adopting and ratifying the Confederate Constitution, these two latter to be voted on by the people on June 8th, were passed.

June 24th. Gov. Isham G. Harris declared Tennessee out of the Union, the vote for Separation being 104,019 against 47,238.

#### VIRGINIA.

January 7th, 1861. Legislature convened. 8th. Anti-coercion resolution passed.

9th. Resolution passed, asking that the

status quo be maintained. 10th. The Governor transmitted a despatch from the Mississippi Convention, announcing its unconditional secession from the Union, and desiring on the basis of the old Constitution to form a new union with the seceding States. The House adoptedyeas 77, nays 61,-an amendment submitting to a vote of the people the question of referring for their decision any action of the Convention dissolving Virginia's connection with the Union, or changing its organic law. The Richmond Enquirer denounced "the emasculation of the Convention Bill as imperilling all that Virginians hold most sacred and dear."

16th. Commissioners Hopkins and Gilmer of Alabama received in the Legislature.

17th. Resolutions passed proposing the Crittenden resolutions as a basis for adjustment, and requesting General Government to avoid collision with Southern States. Gov. Letcher communicated the Resolutions of the Legislature of New York, expressing the utmost disdain, and saying that "the threat conveyed can inspire no terror in freemen." The resolutions were directed to be returned to the Governor of New York. \$1,000,000 appropriated for the 18th.

defence of the State.

Passed resolve that if all efforts 19th. to reconcile the differences of the country

\* The following is the vote in the Senate on the adoption

fail, every consideration of honor and interest demands that Virginia shall unite her destinies with her sister slaveholding States. Also that no reconstruction of the Union can be permanent or satisfactory, which will not secure to each section self-protecting power against any invasion of the Federal Union upon the reserved rights of either. (See Hunter's proposition for adjustment.)

21st. Replied to Commissioners Hopkins and Gilmer, expressing inability to make a definite response until after the meeting of the State Convention.

22d. The Governor transmitted the resolutions of the Legislature of Ohio, with unfavorable comment. His message was tabled by a small majority.

30th. The House of Delegates to-day tabled the resolutions of the Pennsylvania Legislature, but referred those of Tennessee to the Committee on Federal Relations.

February 20th. The resolutions of the Legislature of Michigan were returned without comment.

28th. Ex-President Tyler and James A. Seddon, Commissioners to the Peace Congress, presented their report, and denounced the recommendation of that body as a delusion and a sham, and as an insult and an offense to the South.

**Proceedings of Virginia Convention.** February 4th. Election of delegates to the Convention.

13th. Convention met.

14th. Credentials of John S. Preston, Commissioner from South Carolina, Fulton Anderson from Mississippi, and Henry L. Benning from Georgia, were received.

18th. Commissioners from Mississippi and Georgia heard; both pictured the danger of Virginia remaining with the North; neither contemplated such an event as reunion.

19th. The Commissioner from South Carolina was heard. He said his people believed the Union unnatural and monstrous, and declared that there was no human force —no sanctity of human touch,—that could re-unite the people of the North with the people of the South—that it could never be done unless the economy of God were changed.

20th. A committee reported that in all but sixteen counties, the majority for submitting the action of the Convention to a vote of the people was 52,857. Numerous resolutions on Federal Relations introduced, generally expressing attachment to the Union, but denouncing coercion.

26th. Mr. Goggin of Bedford, in his speech, denied the right of secession, but admitted a revolutionary remedy for wrongs committed upon a State or section, and said wherever Virginia went he was with her.

March 2d. Mr. Goode of Bedford offered a resolution that, as the powers delegated to the General Government by Virginia had and erases the clause relative to the securing

been perverted to her injury, and as the Crittenden propositions as a basis of adjustment had been rejected by their Northern confederates, therefore every consideration of duty, interest, honor and patriotism requires that Virginia should declare her connection with the Government to be dissolved.

5th. The thanks of the State were voted to Hon. John J. Crittenden, by yeas 107, nays 16, for his efforts to bring about an honorable adjustment of the national difficulties. Mr. Harvie of Amelia offered a resolution, requesting Legislature to make needful appropriations to resist any attempt of the Federal authorities to hold, occupy or possess the property and places claimed by the United States in any of the seceded States, or those that may withdraw or collect duties or imposts in the same.

9th. Three reports were made from the Committee on Federal Relations. The majority proposed to submit to the other States certain amendments to the Constitution, awaiting the response of non-slaveholding States before determining whether "she will resume the powers granted by her under the Constitution of the United States, and throw herself upon her reserved rights; meanwhile insisting that no coercion be attempted, the Federal forts in seceded States be not reinforced, duties be not collected, etc.," and proposing a Convention at Frankfort, Kentucky, the last Monday in May, of the States of Delaware, Maryland, North Carolina, Tennessee, Kentucky, Missouri and Arkan-Henry A. Wise differed in details, and Sas. went further in the same direction. Messrs. Lewis E. Harvie, Robert L. Montague and Samuel C. Williams recommended the immediate passage of an Ordinance of Secession. Mr. Barbour of Culpeper insisted upon the immediate adoption by the non-slaveholding States of needed guarantees of safety, and provided for the appointment of three Commissioners to confer with the Confederate authorities at Montgomery.

19th. Committee on Federal Relations reported proposed amendments to the Constitution, which were the substitute of Mr. Franklin of Pa, in "Peace Conference," changed by using the expression "involuntary servitude" in place of "persons held toservice." The right of owners of slaves is not to be impaired by congressional or territorial law, or any pre-existing law in territory hereafter acquired.

Involuntary servitude, except for crime, to be prohibited north of 36°30', but shall not be prohibited by Congress or any Territorial legislature south of that line. The third section has some verbal alterations, providing somewhat better security for property in transit. The fifth section prohibits the importation of slaves from places beyond the limits of the United States. The sixth makes some verbal changes in relation to remuneration for fugitives by Congress, and erases the clause relative to the securing of privileges and immunities. The seventh forbids the granting of the elective franchise and right to hold office to persons of the African race. The eighth provides that none of these amendments, nor the third paragraph of the second section of the first article of the Constitution, nor the third paragraph of the second section of the fourth article thereof, shall be amended or abolished without the consent of all the States.

25th. The Committee of the Whole re-fused (yeas 4, nays 116) to strike out the majority report and insert Mr. Carlile's "Peace Conference" substitute.

26th. The Constitution of the "Confederate" States, proposed by Mr. Hall as a substitute for the report of the committee, rejected—yeas 9, nays 78. 28th. The first and second resolutions

reported by the committee adopted.

April 6th. The ninth resolution of the majority report came up. Mr. Bouldin offered an amendment striking out the whole, and inserting a substitute declaring that the independence of the seceded States should be acknowledged without delay, which was lost -yeas 68, nays 71.

9th. Mr. Wise's substitute for the tenth resolution, to the effect that Virginia recognizes the independence of the seceding States, was adopted—yeas 128, nays 20. April 17. Ordinance of Secession passed

in secret session—yeas 88, nays 55, one excused, and eight not voting.\*

\* The injunction of secrecy has never been removed from this vote, but the tally was recently discovered among the papers of Lewis T. Kinser, Esq., deceased, former law partner of George W. Brent, delegate from Alexandria, Va., and is as follows, as published in the Washington Stor. Washington Star :

former law partner of George W. Breni, delegate from Mexandria, Ya., and is as follows, as published in the Washington Star: TEAS.-WM. M. Ambler, Wm. B. Aston, Jas. Barbonr, Agus R. Blakey, George Blow, Jr., James Boisseau, Peter B. Boost, Wood Bouldin, Wm. W. Boyd, Thomas Branch, James C. Bruce, Fred, M. Cabell, John A. Camp-bell, Allen T. Caperton, William P. Cicil, John R. Chan-bliss, Manilus Chapman, Sam'l A. Coffman, Raphael M. Gonn, James H. Cox, Richard H. Cox, John Crither, Harvy Deskins, James B. Dorman, John Echols, Miers W. Fisher, Thos. S. Flournoy, Wm. W. Forbes, Napoleon B. French, Sam'l M. Garland, H. L. Gillespie, Sam'l L. Graham, Fendal Gregory, Jr., Wm. L. Gorgin, John Goode, Jr., Thos. F. Goode, F. L. Hale, Cyrns Hall, L. S. Hall, Lewis E. Harvie, James P. Holeombe, John N. Hagbes, Eppa Hunton, Lewis D. Isbell, Marmaduke John-son, Peter C. Johnston, Robert C. Kent, John J. Kindred, Charles K. Mallory, Jas. B. Mallory, John L. Mayre, Sr., Fleming B. Miller, Horatio G. Moffett, Robert L. Mon-tregue, Edward T. Morris, Jeremiah Morton, William J. Neblett, Johnson Orrick, Wm. C. Parks, Wm. Ballard Preston, George W. R. Radolph, George W. Eichardson, Timothy Rives, Robert E. Scott, William C. Scott, John Y. Seawell, James W. Sheffer, Charles R. Slaughter, Janes Law, Num. M. Treadway, Robert H. Turner, Franklin P. Turner, John Tyler, Edward Waller, Robert Wittfeld, Sam'l C. Williams, Henry A. Wise, Sam'l Woods, Beujamin F. Wysor-SE. Mars, C. B. Conrad, 'Robert Y. Conrad, James H. Couch, WH. B. Custis, Marshal M. Dent, Wm. H. Dulany, Jubal A., Charles, C. Berlin, Caleb Boggess, Geo. W. Brent, Wyrne, John S. Carlile, John A. Carter, Sherrard Clem-ens, C. B. Conrad, 'Robert Y. Conrad, James H. Couch, WH. B. Custis, Marshal M. Dent, Wm. H. Dulany, Jubal A. Earley, Colbert C. Fugate, Peyton Gravely, Algernon Seray, Ephraim B. Hall, Allen C. Hammond, Alpher Haymond, James W. Hodge, J. G. Hollady, Chester H. Hubbard, George W. Hall, John J. Jackson, John F.

Same day the Commissioners adopted and ratified the Constitution of the Provisional Government of the "Confederate" States of America, this ordinance to cease to have legal effect if the people of Virginia voting upon the Ordinance of Secession should reiect it.

Lewis, Wm. McComas, James C. McGrew, James Mar-shall, Henry M. Masters, Samuel McD. Moore, Hugh M. shaii, Henry M. Masters, Samuel McD. Moore, Hugh M. Nelson, Logan Osburn, Spicer Patrick, Edmund Pendle-ton, George McC. Porter, Samuel Price, David Pugh, John D. Sharp, Thomas Sitlington, Burwell Spurloch, Alex-ander H. H. Stuart, Chapmau J. Stuart, Geo. W. Sum-mers, Campbell Tarr, William White, William C. Wick-ham, Waitman T. Willey, Johu Janney (President)-55. Excused-Benjamin Wilson. The following mombute appear not to have been pres-

The following members appear not to have been pres-

eti, as there is no tally opposite their names in the list from which we quole: Alfred Barbour, Robert E. Grant, Addison Hale, John R. Kilby, Paul McNeil, John Q. Marr, Thomas Martin, Peter Saunders, Sen.

Pending the vote on ratifying the Ordinance of Seces-sion, senator J. M. Mason published the following letter: Winchester, Virginia, May 16, 1861.

Winchester, Firginia, May 16, 1861. To the Editor of the Winchester Virginia. The question has been frequently put to me, what po-sition will Virginia occupy should the Ordinance of Se-cession be rejected by the people at the approaching election? And the frequency of the question may be an excuse for giving publicity to the answer. The Ordinance of Secession withdrew the State of Vir-ginia from the Union, with all the consequences resulting from the separation. It auxulted the Constitution and the laws of the United States within the limits of this State, and absolved the citizens of Vircinia from all obli-

State, and absolved the citizens of Virginia from all obli-

State, and absolved the citizens of virginia from all obli-gations of obedience to them. Hence it follows, if this Ordinance be rejected by the people, the State of Virginia will remain bonnd by the Con-stitution of the United States, and obedience to the Gov-ernment and laws of the United States will be fully and rightfully enforced against them. It follows, of course, that in the war now carried on by the Government of the United States acainst the second

the Government of the United States against the seeded States, Virginia must immediately change sides, and under the orders of that Government turn her arms against her southern sisters.

from this there can be no escape. As a member of the Union, all her resources of men and money will be at once at the command of the Government of the Union.

Again : for mutual defence, immediately after the Ordi-nance of Secession passed, a treaty or "military league" was formed by the convention, in the name of the people of Virginia, with the "Confederate States" of the South, by which the latter were bound to march to the aid of our State against the invasion of the Federal Government our State against the invasion of the Federal Government. And we have now in Uriginia, at Harper's Ferry and at Norfolk, in face of the common foe, several thousand of the galant sons of South Carolina, of Alabama, of Lou-isiana, Georgia, and Mississippi, who hastened to fulfil the covenant they made, and are ready and eager to lay down their lives, side by side with our sons, in defence of the soil of Virginia.

If the Ordinance of Secession is rejected, not only will this "military league" be annulled, but it will have been made a trap to inveigle our generous defenders into the hands of their enemies.

Virginia remaining in the Union, duty and loyalty to her obligations to the Union will require that those southern forces shall not he permitted to leave the State, but shall be delivered up to the government of the Union ; and those who refuse to do so will be guilty of treason, and be justly dealt with as traitors.

Treason against the United States consists as well "in adhering to its enemies and giving them aid," as in levying war.

If it be asked, What are those to do who in their con-sciences cannot vote to separate Virginia from the United States? the answer is simple and plain. Honor and daty alike require that they should not vote on the question; and if they retain such opinions, they must leave the State.

None can doubt or question the truth of what I have written, and none can vote against the Ordinance of Se-cession who do not thereby (whether ignorantly or other-wise) vote to place himself and his State in the position I have indicated. J. M. Masox.

25th. A Convention was made between Commissioners of Virginia, chosen by the Convention, and A. H. Stephens, Commissioner for "Confederates," stipulating that Virginia until she became a member of the Confederacy should place her military force under the direction of the President of the "Confederate" States; also turn over to "Confederate" States all her public property, naval stores, and munitions of war. Signed by J. Tyler, W. B. Preston, S. McD. Moore, Jas. P. Holcombe, Jas. C. Bruce, Lewis E. Harvie-for Virginia; and A. H. Stephens for "Confederate" States.

June 25th. Secession vote announced as 128,884 for, and 32,134 against.

July. The Convention passed an ordinance to the effect, that any citizen of Virginia holding office under the Government of the United States after the 31st of July, 1861, should be forever banished from the State, and be declared an alien enemy. Also that any citizen of Virginia, hereafter undertaking to represent the State of Virginia in the Congress of the United States, should in addition to the above penalties be considered guilty of treason and his property be liable to confiscation. A provision was inserted ex-empting from the penalties of the act all officers of the United States outside of the United States, or of the Confederate States, until after July 1st, 1862.

#### KENTUCKY.

December 12th, 1860. Indiana militia offer their services to quell servile insurrection. Gov. Magoffin declines accepting them.

January 17th, 1861. Legislature convened.

22d. The House by a vote of 87 to 6 resolved to resist the invasion of the South at all hazards.

27th. Legislature adopted the Virginia resolutions requiring the Federal Government to protect Slavery in the Territories and to guarantee the right of transit of slaves through the Free States.

February 2d. The Senate passed by a vote of 25 to 11, resolutions appealing to the Southern States to stop the revolution, protesting against Federal coercion and providing that the Legislature reassemble on the 24th of April to hear the responses from sister States, also in favor of making an application to call a National Convention for proposing amendments to the Constitution of the United States, also by a vote of 25 to 14 declared it inexpedient at this time to call a State Convention. 5th. The House by a vote of 54 to 40

passed the above resolutions.

March 22d. State Rights Convention as-Adopted resolutions denouncing sembled. any attempt on the part of the Government to collect revenue as coercion; and affirming that, in case of any such attempt, the border States should make common cause with the Southern Confederacy. They also recommended a border State Convention.

April 24th. Gov. Magoffin called an extra session of the Legislature.

May 20th. Gov. Magoffin issued a neutrality proclamation.

September 11th. The House of Representatives by a vote of 71 to 26, adopted a re-solution directing the Governor to issue a proclamation ordering the Confederate troops to evacuate Kentucky soil. The Governor vetoed the resolution, which was afterwards passed over his veto, and accordingly he issued the required proclamation.

October 29th. Southern Conference met at Russellville. H. C. Burnett elected Chairman, R. McKee Secretary, T. S. Bryan Assistant Secretary. Remained in secret session two days and then adjourned sine die. A series of resolutions reported by G. W. Johnson were adopted. They recite the unconstitutional and oppressive acts of the Legislature, proclaim revolution, provide for a Sovereignty Convention at Russellville, on the 18th of November, recommend the organization of county guards, to be placed in the service of and paid by the Confederate States Government; pledge re-sistance to all Federal and State taxes, for the prosecution of the war on the part of the United States; and appoint Robert McKee, John C. Breckinridge, Humphrey Mar-shall, Geo. W. Ewing, H. W. Bruce, Geo. B. Hodge, Wm. Preston, Geo. W. Johnson, Blanton Duncan, and P. B. Thompson to carry out the resolutions.

November 18th. Convention met and remained in session three days.

20th. It passed a Declaration of Independence and an Ordinance of Secession. A Provisional Government consisting of a Governor, Legislative Council of ten, a Treasurer, and an Auditor were agreed upon. Geo. W. Johnson was chosen Governor. Legislative Council were : Willis B. Machen, John W. Crockett, James P. Bates, Jas. S. Chrisman, Phil. B. Thompson, J. P. Burnside, H. W. Bruce, J. W. Moore, E. M. Bruce, Geo. B. Hodge.

#### MARYLAND.

November 27th, 1860. Gov. Hicks declined to call a special session of the Legislature, in response to a request for such convening from Thomas G. Pratt, Sprigg Harwood, J. S. Franklin, N. H. Green, Llewellyn Boyle, and J. Pinkney.

December 19th. Gov. Hicks replied to A. H. Handy, Commissioner from Mississippi, declining to accept the programme of Secession.

20th. Wm. H. Collins, Esq., of Baltimore, issued an address to the people, in favor of the Union, and in March a second address.

31st. The "Clipper" denied the existence of an organization in Maryland to prevent

the inauguration of President Lincoln. A. H. Handy of Mississippi addressed citizens of Baltimore in favor of disunion.

January 3d, 1861. Henry Winter Davis issued an address in favor of the Union.

3d. Numerous Union meetings in various parts of the State. Gov. Hicks issued an address to the people against secession.

11th. John C. Legrand in a letter to Hon. Reverdy Johnson replied to the Union speech of the latter.

14th. James Carroll, former Democratic candidate for Governor, announced his desire to go with the seceding States.

16th. Wm. A. Spencer, in letter to Wal-ter S. Cox, Esq., declared against the right of Secession but for a Convention.

16th. Marshal Kane, in a letter to Mayor Berrett, denied that any organization exists to prevent the inauguration of President Lincoln, and said that the President elect would need no armed escort in passing through or sojourning within the limits of Baltimore and Maryland. 24th. Coleman Yellott declared for a

Convention

30th. Messrs. John B. Brooke, President of the Senate, and E. G. Kilbourn, Speaker of the House of Delegates, asked the Governor to convene the Legislature in response to public meetings. Scnator Kennedy published his opinion that Maryland must go with Virginia.

February 18th. State Conference Convention held, and insisted upon a meeting of the Legislature. At a meeting in Howard Co., which Speaker E. G. Kilbourn addressed, a resolution was adopted that "immediate steps ought to be taken for the establishment of a Southern Confederacy, by consultation and co-operation with such other Southern and Slave States as may be ready therefor.'

April 21st. Gov. Hicks wrote to Gen. Butler, advising that he do not land his troops at Annapolis. Butler replied that he intended to land there and march thence to Washington. Gov. Hicks protested against this and also against his having taken forcible possession of the Annapolis and Elkridge railroad.

24th. A special election of ten delegates to the Legislature took place at Baltimore. The total vote cast in all the wards was 9,249. The total vote cast at the Presidential election in November, 1860, was 30,148.

26th. Legislature reassembled at Frederick, Annapolis being occupied by Union troops.

29th. Gov. Hicks sent a message to the Legislature communicating to them the correspondence between himself and Gen. Butler and the Secretary of War relative to the landing of troops at Annapolis. The House of Delegates voted against

Secession, 53 to 13. Senate unanimously. May 2d. The Committee on Federal Relations, "in view of the seizure of the railroads by the General Government and the erection of fortifications," presented resolutions appointing Commissioners to the President to ascertain whether any becoming from the United States Government.

arrangements with the General Government are practicable, for the maintenance of the peace and honor of the State and the security of its inhabitants. The report was adopted, and Otho Scott, Robt. M. McLane. and Wm. J. Ross were appointed such Commissioners.

Mr. Yellott in the Senate introduced a bill to appoint a Board of Public Safety. The powers given to the Board included the expenditure of the two millions of dollars proposed by Mr. Brune for the defence of the State, and the entire control of the military, including the removal and appointment of commissioned officers. It was ordered to a second reading by a vote of 14 to 8. The Board was to consist of Ezekiel F. Chambers, Enoch Louis Lowe, John V. L. MacMahon, Thomas G. Pratt, Walter Mitchell, and Thomas Winans. Gov. Hicks was made *ex-officio* a member of the Board. This measure was strongly pressed by the Dis-unionists for a long time, but they were finally compelled to give way, and the bill never passed.

The Commissioners reported the 6th. result of their interview with the President, and expressed the opinion that some modification of the course of the General Government towards Maryland ought to be expected.

The House of Delegates passed a 10th. series of resolutions reported by the Committee on Federal Relations by a vote of 43 to 12. The resolutions declare that Maryland protests against the war, and does earnestly beseech and implore the President of the United States to make peace with the

" Confederate" States; also, that "the State of Maryland desires the peaceful and immediate recognition of the independence of the Confederate States." Those who voted in the negative are Messrs. Medders, Law-son, Keene, Routzahn, Naill, Wilson of Harford, Bayless, McCoy, Fiery, Stake, McCleary, and Gorsuch.

13th. Both Houses adopted a resolution providing for a committee of eight members, (four from each House) to visit the President of the United States and the President of the Southern Confederacy. The committee to visit President Davis were instructed to convey the assurance that Maryland sympathizes with the Confederate States, and that the people of Maryland are enlisted with their whole hearts on the side of reconciliation and peace.

June 11th. Messrs. McKaig, Yellott and Harding, Commissioners to visit President Davis, presented their report; accompanying which is a letter from Jefferson Davis, expressing his gratification to hear that the State of Maryland was in sympathy with themselves, was enlisted on the side of peace and reconciliation, and avowing his perfect willingness for a cessation of hostilities, and a readiness to receive any proposition for peace

20th. The House of Delegates, and June 22d, the Senate adopted resolutions unqualifiedly protesting against the arrest of Ross Winans and sundry other eitizens of Maryland, as an "oppressive and tyrannical assertion and exercise of military jurisdiction within the limits of Maryland, over the persons and property of her citizens, by the Government of the United States."

#### MISSOURI.

January15th,1861. Senate passed Convention Bill-yeas 31, nays 2. Passed House also.

February 28th. Convention met; motion to go into secret session, defeated. A resolution requiring members to take an oath to support the Constitution of the United States and the State of Missouri, was lost— 65 against 30.

March 4th. Resolution passed, 64 yeas, 35 nays, appointing committee to notify Mr. Glenn, Commissioner of Georgia, that the Convention was ready to hear any communication from his State. Mr. Glenn was introduced, read Georgia's articles of secession, and made a speech urging Missouri to join her.

5th. Resolutions were read, ordering that the protest of St. Louis against coercion be reduced to writing, and a copy sent to the President of the United States; also, resolutions were adopted informing Commissioner from Georgia that Missouri dissented from the position taken by that State, and refused to share the honors of secession with her.

6th. Resolutions were offered by several members and referred, calling a Convention of the Southern States which have not seceded, to meet at Nashville, April 15th, providing for such amendments to the Constitution of the United States as shall secure to all the States equal rights in the Union, and declaring strongly against secession.

9th. The Committee on Federal Relations reported a series of resolutions, setting forth that at present there is no adequate cause to impel Missouri to leave the Union, but that on the contrary she will labor for such an adjustment of existing troubles as will secure peace and the rights and equality of all the States; that the people of Missouri regard the amendments to the Constitution proposed by Mr. Crittenden, with their extension to territory hereafter to be acquired, a basis of adjustment which would forever remove all difficulties; and that it is expedient for the Legislature to call a Convention for proposing amendments to the Constitution.

The Senate passed resolutions that their Senators be instructed, and their Representatives requested, to oppose the passage of all acts granting supplies of men and money to coerce the seceding States into submission or subjugation; and that, should such acts be passed by Congress, their Senators be instructed, and their Repre-

20th. The House of Delegates, and sentatives requested, to retire from the halls me 22d, the Senate adopted resolutions of Congress.

16th. An amendment to the fifth resolution of the majority report of the Committee on Federal Relations, asserting that Missouri would never countenance nor aid a seceding State in making war upon the General Government, nor provide men and money for the purpose of aiding the General Government to coerce a seceding State, was voted down.

*Resolved*, That it is inexpedient for the General Assembly to take any steps for calling a National Convention to propose amendments to the Constitution, as recommended by the State Convention.

July 22d. The Convention reassembled.

23d. Resolution passed, by a vote of 65 to 21, declaring the office of President, held by General Sterling Price at the last session of the Convention, vacant. A committee of seven were appointed to report what action they deem it advisable to take in the dislocated condition of the State.

25th. The committee presented their report. It alludes at length to the present unparalleled condition of things, the reckless course of the recent Government, and flight of the Governor and other State officers from the capital. It declares the offices of Governor, Lieutenant-Governor, and Secretary of State vacant, and provides that their vacancies shall be filled by the Convention, the officers so appointed to hold their positions till August, 1862, at which time it provides for a special election by the people. It repeals the ninth section of the sixth article of the Constitution, and provides that the Supreme Court of the State shall consist of seven members; and that four members, in addition to the three now comprising the Court, shall be appointed by the Governor chosen by this Convention to hold office till 1862, when the people shall decide whether the change shall be permanent. It abolishes the State Legislature, and ordains that in case, before the 1st of August, 1862, the Governor chosen by this Convention shall consider that the public exigencies demand, he shall order a special election for members of the State Legislature. It recommends the passage of an ordinance repealing the following bills, passed by the Legislature in secret session, in May last: The military fund bill, the bill to suspend the distribution of the school fund, and the bill for cultivating friendly relations with the Indian tribes. It repeals the bill authorizing the appointment of one major-general of the Missouri militia, and revives the militia law of 1859.

A resolution was passed that a committee of seven be appointed by the President to prepare an address to the people of the State of Missouri.

November 26th. Jefferson Davis transmit-

ted to the "Confederate" Congress a message | met at Montgomery, Alabama, February concerning the secession of Missouri. It was | 4th, 1861, to organize a Southern Confederaccompanied by a letter from Governor Jackson, and also by an act dissolving the union with the United States, and an act ratifying the Constitution of the Provisional Government of the Confederate States; also, the Convention between the Commissioners of Missouri and the Commissioners of the Confederate States. Congress unanimously ratified the Convention entered into between the Hon. R. M. T. Hunter for the rebel Government and the Commissioners for Missouri.

#### Inter-State Commissioners.

The seceding States, as part of their plan of operation, appointed Commissioners to visit other slaveholding States. They were as follows, as announced in the newspapers :

#### SOUTH CAROLINA.

- To Alabama, A. P. Calhoun.
- To Georgia, James L. Orr, Ex-M.C.
- To Florida, L. W. Spratt.
- To Mississippi, M. L. Bonham, Ex-M. C.
- To Louisiana, J. L. Manning.
- To Arkansas, A. C. Spain. To Texas, J. B. Kershaw.
- To Virginia, John S. Preston.

#### ALABAMA.

- To North Carolina, Isham W. Garrett. To Mississippi, E. W. Pettus. To South Carolina, J. A. Elmore. To Maryland, A. F. Hopkins. To Virginia, Frank Gilmer. To Tennessee, L. Pope Walker. To Kentucky, Stephen F. Hale. To Arkansas, John Anthony Winston.

#### GEORGIA.

To Missouri, Luther J. Glenn. To Virginia, Henry L. Benning.

#### MISSISSIPPI.

- To South Carolina, C. E. Hooker. To Alabama, Jos. W. Matthews, Ex-Gov. To Georgia, William L. Harris.
- To Louisiana, Wirt Adams. To Texas, H. H. Miller.
- To Arkansas, George R. Fall.
- To Florida, E. M. Yerger.
- To Tennessee, T. J. Wharton, Att'y-Gen. To Kentucky, W. S. Featherstone, Ex-M.C. To North Carolina, Jacob Thompson, Ex-
- M. C.

  - To Virginia, Fulton Anderson. To Maryland, A. H. Handy, Judge.
  - To Delaware, Henry Dickinson.
  - To Missouri, ---- Russell.

#### Southern Congress.

This body, composed of Deputies elected by the Conventions of the Seceding States,

acy. Each State had a representation equal to the number of members of the Thirty-sixth Congress. The members were:

#### SOUTH CAROLINA.

Robert W. Barnwell, Ex-U. S. Senator. R. Barnwell Rhett, 66 " " James Chesnut, Jr., Lawrence M. Keitt, Ex-M. C. William W. Boyce, " Wm. Porcher Miles, " C. G. Memminger. Thomas J. Withers.

#### ALABAMA.

W. P. Chilton. Stephen F. Hale. David P. Lewis. Thomas Fearn. Richard W. Walker. Robert H. Smith. Colin J. McRae. John Gill Shorter. J. L. M. Curry, Ex-M. C.

#### FLORIDA.

J. Patten Anderson, Ex-Delegate from Washington Territory. Jackson Morton, Ex-U. S. Senator. James Powers.

#### MISSISSIPPI.

W. S. Wilson. Wiley P. Harris, Ex-M. C. James T. Harrison. Walter Brooke, Ex-U. S. Senator. William S. Barry, Ex-M. C. A. M. Clayton.

#### GEORGIA.

Robert Toombs, Ex-U. S. Senator. Howell Cobb, Ex-M. C. Martin J. Crawford, 44 Augustus R. Wright, " Augustus H. Keenan. Benjamin H. Hill. " Francis S. Bartow. E. A. Nisbet. Thomas R. R. Cobb. Alexander H. Stephens, Ex-M. C. LOUISIANA.

Duncan F. Kenner. Charles M. Conrad, Ex-U. S. Senator. Henry Marshall. John Perkins, Jr. G. E. Sparrow. E. de Clouet.

TEXAS.-(Admitted March 2d, 1861.) Louis T. Wigfall, Ex-U. S. Senator. John Hemphill, John H. Reagan, Ex-M. C T. N. Waul. John Gregg. W. S. Oldham. W. B. Ochiltree.

#### Proceedings of the Southern Congress.

February 4th, 1861. Howell Cobb of Georgia elected President, Johnson J. Hooper of Alabama, Secretary. Mr. Cobb announced that secession "is now a fixed and irrevocable fact, and the separation is perfect, complete and perpetual.

6th. David L. Swain, M. W. Ransom and John L. Bridgers, were admitted as Commissioners from North Carolina, under resolutions of the General Assembly of that State, passed January 29, 1861, "to effect an honorable and amicable adjustment of all the difficulties that disturb the country, upon the basis of the Crittenden resolutions, as modified by the Legislature of Virginia," and to consult with the delegates to the Southern Congress for their "common peace, honor and safety."

7th. Congress notified that the State of Alabama had placed \$500,000 at its disposal, as a loan to the provisional government of the Confederacy of Seceding States.

8th. The Constitution of the Provisional Government adopted.\*

\* The Provisional Constitution adopted by the Secended States differs from the Constitution of the United States in several important particulars. The alterations and additions are as follows:

#### ALTERATIONS.

1st. The Provisional Constitution differs from the other In this: That the legislative powers of the Provisional Gov-ernment are vested in the Congress now assembled, and this body exercises all the functions that are exercised by either or both branches of the United States Government, 2d. The Provisional President holds his office for one year, unless sooner superseded by the establishment of a

permanent Government

3d. Each State is erected into a distinct judicial district, the judge having all the powers heretofore vested in the district and circuit courts; and the several district judges together compose the supreme bench-a majority of them constituting a quorum.

4th Wherever the word "Union" occurs in the United States Constitution the word "Confederacy" is substitnted.

#### THE FOLLOWING ARE THE ADDITIONS.

alst. The President may veto any separate appropria-tion without vetoing the whole bill in which it is contained.

2d. The African slave-trade is prohibited. 3d. Congress is empowered to prohibit the introduction of slaves from any State not a member of this Confederacy. 4th. All appropriations must be upon the demand of the President or heads of departments.

#### OMISSIONS,

lst. There is no prohibition on members of Congress holding other offices of honor and emolument under the Provisional Government.

2d. There is no provision for a neutral spot for the loca-tion of a seat of government, or for sites for forts, arsenals, and dock-yards; consequently there is no reference made to the territorial powers of the Provisional Government. 3d. The section in the old Constitution in reference to

capitation and other direct tax is omitted ; also, the section providing that no tax or duty shall be laid on any exports.

4th. The prohibitiou on States keeping troops or ships of war in time of peace is omitted.

5th. The Constitution being provisional merely, no pro-vision is made for its ratification.

#### AMENDMENTS.

1st. The fugitive slave clause of the old Constitution is Bo amended as to contain the word "slave," and to pro-vide for full compensation in cases of abduction or forci-ple rescue on the part of the State in which such abduc-tion or rescue may take place.

9th. Jefferson Davis of Mississippi, elected Provisional President of the Confederate States of America, and Alexander H. Stephens of Georgia, Vice President. The question of attacking Fort Sumter has been referred to the Congress.

11th. Mr. Stephens announced his accept-Committee appointed to prepare a ance. permanent Constitution.

12th. The Congress assumed "charge of all questions and difficulties now existing between the sovereign States of this Confederacy and the Government of the United States, relating to the occupation of forts, arsenals, navy yards, custom-houses, and all other public establishments." The resolution was directed to be communicated to the Governors of the respective States of the Confederacy.

15th. Official copy of the Texas Ordinance of Secession presented.

16th. President Davis arrived and received with salute, etc.

18th. President Davis inaugurated.19th. Tariff law passed.21st. Robert Toombs appointed Secretary of State; C. G. Memminger, Secretary of the Treasury; L. Pope Walker of Alabama, Secretary of War; Stephen R. Mallory, Secretary of the Navy; Judah P. Benjamin, Attorney-General, and John H. Reagan. Postmaster-General; Philip Clayton of Concrete approximated Acceptor Georgia appointed Assistant Secretary of the Treasury, and Wm. M. Browne, late of the Washington Constitution, Assistant Secretary of State. March 2d. The Texas Deputies received.

#### The Justifying Causes of Secession.

In justification of the passage of an ordinance of Secession, the Convention of South Carolina adopted two papers, one reported by Mr. R. B. Rhett, being styled "The Address of the people of South Carolina, assembled in Convention, to the people of the Slaveholding States of the United States,' and the other, reported by Mr. C. G. Memminger, being styled "Declaration of the causes which justify the Secession of South Carolina from the Federal Union." As these official papers have historic value, they are inserted in full.

The former of these two papers is as follows:

It is now seventy-three years since the union hetween the United States was made by the Constitution of the

2d. Congress, by a vote of two-thirds, may at any time alter or amend the Constitution.

#### TEMPORARY PROVISIONS.

1st. The Provisional Government is required to take immediate steps for the settlement of all matters between the States forming it and their other late confederates of the United States in relation to the public property and

the public deht. 2d. Montgomery is made the temporary seat of government

3d. This Constitution is to continue one year, unless altered by a two-thirds vote or superseded by a permanent Government.

#### 12

United States. During this period their advance in ation without a representation adequate to protection, wealth, prosperity, and power, has been with scarcely there was no difference. By neither would the colonies a parallel in the history of the world. The great object of their union was external defence from the aggressions haid by the British Parliament. of more powerful nations; now complete, from their mere progress in power, thirty-one millions of people, with a commerce and navigation which explores every sea, and of agricultural productions which are necessary to every civilized people, command the friendship of the world. But, unfortunately, our internal peace has not grown with our external prosperity. Discontent and contention has moved in the bosom of the Confederacy contention has moved in the bosom of the Confederacy for the last thirty-five years. During this time South Carolina has twice called her people together in solemn convention, to take into consideration the aggressions and unconstitutional wrongs perpetrated by the people of the North on the people of the South. These wrongs were submitted to by the people of the South, under the hope and expectation that they would be final. But these hopes and expectations have proved to be void. Instead of being incentives to forbearance, our submis-sion has only instigated to new forms of aggressions and outrage, and South Carolina, again assembling her people in convention, has this day dissolved her connection with the States constituting the United States. con-

The one great evil from which all other evils have flowed, is the overthrow of the Constitution of the United States. The Government of the United States is no longer the government of a confederate republic, Is no longer the government of a confederate republic, but of a consolidated democracy. It is no longer a free government, but a despotism. It is, in fact, such a government as Great Britain attempted to set over our fathers, and which was resisted and defeated by a seven years struggle for independence.

even years struggle for independence. The revolution of 1776 turned upon one great princi-ple, self-government, and self-taxation the criterion of self-government. Where the interests of two people united together under one Government are different, each must have the power to protect its interests by the organization of the Government, or they cannot be free. The interests of Great Britain and of the colo-nies were different and antagonistic. Great Britaiu was desirous of carrying out the policy of all nations toward their colonies of making them tributary to their was desirous of carrying out the policy of all nations toward their colonies of making them tributary to their wealth and power. She had vast and complicated rela-tions with the whole world. Her policy toward her North American colonies was to identify them with her in all these complicated relations, and to make them bear, in common with the rest of the empire, the full burden of her obligations and necessities. She had them bear, in common with the rest of the empire, the full burden of her obligations and necessities. She had a vast public debt; she had a European policy and an Asiatic policy, which had occasioned the accumulation of her public debt, and which kept her in continual wars. The North American colonies saw their inter-ests, policical and commercial, sacrificed by such a policy. Their interests required that they should not be identified with the burdens and wars of the mother country. They had been settled under charters which gave them self-government, at least so far as their property was concerned. They had taxed themselves, and had never been taxed by the Government of Great Britain. To make them a part of a consolidated en-pire the Parliament of Great Britain determined to as-sume the power of legislating for the colonies in all Government of Great Britain. The Southern States had on ancestors in the colonies did toward Great Britain. The North-ern States, having the majority in Congress, claim the Det

position toward the Northern States that our ancestors in the colonies did toward Great Britain. The North-ern States, having the majority in Congress, claim the same power of omnipotence in legislation as the Brit-ish Parliament. "The general welfare" is the only limit to the legislation of either; and the majority in Congress, as in the British Parliament, are the sole judges of the expediency of the legislation this "general welfare". requires. Thus the Government of the United States has become a consolidated Govern-ment, and the people of the Southern States are com-pelled to meet the very despotism their fathers threw off in the Revolution of 1776. The consolidation of the Government of Great fixin over the colonies was attempted to be carried out by the taxes. The British Parliament undertook to tax the colonies to promote British interests. Our fathers resisted this pretension. They claimed the right of self-taxation through their Colonial Legisla-tures. They were not represented in the British Par-tiament; and therefore could not rightfully be taxed by its Legislature. The British Government, how-over, offered them a representation in the British Par-tiament; but it was not sufficient to enable them to protect themselves from the majority, and they refused it.

laid by the British Parliament. The Southern States now stand in the same relation toward the Northern States, in the vital matter of taxation, that our ancestors stood toward the peo-ple of Great Britain. They are in a minority in Con-gress. Their representation in Congress is useless to protect them against unjust taxation, and they are taxed by the people of the North for their benefit ex-actly as the people of Great Britain taxed our ancestors in the British Parliament for their benefit. For the last forty years the taxes laid by the Congress of the United States have been laid with a view of subserving the interests of the North. The people of the South have the interests of the North. The people of the South have been taxed by duiles on imports not for revenue, but for an object inconsistent with revenue-to promote, by prohibitions, Northern interests in the productions of their mines and manufactures.

There is another evil in the condition of the Southern toward the Northern States, which onr ancestors re-fused to bear toward Great Britian. Our ancestors not only taxed themselves, but all the taxes collected from them were expended among them. Had they submitted to the pretensions of the British Government, the ted to the pretensions of the British Government, the taxes collected from them would have been expended on other parts of the British Empire. They were fully aware of the effect of such a policy in impoverishing the people from whom taxes are collected, and in enriching those who receive the benefit of their ex-penditure. To prevent the evils of such a policy was one of the motives which drove them on to revolution. Not this British solica hose hose fully realized taward Yet this British policy has been fully realized toward the Southern States by the Northern States. The people of the Southern States are not only taxed for the benefit of the Northern States, but after the taxes are collected three-fourths of them are expended at the North. This cause, with others connected with the operation of the General Government, has provincial-ized the cities of the South. Their growth is paralyzed, while they are the mere suburbs of Northern cities. The bases of the foreign commerce of the United States are the accounting productions of the South wat The bases of the foreign commerce of the United States are the agricultural productions of the South; yet Southern cities do not carry it on. Our foreign trade is almost annihilated. In 1740 there were five ship-yards in South Carcolina to build ships to carry on our direct trade with Europe. Between 1740 and 1779 there were built in these yards twenty-five square-rigged vessels, beside a great number of sloops and schooners to carry on our coast and West India trade. In the half century immediately preceding the Revolu-tion, from 1725 to 1775, the population of South Caro-lina increased seven-fold. No ma can for a moment believe that our ancestors

No man can for a moment believe that our ancestors No man can for a moment believe that our abcessors intended to establish over their posterity exactly the same sort of Government they had overthrown. The great object of the Constitution of the United States, in its internal operation, was, doubtless, to secure the great end of the Revolution—a limited free Govern-ment—a Government limited to those matters only which are the down and the matters of the ment-a Government limited to those matters only which were general and common to all portions of the United States. All sectional or local interests were to be left to the States. By no other arrangement would they obtain free government by a Constitution common to so vast a Confederacy. Yet, by gradual and steady encroachments on the part of the North, and submis-sion on the part of the South, the limitations in the Constitution have been swept away, and the Govern-ment of lumitless powers in its operations It is nottat all surprising. while such is the character

It is not at all surprising, while such is the character of the Government of the United States, that it should assume to possess power over all the institutions of the country. The agitations on the subject of Slavery in the South are the natural results of the consolidation of the Government. Responsibility follows power; and if the people of the North have the power by Congress "to promote the general welfare of the United States," by any means they deem expedient, why should they not assail and overthrow the institution of Slavery in the South? They are responsible for its continuance or existence, in proportion to their power. A majority in Congress, according to their interested and perverted views, is omnipotent. The inducements to act upon the subject of Slavery, nucles such counstances, were so imperious as to amount almost to a moral necessity. To make, however, their numerical power available to rule the Union, the North must consolidate their pow-er. It would not be united on any matter common to the mbole Union. the whole Union—in other words; on any inster constitution-al subject—for on such subjects divisions are as likely to exist in the North as in the South. Slavery was strictly a sectonal interest. If this could be made the criterion of parties at the North, the North could be united in its power, and thus carry ont its measures of sectional ambition, encroachement, and aggrandizement. To build up their sectional predominance in the Union, the Constitution must be first abolished by constructions; but that being done, the consolidation of the North to rule the South, by the tariff and Slavery issnes, was in the obvious course of things.

The Constitution of the United States was an exper-nent. The experiment consisted in uniting under one iment. tment. The experiment consisted in unting under one Government different peoples, living in different cli-mates, and having different pursuits of industry and institutions. It matters not how earefully the limita-tions of such a government be luid down in the consti-tutiou-its success must at least depend upon the good faith of the parties to the constitutional compact in enforcing them. It is not in the power of human language to exclude false inferences, constructions, and language to exclude false inferences, constructions, and perversions, in any constitution; and when vast sec-tional interests are to be subserved involving the ap-propriation of countless millions of money, it has not been the usual experience of mankind that words on parchment can arrest power. The Constitution of the United States, irrespective of the interposition of the States, rested on the assumption that power would yield to faith—that integrity would be stronger than interest, and that thus the limitations of the Constitu-tion would be observed. The experiment has been fairly made. The Southern States, from the commencetion would be observed. The experiment has been fairly made. The Southern States, from the commence-ment of the Government, have striven to keep it with-in the orbit prescribed by the Constitution. The ex-periment has failed. The whole Constitution, by the constructions of the Northern people, has been swal-lowed up by a few words in its preamble. In their reckless lust for power they seem unable to compre-heud that seeming paradox, that the more power is given to the General Government the weaker it be-comes. Its strength consists in its generality and limitations. To extend the scope of its power over sectional or local interests is to raise up against it op-position and resistance. In all such matters the Gen-eral Government must necessarily be a despotism, because all sectional or local interests must ever be rep-resented by a minority in the councils of the General -having up power to protect itself against Governmentthe rule of the majority. The majority, constituted from those who do not represent these sectional or local interests, will control and govern them. A free people caunot submit to such a Government; and the more it enlarges the sphere of its power the greater must be the dissatisfaction it must produce, and the weaker it must become. On the contrary, the more it abstains from usurped powers, and the more faithfully it adheres to the limitations of the Constitution, the stronger it is made. The Northern people have had neither the wisdom nor the faith to perceive that to observe the limitation of the Constitution was the only way to its perpetuity. - 20

Under such a Government there must, of course, he many and endless "irrepressible conflicts," between the two great sections of the Union. The same faithlessness which has abolished the Constitution of the United States, will not fail to carry out the sectional purposes for which it has been abolished. There must be conflict; and the weaker section of the Uuion can ouly find peace and liberty in an independence of the North. The repeated efforts made by South Carolina, in a wise conservatism, to arrest the progress of the General Government in its fatal progress to cousolidation, have been uusapported and denounced as faithcless to the obligations of the Constitution by the very men and States who were destroying it by their usurpations. It is now too late to reform or restore the Government of the United States. All confidence in the North is lost in the South. The faithlessuess of half a century has opened a gulf of separation between them which no promises or engagements can fill.

It cannot he believed that our ancestors would have assented to any union whatever with the people of the North if the feelings and opinions now existing among them had existed when the Constitution was framed. There was then no tariff—no negro fanaticism. It was the delegates from New England who proposed in the Convention which framed the Constitution, to the delegates from South Carolina and Georgia, that if they would agree to give Congress the power of regulating commerce by a majority, that they would support the extension of the African slave-trade for twenty years. African Slavery existed in all the States but one. The dea that they would be made to pay that tribute to their Northern confederates which they had refused to pay to Great Britain, or that the institution of African Slavery would be made the grand basis of a sectional Organization of the North to rule the South, never crossed

their imaginations. The Union of the Constitution was a Union of slaveholding States. It rests on Slavery, by prescribing a representation in Congress for three-fifths of our slaves. There is nothing in the proceedings of the Convention which framed the Constitution to show that the Southern States would have formed any other nnion; and still less that they would have formed a union with more powerful non-slaveholding States, having a majority in both branches of the Legislature of the Government. They were guilty of no such folly. Time and the progress of things have totally altered the relations between the Northern and Southern States since the Union was first established. That identity of feeling, interests, and institutions which once existed is gone. They are now divided between agricultural and manufactaring and commercial States.—between slaveholding and non-slaveholding States, their institutions and industrial pursnits have made them totally different peoples. That equality in the Government between the two sections of the Union which once existed, no longer exists. We hut imitate the policy of our fathers in dissolving a nuion with non-slaveholding States.

Experience has proved that slaveholding States can-not be safe in subjection to non-slaveholding States. Indeed, no people ever expect to preserve their rights and liberties nuless they are in their own custody. To plunder and oppress where plunder and oppression can pinnder and oppress where plander and oppression can be practiced with impunity, seems to be the natural order of things. The fairest portions of the world have been turned into wildernesses, and the most civilized and prosperous communities have been impoverished and ruined by Anti-Slavery fauaticism. The people of and ruined by Anti-Slavery fauaticism. The people of the North have not left us in doubt as to their designs and policy. United as a section in the late Presidential election, they have elected as the exponent of their policy one who has openly declared that all the States of the United States must be made Free States or Slave States. It is true that among those who aided in this States. It is the that along hose who alore in this election, there are various shades of Anti-Slavery hos-tility. But if African Slavery in the Southern States be the evil their political combinations affirm it to be, the requisitions of an inexorable logic must lead them to emancipation. If it is right to preclude or abolish Slavery in a territory, why should it be allowed to remain in the States? The one is not at all more nuconstitutional than the other, according to the decisions of the Supreme Court of the United States. And when it is considered that the Northern States will soon have the power to make that Court what they please, and that the Constitution never has been any barrier whatthat the Constitution never has been any barrier what-ever to their exercise of power, what check can there be in the unrestrained councils of the North to eman-cipation? There is sympathy in association, which carries men along without principle; but when there is principle, and that principle is fortified by long ex-isting prejudices and feelings, association is omnipotent in party influences. In spite of all disclaimers and proin party infinences. In spite of all disclaimers and pro-fessions there can be but one end to the submission by the South to the rule of a sectional Anti-Slavery Gov-ernment at Washington; and that eud, directly or indirectly, must be the emancipation of the slaves of the South. The hyporrisy of thirty years—the faith-lessness of their whole course from the commencement lessness of their whole course from the commencement of our mion with them—show that the people of the non-slaveholding North are not and cannot be safe associates of the slaveholding South under a com-mon Government. Not only their fanaticism, but their erroneous views of the principles of free gov-eroments, reuder it doubtful whether, separated from the South, they can maintain a free Government among themselves. Brute numbers with them is the great elethemselves. Brute numbers with them is the great ele-ment of free Government. A majority is infallible and omnipotent. "The right divine to rule in kings" is only transferred to their majority. The very object of all con-stitutions, in free, popular Governments, is to restrain the majority. Constitutions, therefore, according to their theory, must be most narrighteous inventions, restricting theory, must be most narranteous inventions, restricting likerty. None ought to exist, but the body politic ought simply to have a political organization, to bring ont and enforce the will of a majority. This theory may be harmless in a small community, having an identity of interests and pursuits, but over a vast State—still more, our or need conclusion but over a vast State. interests and pursuits, but over a vast State-still more, over a vast Confederacy, having various and conflicting interests and pursuits-it is a remorseless despotism. In resisting it, as applicable to ourselves, we are vindicating the great cause of free government, more important, per-haps, to the world than the existence of the United States. Nor in resisting it, do we intend to depart from the safe instrumentality the system of government we have established with them requires. In separating from them we invade no rights-mo interest of theirs. We violate no obligation of duty to them. As separations independent States in Convention, we made the Constitu-We violate no obligation of duty to them. As separate, independent States in Convention, we made the Constitution of the United States with them; and as separate, independent States, each State acting for itself, we adopted it. South Carolina, acting in her sovereign capacity, now thinks proper to seecede from the Union. She did not part with her sovereignty in adopting the Constitution. The last thing a State can be, presumed to have surrendered is her sovereignty. Her sovereignty is her life. Nothing but a clear, express grant, can alienate it. Inference should be dumb. Yet it is not at all surprising that those who have construed away all the limitations of the Constitution, should also by construction claim the annihilation of the sovereignty of the States. Having abolished all barriers to their omnipotence by their faithless constructions in the operations of the General Government, it is most natural that they should endeavor to do the same toward ns in the States. The truth is, they having violated the express provisions of the Constitution, it is at an end as a compact. It is morally obligatory only on those who choose to accept its perverted terms. South Carolina, deeming the compact not only violated in particular features, but virtually abolished by her Northern confederates, withdraws herself as a party from its obligations. The right to do so is deuied by her Northern confederates. They desire to establish a despotism, not only omnipotent in Congress, but omnipotent over the States; and as if to manifest the imperions necessity of our secession, they threaten us with the sword, to coerce submission to their rule.

Citizens of the slaveholding States of the United States, circumstances beyond our courtol have placed us in the van of the great controversy between the Northern and Southern States. We would have preferred that other States should have assumed the position we now occupy. Independent ourselves, we disclaim any design or desire to lead the councils of the other Southern States. Provi-dence has cast our lot together, by extending over us an identity of pursuits, interests, and institutions. South Carolina desires no destiny separated from yours. To be Carolina desires no destiny separated from yours. To be one of a great slaveholding confederacy, stretching its arms over a territory larger than any Power in Europe possesses—with population four times greater than that of the whole United States when they achieved their inde-pendence of the British Empire—with productions which make our existence more important to the world than that of any other people inhabiting it—with common institu-tions to defend, and common dangers to encounter—we ask your sympathy and confederation. While constituting a portion of the United States, it has been your statesmanship which has guided it in its mighty strides to power and since which has guided it its might strides to power and expansion. In the field, as in the Cabinet, you have led the way to its renown and grandeur. You have loved the Union, in whose service your great statesmen have la-bored, and your great soldiers have fought and conquered --not for the material benefits it conferred, but with the faith of a generous and devoted chivalry. You have long lingered and hoped over the shattered remains of a broken Source and the second s the North and the South is lost, or has been converted into hate; and we of the South are at last driven together by the stern destiny which controls the existence of nations. Your bitter experience of the faithlessness and rapacity of your Northern confederates may have been necessary to evolve those great principles of free government, upon which the liherties of the world depend, and to prepare which the liherities of the world depend, and to prepare you for the grand mission of vindicating and re-establish-lishing them. We rejoice that other nations should be satisfied with their institutions. Self-complacency is a great element of happiness, with nations as with indi-viduals. We are satisfied with ours. If they prefer a system of industry in which capital and labor are in per-petual conflict—and chronic starvation keeps down the natural increase of population—and a mai is worked out in eight years—and the law ordains that children shall be worked out her hours a day—and the sabre and bara he worked only ten hours a day—and the sabre and bayo-net are the instruments of order—be it so. It is their affair, not ours. We prefer, however, our system of in-dustry, by which labor and capital are identified in inter-est, and capital, therefore, protects labor; by which our capital days and a construct mome: by which our est, and capital, therefore, protects labor; by which our population doubles every twenty years; by which starva-tion is unknown, and abundance crowns the land; by which order is preserved by an unpaid police, and the most fertile regions of the world where the Caucasian cannot labor are bronght into usefulness by the labor of the African, and the whole world is blessed by our own productions. All we demand of other peoples is to be let alone to work out our own high destinies. United to-ether and we must be the most indemedant as we are gether, and we must be the most independent, as we are the most important among the nations of the world. United together, and we require no other instrument to conquer peace than our beneficent productions. United together, and we must be a great, free and prosperous people, whose renown must spread throughout the civil-

ized world, and pass down, we trust, to the remolest ages. We ask you to join us in forming a confederacy of Slaveholding States.

#### The latter paper is as follows :

DECLARATION OF INDEPENDENCE OF SOUTH CAROLINA, DONE IN CONVENTION, DECEMBER 24, 1860. The State of South Carolina having determined to re-

The State of South Carolina having determined to resume her separate and equal place among nations, deems it due to herself, to the remaining United States of North America, and to the nations of the world, that she should declare the causes which have led to this act.

In the year 1765, that portion of the British Empire embracing Great Britain undertook to make laws for the government of that portion composed of the thirteen American Colonies. A struggle for the right of self-government ensmed, which resulted on the 4th of July, 1776, in a Declaration by the Colonies "that they are, and of right ought to be, free and independent States, and that as free and independent States they have full power to levy war, conclude peace, contract alliances, establish commerce, and to do all other acts and things which independent States may of right do."

They further solemnly declared that whenever any "form of government becomes destructive of the ends for which it was established, it is the right of the people to alter or abolish it, and to institute a new government." Deeming the Government of Great Britain to have become destructive of these ends, they declared that the Colonies "are absolved from all allegiance to the British Grown, and that all political connection between them and the States of Great Britain is, and ought to be totally dissolved."

In pursuance of this Declaration of Independence, each of the thirteen States proceeded to exercise its separate sovereignty; adopted for itself a Constitution and appointed officers for the administration of government in all its departments-melgislative, executive, and judicial. For purposes of defence, they united their arms and their counsels; and, in 1778, they entered into a league known as the Articles of Confederation, whereby they agreed to intrust the administration of their external relations to a common agent, known as the Congress of the United States, expressly declaring, in the first article, "that each State retains its sovereignty, freedom, and independence, and every power, jurisdiction, and right which is not, by this confederation, expressly delegated to the United States is.

Under this confederation the war of the Revolution was carried on, and on the 3d of September, 1783, the contest ended, and a definite treaty was signed by Great Britain, in which she acknowledged the independence of the Colonies in the following terms:

"Article 1. His Britannic Majesty acknowledges the said United States, viz.: New Hampshire, Massachusetts Bay, Rhode Island aud Providence Plantations, Connecticut, New York, New Jersey, Pennsylvania, Delaware, Maryland, Virginia, North Carolina, South Carolina, and Georgia, to be free, sovereign, and independent States; that he treats with them as such; and for himself, his heirs and successors, relinquishes all claims to the government, proprietary, and territorial rights of the same and every part thereof."

Thus was established the two great principles asserted by the Colonies, namely, the right of a State to govern itself, and the right of a people to abolish a government when it becomes destructive of the ends for which it was instituted. And concurrent with the establishment of these principles was the fact that each Colony became and was recognized by the mother country as a free, sovereign, and independent State.

In 1757, deputies were appointed by the States to revise the Articles of Confederation, and on 17th of September, 1757, these deputies recommended for the adoption of the States the articles of union known as the Constitution of the United States.

The parties to whom this Constitution was submitted were the several sovereign States. They were to agree or disagree; and when nine of them agreed, the compact was to take effect among those concurring, and the General Government, as the common agent, was then to be invested with their authority.

Government, as the common agent, was then to be invested with their anthority. If only nine of the thirteen States had concurred, the other four would have remained as they then wereseparate, sovereign Stales, independent of any of the provisions of the Constitution. In fact, two of the States did not accede to the Constitution until long after it had gone into operation among the other eleven, and during that interval they exercised the functions of an independent nation.

By this Constitution, certain duties were charged on the several States, and the exercise of certain of their powers restrained, which necessarily implied their continued existence as sovereign States. But, to remove all doubt, an amendment was added, which declared that the powers not delegated to the United States by the Constitution, nor not delegated to the United States by the Constitution, nor prohibited by it to the States, are reserved to the States respectively, or to the people. On 23d May, 1758, South Carolina, by a Convention of her people, passed an ordi-nance assenting to this Constitution, and afterward altered her own Constitution, to conform herself to the obligations she had undertaken.

Thus was established, by compact between the States, a Government, with defined objects and powers, limited to the express words of the grant, and to so much more only as was necessary to execute the power granted. This limitation teft the whole remaining mass of power subject to the clause reserving it to the States or to the people, and rendered unnecessary any specification of reserved rights.

We hold that the Government thus established is sub-ject to the two great principles asserted in the Declaration of Independence, and we hold further that the mode of its formation subjects it to a third fundamental principle -namely, the law of compact. We maintain that in every -namely, the law of compact. We maintain that the very compact between two or more parties the obligation is mutual—that the failure of one of the contracting parties to perform a material part of the agreement entirely re-leases the obligations of the other, and that, where no arbiter is provided, each party is remitted to his own judgment to determine the fact of failure with all its consequences.

In the present case that fact is established with cer-tainty, We assert that fifteen of the States have deliber-ately refused for years past to fulfil their Constitutional obligations, and we refer to their own statutes for the proof.

The Constitution of the United States, in its fourth article, provides as follows:

"No person held to service or labor in one State, under the laws thereof, escaping into another, shall, in conse-quence of any law or regulation therein, be discharged from such service or labor, but shall be delivered up, on claim of the party to whom such service or labor may be due.

This stipulation was so material to the compact that without it that compact would not have been made. The greater number of the contracting parties held slaves, and the State of Virginia had previously declared her esti-mate of its value by making it the condition of her cession of the territory which now composes the States north of the Ohio river.

The same article of the Constitution stipulates also for the rendition by the several States of fugitives from jns-tice from the other States.

The General Government, as the common agent, passed laws to carry into effect these stipulations of the States. For many years these laws were executed. But an in-For many years these laws were excented. But an in-creasing hostility on the part of the Northern States to the institution of Slavery has led to a disregard of their obligations, and the laws of the General Government have ceased to effect the objects of the Constitution. The States of Maine, New Hampshire, Vermont, Massachusetts, Con-necticut, Rhode Island, New York, Pennsylvania, Illinois, Indiana, Ohio, Michigan, Wisconsin and Iowa have en-acted laws which either nullify the acts of Congress, or render useless any attempt to excente them. In many of these States the furgitive is discharged from the scrue or these States the fugitive is discharged from the service or labor claimed, and in none of them has the Star Covern-ment complied with the stipulation made in the Constitu-tion. The State of New Jersey, at an early day, passed a law for the rendition of fugitive slaves in conformity with her Constitutional indertaking; but the current of anti-slavery feeling has led her more recently to enact anti-slavery feeling has led her more recently to enact laws which render inoperative the remedies provided by her own law and by the laws of Congress. In the State of New York even the right of transit for a slave has been denied by her tribunals, and the States of Ohio and Iowa have refused to surrender to justice fugitives charged with murder and with inciting servile insurrec-tion in the State of Virginia. Thus the Constitutional compact has been deliberately broken and disregarded by the non-slaveholding States, and the consequence follows that South Carolina is released from its obligations.

The ends for which this Constitution was framed are declared by itself to be "to form a more perfect Union, establish justice, insure domestic tranquility, provide for the common defence, protect the general welfare, and se-cure the blessings of liberty to ourselves and our posterity."

These ends it endeavored to accomplish by a Federal Government, in which each State was recognized as an Government, in which each State was recognized as an equal, and had separate control over its own institutions. The right of property in slaves was recognized by giving to free persons distinct political rights; by giving them the right to represent, and burdening them with direct axes for three-fifths of their slaves; by authorizing the

.

importation of slaves for twenty years; and by stipulating for the rendition of fugitives from labor. We affirm that those ends for which this Government was instituted have been defeated, and the Government itself has been made destructive of them by the action of the non-slaveholding States. Those States have assumed the right of deciding npon the propriety of our domestic institutions, and have denied the rights of property es-tablished in fifteen of the States, and recognized by the Constitution; they have denounced as sinful the institu-ion of Decent they have promitted the approximation of blight Constitution; they have denounced as sintui the institu-tion of Slavery; they have permitted the open establish-ment among them of societies whose avowed object is to disturb the peace and to endanger the property of the citi-zens of other States. They have encouraged and assisted thousands of our slaves to leave their homes, and those who remain have been incited by emissaries, books, and pictures, to servile insurrection.

For seventy-five years this agitation has been steadily increasing, until it has now secured to its aid the power of the common Government. Observing the forms of the of the common Government. Observing the forms of the Constitution, a sectional party has found within that ar-ticle establishing the Executive Department the means of subverting the Constitution itself. A geographical line has been drawn across the Union, and all the States north of that line have united in the election of a man to the better the subscription of the high office of President of the United States, whose opinions and purposes are hostile to Slavery. He is to be intrusted with the administration of the common Government, because he has declared that that "Governernment, because he has declared that that "Govern-ment cannot endure permanently half slave, half free," and that the public mind must rest in the belief that Sla-very is in the course of ultimate extinction. This sectional combination for the subversion of the Constitution has been aided in some of the States by ele-vation to exitemation persons who, by the supreme have of

Constitution has been aided in some of the States by ele-vating to citizenship persons who, by the supreme law of the land, are incapable of becoming citizens, and their votes have been used to inangurate a new policy hostile to the South, and destructive of its peace and safety. On the 4th of March next this party will take posses-sion of the Government. It has announced that the South shall be excluded from the common territory; that the judicial tribunals shall be made sectional, and that a war must be waged against Slavery until it shall cease throughout the United States. The organates of the Constitution will then no longer

The guarantees of the Constitution will then no longer exist; the equal rights of the States will be lost. The Slaveholding States will no longer have the power of self-government or self-protection, and the Federal Gov-ernment will have become their enemies.

Sectional interest and animosity will deepen the irrita-tion, and all hope of remedy is rendered vain by the fact that public opinion at the North has invested a great political error with the sanctions of a more erroneous rellgious belief.

gious belief. We, therefore, the people of South Carolina, by our de-legation in Convention assembled, appealing to the Su-preme Judge of the world for the rectitude of our inten-tions, have solemnly declared that the Union heretofore existing between this State and the other States of North America is dissolved, and that the State of South Carolina has resumed her position among the nations of the world, or other correspondent of the states and the form has resume aner position among the nations of the world, as a free, sovereign, and independent State, with full power to levy war, conclude peace, contract alliances, establish commerce, and to do all other acts and things which independent States may of right do. And, for the support of this declaration, with a firm

reliance on the protection of Divine Providence, we mutually pledge to each other our lives, our fortunes, and our sacred honor.

The debate on the adoption of these papers discloses some interesting facts, and is subjoined.

Upon Mr. Memminger's declaration being read, Mr. Furman and Mr. Inglis raised questions as to the accuracy of certain statements, the former as to whether New Jersey had, as alleged, voted for a "sectional candidate," and the latter as to the allega-tion that Pennsylvania had on her statutebook a "personal liberty law."

Mr. INGLIS said: They (Pennsylvania) have what they call a law to prevent kidnapping, nearly similar to the law of Virginia, which law, owing to the con-dition of public sentiment in Pennsylvania, has no doubt been perverted to this purpose. A document of this kind, and proceeding from a body like this, ought to be exactly accurate in its statements. I should like to ask the Chairman of the Committee if he has satisfied himself with regard to the fact that

nere is any such law as this on the statute-book of ennsylvania? If he has, why then I am satisfied. Mr. MEMMINGER. In reply to the gentleman I ould say that I hold in my hand an elaborate report ade on this point by a Committee of the Legisla-rre of Virginia in which the laws of each State are rofessed to be correctly stated. Mr. INGLIS. Will the gentleman give me the date that report?

Mr. MEMMINGR. It was made at the last session,

nuary 26th, 1860. Mr. INGLIS. To what law do they refer? for Penn-lyania has recently revised her criminal code, and, Munderstand, has omitted some portion of that law. Mr. MEMMINGER. This is all the information I ave on the subject. It confirms what is stated in he report. Mr. ENGLISH read from DeBow's Review an article

a very erroneous one] in support of the assertion ontained in the Declaration, that nearly all the Free tates had refused to sustain the Constitution.

Mr. MAXCY GREGO. The gentleman who just sumed his seat, has pointed out in detail the various sections referred to in this report. He has shown estions referred to in this report. He has shown int things have been said there which ought not to b have been said, and of the correctness of which e have not sufficient evidence. But my objection to the paper is greater than this. It is that, as a State aper, to go out as a new Deckaration of Independ-ince, it is entirely defective and imperfect. It pur-ports to be a declaration of the causes which justify the Saccession of South Carolina from the Federal nion. The causes! And yet in all this declaration of one word is said about the tariff, which for so any years caused a contest in this State against the ederal Government. Not one word is said about le violations of the Constitution in expenditures of authorized by that instrument; but the main ress is laid upon an incomparably unimportant point lative to fugitive slaves, and the laws passed y Northern States obstructing the recovery of fugi-ve slaves. Mr. President, if we undertake to set of the a declaration of the causes which justify our ecession, we ought to publish a complete document a document which might vie in its completeness ith that which was adopted in 1776-not that I mean o say that that is a model cause! that would be to ay a good deal too much. This declaration might p putforth by gentlemen who had no objection what-rer to the lavish and unconstitutional expenditures hich have been made by the Federal Government any years caused a contest in this State against the For to the lavish and unconstitutional expenditures hich have been made by the Federal Government or forty years past. This is not the sort of paper hich, in my opinion, ought to go forth to justify ur action. A correct designation of this paper ould be a declaration of some of the causes which the secession of South Carolina from the ederal Union. If it is proper to set forth in a solemn eclaration some of the causes, why let the title be thered, and, if the Convention think proper, let it o forth; but if we undertake to set forth all the tates do we not dishonor the memory of all the tatesmen of South Carolina, now departed, who ommenced forty years ago a war against the tariff and against internal improvements, saying nothing of the nited States Bank and other measures which may now e regarded as obsolete. Many of the acts of the mited States Bank and other measures which may now e regarded as obsolete. Many of the acts of the on-slaveholding States obstructing the recovery of igitive slaves have been passed since 1852—1 think ie majority of them; but 1 do not regard it as a matter f any importance. But when the people of South arolina, eight years since, declared that the causes nen existing fully justified the State in seceding, did rey confine themselves to these miserable fugitive lave laws? No I Sir, I regard it as unworthy of the tate of South Carolina to send forth a new declara-on now, and in it to say nothing about any other ause justifying their action but fugitive slaves. I m in favor of laying this report on the table, or re-ommitting it. ommitting it.

militing it. Mr. KEITT. I agree with the gentleman that the ower of taxation is the central power of all Gov-nments. If you put that into my hands, I do not me what the form of Government may be, I will ontrol your people through it. But that is not the vestion in this address. We have instructed the bommittee to draw up a statement of the reasons high influenced us in the present case in our with-rawal. My friend suggests that sufficient notice has not the your members of the House of Representatives roted for the present tariff. If the gentleman had been there he would loo have outed for it. Laughter.] The question of he tariff did agitate us in 1832, and did array this tate against the Federal Government. And I main-2

tain that this State did triumph then. Mr. Clay said, before the nullification, that the tariff system had been established for all time. After the nullification ordinance Mr. Clay said that that ordinance abol-ished the American system, and that the State had triumphed. It is true that we were cheated in the compromise. The tariff is not the question which has brought us up to our present attitude. We are giving a list of the causes to the world—to the Southern States. Let them not quarrel with us now, when we are brought up to a dissolution of the Union, by the discussion of debatable and doctrinal points. The Whig party, throughout all the States, have been pro-tective tariff men, and they cling to that old issue with all the passion incident to the pride of human opinions. Are we to go off now, when other Southopinions. Are we to go off now, when other South-ern States are bringing their people up to the true mark-are we to go off on debatable and doctrinal points? Are we to go back to the consideration of this question, of this great controversy; go back to points? Are we to go back to the consideration of this question, of this great controversy; go back to that party's politics around which so many passions cluster! Names, sir, are much. Opinions, preju-diced passions, cluster around names. Our people have come up to this great act. I am willing in this issue to rest disanion upon the question of slowery. It is the great central point from which we are now proceeding. I believe, sir, that the reference to other States in this address is all correct. The gentleman from Chester-field says that a certain construction of one act of the Pennsylvania code is denied by the citizens of that State. I myself have very great doubts about the propriety of the fugitive slave law. The Constitution was, in the first place, a compact between the several States, and in the second a treaty between the sections; and, I believe the fugitive slave law was a treaty between sections. It was the act of sovereign States as sec-tions; and I believe, therefore, and have very great doubts whether it ought not to have been left to the execution of the several States, and, failing of en-forcement, I believe it should have been regarded as a caus belli. I go for the address because I believe it the main primary causes. Mr. Greage. If this address was to be a declaration

the main primary causes. Mr. GREGG. If this address was to be a declaration of the immediate causes which produced the secession of the immediate causes which produced the secession of South Carolina, what the gentleman had said might be applicable, but its title does not say so. Another document has been submitted to this body—an Ad-dress to the Southern States. This is inconsistent with the other. In the latter address all the causes are stated in full. If we wish to find the immediate cause of all is the election of Lincoln. Mr. INGLIS. Will the gentleman inform us whether the statutes of Vignia do not contain a paragraph relating to kidnapping, precisely similar to that of Pennsylvania? A VOICE. It is the case with Georgia.

Pennsylvania? A Voice. It is the case with Georgia. Mr. KEITT. It may be so, sir, but I do not know. Mr. INGLIS. I say, Mr. President, I make no attack upon this report; but I propose to amend it by strik-ing out the word "fifteen" and inserting "many" in-stead; and then to strike out the sentence which con-tains the enumeration of States. It will not disturb the order to omit that.

Mr. DARGAN. I confess my difficulty results from the same sources as the gentleman from Richland. Let me express also my earnest conviction of the eminent propriety of obtaining a concurrence and symmetry in the declaration of the causes which led symmetry in the declaration of the causes which led to the secession of South Carolina, and in the senti-ments enunciated in the Address to the Southern States; and as the Address to the Southern States, which was read here to-day, was made the special order for to-morrow, I move that this document be also made the special order at the same time and in

also made the special order at the same time and in connection with that subject. Mr. MIDDLETON. They are very different matters --the one an address to the Southern people and the other an address to the world. Mr. DARGAN. The subject-matter is the same. The PRESIDENT. The question will be on making the report of the Committee declaring Secession the special order for one o'clock to-morrow, in connection with the report of the Committee on Slaveholding States upon the same subject. States upon the same subject. The question was taken and the motion was agreed

to.

On Monday, December 24th, 1860, the Convention proceeded to consider both the Address and the Declaration, when further debate ensued.

Many verbal amendments having been made to the latter,

Mr. J. J. P. Smith moved to adopt the former for the present, and table the latter.

A DELEGATE. I second the motion, and call for the previous question.

Mr. LOUIS WARDLAW. I trust that this Convention is not going to act hastily. Whatever is done should be done well. This address will reach no one of the be done well. This address will reach no one of the Southern States before the elections, unless it be the State of Georgia. There is, therefore, no special need of hurrying the reference. There is not one single sentence of that address to which I do not heartily subscribe. It is an able and admirable expo-sition of the structure of our Government and its general operation. And yet I do not think it is ex-actly that which an address to our Southern sisters should be I think it treats too much upon some subshould be. I think it treats too much upon some subshould be. I think it treats too much upon some sub-jects, and does not touch others that are very im-portant. From the beginning I have been very anxious that these two papers should be consistent one with the other, and contain all those matters which we confess should operate either upon the opinion of the Southern people or the opinion of the world. Now, sir, my objection to the address to the Southern people is that it does not dilate as it should upon matters connected with the immediate encode upon matters connected with the immediate cause of by on matters connected with the infinite rates of our secession, but on matters connected with sharery. My objection to the other address is, that it dwells too much upon those fugitive slave laws and those personal liberty bills, which give it too much the appear-ance of special pleading. The address which we have under consideration does not set off to the Southern under consideration does not set off to the Southern people, as it should, our defenceless condition. Al-ready our adversaries have the House of Representa-tives; they will soon have the Senate, and then they can make the Judiciary what they please, and thus have entire power over the Government. It does not set forth, as I think it should, that the election of set forth, as I think it should, that the election of Lincoln is, in fact, an edict of emancipation. It does not set forth what would be the deleterious effects of emancing the temanoication would be destrue. emancipation : that emancipation would be destruction to the blacks and degradation to the whites. tion to the blacks and degradation to the whites. Nor does this address set forth the shameless hypo-crisy of the North, who, whilst they cry out against what they call the sin of slavery, do not choose to relieve themselves of that which they assert is an evil by withdrawing from the Confederacy. When these addresses go forth, they go forth as solemn State papers, by which we must be able to stand. For this reason, every word should be nost carefully considered, and nothing superfluous should be con-tained in them; nothing important should be omitted. Mr. MEAUMINGER next fook the floor and defended

tained in them ; nothing important should be omitted. Mr. MEMIMORER next took the floor and defended the address to the nations of the world, which was reported by himself. After reciting its points and the principles it enunciated, he said : We show by law of compact that we are entitled to leave this Government. My friend from Abbeville says, in this regard, he does not exactly approve this document. Allow me to say to the honorable gentlemen that when you take the position that you have a right to break your faith, to destroy an agreement which you have made, to tear off your seal from the document to which it is affixed, you are bound to justify your-self fully to all the nations of the world; for there is nothing that casts such a stain upon the escutcheon serifully that casts such a stain upon the escutcheon of a nation as a breach of faith. Therefore the docu-ment shows fully that both in measure and in spirit our co-States have broken the Constitution and the Union. Not only in letter has this been done, but also in spirit. The common agent which should have acted for our common good has been converted into an instrument for our destruction. And now as a consummating act a sectional President has been elected, whose chief recommendation was that he desires to see slavery abolished. The great objection that we raise is not to Abraham Lincoln himself, but because he is the representative of a hostile opinion,

destructive of every interest of the South. Mr. RHETT next spoke in explanation of the Ad-dress to the Southern States. which was reported by himself. This committee, he said, determined that, whilst setting forth the immediate cause which in-duced South Carolina to seeded, it was not improper to go into previous causes which led to that result. to go into previous causes which led to that result. The secession of South Carolina is not an event of a day. It is not any thing produced by Mr. Lincoln's election, or by the non-execution of the fugitive slave law. It has been a matter which has been gathering head for thirty years. The election of Lincoln and Hamlin was the last straw on the back of the camel. But it was not the

only one. The back was nearly broken before. The point upon which I differ from my friend is this : He says he thought it expedient for us to put this great question before all the world upon this simple matter of wrongs on the question of slavery, and that ques-tion turned upon the fugitive slave law. Now, in retion turned upon the fugitive slave law. Now, in re-gard to the fugitive slave law. Now, in re-gard to the fugitive slave law, I myself doubt its consti-tutionality, and I doubted it on the floor of the Senate, when I was a member of that body. The States, acting in their sovereign capacity, should be respon-sible for the rendition of fugitive slaves. That was our best security. This report has proceeded up our best security. This report has proceeded upon the elaborate discussion of a constitutional question, the elaborate discussion of a constitutional question, about which the very ablest men in this State have doubted. When we go before the world, if we put it upon mere matter of this kind, we do not do justice to our cause. Sir, to whom are we to speak ? Is it simply to the North ? We are about to sunder our relations with that section, and I trust forever. Our treaties, I suppose, will be with the nations of Eu-rope. Do you suppose the nations of Europe will have any sympathy with us, or confidence, or affec-tion, because of the violation of the fugitive slave law? Germany, and France, and England, what do they all say? Sir, in setting up our independence we are not to narrow it down simply to the question of slavery. We do not do ourselves justice. The aggression upon slavery is the last consequence of a great cause, and that great cause is the dissolution a great cause, and that great cause is the dissolution of the Constitution of the United States by the agents of the North. It is that which led them to the ag-gressions upon the taxing power. It is that which led to the aggressions upon the appropriation power. It is that which led to the aggressions on slavery in the District of Columbia. And now the great cause is, that we do not live in a free Government.

is, that we do not live in a free Government. Mr. MEMINGER. The gentleman who has just taken his seat is not as familiar with this document as I am, or he would have been saved the necessity of a good deal he has said. I entirely concur in the opinion that the Constitution of the United States requires the rendition of slaves by the States and not by the General Government; and if any one will read this report he will perceive that that is precisely the eround upon which it proceeds. We there complain ground upon which it proceeds. We there complain that the States have not fulfilled their constitutional obligations—not that the Federal Government has not done its duty. We there complain that when the Federal Government undertakes to do that which the States had obligated themselves to do, they interfere to prevent its faithful execution.

Judge WITHERS said : I have not much to say to this Convention, but the first thing which I desire to submit to them is this : that the addresses which are now upon your table, and which are the subject-matter of a motion for further reference to the two committees reporting them, are, in my understanding, diplomatic papers. I profess not to be much of a diplomat my-self, yet I profess to have a desire that this Conven-tion shall confine itself to the object which it prescribes to itself.

What is the object of the Address to the Southern cople ? Is it not to conciliate the Southern States reopie: Is it not to conciliate the Southern States towards the purposes of a Southern Confederacy; and, as far as we can, to persuade them to enter into a compact with South Carolina? Is that the object of the Address to the Slaveholding States? If not, why should it be issued? People ?

It is said in the discussion that the Address to the Slaveholding States should descant upon the taxing power and the power to lay duties upon imports, as well as the expenditures of money in undue propor-tion upon the part of the Federal Government among tion upon the part of the Federal Government among the Free States, as matters of grievances of the greatest importance; that such topics ought to be found in this paper setting forth the causes of Seces-sion. Well, in an Address to the People of the Slave-holding States is it expedient to dwell and insist upon a topic which will not find favor with all the upon a topic which will not find favor with all the Southern States' I submit to the experience of the able gentleman who prepared that address to say whether, if we declared that we separated from the Confederacy because of the tariff of protection to domestic manufactures, he will find that, be it ever so true, a sentiment corresponding to public opinion in Louisiana, Missouri, or Kentucky? All this matter of the tariff has been enacted while the Confederacy existed, and with South Carolina as a party to the transaction. When it begun in 1816, who was it voted for a tariff highly protective to domestic manufactures? Did not that great man whom we all reverence, both living and dead-I mean Mr. Calhoun-vote for this measure? Did not the Representative in the House from the Congressional

district including Richland vote for the tariff of 1816<sup>†</sup> Has there ever been a time when Louisiana, Missouri, and Kentucky were not in favor of a protective tariff; not only for protection of domestic manufactures, but for protection on the products of sugar and hemp<sup>†</sup> Are you sure they will join you in saying they should dissolve the Union on account of the existing tariff giving protection to domestic manufactures? You believe, and so do I, that there has been a perversion of the Constitution in relation to imposits for the purpose of protection to domestic manufactures. I know of no time, from the period of my entering college in 1823, that I did not believe it was a bold and daring invasion of the Constitution of the United States. Undoubtedly this is my opinion, undoubtedly this is the opinion of South Carolina. Then if I had to draw these papers, if I should present my views and opinions in a common address to the Slaveholding States, I should suggest the propriety of leaving out all topics of that description, when I believe that three of these States differ in sentiment with South Carolina. It appears to me, therefore, we have not exactly hit upon the matter which is the most expedient and proper in an Address to the People of the Slaveholding States. It is a diplomatic document. I shall vote for it. Butat the same time I do not think, as a diplomatic paper, that with respect to the levying of duties on imports, it is likely to find favor with all our slaveholding friends for whom this tariff was designed as well as for the North.

In regard to expenditures by the Federal Government of its income, we all know very well that the great bulk has gone within the Northern States that there has been, on the part of the Federal Government, favorite States.

ernment, favorite States. When we complain in the aggregate, or in general terms, when we say that the grievances of South Carolina are found in the fact that the Treasury has been depleted by illegal means, and in undue proportion administered to the North, I question whether we are quite safe in alleging that as a grievance of South Carolina, without qualification. There has been an unfaithful execution of the Constitution on the part of its own general agent in that respect. But let us not forget to confess the truth under any and all circumstances. What have we ourgelves been doing? And in the city of Charleston, too, where have you bought your supplies, and with whom do you trade? Where has the great surplus of your money been necessarily spent? Where has it gone to? Has it not gone to these people who have received the Federal money? Government and individuals have sought the same market. Why? Because nobody else could furnish the articles each wanted. Can you say, therefore, that the Federal Government is to be alamed for spending a large amount of money in the non-slaveholding States? Where was the Federal Government fill up the ranks of its army and navy? Will you not allow the Government to buy of its own citizens, as we have all done? If by the cunning of these men in the non-slaveholding States they have been able to present to the Government inducements to obtain their supplies, can we complain? Where else could they have been procured? So far, the Government has been obliged to spend its money among the people of the North and Northwest for bacon, lard, and all the supplies of the army and navy. I aubmit these views for the purpose of drawing the attention of the Convention to the fact that we may go too far in this document, and use assertions too

Strong. In respect to the argument of the fugitive slave law, I concur fully. I heard something said here questioning the Constitutionality of the fugitive slave law, as it is called. This is a difficult question. In the case of Prigg and the Commonwealth of Pennsylvania, all the Judges of the United States Courts but two declared that Congress, and Congress alone, ceuld provide legislation to execute the fourth article of the Constitution of the United States. Immediately after that decision the astute Legislatures of the New England States seized upon that decision, and passed their liberty laws, invoking the doctrine announced in the case of Prigg vs. the Commonwealth of Pennsylvania. Could any man say that South Carolina should separate from the United States in consequence of the Congress of the United States passing such a law? A like law was passed in 1793. Did our people object to it then ! I confess I have a reverence for antiquity. I profess to have a veneration of the men of 1783—Christopher Gads-

den, John Rutledge, the Pinckneys, and others. I profess to believe that they were as patriotic as I profess to be. If we made no objection at that time to the power of Congress to pass a fugitive slave law, under the fourth article of the Constitution of the United States, I hold it would be unsafe for the Convention of South Carolina to say that that is a cause for which she separates from the United States. It was a matter, as long ago as 1643, of stipulation between Massachusetts, Rhode Island, and another colony, that they should deliver each other's fugitive slaves. It was a matter between the color nice that each colony should deliver fugitives. As long ago as that period Congress did exercise this power, and we did acquiesce and never voted against it.

If I were to stand here and declare the various causes which led me to subscribe my name to the Act of Secession, I should insist on some other considerations besides those suggested by this address. I would have said that when a citizen of Maryland went to Pennsylvania to recover his fugitive under an act of Congress they murdered him, and his murderer\* escaped from justice in the court of Pennsylvania. Then was the time for Maryland to have demanded justice under the compact. It was then I would have stood up for the rights of that slaveholder. If I chose further to afflict this Convention I could bring before them a long catalogue of grievances. I think if every member of the Convention should draw up an indiciment against the people of the unfaithful Confederate States, and you might have any number of addressstantial objection to the addresses before us, as I think there is not, it is proper to vote for them, and I shall do so.

The papers were both adopted.

A third report was made to the Convention by Judge Withers from the Committee on the Relations of the Slaveholding States of North America, which should be included, to make the catalogue complete:

The Committee on "Relations with the Slaveholding States of North America," beg leave to report that they have carefully considered the three several propositions contained in the resolutions referred to them, which were submitted in Convention by three several members from St. Phillip's and St. Michael's.

All the resolutions referred to the Committee look to the purpose of confederate relations with our sister States of the South, having common interests with us, and every cause, as we trust, to indulge towards us common sympathies, and to contract cordial relations. In such a purpose the Committee entirely and unanimously concur, and they recommend that every proper measure be adopted to accomplish such an end.

piish such an end. Upon this subject so much unanimity prevails, and has long prevailed in this State, that an argument thereupon would be wholly superfluous. All seem to agree that the first step proper to be taken for the purpose of promoting and securing the confederation we seek is the appointment of Commissioners, by the authority of this Convention, to such States of the South as may call Conventions to consider and determine their future political relations. The Committee advise that such step be taken by this Convention, hoping and believing that our sister States of the South will correctly interpret our action in taking the initiative as arising by no means from

The Committee advise that such step be taken by this Convention, hoping and believing that our sister States of the South will correctly interpret our action in taking the initiative as arising by no means from any presumptious arrogance, but from the advance position which circumstances have given to this State in the line of procedure for the great design of securing the rights, the security, and the very existence of the slaveholding South. It has been a subject of anxious consideration with the Committee whether the Commissioners, whose

It has been a subject of anxious consideration with the Committee whether the Commissioners, whose appointment, they recommend, should be instructed to tender any basis of a temporary or Provisional Government to the States to which they may be accredited. The instrument called the Constitution of the United States of America has been suggested as a suitable and proper basis to be offered for a Provisional Government. This suggestion has been commended to the Committee by various considerations which cannot now be set forth in full or at large.

\* In 1847

That the said instrument was the work of minds of the first order in strength and accomplishment.

That it was most carefully constructed by comprehensive views and careful examination of details

hensive views and careful examination of details. That experience has proved it to be a good form of Government for those sufficiently virtuous, intelli-gent, and patriotic to cause it to be fairly and honest-ly construed and impartially administered. That the settled opinion of this State has never been adverse to that plan of government for Confed-erated States, on account of any thing in its structure, hut the dissatisfaction is attributable to the false glosses and dangerous misinterpretation and perversion of sundry of its provisions even to the extent, in one particular, of so covering up the real purposes of certain legislation (meant to protect domestic manufactures in one section) as to estop the Supreme Court in its opinion from judicially perceiving the real design. perceiving the real design.

That is presents a complete scheme of contedera-tion, capable of being speedily put into operation, fa-miliar, by long acquaintance, with its provisions and their true import to the people of the South, many of whom are believed to cherish a degree of veneration for it, and would feel safe under it when in their own hands for interpretation and administration; espe-cially as the portions that have been, by perversion, made potent for mischief and oppression in the hands of adverse and inimical interests, have received a settled construction by the South.

settied construction by the South. That a speedy confederation by the South is desir-able in the highest degree, which, it is supposed, must be temporary at first, (if accomplished as soon as it should be), and no better basis than the Consti-tution of the United States is likely to be suggested or advanted for temporary supposed

That the opinions of those to whom it is designed to offer it would be concliated by the testimory the very act itself would carry that South Carolina meant to seek no selfish advantage nor to indulge the

meant to seek no sense advantage nor to indulge the least spirit of dictation. That such form of government is more or less known to Europe, and, if adopted, would indicate abroad that the seceding Southern States had the forecast and energy to put into operation for thwith a scheme of government and administration compe-tent to produce a processory organization for internal tent to produce a necessary organization for internal necessities, and a sufficient protection of foreign commerce directed hither, as well as to guarantee Foreign powers in the confidence that a new Confederacy had immediately arisen, quite adequate to su-persede all the evils, internal and external, of a par-tial or total interregnum.

tial or total interregnum. That its speedy adoption would work happily as a reviving agency in matters financial and commercial between the States adopting it, and between them as a united power and foreign commercial nations; and at the same time would combine, without delay, a power touching purse and sword, that might bring to a prudent issue the reflections of those who may perchance be contemplating an invasion, or to an issue disastrous to them, the attempted execution of such unloy design. such unholy design.

Such are some of the considerations, very rapidly stated, which address themselves to this subject.

stated, which address themselves to this subject. It is contended that the same limitation of the power to levy duties, and that to regulate commerce, (and perhaps other provisions of the said Constitu-tion,) may be desirable, and are in fact so to some of the committee; yet these modifications may be safely left to a period when the articles of a perma-time, the Constitution referred to will serve the pur-pose of temporary confederation which the compose of temporary confederation, which the com-mittee unite in believing ought to be sought, through all proper measures, most carnestly. It is also submitted that if the tender of the said

Constitution, even as a Provisional Government, should, in the opinion of the Convention, be accom-panied by a condition that it be subject to specific limitations, expositions of ambiguities, or modificalimitations, expositions of ambiguities, or modifica-tions, the committee would respectfully refer to the Convention itself such matters; and this is done, not because the committee would not willingly con-sider and report upon such subject, but because they deem it due to the Convention and the public interest that they should now lay before the Convention the substantial propositions contained in the following resolutions, which the majority of the committee re-commend to the Convention as fit to be adopted, viz : Resoluted First that this Convention do annotint a

Resolved. First, that this Convention as it to be adopted, viz: Resolved. First, that this Convention do appoint a Commissioner to proceed to each of the slarehold-ing States that may assemble in Convention, for the purpose of laying our Ordinance of Secession before the same, and respectfully invite their co-operation

in the formation with us of a Southern Confederacy.

Second. That our Commissioners aforesaid be fur-ther authorized to submit, on our part, the Federal Constitution as the basis of a Provisional Govern-ment for such States as shall have withdrawn their connection with the Government of the United Third. That the said Commissioners be authorized

to invite the seceding States to meet in Convention at such time and place as may be agreed upon, for the purpose of forming a permanent Government for such States.

On the question of sending copies of the Ordinance and the accompanying Declara-tion of Causes and the Address, to the Governors of the slaveholding States, there was a debate, in which Mr. Dargan urged the propriety of notifying the authorities of all the States, which being objected to,

Mr. Dargan said: A statement of the reasons is required, as well as the Ordinance of Secession. Courtesy to our late Confederates, whether enemies or not, calls for the reasons that have actuated us. It is not true, in point of fact, that all the Northern people are hostile to the rights of the South. We have a Spartan band in every Northern It is due to them they should know State. the reasons which influence us. According to our apprehensions the necessity which exists for our immediate withdrawal from association with the Northern States is that this hostile Abolition party have the con trol of the Government, and there is no hope of redress for our grievances.

### Speech of Alexander H. Stephens, November 14th, 1860.

As against these allegations, we insert the speech of Hon. ALEXANDER H. STEPHENS (1 Georgia, before the Legislature of Georgia November 14th, 1860, and an extract fron his speech in the Convention of Georgia, of January, 1861 :

Fellow-Citizens :---I appear before you to-night, at the re-quest of members of the Legislature and others, to speak of matters of the deepest interest that can possibly concern us all of an earthly character. There is nothing-no question or subject connected with this life-that concerns a free people subject connected with this life—that concerns a free people so intimately as that of the government under which they live. We are now, indeed, surrounded by evils. Never, since I entered upon the public stage, has the country been so environed with difficulties and dangers that threatened the public peace, and the very existence of society, as now. I do not now appear before you at my own instance. It is not to gratify a desire of my own that I am here. Had I consulted my own ease and pleasure I should not be before you; but, believing that it is the duty of every good citizen to give his counsels and views whenever the country is in danger, as to the best policy to be pursued. I am here. For danger, as to the best poley to be pursued. I am here. For these reasons, and these only, do I bespeak a calm, patient and attentive hearing. My object is not to stir up strife, but to allay it; not to

My object is not to stir up strife, but to allay it; not to appeal to your passions, but to your reason. Good govern-ments can never be built up or sustained by the impulse of passion. I wish to address myself to your good sense, to your good judgment, and if, after hearing, you disagree, let us agree to disagree, and part as we met, friends. We all have the same object, the same interest. That people should disagree, in republican governments, upon questions of public policy, is natural. That men should disagree upon all matters connected with hunan investigation, whether relating to science or human conduct, is natural. Hence, in free governments, parties will arise. But a free people should express their different opinions with liberality and charity, with no acrimony toward those of their fellows when honestly and sincerely given. These are my feelings to-night.

Let us. therefore, reason together. It is not my purpose the set us derived reacting of the set in the full who reasy be present; and if in the ardency with which I shall express my opinions, I shall say any thing which may be deemed too strong, let it be set down to the zeal with which I advocate my own convictions. There is with me no intention to irritate or offend.

I advocate my own convictions. There is with me no inten-tion to irritate or offend. The first question that presents itself is, shall the people of the South secede from the Union in consequence of the election of Mr. Lincoln to the Presidency of the United States? My countrymen, *Itell you frankly, candidly and ear-nestly, that I do not timke that they gought.* In my judgment the election of no man, constitutionally chosen to that high office, is sufficient cause for any State to separate from the Union. It ought to stand by and aid still in maintaining the Constitution and the country. To make a point of re-sistance to the government, to withdraw from it because a man has been constitutionally elected, puts us in the wrong. We are pledged to maintain the Constitution. Many of us have sworn to support it. Can we, therefore, for the mere election of a man to the Presidency, and that too in accord-ance with the prescribed forms of the Constitution, make a point of resistance to the government without becoming the breakers of that sacred instrument ourselves—with-draw ourselves from it? Would we not be in the wrong? Whatever fate is to befall this country, let it never be laid to the charge of the people of the South, and especially to the people of Georgia, that we were untrue to our national engagements. Let the fault and the wrong rest upon others. or our heads. Let the fanal and the worger stupon others. If all our hopes are to be blasted, if the Republic is to go down, let us be found to the last moment standing on the deck, with the Constitution of the United States waving over our heads. Let the fanalies of the North break the Constitution founds in the it do it of the united states superconstitution, if such is their fell purpose. Let the respon-sibility be upon them. I shall speak presently more of their acts; but let not the South-let us not be the ones to their acts; but let not the South—let us not be the ones to commit the aggression. We went into the election with this people. The result was different from what we wished; but the election has been constitutionally held. Were we to make a point of resistance to the Government and go out of the Union on that account, the record would be made up

of the Union on that account, the record would be made up hereafter against us. But it is said Mr. Lincoln's policy and principles are against the Constitution, and that if he carries them out it will be destructive of our rights. Let us not anticipate a threatened evil. If he violates the Constitution, then will come our time to act. Do not let us break it because, forsooth, he may. If he does, that is the time for us to strike. I think it would be injudicious and unwise to do this sooner. I do not anticipate that Mr. Lincoln will do any thing to jeopard our safety or security, whatever may be his spirit o do it; for he is bound by the constitutional checks which are thrown around him, which at this time render him powerless to do any great mischief. This shows the wisdom of our system. The President of the United States is no Emperor, no dictator—he is clothed with no absolute power. He can do nothing unless he is backed by power in Congress. The House of Representatives is largely in the majority against him. In the Senate he will also be powerless. There will be a majority of four against him. This, after the loss of Bigler, Fitch, and others, by the unfortunate dissensions of the National Democratic party in their States. Mr. Lincoln cannot appoint an officer without the consent of the Senate. He this sooner. I do not anticipate that Mr. Lincoln will do

cannot appoint an officer without the consent of the Senate. He cannot form a cabinet without the same consent. He will be in the condition of George III. (the embodiment of Toryism), who had to ask the whigs to appoint his minis-ters, and was compelled to receive a cabinet utterly op-posed to his views; and so Mr. Lincoln will be compelled to ask of the Senate to choose for him a cabinet, if the De-mocracy of that body choose to put him on such terms. He will be compelled to do this or let the government stop, if the National Democratic men-for that is their name at the North-the conservative men in the Senate, should so de-termine. Then how can Mr. Lincoln obtain a cabinet which would aid him, or allow him to violate the Consti-tution ? tution?

Why, then, I say, should we disrupt the ties of this Union when his hands are tied, when he can do nothing against us? I have heard it mooted that no man in the State of Georgia, who is true to her interests, could hold office under Mr. Lincoln. But, I ask, who appoints to office? Not the President alone; the Senate has to concur. No man can be appointed without the consent of the Senate. Should any appointed without the consent of the Senate. Should any man then refuse to hold office that was given to him by a Democratic Senate? [Mr. Toombs interrupted, and said if the Senate was Democratic it was for Mr. Breckfuridge.] Well, then, continued Mr. S. I apprehend no man could be justly considered untrue to the interests of Georgia, or in-cur any disgrace, if the interests of Georgia required it, to hold an office which a Breckinridge Senate had given him, even though Mr. Lincoln should be President. I truet my countrymen you will be still and silent I

I trust, my countrymen, you will be still and silent. I am addressing your good sense. I am giving you my views in a calm and dispassionate manner, and if any of you differ with me, yon can, on any other occasion, give your views

as I am doing now, and let reason and true patriotism de-cide between us. In my judgment, I say, under such cir-cumstances, there would be uo possible disgrace for a Southern man to hold office. No man will be suffered to be appointed, I have no doubt, who is not true to the Consti-tution, if Southern Senators are true to their trusts, as I cannot permit myself to doubt that they will be. My honorable friend who addressed you last night (Mr. Toombs), and to whom I listened with the profoundest at-tention asks if we would submit to Black Republican rule?

tonion, asks if we would submit to Black Republican rule? I say to you and to him, as a Georgian, I never would sub-mit to any Black Republican *aggression* upon our constitu-tional rights\_\_\_I will never consent myself, as much as I admire this Union for the glories of the past, or the bless-ings of the present—as much as it has done for the people of all these States—as much as it has done for civilization as much as the hopes of the world hang upon it. I would never submit to aggression upon my rights to maintain it longer; and if they cannot be maintained in the Union, standing on the Georgia platform, where I have stood from the time of its adoption, I would be in favor of disrupting every tie which binds the States together. I will have equality for Georgia and for the citizens of

I will have equality for Georgia and for the citizens of Georgia, in this Union, or I will look for new safeguards elsewhere. This is my position. The only question now is, can they be secured in the Union? That is what I am counseling with you to-night about. Can it be secured? In my judgment it may be, but it may not be; but let us do all we can, so that in the future, if the worst come, it may never be said that we were negligent in doing our duty to the last.

In a proter he share that we were highlighter having our tags of the last. My countrymen, I am not of those who believe this Union has been a curse up to this time. True men, men of integ-rity, entertain different right to do so; I would not impugn their motives in so doing. Nor will I undertake to say that this government of our fathers is perfect. There is nothing perfect in this world of a human origin—nothing connected with human nature, from man himself to any of his works. You may select the wisest and best men for your judges, and yet how many defects are there in the administration of justice? You may select the wisest and best men for your legislators, and yet how many defects are apparent in your legislators, and yet how many defects, with all its de-fects, comes nearer the objects of all good governments. But that this government of our fathers, with all its de-fects, comes nearer the objects of all good governments.

than any other on the face of the earth is my settled conviction. Contrast it now with any on the face of the earth. [England, said Mr. Toombs.] England, my friend says. Well, that is the next best, I grant; but I think we have improved that is the next best, I grant; but I think we have improved upon England. Statesmen tried their apprentice hand on the government of England, and then ours was made. Ours sprung from that, avoiding many of its defects, taking most of the good and leaving out many of its errors, and from the whole constructing and building up this model Repub-lic—the best which the history of the world gives any account of.

Compare, my friends, this Government with that of Spain, Mexico, the South American Republics, Germany, Ireland —are there any sons of that down-trodden nation here to-

Mexico, the South American Republics, Germany, Ireland —are there any sons of that down-trodden nation here to night?—Prussia, or, if you travel further east, to Turkey or China. Where will you go, following the sun in its circuit round our globe, to find a government that hetter protects the liherties of its people, and secures to them the blessings we enjoy? I think that one of the evils that beste us is a surfeit of liberty, an exuberance of the priceless blessings for which we are ungrateful. We listened to my honorable friend who addressed you last night (Mr. Toombs) as he recounted the evils of this Government. — The first was the fishing bounties, paid mostly to the salors of New England. Our friend stated that forty-eight years of our government was under the administration of Southern Presidents. Well, these fishing bounties began under the rule of a Southern President, I believe. No one of them during the whole forty-eight years ever set his ad-ministration against the principle or policy of them. It is not for me to say whether it was a wise policy in the begin-ing; it probably was not, and I have nothing to say in its defence. But the reason given for it was to encourage our young men to go to sea and learn to manage ships. We had at the time but a small nary. It was thought best to encourage a class of our people to become acquainted with seafaring life, to become sallors—to man our narval ships. It requires practice to walk the deck of a ship, to furl the salls, tog o aloft, to climb the mast; and it was thought, by offering this hounty, a nursery might be formed in which young men would become perfected in these arts, and it ap-plied to one section of the country as well as to any other. —The result of this was, that in the war of 1812 our sallors,

plied to one section of the country as well as to any other. The result of this was, that in the war of 1812 our sailors, many of whom came from this nursery, were equal to any that England brought against us. At any rate, no small part of the glories of that war were gained by the veteran tars of America, and the object of these bounties was to foster that branch of the national defence. My opinion is that whatever may have been the reason at first, this bounty ought to be discontinued—the reason for it at first

no longer exists. A bill for this object did pass the Senate the last Congress I was in, to which my honorable friend contributed greatly, but it was not reached in the House of Representatives. I trust that he will yet see that he may with honor continue his connection with the government, and that his eloquence, unrivalled in the Senate, may here-after, as heretofore, be displayed in having this bounty, so obnoxious to him, repealed, and wiped off from the statute book.

The next evil which my friend complained of, was the tariff. Well, let us look at that for a moment. About the time I commenced noticing public matters, this question was agitating the country almost as fearfully as the slave question now is. In 1832, when I was in college, South Carolina was ready to nullify or secede from the Union on Carolina was ready to nullify or secence from the Union on this account. And what have we seen? The tariff no longer distracts the public counsels. Reason, has tri-umphed! The present tariff was voted for by Massachu-setts and South Carolina. The lion and the lamb lay down together—every man in the Senate and House from Massachusetts and South Carolina, I think, voted for it, as did my honorable friend himself. And if it be trne, to use the figure of speech of my honorable friend, that every man in the North that works in iron, and brass and wood, has his muscle strengthened by the protection of the govern-ment, that stimulant was given by his vote, and I believe every other Southern man. So we ought not to complain of that.

Mr. TOOMBS. The tariff assessed the duties.

Mr. STEPHENS. Yes, and Massachusetts with unanimity voted with the South to lessen them, and they were made just as low as Southern men asked them to be, and that is Just as now as Southern men asked them to be, and that is the rate they are now at. If reason and argument, with experience, produced such changes in the sentiments of Massachusetts from 1832 to 1857, on the subject of the tariff, may not like changes be effected there by the same means—reason and argument, and appeals to patrictism on the present vexed question? And who can say that by 1875 or 1890 Massachusetts may not vote with South Carolina and Georgia upon all those questions that now distract the country and threaten its peace and existence. I believe in the power and efficiency of truth, in the omnipotence of truth, and its ultimate triumph when properly wielded.

Another matter of grievance alluded to by my honorable friend was the Navigation Laws. This policy was also commenced under the Administration of one of these Southern Presidents who ruled so well, and has been continued through all of them since. The gentleman's views of the policy of these laws and my own do not disagree. We occupied the same ground in relation to them in Con-gress. It is not my purpose to defend them now. But it is proper to state some matters connected with their origin.

One of the objects was to build up a commercial American marine by giving American bottoms the exclusive carryingtrade between our own ports. This is a great arm of national power. This object was accomplished. We have now an amount of shipping, not only coastwise, but to foreign countries, which puts us in the front rank of the nations of the world. England can no longer be styled the Mistress of the Seas. What American is not proud of the result of Whether those laws should be continued is another question. But one thing is certain: no President, Northern or South-But one thing is certain: no resultent, formern of south-ern, has ever yet recommended their repeal. And my friend's efforts to get them repealed were met with but little favor, North or South. These, then, were the true main grievances or grounds

of complaint against the general system of our Government and its workings-I mean the administration of the Federal Government. As to the acts of the Federal States I shall Government. As to the acts of the reached main ones used against the common head. Now, suppose it be admitted that all of these are evils in the system; do they overbalance and outweigh the advantages and great good which this same government affords in a thousand innumerable ways that cannot be estimated? Have we not at the South. as well as the North, grown great, prosperous, and happy under its operations? Has any part of the world ever shown such rapid progress in the development of wealth, and all the material resources of national power and great-Government, notwithstanding all its defects?

Mr. TOOMES. In spite of it. Mr. STEPHENS. My honorable friend says we have, in spite of the General Government ; that without it, I suppose be thinks, we might have done as well, or perhaps better, than we have done this in spite of it. That may be and it may not be; but the great fact that we have grown great and powerful under the Government as it exists—there is and powerful under the Government as it exists—there is no conjecture or speculation about that i it stands out bold, high, and prominent, like your Stone Mountain, to which the gentleman alluded in illustrating home facts in his record—this great fact of our unrivalled prosperity in the Union as it is admitted; whether all this is in spite of the Government—whether we of the South would have been better off without the Government—is, to say the least,

problematical. On the one side we can only put the fact against speculation and conjecture on the other. But even as a question of speculation I differ with my distinguished friend.

What we would have lost in border wars without the Union, or what we have gained simply by the peace it has secured, no estimate can be made of. Our foreign trade, which is the foundation of all our prosperity, has the protection of the navy, which drove the pirates from the waters near our coast, where they had been buccaneering for centuries before, and might have been still had it not been for the American Navy, under the command of such spirits as Commodore Porter. Now that the coast is clear, that our commerce flows freely outwardly, we can not well estimate how it would have been under other circumstances. The influence of the Government on us is like that of the atmosphere around us. Its benefits are so silent and unseen that they are seldom thought of or appreciated.

We seldom think of the single element of oxygen in the air we breathe, and yet let this simple, unseen and unfelt agent be withdrawn, this life-giving element be taken away from this all-pervading fluid around us, and what instant and appalling changes would take place in all organic creation.

It may be that we are all that we are in "spite of the General Government," but it may be that without it we should have been far different from what we are now. It is true there is no equal part of the earth with natural resources superior perhaps to ours. That portion of this country known as the Southern States, stretching from the Chesapeake to the Rio Grande, is fully equal to the picture drawn by the honorable and eloquent Senator last night, in all natural capacities. But how many ages and centuries an fatural expansion but now many ages and continues passed before these capacities were developed to reach this advanced age of civilization? There these same hills, rich in ore, same rivers, same valleys and plains, are as they have been since they came from the hand of the Creator; uneducated and uncivilized man roamed over them for how long no history informs us.

It was only under our institutions that they could be de-veloped. Their development is the result of the enterprise of our people, under operations of the Government and inof our people, under operations of the Government and in-stitutions under which we have lived. Even our people, without these, never would have done it. The organization of society has much to do with the development of the natural resources of any country or any land. The insti-tutions of a people, political and moral, are the matrix in which the germ of their organic structure quickens into the performance of development for the development. life-takes root, and develops in form, nature, and character. Our institutions constitute the basis, the matrix, from which spring all our characteristics of development and which spring all our characteristics of development and greatness. Look at Greece. There is the same fertile soil, the same blue sky, the same inlets and harbors, the same Ægcan, the same Olympus; there is the same land where Homer sung, where Pericles spoke; it is in nature the same old Greece—but it is living Greece no more.

Descendants of the same people inhabit the country; yet what is the reason of this vast difference? In the midst of present degradation we see the glorious fragments of ancient present degradation we see the glorious fragments of ancient works of art—temples, with ornaments and inscriptions that excite wonder and admiration—the remains of a once high order of civilization, which have outlived the lan-guage they spoke—upon them all, Ichabod is written— their glory has departed. Why is this so? I answer, their institutions have been destroyed. These were but the fruits of their forms of government, the matrix from which their grand development sprung; and when once the in-stitutions of a neonle have been destroyed, there is no stitutions of a people have been destroyed, there is no earthly power that can bringback the Promethean spark to kindle them here again, any more than in that ancient

to kindle them here again, any more than in that ancient land of eloquence, poetry, and song. The same may be said of Italy. Where is Rome, once the mistress of the world? There are the same seven hills now, the same soil, the same natural resources; nature is the same, but what a ruin of human greatness meets the eye of the traveller throughout the length and breadth of that most down-trodden land! Why have not the people of that Heaven-favored elime the spirit that azimated their fathers? Why this sad difference? It is the destruction of her institutions that has caused it; and my construmen if we shall in an evi hour rashiv nol

and, my countrymen, if we shall in an evil hour rashly pull down and destroy those institutions which the patriotic band of our fathers labored so long and so hard to build up, and which have done on which for n and the world who and which have done so much for us and the world, who can venture the prediction that similar results will not ensue? Let us avoid it if we can. I trust the spirit is among us that will enable us to do it. Let us not rashly try the experiment, for, if it fails, as it did in Greece and Italy, and in the South American Republics, and in every other place wherever liberty is once destroyed, it may never be restored to us again.

to us again. There are defects in our government, errors in adminis-tion, and short-comings of many kinds; but in spite of these defects and errors, Georgia has grown to be a great State. Let us pause here a moment. In 1850 there was a great crisis, but not so fearful as this; for, of all I have

22

ever passed through, this is the most perilous, and requires to be met with the greatest calmness and deliberatiou.

There were many among us in 1850 zealous to go at once There were many among us in 1850 zealous to go at once out of the Union, to disruptevery tie that binds us together. Now, do you believe, had that policy been carried out at that time, we would have been the same great people that we are to-day? It thay be that we would, but have you any assurance of that fact? Would you have made the same advancement, improvement, and progress in all that constitutes material wealth and prosperity that we have? I notice, in the Comptroller-General's report, that the taxable property of Georgia is \$670,000,000 and upward, an amount not far from double what it was in 1850. I think I may venture to say that for the last ten years the material wealth of the people of Georgia has been nearly if not quite doubled. The same may be said of our advance in educa-

Weath of the people of Georgia has been heary if not duite doubled. The same may be said of our advance in educa-tion and every thing that marks our civilization. Have we any assurance that, had we regarded the earnest but mis-guided patriotic advice, as I think, of some of that day, and disrupted the ties which bind us to the Union, we would have advanced as we have? I think not. Well, then, let us be careful now before we attempt any rash experiment of this sort. I know that there are friend—whose natriof this sort. I know that there are friends-whose patriotism I do not intend to question—who think this Union a curse—and that we would be better off without it. I do not so think, if we can bring about a correction of those wils which threaten and I am not without hope that this may yet be done. This appeal to go out, with all the pro-visions for good that accompany it, I look upon it as a great and I fear a fatal temptation.

When I look around and see our prosperity in every thing, agriculture, commerce, art, science, and every department of education, physical and mental, as well as moral advanceof culcators, paysed and mental, as went as moral as taken ment, and our colleges, I think, in the face of such au ex-hibition, if we can, without the loss of power, or any essen-tial right or interest, remain in the Union, it is our duty to that right of interest, remain in the omon, it is our duty to ourselves and to posterity to—let us not too readily yield to this temptation—do so. Our first parents, the great pro-genitors of the human race, were not without a like tempta-tion when in the garden of Eden. They were led to believe that their condition would be bettered—that their eyes would be opened—and that they would become as gods. They in an evil hour yielded—instead of becoming gods, they only now their ours ackedness. they only saw their own nakedness.

I look upon this country with our institutions as the Eden of the world, the paradise of the universe. It may be that of the world, the paradise of the universe. It may be that out of it we may become greater and more prosperous, but I am candid and sincere in telling you that I fear if we rashly evince passion, and without sufficient cause shall take that step, that instead of heconing greater or more peaceful, prosperous, and happy—instead of heconing gods, we will become demons, and at no distant day commence cutting one another's throats. This is my apprehension. Let us, therefore, whatever we do, meet those difficulties, great as they are, like wise and sensible men, and consider them in the light of all the consequences which may attend our action. Let us see first clearly where the path of duty our action. Let us see first clearly where the path of duty leads, and then we may not fear to tread therein.

I come now to the main question put to me, and on which my counsel has been asked. That is, what the present Legislature should do in view of the dangers that threaten legislature should be in the or of the dangers that the the use and the wrongs that have been done us by several of our confederate States in the Union, by the acts of their legislatures nullifying the fugitive slave law, and in direct disregard of their constitutional obligations. What I shall

legislatures nullifying the fugifive slave law, and in direct disregard of their constitutional obligations. What I shall say will not be in the spirit of dictation; it will be simply my own judgment for what it is worth. It proceeds from a strong conviction that according to it our rights, interests and honor—our present safety and future security—can be maintained without yet looking to the last resort, the "*ultima ratio regum*." That should not be looked to until all else fails. That may come. On this p int I am hopeful, but not sanguine. But let us use every patriotic effort to prevent it while there is ground for hope. If any view that I may present in your judgment be in-consistent with the hest interests of Georgia, I ask you, as patriots, not to regard it. After hearing me and others whom you have advised with, act in the premises according to your own conviction of duty as patriots. I speak now particularly to the members of the legislature present. There are, as I have said, great dangers ahead. Great dan-gers may come from the election I have spoken of. If the policy of Mr. Lincoln and his Republican associates shall be carried out, or attempted to be carried out, no man in Georgia will be more willing or ready than myself to defend our rights, interests and honor, at every hazard and to the our rights, interests and honor, at every hazard and to the

our rights, interests and nonor, at every measure and to the last extremity. What is this policy? It is, in the first place, to exclude us, by an act of Congress, from the Territories with our slave property. He is for using the power of the General Government against the extension of our institutions. Our position on this point is, and ought to be, at all hazards, for perfect equality between all the States, and the citizens of all the States, in the Territories, under the Constitution of the United States. If Congress should exercise its power

against this, then I am for standing where Georgia planted herself in 1850. These were plain propositions, which were then laid down in her celebrated platform as sufficient for the disruption of the Union if the occasion should ever come. On these Georgia has declared that she will go out of the Union; and for these she would be justified by the nations of the earth in so doing.

I say the same; I said it then; I say it now—if Mr. Lin-coln's policy should be carried out. I have told you that I do not think his bare election sufficient cause: but if his policy should be carried out in violation of any of the principles set forth in the Georgia platform, that would be such an act of aggression which ought to be met as therein pro-vided for. If his policy shall be carried out in repealing or modifying the fugitive slave law so as to weaken its or mounying the rugitive slave law so as to weaken its efficacy, Georgia has declared that she will, in the last re-sort, disrupt the ties of the Union—and I say so too. I stand upon the Georgia platform, and upon every plank, and say, if these aggressions therein provided for take place—I say to you and to the people of Georgia, keep your counder dw, and lat your asseints then have loved if y cool powder dry, and let your assailants then have lead, if need be. I would wait for an act of aggression. This is my position.

Now upon another point, and that the most difficult, and deserving your most serious consideration, I will speak. That is the course which this State should pursue towards those Northern States, which by their legislative acts have attempted to nullify the fugitive slave law. I know that in some of these States their acts pretend to be based upon the principles set forth in the case of PRIGG against Penn-That decision did proclaim the doctrine that the sylvania. State officers are not bound to carry out the provisions of a law of Congress-that the Federal Government can not impose duties upon State officials—that they must execute their own laws by their own officers. And this may be true. But still it is the duty of the States to deliver fugitive slaves, as well as the duty of the General Government to see that it is done.

Northern States, on entering into the Federal compact, pledged themselves to surrender such fugitives; and it is in pleaged themselves to surrender such lugitives; and it is in disregard of their obligations that they have passed laws which even tend to binder or obstruct the fulfilment of that obligation. They have violated their plighted faith; what ought we to do in view of this? That is the question. What is to be done? By the law of nations you would have a right to demand the carrying out of this article of agreement, and I do not see that it should be otherwise with respect to the States of this Union; and, in case it be with respect to the States of this Union; and, in case it be not done, we would, by these principles, have the right to commit acts of reprisal on these faithless governments, and seize upon their property, or that of their citizens, wherever found. The States of this Union stand upon the same foot-ing with foreign nations in this respect. But, by the law of nations, we are equally bound, before proceeding to violent measures, to set forth our grievances before the offending Government, to give them an opportunity to re-dress the wrong. Has our State yet done this? I think not.

Suppose it was Great Britain that had violated some compact of agreement with the General Government, what would be first done? In that case our Minister would be directed, in the first instance, to bring the matter to the attention of that Government, or a Commissioner he sent to that country to open negotiations with her, ask for redress, and it would only be when argument and reason had been exhausted, that we should take the last resort of nations. That would be the course toward a foreign government, and toward a member of this Confederacy I would recommend

the same course. Let us, therefore, not act hastily in this matter. Let your Committee on the State of the Republic make out a bill of grievances; let it be sent by the Governor to those faithless States, and if reason and argument shall be tried in vain—all shall fail to induce them to return to their constitutional obligations-I would be for retaliatory measures, such as the Governor has suggested to you. This mode of resistance in the Union is in our power. It might be effect-ual, and, if in the last resort, we would be justified in the eyes of nations, not only in separating from them, but by using force.

[Some one said the argument was already exhausted.]

Mr. Strepters continued. Some friend says that the ar-gument is already exhausted. No, my friend, it is not. You have never called the attention of the Legislatures of those

have never called the attention of the Legislatures of those States to this subject that I am aware of. Nothing has ever been done before this year. The attention of our own people has been called to this subject lately. Now, then, my recommendation to you would be this: In view of all these questions of difficulty, let a convention of the people of Georgia be called, to which they may be all referred. Let the sovereignty of the people speak, Some think that the election of Mr. Lincoln is canse suffi-cient to disolve the Union. Some think those other refers. clear to dissolve the Union. Some think those other griev-ances are sufficient to dissolve the same, and that the Le-gislature has the power thus to act, and ought thus to act. I have no hesitancy in saying that the Legislature is not the

proper body to sever our Federal relations, if that necessity should arise. An honorable and distinguished gentleman, the other night (Mr. T. R. R. Cobb), advised you to take this course - not to wait to hear from the cross-route and groceries. I say to you, you have no power so to act. You must refer this question to the people, and you must wait to hear from the men at the cross-roads and even the groceries; for the people in this country, whether at the cross-roads or the propries in this country, whether in the country of a country is not in the proceeding of the provided of the country. Sovereignty is not in the Legislature. We, the people, are the sovereigns. I am one of them and have a right to be heard, and so has any other citizen of the State. You, Registators—I speak it respectfully—are but our servants. You are the servants of the people, and not their masters. Power resides with the people in this country.

The great difference between our country and all others, such as France and England and Ireland, is, that here there is popular sovereignty while there sovereignty is exercised by kings and favored classes. This principle of popular sovereignty, however much derided lately, is the founda-tion of our institutions. Constitutions are but the channels through which the popular will may be expressed. Our Constitution came from the people. They made it, and they alone can rightfully unmake it.

Mr. TOOMES. I am afraid of conventions.

Mr. STEPHENS. I am not afraid of any convention legally chosen by the people. I know no way to decide great questions affecting fundamental laws except by representa-tives of the people. The Constitution of the United States was made by the representatives of the people. The Constitution of the State of Georgia was made by representatives of the people chosen at the hallot-box. But do not let the question which comes before the people be put to them in the language of my honorable friend who addressed you last might: Will you submit to abolition rule or resist?

Mr. TOOMES. I do not wish the people to be cheated. Mr. STEPHENS. Now, my friends, how are we going to cheat the people by calling on them to elect delegates to a convention to decide all these questions without any dicta-tion or direction? Who proposes to cheat the people by letting them speak their own untrannelled views in the choice of their ablest and best men, to determine upon all these matters involving their peace?

I think the proposition of ny honorable friend had a con-siderable smack of unfairness, not to say cheat. He wished to have no convention, but for the Legislature to submit their vote to the people-submission to abolition rule or resistance? Now who, in Georgia, would vote "submis-sion to abolition rule?"

Is putting such a question to the people to vote on a fair way of getting an expression of the popular will on all these questions? I think not. Now, who in Georgia is these questions? I think not. Now, who in Georgia is going to submit to abolition rule? Mr. TOOMES. The convention will.

.81

Mr. STEPHENS. No, my friend, Georgia will never do it. The convention will never secede from the Georgia Platform. Under that there can be no abolition rule in the form. Under that there can be no about on the m the General Government. I am not afraid to trust the people in convention upon this and all questions. Besides, the Legislature were not elected for such a purpose. They came here to do their duty as legislators. They have sworn to support the Constitution of the United States. They did for submitting all these questions to a convention of the be submitting all takes questions to a convention of the people. Submit the question to the people, whether they would submit to an abolition rule or resist, and then let the Legislature act upon that vote? Such a course would be an insult to the people. They would have to eat their platform, ignore their past history, blot out their records, and take steps backward, if they should do this. I have never eaten my record or words and never will never eaten my record or words, and never will.

But how will it be under this arrangement if they should vote to resist, and the Legislature should reassemble with this vote as their instruction? Can any man tell what sort of resistance will be meant? One man would say secede; another pass retaliatory measures; these are measures of resistance against wrong-legitimate and right-and there would be as many different ideas as there are members on this floor. Resistance don't mean seces-sion—that, in no proper sense of the term, is resistance. Believing that the times require action, I am for presenting the question fairly to the people, for calling together an untrammelled convention, and presenting all the queswhat course of resistance in the Union they may think best, and then let the Legislature act, when the people in their majesty are beard; and I tell you now, whatever that Convention does, I hope and trust our people will abide by. I advise the calling of a convention with the anies by faire the rearring of a convention with the earnest desire to preserve the peace and harmony of the State. I should dislike, above all things, to see violent measures adopted, or a disposition to take the sword in hand, by individuals, without the authority of law.

My honorable friend said last night: "I ask you to giv a me the sword, for if you do not give it to me, as God live. I will take it myself."

Mr. TOOMBS. I will.

Mr. STEPHENS. I have no doubt that my honorable friend feels as he says. It is only his excessive ardor that makes him use such an expressiou; but this will pass off with the excitement of the hour. When the people in their majesty shall speak, I have no doubt that he will bow to their will, whatever it may be, upon the "sober second thought."

second thought." Should Georgia determine to go out of the Union—I speak for one, though my views might not agree with them—whatever the result may be, I shall bow to the will of her people. Their cause is my cause, and their destiny is my destiny; and I trust this will be the ulti-mate course of all. The greatest curse that can befall a free people is civil war. But, as I said, let us call a convention of the people; bit all these matters he authorited to it on a much will

let all these matters be submitted to it, and when the will of a majority of the people has thus been expressed, the whole State will present one unanimous voice in favor of whatever may be demanded; for 1 believe in the power of the people to govern themselves, when wisdom prevails and passion is silent.

Look at what has already been done by them for their advancement in all that eunobles man. There is nothing like it in the history of the world. Look abroad from one extent of the country to the other—contemplate our great-We are now among the first nations of the earth. ness. Shall it be said, then, that our institutions, founded upon principles of self-government, are a failure?

Thus far it is a noble example, worthy of imitation. The far it is a noble example, worthy of imitation. The gentleman, Mr. Cobb, the other night said it had proven a failure. A failure in what? In growth? Look at our expanse in national power. Look at our population and increase in all that makes a people great. A failure? Why, we are the admiration of the civilized world, and present the brightest hopes of mankind.

Some of our public men have failed in their aspirations; that is true, and from that comes a great part of our troubles.

No, there is no failure of this Government yet. We have made great advancement under the Constitution, and I cannot but hope that we shall advance higher still. Let us be true to our cause.

Let us be true to our cause. Now, wheu this convention assembles, if it shall be called, as I hope it may, I would say in my judgment, without dictation, for I am conferring with you freely and frankly, and it is thus that I give my views, I should take into consideration all those questions which distract the public mind; should view all the grounds of secession so far as the election of Mr. Lincoln is concerned, and I have no doubt they would say that the constitutional election of no man is a sufficient cause to break up the Union, but that the State should wait until he at least does some unconstitutional act does some unconstitutional act.

Mr. TCOMBS. Commit some overt act. Mr. STEPHENS. No, I did not say that. The word overt is a sort of technical term connected with treason, which has come to us from the mother country, and it means an open act of rebellion. I do not see how Mr. Lincoln can do this unless he should levy war upon us. I do not, therefore, use the word overt. I do not intend to wait for therefore, use the word overt. I do not intend to wait for that. But I use the words uncoustitutional act, which our people understand much better, and which expresses just what I mean. But as long as he conforms to the Constitution, he should be left to exercise the duties of his office.

In giving this advice I am but sustaining the Constitu-tion of my country, and I do not thereby become a Lincolu aid man either but a Constitutional aid man. But this matter the Convention can determine. As to the other matter, I think we have a right to pass

retallatory measures, provided they be in accordance with the Constitution of the United States, and I think they can be made such. But whether it would be wise for this be made such. But whether it would be wise for this Legislature to do this now is the question. To the Convention, in my judgment, this matter ought to be referred. Before we commit reprisals on New England we should exhaust every means of bringing about a peaceful solution

of the question. Thus did General Jackson in the case of the French. He did not recommend reprisals until he had treated with He did not recommend reprivate that he had the minimum of the frame, and got her to promise to make indemnification, and it was only on her refusal to pay the money which she had promised that he recommended reprisals. It was and it was only on her retusal to pay the money which she had promised that he recommended reprisals. It was after negotiation had failed. I do think, therefore, that it would be best, before going to extreme measures with our Confederate States, to make a presentation of our de-mands, to appeal to their reason and judgment to giver us our rights. Then, if reason should not triumph, it will be time enough to commit reprisals, and we should be justified in the eyes of a civilized world. At least, let the States know what your grievances are, and if they refuse, as I said, to give us our rights under the Constitution of our country, I should be willing, as a last resort, to sever the ties of this Union.

to sever the ties of this Union. My own opiuion is, that if this course be pursued, and they are informed of the consequences of refusal, these States will secede; but if they should not, then let the consequences he with them, and let the responsibility of the consequences rest upon them. Another thing I would have that Convention to do. Reaffirm the Georgia platform with an additional plank in it. Let that plank be the fulfilment of the obligation on the part of those States to repeal these obnoxious laws as a condition of our remaining in the Union. Give them time to consider it, and I would ask all States South to do the same thing.

I am for exhausting all that patriotism can demand before taking the last step. I would invite, therefore, South Carolina to a conference. I would ask the same of all the other Southern States, so that if the evil has got beyond our control, which God, in his meroy, grant may not be the case let us not be divided among ourselves, but, if possible, secure the united co-operation of all the Southern States; and then, in the face of the civilized world, we may justify our action; and, with the wrong all on the other side, we can appeal to the God of battles to aid us in our cause. But let us not do any thing in which any portion of our people may charge us with rash or hasty action. It is certainly a matter of great importance to tear this Government asunder. You were not sent here of r that purpose. I would wish the whole South to be nnited if this is to be done; and I believe if we pursue the policy which I have indicated, this can be effected.

the policy which I have indicated, this can be effected. In this way our sister Southern States can be induced to act with us, and I have but little doubt that the States of New York, and Pennsyivania, and Ohio, and the other Western States, will compet their Legislatures to recede from their hostile attitudes if the others do not. Then with these we would go on without New England if she choese to stay out.

With these we would go on without new Enginee it one chose to stay out. [A voice in the assembly. "We will kick them out."] Mr. STEPRENS. I would not kick them out. But if they chose to stay out, they might. I think, moreover, that these Northern States, being principally engaged in manufactures, would find that they had as much interest in the Union under the Constitution as we, and that they would return to their constitution as we, and that they would return to their constitution at dif the Middle States and Western States do not join us, we should at least have an undivided South I am, as you clearly perceive, for maintaining the Union as it is, if possible. I will exhaust every means thus to maintain it with an equality in it. My principles are these:

Wy principles are these: First, the maintenance of the honor, the rights, the equality, the security, and the glory of my native State in the Union; but if these cannot be maintained in the Union, then I am for their maintenance, at all hazards, out of it. Next to the honor and glory of Georgia, the land of my hirth, I hold the honor and glory of our common country. In Savannah I was made to say, by the reporters—who very often make me say thiugs which I never did say—that I was first for the glory of the whole country, and next for that of Georgia.

dis say—that I was first for the second try, and next for that of Georgia. Isaid the exact reverse of this. I am prond of her history, of her present standing. I am prond even of her motio, which I would have duly respected at the present time by all her sons—Wisdom, Justice, and Moderation. I would have her rights and that of the Southern States maintained now upon these principles. Her position now is just what it was in 1850, with respect to the Southern States. Her platform then has been adopted by most, if not all, the other Southern States. Now I would add but one additional plank to that platform, which I have stated, and one which time has shown to be necessary.

If all this fails, we shall at least have the satisfaction of knowing that we have done our duty and all that patriotism could require.

### From Mr. STEPHENS'S speech in the State Convention of Georgia:

This step (of secession) once taken, can never be recalled; and all the baleful and withering consequences that must follow, will rest on the convention for all coming time. When we and our postority shall see our lovely South desolated by the demon of war, which THIS ACT OF YOURS WILL INEVITABLY INVITE AND CALL FORTH; when our green fields of waving harvest shall be trodden down by the murderous soldiery and flery car of war sweeping over our land; our temples of justice laid in ashes; all the horrors and desolations of war upon us; who BUT THIS CONVENTION WILL BE HELD RESPONSIBLE FOR IT? and who but him who shall have given his vote for this unwise and ill-timed measure, as I honestly think and believe, SHALL BE HELD TO STRICT ACCOUNT FOR THIS

-

SUICIDAL ACT BY THE PRESENT GENERATION, AND PROBABLY SUCIDAL ACT BY THE PRESENT GENERATION, AND PROBABLY CURSED AND EXECRATED BY POSTERITY FOR ALL COMING TIME, for the wide and desolating ruin that will inevi-tably follow this act you now propose to perpetrate? Pause, I entreat you, and consider for a moment what reasons you can give that will even satisfy yourselves in calmer moments—what reasons you can give to your fellow-sufferers in the calamity that it will bring upou s. Wink presents can you give to The NATIONS OF US. WHAT REASONS CAN YOU GIVE TO THE NATIONS OF THE EARTH TO JUSTIFY IT? They will be the calm and deliherate judges in the case; and what cause or one overt act can you name or point, on which to rest the plea of justification? WHAT RIGHT HAS THE NORTH ASSAILED? What interest of the South has been invaded? Assauled ? what interest of the owner has been with the second market claim founded in justice and right has been withheld? Cau either of you to day name one governmental act of wrong, deliber-ately and purposely done by the government of Wash-ington, of which the South has a right to complain? I challenge the answer. While, on the other hand, let me challenge the answer. While, ou the other hand, let me show the facts (and believe me, gentlemen, I am not here the advocate of the North; but I am here the friend, the firm friend, and lover of the South and her institutions, and for this reason I speak thus plainly and faithfully for yours, mine, and every other mau's interest, the words of truth and soberness), of which I wish you to judge, and I will only state facts which are clear and Judge, and I will only state facts which are clear and undeniable, and which now state and as records authentic in the history of our country. When we of the South demanded the slave-trade, or the importation of Africans for the cultivation of our lands, did they not yield the right for twenty years? When we asked a three-fifths representation in Congress for our slaves, was it not granted? When we asked and demanded the return of the representation for our source of those horsen any fugilive from justice, or the recovery of those persons owing labor or allegiance, was it not incorporated in the Constitution, and again ratified and strengthened by the Fugitive Slave Law of 1850? But do you reply that in many instances they have violated this compact, and many instances incy nave violated this compact, and have not been faithful to their engagements? As indi-vidual and local communities, they may have done so; but not by the sanction of Government; for that has always been true to Southern interests. Again, gentlemen, look at another act: when we have asked that more territory should be added, that we might spread the institution of slavery, have they not yielded to our demands in giving us Louisiana, Florida aud Texas, out of which four States have been carved, and ample of which four States have been carved, and ample territory for four more to be added in due time, if you by this unwise and impolitic act do not destroy this hope, and, perhaps, by it lose all, and have your last slave wrenched from you by stern military rule, as South America and Mexico were; or by the vindictive decree of a universal emancipation, which may reasonably be armeted to follow? expected to follow ?

But, again, gentlemen, what have we to gain by this proposed change of our relation to the General Government? We have always had the control of it, and can yet, if we remain in it, and are as united as we have been. We have had a majority of the Presidents chosen from the South; as well as the control and management of most of those chosen from the North. We have had sixty years of Southern Presidents to their twenty-four, thus controlling the Executive department. So of the Judges of the Supreme Court, we have had eighteen from the South, and but eleven from the North; although nearly four-fifths of the judicial busines has arisen in the Free States, yet a majority of the Court has always been from the South. This we have nequired so as to guard against any interpretation of the Constitution unfavorable to us. In like manner we have been equally watchful to guard our interests in the Legislative branch of Governmeut. In choosing the presiding Presidents (pro tem.) of the Senate, we have had twenty-four to their eleven. Speakers of the House, we have had twentythree, and they twelve. While the majority of the Representatives, from their greater population, have always heen from the North, yet we have so generally secured the Speaker, because he, to a great extent, shapes and controls the legislation of the conntry. Nor have we had less control in every other department of the General Government. Attorney-Generais we have had fourteen, while the North have had but five. Foreign ministers we have had eighty-six, and they but fifty-four. While three-fourths of the business which demands diplomatic agents abroad is clearly from the Free States, from their greater commercial interests, yet we have had the principal embassies so as to secure the world-markets for our coton, tobacco, and sugar on the best possible terms. We have bad a vast majority of the higher offices of holt army and navy, while a larger proportion of the soldiers and sailors were drawn from the North. Equally so of Gierks, Auditors, and Com of the three thousand thus employed, we have had more than two-thirds of the same, while we have but one-third of the white populatiou of the Republic.

Again, look at another item, and one, be assured, in which we have a great and vital interest; it is that of revenue, or means of supporting Government. From official documents, we learn that a fraction over threefourths of the revenue collected for the support of Government has uniformly been raised from the North.

Pause now while you can, gentlemen, and contemplate carefully and candidly these important items. Look at another necessary branch of Government, and learn from stern statistical facts how matters stand in that department. I mean the mail and Post-Office privileges that we now enjoy under the General Government as it has been for years past. The expense for the transportation of the mail in the Free States was, by the report of the Postmaster-General for the year 1860, a little over \$13,000,000, while the income was \$19,000,000. But in the Slave States the transportation of the mail was \$3,001,026, leaving a deficit of \$6,704,974, to be supplied by the North for our accommodation, and without it we anust have been entirely cut off from this most essential branch of Government.

Leaving out of view, for the present, the countless millions of dollars you must expend in a war with the North; with tens of tbousands of your sons and brothers slain in battle, and offered up as sacrifices upon the altar of your ambition—and for what, we ask again? I sit for the overthrow of the American Government, established by our common ancestry, cemented and built up by their sweat and blood, and founded on the broad principles of *Right, Justice* and *Humanity?* And, as such, I must declare here, as I have often done before, and which has been repeated by the greatest and wisest of statesmen and patriots in this and other lands, that it is the best and the most aprincip in its principles to levate the race of men, that the sun of heaven ever shone upon. Now, for you to attempt to overthrow such a Government as this, under which we have lived for more than three-quarters of a century—in which we have gained our wealth, our standing as a nation, our domestic safety while the elements of peril are around us, with peace and tranrights unassailed—is the height of madness, folly, and *vickedness*, to which I can neither lend my sanction nor my vote.

In strong contrast with the doleful narrative of the South Carolina Secessionists, are the following extracts touching the point of the security and prosperity of the Slave system:

From the speech of Hon. JAMES H. HAMMOND, U. S. Senator from South Carolina, delivered at Barnwell Court House, October 27, 1858.

From the time that the wise and good Las Casas first introduced into America the institution of African slavery -I say institution, because it is the oldest that exists, aud will, I believe, survive all others that now flourishit has had its enemies. For a long while they were chiefly men of peculiar and eccentric religious notions. Their first practical and political success arose from the convulsions of the French revolution, which lost to that empire its best colony. Next came the prohibition of the slave-trade, the excitement of the Missonri compromise in this country, and then the deliberate emancipation of the slaves in their colonies by the British Government in 1833-'34. About the time of the passage of that act the abolition societies were formed. I remember the time well, and some of you do also. And what then was the state of opinion in the South 7 Washington had emancipated his slaves. Jefferson had bitterly denounced the system, and had done all that he

And what then was the state of opinion in the South ? Washington had emancipated his slaves. Jefferson had bitterly denounced the system, and had done all that he could to destroy it. Our Clays, Marshalls, Crawfords, and many other prominent Southern men, had led off in the colouization scheme. The inevitable effect in the South was that she believed slavery to be an evil—weakness—disgraceful—nay, a sin. She shrunk from the discussion of it. She cowered under every threat. She attempted to apologize, to excuse herself under the plea which was true—that England had forced upon her: and in fear and trembling she awaited a doom that she deemed inevitable. But a few bold spirits took the question up : they compelled the South to investigate it anew and

thoroughly, and what is the result? Why it would be difficult to find now a Southern man who feels the system to be the slightest burden on his conscience; who does not, in fact, regard it as an equal advantage to the master and the slave elevating both, as wealth, strength, and power, and as one of the main pillars and controlling influences of modern civilization, and who is not now prepared to maintain it at every hazard. Such have been for us the happy results of this abolition discussion. So far our gain has been immense from this contest, sawage and matignant as it has been. Nay, we have solved already the question of emancipation by this re-examination and exposition of the false thereise of religion, philanthropy, and political economy which embarrassed our fathers in their day. With our convictions and our strength, emancipation

With our convictions and our strength, emancipation here is simply an impossibility to man, whether by persuasion, purchase, or coercion. The rock of Gibraltar does not stand so firm on its basis as our slave system. For a quarter of a century it has borne the brunt of hurricane as fierce and pitiless as ever raged. At the North and in Europe they cried "havoe," and let loose upon us all the dogs of war. And how stands it now? Why, in this very quarter of a century our slaves have doubled in value. The very negro who as a prime laborer would have brought \$400 in 1828, would now, with thirty more years upon him, eell for \$800. What does all this mean? Why, that we ourselves have settled this question of emancipatoin against all the world, in theory and in practice, and the world must accept our solution.

From the carefully-prepared speech of Hon. ALEX. H. STEPHENS of Georgia, in July, 1859, after his retirement from Congress, and in review of his political course :

Nor am I of the number of those who believe that we have sustained any injury by those agitations. It is true, we were not responsible for them. We were not the aggressors. We acted on the defensive. We repeiled assault, calumny, and aspersion, by argument, by reason and truth. But so far from the institution of African slavery in our section being weakened or rendered less secure by the discussion, my deliberate judgment is, that it has been greatly strengthened and fortified—strengthened and fortified not only to the opinions, convictions, and consciences of men, but by the action of the Government.

From the Charlottesville (Va.) speech of Hon. ROBERT M. T. HUNTER, U. S. Senator from Virginia, at the Breckinridge Democratic State Convention, 1860:

When I first entered the Federal councils, which was at the commencement of Mr. Van Buren's administration, the moral and political status of the slavery question was very different from what it now is. Then the Southern men themselves, with but few exceptions, admitted slavery to be a moral evil, and palliated and excused it upon the plea of necessity. Then there were few men of any party to be found in the non-slaveholding States who did not maintain both the constitutionality and expediency of the anti-lavery resolution, now generally known as the Wilmot Proviso. Had any man at that day ventured the prediction that the Missouri restriction would ever be repealed, he would have been deemed a visionary and theorist of the wildest sort. What a revolution have we not witnessed in all this ! The discussion and the contest on the slavery question have gone on ever since, so as to absorth almost entirely the American mind. Ia many respects the results of that discussion have not been adverse to us. Southern men no longer occupy a depreadory attitude upon the question of negro slavery in this country. While they by no means pretend that slavery is a good condition of things, under any circumstances and in all conduction of things, under any circumstances and in all conduction of things, under any circumstances and in all conduction of things, under any circumstances and in all conduction of things, under any circumstances and in all conduction of things, under any circumstances and in all conduction of things, under any circumstances and in all conduction of things, under any circumstances and in all conduction of things, under any circumstances and in all conduction of things (so each other here, it is best for both that the inferior should be subjected to the soupled. As evidence, too, of the growing change on this subject of the public sentiment of the world, I may refer to the course of France and Great Britain in regard to the coolie and the African apprenticeship system introduced into ther colonie

From the appeal for recognition, made to Earl Russell, by WM. L. YANCEY, P. A. ROST and A. DUDLEY MANN, Rebel Commissioners, dated:

No. 15 Half Moon Street, London, August 14th, 1861. It was from no fear that the Slaves would be liberated that secession took place. The very party in power has proposed to guarantee slavery forever in the States, if the South would but remain in the Union. Mr. Lincoln's mes-sage proposes no freedom to the slave, but announces subsage proposes no freedom to the slave, but announces sub-jection of his owner to the will of the Union, in other words to the will of the North. Even after the battle of Bull Run, both branches of the Congress at Washington passed resolutions that the war is only waged in order to uphold that (Pro-Slavery) Constitution, and to enforce the laws (many of them Pro-Slavery), and out of 172 votes in the lower House they received all but two, and in the Senate all but one vote. As the army commenced its march, the commanding-general issued an order that no slaves should be received into, or allowed to follow, the camp.

The great object of the war, therefore, as now officially annonuced, is not to free the slave, but to keep him in subjection to his owner, and to control his labor through the legislative channels which the Lincolu Government designs to force upon the master. The undersigned, there-fore, submit with confidence that as far as the anti-slavery sentiment of England is concerned, it can have no sympa-thy with the North; nay, it will probably become disgusted with a canting hyporrisy which would enlist those sym-pathies on false pretences. The undersigned are, however, not insensible to the surmise that the Liucolu Government may, under stress of circumstances, change its policy, a policy based at present more upon a wily view of what is to be its effect in rearing up an element in the Confederate States favorable to the reconstruction of the Union erate States favorable to the reconstruction of the Union than upon any houses desire to uphold a Constitution, the main provisions of which it has most shamelessly violated. But they confidently submit to your Lordship's consider-ation that success in producing so abrupt and violent a destruction of a system of labor which has reared up so vast a commerce between America and the great States of Europe, which, it is supposed, now gives bread to 10,000,000 of the population of those States, which, it may be safely assumed, is intimately blended with the basis of the great disting assumed, is initiately included with the basis of the presence of the distin-guishes the age, and probably not the least of the elements of this prosperity, would be visited with results disastrous to the world, as well as to the master and slave.

These Commissioners made a verbal statement to Earl Russell, May 4th, 1861, as appears from the despatch of the Earl to Lord Lyons:

Foreign Office, May 11th, 1861. My Lord: -On Saturday last I received at my house Mr. Yancey, Mr. Mann, and Jurge Rost, the three gentle-men deputed by the Southern Confederacy to obtain their recognition as an independent State.

recognition as an independent State. One of these gentlemen, speaking for the others, dilated on the causes which had induced the Southern States to seede from the Northern. The principul of these causes, he said, was not slavery, but the very high price which, for the sake of protecting the Northern manufraturers, the South were obliged to puy for the munufratured goods which they required. One of the first acts of the Southern Congress was to reduce these duties, and to prove their sincerity he gave as an instance that Louisiana had given up altogether that protection on her snear which is the ensincerity he gave as an instance that Louisiana had given up altogether that protection on her sugar which she en-joyed by the legislation of the United States. As a proof of the riches of the South, he stated that of \$350,000,000 of exports of produce to foreign countries, \$270,000,000 were furnished by the Southern States. I said that I could hold no official communication with the delegates of the Southern States. That, however, when the question of recognition came to be formally dis-cussed there were two points upon which incoming more

ensed, there were two points upon which inquiry must be made—first, whether the body seeking recognition could maintain its position as an independent State; secondly, in what manner it was proposed to maintain relations with foreign States.

After speaking at some leugth on the first of these points, and allading to the news of the secession of Virginia and other intelligence favorable to their cause, these gentleother intelligence havorable to their cause, these gentle-men called my attention to the article in their constitution prohibiting the slave-trade. I said that it was alleged very enruently that if the Slave States found that they could not compete successfully with the cotton of other conntries they would revive the slave-trade for the pur-pose of diminishing the cost of production. They said

this was a suspicion unsupported by any proof. The fact was, that they had prohibited the slave-trade, and did not mean to revive it. They pointed to the new tariff of the United States as a proof that British manufacturers would be nearly excluded from the North, and freely admitted in the South. Other observations were made, but not of very great importance. The delegates concluded by stating that they should re-

main in London for the present, in the hope that the recog-nition of the Southern Confederacy would not be long delayed. I am, etc., J. RUSSELL.

To all these arguments and suggestions, the Earl was deaf, and the Confederate States of America are still unrecognized

### Seizures and Surrenders, From November 4, 1860, to March 4, 1861.

### SOUTH CAROLINA.

December 27th, 1860. Fort Moultrie and Castle Pinckney, light-house tender and schooner William Aiken, surrendered by Captain Coste of South Carolina.

31st. U. S. Arsenal, Post-Office, and Custom-House in Charleston; arsenal containing seventy thousand stand of arms, and other stores.

January 9th, 1861. Steamer Marion at Charleston. Star of the West fired upon.

April 13th. Fort Sumter surrendered.

### GEORGIA.

January 2d, 1861. Forts Pulaski and Jackson and United States Arsenal, by State troops, under advice from Georgia members of Congress.

24th. Arsenal at Augusta, containing two 12-pound howitzers, two cannon, 22,000 muskets and rifles, and large stores of powder, balls, grape, etc. U.S. steamer Ida seized.

February 8th. Brig W. R. Kibby, and four other New York vessels, estimated at \$50,000, seized by order of the Governor of Georgia, to be held until certain guns on board the Monticello, seized by the police of New York, shall be delivered to the agents of Georgia. Collector of the port of Savannah ordered by the Governor of Georgia to retain all moneys from customs in his possession, and make no payment on account of the Federal Government.

21st. Three New York vessels seized at Savannah by order of the Governor.

### FLORIDA.

January 12th, 1861. The Navy Yard, and Forts Barrancas and McRae, taken by Florida and Alabama troops. Jan. 7, Fort Marion and the arsenal at St. Augustine. The Chattahoochee arsenal taken, containing 500,000 rounds of musket cartridges, 300,000 rifle cartridges, and 50,000 pounds of gunpowder, but no arms. Coast survey schooner F. W. Dana seized.

### ALABAMA.

January 4th, 1861. Fort Morgan seized by Mobile troops, containing about 5,000 shot and shell. Also Mt. Vernon Arsenal, containing 20,000 stand of arms, 1,500 barrels of powder, (150,000 pounds), some pieces of cannon, and large amount of munitions of war. Captain Morrison, of the revenue cutter Lewis Cass, surrendered his vessel on the demand of the collector of Mobile, and subsequently took command of it under authority of the State.

#### MISSISSIPPI.

January 20th, 1861. Fort at Ship Island seized, and the U.S. Hospital on the Mississippi river.

### LOUISIANA.

January 11th, 1861. Forts Jackson and St. Philip, on the Mississippi, and Fort Pike, on Lake Pontchartrain, and the arsenal at Baton Rouge, seized by State troops. The arsenal contained 50,000 small arms, 4 howitzers, 20 heavy pieces of ordnance, 2 batteries, 300 barrels of powder, etc. Also, U. S. hospital at New Orleans.

12th. The entire armament of the revenue cutter Lewis Cass stored at Bellville Iron Works.

28th. All quartermasters' and commissary stores in possession of U. S. officials. Revenue cutter McClelland surrendered by Captain Breshwood.

February 1st. Mint and Custom-House containing \$599,303 in gold and silver.

### TEXAS.

January 10th, 1861. U. S. guns and stores seized on steamship Texas by Galveston troops.

February 20th. Forts Chadbourne and Belknap seized by Texans, with all the property of the Overland Mail Company.

25th. General Twiggs surrendered all Government stores in his command, after hearing that he had been superseded in command by Colonel Waite. 'The stores estimated at \$1,300,000 in value, consisting of \$55,000 in specie, 35,000 stand of arms, 26 pieces of mounted artillery, 44 dismounted, ammunition, horses, wagons, forage, etc.

March 2d. Revenue cutter Dodge seized by Texas authorities, in Galveston bay. 6th. Fort Brown surrendered.

### ARKANSAS.

February 8th, 1861. Arsenal at Little Rock seized, containing 9,000 small arms, 40 cannon, and quantities of ammunition, etc.

### NORTH CAROLINA.

January 8th, 1861. Forts Johnson and Caswell seized by the State militia. Governor Ellis ordered them to be surrendered to the United States authorities, with an intimation to the President that if any attempt should be made to reinforce them, they would be again seized and held by the State.

Governor LETCHER of Virginia, in his annual message of December 31st, 1861, thus alluded to Fortress Monroe :

"It is to be regretted that Fortress Monroe is not in our possession; that it was not as easily captured as the Navy Yard and Harper's Ferry. As far back as the his successor.

8th of January last, I consulted with a gentleman whose position enabled him to know the strength of that fortress, and whose experience in military matters enabled him to form an opinion as to the number of men that would be required to capture it. He represented it to be one of the strongest fortifications in the world, and expressed his doubts whether it could be taken, unless assailed by water as well as by land, and simultaneously. He stated, emphatically and distinctly, that with the force then in the fortress, it would be useless to attempt its capture without a large force thoroughly equipped and well appointed. At no time previous to the secession of Virginia had we a military organization sufficient to justify an attempt to take it, and events since that occurrence demonstrate very clearly that with our military organization since, and now existing, it has not been deemed prudent to make the attempt."

### Mr. Buchanan's Cabinet.

December 12th, 1860. LEWIS CASS resigned as Secretary of State, because the President declined to reinforce the forts in Charleston harbor. December 17th. JEREMIAH S. BLACK was appointed his successor.

December 10th. HOWELL COBB resigned as Secretary of the Treasury—"his duty to Georgia requiring it." December 12th. PHILIP F. THOMAS appointed his successor, and resigned, January 11th, 1861, because differing from the President and a majority of the cabinet, "in the measures which have been adopted in reference to the recent condition of things in South Carolina," especially "touching the authority, under existing laws, to enforce the collection of the customs at the port of Charleston." January 11th, 1861. JOHN A. DIX appointed his successor.

29th. JOHN B. FLOYD resigned as Secretary of War, because, after the transfer of Major Anderson's command from Fort Moultrie to Fort Sumter, the President declined "to withdraw the garrison from the harbor of Charleston altogether."

December 31st. JOSEPH HOLT, Postmaster-General, was entrusted with the temporary charge of the War Department, and January 18th, 1861, was appointed Secretary of War.

January 8th, 1861. JACOB THOMPSON resigned as Secretary of the Interior, because "additional troops, he had heard, have been ordered to Charleston" in the Star of the West.

December 17th, 1860. JEREMIAH S. BLACK resigned as Attorney-General, and Edwin M. Stanton, December 20th, was appointed his successor.

January 18th, 1861. JOSEPH HOLT resigned as Postmaster-General, and HORATIO KING, February 12th, 1861, was appointed his successor.

### Correspondence between President Buchanan and the "Commissioners" of South Carolina.

### The "Commissioners" to the President. Washington, Dec. 28th, 1860.

SIR: We have the honor to transmit to you a copy of the full powers from the Convention of the people of South Carolina, under which we are "authorized and empowered to treat with the Government of the United States for the delivery of the forts, magazines, lighthouses, and other real estate, with their appurtenances, in the limits of South Carolina; and also for an apportionment of the public debt, and for a division of all other property held by the Government of the United States as agent of the Confederated States of which South Carolina was recently a member, and generally to negotiate as to all other measures and arrangements proper to be made and adopted in the existing relation of the parties, and for the continuance of peace and amity between this Commonwealth and the Government at Washington."

In the execution of this trust, it is our duty to furnish you, as we now do, with an official copy of the Ordinance of Secession by which the State of South Carolina has resumed the powers she delegated to the Government of the United States, and has declared her perfect sovereignty and independence.

It would also have been our duty to have informed you that we were ready to negotiate with you upon all such questions as are necessarily raised by the adoption of this ordinance, and that we were prepared to enter upon this negotiation with the earnest desire to avoid all unnecessary and hostile collision, and so to inaugurate our new relations as to secure mutual respect, general advantage, and a future of good-will and harmony beneficial to all the parties concerned. But the events of the last twenty-four hours render such an assurance impossible.

We came here the representatives of an authority which could, at any time within the past sixty days, have taken possession of the forts in Charleston harbor, but which, upon pledges given in a manner that we cannot doubt, determined to trust to your honor rather than to its own power. Since our arrival here, an officer of the United States, acting, as we are assured, not only without, but against your orders, has dismantled one fort and occupied another; thus altering, to a most important extent, the condition of affairs under which we came.

Until the circumstances are explained in a manner which relieves us of all doubt as to the spirit in which these negotiations shall be conducted, we are forced to suspend all discussion as to any arrangements by which our mutual interests might be amicably adjusted. And, in conclusion, we would urge upon you the immediate withdrawal of the troops from the harbor of Charleston. Under present circumstances they are a standing menace which renders negotiation impossible, and, as our recent experience shows, threatens speedily to bring to a bloody issue questions which ought to be settled with temperance and judgment.

We have the honor to be, very respectfully, your obedient servants,

	R. W. BARNWELL,
	J. H. Adams,
	JAMES L. ORR,
	Commissioners.
To the President of the	United States.

The President to the "Commissioners." Washington, December 30th, 1860.

GENTLEMEN: I have had the honor to receive your communication of the 28th inst., together with a copy of "your full powers from the Convention of the people of South Carolina," authorizing you to treat with the Government of the United States on various important subjects therein mentioned, and also a copy of the Ordinance, bearing date on the 20th inst., declaring that "the Union now subsisting between South Carolina and other States, under the name of the United States of America, is hereby dissolved."

In answer to this communication, I have to say that my position as President of the United States was clearly defined in the message to Congress on the 3d inst. In that I stated that "apart from the execution of the laws, so far as this may be practicable, the Executive has no authority to decide what shall be the relations between the Federal Government and South Carolina. He has been invested with no such discretion. He possesses no power to change the relations heretofore existing between them, much less to acknowledge the independence of that State. This would be to invest a mere executive officer with the power of recognizing the dissolution of the Confederacy among our thirty-three sover-eign States. It bears no resemblance to the recognition of a foreign de facto Government, involving no such responsibility. Any attempt to do this would, on his part, be a naked act of usurpation. It is, therefore, my duty to submit to Congress the whole question in all its bearings."

Such is my opinion still. I could, therefore, meet you only as private gentlemen of the highest character, and was entirely willing to communicate to Congress any proposition you might have to make to that body upon the subject. Of this you were well aware. It was my earnest desire that such a disposition might be made of the whole subject by Congress, who alone possess the power, as to prevent the inauguration of a civil war between the parties in regard to the possession of the Federal forts in the harbor of Charleston, and I therefore deeply regret that, in your opinion, "the events of the last twenty-four hours render this impossible."

In conclusion, you urge upon me the "immediate withdrawal of the troops from the harbor of Charleston," stating that "under present circumstances they are a standing menace which renders negotiation impossible, and, as our recent experience shows, threatens speedily to bring to a bloody issue questions which ought to be settled with temperance and judgment."

The reason for this change in your position is, that since your arrival in Washington, "an officer of the United States, acting, as we, (you) are assured, not only without, but against your (my) orders, has dismantled one fort and occupied anotherthus altering to a most important extent the condition of affairs under which we (you) came." You also allege that you came here "the representatives of an authority which could, at any time within the past sixty days, have taken possession of the forts in Charleston harbor, but which, upon pledges given in a manner that we (you) cannot doubt, determined to trust to your (my) honor rather than to its power."

This brings me to a consideration of the nature of those alleged pledges, and in what manner they have been observed.

In my message of the 3d of December last, I stated, in regard to the property of the United States in South Carolina, that it "has been purchased for a fair equivalent, by the consent of the Legislature of the State," for the erection of forts, magazines, arsenals, &c., and over these the authority "to exercise exclusive legislation" has been expressly granted by the Constitution to Congress. It is not believed that any attempt will be made to expel the United States from this property by force; but if in this I should prove to be mistaken, the officer in command of the forts has received orders to act strictly on the defensive. In such a contingency, the responsibility for consequences would rightfully rest upon the heads of the assailants.

This being the condition of the parties, on Saturday, 8th December, four of the Representatives from South Carolina called upon me and requested an interview. We had an earnest conversation on the subject of these forts, and the best means of preventing a collision between the parties, for the purpose of sparing the effusion of blood. I suggested, for prudential reasons, that it would be best to put in writing. what they said to me verbally. They did so accordingly, and on Monday morning, the 10th instant, three of them presented to me a paper signed by all the Representatives from South Carolina, with a single exception, of which the following is a copy:

\*: · ·

Washington, December 9th, 1860. "To His Excellency James Buchanan, President of the United States :

"In compliance with our statement to you yesterday, we now express to you our strong convictions that neither the constituted authorities, nor any body of the people of the State of South Carolina, will either attack or molest the United States forts in the harbor of Charleston, previously to the action of the Convention, and we hope and believe not until an offer has been made through an accredited representative to negotiate for an amicable arrangement of all matters between the State and the Federal Government; provided that no reinforcements shall be sent into those forts and their relative military status shall remain as at present.

JOHN MCQUEEN, M. L. BONHAM, W. W. BOYCE,

LAWRENCE M. KEITT."

And here I must, in justice to myself, remark that at the time the paper was presented to me I objected to the word "provided," as it might be construed into an agreement on my part, which I never would make. They said that nothing was farther from their intention; they did not so understand it, and I should not so consider it. It is evident they could enter into no reciprocal agreement with me on the subject. They did not profess to have the authority to do this, and were acting in their individual character. I considered it as nothing more, in effect, than the promise of highly honorable gentlemen to exert their influence for the purpose expressed. The event has proven that they have faithfully kept this promise, although I have never since received a line from any one of them nor from any member of the Convention on the sub-It is well known that it was my deject. termination, and this I freely expressed, not to reinforce the forts in the harbor, and thus produce a collision, until they had been actually attacked or until I had certain evidence that they were about to be at-This paper I received most cortacked. dially, and considered it as a happy omen that peace might be still preserved, and that time might be thus given for reflection. This is the whole foundation for the alleged pledge.

But I acted in the same manner as I would have done had I entered into a positive and formal agreement with parties capable of contracting, although such an agreement would have been on my part, from the nature of my official duties, impossible. The world knows that I have never sent any reinforcements to the forts in Charleston harbor, and I have certainly never authorized any change to be made "in their relative military status." Bearing upon this subject, I refer you to an order issued by the Secretary of War, on the 11th instant, to Major Anderson, but not brought to my notice until the 21st instant. It is as follows:

"Memorandum of verbal instructions to Major Anderson, First Artillery, commanding Fort Moultrie, South Carolina.

"You are aware of the great anxiety of the Secretary of War that a collision of the troops with the people of this State shall be avoided, and of his studied determination to pursue a course with reference to the force and forts in this harbor which shall guard against such a collision. He has therefore carefully abstained from increasing the force at this point, or taking any measure which might add to the present excited state of the public mind, or which would throw any doubt on the confidence he feels that South Carolina will not attempt by violence to obtain possession of the public works, or interfere with their occupancy.

"But, as the counsel and acts of rash and impulsive persons may possibly disappoint these expectations of the Government, he deems it proper that you should be prepared with instructions to meet so unhappy a contingency. He has, therefore, directed me verbally to give you such instructions.

"You are carefully to avoid every act which would needlessly tend to provoke aggression, and for that reason you are not, without necessity, to take up any position which could be construed into the assumption of a hostile attitude; but you are to hold possession of the forts in this harbor, and, if attacked, you are to defend yourself to the last extremity. The smallness of your force will not permit you, perhaps, to occupy more than one of the three forts; but an attack on or attempt to take possession of either of them will be regarded as an act of hostility, and you may then put your command into either of them which you may deem most proper to increase its power of resistance. You are also authorized to take similar steps whenever you have tangible evidence of a design to proceed to a hostile act. D. P. BUTLER,

"Assistant Adjutant-General.

"Fort Moultrie, (S. C.) Dec. 11th, 1860. "This is in conformity to my instructions

to Major Buell.

"JOHN B. FLOYD, Secretary of War." These were the last instructions transmitted to Major Anderson before his removal to Fort Sumter, with a single exception, in regard to a particular which does not in any degree affect the present question.

Under these circumstances it is clear that Major Anderson acted upon his own responsibility, and without authority,—unless, indeed, he "had tangible evidence of a design to proceed to a hostile act" on the part of the authorities of South Carolina, which has not yet been alleged. Still he is a brave and honorable officer, and justice requires that he should not be condemned without a fair hearing. Be this as it may, when I learned that Major Anderson had left Fort Moultrie and

proceeded to Fort Sumter, my first promptings were to command him to return to his former position, and there to await the contingencies presented in his instructions. This could only have been done with any degree of safety to the command by the concurrence of the South Carolina authorities. But before any step could possibly have been taken in this direction, we received information that the "Palmetto flag floated out to the breeze at Castle Pinckney, and a large military force went over last night (the 27th) to Fort Moultrie." Thus the authorities of South Carolina, without waiting or asking for any explanations, and doubtless believing, as you have expressed it, that the officer had acted not only without but against my orders, on the very next day after the night when the removal was made, seized, by a military force, two of the three Federal forts in the harbor of Charleston, and have covered them under their own flag instead of that of the United States.

At this gloomy period of our history, startling events succeed each other rapidly. On the very day, the 27th instant, that possession of these two forts was taken, the Palmetto flag was raised over the Federal custom-house and post-office in Charleston: and on the same day every officer of the customs-Collector, Naval Officer, Surveyor and Appraiser-resigned their offices. And this, although it was well-known from the language of my message that as an ex-ecutive officer I felt myself bound to collect the revenue at the port of Charleston under the existing laws. In the harbor of Charleston we now find three forts confronting each other, over all of which the Federal flag floated four days ago; but now, over two of them this flag has been supplanted, and the Palmetto flag has been substituted in its stead.

It is under these circumstances that I am urged immediately to withdraw the troops from the harbor of Charleston, and I am informed that without this negotiation is impossible. This I cannot do; this I will not do. Such an idea was never thought of by me in any possible contingency. No such allusion had been made in any communication between myself and any human being.

But the inference is that I am bound to withdraw the troops from the only fort remaining in the possession of the United States in the harbor of Charleston, because the officer there in command of all the forts thought proper, without instructions, to change his position from one of them to another.

At this point of writing, I have received information by telegraph from Capt. Humphreys, in command of the arsenal at Charleston, that "it has to-day (Sunday, the 30th) been taken by force of arms." It is estimated that the munitions of war belonging to the United States in this arsenal are worth half a million of dollars.

Comment is needless. After this information, I have only to add that, whilst it is my duty to defend Fort Sumter as a portion of the public property of the United States against hostile attacks, from whatever quarter they may come, by such means as I may possess for this purpose, I do not perceive how such a defence can be construed into a menace against the city of Washington.

With great personal regard, I remain yours, very respectfully,

JAMES BUCHANAN. To Honorable Robert W. Barnwell, James H. Adams, James L. Orr.

January 1st, 1861. The "Commissioners" replied at length, alleging, with reference to the President's declaration, that he could not withdraw the troops from Charleston harbor, that he had in conversation left a different impression upon their minds and the minds of others who had approached him on that subject, and generally reflecting upon the motives of the President. This paper, Mr. BUCHANAN "declined to receive." In the State Convention of South Carolina, December 19th, 1860, upon a proposition of Mr. MAGRATH to appoint a committee to consider the relations of the State to the forts, Mr. W. PORCHER MILES alluded to the interview between the President and the South Carolina representatives in Congress relative to the forts, and "expressed his solemn opinion that the President was not going to attempt to reinforce those forts." Subsequently Mr. MILES and Mr. KEITT made to the Convention a written statement sustaining the offensive allegations of the Commissioners as to the President's good faith to them. These papers are too long for insertion, and have but limited importance.

# Further demand of South Carolina for Fort Sumter. January 11th, 1861. F. W. Pickens, Gover-

nor of South Carolina, demanded of Major Anderson the surrender of Fort Sumter to the authorities of the State of South Carolina, to prevent a "waste of life." Same day Major Anderson replied, announcing his refusal, but suggesting that if at any time prior to a resort to arms, the Governor should deem fit to "refer this matter to Washington," he could with much pleasure depute one of his officers to accompany the Governor's messenger. Same day, Governor Pickens deputed Hon. Isaac W. Hayne, Attorney-General of the State, to proceed to Washington, and demand from the President the delivery of Fort Sumter to the constituted authorities of South Carolina, adding:

"The demand I have made of Major Anderson, and which I now make of you, is suggested because of my earnest desire to avoid the bloodshed which a persistence in

your attempt to retain possession of that fort will cause, and which will be unavailing to secure you that possession, but induce a calamity most deeply to be deplored. Tf consequences so unhappy shall ensue, I will secure for this State, in the demand which I now make, the satisfaction of having exhausted every attempt to avoid it.

"In relation to the public property of the United States within Fort Sumter the Hon. I. W. HAYNE, who will hand you this communication, is authorized to give you the pledge of the State that the valuation of such property will be accounted for by this State, upon the adjustment of its relations with the United States, of which it was a part. "F. W. PICKENS.

"To the President of the United States." Upon Colonel Hayne's arrival, ten U.S. Senators "from States which have already seceded from the United States, or will have done so before the first of February next,"\* requested that he should not present his demand until these States should have formed a Confederacy. Meanwhile, they offered to propose to the President, that Fort Sumter should not be reinforced in the meantime. To this Colonel Hayne consented, and the Senators proposed this arrangement, which the President declined through Hon. JOSEPH HOLT, Secretary of War, as follows:

War Department, January 22d, 1861. To the Honorable Benjamin Fitzpatrick,

S. R. Mallory and John Slidell. GENTLEMEN: The President has received your communication of the 19th instant, with the copy of a correspondence between yourselves and others, "representing States which have already seceded from the United States, or will have done so before the first of February next," and Colonel Isaac W. Hayne, of South Carolina, in behalf of the Government of that State, in relation to Fort Sumter; and you ask the President "to take into consideration the subject of that correspondence." With this request he has complied, and has directed me to communicate his answer.

In your letter to Col. Hayne of the 15th instant, you propose to him to defer the de-livery of a message from the Governor of South Carolina to the President, with which he has been entrusted for a few days, or until the President and Col. Hayne shall have considered the suggestions which you submit. It is unnecessary to refer specially these suggestions, because the letter to addressed to you by Col. Hayne, of the 17th instant, presents a clear and specific answer to them. In this he says: "I am not clothed with power to make the arrangement you suggest; but provided you can

* Being :		
Louis T. Wigfall, John Hemphill,	Texas.	C. C. Clay, Jr., Benjamin Fitzpatrick }Ala.
	Florida	A. Iverson, Georgia.
S. R. Mallory, Jefferson Davis, Mi	ceieeinni	A. Iverson, Georgia. John Slidell, J. P. Benjamin La.
Jenerson Davis, mi	ssissippi,	J. F. Denjamin,

get assurances, with which you are entirely tion. Should his safety, however, require satisfied, that no reinforcements will be sent to Fort Sumter in the interval, and that the public peace will not be disturbed by any act of hostility towards South Carolina, I will refer your communication to the autho-rities of South Carolina, and, withholding the communication with which I am at pre-

sent charged, will await further instructions." From the beginning of the present un-happy troubles, the President has endeav-ored to perform his Executive duties in such a manner as to preserve the peace of the country and to prevent bloodshed. This is still his fixed purpose. You, therefore, do him no more than justice in stating that you have assurances (from his public mes-sages, I presume) that "notwithstanding the circumstances under which Major Anderson left Fort Moultrie and entered Fort Sumter with the forces under his command, it was not taken and is not held with any hostile or unfriendly purpose towards your State, but merely as property of the United States, which the President deems it his duty to protect and preserve," you have correctly stated what the President deems to be his duty. His whole object now is and has been, to act strictly on the defensive, and to authorize no movement against the people of South Carolina, unless clearly justified by a hostile movement on their part. He could not have given a better proof of his desire to prevent the effusion of blood than by forbearing to resort to the use of force under the strong provocation of an attack (happily without a fatal result) on an un-armed vessel bearing the flag of the United States.

I am happy to observe that, in your letter to Col. Hayne, you express the opinion that it is "especially due from South Carolina to our States, to say nothing of other slaveholding States, that she should, as far as she can consistently with her honor, avoid initiating hostilities between her and the United States, or any other Power." To initiate such hositilities against Fort Sumter would, beyond question, be an act of war against the United States.

In regard to the proposition of Col. Hayne, "that no reinforcements will be sent to Fort Sumter in the interval, and that the public peace will not be disturbed by any act of hostility towards South Carolina," it is impossible for me to give you any such assurances. The President has no authority to enter into such an agreement or understanding. As an executive officer, he is simply bound to protect the public property, so far as this may be practicable; and it would be a manifest violation of his duty to place himself under engagements that he would not perform this duty either for an indefinite or a limited period. At the present moment, it is not deemed necessary to reinforce Major Anderson, because he makes no such request, and feels quite secure in his posi- have so severely tried it, be not accepted as

reinforcements, every effort will be made to supply them.

In regard to an assurance from the President "that the public peace will not be disturbed by any act of hostility towards South Carolina," the answer will readily occur to yourselves. To Congress, and to Congress alone, belongs the power to make war, and it would be an act of usurpation for the Executive to give any assurance that Congress would not exercise this power, however strongly it may be convinced that no such intention exists.

I am glad to be assured from the letter of Col. Hayne, that "Major Anderson and his command do now obtain all necessary supplies, including fresh meat and vegetables, and, I believe, fuel and water, from the city of Charleston, and do now enjoy communication, by post and special messenger, with the President, and will continue to do so, certainly until the door to negotiation has been closed." I trust that these facilities may still be afforded to Major Anderson. This is as it should be. Major Anderson is not menacing Charleston; and I am convinced that the happiest result which can be attained is, that he and the authorities of South Carolina shall remain on their present amicable footing, neither party being bound by any obligation whatever except the high Christian and moral duty to keep the peace and to avoid all causes of mutual irritation.

Very respectfully,

Your obedient servant,

J. HOLT,

Secretary of War. January 31st, 1861. Col. Hayne, having received additional instructions from Governor Pickens, reciting the correspondence between the President and the ten Senators, and expressing his dissatisfaction with the terms of the latter's reply, demanded possession of Fort Sumter "as the legal officer of the State, asserting its undoubted right of eminent domain."

The President replied, February 6th. through Secretary Holt, asserting the title of the United States to Fort Sumter as complete and incontrovertible, and declining  $_{\rm the}$ demand, as, "whatever may be the claim of South Carolina to this fort, he has no con-stitutional power to cede or surrender it." The closing paragraph of the President's reply is as follows :

" If the announcement so repeatedly made of the President's pacific purpose in continuing the occupation of Fort Sumter until the question shall be settled by competent authority, has failed to impress the Government of South Carolina, the forbearing conduct of the Administration for the last few months should be received as conclusive evidence of his sincerity. And if this forbearance, in view of the circumstances which

a satisfactory pledge of the peaceful policy of this Administration towards South Carolina, then it may be safely affirmed that neither language nor conduct can possibly furnish one. If, with all the multiplied proofs which exist of the President's anxiety for peace, and of the earnestness with which he has pursued it, the authorities of that State shall assault Fort Sumter, and peril the lives of the handful of brave and loyal men shut up within its walls, and thus plunge our common country into the horrors of civil war, then upon them and those they represent, must rest the responsibility."

### "Commissioner from Alabama."

February 1st, 1861. Hon. C. C. CLAX, Jr., Senator from Alabama, addressed the President, informing him that Hon. THOMAS J. JUDGE of Alabama had arrived, "duly commissioned to negotiate with the Government of the United States in reference to the forts, arsenals and custom-houses in that State, and the debt of the United States," and desiring when he might have an audience "to present his credentials and enter upon the proposed negotiations."

2d. The PRESIDENT replied, stating that he would be happy to receive Mr. JUDGE as a "distinguished citizen of Alabama," and that, in his judgment, he had no power to recognize him in the character ascribed to him. Mr. CLAY, under date of February 5th, volunteered to give the President his views of men and things, in an excited epistle, which scarcely deserves preservation, either for intrinsic merit of style or thought for historical value.

# Transfer of U. S. Arms South in 1859-60.

REPORT (Abstract of) made by Mr. B. Stanton, from the Committee on Military Affairs, in House of Representatives, Feb. 18th, 1861:

The Committee on Military Affairs, to whom was referred the resolution of the House of Representatives of 31st of December last, instructing said committee to inquire and report to the House, how, to whom, and at what price, the public arms distributed since the 1st day of January, A. D. 1860, have been disposed of; and also into the condition of the forts, arsenals, dockyards, etc., etc., submit the following report:

That it appears from the papers herewith submitted, that Mr. Floyd, the late Secretary of War, by the authority or under color of the law of March 3d, 1825, authorizing the Secretary of War to sell any arms, ammunition, or other military stores which should be found unsuitable for the public service, sold to sundry persons and States 31,610 flint-lock muskets, altered to percussion, at \$2.50 each, between the 1st day of January, A. D. 1860, and the 1st day of January, A. D. 1861. It will be seen from the testimony of Colonel Craig and Captain Maynadier, that

they differ as to whether the arms so sold had been found, "upon proper inspection, to be unsuitable for the public service." Whilst the Committee do not deem it im-

Whilst the Committee do not deem it important to decide this question, they say, that in their judgment it would require a very liberal construction of the law to bring these sales within its provisions.

It also appears that on the 21st day of November last, Mr. Belknap made application to the Secretary of War for the purchase of from one to two hundred and fifty thousand United States muskets, flint-locks and altered to percussion, at \$2.15 each; but the Secretary alleges that the acceptance was made under a misapprehension of the price bid, he supposing it was \$2.50 each, instead of \$2.15.

Mr. Belknap denies all knowledge of any mistake or misapprehension, and insists upon the performance of his contract.

The present Secretary refuses to recognize the contract, and the muskets have not been delivered to Mr. Belknap.

Mr. Belknap testifies that the muskets were intended for the Sardinian government.

It will appear by the papers herewith submitted, that on the 29th of December, 1859, the Secretary of War ordered the transfer of 65,000 percussion muskets, 40,000 muskets altered to percussion, and 10,000 percussion rifles, from the Springfield Armory and the Watertown and Watervliet Arsenals, to the Arsenals at Fayetteville, N. C., Charleston, S. C., Augusta, Ga., Mount Vernon, Ala., and Baton Rouge, La., and that these arms were distributed during the spring of 1860 as follows:

To Charleston Arsenal, To North Carolina Arsenal To Augusta Arsenal, To Mount Vernon Arsenal	12,380 9,280	Altered muskets. 5,720. 9,520 7,620 5,720	2,000 2,000 2,000 2,000 2,000
To Baton Rouge Arsenal,	18,580	11,420	2,000

65,000 40,000 10,000

All of these arms, except those sent to the North Carolina Arsenal,\* have been seized by the authorities of the several States of South Carolina, Alabama, Louisiana and Georgia, and are no longer in possession of the United States.

It will appear by the testimony herewith presented, that on the 20th of October last the Secretary of War ordered forty columbiads and four thirty-two pounders to be sent from the Arsenal at Pittsburg to the fort on Ship Island, on the coast of Mississippi, then in an unfinished condition, and seventy columbiads and seven thirty-two pounders to be sent from the same Arsenal to the fort at Galveston, in Texas, the building of which had scarcely been commenced.

This order was given by the Secretary of War, without any report from the Engineer department showing that said works were

\* These were afterwards seized

ready for their armament, or that the guns were needed at either of said points.

It will be seen by the testimony of Captain Wright, of the Engineer department, that the fort at Galveston cannot be ready for its entire armament in less than about five years, nor for any part of it in less than two; and that the fort at Ship Island will require an appropriation of \$85,000 and one year's time before it can be ready for any part of its armament. This last named fort has been taken possession of by the State authorities of Mississippi.

The order of the late Secretary of War (Floyd) was countermanded by the present Secretary (Holt) before it had been fully executed by the shipment of said guns from Pittsburg.\*

It will be seen by a communication from the Ordnance office of the 21st of January last, that by the last returns there were remaining in the United States arsenals and armories the following small arms, viz:

Percussion muskets and muskets

altered to percussion of calibre 69, 499,554 Percussion rifles, calibre 54..... 42,011

Total . . . . 541,565

Of these 60,878 were deposited in the arsenals of South Carolina, Alabama, and Louisiana, and are in the possession of the authorities of those States, reducing the number in possession of the United States to 480,687.

Since the date of said communication, the following additional forts and military posts have been taken possession of by parties acting under the authority of the States in which they are respectively situated, viz:

Fort Moultrie, South Carolina.

Fort Morgan, Alabama.

Baton Rouge Barracks, Louisiana.

Fort Jackson, Louisiana.

Fort St. Philip,

Fort Pike, Louisiana.

Oglethorpe Barracks, Georgia.

. And the department has been unofficially advised that the arsenal at Chattahoochee, Forts McRea and Barrancas, and Barrancas Barracks, have been seized by the authorities of Florida.

To what further extent the small arms in possession of the United States may have been reduced by these figures, your committee have not been advised.

The whole number of the sea-board forts in the United States is fifty-seven; their appropriate garrison in war would require 26,420 men; their actual garrison at this

\* The attempted removal of these heavy guns from Allegheny Arsenal, late in December, 1860, created intense excitement. A monster mass meeting assembled at the call of the Mayor of the city, and citizens of all parties aided in the effort to prevent the shipment. Through the interposition of Hon. J. K. Moorhead, Hon. R. McKnight, Judge Shaler, Judge Wilkins, Judge Shannon, and others, inquiry was instituted, and a revocation of the order obtained. The Secessionists in Congress bitterly complained of the "mob law" which thus interfered with the routine of governmental affairs. time is 1,334 men, 1,308 of whom are in the forts at Governor's Island, New York; Fort McHenry, Maryland; Fort Monroe, Virginia, and at Alcatraz Island, California, in the harbor of San Francisco.

From the facts elicited, it is certain that the regular military force of the United States is wholly inadequate to the protecticn of the forts, arsenals, dock-yards, and other property of the United States in the present disturbed condition of the country. The regular army numbers only 18,000 men when recruited to its maximum strength, and the whole of this force is required for the protection of the border settlements against Indian depredations. Unless it is the intention of Congress that the forts, arsenals, dock-yards, and other public property, shall be exposed to capture and spoliation, the President must be armed with additional force for their protection. In the opinion of the Committee the law

In the opinion of the Committee the law of February 28th, 1795, confers upon the President ample power to call out the militia to execute the laws and protect the public property. But as the late Attorney-General has given a different opinion, the Committee, to remove all doubt upon the subject, report the accompanying bill, etc.

#### OTHER ITEMS.

Statement of Arms distributed by Sale since the first of January, 1860, to whom sold, and the place whence sold.

	1000.	Arsenais
To whom sold. No.	Date of sale	. Where sold.
J. W. Zacharie & Co4,000	Feb. 3	St. Louis,
James T. Ames1,000	Mar. 14	New York.
Captain G. Barry 80	June 11	St. Louis.
W. C. N. Swift 400	Aug. 31	Springfield.
Do 80	Nov. 13	Do.
State of Alabama1,000	Sep. 27	Baton Rouge.
Do2,500	Nov. 14	Do.
State of Virginia5,000	Nov. 6	Washington.
Phillips county, Ark 50	Nov. 16	St. Louis.
G. B. Lamar10,000	Nov. 24	Watervliet.

The arms were all fint-lock muskets altered to percussion, and were all sold at \$2.50 each, except those purchased by Captain G. Barry and by the Phillips county volunteers, for which \$2 each were paid.

The Mobile Advertiser says: "During the past year 135,430 muskets have been quietly transferred from the Northern arsenal at Springfield alone, to those in the Southern States. We are much obliged to Secretary Floyd for the foresight he has thus displayed in disarming the North and equipping the South for this emergency. There is no telling the quantity of arms and munitions which were sent South from other Northern arsenals. There is no doubt but that every man in the South who can carry a gun can now be supplied from private or public sources. The Springfield contribution alone would arm all the militiamen of Alabama and Mississippi."

General Scott, in his letter of December 2d, 1862, on the early history of the Rebellion, states that "Rhode Island, Delaware and Texas had not drawn, at the end of 1860, their annual quotas of arms for that year, and Massachusetts, Tennessee, and Kentucky only in part; Virginia, South Carolina, Georgia, Florida, Alabama, Louisiana, Mississippi and Kansas were. by order of the Secretary of War, supplied with their quotas for 1861 in advance, and Pennsylvania and Maryland in part."

This advance of arms to eight Southern States is in addition to the transfer, about the same time, of 115,000 muskets to Southern arsenals, as per Mr. Stanton's report.

Governor Letcher of Virginia, in his Message of December, 1861, says, that for some time prior to secession, he had been engaged etc.; purchasing arms, ammunition, in among which were 13 Parrott rifled cannon, and 5,000 muskets. He desired to buy from United States  $\operatorname{Government}$ 10,000the more, when buying the 5,000, but he says "the authorities declined to sell them to us, although five times the number were then in the arsenal at Washington." Had Jefferson Davis' bill relative to the purchase of arms become a law, the result might have been different.

### Sale of Arms to States.

January 9th, 1860. Mr. Jefferson Davis Mississippi introduced to the Senate a of bill "to authorize the sale of public arms to the several States and Territories, and to regulate the appointment of Superintendents of the National Armories."

18th. He reported it from the Military Committee without amendment.

February 21st. Mr. Davis. I should like the Senate to take up a little bill which I hope will excite no discussion. It is the bill to authorize the States to purchase arms from the National Armories. There are a number of volunteer companies wanting to purchase arms, but the States have not a sufficient supply. I move to take up the bill.

The motion was agreed to.

and he is hereby authorized to issue to any State or Territory of the United States, on application of the Governor thereof, arms made at the United States Armories, to such extent as may be spared from the public supplies without injury or inconvenience to the service of the General Government, upon payment therefor, in each case, of an amount sufficient to replace, by fabrication at the national armories, the arms so issued.

Section 2. That so much of the act approved August 5th, eighteen hundred and fifty-four, as authorizes the appointment of a civilian as superintendent of each of the national armories be, and the same is hereby repealed, and that the superintendents of these armories shall hereafter be selected from officers of the ordnance corps.

After a brief discussion, it was made a special order for February 23d.

Fessenden calling for an explanation of the reasons upon which the first section is founded,

Mr. Davis said: "It is, that the volunteer companies of the States desiring arms, may facture. It is a long settled policy—and I think a wise one on the part of the United States-to furnish arms, of the approved pattern for the public service, to the militia. The appropriation which is made to supply the militia with arms, has not been found sufficient. There are constant applications for arms beyond the quota. The Secretary of War has no authority to issue them beyond the fixed allowance to each State, being its pro rata share of the arms which may be made with \$200,000. The Secretary of War, under that pressure, has this year recommended that the appropriation for the arming of the militia should be increased. In the meantime, there are volunteer companies with State appropriations anxious to obtain arms if they will be furnished. If the Congress thinks proper to exclude them from the purchase of arms from the armories, then they must go to private establishments, and get patterns which are not those established by the Government, arms which I believe to be inferior; and arms which, if they were brought into the service of the United States, in the event of the country being involved in war, would not receive the ammunition which the Government supplies. If they are to buy arms at all, it is therefore advantageous that they should buy the Government pattern."-Congressional Globe, 1st Session, 36th Congress, Part 1, p. 862.

March 1st. Its consideration was re sumed.

On motion of Mr. Trumbull of Illinois, ar amendment was inserted in the first section. requiring the payment "in cash, at the time of delivery." It was debated further, without a vote.

5th. Mr. Fessenden moved to add the following to the first section :

Provided, That the whole number of arms which may be sold, as aforesaid, shall be ascertained and determined in each year by the Secretary of War, and no State or Ter-ritory shall be allowed to purchase a numher of arms bearing a greater proportion to the whole number so ascertained and determined, than the Federal population of such State or Territory bears to the aggregate Federal population of all the States and Territories of the Union, according to the census of the United States next preceding such purchase.

16th. Mr. Davis of Mississippi moved the following as a substitute for the above proviso :

"That the sales of each year shall not exceed the increased manufacture which may result from said sales; and that the whole number to be sold, if less than the requisitions made, shall be divided between the States 23d. Its consideration was resumed. Mr. applying to purchase, pro rata, as arms furnicked by the United States are now distributed."

Which was agreed to-yeas 28, nays 18, as follows :

YEAS.—Messrs. Benjamin, Bigler, Bragg, Bright, Brown, Clay, Clingman, Crittenden, Davis, Fitch, Fitzpatrick, Breen, Gwin, Hammond, Hunter, Iverson, Johnson of Arkanasa, Johnson of Tennessee, Kennedy, Latham, Mal-lory, Nicholson, Pearce, Powell, Sebastian, Thomson, Toombs, and Wigfall—28. NATS.—Messrs. Anthony, Bingham, Chandler, Clark, Collamer, Dixon, Doolittle, Fessenden, Foot, Foster, Grimes, Hamlin, Harlan, Ten Eyck, Trumbull, Wade, Wilkinson, and Wilson—18.

A party vote, Democrats in affirmative-Republicans in negative.

The amendment as amended was then adopted.

A motion to strike out the first section

was lost—yeas 20, nays 28. March 26th. The bill passed—yeas 29, nays 18.

NATS.—Anthony, Bingham, Cameron, Chandler, Clark, Collamer, Doolittle, Durkee, Fessenden, Foot, Grimes, Hamlin, Harlan, King, Simmons, Sumner, Ten Eyck, and Wade.

During the debate, on March 28th, between Mr. Simmons of Rhode Island, and Mr. Davis of Mississippi, the latter made these remarks :

"The Senator runs into an error which I find very often prevails, that the militia of the States are not a part of the Army of the United States. It is our glory that the defence of the country rests upon the people. He proposes, then, to arm the militia in time of peace with a weapon which they will aot use in time of war."—Congressional Globe, 1st Session, 36th Congress, Part 2, p. 1351

In the House, the bill was referred to the Committee on Military Affairs, and was not reported.

### How the Telegraph was made to aid in effecting Secession.

Senator Toomes has publicly declared in Georgia that he would, under no circumstances, serve in the Senate after the in-auguration of Mr. Lincoln. He said the same thing in the following telegraphic despatch to Mr. KEITT :

"Macon, November 14th, 1860.

"To Hon, L. M. KEITT: I will sustain South Carolina in secession. I have announced to the Legislature that I will not serve under Lincoln. If you have the power to act, act at once. We have bright prospects here.

"R. Toombs."

#### SOUTHERN MANIFESTO.

Washington, December 13th. At the request of Hon. REUBEN DAVIS of Mississippi, member of the Committee of States, the Southern members of Congress assembled at his rooms to-night and adjourned at eleven o'clock, at which the following declaration was made and signed by those present. It had already been presented to the Committee of Thirty-three:

Washington, December 13th, 1860. To our Constituents: The argument is exhausted. All hope of relief in the Union, through the agency of committees, Congressional legislation, or constitutional amendments, is extinguished, and we trust the South will not be deceived by appearances The Reor the pretence of new guarantees. publicans are resolute in the purpose to grant nothing that will or ought to satisfy the South. We are satisfied the honor, safety, and independence of the Southern people are to be found only in a Southern Confederacy -a result to be obtained only by separate State secession-and that the sole and primary aim of each slaveholding State ought to be its speedy and absolute separation from an unnatural and hostile Union.

Signed by J. L. Pugh, David Clopton, Sydenham Moore, J. L. M. Curry, and J. A. Sydennam Moore, J. L. M. Curry, and J. A. Stallworth of Alabama; Alfred Iverson, J. W. H. Underwood, L. J. Gartrell, and Jas. Jackson, (Senator Toombs is not here, but would sign), John J. Jones, and Martin J. Crawford of Georgia; Geo. S. Hawkins of Florida. It is understood Mr. Yulee will signit. T. C. Hindman of Arkansas. Both Senators will also sign it. A. G. Brown Senators will also sign it. A. G. Brown, Wm. Barksdale, O. R. Singleton, and Reuben Davis of Mississippi; Burton Craige and Thos. Ruffin of North Carolina; J. P. Ben-jamin and John M. Landrum of Louisiana. Mr. Slidell will also sign it. Senators Wigfall and Hemphill of Texas, will sign it.

Mr. Davis made the following statement to the caucus:

Being a member of the Committee of Thirty-three, I state that the above witnessed despatch was communicated to the committee this evening, and a resolution passed proposing no specific relief, eight Northern States dissenting, avowedly intended to counteract the effect of the above despatch, and, as I believe, to mislead the people of the South. From information derived from Republican members of the committee and other Northern Representatives, I fully concur in the above despatch.

#### REUBEN DAVIS.

The manifesto will be immediately communicated to the several constituencies of the gentlemen named by telegraph.

A TELEGRAPHIC MANIFESTO FROM SENATOR TOOMBS.

The Savannah News of Monday, December 24th, publishes the following address to the people of Georgia, telegraphed from Washington, on Saturday, December 22d:

Fellow-Citizens of Georgia: I came here to secure your constitutional rights, or to demonstrate to you that you can get no guarantees for these rights from your Northern Confederates.

The whole subject was referred to a committee of thirteen in the Senate yesterday. I was appointed on the committee and accepted the trust. I submitted propositions, which, so far from receiving decided support

from a single member of the Republicant party on the committee, were all treated with either derision or contempt. The vote was then taken in committee on the amendments to the Constitution proposed by Hon. J. J. CRITTENDEN of Kentucky, and each and all of them were voted against, unanimously, by the Black Republican members of the committee.

In addition to these facts, a majority of the Black Republican members of the committee declared distinctly that they had no guarantees to offer, which was silently acquiesced in by the other members.

The Black Republican members of this Committee of Thirteen are representative men of their party and section, and to the extent of my information, truly represent the Committee of Thirty-three in the House, which on Tuesday adjourned for a week without coming to any vote, after solemnly pledging themselves to vote on all propositions then before them on that date.

That committee is controlled by Black Republicans, your enemies, who only seek to amuse you with delusive hope until your election, in order that you may defeat the friends of secession. If you are deceived by them, it shall not be my fault. I have put the test fairly and frankly. It is decisive against you; and now I tell you upon the faith of a true man that all further looking to the North for security for your constitutional rights in the Union ought to be instantly abandoned. It is fraught with nothing but ruin to yourselves and your posterity.

Secession by the fourth of March next should be thundered from the ballot-box by the unanimous voice of Georgia on the second day of January next. Such a voice will be your best guarantee for LIBERTY, SECURITY, TRANQUILLITY and GLORY.

### ROBERT TOOMBS.

#### IMPORTANT TELEGRAPHIC CORRESPONDENCE.

Atlanta, Georgia, December 26th, 1860. Hon. S. A. Douglas or Hon. J. J. Crittenden:

Mr. Toombs's despatch of the 22d inst. unsettled conservatives here. Is there any hope for Southern rights in the Union? We are for the Union of our fathers, if Southern rights can be preserved in it. If not, we are for secession. Can we yet hope the Union will be preserved on this principle? You Give us are looked to in this emergency. your views by despatch, and oblige

WILLIAM EZZARD. ROBERT W. SIMS. JAMES P. HAMBLETON. THOMAS S. POWELL. S. G. Howell. J. A. HAYDEN. G. W. ADAIR. R. C. HONLESTER.

Washington, December 29th, 1860. In reply to your inquiry, we have hopes

that the rights of the South, and of every State and section, may be protected within the Union. Don't give up the ship. Don't despair of the Republic.

J. J. CRITTENDEN.

S. A. DOUGLAS.

### From the Raleigh Standard Extra of January 2d.

The State Journal of to-day, one of the organs of the disunionists, contains a telegraphic despatch calculated, and no doubt intended, to inflame the public mind and to precipitate North Carolina into revolution. This despatch, most probably sent here from the *Journal* Office, Wilmington, is as follows:

### IMPORTANT !--- IMMEDIATE RETURN OF LEGIS-LATORS TO THEIR POSTS.

#### Wilmington, Dec. 31st, 81 P.M.

The following is the substance of a despatch received here this evening:

"Cabinet broken up in a row; Floyd, Thompson and Thomas have resigned; the President has gone over to the North. Federal troops on their way South. Our fort at the mouth of Cape Fear will shortly be occupied by troops for coercion. The citizens of North Carolina call upon the Legislature for advice and assistance.'

The above produced great excitement in our community. As soon as we saw it we telegraphed to a well-informed and reliable friend in Washington city, whose reply is as follows:

"No troops ordered South. No new ground for excitement known."

#### Special Despatch to the Republican.

### Augusta, Ga., Jan. 1st.

A special despatch to the True Democrat, of this city, dated at Washington, 3 o'clock, P.M., to-day, says :

"The cabinet is broken up, Mr. Floyd, Secretary of War, and Mr. Thompson, Secretary of the Interior, having resigned. A coercive policy has been adopted by the Administration. Mr. Holt, of Kentucky, our bitter foe, has been made Secretary of War. Fort Pulaski is in danger. The Abolitionists are defiant."

This despatch is signed "Robert Toombs."

This spurious and inaccurate despatch had a great influence, it is said, in deciding the wavering vote of Georgia on the question of union or disunion.

The Macon (Georgia) *Telegraph* of the 2d instant, contains the following sensation despatch from the President of the South Carolina Convention. Of course it obtained immediate currency throughout the Southern States :

Charleston, January 1st, 1861. Mayor of Macon: The Convention of South Carolina have directed me to send you the following telegram just received from our Commissioners at Washington : "Holt has been appointed Secretary of War. He is for coercion, and war is inevitable. We believe reinforcements are on the way. Prevent their entrance into the harbor at every hazard. My DEAR SIR

"D. F. JAMISON, "President South Carolina Convention."

### From the Neu Orleans Delta.

THE SOUTH CAROLINA CONVENTION TO THE CITY OF NEW ORLEANS.

The following highly-important despatch from the President of the South Carolina Convention, has been furnished to us for publication by Mayor Monroe, to whom it was addressed :

### Charleston, January 1st.

To the Hon. John T. Monroe, Mayor of New Orleans: The Convention of South Carolina has directed me to send you the following telegram, just received from our Commissioners at Washington:

"Holt has been appointed Secretary of War. He is for coercion, and war, we believe, is inevitable. We believe reinforcements are on the way. We shall prevent their entrance into the harbor at every hazard. "D. F. JAMISON,

"President South Carolina Convention."

### From the National Intelligencer.

In January, when the Crittenden plan of adjustment was voted down in the Senate, rather because of the absence of Southern Senators than by the strength of its opponents, we find from the St. Louis journals, that a despatch was reported to have been straightway sent from Washington to that city by Senators Polk and Green, representing as follows:

"The Crittenden resolutions were lost by a vote of 25 to 23. A motion of Mr. Cameron to reconsider was lost; and thus ends all hope of reconciliation. Civil war is now considered inevitable, and late accounts declare that Fort Sumter will be attacked without delay. The Missouri delegation recommend immediate secession."

We need not say that no such despatch was ever sent by these gentlemen. Yet, says the St. Louis *Republican*, "all over the city (St. Louis) it was spoken of as the despatch from Messrs. Green and Polk."

The temporary rejection of the Crittenden plan was in like manner pressed into the service of the Secessionists in order to accelerate the pace of grave, deliberate, and patriotic North Carolina. The Raleigh *Register* of the 19th instant contains the following despatch, under the signature of Mr. Crittenden himself, published to counteract the disturbing effect of the exaggerated rumors which had been put in circulation from this city:

## " Washington, January 17th, 9 P. M.

"In reply, the vote against my resolutions will be reconsidered. Their failure was the result of the refusal of six Southern words, coercion and war.

Senators to vote. There is yet good hope of success." "JOHN J. CRITTENDEN."

Senate, January 25th, 1861.

My DEAR SIR: Mr. Crittenden is not present, but I can say with confidence that there is hope of adjustment, and the prospect has never been better than now since we first assembled.

## Very truly, your friend,

S. A. DOUGLAS.

We concur in the opinion that there is hope of an adjustment.

> J. J. CRITTENDEN, A. R. BOTELER, JOHN T. HARRIS.

### Hon. JAMES BARBOUR.

In addition to the foregoing testimony on the subject, we insert an extract of a letter from the Hon. John S. Millson to Mr. Barbour to the same effect :

"For myself, I say that I have never had so confident an expectation as I have at this time, of such a termination of the present controversy as would be satisfactory to me, and, I believe, to a large majority of the people of Virginia."

TO THE PEOPLE OF VIRGINIA.

We deem it our duty, as your Representatives at Washington, to lay before you such information as we possess in regard to the probable action of Congress in the prosent alarming condition of the country.

At the beginning of this session, now more than half over, committees were appointed, in both Houses of Congress, to consider the state of the Union. Neither committee has been able to agree upon any mode of settlement of the pending issues between the North and the South.

The Republican members in both committees rejected propositions acknowledging the right of property in slaves; or recommending the division of the Territories between the slaveholding and non-slaveholding States by a geographical line.

In the Senate, the propositions commonly known as Mr. Crittenden's were voted against by every Republican Senator; and the House, on a vote by ayes and noes, refused to consider certain propositions, moved by Mr. Etheridge, which were even less favorable to the South than Mr. Crittenden's.

A resolution giving a pledge to sustain the President in the use of force against the seceding States, was adopted in the House of Representatives by a large majority; and in the Senate every Republican voted to substitute for Mr. Crittenden's propositions resolutions offered by Mr. Clark of New Hampshire, declaring no new concessions, guarantees, or amendments to the Constitution were necessary; that the demands of the South were unreasonable, and that the remedy for the present danger was simply to enforce the laws; in other words, coercion and war. In this state of facts, our duty is to warn you that it is vain to hope for any measures of conciliation or adjustment (from Congress) which you could accept. We are also satisfied that the Republican party designs, by civil war alone, to coerce the Southern States, under the pretext of enforcing the laws, unless it shall become speedily apparent that the seceding States are so numerous, determined and united, as to make such an attempt hopeless.

We are confirmed in these conclusions by our general intercourse here; by the speeches of the Republican leaders here and elsewhere; by the recent refusals of the Legislatures of Vernont, Ohio and Pennsylvania, to repeal their obnoxious Personal Liberty Laws; by the action of the Illinois Legislature on resolutions approving the Crittenden propositions, and by the adoption of the resolutions in the New York and Massachusetts Legislatures (doubtless to be followed by others) offering men and money for the war of coercion.

We have thus placed before you the facts and conclusions which have become manifest to us from this post of observation where you have placed us. There is nothing to be hoped from Congress—the remedy is with you alone, when you assemble in sovereign Convention.

We conclude by expressing our solemn conviction that prompt and decided action, by the people of Virginia in Convention, will afford the surest means under the Providence of God, of averting an impending civil war, and preserving the hope of reconstructing a Union already dissolved.

> J. M. MASON, R. M. T. HUNTER, D. C. DE JARNETTE, M. R. H. GARNETT, SHELTON F. LEAKE, E. S. MARTIN, H. A. EDMUNDSON, ROGER A. PRYOR, THOS. S. BOCOCK, A. G. JENKINS.

### Washington City, January 26th, 1861.

[Owing to the detention of Ex-Governor Smith, at his home in Virginia, by sickness, this address could not be presented to him for his signature. There is no doubt he would have joined in it, if present.]

### STIRRING THE FIRES.

The Richmond *Examiner* of Friday last contains the following despatch, intended to "operate" on the election to be held in Virginia to-day :

"The following despatch fully explains itself. The voters of Virginia cannot now fail to perform their duty:

"' Charleston, January 30th, 1861.

"'To Judge Hopkins, Richmond, Virhighly ginia: Reinforcements have been ordered dated:

to Fort Sumter and elsewnere. Will not Virginia, by her Legislature, interpose to prevent coercion? It will be too late when her Convention meets.

## "J. S. Preston.'"

The Richmond Whig states that a similar despatch was received by another distinguished member of the Legislature, "to which, after consultation with many leading members of the Legislature, a reply was made to the effect that we, here, had heard of no attempt at coercion, but that the President was exerting himself to preserve peace."

### IN FEBRUARY.

From the Nashville Union of February 6th. VIRGINIA DESPATCHES.—Our special despatch with regard to the Virginia election is direct from Richmond, and is from the Editors of the Richmond *Enquirer*. Of course it is more reliable than the despatch sent by the Associated Press of the same date.

Special Despatch to Union and American.

Richmond, (Va.) Feb. 5th.—Resistance men have carried the Convention overwhelmingly. Submission Unionists, not twenty elected ENQUIRER.

### From another column of the same paper. THE OLD DOMINION !—ALL HAIL !

A voice, as from the grave of the immortal Washington, tells us that Virginia will be true to her ancient, ever glorious, historical renown. Throughout the length of her immense territories only twenty Submissionist Union men have been elected. Virginia will before the 4th of March declare herself absolved from all further obligation to a Government, etc. It is eminently proper that the State which was the leader in the Revolution, and the first to proclaim the great doctrine of State-rights in 1799, should lead the column of the Border States.

# From the Nashville Union of February 7th.

IMPORTANT DESPATCHES !

Listen to the following glorious news from old Virginia:

Richmond, Feb. 6th.—To Wm. Williams : The Submissionists will not number thirty in the Convention. The Resistance men have more than one hundred elected. The action of the Convention will be prompt as soon as the Washington Conference adjourns. ENQUIRER.

Will Tennessee elect members to her Convention that wish to wait, wait, wait until Black Republicanism trample us in the dust or kick us out?

We are indebted (says the Charlotte Bulletin) to our much-esteemed Senator, the Hon. T. L. CLINGMAN, for the following highly important-information, by telegraph, dated: Washington, Feb. 18th, 1861. Editor Bulletin: There is no chance whatever for Crittenden's proposition. North Carolina must secede, or aid Lincoln in making war on the South.

## T. L. CLINGMAN.

### SMASHING OF THE PEACE CONFERENCE.

The following is part of a special telegram to the Charleston *Mercury*, dated at Washington, Feb. 21st:

ington, Feb. 21st: "The only hope now is in the smashing up of the Peace Congress and getting Virginia out."

FALSE DESPATCHES.— The Fayetteville (North Carolina) Observer, referring to false telegraphic despatches sent out from Charleston, says: "They are of a piece with the stories circulated in Georgia to affect the election in that State; that which preceded the mission of the Wilmington Committee to Raleigh; and that which resulted in the treasonable seizure of Fort Caswell, whither no troops have been or will be ordered. It does not answer the purposes of those engaged in disunion schemes to allow any period of rest from excitement. The 'Southern heart' must be 'fired,' that the Southmay be 'precipitated into secession.'"

### Other Secessions Proposed.

December 24th, 1860.

Mr. John C. Burch offered the following resolution, which was referred to the Committee on Military affairs:

That the Secretary of War be, and is hereby, directed and required to issue to the State of California quotas of arms for the years 1850 and 1851, equal to the quota issued to said State for the year 1852, of such description as the authorities of said State may require.

## From California.

Fort Kearney, Feb. 5th.—The Pony Express from California on the 19th has arrived. The Governor's message is strong for the Union. The letters from Congressmen Scott and Burch, advocating a Pacific Republic, are severely denounced by the papers.

### From California.

Fort Kearney.—The Pony Express, bringing California news as late as the 29th ultimo, has arrived.

Senator Latham's declaration that California will remain with the Union of the Northern and Western States is generally commended by the newspapers. They undoubtedly present a correct representation of a vast majority of the people of the State.

From the California correspondent of the New York *World*, written in the winter of 1860:

### CALIFORNIA FOR THE UNION.

On Friday last the Assembly passed a piled up the elements along these shores for series of most warm and decided Union a great empire, and if it needs he can make

resolutions, declaring the rebellion in the South to be treasonable, and recommending the use of such force as may be found necessary to vindicate the laws and preserve the Union. The final vote was, ayes 40, nays 22. The official synopsis of the resolutions is as follows:

Resolved, By the Assembly, the Senate concurring, First. The withdrawal of a State from its membership and obligations in the Federal Union, in defiance of the General Government, can only be accomplished by a successful resistance to the whole power of the United States. Second. Decent respect to the opinions of the people of the civilized world, and the instinct of self-preservation demands that the United States Government should use all the power necessary to enforce obedience to its laws and to protect its property. Third. The people of the State of California will sustain and uphold the constitutionally-elected officers of the United States Government, in all constutional efforts to preserve the in-tegrity of the Union, and to enforce obe-dience to the acts of Congress and the decisions of the courts. After the laws have been enforced, and the power and authority of the Constitution and Government of the United States recognized and acknowledged, every feeling of nationality and brotherhood demand that such compromises as are consistent with justice shall be made for the purpose of restoring that harmony which should characterize the people of a common country.

The majority vote was given by moderate Douglas Democrats and Republicans, the minority by Breckenridge men, who did not conceal, but on the contrary loudly boasted their approval of the secession of the South, and spoke with the utmost enthusiasm of the Southern Confederacy and President Davis. The conflict of feeling on this subject in California is more intense than elsewhere, because it is between Northern and Southern men who are contending for ascendancy in the State. Many of the latter, however, refuse to accept the disunion issue, and declare their intention to fight the battle of the South in the Union.

The result was received with most hearty applause, proving the devotion of our people to the Union, and their love of country. Considering the outrageously-treasonable speeches that have been made during the sessions, and the symptoms of approval elicited by them, chiefly from secessionists in the employ of the State, this evidence of patriotic feeling is especially gratifying. A PACIFIC REPUBLIC.—The Shasta Her-

A PACIFIC REPUBPLIC.—The Shasta Herald, of Saturday, in discussing political matters, says:

"If disunion does come, neither North nor South need look for aid and comfort from the Pacific coast. The Almighty has piled up the elements along these shores for a great empire, and if it needs he can make it one. We have no such interest that would demand an alliance with either of the belligerents upon the other side. We are, and have been for ten years, as fully separated from them as though we were but a foreign colony. This coast can stand alone, and if disunion between Norta and South ever comes, it will stand alone and independent. When that day comes—if unfortunately it should ever come—then 'Long live the Pacific Republic !'"

To which the San Francisco Bee responded in opposition:

We have frequently mentioned the existence of an organization in California, having for its object the establishment of a Pacific Republic on this coast, in accordance with the treasonable recommendations of Scott and Burch. We should have taken some pains to learn the names of the leaders of the movement, that they might be held up to the scorn and ridicule of a union-loving people, had we not been satisfied, as we now are, that the scheme would be frowned down the moment it was openly advocated. But two of the ninety journals in California have ventured to offer a word in favor of the movement, and they were actuated, we imagine, more by a spirit of bravado than a hope of success. Aside from a few disappointed office-seekers, there are not to be found to-day in California, outside of the State prison, a thousand voices to give encouragement to the treason. With less to thank the Union for than any other portion of the Confederacy, the people of the Pacific are prepared to make any sacrifice to maintain it—to suffer every neglect rather than sever their connection with it. Still a Pacific Republic is talked of, and a conspiracy, with the focus in San Francisco, is poisoning the land. One of its leading branches, we learn by the Appeal, is in Marysville. It is a secret organization, and is "forming," says the journal, "in a section of country lying back of the Buttes, with a view to establish a military company, pledged to the maintainance of a Pacific Republic. They will apply to the State for arms under the covert pretence of an organization un-der the State militia law. These misguided men have been deluded into the belief by their leaders that the Union is irretrievably gone to pieces, and the only hope for California is the immediate establishment of an independent government." The raising of the "bear flag," at Stockton, some days since, was probably a part of the programme. We are anxious to see the faces of some of these revolutionists-to note the mark with which treason brands its votaries.—San Francisco Mirror.

#### SOUTH AND NORTHWEST.

The Cincinnati Gazette, in the winter of 1860, states:

We are most reliably informed that there are agents of the Gulf States now in the

city endeavoring to create a sentiment among business men favorably to the establishment of a Confederacy composed of the Northern and Southwestern States. A well-known leader among the democracy has been approached, our information goes to show, within the past two or three days, by these agents, with a view of obtaining his influence, but he declined to have any thing to do with so traitorous a scheme, as he is a staunch Union man. The object is to make free-trade the basis of the confederacy -to cut off New York, Pennsylvania, New Jersey, and all the New England States, which are so wedded to a protective tariff. These Southern gentlemen state that there are agents of the Gulf States throughout all the northwestern States, who are making similar overtures, and that it is their aim to spring the issue soon among the citizens of those States.

MAYOR WOOD'S RECOMMENDATION OF THE SECESSION OF NEW YORK CITY, JANU-ARY 6TH, 1861.

### To the Honorable the Common Council:

GENTLEMEN : We are entering upon the public duties of the year under circumstances as unprecedented as they are gloomy and painful to contemplate. The great trading and producing interests of not only the city of New York, but of the entire country, are prostrated by a monetary crisis; and although similar calamities have before befallen us, it is the first time that they have emanated from causes having no other origin than that which may be traced to political disturbances. Truly, may it now be said, "We are in the midst of a revolution bloodless AS YET." Whether the dreadful alternative implied as probable in the conclusion of this prophetic quotation may be averted, "no human ken can di-vine." It is quite certain that the severity of the storm is unexampled in our history, and if the disintegration of the Federal Government, with the consequent destruction of all the material interests of the people shall not follow, it will be owing more to the interposition of Divine Providence, than to the inherent preventive power of our institutions, or the intervention of any other human agency.

It would seem that a dissolution of the Federal Union is inevitable. Having been formed originally on a basis of general and mutual protection, but separate local independence—each State reserving the entire, and absolute control of its own domestic affairs, it is evidently impossible to keep them together longer than they deem themselves fairly treated by each other, or longer than the interests, honor and fraternity of the people of the several States are satisfied. Being a Government created by *opinion*, its continuance is dependent upon the continuance of the sentiment which formed it. It cannot be preserved by coercion or held together by force. A resort to this last dreadful alternative would of itself destroy not only the Government, but the lives and property of the people.

If these forebodings shall be realized, and a separation of the States shall occur, momentons considerations will be presented to the corporate authorities of this city. We must provide for the new relations which will necessarily grow out of the new condition of public affairs.

It will not only be necessary for us to settle the relations which we shall hold to other cities and States, but to establish, if we can, new ones with a portion of our own State. Being the child of the Union, having drawn our sustenance from its bosom, and arisen to our present power and strength through the vigor of our mother—when deprived of her maternal advantages, we must rely upon our own resources and assume a position predicated upon the new phase which public affairs will present, and upon the inherent strength which our geographical, commercial, political, and financial preeminence imparts to us.

With our aggrieved brethren of the Slave States, we have friendly relations and a common sympathy. We have not participated in the warfare upon their constitutional rights or their domestic institutions. While other portions of our State have unfortunately been imbued with the fanatical spirit which actuates a portion of the people of New England, the city of New York has unfalteringly preserved the integrity of its principles in adherence to the compromises of the Constitution and the equal rights of the people of all the States. We have respected the local interests of every section, at no time oppressing, but all the while aiding in the development of the resources of the whole country. Our ships have penetrated to every clime, and so have New York capital, energy and enterprise found their way to every State, and, indeed, to almost every county and town of the American Union. If we have derived sustenance from the Union, so have we in return disseminated blessings for the common benefit of all. Therefore, New York has a right to expect, and should endeavor to preserve a continuance of uninterrupted intercourse with every section.

It is, however, folly to disgnise the fact that, judging from the past, New York may have more cause of apprehension from the aggressive legislation of our own State than from external dangers. We have already largely suffered from this cause. For the past five years, our interests and corporate rights have been repeatedly trampled upon. Being an integral portion of the State, it has been assumed, and in effect tacitly admitted on our part by nonresistance, that all political and governmental power over us rested in the State Legislature. Even the common right of

taxing ourselves for our own government, has been yielded, and we are not permitted to do so without this authority. \* \* \*

Thus it will be seen that the political connection between the people of the city and the State has been used by the latter to our injury. The Legislature, in which the present partizan majority has the power, has become the instrument by which we are plundered to enrich their speculators, lobby agents, and Abolition politicians. Laws are passed through their malign influence by which, under forms of legal enactment, our burdens have been increased, our substance eaten out, and our municipal liberties destroyed. Self-government, though guaranteed by the State Constitution, and left to every other county and city, has been taken from us by this foreign power, whose dependents have been sent among us to destroy our liberties by subverting our political system.

How we shall rid ourselves of this odious and oppressive connection, it is not for me to determine. It is certain that a dissolution cannot be peacefully accomplished, except by the consent of the Legislature itself. Whether this can be obtained or not, is, in my judgment, doubtful. Deriving so much advantage from its power over the city, it is not probable that a partizan majority will consent to a separation-and the resort to force by violence and revolution must not be thought of for an instant. We have been distinguished as an orderly and lawabiding people. Let us do nothing to forfeit this character, or to add to the present distracted condition of public affairs.

Much, no doubt, can be said in favor of the justice and policy of a separation. It may be said that secession or revolution in any of the United States would be subversive of all Federal authority, and, so far as the Central Government is concerned, the resolving of the community into its original elements-that, if part of the States form new combinations and Governments, other States may do the same. California and her sisters of the Pacific will no doubt set up an independent Republic and husband their own rich mineral resources. The Western States, equally rich in cereals and other agricultural products, will probably do the same. Then it may be said, why should not New York city, instead of supporting by her contributions in revenue two-thirds of the expenses of the United States, become also equally independent? As a free city, with but nominal duty on imports, her local Government could be supported without taxation upon her peo-Thus we could live free from taxes, ple. and have cheap goods nearly duty free. In this she would have the whole and united support of the Southern States, as well as all the other States to whose interests and rights under the Constitution she has

It is well for individuals or communities to look every danger square in the face, and to meet it calmly and bravely. As dreadful as the severing of the bonds that have hitherto united the States has been in contemplation, it is now apparently a stern and inevitable fact. We have now to meet it with all the consequences, whatever they may be. If the Confederacy is broken up the Government is dissolved, and it behooves every distinct community, as well as every individual, to take care of themselves.

When Disunion has become a fixed and certain fact, why may not New York disrupt the bands which bind her to a venal and corrupt master—to a people and a party that have plundered her revenues, attempted to ruin her commerce, taken away the power of self-government, and destroyed the Confederacy of which she was the proud Empire City? Amid the gloom which the present and prospective condition of things must cast over the country, New York, as a *Free Cily*, may shed the only light and hope of a future reconstruction of our once blessed Confederacy.

But I am not prepared to recommend the violence implied in these views. In stating this argument in favor of freedom, "peaceably if we can, forcibly if we must," let me not be misunderstood. The redress can be found only in appeals to the magnanimity of the people of the whole State. The events of the past two months have no doubt effected a change in the popular sentiment of the State and National politics. This change may bring us the desired relief, and we may be able to obtain a repeal of the law to which I have referred, and a consequent restoration of our corporate rights.

FERNANDO WOOD, Mayor.

January 6th, 1861.

### Personal Liberty Laws.

Among the grievances mentioned by the South Carolina Convention was the existence in Northern States of "Personal Liberty Laws." We give from the National Intelligencer of December 11th, 1860, a summary of these laws, except that of Massachusetts, which has been condensed more accurately than was there published :

### MAINE.

The laws of this State provide that no sheriff, deputy sheriff, coroner, constable, jailer, justice of the peace, or other officer of the State, shall arrest or detain or aid in so doing, in any prison or building belonging to this State, or in any county or town, any person on account of a claim on him as a fugitive slave, under a penalty not exceeding one thousand dollars, and make it the duty of all county attorneys to repair to the place where such person is held in custody, and render him all necessary and legal assistance in making his defence against said claim.

### NEW HAMPSHIRE.

The law of the State declares that slaves, coming or brought into the State; by or with the consent of the master, shall be free; declares the attempt to hold any person as a slave within the State as a felony, with a penalty of imprisonment of not less than one nor more than five years; provided, that the provisions of this section shall not apply to any act lawfully done by any officer of the United States, or other person, in the execution of any legal process.

### VERMONT.

This State, by her several acts of 1843, 1850 and 1858, provides that no court, justice of the peace or magistrate, shall take cognizance of any certificate, warrant, or process under the fugitive slave law; provides that no officer, or citizen of the State, shall arrest, or aid, or assist in arresting, any person for the reason that he is claimed as a fugitive slave; provides that no officer or citizen shall aid or assist in the removal from the State of any person claimed as a fugitive slave; provides a penalty of one thousand dollars, or imprisonment five years in State prison for violating this act. This act, however, shall not be construed to extend to any citizen of the State acting as a Judge of the Circuit or District Court of the United States, or as Marshal or Deputy-Marshal of the District of Vermont, or to any person acting under the command or authority of said Courts or Marshal. Requires the State's attorneys to act as counsel for alleged fugitives ; provides for issuing habeas corpus and the trial by jury of all questions of fact in issue between the parties; and ordains that every person who may have been held as a slave, who shall come, or be brought, or be in the State, with or without the consent of his or her master or mistress, or who shall come, or be brought, or be involuntarily, or in any way, in this State, shall be free. It is also provided that every person who shall hold, or attempt to hold, in this State, in slavery, or as a slave, any person mentioned as a slave in the section of this act, relating to fugitive slaves, or any free person in any form, or for any time, however short, under the pretence that such person is or has been a slave, shall, on conviction thereof, be imprisoned in the State prison for a term not less than one year nor more than fifteen years, and be fined not exceeding two thousand dollars.

### MASSACHUSETTS. That of 1860.

### The Governor to appoint commissioners in each county to defend alleged fugitives from service or labor, who shall secure to them impartial trials by jury: and the alleged fugitives may have the services of any attorney whose assistance may be de-

sired.

All the expenses of the defence of alleged | fugitives to be paid by the commonwealth.

Persons holding office under this State not to issue process for the arrest of alleged fugitives, nor to serve the same, nor to do any official act in furtherance of the execution of the fugitive slave law of 1793 or that of 1850. Any justice of the peace offending against this provision to forfeit not exceeding one thousand dollars, or be imprisoned, in jail, not exceeding one year, for each offence.

The jails and other prisons of the State not to be used for the detention or imprisonment of any person accused or convicted of any offence created by either of the said acts of Congress; or accused or convicted of resisting any process issued under either of them, or of rescuing or attempting to rescue any person detained under the provisions of either of said acts; nor for the imprisonment of a person arrested on mesne process or execution in a suit for damages in consequence of aid rendered to any fugitive escaping from service or labor.

Whoever removes from this State or assists in so doing, or comes into the State with the intention of removing or assisting in the removing therefrom, or procures or assists in procuring to be so removed, any person who is not "held to service or labor" by the "party" making "claim," or has not "escaped" from the "party" making "claim" or whose "service or labor" is not "due" to the party making "claim" within the meaning of the words of the Constitution of the United States, or the pretence that such person is so held or has escaped or that his "service or labor" is so "due" or with the intent to subject him to such "service or labor," shall be punished by fine not less than one thousand nor exceeding five thousand dollars, and by imprisonment in the State prison not less than one nor more than two years.

And any person sustaining wrong or injury by any proceedings punishable as aforesaid, may maintain an action and recover damages therefore.

Sheriffs, deputy-sherffs and others, state, county, city, district and town officers, and officers of the volunteer militia forbidden to arrest, imprison, detain, or return, or aid in so doing, any person for the reason that he is claimed or adjudged to be a fugitive from service or labor, shall be punished by fine not less than one thousand and not exceeding two thousand dollars, and by imprisonment in the State prision not less than one nor exceeding two years.

The volunteer militia not to aid in the seizure, detention or rendition of a person for the reason that he is claimed or adjudged to be a fugitive from service or labor. Any member thereof offending against this provision, to be punished by fine not less than one thousand nor more than two thousand dollars, and by imprisonment in the State penalty of \$5,000 against any person seizing

prison for not less than one nor more than two years.

The penalties prescribed by the two preceding sections not to apply to any act of military obedience performed by an officer or private of the militia.

The preceding sections not to apply to fugitives from justice. United States judicial officers not to hold any judicial office under the constitution and laws of this State, except that of justice of the peace.

No justice of the peace, while holding the office of a Commissioner of the United States Circuit Court, shall have authority to grant any warrant or to issue any process, civil or criminal, other than summons to witnesses, or hear and try any cause, civil or criminal, under the laws of this State.

### ACT APPROVED MARCH 25TH, 1861.

Sec. 1. Writ of habeas corpus returnable before the Supreme Judicial Court except in cases mentioned in Sections 30 and 32, Chap. 144 of General Statutes.

Sec. 2. On a trial upon a writ of habeas corpus, under sec. 19 of same act, issues to be framed under direction of the court, and rules of evidence, etc., to be those of the common law.

Sec. 3. A person not to be taken by writ of habeas corpus out of the custody of the U. S. Marshal, or his deputy, holding him by legal process issued by court of competent jurisdiction, provided, that the Supreme Judicial Court may investigate and determine upon the validity and legal effect of any process which may be relied upon to defeat the writ, or any other matter properly arising.

Sec. 4. No person to be punished, who, without any false pretence or unlawful intent, claims another person as a fugitive from service or labor.

Sec. 5. The right of any officer or court to call out the militia for the prevention or suppression of any riot or mob, is not prohibited by the 144th chapter aforesaid; the officers or members of the volunteer militia not excused from obeying any lawful order, or liable to any penalty for executing the same; *provided*, that the militia shall not be used to hinder the service of any lawful process of this commonwealth.

#### CONNECTICUT.

The State of Connecticut provides that every person who shall *falscly* and *mali*ciously declare, represent, or pretend that any free person entitled to freedom is a slave, or owes service or labor to any person or persons, with intent to procure or to aid or assist in procuring the forcible removal of such free person from this State as a slave, shall pay a fine of \$5,000 and be imprisoned five years in the Connecticut State prison; requires two witnesses to prove that any person is a slave or owes labor; denounces a

or causing to he seized any free person with intent to reduce him to slavery; depositions not to be admitted as evidence; witnesses testifying falsely, liable to \$5,000 fine and five vears' imprisonment.

### RHODE ISLAND.

This State by her legislation forbids the carrying away of any person by force out of the State; forbids any judge, justice, magis-trate, or court from officially aiding in the arrest of a fugitive slave under the fugitive slave law of 1793 or 1850; forbids any sheriff or other officer from arresting or detaining and person claimed as a fugitive slave; provides a penalty of \$500, or imprisonment not exceeding six months, for violating the act; denies the use of her jails to the United States for the detention of fugitive slaves.

### NEW YORK.

The State of New York has passed no laws having relation to the United States fugitive slave act of 1850. Though pressed frequently upon the Legislature, they have always failed of adoption. The old and obsolete act of 1840, entitled "An act to extend the right of trial by jury," extends the trial by jury to the cases of persons arrested as fugitive slaves; but in the fourth edition of the laws of the State, as prepared and pub-lished by Hon. HIRAM DENIO, at present Chief Justice of the Court of Appeals, may be found appended to the chapter containing this law the following note:

"An Act to Extend the Right of Trial by Jury, passed May 6th, 1840.—The decision of the Supreme Court of the United States, in Prigg vs. the Commonwealth of Penn-sylvania, 16 Peters' R. 539, establishes the doctrine that all State laws calculated to interfere with the third subdivision of section 2, article 4, of the Constitution of the United States are unconstitutional. Since that decision the fugitive slave law (Laws of Congress, 1850, chap. 60) has been passed, containing provisions repugnant to the whole of this act. It is therefore of no force; but, as it never has been repealed, it is here inserted."

#### NEW JERSEY.

The State of New Jersey has no statutes bearing on this subject save those which enjoin upon her State officers the duty of aiding in the recovery of fugitive slaves. Persons temporarily residing in the State are also permitted to bring with them and retain their domestic slaves.

### PENNSYLVANIA.

The State of Pennsylvania has not formally and specially legislated at all against the United States fugitive slave law of 1850, though there was an old statute of 1847 which prohibited any judge, justice of the peace, or alderman from taking cognizance \* March, 1861. A bill to repeal this was of the case of any fugitive from labor, "un-postponed in the House by a vote of 43 to 24.

der a certain act of Congress passed on the 12th day of February, 1793." During the last session of her Legislature the Commissioners appointed to revise and amend the Penal Laws of Pennsylvania (John C. Knox, Edward King, and David Webster) made a report to the Legislature that they had completed their labors, and the result was pre-sented in the shape of a bill entitled "An act to consolidate, revise and amend the Penal Laws of this Commonwealth." That report, on the thirty-first day of March, 1860, was enacted into a law, and by the ninety-fifth section it is enacted as follows:

"No Judge of any of the Courts of this Commonwealth, nor any Alderman or Justice of the Peace of said Commonwealth, shall have jurisdiction or take cognizance of the case of any fugitive from labor from any of the United States or Territories under any act of Congress, nor shall any such Judge, Alderman, or Justice of the Peace of this Commonwealth issue or grant any certificate or warrant of removal of any such fugitive from labor, under any act of Congress; and if any Alderman or Justice of the Peace of this Commonwealth shall take cognizance or jurisdiction of the case of any such fugitive, or shall grant or issue any certificate or warrant of removal, as aforesaid, then, and in either case, he shall be deemed guilty of a misdemeanor in office, and shall, on con-viction thereof, be sentenced to pay, at the discretion of the Court, any sum not exceeding one thousand dollars, the one-half to the party prosecuting for the same, and the other half to the use of this Commonwealth."

The theory of this law, it will be seen, is founded strictly on the decision of the Supreme Court of the United States in the Prigg case, and does not interfere with the functions of the Commissioner appointed under the United States law.

### MICHIGAN.\*

The law of this State requires State's attorneys to act as counsel for fugitive slaves; secures to persons arrested as fugitive slaves the benefits of the writ of habeas corpus, and trial by jury; denies use of State jails for detention of alleged fugitives; requires that identity of fugitive slaves shall be proved by two credible witnesses, or by legal evidence equivalent thereto, and provides a fine of not less than five hundred nor more than one thousand dollars, and imprisonment in State prison for five years, for forcibly seizing, or causing to be seized, any free person, with intent to have such person held in slavery.

#### IOWA.

This State has no legislation on the subject.

\* March, 1861. A bill to repeal this was indefinitely

### WISCONSIN.

The law of this State enjoins on the district attorneys the duty of acting as counsel for alleged fugitive slaves; secures to such persons the benefits of the writ of habeas corpus; provides for appeal to be taken to next stated term of the Circuit Court; secures trial by jury; enjoins a penalty of one thousand dollars and imprisonment of not more than five nor less than one year on all who "falsely and maliciously represent any free person to be a slave"; identity of alleged fugitive slave to be proved by two credible witnesses; no deposition to be received in evidence. It is also provided that—

"No judgment recovered against any person or persons for any neglect or refusal to obey, or any violations of, the act of Congress commonly termed the 'Fugitive Slave Act,' approved September eighteenth, one thousand eight hundred and fifty, or any of the provisions thereof, shall be a lien on any real estate within this State, nor shall any such judgment be enforcable by sale or execution of any real or personal property within this State; but all such sales shall be absolutely void; and in case of seizure or sale of any personal property, by virtue of any execution issued on such judgment, the defendant in said execution may maintain an action in replevin, or other action to secure possession thereof, in the manner provided by law for such actions, on affidavit filed as required by law, and a further statement therein that said execution issued in a judgment rendered under the provisions of the act of Congress aforesaid; and the provisions of this section shall also apply to judgments heretofore rendered."

### OHIO, INDIANA, ILLINOIS, MINNE-SOTA, CALIFORNIA AND OREGON.

We cannot find that these States have any laws in force on the subject.

#### Notes.

In VERMONT Legislature, Nov., 1860, a majority of the Committee of House of Representatives reported in favor of the repeal of the above law, but their proposition was rejected by a vote of two to one.

In MAINE, a repealing bill passed the Senate—yeas 17, yeas 10, and failed in the House.

In MASSACHUSETTS, a modifying bill was passed. For detail, see under "Massachusetts."

In RHODE ISLAND, a repealing bill passed the Senate—yeas 21, nays 9; the House yeas 49, nays 18.

In MICHIGAN, a repealing bill was indefinitely postponed in the House—yeas 43, nays 24.

In PENNSYLVANIA, a repealing bill was introduced into the House, but was not reached, in the course of business, when the firing on Sumter opened the war. A majority of each House was understood to be in favor of it. This law was first passed in 1847, after the Prigg decision; and one section prohibiting the use of State jails was repealed in 1852.

U. S. Senator Simmons of Rhode Island. in one of his speeches, made these remarks:

"Complaint had been made of personal liberty bills. Now, the Massachusetts personal liberty bill was passed by a Democratic House, a Democratic Senate, and signed by a Democratic Governor, a man who was afterwards nominated by Mr. Polk for the very best office in New England, and was unanimously confirmed by a Democratic United States Senate. Further than this, the very first time the attention of the Massachusetts Legislature was called to the propriety of a repeal to this law was by a Republican Governor, Governor Banks. Now, on the other hand, South Carolina had repealed a law imprisoning British colored sailors, but retained the one imprisoning those coming from States inhabited by her own brethren."

## THE PROCEEDINGS OF THE GOVERNMENT IN RELATION SECESSION MOVEMENT. TO THE

### Meeting and Proceedings of Congress.

THIRTY-SIXTH CONGRESS-SECOND SESSION.

Congress met on the first Monday of December, 1860, and was composed of the following persons :---

### SENATE.

JOHN C. BRECKINRIDGE of Kentucky, Vice President.

MAINE-H. Hamlin,\* W. P. Fessenden. New HAMPSHIRE-John P. Hale, Daniel Clark.

VERMONT-Solomon Foot, J. Collamer.

MASSACHUSETTS-Henry Wilson, Charles Sumner.

RHODE ISLAND-James F. Simmons, H. B. Anthony.

CONNECTICUT-L. S. Foster, Jas. Dixon.

NEWYORK-Wm.H.Seward, Preston King. NEW JERSEY-J. C. Ten Eyck, J. R. Thomson.

PENNSYLVANIA-S. Cameron, Wm. Bigler. DELAWARE-J. A. Bayard, W. Saulsbury. MARYLAND-J. A. Pearce, A. Kennedy. VIRGINIA-R. M. T. Hunter, James M.

Mason.

South CAROLINA-James Chesnut,† Jas. H. Hammond.

NORTH CAROLINA-Thomas Bragg, T. L. Clingman.

ALABAMA-B. Fitzpatrick, C. C. Clay, Jr. MISSISSIPPI-A. G. Brown, Jeff. Davis.

LOUISIANA-J. P. Benjamin, John Slidell. TENNESSEE-A. O. P. Nicholson, A. Johnson.

ARKANSAS-R. W. Johnson, W. K. Sebastian.

KENTUCKY-L. W. Powell, J. J. Crittenden.

MISSOURI—Jas. S. Green, Trusten Polk. OHIO—B. F. Wade, Geo. E. Pugh. INDIANA—J. D. Bright, G. N. Fitch. ILLINOIS—S. A. Douglas, L. Trumbull. MICHIGAN—Z. Chandler, K. S. Bingham. FLORIDA—D. L. Yulee, S. R. Mallory.

GEORGIA-Alfred Iverson, Robt. Toombs. TEXAS-John Hemphill, L. T. Wigfall.

WISCONSIN-Charles Durkee, J. R. Doolittle.

Iowa-J. W. Grimes, Jas. Harlan.

CALIFORNIA-M. S. Latham, William M. Gwin.

48

MINNESOTA-H. M. Rice, M. S. Wilkinson. OREGON-Joseph Lane, Edward D. Baker.

HOUSE OF REPRESENTATIVES.

WILLIAM PENNINGTON OF New Jersey, Speaker.

MAINE-D. E. Somes, John J. Perry, E. B. French, F. H. Morse, Israel Washburn, Jr., \$ S. C. Foster.

NEW HAMPSHIRE-Gilman Marston, M. W. Tappan, T. M. Edwards.

VERMONT-E. P. Walton, J. S. Morrill, H. E. Royce.

MASSACHUSETTS-Thos. D. Eliot, James Buffinton, Charles Francis Adams, Alexander H. Rice, Anson Burlingame, John B. Alley, Daniel W. Gooch, Charles R. Train, Eli Thayer, Charles Delano, Henry L. Dawes.

RHODE ISLAND-C. Robinson, W. D. Brayton

CONNECTICUT - Dwight Loomis, John Woodruff, Alfred A.Burnham, Orris S. Ferry. DELAWARE-W. G. Whiteley.

New YORK-Luther C. Carter, James Humphreys, Daniel E. Sickles, W. B. Maclay, Thos. J. Barr, John Cochrane, George Briggs, Horace F. Clark, John B. Haskin, Chas. H. Van Wyck, Wm. S. Kenyon, Chas. L. Beale, Abm. B. Olin, John H. Reynolds, Jas. B. McKean, G. W. Palmer, Francis E. Spinner, Clark B. Cochrane, James H. Graham, Richard Franchot, Roscoe Conkling, R. H. Duell, M. Lindley Lee, Chas. B. Hoard, Chas. B. Sedgwick, M. Butterfield, Emory B. Pottle, Alfred Wells, Wm. Irvine, Alfred Ely. Augustus Frank, Edwin R. Reynolds, Elbridge G. Spaulding, Reuben E. Fenton. New JERSEY-John T. Nixon, John L. N.

Stratton, Garnett B. Adrain, Jetur R. Riggs,

Stratton, Garnett B. Adram, Jettir K. Kiggs, Wm. Pennington (Speaker). PENNSTLVANIA—Thos. B. Florence, E. Joy Morris, John P. Verree, Wm. Millward, John Wood, John Hickman, Henry C. Long-necker, Jacob K. McKenty, Thaddeus Stevens, John W. Killinger, James H. Camp-bell, George W. Scranton, William H. Dim-bell, George W. Scranton, William H. Scranton, William mick, Galusha A. Grow, James T. Hale, Benjamin F.Junkin, Edward McPherson, Samuel S. Blair, John Covode, William Montgomery, James K. Moorhead, Robert McKnight, William Stewart, Chapin Hall, Elijah Babbitt.

‡ Resigned and succeeded January 2d, 1861, by Hon. Stephen Coburn.

<sup>\*</sup> Resigned January 17th, 1861, and succeeded by Hon. Lot M. Morrill. † Did not attend.

MARYLAND-Jas. A. Stewart, J. M. Harris, H. W. Davis, J. M. Kunkel, G. W. Hughes.

VIRGINIA-John S. Millson, Muscoe R. H. Garnett, Daniel C. De Jarnette, Roger A. Pryor, Thomas S. Bocock, William Smith, Alex. R. Boteler, John T. Harris, Albert G. Jenkins, Shelton F. Leake, Henry A. Ed-mundson, Elbert S. Martin, Sherrard Clemens.

South CAROLINA-John McQueen, Wm. Porcher Miles, Lawrence M. Keitt, Milledge L. Bonham, John D. Ashmore, Wm. W. Boyce.

NORTH CAROLINA-W. N. H. Smith, Thos. Ruffin, W. Winslow, L. O'B. Branch, John A. Gilmer, Jas. M. Leach, Burton Craige, Z. B. Vance.

GEORGIA-Peter E. Love, M. J. Crawford, Thos. Hardeman, Jr., L. J. Gartrell, J. W. H. Underwood, James Jackson, Joshua Hill, John J. Jones.

ALABAMA-Jas. L. Pugh, David Clopton, Sydenh. Moore, Geo. S. Houston, W. R. W. Cobb, J. A. Stallworth, J. L. M. Curry.

MISSISSIPPI-L. Q. C. Lamar, Reuben Davis, William Barksdale, O. R. Singleton, John J. McRae.

LOUISIANA-John E. Bouligny, Miles Taylor, T. G. Davidson, John M. Landrum.

Оню—G. H. Pendleton, John A. Gurley, C. L. Vallandigham, William Allen, James M. Ashley, Wm. Howard, Thomas Corwin, Benj. Stanton, John Carey, C. A. Trimble, Chas. D. Martin, Saml. S. Cox, John Sherman, H. G. Blake, William Helmick, C. B. Tompkins, T. C. Theaker, S. Edgerton, Edward Wade, John Hutchins, John A. Bingham.

KENTUCKY-Henry C. Burnett, Green Adams, S. O. Peyton, F. M. Bristow, W. C. Anderson, Robert Mallory, Wm. E. Simms, L. T. Moore, John Y. Brown, J. W. Stevenson

TENNESSEE-T. A. R. Nelson, Horace Maynard, R. B. Brabson, William B. Stokes, Robert Hatton, James H. Thomas, John V. Wright, James M. Quarles, Emerson Etheridge, Wm. T. Avery

INDIANA-Wm. E. Niblack, Wm. H. English, Wm. M'Kee Dunn, Wm. S. Holman, David Kilgore, Albert G. Porter, John G. Davis, James Wilson, Schuyler Colfax, Chas. Case, John U. Pettit.

ILLINOIS-E. B. Washburne, J. F. Farnsworth, Owen Lovejoy, Wm. Kellogg, I. N. Morris, John A. McClernand, James C. Robinson, P. B. Fouke, John A. Logan.

ARKANSAS-Thomas C. Hindman, Albert Rust.

MISSOURI-J. R. Barrett, T. L. Anderson, John B. Clark, James Craig, L. H. Wood-son, John S. Phelps, John W. Noell. MICHIGAN-William A. Howard, Henry Waldron, F. W. Kellogg, De W. C. Leach. FLORIDA-George S. Hawkins.

TEXAS-John H. Reagan, A. J. Hamilton. Iowa-S. R. Curtis, Wm. Vandever.

WISCONSIN-John F. Potter, C. C. Washburne, C. H. Larrabee.

MINNESOTA-Cyrus Aldrich, Wm. Windom. OREGON-Lansing Stout.

KANSAS-Martin F. Conway, (sworn Jan. 30th, 1861).

### President Buchanan's Last Annual Message.

It was delivered on Tuesday, December 4th, 1860.

Mr. Buchanan alluded to the distracted condition of the country, and appealed to the American people. He declared that the election of any one of our fellow-citizens to the office of President does not, of itself, afford just cause for dissolving the Union; and that, to justify a revolutionary resistance the Federal Government must be guilty of "a deliberate, palpable, and dangerous exercise" of powers not granted by the Constitution-which, he alleged, and proceeded to prove, was not at all the He denied that "secession" could be case. justified as a Constitutional remedy, and asserted that the "principle is wholly inconsistent with the history, as well as the char-acter, of the Federal Constitution;" and claimed that such a proposition was not advanced until many years after the origin of the Federal Government, and that then it was met and refuted by the conclusive arguments of General Jackson. He held that this Government is invested with all the attributes of sovereignty over the special subjects to which its authority extends, and then discussed, in the following language, what he 'denominated

#### THE POWER TO COERCE A STATE.

The question, fairly stated, is: Has the Constitution delegated to Congress the power to coerce a State into submission which is attempting to withdraw or has actually withdrawn from the Confederacy? If answered in the affirmative, it must be on the principle that the power has been conferred upon Congress to declare and to make war against a State. After much serious reflection I have arrived at the conclusion that no such power has been delegated to Congress nor to any other department of the Federal Government. It is manifest, upon an inspection of the Constitution, that this is not among the specific and enumerated powers granted to Congress; and it is equally apparent that its exercise is not "necessary and proper for carrying into execution" any one of these powers. So far from this power having been delegated to Congress, it was expressly refused by the Convention which framed the Constitution. It appears, from the proceedings of that body, that on the 31st May, 1787, the clause "authorizing an exertion of the force of

CALIFORNIA-Charles L. Scott, John C. Burch.

the whole against a delinguent State" came up for consideration. Mr. Madison opposed it in a brief but powerful speech, from which I shall extract but a single sentence. He observed : "The use of force against a State would look more like a declaration of war than an infliction of punishment, and would probably be considered by the party attacked as a dissolution of all previous compacts by which it might be bound." Upon his motion the clause was unanimously postponed, and was never, I believe, again presented. Soon afterwards, on the 8th June, 1787, when incidentally adverting to the subject, he said: "Any Government for the United States, formed on the supposed practicability of using force against the unconstitutional proceedings of the States, would prove as visionary and fallacious as the government of Congress," evidently meaning the then existing Congress of the old Confederation

Without descending to particulars, it may be safely asserted that the power to make war against a State is at variance with the whole spirit and intent of the Constitution. Suppose such a war should result in the conquest of a State, how are we to govern it afterwards? Shall we hold it as a province and govern it by despotic power? In the nature of things we could not, by physical force, control the will of the people, and compel them to elect Senators and Representatives to Congress, and to perform all the other duties depending upon their own volition, and required from the free citizens of a free State as a constituent member of the Confederacy.

But, if we possessed this power, would it be wise to exercise it under existing circumstances? The object would doubtless be to preserve the Union. War would not only present the most effectual means of destroying it, but would banish all hope of its peaceable reconstruction. Besides, in the fraternal conflict a vast amount of blood and treasure would be expended, rendering future reconciliation between the States impossible. In the meantime who can foretell what would be the sufferings and privations of the people during its existence?

The fact is, that our Union rests upon public opinion, and can never be cemented by the blood of its citizens shed in civil war. If it cannot live in the affections of the people, it must one day perish. Congress possesses many means of preserving it by conciliation; but the sword was not placed in their hand to preserve it by force.

In this connexion, I shall merely call attention to a few sentences in Mr. Madison's justly celebrated report, in 1799, to the Legislature of Virginia. In this he ably and conclusively defended the resolutions of the preceding Legislature against the strictures of several other State Legislatures. These were mainly founded upon the protest of the shall separate themselves from the Union.

Virginia Legislature against the "alien and sedition acts," as "palpable and alarming infractions of the Constitution." In peinting out the peaceful and constitutional remedies, and he referred to none other to which the States were authorized to resort on such occasions, he concludes by saying, "that the Legislatures of the States might have made a direct representation to Congress with a view to obtain a rescinding of the two offend ing acts, or they might have represented to their respective Senators in Congress their wish that two-thirds thereof would propose an explanatory amendment to the Constitu tion, or two-thirds of themselves, if such had been their option, might, by an application to Congress, have obtained a Convention for the same object."

This is the very course which I earnestly recommend in order to obtain an "explanatory amendment" of the Constitution on the subject of slavery. This might originate with Congress or the State Legislatures, as may be deemed most advisable to attain the object.

The explanatory amendment might be confined to the final settlement of the true construction of the Constitution on three special points:

1. An express recognition of the right of property in slaves in the States where it now exists or may hereafter exist.

2. The duty of protecting this right in all the common Territories throughout their territorial existence, and until they shall be admitted as States into the Union, with or without slavery, as their constitutions may prescribe.

3. A like recognition of the right of the master to have his slave, who has escaped from one State to another, restored and "delivered up" to him, and of the validity of the fugitive-slave law enacted for this purpose, together with a declaration that all State laws impairing or defeating this right are violations of the Constitution, and are consequently null and void.

It may be objected that this construction of the Constitution has already been settled by the Supreme Court of the United States. and what more ought to be required? The answer is, that a very large proportion of the people of the United States still contest the correctness of this decision, and never will cease from agitation and admit its binding force until established by the people of the several States in their sovereign character. Such an explanatory amendment would, it is believed, forever terminate the existing dissensions and restore peace and harmony among the States.

It ought not to be doubted that such an appeal to the arbitrament established by the Constitution itself would be received with favor by all the States of the Confederacy. In any event, it ought to be tried in a spirit of conciliation before any of these States

### Opinion of Attorney-General Black upon the Powers of the President.

ATTORNEY-GENERAL'S OFFICE, November 20, 1860.

SIR: I have had the honor to receive your note of the 17th, and I now reply to the grave questions therein propounded as fully as the time allowed me will permit. Within their respective spheres of action the Federal

Within their respective spheres of action the Federal Government and the Government of a State are both of them independent and supreme, but each is utterly powerless beyond the limits assigned to it by the Constitution. If Congress would attempt to change the law of descents, to make a new rule of personal succession, or to dissolve the family relations existing in any State, the act would be simply void, but not more void than would be a State law to prevent the recapture of fugitives from labor, to forbid the carrying of the mails, or to stop the collection of duties on imports. The will of a State, whether expressed in its constitution or laws, cannot, while it remains in the Confederacy, absolve her people from the duty of obeying the just and constitutional requirements of the Cantral Government. Nor can any act of the Central Government displace the jurisdiction of a State, because the laws of the United States are supreme and binding only so far as they are passed in *pursuance of the Constitution*. I do not say what might be effected by mere revolutionary force. I am speaking of legal and constitutional right. This is the view always takon by the Judiciary, and so

This is the view always taken by the Judiciary, and so universally adopted that the statement of it may seem common-place. The Supreme Court of the United States has declared it in many cases. I need only refer you to the United States vs. Booth where the present Chief Justice, expressing the unanimous opinion of himself and all his brethren, enunciated the doctrine in terms so clear and full that any further demonstration of it can scarcely be required.

required. The duty which these principles devolve not only upon every officer, but every citizen, is that which Mr. Jefferson expressed so compendiously in his first inaugural, namely, "to support the State Governments in all their rights, as the most competent administrations for their domestic concerns, and the surest bulwarks against anti-republican tendencies," combined with "the preservation of the General Government, in its whole constitutional vigor, as the sheet-anchor of our peace at home and safety abroad."

To the Chief Executive Magistrate of the Union is confided the solemn duty of seeing the laws faithfully exceed. That he may be able to meet this duty with a power equal to its performance, he nominates his own subordinates and removes them at his pleasure. For the same reason the land and naval forces are under his orders as their commander-in-chief. But his power is to be used only in the manner prescribed by the legislative department. He cannot accomplish a legal purpose by illegal means, or break the laws himself to prevent them from being violated hy others.

The acts of Congress sometimes give the President a broad discretion in the use of the means by which they are to be executed, and sometimes limit his power so that he can exercise it only in a certain prescribed manner. Where the law directs a thing to be done, without saying how, that implies the power to use such means as may be necessary and proper to accomplish the end of the Legislature. But where the mode of performing a duty is pointed out by statute, that is the exclusive mode, and no other can be followed. The United States have no common law to fall back upon when the written law is defective. If, therefore, an act of Congress declares that a certain thing shall be done by a particular officer, it cannot be done by a different officer. The agency which the law furnishes for its own execution must be used, to the exclusion of all others. For instance, the revenues of the United States are to be collected in a certain way, at certain established ports, and by a certain class of officers; the President has no athority, under any circumstances, to collect the same revenues at other places by a different sort of officers, or in ways not provided for. Even if the machinery furnished by Congress for the collection of the duties should by any cause become so deranged or broken up that it could not be used, that would not be a legal reason for substituting a different stind of machinery in its place. The law requires that all goods imported into the United States within certain cellsetion-distircts shall be entered

The law requires that all goods imported into the United States within certain collection-districts shall be entered at the proper port, and the duty thereon shall be received by the Collector appointed for and residing at that port. But the functions of the Collector may be exercised anywhere at or within the port. There is no law which confines him to the custom-house, or to any other particular spot. If the custom-house were burnt down, he might remove to another building; if he were driven from the shore, he might go on board a vessel in the harbor. If he keeps within the port he is within the law. A port is a place to

which merchandize is imported, and from whence it is exported. It is created by law. It is not merely a harbor or haven, for it may be established where there is nothing but an open roadstead, or on the shore of a navigable river, or at any other place where vessels may arrive and discharge or take in their cargoes. It comprehends the city or town which is occupied by the mariners, merchants, and others who are engaged in the business of importing and exporting goods, navigating the ships and furnishing them with provisions. It includes also so much of the water adjacent to the city as is usually occupied by vessels discharging or receiving their cargoes, or lying at auchor and waiting for that purpose.

The first section of the act of March 2d, 1833, anthorized the President in a certain contingency to direct that the custom-house for any collection district be established and kept in any secure place within some port or harbor of such district, either npon land or on board any vessel. But this provision was temporary, and expired at the end of the session of Congress next afterwards. It conferred upon the Executive a right to remove the site of the custom-house, not merely to any secure place within the legally-established port of entry for the district—that right he had before—but it widened his authority so as to allow the removal of it to any port or harbor within the whole district. The enactment of that law aud the limitation of it to a certain period of time now past, is not therefore, an argument against the opinion above expressed that yon can now if necessary, order the duties to be collected on board a vessel inside of any established port of entry. Whether the first and fifth sections of the act of 1833, both of which were made temporary by the eighth section, should be re-enacted, is a question for the legislative department.

Your right to take such measures as may seem to be necessary for the protection of the public property is very clear. It results from the proprietary rights of the Government as owner of the forts, arsenals, magazines, dockproperty which the United States have bought, built, and paid for. Besides, the Government of the United States is authorized by the Constitution (Art. I, Sec. S) to "exercise exclusive legislation in all cases whatsoever . over all places purchased by the consent of the Legislature of the State in which the same shall be for the erection of forts, magazines, arsenals, dock-yards, and other needful buildings." It is believed that no important public building has been bought or erected on ground where the Legislature of the State, in which it is, has not a passed a law consenting to the purchase of it and ceding the exclusive jurisdiction. This Government, then, is not only the owner of these buildings and grounds, but, by virtue of nany one of an owner's rights is plainer than another, it is that of keeping exclusive possession and repelling intrusion. The right of defending the public property includes also the right of recapture after it has been unlawfully taken by another. President Jefferson held the opinion, and acted upon it, that he could order a military force to take possession of any land to which the United States had title, though they had never occupied it before, though a private party claimed and held it, and though it was not then needed nor proposed to be used for any purpose connected with the operations of the Government. This may have been a stretch of Executive power ; but the right of retaking public property in which the Government has been carrying on its lawful business, and tharper's Ferry in October, 1859, every one acknowledged the legal justiee of it.

the legal justice of it. I come now to the point in your letter which is probably of the greatest practical importance. By the act of 1807 you may employ such parts of the land and naval forces as you shall judge necessary for the purpose of causing the laws to be duly executed, in all cases where it is lawful to use the militia for the same purpose. By the act of 1795 the militia may be called forth "whenever the laws of the United States shall be opposed or the execution thereof obstructed in any State by coubluations too powerful to be suppressed by the ordinary course of judicial proceedings, or by the power vested in the marshals." This imposes upon the President the sole responsibility of deeiding whether the exigency has arisen which requires the use of military force; and in proportion to the magnitude of that responsibility will be his care not to overstep the limits of his legal and just authority. The laws referred to in the act of 1795 are manifestly

The laws referred to in the act of 1795 are manifestly those which are administered by the judges and executed by the ministerial officers of the courts for the punishment of orime against the United States, for the protection of rights claimed under the Federal Constitution and laws, and for the enforcement of such obligations as come

within the cognizance of the Federal Judiclary. To compel obedience to these laws, the Courts have authority to punish all who obstruct their regular administration, and the marshals and their deputies have the same powers the marshals and their deputies have the same powers as sheriffs and their deputies in the several States in exo-cuting the laws of the States. These are the ordinary means provided for the execution of the laws, and the whole spirit of our system is opposed to the employment of any other except in cases of extreme necessity, arising out of great and unusual combinations against them. Their agency must continue to be used notil their in-Their agency must continue to be used notil their in-capacity to cope with the power opposed to them shall be plainly demonstrated. It is only upon clear evidence to that effect that a military force can be called into the field. Even then its operations must be purely defensive. It can suppress only such combinations as are found directly opposing the laws and obstructing the execution thereof. It can do no more than what might and ought to be done by a civil posse, if a civil posse could be raised large enough to meet the same opposition. On such occasions especially the military power must be kept in strict subordination to the civil autherity, since it is only in aid of the latter that the former can act at all.

is only in aid of the latter that the former can act at all. But what if the feeling in any State against the United States should become so universal that the Federal officers States should become so unversal that the Federal officers themselves (including judges, district-attorneys, and mar-shals) would be reached by the same influences, and resign their places? Of course the first step would be to appoint others in their stead, if others could be got to serve. But, in such an event, it is more than probable that great diffi-culty would be found in filling the offices. We can easily conceive how it might become altogether impossible. We are therefore obliged to consider what can be done in case we have no courts to issue judicial 'process, and no minis-terial officers to execute it. In that event troops would certainly be out of place, and their use wholly illegal. If they are sent to aid the courts and marshals, there must be courts and marshals to be oided. Without the carceries of courts and marshals to be aided. Without the exercise of those functions, which belong exclusively to the civil ser-vice, the laws cannot be executed in any event, no matter what may be the physical strength which the Government has at its command. Under such circumstances, to send a military force into any State with orders to act arging the has at its command. Under such circumstances, to send a military force into any State with orders to act against the people would be simply making war upon them. The existing laws put and keep the Federal Government strictly on the defensive. You can use force only to repel

an assault on the public property, and aid the courts in the performance of their duty. If the means given you to col-lect the revenue and execute the other laws be insufficient for that purpose, Congress may extend and make them more effectual to that end. If one of the States should declare her independence,

your action cannot depend upon the rightfulness of the cause upon which such declaration is based. Whether the retirement of a State from the Union be the exercise of a right reserved in the Constitution or a revolutionary moveingut facts to the total you have not in either case the authority to recognize her independence or to absolve her from her Federal obligations. Congress or the other States in convention assembled must take such measures as may be necessary and proper. In such an event I see no course for you but to go straight onward in the path you have hitherto trodden, that is, execute the laws to the extent of the defensive means placed in your hands, and act generally upon the assumption that the present constitutional rela-tions between the States and the Federal Government con-tinue to exist until a new order of things shall be established, either by law or force.

tinue to exist until a new order of things shall be estab-lished, either by law or force. Whether Congress has the constitutional right to make of the Federal Government to carry it on by means of force to be drawn from the other States, is a question for Congress itself to consider. It must be admitted that no such power is expressly given; nor are there any words in the Constitution which imply it. Among the powers rouncrated in article I, section 8, is that "to declare war, grant letters of marque and reprisal, and to make rules concerning captures on land and water." This certainly means nothing more than the power to commence and earry on hostilities against the foreign onemies of the na-tion. Another clause in the same section gives Congress the power "to provide for calling forth the militia," and to use them within the limits of the State. But this power is so restricted by the words which immediately follow, that it can be exercised only for one of the following purposes : 1. To execute the laws of the Union; that is, to add the Federal officers in the performance of their regular duties. 2. To suppress insurrections against the States; but this is confined by article IV section 4, to cases in which the State ensember of the and and the states is what this is confined by article IV section 4, to case in which the State and the all apply for assistance against ther own people. 3. To repel the invasion of a State by enemies who come from abroad to assail her in her own territory. All these provisions are made to protect the States, not to anthorize an attack by one part of the country upon another; to preserve their peace, and not to plunge them into civil war.

Our forefathers do not seem to have thought that war was calculated "to form a more perfect union, establish justice, insure domestic tranquility, provide for the common de-fence, promote the general welfare, and secure the blessings of liberty to ourselves and our posterity." There was un-doubtedly a strong and universal conviction among the men who framed and ratified the Constitution, that military force would not only be useless, but pernicious as a means of holding the States together. If it be true that war cannot be declared, nor a system

of general hostilities carried on by the central government against a State, then it seems to follow that an attempt to do so would be *ipso fucto* an expulsion of such State from the Union. Being treated as an alien and an enemy, she would be compelled to act accordingly. And if Congress would be compelled to act accordingly. And if Congress shall break up the present Union by unconstitutionally putting strife and enmity, and armed hostility, between different sections of the construction was meant to insure, will not all the States be absolved from their Federal obli-gations? Is any portion of the people beund to contribute their money or their blood to carry on a contest like that? The right of the General Government to preserve itself in its whole constitutional vigor by repelling a direct and positive aggression upon its property or its officers, cannot be denied. But this is a totally different thing from an offensive war to punish the people for the political misdeeds of their State governments, or to prevent a threatened vio-

of their State governments, or to prevent a threatened vio-lation of the Constitution, or to enforce an acknowledgment that the government of the United States is supreme. The States are colleagues of one another, and if some of them shall conquer the rest and hold them as subjugated provinces, it would totally destroy the whole theory upon which they are now connected.

If this view of the subject he as correct as I think it is, then the Union must utterly perish at the moment when Congress shall arm one part of the people against another for any purpose beyond that of merely protecting the General Government in the exercise of its proper constitutional I am, very respectfully, yours, etc., J. S. BLACK. functions.

To the President of the United States:

### Committee of Thirty-three.

December 4th. In the House of Representatives, Mr. Boteler of Virginia moved that so much of the President's message as relates to the present perilous condition of the country be referred to a special committee of one from each State, which was agreed to-yeas 145, nays 38, as follows :

was agreed to—yeas 145, nays 38, as follows: YEAS.—Messrs. Adams of Massachusetts, Adams of Ky., Adrain, Aldrich, Allen, Alley, Anderson of Ky., Anderson, of Missouri, Arery, Babbitt, Barr, Barrett, Boock, Boteler, Bouligny, Branch, Brayton, Briggs, Bristow, Brown, Burch, Burnett, Campbell, Carter, Clarke of Miss. Dejar-Burnett, Campbell, Carter, Clarke of Miss. Dejar-nette, Delano, Duell, Dunn, Edunadson, Eliot, Ely, English, Etheridge, Ferry, Florence, Foster, Fouke, Frank, French, Gilmer, Goech, Graham, Gurley, Hale, Hall, Hardeman, Harris of Md., Harris of Va., Haskin, Hatton, Helmick, Hill, Hoard, Holman, Houston, Howard of Ohio, Hughes, Humphrey, Jenkins, Junkin, Kellogg of Illinois, Kenyon, Kilgore, Killinger, Kunkle, Larrabee, Leach of N. Carolina, Marris of Ill, Nelson, Niblack, Nixon, Noel, Palmer, Pend-Morris of Ill, Nelson, Niblack, Nixon, Noel, Palmer, Pend-Morris of Ill, Nelson, Niblack, Nixon, Noel, Palmer, Pen-alder, Eicke, Kings, Robinson of R. I., Robinson of Illinois Royce, Rust, Sickles, Smith of Va., Smith of N. C., Somes, Spaulding, Sinner, Stevenson, Stewart of Md., Stewart of Pa., Stokes, Stoul, Stratton, Thayer, Theaker, Thomas, Train, Trimble, Vallandigham, Vance, Vandever, Walton, Wash-burn of Maine, Webster, Whiley, Window, Wood, Woodruff-145. Nars.-Messrs. Ashley, Beale, Bingham, Blair, Blake,

burn of Maine, Webster, Whiteley, Windom, Winstow, Wood, Woodruff-145. NAY8.-Messrs. Ashley, Beale, Bingham, Blair, Blake, Buffinton, Burlingame, Burnham, Carey, Case, Edgerton, Fenton, Grow, Hickman, Howard of Mich, Hutchins, Irvine, Kellogg of Mich., Leach of Mich., Lee, Loomis, Lovejoy McKean, McKnight, Morse, Perry, Potter, Pottle, Sedgwick, Sherman, Stauton, Stevens, Tappan, Tompkins, Wade, Wash-burn of Wis., Washburne of Ill., Wells-38.

During the vote, Mr. SINGLETON of Mis-sissippi, said he declined to vote because he had not been sent here to make any compromise or patch up existing difficulties, and that a Convention of the people of Mississippi would consider and decide the subject.

Mr. HAWKINS of Florida, said the day of compromise had passed, and that he was opposed, and he believed his State was opposed, to all and every compromise.

Mr. CLOPTON of Alabama, believed in the right of a State to secede, considered that the only remedy for present evils, and would not hold out any delusive hope, or sanction

any temporizing policy. Mr. MILES of South Carolina, said their delegation had not voted on the question because they conceived they had no interest in it. They considered their State as already withdrawn from the Confederacy in every thing except form.

Mr. PUGH of Alabama, said that State intended following South Carolina out of the Union by the 10th of January next, and he paid no attention to any action taken in this The Committee consisted of body.

Mr. Corwin of Ohio. Mr. Millson of Virginia. Mr. Adams of Massachusetts.

Mr. Winslow of North Carolina.

Mr. Humphrey of New York.

Mr. Boyce of South Carolina.

Mr. Campbell of Pennsylvania.

Mr. Love of Georgia.

Mr. Ferry of Connecticut.

Mr. Davis of Maryland.

Mr. Robinson of Rhode Island.

Mr. Whiteley of Delaware.

Mr. Tappan of New Hampshire.

Mr. Stratton of New Jersey.

Mr. Bristow of Kentucky.

Mr. Morrill of Vermont.

Mr. Nelson of Tennessee.

Mr. Dunn of Indiana. Mr. Taylor of Louisiana.

Mr. Davis of Mississippi.

Mr. Kellogg of Illinois.

Mr. Houston of Alabama.

Mr. Morse of Maine.

Mr. Phelps of Missouri.

Mr. Rust of Arkansas.

Mr. Howard of Michigan.

Mr. Hawkins of Florida.

Mr. Hamilton of Texas.

Mr. Washburn of Wisconsin.

Mr. Curtis of Iowa.

Mr. Burch of California.

Mr. Windom of Minnesota.

Mr. Stout of Oregon.

Messrs. Hawkins and Boyce asked to be excused from service on the Committee, but the House refused.

### Propositions Submitted to the Committee.

By Mr. THAVER of Massachusetts. A series of resolutions to the effect that the representatives of the people should devote themselves to the cause of the country, in the spirit of the fathers of the Republic;

that the people should be true to their constitutional obligations; that as our differences had arisen mainly from the acquisition of new territory, no more territory ought ever to be acquired ; affirming the right of self-government in the Territories as independent of Congress or the President; in favor of admission of new States with a population equal to the ratio of representation; that the Government of the United States should never own any more territory, and that annexation of territory in the future should only be by consent of the States; that there should be no Congressional legislation whatever on the subject of slavery, and that every Congressional District should in future be an Electoral District, entitled every four years to elect one Presidential elector.

By Mr. JOHN COCHRANE of New York. A preamble and resolution to the effect that the decision of the Supreme Court in the Dred Scott case should be received as a settlement of the questions therein discussed and decided; also, in favor of amending the Constitution so as to give a right to Congress to establish territorial governments; providing for admission of new States with a population equal to the Federal ratio of representation, with or without slavery, and prohibiting Congress and the people of the territory from impairing the right of property in slaves during its existence as a territory.

Mr. JOHN COCHRANE of New York, also offered amendments to the Constitution in favor of a division of territory on the line of thirty-six degrees thirty minutes; in favor of admission of new States with or without slavery; to prohibit Congress from abolishing the inter-State slave-trade; reaffirming the obligation of the fugitive slave law; guaranteeing a right of transit in free States of persons with slaves, and declaring void all nullifying acts of State or Territorial Legislatures.

Mr. JOHN COCHRANE of New York, also offered a preamble and resolutions to the same effect, as regards the question of slavery in the territories, with his proposition to amend the Constitution, just cited, and coupled with a resolution declaring that the Constitution of the United States existed only by agreement of sovereign States, and that any attempt of the Federal Government to coerce a sovereign State into the observance of the Constitutional compact, would be to levy war upon a substantial power and precipitate a dissolution of the Union.

Mr. HASKIN offered as a substitute to the above, a resolution directing the Committee on the Judiciary to inquire and report as to what action Congress should take in regard to enforcing the Constitution and laws in South Carolina, and what was the duty of the Executive in this regard.

By Mr. MALLORY of Kentucky. Instruct-

ing the Committee of Thirty-three in favor of a division of territory on line of thirtysix degrees thirty minutes, and admission of new States with a population equal to the Federal ratio of representation, with or without slavery, and to prohibit Congress from abolishing slavery in any places within its jurisdiction, or from abolishing the inter-State slave-trade.

By Mr. STEVENSON of Kentucky. To so amend the fugitive slave law as to make it felony to resist the execution of said law

felony to resist the execution of said law By Mr. ENGLISH of Indiana. That said committee be instructed to inquire into the expediency of settling all matters of controversy upon the following basis: 1. Division of Territory between the free and slave States, with provision for admission of new States with a population equal to the Federal ratio of representation. 2. Prohibiting Congress from impairing the right of property in slaves. 3. Making the city, county, or township liable in double the value of fugitive slaves forcibly rescued, etc.

By Mr. KILGORE of Indiana. To give the right of trial by jury, where a fugitive slave claims to be free, with right of appeal on writ of error to either party. Monied compensation in case of rescue by force, etc., and making it a criminal offence to resist the enforcement of the fugitive slave law.

By Mr. HOLMAN of Indiana. Resolutions opposing the right of secession, declaring the duty of the General Government to enforce with temperate firmness and in good faith the provisions of the Constitution, and instructing the Committee of Thirty-three to inquire and report what legislation is needed to thwart any attempted nullification.

By Mr. NIBLACK of Indiana. That the committee be instructed to inquire and report whether Congress has power to provide by law for a payment of money to the owner of a' fugitive slave prevented by violence from recapturing him.

By Mr. JOHN A. MCCLERNAND of Illinois To same effect, and further to inquire and report as to the expediency of establishing a *special* Federal police to execute the laws of the United States, and prevent opposition thereto.

By Mr. NOELL of Missouri. That said Committee be instructed to inquire and report as to the expediency of abolishing the office of President of the United States, and establishing in lieu thereof an Executive Council of three members to be elected by districts composed of contiguous States as nearly as possible—each member to be armed with a veto power; and also as to whether the equilibrium of free and slave States in the United States Senate can be restored and preserved, particularly by a voluntary division of some of the slave States into two or more States.

By Mr. HINDMAN of Arkansas. In favor of amending the Constitution as follows: 1st. An express recognition of slavery

in the States where it exists, and prchibition of right of Congress to interfere therewith or with the inter-State slave-trade.

2d. Expressly requiring Congress to protect slavery in the territories, and in all places under its jurisdiction.

3d. For admission of new States, with or without slavery, as their Constitutions should provide.

4th. Right of transit for persons with slaves through the free States.

5th. To prohibit a right of representation in Congress to any States passing laws to impair the obligations of the fugitive slave law until such acts shall have been repealed.

6th. Giving the slave States a negative upon all acts of Congress relating to slavery.

7th. Making the above amendments, and all provisions of the Constitution relative to slavery unamendable.

8th. Granting to the several States authority to appoint all Federal officers within their respective limits.

By Mr. LARRABEE of Wisconsin. Recommending the several States to call a Convention of all the States to adopt such measures as the existing exigency required.

measures as the existing exigency required. By Mr. ANDERSON of Missouri. In favor of a joint resolution to refer the questions at issue between the free and slave States to the Supreme Court of the United States for their opinion, and when obtained, that Congress should pass all necessary laws for giving effect to the opinion of said court.

By Mr. SMITH of Virginia. In favor of declaring out of the Union every State which shall aim by legislation to nullify an act of Congress.

By Mr. SICKLES of New York. amend the Constitution so as to provide, that whenever a Convention of delegates chosen in any State by the people thereof under a recommendation of its Legislature, shall rescind its ratification of the Constitution, the President shall appoint, with consent of the Senate, three Commissioners to agree with such State regarding the disposition of the public property therein, and the proportion of the public debt which such State ought to assume, which being approved by the President and two-thirds of the Senate, he shall by proclamation declare the assent of the United States to the withdrawal of any such State from the Union.

By Mr. DUNN of Indiana. A resolution in favor of a more effectual execution of the 2d Section of the 4th Article of the Constitution to secure the personal rights of citizens of any State, travelling or sojourning in any other State.

By Mr. ADRAIN of New Jersey. A series of resolutions in substance as follows: Declaring the doctrine of non-intervention of Congress in the territories; the right of the people to be admitted as a State, either

54

with or without slavery, as its Constitution should provide; in favor of the repeal of all enactments of State Legislatures which conflicted with acts of Congress, or the Constitution, affirming the constitutionality of the fugitive slave law; inculcating a kind and fraternal spirit among the people of the different States, and deprecating any interference with the domestic institutions of one another, and declaring that the Constitution could only be preserved by the same spirit of compromise that had governed its formation.

By Mr. MORRIS of Pennsylvania. Instructing the Committee of Thirty-three to report if there are any personal liberty bills of any State in conflict with the fugitive slave law, and to inquire and report if the fugitive slave law is not susceptible of amendment so as to prevent kidnapping, and render more certain the ascertainment of the true character of the fugitive.

By Mr. STEWART of Maryland. A lengthy preamble reciting to the effect that the States were sovereign, independent political organizations originally, and had united from time to time under such form of association or union as was deemed expedient-which form of association had been from time to time changed peaceably as circumstances required; that it was the deliberate opinion of many of the people that our present form of government, from causes either resulting from or in violation of the Constitution, was inadequate for the purposes for which it was created; that certain States were threatening to withdraw their allegiance; and that we had reached a crisis in our history which required an alteration of the present form of government; and he followed with a resolution instructing the committee on the President's message to inquire if any measures could be adopted for preserving the rights of all the States under the Union, and if not, to then inquire as to the best mode of adjusting the rights of the several States in a dissolution of the Union.

By Mr. LEAKE of Virginia. A resolution in favor of the amendment of the Constitution in the following particulars: Making it the duty of Congress to pass laws to protect slavery where it exists; taking away all territorial jurisdiction over the matter; guaranteeing the right of transit for persons with their slaves in any State; reaffirming the fugitive slave clause, with additional provision for compensation in case of failure to return the fugitive.

By Mr. JENKINS of Virginia. A resolution directing the Committee of Thirty-three to inquire into the best mode of amending the fugitive slave law so as to adequately punish its infraction and render compensation when the slave should not be restored; also as to whether the election of a President hostile to the slaveholding interest was not a sufficient reason to justify the slaveholding States to require that their concurrent sanction should be

separately given to every act of the Federal Government, or whether there should be a dual Executive or a dual Senate, or the assent of a majority of the Senators from each section necessary to pass any law, or what other measures were needed for the protection of the slaveholding States.

By Mr. Cox of Ohio. A resolution directing the Committee of Thirty-three to inquire what additional legislation was necessary to enforce the provisions of the Constitution relative to rendition of fugitives, and that such inquiry should be made with special reference to punishing all judges, attorney-generals, executives, and other State officers who should oppose its execution.

By Mr. HUTCHINS of Ohio. A resolution directing said committee to inquire what legislation was needed to give effect to section two of article four of the Constitution. granting to "the citizens of each State all the privileges and immunities of citizens in the several States," and to secure to all the people the full benefit of article four of amendments to the Constitution, which guarantees exemption from unreasonable searches and seizures, etc.

By Mr. SHERMAN of Ohio. A series of resolutions to the effect that the only remedy for existing dissensions was to be found in a faithful observance of all the compromises of the Constitution and the laws made in pursuance thereof; instructing the Committee of Thirty-three to inquire whether any State or the people thereof have failed of their duty in this regard, and if so what remedy should be made therefor, and directing said committee to divide the remaining Territories into States of convenient size with a view to their immediate admission into the Union.

By Mr. BINGHAM of Ohio. A resolution directing the committee to report such additional legislation as might be needed to put down armed rebellion and protect the property of the United States from seizure, and the citizens thereof from unlawful violence.

#### OTHER PROPOSITIONS SUBMITTED TO THE HOUSE.

Mr. ETHERIDGE of Tennessee proposed a series of amendments to the Constitution, in substance as follows :

That Congress shall not interfere with slavery in the States, nor in any forts, arsenals, etc., ceded to the United States by a slave State, nor in the District of Columbia, without the consent of Maryland, Virginia, and the inhabitants of the District, nor without making compensation; nor with the inter-State slave-trade. Foreign slavetrade prohibited. In regard to slavery in Territories, a provision similar to that proposed by Mr. Cochrane, before cited. No foreign territory to be acquired except on a concurrent two-thirds vote of both houses of Congress or by a treaty ratified by two-thirds of the members of the Senate; and an amendment in reference to rendition of fugitives

Mr. DAVIS of Indiana. A preamble and resolution reciting the Ordinance of Secession of South Carolina, and directing the Committee on the Judiciary to inquire and report what legislation had been rendered necessary in consequence thereof.

Mr. HOLMAN of Indiana, offered a series of resolutions, denying the right of secession, affirming it to be the duty of the General Government to collect the revenue and protect the public property, and instructing the Committee on the Judiciary to inquire and report what legislation is needed to enable the Government to discharge its constitutional duty in these regards.

Mr. FLORENCE of Pennsylvania, offered as a substitute for the report of the Committee of Thirty-three a series of amendments to the Constitution, in substance as follows:

1st. Granting the right to hold slaves in all territory south of 36° 30', and prohibiting Congress and the Territorial Legislature from interfering with it therein, or in any other place within the jurisdiction of the United States, without the consent of all the slave States.

2d. Admitting States into the Union with a population equal to the ratio of representation, with or without slavery, as their Constitution shall prescribe.

3d. Prohibiting any alteration of the present basis of representation-declaring the slavery question to be one exclusively for each State; but with proviso that this amendment shall not be construed so as to release the General Government from its obligations to suppress domestic insurrection in any State.

4th. Giving the right to abolish slavery in the District of Columbia exclusively to the State of Maryland.

5th. Prohibiting any State from passing laws to obstruct the rendition of fugitive slaves.

6th. Granting the right of transit with slaves through all the States.

7th. Declaring all slaves brought into any State by permission thereof, and escaping, to be fugitives from labor.

8th. Prohibiting the African slave-trade, and also prohibiting persons of African descent from becoming citizens.

9th. Making all acts of any inhabitants of the United States tending to incite slaves to insurrection, penal offences.

10th. Making the county in which any fugitive slave shall be rescued, liable to pay the value thereof. 11th. Prohibiting slavery in territory

north of 36° 30'.

12th. Giving returned fugitive slaves a trial by a jury in the place to which they shall have been returned.

13th. Provides for rendition of fugitives from justice.

14th. Declaring inviolable the rights of the citizens of any State sojourning in another State.

15th. No State shall retire from the Union without the consent of three-fourths of all the States.

16th. Giving full power to three-fourths of the States at any time to call a Convention to amend or abolish the Constitution.

17th. Declares articles 8, 9 and 10 of these amendments to be unalterable, unless by consent of all the slave States.

Mr. FENTON of New York, offered as a substitute for the report of the Committee of Thirty-three a preamble and resolution, reciting the conflicting differences of opinion as to the causes of the present disturbances, and favoring the calling of a Convention of delegates from the several States.

Mr. Kellogg of Illinois. As a substitute for the report of the Committee, amendments to the Constitution in substance as follows.: Prohibiting slavery in territory north of 36° 30', and permitting it south of that line, and providing for its admission as States with a population equal to the ratio of representation, with or without slavery, as its Constitution should prescribe; pro-hibiting any interference by Congress with the subject in the States, either to abolish or establish slavery; affirmatory of the fugitive slave-clause of the Constitution, with amendment providing for compensation, and pro-

hibiting the foreign slave-trade. By Mr. VALLANDIGHAM of Ohio, the following preamble and resolution :

JOINT RESOLUTION PROPOSING AMENDMENTS TO THE CON-STITUTION.

WHEREAS, the Constitution of the United States is a grant of specific powers delegated to the Federal Govern-ment by the people of the several States, all powers not delegated to it nor prohibited to the States being reserved to the States respectively, or to the people; and whereas it is the tendency of stronger Governments to only ore their is the tendency of stronger Governments to enlarge their powers and jurisdiction at the expense of weaker Govern-ments, and of majorities to usurp and abuse power and oppress minorities, to arrest and hold in check which ten-dency compacts and Constitutions are made; and whereas the only effectual constitutional security for the rights of into only elactual constitutional security for the rights of minorities, whether as people or as States, is the power expressly reserved in Constitutions of protecting those rights by their own action; and whereas this mode of protection, by checks and guarantees, is recognized in the Federal Constitution, as well in the case of the equality of the States in successful and the states of the Federal constitution, as were in the case of the equation of the States in representation and in suffrage in the Senate, as in the provision for overruling the vote of the President, and for amending the Constitution, not to enumerate other examples; and whereas, unhappily, because of the vast extent and diversified interests and institutions of the several States of the Union, sectional divisions can no longer he suppressed; and whereas it concerns the peace and sta-bility of the Federal Union and Government that a division of the States into mere slaveholding and non-slaveholding sections, causing hitherto, and from the nature and neces sity of the case, inflammatory and disastrons controversies upon the subject of slavery, ending already in present dis-raption of the Union, should be forever hereafter ignored : and whereas this important end is best to be attained by the recognition of other sections without regard to slavery

the recognition of other sections without regard to slavery neither of which sections shall alone be strong enough to oppress or control the others, and each be vested with the power to protect itself from aggressions: Therefore, Be it resolved by the Senate and House of Representa-tives of the United States of America in Congress assem-bled, (two-thirds of both Houses concurring,) That the following articles be and are bereby proposed as amend-ments to the Constitution of the United States, which shall be valid to all intents and purposes as part of said Constitution when ratified by Conventions in three-fourths of the several States: Arricer XII, SEC, 1. The United States are divided into

ARTICLE XIII. SEC. 1. The United States are divided into

four sections, as follows: The States of Maine, New Hampshire, Vermont, Mass-achusetts, Rhode Island, Connectiont, New York, New

Jersey, and Pennsylvania, and all new States annexed and admitted into the Union, or formed or erected within the jurisdiction of any of said States, or by the junction of two or more of the same or of parts thereof, or out of ter-ritory acquired north of said States, shall constitute one scoular to be known as the North

section, to be known as the NorrH. The States of Ohio, Indiana, Illinois, Michigan, Wis-consin, Minnesota, Iowa, and Kansas, and all new States annexed or admitted into the Union, or erected within the annexed or admitted into the Union, or erected within the jurisdiction of any of said States, or by the junction of two or more of the same or of parts thereof, or out of territory now held or hereafter acquired north of latitude 36° 30' and east of the crest of the Rocky Mountains, shall con-stitute another section, to be known as the WEST. The States of Oregon and California, and all new States

The States of Oregon and California, and all new States annexed and admitted into the Union, or formed or erected within the jurisdiction of any of said States, or by the junction of two or more of the same or of parts thereof, or out of territory now held or hereafter acquired west of the crest of the Rocky Mountains and of the Rio Grande, shall constitute another section to be known as the PACIFIC. The States of Delaware, Maryland, Virginia, North Caro-lina, South Carolina, Georgia, Florida, Alabama, Missis-sippi, Louisiana, Texas, Arkansas, Tennessee, Kentucky, and Missouri, and all new States annexed and admitted into the Union, or formed or erected within the jurisdic-tion of any of said States, or by the junction of two or more of the same or of parts thereof, or out of territory more of the same or of parts thereof, or out of territory acquired east of the Rio Grande and south of latitude 36° 30', shall constitute another section, to be known as the Same the South.

SEC. 2. On demaud of one-third of the Senators of any one of the sections on any bill, order, resolution, or vote, to which the concurrence of the House of Representatives may be necessary, except on a question of adjournment, a vote shall be had by sections, and a majority of the Sena-tors from each section voting shall be necessary to the pas-sage of such bill, order, or resolution, and to the validity

Sage of such yole.
Sage of such yole.
Such yole.
Such yole.
Such yole.
Such yole.
Such yole.
The logislature thereof may direct for the State at large.
The other electors to which each State may be entitled shall be chosen in the respective Congressional districts which the State may at here result deservation. into which the State may at the regular decennial period have been divided, by the electors of each district having the qualifications requisite for electors of the most numer-ons branch of the State Legislature. A majority of all the electors in each of the four sections in this article estab-lished shall be necessary to the choice of President and Vice President; and the concurrence of a majority of the States of each section shall be necessary to the choice of States of each section shall be necessary to the choice of President by the House of Representatives, and of the Sen-ators from each section to the choice of Vice President by the Senate, whenever the right of choice shall devolve upon them respectively. SEC. 4. The President and Vice President shall hold

their offices each during the term of six years; and neither shall be eligible to more than one term except by the votes of two-thirds of all the electors of each section, or of the States of each section, whenever the right of choice of President shall devolve upon the House of Representatives,

President shall devolve upon the House of Representatives, or of the Senators from each section whenever the right of choice of Vice President shall devolve upon the Senate. SEC. 5. The Congress shall by law provide for the case of a failure by the House of Representatives to choose a Pres-ident, and of the Senate to choose a Vice President, whenever the right of choice shall devolve upon them rewhenever the right of choice shall devolve upon them re-spectively, declaring what officer shall them act as Presi-dent; and such officer shall act accordingly until a Presi-dent be elected. The Congress shall also provide by law for a special election for President and Vice President in such case, to be held and completed within six months from the expiration of the term of office of the last pre-redime Breident and to be availated in all worse the ceding President and to be conducted in all respects as provided for in the Constitution for regular elections of the same officers, except that if the House of Representatives shall not choose a President, should the right of choice devolve upon them, within twenty days from the opening of the certificates and counting of the electoral votes, then the Vice President shall act as President as in the case of the dath or other constitution. of the death or other constitutional disability of the President. The term of office of the President chosen President. The territ of once of the Arestonic clockin inder such special elections shall continue six years from the 4th day of March preceding such election. ARTICLE XIV. No State shall secede without the consent of the Legislatures of all the States of the section to which

the State proposing to secede belongs. The President shall have power to adjust with seceding States all questions arising by reason of their secession ; but the terms of adjustment shall be submitted to the Congress for their

approval before the same shall be valid. ARTICLE XV. Neither the Congress nor a Territorial Legislature shall have power to interfere with the right

of the citizens of any of the States within either of the sections to migrate upon equal terms with the citizens of the States within either of the other sections, to the Territories of the United States; nor shall either have power to destroy or impair any rights of either person or property in the Territories. New States annexed for admission In the Territories. New States annexed for admission into the Union, or formed or erected within the jurisdiction of other States, or by the junction of two or more States or parts of States, and States formed with the consent of the Congress out of any territory of the United States, shall be entitled to admission upon an equal footing with the original States, under any constitution establishing a Government republican in form, which the people thereof may ordain, whenever such States shall contain. within may ordain, whenever such States shall contain, within an area of not less than thirty thousand square miles, a population equal to the then existing ratio of representation for one member of the House of Representatives.

### Report of the Committee of Thirtythree.

January 14th, 1861. Mr. CORWIN reported a series of propositions, with a written statement in advocacy thereof, which closes with the remark that "the Committee were not unanimous on all the resolutions and bills presented, but a majority of a quorum was obtained on them all."

Mr. CHARLES FRANCIS ADAMS OF Massachusetts, made a separate report, which "The general conclusion to closes thus: which the subscriber has arrived from a close observation of the action of the Committee, is this: That no form of adjustment will be satisfactory to the recusant States which does not incorporate into the Constitution of the United States a recognition of this obligation to protect and extend slavery. On this condition, and on this alone, will they consent to withdraw their opposition to the recognition of the constitutional election of the Chief Magistrate. Viewing the mat-ter in this light, it seems unadvisable to attempt to proceed a step further in the way of offering unacceptable propositions. He can never give his consent to the terms demanded.

"For this reason it is that, after having become convinced of this truth, he changed his course and declined to recommend the very measures which he in good faith had offered. It certainly can be of no use to propose as an adjustment that which has no prospect of being received as such by the other party. Hence he feels it his duty now to record his dissent from the action of a majority of his colleagues in introducing any measures whatever for the consideration of the House."

Mr. C. C. WASHBURN of Wisconsin; and Mr. MASON W. TAPPAN of New Hampshire, also submitted a minority report, concluding with this resolution :

Resolved, That the provisions of the Constitution are ample for the preservation of the Union and the protection of all the material interests of our country; that it needs to be obeyed rather than amended; and that our extrication from our present difficulties is to be looked for in efforts to preserve and protect the public property and enforce the laws, rather than in new guarantees for particular interests, or compromises or concessions to unreasonable demands.

Messrs. TAYLOR of Louisiana, PHELPS of Missouri, Rust of Arkansas, Whiteley of Delaware, and WINSLOW of North Carolina, also submitted a minority report, arguing that the present difficulties can only be remedied by amendments to the Constitution, and suggesting the Crittenden proposition. If that cannot be adopted, they recommend a Convention of the States, with a view, if no adjustment can be effected, to peaceable separation, by providing for a partition of the common property of the United States, settling the terms on which the social and commercial intercourse between the separated States shall be conducted, and making a permanent arrangement with respect to the navigation of the Mississippi river.

Messrs. Burch of California and Stout of Oregon submitted a separate report, sustaining the propositions submitted by Mr. CORWIN, but urging, in addition, the assembling of a National Constitutional Convention to consider the whole subject matter of the difficulties.

Mr. NELSON of Tennessee made a report, arguing in favor of the establishment of the line of 36° 30', north of which slavery shall be prohibited, and south of which it shall be protected until any territory shall contain the population requisite for a member of Congress, when, if its form of government be republican, it shall be admitted into the Union, with or without slavery, as the Constitution of such new State may He dissented from the proposition provide. to admit New Mexico, opposed the proposed change of the fugitive slave law, and recommended the Crittenden proposition.

Messrs. Love of Georgia, and HAMILTON of Texas, dissented from the majority report, and recommended the Crittenden proposition.

Mr. FERRY was unable to concur in the report made by MR. CORWIN, and made a statement of his position.

VOTE ON THE FIRST PROPOSITION OF THE COM-MITTEE OF THIRTY-THREE, FEB. 27TH, 1861. DECLARATORY RESOLUTIONS.

Resolved by the Senate and House of Representatives of the United States of America in Congress assembled, That all attempts on the part of the Legislatures of any of the States to obstruct or hinder the recovery and surrender of fugitives from service or labor, are in derogation of the Constitution of the United States, inconsistent with the comity and good neighborhood that should prevail among the several States, and dangerous to the peace of the Union.

Resolved, That the several States be respectfully requested to cause their statutes to be revised, with a view to ascertain if any of them are in conflict with or tend to

laws of the United States, made in pursuance of the second section of the fourth article of the Constitution of the United States for the delivery up of persons held to labor by the laws of any State and escaping therefrom; and the Senate and House of Representatives earnestly request that all enactments having such tendency be forthwith repealed, as required by a just sense of constitutional obligations, and by a due regard for the peace of the Republic; and the President of the United States is requested to communicate these resolutions to the Governors of the several States, with a request that they will lay the same before the Legislatures thereof respectively.

Resolved, That we recognize slavery as now existing in fifteen of the United States by the usages and laws of those States; and we recognize no authority, legally or otherwise, outside of a State where it so exists, to interfere with slaves or slavery in such States, in disregard of the rights of their owners or the peace of society.

Resolved, That we recognize the justice and propriety of a faithful execution of the Constitution, and laws made in pursuance thereof on the subject of fugitive slaves, or fugitives from service or labor, and discountenance all mobs or hindrances to the execution of such laws, and that citizens of, each State shall be entitled to all the privileges and immunities of citizens in the several States.

Resolved, That we recognize no such conflicting elements in its composition, or sufficient cause from any source, for a dissolution of this Government; that we were not sent here to destroy, but to sustain and harmonize the institutions of the country, and to see that equal justice is done to all parts of the same; and finally, to perpetuate its existence on terms of equality and justice to all the States.

Resolved, That a faithful observance, on the part of all the States, of all their constitutional obligations to each other and to the Federal Government, is essential to the peace of the country.

Resolved, That it is the duty of the Federal Government to enforce the Federal laws, protect the Federal property, and preserve the Union of these States.

Resolved, That each State be requested to revise its statutes, and, if necessary, so to amend the same as to secure, without legislation by Congress, to citizens of other States travelling therein, the same protection as citizens of such State enjoy; and also to protect the citizens of other States travelling or sojourning therein against popular violence or illegal summary punishment, without trial in due form of law, for imputed crimes.

Resolved, That each State be also respectfully requested to enact such laws as will prevent and punish any attempt whatever embarrass or hinder the execution of the in such State to recognize or set on foot the lawless invasion of any other State or Territory.

Resolved, That the President be requested to transmit copies of the foregoing resolutions to the Governors of the several States, with a request that they be communicated to their respective Legislatures.

to their respective Legislatures.
YEAS-Méssis. Charles F. Adams, Green Adams, Adrain, Aldrich, William C. Anderson, Babhitt, Barrett, Beale, Boteler, Bra'son, Brayton, Briggs, Bristow, Erourn, Barch, Burnham, Butterfield, Campbell, Carter, John B. Clark, Coburn, Clark B. Cochrane, John Cickrane, Colfax, Conkling, Corwin, Covole, Cox, James Craig, H. Winter Davis, John G. Dawis, Delamo, Dinmirk, Dunn, Edwards, Ely, English, Etheridge, Farnesworth, Ferry, Florence, Foster, Fouke, French, Gilmer, Graham, Grow, Gurley, Hale, Hall, Hamilton, J. Morrison Harris, John T. Harris, Haskin, Hatton, Helmick, Veard, Holman, William H. Howard, William A. Inoward, Humphrey, Irvine, Junkin, Francis W. Kellogg, William Kellogg, Kenyon, Kilgore, Killinger, Larrabee, James M. Leach, Lee, Logan, Longnecker, Loomis, Maday, Mallory, Raaco N. Morris, Nore, Noled, Olin, Palmer, Perry, Pettit, Peyton, Phelps, Potter, Pottle, Quarles, John H. Reynolds, Rice, Riggo, Christopher Robinson, James C. Robinson, Koyce, Scranton, Sherman, Sickles, Stams, William Stewart, Stokes, Stoul, Stratton, Thayer, Tompkins, Train, Trimble, Vallandighum, Vance, Verree, Waldon, Welster, Wilson, Windom, Wood, Woodruff-137. Nars-Messis, Alley, Avery, Bingham, Blair, Blake, Scow, Stranch, Burett, Carey, Case, Conway, Burton Craige, Dawes, De Jamett, Duel, Camund, Score, Hore, Malor, Mood, Woodruff-137. YEAS-Messrs, Charles F. Adams, Green Adams, Adrain,

Bocock, Branch, Boffinton, Burlingame, Burnett, Carey, Case, Conway, Burton Craige, Dawes, De Janrette, Duell, Edmund-son, Eliot, Fenton, Garnett, Gooch, Hickman, Hindman, Hughes, Hutchins, Jenkins, De Witt C. Leach, Leake, Love-joy, Elbert S. Martin, Potter, Pryor, Edwin R. Reynolds, Ruffin, Rust, Sedgwick, William Smith, Somes, Spinner, Stevens, James A. Stevart, Tappan, Thomas, Vandever, Van Wyck, Wade, Cadwalader C. Washburn, Wells, Whiteley, Winstow, Woodson-53.

So the joint resolution was passed.

VOTE ON SECOND PROPOSITION. CONSTITUTIONAL AMENDMENT.

Be it resolved by the Senate and House of Representatives of the United States of America in Congress assembled, two-thirds of both Houses concurring, That the following article be proposed to the Legislatures of the several States as an amendment to the Constitution of the United States. which, when ratified by three-fourths of said Legislatures, shall be valid, to all intents and purposes, as a part of the said Constitution, namely:

ART. XII. No amendment of this Constitution having for its object any interference within the States with the relation between their citizens and those described in section second of the first article of the Constitution as "all other persons," shall originate with any State that does not recognize that relation within its own limits, or shall be valid without the assent of every one of the States composing the Union.

Before the vote was taken Mr. CORWIN offered the following substitute for the above article :

ART. XII. No amendment shall be made to the Constitution which will authorize or give to Congress the power to abolish or interfere, within any State, with the domestic institutions thereof, including that of persons held to labor or service by the laws of said State.

Which was agreed to, yeas 120, nays 61, as follows :

THE GOVERNMENT. 99
TRAS-MeSSTS. Charles F. Adams, Green Adams, Aldrich, William C. Anderson, Avery, Babbitt, Barr, Burrett, Bocok, Boteler, Brahson, Briggs, Bristow, Brown, Burch, Burnett, Botteffield, Campbell, Horace F. Clark, John B. Clark, Colurn, Clark B. Cochrane, John Cochrane, Colfax, Corwin, Cax, James Craig, H. Winter Davis, John G. Davis, Dawes, De Jarnette, Delano, Dimmick, Dunn, Edmundson, English, Etheridge, Florence, Fouke, Gilmer, Hale, Hall, Humilton, J. Morrison Harris, John T. Harris, Hatton, Helmick, Hoard, Holman, William Howard, William A. Howard, Hughes, Iumphrey, Jenkins, Junkin, Francis W. Kellogg, William Kellogg, Kenyon, Kilgore, Killinger, Kunkel, Lar-rabee, James M. Leach, Lcale, Logan, Mallory, Marston, Charles D. Martin, Elbert S. Martin, Maynard, McClernand, McKenty, McKnight, McPherson, Millson, Montgomery, Laban T. Moore, Moorhead, Morrill, Edward Joy Morris, Isaac N. Morris, Nelson, Niblack, Nixon, Noeld, Olin, Palmer, Peyton, Phelps, Porter, Pryor, Quarles, John H. Reynolds, Rice, Riggs, Christopher Robinson, James C. Rob-inson, Ruffin, Rust, Scranton, Sietenson, James A. Slewart, Stokes, Stoud, Stratton, Stevenson, James A. Slewart, Stokes, Stoud, Stratton, Buringame, Burnham, Carey, Carter, Case, Conkling, Conway, Duell, Edgerton, Edwards, Elidt, Ely, Fenton, Ferry, Foster, Frank, Gooch, Graham, Grow, Hickman, Hudman, Hutchins, Irvine, De Witt C. Evach, Lee, Longnecker, Vindom, Worldson, Wright-I20. Nars-Messrs. Alley, Ashley, Beale, Bingham, Blair, Patheke, Brayton, Bufinton, Buringame, Burnham, Carey, Carter, Case, Conkling, Conway, Duell, Edgerton, Edwards, Elidt, Ely, Fenton, Ferry, Foster, Frank, Gooch, Graham, Grow, Hickman, Handman, Hutchins, Irvine, De Witt C. Evach, Lee, Longnecker, Vand Wyck, Wade, Waldron, Cadwal-Schuer, Voiter, Pottle, Edwin R. Reynolds, Roye, Sedwick, Somes, Spinner, Stevens, William Stewart, Tappan, Tomp-Kins, Train, Vandever, Van Wyck, Wade, Waldron, Cadwal-deer C. Washburn, Elihbu E. Washburne, Wells, Wilson, Winslow, Woodruff

The resolution as amended was then negatived, yeas 120, nays 71, two-thirds not voting in the affirmative, as follows:

ing in the affirmative, as follows: YEAS-Messrs. Charles F. Adams, Green Adams, Ad-rain, William C. Anderson, Avery, Babbitt, Barr, Barrett, Bocock, Boteler, Bouligny, Brabson, Branch, Briggs, Bris-tow, Brown, Burch, Burnett, Campbell, Horace F. Clark, John B. Clark, Clark B. Cochrane, Join Cochrane, Colfax, Corwin, Caz, James Craig, Burton Craige, H. Winter Davis, John G. Davis, De Jarnett, Delano, Dimmick, Dunn, Ed-munson, English, Etheridge, Florence, Houke, Garnett, Gilmer, Hale, Hall, Hamilton, J. Morrison Harris, John T. Harris, Hatton, Ilelmick, Hoard, Holman, Wan. Howard, William Kellogg, Kenyon, Killinger, Kunkel, Larrobee, James M. Leach, Leake, Logan, Muclay, Mallory, Charles D. Martin, Elbert S. Martin, Maynard, McClernand, Mc Kenty, McKnight, McPherson, Milson, Mongomery, Laban T. Moore, Moorhead, Morrill, Edward Joy Morris, Isaac N. Morris, Nelson, Niblock, Nixon, Noell, Olin, Peyton, Phelps, Porter, Pryor, Quarles, John H. Reynolds, Rice, Russ, Scranton, Sickles, Simms, William N. H. Smith, Spaulding, Stanton, Stevenson, James A. Stewart, Stokes, Nut, Stretton, Thovar, Thowar, Torwar, Stokes, St

Huge, Chinkyen, Sickles, Simms, William N. H. Smith, Spaulding, Stanton, Sickles, Simms, William N. H. Smith, Spaulding, Stanton, Stevenson, James A. Stewart, Stokes, Stout, Stratton, Thayer, Thomas, Trimble, Vallandigham, Vance, Verree, Webster, Whiteley, Windoni, Winslow, Wood, Wnodson, Wright-120, Nars-Messrs, Aldrich, Alley, Ashley, Beale, Bingham, Blatr, Blake, Brayton, Buffinton, Burlingame, Burnham, Blatr, Blake, Brayton, Buffinton, Euringame, Burnham, Butteffield, Carey, Carter, Case, Coburn, Conkling, Conway, Covode, Dawes, Duell, Edgerton, Edwards, Eliot, Ely, Farnsworth, Fenton, Ferry, Foster, Frank, Gooch, Graham, Grow, Gurley, Hickman, Hutchins, Irvine, Francis W. Kellogg, Kilgore, De Witt C. Leach, Lee, Longnecker, Loonis, Lovejoy, Marston, McKean, Morse, Perry, Petiti, Potter, Pottle, Edwin R. Reynolds, Royce, Sedgwick, Somes, Train, Vandever, Van Wyck, Wade, Waldron, Walton, Cadwalader C. Washburn, Ellihu B. Washburne, Wells, Wilson, Woodruff-71. Wilson, Woodruff-71.

Mr. KILGORE entered a motion to re-consider.

February 28th. This motion was carried, yeas 125, nays 68, and the joint resolution then passed, yeas 133, nays 65, as follows:

LICH PASSECI, Yeas 135, hays 60, as follows: YEAS-Messrs. Charles F. Adams, Green Adams, Ad-ram, Aldrich, William C. Anderson, Avery, Babbitt, Barr, Barrett, Bocock, Boteler, Bouligny, Brabson, Branch, Briggs, Bristow, Brown, Burch, Burnett, Butterfield, Camp-bell, Horace F. Clark, John B. Clark, Clemens, Clark B. Cochrane, John Cochrane, Colfax. Corwin, Cox, James Craig, Burton Craige, H. Winter Davis, John G. Davis, De Jarnette, Delano, Dimmiek, Dunn, Edmundson, English, Etherldge, Florence, Fouke, French, Garnett, Gilmer, Hale, Kun, Hatton, Helmick, Hoard, Holman, William Howard, William A. Howard, Hughes, Humphrey, Jenkins, Junkin,

William Kellogg, Kenyon, Kilgore, Killinger, Kunkel, Larrabee, James M. Leach, Leake, Logan, Maclay, Mallory, Oharles D. Martin, Elbert S. Martin, Maynard, McClernand, McKenky, McKnight, McPherson, Millacon, Montgomery, Laban T. Moore, Moorhead, Morrill, Edward Joy Morris, Isaac N. Morris, Morse, Nelson, Niblack, Nixon, Noell, Olin, Palmer, Pendleton, Peyton, Philps, Porter, Pryor, Quarles, John H. Reynolds, Rice, Riggs, Christopher Robinson, James C. Robinson, Ruffin, Rust, Scott, Scranton, Sherman, Sickles, Simms, William N. H. Smith, Spaulding, Stanton, Netwenson, James A. Stewart, Stokes, Stant, Stratton, Thaver, Stevenson, James A. Slewart, Stokes, Stout, Stratton, Thayer, Theaker, Thomas, Trimble, Vallandigham, Vance, Verree, Webster, Whiteley, Windom, Winslow, Wood, Woodson, Theas. Webster, W

Webster, Whiteley, Windom, Winstow, Woos, Wright-133. Nars-Messrs. Alley, Ashley, Beale, Bingham, Blair, Blake, Brayton, Buffinton, Burlingame, Burnham, Carey, Carter, Case, Coburn, Conkling, Conway, Dawes, Dnell, Edgerton, Edwards, Eliot, Ely, Farnsworth, Fenton, Ferry, Foster, Frank, Gooch, Grow, Gurley, Hickman, Hindman, Hutchins, Irvine, Francis W. Kellogg, DeWitt C. Leach, Lee, Longnecker, Loomis, Lovejoy, Marston, McKean, Pettit, Potter, Pottle, Edwin R. Reynolds, Royce, Sedwick Somes, Spinner, Stevens, William Stewart, Tappan, Tomp-kins, Train, Vendever, Yan Wyck, Wade, Waldron, Walton, Cadwalader C. Washburn, Ellihu E. Washburne, Wells, Wilson, Woodruff-65.

March 2d. The joint resolution passed the Senate, yeas 24, nays 12, as follows:

YEAS-Messrs. Anthony, Baker, Bigler, Bright, Critten-don, Dixon, Douglas, Foster, Grimes, Guin, Harlan, Humter, Johnson of Tennessee, Kennedy, Latham, Mason, Morrill, Nicholson, Polk, Pugh, Rice, Sebastian, Ten Eyck, Thom-

son-24. NAYS-Messrs. Bingham, Chandler, Clark, Doolittle, Dur-kee, Foot, King, Sumner, Trumbull, Wade, Wilkinson, Wilson-12.

So the joint resolution was agreed to, by both Houses.\*

#### VOTE ON THIRD PROPOSITION.

FOR THE ADMISSION OF NEW MEXICO INTO THE UNION.

Whereas, by the act of Congress approved on the 9th of September, in the year 1850, it was provided that the people of New Mexico, when admitted as a State, shall be received into the Union with or without slavery, as their constitution may provide at the time of their admission; and whereas the population of said Territory is now sufficient to constitute a State government : Therefore,

Be it enacted by the Senate and House of Representatives of the United States of America in Congress assembled, That the inhabi-tants of the Territory of New Mexico, including therein the region called Arizona, be, and they are hereby, authorized to form for themselves a constitution of State gov-ernment by the name of the State of New Mexico; and the said State, when formed, shall be admitted into the Union upon the same footing with the original States in all respects whatever. And said constitution shall be formed by a convention of the people of New Mexico, which shall consist of twice the number of members now by law constituting the House of Representatives of the Territory; each representative district shall elect two members to said convention for every member now by law elected in such district to the Territorial House of Representatives; and in such election only those persons shall vote for such delegates as are, by the laws of said Territory now in force,

\* The Legislatures of Ohio and Maryland agreed to the amendment promptly.

entitled to vote for members of the Terri-The electorial House of Representatives. tion for the convention shall be held on the 5th day of August, 1861, by the same officers who would hold an election for members of the said House of Representatives; and those officers shall conform to the law now in force in said Territory for election for members of said House of Representatives in all respects, in holding the election, receiving and rejecting votes, and making the returns of the election for the convention. The convention shall assemble at the city of Santa Fé, on the 2d day of September, 1861, and continue its sessions at that place until its de-liberations shall be closed. The constitution agreed on by the convention shall be submitted to the people of the Territory for their approval or rejection as a whole; at such election on the constitution, all those and others shall be entitled to vote who are now entitled to vote for members of the House of Representatives of said Territory; and such election shall be held by the same officers who conduct, by the present laws, the election for members of the House of Representatives of the Territory, at the same places for voting, and in the same manner in all respects; and such election shall be held on the 4th day of November, 1861, and the returns thereof made to the Governor of the Territory, who shall forthwith sum up and declare the result, and shall send a certificate thereof, together with a copy of the constitution, to the President of the United States. The said State shall be entitled to one member of the House of Representatives of the United States of America, held until the apportionment under the next census.

Mr. HICKMAN moved to lay the bill on the table; which was agreed to, yeas 115, nays 71, as follows:

table; Whilch Was agreeu to, yeas 113, hays
71, as follows:
YEAS-Messrs. Aldrich, Alley, Ashley, Avery, Babbitt, Beale, Bingham, Blair, Blake, Bocock, Boteler, Bonligny, Brabson, Branch, Brayton, Bufinton, Burlingame, Burnett, Burnham, Carey, Carter, Case, Coburn, Colfax, Conway, Covode, Burton Craige, John G. Davis, Dawes, De Jarnetle, Duell, Edgerton, Edmandson, Edwards, Eliot, Ely, Farneworth, Fenton, Ferry, Florence, Foster, Frank, Garnett, Gooch, Graham, Grow, Hale, Haskin, Hatton, Hickman, Hindman, Hoard, William A. Howard, Humphrey, Hutchins, Irvine, Jenkins, Francis W. Kellogg, De Witt C. Leach, Jask M. Leach, Leake, Lee, Longnecker, Loomis, Lovejoy, Maclay, Marston, Elbert S. Martin, Maynard, McKean, Morrill, Morse, Nelson, Niblack, Olin, Palmer, Perry, Phelps, Potter, Pottle, Pryor, Quarles, Edwin R. Gromston, Stores, Spinner, Stevens, William Stewart, Tappan, Thomas, Jompkins, Tunible, Vallandigham, Vance, Vandever, Van Wyck, Wade, Waldron, Walton, Cadwalader C. Washburn, William C. Anderson, Barr, Briggs, Bristow, Brown, Burch, Butterfield, Campbell, Horace P. Clark, John & Clark, Etheridey, Wilson, Yindon, Winslow, Woodruff, Woodson, Wright-115.
NAYS-Messrs, Chas, F. Adams, Green Adams, Adrain, William C. Anderson, Barr, Briggs, Bristow, Brown, Burch, Butterfield, Campbell, Horace P. Clark, John & Clark, Clemens, Clark B. Cochrane, John Cochrane, Conkling, Corvin, Cox, James Craig, H. Winter Davis, Delano, Dimick, Dunn, English, Etheridge, Fouke, Gillmer, J. Moorison, Harris, John T. Harris, Holman, William Howard, Hughes, Junkin, William, Keterson, John Cochrane, Kunket, Larrabee, Logan, Mallory, Chas, D. Martin, McClernand, Keterity, McKningt, McPherson, Millson, Laban T. Moore, Moorhead, Edward Joy Morris, Isaae X. Morris, Nixon, Yoeth, Pendelton, Pettit, Peyton, Portsr, Rice, Riggs, Stekles, Simms, Spandling, Stanton, Stevenson, Jones A. Skwart, Stokes, Stratton, Thayer, Webster, Wood-71.

# FOURTH PROPOSITION.

#### AMENDMENT OF FUGITIVE SLAVE LAW.

Be it enacted by the Senate and House of Representatives of the United States of America in Congress assembled, That every person arrested under the laws of Congress for the delivery up of fugitives from labor shall be produced before a court, judge, or commissioner, mentioned in the law approved the 18th of September, 1850, for the State or Territory wherein the arrest may be made; and upon such production of the person, together with the proofs mentioned in the sixth or the tenth section of the said act, such court, judge, or commissioner, shall proceed to hear and consider the same publicly; and if such court, judge, or commissioner, is of opinion that the person arrested owes labor or service to the claimant according to the laws of any other State, Territory, or District of Columbia, and escaped therefrom, the court, judge, or commissioner shall make out and deliver to the claimant, or his agent, a certificate stating those facts; and if the said fugitive shall, upon the decision of the court, judge, or commissioner being made known to him, aver that he is free, and does not owe service or labor according to the law of the State or Territory to which he is to be returned, such averment shall be entered upon the certificate, and the fugitive shall be delivered by the court, judge, or commissioner to the marshal, to be by him taken and delivered to the marshal of the United States for the State or district from which the fugitive is ascertained to have fled, who shall produce said fugitive before one of the judges of the Circuit Court of the United States for the last mentioned State or district, whose duty it shall be, if said alleged fugitive shall persist in his averment, forthwith, or at the next term of the Circuit Court, to cause a jury to be impanelled and sworn to try the issue whether such fugitive owes labor or service to the person by or on behalf of whom he is claimed, and a true verdict to give according to the evidence, on which trial the fugitive shall be entitled to the aid of counsel and to process for procuring evidence at the cost of the United States; and upon such finding the judge shall render judgment, and cause said fugitive to be delivered to the claimant, or returned to the place where he was arrested, at the expense of the United States, according to the finding of the jury; and if the judge or court be not satisfied with the verdict, he may cause another jury to be impanelled forthwith, whose verdict shall be final. And it shall be the duty of said marshal so delivering said alleged fugitive, to take from the marshal of the State from which said fugitive is alleged to have escaped, a certificate acknowledging that said alleged fugitive had been delivered to him, giving a minute description of said alleged fugitive, which certificate shall be authenticated by the United States district judge, or a commissioner of a United l

States Court for said State from which said fugitive was alleged to have escaped, which certificate shall be filed in the office of the United States District Court for the State or district in which said alleged fugitive was seized, within sixty days from the date of the arrest of said fugitive; and should said marshal fail to comply with the provisions of this act, he shall be deemed guilty of a fine of \$1,000 and imprisoned for six months, and until his said fine is paid.

SEC. 2. And be it further enacted, That no citizen of any State shall be compelled to aid the marshal or owner of any fugitive in the capture or detention of such fugitive, unless when force is employed or reasonably apprehended to prevent such capture or detention, too powerful to be resisted by the marshal or owner; and the fees of the commissioners appointed under the act of 18th of September, 1850, shall be ten dollars for every case heard and determined by such commissioner.

Which was passed, yeas 92, nays 83, as follows:

follows: YEAS--Messrs. Green Adams, Adrain, Aldrich, William C. Anderson, Babbitt, Barr, Blair, Brayton, Briggs, Bristow, Burch, Burlingame, Burnham, Butterfield, Campbell, Carter, Case, Clemens, Coburn, John Cochrane, Collax, Conkling, Corwin, Covode, H. Winter Davis, John G. Davis, Delano, Dimmick, Duun, Edwards, Ely, Ferry, Fouke, French, Gurley, Hale, Hall, J. Morrisou Harris, Hatton, Heimick, Hoard, William Howard, William A. Howard, Humphrey, Junkin, Francis W. Kellogg, William Kellogg, Kenyon, Kilgore, Killinger, Lee, Longnecker, Loomis, Marston, Chas. D. Martin, McClernand, McKean, McKenty, McKnight, McPherson, Millward, Moorhead, Morrill, Edw. Joy Morris, Isaac N. Morris, Nixon, Olin, Palmer, Perry, Petiti, Porter, Pottle, John H. Reynolds, Rice, Riggs, Christopher Robinson, James C. Robinson, Scratton, Sickles, Spaulding, Spinner, Stanton, Stratton, Thayer, Theaker, Tompkins, Train, Trimble, Walton, Windom, Wood, Woodruff-92.

ruff-92. NATS-Messrs. Ashley, Avery, Barrett, Beale, Bingham, Blake, Bocock, Boteler, Bouliguy, Brabson, Branch, Brown, Buffinton, Burnett, Carey, Horace F. Clark, John B. Clark, Conway, Burton Craige, Dawes, Duell, Edgerton, Edmundson, Eliot, Farnsworth, Fenton, Florence, Foster, Frank, Garnett, Gilmer, Gooch, Grow, John T. Harris, Hickman, Hindman, Hughes, Hutchins, Irvine, Jenkins, Kunkel, De Witt C. Leach, James M. Leach, Leake, Lovejoy, Maclay, Mallory, Elbert S. Martin, Maynard, Millson, Laban T. Moore, Nelson, Niblack, Noell, Fendieton, Feyton, Phelps, Potter, Quarles, Reagan, Royce, Rufjin, Sedywick, Simms, Somes, Stevens, Stevenson, Stokes, Tappan, Thomas, Vallandigham, Yance, Van Wyck, Wade, Waldron, Cadvalader C. Washburn, Ellihu B. Washburne, Wells, Whiteley, Wilson, Winslow, Woodson, Wrigli-83.

#### FIFTH PROPOSITION.

# AMENDMENT OF THE ACT FOR THE RENDITION OF FUGITIVES FROM JUSTICE.

Be it enacted by the Senate and House of Representatives of the United States of America in Congress assembled, That every person charged, by indictment or other satisfactory evidence, in any State, with treason felony, or other crime, committed within the jurisdiction of such State, who shall flee or shall have fled from justice and be found in another State, shall, on demand of the executive authority of the State from which he fled upon the district judge of the United States of the district in which he may be found, be arrested and brought before such judge, who, on being satisfied that he is the person charged, and that he was within the

jurisdiction of such State at the time such crime was committed, of which such charge shall be prima facie evidence, shall deliver him up to be removed to the State having jurisdiction of the crime; and if any question of law shall arise during such examination, it may be taken on exceptions by writ of error to the Circuit Court.

Which was rejected, yeas 48, nays 125, as follows:

YEAS-Messrs. Green Adams, Adrain, William C. Ander-son, Barr, Barrett, Bouligny, Brahson, Briggs, Bristow, Burch, John B. Clark, Clemens, John Coctrane, Corwin, Coz, H. Winter Davis, John G. Davis, Etheridge, Foulke, Gilmer, Matten

son, Barr, Barrett, Bouligny, Brahson, Briggs, Bristow, Burch, John B. Clark, Clemens, John Coolrane. Corwin, Cox, H. Winter Davis, John G. Davis, Etheridge, Fouke, Gilmer, Hamilton, J. Morrison Harris, John T. Harris, Hatton, Holman, William Howard, Hughes, Larrabee, James M. Leach, Logan, Mallary, Charles D. Martin, Maynard, Mc-Clernand, McKenty, Millson, Laban T. Moore, Moorhead, Nelson, Nixon, Phelps, Riggs, James C. Robinson, Scranton, Stöckes, Stokes, Webster, Wood-48.
 NArs-Messrs. Charles F. Adams, Alley, Ashley, Avery, Babbitt, Beale, Bingham, Blair, Blake, Bocock, Boteler, Burnham, Butterfield, Campbell, Carey, Carter, Case, Hor-ace F. Clark, Coburn, Colfax, Conkling, Conway, Covode, Burton Craige, Dawes, De Jarnette, Delano, Duell, Dunn, Edgerton, Edwards, Eliot, Ely, Farnsworth, Fenton, Ferry, Floronce, Foster, Frank, French, Garnett, Gooch, Grahann, Grow, Hale, Hall, Helmick, Hickman, Hundman, Hoard, William A. Howard, Humphrey, Hutchins, Irvine, Jenkins, F. W. Kellogg, Kenyon, Kilgore, Kunkel, Leake, Long-necker, Loonis, Lovejoy, Marston, Elbert S. Martin, Morkan, Kean, McKnight, McPherson, Morrill, Edward Joy Morris, Isaaca N. Morris, Morse, Palamer, Pendleton, Peyton, Porter, Potter, Pottle, Proye, Quarles, Edwin R. Reynolds, John H. Reynolds, Christopher Robinson, Royce, Kuffin, Sedg-wick, Sherman, Simms, William H. N. Smith, Somes, Spin-er, Stanton, Stevenson, James A. Stewart, William Stewart, Stratton, Tappan, Theaker, Thomas, Tompkins, Train, Trimble, Yallendigham, Yanee, Yandever, Yan Wyck, Wade, Waldron, Walton, Cadwaiader C. Washburn, Ellihu B. Washburne, Wells, Whiteley, Wilson, Windom, Winslow, Woodruff, Woodson-215.

Neither of these propositions was considered in the Senate, except the proposed Constitutional amendment, which was passed. For vote, see page 60.

Pending this report in the House of Representatives,

#### A CONVENTION.

February 27th. Mr. BURCH moved to add to the declaratory resolutions, one recommending to the several States that they, through their legislatures, request Congress to call a Convention of all the States, to amend it "in such manner with regard to such subjects as will more adequately respond to the wants, and afford more sufficient guarantees to the diversified and growing interests of the Government and of the people composing the same; which was rejected, yeas 74, nays 109, as follows:

rejected, yeas 74, nays 109, as follows: YEAS-Messrs. Green Adams, Carnett B. Adrain, W. C. Anderson, Babbitt, Barr, Bcteler, Brabson, Briggs, Bris-tow, Burch, Burnham, Campbell, Coburn, Clark B. Coch-rane, John Cochrane, Colfax, Cox, Curtis, Duell, Etheridge, Ferry, Fouke, Gilmer, Hall, J. Morrison Harris, John T. Harris, Hatton, Helmick, Hoard, Hotman, Wm. Howard, Hughes, Hunphrey, Junkin, Wm. Keilogg, Kenyon, Kil-Hinger, Larrabee, J. M. Leach, Logan, Loomis, Mallory, Chas. D. Martin, Maynard, McClernand, McKenty, McPher-son, Milson, Monlgomery, Moore, Edward Joy Morris, Isaac N. Morris, Nixon, Noell, Palmer, Porter, Quarles, John H. Reynolds, Riggs, J. C. Robinson, Scranton, Sedgwick, Stan-ton, James A. Stewart, Vm. Stowart, Stokes, Stout, Strat-ton, Thayer, Waldron, Webster, Wells, Wood, Woodruff-74. Nars-Messrs. Chas. F. Adams, Aldrich, Alley, T. L. Ander-ton, Ashley, Avery, Barrett, Beale, Bingham, Blair, Blake, Bozock, Branch, Brayton, Brown, Bufinton, Buringame, Burnett, Butterfield, Carey, Carter, Case, John B. Clark, Cor-win, Covode, Jas. Craig, Burton Craige, H. Winter Davis, J. G. Davis, Dawes, De Jarnette, Delano, Dimmick, Dunn, Eger-

ton, Edmundson, Edwards, Eliot, Ely, English, Farnsworth, Fenton, Florence, Frank, French, Garnett, Gooch, Graham, Haskin, Hickman, Hindmon, W. A. Howard, Hutchins Irvine, Jenkins, F. W. Kellogg, Kilgore, Kunkel, D. C. Leach, Leake, Lee, Longnecker, Lovejoy, Marston, Elbert S. Martin, McKean, McKnight, Moorhead, Morrill, Morse, Nelson, Niblack, Olin, Perry, Pettit, Peyton, Phelps, Potter, Pottle, Pyor, F. R. Reynolds, Rice, C. Robinson, Ruffin, Sickles, Stmms, W. H. N. Smith, Somes, Spaulding, Spinner, Stevens, Stemeson, Tappan, Thomas, Tompkius, Train, Trimble, Vallandigham, Vance, Wade, Walton, C. C. Wash-burn, E. B. Washburne, Whiteley, Wilson, Windom, Wind-low, Woodson, Wright-109.

Same day. Mr. KILGORE moved to lay the whole subject on the table; which was rejected-yeas 14, navs 179. The yeas were :

Messrs. Alley, Beale, Buffinton, Carey, Eliot, Farns-worth, Grow, Kilgore, Potter, Sedgwick, Somes, Waldron, Cadwalader C. Washburn, Windom-14.

The House came to a vote Same day. on the following substitute for it, offered by Mr.WM. KELLOGG, of Illinois :

Strike out all after the word "that," and insert:

The following articles be, and are hereby, proposed and submitted as amendments to the Constitution of the United States, which shall be valid, to all intents and purposes, as part of said Constitution, when ratified by conventions of three-fourths of the several States.

ART. 13. That in all the territory now held by the United States situate north of latitude 36° 30', involuntary servitude, except in the punishment for crime, is prohibited while such territory shall remain under a territorial government; that in all the territory now held south of said line, neither Congress nor any Territorial Legislature shall hinder or prevent the emigration to said ter-ritory of persons held to service from any State of this Union, when that relation exists by virtue of any law or usage of such State, while it shall remain in a territorial condition; and when any territory north or south of said line, within such boundaries as Congress may prescribe, shall contain the population requisite for a member of Con-gress, according to the then Federal ratio of representation of the people of the United States, it may, if its form of government be republican, be admitted into the Union on an equal footing with the original States, with or without the relation of persons held to service or labor, as the constitution of such new State may provide.

ART. 14. That nothing in the Constitution of the United States, or any amendment thereto, shall be so construed as to authorize any department of the Government to, in any manner, interfere with the relation of persons held to service in any State where that relation exists, nor in any manner to establish or sustain that relation in any State where it is prohibited by the laws or constitution of such State. And that this article shall not be altered or amended without the consent of every State in the Union.

ART. 15. The third paragraph of the second section of the fourth article of the Constitution shall be taken and construed

1

to authorize and empower Congress to pass laws necessary to secure the return of persons held to service or labor under the laws of any State, who may have escaped therefrom, to the party to whom such service or labor may be due.

ART. 16. The migration or importation of persons held to service or involuntary servitude, into any State, Territory, or place within the United States, from any place or country beyond the limits of the United States or Territories thereof, is forever prohibited.

ART. 17. No territory beyond the present limits of the United States and the Territories thereof shall be annexed to, or acquired by the United States, unless by treaty, which treaty shall be ratified by a vote of twothirds of the Senate.

This was disagreed to-yeas 33, nays 158.

The YEAS were-Messrs. Adrain, Barr, Briggs, Burch, Horace P. Clark, John Cochrane, Caz, John G. Davis, Eng-kish, Etheridge, Fouke, Gilmer, J. Morrison Harris, Hoiman, William Howard, Junkin, William Kellogz, Larrabee, Logan, Charles D. Martin, McClernand, McKerdy, Mont-gomery, Isaac N. Morris, Nelson, Niblack, Riggs, James C. Robinson, Sickles, Stokes, Stout, Vallandigham, Webster -33.

# MR. CLEMENS'S RESOLUTION.

Mr. SHERRARD CLEMENS of Virginia, then offered a substitute for it, being Mr. Crittenden's proposition as amended on motion of Mr. Powell, with these additions :

ART. 7. Section 1. The elective franchise and the right to hold office, whether Federal, State, Territorial, or Municipal, shall not be exercised by persons who are, in whole or in part, of the African race.

Section 2. The United States shall have power to acquire, from time to time, districts of country in Africa and South America, for the colonization, at the expense of the Federal Treasury, of such free negroes and mulattoes as the several States may wish to have removed from their limits, and from the District of Columbia, and such other places as may be under the jurisdiction of Congress. And the substitution of the words : "the Southern boundary of Kansas and the Northern boundary of New Mexico," for the words: "latitude 36° 30'," in the first sentence of Article 1.

Which was negatived, yeas 80, nays 113, as follows:

as follows: Yaas-Messrs. Adrain, William C. Anderson, Avery, Barr, Barrett, Boook, Boteler, Bouligny, Brabson, Branch, Briggs, Bristow, Brown, Burch, Burnett, Horace F. Clark, John B. Clark, John Cochrane, Cox, James Craig, Burton Oraige, John G. Davis, De Jarnette, Dimmick, Edmundson, English, Florence, Fouke, Garnett, Gilmer, Hamilton, J. Morrison Harris, John T. Harris, Ilutton, Holman, Wm. Howard, Hughes, Jenkins, Kimkel, Larrabee, Janese M. Leach, Leake, Logan, Maclay, Mallory, Charles D. Martin, Kibert S. Martin, Maynard, McClernard, Mc Kenty, Milson, Montgomery, Laban T. Moore, Isaac N. Morris, Nelson, Niblack, Noell, Peyton, Phelps, Pryor, Quarles, Riogs, Jas C Robinson, Rust, Sickles, Simms, William Smith, William N. H. Smith, Stevenson, James A. Stewart, Stokes, Stoat, Homson, Vallandigham, Yance, Webster, Whiteley, Winslow, Woodson, and Wright—SO. MATS-Messrs. Charles F. Adams, Aldrich, Alley, Ashley, pabbit, Beale, Binghan, Blair, Blake, Brayton, Buffinton, Burlingame, Buruham, Butterfield, Campbell, Carey, Car-ter, Case, Coburn, Clark B. Cochrane, Colfax, Conkling,

Conway, Corwin, Covode, H. Winter Davis, Dawes, Delano, Duell, Dunn, Edgerton, Edwards, Eliot, Ely, Etheridge, Farnsworth, Fenton, Ferry, Foster, Frank, French, Gooch, Graham, Grow, Hale, Hall, Helmick, Hickman, Hürdman, Hoard, William A. Howard, Humphrey, Ilntchias, Irvine, Junkin, Francis W. Kellogg, William Kellogg, Kenyon, Kilgore, Killinger, De Witt C. Leach, Loe, Longnecker, Loomis, Lovejoy, Marston, McKean, McKnight, McPherson, Moorhead, Morrill, Morse, Nixon, Olin, Palmer, Perry, Petti, Porter, Potter, Pottle, Edwin R. Reynolds, Kieç Christopher Robinson, Royce, Scranton, Stevens, William Stewart, Stratton, Tappan, Thayer, Theaker, Tompking Stewart, Stratton, Tappan, Thayer, Heaker, Wal-dron, Walton, Cadwalader C. Washburn, Ellihu B. Wash-burne, Wells, Wilson, Wiodon, and Woodruff-113.

# Votes in the Senate.

During the pendency, in the Senate, of the Constitutional amendment reported from the House Committee of Thirty-three, and adopted by the House, Mr. Pugh, of Ohio, offered a substitute, which was the same as Mr. Crittenden's\* (as amended by Mr. Powell) with the omission of the preamble and four resolutions, and the addition of the following at the close of Article 4:

But the African slave trade shall be forever suppressed, and it shall be the duty of Congress to make such laws as shall be necessary and effectual, to prevent the migration and importation of slaves, or persons owing service or labor into the United States  $\cdot$ from any foreign country, place, or jurisdiction whatever.

Section 2. That persons committing crimes against the rights of those who hold persons to service or labor in one State, and fleeing to another, shall be delivered up in the same manner as persons committing other crimes; and that the laws of the States from which such persons flee shall be the test of criminality.

Section 3. Congress shall pass efficient laws for the punishment of all persons in any of the States, who shall in any manner aid and abet invasion or insurrection in any other State, or commit any other act tending to disturb the tranquillity of its people, or government of any other State.

And the insertion of a new article:

ART. 7. Section 1. The elective franchise and the right to hold office, whether Federal, State, Territorial, or Municipal, shall not be exercised by persons who are, in whole or in part, of the African race.

MR. DOOLITTLE'S ON THE RIGHT OF SECESSION.

Mr. Doolittle, of Wisconsin, offered as a substitute for the above the following:

"Under this Constitution, as originally adopted, and as it now exists, no State has power to withdraw from the jurisdiction of the United States; but this Constitution, and all laws passed in pursuance of its delegated powers, are the supreme law of the land, any thing contained in any constitution, ordinance, or act of any State, to the

contrary notwithstanding." March 2, 1861. This was rejected—yeas 18, nays 28, as follows :

\* See page 61.

. 1

YEAS-Messrs. Bingham, Chandler, Clark, Collamer, Doolittle, Durkee, Fessenden, Foot, Grimes, Hale, Harlan, King, Morrill, Simmons, Trumbull, Wade, Wilkinson, and Wilson-18.

Wilson-15. NATS-Messrs. Anthony, Baker, Bayard, Bigler, Bragg, Bright, Cameron, Clingman, Crittenden, Dixon, Douglas, Foster, Gwin, Hemphill, Hunter, Johnson of Tennessee, Kennedy, Lanc, Latham, Mason, Nicholson, Tearce, Powell, Pugh, Rice, Sebastian, Ten Eyck, and Wigfall-23.

#### VOTE ON MR. PUGH'S.

The substitute of Mr. PUGH was then rejected—yeas 14, nays 25, as follows :

YEAS-Messrs. Bayard, Bright, Gwin, Hunter, Johnson of Arkansas, Kennedy, Lane, Muson, Nicholson, Polk, Powell, Pugh, Thomson, and Wigfall-14. NAYS-Messrs. Anthony, Baker, Bigler, Bingham, Chand-ler, Clark, Crittenden, Dixon, Doolittle, Douglas, Fessénden, Foot, Foster, Grimes, Harlan, Johnson of Tennessee, King, Latham, Morrill, Rice, Schastian, Sumner, Teu Eyck, Wil-kinson, and Wilson-25.

#### MR. BINGHAM'S PROPOSITION.

Mr. BINGHAM, of Michigan, offered the following substitute, which was rejected-yeas 13, nays 24, as follows:

"That the provisions of the Constitution are ample for the preservation of the Union, and the protection of all the material interests of the country: that it needs to be obeyed rather than amended; and that an extrication from our present dangers is to be looked for in strenuous efforts to preserve the peace, protect the public property, and enforce the laws, rather than in new guarantees for particular interests, compromises for particular difficulties, or concessions to unreasonable demands.

"Resolved, That all attempts to dissolve the present Union, or overthrow or abandon the present Constitution, with the hope or expectation of constructing a new one, are dangerous, illusory, and destructive ; that in the opinion of the Senate of the United States no such reconstruction is practicable; and therefore, to the maintenance of the existing Union and Constitution should be directed all the energies of all the departments of the Government, and the efforts of all good citizens.

YEAS-Messrs. Bingham, Chandler, Clark, Doolittle, Dur-kee, Fessenden, Foot, King, Sumner, Trumbull, Wade, Wil-kinson, and Wilson-13.

MINSON, And WINSON-10. NATS.-WRESTR. Anthony, Baker, Bigler, Bright, Crit-tenden, Dixon, Douglas, Foster, Gwin, Harlan, Hunter, Johnson of Arkansas, Johnson of Tennessee, Kennedy, Lane, Latham, Mason, Nicholson, Polk, Pugh, Rice, Sebastian, Ten Eyck, and Thomson-24.

#### A CONVENTION.

Mr. GRIMES, of Iowa, offered the following substitute:

"The Legislatures of the States of Kentucky, New Jersey and Illinois, have applied to Congress to call a Convention for proposing amendments to the Constitution of the United States : therefore,

"Be it Resolved by the Senate and House of Representatives of the United States of America in Congress assembled, That the Legislatures of the other States be invited to take the subject of such a Convention into consideration, and to express their will on that subject to Congress, in pursuance of the fifth article of the Constitution."

Which was rejected-yeas 14, nays 25, as follows:

YEAS-Messrs. Bingham, Chandler, Clark, Doolittle, Fes-senden, Foot, Grimes, King, Morrill, Pugh, Summer, Trum-bull, Wilkinson, and Wilson-14. NATS-Messrs. Anthony, Baker, Bright, Crittenden, Dison, Douglas, Durkee, Foster, Gwin, Ilarlan, Hunter, Johnson of Arkansas, Johnson of Tennessee, Kennedy, Lane, Lather Micron Nichelen Bulk Fois Scheriber Ten Freib Latham, Mason, Nicholson, Polk, Rice, Sebastian, Ten Eyck, Thomson, and Wade-24.

# THE PEACE CONFERENCE.

The Peace Conference propositions offered by Mr. Johnson, of Arkansas, were then considered as a substitute, and rejected-yeas 3, nays 34.

The joint resolution of the House of Representatives was then passed.

Nore.—During the above votes, several Senators said in explanation of their negative votes upon propositions, that they had dctermined to vote against all substitutes for the House proposition, and for it, believing it to be the only measure practicable at so late a period in the session. Mr. Douglas, of Illinois, and Mr. TEN EYCK, and others, made this statement.

# Vote on the Crittenden Resolutions, January 16, 1861.

A JOINT RESOLUTION (S. NO. 50) PROPOSING CERTAIN AMEND-MENTS TO THE CONSTITUTION OF THE UNITED STATES.

WHEREAS, serious and alarming dissensions have arisen between the Northern and Southern States, concerning the rights and security of the rights of the slaveholding States, and especially their rights in the common territory of the United States; and whereas it is eminently desirable and the tradition of the states are specified as the security of proper that these dissensions, which now threaten the very existence of this Union, should be permanently quieted and settled by constitutional provisions, which shall do equal justice to all sections, and thereby restore to the people that peace and good-will which ought to prevail between all the citizens of the United States: Therefore,

Resclosed by the Serate and House of Representatives of the United States of America in Congress assembled (two-thirds of both Houses concurring), That the following arti-cles be, and are hereby, proposed and submitted as amend-ments to the Constitution of the United States, which shall be valid to all intents and purposes, as part of said Constitution, when ratified by conventions of three-fourths of the several States

ARTICLE 1. In all the territory of the United States now held, or hereafter acquired, situate north of latitude 36° 30' slavery or involuntary servitude, except as a punishment shavery or involution y service, except as a pullabilities for crime, is prohibited while such territory shall remain under territorial government. In all the territory sonth of said line of latitude, slavery of the African race is hereby recognized as existing, and shall not be interfered with by Congress, but shall be protected as property by all the departments of the territorial government during all the departments of the territorial government during its continuance. And when any Territory, north or south of said line, within such boundaries as Congress may pre-scribe, shall contaiu the copulation requisite for a member of Congress according to the then Federal ratio of repre-sentation of the people of the United States, it shall, if its form of government be republican, be admitted into the Union, on an equal footing with the original States, with or without slavery, as the constitution of such new State may provide.

May provide. ART. 2. Congress shall have no power to abolish slavery in places under its exclusive jurisdiction, and situate within the limits of States that permit the holding of slaves.

ART. 3. Congress shall have no power to abolish slavery within the District of Columbia, so long as it exists in the adjoining States of Virginia and Maryland, or either, nor without the consent of the inhabitants, nor withont just compensation first made to such owners of slaves as do not consent to such abolishment. Nor shall Congress at any time prohibit officers of the Federal Government, or mem bers of Congress, whose duties require them to be in said District, from bringing with them their slaves, and holding them as such during the time their duties may require them to remain there, and afterwards taking them from the District

ART. 4. Congress shall have no power to prohibit or

hinder the transportation of slaves from one State to an-other, or to a Territory in which slaves are by law per-mitted to be held, whether that transportation be by land, particable slaves or how the sec

mitted to be held, whether that transportation be by land, navigable rivers, or by the sea. Arr. 5. That in addition to the provisions of the third paragraph of the second section of the fourth article of the Constitution of the United States, Congress shall have power to provide by law, and it shall be its duty so to pro-vide, that the United States shall pay to the owner who shall apply for it, the full value of his fugitive slave in all cases when the marshal or other officer whose duty it was to arrest said fugitive was prepended from so doing by vio. to arrest said fugitive was prevented from so doing by vio-lence or intimidation, or when, after arrest, said fugitive was rescued by force, and the owner thereby prevented and obstructed in the pursuit of his remedy for the recovery of his fugitive slave under the said clause of the Constitution his fngifive slave under the said clause of the Constitution and the laws made in pursuance thereof. And in all such cases, when the United States shall pay for such fugitive, they shall have the right, in their own name, to sue the connty in which said violence, intimidation, or rescue was committed, and to recover from it, with interest and dam-ages, the amount paid by them for said fngitive slave. And the said acoust activity has maid said amount to the United ages, the amount part of them for said ingitive stave. And the said county, after it has paid said amount to the United States, may, for its indemnity, sue and recover from the wrong-doers or rescuers by whom the owner was prevented from the recovery of his fugitive slave, in like manner as the owner himself might have sued and recovered.

ART. 6. No future amendment of the Constitution shall ART. 6. No future amendment of the Constitution shall affect the five preceding articles; nor the third paragraph of the second section of the first article of the Constitution; nor the third paragraph of the second section of the fourth article of said Constitution; and no amendment shall be made to the Constitution which shall authorize or give to Congress any power to abolish or interfere with slavery in any of the States by whose laws it is, or may be, allowed or permitted. And whereas also hesides these causes of discension em.

And whereas, also, besides these canses of dissension embraced in the foregoing amendments proposed to the Con-stitution of the United States, there are others which come within the jurisdiction of Congress, and may be remedied by its legislative power; and whereas it is the desire of Congress, as far as its power will extend to remove all just cause for the popular discontent and agitation which now disturb the peace of the country, and threaten the stability of its institutions: Therefore,

3. Resolved by the Senate and House of Representatives of the United States of America in Congress assembled. That the laws uow in force for the recovery of fuglive slaves are in strict pursuance of the plain and mandatory provisions of the Constitution, and have been sanctioned as valid and constitutional by the judgment of the Supreme Conrt of the United States; that the slaveholding States are entitled to the faithful observance and execution of those laws, and that they ought not to be repealed, or so modi-fied or changed as to impair their efficiency; and that laws ought to be made for the punishment of those who attempt by rescue of the slave, or other illegal means, to hinder or defeat the due execution of said laws.

hinder or defeat the due execution of said laws. 2. That all State laws which conflict with the fugitive slave acts of Congress, or any other constitutional acts of Congress, or which, in their operation, impede, hinder, or delay the free course and due execution of any of said acts, are null and void by the plain provisions of the Constitu-tion of the United States; yet those State laws, void as they are, have given color to practices, and led to conse-quences which have obstracted the due administration and execution of acts of Congress, and especially the acts for the delivery of fugitive slaves, and have thereby con-tributed much to the discord and commotion now nevailfor the delivery of fugitive slaves, and have thereby con-tributed much to the discord and commotion now prevail-ing. Congress, thereforce, in the present perilous juncture, does not deem it improper, respectfully and earnestly to recommend the repeal of those laws to the several States which have enacted them, or such legislative corrections or explanations of them as may prevent their being used or perverted to such mischierous purposes. 3. That the act of the 1Sth of September, 1850, commonly called the fugitive slave law, ought to be so amended as to make the fee of the commissioner, mentioned in the eighth section of the act, equal in amount, in the cases decided by him, whether his decision be in favor of or against the elaimant. And to avoid misconstruction, the last clause

him, whether his decision he in favor of or against the claimant. And to avoid misconstruction, the last clause of the fifth section of said act, which authorizes the person holding a warrant for the arrest or detention of a fugitive slave, to summon to his aid the *posse comitatus*, and which declares it to be the duty of all good citizens to assist him it is execution, ought to be so amended as to expressly limit the authority and duty to cases in which there shall be resistance or danger of resistance or rescue. 4. That the laws for the suppression of the African slave-trade and especially those prohibiting the importation of

Trade, and especially those prohibiting the importation of slaves in the Unit'd States, ought to be made effectual, and ought to be thoroughly executed; and all further enact-ments necessary to those ends ought to be promptly made.

Mr. Powell moved to amend by insert-5

ing in article 1, after the word "territory," in the second sentence, the words "now held or hereafter to be acquired," so that the clause will read : "In all the territory now held or hereafter to be acquired south of said line of latitude, slavery of the African race is hereby recognized, etc.," which was agreed to as follows :

YEAS--Messrs. Baker, Bayard, Benjamin, Bigler, Bragg, Bright, Clingman, Crittenden, Douglas, Filch, Green, Guin, Hemphill, Hunter, Iverson, Johnson of Tennessee, Kennedy, Lane, Mason, Nicholson, Pearce, Polk, Powell, Pugh, Rice, Saulsbury, Schastian, Slüdell, and Wigfall-29. NAYS-Messrs. Atthony, Bingham, Cameron, Chandler, Clark, Collamer, Dixon, Doolittle, Durkee, Fessenden, Foot, Foster, Grimes, Hale, Harlan, King, Latham, Seward, Sim-mons, Sumner, Ten Eyck, Trumbull, Wade, and Wilson-24.

MR. CLARK of New Hampshire offered an amendment to strike out the preamble, and all the resolutions after the word "resolved, and insert : "That the provisions of the Constitution are ample for the preservation of the Union, and the protection of all the material interests of the country; that it needs to be obeyed rather than amended; and that an extrication from our present danger is to be looked for in strenuous efforts to preserve the peace, protect the public property and enforce the laws, rather than in new guarantees for particular interests, compromises for particular difficulties, or concessions to unreasonable demands.

"Resolved, That all attempts to dissolve the present Union, or overthrow or abandon the present Constitution with the hope or expectation of constructing a new one, are dangerous, illusory and destructive; that in the opinion of the Senate of the United States no such reconstruction is practicable; and therefore, to the maintenance of the existing Union and Constitution should be directed all the energies of all the departments of the Government, and the efforts of all good citizens.'

Which was agreed to :

YEAS-Messrs. Anthony, Baker, Bingham, Cameron, Chandler, Clark, Collamer, Dixon, Doollitle, Durkee, Fes-senden, Foot, Foster, Grimes, Hale, Harlau, King, Seward, Sinmonis, Summer, Ten Eyck, Trumbull, Wade, Wilkinson,

Simmons, Summer, 101 Lyta, August, Arg, Bright, Clingman, and Wilson-25. NAYS-Messrs, Bayard, Bigler, Bragg, Bright, Clingman, Crittenden, Fitch, Green, Gwin, Hunter, Johnson of Tennes-eee, Kennedy, Lane, Latham, Mason, Nicholson, Pearce, Polk, Powell, Pugh, Rice, Saulsbury, and Schastian-23.

Messrs. BENJAMIN of Louisiana, Douglas of Illinois, HEMPHILL of Texas, IVERSON of Georgia, JOHNSON of Arkansas, SLIDELL of Louisiana, and WIGFALL of Texas, who voted on the next preceding question, did not vote on this. MR. DOUGLAS stated afterwards in open Senate that he was accidentally absent in one of the retiring rooms, and asked to record his vote, but was refused. permission.

January 18th, 1861.-MR. CAMERON'S motion to reconsider the vote adopting Mr. Clark's amendment, was agreed to :

YEAS-Messrs. Bayard, Bipler, Bragg, Bright, Chingman, Crittenden, Douglaz, Fitch, Green, Gwin, Hemphill, Hun-ter, Johnson of Arkansus, Johnson of Tennessee, Kennedy, Lane, Latham, Mason, Nicholson, Pearce, Folk, Powell, Pugh, Rice, Saulsbury, Sebastian, and Stidell-27.

NATS-Mesars. Anthony, Baker, bingham, Cameron, Chandler, Clark, Collaner, Dixon, Doolittle, Fessenden, Foot, Foster, Grimes, Hale, Harlen, King, Seward, Sin-mons, Sunner, Ten Eyck, Wade, Wigfall, Wilkinson, and Wilson-24.

March 2d. After the adoption of the House resolution proposing an amendment to the Constitution relative to slavery in the States, the Senate returned to the consideration of the Crittenden proposition,-the question being on the amendment offered by Mr. CLARK of New Hampshire, once adopted and then reconsidered, for which, see above vote, which was rejected-yeas 14, nays 22, as follows:

YFAS-Messrs, Bingham, Chandler, Clark, Doolittle, Purkee, Pessenden, Foot, Harlan, King, Morrill, Sumner, Trumbult, Wade, and Wilkinson.—14. NArs-Messes, Anthony, Baker, Bayard, Bigler, Bright, Crittenden, Dixon, Douglas, Foster, Gwin, Houter, Johnson of Tennessee, Kennedy, Lane, Lathum, Muson, Nicholson, Pull, Punk, River Schwigtum, and Ten Fyrk-202 Polk, Pugh, Rice, Schastian, and Ten Eyck-22.

The Senate adopted, without a division, the amendments offered by Mr. Powell in the Committee of the whole, and agreed to as follows: Insert after "territory," in second sentence in first article, the words "now held or to be hereafter acquired," and add to article fourth the words "But the African slave trade shall be forever suppressed; and it shall be the duty of Congress to make such laws as shall be necessary and effectual to prevent the migration or importation of slaves, or persons owing service or labor, into the United States from any foreign country, place, or jurisdiction whatever. SEC. 2. That persons committing crimes

against the rights of those who hold persons to service or labor in one State, and fleeing to another, shall be delivered up in the same manner as persons committing other crimes; and the laws of the State from which such persons flee shall be the test of criminality.

SEC. 3. Congress shall pass efficient laws for the punishment of all persons in any of the States who shall in any manner aid and abet invasion or insurrection in any other State, or commit any other act tending to disturb the tranquillity of its people, or government of any other State."

Mr. CRITTENDEN moved to substitute for his proposition that recommended by the Peace Congress, which was rejected—yeas 7, nays 28. (For vote, see p. 69.)

The Crittenden proposition, as amended, was then rejected-yeas 19, nays 20, as follows:

YEAS--Messrs. Bayard, Bigler, Bright, Crittenden, Douglas, Gwin, Hunter, Johnson of Tennessee, Kennedy, Lane, Lutham, Muson, Nicholson, Polk, Pugle, Rice, Sebas-tian, Theomson, and Wigfall-19. Navs--Messrs. Anthony, Bingham, Chandler, Clark, Dixon, Doolittle, Durkee, Fessenden, Foot, Foster, Grimes, Harlan, King, Morrill, Sumner, Ten Eyek, Trumbull, Wade, Wilkinson, and Wilson-20.

It was not voted upon in the House of Representatives, except as far as it was embodied in the proposition offered by Mr. Clemens of Virginia-for which, see p. 63.

Respecting the vote of January 16th, on Respecting the vote of January 16th, on the Crittenden proposition in the Senate, 17th, page 39.

ANDREW JOHNSON, Senator from Tennessec, in his speech on the expulsion of Jesse D. Bright, Senator from Indiana. delivered Jan-When uary 31st, 1862, made these remarks. the six Senators refused to vote on Senator Clark's amendment,\* Senator Johnson says:

"I sat right behind Mr. Benjamin, and I am not sure that my worthy friend, (Mr. LATHAM,) was not close by, when he refused to vote, and I said to him. 'Mr. Benjamin, why do you not vote? Why not save this proposition, and see if we cannot bring the country to it?' He gave me rather an abrupt answer, and said he would control his own action without consulting me or anybody else. Said 1, 'Vote and show your-self an honest man.' As soon as the vote was taken, he and others telegraphed South, 'We cannot get any compromise.' Here were six Southern men refusing to vote, when the amendment would have been rejected by four majority if they had voted. Who. then, has brought these evils on the country ? Was it Mr. CLARK? He was acting out his own policy; but with the help we had from the other side of the chamber, if all those on this side had been true to the Constitution, and faithful to their constituents, and had acted with fidelity to the country, the amendment of the Senator from New Hampshire could have been voted down, the defeat of which the Senator from Delaware says would have saved the country. Whose fault was it? Who is responsible for it? I think that it is not only getting the nail through, but clinching it on the other side, and the whole staple commodity is taken out of the speech. Who did it? Southern traitors, as was said in the speech of the Senator from California. They did it. They wanted no compromise. They accomplished their object by withholding their votes; and hence the country has been involved in the present difficulty. Let me read another extract from the speech of the Senator from California, [Mr. LATHAM] :

"'I recollect full well the joy that pervaded the faces of some of those gentlemen at the result, and the sorrow manifested by the venerable Senator from Kentucky, [MR. CRITTENDEN.] The record shows that Mr. Pugh, from Ohio, despairing of any com-promise between the extremes of ultra Republicanism and disunionists, working manifestly for the same end, moved, immediately after the vote was announced, to lay the whole subject on the table. If you will turn to page 443, same volume, you will find, when at a late period, Mr. Cameron, from Pennsylvania, moved to reconsider the vote, appeals having been made to sustain those who were struggling to preserve the peace of the country, that vote was reconsidered; and when, at last, the Crittenden propositions were submitted on the 2d day of March, these Southern States having nearly all se-

ceded, they were then lost by but one vote. Here is the vote :

YEAS-Messrs, Bayard, Bigler, Bright, Crittenden, Doug-las, Gwin, Hunter, Johnson of Tennessee, Kennedy, Lane, Latham, Mason, Nicholsou, Polk, Pugh, Rice, Sebastian, Thomson, and Wigfall-10. NATS-Messrs, Anthony, Bingham, Chandler, Clark, Dixon, Doolittle, Durkee, Fessenden, Foot, Foster, Grimes, Harlan, King, Morrill, Sunner, Ten Eyck, Trumbull, Wade, Wilkinson, and Wilson-20.

"'If these seceded Southern States had remained, there would have passed, by a large vote, (as it did without them.) an amendment, by a two-third vote, forbidding Congress ever interfering with slavery in The Crittenden proposition the States. would have been indorsed by a majority vote, the subject finally going before the people, who have never yet, after consideration, refused justice, for any length of time, to any portion of the country.

"'I believe more, Mr. President, that these gentlemen were acting in pursuance of a settled and fixed plan to break up and destroy the Government.'

"When we had it in our power to vote down the amendment of the Senator from New Hampshire, and adopt the Crittenden resolutions, certain Southern Senators prevented it; and yet, even at a late day of the session, after they had seceded, the Crittenden proposition was only lost by one vote. If rebellion and bloodshed and murder have followed, to whose skirts does the responsibility attach? I summed up all these facts myself in a speech during the last session, but I have preferred to read from the speech of the Senator from California, he being better authority, and having presented the facts better than I could."

# Mr. Lincoln's Opinions on a Compromise.

From the N. Y. Tribune of Jan. 30th, 1861. "We do not hesitate to say that these statements are false and calumnious. We have the best authority for saying that Mr. Lincoln is opposed to all concessions of the We know that his views are fully exsort. pressed in his own language as follows :

"I will suffer death before I will consent or advise my friends to consent to any con-cession or compromise which looks like buying the privilege of taking possession of the Government to which we have a constitutional right; because, whatever I might think of the merit of the various propositions before Congress, I should regard any concession in the face of menace as the destruction of the government itself, and a consent on all hands that our system shall be brought down to a level with the existing disorganized state of affairs in Mexico. But this thing will hereafter be, as it is now, in the hands of the people; and if they desire to call a convention to remove any grievances complained of, or to give new guarantees for the permanence of vested rights, it is not mine to oppose.' "

# The "Peace Conference" and its Proposition.

Commissioners composing the "Peace Conference," held at the request of the Legislature of Virginia. met in Washington. February 4th, 1861, and adjourned February The following gentlemen represented 27th. their States, under appointment of the Governors or Legislatures thereof:

# DELEGATES.

Maine.-William P. Fessenden, Lot M. Morrill, Daniel E. Somes, John J. Perry. Ezra B. French, Freeman H. Morse, Stephen Coburn, and Stephen C. Foster.

Hampshire. — Amos Tuck, Levi New Chamberlain, and Asa Fowler.

Vermont.-Hiland Hall, Levi Underwood. H. Henry Baxter, L. E. Chittenden, and B. D. Harris.

Massachusetts.-John Z. Goodrich, John M. Forbes, Richard P. Waters, Theophilus P. Chandler, Francis B. Crowninshield, Geo. S. Boutwell, and Charles Allen.

Rhode Island.-Samuel Ames, Alexander Duncan, William W. Hoppin, George H. Browne, and Samuel G. Arnold.

Connecticut.-Roger S. Baldwin, Channey F. Cleveland, Charles J. McCurdy, James T. Pratt. Robbins Battelle, and Amos Treat.

New York .- David Dudley Field, Wm. Curtis Noyes, James S. Wadsworth, James C. Smith, Amaziah B. James, Erastus Corning, Greene C. Bronson, William E. Dodge, John A. King, and John E. Wool.

New Jersey.—Charles S. Olden, Peter D. Vroom, Robert F. Stockton, Benjamin Williamson, Joseph F. Randolph, Fred. T. Frelinghuysen, Rodman M. Price, William C. Alexander, and Thomas J. Stryker.

Pennsylvania.—Thomas White, James Pollock, Wm. M. Meredith, David Wilmot, A. W. Loomis, Thomas E. Franklin, and William McKennan.

Delaware.—George B. Rodney, Daniel M. Bates, Henry Ridgely, John W. Honston, and William Cannon.

Maryland.—John F. Dent, Reverdy John-son, John W. Crisfield, Augustus W. Brad-ford, Wm. T. Goldsborough, J. Dixon Roman, and Benjamin C. Howard.

Virginia.—John Tyler, William C. Rives, John W. Brockenborough, George W. Summers, and James A. Seddon.

North Carolina .- George Davis, Thomas Ruffin, David S. Reid, D. M. Barringer, and John M. Morehead.

Tennessee .- Samuel Milligan, Josiah M. Anderson, Robert L. Caruthers, Thomas Martin, Isaac R. Hankins, A. O. W. Totten, Robert J. McKinney, Alvin Cullom, Wm. Hickerson, Geo. W. Jones, F. K. Zollicoffer, and William H. Stevens.

Kentucky.-William O. Butler, James B. Clay, Joshua F. Bell, Charles S. Morchead, James Guthrie, and Charles A. Wickliffe.

Missouri.-John D. Coalter, Alexander

M. Doniphan, Waldo P. Johnson, Aylett H. Buekner, and Harrison Hough.

Ohio.—John C. Wright,\*Salmon P. Chase, William S. Groesbeck, Franklin T. Backus, Reuben Hitchcock, Thomas Ewing, and V. B. Horton.

Indiana.—Caleb B. Smith, Pleasant A. Hackleman, Godlove S. Orth, E. W. H. Ellis, and Thos. C. Slaughter.

Illinois.—John Wood, Stephen T. Logan, John M. Palmer, Burton C. Cook, and Thomas J. Turner.

*Iowa.*—James Harlan, James W. Grimes, Samuel R. Curtis, and William Vandever.

Wisconsin.—James R. Doolittle, C. Durkee, John F. Potter, and C. C. Washburn.

The officers were:

JOHN TYLER OF VIRGINIA, PRESIDENT; CRAFTS J. WRIGHT OF Ohio, Secretary; JAMES M. TOWER OF NEW JERSEY, J. HENRY PULESTON OF PENASYlvania, WM. M. HOPPIN OF Rhode Island, JOHN STRYKER OF NEW YORK, assistants.

They recommended that the following be proposed to the several States as amendments to the Constitution of the United States :

ARTICLE 13. SEC. 1. In all the present territory of the United States, north of the parallel of 36° 30' of north latilinde, involuntary servitude, except in punishment of crime, is prohibited. In all the present Territory south of that line, the stardus of persons held to involuntary service or labor, as it now exists, shall not be changed: nor shall any law be passed by Congress or the Territorial Legislature to hinder or prevent the taking of such persons from any of the States of this Union to said relation; but the same shall be subject to judicial coguizance in the Federal courts, according to the course of the common law. When any Territory north or south of said line, within such boundary as Congress may prescribe, shall contain a population equal to that required for a member of Congress, it shall, if its form of dovernment be republican, be admitted into the Union on au equal footing with the original States, with or without involuntary servitude, as the constitution of such State

SEC. 2. No Territory shall be acquired by the United States, except by discovery and for naval and commercial statious, dopots, and transit routes, without the concurrence of a majority of all the Senators from States which allow involuntary servitude, and a majority of all the Senators from States which prohibit that relation; nor shall Territory he acquired by treaty, unless the votes of a majority of the Senators from each class of States hereinbefore mentioned be cast as a part of the two-thirds majority necessary to the ratification of such treaty.

majority of the senators from each class of states hereinbefore mentioned be cast as a part of the two-thirds majority necessary to the ratification of such treaty. Src. 3. Neither the Constitution nor any amendment thereof shall be construed to give Congress power to regulate, abolish, or control, within any State the relation established or recognized by the laws thereof touching persons held to labor or involuntary service therein, nor to interfere with or abolish involuntary service in the District of Columbia without the consent of Maryland and without the consent of the owners, or making the owners who do not consent just compensation; nor the power to interfere with or prohibit Representatives and others from bringing with them to the District of Columbia, retaining, and taking away, persons so held to labor or service; nor the power to interfere with or abolish involuntary service in places under the exclusive jurisdiction of the United States within those States and Territory were to prohibit the removal or transportation of persons held to labor or involuntary service in any State or Territory of the United States to any other State or Territory of the United states to any other State or arvitor, of the ching at ports, shores, and landings, and of landing in case of distress, shall exist; but not the right of transit in or through any State or Territory, or of sale or transit in or through any State or Territory, or of sale or transit in or through

\* Died during the session, and succeeded by C. P. Wolcott.

thereof. Nor shall Congress have power to authorize any higher rate of taxation on persons held to labor or service than on land. The bringing that the District of Columbia of persons held to labor or service, for sale, or placing them in depots to be afterwards transferred to other places for sale as merchandize, is prohibited.

from in depois to be arterivated invariant reaction to the places for sale as merchandize, is problibiled. SEC, 4. The third paragraph of the second section of the fourth article of the Constitution shall not be construed to prevent any of the States, by appropriate legislation, and through the action of their judicial and ministerial officers, from enforcing the delivery of fugitives from labor to the person to whom such service or labor is duo.

from enforcing the delivery of figuites and industrial oncers, from enforcing the delivery of figuites from labor to the person to whom such service or labor is due. Sec. 5. The foreign slave trade is hereby forever prohibited; and it shall be the duty of Congress to pass laws to prevent the importation of slaves, coolies, or persons held to service or labor, into the United States and the Territorice from places beyond the limits thereof.

The formation of the second section of the fourth article of the second section of the fourth article thereof, shall not be amended or abolished without the consecut of all the States.

article incredit, shall be the alledded of abolished without the consent of all the States. SEC, 7. Congress shall provide by law that the United States shall pay to the owner the full value of his figitivo from labor, in all cases where the marshal or other officer, whose duty it was to arrest such fugitive, was prevented from so doing by violence or intimidation from mobs or riotous assemblages, or when, after arrest, such fugitive was rescued by like violence or intimidation, and the owner thereby deprived of the same; and the acceptance of such payment shall preclude the owner from further claim to such fugitive. Congress shall provide by law for securing to the citizens of each State s.

Section 1 was first lost, yeas 8, nays 11, then re-considered, and agreed to, yeas 9, (Delaware, Illinois, Kentucky, Maryland, New Jersey, Ohio, Pennsylvania, Rhode Island, Tennessee,) nays 8, (Connecticut, Iowa, Maine, Massachusetts, North Carolina, New Hampshire, Vermont, Virginia.)

Section 2 was agreed to, yeas 11, (Delaware, Indiana, Kentucky, Maryland, Missouri, New Jersey, Ohio, Pennsylvania, Rhode Island, Tennessee, Virginia,) nays 8, (Connecticut, Illinois, Iowa, Maine, Massachusetts, North Carolina, New Hampshire, Vermont.)

Section 3 was agreed to, yeas 12, (Delaware, Illinois, Kentucky, Maryland, Missouri, New Jersey, North Carolina, Ohio, Pennsylvania, Rhode Island, Tennessee, Virginia,) nays 7, (Connecticut, Indiana, Iowa, Maine, Massachusetts, New Hampshire, Vermont.)

Section 4 was agreed to, yeas 15, (Connecticut, Delaware, Illinois, Indiana, Kentucky, Maryland, Missouri, New Jersey, North Carolina, Ohio, Pennsylvania, Rhode Island, Tennessee, Vermont, Virginia, ) nays 4, (Iowa, Maine, Massachusetts, New Hampshire.)

Maine, Massachusetts, New Hampshire.) Section 5 was agreed to, yeas 16, (Connecticut, Delaware, Illinois, Indiana, Kentucky, Maryland, Missouri, New Jersey, New York, New Hampshire, Ohio, Pennsylvania, Rhode Island, Tennessee, Vermont, Kansas,) nays 5, (Iowa, Maine, Massachusetts, North Carolina, Virginia.)

Section 6 was agreed to, yeas 11, (Delaware, Illinois, Kentucky, Maryland, Missouri, New Jersey, Ohio, Pennsylvania, Rhode Island, Tennessee, Kansas,) nays 9, (Connecticut, Indiana, Iowa, Maine, Massachusetts, North Carolina, New Hampshire, Vermont, Virginia.) New York was divided.

Section 7 was agreed to, yeas 12, (Dela-

ware, Illinois, Indiana, Kentucky, Maryland, New Jersey, New Hampshire, Ohio, Pennsylvania, Rhode Island, Tennessee, Kansas,) nays 7, (Connecticut, Iowa, Maine, Missouri, North Carolina, Vermont, Virginia.) New York was divided.

By a vote of 12 to 7, (New York divided,) the resolution offered by Mr. FRANKLIN of Pennsylvania, after the adoption of the above proposition, to the effect that no State of this Union has any constitutional right to secede therefrom, or to absolve its citizens from their allegiance to the Government of the United States, was indefinitely postponed, yeas 10, (Delaware, Kentucky, Maryland, Missouri, New Jersey, North Carolina, Ohio, Rhode Island, Tennessee, Virginia.) nays 7, (Connecticut, Illinois, Indiana, Iowa, Maine, Massachusetts, Pennsylvania.)

Pending the vote in the Conference on these propositions,

Mr. DAVID DUDLEY FIELD, of New York, moved to substitute for Section 7, a provision that "no State shall withdraw from the Union without the consent of all the States, given in a Convention of the States, convened in pursuance of an act passed by two-thirds of each house of Congress;" which was rejected-yeas 10, nays 11, as follows:

YEAS --- Connecticut, Illinois, Indiana, Iowa, Maine, Massachusetts, New York, New Hampshire, Vermont, Kansas-10.

NAVS-Delaware, Kentucky, Maryland, Missouri, New Jersey, North Carolina, Ohio, Pennsylvania, Rhode Island, Tennessee, Virginia-11.

Mr. BALDWIN, of Connecticut, moved a substitute recommending that Congress call a Convention for proposing amendments to the Constitution of the United States; which was lost—yeas 8, nays 13, as follows:

YEAS-Connecticut, Illinois, Iowa, Maine, Massachusetts, New York, New Hampshire, Vermont-8.

NAVS—Delaware, Indiana, Kentucky, Maryland, Missouri, New Jersey, North Carolina, Ohio, Pennsylvania, Rhode Island, Tennessee, Virginia, Kansas-13.

Mr. SEDDON'S project-securing in addition to the Crittenden proposition, the right of transit for slaves through Free States-excluding from the elective franchise, or the right to hold Federal, State, Territorial or Municipal office, all persons who are in whole or part of the African race-and in a modified form recognizing the right of peaceable State secession,-was also offered, and was lost-yeas 4, nays 16, as follows:

nessee, besides those above), nays 14 (same as above with Tennessee and Kansas out).

Mr Tuck's proposition, denying the power of Congress, or any branch of the Federal Government, to interfere in any manner with slavery in any of the States, and that either of the great political organizations of the country contemplates a violation of the Constitution, and recommending a Convention to propose amendments to the Constitution of the United States for the redress of whatever grievances exist, was also rejectedyeas 9, nays 11, as follows :

YEAS-Connecticut, Illinois, Indiana, Iowa, Maine, Massachusetts, New York, New Hampshire, Vermont—9.

NAVS-Delaware, Kentucky, Maryland, Missouri, New Jersey, North Carolina.

There were 133 members present upon the adoption of the report, representing the States of

Maine, Massachusetts, New York, Delaware, North Carolina, Missouri, New Hampshire, Rhode Island, New Jersey, Maryland, Tennessee, Ohio, Vermont, Connecticut, Pennsylvania, Virginia, Kentucky, Indiana. Illinois, Iowa, Kansas.

NOTE.-Michigan and Wisconsin, of the loyal States, were not represented in the Conference.

These propositions were offered to the Senate by Mr. JOHNSON of Arkansas, March 2d, pending the House resolution proposing an amendment of the Constitution, and they were rejected—yeas 3, navs 34, as follows:

YEAS-Messrs. Foot, Nicholson, and Pugh-3. NATS-Messrs. Foot, Nicholson, and Pugh-3. NATS-Messrs. Anthony, Baker, Bighar, Bingham, Bright, Chandler, Clark, Crittenden, Dixon, Doolittle, Douglas. Durkee, Fessenden, Foster, Grimes, Harlan, Hunter, Johnson of Tennessee, Kennedy, King, Latham, Mason, Morrill, Polk, Rice, Sebastian, Sunner, Ten Eyck, Trumbull, Wade, Wigfall, Wilkinson, and Wilson -34.

Subsequently (March 2d) they were offered to the Senate by Mr. CRITTENDEN of Kentucky, after the adoption of the House constitutional amendment, and pending the Crittenden propositions, and they were rejected, yeas 7, nays 28, as follows:

YEAS-Messrs. Crittenden, Douglas, Harlan, Johnson of

YEAS-Messrs. Criteneten, Douglas, Harnan, Jonnson of Tennessee, Kennedy, Morrill, and Thomson-7. NAVS-Messrs. Bayard, Bigler, Bingham, Bright, Chand-ler, Clark, Dixon, Fessenden, Foot, Foster, Grimes, Gwin, Hunter, Lane, Latham, Mason, Nicholson, Polk, Pugh, Rice, Sebastian, Summer, Ten Eyck, Trumbull, Wade, Wigfall, Wilkinson, and Wilson-28.

March 1st, 1861. Mr. McCLERNAND moved to suspend the rules for the purpose of receiving the recommendation of the Peace Congress, which was rejected, yeas 93, nays 67, as follows:

lost—yeas 4, nays 16, as follows:
YEAS—Kentucky, Missouri, North Carolina and Virginia.
NAYS—Connecticut, Delaware, Illinois, Indiana, Maine, Massachusetts, Maryland, New Jersey, New York, New Hampshire, Ohio, Pennsylvania, Rhode Island, Tennessee, Vermont, and Kansas—16.
Mr. JAMES B. CLAY'S substitute (much like Mr. Seddon's) was also lost—yeas 5 (TenV. Kas-Messrs. Charles F. Adams, Green Adams, Adrain, Atrain, Canderson, Avery, Barr, Barrett, Bocock, Boteler, Brabson, Branch, Briggs, Bristov, Brown, Burch, Boteler, Brabson, Branch, Briggs, Bristov, Brown, Burch, Boteler, Brabson, Branch, Briggs, Bristov, Brown, Burch, John C. Charles, Convent, John B. Clark, John B. Clark, John B. Clark, John C. Charles, Convin, James Craig, John G. Davis, De Jarnett, Campbell, Horace F. Clark, John B. Clark, John St. Harris, Hastim, Hatton, Hoard, Holman, William Howard, Hughes, Milliam Kellogs, Killinger, Karly, McKnight, McPherson, Millson, Millward, Laban T. Moore, Moorhead, Edward Joy Morris, Nelson, Niblack, Nixon, Olin, Pendleton, Peyton,

Phelps, Porter, Peyor, Quarles, John H. Reynolds, Rice, Riggs, James C. Kohinson, Sickles, Simms, Wn. N. H. Smith, Spaulding, Stevenson, William Stewart, Stokes, Thomas, Vance, Webster, Whiteley, Winslow, Woodson, and Wright -93

—93. NAY3—Messirs Alley, Ashley, Bingham, Illair, Brayton, Buffaton, Burlingamo, Hurnham, Carey, Case, Coburn, Colfax, Conway, Burton Craige, Dawes, Dehano, Duell, Ed-gorton, Eliot, Ely, Fenton, Ferry, Frank, Gooch, Graham, Grow, Gneley, Iledmick, Hickman, Hindmon, William A. Howard, Hutchins, Irvine, Francis W. Kellogg, Kenyon, Loomis, Lovejoy, McKean, Morrill, Morse, Palmer, Perry, Potter, Pottle, Christopher Robinson, Royce, Rudin, Sodge, Wick Sherman, Somes Sainton Stateon Wick, Sherman, Sones, Spinner, Stanton, Stevens, Tappan, Tompkins, Train, Vandever, Van Wyck, Wade, Waldron, Walton, Cadwalader C. Washburn, Ellihu B. Washburne, Wells, Wilson, Windom, and Woodruff-67,

When the report was presented to the Senate and a motion made to fix a time to consider it,

Mr. THOS. L. CLINGMAN of North Carolina, said he was "utterly opposed to the proposition," but was willing to give it the direction its friends desired.

Mr. MILTON S. LATHAM of California, said he "had no confidence in this thing," but would vote for the motion.

When it was presented to the House,

Mr. SAML. H. WOODSON of Missouri, said he would vote to receive it, but was against the proposition.

Mr. BURTON CRAIGE of North Carolina, said he was "utterly opposed to any such wishywashy settlement of our National difficulties.

Mr. SHELTON F. LEAKE "regarded this thing as a miserable abortion, forcibly reminding one of the old fable of the mountain and the mouse; nevertheless, he was willing to let the mouse in, in order to have the pleasure of killing it."

Mr. Thos. C. HINDMAN of Arkansas, believed the report "to be unworthy of the vote of any Southern man."

Mr. Muscoe R. H. GARNETT of Virginia, "intending and desiring to express my abhorrence of these insidious propositions, conceived in fraud and born of cowardice, by giving a direct vote against them, yet from respect for the conference which reported them" was willing to receive them.

The above were the only votes taken upon the recommendation. That in the House was not a test, as some voted to receive who were opposed to the adoption of the report, and probably vice versa.

# Senate Committee of Thirteen.

December 18th, 1860. On motion of Mr. Powell, a Committee of Thirteen were authorized to consider the condition of the

country; which consisted of Mr. Powell, Mr. Seward, Mr. Collamer, Mr. Bigler, Mr. Hunter, Mr. Toombs, Mr. Davis, Mr. Rice, Mr. Crittenden, Mr. Douglas, Mr. Wade, Mr. Doolittle, Mr. Grimes. Dec. 31st. The chairman reported that

the Committee had not been able to agree upon any general plan of adjustment.

# PROCEEDINGS OF THE COMMITTEE.

Dec. 22d, 1860, it was agreed that no proposition shall be reported as adopted, the possession of the claimant for more than

unless sustained by each of the two classes of Senators of the Committee-Senators of the Republican party to constitute one class, and Senators of other parties to constitute the other.

Mr. CRITTENDEN offered his amendments.

The first article was lost-yeas 6, nays 7: YEAS-Messrs. Bigler, Crittenden, Douglas, Hunter, Powell, Rice-6.

NAYS-Messrs. Collamer, Davis, Doolittle, Grimes, Seward, Toombs, Wade-7.

The second was lost under the rule-yeas 8, nays 5. The nays were Messrs. Collamer, Doolittle, Grimes, Seward, and Wade.

The third, fourth, and fifth articles were lost by a similar vote; also the sixth, except that Mr. Grimes did not vote.

Mr. CRITTENDEN then submitted a Joint Resolution, containing four propositions :

1. That the Fugitive Slave law is in strict pursuance of the plain and mandatory provisions of the Constitution, and has been sanctioned as valid and constitutional by the judgment of the Supreme Court of the United States, that the slave-holding States are entitled to its faithful execution, and laws should be made to punish those who attempt, by rescue or other illegal means, to hinder or defeat its execution.

2. Congress recommends the repeal of those State laws which conflict with the Fugitive Slave law or other constitutional acts of Congress, or impede, hinder or delay its free course and execution, or such modification of them as may prevent their being used or perverted to such mischievous purposes.

3. The Fugitive Slave law should be so amended as to make the fee of the Commissioner equal, whether his decision be for or against the claimant, and the authority to call the posse comitatus to his aid should be limited to cases of resistance, or danger of resistance or rescue.

4. The laws for the suppression of the African slave trade should be made effectual and thoroughly executed.

The first was lost under the rule-yeas 8, nays 3-Messrs. Doolittle, Grimes and Wade, voting nay; Mr. Collamer and Mr. Seward not voting.

The second was lost under the rule-yeas 7, nays 4-Messrs. Collamer, Doolittle, Seward, and Wade, voting nay; Mr. Grimes not voting.

The third was unanimously agreed toyeas 13; also the fourth.

Mr. DOOLITTLE moved that the laws should secure to the alleged fugitive slave, when he shall claim that he is not a fugitive slave, a jury trial before he shall be delivered to the claimant.

Mr. Toomes moved to amend by adding the words, "in the State from which he fled." Agreed to, yeas 7, nays 5.

Mr. CRITTENDEN moved to limit this right to cases "where he shall have been out of

two years." Lost-yeas 6, nays 6. Mr. Toombs' motion was then lost—yeas 3 (Big-ler, Crittenden, Grimes), nays 9. Dec. 24th, 1860. Mr. SEWARD submitted

these Resolutions:

1. No amendment shall be made to the Constitution which will authorize or give to Congress the power to abolish or interfere, within any State, with the domestic institutions thereof, including that of persons held to labor or service by the laws of said State.

2. The Fugitive Slave act of 1850 shall be so amended as to give to the alleged fugitive a jury trial.

3. The Legislatures of the several States shall be respectfully requested to review all of their legislation affecting the right of persons recently resident in other States, and to repeal or modify all such acts as may contravene the provisions of the Constitution of the United States, or any laws made in pursuance thereof.

The first was agreed to-yeas 11, nays 2 (Rice and Toombs.)

Mr. Toombs moved to add to the second the words, "in the State from which he fled." Agreed to-yeas 7, nays 5. The resolution as amended was disagreed to-yeas 6, nays 7.

YEAS-Messrs. Collamer, Doolittle, Douglas, Grimes, Seward, Wade.

NAYS-Messrs. Bigler, Crittenden, Davis, Hunter, Powell, Rice, Toombs.

The third was lost, under the rule-yeas 7, nays 5.

YEAS-Messrs. Bigler, Collamer, Crittenden, Doolittle, Grimes, Seward, Wade.

NAYS -- Messrs. Davis, Hunter, Powell Rice, Toombs.

Mr. Toomes'\* propositions were then negatived.

all of its departments, everywhere, which the Constituto confest the power upon it to extend to any other property—this not to interfere with the right of every State to prohibit, abolish, or establish, and protect sla-very within its limits. Lost under the rule by the same vote.

very within its timits. Loss and the rate by the same vote.
3. Persons committing crimes against slave property in one State and fleeing to auother, shall be delivered up as other criminals. Lost by the same vote.
4. Congress to pass laws punishing persons engaged in invasion or insurrection, or other act tending to disturb the tranquillity of any other State. Lost, 6 to 4.
5. Fugitive slaves not to have benefit of writ of habeas corpus or trial by jury. Lost, 7 to 5, as before.
6. Congress to pass un law in relation to slavery in the States or Territories or elsewhere, without the consent of a majority of the Senators and Representatives of the slaveholding States. A mended, on motion of Mr. HUXTER, so as to add: "and also a majority of the Senators and then lost—yeas 5, nays 6.
7. None of these provisions, or others in the Constitution relating to slavery (scept the African slave trade), to be altered without the consent of all the States relatives of the sone of all the States in which slavery exists. Lost, 6 to 5 (the Republican members).

members).

Dec. 26, 1860. Mr. SEWARD moved, That under the 4th section of the 4th article of the Constitution, Congress should pass an efficient law for the punishment of all persons engaged in the armed invasion of any State from another, by combinations of individuals, and punishing all persons in complicity therewith, on trial and conviction in the State or district where their acts of complicity were committed, in the Federal courts.

Mr. Toombs moved to add: "And also all attempts to excite insurrection in any State by the people of any other State." Agreed to-yeas 8, nays 5 (Collamer, Doolittle, Grimes, Seward, Wade).

Mr. DOUGLAS moved to add: "And for the suppression and punishment of conspiracies or combinations, in any State or Territory, with intent to invade, assail, or molest the government, inhabitants, property, or institutions of any other State or Territory of the Union." Agreed to.

On the resolution as amended, the question was divided, and the proposition offered by Mr. Seward was agreed to-yeas 9, nays 3 (Rice, Toombs, Wade); that offered by Mr. Toombs was agreed to; that offered by Mr. Douglas was lost—yeas 6, nays 6 (Collamer, Davis, Doolittle, Grimes, Seward, Wade). The resolution as amended by Mr. Toombs was then lost, under the rule-yeas 7, nays 5 (Collamer, Doolittle, Grimes, Seward, Wade).

MR. DAVIS submitted a proposition, that it shall be declared, by amendment of the Constitution, that property in slaves, recog-nized as such by the local laws of any of the States in the Union, shall stand on the same footing in all constitutional and federal relations as any other species of property so recognized, and like other property shall not be subject to be divested or impaired by the local law of any other State, either in escape thereto or of transit or sojourn of the owner therein; and in no case whatever shall such property be subject to be divested or impaired by any legislative act of the United States, or any of the territories thereof. This was lost—yeas 6, nays 6.

YEAS.-Messrs. Bigler, Davis, Hunter, Powell, Rice, Toombs.

NAYS. — Messrs. Collamer, Crittenden, Doolittle, Grimes, Seward, Wade.

December 28th, 1860. Mr. CRITTENDEN submitted a proposition relative to New Mexico, which was lost—yeas 2, (Crittenden and *Douglas*,) nays 11.

Mr. BIGLER submitted a proposition, proposing seven amendments to the Constitution.

1st. Dividing the territory now owned by the United States by a line from east to west on the parallel of 36° 30'.

2d. Dividing the territory south into four territories, the territory north into eight.

Congress to provide governments 3d. when the inhabitants become sufficiently numerous, and when the bona fide inhabitants in any one shall be equal to the ratio of representation in Congress, it shall be the

<sup>\*</sup> They were—lst. The people of the United States shall have an equal right to emigrate to, and settle in, the preshave an equal right to emigrate to, and settle in, the pres-eut or any future acquired territories, with whatever property they may possess (including slaves), and be securely protected in its peaceable enjoyment, until such Territory may be admitted as a State in the Union, with or without slavery, as she may determine, on an equal-ity with all existing States. Lost under the rule-yeas 7, mays 5 (the Republican members). 2. That property in slaves shall be entitled to the same protection from the Government of the United States in all of its denariments. everywhere, which the Constitu-

duty of the President, by proclamation, to announce the admission of such State into the Union.

4th. In territory south, involuntary servitude, as it now exists in the States south of Mason and Dixon's line, shall be recognized and protected by all the departments of the Territorial Governments; and in all territories north, involuntary servitude except as a punishment of crime shall be prohibited.

5th. Congress shall be denied power to abolish slavery in places now under its jurisdiction within slave States and the District of Columbia, while slavery may exist in Virginia or Maryland.

6th. Non-slaveholding States shall provide efficient laws for the delivery of fugitives from labor to the persons to whom such service or labor may be due.

7th. Neither these amendments nor the third paragraph of the second section of the first article of the Constitution, nor the third paragraph of the second section of the fourth article of the Constitution, shall be liable to further amendment.

This proposition was determined in the negative without a division.

Mr. DougLAS's propositions were then considered. They were:

1st. Congress shall make no law in regard to slavery or servitude in the territories, but the status of such territory in this respect to remain as at present; and that every territory might form a Constitution when its population should be fifty thousand, and exercise all the rights of selfgovernment; and that such State should be admitted into the Union on an equal footing with the original States, with or without slavery, as its Constitution should provide, when such State should contain a population to entitle it to a representation, according to the federal ratio of representation. 2d. No more territory to be acquired ex-

2d. No more territory to be acquired except on avote of two-thirds of both branches of Congress, and the status thereof as to servitude to remain as at the time of its acquisition, until altered by the people in framing a Constitution as aforesaid.

3d. Area of future States to contain not less than 60,000, normore than 80,000 square miles.

4th. Extending provisions of the fugitive slave law to the new States and Territories.

5th. Extending the judicial power of the United States to the new States and Territories.

6th. Forbidding the exercise of the elective franchise to persons of the African race, in whole or in part.

7th. Giving power to Congress to acquire territory in Africa or South America for colonization of free blacks.

8th. Prohibiting Congress from abolishing slavery in any places in any slave State without the consent of such State.

9th. Prohibiting Congress from abolish-

ing slavery in the District of Columbia without the consent of Maryland, or preventing persons having business in the District from bringing their slaves with them and taking them away again.

10th. Prohibits Congress from interfering with the inter-State slave-trade.

11th. Provides a payment in cash for any fugitive slave who cannot be arrested by the marshal through fear of mob violence, etc.

12th. Provides that these amendments and the third paragraph of the second section of the first article, and the third paragraph of the second section of the fourth article of the Constitution, shall never be amended, and prohibits any amendment authorizing Congress ever to abolish slavery in the States.

Mr. DOUGLAS'S propositions were negatived. The vote on the *first* was, yeas 2 (Crittenden and Douglas,) nays 11. On the *second*, yeas 1, nays 10. On the *third*, yeas 2, nays 11. On the *fourth*, lost *under the rule*, yeas 8, nays 5 (the Republican members). The *fifth* was lost, *nem con*. The *sixth*, lost *under the rule*, yeas 8, nays 5 (the Republican members). The *seventh* was adopted, yeas 10, nays 3 (Davis, Hunter, Toombs). The *eighth* and *ninth* were lost, *nem con*. The *tenth* was a mended, and negatived *under the rule*, yeas 8, nays 5 (the Republican members). The *eleventh* and *twelfth* were negatived, *nem con*.

Mr. Rice offered a proposition to settle the territorial question by admitting at once into the Union as a State all the territory lying north of  $36^{\circ}$  30' to be called the State of Washington, and all the territory lying south of  $36^{\circ}$  30' to be called the State of Jefferson; and in each case, whenever any portion of said States shall contain, within an area of not less than 60,000 square miles, 130,000 inhabitants, a new State may be formed—all acts organizing territorial governments to be repealed, and Congress to appropriate money to pay the expenses of the convention to form constitutions.

Mr. SEWARD moved to amend by adding: "except so much of the Territory of Kansas as is contained in the proposed boundary of the Wyandot Constitution." Lost—yeas 6, nays 6, Douglas voting aye with the Republican members.

Mr. RICE'S resolution was lost—yeas 3 (Bigler, Davis, Rice), nays 10 (Collamer, Crittenden, Doolittle, Douglas, Grimes, Hunter, Powell, Seward, Toombs, Wade). On motion of Mr. TOOMBS, the committee

On motion of Mr. TOOMES, the committee then agreed to report to the Senate, with their journal, the fact that they had not been able to agree upon any general plan of adjustment.

# Other Propositions of Adjustment.

Other propositions were made, which do not appear to have been voted on in either House, or a committee of either:

Mr. JOSEPH LANE, Senator from Oregon,

offered a preamble and resolutions reciting the irreconcilable differences which had arisen between the North and South on the question of slavery in the Territories, and the measures that had been from time to time adopted to settle the difficulties; that the Southern States had resolved to dissolve the Union rather than submit to the principles embodied in the Chicago platform, and that the mode of amendment to the Constitution prescribed by that instrument itself was inadequate to the emergency. And he proposed to the several States to send delegates to a general convention to devise a new form of government and requesting the delegates of the Southern States to first meet in convention, being the aggrieved party, to confer and submit their conclusions to the delegates from the Northern States, and declaring it contrary and abhorrent to the religion and civilization of the age to prevent by force any State from taking such course as her safety and prosperity might require.

Mr. ANDREW JOHNSON, Senator from Tennessee, submitted a joint resolution proposing amendments to the Constitution in substance as follows:

1st. To apportion each State into as many districts as should be equal in number to the whole number of Senators and Representatives in Congress to which such State should be entitled, and that in every such district the persons receiving the highest number of votes for President and Vice President should be holden to have received one vote; that the persons receiving the vote of a majority of the districts, provided it should be a majority of the whole number of votes given, should be President and Vice President, but that if no person had received a majority of the whole number of votes as aforesaid, then a second election should be held between the two persons having the highest number of votes at the first election, and providing further for the election of a President from the free States and Vice President from a slave State, and vice versâ, every four years.

2d. Providing for an election of United States Senators by the people instead of by the State Legislatures.

3d. Limiting the tenure of office of the Judges of the Supreme Court of the United States to twelve years, and providing for an election of one-third of said Judges every fourth year.

Mr. JOHNSON also offered a resolution instructing the Committee of Thirteen to inquire into the expediency of amending the Constitution, so as to establish a division line in the Territories; to repeal the legislation of Congress relative to the return of fugitive slaves, and require each State to return them or pay their cash value to the owner; allowing the question of slavery in dock-yards, arsenals, etc., to be regulated by the States in which such places were located; restricting Congress from abolishing slavery in the District of Columbia without the consent of Maryland, and making compensation to the owners; and also from interfering with the three-fifths representation of slaves or the inter-State trade; and declaring that these amendments should never be amended.

# THE BORDER STATES PROPOSITION.

At a meeting of Senators and Representatives from the fourteen border free and slave States, at which about seventy-five gentlemen were present, Mr. CRITTENDEN acting as President, and Messrs. COLFAX and BARRETT as Secretaries, the following propositions were submitted:

By Mr. BARRETT of Missouri. Eleven amendments to the Constitution on the slavery question.

By Mr. PRVOR of Virginia. That an attempt to preserve the Union between the States of this Confederacy by force would be equally unconstitutional, impolitic, and destructive of Republican liberty.

By Mr. VALLANDIGHAM. The Crittenden resolutions.

By Mr. COLFAX. That the laws of the Union should be enforced and the Union of the States maintained; and that it is the duty of the Executive to protect the property of the United States with all the power placed in his hands by the Constitution.

By Mr. MORRIS of Illinois. That in the maturing any plan for adjusting the existing difficulties of the country we will keep steadily in view the preservation of the Union under the Constitution as a paramount consideration.

All of the above propositions after some discussion were referred to a committee consisting of Messrs. CRITTENDEN of Kentucky, HARRIS of Maryland, SHERMAN of Ohio, NIXON of New Jersey, SAULSBURY of Delaware, GILMER of North Carolina, HATTON of Tennessee, PETIT of Indiana, HARRIS of Virginia, McCLERNAND of Illinois, BARRETT of Missouri, SEBASTIAN of Arkansas, VANDE-VER of Iowa, and HALE of Pennsylvania.

The committee after consultation adopted the following propositions:

Recommending a repeal of all personal liberty bills.

An efficient amending of the fugitive slave law, preventing kidnapping, equalizing commissioners' fees, etc.

Amending the Constitution so as to prohibit interference by Congress with slavery in the States, or abolishing it in any places within its jurisdiction without the consent of the State in which such places are located.

Or with the inter-State slave trade.

Perpetual prohibition of the African slave trade.

Division of territory into free and slave on the line of 36° 30', prohibiting Congress or the Territorial Legislature from abolishing or interfering with slavery south of that line, and providing for admission of new States with population sufficient for one member of Congress and 60,000 square miles of area.

# OTHER PROPOSITIONS.

Proposition to the Hon. JAMES HUMPHREY, member of the Committee of Thirty-three from the State of New York, signed by Join Cochrane, William B. Maclay, Daniel E. Sickles, George Briggs, Thomas J. Barr, and Horace F. Clark, members of Congress from said State:

1st. Recommending the territorial line of  $36^{\circ} 30'$ .

2d. Non-interference of Congress with slavery in any places within the jurisdiction of Congress without the consent of the States in which such places are located.

3d. Change in the mode of electing President, by the adoption of the single electoral district system for the election of Presidential electors, with two electors at large for each State; with provision for a second election between the two highest candidates in case no one had received a majority on the first vote.

4th. Limiting the eligibility to the Presidency to one term, and recommending a change in the fugitive slave law so as to provide for the delivery of the fugitive, not to his master, but to the marshal of the district in which the claimant resided, there to be tried as to the right of ownership; and in case of unlawful rescue, etc., his value to be paid by the State or county in which such rescue occurred.

A "distinguished gentleman of Kentucky" proposed to so amend the Constitution as to restrain Congress from passing any law to interfere with slavery in territory south of 36° 30', or to interfere with the inter-State slave trade, or to abolish slavery in places belonging to the General Government within the limits of a slave State, or to permit the reopening of the slave trade, or to interfere with slavery in the States by unequal taxation or otherwise, but authorizing Congress to regulate the transit of slaves through free States, and to authorize the recapture of slaves escaping in such transit, and to give compensation to the owner of the slave in case of violent rescue, etc., making it compulsory to admit new States with slave constitutions, and to prevent and punish unlawful invasions of any State

with a design to excite slave insurrections. Mr. A. H. H. STUART offered a series of resolutions in the Senate of Virginia to the effect that the slaveholding States should be created trustees of the territory south of 36° 30', and the free States trustees of the territory north of that line, with the right to regulate the question of slavery in either case respectively.

The National Intelligencer of January 12th, 1861, proposed a plan of compromise by amendments to the Constitution and Congressional legislation substantially as follows:

To prohibit Congress from interfering with slavery in the States, in places under Federal jurisdiction within the limits of a slave State, or in the District of Columbia, or with the inter-State slave trade, or to reopen the slave trade; from acquiring any additional territory except by a concurrent vote of two-thirds of both Houses of Congress, or by treaty ratified by two-thirds of the Senate, and to adopt the single electoral district system, with two electors at large for each State.

Also, to so amend the fugitive slave law as to give the same compensation to the Commissioner whether the slave was given up or not, and making the State liable for all extraordinary expenses attending the capture of fugitive slaves, and for compensation to the owner in case of violent rescue, etc. To admit all the territory north of 36° 30' at once into the Union as a State to be called "Washington," and all south of that line as a State to be called "Jefferson," with provision for the sub-division of said States into two or more States.

Governor LETCHER, of Virginia, in his message to the Legislature, proposed in sub-stance the following plan: To appoint Commissioners to visit each Northern State that had passed personal liberty bills and urge their repeal, guarantee slavery in the District of Columbia, equal rights in the Territories, free transit for slave property, and the inter-State slave trade, punishment of persons aiding in any way in inciting slave insurrections, and deprive the General Government of the power to appoint to office in slave States persons hostile to the institution of slavery. He also declared strongly against the right of coercion, and that the passage of Federal troops across Virginia for such purpose would be repelled, and expressed his willingness that New England and Western New York should be sloughed off and permitted to ally themselves with Canada.

The Albany Evening Journal proposed the following as a basis of settlement:

1. The passage of an efficient but not revolting fugitive slave law.

2. The repeal of personal liberty laws.

3. Non-intervention by Congress with slavery in the Territories.

4. The admission of Territories as States with or without slavery, as they may decide, whenever they have a population entitling them to a Representative in Congress.

5. Or if the two last propositions are inadmissible, then the restoration of the Missouri compromise line.

souri compromise line. Hon. JOSEPH W. TAYLOR of Alabama proposed the following:

1st. That every Southern State should meet in Convention and adopt a platform expressive of their views and wishes in the present crisis, and then that they should all meet in a Southern Convention and adopt a platform setting forth all that they had to demand, with a request that the people of | the Northern States should meet in a similar Convention to consider said demands.

The New York Sun published as its plan of settlement:

1st. The prompt excision by amendment of the Constitution, of any State which shall within the next ten years, elect by vote of two-thirds of her inhabitants to secede from the Union.

2d. The adoption of State and Federal laws for the indemnification of owners of slaves who should be rescued by force or otherwise unlawfully.

In an article in the Baltimore Exchange of January 7th, 1861, it was proposed as a settlement of the territorial question, to admit New Mexico and Arizona as a State, and then amend the Constitution so as to absolutely prohibit the acquisition in the future of any additional territory whatever, the author strongly alleging that Republican Governments had more to fear from the expansion of territory than from any other cause whatever.

# BY A DISTINGUISHED CITIZEN OF NORTH CARO-LINA.

An amendment of the Constitution in regard to the election of a President and Vice President, providing for a vote of the people throughout each State directly for President and Vice President; then ascertain the number to which the popular vote ought to be increased by reason of the slave population, and let that be divided pro rata between the several sets of electors according to their respective strength, then let the majority of each set of electors cast the majority of each set of electric current of ing: popular vote for President and Vice Presi- ing: "Resolved, &c., That we properly estimate

A writer in the National Intelligencer proposed the following: That Congress, by joint resolution, should declare that the personal liberty bills are unconstitutional, and the States should repeal them; that Congress should not interfere with slavery in the States; that no more territory should be acquired except by consent of three-fourths of the States, and that the Dred Scott decision should govern all our present territory.

In a letter published in the New York Herald, purporting to have been written at Lexington, Kentucky, it was asserted that JOHN C. BRECKINRIDGE was in favor of the Crittenden Compromise, and that the Crittenden amendment had actually been drawn up by Mr. Breckinridge and a Mr. M. C. Johnson of Lexington. prior to Mr. Breckinridge's departure for Washington.

# Miscellaneous Resolutions.

# ON UNCONSTITUTIONAL STATE LAWS.

December 17, 1860. Mr. ADRAIN, of New Jersey, offered the following :

"Whereas the Constitution of the United of any party or any man; that whatever States is the supreme law of the land, and evils or abuses exist under it ought to be

its ready and faithful observance the duty of all good and law-abiding citizens,-

"Resolved. That we deprecate the spirit of disobedience to the Constitution, wherever manifested, and we earnestly recommend the repeal of all statutes, including personal lib-erty bills, so called, enacted by State Legislatures, which are in conflict with and in violation of that sacred instrument, and of the laws of Congress made in pursuance thereof."

Which was agreed to-yeas 153, nays 14, as follows:

as follows: YEAS-Messrs. Adams of Ky., Adrain, Aldrich, Allen, Anderson of Missouri, Anderson of Ky., Ashley, Avery, Babbitt, Bar, Borrett, Beale, Bingham, Biair, Islake, Bocock, Boteler, Bouligny, Branch, Brayton, Briggs, Brown, Buffinton, Burlingame, Burnett, Burnham, Butterfield, Campbell, Carter, Case, Horace F. (Tark, Clark of Missouri, Cobb, John Cochrane, Colfax, Conkling, Covode, Cor, Craig of Missouri, Craige of North Carolina, Crawford, Dovis of Indiana, Dawes, De Jarnette, Belano, Duell, Duom, Elmand-son, Ely, English, Etheridge, Farasworth, Fenton, Florence, Fouke, Frank, Garnett, Gilmer, Gooch, Graham, Grow, Hale, Hall, Hardeman, Harris of Md., Harris of Ya., Hatton, Helmick, Hill, Hindman, Holman, Houston, How-ard of Ohio, Mighes, Humphrey, Irvine, Jenkins, Jones, Junkin, Kellogg of Michigan, Kenyon, Kilgore, Killinger, Landrum, Larrabee, Leach of N. Carolina, Leake, Lee, Lo-gan, Longnecker, Loomis, Lovejor, Mallory, Matria of Ohio, Martin of Ya., McClernand, McKenn, McKenty, NetKnight, McPherson, Milward, Montgamery, Moore of Ky., Moorhead, Morris of Penn'a, Morris of HL, Niblack, Nixon, Noell, Olin, Palmer, Fendleton, Pettit, Fiyton, Por-en, Pottle, Pryor, Quarles, Reagon, J. H. Reynolds, Rice, Riggs, Robinson, Royce, Ruffin, Scott, Scranton, Sleven-son, Stewart of Pa., Stokes, Thayer, Thomos, Tompkins, Train, Trimble, Vance, Verree, Wale, Waldron, Widton, Washlurne of Ill, Webster, Wilson, Wood, Woodson, Wright -153.

NAYS-Messrs. Alley, Carey, Edwards, Eliot, Hutchins, Potter, E. R. Reynolds, Sedgwick, Spinner, Tappan, Vande-ver, Washburn of Wisconsin, Washburn of Maine, Wells-14.

The preamble was agreed to-yeas 156, nays none.

# ON THE UNION.

Mr. MORRIS, of Illinois, offered the follow-

the immense value of our National Union to our collective and individual happiness; that we cherish a cordial, habitual, and immovable attachment to it; that we will speak of it as of the palladium of our political safety and prosperity; that we will watch its pre-servation with jealous anxiety; that we will discountenance whatever may suggest even a suspicion that it can in any event be abandoned, and indignantly frown down the first dawning of every attempt to alienate any portion of our country from the rest, or enfeeble the sacred ties which now link together the various parts; that we regard it as a main pillar of the edifice of our real independence, the support of our tranquillity at home, our peace abroad, our safety, our prosperity, and that very liberty which we so highly prize; that we have seen nothing in the past, nor do we see any thing in the present, either in the election of Mr. Lincoln to the Presidency of the United States, or from any other existing cause, to justify its dissolution; that we regard its perpetuity as of more value than the temporary triumph

corrected within the Union, in a peaceful and Constitutional way; that we believe that it has sufficient power to redress every wrong and enforce every right growing out of its organization or pertaining to its proper functions; and that it is a patriotic duty to stand by it, as our hope in peace and our defence in war."

Which was agreed to-yeas 115, nays 44, as follows:

328 10110WS: YEAS-Messrs. Adams of Kentucky, Adrain, Aldrich, Allen, Alley, Anderson of Kentucky, Ashley, Babbitt, Beale, Bingham, Blair, Blake, Brayton, Briggs, Buffinton, Burlingame, Burnham, Butterfield, Campbell, Carey, Car-ter, Case, Colfax, Conkling, Covode, Cox, Davis of Indiana, Dawes, Delano, Duell, Dunn, Edgerton, Edwards, Eliot, Ely, English, Etheridge, Farnsworth, Fenton, Foster, Frank, French, Gooch, Graham, Grow, Hale, Hall, Harris of Md., Helmick, Hickman, Hoard, Holman, Houard of Ohio, Humphrey, Hutchins, Irvine, Junkin, Kellogg of Mich., Ken-yon, Kilgore, Killinger, Leach of Mich, Lee, Laogan, Longneck-Mennick, Mickinki, Moura, Holman, Houdard of Ohlo,
 Humphrey, Hutchins, Irvine, Junkin, Kellogg of Mich., Ken-yon, Kilgore, Killinger, Leach of Mich., Lee, Logan, Longneck-er, Loonis, Lovejoy, Martin of Ohlo, McClernand, McKean,
 McKnight, McPherson, Montgomery, Moorhead,
 Morris of Pa., Morris of Illinois, Niblack, Nixon, Olin,
 Palmer, Perry, Pettit, Porter, Potter, Potte, E. R. Reynolds,
 J. H. Reynolds, Rice, Riggs, Robinson of Ill., Royce, Scranton,
 Sedgwick, Sherman, Spaulding, Spinner, Stanton, Stewart
 of Pa., Stokes, Tappan, Thayer, Tompkins, Train, Trimble,
 Vandever, Verree, Wade, Waldron, Walton, Washburne of
 Ill., Washburn of Maine, Wells, Wilson, Wood—115.
 Nars-Messrs. Anderson of Missouri, Avery, Barksdale,
 Bouliguy, Branch, Burnett, Cark of Missouri, Clopton,
 Cobb, Crawford, Curry, De Jarnette, Eilmundson, Florence,
 Garnett, Gartrell, Hardeman, Hill, Hindman, Hughes, Jackson, Jonkins, Jones, Landrum, Leach of N. C., Leake, Martin
 of Virginia, Moore of Alabama, Noell, Pryor, Quarles, Real,
 gan, Ruffin, Soott, Sickles, Singleton, Smith of Va., Stallworth, Stevenson, Thomas, Underwood, Vance, Webster,
 Wright—41.

# THE CONSTITUTION AND SLAVERY.

Mr. CRAWFORD, of Georgia, offered the following:

"Resolved, That the Constitution of the United States recognizes property in slaves. That the Congress of the United States has passed laws to aid slaveholders in recapturing their slaves whenever they escape and make their way into the free States. That the Supreme Court of the United States have decided that negroes were not included either in the Declaration of Independence or in the Constitution of the United States except as slaves; and that they cannot become citizens of the United States. And we, the members of this House, hereby sustain and will support this construction of the Constitution, the laws, and said decisions of the Supreme Court."

Dec. 18th. Mr. SHERMAN moved to lay the resolution on the table, which was rejected -yeas 89, nays 92. Subsequently a motion was made to reconsider this vote, and the House refusing—yeas 87, nays 91—to lay the motion to reconsider on the table, the motion of Mr. Sherman to lay the resolution on the table was agreed to-yeas 88, nays 81, as follows :

As follows.
YEAS—Messrs. Aldrich, Allen, Alley, Babbitt, Beale, Bingham, Blair, Blake, Brayton, Buffinton, Burlingame, Butterfield, Carey, Carter, Case, Colfax, Conkling, Covode, Curtis, Dawes, Delano, Duell, Edgerton, Edwards, Eliot, Ely, Farnsworth, Fenton, Foster, Frank, Gooch, Graham, Grow, Gurley, Hale, Hall, Helunick, Hickman, Hoard, Hutchins, Irvine, Junkin, Francis W. Kellogg, Kenyon, Kilgore, Killinger, De Witt C. Leach, Lee, Longnecker, Loomis, Lovejoy, McK an, McKnight, McPherson, Millward, Moor-head, Edward Joy Morris, Morse, Perry, Pettit, Porter, Pottle, Edwin R. Reynolds, John H. Reynolds, Rice, Royce, Sedgwick, Sherman, Spaulding, Spinner, Stanton, William

Stewart, Tappan, Theaker, Tompkins, Train, Trimble, Van-dever, Verree, Wade, Waldron, Walton, Cadwalader C. Washburn, Ellihu B. Washburne, Israel Washburn, Wells, Wilson, Wood-88.

MARS-Messrs. Green Adams, Adrain, Thomas L. Anderson, Ashley, Barksdale, Barrett, Bocock, Boteler, Boyce, Briggs, Brown, Burnett, Horace F. Clark, John B. Clark, Clemens, Clopton, Cobb, John Cachrane, Cax, James Craig, Burton Craige, Crawford, Curry, John G. Davis, De Jarnette, Edmundson, English, Florence, Fouke, Gartrell, Gilmer, Hardeman, J. Morrison Harris, John T. Harris, Hatton, Hill, Hindman, Hughes, Jackson, Jones, Kunkel, Landrum, Larabee, Jas. M. Leach, Leake, Logan, Maclay, Mallory, Elbert S. Martin, McClernand, McKenty, Montgomery, Lahan T. Moore, Sydenham Moore, Isaac N. Moris, Niblack, Noel, Pendleton, Peyton, Pryor, Pugh, Quarles, Reagan, Riggs, Scott, Sückles, Simms, Singleton, Wm. Smith, Suillworth, Stevenson, James A. Stewart, Stokes, Thomas, Underwood, Vance, Webster, Woodson, Wright–S1.

APPROVAL OF MAJOR ANDERSON'S COURSE.

In the House of Representatives, January 7th, 1861, this resolution was offered by Mr. ADRAIN, of New Jersey, and adoptedyeas 125, nays 56.

"Resolved, That we fully approve the bold and patriotic act of Major Anderson in withdrawing from Fort Moultrie to Fort Sumter, and of the determination of the President to maintain that fearless officer in his present position; and that we will support the President in all constitutional measures to enforce the laws and preserve the Union."

#### VOTE.

YEAS—Messrs. Charles F. Adams, Green Adams, Adrain, Aldrich, Allen, Alley, W. C. Anderson, Ashley, Babbitt, Beale, Bingham, Blair, Blake, Brayton, Briggs, Buffinton, Burlin-game, Campbell, Carey, Carter, Clemens, Coburn, J. Cochrane, Colfax, Corwin, Covode, Cox, Curtis, H. W. Davis, J. G. Davis, Dawes, Delano, Dimmick, Duell, Dunn, Edgerton, Edwards, Dick Elly, Etheride, Represent F. Conto, Forther, Forther Dawes, Delano, Dimmick, Duell, Dunn, Edgerton, Edwards, Eliot, Ely, Etheridge, Farnsworth, Fenton, Ferry, Foster, Fouke, Frank, French, Gooch, Grow, Gurley, Hiale, Hall, Haskin, Helmick, Hickman, Hoard, Holman, William Hou-ad, William A. Howard, Humphrey, Hutchins, Irvine, Junkin, Francis W. Kellogg, William Kellogg, Kenyon, Kilgore, Larrabee, De Witt C. Leach, Logan, Longnecker, Loomis, Lovejoy, Maclay, Marston, Charles D. Martin, McClernand, McKean, McKrinly, McKnight, McPherson, Millward, Monigomery, Moorhead, Morrill, Edward Joy Morris, Morse, Nixon, Olin, Palmer, Perry, Pettit, Porter, Pottle, Edwin R. Reynolds, Rice, Riggs, Christopher Robin-son, James C. Robinson, Royce, Sedgwick, Sherman, Sickles, Spaulding, Spinner, Stanton, Stevens, William Stewart, Trimble, Vandever, Wade, Waldron, Walton, Cadwalader C. Washburn, Ellihun B. Washburne, Wells, Wilson, Windom, Wood-125. Wood-125

Wood-125. NAIS-Messrs. Thomas L. Anderson, Avery, Barksdale, Barrett, Bocock, Bouligny, Branch, Brown, Burch, Burnett, John B. Clark, Clopton, Cobb, James Craig, Crawford, De Jarnette, Florence, Garnett, Gartrell, Gilmer, Hamilton, Har-deman, John T. Harris, Hatton, Hawkins, Hill, Hindman, Houston, Hughes, Jones, Kunkel, Landrum, James M. Leach, Lore, Mallory, Elbert S. Martin, Maynard, McRae, Laban T. Moore, Sydenham Moore, Peyton, Phelps, Pryor, Quarles, Reagan, Rust, Scott, Simmis, Singleion, William N. H. Smith, Stevenson, Stokes, Vallandigham, Vance, Webster, Wright -56. -56.

#### COERCION.

December 31st. Mr. PRYOR offered the following resolution:

Resolved, That any attempt to preserve the Union between the States of this Confederacy by force would be impracticable and destructive to republican liberty.

Mr. STANTON moved to lay this upon the table, which was agreed to-yeas 98, nays 55, as follows :

YEAS—Messrs. Adams of Mass., Adams of Ky., Adrain, Aldrich, Allen, Babbitt, Beale, Bingham, Blair, Blake, Bray-ton, Briggs, Buffinton, Burlingame, Burnham, Campbell, Carey, Case, Clemens, Colfax, Conkling, Ox, Davis of Ind., Dawes, Duell, Edgerton, Eliot, Ely, Etheridge, Farnsworth, Fenton, Foster, Fouke, French, Gooch, Grow, Hale, Harris

of Md., Hatton, Helmick, Hill, Holman, Howard of Ohio, Howard of Michigan, Humphrey, Hutchins, Junkin, Kel-logg of Mich., Kenyon, Kilgore, Larrabee, Leach of Mich., Lee, Longnecker, Loomis, Lovejoy, Marston, Martin of Ohio, Maynard, McKeau, McKenty, McKnight, McPherson, Mont-gomery, Moore of Ky., Morris of Pa., Morris of Ill., Morse, Noell, Olin, Palmer, Pundleton, Perry, Pettit, Porter, Quarles, Reynolds, Robinson of Rhode Island, Robinson of U. Boyce Sherman, Snahlfur, Stanton, Stevens, Stokes.

Quarles, Reynolds, Robinson of Rhode Island, Robinson of Ill., Royce, Sherman, Spanlding, Stanton, Stevens, Stokes, Tappan, Thayer, Theaker, Vance, Vandever, Verree, Wade, Walton, Washburn of Wisconsin, Washburne of Illinois, Webster, Wilson, Windom-98. NAYS-Messre. Alley, Anderson of Mo., Ashley, Avery, Burksdale, Barrett, Bocock, Branch, Burch, Clark of Mo., Qinpton, Chob, Join Cochrane, Craig of Mo., Crawford, Curry, Curtis, De Jarnette, Dunn, Edmundson, Florence, Garnett, Gartrell, Hardeman, Harris of Ya., Hindman, Houston, Hughes, Jackson, Jenkins, Jones, Leach of N. O., Logan, Love, Murtin of Ya., McClernand, McRae, Moore of Ala., Niblack, Peyton, Pryor, Pugh, Riggs, Rust, Scott, Sickles, Singleton, Spinner, Thomas, Vallandigham, Wells, Whiteley, Winslow, Woodson, Wright-55.

NON-INTERFERENCE WITH SLAVERY IN THE STATES.

February 11th, 1861. Mr. SHERMAN of Ohio, offered the following :

Resolved, That neither the Congress of the United States, nor the people or governments of the non-slaveholding States have the constitutional right to legislate upon or interfere with slavery in any of the slaveholding States in the Union.

Which were agreed to-yeas 161, nays none.

None.
Trass-Messrs. Charles F. Adams, Green Adams, Aldrich, Allen, Thomas L. Anderson, William C. Anderson, Arery, Babbitt, Barr, Barrett, Bingham, Blair, Blake, Bocock, Boteler, Brabson, Brayton, Briggs, Bristow, Brown, Buffin-ton, Burch, Burlingame, Burnett, Butterfield, Campbell, Carey, Carter, Case, Horace F. (Uark, John B. Olark, Coburn, John Cochrane, Colfax, Conkling, Corwin, Covode, Cox, Jas. Graig, Curtis, Dawes, Delano, Diamnick, Duell, Dunn, Ed-mundson, Edwards, Eliot, Ely, English, Etheridge, Farns-worth, Ferry, Foster, Frank, Garnett, Gilmer, Gooch, Gra-ham, Grow, Gurley, Hall, Hamilton, J. Morrison Harris, John T. Harris, Hatton, Helmick, Hoard, Holman, William Howard, William A. Howard, Hughes, Hunph-rey, Hutchins, Irvine, Jenkins, Junkin, Francis W. Kel-ogg, Kenyon, Killinger, Larrabee, Leake, Lee, Lovejoy, Maday, Mallory, Marston, Elbert S. Martin, Maynard, McClernand, McKeany, Jaban T. Moore, Moorhead, Morrill, Edward Joy Morris, Isaac N. Morris, Nelson, Nib-dack, Nixon, Noel, Palmer, Perry, Pettit, Fytyon, Phelps, Porter, Potter, Pottle, Pryor, Quarles, John H. Reynolds, Strice, Riggs, Christopher Roblinson, Royce, Soranton, Sedewick, Sherman, Sickles, William N. H. Smith, Spaulding, Spinner, Stanton, Skevenson, William Stewart, Stokes, Stoud, Varaton, Madron, Waton, Cadwalader C. Washburne, Ellihu & washburne, Webster, Wells, Whiteley, Wilson, Windom, Winslow, Woodson-161. Nave-None.
BILL TO RE-ORGANIZE THE MILLTIA OF THE

BILL TO RE-ORGANIZE THE MILITIA OF THE DISTRICT OF COLUMBIA.

January 28th, 1860. Mr. STANTON of Ohio, reported from the Military Committee a bill providing that every officer, non-commissioned officer, musician, and private of the militia of the District of Columbia, shall take and subscribe to the oath required by the eighteenth section of the act approved January 11, 1812, the name of any person refusing to be stricken from the roll, or if an officer his commission to be forfeited.

The rules were suspended-yeas 116, nays 41, to permit its introduction, and it was finally passed—yeas 120, nays 42, as follows:

YEAS-Messrs. Charles F. Adams, Green Adams, Adrain, Aldrich, Alley, William C. Anderson, Ashley, Babbitt, Beale, Bingham, Blair, Blake, Brayton, Briggs, Bristow, Bufinton,

Burnham, Butterfield, Campbell, Carey, Case, Coburn, Col-fax, Conkling, Covode, Curtis, H. Winter Davis, John G. Davis, Dawes, Delano, Duell, Edgerton, Eliut, Etheridge, Fenton, Foster, Frank, French, Gilmer, Gooch, Grow, Gurley, Hale, J. Morrison Harris, Helmick, Hoard, Holman, Wil-liam Howard, William A. Howard, Humphrey, Hutchins, Irvine, Junkin, Francis W. Kellogg, William Kellogg, Ken-yon, Killinger, De Witt C. Lench, Lee, Longnecker, Loomis, Lovejoy, Mallory, Marston, Charles D. Martin, McClernand, McKean, McKnight, McPherson, Montgomery, Laban T. Moore, Moorhead, Morrill, Edward Joy Morris, Isaac N. Morris, Norse, Nelson, Niblack, Nixon,Olin, Palmer, Perry, Potter, Pottle, Edwin R. Reynolds, John H. Reynolds, Rice, Stewart, Stokes, Stratton, Tappan, Thayer, Thecker, Tomp-tins, Train, Trimble, Yandever, Yan Wyck, Wade, Waldron, Welton, Cadwalader C. Washburn, Ellihu B. Washburne, Webster, Wells, Wilson, Windom, Woodr, Woodruff-119. NAYS-Messrs. Avery, Bocock, Boteler, Branch, Brown,

NAXS-Messrs, Musch, Whath, Wood, Wood, Wood, Hostin-19. NAXS-Messrs. Avery, Boccok, Boteler, Branch, Brown, Burnett, Horace F. Clark, John B. Clurk, James Craig, Burtom Craige, De Jarnette, Edmundson, Florence, Fouke, Garnett, John T. Harris, Halton, Hindman, Hughes, Kun-kel, James M. Leach, Leake, Logan, Maclay, Elbert S. Mar-tin, Maynard, McKenty, Millson, Pendleton, Peyton, Pryor, Quarles, James C. Robinson, Scott, Simms, William N. H. Smith, Stevenson, Thomas, Vance, Whiteley, Winslow, Wrinht-42. Wright-42.

The bill was debated in the Senate, recommitted, re-reported, but objected to by Mr. Mason, Mr. Green, and others, and not finally disposed of.

#### BILL TO SUPPRESS INSURRECTION.

February 18th. Mr. STANTON of Ohio, reported from the Committee on Military affairs a bill enacting :

That the provisions of an act approved the 28th day of February, in the year 1795, entitled "An act to provide for calling forth the militia to execute the laws of the Union, suppress insurrections, and repel invasions, and to repeal the act now in force for those purposes," and of the act approved the 3d day of March, in the year 1807, entitled "An act authorizing the employment of the land and naval forces of the United States in cases of insurrections," are hereby extended to the case of insurrections against the authority of the United States.

That the President in any case in which it may be lawful to use either the militia or the military and naval force of the United States for the purpose aforesaid, may accept. the services of such volunteers as may offer their services, as cavalry, infantry, or artillery, organized in companies of the maximum standard, squadrons and regiments, respectively, according to the mode pre-scribed for the organization of the respective arms in the military establishment of the United States; and it shall be lawful for the President to commission the officers of such companies, battalions, squadrons, and regiments, in their respective grades, to continue till discharged from the service of the United States; and such volunteers, while in the service of the United States, shall be subject to the rules and articles of war, and shall be entitled to same pay and emoluments as officers and soldiers of the same grade in the regular service.

A motion to reject the bill was negatived -yeas 68, nays 109, as follows:

Ellihu B. Wa Woodruff—109.

19th. A motion to lay it on the table was negatived-yeas 68, nays 105. The yeas were:

YEAS—Messrs. Adrain, Allen, Thomas L. Anderson, Wm. C. Anderson, Avery, Barr, Bocook, Boteler, Bonligny, Brabson, Branch, Bristow, Burch, Burnett, Horace F. Clark, John B. Clark, John Cohrane, Cox, James Craig, Burton Craige, John G. Davis, De Jarnette, Edmundson, English, Etheridge, Florence, Fouke, Garnett, Gilmer, Hamilton, J. Morrison Harris, John T. Harris, Hatton, Hindman, Wm. Howard, Hughes, Kunkel, Larrabee, James M. Leach, Haclay, Mallony, Charles D. Martin, Elbert S. Martin, Maynard, McKenly, Millson, Laban T. Moore, Nelson, Nib-lack, Noell, Peyton, Pryor, Quarles, Riggs, James C. Robin-son, Ruflin, Rust, Scott, Simms, Stokes, Stout, Thomas, Val-landighum, Vance, Winslow, Woodson, Wright—68.

26th. On motion of Mr. Corwin of Ohio, the bill was postponed till the 28th, to take up the report of the Committee of Thirtythree, which was agreed to-yeas 100, nays 74, as follows :

74, as follows: YEAS-Messrs. Green Adams, Adrain, Thos. L. Anderson, Wm. C. Anderson, Avery, Barr, Burrett, Bocock, Boteler, Bouligny, Erabson, Branch, Briggs, Bristow, Brown, Burch, Burnett, Campbell, Horaee P. Clark, John B. Clurk, John Cohnane, Corwin. Coz, Jas. Craig, Burton Croige, 11. Winter Davis, John G. Davis, De Jarnette, Edmundson, Etheridge, Florence. Funke, Garnett, Glimer, Hale, Hall, Hamilton, J. Morrison Itarris, Juo. T. Harris, Hatton, Holman, Wa. Have-ard, Hughes, Jenkins, Junkin, W. Kellogg, Kilgore, Kunkel, Larrabe, James M. Leach, Locak, Logan, Malouy, Charles D. Martin, Elbert S. Martin, Maynard, McClernand, McKenty, McKnight, McPherson, Millson, Milward, Monl Morrison, Lawar, Noore, Moorhead, Edward Joy Morris, Isaac N. Morris, Nelson, Niblack, Nixon, Noell, Peyton, Phelps, Pryor, Quarles, John H. Reynolds, Rigol, James C. Robinson, Rufin, Scranton, Siekles, Simms, William N. H. Sith, Scenson, James A. Stevart, Ym. Stewart, Stokes, Stout, Stratton, Thayer, Thomas, Vallandigham, Vanee, Vertee, Wohsers, Charles F. Adams, Alley, Ashley, Babbitt, Beale, Binghum, Blair, Blake, Brayton, Buffinton, Furlin-game, Curthas, Delanc, Duell, Edgerton, Edwards, Flotz, J., Farasworth, Fenton, Cherry, Goech, Grow, Gurley, Haskin, Hickman, Hoard, William A. Howard, Hutchins, Francis W. Kellogg, Kenyon, De Witt C. Leeol, Grow, Gurley, Haskin, Hickman, Hoard, William A. Howard, Hutchins, Francis W. Kellogg, Kenyon, Cadwalder C. Washburn, Elhe, Bill zoog ned econin, mached

-74.

The bill was not again reached.

 TEAS—MESSIS. Thomas L. Anderson, William C. Anderson, Avery, Barr, Burrett, Bocock, Bouligny, Brabson, Avery, Barr, Burrett, Bocock, Bouligny, Brabson, Tranch, Birstow, Brone, Burrch, Burnett, Honce F, Class, Alexander, Sond, Schurt, Glimer, J. Morrison Barret, Garnet, Gilmer, J. Morrison Harris, John T. Hurris, Hatton, Jihadman, William Howard, Hughes, Kunkel, Laurabe, James C. Robinson, Rieflin, Rust, Stot, Schur, Sullian N. H. Snith, Schur, Pytor, Quarles, James C. Robinson, Rieflin, Rust, Thore, State N. Morris, Nelson, Nillack, Noul, Pryton, Philps, Pryor, Quarles, James C. Robinson, Rieflin, Rust, Toore, State N. Morris, Nelson, Nillack, Noul, Pryton, Philps, Pryor, Quarles, James C. Robinson, Rieflin, Rust, Toore, State N. Morris, Nelson, Nillack, Noul, Pryton, Philps, Pryor, Quarles, James C. Robinson, Rieflin, Rust, Toore, State, Simms, William N. H. Snith, Stokes, Skons, Sullian N. H. Snith, Stokes, Stot, Scher, Sullian, Kenke, Burnham, Batterfield, Campbell, Curey, Carter, Case, Colfax, Conking, Conway, Orrigheres, Kulker, Elong, William A. Howard, Hutching, Krink, Frencis, K. Kellogy, William K. Howard, Hutching, Krink, Frencis, K. Kellogy, William K. Howard, Hutching, Krink, Frencis, K. Kellogy, William K. Howard, Hutching, Krink, Frencis, Kellogy, William K. Howard, Hutching, Krynok, Kilgore, Killinger, Petti, Porter, Potte, Caten, Ackeng, Krinker, Kommer, Meilog, William K. Kellogy, William K. Howard, Hutching, Krink, Frencis, W. Kellogy, William K. Howard, Hutching, Krink, Frencis, W. Kellogy, William K. Howard, Hutching, Krink, Frencis, K. Kellogy, William K. Howard, Hutching, Krink, Krencis, K. Kellogy, William K. Howard, Hutching, Krink, Kenne, Krencis, K. Kellogy, William K. Kashbart, Krencik, Kiler, Ander, McKnigh, Kash, Krencik, Kater, Kashbart, Kashal Beak, Barno, ing; and that, in such cases, it shall be unlawful to take the vessel or cargo from the custody of the proper officer of the customs unless by process from some court of the United States; and that, in case of any attempt otherwise to take such vessel or cargo by any force, or combination, or assemblage of persons, too great to be overcome by the officers of the customs, it shall be lawful for the President of the United States or such person or persons as he shall have empowered for that purpose, to employ such part of the land or naval forces, or militia of the United States, as may be deemed necessary for the purpose of preventing the removal of such vessel or cargo, and protecting the officers of the customs in retaining the custody thereof.

March 2d. On a motion to take up the bill, by a suspension of the rules, the yeas were 103, the nays 62, as follows :

Tob, the hays 6., us follows.
Tyras—Messrs. Charles F. Adams, Adrain, Aldrich, Alley, Ashley, Babbitt, Beale, Bingham. Blake, Brayton, Briggs, Bnflinton, Burlingame, Butterfield, Carey, Carter, Colurn, Colfax, Conkling, Conway, Covode, Dawes, Delano, Duell, Dunn, Edgerton, Eliot, Ely, Farnsworth, Fenton, Ferry, Foster, Frank, French, Gooch, Grabam, Grow, Gurley, Hale, Hall, Haskin, Helmick, Hickman, Hoard, Hoiman, Humph-rey, Hutchins, Irvine, Junkin, Francis W. Kellogg, William Kellogg, Kenvon, Kilgore, Killinger, De Witt C. Leach, Lee, Longnecker, Loomis, Lovejoy, Marston, McKean, McKnight, MePherson, Millward, Edward Joy Morris, Nixon, Olin, Palmer, Pettiti, Porter, Potter, Potte, Rice, Christopher Robinson, Royce, Sedgwick, Sherman, Somes, Spaulding, Spinner, Stanton, Stevens, William Stewart, Stratton, Tap-n, Thayer, Theaker, Tompkins, Train, Trimble, Yandever, Yan Wyck, Verree, Wade, Waldron, Walton, Cadwalader C. Washburn, Ellihu B. Washburne, Wells, Wilson, Windom, Wood, Woodruff-103. Wood, Woodruff-103.

Wood, Woodruff-103.
NATS-Messrs. William C. Anderson, Barr, Barrelt, Brabson, Branch. Bristow, Brown, Burch, Burnelt, Horace F. Clark, John B. Clark, Clemens, John Cochrone, Corwin, Cox, James Craig, John G. Davis, De Jarnette, Edmundson, Etheridge, Florence, Fouke, Garnett, Gilmer, Hamilton, J. Morrison Harris, John T. Harris, Hattou, William Howard, Hughes, Kunkel, Larrabee, James M. Leach, Laake, Mallory, Charles D. Martin, Elbert S. Martin, Maynard, McKenty, Charles D. Martin, Elbert S. Martin, Maynard, McKenty, Killson, Laban T. Moore, Nelson, Niblack, Pendleton, Pey-ton, Phelps, Pryor, Quarles, Rigos, James C. Robinson, Simms, William N. H. Smith, James A. Stewart, Stokes, Thomas, Fallandigham, Vance, Webster, Whileley, Winslow, Woodson, Wright-62.

Two-thirds not having voted in the affirmative, the bill was not taken up, nor was it again reached.

January 30th, 1861. Mr. JOHN COCHRANE from the Select Committee on the President's message reported a bill for the "Collection of Duties on Imports," which was ordered to be printed and re-committed to the committee. It was not again heard from. His views of the subject were expressed as follows:

While fully concurring with the President in the opinion that no State possesses, under and by virtue of the Federal Constitution, any right or authority to secede, or with-draw, or separate itself from the Federal Union, I am equally convinced that, not having been prohibited, the justification of the exercise of such right is referable to the nature and extent of those rights reserved to the States or the people thereof. Therefore all the acts and ordinances of secession which have been or may be enacted by any State or States, acting separately, in my opinion, are, in as far as the same may be carried into effect, to be considered as revolutionary infractions of the supreme law of the land, however they may be regarded as the proper exercise of an indefeasible right of "resisting acts which are plainly unconstitutional and too oppressive to be endured."

I also concur with the President that the Federal Constitution has abstained from conferring upon the Federal Government, or any department thereof, authority to declare and wage oppressive war against a seceding State, in order to coerce the repeal of any act or ordinance of secession which she may have passed, or the renunciation of any purpose of secession which she may entertain, or to compel her to remain nominally, as well as in fact, a member of the Federal Union. On the contrary, the plenary power of offensive war and reprisals, conferred by the Constitution upon Congress, is, in my judgment, designed exclusively to authorize and empower such war by the Federal Government, in its discretion, against such governments and communities as may be rightfully considered foreign to the United States. States which profess to have seceded from the Federal Union, by their separate State action, cannot, in my opinion, be rightfully so considered; and, therefore, a just conception of the constitutional authority of Congress combines with other and, if possible, higher and more commanding motives to prescribe other measures than aggressive and coercive wars to remedy the grave inconveniences, perils, and evils of such secession.

It is, on the other hand, entirely manifest that the Federal Government, throughout the whole extent of its constitutional jurisdiction, both territorial and maritime, is vested with powers which it cannot surrender, and charged with trusts and duties

which it must perform. It holds valuable property in every part of the territory over which its jurisdiction has been, by the solemn mandate of the people of all the States, extended; and it holds every article of this property as the common trustee of all the people of all the States for their common use and benefit; and to them, and to all of them, it is directly responsible for the safe keeping and protection of such property. By constitutional legislation, supreme in its character and irrepealable except by the authority which created it, the Federal Government has established a system of revenue laws which it is bound to maintain, and against all obstructions to the execution of which it is equally bound adequately to provide. No extra constitutional action of any State can possibly release the Federal Gov-ernment in any of its departments from this imperative obligation.

It is to this aspect of the duty of Congress in the unfortunate and unexampled state of facts presented for its consideration by the President, in his late message, that the attention of the committee has been specially directed, and they report herewith a bill designed, as is believed, peaceably and in a spirit of moderation and forbearance fully to protect the entire revenue system of the United States from all the unconstitutional and unlawful obstructions and disturbances with which it is now or may be hereafter threatened.

In framing this measure there has been kept steadily in view, first, the obstacles of every character which oppose any attempt by the Federal Government to coerce a State; and, secondly, the principle upon which, as is conceived, the whole coercive action of our revenue system has, from its inception, been founded.

The just and rightful coercion exercised under the Constitution by the Federal Government in the collection of its revenue from foreign commerce is a coercion, not directed against State authority, nor even against unlawful action by assemblages of persons within any State. That coercion on the contrary, applies itself directly to the bulk and body of the foreign importations upon which the revenue is chargeable, and to the vessels and to the agents by which they are brought into our ports, opened to such commerce only under the protection and by the regulation of the Federal law.

The well-understood and just assumption upon which all this foreign commerce proceeds is, that the Federal Government provides for its ports of entry and deliver in which it may be entered and delivered, and in which the lawful duties chargeable upon it shall be collected and paid to the Federal Government, whose laws have, for that very consideration, established and protected and regulated such entry and delivery. The failure to observe these regulations is, by our whole code of revenue laws, visited in

penalties upon the agents by whom the | when, where, and with whom it was made, merchandize is transported, the vessel in which it is carried, and finally in the form of seizure and forfeiture upon the merchan-All this is a coercion of law dise itself. upon foreign commerce, and not a coercion of force upon any State to which it may be bound. No State can rightfully expect that the United States shall permit foreign commerce to be carried on in ports and places open to it only under their authority without compliance with the constitutional condition under which these ports or places have been thus opened. It is in this view that the bill proposes that vessels from a foreign port bound to a port within the scope of its provisions shall, with its foreign cargo, be liable to seizure and condemnation, and in the same view of applying its restrictions only to foreign commerce on which revenue is by law collected, all vessels lawfully engaged in the coastwise trade are exempted from the operation of its purely remedial and defensive provisions.

# REPORTS RELATIVE TO MILITARY AND NAVAL MATTERS.

January 8th, 1861. The President sent a message relative to the condition of the country, urging upon Congress to devote itself exclusively to the question how the Union can be preserved in peace, and recommending the establishment of a line and "letting the North have exclusive control of the Territory above it, and giving Southern institutions protection below it." He transmits copies of the correspondence between himself and the "commissioners" of South Carolina, pledges himself to preserve the public peace in the District of Columbia, warns his countrymen of the dangers which now surround us, and says that he shall carry to his grave the consciousness that he at least meant well for his country.

9th. Mr. HOWARD of Michigan, offered the following resolution :

*Resolved*, That the message be referred to a special committee of five members of this House, and that they be instructed to report on the same as early as possible; and that said committee make immediate inquiry, and report:

1st. Whether any executive officer of the United States has been or is now treating or holding communication with any person or persons concerning the surrender of any forts, fortresses, or public property of the United States, and whether any demand for such surrender has been made, when and by whom, and what answer has been given.

2d. Whether any officer of this Government has at any time entered into any pledge, agreement, or understanding, with any person or persons, not to send reinforcements to the forts of the United States in the harbor of Charleston, and the particulars of February 14th. The committee reported such agreement, pledge, or understanding; unanimously that "the evidence produced

and on what consideration.

3d. What demand for reinforcements of the said forts has been made, and for what reason such reinforcements have not been furnished.

4th. Where the ships of the United States are now stationed, with what commands, and with what orders.

5th. Whether the custom-house, post-office, arsenal, and other public buildings of the United States at Charleston have been seized and are held in possession by any person or persons, and the particulars of such seizure and possession.

6th. Whether any revenue cutter of the United States has been seized and is now held in possession by any person or persons, and the particulars thereof; and whether any efforts have been made by the head of the Treasury Department to recapture or recover possession of said vessel.

That the committee have power to send for persons and papers, to take testimony, and report from time to time, as facts material to the national safety and national honor may be disclosed by the evidence.

Which was adopted—yeas 133, nays 62, as follows:

as follows: YEAS-Messrs. Charles F. Adams, Green Adams, Adrain, Aldrich, Allen, Alley, William C. Anderson, Ashley, Bab-bitt, Beale, Bingham, Blair, Blake, Brayton, Briggs, Bris-tow, Buffinton, Burch, Burtingame, Burnham, Butterfield, Campbell, Carey, Carter, Clemens, Coburn, John Cohrane, Colfux, Covode, Cor, Curtis, John G. Daviz, Dawes, Delano, Duell, Dunn, Edgerton, Edwards, Eliot, Ely, Eiheridge, Farnsworth, Fenton, Fery, Foster, Fouke, Frank, French, Cilmer, Gooch, Grow, Gurley, Hale, Hall, Hamilton, Haskin, Helmick, Hickman, Hoard, Holman, Wm. Howard, Wm. A. Howard, Humphrey, Hutchins, Irvine, Junkin, Francis W. Kellogg, Wm. Kellogg, Kenyon, Kilgore, Larrabee, De Witt C. Leach, Lee, Logan, Longnecker, Loomis, Lovejoy, Maclay, Marston, McClernand, McKean, McKnight, McPherson, Millward, Mandgomerg, Moorhead, Morrill, Edward Joy Morris, Morse, Nelson, Nixon, Neell, Olin, Palmer, Perry, Pettit, Porter, Pottle, Edwin R. Reynolds, John H. Reynolds, Byinner, Stanton, Stevens, Wm. Stewart, Stokes, Stratton, Tappan, Thayer, Theaker, Tompkins, Train, Trimble, Van-dever, Wade, Waldron, Cadwalader C. Washburn, Elihu B. Washburne, Webster, Wells, Wilson, Windom, Wood, Wood-ruf-133. Nars-Mesers, Thomas L. Anderson Avern Barbedale ruff-133.

ruff-133. NATS-Messrs. Thomas L. Anderson, Avery, Barksdale, Barrett, Bocock, Boteler, Bouligny, Branch, Brown, Burnett, John B. Clark, Clopton, Cobb, Crauford, De Jarnette, Dim-mick, Łdmundson, English, Florence, Garnett, Gartrett, Hardeman, John T. Harris, Hatton, Hawkins, Hill, Hou-ton, Hughes, Jones, Kunkel, Landrum, James M. Leach, Leake, Love, Mallory, Charles D. Mortin, Elbert S. Martin, Maynard, McKenty, McRae, Millson, Laban T. Moore, Syden-ham Moore, Niblack, Pendleton, Peylon, Phelps, Pryon, Quarles, Reagan, Rust, Sickles, Simms, Wm. Smith, Wm. N. H. Smith, Taylor, Vallandigham, Vance, Whiteley, Winslow, Woodson, Wrighl-62.

January 10th. The Committee was announced, consisting of Messrs. Howard of Michigan, BRANCH of North Carolina, JOHN COCHRANE of New York, DAWES of Massachusetts, and HICKMAN of Pennsylvania. Mr. Hickman was subsequently excused from serving, and Mr. JOHN H. REYNOLDS of New York, was substituted.

# SECRET ORGANIZATION TO ATTACK THE CAP-ITAL.

before them does not prove the existence of a secret organization, here or elsewhere, hostile to the Government, that has for its object, upon its own responsibility, an attack upon the Capital, or any of the public prop-erty here, or an interruption of any of the functions of the Government."

The majority of the committee, however, state that certain organizations in the District and in Maryland, formerly political clubs have since become military and are drilling, expecting arms from State authorities or private subscriptions; but that, while they sympathize strongly with secession, there is no proof that they intend to attack the Capital or the District, "unless the surrender should be demanded by a State to which they profess a higher degree of allegiance." "Some of these companies in Baltimore profess to be drilling for the sole purpose of preventing other military companies from passing through the State of Maryland." These clubs are, in no proper sense, secret, and hence not such as are contemplated in the resolution of the House.

Mr. BRANCH of North Carolina, added a brief minority report, chiefly to the point that Lieutenant-General Scott has seven companies of artillery, and one company of sappers and miners of the regular army ordered to and quartered in the City of Washington, and closing with this resolution, which he offered:

Resolved, That the quartering of troops of the regular army in this District and around the Capitol, when not necessary for their protection from a hostile enemy and during the session of Congress, is impolitic and offensive, and, if permitted, may be destructive of civil liberty; and, in the opinion of this House, the regular troops now in this city ought to be forthwith removed therefrom.

On motion of Mr. JOHN COCHRANE, this resolution was laid on the table-yeas 125, nays 35, as follows :

nays 35, as follows: YEAS-Messrs. Charles F. Adams, Green Adams, Aldrich, Aller, Alley, William C. Anderson, Ashley, Babbitt, Beale, Bingham, Blair, Blake, Brayton, Briggs, Bristow, Bufinton, Burlingame, Butterfield, Campbell, Carter, Case, Horace F. (Cark, Coburn, John Cochrane, Colfax, Conkling, Conway, Ovode, Cox, Curtis, H. Winter Davis, John G. Davis, Dawes, Duell, Dunn, Edgerton, Eliot, Ely, Etheridge, Farns-worth, Fenton, Ferry, Foster, Fouke, Frank, French, Gilmer, Gooch, Graham, Grow, Hale, Hall, Huskim, Hatton, Helmick, Hoard, Holman, W. A. Howard, Humphrey, Hutchins, Irvine, Junkin, F. W. Kellogg, W. Kellogg, Kenyon, Kilgore, Larra-bee, De Witt C. Leach, Lee, Logan, Longnecker, Loomis, Lovejoy, Mallory, Marston, McClernand, McKean, McKenty, McKnight, McPherson, Millson, Laban T. Moore, Moore, Nel-son, Niblack, Nixon, Palmer, Perry, Porter, Potter, Pottle Schwin R. Reynolds, Rice, Christopher Robinson, James C. Robinson, Royce, Scranton, Sedgwick, Sherman, Sickles, Strattou, Tappan, Theaker, Tompkins, Train, Vandever, Van Wyck, Wade, Waldon, Walton, Cadwalader C. Wash-unr, Elihus B. Washburne, Webster, Wells, Wilson, Wood-ruf-125. burn, Ell ruff—125.

NAYS-Messrs. Avery, Barr, Barrett, Bocock, Boteler, Bouligny, Brabson, Branch, Burch, Burnett, John B. Clark, Hurton Oraige, De Jarnette, English, Florence, Garnett, John T. Harris, Hindman, Hughes, Kunkel, Maclay, Elbert S. Martin, Maynard, Peyton, Phelps, Pryor, Quarles, Rust, Scott, William N. H. Smith, Stevenson, Thomas, Whiteley, Winslow, Winslow, Barb.

## SECRETARY HOLT'S REPORT.

On this subject, the Secretary of War, under date of February 18th, 1861, made report upon a resolution of inquiry passed by the House of Representatives :

by the House of Representatives : This resolution having been submitted to this De-partment for consideration and report, I have the honor to state, that the body of troops temporarily transferred to this city is not large, as is assumed by the resolution, though it is a well-appointed corps and admirably adapted for the preservation of the public peace. The reasons which led to their being assembled here will now be briefly stated. I shall make no comment upon the origin of the revolution which for the last three months has been in progress in several of the Southern States, nor shall lenumerate the causes which have hastened its advancement or exasperated its temper. The scope of the question submitted by the House will be sufficiently met by dealing with the facts as they have proceeded. That revolution has been distin-guished by a boldness, and completeness of success, rarely equalled in the history of civil commotions. Its overthrow of the Federal authority has not only been sudden and wide-spread, but has been marked by excesses which have alarmed all, and been sources of profound humiliation to a large portion of the American people. Its history is a history of sur-prises and treacheries, and ruthless spoliations. The forts of the United States have been captured and garrisoned, and hostile flags unfurled upon their ramparts. Its arsenals have been sequend and garrisoned, and hostile flags unfurled upon their amput of public arms they contained appropriated to the use of the captors; while more than half a million dollars, found in the mint at New Orleans, has been unscrupulously applied to replenish the coffers of Louisiana. Officers in command of rev-enue cutters of the United States have been prevailed to the is charge; and instead of being branded for on to violate their trusts and surrender the property on to violate their trusts and surrender the property in their charge; and instead of being branded for their crimes, they and the vessels they betrayed have been cordially received into the service of the seceded States. These movements were attended by yet more discouraging indications of immorality. It was genbeen cordially received into the service of the seceded States. These movements were attended by yet more discouraging indications of immorality. It was gen-erally believed that this revolution was guided and urged on by men occupying the highest positions in the public service, and who, with the responsibilities of an oath to support the Constitution still resting upon their consciences, did not hesitate secretly to plan and openly to labor for the dismemberment of evil are always more potent than those of good, this spectacle of demoralization on the part of States and statesmen could not fail to produce the most deplorable consequences. The discontented and the disloyal everywhere took courage. In other States, adjacent to and supposed to sympathize in sense of political wrong with those referred to, revolutionary schemes were set on foot, and forts and arms of the united States seized. The unchecked prevalence of the revolution, and the intoxication which its tri-umphs inspired, naturally suggested wilder and yet more desperate enterprises than the conquest of un-garisoned forts or the plunder of an unguarded mint. At what time the armed occupation of Washington city became a part of the revolutionary programme, is not certainly known. More than six weeks ago the impression had already extensively obtained that a conspiracy for the accomplishment of this guily murose the impression had already extensively obtained that a conspiracy for the accomplishment of this guilty purpose was in process of formation, if not fully ma-tured. The earnest endeavors made by men known to be devoted to the revolution to hurry Virginia and Maryland out of the Union, were regarded as preparatory steps for the subjugation of Washington. This plan was in entire harmony with the aim and spirit of those seeking the subversion of the Govern-ment, since no more fatal blow at its existence could be struck than the permanent and hostile possession of the seat of its power. It was in harmony to with the avowed designs of the revolutionists, which looked to the formation of a confederacy of all the slave States, and necessarily to the conquest of the looked to the formation of a confederacy of all the slave States, and necessarily to the conquest of the capital within their limits. It seemed not very in-distinctly prefigured in a proclamation made upon the floor of the Senate, without qualification, if not exultingly, that the Union was already dissolved—a proclamation which, however intended, was certainly calculated to invite, on the part of men of desperate fortunes, or of revolutionary States, a raid upon the capital. In view of the violence and turbulent disor

ders already exhibited in the South, the public mind could not reject such a scheme as at all improbable. That a belief in its existence was entertained by multitudes, there can be no doubt, and this belief I fully shared. My conviction rested not only on the facts already alluded to, but upon information, some of which was of a most conclusive character, that reached the Government from many parts of the country, not merely expressing the prevalence of the opinion that such an organization had been formed, but also often furnishing the plausible grounds on which the opinion was based. Superadded to these proofs were the oft-repeated declarations of men in indpeditical positions here, and who were known to have intimate affiliations with the revolution, if indeed they did not hold its reins in their hands—to the effect that Mr. Lincoln would not, or should not, be inaugurated at Washington. Such declarations from such men could not be treated as empty bluster. They were the solemn utterances of those who well understood the import of their words, and who, in the exultation of the temporary victories gained over their country's flag in the South, felt assured that events would soon give them the power to verify their predictions. Simultaneously with these prophetic warnings, a Southern journal of large circulation and influence, and which is published near the city of Washington, Advocated its seizure as a possible political necessity.

The nature and power of the testimony thus accumulated may be best estimated by the effect produced upon the popular mind. Apprehensions for the safety of the capital were communicated from points near and remote, by men unquestionably reliable and loyal. The resident population became disquieted, and the repose of many families in the city was known to be disturbed by painful anxieties. Members of Congress too, men of calm and comprehensive views and of undoubted fidelity to their country, frankly expressed their solicitude to the President and to this Department, and formally insisted that the defences of the capital should be strengthened. With such warnings, it could not be forgotten that, had the late Secretary of War heeded the anonymous letter which he received, the tragedy at Harper's Ferry would have been avoided; nor could I fail to remember that, had the early admonitions which reached here in regard to the designs of lawless men upon the forts of Charleston harbor been axted on by sending forward adequate reinforcements before the revolution began, the disastrous political complications that ensued might not have occurred.

Impressed by these circumstances and considerations, I earnestly besought you to allow the concentration at this city of a sufficient military force to preserve the public peace from all the dangers that seemed to threaten it. An open manifestation on the part of the Administration of a determination, as well as of the ability, to maintain the laws, would, I was convinced, prove the surest, as also the most pacific means of baffing and dissolving any conspiracy that might have been organized. It was believed, too, that the highest and most solemn responsibility resting upon a President, withdrawing from the Government, was to secure to his successor a peaceful inauguration. So deeply in my judgment did this duty concern the whole country, and the fair fame of our institutions, that to guarantee its faithful discharge I was persuaded no preparation could be too determined or too complete. The presence of the troops alluded to in the resolution is the result of the conclusion arrived at by yourself and Cabinet, on the proposition submitted to you by this Departiment. Already this display of life and loyalty on the part of your Administration has produced the happiest effects. Public confidence has been restored, and the feverish apprehension which it was so mortifying to contemplate has been banished. Whatever may have been the machinations of deluded, lawless men, the execution of their purpose has been suspended, if not altogether abandoned, in view of preparations which announce, more impressively than words, that this Administration is alike able else troops is necessarily offensive; but those who sincerely love our institutions cannot fail to rejoice thas troops is necessarily offensive; but those who sincerely love our institutions cannot fail to rejoice that by this timely precaution they have possibly escaped the deep dishonor which they must have suffered had the capital, like the forts and arsenals of the South, fallen into the hands of the revolutionists, who have found this great Gov

spirit it has refused to strike, even in its own defence, lest it should wound the aggressor.

I have the honor to be, very respectfully, your obcdient servant, J. Holt, The President. Secretary of War.

#### ON THE DISPOSITION OF UNITED STATES VESSELS.

February 21st. Mr. DAWES from the committee made a report upon the fourth item of investigation, furnishing these statements:

Dismantled ships 28, mounting 874 guns; none could be repaired under several weeks, and many would require six months. No orders have been issued to put any of them in readiness.

The vessels in the East India, Brazil, Pacific, Mediterranean, African, and Home Squadrons are named, with those on special service, and on the way to stations, and the committee then say:

That the entire naval force available for the defence of the whole Atlantic coast, at the time of the appointment of this committee, consisted of the steamer Brooklyn, twenty-five guns, and the storeship Relief, two guns; while the former was of too great draught to permit her to enter Charleston harbor with safety, except at spring tides, and the latter was under orders to the coast of Africa with stores for the African squadron. Thus the whole Atlantic seaboard has been to all intents and purposes without defence during all the period of civil commotion and lawless violence, to which the President has called our attention as "of such vast and alarming proportions" as to be beyond his power to check or control.

It further appears that of the vessels which might have been available for protection or defence in case of any sudden emergency arising at home, now at stations in distant seas, or on the way thither, on the 13th of October last the Richmond left our coast to join the Mediterranean squadron; the Vandalia left on the 21st of December to join the East India squadron; and about the same time the Stratoga to join the African squadron, and others to join the Home squadron, then in the harbor of Vera Cruz supporting one of the revolutionary governments of Mexico.

To the Committee, this disposition of the naval force at this critical time seems most extraordinary. The permitting of vessels to depart for distant seas after these unhappy difficulties had broken out at home; the omission to put in repair, and commission ready for orders, a single one of the twenty-eight ships dismantled and unfit for service in our own ports, and that too while \$646,639.79 of the appropriations for repairs in the Navy the present year remained unexpended, were in the opinion of the Committee grave errors in the administration of the Navy Department—the consequences of which have been manifest in the many acts of lawless violence to which they have called attention. The Committee are of opinion that the Secretary had it in his power, with the present naval force of the country at his command, and without materially impairing the efficiency of the service abroad, at any time after the settled purpose of overthrowing the government had become manifest, and before that purpose had developed itself in overt acts of violence, to station at anchor, within reach of his own orders, a force equal to the protection of all the property and all the rights of the government and the citizen, as well as the flag of the country, from any outrage or insult, at any point on the entire Atlantic seaboard. The failure to do this is without justification or excuse.

The committee proceed respecting the acceptance of resignations in the navy:

The attention of the Committee was also drawn to the resignations which have taken place among the officers in the Navy, caused by the political troubles in which the country is now involved, and the course pursued by the Navy department in reference thereto. It will appear from a "List of resignations" furnished by the Department, and which accompanies this report, that since the election twenty-nine officers in the Navy, citizens of the southern dis-

affected States, have tendered their resignations to the Secretary, all of which have been forthwith and without inquiry accepted by him. The circumstances under which these resignations have been received and accepted, and the effect of that acceptance, de-serves especial notice. That these officers have sought to resign and relieve themselves from the ob-luction to the comparison that the obsought to resign and relieve themselves from the ob-ligation to the government imposed by their commis-sions, because of disaffection, and a desire to join, and in many instances to lead insurgent forces against that government, is notorious. One of them, Lieut. J. R. Hamilton, a citizen of South Carolina, forwarded his resignation from on board the Wyoming, at Pan-ama, dated December 1st, 1860. It did not reach the Department until the 15th of the same month, and without inquiry into his conduct, his purpose in re-signing, his loyalty, or any circumstances connected with so unusual a proceeding at such a time, his resignation was accepted the same day. He imme-diately, from Charleston, South Carolina, issued a letter addressed to all the officers in the Navy from Southern States, urging them to resign and join a hostile force against the government, and that those of them in command should bring with them their of them in command should bring with them their vessels into Southern ports and surrender them to the traitors already in arms, taking new commissions under their authority, and then turning their guns upon their own flag.

upon their own flag. Such conduct is nothing less than treason, and has no parallel since the attempt of Benedict Arnold to deliver over important military posts to the enemies of his country. Had the Secretary declined to accept the resignation thus tendered, this man would have been subject to the trial and punishment of a court-martial, according to the rules which govern the service, and would have met the fate of a traitor. This extraordinary letter was published throughout the United States. After its circulation in the public prints in Washington, V. M. Randolph, a captain in the Navy, a citizen of Alabama, who had been ex-cused from active service for two or three years, be-cause of alleged ill-health, on the 10th of January, 1861, forwarded from Montgomery, Alabama, his resignation to the Secretary. Before twelve o'clock at noon of the 12th, and before his resignation had reached Washington, and while he was still a cap-tain in the Navy, he appeared at the gates of the Pensacola navy-yard, in Florida, at the head of an insurgent force, and demanded its surrender. The yard, with whatever of force it had and the United States stores and other property to a vast amount therein, was unconditionally surrendered to him ; and he is now its commandant, occupying the quar-ters of its late commandant, and granting paroles of honor to such of his prisoners of war as have desired from the late commandant, then a prisoner of war, informing the Secretary of this ignominious surren-der, was received at the Department on the evening of the 13th of January; and the resignation of Cap-tain Randolph, who, on the 12th, was the leader of the insurgents, did not reach the Secretary till the 14th, when, without inquiry or delay, it was immediately accepted. Such conduct is nothing less than treason, and has when, without inquiry or delay, it was immediately accepted.

accepted. E. Farrand, commander in the Navy, and also a citizen of Alabama, was the second in command at the Pensacola navy-yard, the executive officer of the yard. When the attack was made upon the yard, Farrand met the assailants at the gates by previous understanding, admitted them to the yard, and con-ducted their leader to the commanding officer, parti-cipated in the formal capitulation, and immediately nergaged in service under the new commandant of the engaged in service under the new commandant of the yard. This was done while he still held in his pos-session his commission as a commander in the Navy. session his commission as a commander in the Navy. On the 13th or 15th of January, (the Department does not know which.) Farrand forwarded his resignation to the Secretary, but it did not reach him till the 21st of the same month, seven days after official notice of the surrender had been received at the Department. Yet this resignation was immediately, and without invariant.

F. B. Renshaw, a lieutenant in the Navy, and a citizen of Florida, was the first lieutenant of the yard, and actively engaged in securing its surrender. It was by his order that the flag was hauled down mid the jeers and shouts of a drunken rabble. He immediately enrolled himself under the leader of the immediately and additional securing the surrender of the immediately enrolled himself. Immediately enrolled himself under the leader of the insurgents, and present commandant of the yard, and from the day of its surrender has continued under him to discharge the duties of first lieutenant, as be-fore under the United States. Yet he continued to hold his commission as a lieutenant in the Navy till the the the United States. the 16th of January, and the resignation did not reach

the Secretary until the 22d, when, like the others, it was, without inquiry or delay, accepted. The conduct of these officers plainly comes within the constitutional definition of treason against the United States, namely: "Levying war against them, or in adhering to their enemics, giving them aid and comfort." And so long as their resignations were unaccepted by the Secretary, they could be tried and punished by a court-martial as traitors. From this they have been relieved by the Secretary himself. To have done this with a knowledge of their acts, would have been to have involved himself in their crime; would have been to have committed runner run term terms of to know that they have committed runner offence, shows a want of that solicitude for the honor and efficiency of the service which is indispensable to its just ad-ministration. Yet the resignation of Farrand and Renshaw, and also those of the other officers resign-ing at the Pensacola navy-yard, were all received and accepted after the Secretary had already been offi-cially informed that they had surrendered to a law-less band of insurgents, and he had detached them to await orders, having "neither approved nor disap-proved of their conduct, and not proposing to do so, without full information touching their conduct in the surrender of the yard." Why, after having been thus warned, and having taken his position, the Secre-tary did not wait for this "information," the com-tary did not wait for this "information," the comwould have been to have committed treason himself. mittee cannot understand.

mittee cannot understand. Several other resignations of officers who do not appear to have engaged in actual war against the United States before tendering the same, were never-theless accepted by the Secretary, with an unneces-sary haste which neither the purpose of the resigna-tion nor the times would justify or excuse. Some of them were even accepted by telegraph, when it was perfectly apparent that the object in resigning was to relieve themselves, as early as possible, from embar-rassment and the obligation of the oath of office, as well as summary trial and punishment by a court-martial, previous to joining insurgent forces against the constituted authorities of their country. These resignations, thus accepted, have been followed by immediate engagement in a service hostile to the Government. Government

One man, holding the office of civil engineer in the Pensacola navy-yard at the time of its surrender, forwarded his resignation on that day to the Secre-The letter of resignation of a matter of the second tary, inclosed in a letter to Senator Mallory, in which which it was surrounded, at all satisfactory to the committée.

The resignation of the officer in charge of the ma-The resignation of the officer in charge of the ma-rine hospital at Pensacola was accepted by telegraph, and he was thereby enabled to take upon himself the same position under the insurgent force without any interruption. And that of Lieutenant R. T. Chapman, dated on board the Brooklyn, when about to sail under orders, was likewise accepted by telegraph, and he was thus relieved of any inconvenience he would otherwise have experienced in being carried to sea against his wishes.

ried to sea against his wisnes. The course pursued by the Secretary, in thus accept-ing these resignations, appears, under the circum-stances, to be most extraordinary. No custom of the Department, in ordinary times, could justify it. No want of confidence in the loyalty of these officers can excuse it; for, if their previous conduct had justified any such suspicion, it also demanded investigation beforeband which would as to some of them base any such suspicion, it also demanded investigation beforehand, which would, as to some of them, have disclosed to the Secretary their complicity in trea-son, calling for court-martial rather than honorable discharge. A prudent regard for the public safety would no doubt have justified, if not imperiously de-manded, that some of these officers should have been early removed from delicate and responsible post-tions of trust by the substitution of others more retions of trust, by the substitution of others more re-liable. But these very considerations appear to the committee to have forbidden the furnishing of any

such facilities for engaging in hostilities/against the Government, as the relief from the summary trial and punishment of a court-martial, secured by an acceptance of their resignations.

The course pursued by the Secretary has resulted in furnishing those engaged in an attempt to overthrow the Government, with the skill, experience, and discipline which education at the expresse of the Government, and a long service in the Navy have conferred upon our own officers. The committee cannot understand how this course is consistent with a proper discharge of the duties of his office by the Secretary, in this critical juncture of affairs. It appears to them to have been attended with consequences the most serious to the service and the coun-They can find no excuse or justification in the claim trv. set up in behalf of the Secretary, that these resignations have been accepted in ignorance of any misconduct, for no resignation should at any time be accepted until there is reason to know at least that the officer tendering it had been guilty of no unofficerlike conduct deserving a courtmartial.

But the circumstances connected with these resigna but the circuits dates connected with these resigna-tions, the apparent purpose for which they were made, and the hostile attitude which the manner of their tender clearly disclosed, called upon the Secretary to refrain from that haste in their acceptance which permitted of neither delay nor inquiry. The committee cannot approve, but are compelled to condemn such a failure in the dis-

charge of public duty, and they therefore recommend the adoption of the following resolution: *Resolved*, That the Secretary of the Navy, in accepting without delay or inquiry, the resignations of officers of the Navy, who were in arms against the Government when tendering the same, and of those who sought to re-sign that they might he realized from the restriction. when tendering the safe of those who sought to restrain tim-posed by their commissions upon engaging in hostilities to the constituted authorities of the nation, has com-mitted a grave error, highly prejudicial to the discipline of the service, and injurious to the honor and efficiency of the Navy, for which he deserves the censure of this House.

Mr. BRANCH made a minority report, as follows:

An examination of the table accompanying the report, marked "List of the vessels of the United States Navy in commission on the 16th of January, 1861," will disclose the fact that, so far from there heing any ground to believe that the ships of the Navy have been purposely placed out of reach, the foreign squadrons are unusually weak, in pro-portion to the whole force of the Navy, and the home squadron unprecedentedly strong. In his annual report of 3d December, 1857, the Secretary of the Navy said: "The home squadron, under the command of its flag-officer, Hiram Paulding, has consisted of the steam-frigates Walash and Roanoke, the sloops-of-war Saratoga and Cy-ane, and the war steamers Susquelanna and Fulton. The unsatisfactory state of affairs in New Grenada and portions of Central America required the increase of this squadron, fact that, so far from there being any ground to believe that

of Central America required the increase of this squadron, and the almost constant presence of a considerable force in the neighborhood, both in the Atlantic and the Pacific.'

From which it will be seen that, at that time, six ships were considered so large a force for the home squadron as to call for an explanation from Congress. The same squadron, on the 16th of January, 1861, con-

stated of eleven ships, of which seven were steamers; for the eleven, eight were on the coast, or under orders for the coast, and have since arrived. It is difficult to perceive from what portion of the testimony the committee have reached the conclusion that the only ships available for the defense of the coast are the Procellum and one other. It is defence of the coast are the Brooklyn and one other. It is true, they were on the 16th of January the only vessels in our harbors; but the House need not be informed that vessels in commission are not allowed to lie idle in the harbors, Sets in commission are not allowed to be falle in the harbors, but are required to cruise on their stations, coming in occasionally for supplies and for orders. The home squad-ron, as its name imports, is inainly intended to guard and protect our own coasts, and on comparatively short notice it could be concentrated for service at a given point. It will be seen from the abstract of orders issued from the Navy Department, furnished to the committee, and doted dith Lonuary thet note single abile has been had a doted.

dated 24th January, that not a single ship has been ordered to any foreign station since the date of the Presidential election; every order since that date has been to increase the force at home.

The second ground on which the committee condemn the action of the Secretary is, that a large number of ships are dismantled and not ready for immediate service. It is well known that, from the very nature of the ser-rice, only a portion of even the serviceable ships can be

Note, only a portion of even the service and single can be kept in commission. In the midst of a foreign war with a naval power, some of the vessels of the Navy will always be in ordinary. There is nothing in the testimony to show that the number in ordinary at the present time is unusu-

ally large. If, however, it is unusually large, both the testimony and the records of the House furnish a sufficient reason for it. It is stated in the following extract from the paper already alluded to, dated 24th January: 2. All the vessels above named, except the Pawnee and the Constitution, are dismantled. Vessels of war are nearly

the Constitution, are dismantled. Vessels of war are nearly always dismantled immediately after their arrival in port, at the termination of a cruise. It is also customary to commence repairing them as soon as possible after their return, if the appropriations and other work will permit. The appropriations were reduced \$1,000,000 below the esti-mates of the Department; and of the whole sum appro-priated (\$1,523,000) for repairs, etc., for the fiscal year end-ing June 30, 1861, there remained on the 1st January, the close of the first half of the fiscal year, the sum of \$646,639.79 for the zemaining helf?" for the remaining half."

The secretary stated to the Committee that he deemed it his duty to conform to the action of Congress, and that in addition, having already expended more than half the ap-propriation in half the year, he did not think it just to his successor that he should continue expenditures until 4th. March on the same scale, much less to increase them.

A large majority of the members of this House voted at the last session to reduce the appropriation; and of course those who thus voted will not unite with the committee in censuring the Secretary for the inevitable consequence of their own act.

Another equally satisfactory reason for not placing more ships in commission, is found in the letter of 24th January, as follows:

"The other vessels mentioned in the list have not been put in condition for immediate service within the last month, because the number of vessels in commission is governed by the numerical strength of the officers and sea-men of the Navy."

And again, in a communication from the Navy Department dated February 4th, 1861:

"In reply to the second inquiry, there are enough cap-tains, commanders, surgeons, chaplains. In the other grades there are not, nor is there a sufficient number of seamen."

It will be remembered that the number of both officers and seamen to be employed is limited by law, and the Secretary of the Navy cannot exceed that limit.

The Secretary deserves no censure; but should receive the highest commendation for inflexibly obeying the law in the administration of his Department. Every attempt at retrenchment and economy is defeated because heads of Departments will not themselves conform to the law and compel their subordinates to do the same. The facility with which Congress supplies deficiencies created by disregarding the law, has encouraged such practices; and it would be mournful indeed, if in the first well-authenticated case in which a head of Department has faithfully and sternly conformed his expenditures to so radical a reduction of his estimates, the officer should receive a vote of censure instead of a vote of thanks.

stead of a vote of thanks. The Nary seems to have been adequate for all the de-mands made upon it by the wise and peace-preserving policy of the President. If the President who goes into office on the 4th March desires to engage in civil war, he will have an ample naval force with which to begin, even so early as the 5th of March; and there will probably be abundant time for increasing it before the war closes.

### II. RESIGNATION OF OFFICERS.

A list of all the officers of the Navy who have resigned, between the 11th of November and the 24th January, was furnished to the Committee, and the chief clerk of the Navy Department was fully examined as to the circum-

Navy Department was thing exchanged as to the ercom-stances aftending each resignation. The whole number was fifty-six, including eleven from the Naval Academy. It is known that in many, if not most of these cases, the officers, in resigning, have not only given up an honorable profession for which alone they were fitted by education and habits, but have reduced themselves and families to penury. Some powerful motive must have actnated them. If it was selfah, let any one point out a possible advantage they could promise themselves personally. It could not be that they aimed to recommend themselves to the favor and bey anice to recommend in the series to the avoid and patronage of the southern confederacy, for that government does not possess a ship, and cannot, for a long time, provide itself with a nary. When it has done so, it cannot be ex-pected that the officers will be in any better situation with reference to their personal interests than they would have heen if they had remained in the Navy of the United States.

States. It is evident that in resigning they have been actuated by a high eense of duty to the States of which they were respectively citizens; and that, in the time and manner of tendering their resignations, they have consulted a nice sense of honor. In a few instances they have eengaged in the military service of their States; but not until they had resigned their commissions in the Navy of the United States. In no instance does it appear that one of them has be-trayed the trust reposed in him by this government, or

engaged in any hostile service until he had discharged him-self of all the responsibilities imposed by his commission. The testimony taken before the committee, partial and limited as it is on this point, is confidently appealed to for the correctness of this statement.

correctness of this statement. The undersigned would gladly pursue this subject, grate-ful asit is to him to vindicate, against the very harsh asper-sions cast upon them by the committee, the characters of the honorable body of men who have so long and so gal-lantly borne the flag of the Republic. But it is foreigu to the inquiry the committee were instructed to make, and will for that reason be dismissed, except in so far as it is the basis of the resolution reported. The censure of the Sec-retary seems to be based principally, if not entirely, on his acceptance of the resignations of Captain Randolph and the two ficutenants on duty at the Pensacola navy-yard at the time of its surrender. time of its surrender.

time of its surrender. Captain Randolph's resignation was dated at Montgomery, Alahama, on the 10th of January, was received at the De-partment on the 14th, and accepted on the 14th. The Pen-sacola navy-yard was surrendered on the 12th, and Captain Armstrong immediately sent the following telegraphic des-patch, which was received by the Secretary of the Navy on the 13th, at eight o'clock, P.M.:

Hon. Isaac Toucey, Secretary of the Navy :

Hom. Isaac Toucey, Secretary of the Ivary : Commissioners appointed by the Governor of Florida, with aregiment of armed men at the gate, demanded the sur-render of this navy-yard, having previously taken posses-sion of the magazines. I surrendered the place and struck my flag at half-past one o'clock this day. The store-ship Supply sailed for Vera Cruz the moment the yard flag was lowered. JAMES ARMSTRONG, Captain U. S. Navy, late Commandant Navy Yard.

The testimony shows that this was the only information received by the Department previous to the 24th. It is evident, from this statement of facts and dates, that at the time Captain Randolph's resignation was accepted, and for at least ten days thereafter, the Department was

and for at least ten days thereafter, the Department was totally ignorant of his participation in the Pensacola expe-dition. Even if we were at liberty to infer that the Secre-tary had seen his name connected with the expedition in the newspapers earlier than the 24th, we cannot suppose that it had been seen before the 14th, the day on which the resignation was accepted, for it would be impossible for even flying newspaper reports to reach this city from Pen-sacola in so short a time. Lieutenant Farrand, who was first lieutenant and executive officer of the navy-yard, re-signed on the 13th January, and his resignation was ac-cepted on the 21st. Lieutenant Renshaw resigned on the 16th, and it was accepted on the 22d. The testimony discloses nothing on which the Secretary could have refused to accept these resignations, if all the testimony taken before the committee had been in his pos-

testimony taken before the committee had been in his pos-session at the time he acted. But by a comparison of dates it will be seen that they were received, and in due course of business acted on before he had any intelligence from

of business acted on before he had any intelligence from Pensacola except the telegraphic despatch already set forth. The undersigned would hold the Secretary, who should accept the resignation of an officer who before resigning had been guilty of any conduct unbecoming an officer, amenable to the severest censure. An acceptance of a resignation is an honorable discharge from service, which had be a given to bins who have maintained by which which are the server to bins who have maintained by which which are the server to bins who have maintained being which have the server to be a se should only be given to him who has acquitted himself with fidelity and honor, but one who has thus acquitted himself is entitled to be discharged without question as to what he intends to do thereafter. The base man would hold his commission and seek an

opportunity to be ray his Government, so that the fruits of his treason might purchase for him favor and reward. Such was the conduct of Arnold.

Such was the conduct of Arnold. The pure man, who, compelled to select between two claims to his allegiance, chooses that which promises least personal advantage, scrupulously discharges all existing obligations, then voluntarily and openly renounces a cher-ished commission from a sense of duty to his native State, gives the highest proof of devotion to principle and obedi-ence to the dictates of honor. Such men cannot be re-garded as traitors; and to call them so is to arraign manly virtue in the name of patriotism. It has been the proud boast of the American Navy that it has never contained a traitor in its ranks. Will it be cred-ticed that it has suddenly become a hot-bed from which has sprung such a bountiful crop of traitors? Such a supposi-tion would discredit the whole body of its officers; for it could only consist with the existence of some poisonors and traitorous influence pervading all its ranks and corrupting all its members.

all its members.

all its members. The chief clerk of the Navy Department testified that there has been nothing unusual in the course recently pur-sued in regard to resignations; and that the uniform course, from time immemorial, has been to act promptly on resig-nations, unless some special reason existed for taking a case out of the routine of current business. The orderly conduct of business necessitates prompt action on each case in its turn; and in an ill-organized Department, or in one in which the force is insufficient, affairs will soon fall into inextrica-

ble confusion. No consideration of convenience, nor the maintenance of any mere system of business, should be allowed to interfere with the substantial demands of justice and public interest. But in the absence of any reason for deviating from the usual course (and the undersigned says confidently that in the present case no such reason existed), the head of an Executive Department is not censurable for requiring all its business to be regularly and promptly des-

requiring an its business to be regulary support. The undersigned can find in the testimony nothing to detract from the high reputation always borne by the pres-ent Secretary for integrity and patriotism; and therefore I cannot concur with the committee in recommending the passage of a resolution of censure.

March 2d. The resolution appended to the majority report, was agreed to-yeas 95. navs 62, as follows:

nays 62, as follows:
TEAS-Messrs. Charles F. Adams, Aldrich, Alley, Ashley, Beale, Bingham, Blair, Blake, Brayton, Buffintou, Burlingame, Butterfield, Campbell, Carey, Carter, Case, Cohurn, Oolfax, Conway, Covode, H. Winter Davis, Dawes, Delano, Duell, Edgerton, Edwards, Eliot, Ely, Farnsworth, Fenton, Duell, Edgerton, Edwards, Eliot, Ely, Farnsworth, Fenton, W., Goch, Grow, Gurley, Hale, Haskin, Helmick, Wm. A. Howard, Humphrey, Hutchins, Irvine, Junkin, Francis W., Kellogs, Kenyon, De Witt C. Leach, Lee, Longnecker, Lovejoy, Marston, McKean, McKnight, McPherson, Millward, Moorhead, Morrill, Edward Joy Morris, Isaac W., Morris, Nixon, Olin, Palmer, Perry, Pettil, Potter, Pottle, Edwin R. Reynolds, Rice, Christopher Robinson, Royce, Sedgwick, Sherman, Somes, Spaulding, Spinner, Stanton, Stevens, Wm. Stewart, Stratton, Tompkins, Train, Trinhle, Vandever, Van Wyck, Verree, Wade, Waldron, Walton, Cadwalader C. Washburne, Ellihu B. Washburne, Wells, Wilsou, Windom, Wood, Woodruff-95.
NAYS-Messrs. Adrain, Wn. C. Anderson, Barr, Barrett, Boteler, Brabson, Branch, Brown, Burch, Burnett, John B. Clark, Clemens, John Occhrane, Coz, James Oraig, Burton, Oraige, John G. Davis, De Jarnette, English, Ferry, Florence, Fouke, Garnett, Hamilton, John T. Harris, Hatton, McKenty, Millson, Mondgomery, Nelson, Nibluck, Pendleton, Peyton, Phelps, Pryor, Quartes, Riggos, James C. Robinson, Rufin, Stokes, Shmms, William N. H. Smithl, Stevenson, Rufin, Stokes, Stokes, Thomas, Vallandigham, Vance, Whiteley, Winslow, Woodson, Wright-62.
January 30th. Mr. JOHN H. REYNOLDS

January 30th. Mr. JOHN H. REYNOLDS made the majority report, and L.O'B. BRANCH the minority report, upon the bill to authorize the President to call out the militia, which was ordered to be printed and recommitted to the committee.<sup>\*</sup>

February 27th. The majority of the committee presented a report reviewing the correspondence, communicated by the President January 8th, between himself and Isaac W. Hayne, "Special Envoy" of South Carolina, and concluding with a resolution that in the opinion of the House, "the President had no constitutional power to negotiate with the representatives of the State of South Carolina for the surrender of any public property within the limits of that State." Messrs. Cochrane and Branch, as a minority, made a report of dissent. Both reports were ordered to be printed and recommitted, and the subject was not resumed.

28th. Mr. HOWARD of Michigan, made the closing report, on the general subject of Se-cession. Mr. COCHRANE and Mr. BRANCH gave notice of their dissent, and obtained leave to make a minority report.

WITHDRAWAL OF UNITED STATES TROOPS.

January 2d, 1861. Mr. JEFFERSON DAVIS of Mississippi offered this joint resolution, which was read and ordered to be printed :

<sup>\*</sup> Mr. John Cochrane expressed his dissent from the measure "at this particular time."—Congressional Globe, Seond Session, 36th Congress, p. 646.

Joint Resolution in relation to the Militia of the States, the condition of forts, magazines, arsenals and dock-yards; military power; withdrawal of troops from garrisons on the application of a State; and the recognition of the right of a State to keep troops and ships of war by proclamation of the President.

Whereas, by the second and third articles of amendment of the Constitution, it is declared that a well regulated Militia is the security of a free State, and that no soldier shall, in time of peace, be quartered in any house without the consent of the owner; and whereas, by the second and third clauses of the tenth section of the first article of the Constitution, it is indirectly provided that a State may, with the consent of Congress, keep troops and ships of war in time of peace; and whereas, by the seventeenth clause of the eighth section of the first article of the Constitution, the exclusive jurisdiction of the Federal Government over forts, magazines, arsenals, dock-yards, and so forth, is limited to places purchased by the consent of the Legislature of the State in which the same shall be; and whereas, the military powers delegated by the States to the Federal Government were designed for the purposes stated generally, in the preamble to the Constitution, namely: to insure domestic tranquillity, and provide for the common defence; therefore, Be it resolved by the Senate and House of Representatives, etc., that upon the application of a State, either through a Convention or Legislature thereof, asking that the Federal forces of the Army and Navy may be withdrawn from its limits, the President of the United States shall order the withdrawal of the Federal garrisons, and take the needful security for the safety of the public property which may remain in said State.

Sec. 2. And be it further resolved, That whenever a State Convention, duly and lawfully assembled, shall enact that the safety of the State requires it to keep troops and ships of war, the President of the United States be, and he is hereby, authorized and directed to recognize the exercise of that power by the State, and by proclamation to give notice of the fact for the information and government of all parties concerned.

# SUSPENDING CERTAIN LAWS IN SECEDED STATES.

January 19th, 1861. Mr. MASON of Virginia offered the following resolution, which was printed:

"It appearing to Congress that the State of South Carolina has, by an ordinance of the people of that State, in Convention assembled, declared the State separated from the United States, and from the Government thereof, as established under the Constitution; and it further appearing, that by reason of such declared separation, there are no officers of the United States acting under

the authority thereof, in the judiciary department of this Government, or under the laws for the collection of the revenues of the United States; whereby, and in consequence whereof, the laws of the United States are in fact suspended within the limits of said State; therefore, to avoid any hostile collision that might arise between the authorities of the United States and of the State aforesaid, in the attempt to execute laws of the United States, in the absence of those officers required by law to administer and execute said laws:

"Be it resolved by the Senate and House of Representatives, That from and after the passage of this joint resolution, all laws of the United States directing the mode in which the Army and Navy and other public force of the United States shall be used by the President of the United States, in aid of the civil authorites in executing the laws and authorizing the same, and all laws for the collection of revenue shall be, and the same are hereby suspended, and made inoperative in the State of South Carolina for the time being; and should it be made to appear hereafter by the executive authority of any other State or States, that a like ordinance has been passed by the peo-ple of any State, declaring such State or States separated from the United States, then it shall be the duty of the President of the United States to announce such separation by his proclamation, and all the laws of the United States shall, in like manner, be suspended and rendered inoperative in such State last aforesaid."

# RETROCESSION OF FORTS AND ARSENALS.

January 2d, 1861. Mr. HUNTER offered in the Senate this resolution, upon which, on the 11th of January, he gave his opinion on the condition of the country and the needed remedies:

WHEREAS certain forts; magazines, arsenals, dock-yards, and other needful buildings have been placed under the exclusive jurisdiction of the United States by a cession to that effect from certain States, and it may be the desire of one or more of these States to resume the jurisdiction thus ceded: Now, therefore—

Be it Resolved, That the President of the United States ought to be authorized by law, upon the application of the Legislature or of a regular Convention of the people of any such State, to retroce this jurisdiction to such States, upon taking proper security for the safe-keeping and return of all the property of the United States, or for paying for the value of the same, if destroyed or injured by the act of any of the States making such application.

# MR. HUNTER'S PLAN OF ADJUSTMENT.

In his speech, Mr. HUNTER indicated the points of adjustment which he considered indispensable

He thought the Southern people were bound to withdraw from the Government unless they can get constitutional guaran-tees, which shall provide: first, that Congress shall have no power to abolish slavery in the States, in the District of Columbia, in the dock-yards, forts, and arsenals of the United States; second, that it shall not abolish, tax, or obstruct the slave trade between the States; third, that it shall be the duty of each of the States to suppress combinations within their jurisdiction for armed invasions of another; fourth, that States shall be admitted with or without slavery, according to the election of the people; fifth, that it shall be the duty of the States to restore fugitive slaves when within their borders, or to pay the value of the same; sixth, that fugitives from justice shall be deemed all those who have offended against the laws of a State within its jurisdiction, and who have escaped therefrom; seventh, that Congress shall recognize and protect as property whatever is held to be such by the laws or prescriptions of any State within the Territories, dock-yards, forts, and arsenals within the United States, and wherever the United States has exclusive jurisdiction; with the following exceptions: First, it may leave the subject of slavery or involuntary servitude to the people of the Territories when a law shall be passed to that effect with the usual sanction, and also with the assent of a majority of the Senators from the slaveholding States, and a majority of the Senators from the non-slaveholding States. That exception is designed to provide for the case where we might annex a territory almost fully peopled, and whose people ought to have the right of self-government, and yet might not be ready to be admitted as a State into the Union.

The next exception is that "Congress may divide the territories to the effect that slavery or involuntary servitude shall be prohibited in one portion of the territory, and recognized and protected in another; provided the law has the sanction of a majority from each of the sections as aforesaid," and that exception is designed to provide for the case where an unpeopled territory is annexed, and it is a fair subject of division between the two sections.

In addition to these "guarantees of principles," there should be "guarantees of power," without which he did not think permanent peace could be secured. He indicated these as the best: First, the dual Executive, not in the form proposed by Mr. Calhoun, but in one less fairly open to objections. He would provide :

That each section shall elect a President, to be called the first and second President: the first to serve for four years as President, the next to succeed him at the end of four years, and to govern for four other years, and afterwards to be re-eligible.

first President, the second should be President of the Senate, with a casting vote in case of a tie; and that no treaty should be valid which did not have the signatures of both Presidents, and the assent of twothirds of the Senate.

That no law should be valid which did not have the assent of both Presidents, or in the event of a veto by one of them, the as-sent of a majority of the Senators of the section from which he came.

That no person should be appointed to a local office in the section from which the second President was elected, unless the appointment had the assent of that President, or, in the event of his veto, the assent of a majority of the Senators from the section from which he came.

He proposed to change the mode of electing these Presidents as follows :

I would provide that each State should be divided into Presidential electoral districts: that each district should elect one man, and that these representatives from the whole United States should meet in one chamber, and that the two men who, after a certain number of ballots, received the highest number of votes, should be submitted as the candidates to the people, and he should be declared as President who received a majority of the districts-the districts each voting singly. I would do this to destroy the opportunities which are given under our present system of nomination to the formation of corrupt combinations for purposes of plunder and of patronage. I would substitute this instead of the National Conventions, which have already done so much harm in our system.

Further :

I would also diminish the temptation to such corrupt combinations for spoils and patronage by the fact that the President, after the first election, would be elected four years before he commenced his service as President, and in the meantime he would be training as a second President at the head of the Senate, and exercising the veto power. The fact that he was elected four years beforehand would do much to prevent such combinations; but, further than this, the effect of such a division of the Executive power would be to destroy, to a great extent, the miserable system of rotation in office which exists at present, and to make merit the test of the fitness for office, and a guarantee for his permanence in place; for, as the second President would probably keep those in office during his term of President whom he had protected by his veto power before, if they were worthy of the place, the effect would be, at least if this system were introduced, that the rotation principle would be applied, if at all, not once in four years, but once in eight years.

But this plan would have another good effect. It would save us from most of those That during the term of service of the agitations attending a Presidential election

which now disturb the country, which unsettle public affairs, and which are doing so much to demoralize and corrupt the people. The election would take place in one section at a time; it would take place in each section but once in eight years, and in this way we would escape those disturbances which are now dividing and destroying us.

Further, "to secure the proper enforcement of rights which are now without remedies," he proposed "that the Supreme Court should also be adjusted. It should consist of ten judges-five from each section-the Chief Justice to be one of the five. I would allow one State to cite another State before this tribunal to charge it with having failed to perform its constitutional obligations; and if the court decided a State thus cited to be in default, then I would provide, if it did not repair the wrong it had done, that any State might deny to its citizens within its jurisdiction the privileges of citizens in all the States; that it might tax its commerce and the property of its people until it ceased to be in default. Thus I would provide a remedy without bringing the General Government into collision with the States, and without bringing the Supreme Court into collision with them. Whenever international stipulations in regard to the duties imposed on the States, as laid down in the Constitution, are violated, I would remedy the wrong by international remedies. would give a State the right, in such cases, after the adjudication of the court, to deny to the offending State the performance of the mutual obligations which had been created for its benefit. In this way I believe that these wrongs might be remedied without producing collision in the system. Α self-executing process would thus provide a remedy for the wrong, without a jar to the machinery of Government. In order to make this check efficient, it should be provided that the Judges of the Supreme Court in each section shall be appointed by the President from that section, and this is the only original appointing power which I would give to the second President."

In consideration of these changes, he expressed his willingness "to regulate the right of Secession, which I hold to be a right not given in the Constitution, but resulting from the nature of the compact. I would provide that, before a State seceded, it should summon a Convention of the States in the section to which it belonged, and submit to them a statement of its grievances and wrongs. Should a majority of the States in such Convention decide the complaint to be well founded, then the State ought to be per-mitted to secede in peace. For, whenever a majority of States in an entire section shall declare that good cause for secession exists, then who can dispute that it ought to take place? Should they say, however, that no good cause existed, then the moral force of

of those who are bound to the complaining State by identical and homogeneous interests, would prevent it from prosecuting the claim any further. I believe that the system thus adjusted would give us a permanent Union, an efficient, a useful, and just Government. I think our Government would then rank among the most permanent of human institutions. It is my honest opinion that, with a Government thus balanced, and with such capacities for empire as our people possess, we should build up a political system whose power and stability and beneficial influences would be unparalleled in all the history of the past." He believed "this scheme afforded the

best basis of settlement which has yet been devised. There are other schemes upon which I would settle. I would settle upon something which would give only a truce, provided it promised to be a long truce, and then trust to public opinion and the progress of truth to remedy future evils when they might arise. But I would prefer, when we do settle, after all this turmoil and confusion, that we should do so upon some principle which promises us a permanent adjustment, a constant and continuing peace, a safe, an efficient, and a stable Government."

At the close of Mr. Hunter's speech, this significant colloquy took place:

Mr. BAKER. I desire to ask the gentleman from Virginia, if he will allow me, and consider it respectful, one question. Mr. HUNTER. What is the question?

Mr. BAKER. It is this: If a majority of this branch of Congress-the constitutional majority, and a majority of the other branch, also the constitutional majority-shall pass constitutional amendments, to be submitted according to the forms of the Constitution for the consent and approbation of the peo-ple, in that event, if they be such as substantially meet the views of the gentlemen on the other side, will the Senator from Virginia, so far as he can, throw the weight of Virginia, and especially the weight of his own individual character, to maintain the Constitution as it is, the Government as it is, the laws as they now are, with the power of the Government, until the people of the States shall have decided upon those amendments?

Mr. Hunter. The Senator has asked me some questions which I cannot answer. I cannot answer for Virginia; I am not au-thorized to do so. I can only say this: that I will vote for the propositions of the Sen-ator from Kentucky which were presented in committee; and other gentlemen declared that they believed they would be satisfactory; but whether the people, who are now seceding and getting in line together for purposes of common defence, would wait to ascertain whether the State would adopt them, I am not authorized to say.

Mr. BAKER. That is not quite it: I do such a decision, on the part of confederates not make myself understood by the gentleman. Will the gentleman himself, as a Senator-

Mr. HUNTER. If the Senator is not satisfied I cannot satisfy him.

Mr. BAKER. Ah!

For speech and colloquy, see Congressional Globe, second session, 36th Congress, pages 328-332.

No vote was taken upon the resolution.

CONSTRUCTION OF SCREW SLOOPS OF WAR.

February 11th. The Senate, as in Committee of the Whole, adopted an amendment to the Naval Appropriation bill, providing for the construction of seven steam screw sloops of war of the second class, with full steam power, whose greatest draught of water shall not exceed 14 feet, and appro-priating \$1.200.000 for the purpose. The priating \$1,200,000 for the purpose. vote was, yeas 30, nays 18, as follows:

YEAS-Messrs. Anthony, Baker, Bigler, Bingham, Cam-eron, Chandler, Clark, Collamer, Doolittle, Durkee, Fessen-den, Foot, Grimes, Hale, Harlan, Johnson of Tennessee, Kennedy, King, Latham, Morrill, Sebastian, Seward, Sim-mons, Sumner, Ten Eyck, Thomson, Trumbull, Wade, Wil-kinson, and Wilson-30.

Hons, cullet, iei Lyos, Induson, Hambull, Wate, Wi-kinson, and Wilson-30.
NATS-Messrs. Bayard, Bragg, Bright, Clingman, Crit-tenden, Fitch, Gwin, Hemphill, Hunter, Lane, Mason, Nich-olson, Pearce, Polk, Fowell, Rice, Sculsburg, and Wigfall -18.

12th. The amendment made in Senate as in Committee of the Whole, was agreed to, yeas 27, nays 17.

20th. The House adopted the amendment, yeas 112, nays 38, as follows:

yeas 112, nays 38, as follows: YEAS-Messrs. Charles F. Adams, Adrian, Aldrich, Allen, Alley, Wm. C. Anderson, Ashley, Barr, Bingham, Blake, Brayton, Briggs, Bristow, Buffiton, Burlingame, Burnham, Butterfield, Campbell, Carey, Carter, Case, Coburn, John Cochrane, Coffax, Conkling, Conway, Coz, Ourtis, Dawes, Delano, Duell, Dunn, Edgerton, Edwards, Eliot, Ely, Farnsworth, Ferry, Florence, Frank, French, Gouch, Gra-ham, Grow, Gurley, Hall, Hamilton, J. Morrison Harris, Helmick, Hoard, Holman, William Howard, Humphrey, Hutchins, Irvine, Junkin, Francis W. Kellogg, Wm. Kellogg, Wethight, McPherson, Millward, Laban T. Moore, Mochnedd, Morrill, Edward Joy Morris, Isaac N. Morris, Nerse, Nixon, Olin, Palmer, Porter, Potte, Edwin Reynolds, Rice, Christopher Robinson, Royce, Scranton, Sedwick, Sickles, Spaulding, Spinner, Stanton, Stevens, Wm. Stewart, Stration, Tappan, Theaker, Tompkins, Van Wyck, Verree, Wade, Weldron, Cadwalader C. Washburn, Ellinn E. Washburne, Webster, Wells, Wilson, Windom, and woodruff-112. NATS-Messrs. Arery, Barrett, Borock, Branch, Brown, Burch, Burnett, John B. Clark, Burton Craige, John G. Daris, De Jarnette, Edmundson, Etheridge, Garnett, John T. Harris, Hatton, Hindman, Hughes, James M. Leach, Elbert & Martin, Nelson, Niblack, Fhelps, Quarles, Kings, James C. Robinson, Russ, Sott, Sherman, Simas, Win, N. H. Smith, Stevenson, Stokes, Thomas, Vallandigham, Vance, Winslow, and Wright-38.

RECOGNITION OF THE SOUTHERN CONFEDER-ACY.

· February 11, 1861. Mr. BURTON CRAIGE of North Carolina, offered in the House of Representatives, the following resolution, which was referred to the Committee on Foreign Affairs, on his motion :

Whereas, the States of South Carolina, Florida, Alabama, Georgia, Mississippi, and Louisiana have seceded from the Confederacy of the United States, and have estab"the Confederacy of the United States South;" and whereas it is desirable that the most amicable relations should exist between the two Governments, and war should be avoided as the greatest calamity which can befall them :

Resolved by the Senate and House of Representatives of the United States of America in Congress assembled, That the President of the United States be, and is hereby, required to acknowledge the independence of said government as soon as he is informed officially of its establishment; and that he receive such envoy, embassador, or commissioner as may or shall be appointed by said government for the purpose of amicably adjusting the matters in dispute with said Government.

It was not reported from the Committee.

THE TERRITORIAL DIFFICULTIES SETTLED BY CONGRESS.

Congress passed and the President approved bills to provide temporary governments for Colorado, February 28th, Nevada, March 2d, and Dakotah, March 2d, 1861.

These three Territories cover the entire region owned by the United States and not included within the States.

The sixth and sixteenth sections of each bill are as follows :

"That the legislative power of the Territory shall extend to all rightful subjects of legislation consistent with the Constitution of the United States and the provisions of the act; but no law shall be passed interfering with the primary disposal of the soil; no tax shall be imposed upon the property of the United States; nor shall the lands or other property of non-residents be taxed higher than the lands or the property of residents; nor shall any law be passed impairing the right of private property; nor shall any discrimination be made in taxing different kinds of property; but all property subject to taxation shall be in proportion to the value of the property taxed.

"That the Constitution and all laws of the United States, which are not locally inapplicable, shall have the same force and effect within the said Territory of Dakotah as elsewhere within the United States."

The ninth section contains this provision : "And each of the said District Courts shall have and exercise the same jurisdistion in all cases arising under the Constitution and laws of the United States as is vested in the Circuit and District Courts of the United States; and the said Supreme and District Courts of the said Territory, and the respective judges thereof, shall and may grant writs of habeas corpus in all cases in which the same are grantable by the judges of the United States in the District of Columbia."

The provisions of these bills were, as was announced in debate, agreed upon by Mr. GREEN of Missouri, Chairman of the Senate lished a Government under the name of Committee on Territories, and Mr. GROW of Pennsylvania, Chairman of the House Committee on Territories, Mr. Wade and others in consultation, and the bills passed without material opposition in either House. The Colorado bill passed the Senate February 4th, without a division, and the House, February 18th, by a vote of 90 yeas to 44 nays as follows:

Hays as follows: YEAS—Messrs. Charles F. Adams, Aldrich, Alley, Bingham, Elair, Blake, Brayton, Buffinton, Burlin-game, Burnham, Butterfield, Campbell, Carey, Carter, Case, Colfax, Conkling, Conway, Covode, Dawes, Delano, Duell, Edgerton, Edwards, Eliot, Ely, Farnsworth, Fentou, Florence, Frank, French, Gooch, Graham, Grow, Gurley, Hale, Itall, Helmick, William M. Howard, Humphrey, Hutchins, Irvine, Junkin, Francis W. Kellogg, Kenyon, K. K. Karaka, K. Kulinger, De Witt C. Leach, Lee, Longnecker, Loomis, Lovejoy, Marston, McKean, McKenty, McKnight, McPherson, Moorhead, Morrill, Edward Joy Morris, Isaac N. Morris, Morse, Nixon, Palmer, Porter, Pot-ter, Pottle, Edwin R. Reynolds, Rice, Christopher Robinson, Tompkins, Train, Vandever, Wade, Waldron, Walton, Cadwalader C. Washburn, Ellihu B. Washburne, Wells, Windom, Wood, Woodruff-90. NAYS-Messrs. Adrain, Allen, W. C. Anderson. Aery, Bar-ret, Boock, Buroth, Burnett, Horace F. Clark, Jao. B. Clark, Jan Barte, Janes, Janeet, Jonese, Jane, B. Clark, Jao. B. Clark, Janes, Clark, Janes, Chark, Janes, Clark, Janes, Ja Aldrich, Alley, ton Burlin-

rett, Bocock, Burch, Burnett, Horace F. Clark, Jno. B. Clark, Fett, Boccck, Burch, Burnett, Horace F. Clark, Juo. B. Clark, Jno. Cochrane, Corvin, Coz, Burtom Graige, Juo, G. Davis, De Jarnette, Edmundson, Garnett, Gilmer, John T. Harris, Hat-ton, Hindman, Holman, Hughes, Leake, Logan, Maynard, McClernand, Millson, Laban T. Moore, Nelson, Niblack, Pettit, Phelps, Pryor, Quarles, James C. Robinson, Ruffin, Sickles, Simms, William N. H. Smith, Stevenson, Stokes, Phorene - March Stevenson, Stokes, Thomas-14.

The Nevada bill was passed by the House, March 1st-yeas 91, nays 32, and the Dakotah, same day, without a division.

#### ON PROTECTING SLAVERY IN THE TERRITO-RIES.

During the consideration of the Davis resolutions in the United States Senate in May, 1860,

Mr. THOMAS L. CLINGMAN of North Carolina, May 24th, offered an amendment to add to the fourth resolution the following:

"Resolved, That the existing condition of the Territories of the United States does not require the intervention of Congress for the protection of property in slaves."

Mr. ALBERT G. BROWN of Mississippi, moved to strike from the amendment the word "not," so as to assert that the existing condition of the Territories of the United States does require the intervention of Congress for the protection of property in slaves.

Which was rejected May 25th-yeas 5, nays 43, as follows:

YEAS-Messrs. Brown, Clay, Iverson, Johnson of Arkan-

TEAS-Messrs. Brown, Clay, Iverson, Journson of Alkan-eas, Yulee-5. NAYS-Messrs. Benjamin, Bigler, Bingham, Bragg, Bright, Chandler, Chesnut, Clark, Clingman, Collamer, Crittenden, Davis, Dixon, Doolitle, Fitzpatrick, Foot, Green, Gwin, Hale, Hamlin, Hammond, Hemphill, Hunter, Johnson of Pennessee, Kennedy, Lane, Latham, Mallory, Mason, Nich-olson, Pearce, Polk, Powell, Pugh, Rice, Sebastian, Stidell, Ten Eyck, Toombs, Trumbull, Wade, Wigfall, Wilson-43.

The question recurring to Mr. CLINGMAN'S amendment,

Mr. COLLAMER of Vermont, moved to amend it so as to make it read :

Resolved, That the existing condition of the Territories of the United States does not, and in our opinion never will, require,

Which was rejected-yeas 16, navs 33, as follows:

YEAS-Messrs. Bingham, 'Chandler, Clark, Collamer, Crittenden, Dixon, Doolittle, Foot, Itale, Hamlin, Harlan, Simmons, Ten Eyck, Trumbull, Wade, Wilsom-16. NATS-Messrs. Benjamin, Bigler, Bragg, Bright, Brown, Chesnut, Clay, Clingman, Davis, Fitzpatrick, Green, Ham-mond, Hemphill, Hunter, Iserson, Johnson of Arkansaa, Johnson of Tennessee, Lance, Latham, Mallory, Mason, Nicholson, Pearce, Folk, Powell, Pugh, Rice, Saulsbury, Sebastian, Slidell, Toombs, Wigfall, Tulee-33.

The motion of Mr. CLINGMAN was then agreed to-yeas 26, nays 23, as follows :

When the fifth resolution was pending,

Mr. CLINGMAN moved to amend by adding, as follows:

"Provided, That it is not hereby intended to assert the duty of Congress to provide a system of laws for the maintenance of slavery."

Which was rejected—yeas 12, nays 31, as follows:

YEAS-Messrs. Clark, Clingman, Dixon, Foot, Foster, Hale, Hamlin, Latham, Pugh, Ten Eyck, Trumbull, Wilson -12

NATS—Messrs. Benjamin, Bragg, Bright, Brown, Chesnut, Clay, Davis, Fitzpatrick, Green, Hammond, Hemphill, Humter, Iverson, Johnson of Arkansas, Johnson of Tennes-see, Kennedy, Lane, Mallory, Mason, Nicholson, Pearce, Polk, Powell, Rice, Saulsbury, Sebastian, Slidell, Thomson, Toombs, Wigfall, Yulee—31.

Mr. BROWN of Mississippi, then offered a substitute for the fifth resolution, as follows :

"That experience having already shown that the Constitution and the common law, unaided by statutory enactment, do not afford adequate and sufficient protection to slave property-some of the Territories having failed, others having refused to pass such enactments-it has become the duty of Congress to interpose and pass such laws as will afford to slave property in the Territories that protection which is given to other kinds of property."

Which was rejected—yeas 3, nays 42, as follows:

YEAS—Messrs. Brown, Johnson of Arkaneas, Mallory—3. NATS—Messrs. Benjamin, Bigler, Bragg, Bright, Chesnut, Clark, Clay, Olingman, Crittenden, Davis, Dixon, Doolittle, Fülzpatrick, Foot, Foster, Green, Grimes, Gwin, Hamlin, Harlan, Hemphill, Hunter, Iverson, Johnson of Tennessee, Lane, Latham, Mason, Nicholson, Pearce, Polk, Powell, Pugh, Rice, Sebastian, Slidell, Ten Eyck, Thomson, Toombs, Trumbull, Wigfall, Wilson, Yulee—42.

# CONSTITUTION OF THE UNITED STATES,

# AND OF THE

# "CONFEDERATE" STATES.

# WITH AN INDEX TO BOTH.

# Constitution of the United States of America.

WE the People of the United States, in order to form a more perfect Union, establish Justice, insure domestic Tranquillity, provide for the common defence, promote the general Welfare, and secure the Blessings of Liberty to ourselves and our Posterity, do ordain and establish this Constitution for the United States of America.

# ARTICLE I.

# SECTION I.

All legislative Powers herein granted shall be vested in a Congress of the United States, which shall consist of a Senate and House of Representatives.

#### SECTION II.

The House of Representatives shall be composed of Members chosen every second Year by the People of the several States, and the Electors in each State shall have the Qualifications requisite for Electors of the most numerous Branch of the State Legislature.

No Person shall be a Representative who shall not have attained to the Age of twenty five Years, and been seven Years a Citizen of the United States, and who shall not, when elected, be an Inhabitant of that State in which he shall be chosen.

Representatives and direct Taxes shall be apportioned among the several States which may be included within this Union, according to their respective Numbers, which shall be dctermined by adding to the whole Number of free Persons, including those bound to Service for a Term of Years, and excluding Indians not taxed, three fifths of all other Persons. The actual Enumeration shall be made within three Years after the first Meeting of the Congress of the United States, and within every subsequent Term of ten Years, in such Manner as they shall by Law direct. The Number of Representatives shall not exceed one for every thirty Thousand, but each State shall have at Least one Representative; and until such enumeration shall be made, the State of New Hampshire shall be entitled to chuse three, Massachusetts eight, Rhode Island and Providence Plantations one, Connecticut five, New York six, New Jersey four,

Pennsylvania eight, Delaware one, Maryland six, Virginia ten, North Carolina five, South Carolina five, and Georgia three.

When vacancies happen in the Representation from any State, the Executive Authority thereof shall issue Writs of Election to fill such Vacancies.

The House of Representatives shall chuse their Speaker and other Officers; and shall have the sole Power of Impeachment.

# SECTION III.

The Senate of the United States shall be composed of two Senators from each State, chosen by the Legislature thereof, for six Years; and each Senator shall have one Vote.

Immediately after they shall be assembled in Consequence of the first Election, they shall be divided as equally as may be into three Classes. The Seats of the Senators of the first Class shall be vacated at the Expiration of the second Year, of the second Class at the Expiration of the fourth Year, and of the third Class at the Expiration of the sixth Year, so that one-third may be chosen every second Year; and if Yacancies happen by Resignation, or otherwise, during the Recess of the Legislature of any State, the Executive thereof may make temporary Appointments until the next Meeting of the Legislature, which shall then fill such Yacancies.

No Person shall be a Senator who shall not have attained to the Age of thirty Years, and been nine Years a Citizen of the United States, and who shall not, when elected, be an Inhabitant of that State for which he shall be chosen.

The Vice President of the United States shall be President of the Senate, but shall have **no** Vote, unless they be equally divided.

The Senate shall chuse their other Officers, and also a President pro tempore, in the Absence of the Vice President, or when he shall exercise the Office of President of the United States.

The Senate shall have the sole Power to try all Impeachments. When sitting for that Purpose, they shall be on Oath or Affirmation. When the President of the United States is tried, the Chief Justice shall preside: And no Person shall be convicted without the Concurrence of two thirds of the Members present.

Judgment in Cases of Impeachment shall not. extend further than to removal from Office, and Disqualification to hold and enjoy any Office of honour, Trust or Profit under the United States: but the Party convicted shall nevertheless be liable and subject to Indictment, Trial, Judgment and Punishment, according to Law.

#### SECTION IV.

The Times, Places and Manner of holding Elections for Senators and Representatives, shall be prescribed in each State by the Legislature thereof; but the Congress may at any time by Law make or alter such Regulations, except as to the places of chusing Senators.

The Congress shall assemble at least once in every Year, and such Meeting shall be on the first Monday in December, unless they shall by Law appoint a different Day.

### SECTION V.

Each House shall be the Judge of the Elections, Returns and Qualifications of its own Members, and a Majority of each shall constitute a Quorum to do Business; but a smaller Number may adjourn from day to day, and may be authorized to compel the Attendance of absent Members, in such Manner, and under such Penalties as each House may provide.

Each House may determine the Rules of its Proceedings, punish its Members for disorderly Behaviour, and, with the Concurrence of two thirds, expel a Member.

Each House shall keep a Journal of its Proceedings, and from time to time publish the same, excepting such Parts as may in their Judgment require Secrecy; and the Yeas and Nays of the Members of either House on any question shall, at the Desire of one fifth of those Present, be entered on the Journal.

Neither House, during the Session of Congress, shall, without the Consent of the other, adjourn for more than three days, nor to any other Place than that in which the two Houses shall be sitting.

#### SECTION VI.

The Senators and Representatives shall receive a Compensation for their Services, to be ascertained by Law, and paid out of the Treasury of the United States. They shall in all Cases, except Treason, Felony and Breach of the Peace, be privileged from Arrest during their Attendance at the Session of their respective Houses, and in going to and returning from the same; and for any Speech or Debate in either House, they shall not be questioned in any other Place.

No Senator or Representative shall, during the Time for which he was elected, be appointed to any civil Office under the Authority of the United States, which shall have been created, or the Emoluments whereof shall have been encreased during such time; and no Person holding any Office under the United States, shall be a Member of either House during his Continuance in Office.

#### SECTION VII.

All Bills for raising Revenue shall originate in the House of Representatives; but the Senate may propose or concur with Amendments as on other Bills.

Every Bill which shall have passed the House of Representatives and the Senate, shall, before it become a Law, be presented to the President

of the United States; If he approve he shall sign it, but if not he shall return it, with his Objections to that House in which it shall have originated, who shall enter the Objections at large on their Journal, and proceed to recon-If after such Reconsideration two sider it. thirds of that House shall agree to pass the Bill, it shall be sent, together with the Objections, to the other House, by which it shall likewisc be reconsidered, and if approved by two thirds of that House, it shall become a Law. But in all such Cases the Votes of both Houses shall be determined by yeas and Nays, and the Names of the Persons voting for and against the Bill shall be entered on the Journal of each House respectively. If any Bill shall not be returned by the President within ten Days (Sundays excepted) after it shall have been presented to him, the Same shall be a law, in like Manner as if he had signed it, unless the Congress by their Adjournment prevent its return, in which Case it shall not be a Law.

Every Order, Resolution, or Vote to which the Concurrence of the Senate and House of Representatives may be necessary (except on a question of Adjournment) shall be presented to the President of the United States; and before the Same shall take Effect, shall be approved by him, or being disapproved by him, shall be repassed by two thirds of the Senate and House of Representatives, according to the Rules and Limitations prescribed in the Case of a Bill.

#### SECTION VIII.

The Congress shall have Power

To lay and collect Taxes, Duties, Imposts and Excises, to pay the Debts and provide for the common Defence and general Welfare of the United States; butall Duties, Imposts and Excises shall be uniform throughout the United States;

To borrow Money on the credit of the United States;

To regulate Commerce with foreign Nations, and among the several States, and with the Indian Tribes;

To establish an uniform Rule of Naturalization, and uniform Laws on the subject of Bankruptcies throughout the United States;

To coin Money, regulate the Value thereof, and of foreign Coin, and fix the Standard of Weights and Measures;

To provide for the Punishment of counterfeiting the Securities and current Coin of the United States;

To establish Post Offices and post Roads;

To promote the progress of Science and useful Arts, by securing for limited Times to Authors and Inventors the exclusive Right to their respective Writings and Discoveries;

To constitute Tribunals inferior to the supreme Court;

To define and punish Piracies and Felonies committed on the high Seas, and Offences against the Law of Nations;

To declare War, grant Letters of Marque and Reprisal, and make Rules concerning Captures on Land and Water;

To raise and support Armies, but no Appropriation of Money to that Use shall be for a longer Term than two Years;

To provide and maintain a Navy;

To make Rules for the Government and Regulation of the land and naval Forces;

To provide for calling forth the Militia to execute the Laws of the Union, suppress Insurrections and repel Invasions;

To provide for organizing, arming, and disciplining, the Militia, and for governing such Part of them as may be employed in the Service of the United States, reserving to the States respectively, the Appointment of the Officers, and the Authority of training the Militia according to the Discipline prescribed by Congress;

To exercise exclusive Legislation in all Cases whatsoever, over such District (not exceeding ten Miles square) as may, by Cession of particular States, and the Acceptance of Congress, become the Seat of the Government of the United States, and to exercise like Authority over all Places purchased by the Consent of the Legislature of the State in which the Same shall be, for the Erection of Forts, Magazines, Arsenals, Dock-Yards, and other needful Buildings;—And

To make all Laws which shall be necessary and proper for carrying into Execution the foregoing Powers, and all other Powers vested by this Constitution in the Government of the United States, or in any Department or Officer thereof.

### SECTION IX.

The Migration or Importation of such Persons as any of the States now existing shall think proper to admit, shall not be prohibited by the Congress prior to the Year one thousand eight hundred and eight, but a Tax or Duty may be imposed on such Importation, not exceeding ten dollars for each Person.

The Privilege of the Writ of Habeas Corpus shall not be suspended, unless when in Cases of Rebellion or Invasion the public Safety may require it.

No Bill of Attainder or expost facto Law shall be passed.

No Capitation, or other direct, Tax shall be laid, unless in Proportion to the Census or Enumeration herein before directed to be taken.

No Tax or Duty shall be laid on Articles exported from any State.

No Preference shall be given by any Regulation of Commerce or Revenue to the Ports of one State over those of another: nor shall Vessels bound to, or from, one State, be obliged to enter, clear, or pay Duties in another.

No Money shall be drawn from the Treasury, but in Consequence of Appropriations made by Law; and a regular Statement and Account of the Receipts and Expenditures of all public Money shall be published from time to time.

No Title of Nobility shall be granted by the United States: and no Person holding any Office of Profit or Trust under them, shall, without the Consent of the Congress, accept of any present, Emolument, Office, or Title, of any kind whatever, from any King, Prince, or foreign State.

#### SECTION X.

No State shall enter into any Treaty, Alliance, Case, after the Choice of the President, the Peror Confederation; grant Letters of Marque and son having the greatest Number of Votes of the

Reprisal; coin Money; emit Bills of Credit; make any Thing but gold and silver Coin a Tender in Payment of Debts; pass any Bill of Attainder, ex post facto Law, or Law impairing the Obligation of Contracts, or grant any Title of Nobility.

No State shall, without the consent of the Congress, layany Imposts or Duties on Imports or Exports, except what may be absolutely necessary for executing it's inspection Laws: and the net Produce of all Duties and Imposts, laid by any State on Imports or Exports, shall be for the Use of the Treasury of the United States; and all such Laws shall be subject to the Revision and Controul of the Congress.

No State shall, without the Consent of Congress, lay any Duty of Tonnage, keep Troops, or Ships of War in time of Peace, enter into any Agreement or Compact with another State, or with a foreign Power, or engage in War, unless actually invaded, or in such imminent Danger as will not admit of Delay.

# ARTICLE II.

#### SECTION I.

The executive Power shall be vested in a President of the United States of America. He shall hold his Office during the Term of four Years, and, together with the Vice President, chosen for the same Term, be elected, as follows

Each State shall appoint, in such Manner as the Legislature thereof may direct, a Number of Electors, equal to the whole Number of Senators and Representatives to which the State may be entitled in the Congress: but no Senator or Representative, or Person holding an Office of Trust or Profit under the United States, shall be appointed an Elector.

The Electors shall meet in their respective States, and vote by Ballot for two Persons, of whom one at least shall not be an Inhabitant of the same State with themselves. And they shall make a List of all the Persons voted for, and of the Number of Votes for each; which List they shall sign and certify, and transmit sealed to the Seat of the Government of the United States, directed to the President of the The President of the Senate shall, in Senate. the Presence of the Senate and House of Representatives, open all the Certificates, and the Votes shall then be counted. The Person having the greatest Number of Votes shall be the President, if such Number be a Majority of the whole Number of Electors appointed; and if there be more than one who have such Majority, and have an equal Number of Votes, then the House of Representatives shall immediately chuse by Ballot one of them for President; and if no Person have a Majority, then from the five highest on the List the said House shall in like Manner chuse the President. But in chusing the President, the Votes shall be taken by States, the Representation from each State having one Vote; A Quorum for this Purpose shall consist of a Member or Members from twothirds of the States, and a Majority of all the States shall be necessary to a Choice. In every Case, after the Choice of the President, the PerElectors shall be the Vice President. But if there should remain two or more who have equal Votes, the Senate shall chuse from them by Ballot the Vice President.\*]

The Congress may determine the Time of chusing the Electors, and the Day on which they shall give their Votes; which Day shall be the same throughout the United States.

No Person except a natural born Citizen, or a Citizen of the United States, at the time of the Adoption of this Constitution, shall be eligible to the Office of President; neither shall any Person be eligible to that Office who shall not have attained to the Age of thirty five Years, and been fourteen Years a Resident within the United States.

In Case of the Removal of the President from Office, or of his Death, Resignation, or Inability to discharge the Powers and Duties of the said Office, the same shall devolve on the Vice President, and the Congress may by Law provide for the Case of Removal, Death, Resignation, or Inability, both of the President and Vice President, declaring what Officer shall then act as President, and such Officer shall act accordingly, until the Disability be removed, or a President shall be elected.

The President shall, at stated Times, receive for his Services, a Compensation, which shall neither be encreased nor diminished during the Period for which he shall have been elected, and he shall not receive within that Period any other Emolument from the United States, or any of them.

Before he enter on the Execution of his Office, he shall take the following Oath or Affirmation :--

"I do solemnly swear (or affirm) that I will faithfully execute the Office of President of the United States, and will to the best of my Ability, preserve, protect and defend the Constitution of the United States."

# SECTION II.

The President shall be Commander in Chief of the Army and Navy of the United States, and of the Militia of the several States, when called into the actual Service of the United States; he may require the Opinion, in writing, of the principal Officer in each of the executive Departments, upon any Subject relating to the Duties of their respective Offices, and he shall have Power to grant Reprieves and Pardons for Offences against the United States, except in Cases of Impeachment.

He shall have Power, by and with the Advice and Consent of the Senate, to make Treatics, provided two thirds of the Senators present concur; and he shall nominate, and by and with the Advice and Consent of the Senate, shall appoint Ambassadors, other public Ministers and Consuls, Judges of the supreme Court, and all other Officers of the United States, whose Appointments are not herein otherwise provided for, and which shall be established by Law: but the Congress may by Law vest the Appointment of such inferior Officers, as they think proper, in

\* This clause of the Constitution has been annulled. See twelfth article of the amendment, page 96.

the President alone, in the Courts of Law, or in the Heads of Departments.

The President shall have Power to fill up all Vacancies that may happen during the Recess of the Senate, by granting Commissions which shall expire at the End of their next Session.

#### SECTION III.

He shall from time to time give to the Congress Information of the State of the Union, and recommend to their Consideration such Measures as he shall judge necessary and expedient; he may, on extraordinary Occasions, convene both Houses, or either of them, and in Case of Disagreement between them, with Respect to the Time of Adjournment, he may adjourn them to such Time as he shall think proper; he shall receive Ambassadors and other public Ministers; he shall take Care that the Laws be faithfully executed, and shall Commission all the officers of the United States.

#### SECTION IV.

The President, Vice President and all civil Officers of the United States, shall be removed from Office on Impeachment for, and Conviction of, Treason, Bribery, or other high Crimes and Misdemeanors.

# ARTICLE III.

#### SECTION I.

The judicial Power of the United States, shall be vested in one supreme Court, and in such inferior Courts as the Congress may from time to time ordain and establish. The Judges, both of the supreme and inferior Courts, shall hold their Offices during good Behavior, and shall, at stated Times, receive for their Services, a Compensation, which shall not be diminished during their Continuance in Office.

# SECTION II.

The judicial Power shall extend to all cases, in Law and Equity, arising under this Constitution, the Laws of the United States, and Treaties made, or which shall be made, under their Authority;—to all Cases affecting Ambassadors, other public Ministers, and Consuls;—to all Cases of admiralty and maritime Jurisdiction;—to Controversies to which the United States shall be a Party;—to Controversies between two or more States;—between a State and Citizens of another State; —between a State and Citizens of another State; —between Citizens of different States, —between Citizens of the same State claiming Lands under Grants of different States, and between a State, or the Citizens

In all Cases affecting Ambassadors, other public Ministers and Consuls, and those in which a State shall be Party, the supreme Court shall have original Jurisdiction. In all the other Cases before mentioned, the supreme Court shall have appellate Jurisdiction, both as to Law and Fact, with such Exceptions, and under such Regulations as the Congress shall make.

The Trial of all Crimes, except in Cases of Impeachment, shall be by Jury; and such Trial shall be held in the State where the said Crimes shall have been committed; but when not committed within any State, the Trial shall be at such Place or Places as the Congress may by Law have directed,

# SECTION III.

Treason against the United States, shall consist only in levying War against them, or in adhering to their Enemies, giving them Aid and Comfort. No Person shall be convicted of Treason unless on the Testimony of two Witnesses to the same overt Act, or on Confession in open Court.

The Congress shall have Power to declare the Punishment of Treason, but no Attainder of Treason shall work Corruption of Blood, or Forfeiture except during the Life of the Person attainted.

### ARTICLE IV.

#### SECTION I.

Full Faith and Credit shall be given in each State to the public Acts, Records, and judicial Proceedings of every other State. And the Congress may by general Laws prescribe the Manner in which such Acts, Records and Proceedings shall be proved, and the Effect thereof.

#### SECTION II.

The Citizens of each State shall be entitled to all Privileges and Immunities of Citizens in the several States.

A Person charged in any State with Treason, Felony, or other Crime, who shall flee from Justice, and be found in another State, shall on Demand of the executive Authority of the State from which he fled, be delivered up, to be removed to the State having Jurisdiction of the Crime.

No Person held to Service or Labour in one State, under the Laws thereof, escaping into another, shall, in Consequence of any Law or Regulation therein, be discharged from such Service or Labour, but shall be delivered up on Claim of the Party to whom such Service or Labour may be due.

#### SECTION III.

New States may be admitted by the Congress into this Union; but no new State shall be formed or erected within the Jurisdiction of any other State; nor any State be formed by the Junction of two or more States, or Parts of States, without the Consent of the Legislatures of the States concerned as well as of the Congress.

The Congress shall have Power to dispose of and make all needful Rules and Regulations respecting the Territory or other Property belonging to the United States; and nothing in this Constitution shall be so construed as to Prejudice any Cleims of the United States, or of any particular State.

#### SECTION IV. .

The United States shall guarantee to every State in this Union a Republican Form of Government, and shall protect each of them against Invasion; and on Application of the Legislature, or of the Executive (when the Legislature cannot be convened) against domestic Violence.

### ARTICLE V.

The Congress, whenever two thirds of both

Houses shall deem it necessary, shall propose Amendments to this Constitution, or, on the Application of the Legislatures of two thirds of the several States, shall call a Convention for proposing Amendments, which, in either Case, shall be valid to all Intents and Purposes, as Part of this Constitution, when ratified by the Legislatures of three fourths of the several States, or by Conventions in three fourths thereof, as the one or the other Mode of Ratification may be proposed by the Congress; Provided that no Amendment which may be made prior to the Year one thousand eight hundred and eight shall in any Manner affect the first and fourth Clauses in the Ninth Section of the first Article; and that no State, without its Consent, shall be deprived of its equal Suffrage in the Senate.

# ARTICLE VI.

All Debts contracted and Engagements entered into, before the Adoption of this Constitution, shall be as valid against the United States under this Constitution, as under the Confederation.

This Constitution, and the Laws of the United States which shall be made in Pursuance thereof; and all Treaties made, or which shall be made, under the authority of the United States, shall be the supreme Law of the Land; and the Judges in every State shall be bound thereby, any Thing in the Constitution or Laws of any State to the Contrary notwithstanding. The Senators and Representatives before

The Senators and Representatives before mentioned, and the Members of the several State Legislatures, and all executive and judicial Officers, both of the United States and of the several States, shall be bound by Oath or Affirmation, to support this Constitution; but no religious Test shall ever be required as a Qualification to any Office or public Trust under the United States.

# ARTICLE VII.

The Ratification of the Conventions of nine States, shall be sufficient for the Establishment of this Constitution between the States so ratifying the Same.

DONE in Convention by the Unanimous Consent of the States present the Seventeenth Day of September in the Year of our Lord one thousand seven hundred and Eighty seven and of the Independance of the United States of America the Twelfth En Clittness whereof We have hereunto subscribed our Names,

G°: WASHINGTON, Presidt and Deputy from Virginia

NEW HAMPSHIRE. John Langdon, Nicholas Gilman. MASSACHUSETS. Nathaniel Gorham, Rufus King. CONNECTICUT. Wm. Sami. Johnson, Roger Sherman. NEW YORK. Alexander Hamilton. NEW JERSEY. Wil: Livingston, David Brearley, Wm. Paterson, Jona. Dayton. PENNSYLVANIA, B. Franklin, Thomas Mifflin, Robt. Morris, Geo: Clymer, Tho: Fitzsimons, Jared Ingersoll, James Wilson, Gouv: Morris. DELAWARE. Geo: Read, Gunning Bedford, Jun'r,

John Dickinson, Richard Bassett, Jaco: Broom. MARYLAND. James McHenry, Dan: of St. Thos. Jenifer, Danl. Carroll. VIRGINIA.

John Blair, James Madison, Jr., NORTH CAROLINA. Wm. Blount, Rich'd Dobbs Spaight, Hu. Williamson. Attest: SOUTH CAROLINA. J. Rutledge, Charles Cotesworth Pinckney Charles Pinckney, Pierce Butler,

GEORGIA.

William Few,

Abr. Baldwin.

WILLIAM JACKSON, Secretary.

# Amendments.

# ARTICLE I.

Congress shall make no law respecting an establishment of religion, or prohibiting the free exercise thereof; or abridging the freedom of speech, or of the press; or the right of the people peaceably to assemble, and to petition the Government for a redress of grievances.

#### ARTICLE II.

A well regulated Militia, being necessary to the security of a free State, the right of the people to keep and bear Arms, shall not be infringed.

### ARTICLE III.

No Soldier shall, in time of peace be quartered in any house, without the consent of the Owner, nor in time of war, but in a manner to be prescribed by law.

#### ARTICLE IV.

The right of the people to be secure in their persons, houses, papers, and effects, against unreasonable searches and seizures, shall not be violated, and no Warrants shall issue, but upon probable cause, supported by Oath or affirmation, and particularly describing the place to be searched, and the persons or things to be seized.

#### ARTICLE V.

No person shall be held to answer for a capital, or otherwise infamous crime, unless on a presentment or indictment of a Grand Jury, except in cases arising in the land or naval forces, or in the Militia, when in actual service in time of War or public danger; nor shall any person be subject for the same offence to be twice put in jeopardy of life or limb; nor shall be compelled in any Criminal Case to be a witness against himself, nor be deprived of life, liberty, or property, without due process of law; nor shall private property be taken for public use, without just compensation.

#### ARTICLE VI.

In all criminal prosecutions, the accused shall enjoy the right to a speedy and public trial, by an impartial jury of the State and district wherein the crime shall have been committed, which district shall have been previously ascertained by law, and to be informed of the nature and cause of the accusation; to be confronted with the witnesses against him; to have Compulsory process for obtaining Witnesses in his favour, and to have the Assistance of Counsel for his defence.

#### ARTICLE VII.

In Suits at common law, where the value in controversy shall exceed twenty dollars, the right of trial by jury shall be preserved, and no fact tried by a jury shall be otherwise reexamined in any Court of the United States, than according to the rules of the common law.

# ARTICLE VIII.

Excessive bail shall not be required, nor excessive fines imposed, nor cruel and unusual punishments inflicted.

#### ARTICLE IX.

The enumeration in the Constitution, of certain rights, shall not be construed to deny or disparage others retained by the people.

#### ARTICLE X.

The powers not delegated to the United States by the Constitution, nor prohibited by it to the States, are reserved to the States respectively, or to the people.

#### ARTICLE XI.

The Judicial power of the United States shall not be construed to extend to any suit in law or equity, commenced or prosecuted against one of the United States by Citizens of another State, or by Citizens or Subjects of any Foreign State.

#### ARTICLE XII.

The Electors shall meet in their respective states, and vote by ballot for President and Vice President, one of whom, at least, shall not be an inhabitant of the same state with themselves; they shall name in their ballots the person voted for as President, and in distinct ballots the person voted for as Vice-President, and they shall make distinct lists of all persons voted for as President, and of all persons voted for as Vice-President, and of the number of votes for each, which lists they shall sign and certify, and transmit sealed to the seat of the government of the United States, directed to the President of the Sen-ate;—The President of the Senate shall, in presence of the Senate and House of Representatives, open all the certificates and the votes shall then be counted ; -The person having the greatest number of votes for President, shall be the President, if such number be a majority of the whole number of Electors appointed; and if no person have such majority, then from the persons having the highest numhers not exceeding three on the list of those voted for as President, the House of Representatives shall choose immediately, by ballot, the President. But in choosing the President, the votes shall be taken by states, the representation from each state having one vote; a quorum for this purpose shall consist of a member or members from two thirds of the states, and a majority of all the states shall be necessary to a choice. And if the House of Representatives shall not choose a President whenever the right of choice shall devolve upon them, before the fourth day of March next following, then the Vice-President shall

act as President, as in the case of the death or other constitutional disability of the President. The person having the greatest number of votes as Vice-President, shall be the Vice-President, if such number be a majority of the whole number of Electors appointed, and if no person have a majority, then from the two highest numbers on the list, the Senate shall choose the Vice-President; a quorum for the purpose shall consist of two-thirds of the whole number of Senators, and a majority of the whole number shall be necessary to a choice. But no person constitutionally ineligible to the office of President shall be eligible to that of Vice-President of the United States.

PROCEEDINGS OF THE CONVENTION WHICH FORMED THE CONSTITUTION.

### IN CONVENTION.

### MONDAY, September 17, 1787.

Resolved, That the preceding Constitution be laid before the United States in Congress assembled: and that it is the opinion of this Convention that it should afterwards be submitted to a convention of delegates, chosen in each State by the people thereof, under the recommendation of its legislature, for their assent and ratification; and that each convention assenting to and ratifying the same should give notice thereof to the United States in Congress assembled.

Resolved, That it is the opinion of this Convention that, as soon as the conventions of nine States shall have ratified this Constitution, the United States in Congress assembled should fix a day on which electors should be appointed by the States which shall have ratified the same. and a day on which electors should assemble to vote for the President, and the time and place for commencing proceedings under this Constitution; that after such publication, the electors should be appointed, and the Senators and Representatives elected; that the electors should meet on the day fixed for the election of the President, and should transmit their votes, certified, signed, sealed, and directed, as the Constitution requires, to the Secretary of the United States in Congress assembled; that the Senators and Representatives should convene at the time and place assigned; that the Senators should appoint a President of the Senate, for the sole purpose of receiving, opening, and counting the votes for President; and that, after he shall be chosen, the Congress, together with the President, should, without delay, proceed to execute this Constitution.

By the unanimous order of the Convention : G<sup>o</sup>: WASHINGTON, President.

WILLIAM JACKSON, Secretary.

LETTER OF THE CONVENTION TO THE OLD CON-GRESS.

### IN CONVENTION.

SEPTEMBEB 17, 1787.

SIR: We have now the honor to submit to the South Carolina, consideration of the United States in Congress land, Mr. Ross.

act as President, as in the case of the death or assembled, that Constitution which has apother constitutional disability of the President. peared to us the most advisable.

The friends of our country have long seen and desired that the power of making war, peace, and treaties; that of levying money, and regulating commerce, and the correspondent executive and judicial authorities, should be fully and effectually vested in the General Government of the Union; but the impropriety of delegating such extensive trust to one body of men is evident; hence results the necessity of a different organization.

It is obviously impracticable in the federal government of these States to secure all rights of independent sovereignty to each, and yet provide for the interest and safety of all. Individuals entering into society must give up a share of liberty to preserve the rest. The magnitude of the sacrifice must depend as well on situation and circumstance as on the object to be obtained. It is at all times difficult to draw with precision the line between those rights which must be surrendered and those which may be reserved; and on the present occasion, this difficulty was increased by a difference among the several States as to their situation, extent, habits, and particular interests.

In all our deliberations on this subject, we kept steadily in our view that which appears to us the greatest interest of every true American—the consolidation of our Union—in which is involved our prosperity, felicity, safety, perhaps our national existence. This important consideration, seriously and deeply impressed on our minds, led each State in the Convention to be less rigid on points of inferior magnitude than might have been otherwise expected; and thus the Constitution which we now present is the result of a spirit of amity and of that mutual deference and concession which the peculiarity of our political situation rendered indispensable.

That it will meet the full and entire approbation of every State is not, perhaps, to be expected; but each will doubtless consider that, had her interest been alone consulted, the consequences might have been particularly disagreeable or injurious tc others. That it is liable to as few exceptions as could reasonably have been expected, we hope and believe. That it may promote the lasting welfare of that country so dear to us all, and secure her freedom and happiness, is our most ardent wish.

With great respect, we have the honor to be, sir, your excellency's most obedient, humble servants.

By unanimous order of the Convention : G<sup>0</sup>: WASHINGTON, President.

His Excellency the PRESIDENT OF CONGRESS.

## PROCEEDINGS IN THE OLD CONGRESS.

UNITED STATES IN CONGRESS ASSEMBLED.

## FRIDAY, September 28, 1787.

Present—New Hampshire, Massachusetts, Connecticut, New York, New Jersey, Pennsylvania, Delaware, Virginia, North Carolina, South Carolina, and Georgia; and from Maryland, Mr. Ross.

Congress having received the report of the Convention lately assembled in Philadelphia-

Resolved, unanimously, That the said report, with the resolutions and letter accompanying the same, be transmitted to the several legislatures, in order to be submitted to a convention of delegates chosen in each State by the pcople thereof, in conformity to the resolves of the Convention made and provided in that case.

CHARLES THOMSON, Secretary.

# Constitution of the "Confederate" States.

In framing the Constitution of the so-called "Confederate States of America," its authors, in nearly all its parts, have adopted the precise language of the Constitution of the United States, and have followed the same order of arrangement in its articles and sections throughout. The two Constitutions, however, differ in many particulars, and it is designed, in this chapter, to exhibit clearly the points of difference between them.

disguos, in line carper, be standy to the order, in prime to difference between them. The preamble is changed in the following respects: the words "United States," near the beginning, are stricken out, and the following words inserted in their place: "Con-federate States, each State acting in its sourceign and inde-pendent character;" also, the words "nore perfect union" are stricken out, and the words "permanent federal govern-ment" inserted in their place; also, the following words are omitted: "provide for the common defence, promote the general welfare;" also, after the word "posterity," the fol-lowing words are inserted: "invoking the favor and gui-dance of Almighty God;" also, the word "Confederate" takes the place of "God;" preceding the word "States." This last change takes place throughout the entire instru-ment, with a single exception, which will be noted in the appropriate place. appropriate place.

### ARTICLE I.

SECTION 1. The word "granted" is stricken out, and the word "delegated" inserted in its place: also, the word "United," preceding the word "States," is stricken out, and the word "Confederated" inserted in its place.

SECTION 2. This section is changed as follows: in the first paragraph, after the words "in each State shall," the fol-lowing words are inserted: "be citizens of the Confederate States, and"; also, the following words are added to this paragraph: "but no person of foreign birth, and not a citizen of the Confederate States, shall be allowed to vote for any officer, civil or political, State or federal."

any officer, civil or political, State or federal." In the second paragraph, the words "been seven years a citizen of the United" are stricken out, and the words "be a citizen of the United" are stricken out, and the words "be a citizen of the Confederate" inserted in place of them. In the third paragraph the word "Union" is stricken ont, and the word "Confederacy" substituted ; also the words "other persons" are stricken out, and the word "slaves" substituted; also, the word "thirdy," preceding the word "the State of" are stricken out, and the follow-ing words substituted: "South Carolina shall be entitled to choose siz, the State of Georgia ten, the State of Alabama nine, the State of Florida two, the State of Mississippi seven, the State of Louisiana siz, and the State of Texas siz." No change is made in the fourth paragraph, and the fifth

No change is made in the fourth paragraph, and the fifth paragraph is changed by adding the following words: "ex-cept that any judicial or other federal officer resident and acting solely within the limits of any State, may be impeached by a vote of two thirds of both branches of the Legislature thereof." thereof."

SECTION 3. In the first paragraph, after the words "thereof for six years," the words "at the regular session next im-mediately preceding the commencement of the term of service" are inserted.

No change is made in the second paragraph. In the third paragraph the words "been nine years a citizen of the United" are stricken out, and the words "be a citizen of the Confederate" substituted.

of the Confederate" substituted. SECTION 4. The first paragraph of this section is changed by inserting the words "subject to the provisions of this Constitution" between the word "thereof" and the word "but?" Also, the words "times and" are inserted near the close, before the word "places." SECTION 5. In the second paragraph of this section, the words "of the whole number" are inserted, between the word "two-thirds" and the word "expel." SECTION 6. This section is changed by striking out the

word "felony" after the word " treason" in the first paragraph.

The second paragraph is changed by adding thereto the

The second paragraph is changed by adding thereto the following words: "But Congress may, by law grant to the principal officer in earh of the executive departments a seat upon the floor of either Hlouse, with the privilege of discuss-ing any measures appertaining to his department." SECTION 7. The following changes are made in this sec-tion: In the second paragraph the words "the House of Rep-resentatives and the Senate," are stricken out, and the words "both Houses" substituted; also the word "such" is in-sorted between the word "all" and the word "the following words are added to the end of this paragraph. the following words are added to the end of this paragraph : the following words are added to the end of this paragraph : " The President may approve any appropriation and disap-prove any other appropriation in the same bill. In such case he shall, in signing the bill, designate the appropria-tions disapproved, and shall return a copy of such appro-priations, with his objections, to the House in which the bill shall have originated; and the same proceedings shall then be had as in case of other bills disapproved by the Presi-ameta? dent."

The third paragraph is changed by striking out the words The 6 three paragraph is outanged by strains where the state and House of Representatives" in both places where they occur, and inserting in their places, in both cases, the words "both Houses," also by substituting the word "may" for the word "shall" after the words "disapproved by him."

SECTION 8. In the first paragraph, after the word "excises" the words "for revenue necessary" are inserted; also the word "and" after the word "debs" is omitted; also the words "and general welfare of the United States; but" are words "and general weifare of the United States; but" are stricken out, and the following words substituted: "and carry on the government of the Confederate States; but no bounties shall be granted from the treasury, nor shall any duties or taxes on importations from foreign nations be laid to promote or foster any branch of industry; and" The third paragraph is changed by adding thereto tho following words: "but neither this, nor any other clause contained in the Constitution, shall ever be construed to de-eagle the nower to Congress to approximate mome for any

equate the power to Congress to appropriate many for any internal improvement intended to facilitate commerce; ex-cept for the purpose of furnishing lights, beacons, and buoys, and other aids to navigation upon the coasts, and the improveand other and the removing of obstructions in river nanigation; in all such cases such duties shall be laid on the nanigation facilitated thereby, as may be necessary to pay the costs and expenses thereof."

The fourth paragraph is changed by adding thereto the following words: " but no law of Congress shall discharge any debt contracted before the passage of the same."

The seventh piragraph is changed as follows: the last word "roads" is changed to "roates," and the following words added thereto: "but the expenses of the Post-office Department, after the first day of March, in the year of our Lord eighteen hundred and sixty-three, shall be paid out of its commence." its own revenues."

In the fifteenth paragraph, the word "Union" is stricken out, and the words "the Confederate States" substituted. SECTION 9. The whole of the first paragraph is stricken out, and the following words substituted therefor: "The

importation of negroes of the African race from any foreign country other than the slaveholding States or territories of the United States of America is hereby forbidden; and Con-gress is required to pass such laws as shall effectually prevent the same. Congress shalt also have power to prohibit the in-reduction of slaves from any State mot a member of co In the third paragraph, after the word "law," the words

" or law denying or impairing the right of property in negro slaves" are inserted.

The fifth paragraph is changed by adding thereto the following words: "except by a vote of two thirds of both Houses."

The sixth paragraph is changed by the omission of all, after the word "another."

The following clauses are inserted after the seventh The following clauses are inserted after the seventh paragraph? "Congress shall appropriate no money from the Treasury except by a vote of two thirds of both Houses, taken by yeas and none, unless it be asked and estimated for by some one of the heads of departments and submitted to Congress by the President; or for the purpose of paying its own expenses and contingencies; or for the payment of claims against the Confederate States, the justice of which shall have been judi-cially declared by a tribunal for the investigation of claims against the Government subtich it is berehow made the dulu of against the Government, which it is hereby made the duty of Congress to establish." "All bills appropriating momey shall specify in Federal currency the exact amount of each appropriation and the

\* This provision is said to have been adopted as a part of the Permauent Constitution, by the vote of the States of Georgia, Alabama, Louisiana, and Mississippi, against that of South Carolina and Florida.

purposes for which it is made; and Congress shall grant no | extra compensation to any public contractor, officer, agent, or servant, after such contract shall have been made or such service rendered."

This section is further changed by adding to it the first eight amendments to the Constitution of the United States, eight amendments to the constitution of the United States, followed by this paragraph: "Every law or resolution hav-ing the force of law, shall relate to but one subject, and that shall be expressed in the tille." Section 10. The first paragraph is changed by striking out the following words: "emit bills of credit;" also by inserting the word "or" between the word "atlainder"

inserting the word "or" between the word "attainder" and the words "expost facto." The third paragraph is changed by inserting after the word "tonnage," the following words: "except on seagoing vessels, for the improvement of its rivers and harbors navi-gated by the said vessels; but such duties shall not conflict with any treaties of the Confederate States with foreign nawith any treaties of the Confederate States with foreign na-tions; and any surplus of revenue thus derived shall, after making such improvement, be paid into the common Treas-ury; nor shall any State<sup>3</sup>. Also the following clause is added to this paragraph : "But when any river divides or flows through two or more States, they may enter into com-pacts with each other to improve the navigation thereof."

### ARTICLE II.

SECTION 1. The first paragraph is so altered as to read as follows: " The executive power shall be vested in a President of the Confiderate States of America. He and the Vice President shall hold their offices for the term of six years; but the President shall not be re-eligible. The President and Vice President shall be elected as follows?" The third paragraph is stricken out, and the welfth arti-cle of the amendments to the Constitution of the United

cle of the amendments to the Constitution of the United States substituted.

The fifth paragraph is changed in the following respects : the words "or a citizen of the United States" are stricken out, and the words "of the Confederate States, or a citizen out, and the words " of the Confederate States, or a citizen thereof" substituted; also after the word "Constitution," the following words are inserted: " or a citizen thereof born in the United States, prior to the 20th of December, 1860;" also the words "United States" at the close of the paragraph, are stricken out, and the words " limits of the Confederate States, as they may exist at the time of his election" are substituted.

The eighth paragraph is changed by substituting the word "enters" for the word "enter."

The last paragraph is changed by striking out the words "of the United States" at the close, and substituting the word "thereof."

SECTOR 2. The second paragraph is changed by adding thereto the words following: "The principal officer in each of the executive departments, and all persons connected with the diplomatic service, may be removed from office at the pleasure of the President. All other civil officers of the expressure of the Trestuent. All other civil opters of the ex-ecutive department may be removed at any time by the Pres-ident, or other appointing power, when their services are unnecessary, or for dishonesty, incapacity, inefficiency, mis-conduct, or neglect of duty; and when so removed, the re-moval shall be reported to the Senate, together with the reasons therefor?

The third paragraph is changed by dropping the word "up" after the word "fill." SECTION 3. This section is changed in the following res-pects: the first word "He" is stricken out and the words "The President" substituted; also the word "Union" is stricken out and the word "Confederacy" substituted.

### ARTICLE III.

ARTICLE III. SECTION I. In this section the word "supreme" is stricken out and the word "Superior" substituted in the first sen-tence of the first paragraph. SECTION 2. This section is changed as follows: after the clause "citizens of another State," the following clause is inserted: "where the State is plaintiff," also the next simple sentence is changed by inserting the words "claiming lands under grants" between the word "citizens" and the word "of 2" also the following clause is stricken out: "be-tween citizens of the same State claiming lands under grants of different States," also the following clause is added to the first paragraph: "but no State shall be sued by a citi-zen or subject of any foreign State;" also in the first part of this paragraph the following clauses: "the aw and equity," and "arising under this Constitution," are transposed. In the third paragraph, in the clause "where the said crimes shall have been committed," the word "crimes" is changed to the word "crime."

### ARTICLE IV.

SECTION 1. No change. SECTION 2. 'The first paragraph is changed by adding thereto the following clauses: "and shall have the right of

transit and sojourn in any State of this Confederacy, with their slaves and other property; and the right of property in such slaves shall not be thereby impaired."

The third paragraph is changed in the following respects: after the first word "No" the words "slave or" are in-serted; also the words "one State" are stricken out, and series; also the words "one State" are stricken out, and the words "any State or Territory of the Confederate States" are substituted; also, after the word "escaping" the words "or lawfully carried into" are inserted; also the words "slave belongs, or to whom such" are inserted between the word "such" and the word "service," near the close of the paragraph.

SECTION 3. The first paragraph is changed by striking out all the words to and including the word "Union," and substituting the words "Other States may be admitted into this Confederacy by a vote of two thirds of the whole House of Representatives and two thirds of the Senate, the Senate voting by States."

The second paragraph is changed by striking out all after the word "regulations" and inserting the following clauses : "concerning the property of the Confederate States, including the lands thereof."

" The Confederate States may acquire new territory, and "The Confederate States may definite new territory, and Congress shall have power to legislate and provide govern-ments for the inhabitants of all territory belonging to the Confederate States lying without the limits of the several States, and may permit them, at such times and in such manner as it may by law provide, to form States to be ad-mitted into the Confederacy. In all such territory, the in-stitution of negro slavery as it now exists in the Confederate States shall be recognized and protected by Congress and by the territorial government, and the inhabitants of the several Confederate States shall have the rind to take Confederate States and territories shall have the right to take to such territory any slaves lawfully held by them in any of the States or Territories of the Confederate States." SECTION 4. This section is changed as follows: the words

SECTION 4. This section is changed as follows: the words "in this Union" are stricken out, and the words "that now is, or hereafter may become, a member of this Confederacy" inserted: also, after the word "Legislature," the words "cannot be convened." are dropped, and the words "is not in session," substituted; also this entire section is omitted, as a section, and placed at the end of section 3, as a paragraph thereof.

### ARTICLE V.

This article is stricken out entirely, and the following substituted: " Upon the demand of any three States, legally assembled in their several Conventions, the Congress shall summon a Convention of all the States, to take into considerasummon a Convention of all the States, to take into considera-tion such amendments to the Constitution as the said States shall concur in suggesting at the time when the said demand is made; and should any of the proposed amendments to the Constitution be agreed on by the said Convention—volue by States—and the same be ratified by the Legislatures of two-thirds of the several States, or by Conventions in two-thirds thereof as the one or the other mode of ratification may be proposed by the general Convention—they shall henceforward form a part of this Constitution. But no State shall, without its consent, be deprived of its equal representation in the Senate."

### ARTICLE VI.

This article is preceded by the following paragraph: "The Government established by this Constitution is the suc-Ine covernment established by this Constitution is the suc-cessor of the Provisional Government of the Confederate States of America, and all laws passed by the lutter shall continue in force until the same sh. It be repeated or modified ; and all the officers appointed by the same shall remain in office until their successors are appointed and qualified or the offices abolished." "The first numerable is showed by the

The first paragraph is changed by striking out the word "Confederation" and substituting the words "Provisional Government."

In the second paragraph, the words "which shall be" following the word "States" are stricken out. This article is also chauged by adding thereto the 9th and

10th articles of the Amendments to the Constitution of the United States, as two additional paragraphs, after adding to the 9th Article the words: "of the several States."

### ARTICLE VII.

The first paragraph is changed by substituting the word "five" for the word "nine" preceding the word "States." Also the following paragraph is added to the foregoing

as amended :

as amended: "When five States shall have ratified this Constitution, in the manner before specified, the Congress under the Provis-tional Constitution shall prescribe the time for holding the elec-tion of President and Vice President; and for the meeting of the electoral college; and for counting the votes and inaugu-rating the President. They shall also prescribe the time for holding the first election of members of Congress under this Constitution, and the time for assembling the same. Until the

assembling of such Congress, the Congress under the provis-ional Constitution shall continue to exercise the legislative powers granted them, not extending beyond the time timited by the Constitution of the Provisional Government." The final paragraph: "Done in convention by the unan-mous consent," &c., is omitted, and the words following substituted: "Adopted unanimously, March 11, 1861."

# Index to the Constitution of the Uni-ted States, and of the "Confederate" States. U. S. ("C." S.

	U.	s.	" C.	" S.
	Art.	Sec.	Art.	Sec.
А.		-		
Arts and Sciences, to be promoted Acts, records, and judicial proceedings of each State entitled to faith and	1	S	1	8
credit in other States	4	1	4	1
made	5	1	5	1
Apportionment of representatives	2 1	$\frac{2}{2}$	$^{2}_{1}$	$\frac{2}{2}$
Appropriations by law	î	9	î	õ
two years Armies, Congress to raise and support.	1 1	8 8	1	8 8
dame right of people to levy and bear.		m'd. m'd.	1 1	9 9
Assemble, people may. Attainder, bill of, prohibited to Con- gress	1	9	i	9
of treason shall not work corruption	1	10	1	10
of blood, or forfeiture except &c Appropriation bills may be approved in part and disapproved in part	3	3	3	3
		•••••	1	7
B.	• •		1	
Bail, excessive, not to be required Bankruptcy laws to be uniform Balls for registing revenue shall originate	1	m'd. 8	î	9 8
Bills for raising revenue shall originate in the House of Representatives before they become laws, shall be	1	7	1	7
ed by President; or, if disapproved,				
shall be passed by two-thirds of each house not returned in ten days, unless an	1	7	1	7
adjournment intervene, shall be			-	
laws Borrow money, Congress may Bounties cannot be granted from the	1	7 8	1 1	7 8
treasury		·····	1	8
с.				
Capitation tax, apportionment of	1	9 2	$1 \\ 1$	9 2
Census, to be made every ten years Citizens of each State to be entitled to the privileges of citizens in the sev- eral States	4	2	4	2
of one State not to sue those of auoth- er in Confederate courts			3	2
Claims, no prejudice to certain: Court of, to be established	4	3	1	9
Coasting trade, regulated Coins, Congress to fix value of foreign	1 1	9 8	$\frac{1}{1}$	9 8
Commerce, Congress to regulate such regulations to be uniform	1 1	8	$\hat{1}$ 1	8 9
Commissions to be granted by the Pres-	1	9 2	1	9 2
ident Common law recognized and established Congress vested with legislative power	7 Å 1	m'd. 1	ĩ	9 1
may alter the regulations of State legislatures concerning elections of	-	-	Î	-
Senators and Representatives, ex- cept as to <i>place</i> of choosing Sena-				
also, except as to time and place of	1	4		
choosing Senators	ï	4	1 1	4 4
officers of Government cannot be members of	1	6	1	6
death, &c., of President and Vice- President	2	1	2	1
may determine the time of choosing	-			-

·	. U.	U. S.		, s.	
		Art. Sec.			
electors of President and Vice-Pres- ident	2	1	2	1	
may invest the appointment of infe- rior officers in the President alone, in the courts of law, or the heads					
of departments	2	2	2	2	
Supreme Court	3	1	3	1	
may declare the punishment of trca- son	3	3	3	3	
the acts and records of each State	4	1	4	1	
to assent to the formation of new States	4	3	4	8	
may propose amendments to Consti- tution, or call convention	5	1	5	1	
to lay and collect taxes to borrow money	1	-8 -8	1	8	
to regulate commerce to establish uniform laws of bank-	1	8	1	8	
ruptcy and navigation	1	8	1	8,	
to coin money, regulate the value of coin, and fix the standard of	1	0	1	8	
weights and measures to punish counterfeiting	î	8	1	8	
to constitute tribunals inferior to the Supreme Court	1	8	1	8	
to punish piracies, felonics on the high seas, and offences against the					
laws of nations	1	8	1	8	
roads, (or routes) to authorize patents to authors and	1	8	1	8	
inventors	1	8	1	8	
marque, and make rules concern-	1	8	1	8	
ing captures to raise and support armies to provide and maintain a power	1	8 8	1	8	
to provide and maintain a navy to make rules for the government of			-	8	
the army and navy to call out the militia in certain	1	8	1	8	
cases to pass laws necessary to carry the	1	8	1	8	
enumerated powers into effect to dispose of and make rules con-	1	8	1	3	
cerning the territory or other prop- erty of the government	4	3	4	3	
President may convene and adjourn, in certain cases	2	3	4	3	
shall not appropriate money unless it be estimated for by the head of a					
department and submitted by the		-			
President, unless by a vote of two thirds of both Houses—or to pay					
its own expenses, or claims against the Confederacy			1	9	
officers, &c., not to be granted			1	9	
Compacts between two or more States may be entered into, to improve					
navigation of rivers in certain cases			1	10	
Constitution, how amended laws and treaties declared to be the	5	1	5	1	
supreme law number of States to ratify to render	6	1	6	1	
operative	7	1 10	7	1 10	
Conventions for proposing amendments to the Constitution	5	1	5	1	
Counterfeiting, Congress to provide for puuishment of	1	8	1	8	
Court, Supreme, its original and appel-					
late jurisdiction Courts, inferior to the Supreme Court,	3	2	3	2	
may be ordained by Congress Ditto Ditto	1 3	8 1	1 3	8 1	
Crimes, persons accused of, fleeing from justice, may be demanded	4	2	4	2	
How to be tried Criminal prosecutions, proceedings in	3 6 A	2 m'd.	3	2 9	
D.					
		-	-		
Debts against the Confederation to be valid against the Provisional Governm't	6	1			
Duties to be levied by Congress, and to	•••••		6	1	

# INDEX TO CONSTITUTIONS.

	υ.	s.	"C.	" s.	1	U.	S. 1	"C."	, <b>S</b> .
	Art.	Sec.	Art.	Sec.		Art.	Sec.	Art.	Sec.
be uniform	1	8	1	8 9	Jury trial, further regulated	6 A	m'd.	1	9
Duties, further provision respecting cannot be levied by the States	1 1 1	9 10 9	1 1	10	secured in suits at common law where the valuo in controversy shall exceed twenty dollars	17 A			
on exports prohibited on exports may be authorized by	1	•	1	9	L.	'A	m'd.	1	9
two-thirds vote on imports and exports imposed by States, shall enure to the treasury		•••••	1	5	Law, what is declared the supreme	6	1	6	1
of the General Government may not be laid en imports to foster any branch of industry	1	10	1	10 8	common, recognized and established President to see faithfully executed of the Provisional Government to be		m'd. 3	12	9 3
Е.					valid every, shall relate to but one subject,			6	1
Elections of Senators and Representa-					which its title shall express		·	1	9
tives shall be prescribed by the States	1	4	1	4	(See Congress.) Loans, authority to make	1	8	1	8
qualifications and returns of mem- bers of Congress to be determined					M.				
by each House Electors of President and Vice Presi-	1	5	1	5	Marque and reprisal, letters of	1	8 8	111	8
dent, how chosen, and their duties altered (see 12th amendment) to vote	2	1	2	1	Militia, to be called out to be officered by the States to be commanded by the President	$\begin{vmatrix} 1\\ 1\\ 2 \end{vmatrix}$	82	1 2	8 8 2
the same day throughout the Uni- ted (or Confederate) States	2	1	2	1	their right to keep and bear arms secured		m'd.	1	9
no Senator, or Representative, or public officer shall serve as Enumeration every ten years	$^{2}_{1}$	1	$^{2}_{1}$	$\frac{1}{2}$	Money shall be drawn from the treas- ury only by appropriation laws	1	9	1	9
first within three years Executive power vested in a President,		2	i	2	Congress to coin and regulate value of	1	8	1	8
(see President,) Exports, not to be taxed	$\frac{2}{1}$	1 9	2	1	States cannot coin	1	10	1	10
may be, by two-thirds vote	 1		1 1	9 9	N.				
prohibited to States	1	10	1	10	Naturalization, uniform rules of Navy, Congress to provide and govern	1 1	8 8	1	8
<b>F</b> .					Nobility, titles of, not to be granted by the United (or Confederate) States nor by the States	1 1	·9 10	1	9 10
Fines, excessive, prohibited Fugitives, from justice, to be delivered		m'd.	1	9	O.	1	10	1	10
up from service, to be delivered up	4 4	$\frac{2}{2}$	44	$\begin{array}{c} 2\\ 2\end{array}$	Officers of the House of Representa-				
H.			ļ		tives to be chosen by the House of the Senate to be chosen by the	1	2	1	2
Habeas Corpus, writ of, can only be suspended in cases of rebellion or					Scnate	1	8	1	3
invasion House of Representatives—(see Repre-	1	9	1	9	Order of one house requiring the con-	2	4	2	4
sentatives.) Heads of Executive Departments may be			l		currence of the other Oath of the President of the public officers	1 2 6	$\begin{vmatrix} 7\\1\\1 \end{vmatrix}$	1 2 6	1
allowed to have seats in either Honse for certain purposes			1	6				Ŭ	
I.					<b>P.</b> Pardons, President may grant	2	2	2	2
Impeachment to be brought by House of Representatives	1	2	1	2	Patents to be granted to inventors Petition, right of	1	8 m'd.	1	8
tried by the Senate judgment on	1 1	3	1 1	3 3	Persons held to service or labor, their importation or migration into the				
all civil officers liable to any judicial or other federal officers	2	4	2	4	United States may be prohibited after 1808	1	9		
resident and acting solely in any State, liable to, by two-thirds vote					escaping from one State into another shall be delivered np	4	2	4	2
of legislature thereof Importation of Slaves, not prohibited			1	2	or lawfully carried from one State to another			4	2
till 1808 prohibited Internal Improvements, Congress shall	1	9 	1	9	Piracy, Congress to prescribe punish- ment for	1	8	1	8
not have power to appropriate money for			1	8	Post offices and post roads (and routes,) establishing of Post office Department to be self-sustain-	1	8	1	8
J.			1	Ů	ing after 1st of March 1863 Powers not delegated to Congress nor	·····		1	8
Judges shall hold their offices during					prohibited to the States are re- served	10 A	m'd.	6	1
good behavior their compensation	3	1 1	3 3	1 1	legislative,(see Congress ;) executive, (see President ;) judicial, (see Judi-				
Judiciary, tribunals inferior to the Su- preme Court may be erected	1	8	1	8	cial.) Presents from foreign powers to public				
Judicial Power vested in a Supreme Court (or Superior Court) and in-			0		officers prohibited Press, freedom of secured	1 1 A	9 m*d.	1 1	9 9
ferior courts powers of the judiciary restrictions as to suits against a State	3	1 2 m'd.	333	$\begin{array}{c} 1\\ 2\\ 2\end{array}$	President of the United States and of the Confederate States vested with executive power	2	1	2	1
Judicial Proceedings of each State are entitled to credit in all	11 A 4	1 I	4	1	shall be chosen for four years shall be chosen for six years	2	i	2	1
Jury trial secured, and shall be held in the State where the crime shall					how elected	2 12 A	1 m'd.	$\frac{2}{2}$	1
have been committed	3	2	3	2	qualifications for	2	1	2	1

# INDEX TO CONSTITUTIONS.

-	σ.	s.	"C.	" S.
-	Art.	Scc.	Art.	Sec.
President, not re-eligible	2	1	$\frac{2}{2}$	1 1
compensation of	$\frac{2}{2}$	1	22	1
President, commander of army, navy,	2	4	2	4
and militia may require the written opinions of heads of departments	2	2	2 2	2
may make treaties with consent of	2	2	2	$\frac{2}{2}$
the Senate may appoint to office with consent of	2	2	2	2
the Senate shall fill vacancies happening during	2	2	2	2
shall give information to Congress	2	2	2	2
and recommend measures may convene both Houses or either	2	3	2	3
may adjourn them in case of disa-	2	3	2	3
shall receive embassadors and pub-	$\frac{2}{2}$	3	2	3
lic ministers shall take care that the laws are faithfully excented	2	3	$\frac{2}{2}$	3
shall commission all officers Privileges and immunities of members	2	3	2	3
of Congress of citizens, (see Citizens; also, Rights.)	1	6	1	6
<b>Property</b> , Congress to provide for care 1	4	3	4	3
of public shall not be taken for public use without just compensation	5 A	m'd.	1	9
Congress shall pass no law impairing or denying the right of, in slaves			4	2
Punishment, cruel and unusual, pro- hibited	8 A	m'd.	1	9
Q.				
Quorum for business, what shall be a of States, in choosing a President by	1	5	1	5
the House of Representatives Quartered—No soldiers to be quartered	2	1	2	1
upon a citizon	3 A	m'd.	1	9
R.				
Receipts and expenditures, accounts of, to be published	1	9	1	9
Records, how to be authenticated Religion-no law to be made; free ex-	4 1 A	1 m'd.	4 1	1 9
ercise of religious test not required Reprieves granted by the President	6 2	1 2	62	$\overset{\circ}{\overset{\circ}{1}}$
Representatives, House of, members of cbosen every second year	1	2	1	2
qualifications of members of	1 1	$\frac{2}{2}$	1	$^{2}_{2}$
vacancies, how supplied shall choose their officers	1 1	$\frac{2}{2}$	1	$\frac{2}{2}$
shall have the power of impeach- ment	1	2	I	2
shall be the judge of the election and qualification of its members	1	5	1	5
what shall be a quorum any number may adjourn, and com-	1	5	1	5
may determine the rules of proceed- ing	1	5 5	1	5
may punish or expel a member shall keep a journal and publish, ex-	î	5	Î	5
shall not adjourn for more than three	1	5	1	5
days, nor to any other place, with- out the consent of the Senate	1	5	1	5
ene fifth may require the yeas and nays	I	5	1	5
shall originate bills for raising reve- nue	1	7	1	7
privileged from arrest, except in cer-	1	6	1	6
tain cases	1	6	1	6
for speech or debate in the House shall not be appointed to office	1	6	1	6 6
shall not be electors of President	2	1	2	1

Representatives and direct taxos, apportioned according to numbersArt. Sec.Art. Sec.Representation of a State y vacancies in, supplied until a new election by excetive anthority.121Resolution, order, or vote, requiring the concurrence of both Houses or Representatives.171Removal of certain civil officers to be reported by the President to the Senter, with his reasons therefore.171Removal of certain civil officers to be priv- ileges of citizens of the server all for onscience in matters of recigion of speech and of the press.171Removal of speech and of the press.22119to be secure from unreasonable searches and seizures.3m <sup>2</sup> d.19to be compelled to be a winess against himself.5A m <sup>2</sup> d.19to be twice jeoparded for the same offace.5A m <sup>2</sup> d.19private property not to be taken for public use.5A m <sup>2</sup> d.19in criming prosecutions shall have and to be twice jeoparded for the same offace.5A m <sup>2</sup> d.19private property not to be taken for public use.5A m <sup>2</sup> d.19in criming prosecutions shall have and unbus bursterwind in the means necessary for his defence.6A m <sup>2</sup> d.1in criming prosecutions shall have and unbus bursterweights.1515in criming prosecutions shall have and unbus bursterweights.1515in c		U	U.S.		" <b>S</b> .
the contraction of a State ; vacancies in, supplied until a new election by executive authority		Art	Sec.	Art.	Sec.
in, supplied until a new election by exocutive authority	tioned according to numbers	1	2	1	2
undergo the formalities of bills	in, supplied until a new election by executive anthority	1	2	1	2
Reveal of certain civil officers to be reported by the President to the Senute, with his reasons therefor.2Reverve bills to originate in the House of Representatives.171Rights of the citizen declared to be priv- ileges of citizens of the several States.171Rights of the citizen declared to be priv- ileges of citizens of the several States.171Rights of the citizen declared to be priv- ileges of citizens of the several states.171Rights of the citizen declared to be priv- ileges of citizens of the several to be casemble and petition.119to be secure from unreasonable searches and seizures.2A m'd.19to be tree from answering for a 	undergo the formalities of bills	1	7	1	7
Revenue bills to originate in the House of lepresentatives.1717Rights of the citizen declared to be privileges of citizens of the several states.1717Rights of the citizen declared to be privileges of citizens of the several religion.4242Riberty of conscience in matters of religion.1Am'd.19freedom of speech and of the press.1Am'd.19to be scentp from the quartering of scidiers.3Am'd.19to be scentp from unreasonable scarches and seizures.5Am'd.19to be scentp from unreasonable scarches and seizures.5Am'd.19not to be deprived of life, liberty, or property without due process of law.5Am'd.19not to be deprived of life, liberty, or property without due process of law.5Am'd.19in criminal prosecutions shall have speedy trial by jury, with all the means necessary tor bis defence netwide of life, liberty, or protects chail shall not be required, excessive fues inposed, nor cruci and nusual punishments inflicted5Am'd.19Rules, each house shall determine its ownt.15155S.Sat of Government, exclusive legisla- ton.13133Kules, each bouse shall determine its ownt.15155S.Sat of Government, exclusive legisla- ton.13	Renoval of certain civil officers to be				
likes of citizens of the several States4242liberty of conscience in matters of religion1 A m'd.19freedom of speech and of the press. to assemble and petition1 A m'd.19to assemble and petition1 A m'd.19to be exempt from the quartering of soldiers3 A m'd.19to be secure from unreasonable searches and seizures. archment by a jury3 A m'd.19not to be twice jeoparded for the same offence5 A m'd.19not to be deprived of life, liberty, or property without due process of iaw5 A m'd.19private property not to be taken for public use.5 A m'd.19in criminal prosecutions shall have speedy trial by jury, with all the means necessary for his defence in civil cases trial to be by jury, and shall only be re-examined accord- ing to common law7 A 'md.19Exules, each house shall bet crequired, excessive fues and seizures, security against. Senate, composed of two Senators from and qualifications of Senators.151Stati bo the judge of the electious and qualifications of senators.1333istail bo the judge of the electious and qualifications of senators.1515may thermine its rules.15155may determine its rules.15155may determine its rules.15155istal ho the indeg	Rerenue bills to originate in the House of Representatives	1	7		
religion	ileges of citizens of the several States	4	2	4	2
to assemble and petition	religion				
to be exempt from the quartering of soldiers	to assemble and petition	IA	m'd.	1	
by our section from answering for a crime, unless on presentment or indictment by a jury,	to be exempt from the quartering of				9
to bo free from answering for a crime, unless on presentment or indictment by a jury	to be secure from unreasonable	3 A	m'd.	1	9
indictment by a jury	to be free from answering for a crime, unless on presentment or	4 A	m'd.	1	9
same original problem to be compelled to be a witness against himself.     5 A m'd.     1     9       not to be deprived of life, liberty, or property without due process of law	indictment by a jury not to be twice jeoparded for the	5 A	m'd.	1	9
against himself.5 A m'd.19not to be deprived of life, liberty, or property without due process of law.5 A m'd.19private property not to be taken for public use.5 A m'd.19in criminal prosecutions shall have speedy trial by jury, with all the means necessary for his defence.6 A m'd.19in civil cases trial to be by jury, and shall only be re-examined accord- ing to common law.6 A m'd.19excessive bail shall not be required, excessive fues imposed, nor cruel and unusual punishments inficted.7 A 'md.19enumeration of certain rights not to operate against reserved rights9 A m'd.61 <i>Rules</i> , each house shall determine its own1515Searches and seizures, security against Searches and seizures, security against shalt keet higde of the electious and qualifications of Senators.1313bow chosen, classed, and terms of service.13133shall be the judge of the electious and qualifications of senators.1515may tumber may adjourn, and com- pel attendance of absentees.1515may propose amendments to bills for raising revenue.15155may propose amendments to bills 	same onenco	5 A	m'd.	1	9
law	against himself	5 A	m'd.	1	9
public use.5 A m'd.19in criminal prosecutions shall have speedy trial by jury, with all the means necessary for his defence.6 A m'd.19in civil cases trial to be by jury, and shall only be re-examined accord- ing to common law.6 A m'd.19excessive bail shall not be required, excessive bail shall not be required, enumeration of certain rights not to operate against reserved rights.7 A 'md.19Rules, each house shall determine its own9 A m'd.61Rules, each house shall determine its own1515S.8.1515Searches and seizures, security against. searches and seizures, security against. shall be the judge of the electious and qualifications of Senators.131shall be the judge of the electious and qualifications of absentees.1515may pumber may adjourn, and com- pel attendance of absentees.1515may punish or expel a members.1515may punish or expel a members.1515may propose amendments to bills for raising revenue.1515may propose amendments to bills for raising revenue.1515may propose amendments to bills for raising revenue.15151515151and qualifications of the other house.1515ma	law	5 A	m'd.	1	9
a shall only be re-examined according to common law.       7 A 'md.       1       9         excessive ball shall not be required, excessive fines imposed, nor cruel and unusual punishments inflicted.       8 A m'd.       1       9         enumeration of certain rights not to operate against reserved rights.       9 A m'd.       1       9         Rules, each house shall determine its own.       1       5       1       5       1       5         Sear of Government, exclusive legislation conservice.       1       3       1       3       3         Seart of Government, exclusive legislation of service.       1       3 <t< td=""><td>public use in criminal prosecutions shall have</td><td>5 A</td><td>m'd.</td><td>1</td><td>9</td></t<>	public use in criminal prosecutions shall have	5 A	m'd.	1	9
shall only be re-examined according to common law	in civil cases that to be by jury, and	6Λ	m'd.	1	9
excessive fines imposed, nor crucil and unusual punishments inficted enumeration of certain rights not to operate against reserved rights 	shall only be re-examined accord- ing to common law	7 A	'md.	1	9
operate against reserved rights9 A m²d.6Rules, each house shall determine its own9 A m²d.61Sate of Government, exclusive legislationtion15Sear of Government, exclusive legislation15Sear of Government, exclusive legislation15Sear of Government, exclusive legislation18Sear of Government, exclusive legislation18Sear of Government, exclusive legislationSear of Government, exclusive legislationSear of Government, exclusive legislationSear of Government, exclusive legislationSear of Covernment, exclusive legislation13131313131313131313131313131	excessive fines imposed, nor cruel and unusual punishments inflicted	8 A	ın'd.	1	9
own       1       5       1       5         Seat of Government, exclusive legislation       1       8       1       8         Searches and seizuros, security against       4 Å m'd.       1       9         Searches and seizuros, security against       1 Å Å m'd.       1       9         Searches and seizuros, security against       4 Å m'd.       1       9         Searches and seizuros, security against       1 Å Å m'd.       1       9         Searches and seizuros, security against       4 Å m'd.       1       9         Searches and seizuros, security against       4 Å m'd.       1       9         Searches and seizuros, security against       4 Å m'd.       1       9         Searches and seizuros, security against       4 Å m'd.       1       9         gualifications of Senators	operate against reserved rights	9 A	m'd.	6	1
Seat of Government, exclusive legislation181tionSearches and seizures, security against. $4 \text{ A} \text{ m'd.}$ 1Searches and seizures, security against. $4 \text{ A} \text{ m'd.}$ 1Searches and seizures, security against. $4 \text{ A} \text{ m'd.}$ 1Searches and seizures, security against. $4 \text{ A} \text{ m'd.}$ 1Searches and seizures, security against. $4 \text{ A} \text{ m'd.}$ 1Searches and seizures, security against. $1 \text{ S} \text{ I}$ 3how chosen, classed, and terms of131gualifications of Senators.131gualifications of Senators.131and qualifications of its members.151and qualifications of its members.151any number may adjourn, and competation and publish the151any punish or expel a member.151shall kee on a journal and publish the151shall consent of the other house.151oue-fith may require the yeas and151oue-fith may require the yeas and151any propose amendments to bills1131for raising revenue.1313effect of their judgment on impeachment.166may propose and mont.1616members of, privileged from arrest.161members not to be questioned for <t< td=""><td></td><td>1</td><td>5</td><td>1</td><td>5</td></t<>		1	5	1	5
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	s				
Searches and seizures, security against. $4 \ A \ m'd.$ $I$ $9$ Senate, composed of two Senators from each State	Seat of Government, exclusive legisla-	1	8	1	8
how chosen, classed, and terms of service	Searches and seizures, security against Senate, composed of two Senators from	4 A	m'd.	I	9
qualifications of Senators1313Vice-President to be President of the1313shall choose their officers1313shall bo the judge of the elections1313and qualifications of its members1515any number may adjourn, and compel attendance of absentees1515may number may adjourn, and compel attendance of absentees1515may pumber may adjourn for more than three1515shall keep a journal and publish the same, except parts requiring sector cory1515out the consent of the other hous15155may propose amendments to bills15155may propose amendments to bills1717shall try impeachments13133effect of their judgment on impeach- meut16166members of, privileged from arrest16166members not to be questioned for words spoken in debate1616	how chosen, classed, and terms of				
shall be the judge of the elections1313shall be the judge of the elections1515and qualifications of its members1515what number may adjourn, and compelatiendance of absentees1515may determine its rules15155may determine its rules15155may determine its rules15155shall keep a journal and publish the same, except parts requiring sectorecy.1515crecy.151555out the consent of the other house1515nays151555oue-fith may require the yeas and nays1515may propose amendments to bills for raising revenue1313effict of their judgment on impeach- mout16166members of, privileged from arrest16166members not to be questioned for words spoken in debate16166	qualifications of Senators	1	3	1	3
what number may adjourn, and compelation and completetee and compelation and compelation and co	shall choose their officers				
any number may adjourn, and complet attendance of absentees     1     5     1       pel attendance of absentees     1     5     1     5       may determine its rules     1     5     1     5       may punish or expol a member     1     5     1     5       shalk keep a journal and publish the same, except parts requiring sectory     1     5     1     5       shall hot adjourn for more than three days, nor to any other place, without the consent of the other house     1     5     1     5       oue-fifth may require the yeas and nay second the yeas and propose amendments to bils for raising revenue     1     5     1     5       redict of their pudgment on impeachment     1     3     1     3     1       effict of their pudgment on impeachments     1     6     1     6       members of, privileged from arrest     1     6     1     6       members not to be questioned for words spoken in debate     1     6     1     6	and qualifications of its members				5
may determine its rules.I5I5may poins or except parts requiring sector1515shall keep a journal and publish the same, except parts requiring sector1515shall not adjourn for more than three days, nor to any other place, with- out the consent of the other house.1515oue-fifth may require the yeas and for raising revenue.15155may propose amendments to bills for raising revenue.1515addit their judgment on impeach- mout.1313compensation to be ascertained by law.1616members of, privilcged from arrest1616members not to be questioned for words spoken in debate.1616	any number may adjourn, and com-				5
shall keep a journal and publish the same, except parts requiring secrecy			- 1	-	-
crecy.1515shall not adjourn for more than three days, nor to any other place, with- out the consent of the other house151oue fifth may require the yeas and nays.1515may propose amendments to bils for raising revenue1717shall try inpeachments.13133effect of their judgment on impeach- mout1313compensation to be ascertained by law1616members of, privileged from arrest1616members not to be questioned for words spoken in debate1616	shall keep a journal and publish the	1	5	1	5
out the consent of the other house     1     5     1     5       oue-fifth may require the yeas and nays     1     5     1     5       may propose amendments to bils     1     5     1     5       may propose amendments to bils     1     7     1     7       shall try impeachments     1     3     1     3       effect of their judgment on impeach- mont     1     3     1     3       compensation to be ascertained by law     1     6     1     6       members of, privileged from arrest     1     6     1     6       members not to be questioned for words spoken in debate     1     6     1     6	shall not adjourn for more than three	1	5	1	5
nays	out the consent of the other house	1	5	1	5
sharin try inpeacinitents     1     3     1       effect of their judgment on inpeachmont.     1     3     1       compensation to be ascertained by     1     6     1       law     1     6     1     6       members of, privileged from arrest.     1     6     1       members not to be questioned for words spoken in debate     1     6     1	nays.	1	5	1	5
mont     1     3     1     3       compensation to be ascertained by law     1     6     1     6       members of, privileged from arrest     1     6     1     6       members not to be questioned for words spoken in debate     1     6     1     6	shall try inpeachments				
law	ment	1	3	1	3
words spoken in debate 1 6 1 6	law members of, privileged from arrest				
	members not to be questioned for words spoken in debate				

# "CONFEDERATE" CONSTITUTION INTERPRETED.

	v.	s.	"C	" S.
	Art.	Sec.	Art.	Sec.
Senator shall not be elector	2	1	2	1
Senators and Representatives, election of, how prescribed	1	4	1	4
Slaves, their importation may be pro- hibited after 1808	1	9		
their importation prohibited, no law to be passed impairing right of	_			
property in right of transit and sojourn with, in	•••••	•••••	1	9
any State, guarantied introduction of, from any State not a member of the Confederacy, may	•••••	•••••	4	2
be prohibited by Congress	 3 A	 m'd.	11	9 9
Speaker, how chosen	1	2 m'd.	1	29
States prohibited from—			-	Ū
entering into treaty, alliance, or con- federation	1 1	10 10	1 1	10 10
granting letters of marque	1	10	i	10
emitting bills of credit making anything a tender but gold	1	10		10
and silver passing hills of attainder, expost facto	1	10	1	10
laws, or laws impairing contracts granting titles of nobility	$1 \\ 1$	10 10	1	10 10
laying duties on imports and exports laying duties on tonnage	1 1	10 10	1 1	10 10
may lay tonnage duty on sea-going vessels for the improvement of riv-				
ers, &c kceping troops or ships of war in time			1	10
of peace entering into any agreement or com-	1	10	1	10
pact with another State or foreign Power	1	10	1	10
may enter into compact for improve- ment of certain rivers			1	10
engaging in war	1	10	1	10
Union (or Confederacy) new, may be admitted upon two-	4	3	4	3
thirds vote of both Houses, the Sen- ate voting by States			4	3
may be formed within the jurisdic- tion of others, or by the junction		ļ		
Congress and the Legislatures of				
the States concerned State judges bound to consider treaties,	4	3	4	3
the Constitution, and the laws under it, as supreme	6	1	6	`1
State, every, guarantied a republican form of government, protected by				
the United (or Confederate) States Supreme Court—(See Court and Judi- ciary.)	4	4	4	4
Suits at Common Law, proceedings in	7 A	m'd.	1	9
т				
Taz, direct, according to representa-	1	2	1	2
shall be laid only in proportion to census	1	9	1	9
on exports prohibited except by vote of two-thirds of both	1	9		
Houses Tender, what shall be a legal	1	10	1	9 10
Territory, or public property, Congress may make rules concerning	4	3		
Test, religious, shall not be required Titles, (see Nobility.)	6	1	6	1
Title from foreign State prohibited Treason, defined	13	93	1 3	9 3
two witnesses, or confession, neces- sary for conviction	3	3	3	3
punishment of, may be prescribed by Congress	3	3	3	3
Treasury, money drawn from only by		9	1	9
appropriation Treaties, how made	2	211	2 6	2
the supreme law States cannot make		10	1	10
ν.,		1		
Vacancies happening during the re-		Ŀ		<b>I</b> .

	σ.	s.	"C.	" <b>S</b> .
	Art.	Sec.	Art	Sec.
cess may be filled temporarily by the President	$2 \\ 1 \\ 1 \\ 1 \\ 2 \\ 12 \\ 2$	2 2 7 3 1	2 1 1 2 2	2 2 7 3 1
may be removed by impeachment Vote of one house requiring the concur- rence of the other	2 2 1	4 7	2 2 1	1 4 7
w.	ĺ			
War, Congress to declare	1	8 m'd.	1	8
when and how they shall issue Witness in criminal cases, no one com- pelled to be against himself Weights and Measures, standard of		m'd. m'd. 8	1 1 1	9 8
Υ.				
Yeas and Navs entered on journal and				

 
 Yeas and Nays entered on journal, and published.....
 1
 5
 1
 5

### THEIR CONSTITUTION AS INTERPRETED BY THE "CONFEDERATE" VICE PRESIDENT.

1861, March 21, ALEX. H. STEPHENS delivered a speech at Savannah, in explanation and vindication of the Constitution, from which this is a well known extract:

"The new Constitution has put at rest forever all the agitating questions relating to our peculiar institutions— African slavery as it exists among us—the proper status of the negro in our form of civilization. This was the immediate cause of the late rupture and present revolution. Jefferson, in his forecast, had anticipated this as the 'rock upon which the old Union would split? He was right. What was conjecture with him, is now a realized fact. But whether he fully comprehended the great truth upon which that rock stood and stands, may be doubted. The prevailing ideas entertained by him and most of the leading statesmen at the time of the formation of the old Constitution, were that the enslavement of the African was in violation of the laws of nature: that it was wrong in principle, socially, morally, and politically. It was an evil they knew not well how to deal with, but the general opinion of the men of that day was, that somehow or other, in the order of Providence, the institution would be evanescent and pass away. This idea, though not incorporated in the Constitution, it is true, secured every essential guarantee to the institution while is should last, and hence no argument can be justly used against the common sentiment of the day. Those ideas, however, were fundamentally wrong. They vested upon the assumption of the equality of races. This was an error. It was a sandy foundation, and the idea of a government built upon it; when the 'storm came and the

"Our new Government is founded upon exactly the opposite idea; its foundations are laid, its corner stone rests upon the great truth that the negro is not equal to the *v* white man. That slavery—subordination to the superior race, is bis natural and normal condition. This, our new Government, is the first, in the history of the world, based upon this great physical and moral truth. This truth has been slow in the process of its development, like all other truths in the various departments of science. It has been so even amongst us. Many who hear me, perhaps, can recollect well, that this truth was not generally admitted, even within their day. The errors of the past generation still clung to many as late as twenty years ago. Those at the North who still cling to these errors, with a zeal above, howledge, we justly denominate fanatics. \* \*

the North who still Ging to these errors, with a zeal above knowledge, we justly denominate fanatics. \* \* \* "In the conflict thus far, success has been, on our side, complete throughout the length and breadth of the Confederate States. It is upon this, as I have stated, our actual fabric is firmly planted; and I cannot permit myself to doubt the ultimate success of a full recognition of this primciple throughout the civilized and enlightoned world. "As I have stated, the truth of this principle may be slow in development, as all truths are, and ever have been, in the various branches of science. It was so with the prinopposed of the state of the second of the second and his principles of political economy—it was so with Harvey and his theory of the circulation of the blood. It is stated that not a single one of the medical profession, iving at the time of the announcement of the bruths made him, admitted them. Now they are universally achowledged. May we not, therefore, look with confidence to the ultimate universal acknowledgment of the truths upon which our system rests. It is the first government ever instituted upon principles of strict conformity to natract and the ordination of Providence, in furnishing the haterials of human society. Many governments have been violation of nature's laws. The negro, by nature, or by the curse against Canaan, is fitted for that condition which he couples in our system. The architect, in the construction of universal states. The negro, by nature, or by the surve against Canaan, is fitted for that condition which he couples in our system. The architect, in the construction of universal states and the ordination with the proper materials.

the granite; then comes the brick or the marble. The substratum of our society is made of the material fitted by nature for it, and by experience we know that it is best, not only for the superior, but for the inferior race that it is load be so. It is, indeed, in conformity with the ordinance of the Creator. It is not for us to inquire into the wisdom of His ordinances, or to question them. For His own purposes Ha has made one race to differ from another, as He has made 'one star to differ from another star in glory.'

"The great objects of humanity are best attained when conformed to His laws and decrees, in the formation of gorernmonts, as well as in all things clse. Our Confederacy is founded upon principles in strict conformity with these laws. This stone which was first rejected by the first builders 'is become the chief stone of the corner' in our new edifice. "The progress of disintegration in the old Union may be expected to go on with almost absolute certainty. We are

"The progress of disintegration in the old Union may be expected to go on with almost absolute certainty. We are now the nucleus of a growing power, which, if we are true to ourselves, our destiny, and high mission, will become the controlling power on this continent. To what extent accessions will go on in the process of time, or where it will end, the future will determine."

# ADMINISTRATION OF ABRAHAM LINCOLN.

# Abraham Lincoln's Inaugural Ad- | consistently with the Constitution and the laws, dress, March 4, 1861.

Fellow-citizens of the United States: In compliance with a custom as old as the Government itself, I appear before you to address you briefly, and to take in your presence the oath prescribed by the Constitution of the United States to be taken by the President "before he enters on the execution of his office."

I do not consider it necessary at present for me to discuss those matters of administration about which there is no special anxiety or excitement.

Apprehension seems to exist among the people of the Southern States that by the accession of a Republican Administration their property and their peace and personal security are to be endangered. There has never been any reasonable cause for such apprehension. Indeed, the most anple evidence to the contrary has all the while existed and been open to their inspection. It is found in nearly all the published speeches of him who now addresses you. I do but quote from one of those speeches when I declare that "I have no purpose, directly or indirectly, to interfere with the institution of slavery in the States where it exists. I believe I have no lawful right to do so, and I have no inclination to do so." Those who nominated and elected me did so with full knowledge that I had made this and many similar declarations, and had never recanted them. And more than this, they placed in the platform for my acceptance, and as a law to themselves and to me, the clear and emphatic resolution which I now read :

"Resolved, That the maintenance inviolate of the rights "Resolved, That the maintenance inviolate of the rights of the States, and especially the right of each State to order and control its own domestic institutions according to its own judgment exclusively, is essential to the balance of power on which the perfection and endurance of our polit-ical fabric depend, and we denounce the lawless invasion by armed force of the soil of any State or Territory, no matter under what pretext, as among the gravest of crimes."

I now reiterate these sentiments; and, in doing so, I only press upon the public attention the most conclusive evidence of which the case is susceptible, that the property, peace, and security of no section are to be in anywise endangered by the now incoming Administration. I add, too, that all the protection which, the Constitution or laws by any hypercritical

can be given, will be cheerfully given to all the States when lawfully demanded, for whatever cause — as cheerfully to one section as to another.

There is much controversy about the delivering up of fugitives from service or labor. The clause I now read is as plainly written in the Constitution as any other of its provisions :

"No person held to service or labor in one State, under the laws thereof, escaping into another, shall, in conse-quence of any law or regulation therein, be discharged from such service or labor but shall be delivered up on claim of the party to whom such service or labor may be due

It is scarcely questioned that this provision was intended by those who made it for the reclaiming of what we call fugitive slaves; and the intention of the law-giver is the law. A 11 members of Congress swear their support to the whole Constitution-to this provision as much as any other. To the proposition, then, that slaves, whose cases come within the terms of this clause, "shall be delivered up," their oaths are unanimous. Now, if they would make the effort in good temper, could they not, with nearly equal unanimity, frame and pass a law by means of which to keep good that unanimous oath?

There is some difference of opinion whether this clause should be enforced by national or by State authority; but surely that difference is not a very material one. If the slave is to be surrendered, it can be of but little consequence to him, or to others, by which authority And should any one, in any case, it is done. be content that his oath shall go unkept, on a merely unsubstantial controversy as to how it shall be kept?

Again, in any law upon this subject, ought not all the safeguards of liberty known in civilized and humane jurisprudence to be introduced, so that a free man be not, in any case, surrendered as a slave? And might it not be well at the same time to provide by law for the enforcement of that clause in the Constitution which guaranties that "the citizens of each State shall be entitled to all privileges and immunities of citizens in the several States ?"

I take the official oath to-day with no mental reservations, and with no purpose to construe rules. And while I do not choose now to specify particular acts of Congress as proper to be enforced, I do suggest that it will be much safer for all, both in official and private stations, to conform to and abide by all those acts which stand unrepealed, than to violate any of them, trusting to find inpunity in having them held to be unconstitutional.

It is seventy-two years since the first inauguration of a President under our National Constitution. During that period fifteeu different and greatly distinguished eitizens have, in succession, administered the Executive branch of the Government. They have conducted it through many perils, and generally with great success. Yet, with all this scope for precedent, I now enter upon the same task for the brief constitutional term of four years under great and peculiar difficulty. A disruption of the Federal Union, heretofore only menaced, is now formidably attempted.

I hold that, in contemplation of universallaw, and of the Constitution, the Union of these States is perpetual. Perpetuity is implied, if not expressed, in the fundamental law of all National Governments. It is safe to assert that no Government proper ever had a provision in its organic law for its own termination. Continue to execute all the express provisions to our National Constitution, and the Union will endure forever--it being impossible to destroy it, except by some action not provided for in the instrument itself.

• Again, if the United States be not a Government proper, but an association of States in the nature of the contract merely, can it, as a contract, be peaceably unmade by less than all the parties who made it? One party to a contract may violate it—break it, so to speak; but does it not require all to lawfully rescind it?

Descending from these general principles, we find the proposition that, in legal contemplation, the Union is perpetual, confirmed by the history of the Union itself. The Union is much older than the Constitution. It was formed in fact by the Articles of Association in 1774. It was matured and continued by the Declaration of Independence in 1776. It was further matured, and the faith of all the then thirteen States expressly plighted and engaged that it should be perpetual, by the Articles of Confederation in 1778. And, finally, in 1787, one of the declared objects for ordaining and establishing the Constitution was "to form a more perfect union."

But if destruction of the Union, by one, or by a part only, of the States, be lawfully possible, the Union is *less* perfect than before, the Constitution having lost the vital element of perpetuity.

It follows, from these views, that no State, upon its own mere motion, can lawfully get out of the Union; that resolves and ordinances to that effect are legally void, and that acts of violence, within any State or States, against the authority of the United States, are insurrectionary or revolutionary, according to circumstances.

I, therefore, consider that, in view of the Happily the human mind is so constituted that Constitution and the laws, the Union is un- no party can reach to the audacity of doing

broken, and, to the extent of my ability, I shall take care, as the Constitution itself expressly enjoins upon me, that the laws of the Union be faithfully executed in all the States. Doing this I deem to be only a simple duty on my part; and I shall perform it, so far as practicable, unless my rightful masters, the American people, shall withhold the requisite means, or, in some authoritative manner, direct the contrary. I trust this will not be regarded as a menace, but only as a declared purpose of the Union that it will constitutionally defend and maintain itself.

In doing this there need be no bloodshed or violence; and there shall be none, unless it be forced upon the national authority. The power confided to me will be used to hold, occupy, and possess the property and places belonging to the Government, and to collect the duties and imposts; but, beyond what may be necessary for these objects, there will be no invasion, no using of force against or among the people anywhere. Where hostility to the United States, in any interior locality, shall be so great and universal as to prevent competent resident citizens from holding the Federal offices, there will be no attempt to force obnoxious strangers among the people for that object. While the strict legal right may exist in the Government to enforce the exercise of these offices, the attempt to do so would be so irritating, and so nearly impracticable with all, I deem it better to forego, for the time, the uses of such offices.

The mails, unless repelled, will continue to be furnished in all parts of the Union. So far as possible, the people everywhere shall have that sense of perfect security which is most favorable to calm thought and reflection. The course here indicated will be followed, unless current events and experience shall show a modification or change to be proper, and in every case and 'exigency my best discretion will be exercised, according to circumstances actually existing, and with a view and a hope of a peaceful solution of the national troubles, and the restoration of fraternal sympathies and affections.

That there are persons in one section or another who seek to destroy the Union at all events, and are glad of any pretext to do it, I will neither affirm nor deny; but if there be such I need address no word to them. To those, however, who really love the Union, may I not speak?

Before entering upon so grave a matter as the destruction of our national fabric, with all its benefits, its memories, and its hopes, would it not be wise to ascertain precisely why we do it? Will you hazard so desperate a step while there is any possibility that any portion of the ills you fly from have no real existence? Will you, while the certain ills you fly to are greater than all the real ones you fly from—will you risk the commission of so fearful a mistake?

All profess to be content in the Union, if all constitutional rights can be maintained. Is it true, then, that any right, plainly written in the Constitution, has been denied? I think not. Happily the human mind is so constituted that no party can reach to the audacity of doing

this. Think, if you can, of a single instance 'may be erroneous in any given case, still the in which a plainly written provision of the Constitution has ever been denied. If, by the mere force of numbers, a majority should deprive a minority of any clearly written con-stitutional right, it might, in a moral point of view, justify revolution-certainly would if such right were a vital one. But such is not our case. All the vital rights of minorities and of individuals are so plainly assured to them by affirmations and negations, guarantees and prohibitions in the Constitution, that controversies never arise concerning them. But no organic law can ever be framed with a provision specifically applicable to every question which may occur in practical administration. No foresight can anticipate, nor any document of reasonable length contain express provisions for all possible questions. Shall fugitives from labor be surrendered by National or by State authority? The Constitution does not expressly say. May Congress prohibit slavery in the Territories? The Constitution does not expressly say. Must Congress protect slavery in the Territories? The Constitution does not expressly say.

From questions of this class spring all our constitutional controversies, and we divide upon them into majorities and minorities. If the minority will not acquiesce the majority must, or the Government must cease. There is no other alternative; for continuing the Government is acquiescence on one side or the other. If a minority in such case will secede rather than acquiesce they make a precedent which, in turn, will divide and ruin them; for a minority of their own will secede from them whenever a majority refuses to be controlled by such minority. For instance, why may not any portion of a new Confederacy, a year or two hence, arbitrarily secede again, precisely as portions of the present Union now claim to All who cherish disunion secede from it? sentiments are now being educated to the exact temper of doing this.

Is there such perfect identity of interests among the States to compose a new Union, as to produce harmony only, and prevent renewed secession?

Plainly, the central idea of secession is the essence of anarchy. A majority held in restraint by constitutional checks and limitations and always changing easily with deliberate changes of popular opinions and sentiments, is the only true sovereign of a free people. Whoever rejects it, does, of necessity, fly to anarchy or to despotism. Unanimity is impossible; the rule of a minority, as a permanent arrangement, is wholly inadmissible; so that, rejecting the majority principle, anarchy or despotism in some form is all that is left.

I do not forget the position assumed by some, that constitutional questions are to be decided by the Supreme Court; nor do I deny that such decision must be binding, in any case, upon the parties to a suit, as to the object of that suit, while they are also entitled to very high respect and consideration in all parallel cases by all other departments of the Government. And stitution amended. While I make no recomwhile it is obviously possible that such decision | mendation of amendments, I fully recognize the

evil effect following it, being limited to that particular case, with the chance that it may be overruled, and never become a precedent for other cases, can better be borne than could the evils of a different practice. At the same time the candid citizen must confess that if the policy of the Government upon vital questions, affecting the whole people, is to be irrevocably fixed by decisions of the Supreme Court, the instant they are made in ordinary litigation between parties in personal actions the people will have ceased to be their own rulers, having to that extent practically resigned their government into the hands of that eminent tribunal.

Nor is there in this view any assault upon the Court or the Judges. It is a duty from which they may not shrink to decide cases properly brought before them, and it is no fault of theirs if others seek to turn their decisions to political purposes. One section of our country believes slavery is right, and ought to be extended, while the other believes it is wrong, and ought not to be extended. This is the only substantial dispute. The fugitive slave clause of the Constitution, and the law for the suppression of the foreign slave trade, are each as well enforced, perhaps, as any law can ever be in a community where the moral sense of the people imperfectly supports the law itself. The great body of the people abide by the dry legal obligation in both cases, and a few break over in each. This, I think, cannot be perfectly cured; and it would be worse in both cases after the separation of the sections than before. The foreign slave trade, now imperfectly sup-pressed, would be ultimately revived without restriction in one section ; while fugitive slaves, now only partially surrendered, would not be surrendered at all, by the other.

Physically speaking, we cannot separate. We cannot remove our respective sections from each other, nor build an impassable wall between them. A husband and wife may be divorced, and go out of the presence and beyond the reach of each other; but the different parts of our country cannot do this. They cannot but remain face to face; and intercourse, either amicable or hostile, must continue between them. Is it possible, then, to make that intercourse more advantageous or more satisfactory after separation than before? Can aliens make treaties easier than friends can make laws? Can treaties be more faithfully enforced between aliens than laws can among friends? Suppose you go to war, you cannot fight always; and when after much loss on both sides, and no gain on either, you cease fighting, the indentical old questions, as to terms of intercourse, are again upon you.

This country, with its institutions, belongs to the people who inhabit it. Whenever they shall grow weary of the existing Government they can exercise their constitutional right of amending it, or their revolutionary right to dismember or overthrow it. I cannot be ignorant of the fact that many worthy and patriotic citizens are desirous of having the National Con-

rightful authority of the people over the whole subject, to be exercised in either of the modes prescribed in the instrument itself; and I should under existing circumstances, favor rather than oppose a fair opportunity being afforded the people to act upon it. I will venture to add that to me the convention mode seems preferable, in that it allows amendments to originate with the people themselves, instead of only permitting them to take or reject propositions originated by others, not especially chosen for the purpose, and which might not be precisely such as they would wish to either accept or refuse. I understand a proposed amendment to the Constitution-which amendment, however, I have not seen-has passed Congress, to the effect that the Federal Government shall never interfere with the domestic institutions of the States, including that of persons held to service. To avoid misconstruction of what I have said, I depart from my purpose not to speak of particular amendments so far as to say that, holding such a provision now to be implied constitutional law, I have no objection to its being made express and irrevocable.

The Chief Magistrate derives all his authority from the people, and they have conferred none upon him to fix terms for the separation of the States. The people themselves can do this also if they choose; but the Executive, as such, has nothing to do with it. His duty is to administer the present Government, as it came to his hands, and to transmit it, unimpaired by him, to his successor.

Why should there not be a patient confidence Is there in the ultimate justice of the people? any better or equal hope in the world? In our present differences is either party without faith of being in the right? If the Almighty Ruler of Nations, with his eternal truth and justice, be on your side of the North, or on yours of the South, that truth and that justice will surely prevail, by the judgment of this great tribunal of the American people.

By the frame of the Government under which we live, this same people have wisely given their public servants but little power for mischief; and have, with equal wisdom, provided for the return of that little to their own hands at very short intervals. While the people retain their virtue and vigilance, no Administration, by any extreme of weakness or folly, can very seriously injure the Government in the short space of four years.

My countrymen, and all, think calmly and well upon this whole subject. Nothing valuable can be lost by taking time. If there be an object to hurry any of you, in hot haste, to a step which you would never take *deliberately*, that object will be frustrated by taking time; but no good object can be frustrated by it. Such of you as are now dissatisfied, still have the eld Constitution unimpaired, and, on the sensitive point, the laws of your own framing under it; while the new Administration will have no immediate power, if it would, to change either. If it were admitted that you who are dissatisfied hold the right side in the dispute, there still is no single good reason for precipitate action. Intelligence, patriotism, christianity, and a firm overtures for the opening of negotiations, assuring this

reliance on Him who has never yet forsaken this favored land, are still competent to adjust, in the best way, all our present difficulty.

In your hands, my dissatisfied fellow-countrymen, and not in mine, is the momentous issue of civil war. The Government will not assail you. You can have no conflict without being your-selves the aggressors. You have no oath registered in Heaven to destroy the Government, while I shall have the most solemn one to "preserve, protect, and defend it."

I am loth to close. We are not enemies, but friends. We must not be enemies. Though passion may have strained, it must not break our bonds of affection. The mystic chords of memory, stretching from every battle-field and patriot grave to every living heart and hearthstone, all over this broad land, will yet swell the chorus of the Union, when again touched, as surely they will be, by the better angels of our nature.

PRESIDENT LINCOLN'S CABINET.

- Secretary of State-WILLIAM H. SEWARD, of New York.
- Secretary of the Treasury-SALMON P. CHASE, of Ohio; succeeded July 5, 1864, by WM. PITT FESSENDEN, of Maine.

Secretary of War-SIMON CAMERON, of Pennsylvania; succeeded Jan. 11, 1862, by EDWIN M. STANTON, of Ohio.

Secretary of the Navy-GIDEON WELLES, of Connecticut.

Secretary of the Interior-CALEB B. SMITH, of Indiana; succeeded January 8, 1863, by JOHN P. USHEE, of Indiana.

Attorney General-EDWARD BATES, of Missouri. Postmaster General-MONTGOMERY BLAIR, of Maryland.

MR. SEWARD AND "THE COMMISSIONERS OF THE SOUTHERN CONFEDERACY."

[Memorandum.] DEPARTMENT OF STATE, WASHINGTON, March 15, 1861. Mr. John Forsyth, of the State of Alabama, and Mr. Mar-tin J. Crawford, of the State of Georgia, on the 11th instant, through the kind offices of a distinguished Senator, submit-ted to the Secretary of State their desire for an unofficial interview. This request was, on the 12th instant, upon

near new initial request was, on the Licht instant, thom exclusively public considerations, respectfully declined. On the 13th instant, while the Secretary was preceenpied. Mr. A. P. Banks, of Virginia, called at this Department and was received by the Assistant Secretary, to whom he deliv-ered a scaled communication, which he had been charged by Messrs. Forsyth and Crawford to present to the Secre-

In that communication Messrs. Forsyth and Crawford inform the Secretary of State that they have been duly ac-credited by the Government of the Confederate States of America as Commissioners to the Government of the United America as Commissioners to the Government of the United States, and they set forth the object of their attendance at Washington. They observe that seven States of the Amer-ican Union, in the exercise of a right inherent in every freo people, have withdrawn, through conventions of their peo-ple, from the United States, reassumed the attribute of sov-ereign power, and formed a Government of their own, and that those Confederate States now constitute an independ-ent nation *de facto* and *de jure*, and possess a Government perfect in all its parts, and fully endowed with all the means of selfsupport. means of self-support.

Messrs. Forsyth and Crawford, in the aforesaid communication, thereupon proceed to inform the Secretary that, with a view to a speedy adjustment of all questions growwith a view to a speedy adjustment of all questions grow-ing out of the political separation thus assumed, upon such terms of amity and good will as the respective interests, geographical contiguity, and the future welfare of the sup-posed two nations might render necessary, they are in-structed to make to the Government of the United States Government that the President, Congress, and people of the Confederate States earnestly desire a peaceful solution of these great questions, and that it is neither their interest nor their wish to make any demand which is not founded in strictest justice, nor to do any act to injure their late confederates.

After making these statements, Messrs. Forsyth and Crawford close their communication, as they say, in obe-dience to the instructions of their Government, by requesting the Secretary of State to appoint as early a day as possible in order that they may present to the Pres-ident of the United States the credentials which they bear, and the objects of the mission with which they aro charged.

charged. The Secretary of State frankly confesses that he under-stands the events which have recently occurred, and the condition of political affairs which actually exists in the part of the Union to which his attention has thus been directed, very differently from the aspect in which they are presented by Messrs. Forsyth and Crawford. He sees in them not a rightful and accomplished revolution and an independent nation, with an established Government, but rather a perversion of a temporary and partisan excitement to the inconsiderate purpose of an unjustifiable and un-constitutional aggression upon the rights and the authority constitutional aggression upon the rights and the authority vested in the Federal Government, and hitherto benignly exercised, as from their very nature they always must be so exercised, for the maintenance of the Union, the pre-servation of liberty, and the security, peace, welfare, happiness, and aggrandizement of the American p.ople. The Secretary of State, therefore, avows to Messrs. Forsyth and Crawford that he looks patiently but confidently for the source of which which here were the form woording as ma

cure of evils which have resulted from proceedings so un-necessary, so unwise, so unusual, and so unnatural, not to irregular negotiations, having in view new and untried relations with agencies unknown to and acting in derogation lations with agencies unknown to and acting in derogation of the Constitution and laws, but to regular and considerate action of the people of those States, in co-operation with their brethren in the other States, through the Congress of the United States, and such extraordinary Conventions, if there shall be need thereof, as the Federal Constitution contemplates and authorizes to be assembled. It is, how-

contemplates and authorizes to be assembled. It is, how-ever, the purpose of the Secretary of State on this occa-sion not to invite or engage in any discussion of these sub-jects, but simply to set forth his reasons for declining to comply with the request of Messrs. Forsyth and Crawford. On the 4th of March instant the then newly elected President of the United States, in view of all the facts bearing on the present question, assumed the Executive administration of the Government, first delivering, in ac-ordance with an early and horored custom, an inaugural address to the people of the United States. The Secretary of State respectfully submits acony of this address to Vessrs. address to the people of the United States. The Secretary of State respectfully submits a copy of this address to Messrs. Forsyth and Crawford. A simple reference to it will be sufficient to satisfy those gentlemen that the Secretary of State, guided by the principles therein announced, is prevented allogether from admitting or assuming that the States referred to by them, have, in law or in fact, withdrawn from the Federal Union, or that they could do so in the manner described by Messrs. Forsyth and Crawford, or in any other manner than with the consent and concert of the people of the United States, to be given through a Na-The people of the United States, to be given through a Xa-tional Convention, to be assembled in conformity with the provisions of the Constitution of the United States. Of course the Secretary of State cannot act upon the assump-tion, or in any way admit that the so-called Confiderate States constitute a foreign Power, with whom diplomatic volctionr courst to be established. relations ought to be established.

Under these circumstances the Secretary of State, whose official duties are confined, subject to the direction of the President, to the conducting of the foreign relations of the country, and do not at all embrace domestic questions or questions arising between the several States and the Federal Government, is unable to comply with the request of Messrs. Forsyth and Crawford, to appoint a day on which they may present the evidences of their authority and the object of their visit to be President of the United States. On the contrary, he is obliged to state to Mossas Forsyth and Crawford that he has no authority, nor is he at liberty to recognize them as diplomatic agents, or hold correspondence or other communication with them.

Finally, the Secretary of State would observe that, al-though he has supposed that he might safely and with pro-priety have adopted these conclusions without making any reference of the subject to the Executive, yet so strong has been his desire to practice entire directness and to act in a spirit of perfect respect and candor towards Messrs. Forsyth and Crawford, and that portion of the people of rersym and crawtord, and that period of the people of the Union in whose name they present themselves before him, that he has cheerfully submitted this paper to the President, who coincides generally in the views it expresses, and sanctions the Secretary's decision de-clining official intercourse with Messrs. Forsyth and Crawford.

APRL 8, 1861. The foregoing memorandum was filed in this Department on the 15th of March last. A delivery of the same, how-ever, to Messrs. Forsyth and Crawford was delayed, as was understood, with their consent. They have now, through their secretary, communicated their desire for a definitivo disposition of the subject. The Secretary of State there-fore directs that a duly verified copy of the paper be now delivered APRIL 8, 1861. delivered.

A true copy of the original, delivered to me by Mr. F. M. Seward, Assistant Secretary of State of the United States, on April 8th, 1861, at 2.15 P. M., in blank envelope. ATTEST: J. T. PICKETT,

Secretary to the Commissioners.

### The Commissioners in reply to Mr. Seward.

WASHINGTON, April 9, 1861. Hon. WM. H. SEWARD, Secretary of State of the United

Hon. WM. H. SEWARD, Secretary of State of the United States, Washington. The "memorandum," dated Department of State, Wash-ington, March 15, 1861, with postscript under date of sith instant, has been received through the hands of Mr. J. T. Pickett, secretary to this commission, who, by the instruc-tions of the undersigned, called for it on yesterday at the Department.

Department. In that memorandum you correctly state the purport of the official note addressed to you by the undersigned on the 12th ultime. Without repeating the contents of that note in full, it is enough to say here that its object was to invite the Government of the United States to a friendly con-sideration of the relations between the United States and the seven States lately of the Federal Union, but now separated from it by the sovereign will of their people, growing out of the pregnant and undeniable fact that these needed here are stated. growing out of the pregnant and undeniable fact that these people have rejected the authority of the United States and established a Government of their own. Those rela-tions had to be friendly or hostile. The people of the old and new Governments, occupying contiguous territories, had to stand to each other in the relation of good neigh-bors, each seeking their happiness and pursuing their national destinies in their own way, without interference with the other, or they had to be rival and hostile nations. The Government of the Confederate States had no hesite-tion in electing its choice in this alternative. Frankly and The tovernment of the Confederate States had no nesta-tion in electing its choice in this alternative. Frankly and unreservedly, seeking the good of the people who had en-trusted them with power, in the spirit of humanity, of the Christian civilization of the age, and of that Americanism which regards the true welfare and happiness of the people, the Gavement of the Confederate States among its first the Government of the Confederate States, among its first acts, commissioned the undersigned to approach the Gov-crument of the United States with the olive branch of peace, and to offer to adjust the great questions pending between them in the only way to be justified by the con-sciences and common sense of good men who had nothing but the welfare of the people of the two Confederacies at heart.

Your Government has not chosen to meet the undersigned in the conciliatory and peaceful spirit in which they are commissioned. Persistently wedded to those fatal theories of construction of the Federal Constitution always rejected by the statesmen of the South, and adhered to by rejected by the statesmen of the South, and autored to by those of the Administration school, until they have pro-duced their natural and often predicted result of the de-struction of the Union, under which we might have con-tinued to live happily and glorionsly together had the spirit of the ancestry who framed the common Constitution spirit of the ancestry who framed the common Constitution animated the hearts of all their sons, you now, with a per-sistence untaught and uncured by the ruin which has been wrought, refuse to recognize the great fact presented to you of a completed and successful revolution; you close your eyes to the existence of the Government founded upon it, and ignore the high duties of moderation and humanity which attach to you in dealing with this great fact. Had you met these issues with the frankness and manliness with which attach updays impured at the present them to which the undersigned were instructed to present them to you and treat their, the undersigned had not now the mel-ancholy duty to return home and tell their Government and their countrymen that their earnest and ceaseless efforts in behalf of peace had been futile, and that the Government of the United States meant to subjugate them by force of arms. Whatever may be the result, impartial history will record the innocence of the Government of the Confederate States, and place the responsibility of the blood and mourning that may ensue upon those who have denied the great fundamental doctrine of American liberty, that "Governments derive their just powers from the con-sent of the governed," and who have set naval and land armaments in motion to subject the people of one portion of this land to the will of another portion. That that can never be done while a freeman survives in the Confederate States to wield a weapon, the undersigned appeal to past you and treat them, the undersigned had not now the mel-States to wield a weapon, the undersigned appeal to past history to prove. These military demonstrations against the people of the second States are certainly far from

being in keeping and consistency with the theory of the Secretary of State, maintained in his memorandum, that these States are still component parts of the late American Union, as the undersigned are not aware of any constitutional power in the President of the United States to levy war, without the consent of Congress, upon a foreign people, much less upon any portion of the people of the United States.

The undersigned, like the Secretary of State, have no purpose to "invite or engage in discussion" of the subject on which their two Governments are so irreconcilably at variance. It is this variance that has broken up the old Union, the disintegration of which has only begun. It is proper, however, to advise you that it were well to dismiss the hopes you seem to entertain that, by any of the modes indicated, the people of the Confederate States will ever be brought to submit to the authority of the Government of the United States. You are dealing with delusions, too, when you seek to separate our people from our Govern-ment and to characterize the deliberate, sovereign act of that people as a "perversion of a temperary and partian excitement." If you cherish these dreams you will be awakened from them and find them as unreal and unsubstantial as others in which you have recently indulged. The undersigned would omit the performance of an obvious duty were they to fail to make known to the Government of the United States that the people of the Confederate States have declared their independence with a full knowledge of all the responsibilities of that act, and with as firm a determination to maintain it by all the means with which nature has endowed them as that which sustained their fathers when they threw off the authority of the British

The undersigned clearly understand that you have de-clined to appoint a day to enable them to lay the objects President of the United States, because so to do would be to recognize the independence and separate nationality of the Confederate States. This is the vein of thought that pervades the memorandum before us. The truth of history requires that it should distinctly appear upon the record that the undersigned did not ask the Government of the United States to recognize the independence of the Confederate States. They only asked audience to adjust, in a spirit of amity and peace, the new relations springing from a manifest and accomplished revolution in the Gov-ernment of the late Federal Union. Your refusal to entertain these overtures for a peaceful solution, the active naval and military preparation of this Government, and a formal forces in the harbor of Charleston that the President in-tends to provision Fort Sunter by forcible means, if necessary, are viewed by the undersigned, and can only be received by the world, as a declaration of war against the Confederate States; for the President of the United States knows that Fort Sumter cannot be provisioned without the effusion of blood. The undersigned, in behalf of their Government and people, accept the gage of battle thus thrown down to them, and, appealing to God and the judgment of mankind for the righteousness of their cause, the people of the Confederate States will defend their liberties to the last against this flagrant and open attempt at their subjugation to sectional power.

This communication cannot be properly closed without adverting to the date of your memorandum. The official note of the undersigned, of the 12th March, was delivered to the Assistant Secretary of State on the 13th of that menth, the gentleman who delivered it informing him that the Secretary of this commission would call at twelve bio Sectoral y of this commission would can be been a o'clock, noon, on the next day for an answer. At the ap-pointed hour Mr. Pickett did call, and was informed by the Assistant Secretary of State that the engagements of the Societary of State had prevented him from giving the note his attention. The Assistant Secretary of State then asked for the address of Messrs. Crawford and Forsyth, the members of the Commission then present in this city, took note of the address on a card, and engaged to send whatever reply might be made to their lodgings. Why this was not done it is proper should be here explained. The memorandum is dated March 15, and was not delivered until April 8. Why was it withheld during the interveuntil April 8. why was it withheid during the interve-ning twenty-three days? In the postcript to your memo-randum you say it "was delayed, as was understood, with their (Messrs. Forsyth and Crawford's) consent." This is true; but it is also true that on the 15th of March Messrs. Forsyth and Crawford were assured, by a person occupying a high official position in the Government, and occupying a night onleaf posterior in the Government, and who, as they believed, was speaking by authority, that Fort Sumter would be evacuated within a very few days, and that no measure changing the existing status prein-dicially to the Confederate States, as respects Fort Pickens, quently repeated, with the addition that any contemplated change as respects Fort Pickens would be notified to us. On the 1st of April we were again informed that there might be an attempt to supply Fort Sumter with pro-visions, but that Governor Pickens should have previous notice of this attempt. There was no suggestion of a re-inforcement. The undersigned did not hesitate to believe that these assurances expressed the intention of the Adthat these assurances expressed the intention of the Ad-ministration at the time, or at all events of prominent members of the Administration. This delay was assented to for the express purpose of attaining the great end of the mission of the undersigned, to wit, a specific solution to existing complications. The inference deducible from the date of your memorandum that the undersigned had, of their own volition and without canse, consented to this long hiatns in the grave duties with which they were charged is therefore not consistent with a just exposition of the facts of the case.

The intervening twenty-three days were employed in active unofficial efforts, the object of which was to smooth the path to a pacific solution, the distinguished personage al-luded to co-operating with the undersigned, and every step of that effort is recorded in writing, and now in possession of the undersigned and of their Government. It was only when all these anxious efforts for peace had been ex-hausted, and it became clear that Mr. Lincoln had determined to appeal to the sword to reduce the people of the Con-federate States to the will of the section or party whose President he is, that the undersigned resumed the official negotiation temporarily suspended, and sent their secre-tary for a reply to their official note of March 12.

It is proper to add that, during these twenty-three days, two gentlemen of official distinction as high as that of tho personage hitherto alluded to, aided the undersigned as in-termediaries in these unofficial negotiations for peace.

The undersigned, Commissioners of the Confederate States of America, having thus made answer to all they deem material in the memorandum filed in the Department on the 15th of March last, have the honor to be,

JOHN FORSYTH, MARTIN J. CRAWFORD, A. B. ROMAN.

A true copy of the original by one delivered to Mr. F. W. at eight o'clock in the evening of April 9, 1861. Attest: J. T. PICKETT, Secretary, &c.

### Mr. Seward in reply to the Commissioners.

DEPARTMENT OF STATE,

WASHINGTON, April 10, 1861. Messrs. Forsyth, Crawford, and Roman, having been ap-prized by a memorandum which has been delivered to them that the Secretary of State is not at liberty to hold official intercourse with them, will, it is presumed, expect no notice from him of the new communication which they have addresssed to him under date of the 9th instant, beyond the

autesset to be a state of the set of the commission-ers of the Confederate States this 10th day of April, 1861. Attest: J. T. PICKETT, Secretary, &c.

### JUDGE CAMPBELL'S STATEMENT RESPECTING HIS PART IN THE NEGOTIATION.

WASHINGTON CITY, April 13, 1861. SIR: On the 15th of March ultimo I left with Judge Crawford, one of the Commissioners of the Confederate States, a note in writing to the effect following: "I feel entire confidence that Fort Sumter will be evacu-

ated in the next five days. And this measure is felt as imposing great responsibility on the Administration. "I feel entire confidence that no measure changing the

existing status, prejudicially to the Southern Confederate States, is at present contemplated.

"I feel an entire confidence that an immediate demand for an answer to the communication of the Commissioners will be productive of evil and not of good. I do not believe that it ought at this time to be pressed."

The substance of this statement I communicated to you the same evening by letter. Five days elapsed and I called with a telegram from General Beauregard to the effect that Sumter was not evacuated, but that Major Anderson was at work making repairs.

The next day, after conversing with you, I communicated to Judge Crawford, in writing, that the failure to evacuate Sumter was not the result of bad faith, but was attributable Sumier was not the result of bad faith, but was attributable to causes consistent with the intention to fulfill the engage-ment, and that, as regarded Pickens, I should have notice of any design to alter the existing status there. Mr. Justice Nelson was present at these conversations, three in num-ber, and I submitted to him each of my written communi-cations to Judge Crawford, and informed Judge C. that they had his (Judge Nelson's) sanction. I gave you, on the 22d of March, a substantial copy of the statement I had made on the 15th.

The 30th of March arrived, and at that time a telegram came from Governor Pickens inquiring concerning Colonel Lamon, whose visit to Charleston he supposed had a con-nection with the proposed evacuation of Fort Sumter. I left that with you, and was to have an answer the follow-ing Monday, (1st of April.) On the 1st of April I received from you the statement in writing: "I am satisfied the Government will not undertake to supply Fort Sumter with-ont giving notice to Governor P.<sup>2</sup> The words "I am satis-fied" were for me to use as expressive of confidence in the fied" were for me to use as expressive of confidence in the remainder of the declaration.

remainder of the declaration. The proposition as originally prepared was, "The Presi-dent may desire to supply Sunter, but will not do so," &c., and your verbal explanation was that you did not believe any such attempt would be made, and that there was no design to reinforce Sumter. There was a departure here from the pledges of the pre-vious month, but, with the verbal explanation, I did not consider it a matter then to complain of. I simply stated to your that L had that assurance nervious

consider it a matter then to complain of. I simply stated to you that I had that assurance previously. On the 7th of April I addressed you a letter on the sub-ject of the alarm that the preparations by the Government had created, and asked you if the assurances I had given were well or ill-founded. In respect to Sumter your reply was, "Faith as to Sumter, fully kept—wait and see." In the morning's paper I read, "An anthorized messenger from President Lincoln informed Governor Pickens and General Beconcered that provisions will be sent to Fart Sumter— President Lincoin informed Governor Pickens and General Bearregard that provisions will be sent to Fort Sumter-peaceably, or otherwise by force." This was the 8th of April, at Charleston, the day following your last assurance, and is the evidence of the full faith I was invited to wait for and see. In the same paper, I read that intercepted dispatches disclosed the fact that Mr. Fox, who had been allowed to visit Major Anderson, on the pledge that his pur-per was energing any large that any the during a place allowed to visit Major Anderson, on the pledge that his pur-pose was pacific, employed his opportunity to devise a plan for supplying the fort by force, and that this plan had been adopted by the Washington Government, and was in pro-cess of execution. My recollection of the date of Mr. Fox's visit carries it to a day in March. I learn he is a near con-nexion of a member of the Cabinet. My connection with the Commissioners and yourself was superinduced by a con-versation with Justice Nelson. He informed me of your strong disposition in favor of peace, and that you were op-pressed with a demand of the Commissioners of the Confed-

pressed with a demand of the Commissioners of the Confed-erate States for a reply to their first letter, and that you desired to avoid it if possible at that time." I told him I might perhaps be of some service in arrang-ing the difficulty. I came to your office entirely at his re-quest and without knowledge of either of the Commis-sioners. Your depression was obvious to both Judge Nelson and myself. I was gratified at the character of the com-sels you were desirous of pursuing, and much impressed with your observation that a civil war might be prevented by the success of my mediation. You read a letter of Mr. Weed to show how irksome and responsible the withdrawal of troops from Sumter was. A portion of my communica-tion to Judge Crawford on the 15th March was founded upon these remarks, and the pledge to evacuate Sumter is upon these remarks, and the pledge to evacuate Sumter is less forcible than the words you employed. These words were: Before this letter reaches you (a proposed letter by me to President Davis) Sumter will have been evacuated.

The Commissioners who received those communications conclude they have been abused and overreached. The Montgomery Government hold the same opinion. The Commissioners have supposed that my communications were with you, and non-the hypothesis were prepared to were with you, and hpon-the hypothesis were prepared to arraign you before the country in connection with the Pres-ident. I placed a peremptory prohibition upon this as being contrary to the term of my communications with them. I pledged myself to them to communicate informa-tion upon what I considered as the best anthority, and they were to confide in the ability of myself, aided by Judge Nelson, to determine upon the credibility of my in-formant formant.

I think no candid man who will read over what I have written, and considers for a moment what is going on at Sumter, but will agree that the equivocating conduct of the Administration, as measured and interpreted in com-nection with these promises, is the proximate cause of the

great calamity. I have a profound conviction that the telegrams of the 8th of April of General Beauregard, and of the 10th of April of General Walker, the Scoretary of War, can be referred to nothing elso than their belief that there has It is under an oppressive sense of the weight of this re-sponsibility that I submit to you these things for your explanation.

Very respectfully, JOHN A. CAMPBELL, Associate Justice of the Supreme Court U.S. Hon. WILLIAM H. SEWARD, Secretary of State.

### Dispatches.

CHARLESTON, April 8, 1861. To L. P. WALKEE, Secretary of War. An authorized message from President Lincoln\* just informed Governor Pickens and myself that provisions will be sent to Fort Sumter peaceably, or otherwise by force. G. T. BEAUREGARD.

# MONTGOMERY, April 10, 1861.

GEN. G. T. BEAUREGARD. GEN. G. T. BEAUREGARD. If you have no doubt as to the authorized character of the agent who communicated to you the intention of the Washington Government to supply Fort Sumtor by force, you will at once demand its evacuation, and if this is re-fused proceed in such manner as you may determine to reduce it. L. P. WALKER.

# Judge Campbell to the Secretary of State.

Judge Campbell to the Secretary of State. WASHINGTON, April 20, 1861. SIR: I inclose yon a letter, corresponding very nearly with one I addressed to you one week ago, (13th April,) to which I have not had any reply. The letter is simply one of inquiry in reference to facts concerning which, I think, I am entitled to an explanation. I have not adopted any opinion in reference to them which may not be modi-fied by explanation; nor have I affirmed in that letter, nor do I in this, any conclusion of my own unfavorable to your integrity in the whole transaction. All that I have said and mean to say is, that an explanation is due from yon to myself. I will not say what I shall do in case this request is not complied with, but I am justified in saying request is not complied with, but I am justified in saying that I shall feel at liberty to place these letters before any

Derson who is entitled to ask an explanation of myself.
 Very respectfully, JOHN A. CAMPBELL, Associate Justice of the Supreme Court U.S.
 Hon. WM. H. SEWARD, Scretary of State.

April 24, 1861 .- No reply has been made to this letter.

### Judge Campbell to General Davis.

MONTGOMERY, (ALA.,) May 7, 1861. SIR: I submit to you two letters that were addressed by me to the Hon. W. H. Seward, Secretary of State of the United States, that contain an explanation of the nature and result of an intervention by me in the intercourse of the Commissioners of the Confederate States with that officer. I considered that I could perform no duty in which the antire American people with the Solerel Unice officer. I considered that I could perform no duty in which the entire American people, whether of the Federal Union or of the Confederate States, were more interested than that of promoting the counsels and the policy that had for their object the preservation of peace. This motive dic-tated my intervention. Besides the interview referred to in these letters, I informed the Assistant Secretary of Stato of the United States, (not being able to see the Secretury,) on the 11th of April ultimo, of the existence of a telegram of that data from Gen. Beauremark to the Commissioners. of that date from Gen. Beauregard to the Commissioners, in which he informed the Commissioners that he had demanded the evacuation of Sumter, and if refused ho would proceed to reduce it. On the same day I had been told that President Lincoln had said that none of the vessels sent to Charleston wore war vessels, and that force was not to be used in the attempt to supply the fort. I had no means of testing the accuracy of this information, but offered that, if the information was accurate, I would send a telegram to the authorities at Charleston, and it might prevent the disastrous consequences of a collision at that fort between the opposing forces. It was the last effort that I would make to avert the calamities of war. The Assistant Secretary promised to give the matter attention, but I had no other intercourse with him or any other person on the subject, nor have I had any reply to the letters submitted to you.

Very respectfully, JOHN A. CAMP Gen. DAVIS, President of the Confederate States. JOHN A. CAMPBELL.

In an article of Mr. THURLOW WEED, in the Albany Evening Journal of May 30, 1861, we find the following statements respecting Judge Campbell's publication :

"If the Secretary of State were at liberty to reply to ex-Judge Campbell, revealing all that passed between them on several occasions, not only no imputation of insincerity would rest upon the Secretary, but the facts would seriously affect Judge Campbell's well established reputation for can-dor and frankness. The revelations would furnish no evi-dence of either the falsehood or the duplicity of Governor Seward, for there was nothing of either in his conversations. "We violate no confidence in saying that Judge Camp-bell balanced long between loyalty and secession, the pre-ponderance, up to a late day, being in favor of the Union.

\* See President Lincoln's First Message to Congress, July 4, 1861.

"If he at any time looked with favor or satisfaction upon secession, he was much and generally misunderstood. If he did not seriously contemplate remaining in the Union Tť and upon the Bench, he was misunderstood. "If during that period of mental trial, he was acting in

harmony with the leading enemies of the Union, he was grossly misunderstood. "That Gov. Seward conversed freely with Judge Camp-

bell, we do not deny, nor do we doubt, that in these con-versations, at one period, he intimated that Fort Sumter

would be evacuated. "Ile certainly believed so, founding his opinion upon a knowledge of Gen. Scott's recommendation.

" Subsequently, the President deemed it his duty to authorize an effort to reinforce and provision that fortress. We do not know whether Gov. Seward met Judge Campbell after that change of purpose; but he was not at liberty, if they did meet, to reveal what was so well kept. "But, whatever Gov. Seward said or intimated to Judge

Campbell was true at the time it was said.

"That Judge Campbell reported to the Confederate President half that he said or intimated, is more than doubtful."

### PRESIDENT LINCOLN'S ANSWER TO THE DELEGATES FROM VIRGINIA.

April 13, 1861. The PRESIDENT had an interview with Wm. Ballard Preston, Alexander H. H. Stuart, and George W. Randolph, who were appointed by the Convention of Virginia then in session, under a resolution recited in the President's reply, which was as follows :

Hon. Messrs. Preston, Stuart, and Randolph:

GENTLEMEN: As a committee of the Virginia Convention. now in session, you present me a preamble and resolution in these words:

"Whereas, in the opinion of this Convention, the uncerwhich as in the opinion of this convention, the inter-tainty which prevails in the public mind as to the policy which the Federal Executive intends to pursue toward the second states is extremely injurious to the industrial and commercial interests of the country, tends to keep up an excitement which is unfavorable to the adjustment of peuding difficulties, and threatens a disturbance of the public peace: Therefore,

"Resolved, That a committee of three delegates be ap-pointed to wait on the President of the United States, present to him this preamble and resolution, and respectfully ask him to communicate to this Convention the policy which the Federal Executive intends to pursue in regard to the Confederate States."

In answer I have to say, that, having at the beginning of my official term expressed my intended policy as plainly as I was able, it is with deep regret and some mortification I now learn that there is great and injurious uncertainty in the public mind as to what that policy is, and what course 1 intend to pursue.

Not having as yet seen occasion to change, it is now my purpose to pursue the course marked out in the inaugural address. I commend a careful consideration of the whole document as the best expression I can give of my purposes. As I then and therein said, I now repeat:

"The power confided to me will be used to hold, occupy, and possess the property and places belonging to the Gov-ernment, and to collect the duties and imposts; but beyond what is necessary for these objects there will be no invasion, no using of force against or among the people anywhere."

By the words "property and places belonging to the Gov-ernment" I chiefly allude to the military posts and property which were in the possession of the Government when it came to my hands.

But if, as now appears to be true, in pursuit of a purpose to drive the United States authority from these places, an unprovoked assault has been made upon Fort Sumter, I shall hold myself at liberty to repossess, if I can, like places which had been seized before the Government was devolved upon me. And, in any event, I shall, to the best of my ability, repel force by force.

In case it proves true that Fort Sumter has been as-saulted, as is reported. I shall perhaps cause the United States mails to be withdrawn from all the States which claim to have seceeded, believing that the commencement of actual war against the Government justifies and possibly demands it.

I scarcely need to say that I consider the military posts and property situated within the States which claim to have

ever, that I may not land a force deemed necessary to

From the fact that I have quoted a part of the inaugural address, it must not be inferred that I repudiate any other part, the whole of which I reaffirm, except so far as what I now say of the mails may be regarded as a modification.

### WHY AND HOW WAR WAS MADE UPON THE UNITED STATES.

In January, the rebel leaders then in Washington prevented an attack upon the forts in Charleston harbor and at Pensacola.

War not breaking out, the conspiracy weakened, and, as expressed by the Mobile Mercury in discussing the position of affairs in those harbors:

"The country is sinking into a fatal apathy, and the spirit and even the patriotism of the people is cozing out under this do-nothing policy. If something is not done pretty soon, decisive, either evacuation or expulsion, the whole country will become so disgusted with the sham of sonthern independence that the first chance the people get at a popu-lar election they will turn the whole movement topsy-turyy so bad that it never on earth can be righted again.

On Wednesday, April 10, 1861, Roger A. Pryor, of Virginia, was serenaded in Charleston, and spoke as follows, as reported in the Mercury :

"Gentlemen, I thank you, especially that you have at last annihilated this accurse Union, [applause,] recking with corruption, and insolent with excess of tyranny. Thank God, it is at last blasted and riven by the lightning wrath of an outraged and indignant people. [Loud ap-plause.] Not only is it gone, but gone forever. [Cries of 'You're right,' and applause.] In the expressive language of Scripture, it is water spilt upon the ground, which cannot be gathered up. [Applause.] Like Lucifer, son of the be gathered up. [Applause.] Like Luchet, son of how morning, it has fallen, never to fise again. [Continued ap-plause.] For my part, gentlemen, if Abraham Lincoln and Hannibal Hamian to-morrow were to abdicate their affect and were to give me a blank sheet of paper to write the condi-tion of reannexation to the defunct Union, I would scornfully more the coverture \* \* \* \* I invoke spurn the overture. I invoke you, and I make it in some sort a personal appeal—personal so far as it tends to our assistance in Virginia—I do invoke you, in your demonstrations of popular opinion, in your exhibitions of official intent, to give no countenance to this idea of reconstruction. [Many voices, emphatically, ' Never, and applause.] In Virginia they all say, if reduced to the dread dilemma of this memorable alternative, they will es-pouse the cause of the South as against the interest of the Northern Confederacy, but they whisper of reconstruction, and they say Virginia must abide in the Union, with the idea of reconstructing the Union which you have annihila-ted. I pray you, gentlemen, rob them of that idea. Proclaim to the world that upon no condition, and under no chrom-stance, will South Carolina ever again enter into political association with the Abolitionists of New England. [Cries of Never, and applause.]

"Do not distrust Virginia. As sure as to-morrow's sun will rise upon us, just so sure will Virginia be a memory of this Southern Confederation. [Applause.] And I will tell you, genilemen, what will put her in the Southern Confedera-tion in less than an hour by Shrewsbury clock—STRIKE A BLOW! [Tremendous applause.] The very moment that blood is shed, old Virginia will make common cause with her sisters of the South. [Applause.] It is impossible she should do otherwise."

Hon. JEREMIAH CLEMENS, formerly United States Senator from Alabama, and a member of the Alabama Seceding Convention who resisted the movement until adopted by the body, at an adjourned Reconstruction meeting held at Huntsville, Ala., March 13, 1864, made this significant statement :

Mr. Clemens, in adjourning the meeting, said he would tell the Alabamians how their State was got out of the Union. "In 1861," said Mr. C., "shortly after the Confed-erate Government was put in operation, I was in the city of Montgomery. One day I stepped into the office of the Secretary of War, General Walker, and found there, en-gaged in a very excited discussion, Mr. Jefferson Davis, Mr. and property situated within the States which claim to have secoded as yet belonging to the Government of the United States as much as they did before the supposed secession. Whatever else I may do for the purpose, I shall not at-termpt to collect the duites and imposts by any armed inva-sion of any part of the country—not meaning by this, howposed. Mr. Gilchrist said to him, 'Sir, unless you sprinkle blood in the face of the people of Alabama they will be back in the old Union in less than ten days!' The next day General Beauregard opened his batteries on Sumter, and Alabama was saved to the Confederacy."

CORRESPONDENCE PRECEDING BOMBARDMENT.

CHARLESTON, APRIL 18, 1861. CHARLESTON, April 8, 1861. To Hon. L. P. WALKER, Sceretary of War, Monigomery: An authorized messenger from President Lincoln\* has just informed Gov. Pickens and myself that provisions will be sent to Fort Sumter peaceably, or otherwise by force. G. T. BEAUREGARD.

MONTGOMERY, April 10, 1861.

To Gen. BEAUREGARD, Charleston : TO Gen. BEAUREGARD, Charleston: If you have no doubt of the authorized character of the agent who communicated to you the intention of the Washington Government to supply Fort Sumter by force, you will at once demand its evacuation, and, if this is re-fused, proceed in such manner as you may determine to reduce it. L. P. WALKER.

CHARLESTON, April 10, 1861

To Hon. L. P. WALKER: The demand will be made to-morrow at 12 o'clock. G. T. BEAUREGARD,

Brigadier General.

MONTGOMERY, April 10, 1861.

To Gen. BEAUREGARD, Charleston: Unless there are special reasons connected with your own condition, it is considered proper that you should make the demand at an earlier hour.

L. P. WALKER.

CHARLESTON, April 10, 1861.

To Hon. L. P. WALKER:

The reasons are special for 12 o'clock. G. T. BEAUREGARD.

CHARLESTON, April 11, 1861.

To Hon. L. P. WALKER: The demand was sent at 2 p. m., and until 6 was allowed or the answer. G. T. BEAUREGARD. for the answer.

MONTGOMERY, April 11, 1861.

Gen. BEAUREGARD, Charleston : en. BEAUREGARD, Charlesson. Telegraph the reply of Major Anderson. L. P. WALKER.

### CHARLESTON, April 11, 1861.

To Hon. L. P. WALKER : Major Anderson replies: "I have the honor to acknowl-Major Anderson replies: "I have the honor to acknowl-edge the receipt of your communication demanding the evacuation of this fort, and to say in reply thereto that it is a demand with which I regret that my sense of honor and of my obligation to my Government prevent my com-pliance." He adds, verbally, "I will await the first shot, and, if you do not batter us to pieces, we will be starved out in a few days." G. T. BEAUREGARD.

### MONTGOMERY, April 11, 1861.

To General BEAUREGARD : Do not desire needlessly to bombard Fort Sumter. If Major Anderson will state the time at which, as indicated by himself, he will exacute, and agree that, in the mean time, he will not use his guns against us unless ours should be employed against Fort Sumter, you are authorized thus to avoid the effusion of blood. If this or its equivalent be refused, reduce the fort, as your judgment decides to be the most practicable. L. P. WALKER.

# HEADQUARTERS PROVISIONAL ARMY C. S. A., CHARLESTON, (S. C.) April 11, 1861, 2 p. m.

Maj. ROBERT ANDERSON, Commanding at Fort Sumter, Charleston Harbor, S. C.: SIR: The Government of the Confederate States has bitherto forborne from any hostile demonstration against Fort Sumter, in the hope that the Government of the United States, with a view to the amicable adjustment of all questions between the two Governments, and to avert

all questions between the two Governments, and to avert the calamities of war, would voluntarily evacuate it. There was reason at one time to believe that such would be the course pursued by the Government of the United States, and, under that impression, my Government has refrained from making any demand for the surrender of the fort. But the Confederate States can no longer delay assum-ing actual possession of a fortification commanding the entrance of one of their harbors, and necessary to its de-fence and security. fence and security. I am ordered by the Government of the Confederate

\*See his first message, July 4, 1861, page 124.

States to demand the evacuation of Fort Sumter. My aids, Col. Chesnut and Capt. Lee, are authorized to make such demand of you. All proper facilities will be afforded for the removal of yourself and command-together with company arms and property, and all private property--to any post in the United States which you may elect. The flag which you have upheld so long, and with so much fortitude, under the most trying circumstances, may be saluted by you on taking it down. Col. Chesnut and Capt. Lee will, for a reasonable time, await your answer.

await your answer.

I am, sir, very respectfully, your obedient servant, G. T. BEAUREGARD, Brig. Gen. Commanding.

HEADQUARTERS FORT SUMTER, S. C.,

April 11. 1861.

To Brig. Gen. G. T. BEAUREGARD, Commanding Provisional Army C. S. A .:

GENERAL: I have the honor to acknowledge the receipt GENERAL: I have the bonor to acknowledge the receipt of your communication demanding the evacuation of this fort, and to say in reply thereto that it is a demand with which I regret that my sense of honor and my obligation to my Government prevent my compliance. Thanking you for the fair, manly, and courteous terms proposed, and for the high compliment paid me, I remain, General, very respectfully, your obedient servant. REMERT ANDERSON

ROBERT ANDERSON Major U. S. Army, Commanding.

HEADQUARTERS PROV'L ARMY C. S. A.,

CHARLESTON, (S. C.) April 11, 1861, 11 P. M. Major ROBERT ANDERSON,

Commanding at Fort Sumter, Charleston Harbor, S. C .: Commanding at fort isomer, on the verbal observations made MAJOR: In consequence of the verbal observations made by you to my aids, Messrs. Chesnut and Lee, in relation to the condition of your supplies, and that you would in a few days be starved out if our guns did not batter you to pieces, or words to that effect, and desiring no useless effu-sion of blood, I communicated both the verbal observation with a supervert on we communication to my and your written answer to my communication to my

Government. If you will state the time at which you will evacuate Fort Sumter, and agree that in the mean time you will not use your guns against us unless ours shall be employed against Fort Sumter, we shall abstain from opening fire-upon you. Col. Chesnut and Capt. Lee are authorized by me to enter into such an agreement with you. You are, therefore, requested to communicate to them an open

I remain, Major, very respectfully, your obedient servant, G. T. BEAUREGARD, Brigadier General Commanding.

HEAEQUARTERS FORT SUMTER, S. C.

2.30 A. M., April 12, 1861.

answer.

To Brig. Gen. G. T. BEAUREGARD, Commanding Provisional Army C. S. A.:

GENERAL: I have the honor to acknowledge the receipt of GENERAL: I have the honor to acknowledge the receipt of your second communication of the 11th instant, by Col. Chesnut, and to state in reply, that, cordially uniting with you in the desire to avoid the useless effusion of blood, I will, if provided with the necessary means of transporta-tion, evacuate Fort Sunter by noon on the 15th instant, should I not receive prior to that time, controlling in-structions from my Government, or additional supplies, and that I will not in the mean time open my fire upon your forces, unless compelled to do so by some hostile act against this fort or the flag of my Government, by the forces under your command, or by some portion of them, or by the perpetration of some act showing a hostile inter-

by the perjectation of some act showing a hostile inten-tion on your part against this fort or the flag it bears. I have the honor to be, General, very respectfully, your obedient sorvant, ROBERT ANDERSON,

Major U.S. A. Commanding.

FORT SUMTER, S. C.,

April 12, 1861, 3.20 A. M.

MAJOR ROBERT ANDERSON, United States Army, Commanding Fort Sumter:

SIR : By authority of Brigadier General Beauregard, Commanding the Provisional Forces of the Confederate States, we have the henor to notify you that he will open the fire of his batteries on Fort Sumter in one hour from this time.

me. We have the honor to be, very respectfully, your obedient rvants, JAMES CHESNUT, Jr., *Aid-de-Camp.* STEPHEN D. LEE, Captain S. C. <u>Army</u> and Aid-de-Camp. servants,

CHARLESTON, April 12, 1861.

To Hon. L. P. WALKER : He would not consent. I write to-day. G. T. BEAUREGARD.

To Hon. L. P. WALKER We opened fire at 4.30.

G. T. BEAUREGARD. Norr.-Intercepted despatches disclose the fact that Mr. Fox, who had been allowed to visit Major Anderson on the pledge that his purpose was pacific, employed his oppor-tunity to devise a plan for supplying the fort by force, and that this plan had been adopted by the Washington Government, and was in process of execution.

# REPORT OF MAJOR ANDERSON TO THE SECRETARY OF WAR.

STEAMSHIP BALTIC, OFF SANDY HOOE, April 18, 1861, 10.30 A. M., via New York. Having defended Fort Sumter for thirty-four hours, until the quarters were entirely burnt, the main gates destroyed by fire, the gorge walls seriously injured, the magazine surrounded by flames, and its door closed from the effects of heat; four barrels and three cartridges of powder only being available, and no provisions remaining but pork, I accepted terms of evacuation offered by General Beauregard -being the same offered by him on the 11th instant, prior to the commencement of hostilities-and marched out of the Fort on Sunday afternoon, the 14th instant, with colors flying and drums beating, bringing away company and private property, and soluting my flag with fifty guns. ROBERT ANDERSON,

Major 1st Artillery, commanding. Hon. SIMON CAMERON, Secretary of War, Washington.

After the surrender, and while the tidings were received throughout the South with joy, Davis and his associates were serenaded, salvos of artillery were fired, and the whole population seemed to be in an exstacy of triumph. The rebel Secretary of War, L. Pope Walker, defiantly said :

"No man can foretell the events of the war inaugurated; "No man can forefell the events of the war hangurated; but I will venture to predict that the dag which now floats on the breeze" (that was his miserable secession flag) "will, before the first of May, float over the dome of the old Capitol at Washington, and if they choose to try South-ern chivalry, and test the extent of Southern resources, will eventually float over Faneuil Hall, in Boston.

The idea spread throughout the South, and such paragraphs as these abounded:

[From the Richmond Enquirer, April 13, 1861.]

"ATENTION, VOLUNTERS!-Nothing is more probable than that President Davis will soon march an army through North Carolina and Virginia to Washington. Those of our volunteers who decide to join the Southern Army as it shall pass through our borders, had better organize at once for that purpose, and keep their arms, accoutrements, uniforms, ammunition, and knapsacks in constant readiness."

[From the New Orleans Picayune, April 18.]

"The first fruits of a Virginia secession will be the renoval of Lincoln and his cabinet, and whatever he can carry away, to the safer neighborhood of Harrisburg or Cincinnati-perhaps to Buffalo or Cleveland."

[From the Richmond Examiner, April 28.]

"There never was half the unanimity among the people before, not a tithe of the zeal upon any subject, that is now manifested to take Washington. From the mountain tops and valleys to the shores of the sea, there is one wild shout of fierce resolve to capture Washington city at all and every human hazard."

### THE "WAR POWER CALLED OUT."

April 15, 1861. The PRESIDENT issued his proclamation for seventy-five thousand troops, as follows :

"Whereas the laws of the United States have been for some time past, and now are opposed, and the execution thereof obstructed, in the States of South Carolina, Georgia, Alabama, Florida, Mississippi, Louisiana, and Texas, by com-binations too powerful to be suppressed by the ordinary course of judicial proceedings, or by the powers vested in the marshals by law; now, therefore, I, ABRAHAN LINCOLN, President of the United States, in virtue of the power in me vested by the Constitution and the laws, have thomoth fit. vested by the Constitution and the laws, have thought fit to call forth, and hereby do call forth, the militia of the pees of subduing her sister Southern States."

several States of the Union to the aggregate number of 75,000, in order to suppress said combinations, and to cause "The details for this object will be immediately com-

municated to the State authorities through the War Department. I appeal to all loyal citizens to favor, facilitate, and ment. I appeal to all loyal citizens to favor, facilitate, and aid this effort to maintain the honor, the integrity, and ex-istence of our national Union, and the perpetuity of popular government, and to redress wrongs already long enough endured. I deem it proper to say that the first service as-signed to the forces hereby called forth, will probably be to repossess the forts, places, and property which have been seized from the Union; and in every event the utmost care will be observed, consistently with the objects aforesaid, to avoid any devastation, any destruction of, or interference with property, or any disturbance of peaceful citizens of any part of the country; and I hereby command the per-

with property, or any disturbance of peaceful citizens of any part of the country; and 1 hereby command the per-sons composing the combinations aforesaid, to disperse and retire peaceably to their respective abodes, within twenty days from this date. "Deeming that the present condition of public affairs pre-sents an extraordinary occasion, I do hereby, in virtue of the power in me vested by the Constitution, convene both Houses of Congress. The Scnators and Representatives are, therefore, summond to assemble at their respective chamtherefore, summoned to assemble at their respective cham-bers at twelve o'clock, noon, on Thursday, the 4th day of July next, then and there to consider and determine such measures as, in their wisdom, the public safety and interest

may seem to domand. "In witness whereof I have hereunto set my hand, and caused the seal of the United States to be affixed.

bore at the scal of the United States to be united in the year of our Lord one thousand eight hundred and sixty-one, and of the independence of the United States "By the President: "ABRAHA "WILLIAM H. SEWARD, Secretary of State."

"ABRAHAM LINCOLN.

The Governors of all the northern States responded with alacrity.

Governor Burton, of Delaware, issued a proclamation, April 26, recommending the formation of volunteer companies for the protection of the lives and property of the people of Delaware against violence of any sort to which they may be exposed, the companies not being subject to be ordered by the Executive into the United States service, the law not vesting him with such authority, but having the option of offering their services to the General Government for the defence of its capital and the support of the Constitution and laws of the countrv

Governor HICKS, of Maryland, May 14, issued a proclamation for the troops, stating that the four regiments would be detailed to serve within the limits of Maryland or for the defence of the capital of the United States.

Governor LETCHER, of Virginia, replied that "The militia of Virginia will not be furnished to the The militia of virginia will not be furmined to the powers at Washington for any such use or purpose as they have in view. Your object is to subjugate the southern States, and a requisition made apon me for such an object —an object, in my judgment, not within the purview of the Constitution or the act of 1795—will not be complied with, Yon have chosen to inaugurate civil war, and having done so we will meet it in a spirit as determined as the Administration has exhibited toward the South."

Governor ELLIS, of North Carolina, replied, April 15:

April 15: "Your dispatch is received, and, if genuine—which its extraordinary character leads me to doubt—I have to say in reply that I regard the levy of troops made by the Ad-ministration, for the purpose of subjugating the States of the South, as in violation of the Constitution and a usurpa-tion of power. I can be no party to this wicked violation of the laws of the country, and to this war upon the liber-ties of a free people. You can get no troops from North Carolina. I will reply more in detail when your call is re-ceived by mail." ceived by mail."

Governor MAGOFFIN, of Kentucky, replied,

April 15: "Your dispatch is received. In answer I say emphatic-ally, Kentucky will furnish no troops for the wicked pur-

# CHABLESTON, April 12, 1861.

Governor HARRIS, of Tennessee, replied, April 18:

"Tennessee will not furnish a single man for coercion, but fifty thousand, if necessary, for the defence of our rights or those of our southern brethren."

Governor JACKSON, of Missouri, replied :

"Your requisition is illegal, unconstitutional, revolution-ary, inhuman, diabolical, and cannot be complied with."

Governor RECTOR, of Arkansas, replied, April 22:

"None will be furnished. The demand is only adding insult to injury."

May 3, 1861-The President called for thirtynine volunteer regiments of infantry and one regiment of cavalry, with a minimum aggregate of 34,506 officers and enlisted men, and a maximum of 42,034; and for the enlistment of 18,000 seamen.

May 3, 1861-The President directed an increase of the regular army by eight regiments of infantry, one of cavalry, and one of artillery-minimum aggregate, 18,054; maximum, 22,714.

August 6-Congress legalized this increase, and all the acts, orders, and proclamations respecting the Army and Navy.

July 22 and 25, 1861-Congress authorized the enlistment of 500,000 volunteers.

September 17, 1861—Commanding officer at Hatteras Inlet, N. C., authorized to enlist a regiment of loyal North Carolinians.

November 7, 1861-The Governor of Missouri was authorized to raise a force of State militia for State defence.

December 3, 1861-The Secretary of War directed that no more regiments, batteries, or independent companies be raised by the Governors of States, except upon the special requisition of the War Department.

July 2, 1862-The President called for three hundred thousand volunteers.

Under the act of July 17, 1862,

August 4, 1862-The President ordered a draft of three hundred thousand militia, for nine months unless sooner discharged; and directed that if any State shall not, by the 15th of August, furnish its quota of the additional 300,000 authorized by law, the deficiency of volunteers in that State will also be made up Wednesday, by special draft from the militia. September 3, was subsequently fixed for the draft.

May 8, 1863—Proclamation issued, defining the relations of aliens to the conscription act, holding all aliens who have declared on oath their intention to become citizens and may be in the country within sixty-five days from date, and all who have declared their intention to become citizens and have voted.

June 15, 1863-One hundred thousand men, for six months, called to repel the invasion of Maryland, West Virginia, Ohio, and Pennsylvania.

October 17, 1863-A proclamation was issued for 300,000 volunteers, to serve for three years or the war, not, however, exceeding three years, to fill the places of those whose terms expire "during the coming year," these being in addition to the men raised by the present January 5, 1864, a draft shall be made on that day.

February 1, 1864-Draft for 500,000 men for three years or during the war, ordered for March 10, 1864.

March 14, 1864-Draft for 200,000 additional for the army, navy and marine corps, ordered for April 15, 1864, to supply the force required for the navy and to provide an adequate reserve force for all contingencies.

April 23, 1864-85,000 onc hundred day men accepted, tendered by the Governors of Ohio, Indiana, Illinois, Iowa, and Wisconsin; 30,000, 20,000, 20,000, 10,000, and 5,000 being tendered respectively.

### Our Military Legislation.

1861, July 22-The President was authorized to accept the services of volunteers, not exceeding five hundred thousand, for a period not exceeding three years. July 27, this authority was duplicated.

1861, July 27-Nine regiments of infantry, one of cavalry, and one of artillery, added to the regular army.

August 5-Passed bill approving and legalizing the orders of the President respecting the army and navy, issued from 4th of March to that date.

1862, July 17-Authorized the President, when calling forth the militia of the States, to specify the period of such service, not exceeding nine months; and if by reason of defects in existing laws or in the execution of them, it shall be found necessary to provide for enroling the militia, the President was authorized to make all necessary regulations, the enrolment to include all able bodied male citizens between eighteen and forty-five, and to be apportioned according to representative popula-He was authorized, in addition to the tion. volunteers now authorized, to accept 100,000 infantry, for nine months; also, for twelve months, to fill up old regiments, as many as may be presented for the purpose.

1863, February 7-Authorized the Governor of Kentucky, by the consent and under the direction of the President, to raise twenty thousand volunteers, for twelve months, for service within the limits of the State, for repelling invasion, suppressing insurrection, and guarding and protecting the public propertytwo regiments to be mounted riflemen. With the consent of the President, these troops may be attached to, and become a part of, the body of three years' volunteers.

1863, March 3-The conscription act passed. It included as part of the national forces, all able bodied male citizens of the United States, and persons of foreign birth who shall have declared on oath their intention to become citizens under and in pursuance of the laws thereof, between the ages of twenty-one and forty-five years, except such as are rejected as physically or mentally unfit for the service; also, the Vice President, the judges of the various courts of the United States, the heads of the various executive departments of the Government, and the Governors of the several States; also, the only son liable to military draft. In States in default under this call, service, of a widow dependent upon his labor

for support; also, the only son of aged or infirm parent or parents, dependent upon his labor for support; also, where there are two or more sons of aged or infirm parents, subject to draft, the father, or if he be dead, the mother, may elect which son shall be exempt; also, the only brother of children not twelve years old, having neither father nor mother, dependent upon his labor for support; also, the father of motherless children under twelve years of age, dependent upon his labor for support; also, where there are a father and sons in the same family and household, and two of them are in the military service of the United States as noncommissioned officers, musicians, or privates, the residue of such family; provided that no person who has been convicted of any felony shall be enrolled or permitted to serve in said forces. It divided the forces into two classes: 1st, those between twenty and thirty-five and all unmarried persons above thirty-five and under forty-five; 2d, all others liable to mili-tary duty. It divided the country into districts, in each of which an enrolment board was established. The persons enrolled were made subject to be called into the military service for two years from July 1, 1863, and to continue in service for three years. A drafted person was allowed to furnish an acceptable substitute, or pay \$300, and be discharged from further liability under that draft. Persons failing to report, to be considered deserters. All persons drafted shall be assigned by the President to military duty in such corps, regiments, or branches of the service as the exigencies of the service may require.

1864, Feb. 24-Provided for equalizing the draft by calculating the quota of each district or precinct and counting the number previously furnished by it. Any person enrolled may furnish an acceptable substitute who is not liable to draft, nor, at the time, in the military or naval service of the United States; and such person so furnishing a substitute shall be exempt from draft during the time for which such substitute shall not be liable to draft, not exceeding the time for which such substitute shall have been accepted. If such substitute is liable to draft, the name of the person furnishing him shall again be placed on the roll and shall be liable to draft in future calls, but not until the present enrollment shall be ex-The exemptions are limited to such hausted. as are rejected as physically or mentally unfit for the service; to persons actually in the military or naval service of the Government, and all persons who have served in the military or naval service two years during the present war and been honorably discharged therefrom.

The separate enrolment of classes is repealed and the two classes consolidated.

Members of religious denominations, who shall by oath or affirmation declare that they are conscientiously opposed to the bearing of arms, and who are prohibited from doing so by the rules and articles of faith and practice of said religious denomination, shall when drafted, be considered non-combatants, and be assigned to duty in the hospitals, or the care of freedmen, or shall pay \$300 to the benefit of sick

and wounded soldiers, if they give proof that their deportment has been uniformly consistent with their declaration.

No alien who has voted in county, State, or Territory shall, because of alienage, be exempt from draft.

"All able-bodied male colored persons between the ages of twenty and forty-five years, resident in the United States, shall be enrolled according to the previsions of this act, and of the act to which this is an amendment, and form part of the national forces; and when a slave of a loyal master shall be drafted and mustered into the service of the United States, his master shall have a certificate thereof; and thereupon such slave shall be free, and the bounty of one hundred dollars, now payable by law for each drafted man, shall be paid to the person to whom such drafted person was owing service or labor at the time of his muster into the service of the United States. The Secretary of War shall appoint a commission in each of the slave States rep-resented in Congress, charged to award to each loyal person to whom a colored volunteer may owe service a just compensation, not exceeding three hundred dollars, for each such colored volunteer, payable out of the fund de-rived from commutations, and every such colored volunteer on being mustered into the service shall be free. And in all cases where men of color have been heretofore enlisted, or have volunteered in the military service of the United States, all the provisions of this act so far as the payment of bounty and compensation are provided, shall be equally or bounty and compensation are provided, shall be equally applicable, as to those who may be hereafter recruited. But men of color, drafted or enlisted, or who may volun-teer into the military service, while they shall be credited on the quotas of the several States, or sub-divisions of States, wherein they are respectively drafted, enlisted, or shall volunteer, shall not be assigned as State troops, but shall be nustered into regiments or companies as United States colored troops."

1864, Feb. 29-Bill passed reviving the grade of Lieutenant General in the army, and Major General Ulysses S. Grant was appointed March 2d.

1864, June 15-All persons of color shall receive the same pay and emoluments, except bounty, as other soldiers of the regular or volunteer army from and after Jan. 1, 1864, the President to fix the bounty for those hereafter mustered, not exceeding \$100.

1864, June 20-The monthly pay of privates and non-commissioned officers was fixed as follows, on and after May 1:

"Sergeant majors, twenty-six dollars; quartermaster and commissary sergeants of Cavalry, artillery, and infantry, twenty-two dollars; first sergeants of Cavalry, artillery, and infantry, twenty-four dollars; sergeants of Cavalry, artill-lery, and infantry, twenty dollars; sergeants of cavalry, artil-lery, and infantry, twenty dollars; sergeants of ordnance, sappers and miners, and pontoniers, thirty-four dollars; corporals of ordnance, sappers and miners, and pontoniers, twenty dollars: privates of norineare and ordnance of the twenty dollars; privates of engineers and ordnance of the twenty dollars; privates of engineers and ordnance of the first class, eighteen dollars, and of the second class, sixteen dollars; corporals of cavalry, artillery, and infantry, eighteen dollars; chief buglers of cavalry, twenty-three dollars; buglers, sixteen dollars; farriers and blacksmiths of cavalry, and artificers of artillery, eighteen dollars; pri-vates of cavalry, artillery, and infantry, sixteen dollars; principal musicians of artillery and infantry, twenty-two dollars; leaders of brigade and regimental bands, seventy-five dollars; musicians, sixteen dollars; hospital stewards of the first class, twenty-five dollars; hospital stewards of the second class, twenty-five dollars." July 4...This hill became a law:

### July 4-This bill became a law:

Be it enacted, &c. That the President of the United States Be it enacted, dc. That the President of the United States may, at his discretion, at any time hereafter call for any number of men as volunteers for the respective terms of one, two, and three years for military service; and any such volunteer, or, in case of draft, as hereinafter provided, any substitute, shall be credited to the town, township, ward of a city, precinct, or election district, or of a county not so subdivided towards the quota of which he may have volunteered or engaged as a substitute; and every volun-teer who is accepted and mustered into the service for a torm of one ware, unless scorer discharged shall receive. teer who is accepted and indistered into the service for a term of one year, unless sooner discharged, shall receive, and be paid by the United States, a bounty of one hundred dollars; and if for a term of two years, unless sooner dis-charged, a bounty of two hundred dollars; and if for a term of three years, unless sooner discharged, a bounty of three hundred dollars; one third of which bounty shall be paid to the soldier at the time of his being mustered into the service, one-third at the expiration of one-half of his

The service, one-third at the expiration of one-half of his term of service, and one-third at the expiration of his term of service. And in case of his death while in service, the residue of his bounty unpaid shall be paid to his widow, if he shall have left a widow; if not, to his children; or if there be none, to his mother, if she be a widow. SEC, 2. That in case the quota, or any part thereof, of any town, township, ward of a city, precinct, or election dis-trict, or of any county not so subdivided, shall not be filled within the space of fifty days after such call, then the President shall immediately order a draft for one year to fill such quota, or any part thereof, which may be unfilled; and in case of any such draft no payment of money shall be accepted or received by the Government as commutation to release any enrolled or drafted man from personal obli-gation to perform military service.

gation to perform military service. SEC. 3. That it shall be lawful for the executive of any of the States to send recruiting agents into any of the States declared to be in rebellion, except the States of Arkansas, Tennessee, and Louisiana, to recruit volunteers under any call under the provisions of this act, who shall be credited to the State, and to the respective subdivisions thereof, which may procure the culistment. SEC. 4. That drafted men, substitutes, and volunteers,

SEC. 4. That drafted men, substitutes, and volunteers, when mustered in, shall be organized into, or assigned to, regiments, batteries, or other organizations of their own States, and, as far as practicable, shall, when assigned, be permitted to select their own regiments, batteries, or other organizations from among those of their respective States which at the time of assignment may not be filled to their maximum number.

SEC. 5. That the twentieth section of the act entitled "An act to amend an act entitled 'An act for enrolling and calling out the national forces, and for other purposes,' " approved out the national forces, and for other purposes, February twenty-four, eighteen hundred and sixty-four, shall be construed to mean that the Secretary of War shall discharge minors under the age of eighteen years under the circumstances and on the conditions prescribed in said sec-tion; and hereafter, if any officer of the United States shall enist or muster into the military service any person under the age of sixteen years, with or without the con-sent of his parent or guardian, such person so enlisted or recruited shall be immediately discharged upon repayment of all bounties received; and such recruiting or mustering officer who shall knowingly enlist any person under sixteen years of age shall be dismissed the service, with forfeiture of all pay and allowances, and shall be subject to such further purishment as a court-martial may direct.

Fect. SEC. 6. That section three of an act entitled "An act to amend an act entitled 'An act for enrolling and calling out the national forces, and for other purposes," approved February twenty-four, eighteen hundred and sixty-four, he, and the same is hereby, amended, so as to authorize and direct district provest marshals, under the direction of the Decreat Warshal General to make a draft for one hundred

direct district provest marshals, under the direction of the Provest Marshal General, to make a draft for one hundred per centum in addition to the number required to fill the quota of any district as provided by said section. SEC. 7. That instead of travelling pay, all drafted persons reporting at the place of rendezvous shall be allowed trans-portation from their places of residence; and persons dis-charged at the place of residence. SEC. 8. That all persons in the naval service of the United States who have entered said service during the present re-bellion, who have not been credited to the quota of any

States who have entered said service during the present re-bellion, who have not been credited to the quota of any town, district, ward, or State, by reason of their being in said service and not enrolled prior to Fehruary twenty-four, eighteen hundred and sixty-four, shall be enrolled and credited to the quotas of the town, wurd, district, or State, in which they respectively reside, upon satisfactory proof of their residence made to the Secretary of War. SEC. 9. That, if any person duly drafted shall be absent from home in prosecution of his usual business, the provost marshal of the district shall cause him to be duly notified

marshal of the district shall cause him to be duly notified as soon as he may be, and he shall not be deemed a deserter, nor liable as such, until notice has been given him, and reasonable time allowed for him to return and report to the provost marshal of his district; but such absence shall not affect his liability under this act. SEC. 10. That nothing contained in this act, shall be con-

sec to that nothing contained in the act, shall be con-struct to alter or in any way affect the provisions of the soventeenth section of an act approved February twenty-fourth, eighteen hundred and sixty-four, entitled "An act to amend an act entitled 'An act for enrolling and calling out the national forces, and for other purposes," approved March third sixteen bundred on sixty-three March third, eighteen hundred and sixty-three. SEC. 11. That nothing contained in this act shall be con-

strued to alter or change the provisions of existing laws relative to permitting persons liable to military service to furnish substitutes.

# "Confederate" Military Legislation.

February 28, 1861, (four days before the in-auguration of Mr. Lincoln)—The "Confederate" Congress passed a bill providing

1st. To enable the Government of the Confederate States to maintain its jurisdiction over all questions of peace and war, and to provide for the public defence, the President be, and he is hereby authorized and directed to assume control of all military operations in every State, having reference to a connection with questions between the said States, or any of them, and Powers foreign to them.

2d. The President was authorized to receive from the several States the arms and munitions of war which have been acquired from the United States.

3d. He was authorized to receive into Government service such forces in the service of the States, as may be tendered, in such number as he may require, for any time not less than twelve months, unless sooner discharged.

March 6, 1861-The President was authorized to employ the militia, military and naval forces of the Confederate States to repel invasion, maintain rightful possession of the territory, and secure the public tranquillity and independence against threatened assault, to the extent of 100,000 men, to serve for twelve months.

May 4, 1861-One regiment of Zouaves authorized.

May 6, 1861-Letters of marque and reprisal authorized.

1861, August 8-The Congress authorized the President to accept the services of 400,000 volunteers, to serve for not less than twelve months nor more than three years after they shall be mustered into service, unless sooner discharged.

The Richmond Enquirer of that date announced that it was ascertained from official data, before the passage of the bill, that there were not less than 210,000 men then in the field.

August 21-Volunteers authorized for local defence and special service.

1862, January—Publishers of newspapers, or other printed matter, are prohibited from giving the number, disposition, movement, or des-tination of the land or naval forces, or description of vessel, or battery, fortification, engine of war, or signal, unless first authorized by the President or Congress, or the Secretary of War or Navy, or commanding officer of post, district, or expedition. The penalty is a fine of \$1,000 and imprisonment not over twelve months.

1862, February-The Committee on Naval Affairs were instructed to inquire into the expediency of placing at the disposal of the President five millions of dollars to build gunboats.

1862-Bill passed to "regulate the destruction of property under military necessity," referring particularly to cotton and tobacco. The authorities are authorized to destroy it to keep it from the enemy; and owners, destroying it for the same purpose, are to be indemnified upon proof of the value and the circumstances of the destruction.

1862, April 16 - The first "conscription"

April 23, gives this abstract of it :

To the law of Congress, as published, there were sub-sequently passed several amendments or auxiliary laws. We present the following synopsis of the law as thus modified:

The conscription law places in the service of the Confed-erate States, for three years, unless the war sooner ends, all while men between eighteen and thirty-five years of age, resident in the Confederate States, and not legally exempt from service.

The law is silent as to exemptions; but an act defining the class of exempts has since been passed, which embraces generally those hitberto exempt, with some additions. All twelve months' men between the prescribed ages, are continued in service for two years from the expiration of their present term, should the war continue so long; and all those under eighteen and over thirty-five, are to be retained or index of their present set. for ninety days after their term expires, unless their places are sooner supplied by recruits. The twelve months' men between eighteen and thirty-

five, who are retained beyond their term of enlistment, and who have not yet received bounty and furlough, shall receive both; the furloughs, however, to be granted in such numbers and at such times as the Secretary of War may deem most compatible with the public interest; and the men may receive in lieu of furlough, the commutation value in money of the transportation granted to furloughed men by the act.

The term of service of those who originally enlisted for the war, or who have since re-enlisted for that period, is not affected by the law.

Men now in service are not permitted to re-enlist in other organizations than those to which they now belong; and all rc-enlistments that have been made from one existing company to another, or into a new company, where the re-enlistment has not been perfected by actual transfer, are in effect canceled.

Companies, battalions, and regiments of twelve months' men, retained in service by the act, shall be entitled, within forty days from the date of the act, on a day to be fixed by the commander of the brigade, to reorganize by electing all their officers whom they had a right heretofore to elect.

Companies, battalions, squadrons, or regiments organized, or in process of organization, by authority from the Secre-tary of War, which may, within thirty days from the pass-age of the act, have the whole number of men necessary to complete their organization actually enrolled, not including, however, in that number persons now in service, shall be mustered into the service of the Confederate States, and be received in that arm of the service in which they were authorized to organize, and elect their company, battalion, and regimental officers.

To enroll the persons contemplated by the act, and not now in service, the President may, with the consent of the Governors of the States, employ State officers; if such con-sent cannot be obtained, Confederate officers shall be appointed by the President.

Persons not now in service, who shall be enrolled, shall be assigned by the Secretary of War to the different companies of the State from which such persons are drawn, until each company is filled to its maximum number.

Seamen and ordinary scamen, enrolled under the act, may, on application of the Secretary of the Navy, be transferred to the naval service.

If, after filling up the companies, regiments, battalions, and squadrons from any State, there shall remain any of the enrolled mcn, the excess shall be kept as a reserve; and at stated intervals, not exceeding three months, details, to be made by lot, shall be drawn from the reserve to keep the companies as nearly full as practicable. The persons so reserved remain at home until called into service, and re-

reserved remain at nome untreaction to the strength of the server on pay until actually mustered in. They are not, while at home in reserve, subject to the rules and articles of war, except that if they wilfully refuse to obey a call of the President, they shall be held as deserters, and punished as such.

Whenever the President shall think that the exigencies of the service require it, he may call into active service the entire reserve, or as much as may be necessary, and they shall be organized under such rules as the Secretary of War may adopt, and shall elect their field and company officers. The reserves from each State, when thus called out, shall

be organized separately.

Every man mustered into service, who shall bring with him a musket, shot gun, rifle or carbine accepted as an efficient weapon, shall receive the value of it as ascertained by the mustering officer under such regulations as the Sec-retary of War may prescribe, or if the owner be unwilling to sell, he shall receive \$1 a month for the use of such arm.

Persons not liable to duty may he received as substi-

bill became a law. The Richmond Enquirer, of | tutes, under such regulations as the Secretary of War may prescribe. Vacancies may be filled by the President from the com-

pany, battalion, squadron or regiment in which such vacancies occur, by promotion according to seniority, ex-cept in cases of disability or other incompetency.

The President may, however, fill a vacancy by promoting any officer of the company, battalion, squadron, or regi-ment, who may have been distinguished in the service by valor and skill, without reference to seniority. Vacancies in the lowest grade of commissioned officers of a company shall be filled by election, or the President may promote to such reservices pro commissioned officers may promote to such vacancies non-commissioned officers who have distinguished themselves by skill and valor in the service. Per-sons not now in service may, before being enrolled, volunteer with existing companies now in service.

# DAVIS'S CONSCRIPTION PROCLAMATION.

### [From the Richmond Enquirer, July 18, 1863.] PROCLAMATION BY THE PRESIDENT.

Whereas, It is provided by an act of Congress, entitled "An Act to further provide for the public defence," ap-proved on the 16th day of April, 1862, and by another act proved on the 16th day of April, 1862, and by another act of Congress, approved on the 27th September, 1862, en-titled "An Act to amend an act antiled 'An Act to pro-vide further for the public defence,'" approved 16th April, 1862, that the President be authorized to call out and place in the military service of the Confederate States, for threo years, unless the war shall have been sooner ended, all white men who are residents of the Confederate States be made, and who are not at such time legally exempted from military service, or such part thereof as in his judgment may be necessary to the public defence; and

may be necessary to the public defence; and Whereas, in my judgment the necessities of the public defence require that every man capable of bearing arms, between the ages aforesaid, should now be called out to do his duty in the defence of his country, and in driving back the invadors now within the limits of the Confederacy:

the invadors now within the limits of the Confederacy: Now, therefore, I, Jefferson Davis, President of the Con-federate States of America, do, by virtue of the powers vested in me as aforesaid, call out and place in the military service of the Confederate States all white men residents of said States, between the ages of eighteen and forty-fivo years, not legally exempted from military service; and I do borehy order and direct that all parson sculicet to this do hereby order and direct that all persons subject to this call, and not now in the military service, do, upon being enrolled, forthwith repair to the conscript camps established in the respective States of which they may be residents, under pain of being held and punished as deserters in the event of their failure to obey this call, as provided in said laws.

And I do further order and direct that the enrolling officers of the several States proceed at once to enroll all persons embraced within the terms of this proclamation, and not heretofore enrolled.

And I do further order that it shall be lawful for any person embraced within this call to volunteer for service before enrollment, and that persons so volunteering be allowed to select the arm of service and the company which they desire to join, provided such company where to join provided such company where to join, provided such company be deficient in the full number of men allowed by isw for its organization. Given under my band and the seal of the Confederate States of America, at the City of Richmond, on this 15th

day of July, in the year of our Lord 1863.

JEFFERSON DAVIS.

By the President: J. P. BENJAMIN, Sec. of State.

ORDERS UNDER THE CONSCRIPTION ACT.

ADJUTANT AND INSPECTOR GENERAL'S OFFICE, RICHMONT, July 20, 1863.—General Orders, No. 98.—I. All white male residents of the Confederate States, between the ages of eighteen and forty-five, not exempted by law, and not already in the service, will be enrolled. Persons liablo to enrollment may be enrolled wherever they may be found.

II. The first paragraph of General Order No. 86, current series, is so amended as to read as follows : companies, batseries, is so amended as to read as follows: companies, bat-talions and regiments compesed of persons not within the age of conscription, (eighteen and forty-five,) will be ac-cepted as volunteers throughout the Confederacy, under the act of August 21, 1861, (No. 209.) for local defence and special service. Those persons belonging to such organiza-tions, who are of conscript age, and neither exempted by law, nor already in the service, will be discharged, and re-ported to the bureau of conscription for enrollment. III. The following regulation will be in addition to those

ported to the bureau of conscription for enrollment. III. The following regulation will be in addition to those heretofore published in regard to substitutes: Hereafter every person furnishing a substitute, in accordance with existing regulations, shall become liable to, and be immedi-ately enrolled for military duty, upon the loss of the ser-

vices of the substitute furnished by him from any cause other than the casualties of war. S. COOPER By order,

Adjutant and Inspector General.

1864, February. The second conscription bill became a law.

The Richmond Sentinel of February 17, 1864, contains a synopsis of what is called the military bill, heretofore forbidden to be printed :

The first section provides that all white men residents of the Confederate States, between the ages of seventeen and

The second sectors between the ages of second end and fifty, shall be in the military service for the war. The second section provides that all between eighteen and forty-five, now in service, shall be continued during the war in the same regiments, battalions, and companies to which In the same regiments, battations, and companies to where they belong at the passage of this act, with the organization, officers, &c., provided that companies from one State organized against their consent, expressed at the time, with regrets, &c., from another State, shall have the privilege of being transferred to the same arm in a regiment from their own State, and men can be transferred to a company from their own State.

Section three gives a bounty eight months hence of \$100 in rebel bonds.

Section four provides that no person shall be relieved from the operations of this act heretofore discharged for disability, nor shall those who furnished substitutes be exempted, where no disability now exists; but exempts religious per-sons who have paid an exemption tax.

Section five provides for the enrolling of all white male residents of the Confederate States between seventeen and residents of the confidential states between sevence and under such regulations as the President may prescribe; time al-lowed east of the Mississippi thirty, and west sixty days; any person failing to enroll without good excuse shall be placed in the field for the war as if he were between eighteen and forth for the mark such as the placed in the field for the war as if he were between eighteen and forty-five. Persons mentioned in this section shall con-stitute a reserve for State defence and detail duty, and not required to perform service out of the State in which they reside.

Section seven provides that any person who shall fail to attend at the place of rendezvous appointed by the Presi-dent, and not excused by him, shall be liable to be placed in the field service for the war.

Section eight declares that hereafter all positions as Section eight decides that detailet an positions as clerks, guards, agents, employees, or laborers on provost, hospital, or ordnance duty, or in the Quartermaster or Com-missary Departments, and all similar duties, shall be filled by such persons between the ages of eighteen and forty-five may be declared by a hoard of examining surgeons to be 38 unfit for active field service, and when these are exhausted, then from those between seventeen and eighteen, and fortyfive and fifty: provided that the President may detail arti-sans, mechanics, or persons of scientific skill to perform indispensable duties in the bureaux herein named.

Indispensable duties in the Dureaux derain handed. The tenth section provides that no person shall be ex-empt except the following: nuinisters, superintendents of deaf, dumb, and blind, or insane asylums; one editor to each newspaper, and such employees as he may swear to be indispensable; the Confederate and State public printers, and the journeymen printers necessary to perform the public printing; one apothecary to each drug store, who was and has been continuously doing business as such since berefore 10 1562: " physicians over 20 years of age of saven was and has been continuously doing business as such since October 10, 1862; physicians over 30 years of age of seven years' practice, not including dentists; presidents and teachers of colleges, academies, and schools, who have not less than 30 papils; superintendents of public hospitals established by law, and such physicians and nurses as may be indispensable for their efficient management.

One agriculturist on such farm where there is no white male adult not liable to duty employing fifteen able-bodied slaves, between 16 and 50 years of age, upon the following conditions:

The party exempted shall give bonds to deliver to the Government in the next twelve months, 100 pounds of bacon, or its equivalent in salt pork, at Government selec-tion, and 100 pounds of beef for each such able-bodied slave employed on said farm, at commissioner's rates.

In certain cases this may be commuted in grain or other provisions.

The person shall further bind himself to sell all surplus The person shall further bind himself to sell all surplus provisons now on hand, or which he may raise, to the Gov-ernment, or the families of soldiers, at commissioner's rates, the person to be allowed a credit of 25 per cent. on any amount he may deliver in three months from the pas-sage of this act; Provided that no enrollment since Feb. 1, 1864, shall deprive the person enrolled from the benefit of this actemption this exemption.

In addition to the above, the Secretary of War is au-thorized to make such details as the public security requires.

The officers and employces of railroad companies engaged in military transportation, not beyond one for each mile used in such transportation, and under certain restrictions. Also, exempts mail contractors and carriers. The eleventh section authorizes the President to detail

The eleventh section authorizes the President to detail either from between 45 and 50 or from the army in the field when necessity requires it, and may, when he thinks proper, revoke such details. Provided, that he shall not exempt or detail any contractor for furnishing supplies, &c., by reason of such contract, except the head of a de-partment shall certify that such exemption is indispensa-ble, the computient except is the xemption of the superble; the exemption to cease if the contractor fails to com-ply with his contract.

The twefth section declares that the Board of Surgeons shall not be appointed from the county or district in which they are required to make examinations.

The vote in the House of Respresentatives was-yeas, 41; nays, 31.

### GUERRILLAS.

1862, April 21-The President was authorized to commission such officers as he may deem proper, with authority to form bands of partisan rangers, in companies, battalions or regiments, either as infantry or cavalry, to receive the same pay, rations, and quarters, and be subject to the same regulations as other soldiers. For any arms and munitions of war captured from the enemy by any body of partisan rangers, and delivered to any quartermaster at designated place, the rangers shall be paid their full value.\*

The following resolution, in relation to partisan service, was adopted by the Virginia Legislature, May 17, 1862 :

Whereas, this General Assembly places a high estimate upon the value of the ranger or partisan service in prosecuting the present war to a successful issue, and regards it as perfectly legitimate; and it being understood that a Federal commander on the northern border of Virginia has intimated his purpose, if such service is not discontinued, to lay waste by fire the portion of our territory at present

under his power. Resolved by the General Assembly, That in its opinion, the policy of employing such rangers and partisans ought to be carried out energetically, both by the authorities of this State and of the Confederate States, without the slightest regard to such threats.

By another act, the President was authorized, in addition to the volunteer force authorized under existing laws, to accept the services of volunteers who may offer them, without regard

\* 1864, February 15-Repealed the above act, but pro-vided for continuing organizations of partisan rangers acting as regular cavalry and so to continue; and authorizing the Secretary of War to provide for uniting all bands of partisan rangers with other organizations and bringing them under the general discipline of the provisional army. The act authorizes the Secretary of War, in his discretion, to exempt from its operation such companies as are serving

within the lines of the enemy. In a late cavalry raid by Col. Lowell, towards Upperville, Va, Lieut. Henry E. Alvord, of the Second Massachusetts Cavalry, captured Major Moseby's private papers, and Cavalry, captured Major Moseby's private papers, and found his commission as major of guerrillas, which is as follows:

CONFEDERATE STATES OF AMERICA, WAR DEPARTMENT, RICHMOND, March 26, 1863.—You are hereby informed that the President has conferred upon you the rank of Major of The resident has conterred upon you the rank of hador of Partisan Rangers, under the act approved April 21, 1862, in the Provisional Army of the Confederate States, to date as such from the 26th of March, 1863. Immediately on receipt thereof please communicate to this Department, to the Adjutant and Inspector General, announcing your accept-ance of said appropriate. With your lattere of cocentry. ance of said appointment. With your letter of acceptance to the Adjutant and Inspector General, you will fill up properly the enclosed oath, and subscribe and swear to it; at the same time state your age, residence, and when ap-pointed, and the State in which you were born. Should you accept, report for duty to General R. E. Lee. JAMES A. SEDDON, Sceretary of War.

Major JOHN MOSEBY.

to the place of enlistment, to serve for and during the existing war.

1862, May 27—Maj. Gen. John B. Floyd was authorized by the Legislature of Virginia, to raise ten thousand men, not now in service or liable to draft, for twelve months.

1862, September 27 — The President was authorized to call out and place in the military service for three years, all white men who are residents, between the ages of thirty-five and forty-five, at the time the call may be made, not legally exempt. And such authority shall exist in the President, during the present war, as to all persons who now are, or hereafter may become eighteen years of age, and all persons between eighteen and forty-five, once enrolled, shall serve their full time.

### THE TWENTY-NEGRO EXEMPTION LAW.

1862, October 11—Exempted certain classes, described in the repealing law of the next session, as follows:

The dissatisfaction of the people with an act passed by the Confederate Congress, at its last session, by which persons owning a certain number of slaves were exempted from the operation of the conscription law, has led the members at the present session to reconsider their work, and already one branch has passed a bill for the repeal of the obnoxious law. This bill provides as follows: "The Congress of the Confederate States do enact, That so

"The Congress of the Confederate States do enact, That so much of the act approved October 11, 1862, as exempts from military service ' one person, either as agent, owner, or overseer, on each plantation on which one white person is required to be kept by the laws or ordinances of any State, and on which there is no white male adult not liable to military service, and in States having no such law, one person, as agenf, owner, or overseer on such plantation of twenty negroes, and on which there is no white male adult not liable to military service; and also the following clause in said act, to wit: 'and furthermore, for additional police of every twenty negroes, on two or more plantations, within five miles of each other, and each having less than twenty negroes, and on which there is no white male adult not liable to military duty, one person, being the oldest of the owners or overseers on such plantations,' be and the same are hereby repealed; and the persons so there to exempted by said clauses of said act are hereby made subject to military duty in the same manner that they would be had said clauses never been embraced in said act."

### RESTRICTIONS UPON VOLUNTEERING.

From the Richmond *Examiner*, January 30, 1864:

General Orders have been issued in the 'Adjutant General's office, instructing the enrolling officers to proceed as rapidly as possible with the new conscription. The privilege of volunteering is restricted in these orders by two important conditions:

"1. The company selected must have been in service on the 16th of April, 1862.

"2. The company selected must be, at the time of volunteering, below the minimum number prescribed by regulations."

### JUDICIAL RULINGS UPON CONSCRIPTION.

The Richmond papers of March. 1864, mention the following decision in reference to the conscription act:

"In the case of J. R. F. Borroughs vs. T. G. Peyton, and L. P. Abrahams vs. the same, the Court of Appeals of Virginia on yesterday rendered a decision. These cases came before the court on *habeas corpus*, the plaintiffs praying to be discharged from the custody of the conscript officer, on the ground that they had furnished substitutes—the one under the State law of February, 1862, the other the Confederate States law.

"The court unanimously rejected the petitions in both cases, and remanded the parties to the military officer. The opinion was delivered by Judge William J. Robertson, and is able and elaborate. It brought under review the constitutionality both of the conscript law and that repealing the exemptions of such conscripts as have furnished substitutes, and it fully sustained the action of Congress in

both instances. The conscript law is a legitimate exercise of the power of Congress to raise armies, which is distinct from and additional to the power to employ the militia of the country. The privilego of putting in substitutes, until recently allowed, was an act of grace and favor to the editzen, and not a contract in any respect to which the Government was a party. "Nor would the Government have had a right to make

"Nor would the Government have had a right to make such contract as in this case is contended for. As an act of grace it was, of course, repealable at the will of Congress. Even if Congress had had the power to makesuch contract, and had exercised it, yet the conditions necessarily attaching would have rendered the contract repealable if, in the judgment of Congress, the exigencies of the country required it. Congress was the sole judge of a public necessity of this nature, and the preamble to the law repealing substitutions recognized this."

### STATE RIGHTS AND PERSONAL LIBERTY.

The Macon (Georgia) *Telegraph* gives the points of the first legal decision made in the State of Georgia under the act of the Confederate Congress repealing the substitute law, and compelling those who had furnished substitutes to go into the army. It is the judgment of Judge Lochrane, of the Superior Court for Macon Circuit, and was delivered on the 11th of February, in the case of Dennis Daley and Philip Fitzgerald, and is interesting as showing what are held to be State-rights and personal liberty in the Confederate States. We quote:

"Judge Lochrane held it was not only the right, but the duty of a nation to protect itself, and that any contract or right flowing out of the operation of law which came in conflict with the preservation of the State, was an unconstitutional act—not obligatory on the law-making power, and within the constitutional power of the Government to repeal.

"That the act allowing substitutes was to be regarded as "That the act allowing substitutes was to be regarded as a contract discharging principals from being called into the service. It was then a contract that the principal should not fight in the defence of the country when it was endangered, and such contract was unauthorized by every principle of constitutional law. If our first Congress had agreed to exempt all men from taxation during the war who paid into the treasury \$500, such exemption could have been set aside by any subsequent legislature, when the public safety and self-preservation of the Government demand it.

"He held that the interest of every citizen was the same as that of the Government of which he formed a part, and the military service rendered by the substitute was just as much rendered to the principal as a citizen of the Government itself—his life, his honor, his property, and his liberty were defended by the act, and the consideration enured to him as a member of the society which composed the Government.

"Contracts and vested rights must all bend to the exigencies of the Government, of which the legislature was the judge, and any act of the Legislature contravening the public interest may be repealed when the safety of the people becomes the supreme law.

"The vested rights of fathers may be annulled over their minor children, to make them soldiers when the public interests demand it, and the law-making power has so declared.

"All rights, all property, all persons who are citizens of a Government, may be used by the Government in time of war, and it was the duty of courts to sustain the Government in the appropriation of the means exercised rightfully by the Legislature to protect the whole people from subjugation and ruin."

1864, March 22 — The Supreme Court of Georgia are reported as having, to-day, unanimously affirmed the constitutionality of the anti-substitute law.

### IN NORTH CAROLINA.

### [From Richmond Sentinel, March 8, 1864.]

HABEAS CORPUS WRITS REFUSED.—The Raleigh Confederate states that Judge Battle has lately refused to issue writs of hadras corpus in a number of cases when applied for by persons who had placed substitutes in the army, on the ground that the writ was suspended in all such cases by the late act of Congress. It also learns that such is the opinion of Judge Manly. Judges Pearson, Battle, and Manly constitute the Supreme Court of North Carolina, and as two of the three judges sustain the Government in the suspension of the matter against the substitute men is only a question of time.

Judge Pearson is said to hold the contrary opinion.

### MARTIAL LAW DECLARED.

March 1, 1862-JEFFERSON DAVIS declared, "by virtue of the powed vested in him by law to declare the suspension of the privileges of the writ of Habeas Corpus in cities threatened with invasion," proclaimed that martial law was extended over Richmond and the adjoining country to the extent of ten miles. He prohibited all distillation of spirituous liquors, and directed that the distilleries be forthwith closed and the establishments for the sale thereof closed. Many Union men were arrested at once and imprisoned.

April 8, 1862, DAVIS issued a proclamation extending martial law over East Tennessee and suspending all civil jurisdiction and the writ of Habeas Corpus. Col. W. M. Churchwell was made provost marshal and was charged with the execution of the proclamation.

May 3, 1862, DAVIS issued a like proclamation with reference to the counties of Lee, Wise, Buchanan, McDowell, and Wyoming in Virginia, under the command of Brig. Gen. Humphrey Marshall.

August 14, 1861, DAVIS issued this

### PROCLAMATION OF BANISHMENT :

Whereas, the Congress of the Confederate States of Amerthe congress of the congress of the conductate bursts of Angust, ica did, by an act approved on the 8th day of August, 1861, entitled "An act respecting alien enemies," make provision that proclamation should be issued by the President in re-lation to alien enemies, and in conformity with the provisions of said act:

Now, therefore, I, Jefferson Davis, President of the Con-federate States of America, do issue this my proclamation: and I do hereby warn and require every male citizen of the and I do hereby warn and require every male citizen of the United States, of the age of fourteen years and upwards, now within the Confederate States, and adhering to the Govern-ment of the United States, and acknowledging the authority of the same, and not being a citizen of the Confederate States, to depart from the Confederate States within forty days from the date of this proclamation. And I do warn all persons above described, who shall remain within the Confederate States after the expiration of said period of forty days, that they right the transfer to soling arguing

States after the 'expiration of said period of forty days, that they will be treated as alien enemies. *Provided, however*, That this proclamation shall not be considered as applicable, during the existing war, to citizens of the United States residing within the Confederate States with intent to become citizens thereof, and who shall make declaration of such intention in due form, acknowledging the authority of this Government; nor shall this proclamation be considered as extending to the States of Delaware, Mary-land, Kentucky, Missouri, the District of Columbia, the Territories of Arizona and New Mexico, and the Indian Ter-ritory south of Kansas, who shall not be chargeable with ritory south of Kansas, who shall not be chargeable with actual hostility or other crime against the public safety, and who shall acknowledge the authority of the Government of the Confederate States.

And I do further proclaim and make known that I have established the rules and regulations hereto annexed in ac-cordance with the provisions of said law.

Cordance with the provisions of said law. Given under my hand and the seal of the Confederate States of America, at the city of Riehmond, on the 14th day of August, A. D. 1861. By the President: JEFFERSON DAVIS. R. M. T. HUNTER, Secretary of State.

TO THE DISAFFECTED PEOPLE OF EAST TENNESSEE.

The undersigned, in executing martial law in this Department, assures those interested who have fled to the enemy's lines, lines, and who are actually in their army, that he will wel-come their return to their homes and families; they are offered annesty and protection if they come to lay down their arms and act as loyal citizens within the thirty days given them by Maj. Gen. E. KERF SMITH to do so. At the end of that time, those failing to return to their homes and accept the annesty thus offered, and provide for

and protect their wives and children in East Tennessee, will have them sent to their care in Kentucky, or beyond the Confederate State lines, at their own expense. All that leave after this date, with a knowledge of the above facts, will have their families sent immediately after

them.

The women and children must be taken care of by husbands

and fathers, either in East Tennessee or in the Lincoln Government. W. M. CHURCHWELL,

Colonel and Provost Marshal. KNOXVILLE, Tenn., April 23, 1862.

# STRENGTH OF THE REBEL ARMY.

1864, Jan. 30 .- In debating the bill to repeal the Substitute Exemption bill, Mr. Wm. N. H. Smith, of North Carolina, said, the "Confed-erates" had at this time four hundred thousand men on their muster roll "of whom probably one half were not there, and it was well known we were unable to feed the fractional part who were, in the field."

The Richmond Whig of Jan. 1, 1864, alluding to the passage of the above named bill, has these comments :

We wish at this time only to make some passing com-ments upon the tone manifested in Congress in the debates upon this measure. A Senator (Mr. Orr of S. C.,) among other objections to the passage of the bill gravely questioned

"if the pending bill becomes a law," said Mr. O., "there will be great difficulty in executing it, by reason of the de-cisions of the Courts in several of the States. We must acquiesce in the decisions of the Courts, or resort to measures which he was not prepared for." A Senator from Missouri (Mr. Clark.) representing a constituency wholly beyond the action and control of our laws, replied, in urging its passage, that "in regard to the action of the Courts, steps may and should be taken to remove the subject beyond their juris-diction." The Senator from Mississippi, (Mr. Brown,) goes further: "We should not defer our legislation to eonsult the views of every State Judge—to ascertain whether he will overthrow it or not by his judicial decision. We have high overthrow it or not by his judicial decision. We have high duties to perform. Let us perform them without reference to State Judges. There was a remedy against the interference of the Courts, in the suspension of the writ of habeas

The vote of Congress upon this measure shows to what extent the provisions of the Constitution are getting to be disregarded in the Legislative branch, and how far revolu-tionary sentiment already prevails in that body. The strength of the provision of the covernment, and the good of the popular respect for our Government, and the good sense of the quiet masses, may for the moment allow such declarations to pass without disturbance. The little respect in which substitute men are held may influence a temporary acquiesence in them. But the intelligent and ardent lover of his country cannot witness such proceedings with indif-ference, nor will he, with submission. There is as much patriotism and intelligence out of the halls of Congress as in it, and the tendency of the legisla-

tive and executive declarations so far this session to a subversion of the liberties of the country and a military despotism is already sowing the seeds of a counter revolution. Our people claim it as their right, as the duty of the General Government to insure to them, as the basis of the compact by which they have associated together, that the Confederacy is but a community of sovereign States. They look to the Constitution as the supreme law of the Confederacy. They regard it as among the blessings for which they are indebted to their anestry that they transmitted to us a written Con-stitution. It received the plighted faith of our fathers. It is the hope of our posterity. To argue questions outside or above it is but to assail the cause of law, of right, and order.

### Meeting of the Thirty-Seventh Congress.

The first session began July 4, 1861, and closed August 6, 1861. The second session began December 2, 1861, and closed

July 17, 1862. • The third session began December 1, 1862, and closed

March 4, 1863.

MEMBERS OF THE THIRTY-SEVENTH CONGRESS, MARCH 4, 1861, TO MARCH 4, 1863.

HANNIBAL HAMLIN, of Maine, President of the Senate.

### SENATORS.

MAINE-Lot M. Morrill, William P. Fessenden.

NEW HAMPSHIRE - John P. Hale, Daniel Clark.

VERMONT-Solomon Foot, Jacob Collamer. MASSACHUSETTS - Charles Sumner, Henry Wilson.

RHODE ISLAND-James F. Simmons, \* Henry B. Anthony.

CONNECTICUT-James Dixon, Lafayette S. Foster.

NEW YORK-Preston King, Ira Harris.

NEW JERSEY-John B. Thomson,\* John C. Ten Eyck.

PENNSYLVANIA-David Wilmot, Edgar Cow-

DELAWARE -James A. Bayard, Willard Saulsburv.

MARYLAND - Anthony Kennedy, James A. Pearce.\*

VIRGINIA.\*

Оню-Benjamin F. Wade, John Sherman.

KENTUCKY-Lazarus W. Powell, John C. Breckinridge.\*

TENNESSEE-Andrew Johnson.

INDIANA - Jesse D. Bright,\* Henry S. Lane. ILLINOIS. - O H. Browning,\* Lyman Trumbull.

MISSOURI-Trusten Polk,\* Waldo P. Johnson.\*

MICHIGAN-Z. Chandler, K. S. Bingham.\*

Iowa-James W. Grimes, James Harlan.

WISCONSIN-James R. Doolittle, Timothy O. Howe.

CALIFORNIA-Milton S. Latham, James A. McDougall.

MINNESOTA-Henry M. Rice, Morton S. Wilkinson.

UREGON-Edward D. Baker,\* James W. Nesmith.

KANSAS-James H. Lane, S. C. Pomeroy.

REPRESENTATIVES.

GALUSHA A. GROW, of Pennsylvania, Speaker of the House.

MAINE-John N. Goodwin, Charles W. Walton,\* Samuel C. Fessenden, Anson P. Morrill, John H. Rice, Frederick A. Pike.

NEW HAMPSHIRE-Gilman Marston, Edward H. Rollins, Thomas M. Edwards.

VERMONT-E. P. Walton, Jr., Justin S. Morrill, Portus Baxter.

MASSACHUSETTS—Thomas D. Eliot, James Buf-finton, Benjamin F. Thomas, Alexander H. Rice, William Appleton,\* John B. Alley, Daniel W. Gooch, Charles R. Train, Goldsmith F. Bailey,\* Charles Delano, Henry L. Dawes.

RHODE ISLAND-William P. Sheffield, George H. Browne.

CONNECTICUT-Dwight Loomis, James E. English, Alfred A. Burnham,\* George C. Woodruff.

NEW YORK-Edward H. Smith, Moses Odell, Benjamin Wood, James E. Kerrigan, William Wall, Frederick A. Conkling, Elijah Ward, Isaac C. Delaplaine, Edward Haight, Charles H. Van Wyck, John B. Steele, Stephen Baker, Abraham B. Olin, Erastus Corning, James B. McKean, William A. Wheeler, Socrates N. Sherman, Chauncey Vibbard, Richard Franchot, Roscoe Conkling, R. Holland Duell, William E. Lansing, Ambrose W. Clark, Charles B.

\* See memorandum at the end of list.

Sedgwick, Theodore M. Pomeroy, Jacob P. Chamberlain, Alexander S. Diven, Robert B. Van Valkenburgh, Alfred Ely, Augustus Frank, Burt Van Horn, Elbridge G. Spalding, Reuben E. Fenton.

New JERSEY-John T. Nixon, John L. N. Stratton, William G. Steele, George T. Cobb, Nehemiah Perry.

PENNSYLVANIA-William E. Lehman, Charles J. Biddle,\* John P. Verree, William D. Kelley, William Morris Davis, John Hickman, Thomas B. Cooper,\* Sydenham E. Ancona, Thaddeus Stevens, John W. Killinger, James H. Camp-bell, Hendrick B. Wright, Philip Johnson, Ga-lusha A. Grow, James T. Hale, Joseph Baily, Edward McPherson, Samuel S. Blair, John Covode, Jesse Lazear, James K. Moorhead, Rob-ert McKnight, John W. Wallace, John Patton, Elijah Babbitt.

DELAWARE-George P. Fisher. MARYLAND-John W. Crisfield, Edwin H. Webster, Cornelius L. L. Leary, Henry May, Francis Thomas, Charles B. Calvert.

VIRGINIA-Charles H. Upton,\* William G. Brown, John S. Carlile,\* Kellian V. Whaley.

Оню—George H. Pendleton, John A. Gurley, Clement L. Vallandigham, William Allen, Jas. M. Ashley, Chilton A. White, Richard A. Har-rison, Samuel Shellabarger, Warren P. Noble, Carey A. Trimble, Valentine B. Horton, Samuel S. Cox, Samuel T. Worcester, Harrison G. Blake, Robert H. Nugen, William P. Cutler, James R. Morris, Sidney Edgerton, Albert G. Riddle, John Hutchins, John A. Bingham.

KENTUCKY - Henry C. Burnett,\* James S. Jackson,\* Henry Grider, Aaron Harding, Chas. A. Wickliffe, George W. Dunlap, Robert Mallory, John J. Crittenden, William H. Wadsworth, John W. Menzies.

TENNESSEE-Horace Maynard,\* Andrew J. Clements,\* George W. Bridges.\*

INDIANA-John Law, James A. Cravens, W. McKee Dunn, William S. Holman, George W. Julian, Albert G. Porter, Daniel W. Voorhees, Albert S. White, Schuyler Colfax, William Mitchell, John P. C. Shanks.

ILLINOIS — Ellihu B. Washburne, Isaac N. Arnold, Owen Lovejoy, William Kellogg, Wil-liam A. Richardson,\* John A. McClernand,\* James C. Robinson, Philip B. Fouke, John A. Logan.\*

MISSOURI-Francis P. Blair, Jr., James S. Rollins, John B. Clark,\* Elijah H. Norton, John W. Reid,\* John S. Phelps,\* John W. Noell.

MICHIGAN-Bradley F. Granger, Fernando C. Beaman, Francis W. Kellogg, Rowland E. Trowbridge.

Iowa—Samuel R. Curtis,\* William Vandever. WISCONSIN-John F. Potter, Luther Hanchett,\* A. Scott Sloan.

MINNESOTA-Cyrus Aldrich, William Windom.

OREGON-Andrew J. Thayer.\*

KANSAS-Martin F. Conway.

MEMORANDUM OF CHANGES.

The following changes took place during the Congress:

\* See Memorandum at end of list.

### IN SENATE.

RHODE ISLAND-1862, Dec. I, Samuel G. Arnold succeeded James F. Simmons, resigned.

MARYLAND-1863, Jan. 14, Thomas H. Hicks, first by appointment and then by election succeeded James A. Pearce, deceased Dec. 20, 1862.

KENTUCKY-1861, Dec. 23, Garrett Davis succeeded John C. Breckinridge, expelled December 4.

INDIANA--1862, March 3, Joseph A. Wright succeeded Jesse D. Bright, expelled Feb. 5. 1863, Jan. 22, David Turpie superseded, by election, Joseph A. Wright.

ILLINOIS—1863, Jan. 30, William A. Richardson superseded, by election, O. H. Browning.

MISSOURI-1861, Jan. 24, R. Wilson succeeded Waldo P. Johnson, expelled Jan. 10. 1862, Jan. 29, John B. Henderson succeeded Trusten Polk, expelled Jan. 10.

MICHIGAN-1862, Jan. 17, Jacob M. Howard succeeded K. S. Bingham, deceased October 5, 1861.

OREGON-1862, Dec. 1, Benjamin F. Harding succeeded Edward D. Baker, deceased Oct. 21, 1862.

IN HOUSE OF REPRESENTATIVES.

MAINE-1862, December 1, Thomas A. D. Fessenden succeeded Charles W. Walton, resigned May 26, 1862.

MASSACHUSETTS-1861, December 1, Amasa Walker succeeded Goldsmith F. Bailey, deceased May 8, 1862; 1861, December 2, Samuel Hooper succeeded William Appleton, resigned.

CONNECTICUT—1861, December 2, Alfred A. Burnham qualified.

PENNSYLVANIA—1861, December 2, Charles J. Biddle qualified; 1862, June 3, John D. Stiles succeeded Thomas B. Cooper, deceased April 4, 1862.

VIRGINIA—1861, July 13, John S. Carlile resigned to take a seat in the Senate; 1861, December 2, Jacob B. Blair succeeded John S. Carlile, resigned; 1862, February 28, Charles H. Upton unseated by a vote of the House; 1862, May 6, Joseph Segar qualified.

KENTUCKY-1862, December 1, George H. Yeaman succeeded James S. Jackson, deceased; 1862, March 10, Samuel L. Casey succeeded Henry C. Burnett, expelled December 3, 1861. TENNESSEE-1861, December 2, Horace May-

TENNESSEE—1861, December 2, Horace Maynard qualified; 1862, January 13, Andrew J. Clements qualified; 1863, February 25, George W. Bridges qualified.

ILLINOIS-1861, December 12, A. L. Knapp qualified, in place of J. A. McClernand, resigned; 1862, June 2, William J. Allen qualified, in place of John A. Logan, resigned; 1863, January 30, William A. Richardson withdrew to take a seat in the Senate. MISSOURI-1862, January 21, Thomas L. Price succeeded John W. Reid, expelled December 2, 1861; 1862, January 20, William A. Hall succeeded John B. Clark, expelled July 13, 1861; 1862, May 9, John S. Phelps qualified.

Iowa—1861, December 2, James F. Wilson suceeded Samuel R. Curtis, resigned August 4, 1861.

WISCONSIN-1863, January 26, Walter D. McIndoe succeeded Luther Hanchett, deceased November 24, 1862.

OREGON-1861, July 30, George K. Shiel succeeded Andrew J. Thayer, unseated.

LOUISIANA - 1863, February 17, Michael Hahn qualified; 1863, February 23, Benjamin F. Flanders qualified.

# President Lincoln's First Message, July 4, 1861.

### Fellow-Citizens of the Senate and House of Representatives :

Having been convened on an extraordinary occasion, as authorized by the **C**onstitution, your attention is not called to any ordinary subject of legislation.

At the beginning of the present Presidential term, four months ago, the functions of the Federal Government were found to be generally suspended within the several States of South Carolina, Georgia, Alabama, Mississippi, Louisiana, and Florida, excepting only those of the Post Office Department.

Within these States, all the forts, arsenals, dock-yards, custom-houses, and the like, including the movable and stationary property in and about them, had been seized, and were held in open hostility to this Government, excepting only Forts Pickens, Taylor, and Jefferson, on and near the Florida coast, and Fort Sumter, in Charleston harbor, South Carolina. The forts thus seized had been put in improved condition; new ones had been built, and armed forces had been organized, and were organizing, all avowedly with the same hostile purpose.

The forts remaining in the possession of the Federal Government in and near these States were either besieged or menaced by warlike preparations, and especially Fort Sumter was nearly surrounded by well-protected hostile batteries, with guns equal in quality to the best of its own, and outnumbering the latter as perhaps ten to one. A disproportionate share of the Federal muskets and rifles had somehow found their way into these States, and had been seized to be used against the Government. Accumulations of the public revenue, lying within them, had been seized for the same object. The navy was scattered in distant seas, leaving but a very small part of it within the immediate reach of the Government. Officers of the Fe-deral army and navy had resigned in great numbers; and of those resigning a large proportion had taken up arms against the Government. Simultaneously, and in connection with all this, the purpose to sever the Federal Union was openly avowed. In accordance with this purpose an ordinance had been adopted in each of these States, declaring the States, respectively, to be separated from the National Union. A formula for instituting a combined government of these States had been promulgated; and this illegal organization, in the character of Confederate States, was already invoking recognition, aid, and intervention from foreign Powers.

Finding this condition of things, and believing it to be an imperative duty upon the incoming Executive to prevent, if possible, the consummation of such attempt to destroy the Federal Union, a choice of means to that end became indispensable. This choice was made, and was declared in the inaugural address. The policy chosen looked to the exhaustion of all peaceful measures before a resort to any stronger ones. It sought only to hold the public places and property not already wrested from the Government, and to collect the revenue ; relying for the rest on time, discussion, and the ballot-box. It promised a continuance of the mails, at Government expense, to the very people who were resisting the Government; and it gave repeated pledges against any disturbance to any of the people, or any of their rights. Of all that which a President might constitutionally and justifiably do in such a case, everything was forborne without which it was believed possible to keep the Government on foot.

On the 5th of March (the present incumbent's first full day in office) a letter of Major Anderson, commanding at Fort Sumter, written on the 28th of February, and received at the War Department on the 4th of March, was, by that Department, placed in his hands. This letter expressed the professional opinion of the writer that reinforcements could not be thrown into that fort within the time for his relief, rendered necessary by the limited supply of provisious, and with a view of holding possession of the same, with a force of less than twenty thousand good and well-disciplined men. This opinion was concurred in by all the officers of his command, and their memoranda on the subject were made enclosures of Major Anderson's letter. The whole was immediately laid before Lieut. General Scott, who at once concurred with Major Anderson in opinion. On reflection, however, he took full time, consulting with other officers, both of the army and the navy, and, at the end of four days, came reluctantly, but decidedly, to the same conclusion as before. He also stated at the same time that no such sufficient force was then at the control of the Government, or could be raised and brought to the ground within the time when the provisions in the fort would be exhausted. In a purely military point of view this reduced the duty of the Administration in the case to mere matter of getting the garrison safely out of the fort.

It was believed, however, that to so abandon that position, under the circumstances, would be utterly ruinous; that the *necessily* under which it was to be done would not be fully understood; that by many it would be construed as a part of a *voluntary* policy; that at home it would discourage the friends of the Union, embolden its adversaries, and go far to insure to the latter a recognition abroad; that in fact jt would be our national destruction consum-

mated. This could not be allowed. Starvation was not yet upon the garrison ; and ere it would be reached Fort Pickens might be re-inforced. This last would be a clear indication of policy, and would better enable the country to accept the evacuation of Fort Sumter as a military necessity. An order was at once directed to be sent for the landing of the troops from the steamship Brooklyn into Fort Pickens. This order could not go by land, but must take the longer and slower route by sea. The first return news from the order was received just one week before the fall of Fort Sumter. The news itself was that the officer commanding the Sabine, to which vessel the troops had been transferred from the Brooklyn, acting upon some quasi armistice of the late Administration, (and of the existence of which the present Administration, up to the time the order was dispatched, had only too vague and uncertain rumors to fix attention,) had refused to land the troops. To now reinforce Fort Pickens, before a crisis would be reached at Fort Sumter, was impossible-rendered so by the near exhaustion of provisions in the latter named fort. In precaution against such a conjuncture the Government had, a few days before, commenced preparing an expedition, as well adapted as might be, to relieve Fort Sumter, which expedition was intended to be ultimately used, or not, according to circumstances. The strongest anticipated case for using it was now presented; and it was resolved to send it forward. As had been intended in this contingency, it was resolved to notify the Governor of South Carolina that he might expect an attempt would be made to provision the fort ;\* and that if the attempt should not be resisted there would be no effort to throw in men, arms, or ammunition, with out further notice, or in case of an attack upon the fort. This notice was accordingly given; whereupon the fort was attacked and bombarded to its fall, without even awaiting the arrival of the provisioning expedition.

It is thus seen that the assault upon and reduction of Fort Sumter was, in no sense a matter of self-defence on the part of the assailants. They well knew that the garrison in the fort could, by no possibility, commit aggression upon them. They knew-they were expressly notified—that the giving of bread to the few brave and hungry men of the garrison was all which would on that occasion be attempted, unless themselves, by resisting so much, should provoke more. They knew that this Government desired to keep the garrison in the fort, not to assail them, but merely to maintain in visible possession, and thus to preserve the Union from actual and immediate dissolutiontrusting, as hereinbefore stated, to time, discussion, and the ballot-box for final adjustment; and they assailed and reduced the fort for precisely the reverse object-to drive out the visible authority of the Federal Union, and thus force it to immediate dissolution. That this was their object the Executive well understood; and having said to them, in the Inaugural Address, "you can have no conflict without being

\* See page 113.

124

yourselves the aggressors," he took pains not | only to keep this declaration good, but also to keep the case so free from the power of ingenious sophistry as that the world should not be able to misunderstand it. By the affair at Fort Sumter, with its surrounding circumstances, that point was reached. Then and thereby the assailants of the Government began the conflict of arms, without a gun in sight or in expectancy to return their fire, save only the few in the fort sent to that harbor years before for their own protection, and still ready to give that protection in whatever was lawful. In this act, discarding all else, they have forced upon the country, the distinct issue: "Immediate dissolution or blood."

And this issue embraces more than the fate of these United States. It presents to the whole family of man the question whether a Constitutional Republic or Democracy-a government of the people by the same people-can or cannot maintain its territorial integrity against its own domestic foes. It presents the question whether discontented individuals. too few in numbers to control administration according to organic law in any case, can always, upon the pretences made in this case, or on any other pretences, or arbitrarily without any pretence, break up their Government, and thus practically put an end to free government upon It forces us to ask: "Is there, in the earth. all republics, this inherent and fatal weakness ?" "Must a Government of necessity be too strong for the liberties of its own people, or too weak to maintain its own existence?"

So viewing the issue, no choice was left but to call out the war power of the Government; and so to resist force, employed for its destruction, by force, for its preservation.

The call was made, and the response of the country was most gratifying, surpassing in unamimity and spirit the most sanguine expectations. Yet none of the States commonly called slave States, except Delaware, gave a regiment through regular State organization. A few regiments have been organized within some others of those States by individual enterprise, and received into the Government service. Of course, the seceded States, so called, (and to which Texas had been joined about the time of the inauguration,) gave no troops to the cause of the Union. The border States, so called, were not uniform in their action; some of them being almost for the Union, while in others-as Virginia, North Carolina, Tennessee, and Arkansas-the Union sentiment was nearly repressed and silenced. The course taken in Virginia was the most remarkable, perhaps the most important. A Convention, elected by the people of that State to consider this very question of disrupting the Federal Union, was in session at the capital of Virginia when Fort Sumter fell. To this body the people had chosen a large major-ity of *professed* Union men. Almost immediately after the fall of Sumter, many members of that majority went over to the original disunion minority, and, with them, adopted an ordinance for withdrawing the State from the Union. Whether this change was wrought by their navy. These measures, whether strictly legal great approval of the assault upon Sumter, or or not, were ventured upon, under what ap-

their great resentment at the Government's resistance to that assault, is not definitely known. Although they submitted the ordinance, for ratification, to a vote of the people, to be taken on a day then somewhat more than a month distant, the Convention, and the Legislature, (which was also in session at the same time and place,) with leading men of the State, not members of either, immediately commenced acting as if the State were already out of the Union. They pushed military preparations vigorously forward all over the State. They seized the United States armory at Harper's Ferry, and the navy-vard at Gosport, near Norfolk. They the navy-yard at Gosport, near Norfolk. They received—perhaps invited—into their State large bodies of troops, with their warlike appointments, from the so-called seceded States. They formally entered into a treaty of temporary alliance and co-operation with the so-called "Confederate States," and sent members to their Congress at Montgomery. And, finally, they permitted the insurrectionary Government

to be transferred to their capital at Richmond The people of Virginia have thus allowed this giant insurrection to make its nest within her borders; and this Government has no choice left but to deal with it where it finds it. And it has the less regret, as the loyal citizens have, in due form, claimed its protection. Those loyal citizens this Government is bound to recognize, and protect, as being Virginia.

In the Border States, so-called-in fact, the Middle States-there are those who favor a policy which they call "armed neutrality;" that is, an arming of those States to prevent the Union forces passing one way, or the disunion the other, over their soil. This would be disunion completed. Figuratively speaking, it would be the building of an impassable wall along the line of separation ; and yet not quite an impassable one; for, under the guise of neutrality, it would tie the hands of the Union men, and freely pass supplies from among them to the insurrectionists, which it could not do as an open enemy. At a stroke it would take all the trouble off the hands of secession, except only what proceeds from the external blockade. It would do for the disunionists that which, of all things, they most desire-feed them well, and give them disunion without a struggle of their own. It recognizes no fidelity to the Constitution, no obligation to maintain the Union; and while very many who have favored it are doubtless loyal citizens, it is, nevertheless, very injurious in effect.

Recurring to the action of the Government, it may be stated that, at first, a call was made for seventy-five thousand militia; and rapidly following this, a proclamation was issued for closing the ports of the insurrectionary districts by proceedings in the nature of blockade. So far all was believed to be strictly legal. At this point the insurrectionists announced their purpose to enter upon the practice of privateering.

Other calls were made for volunteers to serve three years, unless sooner discharged, and also for large additions to the regular army and peared to be a popular demand and a public necessity; trusting then, as now, that Congress would readily ratify them. It is believed that nothing has been done beyond the constitutional competency of Congress. Soon after the first call for militia it was

considered a duty to authorize the Commanding General, in proper cases, according to his discretion, to suspend the privilege of the writ of habeas corpus, or, in other words, to arrest and detain, without resort to the ordinary processes and forms of law, such individuals as he might deem dangerous to the public safety. This authority has purposely been exercised but very sparingly. Nevertheless the legality and propriety of what has been done under it are questioned, and the attention of the country has been called to the proposition that one who is sworn to "take care that the laws be faithfully executed" should not himself violate them. Of course some consideration was given to the questions of power and propriety before this matter was acted upon. The whole of the laws which were required to be faithfully The whole of executed were being resisted and failing of execution in nearly one third of the States. Must they be allowed to finally fail of execution, even had it been perfectly clear that, by the use of the means necessary to their execution, some single law, made in such extreme tenderness of the citizen's liberty that practically it relieves more of the guilty than of the innocent, should to a very limited extent be violated? To state the question more directly, are all the laws but one to go unexecuted and the Government itself go to pieces lest that one be violated? Even in such a case would not the official oath be broken, if the Government should be overthrown, when it was believed that disregarding the single law would tend to preserve it? But it was not believed that this question was presented. It was not believed that any law was violated. The provision of the Constitution that "the privilege of the writ of habeas corpus shall not be suspended unless when, in cases of rebellion or invasion, the public safety may require it," is equivalent to a provision, is a provision, that such privilege may be suspended when, in cases of rebellion or invasion, the public safety does require it. It was decided that we have a case of rebellion, and that the public safety does require the qualified suspension of the privilege of the writ which was authorized to be made. Now it is insisted that Congress, and not the Executive, is vested with this power. But the Constitution itself is silent as to which or who is to exercise the power; and as the provision was plainly made for a dangerous emergency, it cannot be believed the framers of the instrument intended that in every case the danger should run its course until Congress could be called together, the very assembling of which might be prevented, as was intended in this case, by the rebellion.

No more extended argument is now offered, as an opinion at some length will probably be presented by the Attorney General. Whether there shall be any legislation upon the subject, and if any, what, is submitted entirely to the better judgment of Congress.

The forbearance of this Government had been so extraordinary, and so long continued as to lead some foreign nations to shape their action as if they supposed the early destruction of our National Union was probable. While this, on discovery, gave the Executive some concern, he is now happy to say that the sovereignty and rights of the United States are now everywhere practically respected by foreign Powers, and a general sympathy with the country is manifested throughout the world.

The reports of the Secretaries of the Treasury, War, and the Navy will give the information in detail deemed necessary and convenient for your deliberation and action; while the Executive, and all the departments, will stand ready to supply omissions, or to communicate new facts considered important for you to know.

It is now recommended that you give the legal means for making this contest a short and a decisive one; that you place at the control of the Government, for the work, at least four hundred thousand men and four hundred millions of dollars. That number of men is about one tenth of those of proper ages within the regions where, apparently, all are willing to engage; and the sum is less than a twenty-third part of the money value owned by the men who seem ready to devote the whole. A debt of six hundred millions of dollars now, is a less sum per head than was the debt of our Revolution when we came out of that struggle; and the money value in the country now bears even a greater proportion to what it was then than does. the population. Surely, each man has as strong a motive now to preserve our liberties as each had then to establish them.

A right result, at this time, will be worth more to the world than ten times the men and ten times the money. The evidence reaching us from the country leaves no doubt that the material for the work is abundant; and that it needs only the hand of legislation to give it legal sanction, and the hand of the Executive to give it practical shape and efficiency. One of the greatest perplexities of the Government is to avoid receiving troops faster than it can provide for them. In a word, the people will save their Government if the Government itself will do its part only indifferently well.

It might seem, at first thought, to be of little difference whether the present movement at the South be called "secession" or "rebellion." The movers, however, well understand the difference. At the beginning they knew they could never raise their treason to any respectable magnitude by any name which implies violation of law. They knew their people possessed as much of moral sense, as much of devotion to law and order, and as much pride in and reverence for the history and Government of their common country as any other civilized and patriotic people. They knew they could make no advancement directly in the teeth of these strong and noble sentiments. Accordingly they commenced by an insidious debauching of the public mind. They invented debauching of the public mind. an ingenious sophism, which, if conceded, was followed by perfectly logical steps, through all

the incidents, to the complete destruction of the Union. The sophism itself is, that any State of the Union may, consistently with the national Constitution, and therefore lawfully and peacefully, withdraw from the Union without the consent of the Union or of any other State. The little disguise that the supposed right is to be exercised only for just cause, themselves to be the sole judge of its justice, is too thin to merit any notice.

With rebellion thus sugar-coated, they have been drugging the public mind of their section for more than thirty years; and until at length they have brought many good men to a willingness to take up arms against the Government the day after some assemblage of men have enacted the farcical pretence of taking their State out of the Union who could have been brought to no such thing the day before.

This sophism derives much, perhaps the whole, of its currency from the assumption that there is some omnipotent and sacred supremacy pertaining to a State-to each State of our Federal Union. Our States have neither more nor less power than that reserved to them, in the Union, by the Constitution-no one of them ever having been a State out of the Union. The original ones passed into the Union even before they cast off their British colonial dependence; and the new ones each came into the Union directly from a condition of dependence, excepting Texas. And even Texas, in its temporary independence, was never designated a State. The new ones only took the designation of States on coming into the Union, while that name was first adopted for the old ones in and by the Declaration of Independence. Therein the "United Colonies" were declared to be "free and independent States;" but, even then, the object plainly was not to declare their independence of one another, or of the Union, but directly the contrary, as their mutual pledge and their mutual action, before, at the time, and afterwards, abundantly show. The express plighting of faith, by each and all of the original thirteen, in the Articles of Confederation, two years later, that the Union shall be perpetual, is most conclusive. Having never been States, either in substance or in name, outside of the Union, whence this magical omnipotence of "State rights," asserting a claim of power to lawfully destroy the Union itself? Much is said about the "sovereignty" of the States; but the word even is not in the National Constitution; nor, as is believed, in any of the State constitutions. What is a "sovereignty," in the political sense of the term? Would it be far wrong to define it " a political community without a political superior ?" Tested by this, no one of our States, except Texas, ever was a sovereignty. And even Texas gave up the character on coming into the Union; by which act she acknowledged the Constitution of the United States, and the laws and treaties of the United States made in pursuance of the Constitution, to be, for her, the supreme law of the land. The States have their status in the Union, and they have no other legal *status*. If they break from this, they can only do so against law and by revolution. The Union, and not themselves admits of secession. They have assumed to

separately, procured their independence and their liberty. By conquest, or purchase, the Union gave each of them whatever of independ-ence and liberty it has. The Union is older than any of the States, and, in fact, it created them as States. Originally some dependent col-onies made the Union, and, in turn, the Union threw off their old dependence for them, and made them States, such as they are. Not one of them ever had a State constitution independent of the Union. Of course it is not forgotten that all the new States framed their constitutions before they entered the Union; nevertheless, dependent upon and preparatory to coming into the Union.

Unquestionably the States have the powers and rights reserved to them in and by the National Constitution; but among them surely are not included all conceivable powers, however mischievous or destructive, but, at most, such only as were known in the world at the time as governmental powers; and certainly a power to destroy the Government itself had never been known as a governmental, as a merely administrative power. This relative matter of national power and State rights, as a principle, is no other than the principle of generality and locality. Whatever concerns the whole should be confided to the whole-to the General Government; while whatever concerns only the State should be left exclusively to the State. This is all there is of original principle about it. Whether the National Constitution. in defining boundaries between the two, has applied the principle with exact accuracy, is not to be questioned. We are all bound by that defining, without question.

What is now combatted is the position that secession is consistent with the Constitution-is lawful and peaceful. It is not contended that there is any express law for it; and nothing should ever be implied as law which leads to unjust or absurd consequences. The nation purchased with money the countries out of which several of these States were formed. Is it just that they shall go off without leave, and without refunding? The nation paid very large sums (in the aggregate, I believe, nearly a hundred millions) to relieve Florida of the aboriginal tribes. Is it just that they shall now be off without consent, or without making any re-The nation is now in debt for money turn? applied to the benefit of these so-called seceding States, in common with the rest. Is it just either that creditors shall go unpaid or the remaining States pay the whole? A part of the present national debt was contracted to pay the old debts of Texas. Is it just that she shall leave and pay no part of this herself?

Again, if one State may secede, so may another; and when all shall have seceded, none is left to pay the debts. Is this quite just to creditors ? Did we notify them of this sage view of ours when we borrowed their money? If we now recognize this doctrine by allowing the seceders to go in peace, it is difficult to see what we can do if others choose to go, or to extort terms

make a national constitution of their own, in which, of necessity, they have either discarded or retained the right of secession as, they insist, it exists in ours. If they have discarded it they thereby admit that, on principle, it ought not to be in ours. If they have retained it by their own construction of ours, they show that to be consistent they must secede from one another whenever they shall find it the easiest way of settling their debts, or effecting any other selfish or unjust object. The principle itself is one of disintegration, and upon which no Government can possibly endure.

If all the States save one should assert the power to drive that one out of the Union, it is presumed the whole class of seceder politicians would at once deny the power and denounce the actas the greatest outrage upon Staterights. But suppose that precisely the same act, instead of being called "driving the one out," should be called " the seceding of all the others from that one," it would be exactly what the seceders claim to do; unless, indeed, they make the point that the one, because it is a minority, may rightfully do what the others, because they are a majority, may not rightfully do. These politicians are subtle and profound on the rights of minorities. They are not partial to that power which made the Constitution, and speaks from the preamble, calling itself "We, the People."

It may well be questioned whether there is to-day a majority of the legally qualified voters of any State, except perhaps South Carolina, in favor of disunion. There is much reason to believe that the Union men are the majority in many, if not in every other one, of the socalled Seceded States. The contrary has not been demonstrated in any one of them. It is ventured to affirm this, even of Virginia and Tennessee; for the results of an election, held in military camps, where the bayonets are all on one side of the question voted upon, can scarcely be considered as demonstrating popular sentiment. At such an election all the large class who are, at once, for the Union, and against coercion, would be coerced to vote against the Union.

It may be affirmed, without extravagance, that the free institutions we enjoy have developed the powers and improved the condition of our whole people beyond any example in the world. Of this we now have a striking and an impressive illustration. So large an army as the Government has now on foot was never before known, without a soldier in it but who had taken his place there of his own free choice. But more than this: there are many single regiments whose members, one and another, possess full practical knowledge of all the arts, sciences, professions, and whatever else, whether useful or elegant, is known in the world; and there is scarcely one from which there could not be selected a President, a Cabinet. a Congress, and perhaps a Court, abundantly competent to administer the Government itself. Nor do I say this is not true also in the army of our late friends, now adversaries, in this contest; but if it is, so much better the reason why the Government, which has conferred such benefits on both them and us, should not be broken up.

Whoever, in any section, proposes to abandon such a Government would do well to consider in deference to what principle it is that he does it, what better he is likely to get in its stead, whether the substitute will give or be intended to give so much of good to the people. There are some foreshadowings on this subject. Our adversaries have adopted some declaration of independence, in which, unlike the good old one penned by Jefferson, they omit the words "all men are created equal." Why? They have adopted a temporary National Constitution, in the preamble of which, unlike our good old one signed by Washington, they omit "We, the people," and substitute, "We, the deputies of the sovereign and independent States." Why? Why this deliberate pressing out of view the rights of men and the authority of the people?

This is essentially a People's contest. On the side of the Union it is a struggle for maintaining in the world that form and substance of government whose leading object is to elevate the condition of men, to lift artificial weights from all shoulders, to clear the paths of laudable pursuit for all, to afford all an unfettered start and a fair chance in the race of life. Yielding to partial and temporary departures from necessity, this is the leading object of the Government for whose existence we contend.

I am most happy to believe that the plain people understand and appreciate this. It is worthy of note that, while in this the Government's hour of trial, large numbers of those in the Army and Navy who have been favored with the offices have resigned, and proved false to the hand which had pampered them, not one common soldier or common sailor is known to have deserted his flag.

Great honor is due to those officers who remained true, despite the example of their treacherous associates; but the greatest honor and most important fact of all is the unanimous firmness of the common soldiers and common sailors. To the last man, so far as known, they have successfully resisted the traitorous efforts of those whose commands but an hour before they obeyed as absolute law. This is the patriotic instinct of plain people. They understand, without an argument, that the destroying the Government which was made by Washington means no good to them.

Our popular Government has often been Two points in it our called an experiment. people have already settled-the successful establishing and the successful administering of it. One still remains-its successful maintenance against a formidable internal attempt to overthrow it. It is now for them to demonstrate to the world that those who can fairly carry an election can also suppress a rebellion; that ballots are the rightful and peaceful successors of bullets; and that when ballots have fairly and constitutionally decided, there can be no successful appeal back to bullets; that there can be no successful appeal except to ballots themselves at succeeding elections. Such will be a great lesson of peace, teaching men that what they cannot take by an election, neither can they take it by a war; teaching all the folly of being the beginners of a war.

Lest there be some uneasiness in the minds of candid men as to what is to be the course of the Government towards the Southern States *after* the rebellion shall have been suppressed, the Executive deems it proper to say, it will be his purpose then, as ever, to be guided by the Constitution and the laws; and that he probably will have no different understanding of the powers and duties of the Federal Government relatively to the rights of the States and the people, under the Constitution, than that expressed in the inaugural address.

He desires to preserve the Government, that it may be administered for all, as it was administered by the men who made it. Loyal citizens everywhere have the right to claim this of their Government, and the Government has no right to withhold or neglect it. It is not perceived that, in giving it, there is any coercion, any conquest, or any subjugation, in any just sense of those terms.

The Constitution provides, and all the States have accepted the provision, that "the United States shall guaranty to every State in this Union a republican form of government." But, if a State may lawfully go out of the Union, having done so, it may also discard the republican form of government; so that to prevent its going out is an indispensable means to the end of maintaining the guarantee mentioned; and when an end is lawful and obligatory, the indispensable means to it are also lawful and obligatory.

It was with the deepest regret that the Executive found the duty of employing the war power, in defence of the Government, forced upon him. He could but perform this duty or surrender the existence of the Government. No compromise by public servants could in this case be a cure; not that compromises are not often proper, but that no popular government can long survive a marked precedent, that those who carry an election can only save the Government from immediate destruction by giving up the main point upon which the people gave the election. The people themselves, and not their servants, can safely reverse their own deliberate decisions.

As a private citizen the Executive could not have consented that these institutions shall perish; much less could he in betrayal of so vast and so sacred a trust as these free people had confided to him. He felt that he had no moral right to shrink, nor even to count the chances of his own life, in what might follow. In full view of his great responsibility he has, so far, done what he has deemed his duty. You will now, according to your own judgment, perform yours. He sincerely hopes that your views and your action may so accord with his as to assure all faithful citizens, who have been disturbed in their rights, of a certain and speedy restoration to them under the Constitution and the laws.

And having thus chosen our course, without guile and with pure purpose, let us renew our trust in God, and go forward without fear and with manly hearts.

July 4, 1861.

## President Lincoln's First Annual Message, Dec. 3, 1861.

## Fellow-Citizens of the Senate

and House of Representatives:

In the midst of unprecedented political troubles, we have cause of great gratitude to God for unusual good health, and most abundant harvests.

You will not be surprised to learn that in the peculiar exigencies of the times, our intercourse with foreign nations has been attended with profound solicitude, chiefly turning upon our own domestic affairs.

A disloyal portion of the American people have, during the whole year, been engaged in an attempt to divide and destroy the Union. A nation which endures factious domestic division, is exposed to disrespect abroad; and one party, if not both, is sure, sooner or later, to invoke foreign intervention.

Nations thus tempted to interfere, are not always able to resist the counsels of seeming expediency and ungenerous ambition, although measures adopted under such influences seldom fail to be unfortunate and injurious to those adopting them.

The disloyal citizens of the United States who have offered the ruin of our country, in return for the aid and comfort which they have invoked abroad, have received less patronage and encouragement than they probably ex-If it were just to suppose, as the pected. insurgents have seemed to assume, that foreign nations, in this case, discarding all moral, social, and treaty obligations, would act solely, and selfishly, for the speedy restoration of commerce, including, especially, the acquisition of cotton, those nations appear, as yet, not to have seen their way to their object more directly or clearly through the destruction, than through the preservation, of the Union. If we could dare to believe that foreign nations are actuated by no higher principle than this, I am quite sure a sound argument could be made to show them that they can reach their aim more readily and easily by aiding to crush this rebellion than by giving encouragement to it.

The principal lever relied on by these insurgents for exciting foreign nations to hostility against us, as already intimated, is the embarrassment of commerce. Those nations, however, not improbably, saw from the first that it was the Union which made as well our foreign as our domestic commerce. They can scarcely have failed to perceive that the effort for disunion produces the existing difficulty; and that one strong nation promises more durable peace, and a more extensive, valuable, and reliable commerce, than can the same nation broken into hostile fragments.

It is not my purpose to review our discussions with foreign States; because whatever might be their wishes or dispositions, the integrity of our country, and the stability of our Government, mainly depend, not upon them, but on the loyalty, virtue, patriotism, and intelligence of the American people. The correspondence itself, with the usual reservations, is herewith submitted. I venture to hope it will appear that we have practiced prudence and liberality toward foreign Powers, averting causes of irritation, and, with firmness, maintaining our own rights and honor.

Since, however, it is apparent that here, as in every other State, foreign dangers necessarily attend domestic difficulties, I recommend that adequate and ample measures be adopted for maintaining the public defenses on every side. While, under this general recommendation, provision for defending our sea-coast line readily occurs to the mind, I also, in the same connection, ask the attention of Congress to our great lakes and rivers. It is believed that some fortifications and depots of arms and munitions, with harbor and navigation improvements, all at well selected points upon these, would be of great importance to the national defense and preservation. I ask attention to the views of the Secretary of War, expressed in his report, upon the same general subject.

I deem it of importance that the loyal regions of East Tennessee and Western North Carolina should be connected with Kentucky, and other faithful parts of the Union, by railroad. I therefore recommend, as a military measure, that Cougress provide for the construction of such road, as speedily as possible. Kentucky, no doubt, will co-operate, and, through her Legislature, make the most judicious selection of a line. The northern terminus must connect with some existing railroad; and whether the route shall be from Lexington, or Nicholasville, to the Cumberland Gap; or from Lebanon to the Tennessee line, in the direction of Knoxville ; or on some still different line, can easily be determined. Kentucky and the General Government co-operating, the work can be completed in a very short time; and when done, it will be not only of vast present usefulness, but also a valuable permanent improvement, worth its cost in all the future.

Some treaties, designed chiefly for the interests of commerce, and having no grave political importance, have been negotiated, and will be submitted to the Senate for their consideration.

Although we have failed to induce some of the commercial Powers to adopt a desirable melioration of the rigor of maritime war, we have removed all obstructions from the way of this humane reform, except such as are merely of temporary and accidental occurrence.

I invite your attention to the correspondence between her Britannic Majesty's minister accredited to this Government and the Secretary of State, relative to the detention of the British ship Perthshire, in June last, by the United States steamer Massachusetts, for a supposed breach of the blockade. As this detention was occasioned by an obvious misapprehension of the facts, and as justice requires that we should commit no belligerent act not founded in strict right, as sanctioned by public law, I recommend that an appropriation be made to satisfy the reasonable demand of the owners of the vessel for her detention.

I repeat the recommendation of my predecessor, in his annual message to Congress in December last, in regard to the disposition of the

surplus which will probably remain after satisfying the claims of American citizens against China, pursuant to the awards of the commissioners under the act of the 3d of March, 1859. If, however, it should not be deemed advisable to carry that recommendation into effect, I would suggest that authority be given for investing the principal, over the proceeds of the surplus referred to, in good securities, with a view to the satisfaction of such other just claims of our citizens against China as are not unlikely to arise hereafter in the course of our extensive trade with that empire.

By the act of the 5th of August last, Congress authorized the President to instruct the commanders of suitable vessels to defend themselves against, and to capture pirates. This authority has been exercised in single instances For the more effectual protection of our only. extensive and valuable commerce, in the eastern scas especially, it seems to me that it would also be advisable to authorize the commanders of sailing vessels to recapture any prizes which pirates may make of United States vessels and their cargoes, and the consular courts, now established by law in eastern countries, to adjudicate the case, in the event that this should not be objected to by the local authorities.

If any good reason exists why we should persevere longer in withholding our recognition of the independence and sovereignty of Hayti and Liberia, I am unable to discern it. Unwilling, however, to inaugurate a novel policy in regard to them without the approbation of Congress, I submit for your consideration the expediency of an appropriation for maintaining a charge d'affaires near each of those new States. It does not admit of doubt that important commercial advantages might be secured by favorable treaties with them.

The operations of the Treasury during the period which has elapsed since your adjournment have been conducted with signal success. The patriotism of the people has placed at the disposal of the Government the large means demanded by the public exigencies. Much of the national loan has been taken by citizens of the industrial classes, whose confidence in their country's faith, and zeal for their country's deliverance from present peril, have induced them to contribute to the support of the Government the whole of their limited acquisitions. This fact imposes peculiar obligations to economy in disbursement and energy in action.

The revenue from all sources, including loans, for the financial year ending on the 30th June, 1861, was \$86,835,900 27, and the expenditures for the same period, including payments on account of the public debt, were \$84,578,834 47; leaving a balance in the Treasury, on the 1st July, of \$2,257,065 80. For the first quarter of the financial year, ending on the 30th September, 1861, the receipts from all sources, including the balance of 1st of July, were \$102,532,-509 27, and the expenses \$98,239,733 09; leaving a balance on the 1st of October, 1861, of \$4,292,776 18.

Estimates for the remaining three quarters of the year, and for the financial year 1863, together with his views of ways and means for meeting the demands contemplated by them, will be submitted to Congress by the Secretary of the Treasury. It is gratifying to know that the expenditures made necessary by the rebellion are not beyond the resources of the loyal people, and to believe that the same patriotism which has thus far sustained the Government will continue to susta'n it till peace and union shall again bless the land.

I respectfully refer to the report of the Secretary of War for information respecting the numerical strength of the Army, and for recommendations having in view an increase of its efficiency and the well-being of the various branches of the service intrusted to his care. It is gratifying to know that the patriotism of the people has proved equal to the occasion, and that the number of troops tendered greatly exceeds the force which Congress authorized me to call into the field.

I refer with pleasure to those portions of his report which make allusion to the creditable degree of discipline already attained by our troops, and to the excellent sanitary condition of the entire army.

The recommendation of the Secretary for an organization of the militia upon a uniform basis, is a subject of vital importance to the future safety of the country, and is commended to the serious attention of Congress.

The large addition to the regular army, in connection with the defection that has so considerably diminished the number of its officers, gives peculiar importance to his recommendation for increasing the corps of cadets to the greatest capacity of the Military Academy. By mere omission, I presume, Congress has

By mere omission, I presume, Congress has failed to provide chaplains for hospitals occupied by volunteers. This subject was brought to my notice, and I was induced to draw up the form of a letter, one copy of which, properly addressed, has been delivered to each of the persons, and at the dates respectively named and stated, in a schedule, containing also the form of the letter, marked A, and herewith transmitted.

These gentlemen, I understand, entered upon the duties designated, at the times respectively stated in the schedule, and have labored faithfully therein ever since. I therefore recommend that they be compensated at the same rate as chaplains in the army. I further suggest that general provision be made for chaplains to serve in hospitals, as well as with regiments.

The report of the Secretary of the Navy presents in detail the operations of that branch of the service, the activity and energy which have characterized its administration, and the results of measures to increase its efficiency and power. Such have been the additions, by construction and purchase, that it may almost be said a navy has been created and brought into Service since our difficulties commenced.

Besides blockading our extensive coast, squadrons larger than ever before assembled under our flag have been put afloat and performed deeds which have increased our naval renown.

I would invite special attention to the recom- for all the circuits.

mendation of the Secretary for a more perfect organization of the navy by introducing additional grades in the service.

The present organization is defective and unsatisfactory, and the suggestions submitted by the Department will, it is believed, if adopted, obviate the difficulties alluded to, promote harmony, and increase the efficiency of the Navy.

There are three vacancies on the bench of the Supreme Court-two by the decease of Justices Daniel and McLean, and one by the resig-nation of Justice Campbell. I have so far forborne making nominations to fill these vacancies for reasons which I will now state. Two of the outgoing judges resided within the State now overrun by revolt; so that if successors were appointed in the same localities, they could not now serve upon their circuits; and many of the most competent men there probably would not take the personal hazard of accepting to serve, even here, upon the supreme bench. I have been unwilling to throw all the appointments northward, thus disabling myself from doing justice to the South on the return of peace; although I may remark that to transfer to the North one which has heretofore been in the South, would not, with reference to territory and population, be unjust.

During the long and brilliant judicial career of Judge McLean his circuit grew into an empire—altogether too large for any one judge to give the courts therein more than a nominal attendance—rising in population from one million four hundred and seventy thousand and eighteen, in 1830, to six million one hundred and fifty-one thousand four hundred and five in 1860.

Besides this, the country generally has outgrown our present judicial system. If uniformity was at all intended, the system requires that all the States shall be accommodated with circuit courts attended by supreme judges, while, in fact, Wisconsin, Minnesota, Iowa, Kan-sas, Florida, Texas, California, and Oregon, have never had any such courts. Nor can this well be remedied without a change of the system; because the adding of judges to the Supreme Court, enough for the accommodation of all parts of the country, with circuit courts, would create a court altogether too numerous for a judicial body of any sort. And the evil, if it be one, will increase as new States come into the Union. Circuit courts are useful, or they are not useful. If useful, no State should be deuied them; if not useful, no State should Let them be provided for all, or have them. abolished as to all.

Three modifications occur to me, either of which, I think, would be an improvement upon our present system. Let the Supreme Court be of convenient number in every event. Then, first, let the whole country be divided into circuits of convenient size, the supreme judges to serve in a number of them corresponding to their own number, and independent circuit judges be provided for all the rest. Or, secondly, let the supreme judges be relieved from circuit, duties, and circuit judges be provided for all the circuits. Or, thirdly, dispense with circuit judges altogether, leaving the judicial functions wholly to the district courts and an independent Supreme Court.

I respectfully recommend to the consideration of Congress the present condition of the statute laws, with the hope that Congress will be able to find an easy remedy for many of the inconveniences and evils which constantly embarrass those engaged in the practical administration of them. Since the organization of the Government, Congress has enacted some five thousand acts and joint resolutions, which fill more than six thousand closely printed pages, and are scattered through many volumes. Many of these acts have been drawn in haste, and without sufficient caution, so that their provisions are often obscure in themselves, or in conflict with each other, or at least so doubtful as to render it very difficult for even the best informed persons to ascertain precisely what the statute law really is.

It seems to me very important that the statute laws should be made as plain and intelligible as possible, and be reduced to as small a compass as may consist with the fullness and precision of the will of the legislature, and the perspicuity of its language. This, well done, would, I think, greatly facilitate the labors of those whose duty it is to assist in the administration of the laws, and would be a lasting benefit to the people, by placing before them, in a more accessible and intelligible form the laws which so deeply concern their interest and their duties.

I am informed by some whose opinions I respect that all the acts of Congress now in force, and of a permanent and general nature, might be revised and rewritten, so as to be embraced in one volume (or at most two volumes) of ordinary and convenient size. And I respectfully recommend to Congress to consider of the subject, and, if my suggestion be approved, to devise such plan as to their wisdom shall seem most proper for the attainment of the end proposed.

One of the unavoidable consequences of the present insurrection is the entire suppression, in many places, of all the ordinary means of administering civil justice by the officers, and in the forms of existing law. This is the case, in whole or in part, in all the insurgent States; and as our armies advance upon and take possession of parts of those States, the practical evil becomes more apparent. There are no courts nor officers to whom the citizens of other States may apply for the enforcement of their lawful claims against citizens of the insurgent States; and there is a vast amount of debt constituting such claims. Some have estimated it as high as \$200,000,000, due, in large part, from insurgents, in open rebellion, to loyal citizens, who are, even now, making great sacrifices in the discharge of their patriotic duty to support the Government.

Under these circumstances, I have been urgently solicited to establish, by military power, courts to administer summary justice in such cases. I have thus far declined to do it, not because I had any doubt that the end proposed —the collection of the debts—was just and

right in itself, but because I have been unwilling to go beyond the pressure of necessity in the unusual exercise of power. But the powers of Congress I suppose are equal to the anomalous occasion, and therefore I refer the whole matter to Congress, with the hope that a plan may be devised for the administration of justice in all such parts of the insurgent States and Territories as may be under the control of this Government, whether by a voluntary return to allegiance and order, or by the power of our arms. This, however, not to be a permanent institution, but a temporary substitute, and to cease as soon as the ordinary courts can be re-established in peace.

It is important that some more convenient means should be provided, if possible, for theadjustment of claims against the Government, especially in view of their increased number by reason of the war. It is as much the duty of Government to render prompt justice against itself, in favor of citizens, as it is to administer the same between private individuals. The investigation and adjudication of claims, in their nature belong to the judicial department; besides, it is apparent that the attention of Congress will be more than usually engaged, for some time to come, with great national questions. It was intended by the organization of the Court of Claims mainly to remove this branch of business from the Halls of Congress; but while the court has proved to be an effective and valuable means of investigation, it in a great degree fails to effect the object of its creation, for want of power to make its judgments final.

Fully aware of the delicacy, not to say the danger, of the subject, I commend to your careful consideration whether this power of making judgments final may not properly be given to the court, reserving the right of appeal on questions of law to the Supreme Court, with such other provisions as experience may have shown to be necessary.

I ask attention to the report of the Postmaster General, the following being a summary statement of the condition of the Department:

The revenue from all sources during the fiscal year ending June 30, 1861, including the annual permanent appropriation of \$700,000 for the transportation of free mail matter, was \$9,049,-296 40, being about two per cent. less than the revenue for 1860.

The expenditures were \$13,606,750 11, showing a decrease of more than eight per cent. as compared with those of the previous year, and leaving an excess of expenditure over the rerenue for the last fiscal year of \$4,557,462 71.

The gross revenue for the year ending June 30, 1863, is estimated at an increase of four per cent on that of 1861, making \$8,683,000, to which should be added the earnings of the Department in carrying free matter, viz: \$700,000, making \$9,383,000.

The total expenditures for 1863 are estimated at \$12,528,000, leaving an estimated deficiency of \$3,145,000, to be supplied from the Treasury, in addition to the permanent appropriation.

 tal here, was eminently wise, and consequently that the relinquishment of that portion of it which lies within the State of Virginia was unwise and dangerous. I submit for your consideration the expediency of regaining that part of the District, and the restoration of the original boundaries thereof, through negotiations with the State of Virginia.

The report of the Secretary of the Interior, with the accompanying documents, exhibits the condition of the several branches of the public business pertaining to that Department. The depressing influences of the insurrection have been especially felt in the operations of the Patent and General Land Offices. The cash receipts from the sales of public lands during the past year have exceeded the expenses of our land system only about two hundred thou-The sales have been entirely sand dollars. suspended in the southern States, while the interruptions to the business of the country, and the diversion of large numbers of men from labor to military service, have obstructed settlements in the new States and Territories of the Northwest.

The receipts of the Patent Office have declined in nine months about one hundred thousand dollars, rendering a large reduction of the force employed necessary to make it self-sustaining.

The demands upon the Pension Office will be largely increased by the insurrection. Numerous applications for pensions, based upon the casualties of the existing war, have already been made. There is reason to believe that many who are now upon the pension rolls, and in receipt of the bounty of the Government, are in the ranks of the insurgent army, or giving them aid and comfort. The Secretary of the Interior has directed a suspension of the payment of the pensions of such persons upon proof of their disloyalty. I recommend that Congress authorize that officer to cause the names of such persons to be stricken from the pension rolls.

The relations of the Government with the Indian tribes have been greatly disturbed by the insurrection, especially in the southern superintendency and in that of New Mexico. The Indian country south of Kansas is in the possession of insurgents from Texas and Arkansas. The agents of the United States, appointed since the 4th of March, for this superintendency have been unable to reach their posts, while the most of those who were in office before that time have espoused the insurrectionary cause, and assume to exercise the powers of agents by virtue of commissions from the insurrectionists. It has been stated in the public press that a portion of those Indians have been organized as a military force, and are attached to the army of the insurgents. Although the Government has no official information upon this subject, letters have been written to the Commissioner of Indian Affairs by several prominent chiefs, giving assurance of their loyalty to the United States, and expressing a wish for the presence of Federal troops to protect them. It is believed that upon the repossession of the country by the Federal forces the Indians will readily cease

all hostile demonstrations, and resume their former relations to the Government.

Agriculture, confessedly the largest interest of the nation, has, not a department, nor a bureau, but a clerkship only, assigned to it in the Government. While it is fortunate that this great interest is so independent in its nature as to not have demanded or extorted more from the Government, I respectfully ask Congress to consider whether something more cannot be given voluntarily, with general advantage.

Annual reports, exhibiting the condition of our agriculture, commerce, and manufactures would present a fund of information of great practical value to the country. While I make no suggestions as to details, J venture the opinion that an agricultural and statistical bureau might profitably be organized.

The execution of the laws for the suppression of the African slave trade has been confided to the Department of the Interior. It is a subject of gratulation that the efforts which have been made for the suppression of this inhuman traffic have been recently attended with unusual Five vessels being fitted out for the success. slave trade have been seized and condemned. Two mates of vessels engaged in the trade, and one person in equipping a vessel as a slaver, have been convicted and subjected to the penalty of fine and imprisonment, and one captain, taken with a cargo of Africans on board his vessel, has been convicted of the highest grade of offence under our laws, the punishment of which is death.

The Territories of Colorado, Dakotah, and Nevada, created by the last Congress, have been organized, and civil administration has been inaugurated therein under auspices especially gratifying, when it is considered that the leaven of treason was found existing in some of these new countries when the Federal officers arrived there.

The abundant natural resources of these Territories, with the security and protection afforded by organized government, will doubtless invite to them a large immigration when peace shall restore the business of the country to its accustomed channels. I submit the resolutions of the Legislature of Colorado, which evidence the patriotic spirit of the people of the Territory. So far the authority of the United States has been upheld in all the Territories, as it is hoped it will be in the future. I commend their interests and defence to the enlightened and generous care of Congress.

I recommend to the favorable consideration of Congress the interests of the District of Columbia. The insurrection has been the cause of much suffering and sacrifice to its inhabitants, and as they have no representative in Congress, that body should not overlook their just claims upon the Government.

At your late session a joint resolution was adopted authorizing the President to take measures for facilitating a proper representation of the industrial interests of the United States at the exhibition of the industry of all nations to be holden at London in the year 1862. I regret to say I have been unable to give personal attention to this subject—a subject at once so interesting in itself, and so extensively and intimately connected with the material prosperity of the world. Through the Secretaries of State and of the Interior a plan, or system, has been devised, and partly matured, and which will be laid before you.

Under and by virtue of the act of Congress entitled "An act to confiscate property used for insurrectionary purposes," approved Au-gust 6, 1861, the legal claims of certain persons to the labor and service of certain other persons have become forfeited; and numbers of the latter, thus liberated, are already dependent on the United States, and must be provided for in some way. Besides this, it is not impossible that some of the States will pass similar enactments for their own benefit respectively, and by operation of which, persons of the same class will be thrown upon them for disposal. In such case, I recommend that Congress provide for accepting such persons from such States, according to some mode of valuation, in lieu, pro tanto of direct taxes, or upon some other plan to be agreed on with such States respectively; that such persons, on such ac-ceptance by the General Government, be at once deemed free; and that, in any event, steps be taken for colonizing both classes (or the one first mentioned, if the other shall not be brought into existence,) at some place, or places, in a climate congenial to them. It might be well to consider, too, whether the free colored people already in the United States could not, so far as individuals may desire, be included in such colonization.

To carry out the plan of colonization may involve the acquiring of territory, and also the appropriation of money beyond that to be expended in the territorial acquisition. Having practiced the acquisition of territory for nearly sixty years, the question of constitutional power to do so is no longer an open one with us. The power was questioned at first by Mr. Jefferson, who, however, in the purchase of Louisiana, yielded his scruples on the plea of great expediency. If it be said that the only legitimate object of acquiring territory is to furnish homes for white men, this measure effects that object; for the emigration of colored men leaves additional room for white men remaining or coming here. Mr. Jefferson, however, placed the importance of procuring Louisiana more on political and commercial grounds than on providing room for population.

On this whole proposition, including the appropriation of money with the acquisition of territory, does not the expediency amount to absolute necessity—that, without which the Government itself cannot be perpetuated?

The war continues. In considering the policy to be adopted for suppressing the insurrection, I have been anxious and careful that the inevitable conflict for this purpose shall not degenerate into a violent and remorseless revolutionary struggle. I have, therefore, in every case, thought it proper to keep the integrity of the Union prominent as the primary object of the contest on our part, leaving all questions which are not of vital military importance

give personal attention to this subject—a sub- to the more deliberate action of the Legislaject at once so interesting in itself, and so ex- ture.

In the exercise of my best discretion I have adhered to the blockade of the ports held by the insurgents, instead of putting in force, by proclamation, the law of Congress enacted at the late session for closing those ports.

So, also, obeying the dictates of prudence, as well as the obligations of law, instead of transcending, I have adhered to the act of Congress to confiscate property used for insurrectionary purposes. If a new law upon the same subject shall be proposed, its propriety will be duly considered. The Union must be preserved; and hence, all indispensable means must be employed. We should not be in haste to determine that radical and extreme measures, which may reach the loyal as well as the disloyal, are indispensable.

The inaugural address at the beginning of the Administration, and the message to Congress at the late special session, were both mainly devoted to the domestic controversy out of which the insurrection and consequent war have sprung. Nothing now occurs to add or subtract, to or from, the principles, or general purposes, stated and expressed, in those documents.

The last ray of hope for preserving the Union peaceably expired at the assault upon Fort Sumter; and a general review of what has occurred since may not be unprofitable. What was painfully uncertain then, is much better defined and more distinct now; and the progress of events is plainly in the right direction. The insurgents confidently claimed a strong support from North of Mason and Dixon's line, and the friends of the Union were not free from apprehension on the point. This, however, was soon settled definitely, and on the right side. South of the line, noble little Delaware led off right from the first. Maryland was made to seem against the Union. Our soldiers were assaulted, bridges were burned, and railroads torn up within her limits, and we were many days, at one time, without the ability to bring a single regiment over her soil to the capital. Now her bridges and railroads are repaired and open to the Government; she already gives seven regiments to the cause of the Union, and none to the enemy; and her people, at a regular election, have sustained the Union, by a larger majority, and a larger aggregate vote than they ever before gave to any candidate or any question. Kentucky, too, for some time in doubt, is now decidedly, and, I think, unchangeably, ranged on the side of the Union. Missouri is comparatively quiet, and I believe cannot again be overrun by the insurrectionists. These three States of Maryland, Kentucky, and Missouri, neither of which would promise a single soldier at first, have now an aggregate of not less than forty thousand in the field for the Union; while, of their citizens, certainly not more than a third of that number, and they of doubtful whereabouts, and doubtful existence, are in arms against it. After a somewhat bloody struggle of months, winter closes on the Union. people of Western Virginia, leaving them masAn insurgent force of about fifteen hundred, for months dominating the narrow peninsula region, constituting the counties of Accomac and Northampton, and known as eastern shore of Virginia, together with some contiguous parts of Maryland, have laid down their arms; and the people there have renewed their allegiance to, and accepted the protection of, the old flag. This leaves no armed insurrectionist north of the Potomac, or east of the Chesapeake.

Also we have obtained a footing at each of the isolated points, on the southern coast, of Hatteras, Port Royal, Tybee Island, near Savannah, and Ship Island; and we likewise have some general accounts of popular movements, in behalf of the Union, in North Carolina and Tennessee.

These things demonstrate that the cause of the Union is advancing steadily and certainly southward.

Since your last adjournment, Lieutenant General Scott has retired from the head of the Army. During his long life, the nation has not been unmindful of his merit; yet, on calling to mind how faithfully, ably, and brilliantly he has served the country, from a time far back in our history, when few of the now living had been born, and thenceforward continually, I cannot but think we are still his debtors. I submit, therefore, for your consideration, what further mark of recognition is due to him and to ourselves, as a grateful people.

With the retirement of General Scott came the executive duty of appointing, in his stead, a General-in-Chief of the Army. It is a fortunate circumstance that neither in council nor country was there, so far as I know, any difference of opinion as to the proper person to be The retiring chief repeatedly exselected. pressed his judgment in favor of General Mc-Clollan for the position; and in this the nation seemed to give a unanimous concurrence. The designation of General McClellan is, therefore, in considerable degree, the selection of the country as well as of the Executive ; and hence there is better reason to hope there will be given him the confidence and cordial support thus, by fair implication, promised, and without which, he cannot, with so full efficiency, serve the country.

It has been said that one bad general is better than two good ones; and the saying is true, if taken to mean no more than that an army is better directed by a single mind, though inferior, than by two superior ones at variance, and cross-purposes with each other.

And the same is true, in all joint operations wherein those engaged, can have none but a common end in view, and can differ only as to the choice of means. In a storm at sea, no one on board can wish the ship to sink; and yet, not unfrequently, all go down together, because too many will direct, and no single mind can be allowed to control.

It continues to develop that the insurrection is largely, if not exclusively, a war upon the first principle of popular government — the rights of the people. Conclusive evidence of this is found in the most grave and maturely considered public documents, as well as in the general tone of the insurgents. In those documents we find the abridgment of the existing right of suffrage, and the denial to the people of all right to participate in the selection of public officers, except the legislative, boldly advocated, with labored arguments to prove that large control of the people in government is the source of all political evil. Monarchy itself is sometimes hinted at as a possible refuge from the power of the people.

In my present position, I could scarcely be justified were I to omit raising a warning voice against this approach of returning despotism.

It is not needed, nor fitting here, that a general argument should be made in favor of popular institutions; but there is one point, with its connections, not so hackneyed as most others, to which I ask brief attention. It is the effort to place capital on an equal footing with, if not above, labor, in the structure of government. It is assumed that labor is available only in connection with capital; that nobody labors unless somebody else, owning capital, somehow by the use of it, induces him to labor. This assumed, it is next considered whether it is best that capital shall hire laborers, and thus induce them to work by their own consent, or buy them, and drive them to it without their consent. Having proceeded so far, it is naturally concluded that all laborers are either hired laborers, or what we call slaves. And further, it is assumed that whoever is once a hired laborer is fixed in that condition for life.

Now, there is no such relation between capital and labor as assumed; nor is there any such thing as a free man being fixed for life, in the condition of a hired laborer. Both these assumptions are false, and all inferences from them are groundless.

Labor is prior to, and independent of, capi-Capital is only the fruit of labor, and tal. could never have existed if labor had not first existed. Labor is the superior of capital, and deserves much the higher consideration. Capital has its rights, which are as worthy of protection as any other rights. Nor is it denied that there is, and probably always will be, a relation between labor and capital, producing mutual benefits. The error is in assuming that the whole labor of community exists within that relation. A few men own capital, and that few avoid labor themselves, and with their capital hire or buy another few to labor for them. A large majority belong to neither class -neither work for others, nor have others working for them. In most of the southern States, a majority of the whole people of all colors are neither slaves nor masters; while in the northern, a large majority are neither hirers Men with their families-wives, nor hired. sons, and daughters-work for themselves, on their farms, in their houses, and in their shops, taking the whole product to themse.ves, and asking no favors of capital on the one hand, nor It is of hired laborers or slaves on the other. not forgotten that a considerable number of persons mingle their own labor with capitalthat is they labor with their own hands, and also buy or hire others to labor for them; but

this is only a mixed, and not a distinct class. No principle stated is disturbed by the existence of this mixed class.

Again, as has already been said, there is not, of necessity, any such thing as the free hired laborer being fixed to that condition for life. Many independent men everywhere in these States, a few years back in their lives, were hired laborers. The prudent, penniless beginner in the world labors for wages awhile, saves a surplus for which to buy tools or land for himself, then labors on his own account another while, and at length hires another new beginner to help him. This is the just and generous and prosperous system, which opens the way to all, gives hope to all, and consequent energy and progress, and improvement of condition to all. No men living are more worthy to be trusted than those who toil up from poverty-none less inclined to take or touch aught which they have not honestly earned. Let them beware of surrendering a political power which they already possess, and which, if surrendered, will surely be used to close the door of advancement against such as they, and to fix new disabilities and burdens upon them, till all of liberty shall be lost.

From the first taking of our national census to the last are seventy years; and we find our population at the end of the period eight times as great as it was at the beginning. The increase of those other things which men deem desirable has been even greater. We thus have, at one view, what the popular principle applied to government, through the machinery of the States and the Union, has produced in a given time, and also what, if firmly maintained, it promises for the future. There are already among us those who, if the Union be preserved, will live to see it contain two hundred and fifty millions. The struggle of to-day is not altogether for to-day—it is for a vast future also. With a reliance on Providence, all the more firm and earnest, let us proceed in the great task which events have devolved upon us.

ABRAHAM LINCOLN. WASHINGTON, December 3, 1861.

# President Lincoln's Second Annual Message, December 1, 1862.

# Fellow-citizens of the Senate

and House of Representatives:

Since your last annual assembling another year of health and bountiful harvests has passed. And while it has not pleased the Almighty to bless us with a return of peace, we can but press on, guided by the best light He gives us, trusting that in His own good time, and wise way, all will yet be well.

The correspondence touching foreign affairs which has taken place during the last year is herewith submitted, in virtual compliance with a request to that effect made by the House of Representatives near the close of the last session of Congress.

If the condition of our relations with other nations is less gratifying than it has usually been at former periods, it is certainly more satis still, however, a large and an augmenting isfactory than a nation so unhappily distracted number of doubtful cases upon which the Gov-

as we are, might reasonably have apprehended. In the month of June last there were some grounds to expect that the maritime powers which, at the beginning of our domestic difficultics, so unwisely and unnecessarily, as we think, recognized the insurgents as a belligerent, would soon recede from that position, which has proved only less injurious to themselves than to our own country. But the temporary reverses which afterwards befoll the national arms, and which were exaggerated by our own disloyal citizens abroad, have hitherto delayed that act of simple justice.

The civil war, which has so radically changed, for the moment, the occupations and habits of the American people, has necessarily disturbed the social condition, and affected very deeply the prosperity of the nations with which we have carried on a commerce that has been steadily increasing throughout a period of half a century. It has, at the same time, excited political ambitions and apprehensions which have produced a profound agitation throughout the civilized world. In this unusual agitation we have forborne from taking part in any contreversy between foreign States, and between parties or factions in such States. We have attempted no propagandism, and acknowledged no revolution. But we have left to every nation the exclusive conduct and management of its own affairs. Our struggle has been, of course, contemplated by foreign nations with reference less to its own merits than to its supposed and often exaggerated effects and consequences resulting to those nations themselves. Nevertheless, complaint on the part of this Government, even if it were just, would certainly be unwise.

The treaty with Great Britain for the suppression of the slave trade has been put into operation with a good prospect of complete success. It is an occasion of special pleasure to acknowledge that the execution of it, on the part of her Majesty's Government, has been marked with a jealous respect for the authority of the United States, and the rights of their moral and loyal citizens.

The convention with Hanover for the abolition of the stade dues has been carried into full effect, under the act of Congress for that purpose.

A blockade of three thousand miles of seacoast could not be established, and vigorously enforced, in a season of great commercial activity like the present, without committing occasional mistakes and inflicting unintentional injuries upon foreign nations and their subjects.

A civil war occurring in a country where foreigners reside and carry on trade under treaty stipulations, is necessarily fruitful of complaints of the violation of neutral rights. All such collisions tend to excite misapprehensions, and possibly to produce mutual reclamations between nations which have a common interest in preserving peace and friendship. In clear cases of these kinds I have, so far as possible, heard and redressed complaints which have been presented by friendly Powers. There is still, however, a large and an augmenting number of doubtful cases upon which the Government is unable to agree with the Governments whose protection is demanded by the claimants. There are, moreover, many cases in which the United States, or their citizens, suffer wrongs from the naval or military authorities of foreign nations, which the Governments of those States are not at once prepared to redress. I have proposed to some of the foreign States, thus interested, mutual conventions to examine and adjust such complaints. This proposition has been made especially to Great Britain, to France, to Spain, and to Prussia. In each case it has been kindly received, but has not yet been formally adopted.

I deem it my duty to recommend an appropriation in behalf of the owners of the Norwegian bark Admiral P. Tordenskiold, which vessel was, in May, 1861, prevented by the commander of the blockading force off Charleston from leaving that port with cargo, notwithstanding a similar privilege had, shortly before, been granted to an English vessel. I have directed the Secretary of State to cause the papers in the case to be communicated to the proper committees.

Applications have been made to me by many free Americans of African descent to favor their emigration, with a view to such colonization as was contemplated in recent acts of Congress. Other parties, at home and abroad-some from interested motives, others upon patriotic considerations, and still others influenced by philanthropic sentiments-have suggested similiar measures; while, on the other hand, several of the Spanish-American republics have protested against the sending of such colonies to their respective territories. Under these circumstances, I have declined to move any such colony to any State, without first obtaining the consent of its government, with an agreement on its part to receive and protect such emigrants in all the rights of freemen; and I have, at the same time, offered to the several States situated within the tropics, or having colonies there, to negotiate with them, subject to the advice and consent of the Senate, to favor the voluntary emigration of persons of that class to their respective territories, upon conditions which shall be equal, just, and humane. Liberia and Hayti are, as yet, the only countries to which colonists of African descent from here, could go with certainty of being received and adopted as citizens; and I regret to say such persons, contemplating colonization, do not seem so willing to migrate to those countries, as to some others, nor so willing as I think their interest demands. I believe, however, opinion among them in this respect, is improving; and that, ere long, there will be an augmented, and considerable migration to both these countries, from the United States.

The new commercial treaty between the United States and the Sultan of Turkey has been carried into execution.

A commercial and consular treaty has been negotiated, subject to the Senate's consent, with Liberia; and a similar negotiation is now pending with the republic of Hayti. A considerable improvement of the national commerce is expected to result from these measures.

Our relations with Great Britain, France, Spain, Portugal, Russia, Prussia, Denmark, Sweden, Austria, the Netherlands, Italy, Rome, and the other European States, remain undisturbed. Very favorable relations also continue to be maintained with Turkey, Morocco, China, and Japan.

During the last year there has not only been no change of our previous relations with the independent States of our own continent, but more friendly sentiments than have heretofore existed, are believed to be entertained by these neighbors, whose safety and progress are so intimately connected with our own. This statement especially applies to Mexico, Nicaragua, Costa Rica, Honduras, Peru, and Chile.

The commission under the convention with the republic of New Granada closed its session without having audited and passed upon all the claims which were submitted to it. A proposition is pending to revive the convention, that it may be able to do more complete justice. The joint commission between the United States and the republic of Costa Rica has completed its labors and submitted its report.

I have favored the project for connecting the United States with Europe by an Atlantic telegraph, and a similar project to exend the telegraph from San Francisco, to connect by a Pacific telegraph with the line which is being extended across the Russian empire.

The Territories of the United States, with unimportant exceptions, have remained undisturbed by the civil war; and they are exhibiting such evidence of prosperity as justifies an expectation that some of them will soon be in a condition to be organized as States, and be constitutionally admitted into the Federal Union.

The immense mineral resources of some of those Territories ought to be developed as rapidly as possible. Every step in that direction would have a tendency to improve the revenues of the Government, and diminish the burdens of the people. It is worthy of your serious consideration whether some extraordinary measures to promote that end cannot be adopted. The means which suggests itself as most likely to be effective, is a scientific exploration of the mineral regions in those Territories, with a view to the publication of its results at home and in foreign countries—results which cannot fail to be auspicious.

The condition of the finances will claim your most diligent consideration. The vast expenditures incident to the military and naval operations required for the suppression of the rebellion, have hitherto been met with a promptitude and certainty unusual in similar circumstances; and the public credit has been fully maintained. The continuance of the war, however, and the increased disbursements made necessary by the augmented forces now in the field, demand your best reflections as to the best modes of providing the necessary revenue, without injury to business, and with the least possible burdens upon labor.

The suspension of specie payments by the banks, soon after the commencement of your last session, made large issues of United States notes unavoidable. In no other way could the payment of the troops, and the satisfaction of other just demands, be so economically or so well provided for. The judicious legislation of Congress, securing the receivability of these notes for loans and internal duties, and making them a legal tender for other debts, has made them an universal currency; and has satisfied, partially, at least, and for the time, the long felt want of an uniform circulating medium, saving thereby to the people immense sums in discounts and exchanges.

A return to specie payments, however, at the earliest period compatible with due regard to all interests concerned, should ever be kept in view. Fluctuations in the value of currency are always injurious, and to reduce these fluctuations to the lowest possible point will always be a leading purpose in wise legislation. Convertibility, prompt and certain convertibility into coin, is generally acknowledged to be the best and surest safeguard against them; and it is extremely doubtful whether a circulation of United States notes, payable in coin, and sufficiently large for the wants of the people, can be permanently, usefully, and safely maintained.

Is there, then, any other mode in which the necessary provision for the public wants can be made, and the great advantages of a safe and uniform currency secured?

I know of none which promises so certain results, and is, at the same time, so unobjectionable, as the organization of banking associations, under a general act of Congress, well guarded in its provisions. To such associations the Government might furnish circulating notes, on the security of United States bonds deposited in the Treasury. These notes, prepared under the supervision of proper officers, being uniform in appearance and security, and convertible always into coin, would at once protect labor against the evils of a vicious currency, and facilitate commerce by cheap and safe exchanges.

A moderate reservation from the interest on the bonds would compensate the United States for the preparation and distribution of the notes, and a general supervision of the system, and would lighten the burden of that part of the public debt employed as securities. The public credit, moreover, would be greatly improved, and the negotiation of new loans greatly facilitated by the steady market demand for Government bonds which the adoption of the proposed system would create.

It is an additional recommendation of the measure, of considerable weight, in my judgment, that it would reconcile, as far as possible, all existing interests, by the opportunity offered to existing institutions to reorganize under the act, substituting only the secured uniform national circulation for the local and various circulation, secured and unsecured, now issued by them.

The receipts into the Treasury from all expenditures as compared with the preceding sources, including loans, and balance from the year, and about \$3,750,000 as compared with preceding year, for the fiscal year ending on the the fiscal year 1860. The deficiency in the 30th June, 1862, were \$583,885,247 06, of Department for the previous year was \$4,551,-

which sum \$49,056,397 62 were derived from customs; \$1,705,331 73 from the direct tax; from public lands, \$152,203 77; from miscellaneous sources, \$931,787 64; from loans in all forms, \$529,692,460 50. The remainder, \$2,257,065 80, was the balance from last year.

The disbursements during the same period were for congressional, executive, and judicial purposes, \$5,939,000 29; for foreign intercourse, \$1,339,710 35; for miscellaneous expenses, including the mints, loans, post office deficiencies, collection of revenue, and other like charges, \$14,129,771 50; for expenses under the Interior Department, \$3,102,985 52; under the War Department, \$3,102,985 52; under the War Department, \$42,674,669 66; under the Navy Department, \$42,674,669 66; and for payment of public debt, including reimbursement of temporary loan, and redemptions, \$96,096,022 09; making an aggregate of \$570,841,700 25, and leaving a balance in the Treasury on the first day of July, 1862, of \$13,043,546 81.

It should be observed that the sum of \$96,096,922 09, expended for reimbursements and redemption of public debt, being included also in the loans made, may be properly deducted, both from receipts and expenditures, leaving the actual receipts for the year, \$487,788,324 97; and the expenditures, \$474,744,778 16.

Other information on the subject of the finances will be found in the report of the Secretary of the Treasury, to whose statements and views I invite your most candid and considerate attention.

The reports of the Secretaries of War, and of the Navy, are herewith transmitted. These reports, though lengthy, are scarcely more than brief abstracts of the very numerous and extensive transactions and operations conducted through those departments. Nor could I give a summary of them here, upon any principle, which would admit of its being much shorter than the reports themselves I therefore content myself with laying the reports before you, and asking your attention to them.

It gives me pleasure to report a decided improvement in the financial condition of the Post Office Department, as compared with several preceding years. The receipts for the fiscal preceding years. year 1861 amounted to \$8,349,296 40, which embraced the revenue from all the States of the Union for three quarters of that year. Notwithstanding the cessation of revenue from the so-called Seceded States during the last fiscal year, the increase of the correspondence of the loyal States has been sufficient to produce a revenue during the same year of \$8,299,820 90, being only \$50,000 less than was derived from all the States of the Union during the previous year. The expenditures show a still more favorable result. The amount expended in 1861 was \$13,606,759 11. For the last year the amount has been reduced to \$11,125,364 13, showing a decrease of about \$2,481,000 in the expenditures as compared with the preceding year, and about \$3,750,000 as compared with the fiscal year 1860. The deficiency in the

966 98. For the last fiscal year it was reduced to \$2,112,814 57. These favorable results are in part owing to the cessation of mail service in the insurrectionary States, and in part to a careful review of all expenditures in that Department in the interest of economy. The efficiency of the postal service, it is believed, has also been much improved. The Postmaster General has also opened a correspondence, through the Department of State, with foreign Governments, proposing a convention of postal representatives for the purpose of simplifying the rates of foreign postage, and to expedite the foreign mails. This proposition, equally important to our adopted citizens, and to the commercial interests of this country, has been favorably entertained, and agreed to, by all the Governments from whom replies have been received.

I ask the attention of Congress to the suggestions of the Postmaster General in his report respecting the further legislation required, in his opinion, for the benefit of the postal service.

The Secretary of the Interior reports as follows in regard to the public lands:

"The public lands have ceased to be a source of revenue. From the 1st of July, 1861, to the 30th Scptember, 1862, the entire cash receipts from the sale of lands were \$137,476 26—a sum much less than the expenses of our land system during the same period. The homestead law, which will take effect on the 1st of January next, offers such inducements to settlers, that sales for cash cannot be expected, to an extent sufficient to meet the expenses of the General Land Office, and the cost of surveying and bringing the land into market."

The discrepancy between the sum here stated as arising from the sales of the public lands, and the sum derived from the same source as reported from the Treasury Department arises, as I understand, from the fact that the periods of time, though apparently, were not really, coincident at the beginning point—the Treasury report including a considerable sum now, which had previously been reported from the Interior—sufficiently large to greatly overreach the sum derived from the three months now reported by the Interior, and not by the Treasury.

The Indian tribes upon our frontiers have, during the past year, manifested a spirit of insubordination, and, at several points, have engaged in open hostilities against the white settlements in their vicinity. The tribes occupying the Indian country south of Kansas, renounced their allegiance to the United States, and entered into treaties with the insurgents. Those who remained loyal to the United States were driven from the country. The chief of the Cherokces has visited this city for the purpose of restoring the former relations of the tribe with the United States. He alleges that they were constrained, by superior force, to enter into treaties with the insurgents, and that the United States neglected to furnish the protection which their treaty stipulations required.

In the month of August last the Sioux Indians, in Minnesota, attacked the settlements in

their vicinity with extreme ferocity, killing, indiscriminately, men, women. and children. This attack was wholly unexpected, and, therefore, no means of defence had been provided. It is estimated that no less than eight hundred persons were killed by the Indians, and a large amount of property was destroyed. How this outbreak was induced is not definitely known, and suspicious, which may be unjust, need not be stated. Information was received by the Indian bureau, from different sources, about the time hostilities were commenced, that a simultaneous attack was about to be made upon the white settlements by all the tribes between the Mississippi river and the Rocky Mountains. The State of Minnesota has suffered great injury from this Indian war. A large portion of her territory has been depopulated, and a severe loss has been sustained by the destruction of property. The people of that State manifest much anxiety for the removal of the tribes beyond the limits of the State as a guarantee against future hostilities. The Commissioner of Indian Affairs will furnish full details. I submit for your special consideration whether our Indian system shall not be remodelled. Many wise and good men have impressed me with the belief that this can be profitably done.

I submit a statement of the proceedings of commissioners, which shows the progress that has been made in the enterprise of constructing the Pacific railroad. And this suggests the earliest completion of this road, and also the favorable action of Congress upon the projects\_ now pending before them for enlarging the capacifies of the great canals in New York and Illinois, as being of vital, and rapidly increasing importance to the whole nation, and especially to the vast interior region hereinafter to be noticed at some greater length. I propose having prepared and laid before you at an early day some interesting and valuable statistical information upon this subject. The military and commercial importance of enlarging the Illinois and Michigan canal, and improving the Illinois river, is presented in the report of Colonel Webster to the Secretary of War, and now transmitted to Congress. I respectfully ask attention to it.

To carry out the provisions of the act of Congress of the 15th of May last, I have caused the Department of Agriculture of the United States to be organized.

The Commissioner informs me that, within the period of a few months this department has established an extensive system of correspondence and exchanges, both at home and abroad, which promises to effect highly beneficial results in the development of a correct knowledge of recent improvements in agriculture, in the introduction of new products, and in the collection of the agricultural statistics of the different States.

Also that it will soon be prepared to distribute largely seeds, cereals, plants and cuttings, and has already published, and liberally diffused, much valuable information in anticipation of a more elaborate report, which will in due time be furnished, embracing some valuable tests in chemical science now in progress in the laboratory.

The creation of this department was for the more immediate benefit of a large class of our most valuable citizens; and I trust that the liberal basis upon which it has been organized will not only meet your approbation, but that it will realize, at no distant day, all the fondest anticipations of its most sanguine friends, and become the fruitful source of advantage to all our people.

For the residue of the Message, see chapter on "Confiscation and Emancipation."

# Members of the 38th Congress, 1st Session, Dec. 7, 1863-July 4, 1864. SENATE.

The following are the changes from the list as it stood at the close of the 37th Congress:

RHODE ISLAND-Wm. Sprague, in place of Samuel G. Arnold.

NEW YORK-Edwin D. Morgan, in place of Preston King.

NEW JERSEY-Wm. Wright, in place of James W. Wall.

PENNSYLVANIA -- Charles R. Buckalew, in place of David Wilmot.

DELAWARE-1864, Feb. 2, George Read Riddle, in place of James A. Bayard resigned Jan. 29, 1864.

MARYLAND--Reverdy Johnson, in place of Anthony Kennedy.

VIRGINIA-Lemuel J. Bowden, in place of Waitman T. Willey, term expired. Mr. B. died Jan. 2, 1864. His vacancy not filled.

WEST VIRGINIA-(New Senators) Waitman T. Willey, term to expire March 3, 1865, Peter G. Van Winkle, term to expire March 3, 1869. INDIANA-Thomas A. Hendricks, in place of

David Turpie.

MISSOURI-1863, Dec. 14, B. Gratz Brown, elected for the unexpired term of Waldo P. Johnson, expelled, and in place of R. Wilson, appointed.

CALIFORNIA—John Conness, in place of Milton S. Latham.

MINNESOTA-Alexander Ramsey, in place of Henry M. Rice.

TENNESSEE-Unrepresented.

MENORANDUM — Messrs. Elisha Baxter and Wm. M. Fishback, claiming to represent the State of Arkansas, were not admitted.

HOUSE OF REPRESENTATIVES.

SCHUYLER COLFAX, of Indiana, Speaker.

MAINE-Lorenzo D. M. Sweat, Sidney Perham, James G. Blaine, John H. Rice, Frederick A. Pike.

rick A. Pike. NEW HAMPSHIRE—Daniel Marcy, Edward H. Rollins, James W. Patterson.

VERMONT—Frederick E. Woodbridge, Justin S. Morrill, Portus Baxter.

MASSACHUSETTS-Thomas D. Eliot, Oakes Ames, Alexander H. Rice, Samuel Hooper, John B. Alley, Daniel W. Gooch, George S. Boutwell, John D. Baldwin, William B. Washburn, Henry L. Dawes.

RHODE ISLAND-Thomas A. Jenckes, Nathan F. Dixon.

CONNECTIOUT—Henry C. Deming, James E. English, Augustus Brandegee, John H. Hubbard.

NEW YORK — Henry G. Stebbins, Martin Kalbfleisch, Moses F. Odell, Benjamin Wood, Fernando Wood, Elijah Ward, John W. Chanler, James Brooks, Anson Herrick, William Radford, Charles H. Winfield, Homer A. Nelson, John B. Steele, John V. L. Pruyn, John A. Griswold, Orlando Kellogg, Calvin T. Hulburd, James M. Marvin, Samuel F. Miller, Ambrose W. Clark, Francis Kernan, De Witt C. Littlejohn, Thomas T. Davis, Theodore M. Pomeroy, Daniel Morris, Giles H. Hotchkiss, Robert B. Van Valkenburgh, Freeman Clarkee, Augustus Frank, John Ganson, Reuben E. Fenton.

NEW JERSEY—John F. Starr, George Middleton, William G. Steele, Andrew J. Rogers, Nehemiah Perry,

PENNSYLVANIA—Samuel J. Randall, Charles O'Neill, Leonard Myers, William D. Kelley, M. Russell Thayer, John D. Stiles, John M. Broomall, Sydenham E. Ancona, Thaddeus Stevens, Myer Strouse, Philip Johnson, Charles Denison, Henry W. Tracy, William H. Miller, Joseph Baily, Alexander H. Coffroth, Archibald McAllister, James T. Hale, Glenni W. Scofield, Amos Myers, John L. Dawson, James K. Moorhead, Thomas Williams, Jesse Lazear.

DELAWARE-Nathaniel B. Smithers.

MARYLAND—John A. J. Cresswell, Edwin H. Webster, Henry Winter Davis, Francis Thomas, Benjamin G. Harris.

OHIO-Geo. H. Pendleton, Alexander Long, Robert C. Schenck, John F. McKinney, Francis C. LeBlond, Chilton A. White, Samuel S. Cox, William Johnson, Warren P. Noble, James M. Ashley, Wells A. Hutchins, William E. Finck, John O'Neill, George Bliss, James R. Morris, Joseph W. White, Ephraim R. Eckley, Rufus P. Spalding, James A. Garfield.

KENTUCKY-Lucien Anderson, George H. Yeaman, Henry Grider, Aaron Harding, Robert Mallory, Green Clay Smith, Brutus J. Clay, William H. Randall, William H. Wadsworth. INDIANA-John Law, James A. Cravens, Henry W. Harrington, William S. Holman,

INDIANA—John Law, James A. Cravens, Henry W. Harrington, William S. Holman, George W. Julian, Ebenezer Dumont, Daniel W. Voorhees, Godlove S. Orth, Schuyler Colfax, Joseph K. Edgerton, James F. McDowell.

 Iax, Joseph R. Edgeton, James F. McDowen, ILLINOIS—Isaac N. Arnold, John F. Farns- worth, Ellihu B. Washburne, Charles M. Harris, Owen Lovejoy,\* Jesse O. Norton, John R. Eden, John T. Stuart, Lewis W. Ross, An- thony L. Knapp, James C. Robinson, William R. Morrison, William J. Allen, James C. Allen. Merson, William J. Allen, James C. Allen.

MISSOURI-Francis P. Blair, jr.,\* Henry T. Blow, John G. Scott, Joseph W. McClurg, Sempronius H. Boyd, Austin A. King, Benjamin F. Loan, William A. Hall, James S. Rollins.

MICHIGAN—Fernando C. Beaman, Charles Upson, John W. Longyear, Francis W. Kellogg, Augustus C. Baldwin, John F. Driggs.

Iowa—James F. Wilson, Hiram Price, Wm. B. Allison, Josiah B. Grinnell, John A. Kasson, Asahel W. Hubbard.

WISCONSIN-James S. Brown, Ithamar C.

\*See memorandum at close of list.

Sloan, Amasa Cobb, Charles A. Eldridge, Ezra Wheeler, Walter D. McIndoe.

CALIFORNIA—Thomas B. Shannon, William Higby, Cornelius Cole.

MINNESOTA .- Wm. Windom, Ignatius Donnelly.

OREGON-John R. McBride.

KANSAS-A. Carter Wilder.

WEST VIRGINIA—Jacob B. Blair, William G. Brown, Kellian V. Whaley.

MEMORANDUM OF CHANGES.

These changes occurred during the session : 1864, Jan. 12, Francis P. Blair, jr., took his seat, and June 11th was unseated by vote of the House—yeas 81, nays 33. Same day, the House—yeas 70, nays 53—voted that Samuel Knox, the contestant, was entitled to the seat, and June 15th he was qualified.

1864, March 25, Owen Lovejoy died. May 20, Ebon C. Ingersoll qualified as his successor.

The names of A. P. Fields and Thomas Cottman, claiming seats from Louisiana, were placed upon the roll by the Clerk of the 37th Congress, and voted for Speaker, but their credentials were referred to the Committee on Electionsyeas 100, nays 71-and, Feb. 10, they were voted not entitled to seats, without a division. The names of the Maryland delegation; of Messrs. Blair, Blow, McClurg, Boyd, Loan and Hall, of the Missouri delegation; of Mr. Mc-Bride, of Oregon; of Mr. Wilder, of Kansas; and of the West Virginia delegation, omitted by the Clerk, were ordered by the House, to be placed upon the roll. The claims of Messrs. Chandler, Segar, and Kitchen, of Virginia, were referred to the Committee on Elections, and subsequently reported against-the House concurring in the report. Messrs. A. A. C. Rogers, James M. Johnson, and T. M. Jacks, claiming seats from Arkansas, were not admitted.

# President Lincoln's Third Annual Message, December 8, 1863.

Fellow-citizens of the Senate

and House of Representatives :

Another year of health, and of sufficiently abundant harvests, has passed. For these, and especially for the improved condition of our national affairs, our renewed and profoundest gratitude to God is due.

We remain in peace and friendship with foreign Powers.

The efforts of disloyal citizens of the United States to involve us in foreign wars, to aid an inexcusable insurrection, have been unavailing. Her Britannic Majesty's Government, as was justly expected, have exercised their authority to prevent the departure of new hostile expeditions from British ports. The Emperor of France has, by a like proceeding, promptly vindicated the neutrality which he proclaimed at the beginning of the contest. Questions of great intricacy and importance have arisen out of the blockade and other belligerent operations, between the Government and several of the maritime Powers, but they have been discussed, and, as far as was possible, accommo-

dated in a spirit of frankness, justice, and mutual good will. It is especially gratifying that our prize courts, by the impartiality of their adjudications, have commanded the respect and confidence of maritime Powers.

The supplemental treaty between the United States and Great Britain for the suppression of the African slave trade, made on the 17th day of February last, has been duly ratified and carried into execution. It is believed that, so far as American ports and American citizens are concerned, that inhuman and odious traffic has been brought to an end.

I shall submit, for the consideration of the Senate, a convention for the adjustment of possessory claims in Washington Territory, arising out of the treaty of the 15th June, 1846, between the United States and Great Britain, and which have been the source of some disquiet among the citizens of that now rapidly improving part of the country.

A novel and important question, involving the extent of the maritime jurisdiction of Spain in the waters which surround the island of Cuba, has been debated without reaching an agreement, and it is proposed, in an amicable spirit, to refer it to the arbitrament of a friendly Power. A convention for that purpose will be submitted to the Senate.

I have thought it proper, subject to the approval of the Senate, to concur with the interested commercial Powers in an arrangement for the liquidation of the Scheldt dues upon the principles which have been heretofore adopted in regard to the imposts upon navigation in the waters of Denmark.

The long pending controversy between this Government and that of Chili, touching the seizure of Sitana, in Peru, by Chilian officers, of a large amount in treasure belonging to citizens of the United States, has been brought to a close by the award of his Majesty the King of the Belgians, to whose arbitration the question was referred by the parties. The subject was thoroughly and patiently examined by that justly respected magistrate, and although the sum awarded to the claimants may not have been as large as they expected, there is no reason to distrust the wisdom of his Majesty's decision. That decision was promptly complied with by Chili, when intelligence in regard to it reached that country.

The joint commission, under the act of the last session, for carrying into effect the convention with Peru on the subject of claims, has been organized at Lima, and is engaged in the business intrusted to it.

Difficulties concerning inter-oceanic transit through Nicaragua are in course of amicable adjustment.

In conformity with principles set forth in my last annual message, I have received a representative from the United States of Colombia, and have accredited a minister to that republic.

Incidents occurring in the progress of our civil war have forced upon my attention the uncertain state of international questions touching the rights of foreigners in this country and of United States citizens abroad. In regard to some Governments these rights are at least partially defined by treaties. In no instance, however, is it expressly stipulated that, in the event of civil war, a foreigner residing in this country, within the lines of the insurgents, is to be exempted from the rule which classes him as a belligerent, in whose behalf the Government of his country cannot expect any privileges or immunities distinct from that character. I regret to say, however, that such claims have been put forward, and, in some instances, in behalf of foreigners who have lived in the United States the greater part of their lives.

There is reason to believe that many persons born in foreign countries, who have declared their intentions to become citizens, or who have been fully naturalized, have evaded the military duty required of them by denying the fact, and thereby throwing upon the Government the burden of proof. It has been found difficult or impracticable to obtain this proof, from the want of guides to the proper sources of infor-These might be supplied by requiring mation. clerks of courts, where declarations of intentions may be made or naturalizations effected, to send, periodically, lists of the names of the persons naturalized, or declaring their intention to become citizens, to the Secretary of the Interior, in whose department those names might be arranged and printed for general information.

There is also reason to believe that foreiguers frequently become citizens of the United States for the sole purpose of evading duties imposed by the laws of their native countries, to which, on becoming naturalized here, they at once repair, and, though never returning to the United States, they still claim the interposition of this Government as citizens. Many altercations and great prejudices have heretofore arisen out of this abuse. It is, therefore, submitted to your serious consideration. It might be advisable to fix a limit, beyond which no citizen of the United States residing abroad may claim the interposition of his Government.

The right of suffrage has often been assumed and exercised by aliens, under pretences of naturalization, which they have disavowed when drafted into the military service. I submit the expediency of such an amendment of the law as will make the fact of voting an estoppel against any plea of exemption from military service, or other civil obligation, on the ground of alienage.

In common with other western Powers, our relations with Japan have been brought into serious jeopardy, through the perverse opposition of the hereditary aristocracy of the empire to the enlightened and liberal policy of the Tycoon, designed to bring the country into the society of nations. It is hoped, although not with entire confidence, that these difficulties may be peacefully overcome. I ask your attention to the claim of the minister residing there for the damages he sustained in the destruction by fire of the residence of the legation at Yedo.

Satisfactory arrangements have been made with the Emperor of Russia, which, it is believed, will result in effecting a continuous line as may have arisen under treaties and the pub-

of telegraph through that empire from our Pacific coast.

I recommend to your favorable consideration the subject of an international telegraph across the Atlantic ocean; and also of a telegraph between this capital and the national forts along the Atlantic sea-board and the Gulf of Mexico. Such communications, established with any reasonable outlay, would be economical as well as effective aids to the diplomatic, military, and naval service.

The consular system of the United States, under the enactments of the last Congress, begins to be self-sustaining; and there is reason to hope that it may become entirely so, with the increase of trade which will ensue whenever peace is restored. Our ministers abroad have been faithful in defending American rights. In protecting commercial interests, our consuls have necessarily had to encounter increased labors and responsibilities, growing out of the war. These they have, for the most part, met and discharged with zeal and efficiency. This acknowledgment justly includes those consuls who, residing in Morocco, Egypt, Turkey, Japan, China, and other Oriental countries, are charged with complex functions and extraordinary powers.

The condition of the several organized Territories is generally satisfactory, although Indian disturbances in New Mexico have not been en-tirely suppressed. The mineral resources of Colorado, Nevada, Idaho, New Mexico, and Arizona, are proving far richer than has been heretofore understood. I lay before you a communication on this subject from the Governor of New Mexico. I again submit to your considaration the expediency of establishing a system for the encouragement of immigration. Although this source of national wealth and strength is again flowing with greater freedom than for several years before the insurrection occurred, there is still a great deficiency of laborers in every field of industry, especially in agriculture and in our mines, as well of iron and coal as of the precious metals. While the demand for labor is much increased here, tens of thousands of persons, destitute of remunerative occupation, are thronging our foreign consulates, and offering to emigrate to the United States if essential, but very cheap, assistance can be afforded them. It is easy to see that, under the sharp discipline of civil war, the nation is beginning a new life. This noble effort de-mands the aid, and ought to receive the attention and support of the Government.

Injuries, unforeseen by the Government and unintended, may, in some cases, have been inflicted on the subjects or citizens of foreign countries, both at sea and on land, by persons in the service of the United States. As this Government expects redress from other Powers when similar injuries are inflicted by persons in their service upon citizens of the United States, we must be prepared to do justice to foreigners. If the existing judicial tribunals are inadequate to this purpose, a special court may be authorized, with power to hear and decide such claims of the character referred to as may have arisen under treaties and the public law. Conventions for adjusting the claims by joint commission have been proposed to some Governments, but no definite answer to the proposition has yet been received from any.

In the course of the session I shall probably have occasion to request you to provide indemnification to claimants where decrees of restitution have been rendered, and damages awarded by admiralty courts; and in other cases, where this Government may be acknowledged to be liable in principle, and where the amount of that liability has been ascertained by an informal arbitration.

The proper officers of the Treasury have deemed themselves required by the law of the United States upon the subject, to demand a tax upon the incomes of foreign consuls in this While such a demand may not, in country. strictness, be in derogation of public law, or perhaps of any existing treaty between the United States and a foreign country, the expediency of so far modifying the act as to exempt from tax the income of such consuls as are not citizens of the United States, derived from the emoluments of their office, or from property not situated in the United States, is submitted to your serious consideration. I make this suggestion upon the ground that a comity which ought to be reciprocated exempts our consuls, in all other countries, from taxation to the extent thus indicated. The United States, I think, ought not to be exceptionably illiberal to international trade and commerce.

The operations of the Treasury during the last year have been successfully conducted. The enactment by Congress of a national banking law has proved a valuable support of the public credit; and the general legislation in relation to loans has fully answered the expectations of its favorers. Some amendments may be required to perfect existing laws, but no change in their principles or general scope is believed to be needed.

Since these measures have been in operation, all demands on the Treasury, including the pay of the Army and Navy, have been promptly met and fully satisfied. No considerable body of troops, it is believed, were ever more amply provided, and more liberally and punctually paid; and it may be added, that by no people were the burdens incident to a great war ever more cheerfully borne.

The receipts during the year from all sources, including loans and balance in the Treasury at its commencement, were \$901,125,674 86, and the aggregate disbursements \$895,796,630 65, leaving a balance on the 1st of July, 1863, of \$5,329,044 21. Of the receipts there were derived from customs \$69,059,642 40; from internal revenue, \$37,640,787 95; from direct tax, \$1,485,103 61; from lands, \$167,617 17; from miscellaneous sources, \$3,046,615 35; and from loans, \$776,682,361 57; making the aggregate, \$901,125,674 86. Of the disbursements there were for the civil service, \$23,-253,922 08; for pensions and Indians, \$4,216,-520 79; for interest on public debt, \$24,729,846 51; for the War Department, \$599,298,600 83; for the Navy Department, \$63,211,105 27; for payment of funded and temporary debt, \$181,- war itself.

086,635 07; making the aggregate, \$895,796,-630 65, and leaving the balance of \$5,329,044 21. But the payment of funded and temporary debt, having been made from moneys borrowed during the year, must be regarded as merely nominal payments, and the moneys borrowed to make them as merely nominal receipts; and their amount, \$181,086,635 07, should therefore be deducted both from receipts and disbursements. This being done, there remains as actual receipts \$720,039,039 79, and the actual disbursements \$714,709,995 58, leaving the balance as already stated.

The actual receipts and disbursements for the first quarter, and the estimated receipts and disbursements for the remaining three quarters of the current fiscal year, 1864, will be shown in detail by the report of the Secretary of the Treasury, to which I invite your attention. It is sufficient to say here that it is not believed that actual results will exhibit a state of the finances less favorable to the country than the estimates of that officer heretofore submitted; while it is confidently expected that at the close of the year both disbursements and debt will be found very considerably less than has been anticipated.

The report of the Secretary of War is a document of great interest. It consists of--

1. The military operations of the year, detailed in the report of the General-in-Chief.

2. The organization of colored persons into the war service.

3. The exchange of prisoners, fully set forth in the letter of Gen. Hitchcock.

4. The operations under the act for enrolling and calling out the national forces, detailed in the report of the Provost Marshal General.

5. The organization of the invalid corps; and 6. The operation of the several departments of the Quartermaster General, Commissary General, Paymaster General, Chief of Engineers, Chief of Ordnance, and Surgeon General.

It has appeared impossible to make a valuable summary of this report except such as would be too extended for this place, and hence I content myself by asking your careful attention to the report itself.

The duties devolving on the naval branch of the service during the year, and throughout the whole of this unhappy contest, have been discharged with fidelity and eminent success. The extensive blockade has been constantly increasing in efficiency, as the navy has expanded; yet on so long a line it has so far been impossible to entirely suppress illicit trade. From returns received at the Navy Department it appears that more than one thousand vessels have been captured since the blockade was instituted, and that the value of prizes already sent in for adjudication amounts to over thirteen millions of dollars.

The naval force of the United States consists at this time of five hundred and eighty-eight vessels, completed and in the course of completion, and of these seventy-five are iron-elad or armored steamers. The events of the war give an increased interest and importance to the Navy which will probably extend beyond the war itself.

The armored vessels in our Navy, completed | to the policy of fostering and training seamen, and in service, or which are under contract and approaching completion, are believed to exceed in number those of any other Power. But while these may be relied upon for harbor defence and coast service, others of greater strength and capacity will be necessary for cruising purposes, and to maintain our rightful position on the ocean.

The change that has taken place in naval vessels and naval warfare since the introduction of steam as a motive power for ships of war demands either a corresponding change in some of our existing navy-yards, or the establishment of new ones, for the construction and necessary repair of modern naval vessels. No inconsiderable embarrassment, delay, and public injury have been experienced from the want of such governmental establishments. The necessity of such a navy yard, so furnished, at some suitable place upon the Atlantic seaboard, has on repeated occasions been brought to the attention of Congress by the Navy Department, and is again presented in the report of the Secretary which accompanies this communication. т think it my duty to invite your special attention to this subject, and also to that of establishing a yard and depot for naval purposes upon one of the western rivers. A naval force has been created on those interior waters, and under many disadvantages, within little more than two years, exceeding in numbers the whole naval force of the country at the commencement of the present administration. Satisfactory and important as have been the performances of the heroic men of the Navy at this interesting period, they are scarcely more wonderful than the success of our mechanics and artisans in the production of warvessels which has created a new form of naval power.

Our country has advantages superior to any other nation in its resources of iron and timber, with inexhaustible quantities of fuel in the immediate vicinity of both, all available and in close proximity to navigable waters. Without the advantage of public works the resources of the nation have been developed and its power displayed in the construction of a navy of such magnitude which has, at the very period of its creation, rendered signal service to the Union.

The increase of the number of seamen in the public service, from seven thousand five hundred men, in the spring of 1861, to about thirtyfour thousand at the present time, has been accomplished without special legislation, or extraordinary bounties to promote that increase. It has been found, however, that the operation of the draft, with the high bounties paid for army recruits, is beginning to affect injuriously the naval service, and will, if not corrected, be likely to impair its efficiency, by detaching seamen from their proper vocation and inducing them to enter the army. I therefore respectfully suggest that Congress might aid both the army and naval service by a definite provision on this subject, which would at the same time be equitable to the communities more especially interested.

I commend to your consideration the suggestions of the Secretary of the Navy in regard interest in the early settlement and substantial

and also the education of officers and engineers for the naval service. The Naval Academy is rendering signal service in preparing midshipmen for the highly responsible duties which in after life they will be required to perform. In order that the country should not be deprived of the proper quota of educated officers, for which legal provision has been made at the Naval School, the vacancies caused by the neglect or omission to make nominations from the States in insurrection have been filled by the Secretary of the Navy. The school is now more full and complete than at any former period, and in every respect entitled to the favorable consideration of Congress.

During the past fiscal year the financial condition of the Post Office Department has been one of increasing prosperity, and I am gratified in being able to state that the actual postal revenue has nearly equalled the entire expenditures; the latter amounting to \$11,314,206 84, and the former to \$11,163,789 59, leaving a deficiency of but \$160,417 25. In 1860, the year immediately preceding the rebellion, the deficiency amounted to \$5,656,705 49, the postal receipts of that year being \$2,645,722-19 less than those of 1863. The decrease since 1860 in the annual amount of transportation has been only about 25 per cent., but the annual expenditure on account of the same has been reduced 35 per cent. It is manifest, therefore, that the Post Office Department may become self-sustaining in a few years, even with the restoration of the whole service.

The international conference of postal delegates from the principal countries of Europe and America, which was called at the suggestion of the Postmaster General, met at Paris on the 11th of May last, and concluded its deliberations on the 8th of June. The principles established by the conference as best adapted to facilitate postal intercourse between nations, and as the basis of future postal conventions, inaugurate a general system of uniform international charges, at reduced rates of postage, and cannot fail to produce beneficial results.

I refer you to the report of the Secretary of the Interior, which is herewith laid before you, for useful and varied information in relation to the public lands, Indian affairs, patents, pensions, and other matters of public concern pertaining to this department.

The quantity of land disposed of during the last and the first quarter of the present fiscal years was three million eight hundred and forty-one thousand five hundred and forty-nine acres, of which one hundred and sixty-one thousand nine hundred and eleven acres were sold for cash, one million four hundred and fifty-six thousand five hundred and fourteen acres were taken up under the homestead law, and the residue disposed of under laws granting lands for military bounties, for railroad and other purposes. It also appears that the sale of the public lands is largely on the increase.

It has long been a cherished opinion of some of our wisest statesmen that the people of the United States had a higher and more enduring

cultivation of the public lands than in the amount of direct revenue to be derived from the sale of them. This opinion has had a controlling influence in shaping legislation upon the subject of our national domain. I may cite, as evidence of this, the liberal measures adopted in reference to actual settlers; the grant to the States of the overflowed lands within their limits in order to their being reclaimed and rendered fit for cultivation; the grants to railway companies of alternate sections of land upon the contemplated lines of their roads which, when completed, will so largely multiply the facilities for reaching our distant possessions. This policy has received its most signal and beneficent illustration in the recent enactment granting homesteads to actual settlers. Since the first day of January last the beforementioned quantity of one million four hundred and fifty-six thousand five hundred and fourteen acres of land have been taken up under its provisions. This fact and the amount of sales, furnish gratifying evidence of increasing settlement upon the public lands, notwithstanding the great struggle in which the energies of the nation have been engaged, and which has required so large a withdrawal of our citizens from their accus-I cordially concur in the tomed pursuits. recommendation of the Secretary of the Interior, suggesting a modification of the act in favor of those engaged in the military and naval service of the United States. I doubt not that Congress will cheerfully adopt such measures as will, without essentially chang-ing the general features of the system, secure, to the greatest practicable extent, its benefits to those who have left their homes in the defence of the country in this arduous crisis.

I invite your attention to the views of the Secretary as to the propriety of raising, by appropriate legislation, a revenue from the mineral lands of the United States.

The measures provided at your last session for the removal of certain Indian tribes have been carried into effect. Sundry treaties have been negotiated, which will, in due time, be submitted for the constitutional action of the They contain stipulations for extin-Senate. guishing the possessory rights of the Indians to large and valuable tracts of land. It is hoped that the effect of these treaties will result in the establishment of permanent friendly relations with such of these tribes as have been brought into frequent and bloody collision with our outlying settlements and emigrants. Sound policy and our imperative duty to these wards of the Government demand our anxious and constant attention to their material well-being, to their progress in the arts of civilization, and above all, to that moral training which, under the blessing of Divine Providence, will confer upon them the elevated and sanctifying in-fluences, the hopes and consolations of the Christian faith

I suggested in my last annual message the propriety of remodelling our Indian system. It should, the crisis of the contest would then Subsequent events have satisfied me of its necessity. The details set forth in the report of 10

the Secretary evince the urgent need for immediate legislative action.

I commend the benevolent institutions established or patronized by the Government in this District to your generous and fostering care.

The attention of Congress, during the last session, was engaged to some extent with a proposition for enlarging the water communication between the Mississippi river and the northeastern seaboard, which proposition, however, failed for the time. Since then, upon a call of the greatest respectability, a convention has been held at Chicago upon the same subject, a summary of whose views is contained in a memorial addressed to the President and Congress, and which I now have the honor to lay before you. That this interest is one which, ere long, will force its own way, I do not entertain a doubt, while it is submitted entirely to your wisdom as to what can be done now. Augmented interest is given to this subject by the actual commencement of the work on the Pacific railroad, under auspices so favorable to rapid progress and completion. The enlarged navigation becomes a palpable need to the great road.

I transmit the second annual report of the Commissioner of the Department of Agriculture, asking your attention to the developments in that vital interest of the nation.

When Congress assembled a year ago the war had already lasted nearly twenty months, and there had been many conflicts on both land and sea with varying results. The rebellion had been pressed back into reduced limits; yet the tone of public feeling and opinion, at home and abroad, was not satisfactory. With other signs, the popular elections, then just past, indicated uncasiness among ourselves, while amid much that was cold and menacing, the kindest words coming from Europe were uttered in accents of pity, that we were too blind to surrender a hopeless cause. Our commerce was suffering greatly by a few armed vessels built upon and furnished from foreign shores, and we were threatened with such additions from the same quarter as would sweep our trade from the sea and raise our blockade. We had failed to elicit from European Governments anything hopeful upon this subject. The preliminary emancipa-tion proclamation, issued in September, was running its assigned period to the beginning of the new year. A month later the final proclamation came, including the announcement that colored men of suitable condition would be received into the war service. The policy of emancipation, and of employing black soldiers, gave to the future a new aspect, about which hope, and fear, and doubt contended in uncertain conflict. According to our political system, as a matter of civil administration, the General Government had no lawful power to effect emancipation in any State, and for a long time it had been hoped that the rebellion could be suppressed without resorting to it as a military measure. It was all the while deemed possible that the necessity for it might come, and that if it should, the crisis of the contest would then be presented. It came, and, as was anticipated,

Eleven months having now passed, we are permitted to take another view. The rebel borders are pressed still further back, and by the complete opening of the Mississippi the country dominated by the rebellion is divided into distinct parts, with no practical communication between them. Tennessee and Arkansas have been substantially cleared of insurgent control, and influential citizens in each, owners of slaves and advocates of slavery at the beginning of the rebellion, now declare openly for emancipation in their respective States. Of those States not included in the emancipation proclamation, Maryland and Missouri, neither of which three years ago would tolerate any restraint upon the extension of slavery into new territories, only dispute now as to the best mode of removing it within their own limits.

Of those who were slaves at the beginning of the rebellion, full one hundred thousand are now in the United States military service, about one half of which number actually bear arms in the ranks; thus giving the double advantage of taking so much labor from the insurgent cause, and supplying the places which otherwise must be filled with so many white men. So far as tested, it is difficult to say they are not as good soldiers as any. No servile insurrection, or tendency to violence or cruelty, has marked the measures of emancipation and arming the blacks. These measures have been much discussed in foreign countries, and contemporary with such discussion the tone of public sentiment there is much improved. At home the same measures have been fully discussed, supported, criticised, and denounced, and the annual elections following are highly encouraging to those whose official duty it is to bear the country through this great trial. Thus we have the new reckoning. The crisis which threatened to divide the friends of the Union is past.

Looking now to the present and future, and with reference to a resumption of the national authority within the States wherein that authority has been suspended, I have thought fit to issue a proclamation, a copy of which is herewith transmitted. On examination of this proclamation it will appear, as is believed, that nothing will be attempted beyond what is amply justified by the Constitution. True, the form of an oath is given, but no man is coerced to take it. The man is only promised a pardon in case he voluntarily takes the oath. The Constitution authorizes the Executive to grant or withhold the pardon at his own absolute discretion ; and this includes the power to grant on terms, as is fully established by judicial and other authorities.

It is also proffered that if, in any of the States named, a State government shall be, in the mode prescribed, set up, such government shall be recognized and guarantied by the United States, and that under it the State shall, on constitutional conditions, be protected against invasion and domestic violence. The constitutional obligation of the United States to guaranty to every State in the Union a republican form of government, and to protect the State, in the cases stated, is explicit and full. But why tender the benefits of this provision only to a State government set up in this particular way? This section of the Constitution contemplates a case wherein the element within a State, favorable to republican government, in the Union, may be too feeble for an opposite and hostile element external to or even within the State; and such are precisely the cases with which we are now dealing.

An attempt to guaranty and protect a revived State government, constructed in whole, or in preponderating part, from the very element against whose hostility and violence it is to be protected, is simply absurd. There must be a test by which to separate the opposing elements so as to build only from the sound; and that test is a sufficiently liberal one which accepts as sound whoever will make a sworn recantation of his former unsoundness.

But if it be proper to require, as a test of admission to the political body, an oath of allegiance to the Constitution of the United States, and to the Union under it, why also to the laws and proclamations in regard to slavery? Those laws and proclamations were enacted and put forth for the purpose of aiding in the suppression of the rebellion. To give them their fullest effect, there had to be a pledge for their maintenance. In my judgment they have aided, and will further aid, the cause for which they were intended. To now abandon them would be not only to relinquish a lever of power, but would also be a cruel and an astounding breach of faith. I may add at this point, that while I remain in my present position I shall not attempt to retract or modify the emancipation proclamation; nor shall I return to slavery any person who is free by the terms of that proclamation, or by any of the acts of Congress. For these and other reasons it is thought best that support of these measures shall be included in the oath ; and it is believed the Executive may lawfully claim it in return for pardon and restoration of forfeited rights, which he has clear constitutional power to withhold altogether, or grant upon the terms which he shall deem wisest for the public interest. It should be observed, also, that this part of the oath is subject to the modifying and abrogating power of legislation and supreme judicial decision.

The proposed acquiescence of the national Executive in any reasonable temporary State arrangement for the freed people is made with the view of possibly modifying the confusion and destitution which must at best, attend all classes by a total revolution of labor throughout whole States. It is hoped that the already deeply afflicted people in those States may be somewhat more ready to give up the cause of their affliction, if, to this extent, this vital matter be left to themselves; while no power of the national Executive to prevent an abuse is abridged by the proposition.

The suggestion in the proclamation as to maintaining the political frame-work of the States on what is called reconstruction, is made in the hope that it may do good without danger of harm. It will save labor, and avoid great confusion.

But why any proclamation now upon the suo-

ject? This question is beset with the conflicting views that the step might be delayed too long or be taken too soon. In some States the elements for resumption seem ready for action, but remain inactive, apparently for want of a rallying point-a plan of action. Why shall A adopt the plan of B, rather than B that of A? And if A and B should agree, how can they know but that the General Government here will reject their plan? By the proclamation a plan is presented which may be accepted by them as a rallying point, and which they are assured in advance will not be rejected here. This may bring them to act sooner than they otherwise would.

The objection to a premature presentation of a plan by the national Executive consists in the danger of committals on points which could be more safely left to further developments. Care has been taken to so shape the document as to avoid embarrassments from this source. Saying that, on certain terms, certain classes will be pardoned, with rights restored, it is not said that other classes, or other terms, will never be included. Saying that reconstruction will be accepted if presented in a specified way, it is not said it will never be accepted in any other W9.V

The movements, by State action, for emancipation in several of the States, not included in the emancipation proclamation, are matters of profound gratulation. And while I do not repeat in detail what I have heretofore so earnestly urged upon this subject, my general views and feelings remain unchanged; and I trust that Congress will omit no fair opportunity of aiding these important steps to a great consummation.

In the midst of other cares, however important, we must not lose sight of the fact that the war power is still our main reliance. To that power alone we can look, yet for a time, to give confidence to the people in the contested regions, that the insurgent power will not again overrun them. Until that confidence shall be established, little can be done anywhere for what is called reconstruction. Hence our chiefest care must still be directed to the army and navy, who have thus far borne their harder part so nobly and well. And it may be esteemed fortunate that in giving the greatest efficiency to these indispensable arms, we do also honorably recognize the gallant men, from commander to sentinel, who compose them, and to whom, more than to others, the world must stand indebted for the home of freedom disenthralled, regenerated, enlarged, and perpetuated. **ABRAHAM LINCOLN.** 

December 8, 1863.

#### PROCLAMATION OF AMNESTY.

Whereas, in and by the Constitution of the United States. whereas in and by the constitution of the United States, it is provided that the President "shall have power to grant reprives and pardons for offenses against the United States, except in cases of impeachment;" and whereas a robellion now exists whereby the loyal State governments of several States have for a long time been subverted, and many persons have for a long time been subverted, and many per-sons have committed and are new guilty of treason against the United States; and whereas, with reference to said rehel-lion and treason, laws have been enacted by Congress de-claring forfeitures and confiscation of property and liberation of slaves, all upon terms and conditions therein stated, and also declaring that the President was thereby authorized at

any time thereafter, by proclamation, to extend to persons who may have participated in the existing rebellion, in any State or part thereof, pardon and annesty, with such excep-tions and at such times and on such conditions as he may deem expedient for the public welfare; and whereas the congressional declaration for limited and conditional pardon accords with well established judicial exposition of the pardon-ing power; and whereas, with reference to said rebellion, the President of the United States has issued several proclama-tions with manifesting in several to the the state of the several proclamations, with provisions in regard to the liberation of slaves; and whereas it is now desired by some persons heretofore engaged in said rebellion to resume their allegiance to the United States, and to reinaugurate loyal State governments within and for their respective States:

Therefore, I, ABRAHAM LINCOLN, President of the United States, do proclaim, declare, and make known to all persons States, do proclaim, declare, and make known to all persons who have directly, or by implication, participated in the ex-isting rebellion, except as hereinafter excepted, that a full pardon is hereby granted to them and each of them, with restoration of all rights of property, except as to slaves, and in property cases where rights of third parties shall have in-tervened, and upon the condition that every such person shall take and subscribe an oath, and thenceforward keep and maintain said eath inviolate; and which eath shall be registered for nermanent preservation and shall be of the

and maintain sub occurring the when own sub ow the union of the States thereunder; and that I will, in like manner, abide by and faithfully supportall acts of Congress passed during the existing rebellion with reference to slaves, so long and so far as not repealed, modified, or held yoid by Congress, or by decision of the Supreme Court; and that I will, in like manner, abide by and faithfully support all proclamations of the President made during the existing rebellion having reference to slaves, so long and so far as not modified or declared void by decision of the Supreme Court. So help me God."

The persons excepted from the benefits of the foregoing provisions are all who are, or shall have been, civil or diplo-matic officers or agents of the so-called Confederate Government; all who have left judicial stations under the United ment; all who have left judicial stations under the United States to aid the rebellion; all who are, or shall have been, military or naval officers of said so-called Confederate Gov-ernment above the rank of colonel in the Army, or of lien-tenaut in the Nary; all who left seats in the United States Congress to aid the rebellion; all who resigned commissions in the Army or Navy of the United States, and afterwards aided the rebellion; and all who have engaged in any way in treating colored persons, or white persons in charge of such otherwise than lawfull yas prisoners of war, and which persons may have been found in the United States service as soldiers, seamen, or in any other capacity.

persons may have been tound in the United States service as soldiers, esamen, or in any other capacity. And I do further proclaim, declare, and make known that whenever in any of the States of Arkansas, Texas, Louisi-ana, Mississippi, Tennessee, Alabama, Georgia, Virginia, Florida, South Carolina, and North Carolina, a number of persons, not less than one tenth in number of the votes cast in such State at the presidential election of the year of our Lord one thousand eight hundred and sixty, each having taken the oath aforesaid and not having since violated it, and heing a gualified voter by the election law of the State and being a qualified voter by the election law of the State existing immediately before the so-called act of secession, and excluding all others, shall re-establish a State govern-ment which shall be republican, and in nowise contravening said oath, such shall be recognized as the true government of the State, and the State shall receive thereunder the benefits of the constitutional provision which declares that "the United States shall guaranty to every State in this Union a republican form of government, and shall protect

Union a republican form of government, and shall protect each of them against invasion; and, on application of the Legislature, or the Executive, (when the Legislature can-not be convened,) against domestic violence." And I do further proclaim, declare, and make known that any provision which may be adopted by such State government in relation to the freed people of such State, which shall recognize and declare their permanent free-dom, provide for their education, and which may yet bo consistent as a fermorary arrangement with their resent dom, provide for their education, and which may yet be consistent, as a temporary arrangement, with their present condition as a laboring, landless, and homeless class, will not be objected to by the national Executive. And it is suggested as not improper, that, in constructing a loyal State government in any State, the name of the State, the boundary, the subdivisions, the constitution, and the gen-eral code of laws as before the rabeling the maintined boundary, the subavisions, the constitution, and the gen-eral code of laws, as before the rebellion, be maintained, subject only to the modifications made necessary by the conditions hereinbefore stated, and such others, if any, not contravening said conditions, and which may be deemed expedient by those framing the new State govern-

have all the while been maintained. And for the same reason, it may be proper to further say, that whether members sent to Congress from any State shall be ad-mitted to seats constitutionally, rests exclusively with the respective Houses, and not to any extent with the Executive. And still further, that this proclamation is intended to present the people of the States wherein the national authority has been suspended, and loyal State governments have been subverted, a mode in and by which the national and loyal State governments may be re-established within said States, or in any of them; and, while the mode prescnted is the best the Executive can suggest, with his pres-ent impressions, it must not be understood that no other possible mode would be acceptable.

Given under my hand, at the City of Washington, the eighth day of December, A. D. one thousand eight hun-dred and sixty-three, and of the independence of the United States of America the eighty-eighth.

ABRAHAM LINCOLN.

By the President: WILLIAM H. SEWARD, Secretary of State.

#### AMNESTY DEFINED.

### By the President of the United States-A Proclamation.

Whereas it has become necessary to define the cases in which insurgent enemies are entitled to the benefits of the Proclamation of the President of the United States, which was made on the 8th day of December, 1863, and the manner in which they shall proceed to avail themselves of those ben-efits: and whereas the objects of that proclamation were to suppress the insurrection, and to restore the authority of the United States: and whereas the amnesty therein provided by the President was offered with reference to these objects alone:

Now, therefore, I, ABRAHAM LINCOLN, President of the United States, do hereby proclaim and declare that the said proclamation does not apply to the cases of persons who, at the time when they seek to obtain the benefits thereof by taking the oath thereby prescribed, are in military, naval, or civil confinement or custody, or under bonds or on parole of the civil military, or naval authorities, or agents of the United States as prisoners of war, or persons detained for offences of any kind, either before or after conviction; and that, on the contrary, it does apply only to persons who, being yet at large and free from any arrest, confinement, or duress, shall voluntarily come forward and take the said oath with the purpose of restoring peace and establishing the national authority.

Prisoners excluded from the amnesty offered in the said proclamation may apply to the President for clemency, like all other offenders, and their application will receive due consideration.

I do further declare and proclaim that the oath prescribed in the aforesaid proclamation of the 8th of December, 1863, may be taken and subscribed to before any commissioned officer, civil, military, or naval, in the service of the United States, or any civil or military officer of a State or Territory, not in insurrection, who by the law thereof may be qualified for administering oaths.

All officers who receive such oaths are hereby authorized to give certificates thereon to the persons respectively by whom they are made. And such officers are hereby required to transmit the original records of such oaths at as early a day as may be convenient to the Department of State, where they will be deposited and remain in the archives of the Government.

The Secretary of State will keep a register thereof, and

The Secretary of State will keep a register thereof, and will, on application in proper cases, issue certificates of such records in the customary form of such certificates. In testimony whereof I have hereunto set my hand, and caused the sell of the United States to be affixed. Done at the city of Washington, the 26th day of March, in the year of our Lord one thousand eight hundred and sixty-four, and of the independence of the United States the eighty-sciebth. eighth. ABRAHAM LINCOLN.

By the President: WILLIAM H. SEWARD, Secretary of State.

#### CIRCULAR OF THE ATTORNEY GENERAL.

to slaves and where rights of third parties have intervened, which that proclamation offers and secures.

The President's pardon of a person guilty of acts of rebel-lion will of course relieve that person from the penalties incurred by his crime, and, where an indictment is pending against him therefor, the production of the pardon signed by the President, or of satisfactory evidence that he has complied with the conditions on which the pardon is offered, (if he be not of the class excepted from the benefits of the proclamation,) will be a sufficient reason for discontinuing such criminal proceedings, and discharging him from custody therein.

Nor is it less doubtful that a bona fide acceptance of the terms of the President's Proclamation, by persons guilty of acts of rebellion, and not of the excepted class, will secure to such persons a restoration of all the rights of property, to such persons a resonation of an the rights of third parties except as to slaves and where the rights of third parties shall have intervened, notwithstanding such property may, by reason of those acts of rebellion, have been subject to confiscation under the provisions of the Confiscation act of of 6th August, 1861, chap. 60, and 17th July, 1862, chap. 195. For, without adverting to any other source of power in the President to restore or protect their rights of prop-erty, the 13th section of the act of 17th of July, 1862, authorizes the President at any time thereafter, by proclamation, to extend to persons who may have participated in the existing rebellion in any State or part thereof, pardon and amnesty, with such exceptions, and at such time and on such conditions, as he may deem expedient for the public welfare. It will hardly be questioned, I suppose, that the purpose of this section, inserted in a law mainly intended to reach the property of persons engaged in rebellion, was to vest the President with power to relieve such persons, on such conditions as he should prescribe, from the penalty of loss of their property by confiscation. Although the pro-ceedings for confiscation under the acts of Aug. 6, 1861, and July 17, 1862, are *in rem*, against the property seized, yet, under both acts, the ground of condemnation is the personal guilt of the owner in aiding the rebeilion. By the sonal guilt of the owner in aiding the rebellion. By the pardon and annesty, not only is the punishment of that personal guilt remitted, but the offence itself is effaced, that being the special effect of an act of annesty by the Government. Of course it arrests and puts an end to all penal proceedings founded thereon, whether they touch the persons or the property of the offender.

There is, therefore, no case of judicial proceedings to enforce the penalties of acts of rebellion which cannot be reached and cured by the constitutional or statutory power of the President to grant pardon and amnesty, whether those proceedings be against the person of the offender by criminal indictment or against his property under the confiscation acts referred to.

The President has accordingly directed me to instruct you that, in any case where proceedings have been commenced and are pending and undetermined in the District or Circuit Court of the United States for your district against a person charged with acts of rebellion, and not of the excepted class, whether they be by indictment or by seizure and libel of his whether they be by indictment or by seizure and libel of his property for confiscation, (the rights of other parties not having intervened,) you will discontinue and put an end to those proceedings, whenever the person so charged shall produce evidence satisfactory to you that he has, in good faith, taken the oath and complied with the conditions pre-scribed by the President's proclamation of the 8th December, 1863. Nor is it necessary that the evidence which he pro-duces should be a deed of pardon signed by the President.

duces should be a deed of pardon signed by the President. It would be quite impossible for the President to furnish the multitudes who are now availing themselves of the bene-This of the Proclamation, and who are likely to do so here-after, with his formal evidence of pardon. It will be suff-cient to justify your action if the party seeking to be relieved from further proceedings shall prove to your tull satisfaction that he has, in good faith, taken the oath and brought himself within the conditions of pardon and amnesty set forth in the Proclamation.

In the Proclamation. If, in any case, you have good reason to believe that the oath has been taken for the mere purpose of obtaining the possession of personal property seized under the Confisca-tion acts, with intent to remove it from the subsequent reach of the officers of the law, you will make report of the facts and reasons for your belief to this office before discontinuing the second lines or neutring many hyperperiod. WASHINGTON, February 19, 1864—The follow-ing important circular letter has been ad-dressed to United States District Attornies: Sn: Many persons, against whom criminal indictments, or against whose property proceedings under the confisca-tion laws are pending in the courts of the United States, string rebellion, have, in good faith, taken the oath escribed by the proclamation of the President's of Sth Decem-seribed by the proclamation of all rights of property, except as are always officient instruments and legitimate effects of public war.

But although the remission of forfeitures under the act of July, 1861, are thus not within the scope of the Proclamation of pardon, still ample power is conferred on the Secretary of the Treasury by the eighth section of that act to mitigate or remit all forfeitures and penalties incurred under the act. And it is not to be doubted that in all proper cases under And to is not to be doubted that in an proper cases tinder that act, where the owner of the property residing in the territory in rebellion, complies with the conditions of the Proclamation, the Sceretary of the Treasury will exer-cise the power of remission of such forfcitures in the same spirit of generous forbearance and liberality which inspired and characterizes the Proclamation

TITIAN J. COFFEY, Very respectfully, &c., Acting Attorncy General. To -----, United States District Attorney at -

# Other Proclamations of the President.

## THE BLOCKADE.

Whereas an insurrection against the Government of the United States has broken out in the States of South Carolina, Georgia, Alabama, Florida, Mississippi, Louisiana, and Texas, and the laws of the United States for the collection of the revenue cannot be effectually executed therein con-formably to that provision of the Constitution which re-quires duties to be uniform throughout the United States:

And whereas a combination of persons engaged in such insurrection have threatened to grant pretended letters of marque to authorize the bearers thereof to commit assaults on the lives, vessels, and property of good citizens of the country lawfully engaged in commerce on the high seas, and in waters of the United States:

And whereas an executive proclamation has been already issued, requiring the persons engaged in these disorderly proceedings to desist therefrom, calling out a militia force for the purpose of repressing the same, and convening Con-gress in extraordinary session to deliberate and determine thereon:

Now, therefore, I, ABRAHAM LINCOLN, President of the United States, with a view to the same purposes before men-tioned, and to the protection of the public peace, and the lives and property of quict and orderly citizens pursting their lawful occupations, until Congress shall have assem-bled and dollbursted on the geid unlowful uncoechings of bled and deliberated on the said unlawful proceedings, or until the same shall have ceased, have further deemed it advisable to set on foot a blockade of the ports within the States afore to set on hoot a blockade of the ports within the States aforesaid, in pursuance of the laws of the United States and of the law of nations in such case provided. For this purpose a competent force will be posted so as to prevent entrance and exit of vessels from the ports afore-said. If, therefore, with a view to violate such blockade, a vessel shall approach, or shall attempt to leave either of the vessel shall approach, or shall attempt to leave either of the said ports, she will be duly warned by the commander of one of the blockading vessels, who will indorse on her re-gister the fact and date of such warning; and if the same vessel shall again attempt to enter or leave the blockaded port, she will be captured and sent to the nearest conveni-ent port for upon preceding to reach one or nearest ent port, for such proceedings against her and her cargo as prize as may be deemed advisable.

And I hereby proclaim and declare that if any person, under the pretended authority of the said States, or under any other pretence, shall molest a vessel of the United States, or the persons or cargo on board of her, such person will be held amenable to the laws of the United States for the prevention and punishment of piracy. ABRAHAM LINCOLN.

By the President:

WILLIAM H. SEWARD, Secretary of State. WASHINGTON, April 19, 1861.

## OF VIRGINIA AND NORTH CAROLINA.

On the 27th of April, the following additional proclamation, extending the blockade, was issued:

Whereas, for the reasons assigned in my proclamation of the 19th instant, a blockade of the ports of the States of South Carolina, Georgia, Florida, Alabama, Louisiana, Missis-sippi, and Texas, was ordered to be established: and whereas, since that date, public property of the United States has been seized, the collection of the revenue obstructed, and duly commissioned officers of the United States, while engaged in executing the orders of their superiors, have been arrested and held in custody as prisoners, or have been impeded in the discharge of their official duties, without due legal process, by persons claiming to act under authority of the States of Virginia and North Carolina : An efficient blockade of the

ports of those States will also be established. In witness whereof, I have hereunto set my hand, and caused the seal of the United States to be affixed.

Done at the city of Washington, this 27th day of April, in the year of our Lord 1861, and of the independence of the United States the eighty-fifth. ABRAHAM LINCOLN.

By the President: WILLIAM H. SEWARD, Secretary of State.

1862, May 12-The PRESIDENT issued a proclamation opening the ports of Beaufort, Port Royal, and New Orleans.

1863, September 24-The PRESIDENT issued a proclamation opening the port of Alexandria, Virginia.

1864, February 18-The PRESIDENT issued a proclamation opening the port of Brownsville, Texas.

#### COMMERCIAL INTERCOURSE WITH THE REBELLIOUS STATES PROHIBITED

Whereas, on the 15th day of April, 1861, the President of the United States, in view of an insurrection against the laws, Constitution, and Government of the United States, which had broken out within the States of South Carolina, Georgia, Alabama, Florida, Mississippi, Louisiana, and Texas, and in pursuance of the provisions of the act entitled "An act to provide for calling forth the militia to execute the laws of the Union, suppress insurrections, and repel invasions, and to repeal the act now in force for that purpose," approved February 28, 1795, did call forth the militia to suppress said insurrection and to cause the laws of the Union to be duly executed, and the insurgents have failed to disperse by the time directed by the President; and whereas such insurrection has since broken out and yet exists within the States of Virginia, North Carolina, Tennessee, and Arkansas; and whereas the insurgents in all the said States claim to act under the authority thereof, and such claim is not disclaimed or repudiated by the persons exercising the functions of gov-ernment in such State or States, or in the part or parts thereof, in which such combinations exist, nor has such

Insurrection been suppressed by said States: Now, therefore, I, ABRAHAM LINCOLN, President of the United States, in pursuance of the act of Congress approved July 13, 1861, do hereby declare that the inhabitants of the said States of Georgia, South Carolina, Virginia, North Caro-lina, Tennessee, Alabama, Louisiana, Texas, Arkanasa, Mis-fesimin and Floride (corone the inhabitants of the next of sissipi, and Florida, (except the inhabitants of that part of the State of Virginia lying west of the Alleghany Moun-tains, and of such other parts of that State and the other States hereinbefore named as may maintain a loyal adhe-sion to the Union and the Constitution, or may be from time to time accurate and the formation of the former of the time to time occupied and controlled by the forces of the United States engaged in the dispersion of said insurgents) are in a state of insurrection against the United States, and that all commercial intercourse between the same and the inhabitants thereof, with the exceptions aforesaid, and the citizens of other States and other parts of the United States, is unlawful and will remain unlawful until such insurrection shall case or has been suppressed; that all goods and chattels, wares and merchandise, coming from any of said States, with the exceptions aforesaid, into other parts of the States, with the exceptions aforesaid, into other parts or one United States, without the special license and permission of the President, through the Scoretary of the Treasury, or pro-ceeding to any of said States, with the exceptions aforesaid, by land or water, together with the vessel or vehicle con-veying the same or conveying persons to or from the said States, with said exceptions, will be forfeited to the United States; and that, from and after fifteen days from the issu-ice of this predemention all shins and vessels belonging, in ing of this proclamation, all ships and vessels belonging, in whole or in part, to any citizen or inhabitant of any of said States, with said exceptions, found at sea or in any port of the United States, will be forfeited to the United States; and I hereby enjoin upon all district attorneys, States; and I hereby enjoin upon all district attorneys, marshals, and officers of the revenue and of the military and naval forces of the United States, to be vigilant in the exe-cution of said act, and in the onforcement of the penal-ties and forfeitures imposed or declared by it; leaving any party who may think himself aggrieved thereby to his ap-plication to the Secretary of the Treasury for the remission of any penalty or forfeiture, which the said Secretary is authorized by law to grant, if, in his judgment, the special circumstances of any case shall require such remission. In witness whereof L have heremoto set my hand and

In witness whercof I have hereunto set my hand, and caused the seal of the United States to be affixed.

Done at the city of Washington, this sixteenth day of Au-gust, in the year of our Lord one thousand cight hundred and sixty-one, and of the independence of the United States ABRAHAM LINCOLN. of America the eighty-sixth. By the President:

WILLIAM H. SEWARD, Secretary of State.

1863, April 2-The PRESIDENT issued a proc-

forcement of said act of July 13, 1861, and the proper regu-lation of the commercial intercourse authorized by said act with the loyal citizens of said States:

Now, therefore, I, ABRAHAM LINCOLN, President of the United States, do hereby revoke the said exceptions, and declare that the inhabitants of the State Steeptions, and declare that the inhabitants of the States of Georgia, South Carolina, North Carolina, Tennessee, Alabama, Louisiana, Texas, Arkansas, Mississipi, Florida, and Virginia, (except the forty-eight counties of Virginia designated as West Virginia, and except, also, the ports of New Orleans, Key West, Port Royal, and Beaufort, in North Carolina,) are in a state of insurrection against the United States, and that all commercial intercourse, not licensed and conducted as provided in said act, between the said States and the inhabitants thereof, with the exceptions aforesaid, and the citizens of other States and other parts of the United States, is unlawful, and will remain unlawful until such insurrection shall cease or has been suppressed, and notice thereof has been duly given by proclamation; and all cotton, tobacco, and other products, and all other goods and chattels, wares, and merchandise coming from any of said States, with the exceptions aforesaid, into other parts of the United States, or proceeding to any of said States, with the exceptions afore-said, without the license and permission of the President, through the Secretary of the Treasury, will, together with the vessel or vehicle conveying the same, be fordited to the United States.

#### DECLARING BOUNDARIES OF THE INSURRECTION.

Whereas, in and by the second section of an act of Con-gress passed on the 7th day of June, A. D. 1862, entitled "An act for the collection of direct taxes in insurrectionary build of the collection of direct taxes in insurrectionary of the numbers." districts within the United States, and for other purposes it is made the duty of the President to declare, on or before the 1st day of July then next following, by his proclamation, in what States and parts of States insurrection exists:

Now, therefore, be it known that I, ABRAHAM LINCOLN, President of the United States of America, do hereby declare and proclaim that the States of South Carolina, Florida, Georgia, Alabama, Louisiana, Texas, Mississippi, Arkansas, Tennessee, North Carolina, and the State of Virginia, except Telnessee, North Carolina, and the State of Wighing except the following counties: Hancock, Brooke, Ohio, Marshall, Wetzel, Marion, Monongalia, Preston, Taylor, Pleasants, Tyler, Ritchie, Doddridge, Harrison, Wood, Jackson, Wirt, Roane, Calhoun, Gilmer, Barbour, Tucker, Lewis, Braxton, Roane, Calnoun, Gilmer, Baroour, Jucker, Lewis, Braxton, Upshur, Randolph, Mason, Putnam, Kanawha, Clay, Nicho-las, Cabell, Wayne, Boone, Logan, Wyoming, Webster, Fay-ette, and Raleigh, are now in insurrection and rebellion, and by reason thereof the civil authority of the United States is obstructed so that the provisions of the "Act to provide in-creased revenue from imports, to pay the interest on the public debt, and for other purposes," approved August 5, 1861, cannot be peaceably executed, and that the taxes le-relly charge-ble purpone real estate nuder the act last afore. goaly chargeable upon real estate under that the taxes lo-gally chargeable upon real estate under the act last afore-said, lying within the States and parts of States as aforesaid, together with a penalty of fifty per centum of said taxes, shell be a lien upon the tracts or lots of the same, severally charged, till paid.

In witness whereof, I have hereunto set my hand, and caused the seal of the United States to be affixed.

Done at the city of Washington, this 1st day of July, in the year of our Lord one thousand eight hundred and sixty-two, and of the independence of the United States of America the ABRAHAM LINCOLN. eighty-sixth. By the President:

F. W. SEWARD, Acting Secretary of State.

### PROCEEDINGS OF CONGRESS.

## ACTS OF THE PRESIDENT APPROVED - FIRST SESSION, THIRTY-SEVENTH CONGRESS.

1861, August 5-Congress passed a bill, the third section of which was as follows:

That all the acts, proclamations, and orders of the Presi-dent of the United States after the 4th of March, 1861, respecting the army and navy of the United States, and calling out or relating to the militia or volunteers from the States, are hereby approved and in all respects legal-ized and made valid, to the same intent and with the same effect, as if they had been issued and done under the previous express authority and direction of the Congress of the United States.

In SENATE, the bill passed-yeas 33, nays 5, as follows:

YEAS-Messrs. Anthony, Baker, Browning, Carlile,

lamation modifying the above, and reciting his reasons therefor: And whereas experience has shown that the exceptions made in and by sud proclamation embarrass the due en-formement of red not a first 1861 ord the proper whereas the due en-summer, Ten Eyck, Trumbull, Wade, Wilkinson, Wilson-

NAVS-Messrs. Breckinridge, Kennedy, Polk, Powell, Saulsbury-5.

In House, the motion to strike from the bill the ratifying section, was lost-yeas 19, nays 74; as follows;

74; as follows; YEAS-Messrs. Allen, Ancona, George H. Browne, Cal-vert, Cox, Crisfield, Jackson, Johnson, May, Noble, Pendleton, James S. Rollins, Shiel, Smith, Yallandigham, Voorhees, Wadsworth, Ward, Webster-19. NAYS-Messrs. Aldrich, Alley, Arnold, Ashley, Gold-smith F. Bailey, Baxter, Beaman, Francis P. Blair, Samuel S. Blair, Blake, Buffnton, Cobb, Colfax, Frederick A. Conkling, Conway, Covode, Diven, Duell, Dunn, Edwards, Eliot, Fenton, Fessenden, Frank, Goodwin, Granger, Gur-ley, Haight, Hale, Hanchett, Harrison, Horton, Hutchins, Julian, Kelley, Francis W. Kellorgs, William Kellogs, Lan-sing, Leary, Loomis, Lovejoy, McKean, McKnight, Mc-Pherson, Mitchell, Moorhead, Anson P. Morrill, Justin S. Morrill, Olin, Pike, Porter, Alexander H. Rice, John H. Rice, Riddle, Edward H. Rollins, Sedgwick, Shanks, Sheffield, Shellabarger, Sherman, Stevens, Benjamin F. Thomas, Francis Thomas, Train, Trowbridge, Van Horn, Verree, Wall, Charles W. Walton, E. P. Walton, Albert S. Verree, Wall, Charles W. Walton White, Windom, Worcester-74.

The bill then passed without a division.

## African Slave Trade.

1862, April 7-A treaty was made between William H. Seward and Lord Lyons for the suppression of the African Slave Trade, which was ratified by the Senate, and announced by proclamation of the President, June 7, 1862, Article 1 of which is as follows:

The two high contracting parties mutually consent that those ships of their respective navies which shall be provided with special instructions for that purpose, as hereinafter mentioned, may visit such merchant vessels of the after mentioned, may visit such internative vesses of the two nations as may, upon reasonable grounds, be suspected of being engaged in the African slave trade, or of having been fitted out for that purpose; or of having, during the voyage on which they are met by the said cruisers, been engaged in the African slave trade, contrary to the provision of this treaty; and that such cruisers may detain, and send or carry away, such vessels, in order that they may be brought to trial in the manner hereinafter agreed upon.

In order to fix the reciprocal right of search in such a manner as shall be adapted to the attainment of the object of this treaty, and at the same time avoid doubts, disputes, and complaints, the said right of search shall be understood

in the manner and according to the rules following: First. It shall never be exercised except by vessels of war, authorized expressly for that object, according to the stipulations of this treaty.

Secondly. The right of search shall in no case be exercised with respect to a vessel of the navy of either of the two Powers, but shall be exercised only as regards merchant vessels; and it shall not be exercised by a vessel of war of either contracting party within the limits of a settlement or port, nor within the territorial waters of the other party.

Thirdly. Whenever a merchant vessel is searched by a ship of war, the commander of the said ship shall, in the act of so doing, exhibit to the commander of the merchant vessel the special instructions by which he is duly author-ized to search; and shall deliver to such commander a certificate, signed by himself, stating his rank in the naval service of his country, and the name of the vessel he commands, and also declaring that the only object of the search is to ascertain whether the vessel is employed in the African slave trade, or is fitted up for the said trade. When the search is made by an officer of the cruiser who is not the commander, such officer shall exhibit to the captain of the merchant vessel a copy of the before-mentioned special instructions, signed by the commander of the cruiser; and he shall in like manner deliver a certificate signed by himself, stating his rank in the navy, the name of the com-mander by whose orders he proceeds to make the search, that of the cruiser in which he sails, and the object of the search, as above described. If it appears from the search that the papers of the vessel are in regular order, and that it is employed on lawful objects, the officer shall enter in the log-book of the vessel that the search has been made in pursuance of the aforesaid special instructions; and the vessel shall be left at liberty to pursue its voyage. The rank of the officer who makes the search must not be less than that of licutenant in the navy, unless the command, either by reason of death or other cause, is at the time held by an officer of inferior rank.

Fourthly. The reciprocal right of search and detention routing, the reciprocal right of search and detention shall be exercised only within the distance of two hundred miles from the coast of Africa, and to the southward of the thirty-second parallel of north latitude, and within thirty leagues from the coast of the Island of Cuba.

1862, June 26-In Senate, the bill to carry into effect this treaty, by providing for the officials, &c., passed-yeas 34, nays 4, (Messrs. Carlile, Kennedy, Powell, Saulsbury.)

July 7-The bill passed the House without a division.

1863, February 17-The treaty was modified by the addition of an article authorizing the exercise of the reciprocal right of visit and detention within thirty leagues of the island of Madagascar, within thirty leagues of the island of Puerto Rico, and within thirty leagues of the island of San Domingo.

THE "CONFEDERATE" RECORD.

Soon after the adoption of their Constitution in 1861, the rebel Congress passed a bill in relation to the slave trade, which JEFFERSON DAvis returned, with his objections, as follows:

### EXECUTIVE DEPARTMENT February 28, 1861.

GENTLEMEN OF CONGRESS: With sincere deference to the judgment of Congress. I have carefully considered the bill in relation to the slave trade, and to punish persons offend-ing therein, but have not been able to approve it, and, therefore, do return it with a statement of my objections. The Constitution—section seven, article one—provides

that the importation of African negroes from any foreign country other than slaveholding States of the United States is hereby forbidden, and Congress is required to pass such laws as shall effectually prevent the same. The rule herein given is emphatic, and distinctly directs the legislation which black effectually prevent the income of the same set of which shall effectually prevent the importation of African negroes. The bill before medenounces as high misdemeanor the importation of African negroes, or other persons of color, either to be sold as slaves or to be held to service or color, either to be sold as slaves or to be held to service or labor, affixing heavy, degrading penalties on the act if door with such intent. To that extent it accords with the re-quirements of the Constitution, but in the sixth section of the bill provision is made for the transfer of persons who may have been illegally imported into the Confederate States to the custody of foreign States or societies, upon condition of deportation and future freedom, and, if the proposition thus to surrender them shall not be accepted, it is then made the duty of the President to cause said ne-groes to be sold at public outry to the highest bidder in any one of the States where such sale shall not be incom-sistent with the laws thereof. This provision seems to me to be in opposition to the policy declared in the Constitu-tion—the prohibition of the importation of African negroes —and in derogation of its mandate to legislate for the effec-tuation of that object. Wherefore the bill is returned to you for your further consideration, and, together with the objections, most respectfully submitted.

# JEFFERSON DAVIS.

This veto was sustained by the following vote -the question being, "Shall the bill pass notwithstanding the President's objections ?"

YEAS-Messrs. Curry and Chilton, of Alabama; Morton and Owens, of Florida; Toombs, H. Cobb, T. R. R. Cobb, Bartow, Nisbet, and Kenan, of Georgia; Rhett, Barnweil, Keitt, and Miles, of South Carolina; Ochiltree, of Texas-15. NAYS-Messrs. Smith, Hale, Shorter, and Fearn of Alaba-ma; Wright and Stephens, of Georgia; DeClouet, Conrad, Kenner, Sparrow, and Marshall, of Louisiana; Harris, Brooke, Wilson, Clayton, Barry, and Harrison, of Mississippi; Chesnut, Wilthers, and Boyce, of South Carolina; Reagan, Waul, Gregg, and Oldham, of Texas-24.

## INTERCEPTED "CONFEDERATE" DESPATCH UPON THE AFRICAN SLAVE TRADE.

As showing the temper of the "Confederate" Government upon the revival of the African

slave trade, it is instructive to read the letter of J. P. Benjamin, Secretary of State of the "Confederate" Government, to L. Q. C. Lamar, "Confederate" Commissioner at St. Petersburg, which was intercepted and transmitted from St. Petersburg, March 3, 1863, by Bayard Taylor, Chargé d' Affaires :

## CONFEDERATE STATES OF AMERICA

DEPARTMENT OF STATE, RICHMOND, January 15, 1863. Sir: It has been suggested to this Government, from a Site: It has been suggested to this Government, nom a source of unquestioned authenticity, that after the recogni-tion of our independence by the European Powers, an ex-pectation is generally entertained by them, that in our treaties of amity and commerce a clause will be introduced making stipulations against the African slave trade. It is even thought that neutral Powers may be inclined to insist

upon the insertion of such a clause as a sine qua non. You are well aware how firmly fixed in our constitution is the policy of this Confederacy against the opening of that trade; but we are informed that false and insidious suggestions have been made by the agents of the United States at European courts of an intention to change our constitution as soon as peace is restored, and of authorizing the import-ation of slaves from Africa. If, therefore, you should find in your intercourse with the Cabinet to which you are accredited that any such impressions are entertained, you will use every proper effort to remove them; and if an attempt is made to introduce into any treaty which you may be charged with negotiating stipulations on the subject just the position which, under the direction of the President, I

now proceed to develop. The constitution of the Confederate States is an agreement made between independent States. By its terms all the powers of government are separated into classes as follows, viz:

1. Such powers as the States delegate to the General Government. 2. Such powers as the States agree to refrain from exer-

cising, although they do not delegate them to the General Government.

 Such powers as the States, without delegating them to the General Government, thought proper to exercise, by di-rect agreement between themselves contained in the constitution.

4. All remaining powers of sovereignty which, not being delegated to the Confederate States by the constitution, nor prohibited by it to the States, are reserved to the States, respectively, or to the people thereof. On the formation of the constitution, the States thought proper to percent all possible future discussions on the sub-

proper to prevent all possible future discussions on the sub-ject of slavery, by the direct exercise of their own power, and delegated no authority to the Confederate Government, save immaterial exceptions, presently to be noticed. Especially in relation to the importation of African ne-

groes was it deemed important by the States that no power to permit it should exist in the Confederate Government. to permit it should exist in the Confederate Government. The States, by the Constitution, (which is a treaty between themselves of the most solemn character that States can make.) unanimously stipulated "that the importation of negroes of the African race, from any foreign country other than the slaveholding States or Territories of the United States of America, is hereby forbidden; and Congress is regained to pass such laws as shall effectually prevent the same." (Art. 1, sec. 9, par. 1.) It will thus be seen that no power is delegated to the Confederate Government over this subject, but that it is in-

cluded in the third class above referred to, of power exer-

cised directly by the States. It is true that the *duty* is imposed on Congress to pass laws to render effectual the prohibition above quoted. But laws to render effectual the prohibition above quoted. But this very imposition of a duty on Congress is the strongest proof of the absence of power in the President and Senate alone, who are vested with anthority to make treaties. In a word, as the only provision on the subject directs the two branches of the legislative department, in connection with the President, to pass *laws* on this subject, it is out of the power of the President, aided by one branch of the legisla-tive department, to control the same subject by treaties; for there is not only an absence of express delegation of authority to the treaty-making power which alone would authority to the treaty-making power, which alone would suffice to prevent the exercise of such authority, but there is the implied prohibition resulting from the fact that all duty on the subject is imposed on a different branch of the Government.

I need scarcely enlarge upon the familiar principle, that authority expressly delegated to Congress cannot be as-sumed in our Government by the treaty-making power. The authority to levy and collect taxes, to coin money, to declare war, &c., &c., are ready examples, and you can be at no loss for argument or illustration in support of so well recognized a principle.

d.

The view above expressed is further enforced by the clause in the Constitution which follows immediately that which has already been quoted. The second paragraph of the same section provides that "Congress shall also have power to prohibit the introduction of slaves from any State not a member of, or territory not belonging to, this Confed-Here there is no direct exercise of power by the eracy." States which formed our Constitution, but in express dele-gation to Congress. It is thus seen that while the States were willing to trust Congress with the power to prolibit the introduction of African slaves from the United States, they were not willing to trust it with the power of prohibiting their introduction from any other quarter, but deter-mined to insure the execution of their will by a direct in-

Moreover, any attempt on the part of the treaty-making power of this Government to prohibit the African slave trade, in addition to the insuperable objections above suggested, would leave open the implication that the same power has authority to permit such introduction. No such implication can be sanctioned by us. This Government un-equivocally and absolutely denies its possession of any power whatever over the subject, aud cannot entertain any proposition in relation to it. While it is totally beneath the dignity of this Govern-

ment to give assurances for the purpose of vindicating it-self from any unworthy suspicions of its good faith on this subject, that may be disseminated by the agents of the United States, it may not be improper that you should point ont the superior efficacy of our constitutional provision to any treaty stipulations we could make. The constitution is itself a treaty between the States, of such binding force, that it cannot be changed or abrogated without the deliberate and concurrent action of nine out of the thirteen States that compose the Confederacy. A treaty might be abrogated by a party temporarily in power in our country, at the sole risk of disturbing amicable relations with a foreign Power. The Constitution, unless by approach to unanimity, could not be changed without the destruction of this Government itself or a complexibility to provide the properties of the provide treated on the sole of the so itself; and even should it be possible hereafter to procure the consent of the number of States necessary to change it, the forms and delays, designedly interposed by the framers to check rash innovations, would give ample time for the most mature deliberation, and for strenuous resistance on the part of those opposed to such a change.

After all, it is scarcely the part of wisdom to attempt to impose restraint on the actions and conduct of men for all future time. The policy of the Confederacy is as fixed and immutable on this subject as the imperfection of human nature permits human resolve to be. No additional agree-ments, treaties, or stipulations can commit these States to the prohibition of the African slave trade with more binding efficacy than those they have themselves devised. A just and generous confidence in their good faith on this subject, exhibited by friendly Powers, will be far more effica-cious than persistent efforts to induce this Government to clous than persistent enorts to induce this Government to assume the exercise of powers which it does not possess, and to bind the Confederacy by ties which would have no consti-tutional validity. We trust, therefore, that no unnecessary discussion on this matter will be introduced into your negotiations. If, unfortunately, this reliance should prove ill founded your will decline continuing negotiations on ill-founded, you will decline continuing negotiations on your side, and transfer them to us at home, where, in such event, they could be conducted with greater facility and advantage, under the direct supervision of the President.

With great respect, your obedient servant, J. P. BENJAMIN, Secretary of State.

Hon. L. Q. C. LAMAR, Commissioner, &c., &c., St. Petersburg, Russia.

# Arrests of Citizens, and the writ of Habeas Corpus.

June 27, 1861, Major General N. P. BANKS, commanding the Department of Annapolis, had George P. Kane, Chief of Police of Baltimore, arrested for being, in contravention of his duty and in violation of law, by direction or indirection, both witness and protector to transactions hostile to the authority of the Government, and to conspirators avowedly its enemies.

Same day, Charles Howard, Wm. H. Gatchell, Charles D. Hinks, John W. Davis, and George Wm. Brown, Mayor and ex-officio member of Board of Police of Baltimore, protested against the arrest of Marshal Kane, and the suspension of the Board of Police, by a military provost marshal.

July 1, General Banks, in pursuance of orders from the Headquarters of the Army at Washington, arrested the four first-named members. for these reasons:

The incidents of the past week afforded full justification for this order. The headquarters, under the charge of the board, when abandoned by the officers, resembled in some respects, a concealed areenal. After public recognition and protest against the "suspinsion of their functions," they continued their sessions daily. Upon a forced and unvarrantable construction of my proclamation of the 27th ultimo, they declared that the police law was suspended, and the police officers and men put off duty for the present, intend-ing to leave the city without any police protection what-ever. They refused to recognize the officers or men neces-willy refused by the provide merched in the second sarily selected by the provest marshal for its protection, and hold subject to their orders, now and hereafter, the old police force, a large body of armed men, for some purpose not known to the Government, and inconsistent with its peace or security. To anticipate any intentions or orders on their part, I have placed temporarily a portion of the force under my command within the city. I disclaim on the part of the Government I represent, all desire, intention, and purpose to interfere, in any manner whatever, with the ordinary municipal affairs of the city of Baltimore. Whenever a loyal citizen cau be named who will execute its police laws with impartiality and in good faith to the United States, the military force will be withdrawn from the central parts of the municipality at once. No soldier will be permitted in the city, except under regulations satisfactory to the marshal; and if any so admitted violate the municipal law, they shall be punished according to the civil law, by the civil tribunals.

They were transferred to Fort Lafayette, and on the 6th of August, Judge Garrison of Brooklyn, issued a writ directing Col. Burke to pro-duce the persons in court. Col. Burke declined on the authority of an order from Lieut. Gen. Scott. Col. Burke was then cited to answer for contempt of court, but he did not appear, and August 22, Judge Garrison, "submitting to inevitable necessity," dismissed the proceedings. They were subsequently released. See President's Orders, p. 154.

### ARREST OF MEMBERS OF THE LEGISLATURE OF MARYLAND.

#### NEWSPAPER ACCOUNT.

## By Telegraph to the Associated Press.

BALTIMORE, Sept. 13 .- The Provost Marshal, George P. BALTIMORE, Sept. 13.—The Provost Marshail, George P. Dodge, this morning, before day, arrested the Mayor of Baltimore, Mr. Brown, and Messrs. Chas. S. Pitts, Lawrence Sangston, S. Teackle Wallis, T. Parkin Scott, and Ross Winans, membersof the Maryland Legislature of Baltimore city, and F. K. Howard, the editor of the *Exchange* news-paper. They were taken to Fort Mellenry. BALTIMORE, Sopt. 13.—The following additional arrests have been made: Messrs. Dennison, Quinlau, and Dr. Lynch, members of the Legislature from Baltimore county; and Messrs. Henry M. Warfield, Dr. J. Hauson Thomas, John C. Brune, city members; also, Thomas W. Hall, editor of the *South* newspaner.

South newspaper.

The day of the meeting of the Legislature is Tuesday next, when, it is suspected, further legislation hostile to the Government was to be attempted. All the arrests made were under orders direct from the War Department.

BALTINORE, Sept. 13.—I just hear of the arrest of Henry May, member of Congress; also, Henry M. Morfit and W. G. Harrison, members of the Legislature from this city. These, with the names previously sent, complete the ten city delegates. Upon the arrest of Gordon, (member of the Maryland Legislature,) some days ago, papers were found in his baggage reading like amendments to be offered to a proposed secession ordinance to be brought up at the coming

meeting of the Legislature. BALTIMORE, Sept. 18.—The police are arresting secession members of the Legislature as fast as they reach this city, on their way to Frederick. To-morrow is the day set apart for the meeting of the Legislature, but there can be no quothe two meeting of the Degistance, out into Gill be in O glio-rum present, as nearly three fourths of both Houses are secessionists, all of whom it is presumed, will be arrested. This evening Messrs. Dennis and Heckart, of the Senate, and Messrs. Landing and Raisin, of the House of Delegates, were arrested. There are now fifteen members of the House were arrested. There are now international of the mem-and three of the Senate under arrest. Many of the mem-bers cannot be found, and have, it is said, fied from the State.

1

### [From the Baltimore American.]

We are not advised of the specific charges against those members of the Legislature and others prominent as public functionaries who have been arrested in the State by order of the General Government; but, from what has already occurred, the inference seems safe enough that the reasons were such as were fully justified in the needful preservation were such as were fully justified in the needfal preservation of the peace of the State. One thing is certain—that the majority of the distinguished body thus interfered with was thoroughly disloyal; and judging from what they already have done, and their persistant waiting for something to hap-pen to give them a chance to do something more in the di-rection of "State sovereignty," we believe they thought the time might come when they might follow other illustrious examples, and treat the State to that outrage upon the peo-ple, the Constitution, and the Union, a "secession ordinance." They are effectually estopped from such a purpose now, and will have a chance to reflect at their leisure on their utter will have a chance to reflect at their leisure on their utter disregard of the wishes of the people in their doings. The astounding disregard of popular sentiment, definitely shown, has been the chief feature in the secession movement,

and Maryland has only escaped the worst consequences of it by the firm action of the Governor in the first instance, and now by the interposition of the strong hand of the General Government.

#### COPELAND'S REPORT.

FREDERICK, Md., September 18, 1861. To Major General BANKS, Darnestown:

SIR: I have just telegraphed to General Dix that we have seized seven members of the house of a very bitter character, and four officers, clerks, &c., who are intensely bitter, and are said to have been very forward and to have kept some of the weaker men up to the work. Several arrests were made of violent or resisting persons, whom I shall let go after the others are gone. I shall send four men at least to General Dir, at Baltimore, who are very bad men. I have advised Colonel Ruger to send to Sharpsburg Landing to seize 500 sacks of salt, which are waiting for the Southerners to come and take them. They have tried twice to do it. We have also heard of some arms which the colonel will look up. also heard of some arms which the colonel will look up. There is a very bitter man here, a Mr. Sinn, who is currently reported by General Shriver and others to be the medium of communication with the Southern Confederacy. The names of the members are: B. S. Salmon, R. C. McCubbin, J. H. Gordon, C. J. Durant, Thomas Cleggett, Andrew Kessler, and Bernard Mills. We shall get T. Lawrence Jones. The officers of the Legislature: J. N. Brewer, Chief Clerk Sen-ate; Thomas Moore, reading do; Samuel Penrose, jr., As-sistant; N. Kilgore, reading do; Simutel Penrose, jr., (very bad.) Printer to the House; John Hogan, (very bad.) citizen; Joseph Elkins, do.; Mr. Mason, Folder to the House. We shall leave here for headquarters this after-noon. The arrested were nearly all seized by the police-The arrested were nearly all seized by the policenoon. nien. I am, yours respectfully,

R. MORRIS COPELAND, Aid-de-Camp.

Mr. McCubbin is a person whom I should recommend you to set at large if he takes the oath, which I have no doubt he will. He is brother-in-law to General Hammond, and a man much respected; also a man of rather timid nature, and greatly troubled by his arrest. General Shriver has been very active for us, and is very carnest that we should let him go on these terms. If you can do it, it will be well to telegraph to Annapolis to have the oath tendered, and release him. I should do it under my instructions, only that Colonel Ruger thinks he has no authority to allow any man on the list any liberty. R. M. C.

## OTHER ARRESTS.

PHILADELPHIA, August 19. Pierce Butler was arrested this afternoon by the United States Marshal by order of Secretary Cameron. He was taken to New York this evening, en route to Fort Hamilton.

LOUISVILLE, September 19. Barly this morning the United States Marshal seized the office of the Louisville Courier, arrested ex-Governor More-head, Reuben T. Murrett, one of the proprietors of the Courier, and Martin W. Barr, telegraphic news reporter for the New Orleans press, on charges of treason or complicity with treason. The prisoners were all carried to Jefferson-tillo preferred to the outperfured for during the Warshal with treason. The prisoners were all carried to Jefferson-ville, and will be transferred to the custody of the Marshal of Indiana district.

CINCINNATI, September 26. Yesterday afternoon Lieutenant Colonel Letcher, with a detachment of Colonel Woodward's regiment, captured James B. Clay, with sixteen of his men, while on the way to join Zollicoffer. They were taken to Camp Dick Robin-son. John C. Breckinridge was with their party in this city but second city, but escaped.

Ex-Mayor James G. Berret, of Washington, was arrested in August, but released September 12, 1861, on taking the oath of allegiance, and resigning the office of mayor.

Ellis B. Schnable, of Pennsylvania, was also arrested late in August.

James W. Wall, of Burlington, N. J., was arrested, and others.

#### ORDERS ON WHICH CERTAIN ARRESTS WERE MADE.

### Secretary of War to General Banks.

WAR DEPARTMENT, Sept. 11, 1961. GENERAL: The passage of any act of secession by the Legislature of Maryland must be prevented. If necessary all, or any part, of the members must be arrested. Exercise your own judgment as to the time and manner, but do the work effectually. offectually. Very respectfully, your obedient servant, SIMON CAMERON, Secretary of War.

#### Gen. McClellan to Gen. Banks.

[CONFIDENTIAL.]

IECONFIDENTIAL.] HEADQUARTERS ARMY OF THE POTOMAC, WASHINGTON, Sept. 12, 1861. GENERAL: After full consultation with the President, Secretaries of State, War, &c., it has been decided to effect the operation proposed for the 17th. Arrangements have been made to have a Government steamer at Annapolis to receive the prisoners and carry them to their destination. Some four or five of the chief men in the affair are to be arrested today. When they meet on the 17th you will

arrested to-day. When they meet on the 17th, you will please have everything prepared to arrest the whole party,

and be sure that none escape. It is understood that you arranged with Gen. Dix and Gov. Seward the modus operandi. It has been intimated to me that the meeting might take place on the 14th; please be prepared. I would be glad to have you advise me fre-quently of your arrangements in regard to this very im-portant matter.

If it is successfully carried out it will go far toward breaking the backbone of the rebellion. It would prohably be well to have a special train quietly prepared to take prisoners to Annapolis.

I leave this exceedingly important affair to your tact and discrction—and have but one thing to impress upon you— the absolute necessity of secrecy and success. With the highest regard, I am, my dear General, your sincere friend, GEO. B. McCLELLAN,

Maj. Gen. U. S. A.

Copy of Gen. Banks's instructions concerning the Legislature.

[IMPORTANT AND CONFIDENTIAL.]

HEADQUARTERS, CAMP NEAR DARNESTOWN,

September 16, 1863.

Lieut. Col. RUGER, Commanding Third Wisconsin regiment.

on special service at Frederick: SIR: The Legislature of Maryland is appointed to meet in special session to-morrow, Tuesday, Sept. 16. It is not imspecial session to-morrow, Tuesday, Sept. 16. It is not im-possible that the members, or a portion of them may be deterred from meeting there on account of certain arrests recently made in Baltimore. It is also quite possible that on the first day of meeting the attendance may be small. Of the facts, as to this matter, I shall see that you are well informed, as they transpire. It becomes necessary that our meeting of this I arrithtee the theory of the facts that any meeting of this Legislature, at any place or time, shall be prevented.

You will hold yourself and your command in readiness to arrest the members of both Houses; a list of such as you are to detain will be enclosed to you, herewith, among whom

are to detain will be enclosed to you, herewith, among whom are to be specially included the presiding officers of the two houses, secretaries, clerks, and all subordinate officials. Let the arrests be certain, and allow no chance of failure. The arrests should be made while they are in session, I think. You will, upon the receipt of this, quictly examine the premises. I am informed that escape will be impossible, if the entrance to the building be held by you, of that you will judge upon examination. If no session is to be held, you will arrest such members as can be found in Frederick. The entrance is a the source being build be to enter both Houses at the same process of arrest should be to enter both Houses at the same time, announcing that they were arrested by orders of the Government, command them to remain as they are, subject to your orders.

Any resistance will be forcibly suppressed, whatever the consequences. Upon these arrests being effected, the mem-bers that are to be detained will be placed on board a special train for Annapolis where a steamer will await them.

Everything in the execution of these orders is confided to your secrecy, discretion, and promptness.

The Case of Algernon S. Sullivan.

#### DEPARTMENT OF STATE WASHINGTON, September 10, 1861.

To DANIEL LORD, Esq., New York: SIR: I have received your letter of yesterday relating to Algernon S. Sullivan, a political prisoner now in custody at Fort Lafayette. This Department is possessed of treason-able correspondence of that person which no right or privi-lege of a lawyer or counsel can justify or excuse. The public safety will not admit of his being discharged.

In view of the many representations made to me in this case, I pray your excuse for giving this letter to the public. With great respect, sir, your obcdient servant,

WILLIAM H. SEWARD.

# The Case of Robert Elliot. DEPARTMENT OF STATE

# WASHINGTON, October 4, 1861.

To his Excellency ISRAEL WASHBURN, Augusta, Me .: GOVERNOR: Application has been made to the President for the release of Robert Elliot, a political prisoner held in custody at Fort Lafayette.

The evidence taken in his case shows that he had not only conceived the purpose of treasonable co-operation in the State of Maine with the insurrectionary citizens arrayed in arms, in other States, for the overthrow of the Government and the Union, but that he had even gone to the extreme length of getting up an unlawful armed force to operate in Maine against the lawful action of the State and of the Fed-eral Government. His associates in that treasonable enterprise, since his arrest, have taken an oath of allegiance to the United States. This proceeding is very proper in itself. But the representations they make, that they and he were loyal to the Union at the time when they were combining in arms against it, cannot be accepted, at least in his behalf. It appears that he is too intelligent to misunderstand the legitimate tendency of his criminal acts. He cannot be released. On the contrary, your vigilance in ferreting out the conspiracy and in arresting it, by denouncing it to the Government and the country, is deemed worthy of especial commendation.

If any of the other offenders are still persisting in their treasonable course, you will, I am sure, not fail to give in-formation to this Department.

I have the honor to be, very respectfully, your obedient ervant, WILLIAM H. SEWARD. servant,

## THE PRESIDENT'S ORDERS.

1862, February 14-The PRESIDENT issued an order reciting the circumstances of the country, the defection of officials in every department, the treason which pervaded and paralyzed every branch of the service, in justification of the resort to extraordinary measures, and adds :

Meantime a favorable change of public opinion has ocdefined; the whole structure of the Government is firm and stable; apprehensions of public danger and facilities for treasonable practices have diminished with the passions which prompted heedless persons to adopt them. The inwhich prompted heedless persons to adopt them. surrection is believed to have culminated and to be declin-

ing. The President, in view of these facts, and anxious to favor a return to the normal course of the Administration, as far as regard for the public welfare will allow, directs that all political prisoners or State prisoners now held in military custody, be released on their subscribing to a parole engaging them to render no aid or comfort to the enemies in hos-tility to the United States.

The Secretary of War will, however, at his discretion, except from the effect of this order any persons detained as spies in the service of the insurgents, or others whose release at the present moment may be deemed incompatible with the public safety

To all persons who shall be so released, and who shall keep their parole, the President grants au amnesty for any past offences of treason or disloyalty which they may have committed.

Extraordinary arrests will hereafter be made under the direction of the military authorities alone. By order of the President:

EDWIN M. STANTON, Secretary of War.

1862, February 27-The PRESIDENT issued this order :

Executive Order No. 2-In Relation to State Prisoners.

WAR DEPARTMENT, WASHINGTON CITY, Feb. 27, 1862. It is ordered :

First. That a special commission of two persons, one of military rank and the other in civil life, be appointed to examine the cases of the State prisoners remaining in the mil-itary custody of the United States, and to determine whether in view of the public safety and the existing rebellion, they should be discharged, or remain in military custody, or be

remitted to the civil tribunals for trial. Second. That Major General John A. Dix, commanding in Baltimore, and the Hon. Edwards Pierreport, of New York, be and they are hereby appointed Commissioners for the purposes above mentioned, and they are authorized to examine, hear, and determine the cases aforesaid *ex parte* and in a summary manner, at such times and places as in their discretion they may appoint, and make full report to the War Department.

By order of the President:

EDWIN M. STANTON. Secretary of War.

#### ORDER OF THE SECRETARY OF WAR.

WAR DEPARTMENT, WASHINGTON, November 22, 1962. Ordered-1. That all persons now in military custody, who have been arrested for discouraging volunteer enlistments, opposing the draft, or for otherwise giving aid and comfort to the enemy, in States where the draft has been made, or the quota of volunteers and militia has been furnished, shall be discharged from further military restraint. 2. The persons who, by the authority of the military com-

and or or governor in rebcl States, have been arrested and sent from such State for disloyalty or hostility to the Gov-ernment of the United States, and are now in military cus-tody, may also be discharged upon giving their parole to do no act of hostility against the Government of the United States, nor render aid to its enemies. But all such persons shall remain subject to military surveillance and liable to arrest on breach of their parole. And if any such persons shall prefer to leave the loyal States on condition of their not returning again during the war, or until special leave for that purpose be obtained from the President, then such person shall, at his option, be released and depart from the United States, or be conveyed beyond the military lines of the United States forces.

3. This order shall not operate to discharge any person who has been in arms against the Government, or by force and arms has resisted or attempted to resist the draft, nor relieve any person from liability to trial and punishment by civil tribunals, or by court-martial or military commis-sion, who may be amenable to such tribunals for offences committed. mmitted. By order of the Secretary of War: E. D. TOWNSEND,

Assistant Adjutant General.

## ARREST OF JOHN MERRYMAN AND PROCEEDINGS THEREON.

1861, May 25-John Merryman, of Baltimore county, Md., was arrested, charged with holding a commission as lieutenant in a company avowing its purpose of armed hostility against the Government; with being in communication with the rebels, and with various acts of trea-He was lodged in Fort McHenry, in comson. mand of Gen. Geo. Cadwalader. Merryman at once forwarded a petition to Chief Justice Roger B. Taney, reciting his arrest, and praying for a writ of habeas corpus and a hearing. The writ was issued for the 27th, to which General Cadwalader declined to respond, alleging, among other things, that he was duly authorized by the President of the United States to suspend the writ of habeas corpus for the public safety. May 27, the Chief Justice issued a writ of attachment, directing United States Marshal Bonifant to produce the body of General Cadwalader on Tuesday, May 28th, "to answer for his contempt in refusing to produce the body of John Merryman." May 28th, the Marshal replied that he proceeded to the fort to serve the writ, that he was not permitted to enter the gate, and that he was informed "there was no answer to his writ."

#### CHIEF JUSTICE TANEY'S REMARKS.

I ordered the attachment yesterday, because upon the ace of the return the detention of the prisoner was unlaw-

hee of the return the detention of the prisoner was unlaw-ful upon two grounds. 1. The President, under the Constitution and laws of the United States, cannot suspend the privilege of the writ of *indeas corpus*, nor authorize any military officer to do so. 2. A military officer has no right to arrest and detain a person, not subject to the rules and articles of war, for an offence again, the laws of the United States, except in and of the judicial authority and subject to its control; and if the party is arrested by the military, it is the duty of the fficer to deliver him over immediately to the civil author-ty to be dealt with according to law.

ty, to be dealt with according to law. I forbore yesterday to state orally the provisions of the Constitution of the United States which make these princonsultation of the United States which make these prin-iples the fundamental law of the Union, because an oral statement might be misunderstood in some portions of it, and I shall therefore put my opinion in writing and file it a the office of the Clerk of the Circuit Court, in the course of this week.

After reading the above, the Chief Justice rally remarked :

In relation to the present return, I propose to say that the marshal has legally the power to summon out the *posse positiztus* to seize and bring into court the party named in the attachment; but it is apparent he will be resisted in the discharge of that duty by a force notoriously superior of the *posse comitatus*, and each being the case, the Court has no power under the law to order the necessary force to compel the appearance of the party. If, however, he was sefore the Court, it would then impose the only punish-nent. nent.

ment. Under these circumstances the Court can barely say, to-lay, I shall reduce to writing the reasons under which I nave acted and which have led me to the conclusions ex-pressed in my opinion, and shall report them with these proceedings to the President of the United States, and call apon him to perform his constitutional duty to enforce the the proceedings of the concerned the proceed of the United States and call apon him to perform the concerned of the Court aws; in other words, to enforce the process of this Court. This is all this Court has now the power to do.

Subjoined is the opinion of the Chief Justice:\*

## Ex parte JOHN MERRYMAN.

Before the Chief Justice of the Supreme Court of the United States, at Chambers.

United States, at Chambers. The application in this case for a writ of habeas corpus is made to me under the 14th section of the judiciary act of 1789, which renders effectual for the citizen the consti-tutional privilege of the writ of habeas corpus That act zives to the courts of the United States, as well as to each Justice of the Supreme Court, and to every District Judge, ower to grant writs of habeas corpus, for the purpose of an inquiry into the cause of commitment. The petition was presented to me at Washington, under the impression that I would order the prisoner to be brought before me there, but as he was confined in Fort McHenry, at the city of Baltimere, which is in my circuit, I resolved to hear it in the latter city, as obedience to the writ, under such preumstances, would not withdraw Gen. Cadwalader, who

In the latter city, as obechence to the writ, under such bircumstances, would not withdraw Gen. Cadwalader, who had him in charge, from the limits of his military command. The petition presents the following case: The petitioner resides in Maryland, in Baltimore county. While peaceably in his own house, with his family, it was at two o'clock, on the morning of the 25th of May, 1861, entered by an armed force, professing to act under military orders. He was then compelled to rise from his bed, taken into cus-tody, and conveyed to Fort McHenry, where he is im-prisoned by the commanding officer, without warrant from any lawful authority. The commander of the fort, Gen. George Cadwalader, by whom he is detained in confinement, in his return to the writ, does not deny any of the facts alleged in the petition. He states that the prisoner was arrested by order of Gen. Keim, of Pennsylvania, and conducted as a prisoner to Fort McHenry by his order, and placed in his (Gen. Cad-walader's) custody, to be there detained by him as a prisoner. A copy of the warrant, or order, under which the prisoner was arrested, was demanded by his counsel, and refused. And it is not alleged in the yeturn that any specificact, con-stituting an offence against the laws of the United States, has been charged against him upon oath; but he appears

has been charged against him upon oath; but he appears has been charged against min upon general charges of treason and robellion, without proof, and without giving the names of the witnesses, or specifying the acts, which, in the judg-ment of the military officer, constituted these crimes. And

\* For a reply by Hon. Reverdy Johnson, see Moore's Rebellion Record, vol. 2, p. 185.

having the prisoner thus in custody upon these vague and unsupported accusations, he refuses to obey the writ of *habeas corpus*, upon the ground that he is duly authorized by the President to suspend it. The case, then, is simply this: A military officer residing

The case, then, is simply thts: A military officer residing in Pennsylvania issues an order to arrest a citizen of Mary-land, upon vague and indefinite charges, without any proof, so far as appears. Under this order his house is entered in the night; he is seized as a prisoner, and conveyed to Fort McHenry, and there kept in close confinement. And when a hadees corpus is served on the commanding officer, requir-ing him to produce the prisoner before a Justice of the Su-preme Court, in order that he may examine into the legality of the imprisonment, the answer of the officer is that he is authorized by the President to susmed the writ of holes authorized by the President to suspend the write of habeas corpus at his discretion, and, in the exercise of that discre-tion, suspends it in this case, and on that ground refuses obedience to the writ.

As the case comes before me, therefore, I understand that the President not only claims the right to suspend the writ of *habeas corpus* himself, at his discretion, but to delegate that discretionary power to a military officer, and to leave it to him to determine whether he will or will not obey ju-

if to him to determine whether he will or will not obey ju-dicial process that may be served upon him. No official notice has been given to the courts of justice, or to the public, by proclamation or otherwise, that the President claimed this power, and had exercised it in the manner stated in the return. And I certainly listened to it with some surprise, for I had supposed it to be one of those points of constitutional law upon which there was no differ-ence of opinion, and that it was admitted on all hands that the privilege of the writ could not be suspended except by act of Congress. act of Congress.

When the conspiracy of which Aaron Burr was the head became so formidable, and was so extensively ramified as to justify, in Mr. Jefferson's opinion, the suspension of the writ, he claimed, on his part, no power to suspend it, but com-numicated his opiniou to Congress, with all the proofs in his nonversion in order that for the conservation of the diame numcated his opiniou to congress, with all the proofs in his possession, in order that Congress might exercise its discre-tion upon the subject, and determine whether the public safety required it. And in the debate which took place upon the subject no one suggested that Mr. Jefferson might exer-cise the power himself, if, in his opinion, the public safety decreaded it demanded it.

demandéd it. Having, therefore, regarded the question as too plain and too well settled to be open to dispute, if the commanding officer had stated that upon his own responsibility, and in the exercise of his own discretion, he refused obedience to the writ, I should have contented myself with referring to the clause in the Constitution, and to the construction it received from every jurist and statesman of that day, when the case of Burr was before them. But being thus officially notified that the privilege of the writ has been suspended under the orders and by the authority of the President, and believing as I do that the President has exercised a power which he does not possess under the Constitution, a proper which he does not possess under the Constitution, a proper respect for the high office he fills requires me to state plainly and fully the grounds of my opinion, in order to show that I have not ventured to question the legality of his act with-out a careful and deliberate examination of the whole subject.

The clause in the Constitution which authorizes the suspension of the privilege of the writ of *habeas corpus* is in the ninth section of the first article. This article is devoted to the Legislative Department of

The print section of the first article. This article is devoted to the Legislative Department of the United States, and has not the slightest reference to the Executive Department. It begins by providing "that all legislative powers therein granted shall be vested in a Con-gress of the United States, which shall consist of a Senate and House of Representatives." And after prescribing the manner in which these two branches of the legislative de-partment shall be chosen, it proceeds to enumerate specifi-cally the legislative powers which it thereby grants, and legislative powers which it expressly prohibits, and, at the conclusion of this specification, a clause is inserted giving Congress "the power to make all laws which may be neces-sary and proper for carrying into execution the foregoing powers, and all other powers vested by this Constitution in the Government of the United States or in any department or office thereef." The power of legislation granted by this latter clanse is by its word carefully confined to the specific objects before enumerated. But as this limitation was unavoidably somewhat indefinite, it was deemed necessary to guard more effectually cortain great cardinal principles essential to the liberty of the citizen and to the rights and equality of the States by denying to Congress, in express terms, any power of legislation over them. It was apprehended, it seems, that such legislation might be attermized under the pretext that it was necessary and proper to carry into execution the powers granted; and it was determined that there should be no room to doubt, where rights of such vitals immediately followed by an enumeration of cer-tain subjects to which the powers of legislation shall not

extend; and the great importance which the framers of | extend; and the great importance which the trainers of the Constitution attached to the privilege of the writ of habeas corpus, to protect the liberty of the citizen, is proved by the fact that its suspension, except in cases of invasion and rebellicn, is first in the list of prohibited powers; and even in these cases the power is denied and its exercise prohibited unless the public safety shall re-quire it. It is true that in the cases mentioned Congress is of proceeding the hold when the multic safety shall reof necessity the judge of whether the public safety does or does not require it; and its judgment is conclusive. But the introduction of these words is a standing admonition to the legislative body of the danger of suspending it and of the extreme caution they should exercise before they give the Government of the United States such power over the liberty of a citizen.

It is the second article of the Constitution that provides for the organization of the Executive Department, and enumerates the powers conferred on it, and prescribes its duties. And if the high power over the liberty of the citizens now claimed was intended to be conferred on the President, it would undoubtedly be found in plain words in this article. But there is not a word in it that can furnish the slightest ground to justify the exercise of the power.

The article begins by declaring that the Executive power shall be vested in a President of the United States of America, to hold his office during the term of four years, and then proceeds to describe the mode of election, and to spethe process of describe the mode of election, and to spe-cify in precise and plain words the powers delegated to him and the duties imposed upon him. And the short term for which he is elected, and the narrow limits to which his power is confined, show the jealousy and apprehensions of future danger which the framers of the Constitution feit is relation to that denote the Constitution feit in relation to that department of the Government, and how carefully they withheld from it many of the powers belonging to the executive branch of the English Government which were considered as dangerous to the liberty of the subject, and conferred (and that in clear and specific terms) those powers only which were deemed essential to secure the successful operation of the Government.

He is elected, as I have already said, for the brief term of four years, and is made personally responsible, by im-peachment, for malfeasance in office. He is, from necessity, and thenature of his duties, the Commander in Chief of the army and navy, and of the militia, when called into actual service. But no appropriation for the support of the army can be made by Congress for a longer term than two years, so that it is in the power of the succeeding House of Rep-resentatives to withhold the appropriation for its support, and thus disband it, if, in their judgment, the President used or designed to use it for improper purposes. And al-though the militia, when in actual service, are under his command, yet the appointment of the officers is reserved to the States, as a security against the use of the military power for purposes dangerous to the liberties of the people or the rights of the States.

So, too, his powers in relation to the civil duties and authority necessarily conferred on him are carefully restrict-ed, as well as those belonging to his military character. He cannot appoint the ordinary officers of Government, nor make a treaty with a foreign nation or Indian tribe without the advice and consent of the Senate, and cannot ap-point even inferior officers unless he is authorized by an act of Congress to do so. He is not empowered to arrest any one charged with an offence against the United States, any one charged with an oncree against the Umited States, and whom he may, from the evidence before him, believe to be guilty; nor can he authorize any officer, civil or military, to exercise this power, for the fifth article of the amendments to the Constitution expressly provides that no person "shall be deprived of life, liberty, or property with-out due process of law;" that is, judicial process. And even if the privilege of the writ of *habeas corpus* was suspended by act of Congress and a party not subject to the rules and by act of Congress, and a party not subject to the rules and articles of war was afterwards arrested and imprisoned by regular judicial process, he could not be detained in prison or brought to trial before a military tribunal, for the article in the Amendments to the Constitution immediately following the one above referred to—that is, the sixth article—pro-vides that, "in all criminal prosecutions, the accused shall en-joy the right to a speedy and public trial by an impartial jury of the State and district wherein the crime shall have been committed, which district shall have been previously ascertained by law, and to be informed of the nature and cause of the accusation; to be confronted with the witnesses against him; to have compulsory process for obtaining witnesses against his favor, and to have the assistance of counsel for his defence."

And the only power, therefore, which the President pos-sesses, where the "life, liberty, or property" of a private citizen is concerned, is the power and duty prescribed in the third section of the second article, which requires "that he shall take care that the laws be faithfully executed." He is not authorized to execute them himself, or through agents or officers, civil or military, appointed by himself, but he is

to take care that they be faithfully carried into execution as they are expounded and adjudged by the coordinate branch of the Government to which that duty is assigned by the Constitution. It is thus made his duty to come in aid of the judicial authority, if it shall be resisted by a force too strong to be overcome without the assistance of the Executive arm. But in exercising this power, he acts in subordination to judicial authority, assisting it to execute its process and enforce its judgments.

With such provisions in the Constitution, expressed in language too clear to be misunderstood by any one, I can see no ground whatever for supposing that the President, in any emergency or in any state of things, can authorize the suspension of the privilege of the writ of *kabeas corpus*, or arrest a citizen, except in aid of the judicial power. If e cer-tainly dues not faithfully execute the laws if he takes upon himself legislative power by suspending the writ of *habeas* corpus—and the judicial power, also, by arresting and im-prisoning a person without due process of law. Nor can any argument be drawn from the nature of sovereignty, or the necessities of government for self-defense, in times of turnit and danger. The Government of the United States is one of and danger. The Government of the United States is one of delegated and limited powers. It derives its existence and authority altogether from the Constitution, and neither of its branches—executive, legislative, or judicial—can exercise any of the powers of government beyond those-specified and granted. For the tenth article of the amcudments to the Constitution, in express terms, provides that "the powers not delegated to the United States by the Constitution, nor prohibited by it to the States, are reserved to the States, re-Indeed, the security against imprisonment by Executive

authority, provided for in the fifth article of the Amendments of the Constitution, which I have before quoted, is nothing more than a copy of a like provision in the English constitu-tion, which had been firmly established before the Declaration of Independence.

Blackstone, in his Commentaries, (1st vol., 137.) states it in the following words:

To make imprisonment lawful, it must be either by pro-

And the people of the United Colonies, who had them-selves lived under its protection while they were British subjects, were well aware of the necessity of this safeguard for their rescand liberty. And he one are heliow that is for their personal liberty. And no one can believe that in framing the Government intended to guard still more effi-ciently the rights and the liberties of the citizens against executive encroachment and oppression, they would have conferred on the President a power which the history of England had proved to be dangerous and oppressive in the hands of the Crown, and which the people of England had compelled it to surrender after a long and obstinate struggle on the part of the English Executive to usurp and reťain it.

The right of the subject to the benefit of the writ of habeas corpus, it must be recollected, was one of the great points in controversy during the long struggle in England between arbitrary government and free institutions, and must therefore have strongly attracted the attention of statesmen engaged in framing a new and, as they supposed, a freer government than the one which they had thrown off by the Revolution. For, from the earliest history of the common law, if a person was imprisoned—no matter by what authority—he had a right to the writ of *habcas* corpus to bring his case before the King's Bench, and, if of positive offence was charged against him in the warrant of commitment, he was entitled to be forthwith discharged; and if an offence was charged which was hailable in its character the court was bound to set him at liberty on bail. And the most exciting contests between the Crown and the people of England from the time of Magna Charta and the people of high and not not the of this of high Charts were in relation to the privilege of this writ, and they con-tinued until the passage of the statute of 31st Charles 2d, commonly known as the great habeas corpus act. This statute put an end to the struggle, and finally and firmly secured the liberty of the subject from the usurpation and purposed of the convertion oppression of the executive branch of the Government. It nevertheless conferred no new right upon the subject, but only secured a right already existing; for, although the right could not justly be denied, there was often no effectual remedy against its violation. Until the statute of the 13th of William III the judges held their offices at the pleasure of the King, and the influence which he excreted the influence which he excreted becaute of time-serving, and partian judges often induced them, upon some pretext or other, to refuse to discharge the party although he was entitled to it by law, or delayed their decisions from time to time, so as to prolong the im-prisonment of persons who were obnoxious to the King for prisonment of persons who were concrude to the ring of their political opinions, or had incurred his resentment in any other way. The great and inestimable value of the *habeas corpus* act of the 31st Charles II is that it contains provisions which

compel courts and judges, and all parties concerned, to per-

form their duties promptly, in the manner specified in the | ingly, no power in England short of that of Parliament, statute

A passage in Blackstone's Commentaries, showing the an-cient stato of the law upon this subject, and the abuses which were practiced through the power and influence of the Crown, and a short extract from Hallam's Constitu-tional History, stating the circumstances which gave rise to the passage of this statute, explain briefly, but fully, all that is material to this subject.

In presence of the solution examine into its validity, and, according to the circumstances of the case, may discharge, admit to bail, or remand the

"And yet early in the reign of Charles I the Court of "And yet early in the reign of Charles I the Court of King's Bench, relying on some arbitrary precedents, (and those perhaps misunderstood,) determined that they would those perhaps misunderstood,) determined that they would not, upon a habeas corpus either bail or deliver a prisoner, though committed without any cause assigned, in case he was committed by the special command of the King or by the Lords of the Privy Council. This drew on a Parliamen-tary inquiry, and produced the Petition of Rights-3 Chas. I-which recites this illegal judgment, and enacts that no freeman hereafter shall be so imprisoned or detained. But when in the following year Mr. Selden and others were com-mitted by the Lords of the Council in pursuance of his Ma-festy's special command, under a general charge of 'notable mitted by the Lords of the Council in pursuance of his Ma-jesty's special command, under a general charge of 'notable contempts, and stirring up sedition against the King and the Government,' the judges delayed for two terms (including also the long vacation) to deliver an opinion how far such a charge was bailable. And when at length they agreed that it was, they however annexed a condition of finding sureties for their good behavior, which still protracted their impris-omment; the Chief Justice, Sir Nicholas Hyde, at the same time declaring that 'if they were again remanded for that time declaring that if they were again remanded for that cause perhaps the court would not afterward grant a *habeas corpus* being already acquainted with the cause of the im-prisonment. But this was heard with indignation and ascorpus being already acquainted with the cause of the im-prisonment.<sup>3</sup> But this was heard with indignation and as-tonishment by every lawyer present, according to Mr. Sel-den's own account of the matter, whose resentment was not cooled at the distance of four and twenty years.<sup>3</sup> It is worthy of remark that the offences charged against the prisoner in this case, and relied on as a justification for

his arrest and imprisonment, in their nature and character, his arrest and imprisonment, in their nature and character, and in the loose and vague manner in which they are stated, bear a striking resemblance to those assigned in the war-rant for the arrest of Mr. Selden. And yet, even at that day, the warrant was regarded as such a flagrant violation of the rights of the subject that the delay of the time-serving judges to set him at liberty upon the *kabeas corpus* issued in his behalf excited universal indignation at the bar. The extract from Hallam's Constitutional History is equally impressive end equally in point. It is in vol. 4, p. 14: "It is a very common mistake, and not only among for-einens. but many from whom some knowledge of our con-

"It is a very common mistake, and not only among for-eigners, but many from whom some knowledge of our con-stitutional laws might be expected, to suppose that this statute of Charles II enlarged in a great degree our liber-ties, and forms a sort of epoch in their history. But though a very beneficial enactment, and eminently remedial in many cases of illegal imprisonment, it introduced uo new principle, nor conferred any right upon the subject. From the earliest records of the English law, no freeman could be dutated in prison averat upon a criminal charge or conthe earnest records of the English law, ho freeman could be detained in prison, except upon a criminal charge or con-viction, or for a civil debt. In the former case it was always in his power to demand of the Court of King's Bench a writ of habeas corpus ad subjiciendum directed to the person de-taining him in custody, by which he was enjoined to bring up the body of the prisoner with the warrant of commit-ment that the court night index of its sufficiency and the ment, that the court might judge of its sufficiency, and re-mand the party, admit him to bail, or discharge him, accord-ing to the nature of the charge. This writ issued of right, and could not be refused by the court. It was not to bestow an immunity from arbitrary imprisonment, which is abun-dantly provided for in *Magna Charta*, (if indeed it was not more ancient.) that the statute of Charles II was enacted, but to cut off the abuses by which the Government's lust of bower and servile subtlety of Crown lawyers had impaired so fundamental a privilege." While the value set upon this writ in England has been so great that the removal of the abuses which embarrassed

is enjoyment have been locked upon as almost a new grant of liberty to the subject, it is not to be wondered at that the continuance of the writ thus made effective should have been the object of the most jealous care. Accord-

ingly, no power in England short of that of Parliament, can suspend or authorize the suspension of the writ of *habeas corpus.* I quote again from Blackstone (I Comm., 136:) "But the happiness of our Constitution is, that it is not left to the executive power to determine when the danger of the State is so great as to render this measure expedient. It is the Parliament only or legislative power that, whenever it sees proper, can authorize the Crown, by suspending the *habeas corpus* for a short and limited time, to imprison suspected persons without giving any reason for so doing." And if the President of the United States may suspend the writ, then the Constitution of the United Bay subject the wire, then the constitution of the original States has conferred upon him more regal and absolute power over the liberty of the citizen than the people of England have thought it safe to entrust to the Crown—a power which the Queen of England cannot exercise at this day, and which could not have been lawfully exercised by the sovereign even in the reign of Charles the First. But I am not left to form my judgment upon this great question from analogies between the English Government

question from analogies between the English (dovernment and our own, or the commentaries of English jurists, or the decisions of English courts, although upon this subject they are entitled to the highest respect, and are justly re-garded and received as authoritative by our courts of justice. To guide me to a right conclusion, I have the Commentaries on the Constitution of the United States of the late Mr. Justice Story, not only one of the most eminent jurists of the age, but for a long time one of the brightest ornaments of the Supreme Court of the United States, and also the clear and authoritative decision of that Court issol, given more than half a century since, and conclusively establishing the principles I have above stated. Mr. Justice Story, speaking in his Commentaries of the *habeas corpus* clause in the Constitution, says:

"It is obvious that cases of a peculiar emergency may arise, which may justify, and, even require, the temporary suspension of any right to the writ. But as it has fre-quently happened in foreign countries, and even in Eng-land, that the writ has, upon various pretexts and occasions, been suspended, whereby persons apprehended upon sus-picion have suffered a long imprisonment, sometimes from design, and sometimes because they were forgotten, the right to suspend it is expressly confined to cases of rebel-lion or invasion, where the public safety may require it. A very just and wholesome restraint, which cuts down at a blow a fruitfal means of oppression, capable of being abused in bad times to the worst of purposes. Hitherto no suspen-sion of the writ has ever been authorized by Congress since the establishment of the Constitution. It would seem, as the

The establishment of the Constitution. It would seem, as the power is given to Congress to suspend the writ of *habeas corpus* in cases of relclino or invasion, that the right to judge whether the exigency had arisen must exclusively belong to that body."—3 Story's Com. on the Constitution, section 1,336. And Chief Justice Marshall, in delivering the opinion of the Supreme Court in the case *ex parte* Bollman and Swart-wout, uses this decisive language, in 4 Cranch, 95: "It may be worthy of remark, that this act, (speaking of the one under which I am proceeding.) was passed by the First Congress of the United States, sitting under a Consti-tution which had declared 'that the privilege of the writ of *habeas corpus* should not be suspended unless when, in cases of rebellion or invasion, the public safety may require it." Acting under the immediate influence of this injunc-tion, they must have felt with peculiar jorce the obligation tion, they must have felt with peculiar force the obligation of providing efficient means by which this great constitu-tional privilege should receive life and activity; for if the means be not in existence, the privilege itself would be lost, means be not in existence, the privilege itself would be loss, although no law for its suspension should be enacted. Un-der the impression of this obligation they give to all the courts the power of awarding writs of habeas corpus." And again, in page 101: "If at any time the publicsafety should require the sus-provide of the power verted by this set in the courts of the

In at any time the public safety should require the sus-pension of the powers vested by this act in the courts of the United States, it is for the Legislature to say so. That question depends on poltical considerations, on which the Legislature is to decide. Until the legislative will be ex-pressed, this court can only see its duty, and must obey the laws."

I can add nothing to these clear and emphatic words of my great predecessor.

But the documents before me show that the military au-thority in this case has gone far beyond the mere suspension of the privilege of the writ of *habeas corpus*. It has, by force of arms, thrust aside the judicial authorities and officers to whom the Constitution has confided the power and duty of intermeting and equipictorize the large and subtituted to whom the Constitution has confided the power and duty of interpreting and administering the laws, and substituted a military government in its place, to be administered and executed by military officers. For at the time these pro-ceedings were had against John Merryman, the District Judge of Maryland—the commissioner appointed under the act of Congress—the District Attorney and the Marshal, all resided in the city of Baltimore, a faw miles only from the home of the prisoner. Up to that time there had never been the slightest resistance or obstruction to the process of any court or judicial officer of the United States in Maryland, except by the military authority. And if a military officer, or any other person, had reason to believe that the prisoner had committed any offense against the laws of the United States, it was his duty to give information of the fact and the evidence to support it to the District Attorney, and it would then have become the duty of that officer to bring the matter before the District Judge or Commissioner, and if there was sufficient legal evidence to justify his arrest, the Judge or Commissioner would have invested his wavent to the Murchel Commissioner would have issued his warrant to the Marshal to arrest him, and, upon the hearing of the party, would have held him to bail, or committed him for trial, according to the character of the offense as it appeared in the testi-mony, or would have discharged him immediately if there was not sufficient evidence to support the accusation. There was no danger of any obstruction or resistance to the action of the civil authorities, and therefore no reason whatever for of the civil authornies, and therefore ho reason whatever for the interposition of the military. And yet, under these cir-cumstances, a military officer, stationed in Pennsylvania, without giving any information to the District Attorney, and without any application to the judicial authorities, as-sumes to himself the judicial power in the district of Mary-land; undertakes to decide what constitutes the crime of transcene metholisms, what endeance (if indeed he required treason or rebellion; what evidence (if, indeed, he required any) is sufficient to support the accusation and justify the commitment; and commits the party, without having a bear-ing even before himself, to close custody in a strongly gar-risoned fort, to be there held, it would seem, during the pleasure of those who committed him.

The Constitution provides, as I have before said, that "no person shall be deprived of life, liberty, or property, without due process of law." It declares that "the right of the people to be secure in their persons, houses, papers, and effects against unreasonable searches and seizures shall not be violated, and no warrant shall issue but upon probable cause, supported by oath or affirmation, and par-ticularly describing the place to be searched and the per-sons or things to be seized." It provides that the party accused shall be entitled to a speedy trial in a court of iustice.

And these great and fundamental laws, which Congress And these great and inhumentaria laws, which congress fiself could not suspend, have been disregarded and sus-pended, like the writ of *habeas corpus*, by a military order, supported by force of arms. Such is the case now before me; and I can only say that if the authority which the Constitution has confided to the judiciary department and iddicid forces mer they more than protect on under any Constitution has connect to the function y department and judicial officers may thus upon any pretext or under any circumstances be usurped by the military power at its dis-cretion, the people of the United States are no longer living under a Government of laws, but every citizen holds life, liberty, and property at the will and pleasure of the army officer in whose military district he may happen to be found be found.

In such a case my duty was too plain to be mistaken. In such a case my duty was too plain to be mistaken. I have exercised all the power which the Constitution and laws confer on me, but that power has been resisted by a force too strong for me to overcome. It is possible that the officer who had incurred this grave responsibility may have misunderstood his instructions, and exceeded the authority intended to be given him. I shall, therefore order all the proceedings in this case, with my opinion, to be filed and recorded in the Circuit Court of the United Stores for the District of Maryland and direct the coler to be filed and recorded in the Circuit Court of the United States for the District of Maryland, and direct the clerk to transmit a copy, under seal, to the President of the United States. It will then remain for that high officer, in fulfi-ment of his constitutional obligation to "take care that the laws be faithfully executed," to determine what meas-ures he will take to cause the civil process of the United States to be respected and enforced. B. B. TANEY

#### R. B. TANEY, Chief Justice of the Supreme Court of the United States.

### OPINION OF ATTORNEY GENERAL BATES ON THE PRESIDENT'S POWER TO ARREST SUSPECTED PERSONS, AND SUSPEND THE WRIT OF HABEAS CORPUS.

### ATTORNEY GENERAL'S OFFICE, July 5, 1861.

SIR: You have required my opinion in writing upon the

"I. In the present time of a great and dangerous insur-rection, has the President the discretionary power to cause to be arrested and held in custody persons known to have criminal intercourse with the insurgents, or persons against whom there is probable cause for suspicion of such criminal complicity?

"II. In such cases of arrest is the President justified in of a judge, requiring him or his agent to produce the body of the prisoner, and show the cause of his caption and detention, to be adjudged and disposed of hy such court or judge?"

To make my answer to these questions at once consistent and plain, I find it convenient to advert to the great principle of government as recognized and acted upon in most, if not all, the countries in Europe, and to mark the differ-ence between that principle and the great principle which lies at the bottom of our National Government.

Most European writers upon government assume, ex-pressly or by implication, that every national Government is, and must be, the full expression and representation of the nation which it governs, armed with all its powers, and able to assert all its rights. In England, the form of whose Government more nearly approximates our own, and where the rights, interests, and powers of the people are more respected and cared for than in most of the nations of the European continent, it has grown into an axiom that "the Parliament is omnipotent," that is, that it can do any-thing that is possible to be done by legislation or by judg-ment. For all the ends of the Covernment the Parliament is the nation. Moreover, in Europe generally, the sovereignty is vested visibly in some designated man or set of men, so that the subject people can see their sovereign as well as feel the workings of his power. But in this coun-try it has been carefully provided otherwise. In the for-mation of our national Government our fathers were surrounded with peculiar difficulties, arising out of their novel, rounded with peculiar difficulties, arising out of their novel, I may say unexampled, condition. In resolving to break the ties which had bound them to the British Empire, their complaints were leveled chiefly at the King, not the Parliament nor the people. They seem to have been ac-tuated by a special dread of the unity of power, and hence, in framing the Constitution, they preferred to take the risk of leaving some good undore, for lack of power in the organt rether than ary, any covernmental officer with such agent, rather than arm any governmental officer with such great powers for evil as are implied in the dictatorial charge to "see that no damage comes to the commonwealth."

wealth." Hence, keeping the sovereignty always out of sight, they adopted the plan of "checks and balances," forming separate departments of government, and giving to each department separate and limited powers. These departments are co-ordinate and coequal; that is, neither being sovereign, each is independent in its sphere, and not subordinate to the others, either of them or both of them together. We have three of these co-ordinate departments. Now, if we allow one of the three to determine the extent of its own powers, and also the extent of the powers of the other two, that one can control the whole Government, and has in fact achieved can control the whole Government, and has in fact achieved he sovereighty.

We ought not to say that our system is perfect, for its de-fects (perhaps inevitable in all human things) are obvious. Our fathers having divided the Government into co-ordinate departments, did not even try (and if they had tried would probably have failed) to create an arbiter among them to adjudge their conflicts and keep them within their respective-bounds. They were left, by design, I suppose, each indepen-dent and free to act out its own granted powers, without any ordnined legal superior possessing the power to revise and reverse the action. And this with the hope that the three departments, mutually coequal and independent, would keep each other within their proper spheres by their mutual antagonism; that is, by the system of checks and balances to which our fathers were driven at the beginning by their fear of the unity of power.

In this view of the subject, it is quite possible for the same identical question (not case) to come up legitimately before each one of the three departments, and be determined in three different ways, and each decision stand irrevocable, binding upon the parties to each case; and that for the simple reason that the departments are co-ordinate, and there is no ordained legal superior with power to revise and reverse their decisions.

To say that the departments of our Government are coordinate is to say that the judgment of one of them is not binding upon the other two as to the arguments and princibinding upon the other two as to the arguments and princi-ples involved in the judgment. It binds only the parties to the case decided. But if, admitting that the departments of Government are co-ordinate, it be still contended that the principles adopted by one department, in deciding a case properly before it, are binding upon another department, that obligation must of necessity be reciprocal; that is, if the President be bound by the principles laid down by the Judi-ciary, so also is the Judiciary bound by the principles laid down by the President. And thus we shall have a theory of constitutional covernment failty contradicting itself. Penertconstitutional government flatly contradicting itself. Departconstitutional government flatly contradicting itself. Depart-ments co-ordinate and coequal, and yet reciprocally subordi-nate to each other! That cannot be. The several depart-ments, though far from sovereign, are free and independent, in the exercise of the limited powers granted to them respect-ively by the Constitution. Our Government, indeed, as a whole, is not vested with the sovereignty and does not pos-sess all the powers of the nation. It has no powers but such as are granted by the Constitution; and many powers are expressly withheld. The nation certainly is coequal with all other nations, and has equal powers, but it has not chosen to delegate all its powers to this Government, in any or all its

departments. The Government, as a whole, is limited; and limited in all its departments. It is the especial function of the judiciary to hear and determine cases, not to "establish principles," nor "settle questions," so as to conclude any person but the parties and privies to the cases adjudged. specially granted and defined by the Constitution, article 3, section 2:

"The judicial power shall extend to all cases in law and equity arising under this Constitution, the laws of the Unied States, and treaties made and which shall be made under their authority; to all cases affecting ambassadors, other ministers, and consuls; to all cases of admiralty and mariministers, and consuls, to an cases of animity and the United States shall be a party; to controversies to which the United States; hetween States and citizens of other States; between citizens of different States; between citizens of the same State claiming lands under grants of different States, and between a State, or the citizens thereof, and foreign States, citizens, or subjects."

And that is the sum of its powers, ample, and efficient for all the purposes of distributive justice among individual parties, but powerless to impose rules of action and of judgment upon the other departments. Indeed, it is not itself bound by its own decisions, for it can, and often does, overrule and disregard them, as in common honesty it ought to do, when-ever it finds, by its after and better lights, that its former judgments were wrong. Of all the departments of the Government the President

is the most active, and the most constant in action. He is "the Executive;" and so in fact he is, and much more called also, for the Constitution has imposed upon him many important duties, and granted to him great powers which are in their nature not executive—such as the veto power, the power to send and receive ambassadors, the power to make treaties, and the power to appoint officers. This last is not more an executive power when used by the President than it is when it is exercised by either House of Congress, by the courts of justice, or by the people at large. The President is a department of the Government, and,

although the only department which consists of a single man, he is charged with a greater range and variety of pow-ers and duties than any other department. He is a civil magistrate, not a military chief; and in this regard we see a striking proof of the generality of the sentiment prevailing in this country at the time of the formation of our Government, to the effect that the military ought to be held in strict subordination to the civil power; for the Constitution, while it grants to Congress the unrestricted power to de clare war, to raise and support armies, and to provide and maintain a navy, at the same time guards carefully against the abuse of that power, by withholding from Congress and from the army itself the authority to appoint the chief commander of a force so potent for good or evil to the State. The Constitution provides that "the President shall be Commander-in-Chief of the Army and Navy of the United States, and of the militia of the several States when called into the actual service of the United States." And why is this? Surely not because the President is supposed to be, or commonly is, in fact, a military man, a man skilled in the art of war, and qualified to marshal a host in the field of battle. No, it is for quite a different reason; it is, that whatever skillful soldier may lead our armies to victory sgainst a foreign foe, or may quell a domestic insurrection, however high he may raise his professional renown, and whatever martial glory he may win, still he is subject to the orders of the civil magistrate, and he and his army are always "subordinate to the civil power."

And hence it follows that whenever the President, (the civil magistrate,) in the discharge of his constitutional duty to "take care that the laws be faithfully executed," has occassing to use the army to aid him in the performance of that duty, he does not thereby lose his civil character and become a soldier, subject to military law, and liable to and become a souther, subject to minute y faw, and name to be tried by a court-martial, any more than does a civil court lose its legal and pacific nature, and become military and belligerent by calling out the power of the country to enforce its decrees. The civil magistrates, whether judicial or executive, must of necessity employ physical power to aid them in enforcing the laws, whenever they have to deal with disobedient and refractory subjects, and their legal power and right to do so is unquestionable. The right tegat power and right to do so is inducestonable. In Figure of the courts to call out the whole power of the country to enforce their judgments is as old as the common law; and the right of the President to use force in the performance of his legal duties is not only inherent in his office, but has been frequently recognized and aided by Congress. One striking example of this is the act of Congress of March 3, 1807, (2 Stat., 445.) which empowered the Presi-dent, without the intervention of any court, to use the marshal, aud, if he he insufficient, to use thearmy, summa-raily to coupl intervention of any to use the army summararily to expel intruders and squatters upon the public lands. And that power has been frequently exercised,

without, as far as I know, a question of its legality. To call as is sometimes done, the judiciary the civil power, and the President the military power, seems to me at once a mistake of fact and an abuse of language.

While the judiciary and the President, as departments of the General Government, are co-ordinate, equal in dig-nity and power, and equally trusted by the law in their respective spheres, there is, nevertheless, a marked diverin the character of their functions and their modes of sit action. The judiciary is, for the most part, passive. It rarely, if ever, takes the initiative; it seldom or never berarely, if ever, takes the initiative; it soldom or never be-gins an operation. Its great function is judgment, and, in the exercise of that function, it is confined almost exclu-sively to cases not selected by itself, but made and submit-ted by others. The President, on the contrary, by the very nature of his office, is active; he must often take the ini-tiative; he must begin operations. His great function is execution, for he is required by the Constitution (and he is the only department that is so required) to "take care that the laws (all the laws) be faithfully executed;" and in the exercise of that function his duties are co-extensive with the laws of the lawd. the laws of the land.

Often he comes to the aid of the judiciary in the execution of its judgments; and this is only a part, and a small part, of his constitutional duty, to take care that the laws he faithfully executed. I say it is a small part of his duty, because for every instance in which the President executes the judgment of a court there are a hundred instances in which he executes the law without the intervention of the judiciary, and without referring at all to its functions.

I have premised this much in order to show the separate and independent character of the several departments of our Government, and to indicate the inevitable differences in their modes of action and the characteristic diversity of the subjects upon which they operate; and all this as a foundation for the answers which I will now proceed to give to the particular questions propounded to me.

As to the first question, I am clearly of opinion that, in a time like the present, when the very existence of the na-tion is assailed by a great and dangerous insurrection, the President has the lawful discretionary power to arrest and hold in custody persons known to have criminal intercourse with the insurgents, or persons against whom there is probable cause for suspicion of such criminal couplicity. And I think this position can be maintained, in view of the principles already laid down, by a very plain argument.

The Constitution requires the President, before he enters upon the execution of his office, to take an oath that he "will faithfully execute the office of President of the United States, and will, to the best of his ability, pre-serve, protect, and defend the Constitution of the United States."

The duties of the office comprehend all the executive power of the nation, which is expressly vested in the President by the Constitution, (article 2, section 1,) and also all the powers which are specially delegated to the President, and yet are not, in their nature, executive powers; for example, the veto power, the treaty-making power, the appointing power, the pardoning power. These belong to inherent in the Crown. And yet the framers of our Constitution thought proper to preserve them and to vest them in the President, as necessary to the good govern-ment of the country. The executive powers are granted ment of the country. The executive powers are granted generally and without specification; the powers not executive are granted specially, and for purposes obvious in the context of the Constitution. And all these are embraced within the dutics of the President, and are clearly within that clause of his oath which requires him to "faithfully execute the office of President."

The last clause of the oath is peculiar to the President. All the other officers of Government are required to swear An the other oncers of dovernment are required to sweat must swear to "preserve, protect, and defend" it, which im-plies the power to perform what he is required in so solemn a manner to undertake. And then follows the broad and compendious injunction to "take are that the laws be faith-fully executed." And this injunction, embracing as it does And this injunction, embracing as it does all the laws-Constitution, treaties, statutes-is addressed to the President alone, and not to any other department or office of the Government. And this constitutes him, in a peculiar manner, and above all other officers, the guardian

of the Constitution-its preserver, protector, and defender. It is the plain duty of the President (and his peculiar duty, above and beyond all other departments of the Government) to preserve the Constitution and execute the laws all over the nation; and it is plainly impossible for him to perform this duty without putting down rebellion, insurrec-tion, and all unlawful combinations to resist the General The duty to suppress the insurrection being Government. obvious and imperative, the two acts of Congress, of 1795 and 1807, come to his aid, and furnish the physical force which he needs to suppress the insurrection and execute the laws. Those two acts authorize the President to employ for that purpose the militia, the army, and the navy. The argument may be briefly stated thus: It is the Pres-

The argument may be briefly stated thus: It is the Pres-ident's bounden duty to put down the insurrection, as, in the language of the act of 1795, the "combinations are too powerful to be suppressed by the ordinary course of judicial proceedings, or by the powers vested in the marshals." And this duty is imposed upon the President for the very reason that the courts and the marshals are too weak to perform it. The manner in which he shall perform that duty is not prescribed by any law, but the means of performing it are given in the plain language of the statutes, and they are all means of force—the militia, the Army, and the Navy. The end, the suppression of the insurrection, is required of him; the means and instruments to suppress it are lawfully in his hands; but the manner in which he shall use them is not prescribed, and could not be prescribed, without a forc-knowledge of all the future changes and contingencies of the insurrection. He is therefore necessarily thrown upon his discretion as to the manner in which he will use his means to meet the varying exigencies as they arise. If the insurgents assail the nation with an army he may find it best to meet them with an army ne may near the source-tion on the field of battle. If they seek to prolong the re-bellion and gather strength by intercourse with foreign nations, he may choose to guard the coast and close the ports with a navy, as one of the most efficient means to sup-mess the insurance in and the transmission of an press the insurrection. And if they employ spies and emis-saries to gather information, to forward secret supplies, and to excite new insurrections in aid of the original rebellion, he may find it both prudent and humane to arrest and im-prison them. And this may be done either for the purpose of bringing them to trial and condign punishment for their crimes, or they may be held in custody for the milder end of rendering them powerless for mischief until the exigency is past. In such a state of things the President must, of necessity,

be the sole judge, both of the exigency which requires him to act, and of the manner in which it is most prudent for him to employ the powers entrusted to him, to enable him to discharge his constitutional and legal duty; that is, to suppress the insurrection and excente the laws. And this Suppress the insurrection and excente the laws. And this discretionary power of the President is fully admitted by the Supreme Court in the case of Martin vs. Mott. (12 Whenton's Reports, page 19; 7 Curtis, 10.) This is a great power in the hands of the Chief Magis-trate; and because it is great, and is capable of being per-verted to avil ands its gristering has been doubted or do-

verted to evil ends, its existence has been doubted or denied. It is said to be dangerous in the hands of an ambinied. It is suid to be dangerous in the hands of an ambi-tious and wicked President, because he may use it for the purposes of oppression and tyranny. Yes, certainly it is dangerous; all power is dangerous, and for the all-pervad-ing reason that all power is liable to abuse; all the re-cipients of human power are men, not absolutely virtuous and wise. Still it is a power necessary to the piece and safety of the country, and undeniably belongs to the Gov-ernment, and therefore must be exercised by some depart-ment or officer thereof. ment or officer thereof.

Why should this power be denied to the President, on the ground of its liability to abuse, and not denied to the other departments on the same ground? Are they more exempt than he is from the frailties and vices of humanity? Or are they more trusted by the law than he is trusted, in their several spheres of action? If it be said that a President may be ambitious and unscrupulous, it may be said with equal truth that a legislature may be factious and with equal truth that a legislature may be factious and unprincipled, and a court may be venal and corrupt. But these are crimes never to be presumed, even against a private man, and much less against any high and highly trusted public functionary. They are crimes, however, recognized is such, and made punishable by the Constitu-tion; and wheever is guilty of them, whether a president, asenator, or a judge, is liable to impeachment and condem-netion. nation.

As to the second question : Having assumed, in answering the first question, that the President has the legal discretionary power to arrest and imprison persons who are guilty of holding criminal intercourse with men engaged in a great and dangerous insurrection, or persons suspected with "probable cause" of such criminal complicity, it might seem unnecessary to go into any prolonged argument to prove unnecessary to go into any prolonged argument to prove that in such a case the President is fully justified in refusing to obey a writ of habers corpus, issued by a court or judge, commanding him to produce the body of his prisener, and state when he took him, and by what authority, and for what cause he detains him in custody, and then yield him-self to judgment, "to do, submit to, and receive whatsoever the judge or court, awarding the writ, shall consider in that behaft?"

If it is true, as I have assumed, that the President and the Judiciary are co-ordinate departments of Government, and the one not subject to the other, I do not understand how it can be legally possible for a judge to issue a command to the President to come before him *ad subjiciendum*, that is to

submit implicitly to his judgment, and, in case of disobe-dience, treat him as a criminal, in contempt of a superior authority, and punish him as for a misdemeanor, by fine and imprisonment. It is no answer to say, as has sometimes been said, that although the writ of habcas corpus cannot be issued and enforced against the President limself, yet that it can be against any of inssubordinates; for that abandons the princi-ple assumed, of giving relief in "all cases" of imprisonment, by color of authority of the United States, and attempts to take an untenable distinction between the person of the Presdient and his office and legal power. The law takes no such distinction, for it is no respecter of persons. The President, in the arrest and imprisonment of men, must, almost always, act by subordinate agents; and yet the thing done is no less his act than if done by his own hand. But it is possible for Ins act than it done by his own hand. But it is possible for the President to be in the actual custody of a prisoner, taken in eivil war, or arrested on suspicion of being a secret agent and abettor of rebellion, and in that case the writ must be unavailing, unless it run against the President himself. Be-sides the whole subject-matter is political, and not judicial. The insurrection itself is purely political. Its object is to determ the valified accomments of this rates. destroy the political government of this nation, and to es-tablish another political government upon its ruins. And the President, as the chief civil magistrate of the nation, and the most active department of the Government, is emiand the host active department of the covernment, is emi-nently and exclusively political in all its principal functions. As the political chief of the nation, the Constitution charges him with its preservation, protection, and defense, and re quires him to take care that the laws be faithfully excented And in that character, and by the aid of the acts of Congress of 1205 cm 1407. of 1795 and 1807, he wages open war against armed rebelbio 199 and 1807, he wages open war against armed rebel-lion, and arrests and holds in safe custody those whom, in the exercise of his political discretion, he believes to be friends of and accomplices in the armed insurrection, which it is his especial political duty to suppress. He has no judi-cial powers. And the Judiciary Department has no politi-eal powers, and claims none, and therefore (as well as for the powers already agains of the armer is dense to be

call powers, and claims none, and therefore (as well as for other reasons already assigned) no court or judge can take cognizance of the political acts of the President, or under-take to revise and reverse his political decisions. The jurisdiction exercised under the writ of *habeas cor-pus* is, in the nature of an appeal, (4 Cr. 75.) for, as far as concerns the right of the prisoner, the whole object of the process is to re-examine and reverse or affirm the acts of the seriously affirmed that a judge at chambers can enterbe seriously affirmed that a judge, at chambers, can enter-tain an appeal, in any form, from a decision of the President of the United States, and especially in a case purely political.

There is but one sentence in the Constitution which mentions the writ of habeas corpus-art. 1. sec. 9, clanse 2which is in these words:

which is in these words: "The privileges of the writ of *habeas corpus* shall not be suspended unless when, in cases of rebellion or invasion, the public safety may require it." Very learned persons have differed widely about the meaning of this short sentence, and I am by no means con-fident that I fully understand it myself. The sententious language of the Constitution, in this particular, must, I suppose, be interpreted with reference to the origin of our neonle their historical relations to the mother construpeople, their historical relations to the mother country, and people, their historical relations to the mother country, and their inchoate political condition at the moment when onr Constitution was formed. At that time the United States, as a nation, had no common law of its own, and no statu-tory provision for the writ of *habeas corpus*. Still the peo-ple, English by descent, even while in open rebellion against the English Crown, claimed a sort of historical right to the forms of English law and the guarantees of English freedom. They knew that the English Government had, more than once assumed the nower to imprison whom had, more than once, assumed the power to imprison whom it would, and hold them, for an indefinite time, beyond the reach of judicial examination; and they desired, no doubt, reach of judicial examination; and they desired, no doubt, to interpose a guard against the like abuses in this coun-try; and hence the clause of the Constitution now under consideration. But we must try to construe the words, wague and indeterminate as they are, as we find them. "The privilege of the writ of habeas corpus shall not be suspended," &c. Does that mean that the writ itself shall not be issued, or that, being issued, the party shall derive no bonefit from it? Suspended—does that mean delayed, hung up for a time, or altogether denied? The writ of habeas corpus—which writ? In England there were many write called by that name, and used by the courts for the more convenient exercise of their various powers; and our more convenient exercise of their various powers; and our own courts now, by acts of Congress—the Judiciary act of 1789, sec. 14, and the act of March 2, 1833, sec. 7—have, I

1759, sec. 14, and the act of March 2, 1000, sec. i-mave, 1believe, equivalent powers. It has been decided by the Supreme Court, and I doubt not correctly—see Bollman Swarkwont's cuse (4 Cr, 93)— that "for the meaning of the term habeas corpus resort must be had to the common law, but the power to award the writ by any of the courts of the United States must be given by written law." And the same high Court, judging, no doubt, by the history of our people and the circumstances

of the times, has also decided that the writ of habeas corpus mentioned in the Constitution is the great writ ad subjiciendum.

That writ, in its nature, action, and objects, is tersely and accurately described by Sir William Blackstone. I adopt bis language, as found in his Commentaries, Book 3, p. 131:

"But he great and efficacious writ, in all manner of illegal confinement, is that of habeas corpus ad subjiciendum, directed to the person detaining another and commanding him to produce the body of the prisoner, with the day and cause of his caption and detention, ad faciendum, subjiciendum et recipiendum, to do, submit to, and receive whatsoever the judge or court awarding such writ shall consider in that behalf. This is a high prerogative writ, and therefore by the common law, issuing out of the Court of King's Bench, not only in term time, but also during the vacation by a flat from the Chief Justice or any other of the judges, and running into all parts of the King's dominions; for the King is at all times entitled to have an account why the liberty of any of his subjects is restrained, wherever that restraint may be inflicted."

Such is the writ of habeas corpus, of which the Constitution doclares that the privilege thereof shall not be suspended except when, in cases of reliellion or invasion, the public safety may require it. But the Constitution is silent as to who may suspend it when the contingency happens. I am aware that it has been declared by the Supreme Court that "if, at any time, the public safety should require the suspension of the powers vested by this act (meaning the judiciary act of 1789, section 14) in the courts of the United States, it is for the Legislature to say so. That question depends upon political considerations, on which the Legislature is to decide." Upon this I remark only that the Constitution is older than the judiciary act, and yet it speaks of the privilege of the writ of habeas corpus as a thing in existence; it is in general terms, and does not speak with particular reference to powers which might or might not be granted by a future act of Congress. Besides, I take it for certain that, in the common course of legislation, Congress has power, at any time, to repeal the judiciary act of 1789 and the act of 1833 (which grants to the courts and to the judges the power to issue the writs) withult waiting for a rebellion or invasion, and a consequent public necessity, to justify, under the Constitution, the suspension of the privilege of the writ of habeas corpus. The court does not speak of suspending the privilege of

The court does not speak of suspending the privilege of the writ, but of suspending the powers vested in the court by the act. The power to issue a writ can hardly be called a privilege; yet the right of an individual to invoke the protection of his Government in that form may well be designated by that name. And I should infer, with a good deal of confidence, that the Court meant to speak only of its own powers, and not of the privilege of individuals, but for the fact that the court ascribe the power to suspend to the Legislature upon political grounds. It says "that question depends upon political considerations, on which the Legislature is to decide." Now, I had supposed that questhe depend upon political considerations, inasmuch as the President, in his constitutional and official duties, is quite as political as is the Congress, and has daily occasion in the common routine of affairs to determine questions upon political considerations alone.

If by the phrase the suspension of the privilege of the writ of habeas corpus, we must understand a repeal of all power to issue the writ, then I freely admit that none but Congress can do it. But if we are at liberty to understand the phrase to mean, that in case of a great and dangerous rebellion like the present, the public safety requires the arrest and confinement of persons implicated in that rebellion. I as freely declare the opinion that the President has lawful power to suspend the privilege of persons arrested under such circumstances; for he is especially charged by the Constitution with the "public safety," and he is the sole judge of the emergency which requires his prompt action. This power in the President is no part of his ordinary duty in time of peace; it is temporary and exceptional, and was

This power in the President is no part of his ordinary duty in time of peace; it is temporary and exceptional, and was intended only to meet a pressing emergency, when the judiciary is found to be too weak to insure the public safety when (in the language of the act of Congress) there are "combinations too powerful to be suppressed by the ordinary course of judicial proceedings, or by the powers vested in the marshals." Then and not till then, has he the lawful authority to call to his aid the military power of the nation, and with that power perform his great legal and constitutional duty to suppress the insurrection. And shall it be said that when he has fought and captured the insurgent army, and has seized their secret spies and emissaries, he is bound to bring their bodies before any judge who may send him a writ of *habeas corpus*, "to do, submit to, and receive whatsoever the said judge shall consider in that behalf?"

I deny that he is under any obligation to obey such a writ, issued under such circumstances. And in making this denial I do but follow the highest judicial authority of 11

1

the nation. In the case of Luther vs. Borden, (commonly called the Rhode Island case,) reported in 7 Howard, page I, the Supreme Court discussed several of the most important topics treated of in this opinion, and among them the power of the President alone to decide whether the exigency exists anthorizing him to call out the militia under the act of 1795. The court affirmed the power of the Court dent in that respect, and denied the power of the Oresident in that respect, and denied the power of the ourt to examine and adjudge his proceedings. The opinion of the court, delivered by the learned Chief Justice Taney, declares that if the court had that power, "then it would become the duty of the court (provided that it came to the conclusion that the President had decide incorrectly) to discharge those who were arrested or detained by the troops in the service of the United States, or the Government which the President was endeavoring to maintain. If (says that learned court) the judicial power extends so far, the guarantee contained in the Constitution of the United States (meaning, of course, protection against insurrection) is a guarantee of anarchy and not of order."

Whatever I have said about the suspension of the privilege of the writ of *habeas* corgues has been said in deterence to the opinions of others, and not because I myself thought it necessary to treat of that subject at all in reference to the present posture of our national affairs. For, not doubting the power of the President to capture and hold by force insurgents in open arms against the Government, and to arrest and imprison their suspected accomplices, I never thought of first suspending the writ of *habeas corpus* any more than I thought of first suspending the writ of replevin before seizing arms and munitions destined for the euemy.

The power to do these things is in the hand of the President, placed there by the Constitution and the statute law as a sacred trust, to be used by him in his best discretion in the performance of his great first duty—to preserve, protect, and defend the Constitution. And for any breach of that trust he is responsible before the high court of impeachment, and before no other human tribunal.

The powers of the President falling within this general class have been several times considered by the judiciary, and have, I believe, been uniformly sustained, without materially varying from the doctrines laid down in this opinion. I content myself with a simple reference to the cases, without encumbering this document, already too long, with copious extracts. (The Rhode Island case, 7 Howard, page 1; Fleming vs. Page, 9 Howard, page 615; Gross vs. Harrison, 16 Howard, page 189; the Santissima Trinidad, 7 Wheaton, page 305; Martin vs. Mott, 12 Wheaton, page 29.)

To my mind it is not very important whether we call a particular power exercised by the President a peace power or a war power, for undeubtedly he is armed with both. He is the chief civil magistrate of the nation, and, being such, and because he is such, he is the constitutional commander-in-chief of the army and navy; and thus, within the limits of the Constitution, he rules in peace and commands in war, and at this moment he is in the full exercise of all the functions belonging to both these characters. The evil administration is still going on in its peaceful course, and yet we are in the midst of war—a war in which the enemy is, for the present, dominant in many States, and has his secret allies and accomplices scattered through many other States which are still loyal and true; a war all the more dangerous, ad more needing jealous vigilance and prompt action, because it is an internecino and not an international war.

This, sir, is my opinion, the result of my best reflections upon the questions propounded by you. Such as it is, it is submitted with all possible respect, by your obedient servant, EDWARD BATES, Attorney General.

To the President.

#### VIEW OF HORACE BINNEY.

From his pamphlet entitled "The Privilege of the Writ of *Habeas Corpus* under the Constitution," pages 51, 52:

In this matter of suspension of the privilege of the Writ of habeas corpus, the Constitution of the United States stands in the place of the English act of Parliament. It ordains the suspension in the conditioned cases, by the act of the competent department—as Parliament does from time to time. Neither is mandatory in suspending, but only authoritative. Each leaves discretion to the executive power. The difference is, that Parliament limits a timeand provides for the effect by technical terms. The Constitution connects the suspension with the time of rebellion, and provides for the effect, as it did for the privilege, by words that comprehend the right, and deny for a season the eniovment of it.

enjoyment of it. It is further objected, that this is a most dangerous power. It is, fortunately, confined to most dangerous times. In such times the people generally are willing, and are often compelled, to give up for a season, a portion of their freedom to preserve the rest: and fortunately again, it is that portion of the people, for the most part, who like to live on the margin of disobedience to the laws, whose freedom is most in danger. The rest are rarely in want of a *kabeas corpus*.

VIEW OF PROFESSOR THEOPHILUS PARSONS.

The Boston Daily Advertiser of June 5, 1861, contained a summary of the lecture on the question raised respecting the right and power of the Executive branch of the Government to suspend the writ of haleas corpus in certain emergencies, of which this is a part:

The Constitution of the United States, art. 1, sec. 9, n. 2provides that "the privilege of the writ of habeas corpus shall not be suspended unless when, in cases of rebellion or invasion, the public safety may require it." And many of the State constitutions have similar provisions. A fair inference from this is that the right to habeas corpus may be suspended, or, what is the same thing, martial law may be declared and exercised "in cases of rebellion or invasion, when the public safety may require it."

The first and most important question is, who may decide when the exigency occurs, and who may, it is occurs, dechare martial law. On this point I have myself no doubt. The clause on this subject is contained in the first article of the Constitution, and this article relates principally to Congress. Nor can there be any doubt that Congress may, when the necessity occurs, suspend the right to the writ of *habeas* corpus, or, which is the same thing, declare or authorize martial law. The question is, has the President this power? The Constitution does not expressly give this power? The Constitution does not expressly give this it does make this express reservation as to some of the provisions contained in the article. This may be a mere accidental omission, but it seems to me more reasonable and more consonant with the principles of legal interpretation to infer from it an absence of intention to confine it to Congress. And I am confirmed in this opinion by the nature of the case.

The very instances specified as those in which the right to *habcas corpus* may be suspended (invasion and rebellion) are precisely those in which the reasons for doing so may come suddenly, the necessity of determination be immediate, and a certainty exist that the suspension will be useless, and the whole mischief which the suspension might prevent take place if there be any delay. To guard against the suspension by limiting the cases, as is done, securs to me wise; to obstruct it by requiring the delay necessarily arising from legislattive action would seem to be unreasonable. It is true that my construction gives to the President, in the two cases of rebellion and invasion, a vast power; but so is all military power. It is a vast power to send into a rebellions district 15,000 soldiers, as Washingrou did, whose duty it would be to meet the rebels, and, if necessary, kill as many as they could. Ent it was a power which belonged to him, of necessity, as President; and so, I think, did the power of martial law. If it did not, then, when his troops had captured the armed rebels whom they were sent to subdue, the nearest magistrate who could issue a writ of *halocas corpus* might have summoned the officer having them in charge to bring them before lim, and might have liberated them at once to fight again, and this as often as they were captured, until a law could be passed by Congress.

If the power belongs to the President, he may exercise it at his discretion, when either invasion or rebellion occurs, subject, however, to two qualifications. One, a universal one, applicable to his exercise of every power. If he abuses it, or exercises it wrongfully, he is liable to impeachment. The other is more a matter of discretion or propriety. I suppose that he would of course report his doings in such a matter to Congress when he could, and be governed by their action.

My conclusion is, therefore, that in case of invasion from abroad or rebellion at home, the President may declare, or exercise or authorize, martial law at his discretion.

# ARREST OF CLEMENT L. VALLANDIGHAM.

Major General BURNSIDE, commanding Department of the Ohio, issued on the 13th of April, 1863, General Order No. 38, announcing that hereafter "all persons found within our lines who commit acts for the benefit of the enemies of our country will be tried as spies or traitors, and, if convicted, will suffer death."

It was added: "The habit of declaring sympathies for the enemy will not be allowed in this department. Persons committing such offences will be at once arrested, with a view to being tried, as above stated, or sent beyond our lines into the lines of their friends. It must be distinctly understood that treason, expressed or implied, will not be tolerated in this department."

1863, May 4—Mr. VALLANDIGHAM was arrested for violation of this order—charged with "publicly expressing sympathy for those in arms against the Government of the United States, and declaring disloyal sentiments and opinions, with the object and purpose of weakening the power of the Government in its efforts to suppress an unlawful rebellion." The specification alluded to his speech on or about May 1, 1863, at Mount Vernon, O.

May 16—The evidence having been heard, the Court—Brigadier General R. B. Potter presiding—found him guilty of the charge, and not guilty as to part, and guilty as to part, of the specification.

He was sentenced to be placed in close confinement in some fortress of the United States, to be designated by the commanding officer of this Department, there to be kept during the continuance of the war. General BURNSIDE designated Fort Warren, Boston harbor.

May 19—The PRESIDENT directed \* that Mr. Vallandigham be taken, under secure guard, to the headquarters of General Rosecrans, to be put by him beyond our military lines; and that, in case of his return within our lines, he be arrested and kept in close custody for the term specified in his sentence.

This order was executed, but Mr. Vallandigham very soon ran the blockade at Wilmington, N. C., and went to Canada, remaining at Windsor.

On the 5th of May, 1863, Mr. Vallandigham applied through counsel to Judge Leavitt, of the Circuit Court of the United States at Cincinnati, for a writ of *habeas corpus*, to which Gen. Burnside responded with a letter detailing the case and justifying his action. The application was argued at length, and was refused by the judge, who said that the legality of the arrest depends upon the extent of the necessity for making it, and that was to be determined by the military commander. He added:

Mon should know and lay the truth to heart, that there is a course of couldet not involving overt treason, and not therefore subject to punishment as such, which, nevertheless, implies moral guilt, and a gross offence against the country. Those who live under the protection and enjoy<sup>4</sup> the blessings of our benignant Government, must learn that they cannot stab its vitals with impunity. If they cherish

\* This is a copy of the order: UNITED STATES MILITARY TELEGRAPH, [Cipher.] May 19, 1863.

[By telegraph from Washington, 9.40 p. m., 1863.] To Maj. Gen. BURNSIDE,

Commanding Department of Ohio:

SIR: The President directs that, without delay, you send C. L. Vallandigham, under secure guard, to the headquarters of General Rosecrans, to be put by him beyond our military lines, and in case of his return within our lines he be arrested and kept in close custody for the term specified in his sentence.

By order of the President:

E. R. S. CANBY, Brigadier General and A. A. G.

hatred and hostility to it, and desire its subversion, let them withdraw from its jurisdiction, and seek the fellowship and windraw from its jurisdiction, and seek the feilowship and protection of those with whom they are in sympathy. If they remain with us, while they are not of us, they must be subject to such a course of dealing as the great law of self-preservation prescribes and will enforce. And let them not complain if the stringent doctrine of military necessity should find them to be the legitimate subjects of its action. I have no fear that the recognition of this doctrine will lead to an arbitrary invasion of the nersonal scourity or preserval to an arbitrary invasion of the personal security or personal To an around the citizen. It is not not be personal security of personal liberty of the citizen. It is rare indeed that a charge of dis-loyalty will be made on insufficient grounds. But if there should be an occasional mistake, such an occurrence is not to be put in competition with the preservation of the nation; and I confess I am but little moved by the eloquent appeals of those who, while they indignantly denounce violation of personal liberty, look with no horror upon a despotism as unmitigated as the world has ever witnessed.

CORRESPONDENCE BETWEEN NEW YORK DEMO-CRATS AND PRESIDENT LINCOLN.

Letter of the Committee and Resolutions. ALBANY, May 19, 1863.

To his Excellency the PRESIDENT OF THE UNITED STATES:

The undersigned, officers of a public meeting held at the city of Albany on the 16th day of May instant, herewith transmit to your Excellency a copy of the resolutions adopted at the said meeting, and respectfully request your earnest consideration of them. They deem it proper on their personal responsibility to state that the meeting was one of the most respectable as to numbers and character, and one of the most earnest in the support of the Union, ever held in this city.

Yours, with great regard.

b. A. J.
b. B. A. STUS CORNING, President.
ELI PERRY, Vice President.
PETER GANSEVOORT, Vice President.
PETER MONTEITII, Vice President.
SAMUEL W, GIBBS, Vice President.
J. B. SANDERS, Vice President.
WILLIAM SEYMOUR, Vice President.
WILLIAM SEYMOUR, Vice President.
J. B. SANDERS, Vice President.
J. W. N. RADCLIFFE, Vice President.
D. V. N. RADCLIFFE, Vice President.
WILLIAM J. R. RCE, Sceretary.
EDWARD NEWCOMB, Secretary.
M. A. NOLAN, Sceretary.
JOHN R. NESSEL, Secretary.
C. W. WEEKS, Sceretary.
Med at the Meeting held in Albann, N. F.

Resolutions adopted at the Meeting held in Albany, N. Y., on the 16th day of May, 1863.

Resolved. That the Democrats of New York point to their uniform course of action during the two years of civil war uniform course of action during the two years of civil war through which we have passed, to the alacrity which they have evinced in filling the ranks of the army, to their con-tributions and sacrifices; as the evidence of their patriot-ism and devotion to the cause of our imperilled country. Never in the history of civil wars has a government been sustained with such ample resources of means and men as the need have relutrivily need in the hand of the the people have voluntarily placed in the hands of this Administration.

Resolved, That, as Democrats, we are determined to maintain this patriotic attitude, and despite of adverse and disheartening circumstances, to devote all our energies to sustain the cause of the Union; to secure peace through victory, and to bring back the restoration of all the States under the safegnard of the Constitution. Resolved, That while we will not consent to be misappre-

Acsourced, That while we will not consent to be misappre-bended upon these points, we are determined not to be misunderstood in regard to others not less essential. We demand that the Administration shall be true to the Con-stitution; shall recognize and maintain the rights of the States and the liberties of the citizen; shall everywhere, States and the inderfues of the critical's shall everywhere, outside of the lines of necessary military occupation and the scenes of insurrection, exert all its powers to maintain the supremacy of the civil over military law. *Resolved*, That, in view of these principles, we denounce the recent assumption of a military commander to seize and try a citizen of Ohio, Clement L. Vallandigham, for no

other reason than words addressed to a public meeting, in criticism of the course of the Administration and in condemnation of the military orders of that general.

demnation of the minitary orders of that general. *Resolved*, That this assumption of power by a military tribunal, if successfully asserted, not only abrogates the right of the people to assemble and discuss the affairs of government, the liberty of speech and of the press, the right of trial by jury, the law of evidence, and the privilege of *habeas corpus*, but it strikes a fatal blow at the supremaey of law and the authority of the State and Federal Constitutions.

Resolved, That the Constitution of the United States— the supreme law of the land—has defined the crime of treason against the United States to consist "only in levying war against them, or adhering to their enemies, giving them aid and comfort," and has provided that "no person shall be convicted of treason, unless on the testimony of two witnesses to the same overt act, or on confession in open court." And it further provides that "no person shall be lock it operations are agained or otherwise informers ensign be held to answer for a capital or otherwise infamous crime, be need to answer for a capital or otherwise infamous crime, unless on a presentment or indictment of a grand jury, ex-cept in cases arising in the land and naval forces, or in the militia, when in actual service in time of war or public danger;" and further, that "in all criminal prosecutions, the accused shall enjoy the right of a speedy and public trial by an impartial jury of the State and district wherein the crime was committed." *Evolution* That these enformance of the with

Resolved, That these safeguards of the rights of the citi-Resource, that these sateguards of the rights of the chi-zen against the pretensions of arbitrary power were in-tended more especially for his protection in times of civil commotion. They were secured substantially to the Eng-lish people, after years of protracted civil war, and were adopted into our Constitution at the close of the revolu-tion. They have stool the test of seventy-six years of trial, under our republican system, under circumstances which show that, while they constitute the foundation of all free government, they are the elements of the enduring stability of the republic. Resolved, That in adopting the language of Daniel Webs-

ter, we declare, "it is the ancient and undoubted prerogative of this people to canvass public measures and the merits of public men." It is a "homebred right," a fireside and cabin in the nation. It is as undoubted as the right of and cabin in the nation. It is as undoubted as the right of breathing the air or walking on the earth. Belonging to private life as a right, it belongs to public life as a duty, and it is the last duty which those whose representatives we are shall find us to abandon. Aiming at all times to be courteous and temperate in its use, except when the right itself is questioned, we shall place ourselves on the extreme beam denue if one one wight course life and a submer boundary of our own right and bid defiance to any arm that would move us from our ground. "This high consti-tutional privilege we shall defend and exercise in all places -in time of peace, in time of war, and at all times. Living, we shall assert it; and should we leave no other inheritance to our children, by the blessing of God we will leave them the inheritance of free principles, and the example of a manly, independent, and constitutional defence of them." *Resolved*. That in the election of Governor Seymour, the

people of this State, by an emphatic majority, declared their condemnation of the system of arbitrary arrests and their determination to stand by the Constitution. That the revival of this lawless system can have but one result: to di-vide and distract the North, and destroy its confidence in the purposes of the Administration. That we deprecate it as an element of confusion at home, of weakness to our as an element of contact of a lower the estimate of American character and magnify the apparent peril of our cause abroad. And that, regarding the blow struck at a citizen of Oldo as aimed at the rights of every citizen of the North, we denounce it as against the spirit of our laws and Constitution, and most earnestly call upon the Presi-dent of the United States to reverse the action of the military tribunal which has passed a "cruel and unusual punishment" upon the party arrested, prohibited in terms by the Constitution, and to restore him to the liberty of which

the Constitution, and to resolve that is the constitution, and to resolve the has been deprived. Resolved, That the president, vice-presidents, and secre-tary of this meeting be requested to transmit a copy of these resolutions to his excellency the President of the United States, with the assurance of this meeting of their hearty and earnest desire to support the Government in every con-dimensional and heaving measure to supports the existing stitutional and lawful measure to suppress the existing rebellion.

## PRESIDENT LINCOLN'S REPLY.

EXECUTIVE MANSION, WASHINGTON, June 12, 1863.

Hon. ERASTUS CORNING, and others :

GENTLEMEN : Your letter of May 19, inclosing the resolutions of a public meeting held at

Albany, New York, on the 16th of the same month, was received several days ago.

The resolutions as I understand them are resolvable into two propositions-first, the expression of a purpose to sustain the cause of the Union, to secure peace through victory, and to support the Administration in every constitutional and lawful measure to suppress the rebellion; and secondly, a declaration of censure upon the Administration for supposed unconstitutional action, such as the making of military arrests. And, from the two propositions, a third is deduced, which is, that the gentlemen composing the meeting are resolved on doing their part to maintain our common Government and country, despite the folly or wickedness, as they may conceive, of any Administration. This position is eminently patriotic, and as such, I thank the meeting, and congratulate the nation for it. My own purpose is the same; so that the meeting and myself have a common object, and can have no difference, except in the choice of means or measures for effecting that object.

And here I ought to close this paper, and would close it, if there were no apprehension that more injurious consequences than any merely personal to myself might follow the censures systematically cast upon me for doing what, in my view of duty, I could not forbear. The resolutions promise to support me in every constitutional and lawful measure to suppress the rebellion; and I have not knowingly employed, nor shall knowingly employ, any other. But the meeting, by their resolutions, assert and argue that certain military arrests, and proceedings following them, for which I am ultimately responsible, are unconstitutional. I think they are not. The resolutions quote from the Constitution the definition of treason, and also the limiting safeguards and guarantees therein provided for the citizen on trials of treason, and on his being held to answer for capital or otherwise infamous crimes, and, in criminal prosecutions, his right to a speedy and public trial by an impartial jury. They proceed to resolve "that these safeguards of the rights of the citizen against the pretensions of arbitrary power were intended more especially for his protection in times of civil commotion." And, apparently to demonstrate the proposition, the resolutions proceed: "They were secured substantially to the English people after years of protracted civil war, and were adopted into our Constitution at the close of the revolution." Would not the demonstration have been better, if it could have been truly said that these safeguards had been adopted and applied during the civil wars and during dur revolution, instead of after the one and at the close of the other ? I, too, am devotedly for them after civil war, and before civil war, and at all times, "except when, in cases of rebellion or invasion, the public safety may require" their suspension. The resolutions proceed to tell us that these safeguards "have stood the test of seventy-six years of trial, under our republican system, under circum-stances which show that while they constitute

the elements of the enduring stability of the republic." No one denies that they have so stood the test up to the beginning of the present rebellion, if we except a certain occurrence at New Orleans; nor does any one question that they will stand the same test much longer after the rebellion closes. But these provisions of the Constitution have no application to the case we have in hand, because the arrests complained of were not made for treason-that is, not for the treason defined in the Constitution, and upon the conviction of which the punishment is death-nor yet were they made to hold persons to answer for any capital or otherwise infamous crimes; nor were the proceedings following, in any constitutional or legal sense, "criminal prosecutions." The arrests were made on totally different grounds, and the proceedings following accorded with the grounds of the arrests. Let us consider the real case with which we are dealing, and apply to it the parts of the Constitution plainly made for such cases.

Prior to my installation here it had been inculcated that any State had a lawful right to secede from the national Union, and that it would be expedient to exercise the right whenever the devotees of the doctrine should fail to elect a President to their own liking. I was elected contrary to their liking; and, accordingly, so far as it was legally possible, they had taken seven States out of the Union, had seized many of the United States forts, and had fired upon the United States flag, all before I was inaugurated, and, of course, before I had done any official act whatever. The rebellion thus began soon ran into the present civil war; and, in certain respects, it began on very unequal terms between the parties. The insurgents had been preparing for it more than thirty years, while the Government had taken no steps to resist them. The former had carefully considered all the means which could be turned to their account. It undoubtedly was a well-pondered reliance with them that in their own unrestricted efforts to destroy Union, Constitution, and law, all together, the Government would, in great degree, be restrained by the same Constitution and law from arresting their progress. Their sympathizers pervaded all departments of the Government and nearly all communities of the people. From this material, under cover of "liberty of speech," "liberty of the press," and "habeas corpus," they hoped to keep on foot amongst us a most efficient corps of spies, informers, suppliers, and aiders and abettors of their cause in a thousand ways. They knew that in times such as they were inaugurating, by the Constitution itself, the "habeas corpus" might be suspended; but they also knew they had friends who would make a question as to who was to suspend it; meanwhile their spies and others might remain at large to help on their cause. Or if, as has happened, the Executive should suspend the writ, without ruinous waste of time, instances of arresting innocent persons might occur, as are always likely to occur in such cases; and then a clamor could be raised in regard to this, which might be, at the foundation of all free government, they are | least, of some service to the insurgent cause. It

164

needed no very keen perception to discover this | part of the enemy's programme, so soon as by open hostilities their machinery was fairly put Yet, thoroughly imbued with a in motion. reverence for the guaranteed rights of individuals, I was slow to adopt the strong measures which by degrees I have been forced to regard as being within the exceptions of the Constitution, and as indispensable to the public safety. Nothing is better known to history than that courts of justice are utterly incompetent to such cases. Civil courts are organized chiefly for trials of individuals, or, at most, a few individuals acting in concert; and this in quiet times, and on charges of crimes well defined in the law. Even in times of peace bands of horse-thieves and robbers frequently grow too numerous and powerful for ordinary courts of justice. But what comparison, in numbers, have such bands ever borne to the insurgent sympathizers even in many of the loyal States? Again, a jury too frequently has at least one member more ready to hang the panel than to hang the traitor. And yet, again, he who dissuades one man from volunteering, or induces one soldier to desert, weakens the Union cause as much as he who kills a Union soldier in battle. Yet this dissuasion or inducement may be so conducted as to be no defined crime of which any civil court would take cognizance.

Ours is a case of rebellion-so called by the resolutions before me-in fact, a clear, flagrant, and gigantic case of rebellion; and the provi-sion of the Constitution that "the privilege of the writ of habcas corpus shall not be suspended, unless when in cases of rebellion or invasion, the public safety may require it," is the provision which specially applies to our present case. This provision plainly attests the understanding of those who made the Constitution, that ordinary courts of justice are in-adequate to "cases of rebellion"—attests their purpose that, in such cases, men may be held in custody whom the courts, acting on ordinary rules, would discharge. Habeas corpus does not discharge men who are proved to be guilty of defined crime; and its suspension is allowed by the Constitution on purpose that men may be arrested and held who cannot be proved to be guilty of defined crime, "when, in cases of rebellion or invasion, the public safety may require it."

This is precisely our present case—a case of rebellion, wherein the public safety does require the suspension. Indeed, arrests by process of courts, and arrests in cases of rebellion do not proceed altogether upon the same basis. The former is directed at the small percentage of ordinary and continuous perpetration of crime, while the latter is directed at sudden and extensive uprisings against the Government, which, at most, will succeed or fail in no great length of time. In the latter case, arrests are made, not so much for what has been done, as for what probably would be done. The latter is more for the preventive and less for the vindictive than the former. In such cases the purposes of men are much more easily understood than in cases of ordinary crime. The man who stands by and says nothing when the reason. Mr. Vallandigham avows his hostility

peril of his Government is discussed, cannot be misunderstood. If not hindered, he is sure to help the enemy; much more, if he talks ambiguously-talks for his country with "buts" and "ifs" and "ands." Of how little value the constitutional provisions I have quoted will be rendered, if arrests shall never be made until defined crimes shall have been committed, may be illustrated by a few notable examples. General John C. Breckinridge, General Robert E. Lee, General Joseph E. Johnston, General John B. Magruder, General William B. Preston, General Simon B. Buckner, and Commodore Franklin Buchanan, now occupying the very highest places in the rebel war service, were all within the power of the Government since the rebellion began, and were nearly as well known to be traitors then as now. Unquestionably if we had seized and held them, the insurgent cause would be much weaker. But no one of them had then committed any crime defined in the law. Every one of them, if arrested, would have been discharged on habeas corpus were the writ allowed to operate. In view of these and similar cases, I think the time not unlikely to come when I shall be blamed for having made too few arrests rather than too many.

By the third resolution the meeting indicate their opinion that military arrests may be constitutional in localities where rebellion actually exists, but that such arrests are unconstitutional in localities where rebellion or insurrection does not actually exist. They insist that such arrests shall not be made "outside of the lines of necessary military occupation, and the scenes of insurrection." Inasmuch, however, as the Constitution itself makes no such distinction, I am unable to believe that there is any such constitutional distinction. I concede that the class of arrests complained of can be constitutional only when, in cases of rebellion or invasion, the public safety may require them; and I insist that in such cases they are constitutional wherever the public safety does require them; as well in places to which they may prevent the rebellion extending as in those where it may be already prevailing; as well where they may restrain mischievious interference with the raising and supplying of armies to suppress the rebellion, as where the rebellion may actually be; as well where they may restrain the enticing men out of the army, as where they would prevent mutiny in the army; equally constitutional at all places where they will conduce to the public safety, as against the dangers of rebellion or invasion. Take the peculiar case mentioned by the meeting. It is asserted, in substance, that Mr. Vallandigham was, by a military commander, seized and tried "for no other reason than words addressed to a public meeting, in criticism of the course of the Administration, and in condemnation of the military orders of the general." Now, if there be no mistake about this; if this assertion is the truth and the whole truth; if there was no other reason for the arrest, then I concede that the arrest was wrong. But the arrest, as I understand, was made for a very different

arrest was made because he was laboring, with some effect, to prevent the raising of troops; to encourage desertions from the army; and to leave the rebellion without an adequate military force to suppress it. He was not arrested bccause he was damaging the political prospects of the Administration, or the personal interests of the commanding general, but because he was damaging the army, upon the existence and vigor of which the life of the nation depends. He was warring upon the military, and this gave the military constitutional jurisdiction to lay hands upon him. If Mr. Vallandigham was not damaging the military power of the country, then his arrest was made on mistake of fact, which I would be glad to correct on reasonably satisfactory evidence.

I understand the meeting, whose resolutions I am considering, to be in favor of suppressing the rebellion by military force—by armies. Long experience has shown that armies cannot be maintained unless desertion shall be punished by the severe penalty of death. The case requires, and the law and the Constitution sanction, this punishment. Must I shoot a simple-minded soldier boy who deserts, while I must not touch a hair of a wily agitator who induces him to desert? This is none the less injurious when effected by getting a father, or brother, or friend, into a public meeting, and there working upon his feelings till he is persuaded to write the soldier boy that he is fighting in a bad cause, for a wicked Administration of a contemptible Government, too weak to arrest and punish him if he shall desert. I think that, in such a case, to silence the agitator and save the boy is not only constitutional, but withal a great mercy.

If I be wrong on this question of constitutional power, my error lies in believing that certain proceedings are constitutional when, in cases of rebellion or invasion, the public safety requires them, which would not be constitutional when, in absence of rebellion or invasion, the public safety does not require them: in other words, that the Constitution is not, in its application, in all respects the same, in cases of rebellion or invasion involving the public safety, as it is in times of profound peace and public security. The Constitution itself makes the distinction; and I can no more be persuaded that the Government can constitutionally take no strong measures in time of rebellion, because it can be shown that the same could not be lawfully taken in time of peace, than I can be persuaded that a particular drug is not good medicine for a sick man because it can be shown to not be good food for a well one. Nor am I able to appreciate the danger apprehended by the meeting, that the American people will, by means of military arrests during the rebellion, lose the right of public discussion, the liberty of speech and the press, the law of evidence, trial by jury, and habeas corpus, throughout the indefinite peaceful future, which I trust lies before them, any more than I am able to believe that a man could contract so strong an appetite for emetics during tem-

to the war on the part of the Union; and his | them during the remainder of his healthful life.

In giving the resolutions that earnest considcration which you request of me, I cannot overlook the fact that the meeting speak as "Democrats." ' Nor can I, with full respect for their known intelligence, and the fairly presumed deliberation with which they prepared their resolutions, be permitted to suppose that this occurred by accident, or in any way other than that they preferred to designate themselves "Democrats" rather than "American citizens." In this time of national peril I would have preferred to meet you on a level one step higher than any party platform; because I am sure that, from such more elevated position, we could do better battle for the country we all love than we possibly can from those lower ones where, from the force of habit, the prejudices of the past, and selfish hopes of the future, we are sure to expend much of our ingenuity and strength in finding fault with, and aming blows at, each other. But, since you have de-nied me this, I will yet be thankful, for the country's sake, that not all Democrats have done so. He on whose discretionary judgment Mr. Vallandigham was arrested and tried is a Democrat, having no old party affinity with me; and the judge who rejected the constitutional view expressed in these resolutions, by refusing to discharge Mr. Vallandigham on habeas corpus, is a Democrat of better days than these, having received his judicial mantle at the hands of President Jackson. And still more, of all those Democrats who are nobly exposing their lives and shedding their blood on the battle-field, I have learned that many approve the course taken with Mr. Vallandigham, while I have not heard of a single one condemning it. I cannot assert that there are none such. And the name of President Jackson recalls an instance of pertinent history. After the battle of New Orleans, and while the fact that the treaty of peace had been concluded was well known in the city, but before official knowledge of it had arrived, General Jackson still maintained martial or military law. Now, that it could be said the war was over, the clamor against martial law, which had existed from the first, grew more furious. Among other things a Mr. Louaillier published denunciatory newspaper article. General Jackson arrested him. A lawyer by the name of Morel procured the U.S. Judge Hall to order a writ of habeas corpus to relieve Mr. Louaillier. General Jackson arrested both the lawyer and the judge. A Mr. Hollander ventured to say of some part of the matter that "it was a dirty trick." General Jackson arrested him. When the officer undertook to serve the writ of habeas corpus, General Jackson took it from him, and sent him away with a copy. Holding the judge in custody a few days, the general sent him beyond the limits of his encampment, and set him at liberty, with an order to remain till the ratification of peace should be regularly an-nounced, or until the British should have left the southern coast. A day or two more elapsed, the ratification of the treaty of peace was regularly announced, and the judge and others were porary illness as to persist in feeding upon fully liberated. A few days more and the judge

called General Jackson into court and fined him | \$1,000 for having arrested him and the others named. The General paid the fine, and there the matter rested for nearly thirty years, when Congress refunded principal and interest. The late Senator Douglas, then in the House of Representatives, took a leading part in the debates, in which the constitutional question was much discussed. . I am not prepared to say whom the journals would show to have voted for the measure.

It may be remarked : First, that we had the same Constitution then as now; secondly, that we then had a case of invasion, and now we have a case of rebellion ; and, thirdly, that the permanent right of the people to public discussion, the liberty of speech and of the press, the trial by jury, the law of evidence, and the habeas corpus, suffered no detriment whatever by that conduct of Gen. Jackson, or its subsequent approval by the American Congress.

And yet, let me say, that in my own discre-tion, I do not know whether I would have ordered the arrest of Mr. Vallandigham. While I cannot shift the responsibility from myself, I hold that, as a general rule, the commander in the field is the better judge of the necessity in any particular case. Of course, I must practice a general directory and revisory power in the matter.

One of the resolutions expresses the opinion of the meeting that arbitrary arrests will have the effect to divide and distract those who should be united in suppressing the rebellion, and I am specifically called on to discharge Mr. Vallandigham. I regard this as, at least, a fair appeal to me on the expediency of exercising a constitutional power which I think exists. Tn response to such appeal I have to say, it gave me pain when I learned that Mr. Vallandigham had been arrested-that is, I was pained that there should have seemed to be a necessity for arresting him-and that it will afford me great pleasure to discharge him so soon as I can, by any means, believe the public safety will not suffer by it.

I further say, that as the war progresses, it appears to me, opinion and action, which were in great confusion at first, take shape and fall into more regular channels, so that the necessity for strong dealing with them gradually de-creases. I have every reason to desire that it should cease altogether, and far from the least is my regard for the opinions and wishes of those who like the meeting at Albany, declare their purpose to sustain the Government in every constitutional and lawful measure to suppress the rebellion. Still, I must continue to do so much as may seem to be required by the A. LINCOLN. public safety.

CORRESPONDENCE BETWEEN OHIO DEMOCRATS AND PRESIDENT LINCOLN.

# The Letter to the President.

WASHINGTON, June 26, 1863.

## To his Excellency the PRESIDENT OF THE UNITED STATES:

The undersigned, having been appointed a committee, under the authority of the resolutions of the State Convention held at the city of Columbus, Ohio, on the 11th instant, to

communicate with you on the subject of the arrest and banishment of Clement L. Vallandigham, most respectfully submit the following as the resolutions of that Convention bearing upon the subject of this communication, and ask of your Excellency their earnest consideration. And they deem it proper to state that the Convention was one in which all parts of the State were represented, one of the most respectable as to numbers and character, and one the most earnest and sincere in support of the Constitution and the Union, ever held in that State:

Resolved, That the will of the people is the foundation of all free government; that, to give effect to this free will, free thought, free speech, and a free press are absolutely indispensable. Without free discussion there is no certainty of sound judgment: without sound judgment there can be no wise government. 2. That it is an inherent and constitutional right of the

2. That it is an inherent and constitutional right of the people to discuss all measures of their Government, and to approve or disapprove, as to their best judgment seems right. That they have a like right to propose and advocate that policy which in their judgment is best, and to argue and vote against whatever policy seems to them to violate the Constitution, to impair their liberties, or to be detri-mented at their works. mental to their welfare.

3. That these and all other rights guarantied to them by their Constitutions are their rights in time of war as well as in time of peace, and of far more value and necessity in war than in peace, for in peace liberty, security, and prop-

war infait in peace for in peace horry, security, such prop-erty are seldom endangered; in war they are ever in peril. 4. That we now say to all whom it may concern, not by way of threat, but calculy and firmly, that we will not sur-render these rights, nor submit to their forcible violation. We will obey the laws ourselves, and all others must obey

We will obey the laws ourselves, and all others must obey them. 11. That Ohio will adhere to the Constitution and the Union as the best—it may be the last—hope of popular free-dom, and for all wrongs which may have been committed, or evils which may exist, will seek redress, under the Con-stitution and within the Union, by the peaceful but power-ful agency of the suffrages of a free people. 14. That we will earnestly support every constitutional measure tending to preserve the Union of the States. No men have a greater interest in its preservation than we have, none desire it more; there are none who will make greater sacrifices or endure more than we will to accomp-

greater sacrifices or endure more than we will to accomp-lish that end. We are, as we ever have been, the devoted friends of the Constitution and the Union, and we have no

friends of the Constitution and the Union, and we have no sympathy with the enemies of either. 15. That the arrest, imprisonment, pretended trial, and actual banishment of Clement L. Vallandigham, a citizen of the State of Ohio, not belonging to the land or naval forces of the United States, nor to the militia in actual service, by alleged military authority, for no other pre-tended crime than that of uttering words of legitimate crit-icism upon the conduct of the Administration in power, and of appealing to the ballot-box for a charge of policy— said arrest and military trial taking place where the courts of law are open and unobstructed, and for no act done within the sphere of active military operations in carrying on the war-we regard as a palpable violation of the following provisions of the Constitution of the United States:

1. "Congress shall make no law abridging the freedom of

"Congress shall make no law abridging the freedom of speech or of the press, or the right of the people peaceably to assemble and to petition the Government for a redress of grievances."
 "The right of the people to be secure in their persons, houses, papers, and effects, against unreasonable searches and seizures, shall not be violated; and no warrants shall issue but upon probable cause, supported by oath or affirm-ation, and particularly describing the place to be searches and the persons or things to be seized."
 "No person shall be held to answer for a capital or otherwise infauous crime. unless on a presentment or in-

3. "No person shall be held to answer for a capital or otherwise infanuous crime, unless on a presentment or indictment of a grand jury, except in cases arising in the land or naval forces, or in the militia, when in actual service in time of war or public danger."
4. "In all criminal prosecutions, the accused shall enjoy the right to a speedy and public trial, by an impartial jury of the State and district wherein the crime shall have been committed, which district shall have been previously ascertained by law."

And we furthermore denounce said arrest, trial, and banishment as a direct insult offered to the sovereignty of the people of Ohio, by whose organic law it is declared that no person shall be transported out of the State for any offence committed within the same.

16. That Clement L. Vallandigham was, at the time of

his arrest, a prominent candidate for nomination by the Insarrest, a prominent candidate for nonmation by the Democratic party of Ohio for the office of Governor of the State; that the Democratic party was fully competent to decide whether he is a fit man for that nomination, and that the attempt to deprive them of that right, by his ar-rest and banishment, was an unmerited imputation upon their intelligence and loyalty, as well as a violation of the Constitution.

17. That we respectfully, but most carnestly, call upon the President of the United States to restore Clement L. Vallandigham to his home in Ohio, and that a committee of one from each Congressional district of the State, to be selected by the presiding officer of this Convention, is hereby appointed to present this application to the President.

The undersigned, in the discharge of the duty assigned them, do not think it necessary to reiterate the facts connected with the arrest, trial, and banishment of Mr. Vallandigham--they are well known to the President, and are of public history-nor to enlarge upon the positions taken by the Convention, nor to recapitulate the constitutional provisions which it is believed have been contravened; they have been stated at length, and with clearness, in the resolutions which have been recited. The undersigned content themselves with brief reference to other suggestions pertinent to the subject.

They do not call upon your Excellency as suppliants, praying the revocation of the order banishing Mr. Vallandigham as a favor; but, by the authority of a Convention representing a majority of the citizens of the State of Ohio, they respectfully ask it as a right due to an American citizen, in whose personal injury the sovereignty and dignity of the people of Ohio, as a free State, have been offended. And this duty they perform the more cordially from the consideration that, at a time of great national emergency, pregnant with danger to our Federal Union, it is all important that the true friends of the Constitution and the Union, however they may differ as to the mode of administering the Government, and the measures most likely to be successful in the maintenance of the Constitution and the restoration of the Union, should not be thrown into conflict with each other.

The arrest, unusual trial, and banishment of Mr. Vallandigham have created wide-spread and alarming disaffection among the people of the State, not only endangering the harmony of the friends of the Constitution and the Union, and tending to disturb the peace and tranquillity of the State, but also impairing that confidence in the fidelity of your Administration to the great landmarks of free government essential to a peaceful and successful enforcement of the laws in Ohio.

You are reported to have used in a public communication on this subject, the following language :

"It gave me pain when I learned that Mr. Vallandigham had been arrested—that is, I was pained that there should have seemed to be a necessity for arresting him; and that it will afford me great pleasure to discharge him, so soon as I can by any means believe the public safety will not suffer by it."

The undersigned assure your Excellency, from our own personal knowledge of the feelings of the people of Ohio, that the public safety will be far more endangered by continuing Mr. Vallandigham in exile than by releasing him. It may be true that persons differing | ing to prevent the raising of troops or to en-

from him in political views may be found in Ohio, and elsewhere, who will express a different opinion. But they are certainly mistaken.

Mr. Vallandigham may differ with the President, and even with some of his own political party, as to the true and most effectual means of maintaining the Constitution and restoring the Union; but this difference of opinion does not prove him to be unfaithful to his duties as an American citizen. If a man, devotedly attached to the Constitution and the Union, conscientiously believes that, from the inherent nature of the Federal compact, the war, in the present condition of things in this country, cannot be used as a means of restoring the Union; or that a war to subjugate a part of the States, or a war to revolutionize the social system in a part of the States, could not restore, but would inevitably result in the final destruction of both the Constitution and the Union, is he not to be allowed the right of an American citizen to appeal to the judgment of the people for a change of policy by the constitutional remedy of the ballot-box?

During the war with Mexico many of the political opponents of the Administration then in power thought it their duty to oppose and denounce the war, and to urge before the people of the country that it was unjust and prosecuted for unholy purposes. With equal reason it might have been said of them that their discussions before the people were calculated to discourage enlistments, "to prevent the raising of troops," and to induce desertions from the army, and leave the Government without an adequate military force to carry on the war.

If the freedom of speech and of the press are to be suspended in time of war, then the essential element of popular government to effect a change of policy in the constitutional mode is at an end. The freedom of speech and of the press is indispensable, and necessarily incident to the nature of popular government itself. If any inconvenience or evils arise from its exer-If cise, they are unavoidable.

On this subject you are reported to have said further:

"It is asserted, in substance, that Mr. Vallandigham was, by a military commander, seized and tried 'for no other reason than words addressed to a public meeting in criticism of the course of the Administration, and in condemnation of the military order of the general. Now, if there be no mis-take about this, if there was no other reason for the arrest, then I concede that the arrest was wrong. But the arrest, I understand, was made for a very different reason. Mr. Vallandigham avows his hostility to the war on the part of the Union, and his arrest was made because he was laboring, the Union, and his arrest was made because he was laboring, with some effect, to prevent the raising of troops, to encour-age desertions from the army, and to leave the rebellion without an adequate military force to suppress it. He was not arrested because he was damaging the political pros-pects of the Administration or the personal interests of the commanding general, but because he was damaging the army, upon the existence and vigor of which the life of the nation depends. He was warring upon the military, and this gave the military constitutional jurisdiction to lay hands upon him. If Mr. Vallandigham was not damaging the military power of the country, then his arrest was made on mistake of facts, which I would be glad to correct on reasonable satisfactory evidence."

In answer to this permit us to say, first, that neither the charge nor the specifications in support of the charge on which Mr. Vallandigham was tried impute to him the act of either laborcourage desertions from the army. Secondly, no evidence on the trial was offered with a view to support, or even tended to support, any such charge. In what instance, and by what act, did he either discourage enlistments or encourage desertion in the army? Who is the man who was discouraged from enlisting, and who encouraged to desert, by any act of Mr. Vallandigham? If it be assumed that perchance some person might have been discouraged from enlisting, or that some person might have been encouraged to desert, on account of hearing Mr. Vallandigham's views as to the policy of the war as a means of restoring the Union, would that have laid the foundation for his conviction and banishment? If so, upon the same grounds every political opponent of the Mexican war might have been convicted and banished from the country.

When gentlemen of high standing and extensive influence, including your Excellency, opposed, in the discussions before the people, the policy of the Mexican war, were they "warring upon the military," and did this "give the military constitutional jurisdiction to lay hands upon" them? And, finally, the charge in the specifications upon which Mr. Vallandigham was tried entiled him to a trial before the civil tribunals, according to the express provisions of the late acts of Congress, approved by yourself July 17, 1862, and March 3, 1863, which were manifestly designed to supersede all necessity or pretext for arbitrary military arrests.

The undersigned are unable to agree with you in the opinion you have expressed that the Constitution is different in time of insurrection or invasion from what it is in time of peace and public security. The Constitution provides for no limitation upon or exceptions to the guarantees of personal liberty, except as to the writ of *habeas corpus*. Has the President, at the time of invasion or insurrection, the right to engraft limitations or exceptions upon these constitutional guarantees whenever, in his judgment, the public safety requires it?

True it is, the article of the Constitution which defines the various powers delegated to Congress declares that "the privilege of the writ of habeas corpus shall not be suspended unless when, in cases of rebellion or invasion, the public safety may require it." But this qualification or limitation upon this restriction upon the powers of Congress has no reference to or connection with the other constitutional guarantees of personal liberty. Expunge from the Constitution this limitation upon the power of Congress to suspend the writ of habeas corpus, and yet the other guarantees of personal liberty would remain unchanged.

Although a man might not have a constitutional right to have an immediate investigation made as to, the legality of his arrest upon habeas corpus, yet his "right to a speedy and public trial by an impartial jury of the State and district wherein the crime shall have been committed" will not be altered; neither will his right to the exemption from "cruel and unusual punishment;" nor his right to be secure in his pergon, houses, papers, and effects, against unreasonable seizures and searches;

nor his right not to be deprived of life, liberty, or property, without due process of law; nor his right not to be held to answer for a capital or otherwise infamous offence, unless on presentment or indictment of a grand jury, be in anywise changed.

And certainly the restriction upon the power of Congress to suspend the writ of *habeas* corpus, in time of insurrection or invasion, could not affect the guaranty that the freedom of speech and of the press shall not be abridged. It is sometimes urged that the proceedings in the civil tribunals are too tardy and ineffective for cases arising in times of insurrection or invasion. It is a full reply to this to say that arrests by civil process may be equally as expeditious and effective as arrests by military orders.

True, a summary trial and punishment are not allowed in the civil courts. But if the offender be under arrest and imprisoned, and not entitled to a discharge on writ of habeas corpus before trial, what more can be required for the purposes of the Government? The idea that all the constitutional guarantees of personal liberty are suspended throughout the country at a time of insurrection or invasion in any part of it places us upon a sea of uncertainty, and subjects the life, liberty, and property of every citizen to the mere will of a military commander, or what he may say that he considers the public safety requires. Does your Excellency wish to have it understood that you hold that the rights of every man throughout this vast country are subject to be annulled whenever you may say that you consider the public safety requires it, in time of invasion or insurrection ?

You are further reported as having said that the constitutional guarantees of personal liberty have "no application to the present case we have in hand, because the arrests complained of were not made for treason-that is, not for the treason defined in the Constitution, and upon the conviction of which the punishment is death-nor yet were they made to hold persons to answer for capital or otherwise infamous crimes; nor were the proceedings following in any constitutional or legal sense 'criminal prosecutions.' The arrests were made on totally different grounds, and the proceedings following accorded with the grounds of the arrests," &c.

The conclusion to be drawn from this position of your Excellency is, that where a man is liable to "a criminal prosecution," or is charged with a crime known to the laws of the land, he is clothed with all the constitutional guarantees for his safety and security from wrong and injustice; but that, where he is not liable to "a criminal prosecution," or charged with any crime known to the laws, if the President or any military commander shall say that he considers that the public safety requires it, this man may be put outside of the pale of the constitutional guarantees, and arrested without charge of crime, imprisoned without knowing what for, and any length of time, or be tried before a court-martial and sentenced to any kind of punishment, unknown to the laws of the land, which the President or the military commander may see proper to impose.

Did the Constitution intend to throw the shield of its securities around the man liable to be charged with treason as defined by it, and yet leave the man not liable to any such charge unprotected by the safeguards of personal liberty and personal security? Can a man not in the military or naval service, nor within the field of the operations of the army, be arrested and imprisoned without any law of the land to authorize it? Can a man thus, in civil life, be punished without any law defining the offence aud describing the punishment? If the President or a court-martial may prescribe one kind of punishment unauthorized by law, why not any other kind? Banishment is an unusual punishment, and unknown to our laws. If the President has the right to prescribe the punishment of banishment, why not that of death and confiscation of property? If the President has the right to change the punishment prescribed by the court-martial from imprisonment to banishment, why not from imprisonment to torture upon the rack or execution upon the gibbet ?

If an indefinable kind of constructive treason is to be introduced and engrafted upon the Constitution, unknown to the laws of the land and subject to the will of the President whenever an insurrection or invasion shall occur in any part of this vast country, what safety or security will be left for the liberties of the people?

The constructive treasons that gave the friends of freedom so many years of toil and trouble in England were inconsiderable compared to this. The precedents which you make will become a part of the Constitution for your successors, if sanctioned and acquiesced in by the people now.

The people of Ohio are willing to co-operate zealously with you in every effort warranted by the Constitution to restore the Union of the States, but they cannot consent to abandon those fundamental principles of civil liberty which are essential to their existence as a free people.

In their name we ask that, by a revocation of the order of his banishment, Mr. Vallandigham may be restored to the enjoyment of those rights of which they believe he has been unconstitutionally deprived.

We have the honor to be respectfully, yours, &c.,

M. BIRCHARD, Chairman, 19th Dist. DAVID A. HOUK, Secretary, 3d Dist. GEO. BLISS, 14th Dist. W. BARTLEY, 8th Dist. W. J. GORDON, 18th Dist. JOHN ONEILL, 18th Dist. W. E. FINCK, 12th Dist. W. E. FINCK, 12th Dist. ALEXANDER LONG, 2d Dist. J. W. WHITE, 16th Dist. GEO. S. CONVERSE, 7th Dist. GEO. S. CONVERSE, 7th Dist. WARREN P. NOBLE, 9th Dist. GEO. H. DENDLETON, 1st Dist. W. A. HUTCHINS, 11th Dist. J. F. MCKINNEY, 4th Dist. J. F. MCKINNEY, 4th Dist. F. C. LEBLOND, 5th Dist. LOUIS SHAEFER, 17th Dist.

## Letter from President Lincoln.

WASHINGTON, June 29, 1863.

Messes. M. Birchard, David M. Houk, George Eliss, T. W. Bartley, W. J. Gordon, John O'Niell, C. A. White, W. E. Finck, Alexander Long, J. W. White, George H. Pendleton, George S. Converse, Warren P. Noble, James R. Morris, W. A. Hutchins, Abner L. Backus, J. F. McKinney, F. C. LeBlond, Louis Schaefer:

GENTLEMEN: The resolutions of the Ohio Democratic State Convention, which you present me, together with your introductory and closing remarks, being in position and argument mainly the same as the resolutions of the Democratic meeting at Albany, New York, I refer you to my response to the latter as meeting most of the points in the former.

This response you evidently used in preparing your remarks, and I desire no more than that it be used with accuracy. In a single reading of your remarks, I only discovered one inaccuracy in matter which I suppose you took from that paper. It is where you say, "The undersigned are unable to agree with you in the opinion you have expressed that the Constitution is different in time of insurrection or invasion from what it is in time of peace and public security."

A recurrence to the paper will show you that I have not expressed the opinion you suppose. I expressed the opinion that the Constitution is different *in its application* in cases of rebellion or invasion, involving the public safety, from what it is in times of profound peace and public security; and this opinion I adhere to, simply because by the Constitution itself things may be done in the one case which may not be done in the other.

I dislike to waste a word on a merely personal point, but I must respectfully assure you that you will find yourselves at fault should you ever seek for evidence to prove your assumption that I "opposed in discussions before the people the policy of the Mexican war."

You say: "Expunge from the Constitution this limitation upon the power of Congress to suspend the writ of habeas corpus, and yet the other guarantees of personal liberty would remain unchanged." Doubtless if this clause of the Constitution, improperly called, as I think, a limitation upon the power of Congress, were expunged, the other guarantees would remain the same; but the question is, not how those guarantees would stand with that clause out of the Constitution, but how they stand with that clause remaining in it, in case of rebellion or invasion, involving the public safety. If the liberty could be indulged in expunging that clause, letter and spirit, I really think the constitutional argument would be with you.

My general view on this question was stated in the Albany response, and hence I do not state it now. I'only add that, as seems to me, the benefit of the writ of habeas corpus is the great means through which the guarantees of personal liberty are conserved and made available in the last resort; and corroborative of this view is the fact that Mr. Vallandigham, in the very case in question, under the advice of able lawyers, saw not where else to go but to the habeas corpus. But by the Constitution the benefit of the writ of habeas corpus itself may be suspended, when, in case of rebellion or invasion, the public safety may require it.

You ask, in substance, whether I really claim that I may override all the guarantied rights of individuals, on the plea of conserving the public safety-when I may choose to say the public safety requires it. This question, divested of the phraseology calculated to represent me as struggling for an arbitrary personal prerogative, is either simply a question who shall decide, or an affirmation that nobody shall decide, what the public safety does require in cases of rebellion or invasion. The Constitution contemplates the question as likely to occur for decision, but it does not expressly declare who is to decide it. By necessary implication, when rebellion or invasion comes, the decision is to be made, from time to time; and I think the man whom, for the time, the people have, under the Constitution, made the Commander-in-Chief of their Army and Navy, is the man who holds the power and bears the responsibility of making it. If he uses the power justly, the same people will probably justify him; if he abuses it, he is in their hands, to be dealt with by all the modes they have reserved to themselves in the Constitution.

The earnestness with which you insist that persons can only, in times of rebellion, be lawfully dealt with, in accordance with the rules for criminal trials and punishments in times of peace, induces me to add a word to what I said on that point in the Albany response. You claim that men may, if they choose, embarrass those whose duty it is to combat a giant rebellion and then be dealt with only in turn as if there were no rebellion. The Constitution itself rejects this view. The military arrests and detentions which have been made, including those of Mr. Vallandigham, which are not different in principle from the other, have been for prevention, and not for punishment-as injunctions to stay injury, as proceedings to keep the peace-and hence, like proceedings in such cases and for like reasons, they have not been accompanied with indictments, or trials by juries, nor in a single case by any punishment whatever beyond what is purely incidental to the prevention. The original sentence of imprisonment in Mr. Vallandigham's case was to prevent injury to the military service only, and the modification of it was made as a less disagreeable mode to him of securing the same prevention.

I am unable to perceive an insult to Ohio in the case of Mr. Vallandigham. Quite surely nothing of this sort was or is intended. I was wholly unaware that Mr. Vallandigham was, at the time of his arrest, a candidate for the Democratic nomination for Governor, until so informed by your reading to me the resolutions of the Convention. I am grateful to the State of Ohio for many things, especially for the brave soldiers and officers she has given in the present national trial to the armies of the Union.

You claim as I understand, that according to my own position in the Albany response, Mr. Vallandigham should be released; and this because, as you claim, he has not damaged the military service by discouraging enlistments,

encouraging desertions, or otherwise; and that if he had, he should be turned over to the civil authorities under the recent acts of Congress. I certainly do not know that Mr. Vallandigham has specifically and by direct language advised against enlistments, and in favor of desertion and resistance to drafting. We all know that combinations, armed in some instances, to resist the arrest of deserters, began several months ago; that more recently the like has appeared in resistance to the enrollment preparatory to a draft; and that quite a number of assassinations have occurred from the same These had to be met by military animus. force, and this again has led to bloodshed and death. And now, under a sense of responsibility more weighty and enduring than any which is merely official, I solemnly declare my belief that this hindrance of the military, including maiming and murder, is due to the course in which Mr. Vallandigham has been engaged, in a greater degree than to any other cause ; and it is due to him personally in a greater degree than to any other man.

These things have been notorious, known to all, and of course known to Mr. Vallandigham. Perhaps I would not be wrong to say they originated with his especial friends and adherents. With perfect knowledge of them he has frequently, if not constantly, made speeches in Congress and before popular assemblies; and if it can be shown that, with these things staring him in the face, he has ever uttered a word of rebuke or counsel against them, it will be a fact greatly in his favor with me, and one of which, as yet, I am totally ignorant. When it is known that the whole burden of his speeches has been to stir up men against the prosecution of the war, and that in the midst of resistance to it he has not been known in any instance to counsel against such resistance, it is next to impossible to repel the inference that he has counselled directly in favor of it.

With all this before their eyes, the Convention you represent have nominated Mr. Vallandigham for Governor of Ohio, and both they and you have declared the purpose to sustain the National Union by all constitutional means. But, of course, they and you, in common, reserve to yourselves to decide what are constitutional means, and, unlike the Albany meeting, you omit to state or intimate that, in your opinion, an army is a constitutional means of saving the Union against a rebellion, or even to intimate that you are conscious of an existing rebellion being in progress with the avowed object of destroying that very Union. At the same time, your nominee for Governor, in whose behalf you appeal, is known to you and to the world to declare against the use of an army to suppress the rebellion. Your own attitude, therefore, encourages desertion, resistance to the draft, and the like, because it teaches those who incline to desert and to escape the draft to believe it is your purpose to protect them and to hope that you will become strong enough to do so.

After a short personal intercourse with you, gentlemen of the committee, I cannot say I think you desire this effect to follow your atti-

tude; but I assure you that both friends and enemies of the Union look upon it in this light. It is a substantial hope, and by consequence a real strength to the enemy. It is a false hope, and one which you would willingly dispel. will make the way exceedingly easy. I send you duplicates of this letter, in order that you, or a majority, may, if you choose, endorse your names upon one of them, and return it thus endorsed to me, with the understanding that those signing are hereby committed to the following propositions, and to nothing else :

1. That there is now a rebellion in the United States, the object and tendency of which is to destroy the National Union; and that, in your opinion, an army and navy are constitutional means for suppressing that rebellion.

2. That no one of you will do anything which, in his own judgment, will tend to hinder the increase or favor the decrease or lessen the efficiency of the army and navy, while engaged in the effort to suppress that rebellion; and

3. That each of you will, in his sphere, do all he can to have the officers, soldiers, and seamen of the army and navy, while engaged in the effort to suppress the rebellion, paid, fed, clad, and otherwise well provided for and supported.

And with the further understanding that upon receiving the letter and names thus endorsed, I will cause them to be published, which publication shall be, within itself, a revocation of the order in relation to Mr. Vallandigham.

It will not escape observation that I consent to the release of Mr. Vallandigham upon terms not embracing any pledge from him or from others as to what he will or will not do. I do this because he is not present to speak for himself, or to authorize others to speak for him; and hence I shall expect that on returning he would not put himself practically in antagonism with his friends. But I do it chiefly because I thereby prevail on other influential gentlemen of Ohio to so define their position as to be of immense value to the army-thus more than compensating for the consequences of any mistake in allowing Mr. Vallandigham to return, so that, on the whole, the public safety will not have suffered by it. Still, in regard to Mr. Vallandigham and all others, I must hereafter, as heretofore, do so much as the public service may seem to require.

I have the honor to be, respectfully, yours, &c. A. LINCOLN.

#### The Committee's Rejoinder.

#### NEW YORK, July 1, 1863.

To his Excellency the PRESIDENT OF THE UNITED STATES:

SIR: Your answer to the application of the undersigned for a revocation of the order of banishment of Clement L. Vallandigham requires a reply, which they proceed with as little delay as possible to make.

They are not able to appreciate the force of the distinction you make between the Constitution and the application of the Constitution, whereby you assume that powers are delegated

surrection, in derogation of the plain language of the Constitution. The inherent provisions of the Constitution remaining the same in time of insurrection or invasion as in time of peace, the President can have no more right to disregard their positive and imperative requirements at the former time than at the latter. Because some things may be done by the terms of the Constitution at the time of invasion or insurrection which would not be required by the occasion in time of peace, you assume that any thing whatever, even though not expressed by the Constitution, may be done on the occasion of insurrection or invasion, which the President may choose to say is required by the public safety. In plainer terms, because the writ of habeas corpus may be suspended at time of invasion or insurrection, you infer that all other provisions of the Constitution having in view the protection of the life, liberty, and property of the citizen, may be in like manner suspended.

The provision relating to the writ of habeas corpus being contained in the first part of the Constitution, the purpose of which is to define the powers delegated to Congress, has no connection in language with the Declaration of Rights, as guarantees of personal liberty, contained in the additional and amendatory articles, and inasmuch as the provision relating to habeas corpus expressly provides for its suspension, and the other provisions alluded to do not provide for any such thing, the legal conclusion is that the suspension of the later is unauthorized. The provision for the writ of habeas corpus is merely intended to furnish a summary remedy, and not the means whereby personal security is conserved in the final resort; while the other provisions are guarantees of personal rights, the suspension of which puts an end to all pretence of free government. It is true Mr. Vallandigham applied for a writ of habeas corpus as a summary remedy against oppression. But the denial of this did not take away his right to a speedy public trial by an impartial jury, or deprive him of his other rights as an American citizeu. Your assumption of the right to suspend all the constitutional guarantees of personal liberty, and even of the freedom of speech and of the press, because the summary remedy of habeas corpus may be suspended, is at once startling and alarming to all persons desirous of preserving free government in this country.

The inquiry of the undersigned, whether "you hold the rights of every man throughout this vast country, in time of invasion or insurrection, are subject to be annulled whenever you may say that you consider the public safety requires it ?" was a plain question, undisguised by circumlocution, and intended simply to elicit information. Your affirmative answer to this question throws a shade upon the fondest anticipations of the framers of the Constitution, who flattered themselves that they had provided safeguards against the dangers which have ever beset and overthrown free government in other ages and countries. Your answer is not to be disguised by the phraseology that the question to the President at the time of invasion or in- "is simply a question who shall decide, or an

affirmation that nobody shall decide, what the public safety does requine in case of rebellion or invasion." Our Government was designed to be a Government of *law, settled* and *defined*, and not of the arbitrary will of a single man. As a safeguard, the powers were delegated to the legislative, executive, and judicial branches of the Government, and each made co-ordinate with the others, and supreme within its sphere, and thus a mutual check upon each other in case of abuse of power.

It has been the boast of the American people that they had a written Constitution, not only expressly defining, but also limiting the powers of the Government, and providing effectual safeguards for personal liberty, security, and property. And, to make the matter more positive and explicit, it was provided by the amendatory articles nine and ten that "the enumeration in the Constitution of *certain rights* shall not be construed to *deny* or *disparage* others retained by the people," and that "the powers not dele-gated to the United States by the Constitution, nor prohibited by it to the States, are reserved to the States respectively or to the people." With this care and precaution on the part of our forefathers who framed our institutions, it was not to be expected that, at so early a day as this, a claim of the President to arbitrary power, limited only by his conception of the requirements of the-public safety, would have been asserted. In derogation of the constitutional provisions making the President strictly an executive officer, and vesting all the delegated legislative powers in Congress, your position, as we understand it, would make your will the rule of action, and your declaration of the requirements of the public safety the law of the land. Our inquiry was not, therefore, "simply a question who shall decide, or the affirmation that nobody shall decide, what the public safety requires." Our Government is a Government of law, and it is the law-making power which ascertains what the public safety requires, and prescribes the rule of action; and the duty of the President is simply to execute the laws thus enacted, and not to make or annul laws. If any exigency shall arise, the President has the power to convene Congress at any time to provide for it; so that the plea of necessity furnishes no reasonable pretext for any assumption of legislative power.

For a moment contemplate the consequences of such a claim to power. Not only would the dominion of the President be absolute over the rights of individuals, but equally so over the other departments of the Government. If he should claim that the public safety required it, he could arrest and imprison a judge for the conscientious dischage of his duties, paralyze the judicial power, or supercede it by the substitution of courts-martial, subject to his own will, throughout the whole country. If any one of the States, even far removed from the rebellion, should not sustain his plan for prosecuting the war, he could, on the plea of public safety, annul and set at defiance the State laws and authorities, arrest and imprison the Governor of the State or the members of the Legislature, while in the faithful discharge of their

duties, or he could absolutely control the action, either of Congress or of the Supreme Court, by arresting and imprisoning its members, and upon the same ground he could suspend the elective franchise, postpone the elections, and declare the perpetuity of his high prerogative. And neither the power of impeachment nor the elections of the people could be made available against such concentration of power.

Surely it is not necessary to subvert free government in this country in order to put down the rebellion; and it cannot be done under the pretence of putting down the rebellion. Indeed, it is plain that your Administration has been weakened, and greatly weakened, by the assumption of power not delegated in the Constitution.

In your answer you say to us: "You claim that men may, if they choose, embarras those whose duty it is to combat a giant rebellion and then be dealt with in terms as if there were no rebellion." You will find yourself in fault, if your will search our communication to you for any such idea. The undersigned believe that the Constitution and laws of the land, properly administered, furnish ample power to put down an insurrection without the assumption of powers not granted. And if existing legislation be inadequate, it is the duty of Congress to consider what further legislation is necessary, and to make suitable provision by law.

You claim that the military arrests made by your Administration are merely preventive remedies, "as injunctions to stay injury, or proceedings to keep the peace, and not for punishment." The ordinary preventive remedies alluded to are authorized by established law, but the preventive proceedings you institute have their authority merely in the will of the Executive or that of officers subordinate to his authority. And in this proceeding a discretion seems to be exercised as to whether the prisoner shall be allowed a trial or even be permitted to know the nature of the complaint alleged against him, or the name of his accuser. If the proceedings be merely preventive, why not allow the prisoner the benefit of a bond to keep the peace? But if no offence has been committed, why was Mr. Vallandigham tried, convicted, and sentenced by a court-martial? And why the actual punishment by imprisonment or banishment, without the opportunity of obtaining his liberty in the mode usual in preventive remedies, and yet say it is not for punishment?

You still place Mr. Vallandigham's conviction and banishment upon the ground that he had damaged the military service by discouraging enlistments and encouraging desertions, &c., and yet you have not even pretended to controvert our position that he was not charged with, tried, or convicted for any such offence before the court-martial.

rebellion, should not sustain his plan for prosecuting the war, he could, on the plea of public safety, annul and set at defiance the State laws and authorities, arrest and imprison the Governor of the State or the members of the Legislature, while in the faithful discharge of their

resistance to drafting," &c., and yet, in a subsequent part of your answer, after speaking of certain disturbances which are alleged to have occurred in resistance of the arrest of deserters and of the enrollment preparatory to the draft, and which you attribute mainly to the course Mr. Vallandigham has pursued, you say that he has made speeches against the war in the midst of resistance to it; that "he has never been known, in any instance, to counsel against such resistance;" and that "it is next to impossible to repel the inference that he has counselled directly in favor of it." Permit us to say that your information is most grievously at fault.

The undersigned have been in the habit of hearing Mr. Vallandigham speak before popular assemblages, and they appeal with confidence to every truthful person who has ever heard him for the accuracy of the declaration, that he has never made a speech before the people of Ohio in which he has not counselled submission and obedience to the laws and the Constitution, and advised the peaceful remedies of the judicial tribunals and of the ballot-box for the redress of grievances and for the evils which afflict our bleeding and suffering country. And, were it not foreign to the purposes of this communication, we would undertake to establish to the satisfaction of any candid person that the disturbances among the people to which you allude, in opposition to the arrest of deserters and the draft, have been occasioned mainly by the measures, policy, and conduct of your Administration, and the course of its political friends. But if the circumstantial evidence exists, to which you allude, which makes "it next to impossible to repel the inference that Mr. Vallandigham has counselled directly in favor" of this resistance, and that the same has been mainly attributable to his conduct, why was he not turned over to the civil authorities to be tried under the late acts of Congress? If there be any foundation in fact for your statements implicating him in resistance to the constituted authorities, he is liable to such prosecution. And we now demand, as a mere act of justice to him, an investigation of this matter before a jury of his country; and respectfully insist that fairness requires either that you retract these charges which you make against him, or that you revoke your order of banishment and allow him the opportunity of an investigation before an impartial jury.

The committee do not deem it necessary to repel at length the imputation that the attitude of themselves or of the Democratic party in Ohio "encourage desertions, resistance to the draft, and the like." Suggestions of that kind are not unusual weapons in our ordinary political contests. They rise readily in the minds of politicians heated with the excitement of partisan strife. During the two years in which the Democratic party of Ohio has been constrained to oppose the policy of the Administration, and to stand up in defence of the Constitution and of personal rights, this charge has been repeatedly made. It has fallen harmless, however, at the feet of those whom it was intended to injure. The committee believe it

this paper, they might suggest that the measures of the Administration, and its changes of policy in the prosecution of the war, have been the fruitful sources of discouraging enlistments and inducing desertions, and furnish a reason for the undeniable fact that the first call for volunteers was answered by very many more than were demanded, and that the next call for soldiers will probably be responded to by drafted men alone.

The observation of the President in this connection, that neither the Convention in its resolutions, nor the committee in its communication, intimate that they " are conscious of an existing rebellion being in progress with the avowed object of destroying the Union," needs, perhaps, no reply. The Democratic party of Ohio has felt so keenly the condition of the country, and been so stricken to the heart by the misfortunes and sorrows which have befallen it, that they hardly deemed it necessary by solemn resolution, when their very State exhibited everywhere the sad evidences of war, to remind the President that they were aware of its existence.

In the conclusion of your communication you propose that, if a majority of the committee shall affix their signatures to a duplicate copy of it, which you have furnished, they shall stand committed to three propositions, therein at length set forth, that he will publish the names thus signed, and that this publication shall operate as a revocation of the order of The committee cannot refrain banishment. from the expression of their surprise that the President should make the fate of Mr. Vallandigham depend upon the opinion of this committee upon these propositions. If the arrest and banishment were legal, and were deserved; if the President exercised a power clearly delegated, under circumstances which warranted its exercise, the order ought not to be revoked, merely because the committee hold, or express, opinions accordant with those of the President. If the arrest and banishment were not legal, or were not deserved by Mr. Vallandigham, then surely he is entitled to an immediate and unconditional discharge.

The people of Ohio were not so deeply moved by the action of the President merely because they were concerned for the personal safety and convenience of Mr. Vallandigham, but because they saw in his arrest and banishment an attack upon their own personal rights; and they attach value to his discharge chiefly as it will indicate an abandonment of the claim to the power of such arrest and banishment. However just the undersigned might regard the principles contained in the several propositions submitted by the President, or how much soever they might, under other circumstances, feel inclined to indorse the sentiments contained therein, yet they assure him that they have not been authorized to enter into any bargains, terms, contracts, or conditions with the President of the United States to procure the release of Mr. Vallandigham. The opinions of the undersigned touching the questions involved in these propositions are well known, will do so again. If it were proper to do so in have been many times publicly expressed, and

are sufficiently manifested in the resolutions of the convention which they represent, and they cannot suppose that the President expects that they will seek the discharge of Mr. Vallandigham by a pledge implying not only an imputation upon their own sincerity and fidelity as citizens of the United States, and also carrying with it by implication a concession of the legality of his arrest, trial, and banishment, against which they and the convention they represent have solemnly protested. And, while they have asked the revocation of the order of banishment not as a favor, but as a right due to the people of Ohio, and with a view to avoid the possibility of conflict or disturbance of the public tranquillity, they do not do this, nor does Mr. Vallandigham desire it, at any sacrifice of their dignity and self-respect.

The idea that such a pledge as that asked from the undersigned would secure the public safety sufficiently to compensate for any mistake of the President in discharging Mr. Vallandigham is, in their opinion, a mere evasion of the grave questions involved in this discussion, and of a direct answer to their demand. And this is made especially apparent by the fact that this pledge is asked in a communication which concludes with an intimation of a disposition on the part of the President to repeat the acts complained of.

The undersigned, therefore, having fully discharged the duty enjoined upon them, leave the responsibility with the President.

bity with the President.
M. BIRCHARD, 19th Dist., Chairman.
DAVID HOUK, See'y, 3d Dist.
GEO. BLISS, 14th Dist.
T. W. BARTLEY, 8th Dist.
W. J. GORDON, 18th Dist.
JOHN O'NEILL, 13th Dist.
C. A. WHITE, 6th Dist.
W. E. FINCK, 12th Dist.
GEO. S. CONVERSE, 15th Dist.
GEO. S. CONVERSE, 1th Dist.
GEO. H. PENDLETON, 1st Dist.
J. F. MCKUNS, 11th Dist.
J. F. MCKUNNEY, 4th Dist.
J. W. MHITE, 16th Dist.
J. W. WHITE, 16th Dist. F. C. LEBLOND, 5th Dist. LOUIS SCHÆFFER, 17th Dist. WARREN P. NOBLE, 9th Dist.

# The Case before the United States Supreme Court.

WASHINGTON, February 15, 1864. The case of Mr. Vallandligham, ex parte, was decided in the Supreme Court of the United States to-day. The peti-tioner asked that the writ of certiorari be directed to the Judge Advocate General for a revision of the proceedings of the Allitary Commission which tried him, the jurisdiction of which was denied as extending to the case of a civilian, and the object being to have the sentence annulled, on the ground of illegality. The Judge Advocate, Col. Holt, had responded in a written argument that the Court might with as much propriety be called upon to restrain, by injunction, the proceedings of Congress, as to revise by certiorari and reverse the proseedings of the military offences, according to the usages of civilized nations and the power given by the Constitution and laws of the United States for the common defence and public safety. WASHINGTON, February 15, 1864. common defence and public safety.

common detence and public safety. Justice Wayne to day delivered the opinion of the Court, refusing the writ, on the greund that even if the arrest, trial, and punishment of Vallan,ligham were illegal, there is no authority in the Court to grant relief in this mode, and that there is no law by which any appeal or proceedings in the nature of an appeal from a Military Commission to the Supreme Court can be taken.

#### His Letter on "Retaliation."

WINDSOR, C. W., March 7, 1864. Messrs. HUBBARD AND BROTHEE, Dayton, Ohio: GENTLEMEN: I read, several days ago, the telegraphic

announcement of the "riddling" of the Empire office by "furloughed soldiers." I offer you no sympathy, for that will avail nothing now or hereafter. I do express to you my profound regret that you were not prepared to inflict on the spot, and in the midst of the assault, the complete punishment which the assailants deserved; but am grati-fied to learn that some of them did soon after receive their deserts. But these cowardly acts cannot always be guarded against. And they do not primarily come from the "sol-diers." There is, therefore, but one remedy for past and preventive of futures, interforce, but one tenday for plast and mary, and ample reprisals upon the persons and property of the men at home who, by language and conduct, are al-ways exciting these outrages.

Ways exciting these outrages. No legal nor military punishment is ever inflicted upon the immediate instruments. Retaliation, therefore, is the only and rightful remedy in times like these. I speak ad-visedly, and recommend it in all cases hereafter. It is of no avail to annonnee the fulschood that "both partics con-demn it," after the destruction has been consummated. The time here one we for sheddings without restories. demn it," dyter the destruction has been consummated. The time has gone by for obedience without protection. I speak decided language; but the continual recurrence of these outrages—frequently attended with murder, and always without referess—demands it. They must be stopped, let the consequences be what they may. Reprisals in such cases are now the only way left for a return to law and order. C. L. VALLANDIGHAM.

# Very truly,

#### Mr. Vallandigham's Return and Address.

1864, June 15-Mr. Vallandigham returned to Ohio, and that day addressed the Democratic Convention at Hamilton, Ohio, as follows:

MEN OF OHIO: To-day I am again in your midst and upon the soil of my native State. To-day I am once more within the district which for ten years extended to me the highest confidence, and three times honored me as its Representa-tive in the Congress of the United States. I was accused of no crime against the Constitution or laws, and guilty of none. But whenever and wherever thus charged upon due process of law, I am now here ready to answer before any civil court of competent jurisdiction, to a jury of my coun-trymen, and in the meantime to give bail in any sum which any judge or court, State or Federal, may affix, and you, the 186,000 Democrats of Ohio, I offer as my sureties.

186,000 Democrats of Ohio, I offer as my survities. Never for one hour have I remained in exile because I recognized any obligation of obedience to the nucconstitu-tional and arbitrary edict. Neither did personal fear ever restrain me. And to-day I return of my own act and pleas-ure, because it is my constitutional and legal right to return. Only by an exertion of arbitrary power, itself against the Constitution and law, and consummated by military force, I was abducted from my home and forced into benishment. was abducted from my home and forced into banishment. was abducted from my home and forced into banishment. The assertion or insimuation of the President that I was ar-rested "because laboring, with some effect, to prevent the raising of troops, and to encourage desertions from the army," and was responsible for numerous acts of resistance to the draft and to the arrest of deserters, causing "assassi-nation, maining, and murder," or that at any time, in any way, I had disobeyed or failed to counsel obedience to tho lawful authority, or even to the semblance of law, is abso-lutely false. lutely false.

Intely false. I appeal for the proof to every speech I ever made upon those questions, and to the very record of the mock Military Commission by the trial and sentence of which I was out-raged. No; the sole offence then laid to my charge was words of criticism of the public policy of the Administra-tion, addressed to open and public policy of the Administra-tion, addressed to open and public policy of the Administra-tion, addressed to open and public policy of the Administra-tion, addressed to open and public policy of the Administra-tion, addressed to open and public policy of the Administra-tion, addressed to open and public policy of the Administra-tion, addressed to open and public policy of the Administra-tion, addressed to open and public policy of the Administra-tion, addressed to open and public policy of the Administra-artested, and no newspaper suppressed within the States adhering still to the Union, for the expression of political opinion; while hundreds, in public assembly and through the press, have, with a license and violence in which I never induiged, criticised and condemned the acts and policies of the Administration, and denounced the war, maintaining even the propriety and necessity of the recognition of sonthern independence. Independence.

Indorsed by nearly two hundred thousand freemen of the Democratic party of my native State at the late elections, and still with the sympathy and support of millions more, I do not mean any longer to be the only man of that party who is to be the victim of arbitrary power. If Abraham Lucoln seeks my life, let him so declare; but he shall not again restrain me of my personal liherty except upon "due process of law." The unconstitutional and monstrous "Order 35," under which alone I was arrested thirteen convention of 1863, by the gallant gentleman who bore the standard as your candidate for Lieutenant Governor, and by every Democratic press and public speaker ever since. It is dead. From the first it was against the Constitution and laws, and without validity; and all proceedings under it were and are utterly null and void and of no effect.

The indignant voice of condemnation long since went The indignant voice of condemnation long since went forth from the vast majority of the people and presses of America and from all free countries in Europe with entire unanimity. And more recently, too, the "platform" of an earnest, numerous, and most formidable Convention of the sincere Republicans, and still further, the emphatic letter of acceptance by the candidate of that Convention, Gen. John C. Fremont, the first candidate also of the Re-publican party for the Presidency eight years ago, upon the rallying cry of "Free speech and a free Press," give re-neenewed hope that at last the reign of arbitrary power is about to be brought to an end in the United States. It is neither just nor fit, therefore, that the wrongs inflicted is neither just nor fit, therefore, that the wrongs inflicted under "Order 38," and the other edicts and acts of such power, should be any longer endured-certainly not by me alone.

But every ordinary means of redress has first been ex-hausted; yet either by the direct agency of the Adminis-tration and its subordinates, or through its influence or intimidation, or because of want of jurisdiction in the civil courts which no American in former times conceived to be possible here, all have failed. Counsel applied in my behalf to an unjust judge for the writ of *habcas corpus*. It was denied; and now the privilege of that writ is suspended by act of Congress and Executive order in every State. The Democratic Convertion of Ohio, one year ago, by a resolu-tion formally presented through a committee of your best and ablest men in person, at Washington, demanded of the President, in behalf of a very large minority of the people, a revocation of the edict of banishment.

people, a revocation of the edict of banishment. Pretending that the public safety then required it, he refused; saying, at the same time, that "it would afford him pleasure to comply as soon as he could by any means be made to believe that the public safety would not suffer by it." One year has elapsed, yet this hollow pretence is still tacitly asserted, and to-day I am here to prove it unfounded in fact. I appealed to the Supreme Court of the United States; and because Congress had never conferred jurisdic-tion in behalf of a citizen tried by a tribunal unknown for tion in behalf of a citizen tried by a tribunal unknown for such purpose, to the laws, and expressly forbidden by the Constitution, it was powerless to redress the wrong. The time has, therefore, arrived, when it becomes me as a citi-zen of Ohio and of the United States, to demand, and, by my own act, to vindicate the rights, liberties, and privileges which I never forfeited, but of which, for so many months, I have been deprived.

Wherefore, men of Ohio, I am again in your midst to-day. I owe duties to the State, and am here to discharge them. I have rights as a citizen, and am here to assert them; a wife, and child, and home, and would enjoy all the pleas-ures which are implied in those cheerful words. But I am here for peace, not disturbance; for quiet not convulsion; for order and law, not anarchy. Let no mau of the Demo-cratic party begin any act of violence or disorder; but let Crate party begin any act of violence or insorder; but let none shrink from any responsibility, however urgent, if forced upon him. Careful of the rights of others, let him see to it that he fully and fearlessly exacts his own. Sub-ject to rightful authority in all things, let him submit to excess or usurpation in nothing. Obedient to the Consti-tion and law, let him demand and have the full measure of variations which law and Careful thim course to him.

protection which law and Constitution secure to him. Men of Ohio: You have already vindicated your right to *hear*, it is now my duty to assert my right to *speak*. Where-fore, as to the sole offence for which I was arrested, imprisoned, and banished-free speech in criticism and condemnation of the Administration—an Administration fitly described in a recent public paper by one of its early supporters, "marked at home by disregard of constitutional rights, by its violation of personal liberty and the liberty of the press, and, as its crowning shame, by its abandonment of the right of asylum, a right especially dear to all free nations abroad," I repeat it here to-day, and will again and yet again so long as I live, or the Constitution and our present form of Government shall survive.

The words then spoken and the appeal at that time made, and now enforced by one year more of taxation and debt, and of blood and disaster, entreating the people to change the public servants and their policy, not by force, but peaceably, through the ballot, I now and here reiterate in their utmost extent, and with all their significancy I repeat them, one and all, in no spirit of challenge or bravado, but as earnest, sober, solem truth and warning to the people. Upon another subject allow me here a word:

A powerful, widely-spread and very dangerous secret, oath-A powerful, when y spreau and very dangerous secret, oath-bound combination among the friends of the Administra-tion, known as the "Loyal Union League," exists in every State, yet the very men who control it charge persistently upon the members of the Democratic party, that they have organized—especially in the North West—the "Order of Knights of the Golden Circle," or some other secret society,

treasonable or "disloyal" in its character, affiliated with the South, and for the purpose of armed resistance to the Fed-eral and State Governments. Whether any such ever existed I do not know; but the charge that organizations of that sort, or having any such purpose, do now exist among mem-

sort, of having any such purpose, do how exist throng mem-bers of that party in Ohio or other non-slaveholding States, is totally and positively false. That lawful political or party associations have been es-tablished, having, as their object, the organizing and strengthening of the Democratic party, and its success in the ceming Presidential election, and designed as a connter-movement to the so-called "Union Leagues," and, therefore, secret in their proceedings, is yeary norballe, and however movement to the so-called "Union Leagues," and, therefore, secret in their proceedings, is very probable, and however objectionable hitherto, and in ordinary times; I recognize to the fullest extent, not the lawfulness only, but the pro-priety and necessity of such organizations—for "when bad men combine good men must associate." But they are no conspiracy against the Government, and their members are not correspondent but notifier unon not how much term conspirately against the Government, and their memoers are not conspirators, but patriots; men not leagued toge-ther for the overthrow of the Constitution or the laws, and still less, of liberty, but firmly united for the preservation and support of these great objects. There is, indeed, a "conspiracy" very powerful, very an-cient, and I trust that before long I may add, strongly

to be triumphant—a conspiracy known as the Democratic party, the present object of which is the overthrow of the Administration in November next, not by force but through the ballot-box, the election of a President who shall be true the ballot-box, the election of a President who shall be true to his oath, to Liberty and the Constitution. This is the sole conspiracy of which I know anything; and I am proud to be one of the conspirators. If any other exist, looking to unlawful armed resistance to the Federal State authorities anywhere, in the exercise of their legal and constitutional rights, I admonish all persons concerned, that the act is treason, and the penalty death. But I war also the men in power that there is a rast mul-

But I warn also the men in power that there is a vast multitude, a host whom they cannot number, bound together by the strongest and holiest ties, to defend, by whatever means the exigencies of the times shall demand, their natural and constitutional rights as freemen, at all hazards and to the

Last extremity. Three years have now passed, men of Ohio, and the great issue-constitutional liberty and free popular government-is still before you. To you I again commit it, confident that in this, the time of their greatest peril, you will be found worthy of the ancestors who for so many ages, in England and Appria on the field in price and user the affeld and America, on the field, in prison, and upon the scaffold, defended them against tyrants and usurpers, whether in councils or in arms.

June 17-He is reported to have thus spoken in response to a serenade, in Dayton :

MY FRIENDS: I greet you to-night as you greet me, and I can truly say, that from this demonstration it is evident you are determined to support those principles which I have advocated and have suffered for. To me, this demonstration was unexpected, and I appear only to make my renewed acknowledgment to you for this continued expression of kindly feeling.

He would make no threats, but he did not come from a foreign country without a deliberate calculation of the cause and the consequences, and a deliberate calculation of the cause and the consequences, and a deliberate preparation to meet them. He could be taken by any due civil process, by any crippled constable, but without that no force could do it. Three hundred men, armed to the teeth, would not again find him in his honse after the door had been buttoned down but they would be deliber the rest. down, but they would find him the next day and not far off, [immense cheers,] and if any military commander of the President were to undertake such an arrest he warned them 

home from which he had been torn thirteen months before, and to receive in quiet the calls of his friends. He did not expect to be again molested unless by men in this city, and the former scenes revived. "If this be done," he said, "I warn them that the result will be such as compared to it, the other was but dust in the scale."

He then reviewed his personal and political history, defy-ing any person to show wherein he had merited the treat-ment he had received. He again repeated that he desired no disturbance, and believed there would be none. He did no believe there would be any attempt to arrest him again, but should there be, he repeated his warning, not, as he said, in a spirit of bravado, but to let his friends know that he and his friends were prepared for any emergency. This he several times repeated. He then announced his intention of keeping his mouth closed until after the Democratic Convention at Chicago, when he would make his purpose known.

CINCINNATI, June 17.

A despatch from Dayton to the Commercial says: "There is but little doubt that Vallandigham's arrival was unex-pected to his friends. His house was open yesterday and a large number of his friends called on him."

#### Vallandigham's Return.

Vallandigham's Return. A Washington dispatch to the New York Herald says: "A key to the policy of the President to be pursued to-ward Vallandigham has been recently given in a meeting between the Kentucky delegation in Congress and Mr. Lin-coln relative to the case of Colonel Wolford. This officer, it will be remembered, was arrested by General Burbridge and sent to Washington, where he has since remained, re-porting daily to the War Department. In answer to the request that the order of General Burbridge be rescinded, the President replied that he should not depart from the policy before pursued concerning Vallandigham. Mr. Mal-lory remarked that the Vallandigham order was inoperative, that fudividual having returned to Ohio. Mr. Lincoln re-plied, in substance, that he had no official knowledge of Vallandigham's return, and that when Mr. Vallandigham made his presence known by objectionable acts, the Execu-tive would be prepared to act. The application in favor of Golonel Wolford was not granted."

# SUSPENSION OF THE WRIT OF HA-BEAS CORPUS.

1861.

April 27-The PRESIDENT issued to Lieut. General Scott this order:

You are engaged in suppressing an insurrection against the laws of the United States. If at any point on or in the vicinity of any military line which is now or which shall be used between the city of Philadelphia and the city of Washington, you find resistance which renders it necessary to suppress the prior of balance or the public software. to suspend the writ of habeas corpus for the public safety, you personally, or through the officer in command, at the point at which resistance occurs, are authorized to suspend that writ. ABRAHAM LINCOLN.

By the President: WM. H. SEWARD, Secretary of State.

July 2-This order was extended to the military line between New York and Washington.

May 10-The PRESIDENT issued a proclamation authorizing the commander of the forces of the United States on the Florida coast, "if he shall find it necessary, to suspend there the writ of habeas corpus, and to remove from the vicinity of the United States fortresses all dangerous or suspected persons."

#### 1862.

### WASHINGTON, September 24.

Whereas, it has become necessary to call into service, not only volunteers, but also portions of the militia of the State by draft, in order to suppress the insurrection existing in the United States, and disloyal persons are not adequately restrained by the ordinary processes of law from hindering this measure, and from giving aid and comfort in various ways to the insurrection :

Now, therefore, be it ordered :

First. That during the existing insurrection, and as a necessary measure for suppressing the same, all rebels and insurgents, their aiders and abettors, within the United States, and all persons discouraging volunteer enlistments, resisting military drafts, or guilty of any disloyal practice affording aid and comfort to the rebels against the authority of the United States, shall be subject to martial law, and liable to trial and punishment by courts-martial or military commission.

Second. That the writ of habeas corpus is suspended in respect to all persons arrested, or who

12

are now, or hereafter during the rebellion shall be, imprisoned in any fort, camp, arsenal, military prison, or other place of confinement, by any military authority, or by the sentence of any court-martial or military commission.

In witness whereof, I have hereunto set my hand and caused the seal of the United States to be affixed.

Done at the city of Washington, this twentyfourth day of September, in the year of our Lord one thousand eight hundred and sixty-two, and of the independence of the United States the eighty-seventh.

#### ABRAHAM LINCOLN.

By the President:

WM. H. SEWARD, Secretary of State. 1863.

#### GENERAL SUSPENSION OF THE WRIT.

Whereas the Constitution of the United States has ordained that the privilege of the writ of habeas corpus shall not be suspended unless when in cases of rebellion or invasion the public safety may require it; and whereas, a rebellion was existing on the 3d day of March, 1863, which rebellion is still existing: and whereas by a statute which was approved on that day it was enacted by the Senate and House of Representatives of the United States in Congress assembled, that during the present insurrection the President of the United States, whenever in his judgment the public safety may require, is authorized to suspend the privilege of the writ of habeas corpus in any case throughout the United States, or any part thereof; and whereas in the judgment of the President the public safety does require that the privilege of the said writ shall now be suspended throughout the United States in the cases where, by the authority of the President of the United States, military, naval, and civil officers of the United States, or any of them, hold persons under their command or in their custody either as prisoners of war, spies, or aiders or abettors of the enemy, or officers, soldiers, or seamen enrolled or drafted or mustered or enlisted in or belonging to the land or naval forces of the United States, or as deserters therefrom, or otherwise amenable to the military law or the Rules and Articles of War, or the rules or regulations prescribed for . the military or naval services by authority of the President of the United States, or for resisting a draft, or for any other offence against the military or naval service :

Now, therefore, I, ABRAHAM LINCOLN, President of the United States, do hereby proclaim and make known to all whom it may concern, that the privilege of the writ of habeas corpus is suspended throughout the United States in the several cases before mentioned, and that this suspension will continue throughout the duration of the said rebellion, or until this proclamation shall, by a subsequent one to be issued by the President of the United States, be modified or revoked. And I do hereby require all magistrates, attorneys, and other civil officers within the United States, and all officers and others in the military and naval services of the United States, to take distinct notice of this suspension, and to give it full effect, and all

citizens of the United States to conduct and govern themselves accordingly and in conformity with the Constitution of the United States and the laws of Congress in such case made and provided.

In testimony whereof, I have hereunto set my hand and caused the seal of the United States to be affixed, this 15th day of September, 1863, and the independence of the United States of America the eighty-eighth.

#### ABRAHAM LINCOLN.

By the President:

#### WM. H. SEWARD, Secretary of State.

#### 1864.

# SUSPENSION OF THE WRIT IN KENTUCKY.

Whereas, by a proclamation which was issued on the 15th day of April, 1861, the President of the United States announced and declared that the laws of the United States had been for some time past, and then were, opposed and the execution thereof obstructed, in certain States therein mentioned, by combinations too powerful to be suppressed by the ordinary course of judicial proceedings, or by the powers vested in the marshals by law; and whereas, immediately after the issuing of the said proclamation, the land and naval forces of the United States were put into activity to suppress the said insurrection and rebellion; and whereas the Congress of the United States, by an act approved on the 3d day of March, 1863, did enact that during the said rebellion the President of the United States, whenever in his judgment the public safety may require it, is authorized to suspend the privilege of the writ of habeas corpus in any case throughout the United States, or in any part thereof; and whereas the said insurrection and rebellion still continue, endangering the existence of the Constitution and Government of the United States; and whereas the military forces of the United States are now actively engaged in suppressing the said insurrection and rebellion in various parts of the States where the said rebellion has been successful in obstructing the laws and public authorities, especially in the States of Virginia and Georgia;

And whereas, on the fifteenth day of September last, the President of the United States duly issued his proclamation, wherein he declared that the privilege of the writ of habeas corpus should be suspended throughout the United States in cases where, by the authority of the President of the United States, military, naval, and civil officers of the United States, or any of them, hold persons under their command or in their custody, either as prisoners of war, spies, or aiders or abettors of the enemy, or officers, soldiers, or seamen enrolled or drafted or mustered or enlisted in or belonging to the land or naval forces of the United States or as deserters therefrom, or otherwise amenable to military law or the rules and articles of war, or the rules or regulations prescribed for the military or naval service by authority of the President of the United States, or for resisting a draft, or for any other offence against the military or naval service;

And whereas many citizens of the State of

Kentucky have joined the forces of the insurgents, and such insurgents have on several occasions entered the said State of Kentucky in large force, and, not without aid and comfort furnished by disaffected and disloyal citizens of the United States residing therein, have not only greatly disturbed the public peace, but have overborne the civil authorities and made flagrant civil war, destroying property and life in various parts of that State;

And whereas it has been made known to the President of the United States by the officers commanding the national armies that combinations have been formed in the said State of Kentucky with a purpose of inciting rebel forces to renew the said operations of civil war within the said State, and thereby to embarass the United States armies now operating in the said States of Virginia and Georgia, and even to endanger their safety:

to endanger their safety: Now, therefore, I, ABRAHAM LINCOLN, Presi-dent of the United States. by virtue of the authority vested in me by the Constitution and laws, do hereby declare that, in my judgment, the public safety especially requires that the suspension of the privilege of the writ of habeas corpus, so proclaimed in the said proclamation of the 15th of September, 1863, be made effectual and be duly enforced in and throughout the said State of Kentucky, and that martial law be for the present established therein. I do, therefore, hereby require of the military officers in the said State that the privileges of the writ of habeas corpus be effectually suspended within the said State, according to the aforesaid proclamation, and that martial law be established therein, to take effect from the date of this proclamation, the said suspension and establishment of martial law to continue until this proclamation shall be revoked or modified, but not beyond the period when the said rebellion shall have been suppressed or come to an end. And I do hereby require and command, as well all military officers as all civil officers and authorities existing or found within the said State of Kentucky, to take notice of this proclamation and to give full effect to the same.

The martial law herein proclaimed, and the things in that respect herein ordered, will not be deemed or taken to interfere with the holding of lawful elections, or with the proceedings of the constitutional Legislature of Kentucky, or with the administration of justice in the courts of law existing therein between citizens of the United States in suits or proceedings which do not affect the military operations or the constituted authorities of the Government of the United States.

In testimony whereof I have hereunto set my hand and caused the seal of the United States to be affixed.

Done at the city of Washington, this 5th day of July, in the year of our Lord, 1864, and of the independence of the United States the eighty-ninth.

ABRAHAM LINCOLN. By the President: WILLIAM H. SEWARD, Secretary of State.

The Military Governors, appointed by the

President, were clothed with like power. This | is the letter of appointment:

WAR DEPARTMENT, WASHINGTON CITY, D. C., May 19, 1862. SIR: You are hereby appointed military governor of the State of North Carolina, with authority to exercise and perform within the limits of that State all and singular the perform when the limits of that State all and singular the powers, duties, and functions pertaining to the office of mili-iary governor, (including the power to establish all neces-sary offices and tribunals, and suspend the writ of habeas corpus,) during the pleasure of the Prosident, or until the loyal inhabitants of that State shall organize a civil gov-ernment in conformity with the Constitution of the United States FLANDAN EDWIN M. STANTON. States. Secretary of War.

Note.—Major General McCLELLAN authorized General BANKS to suspend the writ of habeas corpus, if necessary in carrying out the instructions of the former for the arrest of certain members of the Maryland Legislature in October, 1861. (See "Military Orders respective Flortion.")

# ACTION OF CONGRESS. First Session, Thirty-Seventh Congress. IN SENATE.

1861, July 29-Mr. TRUMBULL, from the Committee on the Judiciary, reported back the memorial of Charles Howard and others, Police Commissioners of Baltimore, arrested and confined as prisoners in Fort McHenry, and asked to be discharged from the subject, the Committee being of opinion that no legislation by Congress is practicable with reference to the matter set forth in the memorial.\*

Mr. BAYARD proposed to amend the report by substituting:

Resolved, That the members of the police board of the city of Baltimore ought to be either surrendered to the civil

city of Baltimore ought to be either surrendered to the civil authorities on some charge sufficient in law for their arrest and detention, or be discharged from confinement at Fort McHenry, and suffered to resume their official functions. *Resolved*. That the control of the municipal police of Bal-timore ought to be restored to those civil officers to whom, by the laws of Maryland, it is intrusted. *Resolved*. That George P. Kane, marshal of police in the civil authorities on some charge sufficient in law to hold him in custody, or be discharged from confinement in Fort McHenry. McHenry.

The subject was postponed.

Aug. 6-Mr. Powell proposed in the Senate a resolution similar to Mr. May's in the House, but the motion to take it up was lost-yeas 7, nays 33. The seven were Messrs. Breckinridge, Bright, Johnson of Missouri, Latham, Polk, Powell, Saulsbury.

### IN HOUSE.

1861, July 31-Mr. MAY offered this resolution :

Whereas the Constitution of the United States declares that no warrant shall issue but upon probable cause, sup-ported by oath or affirmation; it nat no citizen shall be de-prived of his liberty without due process of law; and that the accused shall enjoy the right of a speedy trial by a jury of the district where the offence was committed: and whereas Charles Howard, William H. Gatchell, and John W. Davis, citizen soft Baltimore; in the State of Maryland, were, on the 1st day of July, 1861, seized without warrant, and without any process of law whatever, by a body of soldiers from the Army of the United States, by order of Major General Banks, alleged to have been made in pursu-ance of orders issued from the headquarters of the Army at Washington, and were removed by force and against their will from their homes to Fort McHenry, where they have ever since been, and now are, confined as prisoners; and whereas the said military officer, without warrant or au-thority of law, superceded and suspended the official func-tions of hols of Ochrises Howard and others, members of the board of police of Baltimore; and whereas, since their sold. Whereas the Constitution of the United States declares board of police of Baltimore : and whereas, since their said arrest, a grand jury attending the United States district

\* For further facts, see "The Conspiracy in Maryland."

court in Baltimore, and selected and summoned by a mar-shal appointed by the present Executive of the United States, having jurisdiction in the premises, and having fully investigated all cases of alleged violation of law, has finally adjourned its session without finding any presentment or indictment or other proceeding against them, or either of them; and the President of the United States, being re-quested by a resolution of the House of Representatives to communicate the grounds, reasons and evidence for their arrest and imprisonment, has declined so to do, because he is advised that it is incomnatible with the nublic interests: is advised that it is incompatible with the public interests: and whereas, since these proceedings, the said citizens, with others, have been, by force and against their wills, trans-ferred by the authority of the Government of the United States beyond the State of Maryland and the jurisdiction of that court which it is their constitutional right to claim, and are to be subjected to an indefinite, a hopeless, and cruel imprisonment in some fort or military place, unfit for the confinement of the citizen, at a remote distance from their confinement of the citizen, at a remote distance from their families and friends, and this without any accusation, inves-tigation, or trial whatever: and whereas the constitutional privilege of the writ of *habcas corpus* has been treated with contempt, and a military officer (the predecessor of General Banks) has taken upon himself the responsibility of wilful disobedience to the writ, and the privilege of the same now continues suspended, thereby subordinating the civil to the military nover thus violating and overthrowing the the military power, thus violating and overthrowing the Constitution of the United States, and setting up in its stead a military despotism : and whereas the Congress of the United a initiary despotsini and whereas the congress of the Dinted States regards the acts aforesaid as clear and palpable vio-lations of the Constitution of the United states, and de-structive to the liberties of a free people: Therefore, *Resolved*, That the arrest and imprisonment of Charles Howard, William H. Gatchell, and John W. Davis, and

others, without warrant and process of law, is flagrantly unconstitutional and illegal; and they should, without de-lay, be released, or their case remitted to the proper judicial tribunals, to be lawfully heard and determined.

Ruled out of order under the rule regulating the business of the session.

# Second Session, Thirty-Seventh Congress. IN SENATE.

1861, Dec. 16-Mr. TRUMBULL offered the following resolution :

Resolved, That the Secretary of State be directed to inform the Senate whether, in the loyal States of the Union, any person or persons have been arrested and imprisoned. and are now held in confinement by orders from him or his Department; and if so, under what law said arrests have been made, and said persons imprisoned.

Which was referred to the Judiciary Committee-yeas 25, nays 17, as follows :

YEAS-Messrs. Anthony, Browning, Chandler, Clark, Col-lamer, Cowan, Dixon, Doclittle, Fessenden, Foot, Foster, Harris, Howe, Johnson of Tennessee, King, Lanc of Indiana,

Marill, Bowe, Jourson of Tennessee, King, Lake of Indulate, Morrill, Pomeroy, Rice, Sherman, Simmons, Sumner, Ten Eyck, Wade, Wilson-25. NAYS-Messes. Bayard, Bright, Carlile, Grimes, Hale, Harlan, Kennedy, Latham, McDougall, Nesmith, Pearce, Powell, Saulsbury, Thomson, Trumbull, Wilkinson, Willey--17.

1861, December 23-Mr. King offered this resolution, which was referred to the Commit-tee on the Judiciary:

Resolved, That the President be requested to cause pro-ceedings to be instituted in the courts of law against perceedings to be instituted in the courts of law against per-sons who have been arrested by executive authority or order since the breaking out of the present insurrection, and who are now detained in custody, so that a judicial examination may be had in each case to ascertain who, if any of them, may be allowed to take the each of allegiance to the United States, and be discharged, and who shall be detained for a further examination or be prosecuted for treason or other crime; and to communicate the names of all persons that have been so arrested, and the date of arrest, to the Senate.

# CASE OF GENERAL CHARLES P. STONE.

1862, April 11-Mr. McDougall offered this resolution :

Resolved, That the Secretary of War be requested to in-form the Senate at once on the following points, namely: 1. Whether or not Brigadier General Charles P. Stone has been arrested by any person in authority in the War De-partment or in the Army of the United States; and if he has been so arrested, from whom the order for General Stone's

arrest originally proceeded—whether the Secretary himself or the general then commanding the army of the Potomac. 2. Also, whether at the time of such arrest General Stone was not subject to the Articles of War, and entitled to the benefit of them; and if he was so subject and entitled, whether or not he was arrested for a violation of any and which of those articles; and on whose complaint General Stone was arrested, and by whom, if by any persons, charges have been preferred against him; and that the Secretary of War be requested to communicate to the Senate the speci-fications under such charges as fully as his present infor-mation will enable him to state them. 3. Also, whether any, and if any, what, steps have been taken toward the paration of such charges and specifications; and if any buch steps have been so taken, whether or not the prose-cution of the matter has been intrusted to the judge advo-cate general of the Army, or of the army of the Potomac, or to some other, and what other, special judge advocate; and if not to either of said judges advocate general, why the case of General Stone did not take the customary course when a general officer is arrested, and whether or not either of the judges advocate above specially named has been, and when first, consulted in this matter. 4. Also, whether or not General Stone has at any time, and when, and how often, in person or by counsel, applied for an immediate trial; and whether he has not represented to the Secretary of War the injustice which he supposed would result to him from deferring his trial by reason of the death of important wit-nesses in any manner connected with the administration of nesses in any manner connected with the administration of his late command upon the Potomac; and what answer, if any, has been made to such representation by or on behalf of General Stone. 5. Also, whether or not the substance of said charges, more or less, has been in any, and what way, and upon whose application, communicated to Gen-eral Stone; and if not, why not; and if not, whether or not General Stone has applied directly or indirectly for such charges. 6. Also, whether any, and what, privileges have inured to General Stone under the Articles of War Nos. 74, 79, 80, and 82, and Nos. 221 and 223 of the Revised Reg-ulations of the Army, and what degree of confinement was originally ordered in reference to General Stone, and when, whether any and what change has been made, and when, from its original severity. 7. Also, if General Stone was from its original severity. 7. Also, if General Stone was not arrested for some alleged violation of the Articles of War, upon what pretence is he kept in close custody.

April 22-On motion of Mr. WILSON, the resolution was amended so as to read thus, and passed :

That the President of the United States be requested to communicate to the Senate any information touching the arrest and imprisonment of Brigadier General Stone, not deemed incompatible with the public interest.

May 2-The PRESIDENT transmitted this message in reply:

#### EXECUTIVE MANSION WASHINGTON, May 1, 1862.

To the Senate of the United States :

In answer to the resolution of the Senate in relation to Brigadier General Stone, I have the honor to state that he Magazier General Picher, inter a moder in general authority, and upon evidence which, whether he be guilty or inno-cent, required, as appears to me, such proceedings to be had against him for the public safety. I deem it incompatible with the public interest, as also, perhaps, unjust to General

Stone, to make a more particular statement of the evidence. He has not been tried because, in the state of military operations at the time of his arrest and since, the officers to constitute a court-martial and for witnesses could not be withdrawn from duty without serious injury to the service. He will be allowed a trial without any unnecessary delay; the charges and specifications will be furnished him in due season, and every facility for his defence will be afforded him by the War Department.

# ABRAHAM LINCOLN.

1862, May 14-Mr. Powell offered the following

Resolved, That the Secretary of State be directed to in-form the Senate, how many citizens of Kentucky have been arrested and confined outside the limits of the State by his order since 1st September last, and state the names of such citizens, places where imprisoned, and how long confined, and also the number and names of persons released and where imprisoned.

Mr. SUMNER offered the following as a substitute :

Resclved, That the President of the United States be requested to communicate to the Senate, if not incompatible with the public interests, any information in his possession

touching the arrest of persons in Kentucky since the 1st of September, 1861, and their imprisonment beyond the limits of that State.

Which was adopted—yeas 30, nays 7, as follows:

YEAS—Messrs. Anthony, Browning, Chandler, Clark, Col-lamer, Dixon, Doolittle, Fessenden, Foot, Foster, Grimes, Harlan, Harris, Howard, Howe, King, Lane of Indiana, Lane of Kansas, Morrill, Pomeroy, Sherman, Simmons, Sunner, Ten Eyck, Trumbul, Wade, Wilkinson, Wilmot, Wilson of Massachusetts, Wright—30. NAYS—Messrs. Davis, McDougal, Nesmith, Powell, Sauls-bury, Willey, Wilson of Missouri—7.

#### IN HOUSE.

1861, December 10-Mr. BINGHAM, from the Judiciary Committee, asked to be discharged from the further consideration of the memorial of Messrs. Howard, Gatchell, and Davis, Police Commissioners of Baltimore.

Mr. PENDLETON moved to recommit the report, with these instructions :

Resolved, That the Congress alone has the power, under the Constitution of the United States, to suspend the privi-lege of the writ of habeas corpus; that the exercise of that power by any other department of the Government is a usurpation, and therefore dangerous to the liberties of the usurpation, and therefore dangerous to the herites of the people; that it is the duty of the President to deliver Charles Howard, William H. Gatchell, and John W. Davis to the custody of the marshal of the proper district, if they are charged with any offence against the laws of the United States, to the end that they may be indicted, and "enjoy the right of a speedy and public trial by an impartial jury of the State and district wherein the crime" is alleged to have been committed.

Mr. BINGHAM moved to lay the whole subject on the table, which was agreed to-yeas 108, nays 26, as follows :

 Mays 26, as 10100vs.
 YEAS-Messrs. Aldrich, Alley, Arnold, Babbitt, Goldsmith F. Baily, Baker, Baxter, Beaman, Bingham, Francis P. Blair, Jacob B. Blair, Samuel S. Blair, Blake, Bufinton, Burnham, Calvert, Campbell, Chamberlin, Clark, Cobh, Colfax, Frede-rick A. Conkling, Roscoe Conkling, Cravers, Davis, Dawes, Delano, Diven, Duell, Dunlay, Dunn, Edgerton, Edwards, Eliot, English, Fenton, Fessenden, Franchot, Frank, Goadwin, Granger, Grider, Huight, Hale, Hanchett, Harrison, Hohman, Hooper, Horton, Hutchins, Julian, Kelley, Francis W. Kel-logg, William Kellogg, Killinger, Lansing, Law, Loomis, Lovejoy, McPherson, Mallory, Maynard, Menzles, Moorhead, Anson P. Morrill, Justin S. Morrill, Nixon, Noell, Olin, Pat-ton, T. G. Phelps, Pile, Pomeroy, Porter, Potter, Alexander H. Rice, Riddle, Edward H. Rollins, Sargent, Sedgwick, Shanks, Shefield, Shellabarger, Sherman, Sloan, Smith, Ste-vens, Stratton, Benjamin F. Thomas, Francis Thomas, Train, Trimble, Trowbridge, Upton, Van Horn, Van Wyck, Verres, Wall, Wallace, Charles W. Walton, E. P. Walton, Wash-burne, Wheeler, Albert S. White, Wilson, Woodruff, Wor-cester, Wight-108.
 MArs-Messers, Allen, Ancona, Joseph Baily, Biddle, George H. Broome, William G. Brown, Cooper, Fouke, Harding, Johnson, Lazear, May, Morris, Noble, Norton, Pendleton, Perry, Robinson, Shiel, John B. Steele, William G. Steele, Vallamakan, Wadsworth, Ward, Chilton A. White, Wiek-Vighan, Wadsworth, Ward, Chilton A. White, Wiek-Vighe-26. YEAS-Messrs. Aldrich, Alley, Arnold, Babbitt, Goldsmith

liffe-26.

# Third Session, Thirty-Seventh Congress. IN SENATE.

### 1862, December 2-Mr. Powell, offered the following resolution :

Resolved, That the President be requested to inform the Senate the number and the names of citizens of Kentucky who have been, and who are now confined in the military prisons and camps of the United States, outside the limits of said State; what are the charges against them, by whom made, and by whose order the arrests were made.

December 5-Mr. CLARK offered an amendment : to insert the words "if not incompatible with the public service;" which was agreed to, and the resolution as amended was adopted.

December 16-Mr. SAULSBURY offered the following resolution :

Resolved, That the Secretary of War be and is hereby di-

rected to inform the Senate whether Dr. John Law and Whiteley Meredith, or either of them, citizens of the State of Delaware, have been arrested and imprisoned in Fort Delaware; when they were arrested and so imprisoned; the sharges against them; by whom made; by what orders they were arrested and imprisoned; and that he communicate to the Senate all papers relating to their arrest and imprison-ment. ment

. Which was laid upon the table-yeas 29, nays 13, as follows:

YEAS - Messrs. Anthony, Arnold, Browning, Chandler, Clark, Collamer, Dixon, Doolittle, Fessenden, Field, Foet, Foster, Grimes, Hale, Harlan, Harris, Howard, Howe, King, Lane of Kanasa, Morrill, Sunner, Ten Eyck, Trumbull, Wade, Wilkinson, Wilmot, Wilson of Massachusetts, Wright

NAYS-Messrs. Bayard, Carlile, Cowan, Davis, Harding, Henderson, Kennedy, Nesmith, Powell, Rice, Saulsbury, Willey, Wilson of Missouri-13.

December 3-Mr. Powell offered the following joint resolution :

ing joint resolution : Whereas, many citizens of the United States have been seized by persons acting, or pretending to be acting, under the authority of the United States, and have been carried ont of the jurisdiction of the States of their residence and imprisoned in the military prisons and camps of the United States without any public charge being preferred against them, and without any opportunity being allowed to learn or disprove the charges made or alleged to be made against them; and whereas, it is the sacred right of every citizen that he shall not be deprived of liberty without due process of law, and when arrested shall have a speedy and public trial by an impartial jury : Therefore Be it resolved by the Senate and House of Representatives of the United States, and a usurpation of power never given by the people to the President or any other official. All such arrests are nevery constitution and taws of the United States, and a usurpation of power never pications of the Constitution of the United States; and the shall be made again the shall have a the shall hereafter case, and that all persons so arrested and yet held should

case, and that all persons so arrested and yet held should have a prompt and speedy public trial according to the pro-visions of the Constitution, or should be immediately released.

Laid on the table and printed.

1863, February 26-Mr. Powell offered the following resolution :

following resolution: Resolved, That a committee of three be appointed to in-vestigate the conduct of Colonel Gilbert, who, in command of a regiment of United States soldiers, dispersed a Demo-cratic Convention of peaceable citizens of the State of Ken-tacky, assembled at the capital of that State, on the 18th of February, 1863, for the purpose of nominating candidates for Governor and Lieutenant Governor and other State offi-cers. That said committee investigate all the facts com-metede with the aforesaid action of Colonel Gilbert and the officers and soldiers under his command; and the said com-mittee are hereby authorized to send for persons and papers, to examine witnesses; and that they authorized to ad-minister oaths to witnesses; and that said committee be authorized to hold sessions in the State of Kentucky or elsewhere, and to employ a reporter to take down testi-mony; and that they report, &c.

March 3-The Senate refused to take up the resolution-yeas 10, nays 25, as follows :

YEAS-Messrs. Carlile, Cowan, Davis, Lane of Kansas, Latham, Nesmith, Powell, Saulsbury, Wall, Wilson of Mis-

Landmit, Jesmith, Jones, Jones, Januards, J., Hais, Hais, Hais, Jouri-10. NATS-Messrs, Anthony, Arnold, Clark, Collamer, Dix-on, Fessenden, Foot, Foster, Grimes, Harding, Harlm, Har-ris, Henderson, Hicks, Howe, Morril, Pomeroy, Sumner, Ten Eyck, Wade, Wilkinson, Willey, Wilmot, Wilson of Massachusetts-25.

THE CASE OF MADISON Y. JOHNSON.

Feb. 2-Mr. RICHARDSON offered this resolution :

Resolved, That a committee of three be appointed to in-vestigate the facts in reference to the arrest and imprison-ment of Madison Y. Johnson, and that said committee have the power to send for persons and papers, to examine wit-nesses under oath, and administer oaths to said witnesses.

The memorial of Madison Y. Johnson was read when Mr. Hows, of Wisconsin, moved to

lay the whole subject on the table ; which was agreed to-yeas 22, nays 16, as follows:

YEAS—Wessrs. Anthony, Arnold, Chandler, Clark, Fessen-den, Foot, Fostor, Grimes, Hale, Harlan, Harris, Hicks, Howard, Howe, King, Lane of Kansas, Morrill, Pomeroy, Sumner, Wade, Wilkinson, Wilson of Massachusetts—22. NATS—Messrs. Bayard, Carlile, Davis, Harding, Hender-son, Kennedg, Latham, McDougall, Fowell, Rice, Richardson, Saulsbury, Turpie, Wall, Willey, Wilson of Missouri—16.

#### IN HOUSE.

1862, Dec. 1-Mr. Cox offered the following preamble and resolution :

Whereas, many citizens of the United States have been seized by persons acting, or pretending to be acting, under the authority of the United States, and have been carried out of the jurisdiction of the States of their residence, and imprisoned in the military prisons and camps of the United States, without any opportunity being allowed to learn or disprove the charges made, or alleged to be made, against them, and without any opportunity being allowed to learn or disprove the charges made, or alleged to be made, against them: and whereas, such arrests have been made in States where there was no insurrection or rebellion, or pretence thereof, or any other obstruction against the authority of the Government: and whereas, it is the sacred right of every citizen of the United States, that he shall not be de-prived of liberty without due process of law, and when ar-rested, that he shall have a speedy and public trial by an impartial jury of his conntrymen: Therefore, *Resolved*, That the House of Representatives do hereby condemn all such arrests as unwarranted by the Constitu-tion and laws of the United States, and as a usurgation of power never given up by the people to their rulers, and do hereby demand that all such arrests shall hereafter cease, and that all persons so arrested and yet held should have a con-Whereas, many citizens of the United States have been

and that all persons so arrested and yet held should have a prompt and public trial, according to the provisions of the Constitution.

Which was laid on the table -- yeas 80, nays The NAYS were: 40.

Messrs. Ancona, Baily, Biddle, Jacob B. Blair, Cal-vert, Corning, Cox, Crittenden, English, Fouke, Granger, Grider, Haight, Hall, Harding, Holman, Knopp, Law, La-zear, Menzies, Morris, Noble, Norton, Nugen, Odell, Price, Richardson, Shefield, Shiel, John B. Steele, Wilkiam G. Steele, Stiles, Benjamin F. Thomas, Francis Thomas, Val-landigham, Word, Chillon A. White, Wickliffe, Wright, Yeanan-40.

December 1-Mr. RICHARDSON offered the following resolution :

Resolved, That the President of the United States be re-quested to inform this House what citizens of Illinois are now confined in the Forts Warren, Lafayette, and Delaware, now contined in the Forts Warren, Lafayette, and Delaware, or the Old Capitol prison, and any other forts or places of confinement; what the charges are against said persons; also the places where they were arrested. That the Presi-dent be further requested to inform this House of the names of the persons that have been arrested in Illinois and taken to and confined in prisons outside of the limits of said State, and who have been released, what were the charges against each of them, by whom the charges were made, also by whose order said arrests. law for such arrests.

Which was laid upon the table-yeas 74, The NAYS werenays 40.

Messrs. Ancona, Baily, Biddle, Calvert, Roscoe Conkling, Conway, Corning, Coz, Crittenden, Dunn, English, Fouke, Granger, Grider, Hall, Harding, Holman, William Kellogg, Knapp, Law, Lazear, Leary, Menzies, Morris, Noble, Nor-ton, Nugen, Odel, Porter, Price, Richardson, Shiel, John B. Steele, William G. Steeles, Stiles, Benjamin F. Thomas, Va-landigham, Ward, Chilton A. White, Wright-40.

Dec. 22-Mr. MAY offered the following resolution :

Resolved, That the Secretary of State be requested to Resolved, That the Scoretary of State be requested to communicate to this House a copy of an order which, on or about the 28th of November, 1861, he caused to be read to State prisoners confined in Fort Warren, wherehy they were forbidden to employ connsel in their behalf, and informed that such employment of counsel would be regarded by the Government and by the State Department as a reason for prolonging the term of their imprisonment.

Which was laid upon the table-yeas 63, nays 48. The NAYS were-

Messrs. William Allen, William J. Allen, Ancona, Biddle,

Burnham, Calvert, Clemens, Cobb, Coz, Cravens, Crittenden, Dunn, English, Granger, Grider, Hale, Harding, Johnson, William Kellogg, Kerrigan, Knapp, Law, Lazear, Leary, May, Morris, Noble, Norton, Nugen, Pendleton, Price, Rob-inson, James S. Rollins, Shiel, Smith, Benjamin F. Thomas, Francis Thomas, Vallandipham, Vibbard, Voorhees, Wads-worth, Ward, Chilton A. White, Wickliffe, Woodruff, Wor-cester, Wright, Feaman-48.

December 15-Mr. PENDLETON offered the following resolution :

Resolved, That the President be requested to inform this House, if in his opinion not inconsistent with the public interest, whether in any oath of allegiance or parole required to be taken by any prisoner held in custody as a socalled political prisoner, there has been inserted a clause to the effect that he should not bring suit for the recovery of damages for such imprisonment, or that he should not op-pose, by speech or otherwise, the war measures of the Administration.

Which was laid on the table by the following vote-yeas 77, nays 43.

March 3-Mr. MAY offered the following resolution :

Whereas it is represented that Major General Schenck, Whereas it is represented that Major General Schenck, commanding the forces of the United States stationed in Baltimore, Maryland, has ordered, as a condition to be an-nexed to the worship of Almighty God by certain religious societies or congregations of the Methodist Church of that city, that the flag of the United States shall be conspicu-ously displayed at the time and place of such worship: and whereas the said order is a plain violation of the inalien-able right to worship for a condinate to the distates of such able right to worship God according to the dictates of every one's conscience, as it is asserted by the said congregations, and also by our declarations of fundamental rights and se-cured by our State and Federal Constitutions : and whereas a minister of the said congregation, the Rev. John H. Dash-iell, having, on Monday, the 15th ultimo, removed the said flag from his own premises, which was also the place of worship of one of said congregations, where the said flag had been placed surreptitionsly by some evil-minded per-son, and for so doing was arrested by order of the said Gen-eral Schenck and held as a prisoner: Therefore, *Be it resolved*, That the Judiciary Committee be, and

hereby is, instructed to inquire into the allegations afore-said, and ascertain by what authority the said General Schenck exercises a power to regulate or interfere with the privileges of divine worship, and also to arrest and detain as a prisoner the said minister of the Gospel, as aforesaid; and, further, that said committee be instructed to report upon the same at an early day.

The House refused to suspend the rules to get the resolution before the House-ayes 28, noes 79, (yeas and nays not called )

### First Session, Thirty-Eighth Congress.

1863, December 17-Mr. HARRINGTON offered this resolution :

Whereas the Constitution of the United States (article one, section nine) provides: "The privilege of the writ of habeas corpus shall not be suspended, unless when, in cases of rebellion or invasion, the public safety may require it:" and whereas such provision is contained in the portion of the Constitution defining legislative powers, and not in the provisions defining executive power; and whereas the Constitution (article four of Amendments) further provides: "The right of the people to be secure in their persons, houses, papers, and effects, against unreasonable searches and seizures, shall not be violated," &c.; and whereas the Thirty-Seventh Congress did by act, claim to confer upon the President of the United States the power, at his will and pleasure, to suspend the privilege of the writ of *habeus* corpus throughout the United States, without limitations or conditions; and whereas the President of the United States, by proclamation, has assumed to suspend such priv-ileges of the citizen in the loyal States; and whereas the people of such States have been publicated to orbitry and people of such States have been ensubjected to arbitrary ar-rests without process of law, and to unreasonable search and seizures, and have been denied the right to a speedy trial and investigation, and have languished in prisons at the arbitrary pleasure of the Chief Executive and his mili-

the arbitrary pleasure of the Unit Executive and his min-tary subordinates: Now, therefore, *Resolved by the House of Representatives of the United States*, That no power is delegated by the Constitution of the United States, either to the legislative or executive power, to suspend the privileges of the writ of habeas corpus in any State loyal to the Constitution and Government not

invaded, and in which the civil and judicial powers are in full operation. 2. Resolved, That Congress has no power under the Con-

stitution to delegate to the President of the United States the authority to suspend the privilege of the writ of habeas corpus, and imprison at his pleasure, without process of law or trial, the citizens of the loyal States.

 Resolved, That the assumption of the right by the Ex-ecutive of the United States to deprive the vitizens of such loyal States of the benefits of the writ of habeas corpus, and to imprison them at his pleasure, without process of and to imprison them at his pleasure, without process of law, is unworthy the progress of the age, is consistent only with a deepotic power unlimited by constitutional obliga-tions, and is wholly subversive of the elementary principles of freedem, upon which the Government of the United States and of the several States is based. 4. *Resolved*, That the Judiciary Committee be instructed to prepare and report a bill to this Huwas next the wide the

prepare and report a bill to this House protecting the righte of the citizens in the loyal States, in strict accordance with the foregoing provisions of the Constitution of the United States.

Which was negatived-yeas 67, nays 90, as follows:

YEAS-Messrs. James C. Allen, William J. Allen, Ancona, Augustus C. Baldwin, Bliss, Brooks, Brown, Chanler, Coj froth, Cox, Cravens, Dawson, Denison, Eden, Edgerton, E Ĕl-John, Cal, Gavens, Dawson, Denison, Luen, Lugron, Angelon, dridge, English, Finck, Ganson, Grider, Hall, Harding, Harrington, Benjamin G. Harris, Herrick, Holman, Wi-liam Johnson, Kernan, King, Knapp, Law, Le Blond, Long, Mallory, Marcy, McAllister, McDowell, McKinney, Middle-ton, William H. Miller, James R. Morris, Morrison, Nelson, Nath Odd, Luber Offert Devidence Device Pathwar Some

 Mallory, Marcy, McAlitser, McDowell, McKinney, Middle-ton, William H. Miller, James R. Morris, Morrison, Nelson, Noble, Odell, John O'Neill, Pendleton, Perry, Radford, Sam-uel J. Randall, Robinson, Rogers, Ross, Scott, John B Steele, William G. Steele, Stitles, Strouse, Sweat, Voorhees, Wads-worth, Ward, Wheeler, Chilton A. White, Joseph W. White, Winfield, Wood-eff.
 NArs-Messrs. Alley, Allison, Ames, Arnold, Ashley, John D. Baldwin, Beaman, Blaine, Blow, Boutwell, Brandegee, Broomall, William G. Brown, Ambrose W. Clark, Freeman Clarke, Clay, Cobb, Cole, Creswell, Henry Winter Davis, Thomas T. Davis, Dawes, Dixon, Donnelly, Driggs, Dumont, Eckley, Eliot, Farnsworth, Fenton, Frank, Garfield, Gooch, Grinnell, Hale, Higby, Hooper, Hotchkiss, Ashele W. Hub-bard, John H. Hubbard, Hulburd, Jenckes, Julian, Kasson, Kelley, Francis W. Kellogg, Orlando Kellogg, Loan, Long-year, Lovejoy, Marrin, McBride, McClurg, McIndoe, Samuel F. Miller, Moerhead, Morrill, Daniel Morris, Ames Myers, John H. Rice, Edward H. Rollins, Schenck, Scofield, Shan-non, Sloan, Smithers, Spalding, Stevens, Thayer, Tracy, Van Valkenburgh, Ellihu B. Washburne, William B. Wash-burn, Whaley, Williams, Wilder, Wilson, Windom, Wood-burd, Suley, Suillams, Wilder, Wilson, Windom, Wood-burd, Son. bridge-90.

1864, February 29-Mr. PENDLETON offered the following resolution :

Resolved, (as the sense of this House.) That the military arrest, without civil warrant, and trial by military com-mission without jury, of Clement L. Vallandigham, a citizen of Ohio, not in the land or naval forces of the United States or the milita in actual service, by order of Major General Burnside, and his subsequent banishment by order of the President, executed by military force, were acts of mere arbitrary power, in palpable violation of the Constitution and laws of the United States.

Which the House refused to table-yeas 33, nays 84, and then rejected-yeas 47, nays 77, as follows :

as follows: YEAS-MESSTS. James C. Allen, Ancona, Augustus C. Bald-win, Brocks, Chanler, Coffroth, Cox, Dawson, Denison, Eden, Eldridge, Finck, Ganson, Harding, Harrington, Her-rick, Holman, Hutchins, Kernan, Knapp, Law, Long, Marcy, McDowell, McKinney, William H. Miller, Morrison, Nelson, Noble, John O'Neill, Pendleton, Radford, Samuel J. Randall, Rogers, Ross, Stott, Stebbins, John B. Steele, Wil-tiam H. Wadsworth, Chilton A. White, Winfield-47. NAYS-Messrs. Alley, Allison, Anderson, Arnold, Baily, John D. Baldwin, Baxter, Francis P. Blair, jr., Blow, Bort-well, Boyd, Brandegee, Ambrose W. Clark, Freeman Clarke, Clay, Cobb, Cole, Creswell, Henry Winter Davis, Dawes, Deming, Dixon, Donnelly, Driggs, Dumont, Eckley, Eliot, Farasworth, Frank, Grinnell, Hale, Higby, Hooper, Hotch-kiss, Asahel W. Hubbard, John H. Hubbard, Jenckes, Julian, Kelley, Francis W. Kellogg, Orlando Kellogg, Loan, Marvin, McBride, McClurg, Moorhead, Morrill, Daniel Mor-ris, Amos Myers, Leonard Myers, Norton, Charles O'Neill, Patterson, Perham, Pomeroy, Price, Willian H. Randall, John H. Rice, Schenck, Scofield, Shannon, Sloan, Smithers,

Starr, Stevens, Thayer, Thomas, Upson, Van Valkenburgh, Ellihu B. Washburne, William B. Washburn, Whaley, Williams, Wilder, Wilson, Windom, Woodbridge—77.

January 25-Mr. McDowell offered the following resolutions, which were laid over under the rule:

Resolved, 1. That the House fully recognizes the great fundamental provision of the Constitution of the United States which guarantees the freedom of speech to every American citizen; and that neither the President, nor any person acting in a subordinate capacity to him, has the rightful authority to arrest and imprison a citizen of the loyal States for the utterance of sentiments distateful to the men in power. 2. That we recognize in the freedom of the press the great human to divid Bharter and that these neurons for a state of the sentence of sentiments and the three mersons for a state of the sentence of

2. That we recognize in the freedom of the press the great bulwark of civil liberty; and that those persons temporarily intrusted with power have not the rightful authority, in those States not in rebellion, to subvert this great constitutional guarantee by issuing military orders, or by a resort to any other means unknown to the laws of the country.

3. That the right to security of person from arrest in the loyal States, when no crime is charged, is a sacred right guaranteed to every citizen; and that neither the President, nor any one acting by his authority, has the legal right to arrest, imprison, or transport our people without "due process of law," requiring affidavit, warrant, arrest, and trial by a jury of the country, impartially selected. 4. That the privilege of the writ of *labeas corpus* is a fundamental and inherent right belonging to the American neonle, solenuly, guaranteed by express unovision of the

4. That the privilege of the writ of *habeas corpus* is a fundamental and inherent right belonging to the American people, solennily guaranteed by express provision of the Constitution, that cannot be denied to the citizens of the loyal States, where the courts are open and the administration of justice is unobstructed, and "invasion and rebellion" do not exist.

5. That the Constitution of the United States is one of expressed and limited powers, and that neither Congress nor the Executive have the "lawful right" to interfere with the established rights and domestic institutions of the several States. 6. That we reaffirm our unalterable devotion to the Con-

6. That we reaffirm our unalterable devotion to the Constitution of the United States, and to each and every provision thereof, as framed by the fathers, including those provisions relating to the rights of property and the inviolability of contracts, as understood and interpreted by the Supreme Court of the United States.

March 21-Mr. ELDRIDGE offered this resolution, which was laid over under the rule :

Resolved, That the President of the United States be respectfully requested, and that the Secretary of State and the Secretary of War be directed, to report and furnish to this Honse the names of all persons, if any there are, arrested and held in prison or confinement in any prison, fort, or other place whatsoever, for political offences, or any other alleged offence against the Government or authority of the United States, by the order, command, consent, or knowledge of them or either of them, respectively, and who have not been charged, tried, or convicted before any civil or criminal (not inilitary) court of the land, together with the charge against such person, or cause for such arrest and imprisonment, if there be any, and the name of the prison, fort, or place where they are severally kept or confined. Also, whether any person or persons, for any alleged like offence, have been banished or sent from the United States, or from the States not in relellion to the rebellious States; and the names, times, alleged offence or cause thereof; and whether with or without trial; and if tried, before what court.

April 4---The resolution, on motion of Mr. EDWARD H. ROLLINS, was laid upon the table---yeas 62, nays 40. The NAYS were---

Messrs, James C. Allen, Ancona, A. C. Baldwin, Bliss, J. S. Brown, Chanler, Coz, Cravens, Dawson, Denison, Eden, Eldridge, English, Finck, Grider, Griswold, Harrington, Herrick, Holman, P. Johnson, Kalufteisch, Law, Lazear, Long, Mallory, Marcy, McKinney, Middielon, J. R. Morriso, Morrison, J. O'Neill, Pruyn, S.J. Randall, Robinson, Rogers, J. B. Steele, Wheeler, C. A. While, Winfield, Yeaman-40.

June 20-Mr. Ross offered the following resolution, which went over under the rule :

Resolved, That all persons not in the military or naval service of the United States who have been arrested and imprisoned by the agents of the Government without process of law, and released without trial or examination, are entitled to the same pay and mileage for the time they were deprived of their liberties as members of Congress; and the Committee of Claims are hereby instructed to report a bill at an early day for that purpose.

Same day, in SENATE-Mr. MORBILL offered

this bill, which was referred to the Committee on the Judiciary :

Be it enacted, dc., That upon all arrests under section 6, of chapter 200, of an act approved the 17th of July, 1862, bail shall be admitted, and such bail, on the demand of the party so arrested, may be taken before any judge of the United States, any chancellor, judge of a supreme or superior court, or chief or first judge of court of common pleas of any State, who shall exercise their discretion therein, regarding the nature and circumstances of the offence, and of the evidence and the usages of the law.

# Third Session, Thirty-Seventh Congress. THE ACT OF INDEMNITY OF MARCH 3, 1863.

The Act of Robernitt of Maken 8, 1000.

Section 1 provides: That, during the present rebellion, the President of the United States, whenever, in his judgment, the public safety may require it, is authorized to suspend the privilege of the writ of habeas corpus in any case throughout the United States, or any part thereof. And whenever and wherever the said privilege shall be suspended, as aforesaid, no military or other officer shall be compelled, in answer to any writ of habeas corpus, to return the body of any person or persons detained by him by authority of the President; but upon the certificate, under oath, of the officer having charge of any one so detained that such person is detained by him as a prisoner under authority of the President, further proceedings under the writ of habeas corpus shall be suspended by the judge or court having issued the said writ, so long as said suspension by the President shall remain in force, and said rebellion continue.

Section 2 directs the Secretary of State and the Secretary of War to furnish to the judges of the circuit and district courts of the United States and of the District of Columbia, a list of the names of all persons, citizens of loyal States, held as State or political prisoners of the United States, in any fort, arsenal, or other place; and provides that where a grand jury has adjourned without finding an indictment against any such person, the judge shall forthwith make an order that any such prisoner desiring a discharge be brought before him to be discharged, and every officer of the United States is directed immediately to obey this order, under penalty of fine and imprisonment -the party first to take a prescribed oath of allegiance. Another section provides: That any order of the President, or under his authority, made at any time during the existence of the present rebellion, shall be a defence in all courts to any action or prosecution, civil or criminal, pending, or to be commenced, for any search, seizure, arrest, or imprisonment, made, done, or committed, or acts omitted to be done, under and by virtue of such order, or under color of any law of Congress, and such defence may be made by special plea, or under the general issue.

Suits begun in State courts may be transferred to United States Courts under circumstances described. Any suit described in this act may be carried on writ of error to the Supreme Court of the United States, and all suits or prosecutions for any arrest or imprisonment or other trespasses or wrongs, shall be commenced within two years.

This bill passed the House of Representa-

tives, March 2, 1863-yeas 99, nays 45, as imprisoned; and that for them redress might be had in the follows :

 Yras-Messrs. Aldrich, Arnold, Ashley, Babbitt, Baker, Baxter, Beaman, Bingham, Jacob B. Blair, Samuel S. Blair, Blake, William G. Brown, Buffinton, Campbell, Casey, Chamberlain, Clark, Colfax, Frederick A. Conkling, Roscoe Conkling, Conway, Cutler, Davis, Dawes, Delano, Jonn, Edgerton, Ellot, Ely, Fenton, Samuel C. Fessenden, Thomas A. D. Fessenden, Fisher, Flanders, Franchot, Frank, Goodwin, Gurley, Hahn, Hale, Harrison, Hooger, Horton, Hutchins, Julian, Kelley, Francis W. Kollogg, William Kellogg, Killinger, Lansing, Leary, Lehman, Loomis, Low, McIndoe, McKean, McKnight, McPherson, Marston, Maynard, Mitchell, Moorhead, Anson P. Morrill, Nixon, Olin, Patton, Timothy G. Phelps, Pike, Pomeroy, Porter, John H. Rice, Riddle, Edward H. Rollins, Sargent, Segwick, Segar, Shanks, Shellabarger, Sherman, Sloan, Spaulding, Stevens, Stratton, Francis Thomas, Trinble, Trowbridge, Van Horn, Van Valkenburgh, Van Wyck, Verree, Walker, Wall, Wallace, Washburne, Wheeler, Albert S. White, Willion, Wincester-99.
 MARS-Messrs. William Allen, William J. Allen, Ancona, Biddle, Calvert, Creaters, Crigield, Delandan, Johnson, Kerrigan, Knapp, Law, Mallory, Moy, Menzies, Morris, Note, North, Nang, Pendelton, Perry, Price, Robinson, Shiel, Smith, John B. Steele, William G. Steele, Stiles, Benjamin P. Thomas, Vallandinkam, Yoonkes, Wadsworth, Ward, Chilton A. White, Wickliffe, Wood, Woodruff, Yeanam-45. -Messrs. Aldrich, Arnold, Ashley, Babbitt, Baker,

Same day, the bill passed the SENATE, without a record of yeas and nays, owing to a misunderstanding respecting the putting of the vote.

March 3-Mr. BAYARD moved that the Secretary of the Senate be directed to request the House of Representatives, to return to the Senate the above report of the Committee of Conference; which was rejected -yeas 13, navs 25, as follows:

YEAS-Messrs. Bayard, Carlile, Davis, Henderson, Latham, Nesmith, Powell, Rice, Richardson, Saulsbury, Turpie, Willey, Wilson, of Missouri-13.

Turpie, Willey, Wilson, of Massouri-13.
NAvs-Messrs. Anthony, Chandler, Clark, Dixon, Doo-little, Foster, Grimes, Harlan, Harris, Hicks, Howard, Howe, King, Lane of Indiana, Lane of Kausas, Morrill, Pomeroy, Sherman, Sumner, Ten Eyck, Trumbull, Wade, Wilkinson, Wilmot, Wilson of Massachusetts-25.

While this subject was pending before Congress, the House, December 8, 1862, passed an indemnity bill—yeas 90, nays 45, against which, on the 22d of December, thirty-six members of the House moved to enter on the journal this protest:

Resolved, That the following protest of thirty-six mem-bers of this House against the passage of the House bill No. 591 be entered upon the Journal :

On the 8th day of December, A. D. 1862, and during the present session of Congress, Mr. STEVENS, of Pennsylvania, introduced the bill No. 591, entitled "An act to indemnify the President, and other persons, for suspending the privilege of the writ of habeas corpus, and acts done in pursu-ance thereof," and after its second reading moved that its consideration be made the special order for the Thursday then next ensuing, which motion being objected to, he moved the previous question, and this being sustained, un-der the operation thereof the bill was read a third time, and passed.

This bill involves questions of the gravest importance. It provides that all suspensions of the privilege of the writ of habeas corpus, all arrests and imprisonments upon whatever pretexts or by whomsover made, under the anthority of the President, however arbitrary or tyrannical or unjust, are confirmed and made valid; and that all persons who advised or executed or assisted in the execution of any such advised or executed or assisted in the execution of any such acts are discharged from all liability, whether to the State or to individuals "in respect thereof;" and that all proceed-ings against them of any nature, whether for the recovery of damages or for the infliction of punishment "commenced or to be commenced," are discharged and made void. It also provides that the President may, during the existence of this rebailion at any time and anywhere therechoust care and provides that the resident may, during the existence of this rebellion, at any time and anywhere throughout any of the United States, and as to any person, suspend the privilege of the writ of *habeas corpus*. The bill is framed upon the idea that the acts recited

courts of the United States, by resort to the peaceful, reg ular, and ordinary administration of the law. It is framed upon the idea that the citizen was arrested without the existence of crime on his part, or even probable cause to suspect it, and that in making such arrests, the substance, as well as the form, of those provisions of law intended to secure personal *laberty* were entirely disregarded. It makes no exception of those cases in which the arrests have been made with malice, and the imprisonments have have been made with mance, and the imprisonments have been inflicted with circumstances of brutality and crulety, in which the "public good" has been made the cloak wherewith to cover the gratification of political animosity or private harred. It distinguishes in nothing between the cases in which an honest mistake has been followed by its immediate correction, and cases in which malignity has been enabled, by false pretences, to procure the arrest and to prolome the imprisonment to the loss of propart the to prolong the imprisonment, to the loss of properts and destruction of health, and, in some instances, the insanity, suicide, or lingering death of the unhappy victim. It dis-tinguishes in nothing between the active officer, zealous in the full discharge of his official duties, and the base mis-creant who volunteers to assume the degrading character of grue and informer that he may with more affort and of spy and informer, that he may, with more effect and security, use the falsehood which the venom of his heart prompted him to invent. It proposes to condone all of-fences, to protect all offenders, and to take away all redress for injuries, however great, or with whatever circumstances of accouncing on the direction inflated of aggravation or bad motive inflicted.

If these acts had been done in all cases from the purest motives, with an eye single to the public good, with as little aggressiou as possible on private rights, with all circum-spection and care that only those who were really guilty should suffer such confinement as would prevent the commission of an unlawful act-if the public good were in fact subserved by them—it might be proper to protect the Pres-ident, and those acting under his authority, from criminal prosecution and penal sentence; it might be proper to pro-tect them from neuroinview theory means from the public prosecution and penal sentence; it might be proper to pro-tect them from pecuniary loss, by the payment, from the pub-lic Treasury, of the damages assessed against them. Even then, whilst admitting that circumstances like these would in seasons of great public dangers negative all wrongful in-tent in the commission of these illegal acts, it would be the duty of the Representatives of the neople to affirm that at duty of the Representatives of the people to affirm that at all times the President of the United States, before all other men, should adhere most strictly to the forms of legal proecdure when directing his powers against the personal lib-erty of the citizen. It could never be proper to indemnify the President, and those acting under his authority, at the expense of the citizen whom they had injured, or to add to their security by the destruction of his remedies.

The Constitution of the United States guards most care-fully the rights of the citizen; it was ordained "to estab-lish justice, insure domestic tranquillity," and to "secure the blessings of liberty;" and ao steadily was this object kept in view, that in addition to the reservation of all powers not granted, there are special prohibitions of seizures withwarrant, detentions without indictment, imprisonment out without a speedy and public trial, and depirvation of life, liberty, or property without due process of law; and there are clauses which extend the judicial power of the United States to all controversies between citizens of different States, and secure a trial by jury in all cases in which the value in controversy exceeds twenty dollars. Congress has hitherto uniformly maintained, and, as far as was neces-sary, has perfected by its legislation these guarantees of per-sonal liberty, and the courts have enforced them by the assessment of damages for their infraction. This bill proposes to deprive the courts of the power to afford such protection. It will, if carried out into practical and general operation, release the people from the duty of appealing to such peace-ful and legal means of redress, and will provoke more sum-mary and less constitutional measures. Yet this bill, without may and less constitutional measures. For this bill, where the precedent in our history, suggesting such grave questions of constitutionality and expediency, believed by many members to be utterly subversive of the rights of the citizen and of the express provisions of the Constitution, by the force of mere numbers and against the remonstrance of the minority, was passed within one hour of its first introduction, without having been printed, without having been referred to any committee, select or standing, and without any op-portunity for consideration or discussion.

The undersigned, members of the House of Representa-They notest again the House of the House of the resentatives, do therefore most solemily remonstrate against this action of the House, and respectfully ask that this their protest may be entered upon the Journal. They protest against the refusal of the House to permit consideration and discussion of the bill as an arbitrary ex-

consideration and uncession of the only as an arbitrary ex-ercise of power by the majority, unjust to the members, unjust to their constituents, and derogatory to its character as a deliberative legislative body. They protest against the passage of the bill— 1. Because it purports to deprive the citizen of all exist-ing proceeds legislative produces for admitted uncess

were illegal, and without just cause or excuse; that they 1. Because it purports to deprive the citizen of all exist-were violations of the rights of the persons arrested and ing, peaceful, legal modes of redress for admitted wrags,

184

and thus constrains him tamely to submit to the injury in-flicted or to seek illegal and forcible remedies. 2. Because it purports to indemnify the President and all acting under his authority for acts admitted to be wrongful, at the expense of the citizen upon whom the wrongful acts have been perpetrated, in violation of the plainest principles of justice, and the most familiar precepts of constitutional law.

3. Because it purperts to confirm and make valid, by act

3. Because it purports to confirm and make valid, by act of Congress, arrests and imprisonments which were not only not warranted by the Constitution of the United States, but were in palpable violation of its express prohibitions.
4. Because it purports to authorize the President, during this rebellion, at any time, as to any person, and everywhere throughout the limits of the United States to suspend the privilege of the writ of *habas corpus*, whereas hy the Constitution the power to suspend the privilege of that writ is confided to the discretion of Congress alone, and is united to the needs the store of the privilege of the store of the privilege of the the discretion of Congress alone, and is united to the needs the abace of the version. limited to the places threatened by the dangers of invasion or insurrection.

5. Because, for these and other reasons, it is unjust and unwise, an invasion of private rights, an encouragement to lawless violence, and a precedent full of hope to all who

lawless violence, and a precedent uni of hope to all who would usurp despotic power and perpetuate it by the arbi-trary arrest and imprisonment of those who oppese them. 6. And finally, because in both its sections it is "a delib-erate, palpable, and dangerous" violation of the Constitu-tion, "according to the plain sense and intention of that in-strument," and is therefore utterly null and void.

George H. Pendleton, W. A. Richardson, J. C. Robinson, P. B. Fouke, C. A. Wickliffe, Charles J. Biddle, J. A. Cravens, Elijah Ward, Philip Johnson, James R. Morris, A. L. Knapp, C. L. Vallandigham, C. A. White, John D. Stiles, D. W. Voorhees, G. W. Dunlap, Hendrick B. Wright, Warren P. Noble, W. Allen, William J. Allen, H. Grider, W. H. Wadsworth, A. Harding, Charles B. Calvert, James E. Kerrigan, Henry May, R. H. Nugen, Caoper, H. Yaoppan S. S. Cox, E. H. Norton, George K. Shiel, S. E. Ancona, J. Lazear, Nehemiah Perry, George H. Yeaman, B. F. Grauger. C. Vibbard, John Law,

The motion to enter this protest was tabledyeas 75, nays 41.

The above bill of Mr. STEVENS was amended in the Senate, and finally passed that body-yeas 33, nays 7, as follows :

yeas 35, Hays 1, as follows. YEAS-Messrs. Anthony, Arnold, Erowning, Chandler, Clark, Collamer, Cowan, Dixon, Doolittle, Fessenden, Foot, Foster, Grimes, Hale, Harlan, Harris, Henderson, Hicka, Howard, King, Lane of Indiana, Lane of Kansas, Morrill, Pomeroy, Sherman, Sumner, Ten Eyck, Trumbull, Wade, Wilkinson, Willey, Wilmot, Wilson of Massachusetts-33. NATS-Messrs. Bayard, Carlile, McDougall, Powell, Turpie, Wall Wilcom of Missouri-7.

Wall, Wilson of Misseuri-7.

The House non-concurred in the amendments, and a Committee of Conference having met, agreed upon a report, which was agreed to in both Houses as stated before.

VOTE ON SUSPENSION OF HABEAS CORPUS.

Pending the consideration of the original House bill in the Senate,

1853, Feb. 19-Mr. Powell moved to strike out the third section authorizing the President to suspend, by proclamation, the writ of habeas corpus in certain contingencies; which was rejected-yeas 13, nays 27, as follows:

YEAS-Messrs. Bayard, Carlile, Cowan, Kennedy, Latham, Nesmith, Powell, Rice, Richardson, Saulsbury, Turpie, Willey, Wilson of Missouri-13. NATS-Messrs. Anthony, Arnold, Chandler, Clark, Davis,

MAYE MESSIS AT MISSORI -15. NAYE MESSIS. ATTONY, Arnold, Chandler, Clark, Davis, Dixon, Deolittle, Fessenden, Foot, Grimes, Harris, Hender-son, Hicks, Howard, Howe, King, Lane of Indiana, Lane of Kansas, Morrill, Pomeroy, Sherman, Sumner, Ten Eyck, Trumbull, Wilkinson, Wilson of Massachusetts-27.

#### VOTE ON ARRESTS.

Feb 23-Mr. CARLILE moved this substitute for the bill :

From and after the passage of this act, and during the

present rebellion, it shall not be lawful for any officer or servant of the United States to arrest or detain any citizen of the United States who may be supposed or alleged to be disloyal thereto, or for any other cause, except upon oath or affirmation of some person or persons well known to be loyal to the United States, and particularly describing in said oath or affirmation the act of disloyalty or other cause

loyal to the United States, and particularly describing in said oath or affirmation the act of disloyalty or other cause for which the said citizen should be arrested and detained. Sec. 2. That any and every officer or servant of the United States who shall arrest or detain any citizen of the United States in contravention of the provisions of the first section of this act shall, on cornviction thereof in any court having jurisdiction in the case, suffer a fine of not less than \$10,000, or imprisonment in the penitentiary for a term not less than five years. SEC. 3. That all persons arrested under the provisions of this act upon the charge of disloyalty to the Government of the United States, or for any other cause, shall have the privilege of the writ of habeas corpus; and the said writ shall not be suspended at any time so far as the same may relate to persons arrested as aforesaid. SEC. 4. That nothing in this act shall be so construed as to prevent the arrest of any person, a citizen of any or either of the United States, who may be charged with trea-sen or disloyalty thereto: *Hrowided*, That all arrests of such persons shall be made as provided in the first section of this act, or upon the precept of the President of the United States. of this act, or upon the precept of the President of the United States.

Which was rejected-yeas 7, (Messrs. Carlile, Kennedy, Powell, Richardson, Saulsbury, Turpie, Wall,) nays 29.

#### THE ACT SUSTAINED BY THE COURTS.

The important case of George W. Jones, ex-Minister to Bogota, vs. William H. Seward, has been decided in New York by the Supreme Court. Mr. Jones was arrested on a telegraphic dispatch from Secretary Seward, and imprisoned at Fort Lafayette. When released he brought a suit for \$5,000 damages for false imprisonment. Mr. Seward, by counsel, moved to transfer the case to the United States Circuit Court, under the act of March, 1863. The motion was denied, and the General Term decided an appeal which was taken to it. The majority of the judges affirm the act; one, Clerke, dissented.

#### George W. Jones vs. William H. Seward.

LEONARD, J.—The question is not whether the 4th section of the act of Congress passed March 3, 1863, affords a valid defence to the action. The true question is this: Is it in the power of Congress to give the circuit court jurisdiction of the case?

The Constitution extends the judicial power of the Union

The Constitution extends the judicial power of the Union to all cases in law and equity arising under the Constitu-tion, laws, and treaties of the United States. The defence in this case arises under the act of Congress, and the validity of that act, considered in the light afforded by the Constitution, will be one of the principal subjects to be determined at the trial. It has been decided that a case arises within the meaning of the Constitution as well when the defendant seeks protection under a law of Congress, as when a plaintiff comes into court to demand some wight when a plaintiff comes into court to demand some right conferred by law.

It has been objected that the original jurisdiction of all actions may be drawn into the Federal courts, by similar enactments of Congress, and that the case arises within the meaning of the Constitution only after a trial and judgment in this court, when the action can be referred by a writ of error or appeal, and brought before the Federal courts for review.

The power of transferring causes to the United States Circuit in a similar manner, where the question involved was of an appellate and not original jurisdiction, has long been sustained.

been sustained. Chief Justice Marshall says, in the case of Osborn vs. The Bank of the United States (9 Wheaton, 821): "We perceive no ground on which the proposition can be maintained, that Congress is incapable of giving the circuit courts original jurisdiction, in any case to which the appellate jurisdiction extenders extends."

Congress has enacted that the defendant may interpose in his defence the orders, &c., of the President, and has directed the transfer of cases involving such a defence, in the manner prescribed, into the circuit court. According to the statements of the defendant such a case.

has arisen. We have nothing to do with the validity of the law as a defence to the action. It is sufficient for the State court that the defence involves the construction and effect of a law of Congress. The case has then arisen when the courts of the United States may have jurisdiction, if Con-gress so directs. If the law does not afford a constitutional or valid defonce, it cannot now be doubted that the learned justices of the United States Courts will so declare it, when the jurisdiction of such cases will remain in the State courts, as before the enactment of the law. It is not our duty to assert the independence of our State sovereignty and jurisdiction; for the final construction and effects of all acts of Congress may be brought before the United States

Courts by the express provision of the Constitution. The manner of taking the cause to those courts is of con-sequence. The Supreme Court of the Union must be relied on to prevent its jurisdiction from being unlawfully extended by Congress. I am of the opinion, therefore, that Congress has the power to direct the transfer of such cases.

In my opinion this application was necessary in order to vest the U.S. Circuit Court with the possession of the ac-tion, but the discussion has not been lost, inasmuch as it will be now settled that this Court will not, in this judicial district, take further cognizance of cases which have been transferred under this act of Congress. It is very proper that an order be entered transferring the cause to the U.S. Circuit, as it affords the evidence in the Court of the disposition made of it.

In arriving at my conclusions I have consulted Story's Com.on the Constitution, chap. 33, 22003, 906, &c.; 1 Wheat., Mardin vs. Hunter; 6 Wheat., Cohen vs. The State of Vir-ginia; 9 Wheat., Osborn vs. The Bank of United States. As a rule of practice I think the Court should not ap-nova any sugging subjest the amount of the band is courd

prove any surcties unless the amount of the bond is equal to the sum in which the defendant in the action has been held to bail, if bail has been required in the State Court. This fact should be made to appear to the satisfaction of the judge to whem the bond is presented for approval. The decision in this case will also embrace the case of

Gudeman vs. Wool, argued at the same general term as the present case.

The order appealed from should be reversed, and the motion below should be granted without costs.

SUTHERLAND, J .- The question is not as to the constitutionality of the fourth section of the act declaring that the order or authority of the President, during the rebellion, order or authority of the President, using the rependent shall be a defence in all Courts, to any order for any arrest, imprisonment, or act done, or omitted to be done, under or by color of the President's order, or of any law of Con-gress; but the question is as to the constitutionality of the fifth section of the act, authorizing the defendant in any such estimates around the grame from the State Court for the Circ. action, to remove the same from the State Court to the Cir-cuit of the United States for the district where the suit is brought for trial, on complying with certain requirements specified in the section; that is, on entering his appearance. filing his petition stating the facts, offering good and sufficient surety, &c. The question presented by this appeal is not as to the con-

stitutional power of the President to order the arrest, imprisonment, &c., or as to the constitutional power of Congress to authorize the President to order the arrest, imprisoument, &c.; but the question presented by the appeal is as to the constitutional power of Congress to give the Cir-cuit Courts of the United States, primary or original, and (as to the State Courts) exclusive jurisdiction, of the trial of actions for such arrests, imprisonments, &c.

In determining the question as to the constitutionality of the sixth section of the act, we must assume, I think, that the trial of this action will involve the determination of the question as the constitutionality of the fourth section ; that Congress, in passing the act, considered that the trials of the actions to be removed to the Circuit Courts of the United States under it would involve the determination of the question as to the constitutionality of the fourth section, whether tried in the State or United States Courts; and that Congress intended by the fifth section to take from the State Courts, and give it to the Circuit Courts of the United States, the right and power to determine that question.

Had Congress the constitutional power to do this? That is the question.

If Congress had the power, then the order appealed from denying the defendant's motion to remove the action and all proceedings therein to the Circuit Court of the United States for the Southern District of New York should be reversed, and I think an order made directing such removal. If Congress had not the power, then the order appealed from should be affirmed.

If no steps had been taken for the removal of the action from this Court, and the action should be tried in this Court, and the question as the constitutionality of the fourth section

the Judiciary Act; 1 Stat. at Large, 85: Cohen vs. Virginia, 6 Wheaton, 264; Miller vs. Nicholls, 4 Wheaton, 311.) Cannot Congress give the Circuit Court of the United States original jurisdiction in any case to which this app.1late jurisdiction extends?

In Osborn es United States Bank, 9 Wheaton, cited by Judgo Leonard, Chief Justice Marshall said he could per-ceive no ground for saving that Congress could not.

In that case one of the questions was whether Congress could constitutionally confer on the Bank the right to sue and be sued "in every Circuit Court of the United States."

It was held that such a suit was a case arising under a law of the United States, consequently that it was within the judicial power of the United States, and Congress could confer upon the Circuit Court jurisdiction over it.

See, also, Curtiss's Com. on the Jurisdiction, &c., of the Courts of the United States, sections 12 and 13; the latter section, containing a quotation from another portion (p. 865) of the opinion of Chief Justice Marshall in Osborn vs. the Bank of the United States, apparently quite pertinent to the question in this case.

I concur, then, in the conclusion of Judge Leonard, that Congress had the power to direct the transfer to the Circuit Court of the United States.

Probably an order of this Court directing such transfer is net absolutely neccessary, but to make one would be in ac-cordance with usage in like cases; and besides, such an order would be the best evidence of the determination of this Court, that it no longer had jurisdiction of this action.

It appearing that the defendant has complied with the requirements of the act for such transfer, the order appealed from should be reversed, and an order made by this Court for the removal of the action and all proceedings therein to the Circuit Court of the United States.

#### Dissenting Opinion.

CLERKE, J .-- I see nothing whatever in the arguments of CLERKE, 3.—1 see forming whatever in the arguments of my brethren, or in those of other judges on the same sub-ject, to induce me to recede from the position which I have attempted to maintain at Special Term. They have all alike, in my very humble judgment, unaccountably over-looked the only point claiming consideration on this great constitutional subject.

According to the doctrine upheld by my brethren, we can scarcely conceive of any act committed by any officer of the General Government, under color of any authority of the General Government, under color of any authority derived from or under the President, which may not con-stitute a genuine, veritable case arising under the Consti-tution of the United States, and which, therefore, may not rightly come within the cognizance of their judicial power. It is only necessary to claim that it was committed under color of that authority, and was, therefore, justified by the Constitution, however monstrous and appalling the act may be, to make it, according to this doctrine, a case arising under that Constitution. For, of course, according to the terms of that claim, the claimant appeals through this remarkable statute, to the Constitution for his justification, and, however palpably frivolous such a claim may be-however palpably manifest may be the conviction that the Censtitution no more sanctions such an act than it sanctions the burning of the Capitol, the dispersion of Congress, and the burning of the Capitol, the dispersion of Congress, and the shooting, imprisonment or exile of the mea of when it is composed, yet it is claimed to present a question, and, therefore, a case arising under the Great Charter of Consti-sutional Liberty in America—the perpetrator of the out-rage making that a question which is unquestionably no question; and the judicial power of the State is oused of its legitimate jurisdiction. Thus, this extraordinary statute measures are not only that the obvergets. its legitimate jurisdiction. Thus, this extraordinary statute prescribes not only that the character, but the mere assertion of the wrong-deer shall determine jurisdiction, and that the subject-matter, which has been always held, ex-cept in cases affecting Embassadors, other diplomatic Ministers and Consuls, as alone the criterion of jurisdiction, shall be excluded from consideration. Surely, if this can be done by Congress, the Government of the United States of America, is not as all men have heretofore supposed, incontestably a Government of limited powers and duties, and is, if not one of unlimited powers and duties, neverthe-less, of very accommodating expansibility. This is a novel and strange theory of development in America

But, it is asserted as the appellate power of the Supreme Court of the United States extends in certain cases to State Court of the United States extends in certain cases to State tribunals, that this case would, after judgment, reach the Federal jurisdiction, and that, therefore, it may as well be transferred to the United States Circuit Court before judg-ment. Even if the Supreme Court of the United States would entertain such a case on appeal this is no controlling reason why it should, necessarily, be transferred to the United States Circuit for adjudication in the first instance. For, the only question to be determined by us on this motion, is whether Congress has the power to transfer cases of this description to the Circuit Court of the United of the act should be decided adversely to the identified by the fourt of Appeals of this State, the Supreme Court of the United States would have final and conclusive appellate jurisdiction of the question. (Const. U. S. Art. 3; see 25 of jurisdiction of the United States Supreme Court.

The act of Congress, passed in 1789, "to establish the ju-dicial courts of the United States," no doubt provides that a final judgment or decree in any suit in the highest court of law or equity of a State, where is drawn in question the validity of a statute of the United States, and the decision is against its validity, may be re-examined and revised or affirmed in the Supreme Court of the United States. But, affirmed in the Supreme Court of the United States. But, if it is too clear for controversy that the statute is an out-rage on the Constitution, if it is palpably usurpation, if it is plain to the most unlettered citizen, that the statute is an attempt to subvert all the securities which the founders of the Government have provided for the preservation of personal liberty, and to invest one man with unlimited dic-tatorial power, and, therefore, that the appeal was palpably frivalous L pressure the court would hear ne argument on frivolous, I presume the court would hear no argument on would hey, for instance, hearken to an appeal involv-ing the validity of an act of Congress giving the President

or any other member of the Government power, by a coup d'ata, to est inguish the legislative branch, as Cromwell did the Long Parliament, and substitute a Barebones Legisla-ture in itsplace? Surely not: if they, too, were not struck down, and were not (if said debasement can be imagined) by force, by fear, or by corrupt appliances or selfsh aspira-tions, robbed of independence. So that the consideration whether the act is not palpably void must present itself on appeal as it now presents itself to us on this motion; and, if it is palpably void, I repeat it would not be treated on appeal as worthy of being for a moment entertained.

I still consider the defence in this case just as destitute of color as the case which I have imagined. Whether, unof color as the case which I have imagined. Whether, un-der the pretext of authority from the President of the United States, any one citizen, at his mere will and pleasure, with-ont any intervention of the judicial tribunals, can incar-cerate another citizen not subject to military law in a loath-Bome dungeon, for many months, or for a day or an hour, cannot, under any circumstances in which the nation may be placed, be treated as a question constituting a case arising under the Constitution; and any statute which declares the contrary is palpably void. The order at Special Term should be affirmed with costs.

### MILITARY ARRESTS.

The following order has been issued by General Augur :

HEADQUARTERS DEPARTMENT OF WASHINGTON,

22D ARMY CORPS, June 20, 1864. GENERAL ORDER NO. 51.—First: Hereafter no citizen, commissioned officer, or enlisted man, will be arrested on the report of a detective employed by any officer subject to the jurisdiction of this department, except in extreme cases where there is no doubt of guilt, and immediate action is needed, until the report has first been forwarded for action at these headquarters.

Scowal: All officers serving in this department employing detectives will send with as little delay as possible a list of those employed to these headquarters, specifying the author-ity by whom employed; and they are notified that they will be held responsible for improper action or abuse of authority or the part of their employed. ity by whom energy be held responsible for improper action on the part of their employees. By command of Major General C. C. AUGUR: J. H. TAYLOR, Chief of Staff, A. A. G.

# "Confederate" Legislation.

[From the Richmond Sentinel, Feb. 17, 1864.]

SUSPENSION OF THE WRIT OF HABEAS CORPUS.

The following bill passed both Houses of Congress:

A Bill to suspend the privilege of the writ of habeas corpus

in certain cases. Whereas, the Constitution of the Confederate States of Whereas, the Constitution of the Confederate States of America provides, in article 1, section 9, paragraph 3, that "the privilege of the writ of habeas corpus shall not be suspending the privilege of said writ, as recognized in said article 1, is vested solely in the Congress, which is the ex-clusive judge of the necossity of such suspension, and the in your rigilant execution the investion of these States by the armies of the United States; and whereas, the President has asked for the sus-pension of the writ of habeas corpus, and informed Congres, are alone to be of the writ a measure proper for the public defonce against invasion and insurrection : Now, therefore, I. That during the present invasion of the Confederate States, the privilege of the writ of habeas corpus be and the

same is hereby suspended; but such suspension shall apply only to the cases of persons arrested or detained by order of the President, Secretary of War, or the general officer com-manding the Trans-Dississippi Military Department, by the authority and under the control of the President. It is hereby declared that the purpose of Congress in the passage of this act is to provide more effectually for the public safety by suspending the writ of habeas corpus in the following

asses, and no other: 1. Of treason, or treasonable efforts or combinations, to subvert the Government of the Confederate States.

2. Of conspiracies to overthrow the Government, or conspiracies to resist the lawful authority of the Confederate States.

3. Of combining to assist the enemy, or of communi-cating intelligence to the enemy, or giving him aid and comfort.

4. Of conspiracies, preparations and attempts to incite servile insurrection.

5. Of desertions or encouraging desertions, or harboring b. Of desertions of encouraging desertions, of hardoring deserters, and of attempts to avoid military service: *Pro-vided*, That in cases of palpable wrong and oppression by any subordinate officer, upon any party who does not le-gally owe military service, his superior officer shall grant prompt relief to the oppressed party, and the subordinate boll he diminered forme officer shall be dismissed from office.

6. Of spies and other emissaries of the enemy.

7. Of holding correspondence or intercourse with the enemy, without necessity, and without the permission of the Confederate States.

8. Of unlawful trading with the enemy and other offences against the laws of the Confederate States, enacted to

9. Of conspiracies, or attempts to liberate prisoners of war held by the Confederate States.

10. Of conspiracies, or attempts or preparations to aid the enemy.

11. Of persons aiding or inciting others to abandon the Confederate cause, or to resist the Confederate States, or to

adhere to the enemy. 12. Of unlawful burning, destroying or injuring, or at-tempting to burn, destroy or injure auy bridge or railroad, or telegraph line of communication, or other property with the intent of aiding the enemy.\*

13. Of treasonable designs to impair the military power of the Government by destroying or attempting to destroy the vessels or arms, or munitions of war, or arsenals, foun-deries, workshops, or other property of the Confederate States.

The remaining sections are unimportant. The act to continue in force 90 days after meeting of next Congress.

EXTENT OF POWER CLAIMED BY THE GOVERNMENT.

The instructions of the War Department with respect to proceedings under the law making a limited or quasi susproceedings while the taw making winneed of quase sus-pension of the halosas corpus, remove many grounds of clamor, and propose what may be considered as a very moderate execution of the law. Parties arrested, in the cases specified in the law, will not be denied a trial, but their cases will be investigated by commissioners, who will

\*Respecting this offence, this order was issued in 1861: WAR DEPARTMENT, RICHMOND, November 25, 1861.

SIR: Your report of the 20th instant is received, and I now proceed to give you the desired instructions in relation to the prisoners taken by you among the traitors of East Tennessee.

Tennessee. First. All such as can be identified in having been en-gaged in bridge burning are to be tried summarily by drum-head court-martial, and, if found guilty, executed on the spot by hanging. It would be well to leave their bodies hanging in the vicinity of the burnt bridges. Second. All such as have not been so engaged are to be treated as prisoners of war, and sent with an armed guard to Tasceloses. Albaem, there to be knot imprisoned of the

to Tuscaloosa, Alabama, there to be kept imprisoned at the depot selected by the Government for prisoners of war. Whenever you can discover that arms are concentrated

by these traitors, you will send out detachments, search for and seize the arms. In no case is one of the men known to have been up in arms against the Government to be re-leased on any pledge or oath of allegiance. The time for such measures is past. They are to be held as prisoners of war, and held in jail till the end of the war. Such as come in voluntarily, take the oath of allegiance, and surrender their arms, are alone to be treated with leniency.

Your vigilant execution of these orders is earnestly urged

by the Government. J. P. BENJAMIN, Secretary of War. Col. W. B. Woon, Knaxville, Tenn. P. S.-Judge Patterson, Col. Pickens, and other ring-leaders of the same class, must be sent at once to Tuscaloosa to tail as wisconce of two. 4

tigate the same. If, upon examination, a reasonable and ngue the same for detention does not appear, he will certify the fact to the general or other officer in command, who will immediately discharge the prisoner from arrest. But if a reasonable and probable cause does appear, the com-missioner will forthwith transmit to the War Department a copy of the evidence taken in the case, with his opinion thereon, for instructions, and meanwhile the prisoner will remain in custody.

The cases where persons not belonging to the military ser-vice shall apply to any court or officer in the Confederate States for a writ of *habeas corpus*, it will be the duty of the officer having the command or custody of such person forth-with to report the case, with all the relevant facts, to the War Department, for instructions as to the proper answer to be made to such writ.

#### Newspaper Exclusion and Suppression.

August 16, 1861-In the United States Circuit Court of New York the grand jury presented the Journal of Commerce, the Daily News, the Freeman's Journal, and the Brooklyn Eagle as aiders and abettors of treason, in terms following:

To the Circuit Court of the United States for the Southern District of New York: The Grand Inquest of the United States of America for the Southern District of New York beg leave to present the following facts to the Court and ask its advice thereon:

There are certain newspapers within this district which are in the frequent practice of encouraging the rebels now in arms against the Federal Government by expressing sym-pathy and agreement with them, the duty of acceding to their demands, and dissatisfaction with the employment of their demands, and dissatistation with the employment of force to overcome them. These papers are the New York daily and weekly Journal of Commerce, the daily and weekly News, the daily and weekly Day-Book, the Freeman's Jour-nal, all published in the city of New York, and the daily and weekly Eagle, published in the city of Brooklyn. The first named of these has just published a list of newspapers in the free States opposed to what it calls "the present unhold war"—a war in the defence of our country and its institutions, and our most sacred rights, and carried on solely for the restoration of the authority of the Government. The Grand Jury are aware that free Governments allow

The Grand Jury are aware that free Governments allow liberty of speech and of the press to their utmost limit, but there is nevertheless a limit. If a person in a fortress or an army were to preach to the soldiers submission to the enemy he would be treated as an offender. Would he be more culpable than the citizen who, in the midst of the most formidable conspiracy and rebellion, tells the con-spirators and rebels that they are right, encourages them to persevere in resistance, and condemns the effort of loyal citizens to overcome and punish them as an "unholy war?" If the utterance of such language in the streets or through the press is not a crime, then there is a great defect in our laws, or they were not made for such an emergency.

The conduct of these disloyal presses is of course condemned add abhorred by all loyal men; but the Grand Jury will be glad to learn from the Court that it is also subject to indictment and condign punishment. All which is respectfully presented. OHARLES GOULD, Foreman.

NEW YORK, August 16, 1861.

[Signed by all the Grand Jurors.]

### ORDER OF THE POSTMASTER GENERAL.

POST OFFICE DEPARTMENT, August 22, 1861. SIR: The Postmaster General directs that from and after your receipt of this letter none of the newspapers pub-lished in New York city which were lately presented by the grand jury as dangerous, from their disloyalty, shall be forwarded in the mails.

I am, respectfully, your obedient servant, T. B. TROTT, Chief Clerk. To the POSTMASTER, New York City.

### SEIZURES OF NEWSPAPERS.

be appointed for these duties in the different military de-partments. Information of all arcests under the law will be given by the department commander as soon as practicable after they are made, and the commissioner will proceed to inves-tivate the arce and the unanged in the second to a second the association of the different in consequence of a late violent article on the "na-tivate the many later and the second to a second to holy war."

> Other newspapers were similarly excluded from the mails, and in due time, the subject engaged the attention of Congress; resolutions of inquiry having been offered in the Senate, January 14, 1863, by Mr. CARLILE, and in the House, December 1, 1862, by Mr. VALLANDIGHAM. 1863, January 20—The Committee on Judici-ary of the House of Representatives made a

> report, which embodied the following letter of the Postmaster General:

# POST OFFICE DEPARTMENT, January 5, 1863.

SR: I have the honr to acknowledge the receipt of the communication signed by you in behalf of the Judiciary Committee, embracing a copy of the resolution of the House of Representatives, in the following words: *"Resolved*, That the Committee on the Judiciary be in-

structed to inquire and report to the House at an early day by what authority of Constitution and law, if any, the Postmaster General undertakes to decide what newspapers may and what shall not be transmitted through the mails of the United States.

On the first day of the last session of Congress, being the earliest opportunity after the action to which the resolution relates, I submitted to Congress a statement of my action, and of the general reasons and authority for the same, in

the following language: "Various newspapers, having more or less influence within the sphere of their circulation, were represented to be, and were, in fact, devoting their columns to the furtherance of the schemes of our national enemies. These efforts ance of the schemes of our national enemies. These efforts were persistently directed to the advancement of hostile interests, to thwart the efforts made to preserve the integ-rity of the Union, and to accomplish the results of open treason without incurring its judicial penaltics. To await the results of slow judicial prosecution was to allow crime to be consummated, with the expectation of subsequent punishment, instead of preventing its accomplishment by prompt and direct interference. "The freedom of the press is secured by a high constitu-tional sarction. But it is freedom ond not license that is

But it is freedom and not license that is tional sanction. It is to be used only for lawful purposes. guaranteed. guaranteed. It's to be used only for fawful purposes. It cannot aim blows at the existence of the Government, the Constitution, and the Union, and at the same time claim its protection. As well could the assassin strike his blow at human life, at the same time claiming that his victim should not commit a breach of the peace by a counter blow. While, therefore, this department neither enjoyed by a claimed the course to survey on the treasure burn but blow. While, therefore, and department relater enjoyed nor claimed the power to suppress such treasonable publi-cations, but left them free to publish what they pleased, it could not be called upon to give them circulation. It could not and would not interfere with the freedom secured by law, but it could and did obstruct the dissemination of that license which was without the pale of the Constitution and law. The mails established by the United States Government could not, upon any known principle of law or public right, be used for its destruction. As well could the com-mon carrier be legally required to transport a machine designed for the destruction of the vehicle conveying it, or an innkeeper be compelled to entertain a traveller whom

all limit be to be intending to commit a robust when "I find these views supported by the high authority of the late Justice Story, of the Supreme Court of the United States. He says, in commenting on that clause of the Constitution securing the freedom of the press:

"'That this amendment was intended to secure to every citizen an absolute right to speak, or write, or print what-soever he might please, without any responsibility, public or private, therefor, is a supposition too wild to be indulged in by any rational man. This would be to allow to every citizen the right to destroy at his pleasure the reputation, the peace, the property, and even the personal safety of evthe peace, the property, and even the personal safety of ev-ery other citizen. A man might, out of mere malice or revenge, accuse another of the most infamous crimes; might excite against him the indignation of all his fellow-citizens by the most atrocious calumnies; might disturb, nay, overturn all his domestic peace and embitter his par-ental affections; might inflict the most distressing punish-ment upon the weak, the timid, and the innocent; might prejudice all a man's civil and political and private rights; and might stir up sedition, rebellion, and treason, even against the Government itself, in the wantonness of his passions, or the corruption of his heart. Civil society could not go on under such circumstances. Men would then be obliged to resort to private vengeance to make up the defie-PHILADELPHIA, August 22—On the arrival of the New York train this morning Marshal Millward, and his officers, examined all the bundles of papers and seized every copy of the New York Daily News. The sale of this paper is totally suppressed in this city. Marshal Millward als seized all the bundles of the Daily News at the Express ciency of the law; and assassinations and savage cruelties would be perpetrated with all the frequency belonging to barbarous and cruel communities. It is plain, then, that the language of this amendment imports no more than that every man has a right to speak, write, and print his opinions upon any subject whatever, without any prior restraint, so always that he does not injure any other person in his rights, person, property, or reputation; and so always that he does not likereby disturb the public peace, or attempt to subsert the Government."

"Of the cases presented for my action, upon the principles above named, I have by order excluded from the mails tweive of these treasonable publications, of which several had been previously presented by the grand jury as incendiary and hostile to constitutional authority."

I am not aware that at any time, nor from any quarter, during that long session, any inquiry or complaint was made, or objection taken touching that action, or the considerations then presented in support of it. From this it was fairly inferred that Congress then unanimously recognized the action as not only in harmony with, but in direct aid of, the Constitution of the United States, then shaken by the assaults of its avowed enemies.

The immediate occasion of the orders excluding certain newspapers from the mails was a communication to this department of the action of a grand jury of the United States circuit court for the Southern District of New York. Their presentment was in the following words. [See ante for presentment.]

This authoritative exhibition of the character of these papers, as disseminators of treason and instigators of the inghest crime known to our laws, could not be disregarded, accompanied, as it was, by representations of the fundancerous effect upon the military operations of the country. Entertaining the highest possible regard for the liberty of the press, distinguished from its uncontrolled and criminal license, I would not, except in time of war, have adopted the arguments of my predecessors in office, in justification of the on-delivery of printed matter sent through the mails. The question has been repeatedly presented to my predecessors in time of peace in relation to printed matter styled "incendiary," or "abolition in its character," and in respect to the States now in insurrection. While justifying postmasters in their refusal to receive or forward mail matter described by the general terms of the postal laws as "mailable matter," an eminent Postmaster General of the administration of General Jackson, under date of August 22, 1835, addressed a letter to the postmaster at New York giving his views upon the question under discussion. The New York postmaster had assumed to decide that certain newspapers, placed in that post office for conveyance in the mails, were incendiary in their character, and calculated to promote insurrection. He refused to forward them. The Postmaster General, declining himself to decide upon the character of the publications in question, and refusing to make the orders thereon, justified his deputy postmaster in the decision made by him, and supported him by the following arguments, extracted from his letter of that date, to which ergth and the policy of excluding such alleged treasonable and insurrectionary publications from the mails.

### POST OFFICE DEPARTMENT, August 22, 1835.

\* "Postmasters may lawfully know, in all cases, the contents of newspapers, because the law expressly provides that they shall be so put up that they may be readily examined; and if they know those contents to be calculated and designed to produce, and, if delivered, will certainly produce, the commission of the most aggravated erimes upon the property and persons of their fellow citizens, it cannot be doubted that it is their duty to detain them, if not even to hand them over to the civil authorities. \*

if not even to hand that it is then duty to useful them, if not even to hand that it is then duty to useful them, \* " If it be justifiable to detain papers passing through the mail, for the purpose of preventing or punishing isolated crimes against individuals, how much more important is it that this responsibility should be assumed to prevent insurrections and save communities? If, in time of war, a postmaster should detect a letter of an enemy or spy passing through the mail, which, if it reached its destination, would expose his country to invasion and her armies to destruction, ought he not to arrest it? Yet, where is the legal power to do so?

legal power to do so? "As a measure of great public necessity, therefore, you and the other postmasters who have assumed the responsibility of stopping these inflammatory papers will, Have no doubt, stand justified in that step before your country and all mankind. " " " " Are the officers of the United States compelled by the Constitution and laws to become the instruments and accomplices of those who design to baffle and make nugatory the constitutional laws of the States; to fill them with sedition and murder and insurrection; to overthrow those institutions which are recognized and guarantied by the Constitution itselft <u>\*</u> In these con-

siderations there is reason to doubt whether the abolitionists have a right to make use of the mails of the United States to convey their publications into States where their circulation is forbidden by law, and it is by no means certain that mail-carriers and postmasters are secure from the penalties of that law, if they knowingly carry, distribute, or hand them out. \* \* As well may the counterfeiter and robber demand the use of the mails for consummating their crimes, and complain of a violation of their rights when it is denied.

"Upon these grounds a postmaster may well hesitate to be the agent of the abolitionists in sending their incendiary publications into States where their circulation is prohibited by law; and much more may postmasters residing in those States refuse to distribute them. \* \* I do not desire to be understood as affirming that the suggestions here thrown out ought, without the action of higher authority, to be considered as the settled construction of the law, or regarded by postmasters as the rule of their future action. It is only intended to say that in a sudden emergency, involving principles so grave and consequences so serious, the safest course for postmasters and the best for the country is that which you have adopted. \* \* \* You prevent your Government from being the unwilling agent and abettor of crimes against the States which strike at their very existence, and give time for the proper authorities to discuss the principles involved and digest a safe rule for the future guidance of the department. "While persisting in a course which philanthropy re-

"While persisting in a course which philanthropy recommends and patriotism approves, I doubt not that you and the other postmasters who have assumed the responsibility of stopping these inflammatory papers in their passage to the South will perceive the necessity of performing your duty in transmitting and delivering ordinary newspapers, magazines, and pamphlets, with perfect punchality. Occasion must not be given to charge the postmasters with carrying their precautions beyond the necessities of the case, or capriciously applying them to other cases in which there is no necessity; and it would be the duty, as well as the inclination, of the department to punish such assumption with unwonted severity. This suggestion I do not make because I have any apprehension that it is needed for your restraint, but because I wish this paper to bear upon its face a complete explanation of the views which I

take of my own duty in the existing emergency." The question was afterwards repeatedly presented in this department. In February, 1857, it was brought before Postmaster General Campbell, in connection with the exclusion of the Cincinnati Gazette from postal privileges in Mississippi. A certain postmaster at Yazoo had denied it the privilege of his post office. Mr. Campbell referred the question to the Attorney General of President Pierce's administration. Under date of March 2, 1857, the Attorney General, as the law officer of the Government, replied officially to the Postmaster General, justifying such action on the part of postmasters, and asserting, among others, the following arguments and conclusions:

#### "ATTORNEY GENERAL'S OFFICE, "March 2, 1857.

\* \* "With these premises we have the main question very much simplified. It is this: Has a citizen of one of the United States plenary indiputable right to employ the functions and the officers of the Union as the means of enabling him to produce insurrection in another of the United States? Can the officers of the Union lawfully lend its functions to the citizens of one of the States for the purpose of promoting insurrection in another State?

"Taking the last of these questions first, it is obvious to say that, inasmuch as it is the constitutional obligation of the United States to protect each of the States against 'domestic violence,' and to make provision to 'suppress insurrections,' it cannot be the right of the United States, or of any of its officers, and, of course, it cannot be their duty to promote, or be the instrument of promoting, insurrection in any part of the United States. "As to the first question, likewise, it seems obvious to say

any part of the United States. "As to the first question, likewise, it seems obvious to say, that, as insurrection in any one of the States is violation of law, not only so far as regards that State itself, but also as regards the United States, therefore no citizen of the Union can lawfully incite insurrection in any one of the States. \* \* \* It would be proported proported for the States.

can lawruny incide insurrection in any one of the States. \* \* \* It would be preposterous to suppose that any citizen of the United States has lawful right to do that which he is bound by law to prevent when a thempted by any and all others; and monstrous to pretend that a citizen of one of the States has a moral right to promote or commit insurrection or domestic violence, that is, robbery, burglary, arson, rape, and murder, by wholesale, in another of the States.

"These considerations, it seems to me, are decisive of the question of the true construction of the act of Congress. Of that it is impossible for me to doubt. Its enactment is, that 'if any postmaster shall unlawfully detain', he shall be subject to fine, imprisonment, and disqualification. Then, if the thing be of lawful delivery, it cannot be law-fully detained; while, on the other hand, it cannot be un-lawful to detain that which it is unlawful to deliver. Such is the plain language and the manifest import of the act of

B the plant initialized on the international state of the plant initialized of the congress. "I do not mean to be understood that the word 'unlaw-fully' of the act determines the case: on the contrary, my conclusion would be the same, though that word had not been here inserted. By employing it, indeed, the act ex-ercises of determines been here inserted. By employing it, indeed, the attest pressly admits that there may be lawful cause of detertion. But such lawful cause would not the less exist, although its existence were not thus expressly recognized. And, of all conceivable causes of detention, there can be none more operative than treasonableness of character, for in every private the arbits existing any environment of form society the public safety is the supremest of laws.

society the public safety is the supremest of laws. "Nay, if, instead of expressly admitting lawful causes of detention, the act had undertaken to exclude them—if, for instance, it had in terms required the postmasters to circu-late papers, which, in tendency and purpose, are of character to incite insurrection in any of the States—still my conclu-sion would be the same. I should say of such a provision of law it is a nullity, it is unconstitutional; not so by reason of conflict with any State law, but because incon-sistent with the Constitution of the United States. "The Constitution forbids insurrection: it imposes on

"The Constitution forbids insurrection; it imposes on Congress and the President the duty of suppressing insur-rection; this obligation descends through Congress and the President to all the subordinate functionaries of the Union, civil and military; and any provision of an act of Congress requiring a Federal functionary to be the agent or minister of insurrection in either of the States would violate palpably the positive letter, and defcat one of the primary objects, of the Constitution.

"These, my conclusions, apply only to newspapers, pamphlets, or other printed matter, the character of which is of public notoriety, or is necessarily brought to the knowledge of the postmaster by publicity of transmission through the mails unsealed, and as to the nature of which he cannot

plead ignorance. \* \* "It is intimated in one of the documents before \* "It is intimated in one of the documents before me that to permit a deputy postmaster to detain a news-paper because of its imputed unlawfulness would be to crect him into a censor of the press. These are but words of rhe-torical exaggeration. Public journals are a necessary part of our social life, just as much as the steamboat, the rail-way train, or the telegraph. There is not the least reason to apprehend that we shall suffer ourselves to be deprived of them by the intervesition of nulawful immediments to of them by the interposition of unlawful impediments to their circulation.

\* \* "We shall appreciate the true legal relation of the whole question if we consider a supposition which has more than once heretofore been actual fact, and may be such again. Suppose that some European Government---whether in the prosecution of war, or induced by hostility of purpose notyet become war, but tending towards it, or in the spirit of misdirected propagandism of its own particular social or ultitude in the super social states of the second states misdirected propagandism of its own particular social or political opinions—should undertake to produce revolution or insurrection in the United States. Would it, in that case, be the duty, would it be the right, of the Government or officers of the Union to aid the foreign Government in its inimical machinations? To this general inquiry, of course, there can be but one possible answer. It would be the mani-fest duty of every officer of the United States—nay, of every officer of each State—nay, of every citizen of the United States, to resist, and to do everything in his power to de-feat all such machinations; for every citizen of the United States is under engagement, express or implied, to uphold and maintain the Constitution. and maintain the Constitution.

"In the general contingency supposed it is quite immaterial whether foreign attempts to produce revolution con-sist of exhortations to insurrection by word of mouth—that is, the introduction of emissaries of sedition into the conn--or of exhortations to insurrection in the form of handbills, newspapers, or pamphlets. In whatever manner at-tempted, the thing itself would be an act of wrongful or basile attack on our sovereignty and on our national and private peace; defensible as an act of war on the part of an enemy Government, but otherwise against natural law, against public law, against municipal law; and therefore, on all these accounts, requiring to be manfully withstood and compterated by wary southbaserid and true minded and counteracted by every sound-hearted and true-minded citizen of the United States, and more especially by all offi-cers, civil and military, of the Federal Government, from the President down to the humblest village postmaster in the land.

"The general supposition includes printed, equally with oral, exhortations to insurrection. Take now, by itself, the case of printed matter of that description. Is it the legal duty of the Post Office Department knowingly to circulate such matter? Is it the legal duty of deputy postmasters? Or reducing the general supposition down to its narrowest expression in the limited exigencies of the present case, is a deputy postmaster required knowingly to circulate such matter under penalty of indictment, removal from office,

and disqualification? Is the inconvenience which the for-eign Government or its emissary may suffer, in not being able to effect the free circulation of such treasonable matter or the inconvenience which the disaffected person to whom it was addressed suffers, in his not being able to receive and to circulate further such treasonable matter—are these in-conveniences to outweigh the inconvenience to the whole country, as well as to individuals, of insurrection, and of civil or servile war? Is that the true construction of the act of Congress? I think no legal expositor could hesitate

act of Congress? I think no legal expositor could incluse to say, no. "Now in what does the general case supposed, with its all but self-evident conclusions, differ from the specific case under consideration? Simply, that any European Govern-ment possesses the sovereign right, as an act of war, to attack us with attempts to excite insurrection as well as with cannon-subject to be repelled by the sovereign power of the Union-but no citizen of the United States possesses legal right to promote rebellious acts in any part of the country, whether as against the authority of the United States or of the particular State in which he is, or of any other of the States.

to the first that no person in the United States, whether he be citizen, subject, or alien, has the legal right to promote re-bellion \* \* In the foregoing and a foregoing bellion. \* \* In the foregoing sorles of supposi-tions we have reasoned out a conclusion from the premises of the attempt of a foreign Government, by the use of our mails and post offices, to promote insurrection in the United States. \* \* Aud shall not the citizens of one of the States of the Union be held entitled to the same security The states of the Onion be here character to the state security from attempts to promote insurrection among them, on the part of their follow-citizens of other States? \* \* On the whole, then, it seems clear to me that a deputy postmaster, or other officer of the United States, is not re-quired by law to become knowingly the enforced agent or instrument of enemies of the public peace, to disseminate, in their behalf, within the limits of any one of the States of the Union printed matter, the design and tendency of

which are to promote insurrection in such State."\* Again, in 1859, Mr. Holt, then at the head of this depart-ment, in a letter dated the 5th of December of that year, addressed to a postmaster in Virginia, adhered to the precedents, and said:

"One of the most solemn constitutional obligations imposed on the Federal Government is that of protecting the States against 'insurrection' and 'domestic violence;' of course none of its instrumentalities can be lawfully employed in inciting, even in the remotest degree, to the very crime which involves in its train all others, and with the suppression of which it is especially charged."

These clations show that a course of precedents has ex-isted in this department for twenty-five years—known to Congress, not annulled or restrained by act of Congress—in accordance with which newspapers and other printed matter, decided by postal officers to be insurrectionary, or treasonable, or in any degree inciting to treason or insurrection, have been excluded from the mails and post offices of the United States solely by authority of the executive adminis-tration. This, under the rules settled by the Supreme Court of the United States, as applicable to executive construction of laws with whose execution the departments are specially

\* JEFFERSON DAVIS thus expressed himself on this opinion of Attorney General CUSHING a few months after its publication:

#### WASHINGTON, January 4, 1858.

GENTLEMEN: When I last addressed you in answer to your Deter communicating the views and feelings of the citizens of Yazoo City, in relation to the circulation of incendiary matter through the mails of the United States, I promised that you should hear from me further, and gave you assur-ance of such action by the last Administration as would be citizene to you satisfactory to you.

I have thus long delayed the promised communication in expectation of receiving the opinion of the Attorney Gen-eral upon the legal merits of the case, the question having been referred to him by the Postmaster General, the Hon. James Campbell.

The Attorney General, in the opinion enclosed, sustains the conclusion of the President and the Postmaster Gen-eral, and so satisfactorily disposes of the question at issue that I hope that we shall be saved from any further agitation of it.

tion of it. Concurring fully with you in your opinion of the powers of a State, the duty of its citizens, and the obligation of our community in such contingency as that presented by the case reported in your letter, I trust we shall also agree that the matter has been concluded in a manner worthy of the State-Rights Administration under which it arose. With great regard I are your friend and follow-citizen With great regard, I am your friend and fellow-citizen

JEFFERSON DAVIS

To Messrs. Robert Bowman, George B. Wilkinson, and A. M. Harlow, committee, Yazoo City.

charged, would establish my action as within the legal construction of the postal acts authorizing the transportation of printed matter in the United States mails. It would settle the right of this department and its various officers to resist all efforts to make them *particeps criminis* of treason and rebellion, by compelling them to circulate and distribute with their own hands the moral weapons which are to bring civil war to their firesides, with its horrible train of barbarities in the destruction of life and property.

The winn their own halles all horal weapons which are to bring civil war to their firesides, with its horrible train of barbarities in the destruction of life and property. Upon the like considerations I have, at different times, excluded from the mails obscene and scandalous printed matter on exhibition of its criminal immorality. If an unscaled printed publication were offered to the mails, instigating murder, arson, destruction of railroads, or other crimes, and advocating an organization for such purposes, I should, upon the same principles, without hesitation, exclude it from the mails as unlawful matter, in the absence of a contravening act of Congress.

I do not wish to be understood, however, as indorsing, but rather as distinctly dissenting from, some of the arguments and conclusions, and from the extent to which preceding Administrations have gone, as indicated by some of the foregoing citations. The precedents and arguments go far beyond any action which I have taken, or would be willing to take, under the like circumstances. Ist. I reject that portion of the precedents which allows

Ist. I reject that portion of the precedents which allows twenty-eight thousand postmasters of the country to judge, each for himself, what newspapers are lawful and what un-lawful; what may go in the mails and what shall be excluded. I have refused to allow postmasters to sit in final judgment upon all the interests involved, subject as they are to conflicting local prejudices. The Postmaster General, who is more directly responsible to Congress, and more accessible to their inquiries, should alone exercise such authority, in whatever degree it exists, and should not devolve it on subordinates. Whatever control can be law always be exercised or ordered by the chief, under whose direction the law expressly subordinates the postmaster. This is a self-evident proposition. It has, however, been sustained by the official opinion of the Attorney General of the United States, dated March 2, 1857.
2d. I dissent from the extent to which the doctrine has been carried by late administrations, that in time of peace,

2d. I dissent from the extent to which the doctrine has been carried by late administrations, that in time of peace, and in the absence of all hostile or criminal organizations, operating against Constitution or law, either a Postmaster General, or any postmaster, can at will exclude from the mails newspapers and other printed matter which contain discussions obnoxions to some special interest, but not aimed against Government, law, or the public safety. It is too dangerous a discretion to be exercised or desired by any executive officer attached to the constitutional freedom of the press. Such has been, in some cases, the action of this department in late years, and I take this occasion to break the too great uniformity of its decisions in this respect. Even in time of war, the power so long conceded should bo used with great care and delicacy. I say in time of war, because the executive department has powers then which do not attach to it in time of peace. The Constitution provides that no person shall \* be deprived of life, liberty, or property be taken for public use without just compensation." Yet, in time of war, the life, liberty, and property be persons in the United States, being also insurrectionary enemies of the United States, are necessarily taken without any process except that of powder and the bayonet. And no man denies the right as an incident of war. Yet, in peace, it could not be done. These acts are as thoroughly constitutional in war as they are unconstitutional in peace. In harmony with this principle, I would give far greater in a period of peace than would be justifiable in a time of war. This could use the arguing the same incident of the term is the normal in other as they are unconstitution ant mater in a period of peace than would be justifiable in a time of war.

This reply to the inquiry transmitted to me by the committee embraces the following conclusions: *First.* That the exercise of the authority inquired of rests

First. That the exercise of the authority inquired of rests upon the Constitution of the United States, and the definition of mailable matter given in the postal law, as construed by past administrations of this department, enforced by the official opinion of a late Attorney General of the United States, and known to and recognized by former Congresses of the United States.

Second. That a power and a duty to prevent hostile printed matter from reaching the enemy, and to prevent such matter from instigating others to co-operate with the enemy, by the aid of the United States mails, exist in time of war, and in the presence of treasonable and armed enemies of the United States, which do not exist in time of peace, and in the absence of criminal organizations. Third. That the present Postmaster General has restricted

There. That the present Postmaster General has restricted the exercise of the power during this war far within the scope claimed and allowed by former Administrations in periods of national peace.

I have the honor to be, very respectfully, your obedient servant, M. BLAIR, Postmaster General.

The committee review the case, and conclude as follows:

Your committee are not unmindful of the fact that too great caution cannot be exercised in arriving at a conclusion as to what is and what is not lawful mailable matter; or, in other words, what papers, publications, or messages are treasonable in their character, or for other reasons unlawful, and should, therefore, be excluded from the mails. In the case now before the committee the grand jury of one of the federal courts in the State of New York concur-

In the case now before the committee the grand jury of one of the federal courts in the State of New York concurred in opinion with the head of the Post Office Department in the construction of the character of the publications, and the purposes of the publishers, it being, too, in a time when extreme vigilance was demanded in the executive department of the Government to preserve the integrity of the Union. And the object being to secure that noble and patriotic object, your committee believe the act of the Postmaster General was not only within the scope of his powers, but induced solely by considerations of the public good.

Mr. GEO. H. PENDLETON, of Ohio, (of the Judiciary Committee,) in his speech, March 3, 1863, in the House, quoted these two additional paragraphs from Amos KENDALL's opinion of 1835:

"After mature consideration of the subject, and seeking the best advice within my reach, I am confirmed in the opinion that the Postmaster General has no legal authority, by any order or regulation of the Department, to exclude from the mails any species of magazines, newspapers, or pamphlets. Such a power vested in the head of this Department would be fearfully dangerous, and has therefore been withheld. Any order or letter of mine directing or officially sanctioning the step you have taken would, therefore, be utterly powerless and void, and would not in the slightest degree relieve you from its responsibility. \* \* "The Postmaster General has no legal power to prescribe any wules for the covernment of nostmasters in such

"The Postmaster General has no legal power to prescribe any rules for the government of postmasters in such cases; nor has he ever attempted to do so. They act in each case on their own responsibility; and if they improperly detain or use papers sent to their offices for transmission or delivery it is at their peril, and on their heads falls the punishment. If in time of war a postmaster should detect the letter of an enemy or a spy passing through the mail, which, if it reached its destination, would expose his country to invasion and her armies to destruction, ough the not to arrest it? Yet where is his legal power to do so?"

#### He added:

In 1836, Mr. Calhoun, as chairman of a special committee of the Senate, reported a bill making it a penal offence for any postmatter to receive into the mails for transmission to any person within a State, or to deliver out of the mails to any such person, any publication the circulation of which was forbidden by that State. Subsequently the first clause of the bill was stricken out, and the latter, relating to the delivery of such matter, was retained. It gave rise to much discussion, and elicited an extremely able debate from the most eminent members of that then very able body. Mr. Calhoun, the zealous advocate of the bill, contended that a bill of this nature was the only one which Congress had the power to pass; that Congress could not discriminate in reference to character what publications shall or shall not be transmitted through the mail, without abridging the liberty of the press, and subjecting it to the publications as they thought dangerous, and that Congress had the power, and ought to exercise it, of co-operating with the States in repressing the circulation of publications

The circulation of anti-slavery documents, tending to excite servile insurrection, had become a great evil. It had awakened fears of trouble among the slaves, and had therefore exasperated the people. Most of the slaveholding States had passed laws forbiding their circulation under severe penalties. They were still carried through the mails, and it began to be questioned whether the postmasters were not relieved from the penalties of the State law because they were acting under the sanction of Federal law. Great anxiety existed to relieve the apprehensions of the southern people. The President, General Jackson, recommended the subject most earnestly to Congress. He did not pretend that there existed any power of relief in any of the Executive Departments. Senators, almost without exception, expressed a determination to go as far as they could to apply aremedy. But the bill was most strennously opposed. It was said to curtait the freedom of the press. \* \* \* \*

The bill was lost by a majority of seven; Messrs. Benton, Clay, Crittenden, Southard, Wall, Leigh, Goldsborogh, among others from the slaveholding States; and Messrs. Webster, Niles, Ewing, and Davis, with others from the non-slaveholding States voting against it. And yet it is in reference to this discussion and this action that the Postmaster General in his letter to the committee says "that Congress by its inaction seemed to concur in the right and the policy of excluding such alleged treasonable and insur-rectionary publications from the mails." On the contrary, Congress expressly refused to sanction the idea that it had the power; and certainly no other department of the Government has.

Generals commanding departments frequently prohibit the circulation of certain newspapers within the limits of their commands. Major General Wallace, May 18, 1864, suppressed the Baltimore Evening Transcript. Major General Rosecrans, May 26, 1864, prohibited the circulation of the Metropolitan Record in the department of Missouri. The circulation of the Cincinnati Enquirer has recently been prohibited, by the General commanding, in Kentucky.

#### A REMINISCENCE.

#### VIRGINIA AND THE TRIBUNE.

Post Office, Lynchburg, Virginia, December 2, 1859.

MR. HORACE GREELEY : SIR: I hereby inform you that I shall not, in future, de-liver from this office the copies of the *Tribune* which come here, because I believe them to be of that incendiary character which are forbidden circulation alike by the laws of the land, and a proper regard for the safety of society. You will, therefore, discontinue them.

Respectfully, R. H. GLASS, Postmaster.

#### Reply.

NEW YORK, December 9, 1859. MR. POSTMASTER OF LYNCHBURG, VIRGINIA :

SIR: I take leave to assure you that I shall do nothing of the sort. The subscribers to the *Tribune* in Lynchburg have paid for their papers; we have taken their money, and shall fairly and fully earn it, according to contract. If they shall fairly and fully earn it, according to contract. If they direct us to send their papers to some other post office, we shall obey the request; otherwise, we shall send them as originally ordered. If you or your masters choose to steal and destroy them, that is your affair—atall events, not ours; and if there is no law in Virginia to punish the larceny, so much the worse for her and our plundered subscribers. If the Federal Administration, whereaf you are the fool after the Federal Administration, whereof you are the tool, after monopolizing the business of mail-carrying, sees fit to be-come the accomplice and patron of mail-robbery, I suppose the outrage must be borne until more honest and less servile rulers can be put into high places at Washington, or till the people can recover their natural right to carry each other's letters and printed matter, asking no odds of the Government. Go ahead in your own base way. I shall stand steadfast for human liberty and the protection of all natural rights.

Yours, stiffly,

#### HORACE GREELEY.

THE RECENT SUPPRESSION IN NEW YORK.

1864, May 19-By order of the Secretary of War, the offices of the Journal of Commerce and the World—in which papers had appeared a forged proclamation of the President for 400,-000 troops-were seized by the military authorities and held for several days. This led to these proceedings :

Gov. Seymour's Letter to the District Attorney.

STATE OF NEW YORK, EXECUTIVE DEPARTMENT, ALBANY, May 23, 1864.

To A. OAKEY HALL, Esq., District Attorney of the County of New York: SR: I am advised that on the 19th inst., the office of The Journal of Commerce and that of The New York World were entered by armed men, the property of the owners seized, and the premises held by force for several days. It is charged that these acts of violence were done without due legal process, and without the sanction of State or national laws.

If this be true the offenders must be punished.

In the month of July last, when New York was a scene of violence, I gave warning that "the laws of the State must be enforced, its peace and order maintained, and the property of its citizens protected at every hazard." laws were enforced at a fearful cost of blood and life. The

that occasion or against any class of men. It is one of an enduring character, to be assorted at all times and against all condition of citizens, without favor or distinction. Un-less all are made to bow to the law, it will be respected by none.

Unless all are made secure in their rights of person and property, none can be protected. If the owners of the above-named journals have violated State or national laws, above-named journals have violated State or national laws, they must be proceeded against and punished by those laws. Any action against them outside of legal procedures is criminal. At this time of civil war and disorder, the ma-jesty of the law must be upheld or society will sink into anarchy. Our soldiers in the field will battle in vain for constitutional liberty if persons or property, or opinions, are trampled upon at home. We must not give up home freedom, and thus disgrace the American character while our citizens in the army are pouring out their blood to maintain the national honor. They must not find when they come back that their personal and fireside rights have been despoiled. been despoiled.

In addition to the general obligation to enforce the laws In addition to the general obligation to entorce the laws of the land, there are local reasons why they must be up-held in the city of New York. If they are not, its commerce and greatness will be broken down. If this great center of wealth, business, and enterprise is thrown into disorder and bankruptcy, the National Government will be paralyzed. What makes New York the heart of our country? Why are its pulsations felt at the extremities of our land? Not through its position alone, but because of the world-wide belief that property is safe within its limits from waste by made and from snoilation hy Government.

benet that properly is sale within its limits from waste by mobs and from spoliation by Government. The laborers in the workshop, the mine, and in the field, on this continent and in every other part of the globe, send to its merchants, for sale or exchange, the products of their toil. These merchants are made the trustces of the wealth of millions living in every land, because it is believed that in their hands property is safe under the shield of laws ad-This great confidence has grown up in the course of many years by virtue of a painstaking, honest performance of duty by the business men of your city. In this they have been aided by the enforcement of laws based upon the solemnly-recorded pledges that "the right of the people to be secure in their persons, houses, papers, and effects against unreasonable searches and seizures shall not be vio-lated, and that no one shall be deprived of liberty or prop-

lated, and that no one shall be deprived of liberty or prop-erty without due process of law." For more than eighty years have we as a people been building up this universal faith in the sanctity of our jurisprudence. It is this which carries our commerce upon every ocean and brings back to our merchants the wealth of every clime. It is now charged that, in utter disregard of the sensitiveness of that faith, at a moment when the national credit is undergoing a fearful trial, the organs of commerce are seized and held, in violation of constitutional nedges that this act was done in a public mart of your pledges, that this act was done in a public mart of your great city, and was thus forced upon the notice of the com-mercial agents of the world, and they were shown in an offensive way that property is seized by military force and arbitrary orders.

These things are more hurtful to the national honor and strength than the loss of battles. The world will confound such acts with the principles of our Government, and the natural results of the spirit of our institutions. Our State and local anthorities must repel this ruinous inference. If the merchants of New York are not willing to have their harbor scaled up and their commerce paralyzed, they must unite in this demand for the security of persons and proper-erty. If this is not done, the world will withdraw from their keeping its treasures and its commerce.

History has taught all that official violation of law in times of civil war and disorder goes before acts of spoilation and other measures which destroy the safeguards of commerce

I call upon you to look into the facts connected with tho I call upon you to look into the facts connected with tho seizure of *The Journal of Commerce* and of *The New York World*. If these acts were illegal, the offenders must be punished. In making your inquiries and in prosecuting the parties implicated, you will call upon the Sheriff of the county and the heads of the Police Department for any needed force or assistance. The failure to give this by any official under my control will be deemed a sufficient cause for his remeval. for his removal c his removal. Very respectfully yours, &c., HORATIO SEYMOUR.

#### CHARGE OF JUDGE RUSSELL,

Of the New York Court of General Sessions, June 13, to the grand jury, composed as follows :

Cyrus Mason, Foreman, John E. Anderson, Nathaniel W. Carter, Martin L. Delafield, Mathew Hettrick, John J. Hay-The declaration I then made was not intended merely for er, David C. Newell, James H. Pinckney, Wm. Palen, Wm.

Remson, J. Austin Stevens, jr., Amos H. Trowbridge, Samuel A. Beekman, Seabury Brewster, Jacob D. T. Hersey, Benedict Lewis, jr., Willard Phelps, Wm. T. Skeff, W. M. Thurman, John P. Worstell, John Townsend, John D. Welsh, Chris. Zabriskie, jr.

The grand jury having taken the usual oath, Judge Russell delivered a charge in which he thus alluded to the order for the arrest of the proprietors, and for the suppression of their iournals :

The first part of the order was never fully executed. The latter part was, and the forcible possession maintained for several days. The author of the frand, it is said, has been several days. The author of the Trand, it is said, has been discovered, and the newspapers in question have been exconerated from all suspicion of guilt or blame. If this be so, this is an instance of innocent men being summarily interfered with, or trespassed upon, in the sancity of their persons and property. As such, it is a violation of both the Federal and State Constitutions, and it is your duty to examine into it. This is not a self-imposed or self-assumed duty by this court. The facts were communicated to the Executive of this State, and he addressed to the district attorney of this county the communication I now read to you. you

[The Judge then read Governor Seymour's letter.]

Acting yon the duty this Court owes to the laws of this State, which is repeated in the official document I have just read to you, I beg to submit the matter to your calmest and most careful consideration. The Court is convinced that you most careful consideration. The court is convinced that you will deal with it in such a manner as becomes the dutiful and loyal citizens of a dutiful and loyal State. Anything like political bias should be discarded. The question is simply thus: Have the laws of the State, in reference to the protection of person and property, been violated, and if so, who are the parties who have been concerned in if Ne mutic what their statics must answer No matter what their station may be they must answer for the wrong, if there be one. If the President of the United States, or other officer who assumed to issue the on-der, had no such power or authority, those who obeyed and

enforced it are clearly responsible. For the purposes of this occasion, the Court instructs you For the purposes of this occasion, the court instructs you that such an order as has been referred to would not, under the circumstances stated, be any protection to those con-cerned in its execution. This will raise the question at issue between the State and General Government in a legal way. Any attempt to interfere with freedom of speech or liberty of the press has been regarded and watched with the great-est jealousy by the constituents of our Federal and State Governments. These invaluable privileges are protected in both the Federal and State Constitutions. Neither Con-gress nor our State Legislature can make a law abridging either right. In the year 1798, the famous "Sedition law" was passed by Congress, giving the Government extraor-dinary power in reference to publications calculated to weaken its authority. So unpaltable was this law that it was finally repealed. Two of the State Legislatures ex-pressly declared against its constitutionality. At the time it was passed, the Government being in a state of compara-tive infancy, it ought probably to have been more favor-ably regarded; but it involved rights too dear to be trenched upon or surrendered. In reference to the alleged author of the spurious proclamation, you will receive evidence of the spurious proclamation, you will receive evidence of the spurious proclamation and if you are satisfied of it, you will present him for such an offence as, under the advice of Any attempt to interfere with freedom of speech or liberty will present him for such an offence as, under the advice of the district attorney, (to whom you are entitled to appeal for advice,) may be proper. At common law, the "spread-ing false news to make discord between the king and no-bility, or concerning any great man of the realm," was an offence against the public peace, punishable with fine and imprisonment.

imprisonment. It ways be that the elements of the common law will be It may be that the elements of the common law will be Invoked by the district attorney in reference to this offender. In reference to the parties engaged in taking and maintain-ing forcible possession of the newspaper establishments, the court instructs you that if there were three or more of them, they would be liable as for riot, which has been de-fined to be " where three or more actually do an unlawful act of violence, either with or without a common cause or quarrel, or even do a lawful act, as removing a nuisance, in a violent or tumultuous manner."

#### RESPONSE OF THE GEAND JURY.

Resolved, That the grand inquest respectfully represent to the honorable court that in their judgment, it is inexpe-dient to examine into the subject referred to in the commu-nication of the Executive of the State and the charge of the court, namely : the action of the General Government as to court, namely : the action of the city. Certain newspapers in this city. CYRUS MASON, Foreman.

JOHN AUSTIN STEVENS, Jr., Secretary.

#### THE LAWS TO BE ENFORCED.

Letter from Governor Seymour to District Attorney Hall, of New York.

EXECUTIVE CHAMBER, ALBANY, June 25.

A. OAKEY HALL, Esq., District Attorney of the City and County of New York:

SIR: In the matter of the seizure of the offices of The World and Journal of Commerce, the grand jury, in disre-gard of their oaths, "to diligently inquire into and true presentment make of all such matters and things as should be given them in charge," have refused to make such in-quiries, and declare that "it is inexpedient to examine into quiries, and declare that "It is inexpedient to examine into the subject referred to in the charge of the court" with re-spect to such solaures. It becomes my duty, under the ex-press requirements of the constitution, "to take care that the laws of the State are faithfully executed." If the grand jury, in pursuance of the demands of the law and the obli-gations of their oaths, had inquired into the matter given them in charge by the court and the public prosecutor, their decision whetever it might have been would have been decision, whatever it might have been, would have been entitled to respect. As they have refused to do their duty, the subject of the seizure of these journals should at once be brought before some proper magistrate. If you wish any assistance in the prosecution of these investigations, it will be given to you.

As it is a matter of public interest that violations of the laws of the State be punished, the views or wishes of the parties immediately affected must not be suffered to in fluence the action of public officers. If through fear or other motives they are unwilling to aid you in getting at facts, it will be your duty to compel their attendance as witnesses in behalf of the people.

Respectfully yours,

#### HORATIO SEYMOUR

The newspapers give this account of further proceedings:

#### THE ARREST OF GENERAL DIX.

The arrest of General Dix and several other officers on Friday, July 1, was made upon warrants issued by City Judge Russell. Several persons appeared before the city Judge Russell. Several persons appeared before the city judge, in answer to subpenas allowed by him, at the in-stance of District Attorney Hall, and had testified to facts relating to the seizure of *The Workl* and *Journal of Com-merce* newspapers. The letter of Governor Seymour to the district attorney, condemning the grand jury for its return in the case of those newspapers, and saying that "the sub-iest about the hermouth before some prepresented the ject should be brought before some saying that 'the sub-ject should be brought before some proper magistrate,'' is said to have induced the district attorney to procure the af-fidavits to be made before Russell. The district attorney first made an affidavit in the form of a complaint, dated 28th first made an alidavit in the form of a complaint, dated 28th June, in which he declared that he had been informed and believed that "Hon. A. Lincoln" directed "John A. Dix" to do several acts against *The World* and Journal of Com-merce, and the editors of those journals, enumerated in the complaint of the district attorney, and charging that the said Dix "feloniously ordered one William Barstow" (Cap-Solution of the district automoty and charging bias with Cap-said Dix " feloniously ordered one William Barstow" (Cap-tain Barstow) to arrest the editors of the newspapers named, and "mischievously ordered one William Hays" (Acting Assistant Provost Marshal General Hays) to procure the closing up of the newspaper offices; that the arrest of Mr. Hallock was procured, and that gentleman kept for the space of about three hours; that "the said Hays instructed Major Powers, who caused one Fundy" (Captain Fundy) and some commissioned officers and privates, whom the district attorney names, to "go armed and equipped" to take possession of the Journal of Commerce office; and that the said Hays caused similar acts to be done to The World, through Lieutenant Gabriel Tuthil and several other sol-diers. The district attorney then charges that John A. Dix and William Barstow are guilty of kidnapping, and the others, with John A. Dix, of inciting to a riot and foreibly detaining property; and the district a'torney prays that action be taken to sustain the dignity of the State. Judge Russell then issued subpenas, directed to Messrs.

action be taken to sustain the dignity of the State. Judge Russell then issued subpenas, directed to Messrs. William H. Hallock, of the Journal of Commerce; David G. Croly, of The World; William W. Jacobus, John S. Betts, auctioneer, Daniel R. Kirwan, and Washington Hills, Jr., clerk in The World office, who appeared before the judge and made their several affidavits, the district attorney ex-mining the witherees amining the witnesses.

#### ARREST OF GENERAL DIX BY THE SHERIFF.

In accordance with the latter of Governor Seymour, di-recting the matter of the suppressed newspapers to be brought before a magistrate, Mr. A. Oakey Hall commenced taking evidence and submitting testimouy before Judge Russell on Tuesday. After examining the witnesses, Judge Russell came to the conclusion that it was a proper case for him to issue his warrant. Accordingly warrants were placed

in the bands of the sheriff, who arrested Major General Dix, Major Barstow, Captain Fundy, Major Powers, and other officers on guard at the offices of *The World* and *Journal* of *Commerce*.

of Commerce. The military gentlemen very courteously submitted to the arrest, and their counsel, E. Delafield Smith, appeared before Judge Russell. Mr. Smith asked for time to examine into the papers and consult with his associate, ex.Judge Pierrepont, as to the future course to be pursued by them. The matter was then adjourned, the defendants in the meantime being released on their own verbal recognizances.

### First Session, Thirty-Eighth Congress. IN SENATE.

1864, May 26-Mr. PowerL offered this resolution, which went over:

Resolved, That the conduct of the executive authority of this Government, in recently closing the offices and suppressing the publication of The World and Journal of Commerce, newspapers in the city of New York, under circumstances which have been placed before the public, was anact unwarranted in itself, dangerous to the cause of theUnion, in violation of the Constitution, and subversive ofthe principles of civil liberty, and as such is hereby censured by the Senate.

#### IN HOUSE.

May 23-Mr. GRINNELL asked consent to offer this resolution, but it was objected to:

Resolved, That the President be requested to communicate to this House whether, by any order of the Government, or by any officer thereof, *The World and Journal of Commerce*, newspapers in the city of New York, were suspended from being published; and if so, that said order be communicated to this House, and the proceedings in the execution of that order.

May 23-Mr. Pruvn asked consent, on behalf of a portion of the New York delegation, to offer this resolution:

Resolved, That the conduct of the executive authority of the Government in recently closing the offices and suppressing the publication of The World and Journal of Com-

merce, newspapers in the city of New York, under circumstances which have been placed before the public, was an act unwarranted in itself, dangerous to the cause of the Union, in violation of the Constitution, and subversive of the principles of civil liberty, and as such is hereby censured by this House.

Several members objected. At a later hour he moved a suspension of the rules for the purpose of offering it, but this motion was rejected—yeas 54, nays 79, as follows :

Cuc-yeas 54, hays 79, as follows: YEAS-MOSESS. James C. Allen, Augustus C. Baldwin, Bliss, Brooks, James S. Brown, Chanler, Coffroth, Cox, Dawson, Denison, Filen, Edgerton, Eldridge, Funck, Grider, Harding, Charles M. Harris, Herrick, Holman, Hudchins, Philip Johnson, William Johnson, Kalbfleisch, Kernan, King, Knapp, Law, Lazear, Mallory, Marcy, McAlbister, McDowell, William H. Miller, Morrison, Nelson, Noble, John O'Nell, Pendleton, Prugn, Radford, Samuel J. Randall, Robinson, James S. Rollins, Ross, Scott, John B. Steele, William G. Sleele, Strouse, Yoorhees, Wadsworth, Ward, Wheeler, Joseph W. White, Fernando Wood-54.

Fernando rood-or.
NAYS-Messrs. Alley, Ames, Arnold, John D. Baldwin, Baxter, Beaman, Jacob B. Blair, Blow, Boutwell, Boyd, Broomall, William G. Brown, Ambrose W. Clark, Freeman Clarke, Cobb, Cole, Creswell, Henry Winter Davis, Thomas T. Davis, Dawes, Deming, Dixon, Donnelly, Driggs, Eckley, Eliot, Farnsworth, Garfield, Gooch, Grinnell, Higby, Hooper, Hotchkiss, Asahel W. Hubbard, Ingersoll, Jenckes, Julian, Kelley, Francis W. Kellogg, Loan, Longyear, Marvin, Mc-Bride, McClurg, Samuel F. Miller, Moornead, Morrill, Amos Myers, Leonard Myers, Charles O'Neill, Orth, Patterson, Perham, Pike, Pomeroy, Price, William H. Randall, John H. Rice, Edward H. Rollins, Schenck, Scofield, Shannon, Upson, Elihu B. Washburne, William B. Washburn, Webster, Whaley, Williams, Wilder, Wilson, Windom, Woodbridge-79.

Same day, Mr. ARNOLD offered this resolution, which was adopted :

Resolved, That the Committee on the Judiciary be instructed to inquire and report what, if any, additional legislation may be necessary to punish the forgery and publication of official documents, and what legislation is necessary to punish those who through the press or otherwise give information, aid, or comfort to the rebels.

# CONFISCATION AND EMANCIPATION.

# CONFISCATION.

### First Session, Thirty-Seventh Congress.

1861, August 16-A bill was approved, of which these are the first and fourth sections:

That if, during the present or any future insurrection against the Government of the United States, after the President of the United States shall have declared, by proclamation, that the laws of the United States are op-posed, and the execution thereof obstructed, by combina-tions too powerful to be suppressed by the ordinary course of judicial proceedings, or by the power vested in the mar-bale bulker any nersons on berown bits are other agent of judicial proceedings, or by the power vested in the mar-shals by law, any person or persons, his, her, or their agent, attorney, or employee, shall purchase or acquire, sell or give, any property of whatsoever kind or description, with intent to use or employ the same, or suffer the same to be used or employed, in aiding, abetting, or promoting such insurrection or resistance to the laws, or any persons en-gaged therein; or if any person or persons, being the owner or owners of any such property, shall knowingly use or employ, or consent to the use or employment of the same as aforesaid, all such property is hereby declared to be lawful subject of prize and capture wherever found; be lawful subject of prize and capture wherever found; and it shall be the duty of the President of the United States to cause the same to be seized, confiscated, and condemned.

SEC. 4. That whenever hereafter, during the present insurrection against the Government of the United States, any person claimed to be held to labor or service under the law of any State shall be required or permitted by the person to whom such labor or service is claimed to be due, or by the lawful agent of such person, to take up arms against the United States; or shall be required or permitted by the person to whom such labor or service is claimed to be due, or his lawful agent, to work or to be employed in or upon any fort, navy-yard, dock, armory, ship, entrenchment, or in any military or naval service whatsoever, against the Government and lawful authority of the United States, then, and in every such case, the person to whom such la-bor or service is claimed to be due shall forfeit his claim to such labor, any law of the State or of the United States to the contrary notwithstanding. And whenever thereafter the person claiming such labor or service shall seek to en-force his claim, it shall be a full and sufficient answer to such claim that the person whose service or labor is claimed had been employed in hostile service against the Government of the United States, contrary to the provisions of this act.

This bill, as reported from the Judiciary Committee of the Sevate, did not contain the fourth section, and while it was pending in the Senate Mr. TRUMBULL moved to add this as a new section .

That whenever any person claiming to be entitled to the Service or labor of any other person under the laws of any State, shall employ such person in aiding or promoting any insurrection, or in resisting the laws of the United States, or shall permit him to be so employed, he shall forfeit all right to such service or labor, and the person whose labor or service is thus claimed shall be thenceforth discharged therefrom, any law to the contrary notwithstanding.

Which was agreed to-yeas 33, nays 6, as follows:

YEAS-Messrs. Anthony, Bingham, Browning, Chandler, Clark, Collamer, Cowan, Dixon, Doolittle, Fessenden, Foot, Foster, Grimes, Hale, Harlan, Harris, Howe, Johnson of Ten-

nessee, King, Lane of Kansas, McDougall, Morrill, Nesmith, Pomeroy, Sherman, Simmons, Sumner, Ten Eyck, Trum-bull, Wade, Wilkinson, Wilmon-33. NATS-Messrs, Breckinridge, Johnson of Missouri, Kenne-

dy, Pearce, Polk, Powell-6.

The bill then passed without a division.

# IN HOUSE.

August 2-The House Committee on the Judiciary reported a substitute for the bill, which provides that whenever hereafter, during the existence of the present insurrection against the Government of the United States, any person held to labor or service under the laws of any State shall be required or permitted, by the person to whom such labor or service is due, or his legal agent, to take up arms against the United States, or to work, or be employed in or about any fort, navy-yard, armory, dock-yard, ship, or in any military or naval service, against the Government of the United States, or as the servant of any person engaged in active hostilities against the United States, then the person to whom such labor is due shall forfeit all claim to such service or labor, any law of any State, or of the United States, to the contrary notwithstanding; and, in case of a claim for such labor, such facts shall be a full and sufficient answer.

Which was rejected without a division; when, after debate, the bill was recommitted to the committee.

August 3-The committee reported the Senate bill with a substitute for section four, adopted above in the Senate, being the fourth section of the act as approved.

A motion to table the bill was lost-yeas 47, nays 66; and the amendment was agreed to, and the bill passed-yeas 60, nays 48, as follows:

lows: YEAS-Messrs. Aldrich, Alley, Arnold, Ashley, Babbitt, Baxter, Beaman, Bingham, Francis P. Blair, Samuel S. Blair, Blake, Buffinton, Chamberlain, Clark, Colfax, Fred-erick A. Conkling, Covode, Duell, Edwards, Eliot, Fenton, Fessenden, Franchot, Frank, Granger, Gurley, Hanchett, Harrison, Hutchins, Julian, Kelley, Francis W. Kellogg, William Kellogg, Lansing, Loomis, Lovejoy, McKean, Mitchell, Justin S. Morrill, Olin, Potter, Alexander H. Rice, Edward H. Rollins, Sedgwick, *Shefield*, Shellabarger, Sher-man, Sloan, Spaulding, Stevens, Benjamin F. Thomas, Train, Van Horn, Verree, Wallace, Charles W. Walton, E. P. Walton, Wheeler, Albert S. White, Windom-60. NATS-Messrs. Allen, Ancona, Joseph Baily, George H. Broume, Burnett, Calvert, Cox, Cravens, Crisjeld, Critten-den, Diven, Dunlap, Dunn, English, Fouke, Grider, Haight, Hale, Harding, Holman, Horton, Jackson, Johnson, Law, 195

195

May, McClernand, McPherson, Mallory, Menzies, Morris, Noble, Norton, Odell, Pendleton, Porter, Reid, Robinson, James S. Rollins, Shiel, Smith, John B. Steele, Stratton, Francis Thomas, Vallandigham, Voorhees, Wadsworth, Webster, Wickliffe-48.

August 5—The Senate concurred in the amendment of the House—yeas 24, nays 11, as follows:

YEAS-Messrs. Anthony, Bingham, Browning, Clark, Collamer, Dixon, Doolittle, Fessenden, Foot, Foster, Grimes, Hale, Harrie, King, Lane of Indiana, Lane of Kansas, Mo-Dougall, Sherman, Simmons, Sumner, Ten Eyck, Trumbull, Wade, Wilson-24.

NAVS-Messrs. Breckinridge, Bright, Carlile, Cowan, Johnson of Missouri, Latham, Pearce, Polk, Powell, Rice, Saulsbury-11.

#### Second Session, Thirty-Seventh Congress.

1862, July 17—A bill was approved to "suppress insurrection, to punish treason and rebellion, to seize and confiscate the property of rebels, and for other purposes," which contains these provisions:

That every person who shall hereafter commit the crime of treason against the United States, and shall be adjudged guilty, shall suffer death, or be imprisoned not less than five years and fined not less than \$10,000, (the fine to be levied on all property excluding slaves,) and all his slaves, if any, shall be declared and made free.

That if any person shall hereafter incite, set on foot, assist, or engage in any rebellion or insurrection against the authority of the United States, or the laws thereof, or shall give aid or comfort thereto, or shall engage in, or give aid and comfort to, any such existing rebellion or insurrection, and be convicted thereof, such person shall be punished by imprisonment for a period not exceeding ten years, or by a fine not exceeding ten thousand dollars, and by the liberation of all his slaves, if any he have; or by both of said punishments, at the discretion of the court.

That every person guilty of either of the offences described in this act shall be forever incapable and disqualified to hold any office under the United States.

That this act shall not be construed in any way to affect or alter the prosecution, conviction, or punishment of any person or persons guilty of treason against the United States before the passage of this act, unless such person is convicted under this act.

That, to insure the speedy termination of the present rebellion, it shall be the duty of the President of the United States to cause the seizure of all the estate and property, money, stocks, credits, and effects of the persons hereinafter named in this section, and to apply and use the same and the proceeds thereof for the support of the army of the United States, that is to say:

First. Of any person hereafter acting as an officer of the army or navy of the rebels in arms against the Government of the United States.

Secondly. Of any person hereafter acting as President, Vice President, member of Congress, judge of any court, cabinet officer, foreign minister, commissioner or consul of the socalled Confederate States of America.

Thirdly. Of any person acting as Governor

of a State, member of a convention or legislature, or judge of any court of any of the socalled Confederate States of America.

Fourthly. Of any person who, having held an office of honor, trust, or profit in the United States, shall hereafter hold an office in the socalled Confederate States of America.

Fifthly. Of any person hereafter holding any office or agency under the Government of the so-called Confederate States of America, or under any of the several States of the said Confederacy, or the laws thereof, whether such office or agency be national, State, or municipal in its name or character: *Provided*, That the persons, thirdly, fourthly, and fifthly above described, shall have accepted their appointment or election since the date of the pretended ordinance of secession of the State, or shall have taken an oath of allegiance to, or to support the Constitution of the so-called Confederate States.

Sizthly. Of any person who, owning property in any loyal State or Territory of the United States, or in the District of Columbia, shall hereafter assist and give aid and comfort to such rebellion; and all sales, transfers, or conveyances of any such property shall be null and void; and it shall be a sufficient bar to any suit brought by such person for the possession or the use of such property, or any of it, to allege and prove that he is one of the persons described in this section.

That if any other persons in armed rebellion, or abetting it, shall not, within sixty days after proclamation to be made, return to their allegiance, all their estate and property shall be liable to seizure, and it shall be the duty of the President to seize and use them as aforesaid.

To secure condemnation and sale of such property, proceedings in rem shall be instituted in the name of the United States in any district or territorial court thereof within which the property may be found, &c.

That all slaves of persons who shall hereafter be engaged in rebellion against the Government of the United States, or who shall in any way give aid or comfort thereto, escaping from such persons and taking refuge within the lines of the army; and all slaves captured from such persons or deserted by them and coming under the control of the Government of the United States; and all slaves of such persons found on [or] being within any place occupied by rebel forces and afterwards occupied by the forces of the United States, shall be deemed captives of war, and shall be forever free of their servitude and not again held as slaves.

That no slave escaping into any State, Territory, or the District of Columbia, from any other State, shall be delivered up, or in any way impeded or hindered of his liberty, except for crime or some offence against the laws, unless the person claiming said fugitive shall first make oath that the person to whom the labor or service of such fugitive is alleged to be due is his lawful owner, and has not borne arms against the United States in the present rebellion, nor in any way given aid and comfort thereto; and no person engaged in the military or naval service of the United States shall. under any pretence whatever, assume to decide | on the validity of the claim of any person to the service or labor of any other person, or surrender up any such person to the claimant, on pain of being dismissed from the service.

That the President of the United States is authorized to employ as many persons of African descent as he may deem necessary and proper for the suppression of this rebellion, and for this purpose he may organize and use them in such manner as he may judge best for the public welfare.

That the President of the United States is hereby authorized to make provision for the transportation, colonization, and settlement, in some tropical country beyond the limits of the United States, of such persons of the African race made free by the provisions of this act as may be willing to emigrate, having first obtained the consent of the Government of said country to their protection and settlement within the same, with all the rights and privileges of freemen.

That the President is hereby authorized, at any time hereafter, by proclamation, to extend to persons who may have participated in the existing rebellion in any State or part thereof, pardon and amnesty, with such exceptions and at such time and on such conditions as he may deem expedient for the public welfare.

Same day-This joint resolution was approved :

That the provisions of the third clause of the fifth section of "An act to suppress insurrection, to punish treason and rebellion, to seize and conflicate the property of rebels, and for other purposes," shall be so construed as not to apply to for other purposes," shall be so construct as not to apply to any act or acts done prior to the passage thereof; nor to in-clude any member of a State Legislature, or judge of any State court, who has not in accepting or entering upon his office, taken an oath to support the Constitution of the so-called Confederate States of America; nor shall any punish-ment or proceedings under said act be so construct as to work a forfeiture of the real estate of the offender beyond big notwer big. his natural life.

#### VOTE ON THE BILL.

July 11-The bill, being the report of a committee of conference, passed the House-yeas 82, nays 42, as follows:

82, nays 42, as follows: YEAS-Messrs. Aldrich, Alley, Arnold, Ashley, Babbitt, Baxter, Beaman, Bingham, Jacob B. Blair, Samuel S. Blair, Blake, Buffinton, Campbell, Casey, Clark, Colfax, Frederick A. Conkling, Roscoe Conkling, Covode, Cutler, Davis, Dawes, Duell, Dunn, Edwards, Eliot, Ely, Fenton, Fessenden, Fisher, Frank, Gooch, Goodwin, Gurley, Hale, Hooper, Hutchins, Julian, Kelley, Francis W. Kellogg, William Kellogg, Lan-bing, Loomis, Lovejoy, Low, McKnight, McPherson, May-nard, Mitchell, Moorhead, Anson P. Morrill, Justin S. Mor-rill, Nixon, Patton, Timothy G. Phelps, Pite, Porter, Potter, Alexander H. Rice, John H. Rice, Riddle, Edward H. Rol-lins, Sargent, Sedgwick, Shanks, Shellabarger, Sherman, Stevens, Stratton, Trimble, Trowbridge, Van Horn, Verree, Wall, Wallace, Walton, Washburne, Wheeler, Albert S. White, Wilson, Windom, Worcester-82. Navs-Messre, William Allen, William J. Allen, Ancona,

WINTE, Wilson, Windom, Worcester—82. NATS-MESSTS. Wilkiam Allen, William J. Allen, Ancona, Baily, Biddle, George H. Broune, Clements, Cobb, Coz, Cris-field, Crittenden, Dunlap, Fouke, Granger, Grider, Haight, Hall, Harding, Holman, Kerrigan, Knapp, Law, Lazear, Lehman, Mallory, Menzies, Morris, Nugen, Odell, Pendleton, James S. Rollins, Segar, Shiel, John B. Steele, William G. Skeele, Stiles, Benjamin F. Thomas, Francis Thomas, Ward, Webster, Wickliffe, Wood-42.

July 12-The bill passed the Senate-yeas 28, nays 13, as follows:

senden, Fost, Foster, Grimes, Hale, Harlan, Harris, Howard, Howe, King, Lane of Indiana, Lano of Kansas, Morrill, Pomeroy, Sherman, Simmons, Sumner, Ten Eyck, Trum-bull, Wade, Wilkinson, Wilmot, Wilson of Massachusetts, Wright-28. YEAS-Messrs. Anthony, Chandler, Clark, Doolittle, FesNAVE-Messrs. Bayard, Browning, Carlile, Cowan, Davis, Henderson, Kennedy, McDougall, Powell, Saulsbury, Stark, Willey, Wilson of Missouri-13.

### THE JOINT RESOLUTION.

July 15-Mr. MAYNARD, on a suspension of the rules, (yeas 68, nays 33,) introduced the joint resolution, which is the same as the above, omitting the last clause, and which passed the House without a division.

July 16-In SENATE,

Mr. CLARK moved to add the last clause.

Mr. POWELL moved to strike out the word "real"; which was rejected—yeas 6, nays 31, as follows :

YEAS-Messrs. Davis, Henderson, Powell, Saulsbury, Stark, Wilson of Missouri-6.

Stark, Wilson of Missouri-6. NAYS-Messrs. Anthony, Browning, Chandler, Clark, Cowan, Doolittle, Fessenden, Foot, Foster, Grimes, Hale, Harlan, Harris, Howard, Howe, King, Lane of Indiana, Lane of Kansas, McDougall, Morrill, Pomeroy, Sherman, Simmons, Sumner, Ten Eyck, Trumbull, Wade, Wilkinson, Wilmot, Wilson of Massachusetts, Wright-31.

The amendment of Mr. CLARK was agreed to—yeas 25, nays 15, as follows:

EAS — Messrs. Analysis, as forwing, Chandler, Clark, Collamer, Cowan, Doolittle, Fessenden, Foot, Foster, Hale, Harris, Henderson, Howard, Howe, Lane of Kansas, Morrill, Pomeroy, Sherman, Simmons, Sunner, Ten Eyck, Willey, Wilson of Massachusetts, Wright—25. NAYS—Messrs. Carlie, Davis, Grimes, Harlan, Kennedy, King, Lane of Indiana, Powell, Szulsbury, Stark, Trumbull, Wade, Wilkinson, Wilmot, Wilson of Missouri—15.

July 17-The PRESIDENT sent this message to Congress:

Fellow-Citizens of the Senate and

House of Representatives:

Considering the bill for "An act to suppress. insurrection, to punish treason and rebellion, to seize and confiscate the property of rebels, and for other purposes," and the joint resolution explanatory of said act, as being substantially one, I have approved and signed both.

Before I was informed of the resolution, I had prepared the draft of a message, stating objections to the bill becoming a law, a copy of which draft is herewith submitted.

#### ABRAHAM LINCOLN.

July 12, 1862.

#### [Copy.]

Fellow-Citizens of the House of Representatives: I herewith return to your honorable body, in which it originated, the bill for an act entitled "An act to suppress originated, the bill for an act entitled "An act to suppress treason and rebellion, to seize and confiscate the property of rebels, and for other purposes," together with my objec-tions to its becoming a law. There is much in the bill to which I perceive no objec-tion. It is wholly prospective; and it touches neither person or property of any loyal citizen, in which particular it is just and proper

it is just and proper.

The first and second sections provide for the conviction and punishment of persons who shall be guilty of treason, and persons who shall "incite, set on foot, assist, or engage and punishment of persons who shall be guilty of treason, and persons who shall "incite, set on foci, assist, or engage in any rebellion or insurrection against the anthority of the United States, or the laws thereof, or shall give aid or comfort thereto, or shall engage in or give aid and comfort to any such existing rebellion or insurrection." By fair construction, persons within those sections are not punished without regular trials in duly constituted courts under the forms and all the substantial provisions of law and the Constitution applicable to their several cases. To this I perceive no objection ; especially as such persons would be within the general pardoning power, and also the special provision for pardon and amnesty contained in this act. It is also provided that the slaves of persons convicted under these sections shall be free. I think there is an un-fortunate form of expression, rather than a substantial ob-jection, in this. It is startling to say that Congress can free-a slave within a State, and yet if it were said the ownership of the slave had first been transferred to the nation, and Congress had then liberated him, the difficulty would at

Congress had then liberated him, the difficulty would at once vanish. And this is the real case. The traitor against the General Government forfeits his slave at least as justly

as he does any other property; and he forfeits both to the Government against which he offends. The Government, so far as there can be ownership, thus owns the forfeited slaves, and the question for Congress in regard to them is, "shall they be made free or sold to new masters?" I perceive no they be made free or sold to new masters?" I perceive no objection to Congress deciding in advance that they shall be free. To the high honor of Kentucky, as I an informed, she is the owner of some slaves by *escheat*, and has sold none, but liberated all. I hope the same is true of some other States. Indeed, I do not believe it will be physically possible for the General Government to roturn persons so circum-stanced to actual slaver. The liberat there would be hybride for the General Government to return persons so the char-stanced to actual slavery. I believe there would be physical resistance to it, which could neither be turned aside by ar-gument nor driven away by force. In this view I have no objection to this feature of the bill. Another matter involved in these two sections and running through other parts of the act will be noticed hereafter.

I perceive no objections to the third and fourth sections.

So far as I wish to notice the fifth and sixth sections, they so far as I wish to notice the intra and sixin sections, they may be considered together. That the enforcement of these sections would do no injustice to the persons embraced within them is clear. That those who make a causeless war should be compelled to pay the cost of it is too obviously just to be called in question. To give governmental protec-tion to the property of persons who have abandoned it and gone on a crusade to overthrow that same states about about if considered in the mere light of justice. The severest justice may not always be the best policy. The prin-ciple of seizing and appropriating the property of the person embraced within these sections is certainly not very objectionable, but a justly discriminating application of it would be very difficult, and, to a great extent impossible. be very difficult, and, to a great extent imposition would it not be wise to place a power of remission some-where, so that these persons may know they have something unitary and something to rain by desisting? I And to lose by persisting and something to gain by desisting? I am not sure whether such power of remission is or is not within section thirteen. Without any special act of Congress, I think our military commanders, when, in military phrase, "they are within the enemy's country," should, in an orderly manner, seize and use whatever of real or per-sonal property may be necessary or convenient for their commands; at the same time preserving, in some way, the evidence of what they do.

What I have said in regard to slaves while commenting on the first and second sections is appplicable to the minth, with the difference that no provision is made in the whole act for determining whether a particular individual slave does or does not fall within the classes defined in that section. He is to be free upon certain conditions; but whether

those conditions do r do not pertain to him, no mode of ascertaining is provided. This could be easily supplied. To the tenth section I make no objection. The oath therein required seems to be proper, and the remainder of the section is substantially identical with a law already existing.

The eleventh section simply assumes to confer discretionary power upon the Executive. Without the law, I have no hesitation to go as far in the direction indicated as I may at any time deem expedient. And I am ready to say now, I think it is proper for our military commanders to employ, as laborers, as many persons of African descent as can be used to advantage.

The twelfth and thirtcenth sections are something better than objectionable; and the fourteenth is entirely proper, if all other parts of the act shall stand.

That to which I chiefly object pervades most part of the act, but more distinctly appears in the first, second, seventh, and eighth sections. It is the sum of those provisions which results in the divesting of title forever.

For the causes of treason and ingredients of treason, not amounting to the full crime, it declares for frequents of treason, not amounting to the full crime, it declares forfeiture extending beyond the lives of the guilty parties; whereas the Consti-tution of the United States declares that "no attainder of treason shall work corruption of blood, or forfeiture except during the life of the person attainted." True, there is to be nonpresent attainted in this scene still. be no formal attainder in this case; still, I think the greater punishment cannot be constitutionally inflicted, in a different form, for the same offence.

With great respect I am constrained to say I think this feature of the act is unconstitutional. It would not be difficult to modify it. I may remark that the provision of the Constitution, put

I may remark that the provision of the Constitution, put in language borrowed from Great Britain, applies only in this country, as I understand, to real or landed estate. Again, this act, in rem, forfeits property for the ingredi-ents of treason without a conviction of the supposed crim-inal, or a personal hearing given him in any proceeding. That we may not touch property lying within our reach because we cannot give personal notice to an owner who is absent endeavoring to destroy the Government is cer-tainly satisfactory. Still, the owner may not be thus en-gaged; and I think a reasonable time should be provided for such parties to appear and have personal hearings. Simfor such parties to appear and have personal hearings. Sim-

ilar provisions are not uncommon in connection with proceedings in rem.

For the reasons stated, I return the bill to the House in which it originated.

# Other Proceedings. IN HOUSE.

1862, May 14-The Select Committee consisting of Messrs. Eliot of Mass., Sedgwick of New York, Noell of Missouri, Hutchins of Ohio, Mallory of Kentucky, Beaman of Michigan, and Cobb of New Jersey, reported two bills-one "to confiscate the property of rcbels for the pay-ment of the expenses of the present rcbellion, and for other purposes;" and the other "to free from servitude the slaves of rebels engaged in or abetting the existing rebellion against the Government of the United States."

The former bill was first considered. Tt provided "that all the estate and property, money, stocks, credits, and effects of the persons hereafter named in this section, are hereby forfeited to the Government of the United States, and are declared lawful subjects of seizure and of prize and capture wherever found, for the indemnity of the United States against the expenses of suppressing the present rebellion."

The classes are substantially as stated in the act, and the provisions of the bill 'are, in large part, those of the act.

Pending its consideration,

May 26-Mr. Roscoe Conkling moved the addition of the proviso to the fifth clause of the fifth section, which was agreed to-yeas 100, nays 50-as follows :

100, nays 50—as follows : YEAS—Messrs. Aldrich, Alley, Arnold, Ashley, Babbitt, Ba-ker, Baxter, Beaman, Francis F. Blair, Jacob B. Blair, Samuel S. Blair, Blake, William G. Brown, Buffinton, Campbell, Chamberlain, Clark, Clements, Colfax, Frederick A. Conkling, Roscoe Conkling, Cutler, Davis, Dawes, Delano, Duell, Dunn, Edgerton, Edwards, Eliot, Ely, Fenton, Fessenden, Fishri, Frank, Gooch, Goodwin, Granger, Gurley, Hanchett, Har th-son, Hickman, Hooper, Horton, Hutchins, Julian, Kelley, Francis, Ev, Kellogg, William Kellogg, Killinger, Lansiug, Loomis, Lovejoy, McKnight, McPherson, Michell, Moor-head, Anson P. Morrill, Justin S. Morrill, Nixon, Noell, Olin, Patton, Timothy G. Phelps, Pike, Pomeroy, Porter, Potter, Price, Alexander H. Ricc, John H. Rice, Riddle, Edward H. Spaulding, Stevens, Stratton, Benjamin F. Thomas, Francis, Thomas, Train, Trimble, Trowbridge, Van Horn, Van Val-kenburgh, Vernee, Wall, Wallace, E. P. Walton, Washburne, Wheeler, Whaley, Albert S. White, Wilson, Windom, Wor-cester-100. Wheeler, W cester-100.

cester-100. NATS-Messrs. Allen, Ancona, Baily, Biddle, George H. Browne, Calvert, Cobb, Corning, Cox. Cravens, Crisfield, Crittenden, Dunlap, English, Grider, Haight, Hall, Harding, Holman, Johnson, Kerrigan, Knapp, Law, Lazear, Leary, Leionan, Mallory, Maynard, Mensies, Noble, Norton, Nugen, Odell, Fendleton, Perry, Join S. Fleips, Richardson, Rob-inson, James S. Rollins, Shiel, John B. Steele, William G. Steele, Vallandigham, Foorhees, Wadsworth, Ward, Webster, Wickliffe, Wood, Woodruff-50.

Mr. MAYNARD submitted a substitute, defining the offence of treason and affixing the penalty of death by hanging; "and if the offender shall theretofore have held any office under the Government of the United States, of honor or profit, whether military, naval, or civil, he shall be adjudged in a fine at least equal to the proven value of his entire estate." It also defines misprision of treason, and affixes the ' It also punishment of confinement at hard labor for not less than five years and a fine equal to his or her entire estate. It makes other offences. described, high misdemeanors, punishable, on conviction, with fine not less than \$100, and

It proimprisonment not less than fifty days. vides for trying persons charged with the described offences upon the presentment or the indictment of a grand jury in any of the circuit or district courts within the judicial district in which the crimes shall be alleged to have been committed, or authorizes a writ of capias to the marshals of the districts, to be furnished, on application to the President, with sufficient military force to execute the writ. An oath is prescribed for all officials, jurors, &c., and the President is authorized to grant an amnesty to all offenders within the act, except those who, having held office under the United States, have at any time engaged in rebellion and held office.

This was rejected—yeas 9, (Messrs. J. B. Blair, Clements, Diven, Fisher, Harrison, Horton, Lazear, Maynard, Mitchell.) nays 140.

Mr MORRILL, of Vermont, offered a substistute whose principal difference was the omission of this the second section of the reported bill:

That if any person within any State or Territory of the United States, other than those named as the aforesaid, after the passage of this act, being engaged in armed rehelion against the Government of the United States, or aiding or abeting such rehelion, shall not, within sixty days after public warning and proclamation duly given and made by the President of the United States, cease to aid, countenance, and abet such rehelion, and return to his allegiance to the United States, all the estate and property, moneys, stocks, and credits of such person are hereby forficitel thereafterwards to the Government of the United States, and the same are declared lawful subjects of solaure and of prize and capture wherever found; and the President of the United States shall cause the same to be seized, to the end that they may be confiscated and condemned, as hereinafter provided, to the use of the United States; and all sales, transfers, or conveyances, of any such property after the expiration of the said sixty days from the date of such warning and proclamation shall be null and void; and it shall be a sufficient har to any suit brought by such person for the possession or the use of such property, or any of it, to allege and prove that he is one of the persons described in this section.

#### And this addition :

That every person guilty of the acts described in the first section shall, in addition to the forfeitures, be incapable of voting for President or Vice President, and not be an elector of the United States, and disqualified from holding the office of President or Vice President, or holding any office by appointment from the President.

The substitute was rejected—yeas 25, nays 122. The YEAS were:

Messrs. Buxter, Jacob B. Blair, Roscoe Conkling, Diven, Dunn, Fisher, Goodwin, *Haight*, William Kellogg, Killinger, McKnight, McPherson, Maynard, Mitchell, Moorhead, Justin S. Morrill, Nixon, Timothy G. Phelps, Porter, *Sheffield*, Stratton, Trimble, E. P. Walton, Worcester—25.

The bill reported by the committee was then passed—yeas 82, nays 68, as follows:

passed—yeas 82, nays 68, as follows: YEAS—Messrs. Aldrich, Alley, Arnoid, Ashley, Babbitt, Baker, Baxter, Beauma, Francis P. Blair, Sanuel S. Blair, Blake, Katter, Beauma, Francis P. Blair, Sanuel S. Blair, Blake, William G. Brown, Buffinton, Campbell, Chamberlain, Clark, Colfax, Frederick A. Conkling, Roscoe Conkling, Cutler, Davis, Duell, Dunn, Edgerton, Edwards, Eliot, Ely, Fenton, Fessenden, Frank, Gooch, Goodwin, Gurley, Hickman, Hooper, Hutchins, Julian, Kelley, Francis W. Keilogg, William Kellogg, Killinger, Lansing, Loomis, Lovejoy, McKnight, McPherson, Mitchell, Moorhead, Anson P. Morrill, Justin S. Morrill, Noell, Olin, Patton, Pike, Pomeroy, Perter, Potter, John H. Rice, Riddle, Edward H. Rollins, Sargent, Sedgwick, Shanks, Sloan, Spaulding, Stevens, Stratton, Trimble, Trowbridge, Van Horn, Van Valkenburgh, Verree, Wall, Wallace, E. P. Walton, Washburne, Weceter, Waley, Albert S. White, Wilson, Windom, Weceter, Sa.

Wrotester, Wratey, Alocte & Hand, Baily, Biddle, Jacob B. NATS-Messrs. Allen, Ancona, Baily, Biddle, Jacob B. Blair, Gorge H. Broune, Calvert, Clements, Cobb, Corning, Coz, Cravens, Crisfield, Crittenden, Dawes, Delano, Diven, Dunlag, English, Fisher, Granger, Grider, Haight, Hall,

Harding, Harrison. Holman, Horton, Johnson, Kerrigan, Knapp, Law, Lazear, Leary, Lelman, Mallory, Maynard, Manzies, Nixon, Noble, Norton, Nugan, Oddl, Pendieton, Perry, John S. Pholps, Price, Alexander H. Rice, Richardson, Robinson, James S. Rollins, Segar, Shejfield, Shiel, Smith, John B. Steele, William G. Steele, Benjamin F. Thomas, Francis Thomas, Train, Fallandigham, Joorhees, Wadsworth, Ward, Webster, Wickliffe, Wood, Woodruff-68.

#### AN EMANCIPATION BILL.

May 14—The Select Committee reported this bill :

That if any person within any State or Territory of the United States shall, after the passage of this act, wilfully engage in armed rebellion against the Government of the United States, or shall wilfully aid or abet such rebellion, or adhero to those engaged in such rebellion, giving them aid and comfort, every such person shall thereby forfeit all claim to the service or labor of any persons, commonly known as slaves; and all such slaves are hereby declared free and forever discharged from such servitude, anything in the lws of the United States or of any State to the contrary notwithstanding. And whenever thereafter any person claiming the labor or service of any such slave shall seek to enforce his claim, it shall be a sufficient defence thereto that the claimant was engaged in the said rebellion, or aided or abetted the same, contrary to the provisions of this act.

That whenever any person claiming to be entitled to the service or labor of any other person shall seek to enforce such claim, he shall, in the first instance, and before any order shall be made for the surrender of the person whose service or labor is claimed, establish not only his claim to such service or labor, but also that such claimant has not in any way added, assisted, or countenanced the existing rebellion against the Government of the United States.

May 26, pending its consideration,

Mr. FRANCIS P. BLAIR Jr., offered an amendment, to add to the bill sections providing for the appointment by the President of commissioners for each slave States, to make lists of the names of slaves held by persons described in the first section, to be reported to the district court of the United States, of the proper district. If the slaves are not claimed by any one, they will be declared free by the court; if they are claimed, the claimant must prove that he has not been engaged in, nor aided and abetted the rebellion, nor given aid and comfort to those engaged; or if engaged under compulsion this must be proved. In failure of proof, the slaves shall be declared free, and be given a certificate of freedom, and to be employed under the direction of the commissioners, in cultivating lands belonging to the United States, or other useful labor, and may be employed by the commanding officers of the army, with the consent of the commissioners, as may be agreed upon. Commissioners authorized to bind them as apprentices to loyal proprietors of lands or mechanics, for not over five years where the slaves are over twenty-one, and if under, not to extend beyond their twenty-fifth year. The President was authorized to purchase lands in Mexico, Central or South America, or islands in the Gulf of Mexico, to be removed with their own consent, they to receive not exceeding fifty acres to an individual, or eighty to the head of a family, and be guaranteed the civil and political rights secured to all other citizens in said countries-the proceeds of confiscation and the earnings of those liberated persons to be applied to the payment of those expenses.

This was rejected—yeas 52, nays 95, as follows:

YEAS-Messrs. Aldrich, Arnold, Baker, Baxter, Francis P. Bleir, Campbell, Clements, Coifax, Roscoe Conkling, Dawes, Delano, Diven, Durin, Edgerton, Edwards, Fenton, Fisher, Frank, Gooch, Goodwin, Gurley, Kelley, Killinger, McKnight, McPherson, Maynard, Moorhead, Anson P. Morrill, Justin S. Morrill, Nixon, Olin, Patton, Timothy G. Phelps, Pike, Pomeroy, Porter, Alexander H. Rice, Riddle, Elward H. Rollins, James S. Rollins, Shanks, Stratton, Train. Trim-ble, Verree, Wallace, E. P. Walton, Washburne, Wheeler, Whaley, Albert S. White, Worcsster-52. Nars-Messrs. Allen, Alley, Ancona, Ashley, Babbitt, Eaily, Beaman, Biddle, Jacob B. Flair, Samuel S. Blair, Blake, George H. Browne, William G. Brown, Buffatton, Calmert Clamberlain, Clark, Cobb, Frederick A. Conkling.

Blake, George H. Browne, William G. Brown, Buffuton, Calvert, Chamberlain, Clark. Cobb, Frederick A. Conkling, Corning, Cox, Craceras, Crisfield, Crittenden, Gutler, Davis, Duell, Dualap, Eliot, English, Fessenden, Granger, Grüder, Haight, Hall, Hanchett, Harding, Harrison, Hickman, Hol-man, Hooper, Horton, Hutchins. Johnsom, Julian, Francis W. Kellogg, William Kellogg, Kerrigan, Knapp, Lansing, Law, Lazear, Leary, Lechnan, Lovejoy, Mullory, Nöble, Noell, Norton, Nugen, Odell, Pendleton, Perry, John S. Phelps, Potter, Price, John H. Rice, Richardson, Robinson, Sargent, Sedgwick, Segar, Sheffield, Skiel, Sloan, Spaulding, John B. Steele, William G. Steele, Stevens, Benjamin F. Thomas, Francis Thomas, Trowbridge, Fallandigham, Van Horn, Van Valkenburgh, Foorhees, Wadsworth, Wail, Ward, Web-ster, Wickliffe, Wilson, Windom, Woodr, Woodruff-95.
Mr. Sznowyney's enbstituto as follows:

#### Mr. SEDGWICK's substitute, as follows

And whereas the several States of Virginia, North Caro-And whereas the several states of virginia, North Carolina, Georgia, Tennessee, Alabama, Missie-sippi, Louisiana, Florida, Texas, and Arkansas, wickedly and unlawfully combining under the title of the Confederate States of America, have, together, made war upon and re-belled against the Government of the United States, and continue in such state of war and rebollion: Therefore,

SEC. 3. Be it further enaded, That every commanding military or naval officer whose military district shall em-brace any portion of the above-named States may, and it shall be his duty, by proclamation or otherwise, to invite all loyal persons to come within his liues and be enrolled in the service of the United States; and it shall be his duty to enroll every such loyal person and to employ such of them as may be necessary in the service of the United States, and no person so enrolled and declaring his loyalty to the United States shall ever thereafter be held to involuntary service or labor, (excepting as a punishment for crime,) any law or regulation of any State to the contrary notwithstanding: *Provided*, That if the slaves of any person or persons who have been and continued loyal to the Government of the United States shall be made free by the operation of this section, such loyal citizen or citizens shall be entitled to just and reasonable compensation for his claim to the service or labor of such slave: And provided further, That if the slaves of any person or persons who are inters or married women shall be made free by the opera-tion of this section, they shall also be entitled to just and reasonable compensation for their claim to the service or labor of such slaves.

Was rejected-yeas 32, nays 116; as follows:

YEAS-Messrs. Aldrich, Babbitt, Baker, Beaman, Samuel S. Blair, Blake, Buffinton, Clark, Davis, Duell, Edgerton, Eliot, Fessenden, Frank, Hickman, Hutchins, Julian, Fran-cis W. Kellogg, Lansing, Lovejoy, Anson P. Morrill, Pike, Potter, John H. Ricz, Riddle, Edward H. Rollins, Sedgwick, Sloan, Trowbridge, Van Valkenburgh, E. P. Walton, Wilson, Window Windom—32.

NATS-Messrs. Allen, Ancoma, Arnold, Ashley, Baily, Bax-ter, Biddle, Francis P. Blair, Jacob B. Blair, George H. Browne, William G. Brown, Calvert, Campbell, Chamberlain, Art, Duance, Frinners F. Dhau, Shound B. Dhair, George H. Browne, William G. Brown, Calvert, Campbell, Chamberlain, Clements, Cobb, Colfax, Frederick A. Conkling, Roscoe Conk-ling, Corning, Cox, Cravens, Criefield, Crittenden, Cutler, Dawes, Delano, Diven, Dunlan, Dunn, Edwards, Ely, Eng-lish, Fenton, Fisher, Gooch, Goodwin, Granger, Grider, Gur-ley, Haight, Halt, Hanchett, Harding, Harrison, Holman, Horton, Johnson, Kelley, William Kellogg, Kerrigan, Kil-linger, Knapp, Law, Lazear, Leary, Lehman, McKnight, McPherson, Mallory, Maynard, Menzies, Mitchell, Moorhead, Justin S. Morrill, Nixon, Noble, Noell, Norton, Nugen, Odeil, Olin, Patton, Pendleton, Perry, John S. Phelps, Timothy G. Phelps, Pomeroy, Porter, Price, Alexander H. Rice, Rich-ardson, Robinson, Jones S. Rollins, Sargent, Segar, Shanks, Shefield, Shiel, Smith, Spanlding, John B. Steele, William G. Yoorhees, Woalsworth, Wall, Wallace, Ward, Washburne, Webster, Wheeler, Whaley, Albert S. White, Wickliffe, Wood, Woodruff, Worcester-Hl6.

# Mr. E. P. WALTON'S substitute provides:

SEC. 1. That every person who shall hereafter commit treason, shall suffer death, and all his slaves made free; or be imprisoned not less than five years, and fined not less than \$10,000, and all his slaves made free; said fine shall be levied

SEC. 2. That if any person shall engage in rebellion against the United States, or give aid and comfort thereto,

shall be punished by the forfeiture of his personal property, including choses in action, and his life estate in any real property within the United States, and by the liberation of his slaves.

SEC. 3. That every such person described disqualified to bold any office under the Government. SEC. 4. This act shall not be construed as to affect prose-

cution or conviction of persons guilty of treason before the passage of this act, unless convicted under it.

SEC. 5. The duty of the President, by his marshals or other officers, to seize and sequester property of every kind, of persons engaged in rebellion and especially of officers of the rebel army and navy, and of the President and other officers, military, naval, or civil, of persons formerly hold-ing office under the United States and taking up arms, or give aid and comfort to the rebellion, or persons owning property in loyal portions of the country hereafter engaged appearance and trial of the offender. No slave to be seized under this act, but the United States to have a lien on all slaves of the persons described to answer such order as may be made touching their liberation, and no sale of any force made after the commission of the offence.

Sic. 6. That the property so seized and sequestered shall be held or rented until the owners can be proceeded against, and if convicted, said property shall be forfeited and all perishable property may be sold and proceeds paid into the Treasury, and if owner discharged on trial shall be returned to said owner. returned to said owner.

SEC. 7. That if the owner of any property seized shall fiee from justice, and cannot be brought to trial, an order Schall be made by the court requiring such person to appear at a certain time, and if he do not, all his estate shall be forfeited, and the liberation of his slaves, and himself and heirs forever barred from recovery thereof.

SEC. 8. That the President of the United States, when he deems it necessary that any personal property seized by the army and havy, and helonging to a person hereafter en-gaged in the rebellion or given aid and comfort thereto should be confiscated, may cause proceedings *in rem* against such property, and if found to belong to a person engaged in the rebellion, said property shall be forfeited. SEC. 9. President may by proclamation command insur-

gents to lay down their arms within sixty days, and if they do not, their property shall be confiscated and slaves freed. SEc. 10. That no person discharged from labor under this act, nor the descendants of any one, shall ever be reduced

to involuntary servitude, (except as a punishment for crime,)

to involuntary service (excepts a punishment or trune,) and entitled to be discharged on *habeas corpus*. SEC. 11. That whenever any person shall be discharged from service or labor owing to another, the court shall give such person a certificate of discharge, under seal of the court, and conclusive evidence of his freedom, and if thereafter seized, shall be discharged on habeas corpus, and if the person so holding the freed man shall be convicted, he shall be punished with imprisonment for not less than one year or more than five years.

SEC. 12. That no slave escaping from one State or terri-tory to another shall be delivered up, (except for crime,) unless the owner make oath that he has not been engaged in the rebellion or aided and abetted it; and no person in the military or naval service of the country shall assume to pass on the validity of any claim of one person to the services of another.

SEC. 13. That the President is authorized to employ negroes for the suppression of rebellion and treason, and

negrotes for the suppression of rependent and reason, and organize and use them as he may deem proper. SEC. 14. And is also authorized to make provision for the colonization of negroes made free by this act as may be willing to emigrate, and obtain the consent of the Gov-ernment of the said country to their protection and settle-cent within the acoust with the widths and mixing each of the Govment within the same, with all the rights and privileges of freemen

SEC. 15. And is also authorized to extend pardon and amnesty to these engaged in the rebellion.

SEC. 16. That the courts shall have full power to carry this act into effect.

Which was rejected-yeas 29, nays 121. The YEAS were

Messrs. Baxter, Francis P. Blair, Dawes, Delano, Dunn, Fisher, Frank, Gooch, Goodwin, Granger, Killinger, Mc-Knight, Maynard, Mitchell, Moorhead, Justin S. Morrill, Nixon, Olin, Patton, Timothy G. Phelps, Porter, Alexander H. Rice, *Sheffield*, Stratton, Train, Trimble, Verree, E. P. Walton, Worcester.

Mr. MORRILL, of Vermont, offered a substitute providing

SEC. 2. That the President shall appoint commissioners for each State by whose laws persons are held to service, who shall make a list of the names of slaves and their owners and return it to the district where the slave resides; which

list shall be published, requiring all persons to appear and show cause why certain persons held to labor should not be discharged; and on failure of such persons to appear, their slaves shall be declared free, and on appearance the claim-ant shall file an affidavit that be has not been engaged in rebellion, or aided and abetted it, or if engaged under com-pulsion must show it, and if proved that he was not engaged in rebellion his slaves shall be returned to him; but on fail-me of such proof or on the failure or refusal to take the ure of such proof, or on the failure or refusal to take the foregoing affidavit to prove the compulsion when alleged as an excuse, the court shall declare the person so claimed as free, and grant him a certificate of the same under seal, and shall be conclusive evidence of his freedom. And all persons so declared free, if seized, shall be forthwith discharged on habeas corpus; and the court, acting on said writ, shall commit for trial for kidnapping the person so holding the freed man, and on conviction imprisoned for not less than one or more than five years, and any one swearing falsely

shall be guilty of perjury. SEC. 3. That if any person held to labor by one engaged in rebellion, if omitted from the commissioner's list may, on summary application to the district court, be placed on the list, and to be treated in every way the same as if his name had been placed on the list by the commissioner.

Commissioners shall have ample time to complete their lists.

SEC. 4. That no such person or his descendants shall ever again be reduced to involuntary servitude, and every such person shall be entitled to discharge from such service on habeas corpus.

Which was rejected-yeas 16, nays 126. The YEAS were-

Messrs. Roscoe Conkling, Dunn, Frank, Goodwin, Kill-inger, Loomis, McKnight, Justin S. Morrill, Nixon, Olin, Patton, Timothy G. Phelps, Porter, Stratton, E. P. Walton, Worcester-16.

The original bill reported from the committee was then negatived-yeas 74, nays 78; as follows:

IOIIOWS: YEAS-Messrs. Aldrich, Alley, Arnold, Ashley, Babbitt, Baker, Baxter, Beaman, Francis P. Blair, Samuel S. Blair, Blake, Buffinton, Campbell, Chamberlain, Clark, Colfax, Frederick A. Conkling, Roscoe Conkling, Cutler, Davis, Duell, Edgerton, Edwards, Eliot, Ely, Fenton, Fessenden, Frank, Gooch, Goodwin, Gurley, Hanchett, Hickman, Hooper, Hutchins, Julian, Kelley, Francis W. Kellogg, Lansing, Loomis, Lovejoy, McKnight, McPherson, Moor-head, Anson P. Morrill, Justin S. Morrill, Olin, Patton, Timothy G. Phelps, Pike, Pomeroy, Potter, John H. Rice, Riddle, Edward H. Rollins, Sargent, Sedgwick, Shanks, Sloan, Spaulding, Stevens, Trowbridge, Van Horn, Van Valkenburgh, Verree, Wall, Wallace, E. P. Walton, Wash-burne, Wheeler, Albert S. White, Wilson, Windom, Wor-extr-74. cester-74.

cester-74. NATS-Mesers. Allen, Ancona, Baily, Biddle, Jacob B. Blair, George H. Browne, William G. Brown, Calvert, Clem-ents, Cobb, Corning, Cox, Cravens, Crisfield, Crittenden, Dawes, Delano, Divon, Dunlar, Dunn, English, Fisher, Granger, Grider, Haight, Hall, Harding, Harrison, Holman, Horton, Johnson, William Kellogg, Kerrigan, Killinger, Knapp, Law, Lazear, Leary, Lehman, Mallory, Maynard, Menzies, Mitchell, Nixon, Noble, Noell, Norton, Nugen, Odell, Pendleton, Perry, John S. Phelps, Porter, Price, Alex-ander H. Rice, Richardson, Robinson, James S. Rollins, Se-gar, Sheffield, Shiel, Smith, John B. Steele, William G. Steele, Stratton, Benjamin F. Thomas, Francis Thomas, Train, Trimble, Yullandigham, Voorhees, Wadsworth, Ward, Web-ster, Whaley, Wickliffe, Wood, Woodruff-78. May 27-Mr. PORTEE moved to reconsider

May 27-Mr. PORTER moved to reconsider this vote. A motion to table the motion to reconsider was lost-yeas 69, nays 73.

June 4-The motion to reconsider was agreed to-yeas 84, nays 65; and

Mr. PORTER moved that the bill be recommitted to the Committee with instructions to prepare a substitute, providing that the slaves of the persons included in the classification in the confiscation law, are declared forever distollows: tollows: Transport during the definition faw, are defined for the providing for an enrolment of them by commissioners, and the action of United States judges, as indi-cated in other amendments, for colonization of them in Mexico, Central or South America, or the Gulf Islands, with homesteads, and declar-ing every person embraced in the classification the difference of the providing for the provided of the provided for the provided of the provided

incapable of holding or exercising any office of honor, trust, or profit under the Government of the United States; which was agreed to-yeas 84, nays 66, as follows:

84, nays 66, as follows: Yras-Messrs. Aldrich, Alley, Arnold, Ashley, Babbitt, Baker, Baxter, Beanan, Bingham, Francis P. Blair, Blake, Bufinton, Burnham, Chamberlain, Clark, Colfax, Federick A. Conkling, Roscoe Conkling, Covode, Davis, Dawes, Dnell, Junn, Edgerton, Edwards, Ely, Fessenden, Fisher, Fran-chot, Frank, Gooch, Goodwin, Gurley, Hale, Hancbett, Hickman, Hooper, Hutchins, Julian, Kelley, Francis W. Kellogg, Killinger, Lansing, Loomis, Lovejoy, Low, Mc-Knight, McPherson, Mitcheil, Moorhead, Anson P. Morrill, Justin S. Morrill, Nixon, Timothy G. Phelps, Pike, Pome-roy, Porter, Potter, Alexander H. Rice, John H. Rice, Rid-dle, Edward H. Rollins, Sargent, Sedgwick, Shanks, Shella arger, Sican, Spaulding, Stevens, Stratton, Train, Trimble, Trowbridge, Van Horn, Van Valkenburgh, Verree, Wall, Wallace, Washburne, Albert S. White, Wilson, Windom, Worcester-S4. Worcester. -84.

Mars-Messel Messel William J. Allen, Baily, Biddle, Jacob B. Blair, George H. Browne, William G. Brown, Calvert, Clem-ents, Colb, Corning, Cox, Crawens, Crittenden, Delano, Dela-plaine, Dunlap, English, Fouke, Grauger, Grider, Haight, Marding, Harrison, Holman, Horton, Johnson, William Kel-ogg, Kerrigan, Knapp, Law, Lazear, Leary, Lehman, Mal-Harding, Harrison, Holman, Horton, Johnson, William Kel-logg, Kerrigan, Knapp, Law, Laeear, Leary, Lehman, Mal-lory, May, Maynard, Menzies, Noble, Noell, Norton, Nugen, Pendleton, Joins. P. Heips, Price, Robinson, James S. Rollins, Segar, Sheffield, Shiel, Smith, John B. Steele, William G. Steele, Stiles, Benjamin F. Thomas, Francis Thomas, Val-landigham, Vibbard, Voorhees, Wadsworth, Ward, Webster, Chilton A. White, Wickliffe, Woodruff, Wright-66.

June 18-The House passed a bill reported in pursuance of these instructions-yeas 82, nays 34. It was not considered in the Senate; but emancipation clauses were inserted in the confiscation bill, and agreed to by both Houses.

During the pendency of another bill (107) to forfeit the property and slaves of persons who shall engage in, or aid and abet, armed rebellion against the United States,

1862, April 22-Mr. BINGHAM-offered a substitute-

That if any person or persons, within any State or Terri-tory of the United States, shall, after the taking effect of this act, engage in armed rebellion against the Government of the United States, or shall aid or abet such rebellion, all the property moneys, stocks, credits, and effects of such person or persons are hereby declared lawful subjects of prize and capture, wherever found, for the indemnity of the United States against the expenses of suppressing such re-bellion; and it is hereby made the duty of the President of the United States for an expense of suppressing such rethe United States to cause all such property, wherever found, to be seized, to the end that the same may be confiscated and condemned as hereinafter provided for the use of the United States.

SEC. 2. That all property so captured or seized shall be condemned in the district courts of the United States, and that the proceedings of condemnation shall be *in rem*, and shall be instituted and prosecuted in the name of the United States in any district court of the United States within any district in which the same may be seized or situate, or into district in which the same may be seized or situate, or into which the same may be taken and proceedings first insti-tuted, and which proceedings shall conform as nearly as may be to proceedings in prize cases, or to cases of forfeit-ure arising under the revenue laws; and in all cases the property so seized and condemned, whether real or per-sonal, shall be sold pursuant to such rules as the Secretary of the Treasury may prescribe, and the proceeds deposited in the Treasury of the United States for the sole use of the United States. United States.

SEC. 3. That the Attorney General or any district attorney of the United States of any district in which the said property or effects may at the time be, or into which the same may be taken, shall institute the proceedings of condemna-tion as hereinbefore provided.

Which was agreed to-yeas 62, nays 48; as follows:

labarger, Stevens, Stratton, Trimble, Trowbridge, Van Horn, Van Valkenburgh, Albert S. White, Wilson, Windom, Worcester-62

cester-62. NAYS-MESSTS. Allen, Joseph Baily, Baxter, Biddle, Francis P. Blair, Jacob B. Blair, George H. Browne, William G. Brown, Cuhert, Cascy, Clements, Cox, Crisfield, Crittenden, Diven, Dunlap, Grider, Hall, Harding, Harrison, Hickman, Kerrigan, Knapp, Law, Lehman, Mallory, May, Menzies, Justin S. Morrill, Morris, Noble, Norton, Odell, Olin, James S. Rollins, Shefjield, Smith, John B. Steele, Benjamin F. Thomas, Francis Thomas, Vallandigham, Vibbard, Voorhees, E. P. Walton, Ward, Chillon A. White, Woodruff, Wright 48 18

Mr. E. P. WALTON offered a substitute, defining the crime of treason, and affixing a penalty of death, or imprisonment and fine, on conviction or confession, and his slaves shall be free. The President is authorized to appoint commissioners to sequester and seize the property, real and personal, of persons bearing arms against the United States, or giving them aid and comfort, and is also authorized to grant pardon and amnesty. These are the concluding sections :

SEC. 6. And be it further enacted, That if any State, or part thereof, in which the inhabitants have by the President part thereof, in which the inhabitants have by the President been declared in a state of insurrection, the said insurrec-tion shall have continued for a period of six months, then and in that case the President is hereby authorized, if in his opinion it shall be necessary to the successful suppres-sion of said insurrection, by proclamation to fix and appoint a day when all persons holden to service or labor in any such State car next thereof as headed declars, where service such State, or part thereof as he shall declare, whose service or labor is by the law or custom of said State due to any person or persons, who, after the day so fixed by said proclamation, shall levy war or participate in insurrection against the United States, or give aid to the same, shall be free and discharged from all such claim to labor or service; and thereupon said person shall be forever free and dis charged from said labor and service, any law or custom of said State to the contrary notwithstanding.

SEC. 7. That whenever any person claiming to be entitled to the service of any other person as a slave shall seek to enforce such claim, he shall, in the first instance, and as preliminary to the trial of such claim, show satisfactorily that he and the person to whom said service was claimed to be due during the period of insurrection or rebellion was loyal to the United States.

Which was rejected-yeas 33, nays 69, as follows:

follows:
YEAS-Messrs. Baxter, Francis P. Blair, William G. Brown, Calvert, Casey, Clements, Roscoo Conkling, Critterden, Dun-lap, Dunn, Fisher, Goodwin, Granger, Grider, Gurley, Hard-ing, Harrison, McKnight, McPherson, Mallory, Menzies, Mitchell, Moorhead, Justin S. Morrill, Nixon, Olin, Edward H. Rollins, James S. Rollins, Shefjield, Shellabarger, Strat-ton, E. P. Walton, Worcester-33.
NATS-Messrs. Aldrich, Allen, Arnold, Ashley, Bablitt, Joseph Baily, Baker, Beaman, Biddle, Bingham, Jacob B. Blair, Samuel S. Blair, Blake, George H. Browne, Bufinton, Chamberlain, Colfax, Frederick A. Conkling, Covode, Cas, (utler, Davis, Duell, Eliot, Ely, Fenton, Franchot, Frank, Gooch, Hickman, Julian, Kelley, Francis W. Kellogg, Ker-irgan, Krapp, Lansing, Law, Loomis, Lovejoy, May, Morris, Noble, Noell, Odell, Patton, Timothy G. Phelps, Pike, Potter, John H. Rice, Riddle, Sargent, Shanks, Smith, John B. Stele, Palandigham, Van Horn, Van Valkenburgh, Voorhees, Vall, Ward, Albert S. White, Chillon A. White, Wilson, Windom, Woodruff, Wright-69. Woodruff, Wright-69.

Mr. PORTER also proposed a substitute, which was rejected—yeas 30, nays 72.

April 23-The bill was tabled-yeas 59, nays 52

Subsequently a new bill was prepared by the Select Committee, to whom, April 24, the whole subject was referred.

#### IN SENATE.

Pending the consideration of a bill to confiscate the property and free the slaves of rebels, 1862, April 24-Mr. SHERMAN moved to strike

from the first section the words: Belonging to any person or persons beyond the jurisdic-

tion of the same, or to any person or persons in any State or district within the United States, now in a State of in-surrection and rebellion against the authority thereof, so that in either case the ordinary process of law cannot be served upon them, who shall during the present rebellion be found in arms against the United States, or giving aid and comfort to said rebellion.

#### And to insert in lieu thereof the following:

First. Persons hereafter acting as officers of the army or navy of the rebels now or hereafter in arms against the United States. Second. Persons hereafter acting as President, Vice President, member of Congress, judge, foreign minister, consul, or commissioner of the so-called Confed-erato States. Third. Persons hereafter acting as an officer, erato States. Innra, Fersons nerealter acting as an oncer, whether civil, military, or naval, of any State or Territory who by the Constitution of the so-called Confederate States is required to take an oath to support said Constitution. Fourth. Persons who having held an office of honor, trust, or profit under the United States, shall hereafter take up arms against the United States. Fifth. Persons who, own-ing property in the loyal States or Territories, or the loyal portions of dislowed States are solved as or the loyal portions of disloyal States, shall hereafter assist or give aid and comfort to the present rebellion.

Mr. King moved to amend by adding:

Sixth. Persons in the present insurrection levying war against the United States or adhering to their enemies, giving them aid and comfort.

Which was rejected-yeas 7, nays 32, as follows:

YEAS-Messrs. Chandler, Grimes, King, Pomeroy, Trum-

YEAS--Messrs. Chandler, Grimes, King, Pomeroy, Tram-bul, Wade, Wilkinson-7. NArS--Messrs. Anthony, Browning, Carlile, Clark, Cowan, Davis, Dixon, Doolittle, Fessenden, Foot, Foster, Hale, Henderson, Howard, Lane of Indiana, Lane of Kansas, La-tham, Morrill, Nesmith, Powell, Sculsbury, Sherman, Sim-mons, Stark, Summer, Ten Eyek, Thomson, Willey, Wilson of Massachusetts, Wilson of Missouri, Wright-32.

The amendment of Mr. SHERMAN was then agreed to-yeas 26, nays 11, as follows:

YEAS-Messrs. Anthony, Browning, Chandler, Clark, Cowan, Davis, Dixon, Doolittle, Fessenden, Foot, Foster, Grimes, Hale, Henderson, Howard, Howe, Lane of Indiana,

Willey, Wilson of Massachusetts, Wright-26. NAVS-Mickster, Carbie, King, Lane of Kansas, Morrill, Pomeroy, Saulsbury, Stark, Trumbull, Wade, Wilkinson, Wilson of Missouri-11.

The bill was referred to a select committee, and was not reported; but this classification was introduced into the confiscation act.

SECOND AMENDATORY JOINT RESOLUTION.

First Session, Thirty-Eighth Congress.

#### IN HOUSE.

1864, February 5-The House passed this joint resolution :

That the last clause of a "joint resolution explanatory of 'An act to suppress insurrection, to punish treason and for other purposes," approved July 17, 1862, be, and the same hereby is, so amended as to read: "nor shall any punishment or proceeding under said act be so construed as to ishment or proceeding under said act be so construed as to work a forfeiture of the estate of the offender contrary to the Constitution of the United States: *Provided*, That no other public warning or proclamation under the act of July 17, 1862, chapter ninety-five, section six, is or shall be re-quired than the proclamation of the President, made and published by him on the 25th day of July, 1862, which proclamation so made shall be received and held sufficient in all cases now pending, or which may hereafter arise under said act."

By a vote of yeas 83 to nays 76, as follows:

By a vote of yeas 85 to mays to, as follows: YEAS--Messrs. Alley, Allison, Ames, Anderson, Arnold, Ashley, John D. Baldwin, Baxter, Beaman, Blow, Boutwell, Boyd, Brandegee, Broomall, Ambrose W. Clark, Freemau Clarke, Cobb, Cole, Creswell, Henry Winter Davis, Thomas T. Davis, Dawes, Deming, Donnelly, Driggs, Ellot, Farns-worth, Fenton, Frank, Garfield, Gooch, Grinnell, Higby, Hooper, Hotchkiss, Sashel W. Hubbard, Join H. Hubbard, Hulburd, Jenckes, Julian, Kasson, Kelley, Francis W. Kel-logg, Orlando Kellogg, Loan, Longvear, Marvin, McBride, McClurg, McIndoe, Sanuel F. Miller, Moorhead, Morrill, Daniel Morris, Amos Myers, Leonard Myers, Norton, Charles O'Neill, Orth, Patterson, Perham, Pike, Pomeroy, Alexander

H. Rice, John H. Rice, Edward H. Rollins, Schenck, Scofield,

H. Rice, John H. Rice, Edward H. Rollins, Schenck, Scofield, Shannon, Sloan, Smithers, Spalding, Stevens, Thayer, Tracy, Opson, Yan Valkenburgh, Ellihu B. Washburne, William B. Washburn, Williams, Wilson, Windom, Woodbridge—83. NATS-Messrs. James C. Allen, Ancona, Baily, Augustus C. Baldwin, Jacob B. Blair, Elliss, Brooks, James S. Brown, William G. Brown, Chanler, Clay, Cofroth, Cox, Cravens, Dawson, Denison, Edan, Edgerdon, Eldridge, Finel, Ganson, Grider, Harding, Harrington, Benjamin G. Harris, Herrick, Holman, Hutchins, William Johnson, Kulifleisch, Kernan, King, Knapp, Lave, Lazear, LeBlond, Long, Mallory, Marey, McDowell, McKinney, Midleton, Wan. H. Miller, James R. Morris, Morrison, Nelson, Noble, Odell, John O'Neill, Pen-Randall, Robinson, Rogers, James S. Rollins, Hoss, Scott, John B. Steele, William G. Steele, Stiles, Strouse, Stuart, John B. Steele, William of Steele, Stiles, Strouse, Stuart, John B. Steele, William of Steele, Stiles, Strouse, Stuart, John B. Steele, William Onthesson, Webster, Whaley, Wheeler, Chilton A. White, Joseph W. White, Winfield, Fernando Wood, Teaman-76.

#### IN SENATE.

February 17-Mr. REVERDY JOHNSON, from the Committee on the Judiciary, reported back this joint resolution, without amendment, and with a recommendation that it do not pass.

June 27-Pending the consideration of the bill to establish a bureau of Freedmen's Affairs.

Mr. TRUMBULL offered this as a new section :

And be it further enacted, That the last clause of a joint resolution explanatory of an act to suppress insurrection, to purish treason and rebellion, to seize and confiscate the property of rebels, and for other purposes, approved July 17, 1862, be, and the same is hereby, repealed.

The words proposed to be repealed are, " nor shall any punishment or proceedings under said act be so construed as to work a forfeiture of the real estate of the offender beyond his natural life."

June 28-This was agreed to-yeas 23, nays 15, as follows:

YEAS-Messrs. Anthony, Brown, Chandler, Conness, Foot, Grimes, Hale, Harlan, Harris, Howe, Lane of Indiana, Lane of Kansas, Morgan, 'Morrill, Pomeroy, Sherman, Sprague, Summer, Trumbull, Van Winkle, Wade, Wilkinson, Wilson

NAYS-Messrs. Carlile, Clark, Collamer, Cowan, Davis, Doolittle, Henderson, Hendricks, Hicks, McDougall, Powell, Riddle, Saulsbury, Ten Eyck, Willey-15.

Same day, Mr. HENDRICKS moved to strike out this section; which was disagreed toyeas 13, nays 16, as follows:

YEAS-Messrs. Buckalew, Carlile, Clark, Cowan, Davis, Doolittle, Hendricks, McDougall, Powell, Riddle, Ten Eyck, Van Winkle, Willey—13.

NATS-Messrs. Anthony, Chandler, Conness, Foot, Har-lan, Howe, Lane of Indiana, Morgan, Morrill, Pomeroy, Ransey, Sprague, Sumner, Trumbull, Wilkinson, Wilson -16.

# "Rebel" Sequestration.

1861, August 6-The "Confederate" Congress passed a bill "for the sequestration of the estates, property, and effects of alien enemies, and for the indemnity of citizens of the Confederate States, and persons aiding the same in the existing war against the United States," of which the Richmond Examiner of the follow-

ing Monday gave an abstract. The following is the principal legislative clause :

**Clause:** Be it enacted by the Congress of the Confederate States, That all and every the lands, tenements, and hereditaments, goods and chattels, rights and credits within these Confed-orate States, and every right and interest therein held, owned, possessed, or enjoyed by or for any alien enemy since the twenty-first day of May, 1861, except such debts due to an alien enemy as may have been paid into the Treasury of any one of the Confederate States prior to the passage of this law, be and the same are hereby seques-trated by the Confederate States of America, and shall be

held for the full indemnity of any true and loyal citizen, a resident of these Confederate States, or other verson aiding said Confederate States in the prosecution of the present war between said Confederate States and the United States of America, and for which he may suffer any loss or in-jury under the act of the United States, or of any State thereof, authorizing the seizure or confiscation of the property of citizens or residents of the Confederate States, never, automizing the service or confiscation of the property of citizens or residents of the Confederate States, or other persons aiding said Confederate States, and the same shall be seized and disposed of as provided for in this act: *Provided, however*, When the estate, prop-erty, or rights to be effected by this act were or are within some State of this Confederacy, which has become such since said twenty-first day of May, then the act shall oper-ate upon and as to such estate, property, or rights, and all persons claiming the same from and after the day such State became a member of this Confederacy, and not be-fore: *Provided further*, That the provisions of this act shall not extend to the stocks or other public securities of this Confederace, held or owned by an alien enemy, or to any debt, obligation, or sum due from the Confederate Govern-ment, or any of the States to such alien enemy. *And pro-tieled*, also, That the provisions of this act shall not extend the property of citizens or residents of either of the States of Delaware, May and Kornucker, Wiener and prothe property of citizens or residents of either of the States of Delaware, Maryland, Kentucky, Missouri, or the District of Columbia, or the Territories of New Mexico, Arizona, or the Indian Territory south of Kansas, except such of said citizens or residents as shall commit actual hostilities against

the Confederate States, or aid or abet the United States in the existing war against the Confederate States. Sections 2 to 13 provide for the appointment of receiv-ers in each county, and impose a penalty of \$2,000 on all who may eucleavor to conceal the ownership of property belowing the above county. belonging to aller enemies. Section 14 provides for the appointment of three commissioners to take charge of the sequestration fund, and to hear and decide on all claims against it.

For its enforcement, the then Attorney General, J. P. Benjamin, issued this circular:

DEPARTMENT OF JUSTICE

RICHMOND, September 12, 1861. Instructions to receivers under the act entitled "An act for

the sequestration of the estates, property, and effects of alien enemies, and for the indemnity of citizens of the Confederate States and persons aiding the same in the existing war against the United States," approved August 8th, 1861.

The following persons are subject to the operation of the law as alien enemies :

All citizens of the United States, except citizens or residents of Delaware, Maryland, Kentucky, or Missouri, or the District of Columbia, or the Territories of New Mexico,

Arizona, or the Indian territory south of Kansas. All persons who have a *domicil* within the States with which this Government is at war, no matter whether they be citizens or not: thus the subjects of Great Britain, France, or other neutral nations, who have a domicil, or are carry-ing on business or traffic within the States at war with this

Confederacy, are alien enemies under the law. All such citizens or residents of the States of Delaware, Maryland, Kentucky, or Missouri, and of the Territories of New Mexico, Arizona, and the Indian territory south of Kansas, and of the District of Columbia, as shall commit actual hostilities against the Confederate States, or aid or assist the United States in the existing war against the Confederate States.

Immediately after taking your oath of office, you will take possession of all the property of every nature and kind whatsoever within your district belonging to alien enemies as above defined.

You will forthwith apply to the clerk of the court for writs of garnishment under the Sth section of the law, and will propound to the garmishees the interrogatories of which a form is annexed. These interrogatories you will probuilt to the following persons, viz: 1st. All attorneys and counsellors practicing law within

your district.

2d. The presidents and cashiers of all banks, and principal administrative officers of all railroad and other corpora-

paradiministrative oncess of an intervention of the second utors and administrators of estates, assignees and syndics of insolvent estates, trustees, and generally all persons who are known to do business as agents for others.

In the first week of each month you will exhibit to the judge a statement showing the whole amount of money in your hands as receiver, and deposit the same for safe keep-ing in such bank or other depository as may be selected for that purpose by the judge-reserving only such amount as

may be required for immediate necessary expenditure in the |

may be required for immediate necessary expenditure in the discharge of your duties as receivers. Whenever, in the discharge of your duties, you discover that any attorney, agent, former partner, trustee, or other person holding or controlling any property, rights, or credits of an alien enemy, has wilfully falled to give you information of the same, you will immediately report the fact to the district attorney for your district, to the end that the guilty party may be subjected to the pains and penalties prescribed by the third section of the law.

J. P. BENJAMIN, Attorney General.

The following interrogatories to garnishees have been prepared for your use, together with a note annexed for the information of the garnishee: I. Have you now, or have you had in your possession or under your control, since the twenty-first day of May last, 1961 before the twenty-first day of May last,

(1861,) and if yea, at what time, any laud or lands, tene-ment or tenements, hereditament or hereditaments, chattel or chattels, right or rights, credit or reclits, within the Confederate States of America, held, owned, possessed, or enjoyed for or by an alien enemy, or in or to which any alien enemy had, and when, since that time, any right, title, or interest, either directly or indirectly? 2. If you answer any part of the foregoing interrogatory in the offermating them of forth proceeding and the matting

in the affirmative, then set forth, specifically and particularly, a description of such property, right, title, credit, or interest, and if you have disposed of it in whole or in part, or of the profit, or rent, or interest accruing therefrom, then state when you made such disposition, and to whom,

then state when you made such disposition, and to whom, and where such property now is, and by whom held? 3. Were you, since the twenty-first day of May, 1861, and if yea, at what time, indebted, either directly or indi-rectly, to any alien enemy or alien enemics? If yea, state the amount of such indebtedness, if one, and of each indebt-edness if more than one; give the name or names of the creditor or creditors, and the place or places of residence, and state whether and to what extent such debt or debts have heen discharged and also the time and menner of the have been discharged, and also the time and manner of the discharge.

4. Do you know of any land or lands, tenement or tenements, hereditament or hereditaments, chattel or chattels, right or rights, credit or credits, within the Confederate States of America, or any right or interest held, owned, possessed, or enjoyed, directly or indirectly, by or for one or nore alien examines ince the twenty-first days of May 1861. more alien enemies since the twenty-first day of May, 1861, or in or to which any one or more alien enemies had since that time any claim, title, or interest, direct or indirect? If yea, set forth specially and particularly what and where the property is, and the name and residence of the holder,

the property is, and the name debtor, trustee, or agent. 5. State all else that you know which may aid in carry-ing into full effect the sequestration act of the 30th of Au-gust, 1861, and state the same as fully and particularly as is there entry specially interrogated. A. B., Receiver.

Note—The garnishee in the foregoing interrogatories is specially warned that the sequestratiou act makes it the duty of each and every citizen to give the information asked in said interrogatories.—[Act 30th August, 1861, sec. 2.] And if any attorney, agent, former partner, trustee, or other person holding or controlling any property or interest therein of or for any alien enemy shall fail speedily to inform the receiver of the same, and to render him an ac-account of such property or interest, he shall be guilty of a high misdemeanor and, upon conviction, shall be fined in a sum not exceeding five thousand dollars, and imprisoned not longer than six months, and be liable to pay besides to the Confederate States double the value of the property or interest of the align ememios are held or mybinst to him or interest of the alien enemies so held or subject to his control.- | Sec. 3.7

This act was rigidly enforced. Subjoined are two notices, preserved from the newspapers of the day

Thomas T. Giles, who has been appointed receiver for the eastern district of Virginia, has issued a notice addressed "to all whom it may concern," in the following words: "I hereby notify every attorney, agent, former partner, trustee, or other person holding or controlling any lands, tenement or breditarent code or thetaffer.

trustee, or other person holding or controlling any lands, tenements, or hereditaments, goods or chattels, rights or credits, or any interest therein, within the eastern district of Virginia, of or for any alien enemy of the Confederate States of America, speedily to inform me, appointed receiver for the said district, of the same; and to render to me an account thereof, and, so far as is practicable, to place the same in my hands. Any such person wilfully failing to do so will be guilty of a high misdemeanor, and liable to be indicted, convicted, fined, and imprisoned, as provided by law

interest therein held, owned, possessed, or enjoyed by or for

"My office is on the northern side of Main street, in the city of Richmond, between Tenth and Eleventh streets."

# SEQUESTRATION NOTICE

Merchants and all other persons residing in Nansemond, Norfolk city, city of Portsmouth, Princess Anne, and Isle of Wight who owe debts to alien enemies, or have property of any kind in their possession, or under their control, be-longing to any such alien enemies, and who have failed and neglected to make report thereof, are hereby notified that unless a report of the said debts and information of said property is rendered by them to the undersigned, receiver of this district, on or before Saturday, the 30th instant, they will be reported as delinquents and subject to the fine of five thousand dollars imposed by law. Merchants and all other persons residing in Nansemond, five thousand dollars imposed by law

JOHN T. FRANCIS,

Receiver.

This is a copy of the writ served upon the wife of Hon. Horace Maynard, of Tennessee, while he was in Washington, in attendance upon Congress, quoted in his speech of May 23, 1862:

# To Hon. WEST H. HUMPHREYS, Judge of the District Court of the Confederate States of America for the District of Tennessee :

The Confederate States of America, through Landon C. Haynes, the receiver for the eastern district of Tennessee, respectfully represents unto your honor that within the ju-risdiction of this court there are (various items of property specifically enumerated.] The said States, by said receiver, show that said property, real, and personal, and mixed, be-longs to one Horace Maynard, who has his domicil and who resides in the function of the States at the interior resides in Kentucky, or some one of the States or districts of the United States, and who is in actual hostility to the Government of the Confederate States of America, and who adheres to the enemy of said States by speeches, words, and acts, giving them aid and comfort in Kentucky and other acts, giving them aid and comfort in Kentucky and other places in the United States, and is an alien enemy to said Confederate States. All of which are situate and being in the counties of Knox and Campbell, in the State and within the district of said receiver. And the said Confederate States further represent that the said property, debts, claims, choses in action, are, as said States have been informed by said re-ceiver, under the control and supervision of Mrs. Maynard, wife of said Horace Maynard, and the said W. P. Washburn, who resides in Knox county, in this State, which property is liable to be seized, under the act of the Confederate Con-gress for the seomestration of the estates, property, and efgress for the sequestration of the estates, property, and effects of alien enemies. The said Mrs. Maynard and Washburn are required to set forth, on oath, and specifically to describe said property, and debts, and choses in action, as they have full knowledge of the same, and answer the inter-rogatories herewith filed, on oath, and marked exhibit A, as

a part of this petition. The Confederate States of America, through said receiver, The Confederate States of America, through said receiver, therefore pray that the said persons having supervision and control over said property, as aforesaid, who reside in the county of Knox, be made parties to this petition, and that a copy thereof, together with notice, be issued by the clerk of this court to the marshal or his deputy, to be served on said persons. Said Confederate States further pray that your honor direct said property to beseized and sequestrated, and placed in the hands of said receiver, and by him sold or dis-posed of upon such terms and conditions as your honor may direct. And on final heaving the Confederate States near direct. And on final hearing, the Confederate States pray for all such other, further, and different relief in the prem-ises as may be consistent with the act of the Confederate Congress.

# LANDON C. HAYNES,

Receiver for the Eastern District of Tennessee. A true copy of the original.

William G. McAdoo, Clerk, dc.

Proceedings were also taken against Judge Catron and Andrew Johnson.

#### SUNDRY ITEMS.

The Southern Commercial Convention which met at Macon, Oct. 16, 1861, passed a series of resolutions in relation to the sequestration act, and recommending that the payment of debts sequestered be not required during the war; that claims for indemnity and indebtedness due "I also notify each and every citizen of the Confederate States speedily to give information to me (as he is required by law to do) of any and every lands, tenements, and here-ditaments, goods and chattels, rights and credits, within the said eastern district of Virginia, and of every right and

the disabilities of coverture of infancy be exempted.

#### The Richmond Enquirer of Oct 14th says:

"It was understood at Richmond by a gentleman who is connected with the Department of the Interior, that the returns were being made of debts due to tho alien enemies with the utmost promptitude, and that from the city of Petersburg there would be realized a sum not less than all hundred thousand dollars. As for Richmond, it was difficult to estimate the aggregate, but the lowest estimate placed it at two millions of dollars." The Confederate States Court in Richmond have confis-ented \$15000 in the head of the bondrage Dursell in that

The Confederate States Court in Richmond have confis-eated \$15,000 in the hand of the bankers Purcell, in that city, the property of Ashmead *et als.*, of Philadelphia.

In April, 1863, the "Confederate" Senate considered a bill to confiscate the leasehold interest and shares of stock owned by the American Telegraph Company and other alien enemies in the lines of telegraph in the Confederate States. Mr. Oldham and Mr. Hill favored the bill, and Mr. Johnson, of Arkansas, and Mr. Johnson, of Georgia, opposed it-the former, because many of the stockholders whose property it was proposed to confiscate were not enemies, but friends living in Washington city and Maryland—the latter because he considered confiscation "an act of legal plunder unworthy of the age." The bill was postponed until the next day, and its fate is not known.

A movement was made about the same time in the Legislature of Virginia, appropriating so much of the public debt of the State and other securities held by resident citizens of the United States and the District of Columbia as may be necessary to indemnify the citizens of Virginia who are loyal to the State for losses sustained by them in consequence of any confiscation act of the Congress of the United

States, or any other act growing out of the war. Up to September 30, 1863, the rebel treasury had received from sequestration \$1,862,650 27, as reported by the Secretary of the Treasury.

#### THE "SEQUESTRATION" ACT DECLARED CONSTITU-TIONAL

In the fall of 1861, in the first session of the Confederate States Court, in Charleston, after Judge Magrath charged the Jury, the following proceedings are reported to have taken place :

Mr. Petigru read a writ of garnishment, served upon him, and interrogatories attached, in reference to alien enemy's

property. Mr. Petigru said the objection he had to these interroga-tories was, that no human authority has the right to put these questions to him or any one in the same circumstances. these quesitions to him or any one in the same circumstances, the might recognize the authority of South Carolina to do as proposed by the act, because in a State like South Carolina a sufferer has no security or remedy against those in power, unless from some guarantee in the constitution of the State; for a State may do whatever it is not forbidden to do by the fundamental law of the State. But the Confederate States have no such claim to generality. Their authority is con-fined to the constitution which confers it and the powers delegated to them, and whereas, in the case of a sovereign, we must show a guarantee against the power, in the case of the Confederacy they must show a warrant for their power. There is no article in the Constitution of the Confederate States which authorizes them to set up an *inquisition*, or to

There is no article in the Constitution of the Confederate States which authorizes them to set up an *inquisition*, or to proceed otherwise than according to the laws of the land. In fact, the best authority for this proceeding is Hudson's treatise on the Star Chamber, in *Second Collutanca Juridicia*. It will be found that the method prescribed in this confis-cation act is precisely that of the Star Chamber. They call this a writ of garnishment; Mr. Hudson calls it a subpons. This calls upon me to disclose all the cases in my knowledge of property held by an alien energy. Mr. Hudson requires the party to appear before the Star Chamber, and answer all questions which may be put to him. These are allke in being general. There is no plaintiff. It is an *inquisition*.

a thing be legal? \* \* What is incident to cases of the war power, the grant of the war power covers; but does a thing be legal? \* \* What is incident to cases of the war power, the grant of the war power covers; but does the war power require the creation of a Star Chamber to wrong and harass our people? \* Where is the authority given? Where is the power to call upon the citi-zen in a new and unheard-of manner to answer questions upon oath for the purpose of enforcing the confiscation law? Shall it be said that it is to furnish the means to carry on the war? How can that be said to be necessary, which is absolutely never known to have been dome before? Was there anybody that ever fought bofore Gen. Beauregard? War unfortunately is not a new thing. Its history is found on every page. Was there ever a law like this endured, prac-ticed, or heard of? It certainly was not found among the people from whom we derive the common law. No Eng-lish monarch or Parliament has ever sanctioned or under-taken such a thing. It is interly inconsistent with the common law to require an inquisitorial examination of the subjects of the laws of war. It is no more a part of the law of war than it is a part of the law of peace. \* \* \* All that can be said in favor of the end and object pro-posed can be said in favor of the Star Chamber and the Spanish Inquisition. Torquemada set out on the latter in-souls. He labored most earnestly, in season and out of season; and when high necessity commanded, he burnt their bodies to save their sonls. \* \* We do not con-sider that the end justifies the means in these days, but Torquemada might have burnt Jews and Protestants, with-out calling upon their best friends to inform against them, and making it penal not to do so. \* \* \* What is incident to cases of

In quantum might have our a sense that From against them, and making it penal not to do so. \* \* \* The war power includes as an incident, everything which is necessary or usual, since it never was done before. This is necessary or usual, since it never was done before. This is not the first war that since it never was due to be the subject of wild speculation. Now, the means granted to attain this end are based upon the supposition that the end deserves all commendation; that nothing in the world is more calculated to advance the repute of the country than to be keen in searching out the property of enemies, and than to be keen in searching out the property of enemies, and proceeding against them when they have no opportunity of being heard, and to impoverish them by taking away the earnings of their industry and applying it toother uses.\* \* It would be the most intolerable hardship for me, for a citizen, at every quarter section to be obliged to tell all he knows or suspects against his neighbor. It is pretended that it is an innccent proceeding. How can that be inno-cent which calls upon one to commit a breach of trust? "The law protects every men in keeping silence when a

The law protects every man in keeping silence when a question is asked that involves professional confidence. There can be no greater oppression than to compel a per-son to violate a moral or legal duty. Something should be said about the objects of this law,

Someting should be said about the objects of this law, for there is a very common error in supposing that it ap-plies to the estates of native citizens who are living abroad in an enemy's country. The term alien enemy is the only one used in the act. It is a definite, technical construction. An alien enemy must be born out of the legiance of the sovereign. There can be no dispute about it. He is not an alien enemy if he was born within the domains of the sovereign. A sovereign has a right to require his return. He may call on him to come home. What is it in the sover-eign's power to do, and what he may do with his subject when he refuses to return, is another matter. But until he has been called on by his sovereign to return, a man com-mits no breach of duty in living in an enemy's country, according to law. It is impossible that the makers of the law should not be aware of this, and they seem to have pur-

law should not be aware of this, and they seem to have pur-posely left this open for the interposition of humanity. Mr. Petigru denied that there was any precedent for this law; and a freeman could not be compelled to aid this confiscating law, by informing against both his friends and enemies. It was this which moved those brave men, who not only shook the pillars of monarchy to its base, and abolished the Star Chamber, butdid it with the declaration that no such thing should be tolerated again. Are we going, in the hey-day of our youth, to set an example which has been repudiated by every lover of freedom from the be-ginning of time to this day, which has never found an advo-cate, shocks the conscience, and invades the rights of the private citizen.

It is an extraordinary stretch of power, in an extraordi-The satisfiest of the second of power, in an estimation of the second before the world our right to its respect as an enlightened people a people capable of self-government, and of governing them selves in a manner worthy of the civilization and light of the age; and this act, borrowed from the darkest period of tyranny, is dug up from the very quarters of despotism, and put forward as our sontiments. They are not my sentiments; and sorry will I be if in this sentiment I am solitary and alone. \* \* With regard to that which requires the wild time of professional activity and source the bottor in violation of professional confidence, he must be better in-structed before making up his mind to the order of responsi-bility or not. There are cases when it is dishonor or death

-and death will certainly be chosen by every man who deserves the name.

Mr. Miles, the District Attorney, moved that Mr. Petigrn make a return to the court of garnishment, in which the question stated by him should be raised, that if the first duty which devolved upon his honor since he had put on his robes, and opened the first term of the Confederate court in South Carolina, was to listen to an invective against the Government whose commission he bore, at least so much respect might be paid to the mandate of the court, which issued, with the sanction of his honor's name, that a formal return might be made to it, so that the points made by the respondent, in which not only the constitutionality of the respondent, in which not only the constitutionality of the law passed by the Congress of the Confederate States, but the very authority of that Congress itself, and the validity of the Government which it represents, are dra wn in question, may be at least set down for argument, and not be allowed to be treated only with invective.

He might be pardoned, however, if, in passing, he called the attention of the audience for whose benefit the remarks of the respondent seem to have been made, to the singular position which the succinct respondent to day for the first time occupied. It was not strange that one who had so often distinguished.himself by the undaunted boldness with which he threw himsel in optical the undathed boliness with lic opinion, should be the one who now invoked the aid of the Court to protect those whom the law of Congress desig-nates as alien enemies, but whom he still prides himself on calling "fellow-countrymen," from the tyramy of a Govern-ment which attempts to make their property subject to the rules of war. This was consistent with his past position. But it was consistent with metamon host bat the But it was certainly a remarkable metimorphosis, that the eminent jurist who fearlessly, and almost alone, in his opposition to the political sentiments of the State, should now invoke the strictest and sternest construction of State rights that had ever been contended for even iu South Garolina, in opposition to the power of the Confederate Government to pass a law in relation to a subject-matter expressly intrusted to Congress by the Constitution. It is true that the profession of submission to the au-

thority of the State in this matter was accompanied by the explanation, that such submission would be given only because there could be no successful resistance to the tyranny. But even with this qualification, the acknowledgment of the authority of the State was remarkable from such a quarter.

Subsequently Judge Magrath, refusing to sustain the demurrer of Messrs. Petigru, Mitchell and Whaley, sustained the constitutionality of the act.

JUDICIAL ACTION UNDER THE CONFISCATION LAW.

The law has been enforced. The life estate of Joseph R. Anderson, an officer in the rebel army, in about 50,000 acres of land in Allegany county, Maryland, has been condemned. Also, that of Hunter, Ould, Aiken, Magoffin, Boyce, McQueen, Corcoran, and Flournoy, in lands in Wisconsin. Also of rebel owners, various properties in Washington city, New York, Alexandria, &c.

In Alexandria, Judge John C. Underwood of the United States District Court for the eastern district of Virginia, has condemned "all the right, title, and interest" of parties under his view of the law, which he thus argues:

#### United States vs. the Right, Title, and Interest of Hugh Latham, &c.

The decree in this and similar cases must depend upon the construction given to article third, section third, of the Constitution of the United States, and the legislation of the last Congress for the confiscation of rebel property. As the Supreme Court of the United States has never devided the questione involved in this case, and this court

decided the questions involved in this case, and this court feels constrained to differ from the opinion of the supreme court of the neighboring district in some recent cases, a brief statement of the reasons of such dissent from the able jurists of that court may not be inappropriate. This Court cannot limit a decree to a condemnation of a

This Court cannot timit a decree to a condemnation of a traitor's right, title, and interest in the property forfeited for the term of his own life, with a reversion to his heirs, for the reason that it does not consider such limitation warranted by the section and acts of Congress above re-ferred to. The language of the Constitution is as follows: "The Congress shall have power to declare the punish-ment of treason, but no attainder of treason shall work

corruption of the blood or forfeiture, except during the life of the person so attainted."

The general power to purish treason is expressly granted in the first part of the above constitutional provision. The consequences of attainder of treason mentioned in the The consequences of attainder of treason mentioned in the latter part of the provision would not require notice, as this is not a case of attainder of treason, were it not for the reference made to the limitations of this provision in the joint explanatory resolution passed by Congress July 17, 1862. But that reference and the quotation of the limita-tions made it necessary to inquire what was the "forfeit-ure" to be worked, and when the "work" was to be effected. The authors of the constitutional newision were doubt-

The authors of the constitutional provision were doubt-less profound lawyers, and used the term "forfeiture," in its strict technical and well settled legal meaning. Elackstone gives us a whole chapter on this important word, which he

begins thus: "Forfeiture is a punishment annexed by law to some illegal act or negligence in the owner of lands, tenements, and hereditaments, whereby he loses all his interest therein, and they go to the party injured as a recompense for the wrong sustained."

Again, he enumerates "forfeiture," deed, device, &c., as the modes of absolute conveyance of real estate; and it seems clearly that this was the sense in which it was used in the constitutional provision. Besides being good law-yers, the authors were evidently thorough scholars, fami-har with King James's translation of the Bible and the old liar with King James's translation of the Bible and the old English classics, and employed the word "except," in the phrase—"Except during the life of the person attainted," in a sense now nearly obsolete, though common a hundred years ago, and at the date of the Constitution making it synonymous with the word "unless," as in the Bible decla-rations—"Except ye repent ye shall all likewise perish." "Except a man be born again he cannot see the Kingdom of God." "Except these abide in the ship ye cannot be saved." Shakspeare and other English classics abound in examples of the use of this word as a conditional conjuncexamples of the use of this word as a conditional conjunc-

tion, but one will suffice: The Bishop of Winchester, when accused of sedition in King Henry VI, act III, scene 1, indignantly exclaims-

"And for dissension, who preferreth peace, More than I do-except I be provoked?"

We find a similar definition of the term in Webster's Dictionary.

If we use the word "except" in the above sense in the the constitutional provision or make it read "unless during the life of the person attainted," we shall at once come to the true intent and meaning of the provision, to wit: That the forfeiture was to be perfected *during*, and not *after*, the lifetime of the morie retrieved. lifetime of the party attainted.

Bills of attainder were common in the British Parliament. Several of the Colonial Legislatures passed similar acts during the war of the Revolution. They were summary and sweeping in their character, requiring no process or any action of courts of law, but included in a single bill many persons, and in a single section confiscated all their property, both real and personal. In some cases the dead as well as the living were included in these bills, so tran-scendingly wicked did the crime of treason appear to our ancestors. The Government thus convicted men of this crime without trial, and after they were dead and buried, taking as a punishment estates which had already descended to their heirs or passed to innocent purchasers in good faith. This mode of proceeding was manifestly in violation of the general spirit and principles of English and Ameri-can criminal laws, by which crimes, trespasses, and torts of every description, from the highest to the lowest, are alone punishable during the life of the wrong-doer, whose death before final judgment is a perpetual stay of all proceedings and all legal consequences. According to the maxim: "Actio personalis, cun persona moritor." The Constitu-tional provision was therefore inserted to correct this harsh anomaly, and to bring the punishment of treason into har-mony with that of all other crimes and misdemcanors.

It also abolished entirely the cruel doctrine of "corrup-tion of the blood," which prevented the children of a traitor who had been attainted from ever inheriting from

trator who had been attainted from ever inheriting from or through the father either title or estate. Its limitation was as to time of working forfeiture and not as to the definition of the term or character of the estate forfeited, and it made no distinction between personal and real estate. Nor is there anything in the act of Congress of July 17, 1862, limiting the forfeiture within more narrow bounds than those of the Constitution. On the contrary, the only fear ever expressed in relation to that act was that it transcended the Constitution.

The only ground for restricting the forfeiture then must be found in the joint resolution of the two Houses of Congress of the same date of the act aforesaid.

The first branch of this resolution was evidently intended to explain the act and to prevent its application in such a way as to give it an unconstitutional and *ex post facto* oper-

ation. And a careful consideration of its last clause affords the strongest internal evidence that it also was intended simply to prevent an infringement of the Constitution. Its language is copied almost verbatim from the Constitution. Even the very unusual legal phrase, "work a forfoiture," is borrowed from the same source, and means finish, perfect and complete, the necessary legal proceedings for a forfeit-are or conveyance of estate, as we work out or perfect a problem. The only variance is the substitution of the preposition "beyond" for the conjunction "except" in the phrase "beyond the life," instead of "except during the fife." But the similarity of language is so great as to force a conviction that the last part of the resolution, like the first part, was intended to restrain the courts within the bounds of the Constitution. It cannot be supposed that Congress intended to repeal its own act by the resolution, or so to emaculate it as to make it worse than a nullity. It seems also but reasonable that we should give such a construction to the resolution as will, if possible, bring it into harmony with those more weighty and formal instru-And a careful consideration of its last clause affords

into harmony with those more weighty and formal instru-ments, the act and the Constitution. The variation of a ments, the act and the constitution. The transmission single word in quoting the Constitution is not surprising when we consider the probable hurry and pressure which attended the passage of this resolution among many others on the last day of a long session of Congress. We could attended the passage of this resolution among many others on the last day of a long session of Congress. We could hardly expect the same precision of language under such circumstances as in an act passed with mature considera-tion in committees, and after three separate readings in each house of Congress. And though the construction given by the court of the District of Columbia would seem four noting tables the clause rithest it the separate given by the control the bisite of the obstite of the control we will see when we consider those connections, it will appear that our construction will not only best secure the objects of the law, but entirely harmonizes with the act and Constitution,

law, but entirely harmonizes with the act and Constitution, without the least violence to the language of the resolu-tion. By quoting the language of the Constitution, the conclusion is inevitable that the constitutional authority and power was referred to the action of the courts. If the construction that only a life estate is to be confis-cated shall be established, the bill passed by Congress on the 17th of July, 1862, instead of being true to its title, "An Act to suppress insurrection, to punish treason and robellion, to seize and confiscate the property of rebels," would in its effects become an act to prevent the re-settlewould in its checks become an act to prevent the resetue-ment and improvement of the country; an act to promote jealousy, envy, and hatred between the holders of life estates and reversionary interests, and to continue the desolations caused by the rebellion for a whole generation without any benefit to the National Treasury, to the parties interacted on the number of discord interested, or to anybody else except the lovers of discord and officers of law; an act to defeat the leading objects of its own friends in and out of Congress, which objects are well known to be, 1st-by the sale of rebel estates to re-imburse in some degree the immense expenditues incurred in suppressing the rebellion, and 2d—to supply the places of departed rebels with loyal men and bring again into culti-vation and improvement those deserted and wasted estates. But if only a life interest is to he acquired, no purchaser could afford to take on so uncertain a tenure, at even a nominal price, the wasted plantations, and restore the de-stroyed houses and fences by building valuable and permanent structures.

Those of us who remember the condition of the country between the Potomac and the Rappahannock from twenty between the Potomac and the Rappahannock from twenty to forty years ago, the desolation and heart-burning, the agony and despair, the ejectments and expulsions attend-ing the expiration of the onti-republican if le leases of the old proprietor, Lord Fairfax, will certainly pray to be de-livered from a reputition of the scenes which this construc-tion would inevitably produce. The life state theory also involves the absurdity that in proportion to the enormity and certainty of the guilt of any person, should be the shortness of the forfeiture of his estate, for in the most factor the desting the detween conviction and execution is flagrant cases the time between conviction and execution is magrant cases the time between conviction and execution is usually very brief, and forfeiture for that brief period would be both puerile and preposterous. Applying, then, the well sottled rule of interpretation, that where two constructions can be given to a written instrument, that must be pre-sumed which is most just and beneficial to the public in its convertion, we can have, and doubt of the meaning of this operation, we can have no doubt of the meaning of this resolution.

Judging from the many wise enactments of the last Congress, we should be forced to the belief that its intentions gress, we should be torced to the belief that its intentions were uniformly beneficent, and therefore it could not have intended to open the door to such absurdities and calami-ties. We must conclude from its words and connections that the resolution had its origin in an abundant caution againet a violation of the Constitution, but was never in-tended to confine the action of the courts within any limits less axtended than the Constitution, not to give a construc-tion to the Constitution. It was not meant to prevent the tion to the Constitution. It was not meant to prevent the full and just action of all the proper and legitimate powers of the Government for the punishment of treason. Nor has our construction any color of undue harshness and severity

toward the great criminals who have brought upon the country this terrible civil war. In all Christian or civilized nations on the earth, the punishment of treason has been a total forfeiture both of estate and life, and justly so, for it is not in its consequences simply murder, but wholesale murder with all the other crimes of the decalogue added thereto.

And how absurd is the idea that we may take the life of a traitor but may not take that which is less valuable than life, to wit: his property? Does not the greater always include the less?

By this court, then, looking for light and guidance to the tribunal of Eternal Justice, it must beheld that enough loyal blood, and widows' and orphans' tears, have been poured out and mingled with the soil of Virginia to extinguish all rebel rights and establish a perfect title in the Government, for the benefit of the heirs of those who have fallen defending the republican institutions received from our patriot fathers-institutions which we are bound to preserve and to transmit unimpaired and, if possiblo, im-

proved and purified to our posterity. The point raised by the respondent's counsel against the constitutionality of the law under which this suit is brought, constitutionality of the law under which this suit is brought, because it does not provide a jury trial in any case, is one of very grave import. In the judgment of this court it would be much more in the spirit, not only of the Consti-tution, but of the great charter ratified at Runnymede, if a respondent, on putting in his answer and taking the oath of allegiance to the Government, was permitted to demand a trial by jury. But the decisions of the Suprema Court of the United

a trial by jury. But the decisions of the Supreme Court of the United States in the Dred Scott case, and other cases under the fugitive slave laws of Congress, deprived men of liberty, which is decarer than property, without trial by jury, will not permit this court, especially where no cath of allegiance is offered by the respondent, to declare the law unconstitu-tional on this ground. It is to be hoped, however, that all uncertainty arising from different constructions of the courts may be speedily removed by the action of the Supreme Court of the United States, or by the explanatory legislation of Congress now

States, or by the explanatory legislation of Congress now about to be assembled.

The decree of the court in this case is, that all right, title, interest and estate of the said Hugh Latham, in and to the property mentioned in the libel in this case, be forfeited and confiscated to the United States, and that the marshal of the court proceed to sell said right, title, interest and estate at public auction for cash to the highest bidder, after having at public auction for cash to the highest budder, after having given ten days' notice of the time, place, and terms of sale of the personal property, and at least twenty days' notice of the time, place, and terms of sale of the real estate, by advertisement in one or more newspapers printed and pub-hished in the cities of Alexandria and Washington. That and anshal deliver the personal property to the purchaser, and also make and deliver a deed of the real estate to the purchaser on the payment of the purchase money therefor, and that the marshal bring the proceeds of said sale into this court to await the further order thereof.

#### MILITARY ORDERS.\*

#### CONFISCATION OF PROPERTY.

The following important military order has been issued by Major General Wallace:

HEADQUARTERS MIDDLE DEPARTMENT, STH ARMY CORPS, BALTIMORE, MD., April 26, 1864.

[General Orders No. 30.] Many citizens of this department have gone voluntarily into the States in rebellion against the United States, some to join the rebel army, others to aid and encourage the rebellion by their presence and otherwise, who have left prop-erty in real estate, slaves, stocks of various descriptions, and other securities for money in this department. And many citizens of the States in rebellion, who have participated in and encouraged that in rovement, have similar property within this department. It is deemed important that such property should not be under the control of such persons, and hable to be used in whole or in part in the support of the rebellion, and against the interests of the United States

It is therefore hereby ordered that the proceeds of all real estate, the hire of all slaves, the interest on all debts due from persons in this department, the current interest on all private debts, the dividends and interest on all stocks and bonds of railroad companies, banks, turupike need computing manufacturing companies, and arbitic conroad companies, manufacturing companies, and public corporations, howsoever declared and payable, which are the property of the persons above described, and are within this department, shall be withheld by the persons author-ized, and whose duty it is to pay the same, from such per-sons, their representatives, agents, and attorneys, howso-

\* The President has suspended the execution of these orders. Moneys paid under them are ordered repaid.

ever constituted; and that the same shall be paid over to Lieutenant Colonel Alexander Bliss, Quartermaster of this department, or such other agent as the general commanding may authorize and appoint from time to time. All persons having authority over such property will be held re

sons having automicy over such propercy will be left as sponsible for such sums as may be paid in violation of this order, and be otherwise punished by military commission. The hire and the proceeds of the labor of such slaves a are in the counties of Maryland in this department, and belonging to the persons above described, will in due proportion be set apart and reserved for the use of such sla when they shall have been freed by the constitutional law of Maryland, as it is hoped they soon will be. By command of Major General Wallace : SAMUEL B. LAWRENCE, Asst. Adj. Gen.

Official: JAS. R. Ross, A. D. C.

#### HEADQUARTERS MIDDLE DEPARTMENT, 8TH ARMY COEPS, BALTIMORE, MD., May 1, 1864.

[General Orders No. 33.] In order more effectually to carry out the provisions of General Orders No. 30, from these headquarters, it is

ordered

1st-That the President and directors, or other authorized agonts and representatives of all banks, insurance com-panies, railroad, turnpike, ferry, and manufacturing com-panies, and all other monied corporations, institutions, and joint stock companies whatsoever within this military de-Joint stock companies whatsoever while this initial, up partment, shall, with the least possible delay, and not later than the 15th day of June next, forward to the office of Lieut. Col. Alexander Bliss, Quartermaster of this department, a written statement verified under oath by the president, and by the secretary or treasury of such corporation or company, &c., setting forth the names and address of all the proprietors or stockholders, or others having now or within the past year, any interest whatsoever in the capital stock, the bonds or other debt, funded or otherwise of such corporation or company, or in the dividends, interest, pre-miums or other profits whatscever arising therefrom or from its business, who are, or who since April 19th, 1861, have been residents of, or have hved within any of the States now in rebellion, or who now are, or who have been in the rebel army, or in the employment of the rebel government, to the best knowledge and belief of the deponents—the exact amount and nature of the share or other interest of exact amount and nature of the share or other interest of every such person, the date of the commencement of such interest, or of any increase thereof; also, the dates and amounts of all payments of dividends, interest, premiums or other profits by said company since May 1st, 1563, to any such person, or to any one whatsoever on account of any such persons, and the names of those to whom paid. Also the name of any of the said persons to whom any in-terests or profits are now payable, and the amounts and date when due. dates when due.

2d.-It is ordered that all corporations, joint stock com-panies, and all individuals within this department, who now owe, or who since May 1st, 1863, have owed any inter-est upon any mortgage, bond or note, or other security, or who since May let, 1863, have paid any interest upon any mortgage, bond, note, or other security, or any other interest, or profit whatsoever to any person of any other interest, or profit whatsoever to any person of any of the classes enumerated in the preceding paragraph, (or to any agent, attorney, or representative of any of the said persons,) shall, with the least possible delay, and not later than the 15th day of June, 1864, forward to the office of Lientenant Colonel Alexander Bliss, Quartermaster of the Middle department, a written statement verified under oath setting forth the names and address of all persons to whom any such interesthas been risclues of a periods to whom any such interesthas been risclues, or to whom paid, the amount thereof, the amount of the principal upon which it is due or has been paid, the nature of the debt, whether bond, (secured or not by mortgage), or note, or other security or evidence of debt whatsoever, and the date of its maturing. of its maturity.

3d. It is ordered that all corporations and joint stock companies, and all individuals within this department who now occupy or enjoy the use of, or since May 1st, 1863, have occupied or enjoyed the use of any lands, tenements,

buildings, or other real estate whatsoever owned either in whole or in part by, or of which the rent or other proceeds in any manner accrue to, any of the persons of any of the classes above enumorated in paragraph 1 of this Order, shall, with the least possible delay, and not later than the 15th day of June next, forward to the office of Lieutenant Colonel Alexander Bliss, Quartermastor of the Middle department, a written statement, verified under oath, seiting forth their names and address, the location and description of any such lands, buildings, or other real estate, the names of the own-ers, lessees, &c., thereof, the rate of the rent or hire thereof, and any amount now due on account of the rent or hire thereof, or which have since May 1st, 1863, been due, or which have since that date been paid to any person what-soever, and the names and address of all persons to whom said amounts are or have been due or paid, and the date when due or paid.

4th. It is ordered that all persons in the counties of Ma-4th. It is ordered that all persons in the countes of ma-ryland within this department now hiring, or who since January 1st, 1864, have hired slaves, belonging wholly or in part, now or within the past year, to any of the persons of the class enumerated in paragraph 1 of this Order, shall furnish to Lieutenant Colonel Alexander Bliss, Quarter-master of the Middle department, a written statement, ver-mater of the Middle department, a written statement, verified under oath, of their names and address, of the number of such slaves hired by them, the rate of hire, the names of the owners of each, the name of each slave, and the names of of the persons to whom any payments have been since Jan-uary 1st, 1864, made, or are now due on account of such hire, the amounts of such payments, and the amounts now due.

And all slaves are required to furnish themselves such of the above information as is in their power, by appearing personally at the above place for that purpose, or otherwise, as they may be able.

By command of Major General Wallace:

SAMUEL B. LAWRENCE, Asst. Adj. Gen. Official: MAX WOODHULL, A. D. C.

### THE PRESIDENT'S PROCLAMATION UNDER THE CON-FISCATION LAW.

In pursuance of the sixth section of the act of Congress entitled "An act to suppress insurrection, to punish treason and rebellion, to seize and confiscate the property of rebels, and for other purposes," approved July 17, 1862, and which act, and the joint resolution explanatory thereof, are herewith published, I, ABRAHAM LIN-COLN, President of the United States, do hereby proclaim to and warn all persons within the contemplation of said sixth section to cease participating in, aiding, countenancing, or abetting the existing rebellion, or any rebellion, against the Government of the United States, and to return to their proper allegiance to the United States, on pain of the forfeitures and seizures

as within and by said sixth section provided. In testimony whereof, I have hereunto set my hand and caused the seal of the United States to be affixed.

Done at the city of Washington, this twentyfifth day of July, in the year of our Lord one thousand eight hundred and sixty-two, and of the Independence of the United States the eighty-seventh.

### ABRAHAM LINCOLN.

By the President:

WM. H. SEWARD, Secretary of State.

# EMANCIPATION.

# Compensated Emancipation. Second Session, Thirty-Sixth Congress. IN HOUSE.

Feb. 11, 1861-Mr. JAMES B. MCKEAN, of New York, introduced into the House the folowing resolution, which-Mr. BURNETT, of Kentucky, indicating a desire to discuss itwas laid on the table under the rule, and was not again considered :

not again considered: Whereas the "Gulf States" have assumed to secede from the Union, and it is deemed important to prevent the "bor-terslave States" from following their example; and whereas t is believed that those who are inflexibly opposed to any molve, a sacrifice of principle or the extension of slavery, yould nevertheless cheerfully concur in any lawful measure or the emancipation of the slaves: Therefore, *Resolved*, That the select committee of five be instructed o inquire whether, by the consent of the people, or of the State governments, or by compensating the slaveholders, it be practicable for the General Government to procure the mancipation of the slaves in some, or all, of the "border States;" and if so, to report a bill for that purpose.

Second Session, Thirty-Seventh Congress. THE PRESIDENT'S RECOMMENDATION TO CONGRESS.

March 6, 1862-The PRESIDENT sent the fol-

lowing message to Congress :

Fellow-Citizens of the Senate

and House of Representatives :

I recommend the adoption of a joint resolution by your honorable bodies, which shall be substantially as follows :

Resolved, That the United States ought to co-operate with any State which may adopt gradual abolishment of slavery, giving to such State pecuniary aid, to be used by such State in its discretion, to compensate for the incon-reniences, public and private, produced by such change of aretem. system.

If the proposition contained in the resolution loes not meet the approval of Congress and the country, there is the end; but if it does command such approval, I deem it of importance that the States and people immediately interested should be at once distinctly notified of the fact, so that they may begin to consider The Federal whether to accept or reject it. Government would find its highest interest in such a measure, as one of the most efficient means of self-preservation. The leaders of the existing insurrection entertain the hope that this Government will ultimately be forced to acknowledge the independence of some part of the disaffected region, and that all the slave States north of such part will then say, "the Union for which we have struggled being already gone, we now choose to go with the southern section." To deprive them of this hope, substantially ends the rebellion; and the initiation of emancipation completely deprives them of it as to all the States initiating it. The point is not that all the States tolerating slavery would very soon, if at all, initiate emancipation; but that, while the offer is equally made to all, the more northern shall, by such initiation, make it certain to the more southern that in no

event will the former over join the latter in their proposed confederacy. I say "initiation," because, in my judgment, gradual, and not sudden emancipation, is better for all. In the mere financial or pecuniary view, any member of Congress, with the census tables and Treasury reports before him, can readily see for himself how very soon the current expenditures of this war would purchase, at fair valuation, all the slaves in any named State. Such a proposition on the part of the General Government sets up no claim of a right by Federal authority to inrerfere with slavery within State limits, referting, as it does the absolute control of the subject in each case to the State and its people im-mediately interested. It is proposed as a matter of perfectly free choice with them. In the annual message last December, I

thought fit to say, "the Union must be preserved; and hence all indispensable means must be employed." I said this not hastily, but deliberately. War has been made, and continues to be an indispensable means to this end. A practical reacknowledgment of the national authority would render the war unnecessary, and it would at once cease. If, however, resistance continues, the war must also continue; and it is impossible to foresee all the incidents which may attend, and all the ruin which may follow it. Such as may seem indispensable, or may obviously promise great efficiency toward end-

ing the struggle, must and will come. The proposition now made, though an offer only, I hope it may be esteemed no offence to ask whether the pecuniary consideration tendered would not be of more value to the States and private persons concerned, than are the institution, and property in it, in the present aspect of affairs?

While it is true that the adoption of the proposed resolution would be merely initiatory, and not within itself a practical measure, it is recommended in the hope that it would soon lead to important practical results. In full view of my great responsibility to my God and to my country, I earnestly beg the attention of Congress and the people to the subject.

ABRAHAM LINCOLN.

March 6, 1862.

#### IN HOUSE.

March 10-Mr. Roscoe Conkling, of New York, introduced under a suspension of the rules, which was carried-yeas 86, nays 35, this joint resolution :

Resolved, dc., That the United States ought to co-operate with any State which may adopt gradual abolishment of sla-very, giving to such State pecuniary aid, to be used by such State in its discretion, to compensate for the inconveniences, public and private, produced by such change of system.

March 11-The resolution passed the House of Representatives-yeas 97, nays 36, as follows:

YEAS-Messrs. Aldrich, Arnold, Ashley, Babbitt, Baker,

April 2-The resolution passed the Senateyeas 32, nays 10, as follows.

YEAS-Messrs. Anthony, Browning, Chandler, Clark, Collamer, Davis, Dixon, Doolittle, Fessenden, Foot, Foster, Grimes, Hale, Harlan, Henderson, Howard, Howe, King, Lane of Indiana, Lane of Kansan, Morrill, Pomeroy, Sher-man, Sumner, Ten Eyck, Thomson, Trumbull, Wade, Wil-kinson, Willey, Wilmot, Wilson of Massachusetts-32. NAYS-Messrs. Bayard, Carlile, Kennedy, Latham, Nesmith, Powell, Saulsbury, Stark, Wilson, of Missouri, Wright-10.

### MEMORANDUM OF AN INTERVIEW BETWEEN THE PRESIDENT AND SOME BORDER SLAVE STATE REPRESENTATIVES, MARCH 10, 1862, BY HON. J. W. CRISFIELD.

"DEAR SIR: I called, at the request of the President, to ask you to come to the White House to-morrow morning, at nine o'clock, and bring such of your colleagues as are in town."

#### WASHINGTON, March 10, 1862.

Yesterday on my return from church I found Mr. Postmaster General Blair in my room, writing the above note, which he immediately suspended, and verbally communicated the President's invitation; and stated that the President's purpose was to have some conversation with the delegations of Kentucky, Missouri, Maryland, Virginia, and Delaware, in explanation of his message of the 6th inst.

This morning these delegations, or such of them as were in town, assembled at the White House at the appointed time, and after some little delay were admitted to an audience. Mr. Leary and myself were the only members from Maryland present, and, I think, were the only members of the delegation at that time in the I know that Mr. Pearce, of the Senate, city. and Messrs. Webster and Calvert, of the House, were absent.

After the usual salutations and we were seated, the President said, in substance, that he had invited us to meet him to have some conversation with us in explanation of his message of the 6th; that since he had sent it in several of the gentlemen then present had visited him, but had avoided any allusion to the message, and he therefore inferred that the import of the message had been misunderstood, and was regarded as inimical to the interests we represented; and he had resolved he would talk with us, and disabuse our minds of that erroneous opinion.

The President then disclaimed any intent to injure the interests or wound the sensibilities of the slave States. On the contrary, his pur-

Baxter, Beaman, Bingham, Francis P. Blair, Jacob B. Blair, Samuel S. Blair, Blake, William G. Brown, Buffinton, Camp-bell, Chamberlain, Clark, Clements, Cobb, Colfax, Frederick, A. Conkling, Roscoe Conkling, Conway, Covode, Cutter, Davis, Delano, Diven, Duell, Dunn, Edgerton, Edwards, Eliot, Ely, Fessenden, Fisher, Franchot, Frank, Gooch, Goodwin, Hurton, Hutchins, Julian, Kelley, Francis W. Kellogg, Wil-liam Kellogg, Killinger, Lansing, Lehman, Loomis, Lovejoy, McKnight, McPherson, Mitchell, Moorhead, Anson P. Nor-rill, Justin S. Morrill, Nixon, Olin, Patton, Timothy G. Phelps, Pike, Pomeroy, Porter, Alexander H. Rice, John H. Rice, Riddle, Edward II. Rolins, Sargent, Shanks, Skeffeld, Shellabarger, Sloan, Stratton, B. F. Thomas, Trimble, Trow-bridge, Van Valkenburgh, Verree, Wall, Wallace, Charles W. Walton, E. P. Walton, Whaley, Wheeler, Albert S. White, Wilson, Windom, Worcester-97.
NAXS-Messrs. Ancona, Joseph Baily, Biddle, Calvert, Corr-ing, Cox, Gravens, Crisfield, Crittenian, Dunlap, English, Gri-sont, Ward, Chiltom B. Steele, Francis Thomas, Fornees, Wads-worth, Ward, Chiltom A. White, Wickliff, Wood, Woodrugf-26. April 2-The resolution passed the Senatetheir slaves induced to abscond and protected within the lines; these complaints were numerous, loud, and deep; were a serious annoyance to him and embarrassing to the progress of the war; that it kept alive a spirit hostile to the Government in the States we represented strengthened the hopes of the confederates tha at some day the border States would unite with them, and thus tend to prolong the war; and he was of opinion, if this resolution should be adopted by Congress and accepted by our States these causes of irritation and these hopes would be removed, and more would be accomplished towards shortening the war than could be hoped from the greatest victory achieved by Unior armies; that he made this proposition in good faith, and desired it to be accepted, if at all voluntarily, and in the same patriotic spirit in which it was made; that emancipation was a subject exclusively under the control of the States, and must be adopted or rejected by each for itself; that he did not claim nor had this Government any right to coerce them for that purpose; that such was no part of his purpose in making this proposition, and he wished it to be clearly understood; that he did not expect us there to be prepared to give him an answer, but he hoped we would take the subject into serious consideration; confer with one another. and then take such course as we felt our duty and the interests of our constituents required of us.

> Mr. NOELL, of Missouri, said that in his State slavery was not considered a permanent institution; that natural causes were there in operation which would, at no distant day, extinguish it, and he did not think that this prop-osition was necessary for that; and, besides that, he and his friends felt solicitous as to the message on account of the different constructions which the resolution and message had received. The New York Tribune was for it, and understood it to mean that we must accept gradual emancipation according to the plan suggested, or get something worse.

> The PRESIDENT replied, he must not be expected to quarrel with the New York Tribune before the right time; he hoped never to have to do it; he would not anticipate events. In respect to emancipation in Missouri, he said that what had been observed by Mr. Noell was probably true, but the operation of these natural causes had not prevented the irritating

conduct to which he had referred, or destroyed the hopes of the Confederates that Missouri would at some time range herself alongside of them, which, in his judgment, the passage of this resolution by Congress and its acceptance by Missouri would accomplish.

Mr. CRISFIELD, of Maryland, asked what would be the effect of the refusal of the State to accept this proposal, and desired to know if the President looked to any policy beyond the acceptance or rejection of this scheme.

The PRESIDENT replied that he had no designs beyond the action of the States on this particular subject. He should lament their refusal to accept it, but he had no designs beyond their refusal of it.

Mr. MENZIES, of Kentucky, inquired if the President thought there was any power except in the States themselves to carry out his scheme of emancipation.

The PRESIDENT replied, he thought there could not be. He then went off into a course of remark not qualifying the foregoing declaration nor material to be repeated to a just understanding of his meaning.

Mr. CRISFIELD said he did not think the people of Maryland looked upon slavery as a permanent institution; and he did not know that they would be very reluctant to give it up if provision was made to meet the loss, and they could be rid of the race; but they did not like to be coerced into emancipation, either by the direct action of the Government or by indirection, as through the emancipation of slaves in this District, or the confiscation of southern property as now threatened; and he thought before they would consent to consider this proposition they would require to be informed on these points.

The PRESIDENT replied that, "unless he was expelled by the act of God or the Confederate armies, he should occupy that house for three years, and as long as he remained there Maryland had nothing to fear, either for her institutions or her interests, on the points referred to."

Mr. CRISFIELD immediately added: "Mr. President, if what you now say could be heard by the people of Maryland they would consider your proposition with a much better feeling than I fear without it they will be inclined to do."

The PRESIDENT. "That (meaning a publication of what he said) will not do; it would force me into a quarrel before the proper time;" and, again intimating, as he had before done, that a quarrel with the "Greeley faction" was impending, he said "he did not wish to encounter it before the proper time, nor at all if it could be avoided."

Governor WICKLIFFE, of Ky., then asked him respecting the constitutionality of his scheme.

The PRESIDENT replied: "As you may suppose, I have considered that; and the proposition now submitted does not encounter any constitutional difficulty. It proposes simply to co-operate with any State by giving such State pecuniary aid; and he thought that the resolution, as proposed by him, would be considered rather as the expression of a sentiment than as involving any constitutional question." Mr. HALL, of Mo., thought that if this proposition was adopted at all it should be by the votes of the free States, and come as a proposition from them to the slave States, affording them an inducement to put aside this subject of discord; that it ought not to be expected that members representing slaveholding constituencies should declare at once, and in advance of any proposition to them, for the emancipation of slavery.

The PRESIDENT said he saw and felt the force of the objection; it was a fearful responsibility, and every gentleman must do as he thought best; that he did not know how this scheme was received by the members from the free States; some of them had spoken to him and received it kindly; but for the most part they were as reserved and chary as we had been, and he could not tell how they would vote. And in reply to some expression of Mr. Hall as to his own opinion regarding slavery, he said he did not pretend to disguise his antislavery feeling; that he thought it was wrong and should continue to think so; but that was not the question we had to deal with now. Slavery existed, and that, too, as well by the act of the North as of the South; and in any scheme to get rid of it, the North, as well as the South, was morally bound to do its full and equal share. He thought the institution wrong, and ought never to have existed; but yet he recognized the rights of property which had grown out of it, and would respect those rights as fully as similar rights in any other property ; that property can exist, and does legally exist. He thought such a law wrong, but the rights of property resulting must be respected; he would get rid of the odions law, not by violating the right, but by encouraging the proposition and offering inducements to give it up.

Here the interview, so far as this subject is concerned, terminated by Mr. Crittenden's assuring the President that, whatever might be our final action, we all thought him solely moved by a high patriotism and sincere devotion to the happiness and glory of his country; and with that conviction we should consider respectfully the important suggestions he had made.

After some conversation on the current war news, we retired, and I immediately proceeded to my room and wrote out this paper.

J. W. CRISFIELD.

We were present at the interview described in the foregoing paper of Mr. CRISFIELD, and we certify that the substance of what passed on the occasion is in this paper faithfully and fally given.

J. W. MENZIES, J. J. CRITTENDEN, R. MALLORY.

March 10, 1862.

COMPENSATED EMANCIPATION IN THE DISTRICT OF COLUMBIA.

April 16—A bill was passed liberating all persons of African descent held to service or labor within the District of Columbia, and prohibiting slavery or involuntary servitude in the

District except as a punishment for crime. It provided for a commission to appraise the valuation of the slaves liberated, but limited their allowance in the aggregate to an amount equal to \$300 per slave; and appropriated \$1,000,000 to pay loyal owners, and \$100,000 to colonize such of the slaves as desired to emigrate to Hayti or Liberia, this expenditure not to exceed \$100 for each person emigrating.\*

\* Pending the civil appropriation bill in the Senate and House, June 25 and 29, 1864, a section was adopted without a division, repealing this section, and also that part of the first section of the civil appropriation bill of July 16, 1862, which appropriated \$500,000 to colonize emancipated slaves of the District of Columbia, to be repaid to the Treasury out of the proceeds of confiscated property. This report from the Secretary of the Interior, March 7,

1864, to the President, gives a statement of the results of the experiment:

DEPARTMENT OF THE INTERIOR.

WASHINGTON, D. C., March 7, 1864. SIR: I have the honor to return herewith the letter of Sis: I have the honor to return herewith the letter of Hon. Benjamin F. Wade, chairman of the Committee on Territories in the Senate, requesting to be informed how much of the fund appropriated by the Thirty-Seventh Con-gress for colonization purposes has been paid out, for what purpose, and to whom, which you referred to this Depart-ment for report, and to state that as the accounts of all the persons to phone memory was advanced for colonization purp persons to whom money was advanced for colonization purposes have not yet been finally settled at the Treasury, I am not able, at the present time, to state the exact amount actually expended. The total amount drawn from the Treasury up to the present time, however, is S33,226 97.

It was paid or advanced to the following named persons, and for the objects herein stated:

Amount paid to John D. Defrees, Superintendent

- of Public Printing, for binding in paper 5,000 copies "White and African Races"..... Amount paid Green & Williams for six cane-seat chairs..... -----

- Amount advanced Hon. S. C. Pomeroy, special agent for colonization in Central America..... Amount paid Watt J. Smith for traveling ex-penses to New York and back... Amount paid John P. Usher for traveling ex-penses to New York and back, and for advance made to agent sent to the East to obtain in-formation in relation to advance formation in relation to colonization .
- Amount paid James Mitchel, for clerical services of W. B. Smith and Thomas R. Smith; for services of Rev. A. Bemar, (colored minister;) and for advance to J. E. Williams, (colored,) agent
- of A. W. Thompson. Amount advanced Rev. R. R. Gurley, correspond-ing secretary of the American Colonization Society, to aid the Rev. Chancey Leonard, pastor of the First Baptist Church in Washington city,
- of the First Baptist Church in Washington city, in visiting Liberia, with reference to the estab-ment of a colony on the St. John's river....... Amount paid Augustus A. Smith, as clerk...... Amount paid J. Mitchell, for office furniture, and for services of J. W. Menard, in emigration office Amount paid J. W. Menard, for services as clerk in emigration office
- in emigration office..... Amount paid Aug. A. Smith, for services as clerk Amount paid J. W. Fitzhugh, for furniture for
- emigration office ... Amount paid James Mitchell to 31st December,
- ing purchased and sent to colonists at lie à Vache ..... .....
- Amount paid to Hurxthal & Barnum, for clothing purchased and sent to colonists at Ile & Vache.
- Amount paid American Colonization Society, for passage engaged for J. W. Menard to Liberia... Amount paid American Colonization Society, for passage, &c., to Liberia of three colored emi-grants from the District of Columbia...... Amount paid Wilson & Cammann, for draft of D. C. Donnohue, special agent...... Amount paid Hallet Kilbourn, for traveling ex-monent on Nary Novik on smooth arms of the second
- penses to New York as special agent to pur-chase clothing for the relief of the colonists at Ile à Vache.....

April 3-The bill passed the Senate-yeas 29, nays 14; as follows:

YEAS-Messrs. Anthony, Browning, Chandler, Clark, Col-lamer, Dixon, Doolittle, Fessenden, Foot, Foster, Grimes, Hale, Harlan, Harris, Howard, Howe, King, Lane of Indi-ana, Lane of Kansas, Morrill, Pomeroy, Sherman, Summer, Ten Eyck, Trumbull, Wade, Wilkinson, Wilmot, Wilson of Massochusetts-20

Massachusetts-29. NATS-Messrs. Bayard, Carlile, Davis, Henderson, Ken-nedy, Latham, McDougall, Nesmith, Powell, Saulsbury, Stark, Willey, Wilson of Missouri, Wright-14.

April 11-It passed the House-yeas 92, nays 39, as follows :

39, as follows:
YEAS-Messrs. Aldrich, Alley, Arnold, Ashley, Babbitt, Baker, Baxter, Beaman, Bingham, Francis P. Blair, Samuel S. Blair, Slake, George H. Browne, Bufinton, Campbell, Chamberlain, Clark, Colfax, Frederick A. Conkling, Roscoe Conkling, Covode, Davis, Dawes, Delano, Diven, Duell, Dunn, Edgerton, Edwards, Eliot, English, Fenton, Fessenden, Fisher, Franchot, Frank, Gooch, Goodwin, Granger, Haight, Hale, Hanchett, Harrison, Hickman, Hooper, Hutchins, Julian, Kelley, Francis W. Kellogg, Killinger, Lansing, Joomis, Lovejoy, McKnight, McPherson, Mitchell, Moorhead, Anson P. Morrill, Justin S. Morrill, Nixon, Odell, Olin, Patton, Pike, Porter, Potter, Alexander H. Rice, John H. Rice, Riddle, Edward H. Rollins, Sargent, Sedgwick, Shanks, Shefjield, Shellabarger, Sloan, Spaulding, Stevens, Stratton, Veneeler, Albert S. White, Wilson, Windom-92. Nars-Messrs. Allen, Joseph Baily, Biddle, Jacob B. Blair, William G. Brown, Casey, Crittenden, Delaplaine, Dunla, Jories, Wadsworth, Ward, Chilton A. White, Wickliffe, Wright - 29.
April 16-The PRESIPENT sent to Congress

April 16-The PRESIDENT sent to Congress this message :

12 00

\$127 50

750 00

Fellow-Citizens of the Senate and House of Representatives :

The act entitled "An act for the release of 25,000 00 certain persons held to service or labor in the 52 40 District of Columbia," has this day been approved and signed.

I have never doubted the constitutional au-152 40thority of Congress to abolish slavery in this District ; and I have ever desired to see the national capital freed from the institution in some satisfactory way. Hence there has never been 112 00 in my mind any question upon the subject except the one of expediency, arising in view of all the circumstances. If there be matters within and about this act which might have 200 00 taken a course or shape more satisfactory to my 18 33 judgment, I do not attempt to specify them. 60 00 am gratified that the two principles of compensation and colonization are both recognized and 50 00 practically applied in the act. In the matter 100 00 of compensation, it is provided that claims may 326 15 be presented within ninety days from the passage of the act, " but not thereafter ;" and there 2,838 46

It is proper that I should state that the claim of Messre. Forbes & Tuckerman, for trausporting a number of emi-grants to Ile à Vache, is not yet liquidated, and that the colonization fund is also liable for the expenses which have been incurred by the special agent sent by the Government to Ile à Vache to look into and report upon the condition of the colonists there. The agent has not yet returned, and consequently the expenses incurred by him cannot now be ascertained. It is also liable for the expenses of the restel sent on the ordingt to the Iluited 300 00 900 00 1,786 33 95 00 the vessel sent out to return the colonists to the United 285 00 States.

#### IN HOUSE.

April 5, 1864-Mr. Cox asked leave to offer this resolution,

hut it was objected to: Resolved, That the Secretary of State communicate to this House all correspondence between our consul at Aux Cayes and the State Department in regard to our colony of blacks at the Isle & Vache.

Making in all.....

is no saving for minors, femmes covert, insane, or absent persons. I presume this is an omission by mere oversight, and I recommend that it be supplied by an amendatory or supplemental act. ABRAHAM LINCOLN.

An amendatory bill, meeting the President's suggestions, was passed in the House-yeas 69, nays 36; and in the Senate-yeas 28, nays 6, and approved by him.

### COMPENSATED EMANCIPATION IN THE BORDER STATES.

IN HOUSE.

1862, April 7-Mr. ALBERT S. WHITE, of Indiana, offered a resolution for the appointment of a select committee of nine members, to report whether any plan can be proposed and recommended for the gradual emancipation of all the African slaves, and the extinction of slavery in Delaware, Maryland, Virginia, Kentucky, Tennessee, and Missouri, by the people or local authorities thereof, and how far and in what way the Government of the United States can and ought equitably to aid in facilitating either of the above objects.

Mr. MALLORY moved to lay the resolution on the table, which was disagreed to-yeas 51, nays 68; and the resolution was adopted-yeas 67, nays 52, as follows:

nays 52, as follows: YEAS--Messrs. Alley, Arnold, Ashley, Babbitt, Baker, Baxter, Beaman, Bingham, Francis P. Hair, Blake, Buffin-ton, Campbell, Chamberlin, Clark, Colfax, Frederick A. Conkling, Davis, Dawes, Duell, Edgerton, Eliot, Fenton, Fessenden, Gurley, Hanchett, Hickman, Hutchins, Julian, Kelley, Francis W. Kellogg, Lansing, Loomis, Lovejoy, Mo-fnight, McPherson, Moorhead, Anson P. Morrill, Justin S. Morrill, Nixon, Olin, Pike, Pomeroy, Potter, Alexander H. Rice, John H. Rice, Riddle, Edward H. Rollins, Sargent, Stratton, Train, Van Horn, Van Valkenburgh, Verree, Wal-Lee, Charles W. Walton, E. P. Walton, Weshburne, Wheeler, Albert S. White, Wilson, Windom-off. Navs-Messrs. Allen, Biddle, Jacob B. Blair, George H. Browne, William G. Brown, Calvert, Casey, Cobb, Corning, Cox, Cravens, Crittenden, Delano, Delaplaine, Diven, Duraloy, Conglish, Fisher, Critter, Haight, Harding, Harrison, Horton, Kergigan, Kangp, Law, Lazear, Leary, Lehman, Mallory, Witchell Witchell, Neich Neuto, Naven Mender, Mallory,

Bagitsi, Fisher, Grace, Judge, Jarding, Hartong, Harton, Molory, Kerrigan, Knapp, Law, Lazear, Leary, Lehman, Mallory, Menzies, Mitchell, Noble, Noell, Norton, Nugen, Penaleton, Perry, Timothy G. Phelps, Price, Richardson, Shiel, Smith, John B. Steele, William G. Steele, Vallandigham, Yoorhees, Wadsworth, Ward, Webster, Chilton A. White, Wicklifte–52.

April 14-The SPEAKER appointed the committee as follows : Albert S. White of Indiana, Francis P. Blair of Missouri, George P. Fisher of Delaware, William E. Lehman of Pennsylvania, C. L. L. Leary of Maryland, K. V. Whaley of Virginia, James F. Wilson of Iowa, Samuel L. Casey of Kentucky, and Andrew J. Clements of Tennessee.

In July, 1862, the President requested and obtained an interview with the border State Congressmen, the result of which is contained in this statement:

#### THE PRESIDENT'S APPEAL TO THE BORDER STATES.

The Representatives and Senators of the border slaveholding States having, by special invitation of the President, been convened at the Executive Mansion, on Saturday morning last, (July 12,) Mr. Lincoln addressed them as follows from a written paper held in his hand:

"GENTLEMEN: After the adjournment of Congress, now near, I shall have no opportunity of seeing you for several months. Believing

that you of the border States hold more power for good than any other equal number of members, I feel it a duty which I cannot justifiably waive, to make this appeal to you.

"I intend no reproach or complaint when I assure you that, in my opinion, if you all had voted for the resolution in the gradual emancipation message of last March, the war would now be substantially ended. And the plan therein proposed is yet one of the most potent and swift means of ending it. Let the States which are in rebellion see definitely and cer-tainly that in no event will the States you represent ever join their proposed Confederacy, and they cannot much longer maintain the contest. But you cannot divest them of their hope to ultimately have you with them so long as you show a determination to perpetuate the institution within your own States. Beat them at elections, as you have overwhelmingly done, and, nothing daunted, they still claim you as their own. You and I know what the lever of their power is. Break that lever before their faces, and they can shake you no more forever.

"Most of you have treated me with kindness and consideration, and I trust you will not now think I improperly touch what is exclusively your own, when, for the sake of the whole country, I ask, 'Can you, for your States, do better than to take the course I urge?' Discarding *punctilio* and maxims adapted to more manageable times, and looking only to the unprecedentedly stern facts of our case, can you do better in any possible event? You prefer that the constitutional relations of the States to the nation shall be practically restored without disturbance of the institution; and, if this were done, my whole duty, in this respect, under the Constitution and my oath of office, would be performed. But it is not done, and we are trying to accomplish it by war. The incidents of the war cannot be If the war continues long, as it must, avoided. if the object be not sooner attained, the institution in your States will be extinguished by mere friction and abrasion-by the mere incidents of the war. It will be gone, and you will have nothing valuable in lieu of it. Much of its value is gone already. How much better for you and for your people to take the step which at once shortens the war and secures substantial compensation for that which is sure to be wholly lost in any other event! How much better to thus save the money which else we sink forever in the war! How much better to do it while we can, lest the war ere long render us pecuniarily unable to do it! How much better for you, as seller, and the nation, as buyer, to sell out and buy out that without which the war could never have been, than to sink both the thing to be sold and the price of it in cutting one another's throats !

"I do not speak of emancipation at once, but of a decision at once to emancipate gradually. Room in South America for colonization can be obtained cheaply and in abundance, and when numbers shall be large enough to be company and encouragement for one another, the freed people will not be so reluctant to go.

"I am pressed with a difficulty not yet men-

tioned, one which threatens division among | make to the President, and, as the result of a those who, united, are none too strong. An instance of it is known to you. General Hunter is an honest man. He was, and I hope still is, my friend. I valued him none the less for his agreeing with me in the general wish that all men everywhere could be freed. He proclaimed all men free within certain States, and I repudiated the proclamation. He expected more good and less harm from the measure than I could believe would follow. Yet, in repudiating it, I gave dissatisfaction, if not offence, to many whose support the country cannot afford to lose. And this is not the end of it. The pressure in this direction is still upon me, and is increasing. By conceding what I now ask you can relieve me, and, much more, can relieve the country in this important point.

"Upon these considerations I have again begged your attention to the message of March last. Before leaving the Capitol, consider and discuss it among yourselves. You are patriots and statesmen, and as such I pray you consider this proposition; and at the least commend it to the consideration of your States and people. As you would perpetuate popular government for the best people in the world, I bescech you that you do in nowise omit this. Our common country is in great peril, demanding the loftiest views and boldest action to bring a speedy re-Once relieved, its form of government is lief. saved to the world, its beloved history and cherished memories are vindicated, and its happy future fully assured and rendered inconceivably grand. To you, more than to any others, the privilege is given to assure that happiness and swell that grandeur, and to link your own names therewith forever."

At the conclusion of these remarks some conversation was had between the President and several members of the delegations from the border States, in which it was represented that these States could not be expected to move in so great a matter as that brought to their notice in the foregoing address while as yet the Congress had taken no step beyond the passage of a resolution, expressive rather of a sentiment than presenting a substantial and reliable basis of action.

The President acknowledged the force of this view, and admitted that the border States were entitled to expect a substantial pledge of pecuniary aid as the condition of taking into consideration a proposition so important in its relations to their social system.

It was further represented, in the conference, that the people of the border States were interested in knowing the great importance which the President attached to the policy in question, while it was equally due to the country, to the President, and to themselves, that the representatives of the border slaveholding States should publicly announce the motives under which they were called to act, and the considerations of public policy urged upon them and their constituents by the President.

With a view to such a statement of their position, the members thus addressed met in

comparison of opinions among themselves, they determined upon the adoption of a majority and minority answer.

### REPLY OF THE MAJORITY.

The following paper was yesterday sent to the President, signed by the majority of the Representatives from the border slaveholding States :-

### WASHINGTON, July 14, 1862.

To the PRESIDENT :

The undersigned, Representatives of Kentucky, Virginia, Missouri, and Maryland, in the two Houses of Congress, have listened to your address with the profound sensibility naturally inspired by the high source from which it emanates, the earnestness which marked its delivery, and the overwhelming importance of the subject of which it treats. We have given it a most respectful consideration, and now lay before you our response. We regret that want of time has not permitted us to make it more perfect.

We have not been wanting, Mr. President, in respect to you, and in devotion to the Constitution and the Union. We have not been indifferent to the great difficulties surrounding you, compared with which all former national troubles have been but as the summer cloud; and we have freely given you our sympathy and support. Repudiating the dangerous heresies of the secessionists, we believed, with you, that the war on their part is agressive and wicked, and the objects for which it was to be prosecuted on ours, defined by your message at the opening of the present Congress, to be such as all good men should approve. We have not hesitated to vote all supplies necessary to carry it on vigorously. We have voted all the men and money you have asked for, and even more ; we have imposed onerous taxes on our people, and they are paying them with cheerfulness and alacrity; we have encouraged enlistments and sent to the field many of our best men; and some of our number have offered their persons to the enemy as pledges of their sincerity and devotion to the country.

We have done all this under the most discouraging circumstances, and in the face of measures most distasteful to us and injurious to the interests we represent, and in the hearing of doctrines avowed by those who claim to be your friends, must be abhorrent to us and our constituents. But, for all this, we have never faltered, nor shall we as long as we have a Constitution to defend and a Governmeut which protects us. And we are ready for renewed efforts, and even greater sacrifices, yea, any sacrifice, when we are satisfied it is required to preserve our admirable form of government and the priceless blessings of constitutional liberty.

A few of our number voted for the resolution recommended by your message of the 6th of March last, the greater portion of us did not, and we will briefly state the prominent reasons which influenced our action.

In the first place, it proposed a radical change of our social system, and was hurried council to deliberate on the reply they should through both Houses with undue haste, with-

out reasonable time for consideration and debate, and with no time at all for consulta-tion with our constituents, whose interests it deeply involved. It seemed like an interference by this Government with a question which peculiarly and exclusively belonged to our respective States, on which they had not sought advice or solicited aid. Many of us doubted the constitutional power of this Government to make appropriations of money for the object designated, and all of us thought our finances were in no condition to bear the immense outlay which its adoption and faithful execution would impose upon the national Treasury. If we pause but a moment to think of the debt its acceptance would have entailed, we are appalled by its magnitude. The proposition was addressed to all the States, and embraced the whole number of slaves.

According to the census of 1860 there were then nearly four million slaves in the country; from natural increase they exceed that number now. At even the low average of \$300, the price fixed by the emancipation act for the slaves of this District, and greatly below their real worth, their value runs up to the enormous sum of \$1,200,000,000; and if to that we add the cost of deportation and colonization, at \$100 each, which is but a fraction more than is actually paid by the Maryland Colonization Society, we have \$400,000,000 more. We were not willing to impose a tax on our people sufficient to pay the interest on that snm, in addition to the vast and daily increasing debt already fixed upon them by the exigencies of the war, and if we had been willing, the country could not bear it. Stated in this form the proposition is nothing less than the deportation from the country of \$1,600,000,000 worth of producing labor, and the substitution in its place of an interest-bearing debt of the same amount.

But, if we are told that it was expected that only the States we represent would accept the proposition, we respectfully submit that even then it involves a sum too great for the financial ability of this Government at this time. According to the census of 1860—

Slaves.	
Kentucky had	1
Maryland	I
Virginia	
Delaware.,	
Missouri	I
Tennessee	I
	н

And we have the enormous sum of ......\$478,038,133

We did not feel that we should be justified in voting for a measure which, if carried out, would add this vast amount to our public debt at a moment when the Treasury was reeling under the enormous expenditure of the war.

Again, it seemed to us that this resolution was but the annunciation of a sentiment which could not or was not likely to be reduced to an actual tangible proposition. No movement was then made to provide and appropriate the funds required to carry it into effect; and we were not encouraged to believe that funds

would be provided. And our belief has been fully justified by subsequent events. Not to mention other circumstances, it is quite sufficient for our purpose to bring to your notice the fact that, while this resolution was under consideration in the Senate, our colleague, the Secator from Kentucky, moved an amendment appropriating \$500,000 to the object therein designated, and it was voted down with great unanimity. What confidence, then, could we reasonably feel that if we committed ourselves to the policy it proposed, our constituents would reap the fruits of the promise held out; and on what ground could we, as fair men, approach them and challenge their support?

The right to hold slaves is a right appertaining to all the States of this Union. They have the right to cherish or abolish the institution, as their tastes or their interests may prompt, and no one is authorized to question the right or limit the enjoyment. And no one has more clearly affirmed that right than you have. Your inaugural address does you great honor in this respect, and inspired the country with confidence in your fairness and respect for the law. Our States are in the enjoyment of that right. We do not feel called on to defend the institution or to affirm it is one which ought to be cherished; perhaps, if we were to make the attempt, we might find that we differ even among ourselves. It is enough for our purpose to know that it is a right; and, so knowing, we did not see why we should now be expected to yield it. We had contributed our full share to relieve the country at this terrible crisis; we had done as much as had been required of others in like circumstances; and we did not see why sacrifices should be expected of us from which others, no more loyal, were exempt. Nor could we see what good the nation would derive from it.

Such a sacrifice submitted to by us would not have strengthened the arm of this Government or weakened that of the enemy. It was not necessary as a pledge of our loyalty, for that had been manifested beyond a reasonable doubt, in every form, and at every place possible. There was not the remotest probability that the States we represent would join in the rebellion, nor is there now, or of their electing to go with the southern section in the event of a recognition of the independence of any part of the disaffected region. Our States are fixed unalterably in their resolution to adhere to and support the Union. They see no safety for themselves, and no hope for constitutional liberty but by its preservation. They will, under no circumstances, consent to its dissolution; and we do them no more than justice when we assure you that, while the war is conducted to prevent that deplorable catastrophe, they will sustain it as long as they can muster a man or command a dollar. Nor will they ever consent, in any event, to unite with the Southern Confederacy. The bitter fruits of the peculiar doctrines of that region will forever prevent them from placing their security and happiness in the custody of an association which has incorporated in its organic law the seeds of its own destruction.

We cannot admit, Mr. President, that, if we had voted for the resolution in the emancipation message of March last, the war would now be substantially ended. We are unable to see how our action in this particular has given, or could give, encouragement to the rebellion. The resolution has passed; and, if there be virtue in it, it will be quite as efficacious as if we had voted for it. We have no power to bind our States in this respect by our votes here; and, whether we had voted the one way or the other, they are in the same condition of freedom to accept or reject its provisions. No, sir, the war has not been prolonged or hindered by our action on this or any other measure. We must look for other causes for that lamented fact. We think there is not much difficulty, not much uncertainty, in pointing out others far more probable and potent in their agencies to that end.

The rebellion derives its strength from the union of all classes in the insurgent States; and while that union lasts the war will never end until they are utterly exhausted. We know that at the inception of these troubles southern society was divided, and that a large portion, perhaps a majority, were opposed to secession. Now the great mass of southern people are To discover why they are so we must united. glance at southern society, and notice the classes into which it has been divided, and which still distinguish it. They are in arms, but not for the same objects; they are moved to a common end, but by different and even inconsistent reasons. The leaders, which comprehends what was previously known as the State-rights party, and is much the lesser class. seek to break down national independence and set up State domination. With them it is a The other class is war against nationality. fighting, as it supposes, to maintain and preserve its rights of property and domestic safety, which it has been made to believe are assailed by this Government. This latter class are not disunionists per se; they are so only because they have been made to believe that this Administration is inimical to their rights, and is making war on their domestic institution. As long as these two classes act together they will never assent to a peace.

The policy, then, to be pursued, is obvious. The former class will never be reconciled, but the latter may be. Remove their apprehensions; satisfy them that no harm is intended to them and their institutions; that this Government is not making war on their rights of property, but is simply defending its legitimate authority, and they will gladly return to their allegiance as soon as the pressure of military dominion imposed by the Confederate authority is removed from them.

Twelve months ago both Houses of Congress, adopting the spirit of your message, then but recently sent in, declared with singular unanimity the objects of the war, and the country instantly bounded to your side to assist you in carrying it on. If the spirit of that resolution had been adhered to, we are confident that we should before now have seen the end of this deplorable conflict. But what have we seen ?

In both Houses of Congress we have heard doctrines subversive of the principles of the Constitution, and seen measure after measure founded in substance on those doctrines proposed and carried through which can have no other effect than to distract and divide loyal men, and exasperate and drive still further from us and their duty the people of the rebellious States. Military officers, following these bad examples, have stepped beyond the just limits of their authority in the same direction, until in several instances you have felt the necessity of interfering to arrest them. And even the passage of the resolution to which you refer has been ostentatiously proclaimed as the triumph of a principle which the people of the southern States regard as ruinous to them. The effect of these measures was foretold, and may now be seen in the indurated state of southern feel-

To these causes, Mr. President, and not to our omission to vote for the resolution recommended by you, we solemnly believe we are to attribute the terrible earnestness of those in arms against the Government and the continuance of the war. Nor do we (permit us to say, Mr. President, with all respect to you) agree that the institution of slavery is "the lever of their power," but we are of the opinion that "the lever of their power" is the apprehension that the powers of a common Government, created for common and equal protection to the interests of all, will be wielded against the institutions of the southern States.

There is one other idea in your address we feel called on to notice. After stating the fact of your repudiation of General Hunter's proclamation, you add:

"Yet, in repudlating it, I gave dissatisfaction, if not offence, to many whose support the country cannot afford to lose. And this is not the end of it. The pressure in this direction is still upon me and is increasing. By conceding what I now ask, you can relieve me, and, much more, can relieve the country, in this important point."

We have anxiously looked into this passage to discover its true import, but we are yet in painful uncertainty. How can we, by conceding what you now ask, relieve you and the country from the increasing pressure to which you refer? We will not allow ourselves to think that the proposition is, that we consent to give up slavery, to the end that the Hunter proclamation may be let loose on the southern people, for it is too well known that we would not be parties to any such measure, and we have too much respect for you to imagine you would propose it. Can it mean that by sacrificing our interest in slavery we appease the spirit that controls that pressure, cause it to be withdrawn, and rid the country of the pestilent agitation of the slavery question? We are forbidden so to think, for that spirit would not be satisfied with the liberation of 700,000 slaves, and cease its agitation while 3,000,000 remain in bondage. Can it mean that by abandoning slavery in our States we are removing the pressure from you and the country, by preparing for a separation on the line of the cotton States?

We are forbidden so to think, because it is known that we are, and we believe that you

are, unalterably opposed to any division at | all. We would prefer to think that you desire this concession as a pledge of our support, and thus enable you to withstand a pressure which weighs heavily on you and the coun-Mr. President, no such sacrifice is necestry. sary to secure our support. Confine yourself to your constitutional authority; confine your subordinates within the same limits; conduct this war solely for the purpose of restoring the Constitution to its legitimate authority; concede to each State and its loyal citizens their just rights, and we are wedded to you by in-dissoluble ties. Do this, Mr. President, and you touch the American heart and invigorate it with new hope. You will, as we solemnly believe, in due time restore peace to your country, lift it from despondency to a future of glory, and preserve to your countrymen, their posterity, and man, the inestimable treasure of a constitutional government.

Mr. President, we have stated with frank-ness and candor the reasons on which we forbore to vote for the resolution you have mentioned; but you have again presented this proposition, and appealed to us with an earnestness and eloquence which have not failed to impress us, to "consider it, and at the least to commend it to the consideration of our States and people." Thus appealed to by the Chief Magistrate of our beloved country, in the hour of its greatest peril, we cannot wholly decline. We are willing to trust every question relating to their interest and happiness to the consideration and ultimate judgment of our own people. While differing from you as to the necessity of emancipating the slaves of our States as a means of putting down the rebellion, and while protesting against the propriety of any extra-territorial interference to induce the people of our States to adopt any particular line of policy on a subject which peculiarly and exclusively belongs to them, yet, when you and our brethren of the loyal States sincerely believe that the retention of slavery by us is an obstacle to peace and national harmony, and are willing to contribute pecuniary aid to compensate our States and people for the inconveniences produced by such a change of system, we are not unwilling that our people shall consider the propriety of putting it aside.

But we have already said that we regarded this resolution as the utterance of a sentiment, and we had no confidence that it would assume the shape of a tangible, practical proposition, which would yield the fruits of the sacrifice it required. Our people are influenced by the same want of confidence, and will not consider the proposition in its present impalpable form. The interest they are asked to give up is to them of immense importance, and they ought not to be expected even to entertain the proposal until they are assured that when they accept it their just expectations will not be frustrated. We regard your plan as a proposition from the Nation to the States to exercise an admitted constitutional right in a particular manner and yield up a valuable interest. Before they ought to consider the proposition, it should

be presented in such a tangible, practical, efficient shape as to command their confidence that its fruits are contingent only upon their acceptance. We cannot trust anything to the contingencies of future legislation.

If Congress, by proper and necessary legislation, shall provide sufficient funds and place them at your disposal, to be applied by you to the payment of any of our States or the citizens thereof who shall adopt the abolishment of slavery, either gradual or immediate, as they may determine, and the expense of deportation and colonization of the liberated slaves, then will our State and people take this proposition into careful consideration, for such decision as in their judgment is demanded by their interest, their honor, and their duty to the whole country. We have the honor to be, with great respect,

C. A. WICKLIFFE, Ch'n., GARRETT DAVIS, R. WILSON, J. J. CRITTENDEN, JOHN S. CARLILE, J. W. CRISFIELD, J. S. JACKSON, H. GRIDER, JOHN S. PHELPS, FRANCIS THOMAS, CHAS. B. CALVERT, C. L. LEARY, EDWIN H. WEBSTER, R. MALLORY, AARON HARDING, JAMESS ROLLINS, J. W. MENZIES, THOMAS L. PRICE, G. W. DUNLAP, WM. A. HALL.

#### REPLY OF THE MINORITY.

WASHINGTON, July 15, 1862. MR. PRESIDENT: The undersigned, members of Congress from the border States, in response to your address of Saturday last, beg leave to say that they attended a meeting on the same day the address was delivered, for the purpose of considering the same. The meeting appointed a committee to report a response to your address. That report

mittee to report a response to your address. That report was made on yesterday, and the action of the majority indicated clearly that the response, or one in substance the same, would be adopted and presented to you. Inasmuch as we cannot, consistently with our own sense

of duty to the country, under the existing perils which surround us, concur in that response, we feel it to be due to you and to ourselves to make to you a brief and candid answer over our own signatures.

We believe that the whole power of the Government, upheld and sustained by all the influences and means of all loyal men in all sections, and of all parties, is essentially nccessary to put down the rebellion and preserve the Union and the Constitution. We understand your appeal to us to have been made for the purpose of securing this result. A very large portion of the people in the northern States believe that slavery is the "lever-power of the rebellion." It matters not whether this belief be well founded or not. The belief does exist, and we have to deal with things as they are, and not as we would have them be. In consequence of the existence of this belief, we understand that an inunense pressure is brought to bear for the purpose of striking down this institution through the exercise of military authority. The Government cannot maintain this great struggle if the support and influence of the men who entertain these opinions be withdrawn. Neither can the Government hope for early success if the support of that element called "conservative" be withdrawn.

Such being the conservative be winning win. Such being the condition of things, the President appeals to the border State men to step forward and prove their patriotism by making the first sacrifice. No doubt, like appeals have been made to extreme men in the North to meet us half way, in order that the whole moral, political, pecuniary, and physical force of the nation may be firmly and earnestly united in one grand effort to save the Union and the Constitution.

Believing that such were the motives that prompted your address, and such the results to which it looked, we cannot reconcile it to our sense of duty, in this trying hour, to respond in a spirit of fault-finding or querulousness over the things that are past. We are not disposed to seek for the cause of present misfortunes in the errors and wrongs of others who now propose to unite with us in a common purpose. But, on the other hand, we meet your address in the spirit in which it was made, and, as loyal Americans, declare to you and to the world that there is no sacrifice that we are not ready to make to save the Government and institutions of our fathers.

That we, few of us though there may be, will permit no

man, from the North or from the South, to go further than ! man from the scornplishment of the great work before us. That, in order to carry out these views, we will, so far as may be in our power, ask the people of the border States callmy, deliberately, and fairly to consider your recom-mendations. We are the more emboldened to assume this position from the fact, now become history, that the leaders of the southern rebellion have offered to abolish slavery among them as a condition to foreign intervention in favor of their independence as a nation.

If they can give up slavery to destroy the Union, we can surely ask our people to consider the question of emancipa-tion to save the Union.

With great respect, your obedient servants, JOHN W. NOELL, SAMUEL L. CASEY, GEORGE P. FISHER, A. J. CLEMENTS, WILLIAM G. BROWN. JACOB B. BLAIR, W. T. WILLEY.

### REPLY OF MR. MAYNARD.

#### HOUSE OF REPRESENTATIVES, July 16, 1862.

SIR: The magnitude and gravity of the proposition submitted by you to Representatives from the slave States would naturally occasion diversity, if not contrariety, of opinion. You will not, therefore, be surprised that I have not been able to concur in view with the majority of them. This is attributable, possibly, to the fact that my State is not a border State, properly so called, and that my immedinot a border state, property so called, and that my inmediate at constituents are not yet disenthralled from the hostile arms of the rebellion. This fact is a physical obstacle in the way of my now submitting to their consideration this or any other proposition looking to political action, especi-ally such as, in this case, would require a change in the organic law of the State.

But do not infer that I am insensible to your appeal. am not. You are surrounded with difficulties far greater than have embarrarsed any of your predecessors. You need the support of every American citizen, and you ought to have it—active, zealous, and honest The union of every have it—active, zealous, and honest The union of every Union man to aid you in preserving the Union is the duty of the time. Differences as to policy and methods must be subordinated to the common purpose.

In looking for the cause of this rebellion, it is natural that each section and each party should ascribe as little blame as possible to itself, and as much as possible to its opponent section and party. Possibly you and I might not agree ou a comparison of our views. That there should be differences of opinion as to the best mode of conducting onr military operations, and the best men to lead our armies, is equally natural. Contests on such questions weaken our-selves and strengthen our enemies. They are nnprofitable, and possibly unpatriotic. Somebody must yield, or we waste our strength in a contemptible struggle among ourselves.

You appeal to the loyal men of the slave States to sacrifice something of feeling and a great deal of interest. The sacrifices they have already made and the sufferings they The have endured give the best assurance that the appeal will not have been made in vain. He who is not ready to yield all his material interests, and to forego his most cherished sentiments and opinions for the preservation of his country, although he may have periled his life on the battle-field in her defence, is but half a patriot. Among the loyal people that I represent there are no half patriots.

Already the rebellion has cost us much, even to our undoing; we are content, if need be, to give up the rest to suppress it. We have stood by you from the beginning of this struggle, and we mean to stand by you, God willing, till the end of it.

I did not vote for the resolution to which you allude, solely for the reason that at the time I was absent at the capital of my own State. It is right. Should any of the slave States think proper to terminate

that institution, as several of them, I understand, or at least some of their citizens propose, justice and a generous comity require that the country should interpose to aid it in lessening the burden, public and private, occasioned by so radical a change in its social and industrial relations. I will not now speculate upon the effect, at home or

abroad, of the adoption of your policy, nor inquire what action of the rebel leaders has rendered something of the kind important. Your whole administration gives the highest assurance that you are moved, not so much from a desire to see all men everywhere made free, as from a higher desire to preserve free institutions for the benefit of men already free; not to make slaves freemen, but to prevent freemen from being made slaves inot to destroy an institu-tion, which a portion of us only consider bad, but to save institutions which we all alike consider good. I am satis-fied you would not ask from any of your fellow-citizens a

sacrifice not, in your judgment, imperatively required by the safety of the country.

This is the spirit of your appeal, and I respond to it in the same spirit.

I am, very respectfully, your obedient servant. HORACE MAYNARD.

To the PRESIDENT.

#### SENATOR HENDERSON'S REPLY TO THE PRESIDENT.

WASHINGTON CITY, July 21, 1862.

Mr. PRESIDENT: The pressure of business in the Senate during the last few days of the session prevented my at-tendance at the meeting of the border State members, called to consider your proposition in reference to gradual emancipation in our States.

It is for this reason only, and not because I fail to appre-ciate their importance or properly respect your suggestions, that my name does not appear to any of the several pa-pers rubmitted in response. I may also add that it was my intention, when the subject came up practically for consid-eration in the Senate, to express fully my views in regard to it. This of course would have rendered any other re-sponse unnecessary. But the want of time to consider the matter deprived me of that opportunity, and, lest now my silence be misconstrued, I deem it proper to say to you that I am by no means indifferent to the great questions so earnestly, and as I believe so honestly, urged by you upon our consideration.

The border States, so far, are the chief sufferers by this war, and the true Union men of those States have made the greatest sacrifices for the preservation of the Government. This fact does not proceed from mismanagement on the part of the Union authorities, or a want of regard for our peo-ple, but it is the necessary result of the war that is upon us. Our States are the battle-fields. Our people, divided among themselves, maddened by the struggle and blinded by the smoke of battle, invited upon our soil contending armies-the one to destroy the Government, the other to maintain it. The consequence to us is plain. The shock of the contest upturns society and desolates the land. We have made sacrifices, but at last they were only the sacrifices demanded by duty, and unless we are willing to make others, indeed any that the good of the country, involved in the overthrow of treason, may exact at our hands, our title to patriotism is not complete.

When you submitted your proposition to Congress, in March last, "that the United States ought to co-operate with any State which may adopt a gradual abolishment of slavery, giving to such State pecuniary aid, to be used by such State in its discretion, to compensate for the inconsuch State in its discretion, to compensate for the incon-veniences, public and private, produced by such change of system," I gave it a most cheerful support, and I am satis-fied it would have received the approbation of a large majority of the border States delegations in both branches of Congress, if, in the first place, they had believed the war, with its continued evils—the most prominent of which, in a material point of view, is its injurious effect on the instia material point of view, is its injurious elect of the material tution of slavery in our States—could possibly have been protracted for another twelve months; and if, in the second place, they had felt assured that the party having the ma-jority in Congress would, like yourself, be equally prompt in practical action as in the expression of a sentiment. While scarcely any one doubted your own sincerity in ths premises, and your earnest wish speedily to terminate the war, you can readily conceive the grounds for difference of opinion where conclusions could only be based upon coniecture.

Believing, as I did, that the war was not so near its termbelieving, as full, that the war was hot so had to each the term ination as some supposed, and feeling disposed to accord to others the same sincerity of purpose that I should claim for myself under similar circumstances. I voted for the proposition. I will suppose that others were actuated by no sinister motives.

In doing so, Mr. President, I desire to be distinctly underat the time that I was personally making any sacrifice by supporting the resolution, nor that the people of my State were called upon to make any sacrifice, either in considering or accepting the proposition, if they saw fit. I agreed with you in the remarks contained in the message accom-panying the resolution, that "the Union must be pre-served, and hence all indispensable means must be em-ployed." \* \* War has been and continues to be an ployed." \* \* War has been and continues to be an indispensible means to this cal. A practical reacknowl-edgment of the national authority would render the war unnecessary, and it would at once cease. If, however, re-sistance continues, the war must also continue; and it is impossible to foresee all the incidents which may attend and all the ruin which may follow it. It is truly "imposwith the Government of our fathers secure, and constitutional liberty in all its purity guarantied to the white man, the result will be better than that having a place in the fears of many good men at present, and much better than the past history of such revolutions can justify us in expecting

pecifing. In this period of the nation's distress, I know of no hm-man institution too sacred for discussion; no material in-terest belonging to the citizen that he should not willingly place upon the altar of his country, if demanded by the public good. The man who cannot now sacrifice party and put aside selfish considerations is more than half disloyal. Such a man does not descrve the blessings of good govern-ment. Pride of opinion, based upon sectional jealousies, should not he permitted to control the decision of any political question. These remarks are general, but apply with pe-culiar force to the people of the border States at present.

Let us look at our condition. A desolating war is upon us. We cannot escape it if we would. If the Union armies were to-day withdrawn from the border States without first crushing the rebellion in the South, no rational man can doubt for a moment that the adherents of the Union cause in those States would soon be driven in exile from their the people of the border States understand very well the

Infriendly and selfish spirit exercised toward them by the leaders of this cotton State rebellion; beginning some time previous to its outbreak. They will not fail to remember their insolent refusal to counsel with us, and their haughty their insolent refusal to counsel with us, and their haughty assumption of responsibility upon themselves for their mis-guided action. Our people will not soon forget that, while declaiming against coercion, they closed their doors against the exportation of slaves from the border States into the South, with the avowed purpose of forcing us into rebellion through fears of losing that species of property. They knew very well the effect to be produced on slavery by a civil war, especially in those States into which hostile ar-mies might penetrate, and upon the soil of which the great miss might penetrate, and upon the soil of which the great contests for the success of republican government were to be decided. They wanted some intermediate ground for the conflict of arms-territory where the population would be divided. They knew, also, that by keeping slavery in the border States the mere "friction and abrasion," to which you so appropriately allude, would keep up a constant irri-tation, resulting necessarily from the frequent losses to which the owners would be subjected. They also calculated which the owners would be subjected. They also calculated largely, and not without reason, upon the repugnance of non-slaveholders in those States to a free negro population. In the meantime they intended persistently to charge the overthrow of slavery to be the object of the Government, and hostility to this institution the origin of the war. By this means the unavoidable incidents of the strife might easily be charged as the settled purposes of the Government. Again, it was well understood by these men that exemplary conduct on the next of every officer and soldier employed conduct on the part of every officer and soldier employed by the Government could not in the nature of things be expected, and the hope was entertained, upon the most reason-able grounds, that every commission of wrong and every omission of duty would produce a new cause for excitement and a new incentive to rehellion.

By these means the war was to be kept in the border States, regardless of our interests, until an exhausted treas-ury should render it necessary to send the tax-gatherer among our people, to take the little that might be left them from the devastations of war. They then expected a clamor for peace by us, resulting in the interference of France and England, whose operatives in the meantime would be driven to want, and whose aristocracy have ever been ready

driven to want, and whose aristocracy have ever been ready to welcome a dissolution of the American Union. This cunningly-devised plan for securing a Gulf-Confed-eracy, commanding the months of the great western rivers, the Gulf of Mexico, and the southern Atlantic ocean, with their own territory unscathed by the horrors of war, and surrounded by the border States, half of whose population would be left in sympathy with them for many years to come, owing to the irritations to which I have alluded, has so far succeeded too well. In Missouri they have allered wearsed us to lose a third an

In Missouri they have already caused us to lose a third or The mission has have so where the set of the base cannot be more of the slaves owned at the time of the last census. In addition to this, I can make no estimate of the vast amount of property of every character that has been destroyed by military operations in the State. The loss from general depreciation of values, and the utter prostration of every business interest of our people, is wholly beyond calcula-tion. The experience of Missouri is but the experience of other sections of the computer will have situated. tion. The experience of Missouri is but the experience of other sections of the country similarly situated. The ques-tion is therefore forced upon us, "How long is this war to continue; and, if continued, as it has been, on our soil, aided by the treason and folly of our own citizens, acting in con-cert with the Confederates, how long can slavery, or, if you pleaso, any other property interest, survive in our States?" As things now are, the people of the border States yet divided, we cannot expect an immediate termination of the

struggle, except upon condition of southern independence, struggle, except upon condition of southern independence, losing thereby control of the lower Mississippi. For this we in Missouri are not prepared, nor are we prepared to be-come one of the Confederate States, should the terrible calamity of dissolution occur. This, I presume, the Union men of Missouri would resist to the death. And whether they should do so or not, I will not suppose for an instant that the Government of the United States would upon any condition sulmit to the loss of territory so essential to its condition submit to the loss of territory so essential to its future commercial greatness as is the State of Missouri. But should all other reasons fail to prevent such a misfortune to our people of Missouri, there is one that cannot fail. The Confederates never wanted us, and would not have us. I assume, therefore, that the war will not cease, but will be continued until the rebellion shall be overcome. It cannot and will not cease, so far as the people of Missouri are conerned, except upon condition of our remaining in the Union, and the whole West will demand the entire control of the Mississippi river to the Gulf. Our interest is there-fore bound up with the interests of those States maintaining the Union, and especially with the great States of the West, that must be consulted in regard to the terms of any peace that must be consulted in regard to the terms of any peace should they at any time unfortunately depart from their former pacific policy and determine to intervene in our affairs.

The war, then, will have to be continued until the Union shall be practically restored. In this alone consists the fu-ture safety of the border States themselves. A separation of the Union is ruinous to them. The preservation of the Union can only be secured by a continuation of the war. The consequences of that continuation may be judged of by the experience of the last twelve months. The people of my State are as competent to pass judgment in the premises as Lam. These every confidence in their intelligence, their honesty, and their patriotism.

honesty, and their patricusm. In your own language, the proposition you make "sets up no claim of a right by Federal authority to interfere with slavery within State limits, referring, as it dees, the absolute control of the subject in each case to the State and its people immediately interested. It is proposed as a mat-ter of perfectly free choice with them." In this view of the subject I can frankly say to you that, personally, I never could appreciate the objections so fre-quently urged against the proposition. If I understood

quently urged against the proposition. If I understood you properly, it was your opinion, not that slavery should be removed in order to secure our loyalty to the Govern-ment, for every personal act of your administration pre-cludes such as inference, but you believe that the peculiar species of property was in imminent danger from the war in which we reare expected and that commends institute descent which we were engaged, and that common justice demandwhich we were engaged, and that common justice demand-ed remuneration for the loss of it. You then believed, and again express the opinion, that the peculiar nature of the contest is such that its loss is almost inevitable, and lest any pretext for a charge of injustice against the Gov-ernment be given to its enemies, you propose to extend to the people of those States standing by the Union the choice

of payment for their slaves or the responsibility of loss, should it occur, without complaint against the Government. Placing the matter in this light, (a more remunctation for losses rendered inevitable by the casualties of war.) the objection of a constitutional character may be rendered purch loss formidple in the mind of payments. objection of a constitutional character may be requested much less formidable in the minds of northern Represent-atives whose constituents will have to share in the pay-ment of the money; and, so far as the border States are concerned, this objection should be most sparingly urged, is in balance motively of their "own free choice," concerned, this objection should be most sparingly urged, for it being a matter entirely of their "'own free choice," in case of a desire to accept, no serious argument will likely be urged against the receipt of the money, or a fund for colonization. But, aside from the power derived from the operations of war, there may be found numerous prethe operations of war, there may be noted informations pre-cedents in the legislation of the past, such as grants of land and money to the several States for specified objects deemed worthy by the Federal Congress. And in addition to this may be cited a deliberate opinion of Mr. Webster upon this very subject, in one of the ablest arguments of his life.

I allude to this question of power merely in vindication of the position assumed by moin my ote for the resolution of March last. In your last communication to us, you beg of us "to commend this subject to the consideration of our States and people." While I entirely differ with you in the opinion expressed, that had the unehers from the border States approved of your resolution of March last "the war would now be substantially ended," and while I do not regard the suggestion "as one of the most potent and swift means of ending" the war, I am yet free to say that I have the most unbounded confidence in your sin-cerify of purpose in calling our attention to the dangers surrounding us. I am satisfied that you appreciate the troubles of the border States, and that your suggestions are peal, and the logic of surrounding circumstances brings conviction even to an unwilling believer. Having said that, of the position assumed by mo in my vote for the resolution

in my judgment, you attached too much importance to this measure as a means for suppressing the rebellion, it is due to you that I shall explain. Whatever may be the status of the border States in this

Whatever may be the status of the border States in this respect, the war cannot be eoded until the power of the Government is made manifest in the secoded States. They appealed to the sword; give them the sword. They asked for war; let them see its evils on their own soil. They have erected a Government and they force obedience to its behests. This structure must be destroyed; this image, before which an unwilling people have been compelled to bow, must be broken. The authority of the Federal Government must be felt in the heart of the rebellious district. To do this let armies be marched upon them at once, and let them feel what they have inflicted on us in the border. Do not fear our States ; we will stand by the Government in this work.

I ought not to disguise from you or the people of my State that personally I have fixed and unalterable opinions on the subject of your communication. Those opinions I shall communicate to the people in that spirit of frankness that should characterize the intercourse of the representative with his constituents. If I were to-day the owner of the lands and slaves of Missouri, your proposition, so far as that State is concerned, would be immediately accepted. Not a day would be lost. Aside from public considerations, which you suppose to be involved in the proposition, and which no patriot, I agree, should disregard at present, my own personal interest would prompt favorable and immediate action.

But having said this, it is proper that I say something more. The representative is the servant and not the master of the people. It has no authority to bind them to any course of action, or even to indicate what they will or will not do when the subject is exclusively theirs and not his. I shall take occasion, I hope honestly, to give my views of existing troubles and impending dangers, and shall leave the rest to them, disposed, as I am, rather to trust their judgment upon the case stated than my own, and at the same time most cheerfully to acquiesce in their decision.

For you, personally, Mr. President, I think I can pledge the kindest considerations of the people of Missouri, and I shall not hesitate to express the belief that your recommendation will be considered by them in the same spirit of kindness munifested by you in its presentation to us, and that their decision will be such as is demanded "by their interests, their honor, and their duty to the whole country."

I am, very respectfully, your obedient servant, J. B. HENDERSON. To his Excellency A. LINCOLN, President.

December 1, 1862 — The President, in his second annual message, recurs to the subject:

EXTRACT FROM PRESIDENT LINCOLN'S SECOND ANNUAL MESSAGE.

On the 22d day of September last, a proclamation was issued by the Executive, a copy of which is herewith submitted.

In accordance with the purpose expressed in the second paragraph of that paper, I now respectfully recall your attention to what may be called "compensated emancipation."

A nation may be said to consist of its territory, its people, and its laws. The territory is the only part which is of certain durability. "One generation passeth away, and another generation cometh, but the earth abideth forever." It is of the first importance to duly consider and estimate this ever-enduring part. That portion of the earth's surface which is owned and inhabited by the people of the United States, is well adapted to be the home of one national family; and it is not well adapted for two, or more. Its vast extent, and its variety of climate and productions, are of advantage, in this age, for one people, whatever they might have been in former ages. Steam, telegraphs, and intelligence, have brought these to be an advantageous combination for one united people.

In the inaugural address I briefly pointed nificent region sloping west from the Rocky out the total inadequacy of disunion, as a Mountains to the Pacific, being the deepest and

remedy for the differences between the people of the two sections. I did so in language which I cannot improve, and which, therefore, I beg to repeat:

"One section of our country believes slavery is *right*, and ought to be extended, while the other believes it is *wrong*, and ought not to be extended. This is the only substantial dispute. The fugitive slave clause of the Constitution, and the law for the suppression of the foreign slave trade, are each as well enforced, perhaps, as any law can ever be in a community where the moral sense of the people imperfectly supports the law itself. The great body of the people abide by the dry legal obligation in both cases, and a few break over in each. This, I think, cannot be perfectly cured; and it would be worse in both cases *after* the separation of the sections, than before. The foreign slave trade, now imperfectly suppressed, would be ultimately revived without restriction in one section; while fugitive slaves, now only partially surrendered, would not be surrendered at all by the other.

Bit daily sufference by word how be sufficient at an opthe other. "Physically speaking, we cannot separate. We cannot remove our respective sections from each other, nor build an impassable wall between them. A husband and wife may be divorced, and each go out of the presence and beyond the reach of each other; but the different parts of our country cannot do this. They cannot but remain face to face; and intercourse, either amicable or hostile, must continue between them. Is it possible, then, to make that intercourse more advantageous or more satisfactory after separation than before? Can lense make tratices easier than friends can make laws? Can treatics be more faithfully enforced between aliens than laws can among friends? Suppose you go to war, you cannot fight always; and when, after much loss on both sides, and no gain on either, you cease fighting, the indentical old questious as to terms of intercourse are again upon you."

There is no line, straight or crooked, suitable for a national boundary upon which to divide. Trace through, from East to West, upon the line between the free and slave country, and we shall find a little more than one third of its length are rivers, easy to be crossed, and populated, or soon to be populated, thickly upon both sides; while nearly all its remaining length are merely surveyor's lines, over which people may walk back and forth without any consciousness of their presence. No part of this line can be made any more difficult to pass by writing it down on paper or parchment as a national The fact of separat on, if it comes, boundary. gives up on the part of the seceding section the fugitive slave clause, along with all other constitutional obligations upon the section seceded from, while I should expect no treaty stipulation would ever be made to take its place.

But there is another difficulty. The great interior region, bounded east by the Alleghanies, north by the British dominions, west by the Rocky mountains, and south by the line along which the culture of corn and cotton meets, and which includes part of Virginia, part of Tennessee, all of Kentucky, Ohio, Indiana, Michigan, Wisconsin, Illinois, Missouri, Kansas, Iowa, Minnesota, and the Territories of Dakota, Nebraska, and part of Colorado, already has above ten million people, and will have fifty millions within fifty years, if not prevented by any political folly or mistake. It contains more than one third of the country owned by the United States--certainly more than one million square miles. Once half as populous as Massachusetts already is, it would have more than seventy-five million people. A glance at the map shows that, territorially speaking, it is the great body of the Republic. The other parts are but marginal borders to it, the magnificent region sloping west from the Rocky

also the richest in undeveloped resources. In the production of provisions, grains, grasses, and all which proceed from them, this great interior region is naturally one of the most important in the world. Ascertain from the statistics the small proportion of the region which has, as yet, been brought into cultivation, and also the large and rapidly increasing amount of its products, and we shall be overwhelmed with the magnitude of the prospect presented. And yet this region has no seacoast, touches no ocean anywhere. As part of one nation, its people now find, and may forever find, their way to Europe by New York, to South America and Africa by New Orleans, and to Asia by San Francisco. But separate our common country into two nations, as designed by the present rebellion, and every man of this great interior region is thereby cut off from some one or more of these outlets, not, perhaps, by a physical barrier, but by emharrassing and onerous trade regulations.

And this is true, wherever a dividing or boundary line may be fixed. Place it between the now free and slave country, or place it south of Kentucky, or north of Ohio, and still the truth remains, that none south of it can trade to any port or place north of it, and none north of it can trade to any port or place south of it except upon terms dictated by a Government foreign to them. These outlets, east, west, and south, are indispensable to the well-being of the people inhabiting, and to inhabit, this vast Which of the three may be interior region. the best is no proper question. All are better than either; and all of right belong to that people, and to their successors forever. True to themselves, they will not ask where a line of separation shall be, but will vow rather that there shall be no such line. Nor are the marginal regions less interested in these communications to and through them, to the great outside world. They, too, and each of them, must have access to this Egypt of the West without paying toll at the crossing of any national boundary.

Our national strife springs not from our permanent part; not from the land we inhabit; not from our national homestead. There is no possible severing of this, but would multiply and not mitigate evils among us. In all its adaptations and aptitudes it demands union and abhors separation. In fact it would, ere long, force reunion, however much of blood and treasure the separation might have cost.

Our strife pertains to ourselves---to the passing generations of men; and it can, without convulsion, be hushed forever with the passing of one generation.

In this view, I recommend the adoption of the following resolution and articles amendatory to the Constitution of the United States:

"Resolved by the Senate and House of Representatives of the United States of America in Congress assembled, (two thirds of both Houses concurring.) That the following articles be proposed to the Legislatures (or conventions) of the several States as amendments to the Constitution of the United States, all or any of which articles when ratified by three fourths of the said Legislatures (or conventions) to be valid as part or parts of the said Constitution, namely: namely: "ARTICLE -. Every State, wherein slavery now exists,

which shall abolish the same therein, at any time, or times, before the first day of January, in the year of our Lord one thousand nine hundred, shall receive compensation from the United States, as follows, to wit: "The President of the United States shall deliver to every such State bonds of the United States, bearing interest

- per cent. per annum, to an amount at the rate of ---equal to the aggregate sum of \_\_\_\_\_\_for each slave shown to have been therein by the eighth census of the United States, said bonds to be delivered to such States by instalments, or in one parcel, at the completion of the abolish-ment, accordingly as the same shall have been gradual, or at one time, within such State; and interest shall begin to run upon any such bond only from the proper time of its delivery as aforesaid. Any State having received bonds as aforesaid, and afterwards reintroducing or tolerating slavery

therein, shall refund to the United States the bonds so re-ceived, or the value thereof, and all interest paid thereon. "ARTICLE —. All slaves who shall have enjoyed actual freedom by the chances of the war at any time before the end of the rebellion, shall be forever free; but all owners of and, who shall path have been disloved shall be componenof such, who shall not have been disloyal, shall be compensated for them, at the same rates as is provided for States adopting abolishment of slavery, but in such way that no slave shall be twice accounted for.

"ARTICLE -. Congress may appropriate money and oth-erwise provide for colonizing free colored persons, with their own consent, at any place or places without the United States."

I beg indulgence to discuss these proposed articles at some length. Without slavery the rebellion could never have existed; without slavery it could not continue.

Among the friends of the Union there is great diversity of sentiment and of policy in regard to slavery, and the African race among us. Some would perpetuate slavery ; some would abolish it suddenly, and without compensation ; some would abolish it gradually, and with compensation; some would remove the freed people from us, and some would retain them with us; and there are yet other minor diversities. Because of these diversities, we waste much strength in struggles among ourselves. By mutual conces-sion we should harmonize and act together. This would be compromise; but it would be compromise among the friends, and not with the enemies of the Union. These articles are intended to embody a plan of such mutual concessions. If the plan shall be adopted, it is assumed that emancipation will follow, at least, in several of the States.

As to the first article, the main points are: first, the emancipation; secondly, the length of time for consummating it-thirty-seven years; and, thirdly, the compensation.

The emancipation will be unsatisfactory to the advocates of perpetual slavery; but the length of time should greatly mitigate their dissatisfaction. The time spares both races from the evils of sudden derangement-in fact, from the necessity of any derangement-while most of those whose habitual course of thought will be disturbed by the measure will have passed away before its consummation. They will never Another class will hail the prospect of see it. emancipation, but will deprecate the length of time. They will feel that it gives too little to the now living slaves. But it really gives them It saves them from the vagrant destimuch. tution which must largely attend immediate emancipation in localities where their numbers are very great; and it gives the inspiring assurance that their posterity shall be free forever. The plan leaves to each State, choosing to act under it, to abolish slavery now or at the end of the century or at any intermediate time or by degrees, extending over the whole or any part of the period ; and it obliges no two States to proceed alike. It also provides for compensation, and generally, the mode of making it. This, it would seem, must further mitigate the dissatisfaction of those who favor perpetual slavery, and especially of those who are to receive the compensation. Doubtless some of those who are to pay, and not to receive, will Yet the measure is both just and object. economical. In a certain sense, the liberation of slaves is the destruction of property-property acquired by descent, or by purchase, the same as any other property. It is no less true for having been often said, that the people of the South are not more responsible for the original introduction of this property than are the people of the North; and when it is remembered how unhesitatingly we all use cotton and sugar, and share the profits of dealing in them, it may not be quite safe to say that the South has been more responsible than the North for its continuance. If, then, for a common object, this property is to be sacrificed, is it not just that it be done at a common charge?

And if, with less money, or money more easily paid, we can preserve the benefits of the Union by this means than we can by the war alone, is it not also economical to do it? Let us consider it then. Let us ascertain the sum we have expended in the war since compensated emancipation was proposed last March, and consider whether, if that measure had been promptly accepted, by even some of the slave States, the same sum would not have done more to close the war than has been otherwise done. If so, the measure would save money, and, in that view, would be a prudent and economical measure. Certainly it is not so easy to pay something as it is to pay nothing ; but it is easier to pay a large sum than it is to pay a larger one. And it is easier to pay any sum when we are able, than it is to pay it before we are able. The war requires large sums, and requires them at once. The aggregate sum necessary for compensated emancipation of course would be large. But it would require no ready cash, nor the bonds, even, any faster than the emancipation progresses. This might not, and probably would not, close before the end of the thirty-seven years. At that time we shall probably have a hundred million people to share the burden, instead of thirty-one millions, as now. And not only so, but the increase of our population may be expected to continue for a long time after that period as rapidly as before; because our territory will not have become full. I do not state this inconsiderately. At the same ratio of increase which we have maintained, on an average, from our first national census in 1790, until that of 1860, we should, in 1900, have a population of 103,208,415. And why may we not continue that ratio far beyond that period? Our abundant room-our broad national homestead-is our ample resource. Were our territory as limited as are the British Isles, very certainly our population could not expand as stated. Instead of receiving the foreign born, as now, we should be compelled to send part of the native born away.

But such is not our condition. We have two million nine hundred and sixty-three thousand square miles. Europe has three million and eight hundred thousand, with a population averaging seventy-three and one third persons to the square mile. Why may not our country, at some time, average as many? Is it less fertile? Has it more waste surface, by mountains, rivers, lakes, deserts, or other causes? Is it inferior to Europe in any natural advantage? If, then, we are at some time to be as populous as Europe, how soon? As to when this may be, we can judge by the past and the present; as to when it will be, if ever, depends much on whether we maintain the Union. Several of our States are already above the average of Europe-seventy-three and a third to the square mile. Massachusetts has 157; Rhode Island, 133; Connecticut, 99; New York and New Jersey, each, 80. Also two other great States, Pennsylvania and Ohio, are not far below, the former having 63 and the latter 59. The States already above the European average, except New York, have increased in as rapid a ratio, since passing that point, as ever before; while no one of them is equal to some other parts of our country in natural capacity for sustaining a dense population.

Taking the nation in the aggregate, and we find its population and ratio of increase, for the several decennial periods, to be as follows :

1790	3,929,827			
1800	5,305,937	35.02 1	er cent	t. ratio of increase.
1810	7,239,814	36.45	"	**
1820	9,638.131	33.13	46	**
1830	12,866,020	33.49	"	**
1840		32.67	"	**
1850	23,191,876	35.87	"	*6
1860	31,443,790	35.58	46	46

This shows an average decennial increase of 34.69 per cent. in population through the seventy years from our first to our last census yet taken. It is seen that the ratio of increase, at no one of these seven periods, is either two per cent. below or two per cent. above the average; thus showing how inflexible, and, consequently, how reliable, the law of increase, in our case, Assuming that it will continue, gives the is. following results:

1870,	42.323.341
1880	56,967,216
1890	76.677.872
1900	
1910	
1920	
1930	251.080.914

These figures show that our country may be as populous as Europe now is at some point between 1920 and 1930-say about 1925-our territory, at seventy-three and a third persons to the square mile, being of capacity to contain 217,186,000.

And we will reach this, too, if we do not ourselves relinquish the chance by the folly and evils of disunion, or by long and exhausting war springing from the only great element of national discord among us. While it cannot be foreseen exactly how much one huge example of secession, breeding lesser ones indefinitely, would retard population, civilization, and prosperity, no one can doubt that the extent of it would be very great and injurious.

The proposed emancipation would shortem

the war, perpetuate peace, insure this increase | of population, and proportionately the wealth of the country. With these, we should pay all the emancipation would cost, together with our other debt, easier than we should pay our other debt without it. If we had allowed our old national debt to run at six per cent. per annum, simple interest, from the end of our revolutionary struggle until to-day, without paying anything on either principal or interest, each man of us would owe less upon that debt now than each man owed upon it then; and this because our increase of men through the whole period has been greater than six per cent.; has run faster than the interest upon the debt. Thus, time alone relieves a debtor nation, so long as its population increases faster than unpaid interest accumulates on its debt.

This fact would be no excuse for delaying payment of what is justly due; but it shows the great importance of time in this connection—the great advantage of a policy by which we shall not have to pay until we number a hundred millions, what, by a different policy, we would have to pay now, when we number but thirty-one millions. In a word, it shows that a dollar will be much harder to pay for the war than will be a dollar for emancipation on the proposed plan. And then the latter will cost no blood, no precious life. It will be a saving of both.

As to the second article, I think it would be impracticable to return to bondage the class of persons therein contemplated. Some of them, doubtless, in the property sense, belong to loyal owners, and hence provision is made in this article for compensating such.

The third article relates to the future of the freed people. It does not oblige, but merely authorizes, Congress to aid in colonizing such as may consent. This ought not to be regarded as objectionable on the one hand or on the other, insomuch as it comes to nothing, unless by the mutual consent of the people to be deported, and the American voters, through their representatives in Congress:

I cannot make it better known than it already is, that I strongly favor colonization. And yet I wish to say there is an objection urged against free colored persons remaining in the country which is largely imaginary, if not sometimes malicious.

It is insisted that their presence would injure and displace white labor and white la-If there ever could be a proper time borers. for mere catch arguments, that time surely is not now. In times like the present men should utter nothing for which they would not willingly be responsible through time and in eternity. Is it true, then, that colored people can displace any more white labor by being free than by remaining slaves ? If they stay in their old places, they jostle no white laborers; if they leave their old places, they leave them open to white laborers. Logically, there is neither more nor less of it. Emancipation, even without deportation, would probably enhance the wages of white labor, and, very surely would not reduce them. Thus, the customary amount of labor would still have to be performed; the freed peo-

ple would surely not do more than their old proportion of it, and, very probably, for a time would do less, leaving an increased part to white laborers, bring their labor into greater demand, and consequently enhancing the wages of it. With deportation, even to a limited extent, enhanced wages to white labor is mathematically certain. Labor is like any other commodity in the market—increase the demand for it and you increase the price of it. Reduce the supply of black labor by colonizing the black laborer out of the country, and by precisely so much you increase the demand for and wages of white labor.

But it is dreaded that the freed people will swarm forth, and cover the whole land? Are they not already in the land? Will liberation make them any more numerous? Equally distributed among the whites of the whole country, and there would be but one colored to seven whites. Could the one, in any way, greatly disturb the seven? There are many communities now, having more than one free colored person to seven whites; and this, without any apparent consciousness of evil from it. The District of Columbia, and the States of Maryland and Delaware, are all in this condition. The District has more than one free colored to six whites; and yet, in its frequent petitions to Congress, I believe it has never presented the presence of free colored persons as one of its grievances. But why should emancipation south, send the freed people north? People, of any color, seldom run, unless there be something to run from. Heretofore colored people, to some extent, have fled north from bondage, and now, perhaps, from both bondage and destitution. But if gradual emancipation and deportation be adopted, they will have neither to flee from. Their old masters will give them wages at least until new laborers can be procured; and the freed men, in turn, will gladly give their labor for the wages, till new homes can be found for them, in congenial climes, and with people of their own blood and race. This proposition can be trusted on the mutual interests involved. And, in any event, cannot the North decide for itself, whether to receive them?

Again, as practice proves more than theory, in any case, has there been any irruption of colored people northward because of the abolishment of slavery in this District last spring?

What I have said of the proportion of free colored persons to the whites in the District is from the census of 1860, having no reference to persons called contrabands, nor to those made free by the act of Congress abolishing slavery here.

The plan consisting of these articles is recommended, not but that a restoration of the national authority would be accepted without its adoption.

Nor will the war, nor proceedings under the proclamation of September 22, 1862, be stayed because of the *recommendation* of this plan. Its timely *adoption*, I doubt not, would bring restoration, and thereby stay both.

And, notwithstanding this plan, the recommendation that Congress provide by law for compensating any State which may adopt emancipation before this plan shall have been acted npon is hereby earnestly renewed. Such would be only an advance part of the plan, and the same arguments apply to both. freedom to the s

This plan is recommended as a means, not in exclusion of but additional to all others for restoring and preserving the national authority throughout the Union. The subject is presented exclusively in its economical aspect. The plan would, I am confident, secure peace more speedily, and maintain it more permanently, than can be done by force alone; while all it would cost, considering amounts, and manner of payment, and times of payment, would be easier paid than will be the additional cost of the war, if we rely solely upon force. It is much, very much, that it would cost no blood at all.

The plan is proposed as permanent constitutional law. It cannot become such without the concurrence of, first, two thirds of Congress, and afterwards, three fourths of the States. The requisite three fourths of the States will necessarily include seven of the slave States. Their concurrence, if obtained, will give assurance of their severally adopting emancipation at no very distant day upon the the new constitutional terms. This assurance would end the struggle now, and save the Union forever.

I do not forget the gravity which should characterize a paper addressed to the Congress of the nation by the Chief Magistrate of the nation. Nor do I forget that some of you are my seniors, nor that many of you have more experience than I in the conduct of public affairs. Yet I trust that in view of the great responsibility resting upon me, you will perceive no want of respect to yourselves in any undue earnestness I may seem to display.

Is it doubted, then, that the plan I propose, if adopted, would shorten the war, and thus lessen its expenditure of money and of blood? Is it doubted that it would restore the national authority and national prosperity, and perpetuate both indefinitely? Is it doubted that we here-Congress and Executive-can secure its adoption? Will not the good people respond to a united and earnest appeal from us? Can we, can they, by any other means so certainly or so speedily assure these vital objects? We can succeed only by concert. It is not "can any of us imagine better ?" but " can we all do better ?" Object whatsoever is possible, still The the question recurs "can we do better ?" dogmas of the quiet past are inadequate to the stormy present. The occasion is piled high with difficulty, and we must rise with the occasion. As our case is new, so we must think anew and act anew. We must disenthrall ourselves, and then we shall save our country.

Fellow-citizens, we cannot escape history. We, of this Congress and this Administration, will be remembered in spite of ourselves. No personal significance, or insignificance, can spare one or another of us. The fiery trial through which we pass will light us down, in honor or dishonor, to the latest generation. We say we are for the Union. The world will not forget that we say this. We know how to

save the Union. The world knows we do know how to save it. We—even we here—hold the power and bear the responsibility. In giving freedom to the slave we assure freedom to the free—honorable alike in what we give and what we preserve. We shall nobly save, or meanly lose, the last, best hope of earth. Other means may succeed; this could not fail. The way is plain, peaceful, generous, just—a way which, if followed, the world will forever applaud, and God must forever bless.

# ABRAHAM LINCOLN.

COMPENSATED EMANCIPATION IN MISSOURI.\*

#### Third Session, Thirty-Seventh Congress.

#### IN HOUSE.

1863, Jan. 6—Mr. NOELL, of Missouri, from the Select Committee, reported a bill, providing that the following propositions be submitted to the State of Missouri to be accepted or rejected by the Legislature or people thereof:

 The Government of the United States will, npon the passage by said State of a good and valid act of emancipation of all the slaves therein, to take effect within the period hereinafter named, and to be irrepealable unless by the consent of the United States, apply the sum of \$10,000,000, in United States bonds, redeemable in thirty years from their date, and bearing interest, payable somi-annually, at the rate of five per cent. per annum, for the purpose of paying to the loyal owners of such slaves in said State a just compensation for the loss of the services of such slaves; the said bonds to be prepared and issued by the Secretary of the Treasury, under such rules and regulations as he may prescribe to make effective the provisions of this act.
 That in the event of the adoption by the said State of abolishment of slavery therein, as hereinbefore provided, the Government of the United States will employ all reamarked the denortion of such edges of a since of such slaves;

2. That in the event of the adoption by the said State of abolishment of slavery therein, as hereinbefore provided, the Government of the United States will employ all reasonable means for the deportation of such of said emancipated slaves, and for their settlement or colonization in some place outside of the United States, as may consent thereto, so soon as and whenever proper negotiations can be made with any country for such settlement or colonization in a congenial climate, or within a reasonable time therear.

The semi-annual interest to be payable, and the bonds redeemable out of moneys to be hereafter appropriated by Congress, and to be reimbursed to the United States Treasury from the property of rebels confiscated, and no part of the money to be paid to persons who have, at any time engaged in or in any manner aided in the rebellion, or held office under the Confederate States, &c. The bonds to be delivered to the Governor of Missouri when the President shall be satisfied that a valid act of immediate emancipation shall have been passed by the Legislature, irrepealable without the consent of the United States, and within one year from Jan. 1, 1863.

Mr. HOLMAN moved to lay the bill on the table; which was lost—yeas 41, nays 73.

An unimportant amendment was made, when Mr. PENDLETON moved to lay the bill on the table; which was lost—yeas 43, nays 73.

Mr. CRAVENS then moved that the House adjourn : which was lost-yeas 19, pays 82.

journ; which was lost—yeas 19, nays 82. When the bill passed—yeas 83, nays 50, as follows:

Iollows: YEAS-Messrs. Aldrich, Alley, Ashley, Babbitt, Baker, Baxter, Bingham, Jacob B. Blair, Blake, Bufinton, Burnham, Campbell, Chamberlain, Clark, Colfax, Frederick A. Conkling, Roscoe Conkling, Covode, Dnell, Edgerton, Edwards, Ellot, Ely, Fenton, Samuel C. Fessenden, Thomas A. D. Fessenden, Fisher, Franchot, Gooch, Goodwin, Hale, Harrison, Hickman, Hooper, Horton, Hutchins, Julian, Kelley, Francis W. Kellogg, Lansing, Lehman, Loomis, Lovejoy, Low, McKean, McKnight, McPherson, Maynard, Moorhead, Ansou P. Morrill, Justin S. Morrill, Noell, Olin, Patton, Timothy G. Phelps, Pike, Pomeroy, Alexander H. Rice, John H. Rico, Kiddle, Edward H. Rollins, James S. Rollins, Sargent, Sedgwick, Shanks, Sheffield, Shellabarger, Sherman, Spallding, Stevens, Beujamin F. Thomas, Train, Trowbridge, Van Horn,

\* See resolutions of Convention, supra.

Van Valkenburgh, Verree, Walker, Wall, Wallace, Wheeler, Albert S. White, Windom, Worcester-S3. NAYS-Messrs. William Allen, Ancona, Baily, Biddle, Cal-

NAYS-Messrs. William Alten, Ancona, Baiy, Buald, Cal-ert, Clements, Cobb, Corning, Cox, Cravens, Crittenden, Davis, Dunlag, Dunn, Granger, Grider, Haight, Haul, Harding, Hol-man, Johnson, William Kellogg, Kerrigan, Knapp, Law, Malory, Menzies, Morris, Norton, Odell, Fradleton, Perry, Porter, Price, Shiel, Smith, John B. Stele, William G. Steele, Siles, Stratton, Trimble, Fallandigham, Foorhees, Wadsworth, Ward, Webster, Wickliffe, Wilson, Wood, Woodruff-50.

# IN SENATE.

January 14-Mr. TRUMBULL, from the Committee on the Judiciary, reported a substitute as follows:

That whenever satisfactory evidence shall be presented to the President of the United States that the State of Misthe Freshent of the Dinet States that the cute of his souri has adopted a valid and constitutional law, ordinance, or other provision for the gradual or immediate emancipa-tion of all the slaves therein, and the exclusion of slavery forever thereafter from said State, it shall be his duty to prepare and deliver to the Governor of said State, as here-inatter provided, to be used by said State to compensate for the inconveniences produced by such change of system, bonds of the United States to the amount of \$20,000,000, the some to bear interest the rate of five percent. per annum, and payable thirty years after the date thereof: *Provided*, That the said bonds shall not be delivered as herein di-rected unless the act of emancipation shall be adopted by said State within twelve months after the passage of this said State within twelve months after the passage of this act, nor unless said act shall provide that slavery or invol-untary servitude within said State, except in punishment of crime, shall forever cease and determine on some day not later than the 4th day of July, 1856, and never afterwards be introduced therein: And provided further, That said be dain the account concert concert chell not exceed the runn be informed aggregate amount shall not exceed the sum of \$300 for each slave emancipated under the provisions of this act and the act of emancipation to be passed in the State of Missouri in pursuance thereof, not including any slave who shall be introduced into said State subsequently slave who shall be introduced into said state subsequently to the passage of this act, or has been emancipated or shall become free under the provisions of an act entitled "An act to suppress insurrection, to punish treason and rebellion, to seize and confiscate the property of rebels, and for other purposes," approved July 17, 1862. Sec. 2. And be it further enacted, That in the event of the adoption by said State of an act or ordinance for the abolition of slavery therein, as thereinbefore provided, the President shall, from time to time, as they may be re-paired course honds to be prepared and delivered as a fore.

quired, cause bonds to be prepared and delivered, as aforesaid, in such amounts as may be necessary to pay for each slave made free nnder the provisions of this act, and the act to be passed by the State of Missouri in pursuance thereof, not to exceed an average of \$300; said bonds to be delivered only as said slaves are made free, and in the ag-gregate not to exceed \$20,000,000; and not to exceed \$16,000,000, unless the law or act of emancipation to be adopted by said State shall provide for the full and perfect

adopted by said state shall provide for the full and perfect manumission of all the slaves therein on or before the 4th day of July, A. D. 1865, and the future exclusion of the system of slavery from the State forever thereafter. Szc. 3. And be it further enacted, That upon the enact-ment of a law by said State referring to this act, accepting its provisions and adopting laws for the ultimate extinction of slavery therein the fails of the United States and of the of slavery therein, the faith of the United States and of the said State will be pledged to carry out the terms of the compact thus created: *Provided*, That nothing herein con-tained shall prevent the State from making at any time such alterations or amendments in its emancipation laws as do not conflict with this act, or such as shall only de-crease the time during which slavery may exist in the said State.

Feb. 7-Mr. WILSON, of Missouri, offered the following amendment:

That the said bonds shall not be delivered as herein directed unless the act of emancipation shall be adopted by said State within three years after the passage of this act.

Striking out "twelve months" and inserting "three years;" which was disagreed to-yeas 8, nays 27, as follows:

YEAS-MESSES. Carlile, Davis, Kennedy, Nesmith, Powell, Richardson, Wall, Wilson of Missouri-S.
 NAYS-Messes. Anthony, Arnold, Chandler, Clark, Colla-mer, Dixon, Doolittle, Fessenden, Foot, Foster, Grimes, Harding, Harlan, Harris, Henderson, Hicks, King, Lane of Indiana, Laue of Kansas, Morrill, Pomeroy, Sumner, Ten Eyck, Trumbull, Willey, Wilmot, Wilson of Massachusetts -27.

Mr. WILSON, of Missouri, moved to strike out the provision limiting the amount to be paid for each slave, as follows :

And provided further, That said bonds in their aggregate amount shall not exceed the sum of \$300 for each slave emancipated under the provisions of this act and the act of emancipation to be passed by the State of Missouri in pursnance thereof.

Which was disagreed to-yeas 9, nays 27, as follows:

YEAS-Mossrs. Davis, Kennedy, Latham, Nesmith, Powell, Rice, Wall, Willey, Wilson of Missouri-9. Navs-Messrs. Anthony, Arnold, Chandler, Clark, Colla-mer, Dixon, Doolittle, Fessenden, Foot, Foster, Grimes, Harding, Harlan, Harris, Henderson, Hicks, King, Lane of Indiana, Lane of Kansas, Morrill, Pomeroy, Sherman, Sum-ner, Ten Eyck, Trumbull, Wilnot, Wilson of Massachusetts -27

Mr. POWELL moved to strike out of the first section the words "not to exceed \$20,000,000," and insert "\$11,000,000;" which was disagreed to-yeas 13, nays 22, as follows:

YEAS-Messre. Curlile, Clark, Collamer, Daris, Fessenden, Grimes, Harding, Harlan, Lane of Indiana, Nesmith, Pow-ell, Rice, Wall-13. NAVS-Messrs. Anthony, Arnold, Chandler, Dixon, Doo-little, Foot, Foster, Harris, Henderson, Howard, King, Lane of Kansas, Morrill, Pomeroy, Sumner, Ten Eyck, Trumball, Wade, Willey, Wilmot, Wilson of Massachusetts, Wilson of Missouri-22.

Mr. WILSON, of Missouri, offered the following amendment:

And be it further enacted, That no part of the bonds herein specified shall be delivered until the act of the Legislature or the Constitutional Convention of the State of Mis-souri providing for the emancipation of the sluves in said State shall be submitted to a vote of the people and ap-proved by a majority of the legal voters of said State.

Which was disagreed to-yeas 13, nays 27, as follows:

YEAS-Messrs. Carlile, Davis, Hardiny, Kennedy, Nesmith, Powell, Rice, Richardson, Saulsbury, Turpie, Wall, Willey,

diana, Laue of Kansas, Morrill, Pomeroy, Sumner, Ten Eyck, Trumbull, Wade, Wilmot, Wilson of Massachusetts -27.

Mr. SUMNER offered an amendment to strike out "seventy-six," in the first section, and insert "sixty-four," making the act to take effect on the 4th of July, 1864; which was disagreed to-yeas 11, nays 26, as follows:

YEAS-Messrs. Fessenden, Grimes, Harlan, Lane of Indi-ana, Lane of Kansas, Ponieroy, *Rice*, Sumner, Wade, Wil-mot, Wilson of Massachusetts-11.

NATS—Messts. Anthony, Arnold, Chandler, Clark, Cowan, Davis, Dixon, Doolittle, Foot, Foster, Harding, Harris, Hen-derson, Howard, Howe, King, Lutham, Morrill, Nesmith, Fowell, Sherman, Ten Eyck, Trumbull, Wall, Willey, Wilson of Missouri-26.

February 12-Mr. SUMNER offered the following amendment :

That said bonds, in their aggregate amount, shall not exceed the sum of \$200 for each slave emancipated.

Which was agreed to-yeas 19, nays 17, as follows:

YEAS-Messrs. Carlile, Clark, Collamer, Fessenden, Grimes, Harlan, Howe, King, Lane of Indiana, Lane of Kansas, Pomeroy, Rice, Sherman, Sumner, Trumbull, Wade, Wilkin-

Someroy, rice, Sherman, Sumner, Trumbull, Wale, Wilkinson, Wilmot, Wilson of Massachusetts—19. NATS—Messrs. Anthony, Arnold, Cowan, Dixon, Doolittle, Foster, Harris, Henderson, Hicks, Howard, Laiham, Mc-Dougall, Morrill, Richardson, Ten Eyck, Wall, Wilson of Missouri—17.

Mr. SUMNER offered an amendment to strike out of the first section the word "gradual," making the operation of emancipation immediate; which was disagreed to-yeas 11, nays | 27, as follows :

YEAS - Messrs. Carlile, Collamer, Cowan, Fessenden, Grimes, Harlan, Lane of Kansas, Pomeroy, Sumner, Wade, Wilson of Massachusetts-11.

NATS-Messrs. Anthony, Arnold, Chandler, Clark, Davis, Dixon, Doolittle, Foster, Harris, Henderson, Hicks, Howard, Howe, King, Latham, McDougall, Morrill, Nesmith, Pranell, Rice, Richardson, Saulshury, Sherman, Ten Eyck, Trumbull, Wilkinson, Wilmot, Wilson of Missouri-27.

The bill, as amended, then passed the Senate -yeas 23, nays 18, as follows :

YEAS-Messrs. Anthony, Arnold, Chandler, Clark, Col-lamer, Doolittle, Foot, Foster, Harlan, Harris, Henderson, Howard, Howe, King, Lane of Kansas, Morrill, Pomeroy, Summer, Trumbull, Wade, Wilkinson, Wilmot, Wilson of

Sumlet, Tribundi, Wate, Wikinson, Willich, Wilson Of Massachusetts—23. NAVS—Messrs. Curlile, Cowan, Davis, Fessenden, Grimes, Harding, Kennedy, Lane of Indiana, Latham, McDougall, Nesmith, Powell, Richardson, Saulsbury, Ten Eyck, Turpie, Wall, Wilson of Missouri—18.

February 18-The bill was considered in the House of Representatives, and recommitted to the Select Committee-yeas 81, nays 51.

February 25-The committee reported back the bill, but it was not again considered by the House.

March 3-Mr. WHITE, of Indiana, moved to suspend the rules to take it up, but the motion was lost-yeas 62, nays 57, (two thirds being required.)

COMPENSATED EMANCIPATION IN MARYLAND.

1863, January 12-Mr. FRANCIS THOMAS offered the following resolution, which was agreed to:

Resolved, That the Committee on Emancipation and Colonization be instructed to inquire into the expediency of making an appropriation to aid the State of Maryland in a system of emancipation and colonization of persons of color, inhabitants of said State.

January 19-Mr. BINGHAM, of Ohio, intro-duced a bill giving aid to the State of Maryland, for the purpose of securing the abolishment of slavery in said State.

February 25-Mr. WHITE, of Indiana, reported back the bill with amendments. The bill appropriated \$10,000,000, upon the passage of an emancipation bill, to take effect within two years from Jan. 1, 1863, upon substantially the terms set forth in the Missouri bill, as passed by the House.

After some debate, the bill was re-committed to the Select Committee by a vote of-yeas 75, nays 55, as follows:

Nays 55, as follows:
 YEAS-Messrs. Aldrich, Alley, Arnold, Ashley, Babbitt, Bake, William G. Brown, Buffinton, Campbell, Casey, Chamberlain, Clark, Clements, Colfax, Frederick A. Conk-ing, Roscoe Conkling, Cutter, Davis, Dawes, Diven, Ed-gerton, Edwards, Eliot, Ely, Fenton, Samuel C. Fessenden, Thomas A. D. Fessenden, Fisher, Flanders, Frank, Gooch, Gurley, Hahn, Hutchins, Kelley, Lansing, Leavy, Lehman, Loomis, Lovejov, Low, McIndoe, McKean, McKnight, May-nard. Mitchell, Moorhead, Justin S. Morrill, Nixon, Pendle-ton, Timothy G. Phelps, Pomeroy, Alexander II, Nice, John H. Rice, Riddle, Edward H. Rollins, Sargent, Sedgwick, Scofield, Shellabarger, Sherman, Stevens, Francis Thomas, Trowbridge, Verree, Walton, Wallace, Albert S. White, Wilson, Windom-75.
 NAYS-Messrs. William Allen, William, J. Allen, Ancona, Biddle, Calvert, Conway, Coz, Cravens, Crisfield, Delano, Dunlop, Dunn, Fouke, Granger, Haight, Hale, Hall, Hard-ing, Harrison, Holman, Johnson, Kervigan, Knapp, Law, Juarear, NetPerson, Mallory, May, Menzie, Morris, Noble, Notion, Nugen, Perry, Porter, Frie, Robinson, James S. Roltins, Segar, Smith, John B. Steele, William G. Sleele, Stiles, Benjamin F. Thomas, Train, Trimble, Vallandigham, Wadsworth, Ward, Webster, Wickliffe, Wood, Woodruff, Worcester, Feanam-55.
 It was not again reached.

It was not again reached.

### EMANCIPATION IN MARYLAND-ITS DAWNING.

At the meeting of the Union Convention of Baltimore city, May 28, 1862, Mr. P. G. Sauerwein, from the Business Committee, reported these resolutions, which were adopted, seriatim. unanimously:

Whereas, the Union men of the city of Baltimore have recently, by large meetings in their respective wards, chosen this City Convention to represent their political interest for the ensuing year, and it is proper that amid the rapid progress of events, and in these times of emergency and peril, we should more definitely settle and define our position and principles in the face of the county : Therefore.

Resolved, By the Union City Convention of Baltimore, that we reaffirm our unconditional adhesion to the Constitution

ad Government of the United States. 2. That the National Government ought not to lay down its arms until its authority shall be acknowledged and obeyed in every portion of the national territory.

3. That the attempt to revive a political organization under the leadership of men who have, by their votes and speeches, encouraged the rebellion, or carped at the neces-sary measures of the Government in its hour of trial, or failed to sympathize in the great uprising of the people to defend the Union, ought to be disconraged by every true patriot as a selfish and disloyal effort to obtain power in the future by conciliating those who are now in flagrant rebellion.

4. That we cordially approve the firm and vigorous ef-4. That we contain approve the firm and vigorous el-forts of the Administration to maintain the integrity and honor of our country, to crush rebellion, and to anticipate and defeat the acts of traitors; and we are fully persuaded that, so long as the insurrection is raging, leniency is wasted alike upon the robels in arms, and upon their malignant sympathizers at home; and that all efforts at conciliation will prove unavailing till the Government shall have fully domenet and its nover. demonstrated its power.

5. That we approve the wise and conservative policy pro-posed by the President in his message of the 6th March, 1862, and sanctioned by Congress, tendering pecuniary aid to such States as may choose to adopt a system of gradual emancipation; and that it is not only the *duty* of the loyal people of Maryland to meet the Government in this endeavor to relieve them, in the only practical way, from the cyils which armed insurrection has brought upon the land, thus sustaining the Government as well against the treason of secession as against the radical and violent projects of fanatical abolitionists, but that it is likewise for the *interest* of all the people of this State, more especially of the s aveowners, promptly to accept the aid thus tendered, and re-move from our midst an institution which has ceased to be profitable, and is now injurious to our political and material interests, and dangerous to our peace and salety, by inau-gurating such a plan of emancipation and colonization as will be equitable to those interested, and as will tend to secure the industry of the State to the white labor of the State.

State. 6. That the Legislature, at its late session, in failing to adopt stringent measures for the suppression of treason; in failing to make adequate provision for the families of our volunceers; in failing to organize the military of the State; in putting forth unnecessary protests calculated to embar-rass the action of the Government and throw doubt upon the position of the State; and, in neglecting to comply with its constitutional duty, to provide for a vote on the question of a Constitutional duty, to provide for a vote on the question of a constitutional Convention, did not meet the demands of the crisis, or answer the just expectations of the loyal people of Maryland.

Whereas, the principle of equal rights underlies the founda-tion of republican institutions; and whereas notwithstand-ing the eight counties (this city included) lying north of the tion of republican institutions; and whereas neuronanticing the eight counties (this city included) lying north of the Sassafras and Patapaco rivers contain about three fourths of the white population and wealth of this State, and pay more than three fourths of the taxes, yet, according to the basis of representation under the present constitution, the southern counties, containing one fourth of the population and wealth, and paying less than one fourth of the taxes, yet, southern counties, the seventy four delegates, and fourteen out of twerty-two senators, to the Legislature, being an average of one delegate to 3,831 white persons in the southern consties, against one delegate to 9,641 in the northern, and one senator to 9,205 in the former against one senator to 48,205 free white persons in the latter; and whoreas, in the present arrangement of representation, according to to 45,205 free while persons in the atter; and whiereas, in the present arrangement of representation, according to which Baltimore city has no more voice in the Senate than counties containing hardly more voters than one of her wards, and was deprived of fifteen delegates to which she was entitled, which were distributed among the southern

counties exclusively, the object was to secure the power of counties exclusively, the object was to secure the power of the State to the slave-owners, constituting less than sixteen thousand individuals, as is apparent from the fact that the number of slaves in the city is less than one per cent. of the whole population, and that the proportion of slaves to the whites throughout the northern part of the State is less than four of the former to one bundred of the latter, while the pro-unition is the constant of the state to be four of the former to one hundred of the latter, while the pro-portion in the favored counties is fifty-six slaves to one hun-dred white people; and whereas, the Governor, the judges of the court of appeals, and the United States Senators are elect-ed according to a similar unjust arrangement, for the same purpose of confirming and perpetuating this domination of the slaveholding counties; and whereas, in pursuance of the same partial and injuitous scheme, the owners of slave property are eventued from equal taration, while they are property are exempted from equal taxation, while they are compensated to its full value in the case of convict slavesall of which is an intolerable burden to the majority of our people, who are, nevertheless, without remedy under the present constitution : Therefore

Resolved, That the State constitution ought to be changed, so as to correct the present unequal representation of the people in the Legislature, and to secure the rights of representation in proportion to white population to all parts of the State.

#### IN CONSTITUTIONAL CONVENTION.

1864, June 23-The question recurred on the twenty-third article of the Bill of Rights, as reported by the Committee for the New Constitution, as follows :

Hereafter, in this State, there shall be neither slavery nor involuntary servitude, except in punishment of crime, whereof the party shall have been duly convicted, and all persons held to labor as slaves are hereby declared free.

An amendment was pending, offered by Mr. Brown, of Queen Anne's, as follows:

And the Legislature shall make provision from the treas-ury of the State for the comfortable support and maintenance of the helpless and paupers hereby emancipated.

The amendment was rejected-yeas 28, nays 51, as follows:

51, as follows: Yaas-Messrs. Berry of Baltimore county, Berry of Prince George's, Billugsley, Blackiston, Briscoe, Brown, Cham-bers, Clarke, Crawford, Dail, Davis of Charles, Dennis, Du-vall, Edelen, Gale, Harwood, Hollyday, Horsey, Johnson, Lansdale, Lee, Marbury, Mitchell, Miller, Parran, Peter, Smith of Dorchester, Turner-28. NAYS-Messrs. President (H. H. Goldsborough), Ab-bott, Annan, Andoun, Baker, Barron, Carter, Cunuingham, Cushing, Daniel, Davis of Washington, Earle, Ecker, Far-row, Galloway, Greene, Hatch, Hebb, Hofman, Hopkins, Hopper, Jones of Cecil, Keefer, Kennard, King, Larsh, Mace, Markey, Mulhkin, Murray, Negley, Nymau, Parker, Purnell, Ridgely, Robinette, Russell, Sands, Schley, Schlos-ser, Scott, Smith of Carroll, Sneary, Striling, Stockbridge, Sykes, Thomas, Thruston, Valliant, Wickard, Wooden-51. The article was then adopted-yeas 53, nays

The article was then adopted-yeas 53, nays 27, as follows:

Yzas-Messrs. President (H. H. Goldsborough), Abbott, Annan, Audoun, Baker, Barron, Berry of Baltimore county, Carter, Canningham, Cushing, Baniel, Davis of Washing-ton, Earle, Ecker, Farrow, Galloway, Greene, Hatch, Hebb, Moffman, Hoykins, Hopper, Jones of Cecil, Keefer, Ken-nard, King, Larsh, Mace, Markey, McComas, Mullikm, Wurray, Negley, Nyman, Parker, Purnell, Ridgely, Robi-nette, Russell, Sands, Schloy, Schlossor, Scott, Smith of Car-roll, Sneary, Stirling, Stockbridge, Sykes, Thomas, Thrus-ton, Valliant, Wickard, Wooden-53. NATS-Messrs. Berry, of Prince George's, Billingsley, Blackiston, Briscee, Brown, Chambers, Clarke, Crawford, Daid, Daniel, Davis of Charles, Dennis, Duvail, Edelen, Gale, Harwood, Hollyday, Horsey, Johnson, Lansdale, Lee, Mar-bury, Mitchel, Miller, Parran, Peter, Smith of Dorchester, Turner-27. YEAS-Messers. President (H. H. Goldsborough), Abbott,

#### Emancipation.

PROCLAMATIONS OF THE PRESIDENT. Of Sep. 22, 1862.\* I, ABRAHAM LINCOLN, President of the United

\* Major General McClellan's Order respecting it: HEADQUARTERS ARMY OF THE POTOMAC. October 7, 1862.

[General Order No. 163.] The attention of the officers and soldiers of the army of

States of America, and Commander-in-Chief of the army and navy thereof, do hereby proclaim and declare that hereafter, as heretofore, the war will be prosecuted for the object of practically restoring the constitutional relation between the United States and each of the States and the people thereof, in which States that relation is or may be suspended or disturbed.

That it is my purpose, upon the next meeting of Congress, to again recommend the adoption of a practical measure tendering pecuniary aid to the free acceptance or rejection of all slave States, so called, the people whereof may not then be in rebellion against the United States, and which States may then have voluntarily adopted, or thereafter may voluntarily adopt, immediate or gradual abolishment of slavery within their respective limits; and that the effort to colonize persons of African descent with their consent upon this continent or elsewhere, with the previously obtained consent of the Governments existing there, will be continned.

That on the first day of January, in the year of our Lord one thousand eight hundred and sixty-three, all persons held as slaves within any State or designated part of a State, the people whereof shall then be in rebellion against the United States, shall be then, thenceforward, and forever free; and the Executive Government of the United States, including the military and

the Potomac is called to General Order No. 139, War Department, September 24th, 1862, publishing to the army

particular, september 24th, 1002, publishing to the army the President's proclamation of September 22d. A proclamation of such grave moment to the nation, offi-cially communicated to the army, affords to the General Commanding an opportunity of defining specifically to the officers and soldiers under his command the relation borne by all mersons in the millioner service of the United States by all persons in the military service of the United States towards the civil authorities of the Government.

The Constitution coufides to the civil authorities--legisla The Constitution counces to the civit authorities—legisla-tive, judicial, and executive—the power and duty of making, expounding, and executing the Federal laws. Armed forces are raised and supported simply to sustain the civil authorities, and are to be held in strict subordi-nation thereto in all respects. This fundamental law of our political system is essential to the security of our republican institutions, and should be thoroughly understood and observed by every soldier.

The principle upon which and the objects for which armies shall be employed in suppressing rebellion must be determined and declared by the authorities, and the Chief Executive, who is charged with the administration of the national affairs, is the proper and only source through which the views and orders of the Government can be made known to the armies of the nation.

made knowu to the armies of the nation. Discussion by officers and soldiers concerning public measures determined upon and declared by the Govern-ment, when carried beyond the ordnary, temperate, and respectful expression of opnion, tend greatly to impair and destroy the discipline and efficiency of the troops, by sub-stituting the spirit of political faction for the firm, steady, and earnest support of the authority of the Government, which is the highest duty of the American soldier. The remedy for political errors, if any are committed, is to be found only in the action of the people at the polls. In thus calling the attention of this army to the true relation between the soldiers and the Government, the General Commanding merely adverts to an evil against which it has been thought advisable during our whole his-tory to guard the armies of the Republic, and in as doing

tory to guard the armies of the Republic, and in so doing he will not be considered, by any right minded person, as casting any reflection upon that loyalty and good conduct which has been so fully illustrated upon so many battlefields.

In carrying out all measures of public policy this army will, of course, be guided by the same rules of mercy and Christianity that have ever controlled its conduct towards the defenceless.

By command of Maj. Gen. McClellan

JAMES A. HARDIE, Licut. Col. Aid-de-Camp and Act'g Ass't Adjt. General.

naval authority thereof, will recognize and maintain the freedom of such persons, and will do no act or acts to repress such persons, or any of them, in any efforts they may make for their actual freedom.

That the Executive will, on the first day of January aforesaid, by proclamation, designate the States and parts of States, if any, in which the people thereof respectively, shall then be in rebellion against the United States; and the fact that any State, or the people thereof, shall on that day be, in good faith, represented in the Congress of the United States by members chosen thereto at elections wherein a majority of the qualified voters of such State shall have participated, shall, in the absence of strong countervailing testimony, be deemed conclusive evidence that such State, and the people thereof, are not in rebellion against the United States.

That attention is hereby called to an act of Congress entitled "An act to make an additional article of war," approved March 13, 1862, and which act is in the words and figures following:

"Be it enacted by the Senate and House of Representatives of the United States of America in Congress assembled, That hereafter the following shall be promulgated as an ad-ditional article of war, for the government of the army of the United States, and shall be obeyed and observed as such.

"ARTICLE -.. All officers or persons in the military or naval service of the United States are prohibited from em-ploying any of the forces under their respective commands for the purpose of returning fugitives from service or labor who may have escaped from any persons to whom such ser-vice or labor is claimed to be due, and any officer who shall be found guilty by a court-martial of violating this article shall be dismissed from the service. "Sec. 2. And be it further engeded. That this act shall

"SEC. 2. And be it further enacted, That this act shall take effect from and after its passage."

Also to the ninth and tenth sections of an act entitled "An act to suppress insurrection, to punish treason and rebellion, to seize and confiscate property of rebels, and for other purposes," approved July 17, 1862, and which sections are in the words and figures following :

"SEC. 9. And be it further enacted, That all slaves of per-sons who shall hereafter he engaged in rebellion against the Government of the United States or who shall he any way give all or comfort thereto, escaping from such persons and taking refuge within the lines of the army; and all here work when the lines of the army; and all and taking refuge within the index of the army; and all laves captured from such persons or descrited by them, and coming under the control of the Government of the United States; and all slaves of such persons found on[or]being within any place occupied by rebel forces and after-wards occupied by the forces of the United States, shall be deemed captives of war, and shall be forever free of their sometime up and endexing back of a concerned to the states of the states

deemed captives of war, and shall be forever free of their servitude, and not again held as slaves. "Scc. 10. And be it further enacted, That no slave escap-ing into any State, Territory, or the District of Columbia, from any other State, shall be delivered up, or in any way impeded or hindered of his liberty, except for crime, or some offence against the laws, unless the person claiming wid furifue chall for male or the thet the never territory. said fugitive shall first make oath that the person to whom the labor or service of such fugitive is alleged to be due is his lawful owner, and has not borne arms against the United States in the present rebellion, nor in any way given aid and confort thereto; and no person engaged in the military or naval service of the United States shall, under any pretence whatever, assume to decide on the validity of the claim of any person to the service or labor of any other person, or surrender up any such person to the claimant, on pain of being dismissed from the service."

And I do hereby enjoin upon and order all persons engaged in the military and naval service of the United States to observe, obey, and enforce, within their respective spheres of service, the act and sections above recited.

And the Executive will in due time recom-

shall have remained loyal thereto throughout the rebellion shall (upon the restoration of the constitutional relation between the United States and their respective States and people, if that relation shall have been suspended or disturbed) be compensated for all losses by acts of the United States, including the loss of slaves.

In witness whereof, I have hereunto set my hand, and caused the seal of the United States to be affixed.

Done at the city of Washington this twentysecond day of September, in the year of our Lord one thousand eight hundred and sixtytwo, and of the Independence of the United States the eighty-seventh.

### ABRAHAM LINCOLN.

By the President:

### WILLIAM H. SEWARD, Secretary of State.

#### January 1, 1863.

WHEREAS, on the twenty-second day of September, in the year of our Lord one thousand eight hundred and sixty-two, a proclamation was issued by the President of the United States, containing, among other things, the following, to wit:

"That on the first day of January, in the year of our Lord one thousand eight hundred and sixty-three, all per-sons held as slaves within any State or designated part of a State, the people whereof shall then be in rebellion against the United States, shall be then, theneeforward, and forever, free; and the Executive Government of the United States, including the military and naval authority thereof, will recognize and maintain the freedom of such persons, and will do no act or acts to repress such persons, or any of them, in any efforts they may make for their acor any of them, in any efforts they may make for their actual freedom.

"That irredom. "That the Executive will, on the first day of January aforesaid, by proclamation, designate the States and parts of States, if any, in which the people thereof, respectively, shall then be in rebellion against the United States; and the fact that any State, or the people thereof, shall on that day be in good faith represented in the Congress of the United States, by members chosen thereto at elections wherein a majority of the qualified voters of such States shall have participated, shall, in the absence of strong countervailing testimony, be deemed conclusive evidence that such State, and the people thereof, are not then in rebellion against the United States."

Now, therefore, I, ABRAHAM LINCOLN, President of the United States, by virtue of the power in me vested as Commander-in-Chief of the Army and Navy of the United States, in time of actual armed rebellion against the authority and Government of the United States, and as a fit and necessary war measure for snppressing said rebellion, do, on this first day of January, in the year of our Lord one thousand eight hundred and sixty-three, and in accordance with my purpose so to do, publicly proclaimed for the full period of one hundred days from the day first above mentioned, order and designate as the States and parts of States wherein the people thereof, respectively, are this day in rebellion against the United States, the following, to wit:

Arkansas, Texas, Louisiana, (except the parishes of St. Bernard, Plaquemines, Jefferson, St. John, St. Charles, St. James, Ascension, Assumption, Terre Bonne, Lafourche, St. Mary, St. Martin, and Orleans, including the city of New Orleans,) Mississippi, Alabama, Florida, Georgia, South Carolina, North Carolina, and Virginia, (except the forty-eight counties demend that all citizens of the United States who | signated as West Virginia, and also the counties

of Berkeley, Accomac, Northampton, Elizabeth City, York, Princess Ann, and Norfolk, including the cities of Norfolk and Portsmouth,) and which excepted parts are for the present left precisely as if this proclamation were not issued.

And by virtue of the power and for the purpose aforesaid, I do order and declare that all persons held as slaves within said designated States and parts of States are, and henceforward shall be, free; and that the Executive Government of the United States, including the military and naval authorities thereof, will recognize and maintain the freedom of said persons.

And I hereby enjoin upon the people so declared to be free to abstain from all violence, unless in necessary self-defence; and I recommend to them that, in all cases when allowed, they labor faithfully for reasonable wages.

And I further declare and make known that such persons, of suitable condition, will be received into the armed service of the United States to garrison forts, positions, stations, and other places, and to man vessels of all sorts in said service.

And upon this act, sincerely believed to be an act of justice, warranted by the Constitution upon military necessity, I invoke the considerate judgment of mankind and the gracious favor of Almighty God.

In witness whereof, I have hereunto set my hand and caused the seal of the United States to be affixed.

Done at the city of Washington this first day of January, in the year of our Lord one thousand eight hundred and sixty-three, and of the independence of the United States of America the eighty-seventh.

ABRAHAM LINCOLN.

### By the President :

WILLIAM H. SEWARD, Secretary of State.

VOTE ON THE EMANCIPATION PROCLAMATION.

Third Session, Thirty-Seventh Congress.

### IN HOUSE.

1862, December 11-Mr. YEAMAN, of Ky., offered the following resolutions:

Resolved by the House of Representatives, (the Senate concurring.) That the proclamation of the President of the United States, of date the 224 of September, 1862, is not warranted by the Constitution. *Resolved*, That the policy of emancipation, as indicated in that proclamation, is not calculated to hasten the resto-

Resolved, That the policy of emancipation, as indicated in that proclamation, is not calculated to hasten the restoration of peace, was not well chosen as a war measure, and is an assumption of power dangerous to the rights of citizens and to the perpetuity of a free people.

Which were laid on the table—yeas 94, nays 45, as follows:

45, as follows: Yras-Messrs. Aldrich, Alley, Arnold, Ashley, Babbitt, Baker, Baxter, Beaman, Bingham, Samuel S. Blair, Blake, Buffinton, Burnham, Campbell, Casey, Chamberlain, Clark, Covode, Cutler, Davis, Dawes, Duell, Dunn, Edgerton, Eliot, Ely, Fenton, Samuel C. Fessenden, Thomas A. D. Fessenden, Franchot, Frank, Gooch, Goodwin, Gurley, Haight, Hale, Hickman, Horton, Hutchins, Julian, Kelley, Francis W. Kellogg, William Kellogg, Killinger, Lansing, Leary, Lehman, Loomis, Lovejoy, Low, McKnight, McPherson, Mitchell, Muorhead, Anson P. Morrill, Justin S. Morrill, Noelt, Olin, Patton, Timothy G. Phelps, Pike, Pomeroy, Porter, Potter, Alexander H. Rice, John H. Rice, Riddle, Edward H. Rollins, Sargent, Sedgwick, Shanks, Shellabarger, Sloan, Spaulding, Stovens, Stratton, Train, Trimble, Trowbridge, Vandever, Van Horn, Verree, Walker, Wall, Wallace, Washburne, Albert S. White, Wilson, Windom, Worcester-94.

NAYS-Messrs. William J. Allen, Ancona, Baily, Biddle, Clements, Cox, Crisfield, Crittenden, Delaplaine, Dunlap, English, Fouke, Granger, Grider, Hall, Harding, Harrison, Holman, Knapp, Lavo, Lazear, Mallory, May, Maynard, Menzies, Morris, Noble, Norton, Odell, Pendleton, Robinson, Sheffield, Shiel, Smith, John B. Steele, William G. Steele, Stiles, Benjamin F. Thomas, Vallandigham, Voorhees, Wadsworth, Chillon A. White, Wickliffe, Woodruff, Yeaman-45.

1862, December 15-Mr. S. C. FESSENDEN, of Maine, offered the following resolutions :

Resolved, That the proclamation of the President of the United States, of the date of 22d September, 1862, is warranted by the Constitution.

Resolved, That the policy of emancipation, as indicated in that proclamation, is well adapted to hasten the restoration of peace, was well chosen as a war measure, and is an exercise of power with proper regard for the rights of the States, and the perpetuity of free government.

Which were adopted—yeas 78, nays 51, as follows:

follows: YEAS-Messrs. Aldrich, Alley, Arnold, Babbitt, Baker, Beaman, Bingham, Samuel S. Biair, Blake, Bufinton, Burnham, Chamberlain, Clark, Colfax, Frederick A. Conkling, Roscoe Conkling, Cutler, Dawes. Delano, Duell, Edgerton, Edwards, Eliot, Ely, Fenton, Sannel C. Fessenden, Thomas A. D. Fessenden, Fisher, Franchot, Frank, Gooch, Goodwin, Gurley, Haight, Iliekman, Hooper, Hutchins, Julian, Kelley, Francis W. Kellogg, William Kellogg, Killinger, Lansing, Loomis, Lovejoy, Low, McFherson, Mitchell, Moorhead, Anson P. Morrill, Justin S. Morrill, Noell, Patton, Pike, Pomeroy, Porter, Potter, Alexander H. Rice, John II. Rice, Riddle, Edward H. Rollins, Sargent, Sedgvick, Shellabarger, Sloan, Spaulding, Train, Trowbridge, Van Valkenburgh, Van Wyck, Verree, Walker, Wall, Wushburne, Albert S. White, Wilson, Windom, Worcester-78.

ter – 78. NATS – Messrs. William J. Allen, Ancona, Baily, Biddle, Calvert, Cobb, Coz, Cravens, Crisfield, Crittenden, Dunlap, English, Fouke, Granger, Grider, Hall, Harding, Harrison, Holman, Kerrigan, Knapp, Leary, Mallory, Maynard, Menzies, Morris, Noble, Norton, Oddil, Pendleton, Perry, Price, Richardson, Robinson, James S. Rollins, Sheffield, Shiel, Smith, John B. Steele, Stiles, Benjamiu F. Thomas, Francis Thomas, Tallandigham, Wadsworth, Wurd, Whaley, Chilton A. White, Wicklife, Woodruff, Wright, Yeaman–51.

# First Session, Thirty-Eighth Congress. IN SENATE.

1864, Feb. 10-Mr. CLARK introduced to the Senate a bill ratifying the emancipation proclamation and giving it the force of statute; which was referred to the Committee on Slavery and Freedmen. (See Reconstruction bill for a vote of the Senate.).

# IN HOUSE.

1863, Dec. 14—Mr. ARNOLD introduced this bill:

A bill to aid the President of the United States in carrying into more immediate execution the proclamation of emancipation issued by him on the first day of January, A. D. 1863, prohibiting the holding of certain persons as slaves in all that portion of the United States designated thereip.

Whereas the President of the United States, by his proclamation issued on the 1st day of January, in the year 1863, as Commander-in-Chief of the Army thereof, did, as a fit and lawful means of suppressing the rebellion, in accordance with the laws of war and with the dictates of justice and humanity, order, proclaim, and declare that all persons held as slaves within the limits of certain States and parts of States therein designated were, and should thereafter and forever be, free; and that the executive, military, and naval authorities would and should theneeforward recognize and maintain the freedom of all such persons; and whereas by said proclamation and order the President has guarantied to all such persons their freedom, and has pledged the faith and honor of the country that their freedom shall be recognized and forever maintained; and whereas it is the right and the duty of Congress to make all laws which may be necessary and proper for carrying into execution all the powers, whether civil or military, vested by the Constitution in the President as Gommilitary powers is that of making and executing the proc-lamation aforesaid; and whereas all persons heretofore held as slayes, as aforesaid, within said designated States or parts of States are now of right free, and ought to be hereafter forever unmolested in the enjoyment of that freedom which the Government of the United States is bound to "recognize and maintain :"

Now, therefore, for the purpose of carrying into more complete and immediate execution the aforesaid proclamation, and to secure forever the recognition and maintenance of the freedom of all persons designated therein, and thereby to provide more effectually for the suppression of the rehel-lion, the securing of domestic tranquillity, the maintaining of the common defence, and the preservation of the liber-

of the common detends and the preservation of the model ties of the people: Be it enacted by the Senate and House of Representatives of the United States of America in Congress assembled, That in all States and parts of States designated in said procla-mation as in rebollion, the re-enslaving or holding, or attempting to bold, in slavery or in involuntary servitude of any person who shall have been made or declared to be free by said proclamation, or any of their descendants, from and after the date of said proclamation, otherwise that in pun-ishment of crime whereof the accused shall have been duly convicted, is and shall be forever prohibited, any law or regulation of either of such States to the contrary notwithstanding

Which was referred to the Committee on the Judiciary.

#### RESOLUTIONS OFFERED.

# Second Session, Thirty-Seventh Congress. IN HOUSE.

1861, December 3-Mr. SHELLABARGER offered these resolutions:

Whereas the Constitution has wisely withdrawn from Congress the command of the armies of the United States; and. after they have been called forth, organized, and disciplined, under the rules which it is competent for Congress to make, and which it has made for the government of the land and naval forces, has placed that supreme command exclusively in the President of the United States: Therefore.

Resolved, That it is neither within the province of this House, nor would it be wise, to lay down any rules whatever regulating or attempting to control any part of that chief command of our armies in the field, either in the matter as to what "military necessities" do or do not exist as to the treatment or use of slaves within any of the States, nor in any other matter whatever affecting the supreme command of the President

Resolved, That whilst this House refrains from all attempts to assume any of the functions of the Commanderiu-Chief of the Army, and hereby expresses its confidence in the wisdom, prudence, and patriotism of the President, as indicated in the discharge of the most responsible and arduous duties of the present Executive term, and its wil-lingness to continue to him the untrammelled exercise of his powers as the Commander-in-Chief of the Army and Navy, yet the House deems it appropriate to express its earnest sense and conviction that in the prosecution of this war no resort, on the one hand, should be had to a plea of "military necessity" as a cover for any violation of any con-stitutional right of any citizen of the United States, either in slaves or any other right of the citizen; nor, on the other hand, should any privileges or protections whatever be extended by our armise either to the title in slaves or in any property whatever which is incompatible with the safety of the Government itself or with the success of our arms in suppressing the rebellion; but all individual rights of property, when necessary, should, in the prosecution of the war, be made to yield to the paramount right of the Government to re-establish the authority of the Constitution over all the people and States of the Union.

Which were postponed, and not voted upon.

December 4-Mr. CONWAY, on leave, introduced the following joint resolution :

Joint resolution in relation to persons claimed as slaves in the States now in rehellion.

Whereas the belligerent character acquired by the States whereas the configerent contracter acquired by the States now in rebellion confers upon them a recognized status among nations, in contravention of their Federal status, and arising from incompatible relations, which, though not im-plying nationality, is, nevertheless, one from which they derive important advantage in making war upon the United States; and whereas the Federal Government is by this fact fully exonerated from respecting, in any manner whatso-

ever, the constitutional obligations it would otherwise be under to said States : Therefore

Be it resolved by the Scrate and House of Representatives of the United States of America in Congress assembled, That the President be requested not to permit the enforcement of any law or usage under which persons may be claimed as slaves within any of suid States into which the enthority of the United States may be extended, and that he be requested the oniced States may be extended, and that he pereducated to declare, by proclamation, all persons so claimed to be free, and to employ them in the service of the United States in any capacity to which they may be suited, and in such numbers as the public service may require. *Resolved, further*, That the faith of the nation be pledged to all persons sustaining loss through the operation of this measure made are used aball require the invest to the United

measure who are now and shall remain loyal to the United States.

The resolutions were postponed.

Third Session, Thirty-Seventh Congress.

1863, January 12-Mr. Wilson introduced a joint resolution to approve, ratify, and confirm the proclamation issued by the President, as Commander-in-Chief of the Army and Navy of

the United States, dated January 1, 1863. Mr. Cox moved that it be tabled; which was lost-yeas 50, nays 85. It was then referred to the Committee on the Judiciary.

#### First Session, Thirty-Eighth Congress.

1863, December 17-Mr. EDGERTON offered this resolution:

Whereas the proclamations of the President of January 1, 1863, and December 8, 1863, in relation to emancipation, impose conditions of pardon and amnesty to the persons who have participated in the existing rebellion, as well as conditions precedent to the establishment and recognition of State government in the States to which said proclamations apply, which in the judgment of a large number of faithful citizens, have a tendency to give to the rebellion "the advantage of a changed issue," and "to reinvigorate "the advantage of a change issue," and "to reinvigorate the otherwise declining insurrection in the South," and to prolong the war: and whereas this House cannot but re-gard with anxiety the unprecedented and extraordinary claims and assumption of high prerogative by the President in said proclamations, especially in view of the fact that the Decident in his immune address of the dth due of In said proclamations, especially in view of the lact that the Precident, in his inaugural address of the 4th day of March, 1861, declared, "I have no purpose directly or indi-rectly to interfere with the institution of slavery in the States where it exists; I believe I have no right to do so, and I have no inclination to do so." Therefore,

Resolved, As the judgment of this House, that the main-tenance inviolate of the constitutional powers of Congress, and the rights of the States, and especially the right of each State to order and control its own domestic institueach State to order and control its own domestic institu-tions according to its own judgment exclusively, is essential to the balance of power on which the perfection and endu-rance of our political fabric of federal union depends; and; we denounce, as among the gravest of crimes, the invasion, or occupation, by armed force, of any State, under the pre-text or for the purpose of coercing the people thereof to modify or abrogate any of their laws or domestic institu-tions that are consistent with the Constitution of the. United States; and we affirm the principle declared in this resolution to be a law, alike to the President and the peo-ple of the United States.

Mr. GRINNELL moved to table the resolution; which was agreed to-yeas 90, nays 66, as follows:

YEAS-Messrs. Alley, Allison, Ames, Anderson, Arnold, Ashley, John D. Baldwin, Beaman, Blaine, Blow, Boutwell, Ashley, John D. Baldyi Minkol, Minkol, Minkol, Massi Mikol, Massi Massi Mikol, Massi Massi Massi Mikol, Massi Mikol, Massi Mikol, Mikol, Massi Mikol, Massi Mikol, Massi Mikol, Massi Mikol, Massi Mikol, Mikol, Massi Mikol, Mikol, Massi Massi Mikol, Mikol, Mikol, Massi Mikol, Mikol, Mikol, Massi Mikol, Mikol, Massi Mikol, Mikol, Mikol, Mikol, Mikol, Mikol, Mikol, Massi Mikol, Mikol, Mikol, Mikol, Mikol, Mikol, Massi Mikol, Mikol, Mikol, Mikol, Massi Mikol, Windom, Woodbridge-90.

NAVS-Messrs. James C. Allen, Ancona, Augustus C. Baldwin, Bliss, Brooks, James S. Brown, Coz, Cravens, Dawson, Denison, Eden, Edgerton, Eldridge, Finck, Ganson, Grider, Griswold, Hail, Harding, Harrington, C. M. Harris, Herrick, Holman, William Johnson, Kernan, Kindy, Knapp, Law, Lazear, LeBlond, Long, Mallory, Marcy, McDowell, McKinney, Middleton, Wm. H. Miller, James R. Morris, Morrison, Nelson, Noble, John O'Neill, Pendleton, Perry, Radford, Samuel J. Randall, Robinson, Rogers, James S. Rollins, Ross, Scott, John B. Steele, Wm. G. Stele, Siles, Strouse, Stuart, Sweet, Voorhees, Wadsworth, Ward, Wheeler, Chilton A. White, Joseph W. White, Winfield, Fernando Wood, Yeaman-66.

1864, Jan. 25-Mr. EDGERTON offered the following preamble and resolutions, which were laid over under the rule :

Whereas this House on the 17th day of December last adopted, with but one dissentient vote, the following resolution, to wit: "*Resolved*, That we hold it to be the duty of Congress to pass all necessary bills to supply men and money, and the duty of the people to render every aid in their power to the constituted authorities of the Government in the crushing out of the rebellion, and in bringing the leaders thereof to condign punishment;"

Therefore in explanation of the foregoing resolution, and in further expression of the opinion and purpose of this House,

Resolved, That the aid hitherto liberally supplied in men and money by the people of the United States, to enable the Federal Executive to prosecute the existing civil war, has been so supplied by all citizens truly faithfui to the Federal Union and Constitution, for the purpose, and no other, expressed in the resolution adopted by Congress in July, 1861, declarative of the object of the war, and commonly known as "the Crittenden resolution," and public faith, true Christian humanity, and wise statesmanship alike demand strict adherence by the "constituted authorities of the Government" to the purpose or object of the war, as thus declared by Congress and accepted by the people. 2. That the demand of the President, in his proclamation

2. That the demand of the President, in his proclamation of December 8, 1863, that the people of the States wherein rebellion exists shall swear to abide by and support his proclamation of emancipation (in other words, change, or submit to the change, at his dictation, of their State constitutions, local laws, and domestic institutions, not inconstitutions, local laws, and domestic institutions, not inconstitutions, local laws, and domestic institutions before ench States or their people will by him be considered to have ceased to be in rebellion, and entitled to pardon or annesty, and entitled to their constitutional rights of State government, in harmony with the Government of the United States, is, in the judgment of this House, an oppressive and unconstitutional demand, the tendency and effect of which, if persisted in and enforced by war, will be to substantially change the object and character of the war on the part of the Federal Government, from one to preserve, protect, and defend the Constitution of the United States as the supreme law of the land, to a revolutionary war against the constitutional rights and sovereignty of Federal States, and virtually subversive of the constitutional Government of the United States; and of such a war we now record our disapproval.

3. That in view of the immense power of war demanded by the President and supplied to him by a patriotic people, and hitherto wielded by him according to his own will, with little deference or regard to the opioions and convictions of the very large number; if not majority; of faithful Union citizens in the United States who have doubted or disapproved his policy in the conduct of the war and his extraordinary assumptions of executive power, and in view of the dangers to constitutional liberty and the manifold evils that ever attend civil war, we desire peace, and the replacement under its healthful and benign influence, with the least possible further waste of blood and treasure of the people, of all the relations and functions of constitutional government, State and Federal, now disturbed and endar government, State and Federal, now disturbed and endar government, state and a docordially invite and pledge our co-operation in negotiations, proposals, and efforts for peace upon the basis of a restoration of the Federal Union under the Constitution as it is, leaving to the tree constitutional action of the people the questions of annendments of the Federal Constitution as it is, leaving to the tree constitutional action of the people the questions of annendments of the Federal Constitution as it is, leaving to the free constitutional action of the people the questions of annendments of the Federal Constitution, and leaving also to the people of each State, as their unquestionable right, the right, and constitutions, laws, and domestic institutions in their own way, subject only to the Constitution of the United States.

# Respecting the Issue of the Proclamation.

THE PRESIDENT AND THE CHICAGO DEPUTATION.

1862, September 13—The President gave an audience to a deputation from all the religious denominations of Chicago, presenting a memorial for the immediate issue of an emancipation proclamation, which was enforced by some remarks by the chairman. The President replied :

The subject presented in the memorial is one upon which. I have thought much for weeks past, and I may even say for months. I am approached with the most opposite opinions and advice, and that by religious men, who are equally certain that they represent the Divine will. I an sure that either the one or the other class is mistaken in that belief, and perhaps, in some respect, both. I hope it will not be irreverent for me to say that if it is probable that God would reveal his will to others, on a point so connected with my duty, it night be supposed he would reveal it directly to me; for, unless I am more deceived in myself than I often am, it is my earnest desire to know the will of Providence in this matter. And if I can learn what it is I will do it! These are not, however, the days of miracles, and I suppose it will be granted that I am not to expect a direct revelation. I must study the plain physical facts of the case, as cortain what is possible, and learn what appears to be wise and right.

The subject is difficult, and good men do not agree. For instance, the other day four gentlemen of standing and intelligence from New York called as a delegation on business connected with the war; but before leaving two of them earnestly besought me to proclaim general enancipation, upon which the other two at once attacked them. You know also that the last session of Congress had a decided majority of anti-slavery men, yet they could not unite on this policy. And the same is true of the religious people. Why, the rebel soldiers are praying with a great deal more earnestness, I fear, than our own troops, and expecting God to favor their side; for one of our soldiers who had been taken prisoner toil Senator Wilson a few days since that he met nothing so discouraging as the evident sincerity of those he was among in their prayers. But we will talk over the merits of the case.

What good would a proclamation of emancipation from me do, especially as we are now situated? I do not want to issue a document that the whole world will see must neto issue a document that the whole world will see must ne-cessarily be inoperative, like the Pope's bill against the comet! Would my word free the slaves, when I cannot even enforce the Constitution in the rebel States? Is there a single court, or magistrate, or individual that would be influenced by it there? And what reason is there to think it would have any greater effect upon the slaves than the lote here Generative which the sparse of which effect late law of Congress, which I approved, and which offers protection and freedom to the slaves of rebel masters who come within our lines? Yet I cannot learn that that law has caused a single slave to come over to us. And suppose they could be induced by a proclamation of freedom from me to throw themselves upon us, what should we do with them? How can we feed and care for such a multitude? General Butler wrote me a few days since that he was issuing more rations to the slaves who have rushed to him than to all the white troops under his command. They eat, and that is all; though it is true General Butler is feeding the whites also by the thousand; for it nearly amounts to a famine there. If, now, the pressure of the war should call off our forces from New Orleans to defend some other point, what is to prevent the masters from reducing the blacks to slavery again; for I am told that whenever the rebels take any black prisoners, free or slave, they immediately auction them off! They did so with those they took from a boat that was aground in the Tennessee river a few days ago. And then I am very ungenerously attacked for it! For in-stance, when, after the late battles at and near Bull Run, an expedition went out from Washington under a flag of truce to bury the dead and bring in the wounded, and the rebeis seized the blacks who went along to help, and sent them into slavery, Horace Greeley said in his paper that the Gov-ernment would probably do nothing about it. What could I do?

Now, then, tell me, if you please, what possible result of good would follow the issuing of such a proclamation as you desire? Understand, I raise no objections against it on legal or constitutional grounds, for, as Commander-in-Chief of the Army and Navy, in time of war I suppose I have a right to take any measure which may best subdue the enemy, nor do I urge objections of a moral nature, in view of possible consequences of insurrection and massacre at the South. I view this matter as a practical war measure, to be decided on according to the advantages or disadvantages it may offer to the suppression of the rebellion.

The committee replied to these remarks, and the President responded :

I admit that slavery is at the root of the rebellion, or at least its sine qua nom. The ambition of politicians may have instigated them to act, but they would have been im-potent without slavery as their instrument. I will also concede that emancipation would help us in Enrope, and convince them that we are incited by something more than ambition. I grant, further, that it would help somewhat at the North, though not so much, I fear, as you and those you represent imagine. Still, some additional strength would he added in that way to the war, and then, unquestionably, it would weaken the rebels by drawing off their laborers which is of great importance; but I am not so sure we could do much with the blacks. If we were to arm them, I fear that in a few weeks the arms would be in the hands of the rebels; and, indeed, thus far, we have not had arms enough to equip our white troops. I will mention another thing, though it meet only your scorn and contempt. There are 50,000 bayonets in the Union Army from the border slave States. It would be a serious matter if, in consequence of a proclamation such as you desire, they should go over to the rebels. I do not think they all would—not so many, indeed, as a year ago, or as six months ago-not so many to-Every day increases their Union feeling. day as yesterday. They are also getting their pride enlisted, and want to beat the rehels. Let me say one thing more: 1 think you should admit that we already have an important principle to rally and unite the people, in the fact that constitutional government is at stake. This is a fundamental idea going down about as deep as anything.

After further remarks from the committee, the President said:

Do not misunderstand me because I have mentioned these objections. They indicate the difficulties that have thus far prevented my action in some such way us you desire. I have not decided against a proclamation of liberty to the slaves, but hold the matter under advisement. And I can assure you that the subject is on my mind, by day and night, more than any other. Whatever shall appear to be God's will I will do. I trust that in the freedom with which I have canvassed your views I have not in any respect injured your feelings.

#### REMARKS IN THE HOUSE.

### First Session, Thirty-Eighth Congress.

1864, June 25-Mr. BOUTWELL, of Massachusetts: The gentleman from Kentucky [Mr. Mallory] said this morning that the whole policy of the country was changed by the proclamation of the President, and he attributed that proclamation to the meeting of the Governors of certain States at Altoona. I am not here to be put upon the witness stand, but it so happens that I have the means of knowing that the proclamation of September, 1862, was entirely independent of and antecedent to the neeting of the Governors at Altoona. The meeting of the Governors had no connection with the proclamation. The gentleman from Kentucky should remember that prior to the issuing of that we had endured many, many reverses. Lee had battled for four days under the fortifications of the capital, and had finally crossed the Potomae into Maryland. It was not until the country put itself on the side of justice that it had a right to expect the favor of Divine Providence, or any of those successes which have rendered this war glorious in the cause of freedont.

the cause of freedom, truth, and justice. Mr. MALLORY. Will the gentleman state when that convention of Governors assembled at Altoona?

Mr. BOUTWELL. I think it assembled at Altoona previous to the 22d of September, but I assert as within my own knowledge that the issuing of the proclamation was determined upon previous to the meeting at Altoona.

Mr. MALLORY. Can the gentleman inform me when the issuing of that proclamation was determined upon?

Mr. Borrweil. I cannot go far in this matter, I assert distinctly the fact which is within my own knowledge that the President, previous to the meeting of the Governors at Altoona, had decided in a certain contingency, which happened upon the Wednesday preceding the 22d of September, to issue the proclamation, and therefore the inference I draw is in contravention of the declaration of the gentleman from Kentucky that that proclamation was the result of the meeting of the Governors at Altoona.

of the meeting of the Governors at Altoona. Mr. MALLORX. Will the gentleman tell us the contingency on the happening of which that proclamation was to be issued?

Mr. BOUTWELL. I said, Mr. Speaker, when I mentioned

this fact, that I was not to be put upon the stand as a witness. I have made a statement as of a fact within my own knowledge, and history will confirm the statement.

Mr. MALLONY. If the gentleman from Massachusetts does not wish to answer the question or to state the fact, I will not insist.

ADDRESS OF LOYAL GOVERNORS TO THE PRESIDENT,

Adopted at a meeting of Governors of loyal States, held to take measures for the more active support of the Government, at Altoona, Pennsylvania, on the 22d day of September, 1862.

After nearly one year and a half spent in contest with an armed and gigantic rebellion against the national Government of the United States, the duty and purpose of the loyal States and people continue, and must always remain as they were af its origin—namely, to restore and perpetuate the authority of this Government and the life of the nation. No matter what consequences are involved in our fidelity, this work of restoring the Kepublic, preserving the institutions of democratic liberty, and justifying the hopes and toils of our fathers shall not fail to be performed. And we pledge without hesitation, to the President of the United States, the most loyal and cordial support, hereafter as

And we pledge without hesitation, to the President of the United States, the most loyal and cordial support, hereafter as heretofore, in the exercise of the functions of his great office. We recognize in him the Chiof Executive Magistrate of the nation, the Commander-in-Chief of the Army and Navy of the United States, their responsible and constitutional head, whose rightful authority and power, as well as the constitutional powers of Congress, must be rigorensly and religiously guarded and preserved, as the condition on which alone our form of Government and the constitutional rights and liberties of the people themselves can be saved from the wreck of anarcopy or from the gulf of despotism.

In submission to the laws which may have been or which may be duly enacted, and to the lawful orders of the President, co-operating always in our own spheres with the national Government, we mean to continue in the most vigorous exercise of all our lawful and proper powers, contending against treason, rebellion, and the public enemies, and, whether in public life or in private station, supporting the arms of the Union, until its cause shall conquer, until final victory shall perch upon its standard, or the rebel foe shall yield a dutiful, rightful, and unconditional aubmission.

And, impressed with the conviction that an army of reserve ought, until the war shall end, to be constantly kept on foot, to be raised, armed, equipped, and trained at home, and ready for emergencies, we respectfully ask the President to call for such a force of volunteers for one year's service, of not less than one hundred thousand in the aggregate, the quota of each State to be raised after it shall have filled its quota of the requisitions already made, both for volunteers and militia. We believe that this would be a measure of military education of the people.

We hail with heartfelt gratitude and encouraged hope the proclamation of the President, issued on the 22d instant declaring emancipated from their bondage all persons held to service or labor as slaves in the rebel States, whose rebellion shall last until the first day of January now next ensuing. The right of any person to retain authority to compel any portion of the subjects of the national Government to rebel against it, or to maintain its enemies, implies in these who are allowed possession of such authority the right to rebel themselves; and therefore the right to estab-lish martial law or military government in a State or Territory in rebellion implies the right and the duty of the Government to liberate the minds of all men living therein by appropriate proclamations and assurances of protection, in order that all who are capable, intellectually and morally, of loyalty and obedience, may not be forced into treason as the unwilling tools of rebellious traitors. To have continued indefinitely the most efficient cause, support, and stay of the rebellion, would have been, in our judgment, unjust to the loyal people whose treasure and lives are made a willing sacrifice on the altar of patriotism—would have discriminated against the wife who is compelled to surrender her husband, against the parent who is to surrender his child to the hardships of the camp and the perils of battle, in favor of redel masters permitted to retain their slaves. It would have been a final decision alike against humanity, justice, the rights and dignity of the Government, and against sound and wise national policy. The decision of the President to strike at the root of the rebellion will lend new President to strike at the root of inercontain will lead new vigor to the efforts and new life and hope to the hearts of the people. Cordially tendering to the President our re-spectful assurances of personal and official confidence, we trust and believe that the policy now inaugurated will be crowned with success, will give speedy and triumphant vic-tories over our enemies, and secure to this nation and this people the blessing and favor of Almighty God. We believe that the blood of the herces who have already fallen, and

The splendid valor of our soldiers, their patient endur-ance, their manly patriotism, and their devotion to duty, demand from us and from all their countrymen the homage of the sincerest gratitude and the pledge of our constant reinforcement and support. A just regard for these brave men, whom we have contributed to place in the field, and for the importance of the duties which may lawfully pertain for the importance of the titles which they have by growing to to us hereafter, has called us into friendly conference. And now, presenting to our national Chief Magistrate this conclusion of our deliberations, we devote ourselves to our country's service, and we will surround the President with the title they devide the dedicity and real of our constant support, trusting that the fidelity and zeal of the loyal States and people will always assure him that he will be constantly maintained in pursuing with the utmost vigor this war for the preservation of the national life and the home of humanity. the hope of humanity.

A. G. CURTIN JOHN A. ANDREW, RICHARD YATES, ISRAEL WASHBURNE, JR., EDWARD SOLOMON SAMUEL J. KIRKWOOD, 0. P. MORTON, By D. G. Rose, his representative, WM. SPRAGUE. F. H. PEIRPOINT, DAVID TOD, N. S. BERRY AUSTIN BLAIR.

#### LETTER FROM CHARLES SUMNER.

SENATE CHAMBER, June 5, 1862. MY DEAR SIR: Your criticism of the President is hasty I am confident that, if you knew him as I de, you would not make it.

Of course, the President cannot be held responsible for the misfeasances of subordinates, unless adopted or at least tolerated by him. And I am sure that nothing unjust or ungenerous will be tolerated, much less adopted, by him.

ungenerous will be tolerated, much less adopted, by him. I am happy to let you know that he has no sympathy with Stanly in his absurd wickedness, closing the schools, nor again in his other act of turning our camp into a hunt-ing ground for slaves. He reputates both-positively. The latter point has occupied much of his thought; and the convergence here act cane to far in reacredian bio sc the newspapers have not gone teo far in recording his repeated declarations, which I have often heard from his own lips, that slaves finding their way into the national lines are never to be re-enslaved. This is his conviction, expressed without reserve.

Could you have seen the President-as it was my privilege often-while he was considering the great questions on which he has already acted-the invitation to emancination in the States, emancipation in the District of Celumbia, and the acknowledgment of the independence of Hayi and Liberia-even your zeal would have been satisfied, for you would have felt the sincerity of his purpose to do what he could to carry forward the principles of the Declaration of Independence. His whole soul was occupied, especially by the first proposition, which was peculiarly his own. In familiar intercourse with him, I remember nothing more touching than the earnestness and completeness with which he embraced this idea. To his mind, it was just and beneficent while it promised the sure end of slavery. Of course, to me who had already proposed a bridge of gold for the re-treating fiend, it was most we come. Proceeding from the President, it must take its place among the great events of history. If you are disposed to be impatient at any seeming

those who may yet give their lives to their country, will short-comings, think, I pray you, of what has been done in a brief period, and from the past discern the sure arec, their manly patriotism, and their devotion to duty, may, perhaps, derive some assurance from my confidence. I may say to you, therefore, stand by the Administration. If need be, help it by word and act, but stand by it and have faith in it.

I wish that you really knew the President, and had heard the artless expression of his convictions on those questions which concern you so deeply. You might, per-haps, wish that he were less cautious, but you would be Believe me, therefore, you are wrong, and I regret it the more because of my desire to see all our friends stand firmly together.

If I write strongly it is because I feel strongly; for my constant and intimate intercourse with the President, be ginning with the 4th of March, not only binds me pecu-harly to his Administration, but gives me a personal as well as a political interest in seeing that justice is done him.

Believe me, my dear sir, with much regard, ever faithfully yours, CHARLES SUMNER.

#### LETTER FROM OWEN LOVEJOY.

The Boston Liberator publishes a letter from the late Owen Lovejoy, addressed to William Lloyd Garrison, under the date of Washington, February 22, 1864. In this letter Mr. Lovejoy says :

"I write you, although ill-health compels me to do it by the hand of another, to express to you my gratification at the position you have taken in reference to Mr. Lincoln. the position you have taken in feither to say in regard 1 am astifised, as the old theologians nsed to say in regard to the world, that if he is not the best conceivable Presi-dent, he is the best possible. I have known something of the facts inside during his Administration, and I know the facts inside during his Administration, and I know that he has been just as radical as any of his Cabinet. And although he does not do everything that you or I would like, the question recurs, whether it is likely we can elect a man who would. It is evident that the great mass of Unionists prefer him for re-election; and it seems to me certain that the providence of God, during another term, will grind slavery to powder. I believe now that the President is up with the average of the House. "Recurring to the President, there are a great many re-ports concerning him which seem to be reliable and authen-tic, which, after all, are not so. It was currently reported

bors concerning init which seem to be reinable and authen-tic, which, after all, are not so. It was currently reported among the anti-slavery men of Illinois that the emancipa-tion proclamation was extorted from him by the outward pressure, and particularly by the delegation from the Chris-tian Convention that met at Chicago. Now, the fact is this, so I had it from his corp like: Ukaded white the proclaas I had it from his own lips: He had written the procla mation in the snummer, as early as June, I think—but will not be certain as to the precise time—and called his Cabinet together, and informed them he had written it, and meant to make it, but wanted to read it to them for any criticism or remarks as to its features or details. After having done so, Mr. Seward suggested whether it would not be well for him to withhold its publication until after we had gained some substantial advantage in the field, as at that time we had met with many reverses, and it might be considered a cry of despair. He told me he thought the suggestion a wise one, and so held on the proclamation until after the battle of Antietam.

"I mention this as a sample of a great many others."

# THE FUGITIVE SLAVE LAWS. REPEAL OF

# "CONTRABANDS," AND KINDRED SUBJECTS.

The act of 1793 was passed by the following | vote:

### IN SENATE.

January 18, 1793-Without a call of the yeas and nays.

#### IN HOUSE.

February 5, 1793-Yeas 48, nays 7, as follows:

lows: YEAS-Messrs. Fisher Ames, John Baptist Ashe, Abraham Baldwin, Robert Barnwell, Egbert Benson, Elias Boudinot, Shearjashub Bourne, Benjamin Bourne, Abraham Clark, Jonathan Dayton, William Findley, Thomas Fitzsimons, Elbridge Gerry, Nicholas Gilman, Benjamin Goodhue, James Gordon, Christopher Greenup, Andrew Gregg, Samuel Grif-fin, William Barry Grove, Thomas Harlley, James Hill-house, William Hindman, Daniel Huger, Israel Jacobs, Philip Key, Aaron Kitchell, Amasa Learned, Richard Bland Lee, George Leonard, Nathaniel Macon, Andrew Moore, Frederick Angustas Muhlenberg, William Yans Murray, Alexander D. Orr, John Page, Cornelius C. Schoonmaker, Theodore Sedgwick, Peter Silvester, Israel Smith, William Smith, John Steele, Thomas Sumpter, Thomas Tudor Tucker, Jeremiah Wadsworth, Alexander White, Hugh Williamson, Francis Willis-43. Francis Willis-48.

NAYS-Samuel Livermore, John Francis Mercer, Nathan-iel Niles, Josiah Parker, Jonathan Sturges, George Thatcher, Thomas Tredwell-7.

GEORGE WASHINGTON, President, approved it February 12, 1793.

The act of 1850 was passed by the following vote:

### IN SENATE.

August 23, 1850-yeas 27, nays 12, as follows:

10WS: YEAS-Messrs. Atchison, Badger, Barnwell, Eell, Berrien, Butler, Davis of Mississippi, Dawson, Dodge of Iowa, Downs, Foota, Houston, Huuter, Jones, King, Mangum, Mason, Pearce, Rusk, Sebastian, Soulé, Spruance, Sturgeon, Turney, Underwood, Wales, Yulee-27. NAVS-Messrs. Baldwin, Bradbury, Chase, Cooper, Davis of Massachusetts, Dayton, Dodge of Wisconsin, Greene, South Under Winthers Winthers 19

of Massachusetts, Dayton, Dodge of Smith, Upham, Walker, Winthrop-12.

#### IN HOUSE.

September 12, 1850-yeas 109, nays 76, as follows:

IOIIOWS: YEAS-Messrs. Nathaniel Albertson, William J. Alston, Josiah M. Anderson, William S. Ashe, Thomas H. Averett, William V. N. Bay, Thomas H. Bayly, James M. H. Beale, William H. Bissell, Franklin W. Bowdon, Richard I. Bowie, James B. Bowlin, Lynn Boyd, Daniel Breck, Albert G. Brown, William J. Brown, Alexander W. Buel, Armistead Burt, George Alfred Caldwell, Joseph P. Caldwell, Thomas L. Clingman, Williamson R. W. Cobb, William F. Colcock, John R. J. Daniel, Edmund Deberry, Milo M. Dimmick, Cyrus L. Dunham, Henry A. Edmundson, Samuel A. Eliot, Andrew Ewing, Winfield S. Featherston, Thomas J. D. Fuller, Meredith P. Gentry, Elbridge Gerry, Edward Gilbert, Willis A. Gorman, James S. Green, Willard P. Hall, William T. Hamilten, Hugh A. Haralson, Isham G. Harris, S. W. Har-

ris, Thomas L. Harris, Thomas S. Haymond, Harry Hibbard, Henry W. Hilliard, Moses Hoagland, Alexander R. Holla-day, Isaac E. Holmes, John W. Houston, Volney E. Howard, David Hubbard, Samuel W. Inge, Joseph W. Jackson, An-drew Johnson, James L. Johnson, Robert W. Johnson, George W. Jones, David S. Kaufman, John B. Kerr, Emile LaSère, Shepperd Leffler, Nathaniel S. Littlefield, Job Mann, Humphrey Marshall, John C. Mason, John B. Kerr, Emile LaSère, Shepperd Leffler, Nathaniel S. Littlefield, Job Mann, Humphrey Marshall, John C. Mason, John A. McClernand, Joseph E. McDonald, Edward W. McGaughey, James X. McLanahan, Finis E. McLean, Fayette McMullen, John McQueen, William McWillie, Richard K. Meade, John K. Miller, John S. Millson, Jeremiah Morton, James L. Orr, David Outlaw, Allen F. Owen, Richard Parker, Charles H. Peaslee, John S. Phelps, Paulus Powell, William A. Rich-ardson, John Robbins, Jr., Thomas Ross, John H. Savage, James A. Seddon, Augustine H. Shepperd, Edward Stanly, Frederick P. Stauton, Richard H. Stanton, John L. Taylor, James H. Thomas, Jacob Thompson, James Thompson, John B. Thompson, Robert Toombs, Abraham W. Venable, Hiram Walden, Daniel Wallace, Albert G. Watkins, Mar-shall J. Wellborn, Isaac Wildrick, Christopher H. Williams, Joseph A. Woodward, Timothy R. Young-109. Nars-Henry P. Alexander, Charles Allen, Edward, D. Baker, Henry Bennett, Kinsley S. Bingham, Walter Booth, George Briggs, Lorenzo Burrows, Thomas B. Butler, Joseph Cable, Samuel Calvin, Lewis D. Campbell, David K. Cartter, Joseph R. Chandler, Charles E. Clarke, Orasmus Cole, Moses B. Corwin, John Crowell, Jesse C. Dickey, David K. Gartter, Joseph R. Chandler, Charles E. Clarke, Orasmus Cole, Moses

George Briggs, Lorenzo Burrows, Thomas B. Butler, Joseph Cable, Sanuel Calvin, Lewis D. Campbell, David K. Cartter, Joseph R. Chandler, Charles E. Clarke, Orsamus Cole, Moses B. Corwin, John Crowell, Jesse C. Dickey, David T. Disney, Natlan F. Dixon, James Duane Doty, James H. Duncan, Charles Durkee, Nathan Bvans, Graham N. Fitch, Orin Fowler, John Freedly, Joshua R. Giddings, Daniel Gott, Herman D. Gould, Ransom Halloway, Moses Hampton, An-drew J. Harlan, Andrew K. Hay, William Hebard, William Henry, John W. Howe, William F. Hunter, William T. Jackson, George W. Julian, George G. King, James G. King, John A. King, Preston King, Horace Mann, Orsamus B. Matteson, Thomas McKissock, James Meacham, Henry D. Moore, Jonathan D. Morris, William Nelson, John Otis, Charles W. Pitman, Harvey Putnam, Robert R. Reed, John L. Robinson, Joseph M. Root, David Rumsey, jr., William A. Sackett, Cullen Sawtelle, Ab'm M. Schermerhorn, John L. Schoolcraft, Peter H. Silvester, William Sprague, Thad-deus Stevens, Charles Stetson, John R. Thurman, Amos Tuck, Walter Underhill, Samuel F. Vinton, Loren P. Waldo, John Wentworth, William A. Whittlesey, Amos E. Wood, George W. Wright-76. MILLARD FILLMORE, President, approved it,

MILLARD FILLMORE, President, approved it, September 18, 1850.

#### MOVEMENTS FOR ITS REPEAL.

#### First Session, Thirty-Second Congress.

#### IN SENATE.

1852, Aug 26-The civil and diplomatic bill pending,

Mr. SUMNER offered an amendment to add to a section appropriating money to pay ministerial officers of the United States extraordinary expenses incurred, this proviso :

That no such allowance shall be authorized for any expenses incurred in executing the act of September 18, 1850, for the surrender of fugitives from service or labor; which said act is hereby repealed.

lows:

YEAS-Messrs. Chase of Ohio, Hale of N. H., Sumner of Mass., Wade of Ohio-4. NATS-Messrs. Adams of Miss., Badger of N. C., Bayard

Mass., Wade of Ohio, Hale of N. H., Sumler of Mass., Wade of Ohio, -4. Nars.-Messrs. Adams of Miss., Badger of N. C., Bayard of Del., Bell of Tenn, Borland of Ark, Brabhury of Maine, Bright of Ind., Brodhead of Penn., Brooke of Miss., Butler of S. C., Cass of Mich., Charlton of Geo., Clark of R. I., Clemens of Ala., Cooper of Penn., Dawson of Geo., DeSaus-sure of S. C., Dodge of Iowa, Douglass of III., Fitch of Mich., Fish of N. Y., Geyer of Mo., Gwin of Cal., Hamlin of Maine, Houston of Texas, Hunter of Va., James of R. I., Jones of Iowa, King of Ala., Mallory of Florida, Mangum of N. C., Mason of Va., Meriwether of Ky., Miller of N. J., Morton of Fla., Pearce of Md., Pratt of Md., Rusk of Texas, Shields of II., Smith of Conn., Solle of La., Spruance of Del., Toucey of Conn., Underwood of Ky., Upham of Vt., Walker of Wis, Weller of Cal.-47.

### First Session, Thirty-Third Congress. IN HOUSE.

1854, July 28-Mr. THOMAS D. ELIOT, of Massachusetts, asked leave to introduce a bill to repeal the fugitive slave law. Mr. BRIDGES, of Pennsylvania, objected.

Mr. ELIOT moved to suspend the rules for that purpose; which was rejected-yeas 45, nays 120, as follows:

YEAS-Messrs. Edward Ball, Henry Burnett, Samuel P. Benson, L. D. Campbell, David Carpenter, Moses B. Corwin, Benson, L. D. Campbell, David Carpenter, Mosés B. Corwin, Samuel L. Crocker, Thomas Davis, Alexander DeWitt, John Dick, Edward Dickinson, Ben. C. Eastman, J. Wiley Ed-mands, Thomas D. Eliot, William Everhart, Joshua R. Giddings, John Z. Goodrich, Aaron Harlan, Thomas M. Howe, Daniel T. Jones, James Knox, O. B. Matteson, Sam-uel Mayall, Edwin B. Morgan, Jesse O. Norton, Samuel W. Parker, Alexander C. M. Pennington, Benjamin Pringle, David Ritchie, Samuel L. Russell, Alvah Sabin, Russell Sage, William R. Sapp, George A. Simmons, Gerrit Smith, Andrew Start, Benjamin B. Thruston, M. C. Trout, C. W. Upham, Edward Wade, Samuel H. Walley, Ellihu B. Wash-burne, Israel Washburn, jr., Daniel Wells, jr., Tappan Went-

burne, Israel Washburn, Jr., Damer Weile, Jr., Approx. Worth-45.
NAYS--William Aiken, James C. Allen, Willis Allen, Wilsiam S. Ashe, D. J. Bailey, W. S. Barry, T. H. Benton, T. S. Bocock, W. W. Boyce, J. C. Breckinridge, S. A. Bridges, P. S. Brooks, S. Caruthers, E. M. Chamberlain, E. S. Chastain, J. S. Chrisman, W. M. Churchwell, S. Clark, T. L. Clingman, W. R. W. Cobb, J. P. Cook, L. M. Cox, B. Craige, C. B. Curtis, J. G. Davis, J. L. Dawson, D. T. Disney, J.F. Dowdell, A. Denhar. N. Eddy. A. P. Edeerton, H. A. Edmund- Harrison, S. G. Haven, T. A. Hendricks, B. Henn, H. Hibbard, C. S. Hill, G. S. Houston, T. G. Hunt, H. H. Johnson, G. W. Jones, L. M. Keitt, J. Kerr, Z. Kidwell, G. W. Kettridge, W. H. Kurtz, A. W. Lamb, M. S. Latham, J. Letcher, J. J. Lindley, F. McNullen, J. McNair, J. Mc Queen, J. B. Macy, J. Manrice, A. E. Maxwell, J. G. Miller, S. Miller, J. S. Millson, G. W. Morrison, W. Murray, M. H. Nichols, D. A. Noble, E. B. Olds, A. Oliver, J. L. Orr, R. W. Peckham, J. S. Phelps, P. Phillips, J. T. Pratt, W. Preston, R. C. Furyear, D. A. Reese, G. R. Riddle, J. Robbins, jr, S. H. Rogers, T. Ruffin, J. L. Seward, W. Shannon, H. M. Shaw, J. Shower, C. Skelton, S. A. Smith, W. S. Smith, G. W. H. Kogers, I. Runn, J. L. Seward, W. Shannon, H. M. Snaw, J. Shower, C. Skelton, S. A. Smith, W. R. Smith, G. W. Smyth, A. R. Sollers, F. P. Stanton, R. H. Stanton, A. H. Stephens, H. L. Stevens, N. T. Stratton, C. M. Straub, D. Stuart, F. J. Taylor, J. L. Taylor, N.G. Taylor, G. Vail, J. Vansant, H. Walbridge, W. A. Walker, J. Wheeler, W. H. Witte, D. B. Wright, H. B. Wright, F. K. Zollicoffer-120.

# Second Session, Thirty-Seventh Congress.\* IN SENATE.

1861, December 26-Mr. Hows, of Wisconsin,

\* On the 23d of July, 1861, the Attorney General, in answer to a letter from the United States marshal of Kansas, inquiring whether he should assist in the execution of the fugitive slave law, wrote: ATTORNEY GENERAL'S OFFICE, July 23, 1861.

ATTORNEY GENERAL'S OFFICE, July 23, 160. J L. MODOWELL, U. & Marshal, Kanszas: Your letter of the 11th of July, received 19th, (under frank of Senator Lane, of Kansas,) asks advice whether you should give your official services in the execution of the fugitive slave law. It is the President's constitutional duty to "take care that

Which was rejected-yeas 4, nays 47, as fol- | introduced a bill to repeal the fugitive slave law; which was referred to the Committee on the Judiciary.

1862, May 24-Mr. WILSON, of Massachusetts, introduced a bill to amend the fugitive slave law; which was ordered to be printed and lie on the table.

June 10-Mr. WILSON moved to take up the bill; which was agreed to-yeas 25, nays 10, as follows:

YEAS-Messrs. Anthony, Browning, Chandler, Clark, Cowan, Dixon, Doolittle, Fessenden, Foot, Grimes, Hale, Harlan, Harris, Howard, Howe, King, Lane of Kansas, Mor-rill, Pomeroy, Simmons, Sumner, Ten Eyck, Trumbull, Wade, Wilson of Massachusetts-25. NATS-Messrs. Carlie, Davis, Latham, McDougall, Nes-mith, Powell, Saulsbury, Stark, Willey, Wright-10.

The bill was to secure to claimed fugitives a right to a jury trial in the district court for the United States for the district in which they may be, and to require the claimant to prove his loyalty. The bill repeals sections 6, 7, 8, 9, and 10 of the act of 1850, and that part of section 5, which authorizes the summoning of the posse comitatus. When a warrant of return is made either on jury trial or confession of the party in the presence of counsel, having been warned of his rights, the fugitive is to be surrendered to the claimant, or the marshal where necessary, who shall remove him to the bound-ary line of the district, and there deliver him to the claimant. The bill was not further considered.

### IN HOUSE.

1861, December 20-Mr. JULIAN offered this resolution :

Resolved, That the Judiciary Committee be instructed to report a bill, so amending the fugitive slave law enacted in 1850 as to forbid the rocapture or return of any fugitive from labor without satisfactory proof first made that the claimant of such fugitive is loyal to the Government.

Mr. HOLMAN moved to table the resolution, which was disagreed to-yeas 39, nays 78, as follows:

follows: YEAS-Messrs. Ancona, Joseph Baily, Biddle, George H. Browne, Gobb, Cooper, Cox, Cravens, Crittenden, Dunlap, English, Fouke, Grider, Harding, Holman, Johnson, Law, Lazear, Leary, Lehman, Mallory, Morris, Noble, Noell, Nor-ton, Nugen, Odell, Pendleton, Robinson, Shiel, John B. Skele, William G. Skele, Vallandigham, Wadsworth, Webster, Chilton A. White, Wickliffe, Woodruff, Wright-39. NATS-Messrs. Aldrich, Alley, Arnold, Babbitt, Baker, Blair, Blake, Buffinton, Burnhan, Chamberlain, Clark, Col-fax, Frederick A. Conkling, Roscoe Conkling, Cutler, Davis, Dawes, Delano, Duell, Edwards, Eliot, Fessenden, Franchot, Frank, Gooch, Goodwin, Gurley, Hale, Hanchett, Harrison, Hooper, Hutchins, Julian, William Kellogg, Lan-sing, Loomis, Lovejoy, McKnight, McPherson, Marston, Mitchell, Moorhead, Anson P. Morrill, Justin S. Morrill, Olin, Patton, Pike, Pomeroy, Porter, Potter, John H. Rice, Riddle, Edward H. Rollins, Sargent, Sedgwick, Shanks, Shellabarger, Sherman, Sloan, Spaulding, Stevens, Ben-jamin F. Thomas, Train, Vandever, Wall, Wallace, E. P. Walton, Washburne, Wheeler, Whaley, Albert S. White, Wilson, Windom, Worcester-78.

the laws be faithfully executed." That means all the laws. He has no right to discriminate, no right to execute the laws he likes, and leave unexecuted those he dislikes. And of course you and I, his subordinates, can have no wider latitude of discretion than he has. Missouri is a State in the Union. The insurrectionary disorders in Missouri are but, individual crimes, and do not change the legal status of the State, nor change its rights and obligations as a member of the Union.

A refusal by a ministerial officer to execute any law which which I have no doubt the President would take notice. Very respectfully, EDWARD BATES. Very respectfully,

The resolution was then adopted-yeas 78, nays 39.

1862, June 9-Mr. JULIAN, of Indiana, introduced into the House a resolution instructing the Judiciary Committee to report a bill for the purpose of repealing the fugitive slave law; which was tabled—yeas 66, nays 51, as follows:

which was tabled—yeas 66, nays 51, as follows: YEAS—Messrs. William J. Allen, Ancona, Baily, Biddle, Francis P. Blair, Jacob B. Blair, George II. Browne, William G. Brown, Burnham, Calvert, Casey, Clements, Cobb, Corn-ing, Crittenden, Delano, Diven, Granger, Grider, Haight, Hale, Harding, Holman, Johnson, William Kellogg, Kerri-gan, Knapp, Lazear, Low, Maynard, Menzies, Moorhead, Mor-ris, Noble, Noell, Norton, Odell, Pendleton, John S. Phelps, Timothy G. Phelps, Porter, Richardson, Robinson, James S. Rollins, Sargent, Segar, Sheffield, Shiel, Smith, John B. Steele, William G. Steele, Benjamin F. Thomas, Francis Thomas, Trimble, Vallandigham, Verree, Vibbard, Voorhees, Wads-worth, Webster, Chillon A. White, Wickliffe, Wood, Woodruff, Worcester, Wrighl-66.

worth, Webster, Chillon A. White, Wickliffe, Wood, Woodrun, Worcester, Wright-66. "NAYS-Messrs. Aldrich, Alley, Baker, Baxter, Beaman, Bingham, Elake, Buffinton, Chamberlain, Colfax, Frederick A. Conkling, Davis, Dawes, Edgerton, Edwards, Eliot, Ely, Franchot, Good, Goodwin, Hanchett, Hutchins, Julian, Kel-ley, Francis W. Kellogg, Lansing, Lovejoy, McKnight, Mc Pherson, Mitchell, Anson P. Morrill, Pike, Pomeroy, Potter, Alexander II. Rice, John H. Rice, Riddle, Edward H. Rollins, Shellabarger, Sloan, Spaulding, Stevens, Train, Trowbridge, Yan Horn, Van Valkenburgh, Wall, Wallace, Washburne, Albert S. White, Windom-51.

Same day-Mr. COLFAX, of Indiana, offered this resolution:

Resolved, That the Committee on the Judiciary be in-structed to report a bill modifying the fugitive slave law so as to require a jury trial in all cases where the person claimed denies under each that he is a slave, and also re-quiring any claimant under such act to prove that he has been loyal to the Government during the present rebellion.

Which was agreed to-yeas 77, nays 43, as follows:

YEAS-Messrs. Aldrich, Alley, Arnold, Ashley, Babbitt, Ba-ker, Baxter, Beaman, Bingham, Francis P. Blair, Blake, Buffinton, Burnham, Chamberlain, Colfax, Frederick A. Conk-Iing, Davis, Dawes, Delano, Diven, Edgerton, Edwards, Eliot, Ely, Franchot, Gooch, Goodvin, Granger, Gurley, Haight, Hale, Hanchett, Hutchins, Julian, Kelley, Francis W. Kel-logg, William Kellogg, Lansing, Loomis, Lovejoy, Low, McKnight, McPherson, Mitchell, Anson P. Morrill, Justin S. Morrill, Nixon, Timothy G. Phelps, Pike, Pomeroy, Porter, Potter, Alexander H. Rice, John II. Rice, Riddle, Edward H. Rollins, Sargent, Shanks, Sheffield, Sheilabarger, Sloan, Spaulding, Stevens, Stratton, Benjamin F. Thomas, Train, Trimble, Trowbridge, Van Valkenburgh, Verree, Wall, Wal-lace, Waishburne, Albert S. White, Wilson, Windom, Wor-cester-77. ling, Davis, Dawes, Delano, Diven, Edgerton, Edwards, Eliot, cester-77.

cester-77. NAYS-Messrs. William J. Allen, Ancona, Baily, Biddle, Jacob B. Blair, William G. Brown, Cubert, Cascy, Clements, Cobb, Corning, Crittenden, Fouke, Grider, Harding, Hol-man, Joinson, Knapp, Maynard, Menzies, Noble, Noell, Nor-ton, Pendleton, John S. Phelps, Richardson, Robinson, James S. Rollins, Segar, Shiel, Smith, John B. Steele, William G. Steele, Francis Thomas, Vallandigham, Vibbard, Voorhees, Wadsworth, Webster, Chilton A. While, Wickliffe, Wood, Wrinht-13. Wright-13.

# Third Session, Thirty-Seventh Congress. IN SENATE.

1863, February 11-Mr. TEN EYCK, from the Committee on the Judiciary, to whom was referred a bill, introduced by Senator Hows, in second session, December 26, 1861, to repeal the fugitive slave act of 1850, reported it back without amendment, and with a recommendation that it do not pass.

# First Session, Thirty-Eighth Congress. IN HOUSE.

1863, Dec. 14-Mr. JULIAN, of Indiana, offered this resolution :

persons escaping from the service of their masters," approved February 12, 1793, and the act to amend and supplementary to the aforesaid act, approved September 18, 1850

Mr. HOLMAN moved that the resolution lie upon the table, which was agreed to-yeas 81, nays 73, as follows:

nays 73, as follows:
 YEAS-Messrs. James C. Allen, William J. Allen, Ancona, Anderson, Baily, Augustus C. Baldwin, Jacob B. Blair, Bliss, Brooks, James S. Brown, William G. Brown, Clay, Cobb, Coffroth, Cox, Cravens, Creswell, Dawson, Deming, Denison, Eden, Edgertom, Eltridge, English, Frinck, Ganson, Grider, Crisvold, Hall, Harding, Harrington, Benjamin G. Harris, Charles M. Harris, Higby, Ioliman, Hutchins, William John-son, Kernan, King, Knapp, Law, Lazear, Le Blond, Long, Maldory, Marcu, Marvin, McBride, McDowell, McKinney, William H. Miler, James R. Morris, Morrison, Nelsen, Noble, Odell, John O'Neil, Fendleton, William H. Randall, Robin-son, Rogers, James S. Rollins, Ross, Scott, Smith, Smithers, Stebbins, John B. Steele, Stuart, Sweat, Thomas, Voorhees, Wadsworth, Ward, Wheeler, Chiltor A. White, Joseph W. White, Williams, Winfield, Fernando Wood, Yeaman-S1 NAVS-Messrs. Alley, Allison, Ames, Arnold, Ashley, John D. Baldwin, Baxter, Beaman, Blaine, Ellow, Bourvell, Bord, Grandegee, Broomall, Ambrose W. Clark, Freeman Clarke, Ocle, Henry Winter Davis, Dawes, Dixon, Donnelly, Driggs, Dumont, Eckley, Eliot, Farnsworth, Fenton, Frank, Garfield, Gooch, Grinnell, Huburd, Jenckes, Julian, Francis W. Kellogg, Orlando Kellogg, Loan, Longyear, Lovejoy, Mc Clurg, McIndoe, Sanuel F. Miller, Moorhead, Morrill, Amos Myers, Leonard Myers, Norton, Charles O'Neill, Orth, Pat-erson, Pike, Pomerov, Price, Alexander H. Rice, John H.

Wayes, Leonard Myers, Norton, Charles O'Neill, Orth, Pat-terson, Pike, Pomeroy, Price, Alexander H. Rice, John H. Rice, Edward H. Rollins, Schenck, Scofield, Shannon, Spalding, Thayer, Van Valkenburgh, Ellihu B. Washburne, William B. Washburn, Whaley, Wilder, Wilson, Windom, Woodbridge-73.

1864, June 6-Mr. HUBBARD, of Connecticut, offered this resolution :

Resolved, That the Committee on the Judiciary be instructed to report to this House a bill for the repeal of all acts and parts of acts which provide for the rendition of fugitive slaves, and that they have leave to make such report at any time.

Which went over under the rule. May 30, he had made an ineffectual effort to offer it, Mr. HOLMAN objecting.

#### REPEALING BILLS.

1864, April 19-The Senate considered the bill to repeal all acts for the rendition of fugitives from service or labor. The bill was taken upyeas 26, nays 10.

Mr. SHERMAN moved to amend by inserting these words at the end of the bill :

Except the act approved February 12, 1793, entitled "An act respecting fugitives from justice, and persons escaping from the service of their masters."

Which was agreed to-yeas 24, nays 17, as follows:

YEAS-Messrs. Buckalew, Carlile, Collamer, Cowan, Daris, Dixon, Doolittle, Foster, Harris, Henderson, Hondricks, Howe, Johnson, Lane of Indiana, McDougall, Nesmith, Powell, Riddle, Saulsbury, Sherman, Ten Eyck, Trumbull, Van Winkle, Willey-24.

MATS-Messrs. Anthony, Brown, Clark, Conness, Fessen-den, Grimes, Hale, Howard, Lane of Kansas, Morgan, Mor-rill, Pomeroy, Ramsey, Sprague, Sumner, Wilkinson, Wilson-17.

Mr. SAULSBURY moved to add these sections : And be it further enacted, That no white inhabitant of the United States shall be arrested, or imprisoned, or held to answer for a capital or otherwise infamous crime, except in

cases arising in the land or naval forces, or in the militia when in actual service in time of war or public danger, And be it further enacted, That no person engaged in the

And be it further enacted, That no person engaged in the executive, legislative, or judicial departments of the Gov-ernment of the United States, or holding any office or trust recognized in the Constitution of the United States, and no person in military or naval service of the United States, shall, without due process of law, arrest or imprison any white inhabitant of the United States who is not, or has not been on chell not at the time of such ensure or imprison Resolved, That the Committee on the Judiciary be in-structed to report a bill for the repeal of the third and fourth gections of the "act respecting fugitives from justice and "ment be, engaged in levying war against the United States,"

or in adhering to the enemies of the United States, giving them aid and comfort, nor aid, abet, procure, or advise the same, except in cases arising in the land or naval forces, or in the militia when in actual service in time of war or public danger. And any person as aforesaid so arresting, or In changer. And any person as anoresaid so arresting, or imprisoning, or holding, as aforesaid, as in this and the sec-ond section of this act mentioned, or aiding, abetting, or procuring, or advising the same, shall be deemed guilty of felony, and, upon conviction thereof in any court of compe-tent jurisdiction, shall be imprisoned for a term of not less there me are more there for a wears shall more fine of more than one nor more than five years shall pay a fine of not less than 31,000 nor more than \$5,000, and shall be forever incapable of holding any office or public trust under the Governmet of the United Sates.

Mr. HALE moved to strike out the word "white" wherever it occurs; which was agreed to.

The amendment of Mr. SAULSBURY, as amended, was then disagreed to-yeas 9, nays 27, as follows :

YEAS-Messrs. Buckalew, Carlile, Cowan, Davis, Hen-dricks, McDougall, Powell, Riddle, Saulsbury-9. NATS-Messrs. Anthony, Clark, Collamer, Conness, Doo-little, Fessenden, Foster, Grimes, Hale, Harris, Howard, Howe, Lane of Indiana, Lane of Kansas, Morgan, Morrill, Howe, Lane Vargen, Sharman, Dargun, Sunyar, Too, Walt Pomeroy, Ramsey, Sherman, Sprague, Sumner, Ten Eyck, Trumbull, Van Winkle, Wilkinson, Willey, Wilson-27.

Mr. CONNESS moved to table the bill; which was disagreed to-yeas 9, (Messrs. Buckalew, Carlile, Conness, Davis, Hendricks, Nesmith, Powell, Riddle, Saulsbury,) nays 31.

It was not again acted upon.

1864, June 13-The House passed this bill, introduced by Mr. SPALDING, of Ohio, and reported form the Committee on the Judiciary by Mr. MORRIS, of New York, as follows:

Be it enacted, &c., That sections three and four of an act entitled "An act respecting fugitives from justice and per-sons escaping from the service of their masters," passed February 12, 1793, and an act entitled "An act to amend, and supplementary to, the act entitled 'An act respecting fugitives from justice, and persons escaping from their masters; passed February 12, 1793," passed September 18, 1850, be, and the same are hereby, repealed.

Yeas 86, nays 60, as follows:

Yeas 86, nays 60, as follows: Tas-Messrs. Alley, Allison, Ames, Arnold, Ashley, Nohn D. Baldwin, Baxter, Beaman, Blaine, Blair, Blow, Preeman Clarke, Cobb, Cole, Creswell, Henry Winter Davis, The The State of the State of the State of the State Freeman Clarke, Cobb, Cole, Creswell, Henry Winter Davis, Thomas T. Davis, Dawes, Dixon, Donnelly, Driggs, Eckley, Thitty, Higby, Hooper, Hotchkiss, Asahel W. Hubbard, John Hubbard, Hubburd, Ingersoll, Jeuckes, Julian, Kelley, Morrill, Daniel Morris, Amos Myers, Leonard Myers, Nor-or, Charles O'Neill, Orth, Patterson, Perham, Pike, Price, Atomic M. Kellogg, O. Kellogg, Littlejohn, Loan, Longyear, Morrill, Daniel Morris, Amos Myers, Leonard Myers, Nor-Morrill, Daniel Morris, Barnes S. State, Scoffeld, Stan-Morris, Uson, Van Valkenhurgh, Webter, Whaley, Wil-tiams, C. Baldwin, Bliss, Brooks, James S. Erown, Nats-Messrs, James C. Allen, William J. Allen, Ancona, Myenster, Coffroth, Coz, Cravens, Dawson, Denison, Eder Martington, Charles M. Harris, Herrick, Haiman, Herrington, Charles M. Harris, Herrick, Holman, Mather, Alblory, Marcy, McDowell, McKinney, Wm. H. Madford, Robinson, Jas. S. Kollins, Koss, Shuithers, John Szele, Wm. G. Steele, Stiles, Strouse, Staart, Sweat, Marador, Robinson, Jas. S. Kollins, Koss, Shuithers, Jone Marador, Kolins, Koss, Shuithers, Jones, Jones Marador, Kolins, Koss, Shuither, Joseph W. Mather, Janes 22-This bill was taken up in the Sen-

June 22-This bill was taken up in the Senate, when Mr. SAULSBURY moved this substitute:

That no person held to service or labor in one State, under the laws thereof, escaping into another, shall, in conse-quence of any law or regulation therein, be discharged from such service or labor, but shall be delivered up on claim of the party to whom such service or labor may be due; and Congress shall pass all necsssary and proper laws for the rendition of all such persons who shall so, as aforesaid, escape.

Which was rejected-yeas 9, nays 29, as follows:

YEAS-Messars. Buckalew, Carlile, Cowan, Davis, Mc-Daugall, Powell, Richardson, Riddle, Saulsbury-9. NAYS-Messrs. Anthony, Brown, Chandler, Clark, Con-ness, Dixon, Foot, Grimes, Hale, Harlan, Harris, Hicks, Howard, Howe, Johnson, Lane of Indiana, Lane of Kansas, Morgan, Morrill, Pomeroy, Ramsey, Spragne, Summer, Ten Eyck, Trumbull, Van Winkle, Wade, Willey-29.

Mr JOHNSON, of Maryland, moved an amendment to substitute a clause repealing the act of 1850; which was rejected-yeas 17, nays 22, as follows:

YEAS-Messrs. Buckalew, Carlile, Cowan, Davis, Harris, Hicks, Johnson, Lane of Indiana, McDougall, Powell, Rick-ardson, Riddle, Saulsbury, Ten Eyck, Trumbull, Van Winkle, Willey-17. NAYS-Messrs. Anthony, Brown, Chandler, Clark, Con-ness, Dixon, Fessenden, Foot, Grimes, Hale, Harlan, How-ard, Howe, Lane of Kansas, Morgan, Morrill, Pomeroy, Ramsey, Sprague, Sumner, Wade, Wilson-22.

The bill then passed-yeas 27, nays 12, as follows:

YEAS-Messrs. Anthony, Brown, Chandler, Clark, Con-ness, Dixon, Fessenden, Foot, Grimes, Hale, Harlan, Harris, Hicks, Howard, Howe, Lane of Indiana, Lane of Kausas,

Micras, Howard, Howe, Land of Hudata, Labo of Kalsao, Morgan, Morrill, Pomeroy, Ramsey, Sprague, Sumher, Ten Eyck, Trumbull, Wade, Wilson-27. NAYS-Messrs. Buckalew, Carlile, Cowan, Davis, Johnson, McDougall, Powell, Richardson, Riddle, Saulsbury, Van Winkle, Willey-12.

ABRAHAM LINCOLN, President, approved it, June 28, 1864.

### ESCAPE OF FUGITIVE SLAVES, FROM 1850 TO 1860.

The census report shows that notwithstanding all the controversies upon the subject of the fugitive slave law and its enforcement, from 1850 down to 1860, there were less per cent. escapes of fugitive slaves than at any former period of the Government. The report states :

"The number of slaves who escaped from their masters in 1860 is not only much less in proportion than in 1850, but greatly reduced numerically. The greatest increase of escapes appears to have occurred in Mississippi, Missouri, and Virginia, while the decrease is most marked in Dela-

and rights, while the declease is more index in least ware, Georgia, Louisiana, Maryland, and Tennessee. "That the complaint of inscentry to slave property by the escape of this class of persons into the free States, and their recovery impoded, whereby its value has been less-ened, is the result of misapprehension, is evident not only from the avell number who have how here to the its outpart from the small number who have been lost to their owners, but from the fact that up to the present time the number of escapes has been gradually diminishing to such an extent that the whole annual loss to the southern States from this cause bears less proportion to the amount of capital in-volved than the daily variations which in ordinary times occur in the fluctuations of State or Government securities in the city of Now York alone. "From the tables annexed it appears that while there

" from the tables annexed it appears that While there escaped from their masters 1,011 slaves in 1850, or 1 in each 3,165 held in bondage, (being about one thirtieth of one per cent.) during the census year ending June 1, 1860, out of 3,949.557 slaves, there escaped only 803, being 1 to about 5,000, or at the rate of one fiftieth of one per cent.

#### THE NEW ARTICLE OF WAR-MARCH 13, 1862.

# Second Session, Thirty-Seventh Congress. IN HOUSE.

ART. 102. All officers or persons in the military or naval service of the United States are prohibited from employing service of the onneu States are prohibited from employing any of the forces under their respective commands for the purpose of returning fugitives from service or labor who may have escaped from any persons to whom such service or labor is claimed to be due, and any officer who shall be found guilty by a court-martial of violating this article shall be dismissed from the service.

SEC. 2. That this act shall take effect from and after its passage.

1862, February 25-The article passed the House-yeas 95, nays 51, as follows:

House-yeas 95, nays 51, as follows:
 YEAS-Messrs. Aldrich, Alley, Arnold, Ashley, Babbitt,
 Bake, Bukrer, Bingham, Francis P. Blair, Samuel
 Blair, Blake, Buffinton, Campbell, Chamberlain, Clark,
 Colfax, Frederick A. Conkling, Roscoe Conkling, Conway,
 Covode, Cutler, Davis, Dawes, Diven, Duell, Edgerton, Edwards, Eliot, Ely, Fessenden, Franchot, Frank, Gooch, Goodwin, Granger, Gurley, Hale, Hanchett, Harrison, Hickman,
 Hooper, Hutchins, Julian, Kelley, Francis W. Kellogg,
 William Kellogg, Lansing, Loomis, Lovejoy, McKnight,
 McPherson, Mitchell, Moorhead, Anson P. Morrill, Justin
 Shorrill, Nixon, Olin, Patton, Timothy G. Phelps, Pike,
 Pomeroy, Porter, Potter, Alexander H. Rice, John H. Rice,
 Riddle, Edward H. Rollins, Sargent, Sedgwick, Shanks,
 Sheffield, Shellabarger, Sherman, Sloan, Spaulding, Stevens,
 Charles W. Walton, E. P. Walton, Washburne, Wheeler,
 Albert S. White, Wilson, Windom, Worcester-95.
 NATS-Messis. Ancona, Joseph Baily, Biddle, Jacob B.
 Blair, Calvert, Clements, Cobb, Corning, Cox, Cravens, Crisfield, Grittenden, Danlap, English, Grider, Halt, Harding,
 Holman, Johnson, Knapp, Law, Lazer, Leary, Lehman,
 Muloy, May, Maynard, Menzies, Morris, Noell, Norton,
 Naye-Preulidon, Prry, Price, Eobinson, James S. Rollins,
 Sheil, Smith, John B. Steele, William G. Steele, Francis Thomas,
 Yaliandigham, Yibbard, Voorhees, Ward, Webster, Wiektife,
 Woodruff, Wright—51.

#### IN SENATE.

March 10-The article was considered.

Mr. DAVIS, of Kentucky, moved to amend by inserting after the word "due" the words "and also from detaining, harboring, or concealing any such fugitives;" which was disagreed to yeas 10, nays 29, as follows:

YEAS-Messrs. Bayard, Carlile, Davis, Henderson, La-tham, McDougall, Powell, Rice, Saulsbury, Wilson of Mis-Bouri-10.

souri-10. NATS-Messrs. Anthony, Browning, Chandler, Clark, Col-lamer, Cowan, Dixon, Doolittle, Fessenden, Foot, Foster, Grimes, Hale, Harlan, Harris, Howard, Howe, King, Lane of Indiana, Lane of Kansas, Morrill, Pomeroy, Sherman, Sumner, Ten Eyck, Trumbull, Wade, Wilson of Massachu-setts, Wright-29.

Mr. SAULSBURY moved to exempt from the operation of the article, Delaware, Maryland, Missouri, and Kentucky, and elsewhere where the Federal authority is recognized or can be enforced; which was lost-yeas 7, (Bayard, Carlile, Latham, McDougall, Powell, Saulsbury, Wilson of Missouri, ) nays 30.

Mr. SAULSBURY moved to add after the word "due" the words "or for the purpose of enticing or decoying any person held to service or labor from the service of their loyal masters ;" which was lost-yeas 10, nays 29, (same as above.)

The article then passed-yeas 29, nays 9. The affirmative vote was the same as above, except that Mr. Cowan did not vote, and Mr. McDougall voted aye. The negative vote was:

Messrs. Bayard, Carlile, Davis, Henderson Powell, Rice, Saulsbury, Wilson of Missouri-9. Henderson, Latham,

The following action had previously been taken in the House:

### First Session, Thirty-Seventh Congress.

1861, July 9-Mr. LOVEJOY offered the following resolution :

Resolved, That, in the judgment of this Honse, it is no part of the duty of the soldiers of the United States to capture and return fugitive slaves.

Which was agreed to-yeas 92, nays 55, as follows:

YEAS-Messrs. Aldrich, Alley, Arnold, Ashley, Babbitt, Goldsmith F. Bailey, Baker, Baxter, Beaman, Bingham,

Francis P. Blair, Samuel S. Blair, Blake, Buffinton, Camphell, Chamberlain, Ambrose W. Clark, Colfax, Frederick A. Conkling, Roscoe Conkling, Cowary, Covode, Chtler, Davis, Dawes, Delano, Diven, Duell, Dunn, Edgerton, Edwards, Eliot, Ely, Fessenden, Franchot, Frank, Gooch, Grang, Grudey, Hale, Hickman, Itutchins, Julian, Kelley, Francis W. Kellogg, Killinger, Lansing, Loomis, Lovejoy, McKean, McPherson, Mitchell, Moorhead, Anson P. Morrill, Justin S. Morrill, Olin, Patton, Pike, Pomeroy, Porter, Potter, Alexander II. Rice, John H. Rice, Riddle, Edward H. Kollins, Stegwick, Shanks, Shellabarger, Sherman, Sloan, Spaulding, Stevens, Stratton, Benjamin F. Thomas, Thayer, Train, Trimble, Trowbridge, Vandever, Van Horr, Van Vakenburgh, Van Wyck, Verree, Wall, Wallace, Charles W. Walton, E. P. Walton, Washburne, Wheeler, Albert S. White, Windom, Worcester-92.
 MARS-Messes, Allen, Ancona, Joseph Baily, Burnett, Calvert, Carlie, Cobb, Cooper, Coz, Crauens, Crisfield, Crittenden, Duday, English, Fisher, Fouke, Grider, Haight, Harding, Unton, Jackson, Johnson, William Kellogg, Law, Lazear, Logan, McClernand, Millory, Menzies, Morris, Nixon, Noble, Noell, Nugon, Odelt, George H. Pendleton, Kichardsm, Skeffield, Smith, Join B. Steele, Wm. G. Steele, Francis Thomas, Upton, Vallandigham, Wadsworth, Hard, Weishert, Whide, Wiedh-55.
 Dee. 20-Mr. SHANKS offered this resolution,

Dec. 20-Mr. SHANKS offered this resolution, which was referred, Dec. 23, to the Committee on the Judiciary:

Resolved, That the constitutional power to return fugi-tive slaves to their masters rests solely with the civil de-partments of the Government; and that the order of the Secretary of War, under date of December 6, 1861, to Gen-eral Wool, for the delivery of a slave to Mr. Jessup, of Maryland, as well as all other military orders for the return of slaves, are assumptions of the military power over the civil law and the rights of the slave.

Second Session, Thirty-Seventh Congress.

1861, December 23-Mr. WILSON, of Iowa, offered this resolution :

Resolved, That the Committee on Military Affairs be in-structed to report to this House a bill for the enactment of an additional article of war for the government of the army. whereby the officers in the military service of the United States shall be prohibited from using any portion of the forces under their respective commands for the purpose of returning fugitives from service or labor; and providing for the punishment of such officers as may violate said article by dismissal from the service.

Which Mr. NOELL moved to lay on the table; lost-yeas 33, nays 70. The yeas were :

YEAS — Messrs. Joseph Baily, Biddle, Jacob B. Blair, George H. Browne, Calvert, Cobb, Cravens, Dunlap, Dunn, Fouke, Grider, Haight, Harding, Knapp, Law, Leary, Logon, Maynard, Menzies, Noble, Noell, Norton, Nugen, Odell, Perry, Robinson, Sliel, Smith, John B. Steele, Francis Thomas, Upton, Vallandigham, Webster—33.

The resolution was then adopted.

ON EMPLOYMENT OF SLAVES IN DOCK-YARDS, ETC.

# Second Session, Thirty-Seventh Congress. IN SENATE.

1862, June 14-Mr. WILSON, of Massachusetts, offered this additional section to the naval appropriation bill:

That persons held to service or labor, commonly called slaves, shall not be employed in any capacity whatever in any navy-yard, dock-yard, arsenal, magazine, fort, or in the Naval Academy.

Which was agreed to in Committee of the Whole.

June 16-It was rejected in open Senateyeas 17, nays 18, as follows:

YEAS 11, M29 10, as 1010 wr. Dixon, Foot, Grimes, Hale, Harlan, King, Lane of Indiana, Lane of Kansas, Mor-rill, Pomeroy, Summer, Trumbull, Wilmot, Wilson of Mas-sachnsetts, Wright—17. NATS—Messrs. Anthony, Browning, Chandler, Davis, Doolittle, Fessenden, Foster, Henderson, Howard, Howe, Latham, McDougalt, Fowell, Saulsbury, Stark, Ten Eyck, Willey, Wilson of Missouri—18.

### RECOGNITION OF HAYTI AND LIBERIA.

# Second Session, Thirty-Seventh Congress.

The bill to authorize the President to appoint diplomatic representatives to the republics of Hayti and Liberia, respectively, each representative to be accredited as commissioner and consul general, and to receive the pay of commissioners, that at Liberia not to exceed \$4,000 per annum-

1862, April 24-Passed the Senate-yeas 32, nays 7, as follows:

YEAS-Messrs. Anthony, Browning, Chandler, Clark, Col-lamer, Cowan, Dixon, Doolittle, Fessenden, Foot, Foster, Grimes, Hale, Henderson, Howard, Howe, King, Lane of Indiana, Lane of Kansas, Latham, McDougall, Morrill, Pomeroy, Sherman, Simmons, Sumner, Ten Eyck, Trum-bull, Wade, Wilkinson, Wilson of Massachusetts, Wright -32.

NAVS-Messrs. Bayard, Carlile, Davis, Powell, Saulsbury, Stark, Thomson-7.

Previously, an amendment like that offered in the House by Mr. Cox was lost-yeas 8, nays 30.

June 3-It was taken up in the House.

Mr. Cox, of Ohio, offered an amendment providing for the appointment of a consul general to each, with authority to negotiate any treaties of commerce between Hayti and Liberia and this country, and with the compensation of consuls general; which was lost-yeas 40, nays 82.

The bill then passed-yeas 86, nays 37, as follows:

follows: YEAS-Messrs. Aldrich, Alley, Ashley, Babbitt, Baker, Baxter, Beaman, Bingham, Francis P. Blair, Blake, Buffin-ton, Casey, Chamberlain, Clark, Clements, Colfax, Frederick A. Conkling, Roscoe Conkling, Covode, Davis, Dawes, De-lano, Ducli, Dunn, Edgerton, Ely, Fessenden, Fisher, Frank, Gooch, Goodwin, Granger, Gurley, Hale, Hickman, Hooper, Horton, Hutchins, Julian, Kelley, William Kellogg, Lan-ter, Alexander H. Rice, John H. Rice, Romeroy, Por-ter, Alexander H. Rice, John H. Rice, Riddle, Edward H. Rollins, Sargent, Sedgwick, Shanks, Sheffield, Shellabarger, Sloan, Spaulding, Stevens, Stratton, Benjamin F. Thomas, Francis Thomas, Train, Trimble, Trowbridge, Van Horn, Aubert S. White, Wilson, Windom, Worcester-86. NATS-Messrs. W. J. Allen, Ancona, Baily, Biddle, Jacob B. Blair, George H. Browne, Calvert, Cobb, Corning, Cax, Dunlap, Grider, Harding, Holman, Knopp, John S. Phelps, Price, Segar, Smith, John B. Steele, Wilkiam G. Steele, Stiles, Yibbard, Yoonkees, Wadsworth, Ward, Webster, Wickliffe, Wright-37.

#### ROBERT SMALL.

Second Session, Thirty-Seventh Congress.

### IN SENATE.

May 19-Mr. GRIMES brought in a bill for the relief of Robert Small, which passed without a division.

It authorizes the Secretary of the Navy to cause the steam transport boat Planter, recently in the rebel service in the harbor of Charleston, and all the arms, munitions, tackle, and other property on board of her at the time of her delivery to the Federal authorities, to be appraised by a board of competent officers, and, when the value shall be thus as-certained, to cause an equitable apportionment of one half of such value, so ascertained, to be made between Robert of such value, so ascertained, to be made between Robert Small and his associates who assisted in rescuing her from the enemies of the Government. The Secretary of the Navy may, if he deems it expedient, cause the sum of money allotted to each individual under this bill to be invested in United States securities for his benefit, the interest to be paid to him or to his heirs annually until such time as the Secretary of the Navy may deem it expedient to pay to him or his heirs the principal sum.

#### IN HOUSE.

May 26-It passed-yeas 121, nays 9.-(Messrs. Calvert, Dunlap, Harding, Johnson, Kerrigan, Norton, Shiel, Vallandigham, Voorhees.)

TO REMOVE DISQUALIFICATION OF COLOR IN CAR-RYING THE MAILS.

### Second Session, Thirty-Seventh Congress. IN SENATE.

1862, April 11-The Senate considered a bill "to remove all disqualification of color in carrying the mails of the United States." It directed that after the passage of the act no person, by reason of color, shall be disqualified from employment in carrying the mails, and all acts and parts of acts establishing such disqualification, including especially the seventh section of the act of March 3, 1825,\* are hereby repealed.

The vote in the Senate was, yeas 24, nays 11, as follows :

YEAS-Messre. Anthony, Browning, Chandler, Clark, Col-lamer, Dixon, Doolittle, Fessenden, Foot, Foster, Grimes, Hale, Howard, Howe, King, Lane of Kansas, Morrill, Pome-roy, Sherman, Simmons, Summer, Wade, Wilkinson, and Wilson of Massachusetts-24.

MATS-Messrs. Davis, Henderson, Kennedy, Lane of In-diana, Latham, Nemith, Powell, Stark, Willey, Wilson of Missouri, Wright-11.

\* The section referred to is in these words:

That no other than a free white person shall be employed in conveying the mail, and any contractor who shall em-ploy or permit any other than a free white person to con-vey the mail shall, for every such offence, incur a penalty of twenty dollars.

The act of 1825 passed the Senate March 1, and the House March 2, without a division. The suggestion of this meas-ure appears to have been first made in 1802 by Gideon Granger, Postmaster General under President Jefferson, in the following letter:

GENERAL POST OFFICE, March 23, 1802.

SIR: An objection exists against employing negroes, or people of color, in transporting the public mails, of a na-ture too delicate to ingraft into a report which may become public, yet too important to be omitted or passed over withphone, yet too important to be officient or passed over with out full consideration. I therefore take the liberty of mak-ing to the committee, through you, a private representa-tion on that subject. After the scenes which St. Domingo has exhibited to the world, we cannot be too cautious in at-tempting to prevent similar evils in the four southern States, where there are, particularly in the eastern and old settled marts of them, so great a proportion of blacks as to hazard the tranquillity and happiness of the free citizens. Indeed, in Virginia and South Carolina (as I have heen informed) plans and conspiracies have already been concerted by them, more than once, to rise in arms, and subjugate their masters.

Everything which tends to increase their knowledge of natural rights, of men and things, or that affords them an opportunity of associating, acquiring, and communicating sentiments, and of establishing a chain or line of intelli-gence, must increase your hazard, because it increases their means of effecting their object.

The most active and intelligent arc employed as post-The nost active and intelligent are employed as post-riders. These are the most ready to learn, and the most able to execute. By traveling from day to day, and hourly mixing with people, they must, they will acquire informa-tion. They will learn that a mark's rights do not depend on his color. They will, in time, become teachers to their breth-ren. They become acquainted with each other on the line. Whenever the body, or a portion of them, wish to act, they are an organized corps, circulating our intelligence openly, their own privately. their own privately.

Their traveling creates no suspicion, excites no alarm. One able man among them, perceiving the value of this ma-chine, might lay a plan which would be communicated by your post-fiders from town to town, and produce a general and united operation against you. It is easier to prevent the evil than to cure it. The hazard may be small and the prospect remote, but it does not follow that at some day the event would not be certain. With generat and extrem

With respect and esteem, GIDEOI Hon. JAMES JACKSON, Senator from Georgia. GIDEON GRANGER.

### IN HOUSE.

#### May 21-It was considered in the House and laid on the table-yeas 83, nays 43, as follows :

Yaks-Messrs. Allen, Ancona, Babbitt, Baily, Biddle, Francis P. Blair, Jacob B. Blair, William G. Brown, Culverl, Casey, Clements, Colfax, Roscoe Conkling, Corwing, Cox, Cravens, Crisfield, Criltenden, Diven, Dunlap, Dunn, Ely, English, Franchot, Granger, Grider, Haight, Hale, Hall, Hanchett, Harding, Harrison, Holman, Horton, Johnson, Kelley, William Kellogg, Kerrigan, Küllinger, Law, Lazear, Leary, Lehman, McKnight, McPherson, Mullory, Maynard, Menzies, Nixon, Noell, Nugen, Oddl, Patton, Prendleon, Perry, John S. Phelps, Timothy G. Phelps, Porter, Price, Richardson, Robinson, Sargent, Shiel, Smilh, William G. Stratton, Francis Thomas, Trowbridge, Van Valkenburgh, Verree, Yibbard, Foorhees, Walkenorth, Wall, Wurd, Webster, Yuhalev, Albert S. White, Chilton A. While, Wielster, Walley, Noorhees, Park, Steit, Swilh, Will, Wind, Webster, Statton, Prancis Thomas, Trowbridge, Van Valkenburgh, Verree, Yibbard, Foorhees, Walkenorth, Wall, Wurd, Webster, Schlert, Steite, Nach, Statton, Johrens, Mather, Malter, Albert S. Whalev, Albert S. White, Chilton A. White, Wildik, William K. Vibbard, Voorhees, Wudsworth, Wall, Ward, Webster, Whaley, Albert S. White, Chilton A. White, Wieldiffe, Wood-

Whatey Abert S. White, Childon A. White, Weak for Wood-raff, Worcester—S2. NArs—Messrs. Aldrich, Alley, Beaman, Bingham, Sanuel S. Blair, Blake, Bnflinton, Chamberlain, Frederick A. Conk-ling, Cutler, Davis, Dawes, Delano, Duell, Edgerton, Ed-wards, Eliot, Fenton, Fessenden, Frank, Gooch, Goodwin, With the Science Line of Component. Wards, Ehol, Feiton, Fessenden, Frank, Gotel, Godowin, Hooper, Hutchins, Julian, Lansing, Loomis, Pike, Pomeroy, Potter, Alexander II. Rice, Riddle, Edward II. Rollins, Sedgwick, Sheffield, Sloan, Spaulding, Benjamin F. Thomas, Train, Van Horn, Wallace, Charles W. Walton, E. P. Walton, Wilson, Windom—43.

#### First Session, Thirty-Eighth Congress.

1864, February 26-The Senate considered the bill-the question being on agreeing to a new section proposed by the Committee on Post Offices and Post Roads-as follows :

SEC. 2. That in the courts of the United States there shall be no exclusion of any witness on account of color.

Mr. Powell moved to amend by inserting after the word "States" the words: "in all cases for robbing or violating the mails of the United States."

No further progress was made on the bill.

#### NEGRO SUFFRAGE IN MONTANA TERRITORY.

1864, March 18-The House passed, without a division, a bill in the usual form, to provide a temporary government for the Territory of Montana.

March 31-The Senate considered it, when Mr. WILKINSON moved to strike from the second line of the fifth section, (defining the qualifications of voters,) the words "white male inhabitant" and insert the words: "male citizen of the United States, and those who have declared their intention to become such;" which was agreed to-yeas 22, nays 17, as follows :

YEAS-Messrs. Brown, Chandler, Clark, Collamer, Con-ness, Dixon, Fessenden, Foot, Foster, Grimes, Hale, Harlan, Harris, Howard, Howe, Morgan, Morrill, Pomeroy, Sunner, Wade, Wilkinson, Wilson-22. NAYS-Messrs, Buckalew, Carlile, Cowan, Davis, Harding, Henderson, Johnson, Lane of Indiana, Newrith Bound

Henderson, Johnson, Lane of Indiana, Nesmith, Powell, Riddle, Saulsbury, Sherman, Ten Eyck, Trumbull, Van Winkle, Willey-17.

The bill was then passed—yeas 29, nays 8, (Messrs. Buckalew, Davis, Johnson, Powell, Riddle, Saulsbury, Van Winkle, Willey.)

April 15-The Senate adopted the report of the Committee of Conference on the Montana bill, which recommended the Senate to recede from their second amendment, and the House to agree to the first and third amendments of the Senate, (including the above.)

April 15-Mr. BEAMAN presented the report of the Committee of Conference on the Montana bill, a feature of which was that the House should recede from its disagreement to the Senin the description of those authorized to vote. Mr. HOLMAN moved that the report be tabled ;

which was lost by the casting vote of the Speaker-yeas 66, nays 66.

Upon agreeing to the report the yeas were 54, nays 85, as follows:

54, nays 85, as follows:
YEAS-Messrs. Alley, Allison, Ames, Anderson, Ashley, John D. Baldwin, Baxter, Beaman, Blaine, Boyd, Ambrose W. Clark, Cole, Dawes, Deirges, Dumont, Eckley, Farnsworth, Frank, Garfield, Gooch, Higby, Hooper, Hotchkiss, John H. Hubbard, Jenekes, Julian, Kelley, Orlando Kellogg, Loan, Longyear, Marvin, McClurg, McIndoe, Morrill, Daniel Morris, Amos Myers, Leonard Myers, Charles O'Neill, Patterson, Perham, Pike, Price, Alexander H. Rice, John H. Rollins, Shannon, Sloan, Stevens, Upsen, Van Valkenburgh, William B. Washburn, Williams, Windom, Woodbridge-54.
NAYS-Messrs, James C. Allen, William J. Allen, Baily, Augustus C. Boldwin, Francis P. Blair, Eliss, Brooks, James S. Brown, Charler, Clay, Coffroth, Coz, Cravens, Creswell, Henry Winter Davis, Dawison, Deming, Devison, Edar, Eldridge, Finek, Ganson, Grider, Hall, Harding, Benjamin G. Horris, Morrison, Kaliyleisch, Kasson, Francis W. Kellogg, Kernan, Knapp, Law, Lazear, Long, Mallory, Marey, Meride, Morris, Norrison, Nelson, Noble, Odell, Orth, Pendicton, Pomeroy, Pruyn, Radford, Sumues J. Rundall, William H. Randall, Robinson, Rogers, Jonnes S. Moltins, Koss, Schenek, Scotl, Smithers, Stebins, Johns, J. Kuek, Yorkes, Ellihu B. Washburne, Webster, Whaley, Wheeler, Chilton A. Whiel, Joseph W. White, Wilson, Winfield, Fernando Wood, Yeaman-S5.

On a motion to adhere to its amendments, and ask another Committee of Conference, Mr. WEBSTER moved instructions :

And that said committee be instructed to agree to no report that authorizes any other than free white malc citi-zens, and those who have declared their intention to become such, to vote.

Which was agreed to-yeas 75, nays 67, as follows:

YEAS-Messrs. James C. Allen, Wm. J. Allen, Baily, Au-gustus C. Baldwin, Francis P. Blair, Bliss, Brooks, James S. Broum, Wm. G. Brown, Chanler, Clay, Coffroth, Cox, Cravens, Creswell, Henry Winter Davis, Dawson, Denison, Eden, Eldridge, Finck, Ganson, Grider, Halt, Harding, Benjamin G. Harris, Herrick, Holman, Hutchins, William Johnson, Kallfleisch, Kernan, Knopp, Law, Lazear, Long, Mallory, Marcy, McBride, McDouell, McKinney, Wm. H. Miller, James R. Morris, Morrison, Nelson, Noble, Odell, Penditton, Radford, Samuel J. Randall, Wm. H. Randall, Robinson, Rogers, Johnes, Tracy, Voorhees, Webster, Whaley, Wheeler, Chillon A. White, Joseph W. While, Winfield, Fer-nando Wood, Feaman-75. NAYS-Messre, Alley, Alley, Ames, Anderson, Ashley. YEAS-Messrs. James C. Allen, Wm. J. Allen, Baily, Au-

nando Wood, Feaman-75. NAvs-Messrs. Alley, Allison, Ames, Anderson, Ashley, John D. Baldwin, Baxter, Beaman, Blaine, Boutwell, Boyd, Broomall, Ambrose W. Clark, Cobb, Cole, Dawes, Deming, Driggs, Dumont, Farnsworth, Frank, Gooch, Grinnell, Hig-by, Hooper, Hotchkiss, Asahel W. Hubbard, John H. Hub-bard, Jenckes, Julian, Kelley, Francis W. Kellogg, Orlando Kellogg, Loan, Longyear, Marvin, McClurg, McIndoe, Samuel F. Willer Morrill Daniel Morris, Leonard Myers bard, Jenckes, Julian, Kelley, Francis W. Kellogg, Orlando Kellogg. Loan, Longyear, Marvin, McClurg, McIndoe, Samuel F. Miller, Morrill, Daniel Morris, Leonard Myers, Norton, Charles O'Neill, Orth, Patterson, Perham, Pike, Pomeroy, Price, Alexander II. Rice, John II. Rice, Edward II. Rollins, Schenck, Shannon, Sloan, Stevens, Thayer, Upson, Van Valkenburgh, Ellihu B. Washburne, William B. Washburn, Williams, Wilder, Wilson, Windom, Wood-bridge\_O' bridge-67.

April 15-The Senate declined the conference upon the terms proposed by the House resolution of that day.

April 18—The House proposed a further free conference, to which, April 25, the Senate acceded.

May 17-In Senate, Mr. MORRILL submitted a report from the Conference Committee who recommend that qualified voters shall be:

All citizens of the United States, and those who have ate amendment striking out the word "white" declared their intention to become such, and who are otheract of Congress providing for a temporary government for and inserting the words -the Territory of Idaho approved March 3, 1863.\*

The report was concurred in-yeas 26, nays 13, as follows:

YEAS-Messrs. Buckalew, Carlile, Collamer, Cowan, Davis, Doolittle, Foot, Foster, Harding, Harris, Henderson, Hen-dricks, Howard, Johnson, Lane of Indiana, Morrill, Nesmith, Powell, Ramsey, Saulsbury, Ten Eyck, Trumbull, Van Winkle, Wade, Wilkinson, Willey-26. NATS-Messrs, Anthony, Chandler, Clark, Dixon, Grimes, Hale, Herlan, Lange Kayess, Morgan, Powerson, Sciences,

Hale, Harlan, Lane of Kansas, Morgan, Pomeroy, Sprague, Sumner, Wilson-13.

May 20-The above report was made by Mr. WEBSTER in the House, and agreed to-yeas 102, nays 26, as follows :

nays 26, as follows: YEAS-Messrs. James C. Allen, Baily, Beaman, Blaine, Jacob B. Blair. Bliss, Brooks, William G. Brown, Chan-ler, Coffroth, Cox, Cravens, Creswell, Thomas T. Davis, Dawson, Donnelly, Driggs, Edon, Edgerton, Eldridge, Farnsworth, Finck, Grider, Hale, Hall, Harding, Har-rington, Charles M. Harris, Herrick, Holman, Hotchkiss, Asahel W. Hubbard, Hutchins, Ing:rsoll, William John-son, Kalbfeisch, Kasson, Francis W. Kellogg, Kernan, King, Law, Lazear, Long, Longyear, Mallory, Marcy, McAllister, McBride, McDowell, McIndoe, McKinney, Samuel F. Mil-ler, James R. Morris, Merrison, Amos Myers, Nilson, Noble, Norton, Charles O'Niell, John & Niell, Orth. Pendleton, Per-ham, Pike, Pomeroy, Pruyn, Radford, S. J. Randall, Vm. H. Randall, Alexander II. Rice, John H. Riee, Rolins, Snith, H. Kandall, Alexander H. Rice, John H. Rice, *Koomson*, James S. Rollins, Ross, Scofield, Scott, Shannon, Sloau, Smith, Smithers, John B. Steele, Wm. G. Steele, Stiles, Strouse, Stuart, Sweat, Thayer, Tracy, Upson, Van Valkenburgh, Voorhees, Ellihu B. Washburne, William B. Washburn, Webster, Whaley, Wheeler, Wilson, Windon, Fernando Wood, Woodbridge, Feaman-102. Murg. Macron Aluce, Aluce Ange, John B. Daldrin.

NATS-Messrs. Alley, Allicon, Ames, John D. Baldwin, Boutwell, Ambrose W. Clark, Cobb, Cole, Dawes, Dixon, Eliot, Gooch, Grinnell, Higby, John H. Hubbard, Julian, Kelley, Orlando Kellogs, Loan, Moorhead, Morrill, Price, Edward H. Rollins, Spalding, Stevens, Wilder-26.

#### IN WASHINGTON CITY.+

1864, May 6-The Senate considered the bill for the registration of voters in the city of Washington, when

Mr. Cowan moved to insert the word "white" in the first section, so as to confine the right of voting to white male citizens.

May 12-Mr. MORRILL moved to amend the amendment by striking out the words-

And shall have paid all school taxes and all taxes on personal property properly assessed against him, shall be entitled to vote for mayor, collector, register, members of the board of aldermen and board of common council, and assessor, and for every officer authorized to be elected at any election under any act or acts to which this is amendatory or supplementary.

\* This section is as follows:

SEC. 5. And be it further enacted, That every free white male inhabitant above the age of twenty-one years, who shall have been an actual resident of said Territory at the 

proposition to permit negro suffrage without a property qualification. The result in the city was—yeas 1,640, nays 37,471. In the State—yeas 197,503, nays 337,984. In 1864 a like proposition was defeated—yeus 85,406, nays 224,536. In 1862, in August, a vote was had in the State of Illinois, or anyone proceeding soleling to purpose and multinois,

on several propositions relating to negroes and mulattoes, with this result:

For excluding them from the State	71.306	
Against granting them suffrage or right to office.		100,587
For	35,649	176.271

154,524

For the enactment of laws to prohibit them from going to, or voting in, the State...... 198,938 Against 44,414

16

And shall within the year next preceding the election have paid a tax, or been assessed with a part of the revenne of the District, county, or cities therein, or been exempt from taxation having taxable estate, and who can read and write with facility, shall enjoy the privileges of an elector.

May 26-Mr. SUMNER moved to amend the bill by adding this proviso:

Provided, That there shall be no exclusion of any person from the registry on account of color.

May 27-Mr. HARLAN moved to amend the amendment by making the word "person" read "persons," and adding the words-

Who have borne arms in the military service of the United States, and have been honorably discharged therefrom.

Which was agreed to-yeas 26, nays 12, as follows:

YEAS - Messrs. Anthony, Chandler, Clark, Collamer Conness, Dixon, Fessenden, Foot, Foster, Grimes, Hale, Harlan, Harris, Johnson, Lane of Indiana, Lane of Kansas, Martan, Harris, Jonnson, Lane of Indiana, Lane of Kansas, Morgan, Morrill, Pomeroy, Ramsey, Sherman, Ten Eyck, Trumbull, Wade, Willey, Wilson-26. NATS-Messrs. Buckalew, Curlile, Cowan, Davis, Hen-dricks, McDougall, Powell, Richardson, Saulsbury, Sumner, Van Winkle, Wilkinson-12.

May 28-Mr. SUMMER moved to add these words to the last proviso :

And provided further, That all persons, without distinc-tion of color, who shall, within the year next preceding the election, have paid a tax on any estate, or been as-sessed with a part of the revenue of said District, or been exempt from taxation having taxable estate, and who can read and write with facility, shall enjoy the privilege of an elector. But no person now entilled to vote in the said District continuing taxable busics busics. District, continuing to reside therein, shall be disfranchised hereby.

Which was rejected-yeas 8, nays 27, as follows:

YEAS—Messrs. Anthony, Clark, Lane of Kansas, Morgan, Pomeroy, Ramsey, Summer, Wilkinson—8. NATS—Messrs. Buckalew, Carlile, Collamer, Cowan, Davis, Dixon, Fessenden, Foot, Foster, Grimes, Hale, Harlan, Har-ris, Hendricks, Hicks, Johnson, Lane of Indiana, McDougall, Wardiu, Wang, Carl, Januar, The Stark, Theorem. Morrill, Powell, Suulsbury, Sherman, Ten Eyck, Trumbull, Van Winkle, Willey, Wilson-27.

The other proposition of Mr. SUMNER, amended on motion of Mr. HARLAN, was then rejected -yeas 18, nays 20, as follows:

YEAS-Messrs. Anthony, Chandler, Clark, Dixon, Foot, Foster, Hale, Harlan, Howard, Howe, Lane of Kansas, Mor-gan, Pomeroy, Ramsey, Sherman, Sumner, Wilkinson, Wil-

Son-18. NATS-Messrs. Buckalew, Carlile, Cowan, Davis, Grimes, Harris, Hendricks, Hicks, Johnson, Lane of Indiana, Me-Dougall, Morrill, Nesmith, Powell, Richardson, Saulsbury, Ten Eyck, Trumbull, Van Winkle, Willey-20.

The bill then passed the Senate, and afterward the House, without amendment.

EXCLUDING COLORED PERSONS FROM CARS.

Third Session, Thirty-Seventh Congress. IN SENATE.

1863, February 27-Pending a supplement to the charter of the Washington and Alexandria Railroad Company,

Mr. SUMNER offered this proviso to the first section:

That no person shall be excluded from the cars on account : of color.

Which was agreed to-yeas 19, nays 18, as . follows:

YEAS-Messrs. Arnold, Chandler, Clark, Fessenden, Foot, Grimes, Harris, Howard, King, Lane of Kansas, Morrill, Pomeroy, Summer, Ten Eyck, Trumbull, Wade, Wilkinson, Wilmot, Wilson of Massachusetta-19. NATS-Messrs. Anthony, Bayard, Carlile, Cowan, Davis,

Henderson, Hicks, Howe, Kennedy, Lane of Indiana, La-tham, McDougall, Powell, Richardson, Saulsbury, Turpie, Willey, Wilson of Missouri-18.

March 2-The House concurred in the amendment without debate, under the previous question.

# First Session, Thirty-Eighth Congress. IN SENATE.

1864, February 10-Mr. SUMNER offered the following :

Resolved. That the Committee on the District of Columbia be directed to consider the expediency of further pro-viding by law against the exclusion of colored persons from the equal enjoyment of all railroad privileges in the District of Columbia.

Which was agreed to-yeas 30, nays 10, as follows:

YEAS-Messrs. Anthony, Brown, Chandler, Clark, Colla-mer, Conness, Cowan, Dixon, Fessenden, Foot, Foster, Grimes, Hale, Harlan, Harris, Howard, Howe, Lane of Kan-sas, Morzin, Morrill, Pomeroy, Ramsey, Sherman, Sprague, Sanner, Ten Eyck, Trumbull, Wade, Wilkinson, Wilson

NAYS-Messrs. Buckalew, Davis, Harding, Hendricks, Nesmith, Powell, Richardson, Riddle, Saulsbury, Van Winkle-10.

February 24-Mr. WILLEY, from the Committee on the District of Columbia, made this report, and the committee were discharged :

The Committee on the District of Columbia, who were required by resolution of the Senate, passed February 8, 1864, "to consider the expediency of further providing by law against the exclusion of colored persons from the equal enjoyment of all railroad privileges in the District of Co-lumbia," have had the matter thus referred to them under considering and here leave to ensure the

thindia, have been added to report: The act entitled "An act to incorporate the Washington and Georgetown Railroad Company," approved May 17, 1862, makes no distinction as to passengers over said road on account of the color of the passenger, and that in the opinion of the committee colored persons are entitled to all the privileges of said road which other persons have, and to all the remedies for any denial or breach of such privileges which belong to any person.

The committee therefore ask to be discharged from the further consideration of the premises.

March 17-The Senate considered the bill to incorporate the Metropolitan Railroad Company, in the District of Columbia, the pending question being an amendment, offered by Mr. SUMNER, to add to the fourteenth section the words:

Provided, That there shall be no regulation excluding any person from any car on account of color.

Which was agreed to-yeas 19, nays 17, as follows:

YEAS-Messrs. Anthony, Brown, Clark, Conness, Fessen-den, Foot, Foster, Grimes, Harlan, Howe, Lane of Kansas, Morgan, Morrill, Pomeroy, Ramsey, Sumner, Wade, Wilkin-son, Wilson-19.

Norgan, Morini, Tomeroy, Rainsey, Sammer, Wate, Winki-son, Wilson-19. NATS-Messrs. Buckalew, Carlile, Davis, Doolitile, Hard-ing, Harris, Hendricks, Johnson, Lane of Indiana, Powell, Riddle, Saulsbury, Sherman, Ten Eyck, Trumbull, Van Winkle, Willey-17.

The bill then passed the Senate.

June 19-The House refused to strike out the proviso last adopted in the Senate-yeas 60, nays 76, as follows:

Hays 10, 48 IOHOWS: YEAS-Messrs. James C. Allen, William J. Allen, Ancona, Baily, Augustus C. Baldwin, Blair, Eliss, Brooks, James S. Brown, William G. Brown, Chanler, Coffrodh, Gravens, Daw-son, Denison, Eden, Edgerton, Eldridge, Finck, Ganson, Grider, Griswold, Harding, Charles M. Harris, Holman, Hutchins, Philip Johnson, William Johnson, Kernan, Knapp, Law, Lazear, LeBlond, Long, Marcy, McDowell, McKinney, Middleton, Wm. H. Miller, James R. Morrison, Norison, Nelson, Noble, Pendleton, Perry, Pruyn, Radford, Samuel I. Randall, Ross, Scott, John B. Steele, Stiles, Stuart, Thomas, Wadsworth, Ward, Whaley, Wheeler, Joseph W. White, Winfield-60.

NATS-Messrs. Alley, Ames, Anderson, Arnold, Ashley, John D. Baldwin, Baxter, Beaman, Blaine, Boutwell, Boyd, Brandegee, Broomall, Ambrose W. Clark, Freeman Clarke, Cobb, Cole, Dawes, Deming, Dixon, Driggs, Eckley, Eliot, Farnsworth, Fenton, Frank, Garfield, Gooch, Higby, Hooper, Hotchkiss, Asahel W. Hubbard, John H. Hubhard, Hulburd, Ingersoll, Julian, Kelley, Orlando Kellogg, Knox, Loan, Longyear, Marvin, McClurg, McIndoe, Samuel F. Miller, Moorhead, Morrill, Daniel Morris, Amos Myers, Leonard Myers, Norton, Charles O'Neill, Orth, Patterson, Perham, Pike, Pomeroy, Price, Alexander H. Rice, John H. Rice, Edward H. Rollins, Schenck, Scofield, Shannon, Sloan, Smithers, Stevens, Thayer, Upson, Ellihu B. Washburne, William B. Washburn, Williams, Wilder, Wilson, Windom, Woodbridge-70. Woodbridge-76.

And the bill passed the House and was approved by the President.

#### IN SENATE.

June 21-On a supplement to the charter of the Washington and Georgetown Railroad Company, in Committee of the Whole, Mr. SUMNER moved to insert:

Provided, That there shall be no exclusion of any person from any car on account of color.

Which was rejected-yeas 14, nays 16, as follows:

YEAS-Messrs. Anthony, Brown, Chandler, Clark, Colla-mer, Conness, Dixon, Foot, Howard, Morgan, Pomeroy, Summer, Wade, Wilson-14.

NATS-Buckalew, Carlile, Cowan, Davis, Foster, Grimes, Hendricks, Johnson, Lane of Indiana, Powell, Riddle, Sauls-bury, Sherman, Ten Eyck, Trnmbull, Willey-16.

Same day, in the Senate, Mr. SUMNER renewed the amendment, which was agreed to-yeas 17, nays 16:

YEAS-Messrs. Brown, Clark, Conness, Dixon, Foot, Hale, Harlan, Howe, Lane of Kansas, Morgan, Morrill, Pomeroy, Ramsey, Sprague, Sumner, Wade, Wilson-17. NAYS-Messrs. Buckalew, Carlile, Cowan, Doolittle, Foster, Grimes, Johnson, Lane of Indiana, Powell, Riddle, Sauls-bury, Sherman, Ten Eyck, Trumbull, Van Winkle, Wil-ley-16.

The bill then passed the Scnate-yeas 23, nays 8, (Messrs. Buckalew, Carlile, Cowan, Hale, Lane of Indiana, Powell, Riddle, Saulsbury.) June 29-The bill fell, the House having

tabled the report of the Committee of Conference on the disagreeing votes of the two Houses on it.

### COLORED PERSONS AS WITNESSES.

# Second Session, Thirty-Seventh Congress.

#### IN SENATE.

Pending the confiscation bill, June 28, 1862, Mr. SUMNER moved these words as an addition to the 14th section:

And in all proceedings under this act there shall be no exclusion of any witness on account of color.

Which was rejected-yeas 14, nays 25, as follows:

YEAS--Messrs. Chandler, Grimes, Harlan, Howard, King, Lane of Kansas, Morrill, Pomeroy, Sumner, Trumbull, Wade, Wilkinson, Wilmot-14. NAYS--Messrs. Anthony, Browning, Carlile, Clark, Colla-mer, Cowan, Davis, Dixon, Doolittle, Fessenden, Foot, Fos-ter, Harris, Henderson, Lane of Indiana, Nesmith, Pearce, Powell, Sherman, Simmons, Stark, Ten Eyck, Willey, Wil-son of Missouri, Wright-25.

Pending the consideration of the supplement to the emancipation bill for the District of Columbia,

1862, July 7-Mr. SUMNER moved a new section :

That in all the judicial proceedings in the District of Co-lumbia there shall be no exclusion of any witness on account of color.

Which was adopted-yeas 25, nays 11, as follows:

YEAS-Messrs. Anthony, Chandler, Clark, Collamer, Doo-little, Fessenden, Foot, Foster, Grimes, Hale, Harlan, Harris, Howe, King, Lane of Kansas, Morrill, Sherman, Simmons, Sunner, Ten Eyck, Trumbull, Wade, Wilkinson, Wilmot, Wilson of Massachusetts-25. NAYS-Messrs. Browning, Carlile, Cowan, Davis, Hender-son, Kennedy, McDougall, Provell, Rice, Willey, Wright-11.

The bill then passed-yeas 29, nays 6; (Messrs. Carlile, Davis, Kennedy, Powell, Wilson, of Missouri, Wright.)

July 9-The bill passed the House-yeas 69, nays 36. There was no separate vote on the above proposition. The nays were :

NATS-Messrs. William Allen, Ancona, Baily, Biddle, Jacob B. Blair, Clements, Cobb, Corning, Cox, Crisfield, Dunlap, English, Fouke, Grider, Harding, Knapp, Law, Lazear, Mallory, Maynard, Menzies, Morris, Nugen, Fendle-ton, Perry, Richardson, James S. Rollins, Shiel, John B. Steele, William G. Steele, Stiles, Francis Thomas, Voorhees, Ward, Webster, Wood-36.

Pending the consideration in the Senate of the House bill in relation to the competency of witnesses in trials of equity and admiralty,

1862, July 15-Mr SUMNER offered this proviso to the first section :

Provided, That there shall be no exclusion of any witness on account of color.

Which was rejected-yeas 14, nays 23, as follows :

YEAS-Messrs. Chandler, Grimes, Harlan, Howard, Howe, King, Lane of Kansas, Pomeroy, Rice, Summer, Wade, Wil-

RING, Lanc of Kansas, Pomeroy, Ricc, Sumner, Wade, Wil-kinson, Wilmot, Wilson of Massachusetts—14. NAYS—Messrs. Anthony, Bayard, Browning, Clark, Cowan, Davis, Doolittle, Foster, Hale, Harris, Henderson, Kennedy, Lane of Indiana, Powell, Sculkiway, Sherman, Simmons, Stark, Ten Eyck, Trumbull, Willey, Wilson of Missouri, Warnher, Stark, Sta Wright-23.

# First Session, Thirty-Eighth Congress.

1864, June 25-Pending the civil appropriation bill, in Committee of the Whole, Mr. SUM-NER offered this proviso :

Provided, That in the courts of the United States there shall be no exclusion of any witness on account of color.

Mr. BUCKALEW moved to add :

Nor in civil actions because he is a party to or interested in the issue tried.

Which was agreed to; and the amendment as amended was agreed to-yeas 22, nays 16, as follows:

YEAS-Messrs. Anthony, Brown, Chandler, Clark, Colla-mer, Conness, Foot, Foster, Grimes, Hale, Harlan, Howard, Howe, Lane of Kansas, Morgan, Morrill, Pomeroy, Sprague, Summer, Wade, Wilkinson, Wilson-22. NATS-Messrs. Buckaleto, Cartile, Cowan, Davis, Harris, Buckaleto, User, New W. Beng, Bible and A.

Richardson, Hendricks, Hicks, Johnson, Nesmith, Powell, Richardso Saulsbury, Sherman, Trumbull, Van Winkle, Willey-16.

The Senate subsequently concurred in this amendment-yeas 29, nays 10, as follows:

YEAS-Messrs. Anthon y, Brown, Chandler, Clark, Conness, Dixon, Doolittle, Fessenden, Foot, Foster, Grimes, Hale, Harlan, Harris, Howard, Howe, Lane of Indiana, Lane of Kansas, Morgan, Morrill, Pomeroy, Ramsey, Sherman,\* Spragne, Sumner, Ten Eyck, Wade, Wilkinson, Wilson -29.

NAYS-Messrs. Buckalew, Carlile, Hendricks, Hicks, Nes-mith, Powell, Saulsbury, Trumbull, Van Winkle, Willey -10.

\* Before this vote was taken, Mr. SHERMAN said: "It is due to myself to say in explanation that I voted against and opposed this amendment for the sole ground, as I stated, that it ought not to be put upon this bill. That is my deliberate conviction yet; but as the Senate have by a majority vote decided to put the amendment on the bill in spite of my remonstrances and resistance, I feel bound now to vote according to my conviction on the merits of the proposition."

# IN HOUSE.

June 29-The question being on agreeing to the amendment,

Mr. MALLORY moved to add this proviso to the section amended in the Senate:

Provided, That negro testimony shall only be taken in the United States courts in those States the laws of which authorize such testimony.

Which was rejected—yeas 47, nays 66. The amendment of the Senate was then agreed to-yeas 67, nays 48, as follows :

agreed to-yeas 67, nays 48, as follows: YEAS-Messrs. Allison, Ames, Arnold, Ashley, Baily, John D. Baldwin, Beaman, Boutwell, Boyd, Broomall, Cobb, Cole, Thomas T. Davis, Dawes, Deming, Dixon, Don-nelly, Driggs, Eckley, Eliot, Farnsworth, Fenton, Frank, Garfield, Gooch, Higby, Hooper, Hotchkiss, Hulburd, In-gersoll, Jenckes, Francis W. Kellogg, Orlando Kellogg, Knox, Littlejohn, Loan, Longyear, McBride, McClurg, Moorhead, Morrill, Daniel Morris, Amos Myers, Leonard Myers, Norton, Charles O'Neill, Patterson, Perham, Alex-ander H. Rice, John H. Rice, Edward H. Kollins, Schenck, Schofield, Shannon, Sloan, Smithers, Spaulding, Stevens, Thayer, Upson, Van Valkenburgh, Elihu B. Washburne, William B. Washburn, Williams, Wilder, Wilson, Win-dom-67. dom-67.

dom-67. NATS-Messrs. William J. Allen, Ancona, Augustus C. Baldwin, Blair, Bliss, Brooks, William G. Brown, Chanler, Coffroth, Dawson, Dcrison, Eden, Edgerton, Eldridge, Finck, Harding, Eenjamin G. Harris, Charles M. Harris, Herrick, Holman, William Johnson, Knapp, Le Blond, Mallory, Mar-cy, James B. Morris, Morrison, Noble, John O'Neill, Pen-dleton, Perry, Samuel J. Randall, Robinson, Ross, John B. Steele, William G. Sleele, Stiles, Strouse, Shuart, Thomas, Tracy, Wadsworth, Ward, Webster, Whaley, Wheeler, Chil-ton A. White, Joseph W. White-48.

REPEAL OF LAWS REGULATING THE COASTWISE SLAVE TRADE.

#### First Session, Thirty-Eighth Congress.

1863, June 25-Mr. SUMNER offered this additional section, the Senate sitting as in Committee of the Whole, pending the consideration of the civil bill:

And be it further enacted, That sections eight and nine of the act entitled "An act to prohibit the importation of slaves into any port or place within the jurisdiction of the United States from and after the 1st day of January, in the year of our Lord 1808," which sections undertake to regulate the coastwise slave trade, are hereby repealed, and the coastwise slave trade prohibited forever.

Which was rejected-yeas 13, nays 20, as follows:

YEAS—Messrs. Conness, Grimes, Harlan, Howard, Lane of Kansas, Morgan, Morrill, Pomeroy, Ramsey, Sprague, Sumner, Wade, Wilson—13. NATS—Messrs. Buckalew, Carbile, Clark, Collamer, Cowan, Davis, Harris, Hendricks, Hicks, Howe, Johnson, McDou-gall, Nesmith, Powelk, Richardson, Riddle, Saulsbury, Sherman, Trumbull, Van Winkle—20.

Same day, in open Senate,

Mr. SUMMER renewed the amendment, which was agreed to-yeas 23, nays 14, as follows:

YEAS-Messrs. Anthony, Brown, Chandler, Conness, Dixon, Doolitle, "Fessenden, Foot, Harlan, Harris, Howard, Howe, Lane of Kansas, Morgan, Morrill, Pomeroy, Sprague, Summer, Ten Eyck, Wade, Wilkinson, Wilson-23.

NATS-Messrs. Buckalevo, Carlile, Clark, Hendricks, Hicks, Johnson, Lane of Indiana, Nesmith, Powell, Rich-ardson, Saulsbury, Sherman, Trumbull, Van Winkle, Wil-ley-14.

The bill passed the Senate-yeas 32, nays 4, (Messrs. Carlile, Hendricks, Powell, Saulsbury.) June 29-The House agreed to the amendment without a division, after a brief debate.

\* Before the vote was taken, Mr. DOOLITTLE said : "I voted against this amendment before on the ground that I did not like to vote for such measures on appropria-tion bills; but two or three others have been put on, and if this is to be legislated upon, as I am in favor of the aboli-tion of the coastwise slave trade, I shall vote in the affirmative."

#### T. W. HIGGINSON'S PARTICIPATION IN THE BURNS CASE.

1864, March 14-Mr. DAVIS offered this resolution :

Whereas in the history of the attempt to rescue Anthony Burns, a fugitive slave from the State of Virginia, from the custody of the United States officers in Boston, in 1854, it is represented, and it is also generally reported, that T. W. Higginson, now the colonel of a regiment of negro troops in the curric of the United States lad or use an encoded in an Inguistic, now the control of a regiment of negligible trops of the service of the United States, led, or was engaged in, an assault made by a body of men, with force and arms, upon the court-house in Boston, where the said Anthony Burns was held in the custody of the law and officers of the United States, with the intent and purpose of forcibly rescuing him States, with the intent and purpose of forcibly reseming initial from such custody; and whereas it is represented and gen-erally reported that a citizen of the United States, then hav-ing the custody of said Burns, was killed and murdered by said assailants: Therefore, be it *Resolved*, That the president of the Senate appoint a com-mittee of three members of the Senate to investigate whether

the said T. W. Migginson had any connection, and if any, what, with the said attempt to rescue the said Burns, and with the killing and murdering of any person having his custody, and that said committee have power to send for persons and papers.

March 17-Mr. TRUMBULL moved to table it; which was agreed to-yeas 29, nays 10, as follows:

YEAS-Messrs. Anthony, Brown, Chandler, Collamer, Dix-on, Doolittle, Fessenden, Foot, Foster, Grimes, *Harding*, Har-lan, Howard, Howe, Lane of Indiana, Lane of Kansas, Mor-gan, Morrill, Pomeroy, Ramsey, Sherman, Sprague, Sunner, Ten Eyck, Trumbull, Van Winkle, Wilkinson, Willey, Wilson-29.

NATS-Messrs. Buckalew, Carlile, Conness, Davis, Hen. dricks, Johnson, McDougall, Powell, Riddle, Saulsbury-10

#### COLORED SCHOOLS.

June 8-The House passed a bill to provide for the public instruction of youth in Washington city, with an amendment providing for separate schools for the colored children, by setting apart such a proportion of the entire school fund as the number of colored children between the ages of six and seventeen bear to the whole number of children in the District. The bill, with amendments, passed both Houses without a division.

# Military Reports, Orders, and Proclamations.

MAJOR GENERAL M'CLELLAN'S PROCLAMATION IN WESTERN VIRGINIA

> HEADQUARTERS DEPARTMENT OF OHIO, CINCINNATI, May 26, 1861.

To the Union men of Western Virginia: VIRGINIANS: The General Government has long enough VIRGINIANS: The General Government has long enough endured the machinations of a few factious rebels in your midst. Armed traitors have in vain endeavored to deter you from expressing your loyalty at the polls. Having failed in this infamous attempt to deprive you of the exer-cise of your dearest rights, they now seek to inaugurate a reign of terror, and thus force you to yield to their schemes and submit to the yoke of the traitorous conspiracy digni-fied by the name of the Southern Confederacy. They are destroving the property of citizens of your State and ruindestroying the property of citizens of your State and ruin-ing your magnificent railways. The General Government has heretofore carefully ab-

The order to overinnent has heretoric careting ap-stained from sending troops across the Ohio, or even from posting them along its banks, although frequently urged by many of your prominent citizens to do so. It determ-ined to wait the result of the State election, desirous that no one might be able to say that the slightest effort had been made from this side to influence the form correction of results. no one might be able to say that the signest cast had teen made from this side to influence the free expression of your opinions, although the many agencies brought to bear upon you by the rebels were well known. You have now shown, under the most adverse circumstances, that the great mass of the people of Western Virginia are true and loyal to that

Government cannot close its ears to the demand you have made for assistance. I have ordered troops to cross the river. They come as your friends and brothers—as enemies only to armed rehels, who are preying upon you; your homes, your families, and your property are safe under our protection. All your rights shall be religiously respected, netwithstanding all that has been said by the traitors to induce you to believe our advent among you will be signal-ized by an interference with your slaves. Understand one thing clearly: not only will we abstain from all such inter-ference, but we will, on the contrary, with an iron hand crush any attempt a insurrection on their part. Now that binds you to traitors; proclaim to the world that the binds you to traitors; proclaim to the world that the binds you to traitors. The component of the order of the binds of the stars and stripes. G. B. McCLELLAN, Government cannot close its ears to the demand you have

G. B. McCLELLAN, Major General Commanding.

MAJOR GENERAL ROBERT PATTERSON'S PROCLAMA-TION.

> HEADQUARTERS DEPARTMENT OF PENNSYLVANIA CHAMBERSBURG, PA., June 3, 1861.

To the United States Troops of this Department :

The restraint which has necessarily been imposed upon you, impatient to overcome those who have raised their parricidal hands against our country, is about to be re-moved. You will soon meet the insurgents.

You are not the aggressors. A turbulent faction, misled by ambitious rulers, in times of profound peace and national prosperity, have occupied your forts and turned the guns against you; have scized your arsenals and armories, and appropriated to themselves Government supplies; have ar-rested and held as prisoners your companions marching to their homes under State pledge of security, and have cap-tured unseele and newsing columbrially argument to the State tured vessels and provisions voluntarily assured by State legislation from molestation, and now seek to perpetuate a reign of terror over loyal citizens.

They have invaded a loyal State, and entrenched them-selves within its boundaries in defiance of its constituted authorities.

You are going on American soil to sustain the civil power, to relieve the oppressed, and to retake that which is unlawfully held.

You must bear in mind you are going for the good of the whole country, and that, while it is your duty to punish sedition, you must protect the loyal, and, should the occasion offer, at once suppress servile insurrection

Success will crown your efforts; a grateful country and a happy people will reward you. By order of MAJOR GENERAL PATTERSON:

F. J. PORTER, Asst. Adj. General.

#### SECRETARY CAMERON'S LETTER TO GEN. BUTLER. WASHINGTON, May 30, 1861.

SIR: Your action in respect to the negroes who came within your lines from the service of the rebels is approved. The Department is sensible of the embarrassments which must surround officers conducting military operations in a State by the laws of which slavery is sanctioned. The Gov-ernment cannot recognize the rejection by any State of the Federal obligations, nor can it refuse the performance of the Federal obligations resting upon itself. Among these Federal obligations, however, none can be more important than that of suppressing and dispersing armed combina-tions formed for the purpose of overthrowing its whole constitutional authority. While, therefore, you will per-mit no interference by the persons under your command with the relations of persons held to service under the laws of any State, you will, on the other hand, so long as any State must surround officers conducting military operations in a of any State, you will, on the other hand, so long as any State within which your military operations are conducted is under the control of such armed combinations, refrain from sur-rendering to alleged masters any persons who may come within your lines. You will employ such persons in the services to which they may be best adapted, keeping an ac-count of the labor by them performed, of the value of it, and the expenses of their maintenance. The question of their final disposition will be reserved for future deter-mination. mination.

SIMON CAMERON, Secretary of War.

To Major General BUTLER.

MAJOR GENERAL BUTLER TO LIEUTENANT GENERAL SCOTT. HEADQUARTERS DEPARTMENT OF VIRGINIA,

beneficent Government under which we and our fathers lived so long. As soon as the result of the election was known the trai-tors commenced their work of destruction. The General

negroes in the batteries, and are preparing to send the womer and children south. The escapes from them are very numerous, and a squad has come in this morning to my pickets bringing their women and children. Of course these cannot be dealt with upon the theory on which I designed to treat the services of able-bodied men and women signed to treat the services of able-bodied men and women who might come within my lines, and of which I gave you a detailed account in my last despatch. I am in the ut-most doubt what to do with this species of property. Up to this time I have had come within my lines men and wo-men with their children, entire families, each family be-longing to the same owner. I have, therefore, determined to employ, as I can do very profitably, the able-bodied per-sons in the party, issuing proper food for the support of all, and charging against their services the expense of care and ensteance of the non-laborars keeping a strict and exand sustenance of the non-laborers, keeping a strict and so-curate account as well of the services as of the expendi-ture, having the worth of the services and the cost of the ture, naving the worth of the sorvices and the cost of the expenditure determined by a board of survey, to be here-after detailed. I know of no other manner in which to dispose of this subject and the questions connected there-with. As a matter of property to the insurgents, it will be of very great moment, the number that I now have mounting as I am information to what in good times be of very great moment, the number that I now have amounting, as I am informed, to what, in good times, would be of the value of sixty thousand dollars. Twelve of these negroes, I am informed, have escaped from the batteries on Sewall's Point, which, this morning, fired upon my expedition as it passed by out of range. As means of offence therefore in the enemy's hands, these negroes, when able-bodied, are of the last importance. Without them the batteries could not have been erected, at least for many weeks. As a military question, it would at least for many weeks. As a minitary question, it would seem to be a measure of necessity to deprive their masters of their services. How can this be done? As a political question and a question of humanity, can I receive the services of a father and mother, and not take the children ? Of the humanitarian aspect I have no doubt. Of the political one I have no right to judge. I therefore submit all this to your better judgment, and as the questions have wrong in so doing, to duplicate the parts of my despatch relating to this subject, and forward them to the Secretary of War. \* \* \* Very respectfully, your obedient servant, B. F. BUTLER.

Lientenant General SCOTT.

#### SECRETARY CAMERON'S REPLY TO GEN. BUTLER.

WASHINGTON, August 8, 1861. GENERAL: The important question of the proper disposi-tion to be made of fugitives from service in States in insurrection against the Federal Government, to which you have again directed my attention in your letter of July 30, has received my most attentive consideration.

It is the desire of the President that all existing rights, in all the States, be fully respected and maintained. The war now prosecuted on the part of the Federal Govern-ment is a war for the Union, and for the preservation of all constitutional rights of States, and the citizens of the States, in the Union. Hence no question can arise as to fugitives from service within the States and Territorics in which the authority of the Union is fully acknowledged. The ordinary forms of judicial proceeding, which must be respected by military and civil authorities alike, will suffice for the enforcement of all legal claims. But in States wholly or partially under insurrectionary control, where the laws of the United States are so far opposed and resisted that they cannot be effectually enforced, it is obvious that rights dependent on the execution of those laws must, tem-porarily, fail; and it is equally obvious that trights dependent It is the desire of the President that all existing rights, porarily, fail; and it is equally obvious that rights depend-ent on the laws of the States within which military operations are conducted must be necessarily subordinated to the military exigencies created by the insurrection, if not wholly forfeited by the treasonable conduct of parties claiming them. To this general rule rights to services can form

no exception. The act of Congress approved August 6, 1861, declares that if persons held to services all be employed in hostility to the United States, the right to their services shall be forfeited, and such persons shall be discharged therefrom. It follows of necessity that no claim can be recognized by the military authorities of the Union to the services of such persons when fugitives.

and persons when fugitives. A more difficult question is presented in respect to per-sons escaping from the service of loyal masters. It is quite apparent that the laws of the State, under which only the services of such fugitines can be claimed, must needs be wholly, or almost wholly, suspended, as to remedies, by the insurrection and the military measures necessitated by it. And it is equally apparent that the substitution of military for judicial measures for the enforcement of such claims must be attended by creat inconveniences embarrasments. must be attended by great inconveniences, embarrassments, and injuries.

Under these circumstances it seems quite clear that the substantial rights of loyal masters will be best protected by receiving such fugitives, as well as fugitives from disloyal masters, into the service of the United States, and employing them under such organizations and in such occupations as circumstances may suggest or require. Of course a record should be kept showing the name and description of the fugitives, the name and the character, as logal or dis-loyal, of the master, and such facts as may be necessary to a correct understanding of the circumstances of each case after tranquility shall have been restored. Upon the re-tu  $\cdot$  of peace, Congress will, doubtless, properly provide for all the persons thus received into the service of the Union, at for just compensation to loyal masters. In this way on v, it would seem, can the duty and safety of the Governno at and the just rights of all be fully reconciled and harne nized.

You will therefore consider yourself as instructed to govyour future action, in respect to fugitives from service, the principles herein stated, and will report from time time, and at least twice in each month, your action in the premises to this Department. You will, however, neither uthorize nor permit any interference, by the troops under your command, with the servants of peaceful citizens in house or field; nor will you, in any way, encourage such servants to leave the lawful service of their masters; nor will you, except in cases where the public safety may seem to require, prevent the voluntary return of any fugitive, to the service from which he may have escaped.

I am, General, very respectivily, your obcdient servant, SIMON CAMERON, Secretary of War. Maj. Gen. B. F. BUTLER, Commanding Department of Vir ginia, Fortress Monroe. Ur.

#### CONCERNING FUGITIVE SLAVES.

HEADQUARTERS DEPARTMENT OF WASHINGTON. WASHINGTON, July 17, 1861.

[General Orders No 33.]

Fugitive slaves will, under no pretext whatever, be per-mitted to reside or be in any way harbored in the quarters and camps of the troops serving in this Department. Neither will such slaves be allowed to accompany troops on the march.

Commanders of troops will be held responsible for a strict observance of this order. By command of Brigadier General Mansfield

THEO. TALBOT,

Assistant Adjutant General.

Complaint having been made that slaves were abducted by soldiers going North on the cars, the following order was issued:

HEADQUARTERS CITY GUARD, WASHINGTON, August 10, 1861.

To Captain H. DAVIDSON, Commanding Guard at Railroad Depot: SIR: It is directed by the Provost Marshal that you permit no soldiers to leave this city by the railroad who are unable to show that they have been properly discharged from the service of the United States; also, that no negroes, without sufficient evidence of their being free or of their right to travel, are permitted to leave the city upon the cars.

I am, Captain, very respectfully, your obcdient servant, W. W. AVERELL, A. A. A. G.

EMANCIPATION PROCLAMATION OF GEN. FREMONT.\*

HEADQUARTERS OF THE WESTERN DEPARTMENT,

ST. LOTIS, August 31, 1861. Circumstances in my judgment, of sufficient urgency, render it necessary that the commanding general of this

\* This Retaliatory Proclamation was issued by a rebel officer:

HEADQUARTERS FIRST MILITARY DISTRICT M. S. G. CAMP HUNTER, September 2, 1861.

To all whom it may concern:

Whereas Major General John C. Fremont, commanding the minions of Abraham Lincoln in the State of Missonri, has seen fit to declare martial law throughout the whole State, and has threatened to shoot any citizen soldier found in arms within certain limits; also, to confiscate the propat the transformer of the second seco

General of the first infinitely district of Massouri, having not-only the military authority of Brigadier General, but cer-tain police powers granted by Acting Governor Thomas G. Reynolds, and confirmed afterwards by Governor Jackson, do most solemnly promise that for every member of the Missouri State Guard or soldier of our alies, the armies of the Confederate States, who shall be put to death in pursu-

Department should assume the administrative powers of the State. Its disorganized condition, the helplessness of the civil authority, the total insecurity of life, and the devastation of property by bands of murderers and marauders, who in-fest nearly every county of the State, and avail themselves of the public misfortunes and the vicinity of a hostile force to gratify private and neighborhood vengeance, and who find an enemy wherever they find plunder, finally demand the severest measures to repress the daily increasing crimes and outrages which are driving off the inhabitants and ruining the State.

In this condition the public safety and the success of our arms require unity of purpose, without let or hindrance, to the prompt administration of affairs.

In order, therefore, to suppress disorder, to maintain as far as now practicable the public peace, and to give security and protection to the persons and property of loyal citizens, I do hereby extend and declare established martial law throughout the State of Missouri.

The lines of the army of occupation in this State are for the present declared to extend from Leavenworth by way of the posts of Jefferson City, Rolla, and Ironton, to Cape Girardeau, on the Mississippi river.

All persons who shall be taken with arms in their hands within these lines shall be tried by court-martial, and if found guilty will be shot.

The property, real and personal, of all persons, in the State of Missenri, who shall take up arms against the Uni-ted States, or who shall be directly proven to have taken an active part with their gnemies in the field, is declared to be confiscated to the public use, and their slaves, if any they have, are hereby declared free men.

All persons who shall be proven to have destroyed, after the publication of this order, raitroad tracks, bridges, and the publication of this order, raitroad tracks, bridges, or tele-graphs, shall suffer the extreme penalty of the law. All persons engaged in treasnable correspondence, in giving or procuring aid to the enemies of the United States,

in fomenting tumults, in disturbing the public tranquillity by creating and circulating false reports or incendiary documents, are in their own interests warned that they are

exposing themselves to sudden and severe punishment. All persons who have been led away from their allegi-ance are required to return to their homes forthwith; any such absence, without sufficient cause, will be held to be presumptive evidence against them.

The object of this declaration is to place in the hands of the military authorities the power to give instantaneous effect to existing laws, and to supply such deficiencies as the conditions of war demand. But this is not intended to suspend the ordinary tribunals of the country, where the law will be administered by the civil officers in the usual manner, and with their customary authority, while the same can be peaceably exercised.

The commanding general will labor vigilantly for the public welfare, and in his efforts for their safety hopes to obtain not only the acquisseence, but the active support of the loyal people of the country.

J. C. FREMONT. Major General Commanding.

### [SPECIAL MILITARY ORDER.]

HEADQUARTERS WESTERN DEPARTMENT,

ST. LOUIS, August 30, 1861.

ST. LOUIS, August 30, 1861. The commanding general sincerely regrets that he finds it necessary to make any reproach to the patriotic army under his command. He had hoped that the rigid enforce-ment of discipline and the good example of the mass of the enlightened soldiery which he has the honor to lead, would have been sufficient to correct in good time the irregulari-ties and licence of the few who have reflected discredit HDDO OUR cause and correcting. upon our cause and ourselves

But the extension of martial law to all the State of Missouri, rendered suddenly necessary by its unhappy condition, renders it equally imperative to call the army to good order and rigorous discipline. They are reminded that the power

ance of the said order of Gen. Fremont, I will hang, draw, and quarter a minion of said Abraham Lincoln.

While I am anxious that this unfortunate war shall be conducted, if possible that this build thate way shall be conducted, if possible that the most liberal principles of civilized warfare—and every order that I have issued has been with that object—yet, if this rule is to be adopted, (and it must first be done by our enemies.) I intend to ex-ceed Gen. Foremont in bis avcesse and will make the issue (and it must first be done by our enemies.) I intend to ex-ceed Gen. Fremont in his excesses, and will make all tories that come within my reach rue the day that a different policy was adopted by their leaders. Already mills, barns, warebourses, and other private property have been wastefully and wantonly destroyed by the enemy in this district, while we have taken nothing except articles strictly con-traband or absolntely necessary. Should these things he repeated, I will rotaliate ten-fold, so help me God! M. JEFF. THOMPSON, Brig. Gen. Commanding.

to inflict the extraordinary severities of the now governing law is rigidly confined to few, who are to be held strictly accountable for its exercise. They are also reminded that the same necessity which requires the establishment of martial law demands also the enforcement of the military law, which governs themselves with the same sudden severity.

The commanding general therefore strictly prohibits all vexatious proceedings calculated unnecessarily to harass the citizens, and also unauthorized searches, seizures, and destruction of property, except in cases of military neces-sity, and for which the officer authorizing or permitting it will be held strictly and personally responsible.

All officers commanding districts, posts, or detachments are enjoined to use the utmost prudence and circumspec-tion in the discharge of their duties. Under the circumthe stances a strict obedience to orders, close attention to du-ties, and an earnest effort to protect and to avoid harassing innocent persons, is requested and expected everywhere from officers and men.

The commanding general trusts that he will find few occasions to reproach the troops. He hopes and helicves that he will find many to admire and commend them.

J. C. FREMONT,

Major General Commanding.

PRESIDENT LINCOLN'S FIRST LETTER TO GENERAL FREMONT.

# [Private.]

WASHINGTON, D. C., Sept. 2, 1861.

MY DEAR SIR: Two points in your proclamation of August 30th give me some anxiety:

First. Should you shoot a man according to the proclamation, the Confederates would very certainly shoot our best men in their hands, in retaliation; and so, man for man, indefinitely. It is, therefore, my order that you allow no man to be shot under the proclamation without first having my approbation or consent.

Second. I think there is great danger that the closing paragraph, in relation to the con-fiscation of property, and the liberating slaves of traitorous owners, will alarm our Southern Union friends and turn them against us; perhaps ruin our rather fair prospect for Kentucky.

Allow me, therefore, to ask that you will, as of your own motion, modify that paragraph so as to conform to the *first* and fourth sections of the act of Congress entitled, "An act to confiscate property used for insurrectionary purposes," approved August 6, 1861, and a copy of which act I herewith send you.

This letter is written in a spirit of caution, and not of censure.

I send it by a special messenger, in that it may certainly and speedily reach you.

A. LINCOLN. Yours, very truly, Major General FREMONT.

#### GEN. FREMONT'S REPLY.

HEADQUARTERS WESTERN DEPARTMENT, ST. LOUIS, September 8, 1861.

MY DEAR SIR : Your letter of the second, by special messenger, I know to have been written before you had received my letter, and before my telegraphic dispatches and the rapid development of critical conditions here had informed you of affairs in this quarter. I had not written to you fully and frequently; first, because in the incessant change of affairs I would be exposed to give you contradictory accounts; and, secondly, because the amount of the subjects to be laid before you would demand too much of your time.

Trusting to have your confidence, I have

been leaving it to events themselves to show | you whether or not I was shaping affairs here according to your ideas. The shortest communication between Washington and St. Louis generally involves two days, and the employment of two days in time of war goes largely toward success or disaster. I therefore went along according to my own judgment, leaving the result of my movement to justify me with And so in regard to my proclamation of vou. the thirtieth. Between the rebel armies, the provisional government, and home traitors, I felt the position bad, and saw danger. In the night I decided upon the proclamation and the form of it-I wrote it the next morning and printed it the same day. I did it without consultation or advice with any one, acting solely with my best judgment to serve the country and yourself, and perfectly willing to receive the amount of censure which should be thought due, if I had made a false movement. This is as much a movement in the war as a battle, and in going into these I shall have to act according to my judgment of the ground before me, If, upon reflection, as I did on this occasion. your better judgment still decides that I am wrong in the article respecting the liberation of slaves, I have to ask that you will openly direct me to make the correction. The implied censure will be received as a soldier always should the reprimand of his chief. If I were to retract of my own accord, it would imply that I myself thought it wrong, and that I had acted without the reflection which the gravity of the point demanded. But I did not. I acted with full deliberation, and upon the certain conviction that it was a measure right and necessary, and I think so still.

In regard to the other point of the proclamation to which you refer, I desire to say that I do not think the enemy can either misconstrue or urge anything against it, or undertake to make unusual retaliation. The shooting of men who shall rise in arms against an army in the military occupation of a country is merely a necessary measure of defence, and entirely according to the usages of civilized warfare. The article does not at all refer to prisoners of war, and certainly our enemies have no grounds for requiring that we should waive in their benefit any of the ordinary advantages which the usages of war allow to us.

As promptitude is itself an advantage in war, I have also to ask that you will permit me to carry out upon the spot the provisions of the proclamation in this respect.

Looking at affairs from this point of view, I am satisfied that strong and vigorous measures have now become necessary to the success of our arms; and hoping that my views may have the honor to meet your approval,

I am, with respect and regard, very truly

you, upon the ground, could better judge of the necessities of your position than I could at this distance, on seeing your proclamation of August 30th, I perceived no general objection to it. The particular clause, however, in relation to the confiscation of property and the liberation of slaves, appeared to me to be objectionable in its non-conformity to the act of Congress passed the 6th of last August upon the same subjects; and hence I wrote you expressing my wish that that clause should be modi-Your answer, just received, fied accordingly. expresses the preference, on your part, that I should make an open order for the modification, which I very cheerfully do. It is therefore ordered that the said clause of said proclamation be so modified, held, and construed as to conform to and not to transcend the provisions on the same subject contained in the act of Congress entitled "An act to confiscate property used for insurrectionary purposes," ap-proved August 6, 1861, and that said act be published at length with this order.

Your obedient servant,

A. LINCOLN. Maj. Gen. JOHN C. FREMONT.

GENERAL FREMONT'S LETTER TO REBEL COLONEL

TAYLOR, EXPLAINING HIS PROCLAMATION.

HEADQUARTERS WESTERN DEPARTMENT, September 14, 1861.

Col. T. T. TAYLOR, Commanding at Springfield, Mo.:

SIR: Yours of the 8th instant, containing an erroneous construction of my proclamation of the 30th ultimo, has had my attention

I understand the object of your note to be to enquire whether it was my intention to shoot the wounded who might be made prisoners by the forces under my com-mand. The following paragraph, extracted from the pro-clamation, will be strictly enforced within the lines pre-scribed against the class of offenders for whom it was in-

school again the class of obtained is of whom it was in-tended, viz: "All persons who shall be taken with arms in their hands within these lines shall be tried by a court martial, and if found guilty will be shot."

The lines are expressly declared to be those of the army in the military occupation of this State. You have wholly misapprehended the meaning of the proclamation. Without undertaking to determine the condition of any man engaged in this rebellion, I desire it to be clearly understood that the proclamation is intended distinctly to recognize the usual rights of an open enemy in the field, and to be in all respects strictly conformable with the usages of war. It is hardly necessary for me to say that it was not reported with any nurrose to improve say that it was not prepared with any purpose to ignore the ordinary rights of humanity with respect to the wounded men, or those who are humanely engaged in alleviating their sufferings.

J. C. FREMONT, Major General Commanding.

GENERAL INSTRUCTIONS OF THE WAR OFFICE.

WAR DEPARTMENT, October 14, 1861.

SIR: In conducting military operations within States de-clared by the proclamation of the President to be in a state of insurrection, you will govern yourself, so far as persons held to service under the laws of such States are concerned, by the principles of the letters addressed by me to Major General Butler on the 30th of May and the 8th of August, copies of which are herewith furnished to you. As special 

 Tam, with respect and regard, tery truly

 yours.

 The PRESIDENT.
 J. C. FREMONT.

 THE PRESIDENT.
 J. C. FREMONT.

 THE PRESIDENT'S REJOINDER ORDERING A MODI 

 FICATION OF THE PROCLAMATION.

 WASHINGTON, September 11, 1861.

 SIR : Yours of the 8th, in answer to mine of the 2d instant, is just received. Assuming that

 them for military service. You will assure all loyal masters that Congress will provide just compensation to them for the loss of the services of the persons so employed. It is believed that the course thus indicated will best secure the substantial rights of loyal masters, and the benefits to the United States of the services of all disposed to support the Government, while it avoids all interference with the social systems or local institutions of every State, beyond that which insurrection makes unavoidable, and which a restoration of peaceful relations to the Union, under the Constitution, will immediately remove.

SIMON CAMERON.

Secretary of War. Brigadier General T. W. SHERMAN,

Commanding Expedition to the Southern Coast.

In pursuance of these instructions, a proclamation was issued by General Sherman to the people of South Carolina, saying that-

In obedience to the orders of the President of these United States of America, I have landed on your shores with a small force of national troops. The dictates of a duty which, under these circumstances, I owe to a great duty which, under these circumstances, I owe to a great sovereign State, and to a proud and hospitable people, among whom I have passed some of the pleasantest days of my life, prompt me to proclaim that we have come amongst you with no feelings of personal animosity, no desire to harm your citizens, destroy your property, or in-terfere with any of your lawful rights or your social or local institutions, beyond what the causes herein alluded to may render unavoidable.

Major General Dix also issued a proclamation to the people of Accomac and Northampton counties, Virginia, dated November 13, 1861, beginning as follows:

The military forces of the United States are about to other your counties as a part of the Union. They will go enter your counties as a part of the Union. among you as friends, and with the earnest hope that they may not, by your own acts, be forced to become your ene-mies. They will invade no rights of person or property. on the contrary, your laws, your institutions, your usages, will be scrupulously respected. There need be no fear that the quietude of any fireside will be disturbed, unless the disturbance is caused by yourselves.

Special directions have been given not to interfere with the condition of any person held to domestic service; and, in order that there may be no ground for mistake or pretext for misrepresentation, commanders of regiments and corps have been instructed not to permit any such persons to come within their lines.

The same day, Major General Wool issued this order:

HEADQUARTERS DEPARTMENT OF VIRGINIA, FORT MONROE, October 14, 1861.

[Special Orders No. 72.]

All colored persons called coutrabands, employed as ser-vants by officers and others residing within Fort Monroe, croutside of the Fort at Camp Hamilton and Camp Butler will be furnished with their subsistence and at least eight dollars per month for males, and four dollars per month for females, by the officers or others thus employing them.

So much of the above named sums as may be necessary to furnish clothing, to be decided by the Chief Quartermas-ter of the department, will be applied to that purpose, and the remainder will be paid into his hands to create a fund for the support of those contrabands who are unable to work for their own support.

All able-odied colored persons who are under the pro-tection of the troops of this department, and who are not employed as servants, will be immediately put to work in either the Engineer's or Quartermastor's Department. By command of Major General Wool: (Signed) WM. D. WHIPPLE, Assistant Adjutant General.

Again, November 1, 1861 :

HEADQUARTERS DEPARTMENT OF VIRGINIA FORT MONROE, November 1, 1801.

[General Orders No. 34.] The following pay and allowances will constitute the valuation of the labor of the contrabands at work in the Engineer, Ordnance, Quartermaster, Commissary, and Medi-cal Departments at this post, to be paid as hereinafter mentioned:

Class 1st .- Negro men over eighteen years of age, and

able-bodied, ten dollars per month, one ration and the necessary amount of clothing. Class 2d.—Negro boys from 12 to 18 years of age, and sickly and infrm negro men, five dollars per month, one ration, and the necessary amount of clothing.

The quartermaster will furnish all the clothing. The department employing these men will furnish the subsistence specified above, and as an incentive to good behavior (to be withheld at the direction of the chiefs of the departments respectively) each individual of the first class will receive \$2 per month, and each individual of the second class \$1 per month, for their own use. The remainder of the money valuation of their labor will be turned over to the quartermaster, who will deduce tfrom it the cost of the clothing issued to them; the balance will constitute a fund to be expended by the quartermaster under the direction of the commanding officer of the department of Virginia for the support of the women and children and those that are unable to work.

For any unnsual amount of labor performed they may receive extra pay, varying in amount from fifty cents to one dollar, this to be paid by the departments employing them,

to the men themselves, and to be for their own use. Should any man be prevented from working, on account of sickness, for six consecutive days, or ten days in any one month, one half of the money value will be paid. For being prevented from laboring for a longer period than ten Agy in any one month all pay and allowances cease. By command of Major General Wool: (Signed) WM. D. WHIPPLE, Assistant Adjutant General.

#### GENERAL HALLECK'S ORDER OF NOVEMBER 20. 1861.\*

#### HEADQUARTERS DEPARTMENT OF MISSOURI. ST. LOUIS, November 20, 1861.

[General Orders No. 3.]

I. It has been represented that important information, respecting the number and condition of our forces, is con-veyed to the enemy by means of fugitive slaves who are admitted within our lines. In order to remedy this evil, it is directed that no such persons be hereafter permitted to enter the lines of any camp, or of any forces on the march; and that any now within such lines be immediately excluded therefrom.

#### CONTRABANDS IN THE DISTRICT.

DEPARTMENT OF STATE

WASEINGTON, December 4, 1861. To Major General GEO. B. MCCLELLAN, Washington. GENERAL: I am directed by the President to call your

GENERAL: I am directed by the President to call your attention to the following subject: Persons claimed to be held to service or labor under the laws of the State of Virginia, and actually employed in hostile service against the Government of the United States, frequently escape from the lines of the enemy's forces and are received within the lines of the army of the Potomac.

This Department understands that such persons afterwards coming into the city of Washington are liable to be arrested by the city police, upon the presumption, arising from color, that they are fugitives from service or labor. By the 4th section of the act of Congress approved An-

By the 4th section of the act of concress approved Af-gust 6th, 1861, entitled "An act to confiscate property used for insurrectionary purposes," such hostile employ-ment is made a full and sufficient answer to any further claim to service or labor. Persons thus employed and es-caping are received into the military protection of the United States, and their arrest as fugitives from service or labor should be immediately followed by the military arrest of the parties making the seizure.

Copies of this communication will be sent to the Mayor of the city of Washington and to the Marshal of the Dis-

\* General Halleck, in a letter, thus explained the order:

\* General Halleck, in a letter, thus explained the order: HEADQUARTERS DEPARTMENT OF THE MISSOULI, Sr. LOUIS, December 8, 1861. MY DEAR COLONEL: YOURS of the 4th instant is just re-ceived. Order No. 3 was, in my mind, clearly a military necessity. Unauthorized persons, black or white, free or slaves, must be kept out of our camps, unless we are wil-ling to publish to the enemy everything we do or intend to do. It was a military and not a political order. I am ready to curry out any lawful instructions in regard to furtive slaves which my superiors may give me, and to

to fugitive slaves which my superiors may give me, and to to ugnive slaves which my superiors may give me, and to enforce any law which Congress may pass. But I cannot make law, and will not violate it. You know my private opinion on the policy of confiscating the slave property of rebels in arms. If Congress shall pass it, you may be cer-tain that I shall enforce it. Perhaps my policy as to the treatment of rebels and their property is as well set out in Order No. 13, issued the day your letter was written, as I could now describeit. could now describe it.

Hon. F. P. BLAIR, Washington.

Respectfully,

trict of Columbia, that any collision between the civil and the rebels, and their rapine would be defended under the laws of war. While the loyal States thus have all their I am, General, your very obedient,

#### WILLIAM H. SEWARD.

#### FROM THE REPORT OF THE SECRETARY OF WAR, DEC. 1. 1861.

It is already a grave question what shall be done with It is aircady a grave question what shall be done with those shaves who were abandoned by their owners on the advance of our troops into southern territory, as at Beau-fort district, in South Carolina. The number left within our control at that point is very considerable, and similar cases will probably occur. What shall be done with thom? Can we afford to send them forward to their masters, to be by them armed against us, or used in producing supplies to sustain the rebellion? Their labor may be useful to us; by them armen against us to use in boot may be useful to us; withheld from the enemy it lessens his military resources, and withholding them has no tendency to induce the horrors of insurrection, even in the rebel communities. horrors of insurrection, even in the react commutatives. They constitute a military resource, and, being such, that they should not be turned over to the enemy is too plain to discuss. Why deprive him of supplies by a blockade, and voluntarily give him men to produce them? The disposition to be made of the slaves of robels, after the discussion of the slaves of robels, after

the close of the war, can be safely left to the wisdom and patriotism of Congress. The Representatives of the people will unquestionably secure to the loyal slaveholders every right to which they are entitled under the Constitution of the country.

#### SIMON CAMERON Secretary of War.

It is understood that the Report of the Secretary of War, as originally prepared was as follows, and that it was modified at the request of the PRESIDENT :

It has become a grave question for determination what shall be done with the slaves abandoned by their owners on the advance of our trops into southern territory, as in the Beaufort district of South Carolina. The whole white popu-lation therein is six thousand, while the number of negroes exceeds thirty-two thousand. The panic which drove their masters in wild confusion from their homes, leaves them in undisputed possession of the soil. Shall they, armed by their masters, be placed in the field to fight against us, or shall their labor be continually employed in reproducing the means for supporting the armies of rebellion? The war into which this Government has been forced by

rebellious traitors is carried on for the purpose of repossesby the enemies of the Government, and to re-establish the authority and laws of the United States in the places where it is opposed or overthrown by armed insurrection and re-bellion. Its purpose is to recover and defend what is justly its own.

War, even between independent nations, is made to subdue the enemy, and all that belongs to that enemy, by occupying the hostile country, and exercising dominion over all the men and things within its territory. This being true in respect to independent nations at war with each other, it follows that rebels who are laboring by force of arms to overthrow a Government, justly bring upon themselves all the consequences of war, and provoke the destruction merited by the worst of orimes. That Govern-ment would be false to national trust, and would justly excite the ridicule of the civilized world, that would abstain from the use of any efficient means to preserve its own existence, or to overcome a rebellious and traitorous enemy, by sparing or protecting the property of those who are waging war against it.

The principal wealth and power of the rebel States is a peculiar species of property, consisting of the service or labor of African slaves, or the descendants of Africans. This property has been variously estimated at the value of from seven hundred million to one thousand million dollars. Why should this property be exempt from the hazards and consequences of a rebellious war?

It was the boast of the leader of the rebellion, while he yet had a seat in the Senate of the United States, that the southern States would be comparatively safe and free from the burdens of war, if it should be brought on by the con-templated rebellion, and that boast was accompanied by the savage threat that "northern towns and cities would become the victims of rapine and military spoil," and that "northern men should smell southern gunpowder and feel southern steel." No one doubts the disposition of the southern steel," No one doubts the disposition of the rebels to carry that threat into execution. The wealth of northern towns and cities, the produce of northern farms, northern workshops and manufactories would certainly be seized, destroyed, or appropriated as military spoil. No property in the North ward he ward from the hands of

property and possessions at stake, are the insurgent rebels to carry on warfare against the Government in peace and security to their own property?

Reason and justice and self-preservation forbid that such should be the policy of this Government, but demand, such as when the policy of this Soverinitein, but definite on the contrary, that, being forced by traitors and rebels to the extremity of war, all the rights and powers of war should be exercised to bring it to a speedy end. Those who war against the Government justly forfeit

all rights of property, privilege, or security, derived from the Constitution and laws, against which they are in armed rebellion; and as the labor and service of their slaves con-stitute the chiefproperty of the rebels, such property should share the common fate of war to which they have devoted the property of loyal citizens.

While it is plain that the slave property of the South is justly subjected to all the consequences of this rebellious war, and that the Government would be untrue to its trust in not employing all the rights and powers of war to bring it to a speedy close, the details of the plan for doing so, like all other military measures, must, in a great degree, be left to be determined by particular exigencies. The disposition of other property belonging to the rebels that becomes sub-ject to our arms is governed by the circumstances of the The Government has no power to hold slaves, none case. to restrain a slave of his liberty, or to exact his service. It has a right, however, to use the voluntary service of slaves liberated by war from their rebel masters, like any other property of the rebels, in whatever mode may be most effithe war, and the suppression of rebellion. It is clearly a right of the Government to arm slaves when it may become necessary as it is to take gunpowder from the enemy. Whether it is expedient to do so is purely a military ques-tion. The right is unquestionable by the laws of war. The expediency must be determined by circumstances, keeping in view the great object of overcoming the the rebels, re-establishing the laws, and restoring peace to the nation. It is vain and idle for the Government to carry on this

war, or hope to maintain its existence against rebellious force, without employing all the rights and powers of war. As has been said, the right to deprive the rebels of their property in slaves and slave labor is as clear and abso-lute as the right to take forage from the field, or cotton from the warehouse, or powder and arms from the magazine. To leave the enemy in the possession of such property as forage and cotton and military stores, and the means of conage and cotton and ministry stores, and the means of con-stantly reproducing them, would be madness. It is, there-fore, equal madness to leave them in peaceful and secure possession of slave property, more valuable and efficient to them for war than forage, cotton, and military stores. Such policy would be national suicide. What to do with that species of property is a question that time and circumstances will solve, and need not be apticipated further than to reawill solve, and need not be anticipated further than to repeat that they cannot be held by the Government as slaves. It would be useless to keep them as prisoners of war; and self-preservation, the highest duty of a Government, or of individuals, demands that they should be disposed of or employed in the most effective manner that will tend most speedily to suppress the insurrection and restore the authority of the Government. If it shall be found that the men who have been held by the rebels as slaves are capable of bearing arms and performing efficient military service, it is the right, and may become the duty, of this Government to arm and equip them, and employ their services against the rebels, under proper military regulations, discipline, and command.

But in whatever manner they may be used by the Gov-ernment, it is plain that, once liberated by the rebellions act of their masters, they should never again be restored to bondage. By the master's treason and rebellion he forfeits all right to the labor and service of his slave; and the slave of the rebellious master, by his service to the Government, becomes justly entitled to freedom and protection.

The disposition to be made of the slaves of rebels, after the close of the war, can be safely left to the wisdom and patriotism of Congress. The representatives of the people will unquestionably secure to the loyal slaveholders every right to which they are entitled under the Constitution of the country.

#### GENERAL BURNSIDE'S PROCLAMATION.

ROANOKE ISLAND, N. C., February 18, 1862. TO THE PEOPLE OF NORTH CAROLINA:

The mission of our joint expedition is not to invade any of your rights, but to assert the authority of the United States, and to close with you the desolating war brought upon your State by comparatively a few bad men in your midst.

The Government asks only that its authority may be rec-

to impress upon all officers the importance of preserving good order and discipline among their troops as the armies of the West advance into Tennessee and the southern States. \* \* \*

Soldiers I let no excess on your part tarnish the glory of our arms!

The order heretofore issued in this department, in regard to pillaging and maradding, the destruction of private prop-erty, and the stealing or concealment of slaves, must be strictly enforced. It does not belong to the military to decide upon the relation of master and slave. Such questions must be settled by the civil courts. No fogitive slave will therefore be admitted within our lines or camps, except

when specially ordered by the general commanding. \* Military stores and the public property of the enemy must be surrendered, and any attempt to conceal such property, by fraudulent transfer or otherwise, will be punished, but no prirate property will be touched urless by order of the general commanding. Wherever it becomes necessary to general commanding. Wherever it becomes becessary to obtain forced contributions for the supply and subsistence of our troops, such levies will be made as light as possible, and be so distributed as to produce no distress among the people. All property so taken must be receipted and fully accounted for, as heretofore directed.

These orders will be read at the head of every regiment, and all officers are commanded to strictly enforce them. By command of Major General Halleck : N. H. McLEAN, Adjutant General.

### GENERAL BUELL'S LETTER.

HEADQUARTERS DEPARTMENT OF THE OHIO, Nashville, March 6, 1862. DEAR SIR: I have had the honor to receive your commu-nication of the 1st instant on the subject of fugitive slaves in the camps of the army.

It has come to my knowledge that slaves sometimes make their way improperly into our lines, and in some instances they may be enticed there, but I think the number has been magnified by report. Several applications have been made to me by persons whose servants have been found in our camps, and in every instance that I know of the master has recovered his servant and taken him away.

I need hardly remind you that there will always be found some lawless and mischievous persons in every army; but I assure you that the mass of this army is law-abiding, and that it is neither its disposition nor its policy to violate law or the rights of individuals in any particular.

With great respect, your obedient servant, D. C. BUELL,

# Brig. Gen. Commanding Department.

Hon. J.R. UNDERWOOD Chairman Military Committee, Frankfort, Ky.

GENERAL HOOKER'S ORDER IN A FUGITIVE SLAVE CASE.

# HEADQUARTERS, HOOKER'S DIVISION, CAMP BAKER, LOWER POTOMAC, March 26, 1862.

TO BRIGADE AND REGIMENTAL COMMANDERS OF THIS DIVISION : Messrs. Nally, Gray, Dummington, Dent, Adams, Speake, Price, Posey, and Cobey, citizens of Maryland, have negroes supposed to be with some of the regiments of this division; the brigadier general commanding directs that they be permitted to visit all the camps of his command, in search of their property, and if found, that they be allowed to take possession of the same, without any interference whatever. Should any obstacle be thrown in their way by any officer or soldier in the division, they will be at once reported by the regimental commanders to these headquarters.

By command of Brigadier General Hooker: JOSEPH DICKINSON,

Assistant Adjutant General.

The following report was made in relation to said order by direction of General Sickles:

HEADQUARTERS SECOND REGIMENT, EXCELSION BRIGADE. CAMP HALL, March 27, 1862.

LIEUTENANT : In compliance with verbal directions from

ognized; and we repeat, in no manner or way does it desire to interfere with your laws, constitutionally established, your institutions of any kind whatever, your property of any sort, or your usages in any respect. L. M. GOLDSBOROUGH, Flag Officer, Commanding North Carolina Blockading Syuadron. A. E. BUENSIDE, Brigadier General, Numbering, perhaps, fifteen. They presented Commanding Department of North Carolina. GENERAL HALLECK'S PROCLAMATION. HEADQUARTERS DEPARTMENT OF MISSOUL, ST. LOUS, February 23, 1862. The major general commanding the department desires to impress upon all officers the importance of presented disres to impress upon all officers de importance of preserving of the slave-owners. At this time General Sickles opporof the slave-owners. At this time General Sickles opportunely arrived, and instructed me to order them outside the camp, which I did, amidst the loud cheers of our soldiers. It is proper to add, that before entering our lines, and within about seventy-five or one hundred yards of our camp, one of their number discharged two pistol shots at a negro, who was running past them, with an evident intention of taking his life. This justly enraged our men. All of which is respectfully submitted.

JOHN TOLEN Your obedient servant,

Major Commanding Second Regiment, E. B. To Lieutenant J. L. PALMER, Jr., A. D. C. and A. A. A. General.

#### GENERAL DOUBLEDAY'S ORDER.

HEADQUARTERS MILITARY DEFENSES NORTH OF THE POTOMAC, WASHINGTON, April 6, 1862.

SIR: I am directed by General Doubleday to say, in an-swer to your letter of the 2d instant, that all negroes coming into the lines of any of the camps or forts under his command, are to be treated as persons, and not as chattels. Under no circumstances has the commander of a fort or

camp the power of surrendering persons claimed as fugitive slaves, as it cannot be done without determining their character.

The additional article of war recently passed by Congress positively prohibits this.

The question has been asked, whether it would not be better to exclude negroes altogether from the lines. The general is of the opinion that they bring much valuable in-formation, which cannot be obtained from any other source. formation, which cannot be obtained from any other source. They are acquainted with all the roads, paths, fords, and other natural features of the country, and they make excel-lent guides. They also know and frequently have exposed the haunts of secession spies and traitors and the existence of rebel organizations. They will not, therefore, be excluded.

The general also directs me to say that civil process cannot be served directly in the camps or forts of his command, without full authority be obtained from the commanding officer for that purpose.

I am, very respectfully, your obedient servant E. P. HALSTED,

Assistant Adjutant General.

Lieut. Col. JOHN D. SHAUL, Commanding 76th Reg. N. Y. Vols.

PRESIDENT'S PROCLAMATION RESCINDING GENERAL HUNTER'S PROCLAMATION-MAY 19, 1862.

Whereas there appears in the public prints what purports to be a proclamation of Major General Hunter, in the words and figures following, to wit:

HEADQUARTERS DEPARTMENT OF THE SOUTH, HILTON HEAD, S. C., May 9, 1862.

[General Orders No. 11.] The three States of Georgia, Florida, and South Carolina, comprising the military department of the South, having comprising the mintary department of the south, having deliberately declared themselves no longer under the pro-tection of the United States of America, and having taken up arms against the said United States, it becomes a military necessify to declare them under martial law. This was accordingly done on the 25th day of April, 1862. Slavery and martial law in a free country are altogether incompatible; the persons in these three States—Georgia, Florida, and South Carolina—heretofore held as slaves, are therefore declared forever free. therefore declared forever free.

DAVID HUNTER, Major General Commanding.

Official:

ED. W. SMITH, Acting Assistant Adjutant General.

And whereas the same is producing some excitement and misunderstanding,

Therefore, I, ABRAHAM LINCOLN, President of the United States, proclaim and declare, that the Government of the United States had no knowledge, information, or belief, of an intention on the part of General Hunter to issue such a proclamation; nor has it yet any authentic information that the document is genuine. And further, that neither General Hunter, nor any other commander, or person, has been authorized by the Government of the United States to make proclamations declaring the slaves of any State free; and that the supposed proclamation, now in question, whether genuine or false, is altogether void, so far as respects such declaration.

I further make known that whether it be competent for me, as Commander-in-Chief of the Army and Navy, to declare the slaves of any State or States free, and whether, at any time, in any case, it shall have become a necessity indispensable to the maintenance of the Government, to exercise such supposed power, are questions which, under my responsibility, I reserve to myself, and which I cannot feel justified in leaving to the decision of commanders in the field. These are totally different questions from those of police regulations in armies and camps.

On the sixth day of March last, by a special message, I recommended to Congress the adoption of a joint resolution to be substantially as follows :

Resolved, That the United States ought to co-operate with any State which may adopt a gradual abolishment of slavery, giving to such State pecuniary aid, to be used by such State, in its discretion, to compensate for the inconveniences, public and private, produced by such change of system.

The resolution, in the language above quoted, was adopted by large majorities in both branches of Congress, and now stands an authentic, definite, and solemn proposal of the nation to the States and people most immediately interested in the subject matter. To the people of those States I now earnestly appeal-I do not argue-I beseech you to make the argument for yourselves—you cannot, if you would, be blind to the signs of the times—I beg of you a calm and enlarged consideration of them, ranging, if it may be, far above personal and partisan politics. This proposal makes common cause for a common object, casting no reproaches upon any. It acts not the Pharisee. The changes it contemplates would come gently as the dews of Heaven, not rending or wrecking anything. Will you not embrace it? So much good has not been done, by one effort, in all past time, as, in the Providence of God, it is now your high privilege to do. May the vast future not have to lament that you have neglected it.

In witness whereof, I have hereunto set my hand and caused the seal of the United States to be affixed.

Done at the City of Washington this nineteenth day of May, in the year of our Lord one thousand eight hundred and sixty-two, and of the Independence of the United States the eighty-sixth.

#### ABRAHAM LINCOLN.

By the President: WILLIAM H. SEWARD, Secretary of State. AN ORDER OF MAJOR GENERAL M'DOWELL.

HEADQUARTERS DEPARTMENT OF THE RAPPAHANNOCK. OPPOSITE FREDERICKSBURG, VA., May 26, 1862.

[Special Order No. 68.] Colonel Meredith, commanding the fifty-sixth Pennsylva-nia volunteers will furnish from his regiment a guard for the house and property of Mr. L. J. Huffman, who lives near Belle Plain. Colonel Meredith will see that no more own is token from M. Huffmen and these prove forming corn is taken from Mr. Huffman, and that no more fencing is disturbed. The guard will be so placed as to make this sure, even if it should be necessary to place a sentinel over every panel of fence.

By command of Major General McDowell SAMUEL BRECK, A. A. G.

Col. S. A. MEREDITH,

Commanding fifty-sixth Pennsylvania Volunteers. Sent by Mr. Huffman.

I certify that the above is a true copy, E. P. HALSTED, Capt., A. A. G.

AN ORDER OF GENERAL T. WILLIAMS.

HEADQUARTERS SECOND BRIGADE BATON ROUGE, June 5, 1862.

[General Orders No. 46.]

In consequence of the demoralizing and disorganizing tendencies to the troops of harboring runaway negroes, it is hereby ordered that the respective commanders of the camps and garrisons of the several regiments, second brigade, turn all such fugitives in their camps or garrisons out beyond the limits of their respective guards and sentinels. By order of Brigadier General T. Williams:

WICKHAM HOFFMAN, Assistant Adjutant General.

Colonel H. E. Paine, Fourth Wisconsin, declined to have his regiment employed "in violation of law for the purpose of returning fugitives to rebels," and reported that he could not obey the order. He was then placed under arrest, his command devolving on the next ranking officer.

Lieutenant Colonel D. R. Anthony, Seventh Kansas, issued this order:

HEADQUARTERS MITCHELL'S BRIGADE, ADVANCE COLUMN, FIRST BRIGADE, FIRST DIVISION, GENERAL ARMY OF THE MISSISSIPPI

CAMP ETHERIDGE, TENNESSEE, June 18, 1862. [General Orders No. 26.]

1. The impudence and impertinence of the open and armed rebels, traitors, secessionists, and southern-rights men of this section of the State of Tennessee, in arrogantly demanding the right to search our camp for fugitive slaves, has become a nuisance, and will no longer be tolerated. Officers will see that this class of men, who visit our camp for this number of the state of the

5 for this purpose, are excluded from our lines. 2. Should any such person be found within our lines, they will be arrested and sent to headquarters. 3. Any officer or soldier of this command who shall ar-

rest and deliver to his master a fugitive slave shall be summarily and severely punished, according to the laws relative

4. The strong Union sentiment in this section is most gratifying, and all officers and soldiers, in their intercourse with the loyal and those favorably disposed, are requested to act in their usual kind and courteous manner and protect them to the fullest extent.

By order of D. R. Antuony, Long Kansas volunteers, commanding: W. W. H. LAWRENCE, By order of D. R. Anthony, Lieutenant Colonel Seventh

Captain and Assistant Adjutant General.

And was put under arrest.

#### SECRETARY OF WAR TO GENERAL SAXTON.

WAR DEPARTMENT, WASHINGTON CITY, D. C., June 16, 1862.

SIR : You are hereby assigned to duty in the department of the South, to act under orders of the Secretary of War. You are directed to take possession of all the plantations heretofore occupied by the rebels, and take charge of the inhabitants remaining thereon within the department, or which the fortunes of war may hereafter bring into it, with authority to take such measures, make such rules and reg-ulations for the cultivation of the land, and for the protec-tion, employment, and government of the inhabitants as. circumstances may seem to require.

You are authorized to exercise all sanitary and police

powers that may be necessary for the health and security of the persons under your charge, and may imprison or exclude all disorderly, disobedient, or dangerous persons from the limits of your operations.

The major general commanding the department of the South will be instructed to give you all the military aid and protection necessary to enable you to carry out the views of the Government. You will have the power to act upon the decisions of courts-martial which are called for the trial of persons not in the military service to the same ex-tent that a commander of a department has over courts-martial called for the trial of soldiers in his department; and, so far as the persons above described are concerned, you will also have a general control over the action of the provost marshals.

It is expressly understood that, so far as the persons and purposes herein specified are concerned, your action will be independent of that of the other military authorities of the department, and in all other cases subordinate only to the major general commanding.

In cases of need or destitution of the inhabitants, you are In cases of need of destinution of the immanifulty you are directed to issue such portions of the army rations and such articles of clothing as may be suitable to the habits and wants of the persons supplied, which articles will be furnished by the quartermaster and commissary of the de-partment of the South upon requisitions approved by yourself. It is expected that by encouraging industry, skill in the cultivation of the necessaries of life, and general selfimprovement, you will, as far as possible, promote the real well being of all people under your supervision. Medical and ordnance supplies will be furnished by the proper officers, which you will distribute and use according

You will account regularly with the proper bureaus of this department and report frequently—once a week, at Yours, truly, EDWIN M. STANTON least.

Secretary of War.

#### GENERAL INSTRUCTIONS FROM THE PRESIDENT THROUGH THE WAR OFFICE.\*

WAR DEPARTMENT

WAR JEPARTMENT, WASHINGTON, July 22, 1862. First. Ordered that military commanders within the States of Virginia, North Carolina, Georgia, Florida, Ala-bama, Mississippi, Louisiana, Texas, and Arkansas, in an orderly manner seize and use and warmonet. orderly manner seize and use any property, real or personal which may be necessary or convenient for their several commands, for supplies, or for other military purposes; and that while property may be destroyed for proper military objects, none shall be destroyed in wantonness or malice. *Second.* That military and naval commanders shall em-ploy as laborers, within and from said States, so many per-sons of African descent as can be advantageously used for

military or naval purposes, giving them reasonable wages for their labor.

Third. That, as to both property, and persons of African descent, acccounts shall be kept sufficiently accurate and in detail to show quantities and amounts, and from whom both property and such persons shall have come, as a basis upon which compensation can be made in proper cases; and the several departments of this Government shall attend to and perform their appropriate parts towards the execution of these orders.

By order of the President:

EDWIN M. STANTON, Secretary of War.

#### GEN. M'CLELLAN'S PROMULGATION OF IT.

August 9, 1862-Major General McClellan, from his headquarters at Harrison's Landing, promulgated this order, with directions for its observance. We quote several pertinent paragraphs :

Inhabitants, especially women and children, remaining peaceably at their homes, must not be molested; and wherever commanding officers find families peculiarly ex-posed in their persons or property to marauding from this army, they will, as heretofore, so far as they can do with safety and without detriment to the service, post guards for their protection.

In protecting private property, no reference is intended to persons held to service or labor by reason of African de-

\* The issne of this Order is one of the reasons given by JEFFERSON DAVIS for his Order of August 1, 1862, directby oFFFERENCE DATE for ins of the of August 1, 1802, direct-ing that the commissioned officers of Pope's and Steinwehr's commands he not entitled, when captured, to be treated as soldiers, and entitled to the benefit of the cartel of exchange.

scent. Such persons will be regarded by this army, as they heretofore have been, as occupying simply a peculiar legal status under State laws, which condition the military authorities of the United States are not required to regard at all in districts where military operations are made neces-sary by the rehellions action of the State governments.

Persons subject to suspicion of hostile purposes, residing or being near our forces, will be, as heretofore, subject to arrest and detention, until the cause or necessity is re-moved. All such arrested parties will be sent, as usual, to the Provost Marshal General, with a statement of the facts in each case.

The general commanding takes this occasion to remind the officers and soldiers of this army that we are engaged in supporting the Constitution and the laws of the United States and suppressing rebellion against their authority; that we are not engaged in a war of rapine, revenge, or subjugation; that this is not a contest against populations, but against armed forces and political organizations; that it is a strugglo carried on with the United States, and should he conducted by us upon the highest principles known to Christian civilization.

Christian civilization. • Since this army commenced active operations, persons of African descent, including those held to service or labor under State laws, have always been received, protected, and employed as laborers at wages. Hereafter it shall be the duty of the Provost Marshal General to cause lists to be made of all persons of African descent employed in this army as laborers for military purposes—such lists being made sufficiently accurate and in detail to show from whom such persons so aphiect and so employed have always under-

Persons so subject and so employed have hiways under-stood that after being received into the military service of the United States, in any capacity, they could never be re-claimed by their former holders. Except upon such under-standing on their part the order of the President, as to this class of persons, would be inoperative. The general comclass of persons, would be inoperative. The general com-manding therefore feels authorized to declare to all such employees, that they will receive permanent military pro-tection against any compulsory return to a condition of servitude.

rvitude. By command of Major General McClellan : S. WILLIAMS, Assistant Adjutant General.

#### MAJOR GENERAL BUTLER AND BRIGADIER GENERAL PHELPS.

August 2, 1862-Major General Benjamin F. Butler, commanding Department of the Gulf, declined to approve of the conduct of Brigadier General J. W. Phelps, in organizing five companies of negroes, whom he proposed to arm and equip, upon the ground that the President alone had the authority to employ Africans in arms, and that he had not indicated this purpose. General Phelps resigned his commission in consequence.

#### REPLY OF THE WAR DEPARTMENT ON THE CASE.

WAR DEPARTMENT,

WASHINGTON CITY, July 3, 1862. GENERAL: I wrote you last under date of the 29th ultimo, and have now to say that your dispatch of the 18th ultimo, with the accompanying report of General Phelps, concerning certain fugitive negroes that have come to his pickets,

has been considered by the President. He is of opinion that under the law of Congress they can not be seut back to their master; that in common humanity they must not be permitted to suffer for want of food, shelter, or other necessaries of life; that to this end they should be provided for by the quartermaster's and commissary's departments, and that those who are capable of labor should

be set to work and paid reasonable wages. In directing this to be done, the President does not mean, at present, to settle any general rule in respect to slaves or slavery, but simply to provide for the particular case under the circumstances in which it is now presented.

I am, General, very respectfully, your obedient servant, EDWIN M. STANTON,

Secretary of War.

Major General B. F. BUTLER, Commanding, &c., New Orleans, Louisiana.

THE PRESIDENT'S ORDER TO GEN. SCHOFIELD. Ост. 1, 1863.

¥ Under your recent order, which I have approved, you will only arrest individuals, and suppress assemblies or newspapers.

when they may be working palpable injury to the military in your charge; and in no other case will you interfere with the expression of opinion in any form, or allow it to be inter-fered with violently by others. In this you have a discretion to exercise with great caution, calmness, and forbearance. \* \*

I think proper, however, to enjoin upon you the following: Allow no part of the military under your command to be engaged in either returning fugitive slaves, or in forcing or enticing slaves from their homes; and, so far as practicable, enforce the same forbearance upon the people.

Report to me your opinion upon the availability for good of the enrolled militia of the State. Allow no one to enlist colored troops, except upon orders from you, or from here through you.

Allow no one to assume the functions of confiscating property, under the law of Congress, or otherwise, except upon orders from here.

At elections see that those and only those, are allowed to vote, who are entitled to do so by the laws of Missouri, including as of those laws the restrictions laid by the Missouri Convention upon those who may have participated in the rebellion.

# GENERAL TUTTLE'S ORDERS AT NATCHEZ, MISS., MARCH 19, 1864. NATCHEZ, MISS., February 16, 1864.

[General Order No. 2.] \* \* From henceforward, all contraband negroes of Natchez are forbidden from the renting of houses, and living to themselves, but are required to secure legitimate syment with responsible respectable persons, or otherempl wise be sent to the contraband encomponent. Those hiring them are expected to aid in the enforcement of the foregoing regulations by the proper employment of them in their families or messes.

The congregation of so many negroes in one house, with their filthy and lazy habits, (as the undersigned has found to be the case,) if continued, will eventually prove fatal to the health of the city.

All owners or renters of houses are forbidden the renting An owners or renters of houses are formaden the renting or sub-renting of the same to contrahand negroes under a penalty for so doing. Otherwise it will be impossible for the undersigned to carry out a system of sanitary regula-tions essential to the well being of the city. By order of A. W. Kelley, Surgeon and Health Officer: T. A. RALSTON, A. A. Gen. Post.

NATCHEZ, (MISS.,) March 19, 1864. To preserve the general health of the troops stationed in the city of Natchez and of the inhabitants, and to guard against the origination here, and the introduction of pesti-lential diseases the ensuing summer and autumn, it im-peratively requires the prompt, vigorous, and steady enforce-ment of the againty used being home for a provided in ment of the sanitary regulations heretofore prescribed in this city. It is of the first and greatest importance and necessity

that all causes tending to the engendering and dissemination of pestilential diseases here, so soon as their existence is known, shall be at once abated or removed, so far as practicable. It is to be apprehended that serious danger to the health of this city will result from the congregation Within its limits of the large number of idle congregation within its limits of the large number of idle negroes which now throng the streets, lance, and alleys, and overcrowd every hovel. Lazy and profigate, unused to caring for themselves; thriftless for the present, and recklessly im-provident of the future, the most of them loaf idly about the streets and alleys, prowiling in secret places, and lounge lazily in crowded hovels, which soon become dens of noisome fith, the hot-heds fit to engender and rapidly dis-seminate the most loathsome and maliguant diseases. To Drevent these ewil effects it is hereby ordered that

hire any premises in the city for any purpose whatever, and no other person will be allowed to hire such premises for the purpose of evaluate this order, nor allowed to hire with the order presence of the pre will be removed to the contraband encampment.

The word contraband is hereby defined to mean all per-sons formerly slaves who are not now in the employ of their former owners.

Persons drawing rations from the United States Government are not supposed to need any hired servants. The number allowed to each family will be determined by the undersigned.

By order of A. W. Kelly, Surgeon and Health Officer. Approved:

J. M. TUTTLE, Brig. Gen. Com'g District.

HEADQUARTERS OF DEFENCES.

NEW ORLEANS, March 24, 1804. Citizens having colored people in their employ, who are superfluons or insubordinate, will be promptly relieved of them by recording the fact to Col. Hards Constant them by reporting the fact to Col. Hanks, Superintendent of Negro Labor.

# OFFICE PROVOST MARSHAL, PARISHES OF JEFFERSON AND ST. CHARLES, LA. CARROLLTON, March 28, 1864. The Provost Marshal of the Parish of Jefferson is

also charged with the execution of General Order No. 12, Headquarters of Defences of New Orleans, March 24, 1864, so far as it relates to its execution within this district. All persons within this district are requested to report at once to his office the names of their colored servants of either sex, in order that he may give them an employment certifi-cate, and all colored people of either sex who shall not on the first of April have such certificate in their possession will be considered unemployed, the males organized into squads and companies for labor on the parapet, and the females turned over to Col. Hanks.

By command of Brig. Gen. Roberts

R. SKINNER, First Lieut., 10th U. S. Infantry, A. D. C. and A. A. A. G. R. B BROWN, Colonel and Provost Marshal.

#### RESOLUTION OF INQUIRY.

## First Session, Thirty-Eighth Congress. IN HOUSE.

1864, May 16-Mr. GRINNELL offered this resolution which was adopted :

Resolved, That the committee on the conduct of the war be instructed to inquire as to the occasion of the military order of Brigadier General J. M. Tuttle for the government of the city of Natchez, Mississippi, which forbids any con-traband remaining in the city of Natchez who is not employed by some responsible white person; and also forbids any contraband from hiring any house in said city; whereby hundreds of children have been taken from the schools and many of the families of soldiers have been delivered to slavery.

PROPOSED CENSURE OF GEN. HALLECK'S ORDER.

Second Session, Thirty-Seventh Congress.

1861, Dec. 9-Mr. Lovejoy offered a resolution requiring the Secretary of War to revoke the first section of General Halleck's order, No. 3, Nov. 20, 1861.

A motion to table the resolution was lostyeas 63, nays 68, as follows:

within its limits of the large number of idle negroes which now throug the streets, lanes, and alleys, and overcrowd every hoviel. Lazy and profligate, unused to caring themselves; thrifless for the present, and recklessly im-provident of the future, the most of them loai idly about the streets and alleys, prowling in secret places, and loung lazily in crowded hovels, which soon become dens of noisome fith, the hot-beds fit to engender and rapidly but seminate the most loathsome and maliguant diseases. To prevent these evil effects, it is hereby ordered that after the first day of April, 1864, no contraband shall be allowed to remain in the city of Natchez, who is not em-ployed by some responsible while person in some legit mate business, and who does not reside at the domicil of his or her employer; and no contraband will be allowed to the streets. Aller, Ancona, Joseph Baily, Biddle, Jourde Bail, George H. Broune, William G. Brown, Burn-ham, Cobb, Coz, Cravens, Crisfeld, Crittenden, Dawes, De-ano, Dunlap, Dunn, English, Fouke, Criter, Harding, Mallovy, Maynard, Merzies, Nixon, Noble, Noell, Nor-tom, Odell, Olin, Pendleton, Richardson, Robinson, Sheffield, Vallandigham, Wadsworth, Webster, Chillow A. White, Wickiffe, Wood, Woodraff, Wright-B3. Nars-Messrs. Aldrich, Alley, Ashley, Babbitt, Goldsmith F. Bailey, Baker, Baxter, Bingham, Francis P. Blair, Sam-

uel S. Blair, Blake, Buffinton, Chamberlain, Clark, Colfax, Frederick A. Conkling, Roscoe Conkling, Conway, Davis, Diven, Duell, Edgertou, Edwards, Eliot, Fenton, Fessenden, Franchot, Frank, Gooch, Granger, Gurley, Hutchins, Julian, Kelley, Lansing, Loomis, Lovejoy, Moorhead, Anson P. Morrill, Justin S. Morrill, T. G. Pholps, Pomeroy, Porter, Potter, Alexander II. Rice, Riddle, Edward II. Rollins, Sargent Sadewick Shanks, Sherman Sloan, Snanlding, Sargent, Sedgwick, Shanks, Sherman, Sloan, Spaulding, Stevens, Train, Trimble, Trowbridge, Vandever, Van Horn, Van Wyck, Wall, Wallace, Charles W. Walton, E. P. Wal-ton, Washburne, Wheeler, Wilson, Worcester-68.

Mr. LOVEJOY then modified this resolution so as to "request" (instead of "requiring") the Secretary of War to revoke it,

When Mr. LANSING, of New York, offered the following as a substitute, which Mr. LOVEJOY accepted :

Whereas Major General Halleck, of the western depart-ment, has issued an order prohibiting negroes from coming ment, has issued an order prohibiting negroes from coming within the lines of our army, and excluding those already under the protection of our troops, and whereas a different policy and practice prevails in other departments, by the direct sanction of the Administration; and whereas said order is cruel and inhuman, and in the judgment of this House based upon no military necessity: Therefore, *Resolved*. That the President be respectfully requested to direct General Halleck to recall said order, or cause it to conform with the practice of the other departments of the

conform with the practice of the other departments of the army

Dec 11-The whole subject was then laid on the table-yeas 78, nays 64, as follows:

the table—yeas 78, nays 64, as follows: YEAS—Messrs. Allen. Ancona, Joseph Baily, Biddle, Francis P. Blair, Jacob B. Blair, George H. Erowne, William G. Brown, Burnham, Callert, Cook, Cooper, Cox, Cravens, Cris-feld, Crittenden, Delano, Delaplaine, Diven, Dunlap, Dunn, English, Fisher, Foule, Granger, Grider, Haight, Hanchett, Harding, Harrison, Holman, Horton, Johnson, William Kel-logg, Killinger, Law, Lazear, Leary, Lehman, McPherson, Mallory, Maynard, Menzies, Morris, Nixon, Noble, Noell, Norton, Odell, Olin, Pendleton, Perry, T. G. Phelps, Porter, Alexander H. Rice, Richardson, Riddle, Robinson, Shefield, Shellabarger, Shiel, Smith, John B. Steele, William G. Steele, Stratton, Benjamin F. Thomas, Trimble, Upton, Vallan-Gigham, Voorhees, Wickliffe, Woodruff, Wright—78. NAYS—Messrs. Alley, Arnold, Ashley, Babbitt, Goldsmith F. Bailey, Baker, Baxter, Bingham, Blake, Bufinton, Camp-bell, Chamberlain, Clark, Colfax, Frederick A. Conkling, Gooch, Goodwin, Hooper, Hutchins, Julian, Kelley, Fran-cis W. Kellogg, Lausing, Loomis, Lovejoy, Moorhead, Anson P. Morrill, Justin S. Morrill, Patton, Pike, Potter, Edward H. Rollins, Sedgwick, Shanka, Sherman, Sloae, Spaulding, Stevens, Train, Trowbridge, Van Horn, Van Wyck, Wall, Wallace, Charles W. Walton, E. P. Walton, Washburne, Wheeler, Albert S. White, Wilson, Worcester—64.

LEGISLATION PROHIBITING THE EXISTENCE OF SLAVERY IN TERRITORIES, ETC.

### Second Session, Thirty-Seventh Congress. IN HOUSE.

1862, May 12-Mr. LOVEJOY offered this bill, being a substitute for one previously offered by him:

To the end that freedom may be and remain forever the fundamental law of the land in all places whatsoever, so far as it lies within the powers or depends upon the action of the Government of the United States to make it so; Therefore,

Therefore, Be it enacted by the Senate and House of Representatives of the United States of America in Congress assembled, That slavery or involuntary servitude, in all cases whatsoever, (other than in the punishment of crime, whereof the party shall have been duly convicted,) shall henceforth cease, and be prohibited forever in all the Territories of the United States, now existing, or hereafter to be formed or acquired in any way. in any way.

Mr. Cox moved that it be tabled ; which was rejected-yeas 49, nays 81; and the bill was then passed-yeas 85, nays 50, as follows :

YEAS-Messrs. Aldrich, Alley, Arnold, Ashley, Babbitt, Baker, Baxter, Beaman, Bingham, Francis P. Blair, Samuel S. Blair, Blake, Buffinton, Campbell, Chamberlain, Clark,

Colfax, Frederick A. Conkling, Roscoe Conkling, Cutler, Davis, Dawes, Delano, Diven, Duell, Dunn, Edgerton, Edwards, Eliot, Ely, Fenton, Fessenden, Franchot, Frank, Gooch, Granger, Hale, Harrison, Hickman, Hooper, Horton, Hutchins, Julian, Kelley, William Kellogg, Lansing, Loomis, Lovejoy, McKnight, McPherson, Mitchell, Moor-head, Anson P. Morrill, Justin S. Morrill, Olin, Pike, Por-ter, Potter, Alexander H. Rice, John H. Rice, Riddle, Edward H. Rollins, Sargent, Sedgwick, Shanks, Shef-jeld, Shellaharger, Stevens, Stratton, Benjamin F. Thomas, Train, Trimble, Trowbridge, Van Horn, Verree, Wall, Wal-ce, Charles W. Walton, E. P. Walton, Washburne, Wheeler, Albert S. White, Wilson, Windom, Worcester - 85. NATS-Messrs. Allen, Ancona, Joseph Baily, Biddle, Jacob B. Blair, Gorge H. Broome, Wm. G. Brown, Calvett, Casey, Clements, Cobb, Cox, Cravens, Criglield, Crittenden, Dunkap, Kerigan, Knapp, Law, Lazcar, Leary, Lehman, Mallovy, Maynard, Mensies, Morris, Nodl, Odell, Perry, John S. Fhelps, Richardson, Robinson, Segar, Join B. Shele, Iwn. G. Steele, Francis Thomas, Vibbard, Voorheess, Wadsworth, Ward, Wester, Wickliffe, Woodrugf-50.

As originally reported the bill proposed that slavery should cease in all the Territories; the forts, arsenals, dock-yards, &c., of the United States; in all vessels on the high seas, and "in all places where the national Government is supreme, or has exclusive jurisdiction or power."

A motion to table it was rejected-yeas 50, nays 64.

May 9-Mr. Lovejoy offered a substitute, containing the other proposition and this addition:

SEC. 2. That any person now held or attempted to be held hereafter as a slave in any of the places above named is hereby declared to be free, and the right to freedom hereby declared may be asserted in any of the courts of the United States or of the several States, in behalf of the party, or his or her posterity, after any lapse of time.

A motion to table the bill was rejected-yeas 50, nays 65.

May 12-It was modified and passed.

#### IN SENATE.

June 9-The bill was reported amended by inserting this substitute :

That from and after the passage of this act there shall be neither slavery nor involuntary servitude in any of the Ter-ritories of the United States now existing, or which may at any time hereafter be formed or acquired by the United States, otherwise than in punishment of crimes whereof the party shall have been duly convicted.

And was passed-yeas 28, nays 10, as follows:

YEAS-Messrs. Anthony, Browning, Chandler, Clark, Col-lamer, Cowan, Dixon, Fessenden, Foot, Foster, Grimes, Hale, Harlan, Harris, Howard, Howe, King, Lane of Indiana, Pomeroy, Rice, Simmons, Sumner, Ten Eyck, Trumbull, Wade, Wilkinson, Wilmot, Wilson of Massachusetts-28. Nars-Messrs. Carlie, Davis, Kennedy, Latham, McDou-gall, Nesmith, Powell, Saulsbury, Stark, Wright-10.

June 17-The House concurred in the amendment of the Senate-yeas 72, nays 38.

#### IN THE TERRITORY OF MONTANA.

## Third Session, Thirty-Seventh Congress. IN HOUSE.

1863, February 12-Mr. AshLey, from the Committee on Territories, reported a bill to provide a temporary government for the Territory of Montana.

Mr. Cox moved to strike out the proviso to the sixth section, as follows:

Provided, That whereas slavery is prohibited in said Territory by act of Congress of June 19, 1862, nothing herein contained shall be construed to authorize or permit its existence therein.

Which was disagreed to-yeas 39, nays 96, as follows:

YEAS-Messrs. William Allen, Ancona, Baily, Biddle, Cal

pert, Cobb, Coz, Crisfield, Grider, Hall, Harding, Holman, Kerrigan, Knapp, Law, Mallow, Menzies, Marris, Noble, Smith, John B. Steele, Win. G. Steele, Stiles, Railandigham, Woorhees, Wadsworth, Webster, Chilton A. White, Wich-ing, Woodruff, Wright, Feaman-38.
 mars-Messrs. Aldrich, Alley, Arnold, Ashley, Babbitt, Bakte, William G. Brown, Bufinton, Chamberlain, Clark Blake, William Kellogg, Lansing, Lehman, Loomis Lovejoy, Low, McIndoe, McKean, McKinght, Maynaud Moorhead, Anson P. Morrill, Justin S. Morrill, Nixon, Noelf Other, Parander I, Haice, John H. Rice, Kiddle, Edwan, Kollows:
 Mr. SAULSBURY moved to postpone it indefi-nitely; which was disagreed to—yeas 8, nays 31, as follows:
 "Mass-Messrs. Anthony, Chandler, Clark, Collamer, Con-strikes, Franchot, Chark, Sheffield, Shellabarger, Sher-man, Stoan, Spaulding, Stevens, Benjamin F. Thomas, Fram-is Thomas, Train, Trimble, Trowbridge, Van Horn, Van Yalkenburgh, Van Wyck, Verree, Walker, Wallace, Walton
 "Data Mether and the Harlen, Harris, Howard, Howe, Johnson, Narse-Messrs, Anthony, Chandler, Clark, Collamer, Con-anes, Cowan, Dixon, Doolittle, Fessenden, Foot, Foster, Grimes, Hale, Harlan, Harris, Howard, Howe, Johnson, Narse-Messrs, Sherman, Sprague, Summer, Trumbull, Van Winkle, Wilson, Wison, Wildon-96

#### First Session, Thirty-Eighth Congress.

1864, March 17-The House considered the bill to enable the people of Colorado to form a Constitution and State Government, when

Mr. MALLORY moved to strike from the fourth section this clause :

First. That there shall be neither slavery nor involuntary servitude in the said State otherwise than in the punishment of crimes, whereof the party shall have been duly convicted.

Which was disagreed to-yeas 18, nays 87, as follows:

YEAS-Messers. James C. Allen, Ancona, Chanler, Cox, Daw-son, Denison, Eldridge, Hall, Harding, Knapp, Long, Mal-lory, William H. Miller, Morrison, John O'Neill, Rogers, Ross. Stiles-18.

Burg, Watada H. Batter, Borreson, Solar O'Iveat, Rogers, Ross, Stiles-18. Nars-Messrs. Allison, Ames, Anderson, Arnold, Ashley, Baily, Augustus C. Baldwin, Batver, Beaman, Francis P. Blair, jr., Jacob B. Blair, Blow, Bout-well, Boyd, Broomall, James S. Brown, Ambrose W. Clark, Cobb, Cole, Henry Winter Davis, Thomas T. Davis, Dawes, Donnelly, Driggs, Dumont, Eckley, Eliot, Farnsworth, Fen-ton, Frank, Garfield, Gooch, Griswold, Higby, Hotchkiss, John H. Hubbard, Hutchins, Jenckes, Julian, Kasson, Kel-ley, Francis W. Kellogg, Kernan, Longyear, Marvin, Mc-Allister, McBride, McClurg, McIndee, S.F. Miller, Moorhead, Daniel Morris, Amos Myers, Leonard Myers, Norton, Odel, Charles O'Neill, Orth, Perham, Pike, Price, Alexander H. Rice, John H. Rice, James S. Rollins, Scofield, Shannon, Smith, Smithers, Spalding, Starr, Stevens, Sweat, Thayer, Thomas, Tracy, Upson, Van Valkenburgh, Elihu B. Wash-burne, William B. Washburn, Webster, Whaley, Williams, Wilder, Wilson, Windom, Woodridge-S7.

# Proposed Amendment of the Constitution.

# IN SENATE.

1864, January 11-Mr. HENDERSON offered this joint resolution, which was referred to the Committee on the Judiciary, proposing amendments to the Constitution :

Be it resolved, dc., That the following articles be proposed as amendments to the Constitution of the United States, which, when adopted by the Legislatures of three-fourths of the several States, shall be valid, to all intents and pur-poses, as a part of the said Constitution, to wit: ARTICLE 1. Slavery or involuntary servitude, except as a punishment for crime, shall not exist in the United States. ARTICLE 2. The Congress, whenever a majority of the members elected to each house shall deem it necessary, may promose amendments to the Constitution, or, on the appli-

propose amendments to the Constitution, or, on the appli-cation of the Legislatures of a majority of the several States, shall call a convention for proposing amendments, which in either case shall be valid, to all intents and purposes, as part of the Constitution, when ratified by the Legislatures of two-thirds of the several States, or by conventions in twothirds thereof, as the one or the other mode of ratification may be proposed by Congress.

1864, Feb. 8-Mr. SUMNER introduced this

Pomeroy, Ramsey, Sherman, Sprague, Sumner, Trumbull, Van Winkle, Wilson-31.

Feb. 9-Mr. Powell introduced the following joint resolution, (embodying, as he stated, the view of Judge Nicholas, of Kentucky,) which was referred to the Committee on the Judiciary :

Resolved by the Senate and House of Representatives of the United States of America in Congress assembled, That the Constitution of the United States be amended as follows:

#### ARTICLE NO. -

SEC. 1. Congress shall at its first session after the adop-tion of this amendment, and from time to time, thereafter, apportion among the several States the electors of President and Vice President according to the following ratio of population in Federal numbers: One elector to each State having less than a million; two to each State having one but less than two million; four to each having three but less than three million; four to each having three but less then four million; four to each having three but less than four million; five to each having four but less than six million; six to each having six but less than eight million; and seven to each having eight million of population. Each State having but one elector shall be an electoral district, and each of the other States shall be divided by Congress into districts equal to the number of its electors, to be composed of coterminous territory, and as near as may be

the districts to have equality of population. SEC. 2. The voters of each district, qualified to vote for members of the most numerous branch of its Legislature, shall elect an elector.

The election for electors shall be held during the month of October next preceeding the commencement of any presidential term.

The several State Legislatures shall prescribe the time and manner for holding those elections and making returns thereof; also, for deciding them when contested, and making new elections therein; but Congress may discharge this duty, in whole or in part, when deemed necessary. SEC. 3. The electors shall convene in the Senate Chamber

at the seat of Government, at noon of the first Monday in February next preceding the commencement of the ensuing presidential term, and form an electoral college. Two thirds of all the electors elected shall be a quorum

the college.

The Chief Justice of the United States, or in his absence the President of the Senate, or in the absence of both, the Speaker of the House of Representatives shall be the presiding officer of the college.

The presiding officer shall cause all the electors elected, whether present or not, to be listed in the alphabetical order of their names, and in that order divide them into six classes of equal numbers, distributing by lot separately among the several classes such electors at the bottom of the list, if

Any, as are left out in the division. He shall by lot, under the supervision of one from each class, designate the several classes by numbers from one. to six.

When a quorum is present he shall announce that the college is formed, and note the time at which the enuncia-tion is made; but, when necessary, the enunciation shall be postponed until after the verification, by a majority of the electors present, of the returns and qualifications of members.

SEC. 4. After the college is formed the electors present of each class shall choose an elector from the class next succeeding it in number, except class six, which shall choose from class one.

In open session of the college the presiding officer, under the supervision and control of the six so chosen, or a ma-

jority of them, shall cause two of those six to be designated by lot.

From these two the college shall choose one, who shall be President for the next ensuing term of four years, and the other shall be the Vice President for that term.

The voting by class or college shall be viva roce in open session of the college.

In cases of tie, the casting vote shall be given by the presiding officer, who, if he be also an elector, shall not vote except in cases of tie.

The college may adopt rules for expediting a decision by the several classes, and to prevent more than two per-sons from receiving an equality of votes on the final vote of a class.

If there be a failure to choose one of the six from any class within the time prescribed by the college, the mem-bers of that class shall themselves make the choice.

There shall be no reconsideration of a vote given.

SEC. 5. If the college fail, except from exterior violence or intimidation, to make an election of President and Vice President within twenty-four hours from the time when the college was formed, it shall be dissolved, and the offices of its electors vacated.

Thereupon the presiding officer shall order a new election of electors on any day, not less than thirty from the date of his proclamation, and at least thirty before the next month of June, which election shall be held, and the electors chosen shall convene at the time and place designated by the proclamation, and proceed to the election of a President and Vice President as before directed, within twenty-four hours from the time of their formation into a college, and under like penalty for their failure. Should the failure to elect be caused by exterior violence

or intimidation, the functions of the college shall not cease, but it shall reconvene when and where a majority of its members shall by proclamation direct, and make or com-plete an election as before directed, within the time specified, under like penalty. SEC. 6. Should no election of President and Vice Presi-

dent be made by an electoral college before the 1st day of June next ensuing the commencement of a presidential term, the Senate of the United States shall convene in its Chamber at noon of the first Monday in July next thereafter, constitute all its elected members, whether present or not, into an electoral college, as though each Senator had been elected an elector, and proceed in all respects as before directed, within twenty-four hours, to choose a President and Vice President to fill the vacancy.

Should the Senate fail to elect, the discharge of the duties of President and Vice President for the residue of that term shall devolve upon such officers of the Government as Congress shall have theretofore directed.

SEC. 7. No office shall be incompatible with that of an elector except the office of Chief Justice of the United States.

SEC. S. An act or resolution passed by Congress, which shall be returned by the President with his objections, shall be valid without his signature, if repassed by each House of Congress by a vote equal to a majority of all the members elected therete.

SEC. 9. It shall not be deemed compatible with the duty of a President habitually to use the patronage of his office for the special advantage of any particular political party, or to suffer the patronage of any subordinate office so to be used.

Szc. 10. Should a vacancy occur in both the office of Pres-ident and in that of Vice President while there are two years remaining of the then presidential term, the Chief Jugtice of the United States, or in his absence the Secretary of State, shall convene the electoral college after thirty days' notice by proclamation, who shall fill the vacancies for the remainder of the term in all respects as if it were an original election.

SEC. 11. Every elector, before entering on the duties of his office, shall, by eath or affirmation, promise to support the Constitution of the United States, and declare that he has not, and will not, pledge his vote as an elector in favor of any person or toward aiding any political party.

#### AMENDMENT FOR THE EXTINCTION OF SLAVERY.

1864, February 10-Mr. TRUMBULL, from the Committee on the Judiciary reported a joint resolution :

Be it resolved, dc., That the following article be proposed to the legislatures of the several States as an amendment to the Constitution of the United States, which, when ratified by three fourths of said legislatures, shall be valid, to all intents and purposes, as a part of the said Constitution, namely:

#### ARTICLE XIII.

SECTION 1. Neither slavery nor involuntary servitude, except as a punishment for crime, whereof the party shall have been duly convicted, shall exist within the United States, or any place subject to their jurisdiction.

SEC. 2. Congress shall have power to enforce this article by appropriate legislation.

April 5-Mr. Davis moved to strike out the above and insert-

No negro, or person whose mother or grandmother is or was a negro, shall be a citizen of the United States, or be eligible to any civil or military office, or to any place of trust or profit under the United States.

Which was disagreed to-yeas 5, nays 32, as follows:

YEAS-Messrs. Buckalew, Davis, Powell, Riddle, Sauls-

Way-5, Nasales, Justice, Justice, Justice, Carls, Collamer, Con-NAvs-Messrs, Anthony, Chandler, Clark, Collamer, Con-ness, Dixon, Doolittle, Fessenden, Foot, Grimes, Hale, Har-ding, Harlan, Harris, Howard, Howe, Johnson, Lane of In-ding, Harlan, Harris, Howard, Howe, Johnson, Lane of Indiana, Lane of Kansas, Morgan, Morrill, Pomeroy, Ramsey, Sherman, Sprague, Sumner, Ten Eyck, Trumbull, Van Winkle, Wade, Willey, Wilson-32.

Mr. Powell moved to amend by adding to section 1, the words-

No slave shall be emancipated by this article, unless the owner thereof shall be first paid the value of the slave or slaves so emancipated.

Which was disagreed to--yeas 2, (Messrs. Davis and Powell,) nays 34:

Nars — Messrs. Anthony, Euckalev, Carlile, Chandler, Clark, Collamer, Conness, Dixon, Doolittle, Fessenden, Foot, Grinnes, Hale, Harding, Harlan, Harris, Howard, Howe, Johnson, Lane of Indiana, Laue of Kansas, Morgan, Mor-rill, Pomeroy, Ramsey, Sherman, Sprague, Sumner, Ten Eyck, Trumball, Van Winkle, Wade, Willey, Wilson-34.

April 6-Mr. Powell moved to add a new article to the Constitution, as follows:

#### ARTICLE XIV.

The President and Vice President shall hold their offices for the term of six years. A person who has filled the office of President shall not be re-eligible.

Which was disagreed to-yeas 12, nays 32 :

YEAS-Messrs. Brown, Davis, Foster, Grimes, Hendricks, Nesmith, Pomeroy, Powell, Riddle, Saulsbury, Wade, Wilkinson-12.

NATS-Messrs. Anthony, Chandler, Clark, Collamer, Con-ness, Cowan, Dixon, Doolittle, Fessenden, Foot, Hale, Hard-ing, Harlan, Harris, Henderson, Heward, Hoye, Johnson, Lane of Indiana, Lane of Kansas, McDougall, Morgan, Morrill, Ramsey, Sherman, Sprague, Sumner, Ten Eyck, Trum-bull, Van Winkle, Willey, Wilson-32.

Mr. Powell moved to add an article, as follows:

#### ARTICLE XV.

The principal officer in each of the Executive Depart-ments, and all persons connected with the diplomatic ser-vice, may be removed from office at the pleasure of the President. All other civil officers of the Executive Departments may be removed at any time by the President, or other appointing power, when their services are unnecessary, or for dishonesty, incapacity, inefficiency, misconduct, or neg-lect of duty; and when so removed the removal shall be reported to the Senate, together with the reasons therefor.

Which was disagreed to-yeas 6, nays 38, as follows:

YEAS-Messrs. Davis, Hendricks, Powell, Riddle, Saulsbury, Wade-6.

NATS-MESSIS. Anthony, Brown, Chandler, Clark, Colla-mer, Conness, Cowan, Dixon, Doolitile, Fessenden, Foot, Foster, Grimes, Hale, Harding, Harlan, Harris, Henderson, Howard, Howe, Johnson, Lane of Indiana, Lane of Kansas, McDougall, Morgan, Morrill, Nesmith, Pomeroy, Ramsey, Sherman, Sprague, Sumner, Ten Eyck, Trumbull, Van Winkle, Wilkinson, Willey, Wilson-38.

Mr. POWELL offered a new article, as follows : ARTICLE XVI.

Every law, or resolution having the force of law, shall relate to but one subject, and that shall be expressed in the title.

Which was disagreed to-yeas 6, nays 37, as follows:

YEAS-Messrs. Cowan, Davis, Hendricks, Powell, Riddle,

Saulsbury.-6. NATS-Messrs. Anthony, Brown, Chandler, Clark, Con-ness, Dixon, Doolittle, Fessenden, Foot, Foster, Grimes,

Hale, Harding, Harlan, Harris, Henderson, Howard, Howe, Johnson, Lane of Indiana, Lane of Kansas, McDougall, Morgan, Morrill, Nesmith, Pomeroy, Ramsey, Sherman, Sprague, Summer, Ten Eyck, Trumbull, Van Winkle, Wade, Wilkinson, Willey, Wilson-37.

April 8--Mr. SAULSBURY moved to strike out all after the word " Article xiii," and insert the following:

SECTION 1. All persons shall have the right peaceably to ssemble and worship God according to the dictates of their own conscience.

SEC. 2. The use of the public press shall not be obstructed; but criminal publications made in one State against the lawful institutions of another State shall not be allowed.

SEC. 3. The right of citizens to free and lawful speech in built assemblies shall not be denied. Access of citizens to the ballot-box shall not be obstructed either by civil or anilitary power. The military shall always he subordinate to the existing judicial authority over citizens. The privilege of the writ of *habeas corpus* shall never be suspended in the presence of the judicial authority. SEC. 4. The militia of a State or of the United States shall

not be employed to invade the lawful rights of the people of any of the several States; but the United States shall not be hereby deprived of the right and power to defend and protect its property and rights within the limits of any of the States.

SEC. 5. Persons held to service or labor for life, in any State under the laws thereof, may be taken into any Territory of the United States south of north latitude 36º 30' and the right to such service or labor shall not be impaired thereby, and the Territorial Legislature thereof shall have the exclusive right to make and shall make all needful rules and regulations for the protection of such right, and also for the protection of such persons: but Congress or any Territorial Legislature shall not have power to impair or abolish such right of service in the said Territory while in a territorial condition without the consent of all the States south of said latitude which maintain such service.

SEC. 6. Involuntary servitude, except for crime, shall not be permanently established within the district set apart for the seat of government of the United States: but the right of sojourn in such district with persons held to service or labor for life shall not be denied.

SEC. 7. When any Territory of the United States south of north latitude 36° 30' shall have a population equal to the ratio of representation for one member of Congress, and the ratio of representation for one memore of congress, and the people thereof shall have formed a constitution for a re-publican form of government, it shall be admitted as a State into the Union, on an equal footing with the other States; and the people may in such constitution either pro-hibit or sustain the right to involuntary labor or service, and alter or amend the constitution at their will.

SEC. S. The present right of representation in section two, article one, of this Constitution shall not be altered without the consent of all the States maintaining the right to involuntary service or labor south of latitude 36° 30', but nothing in this Constitution or its amendments shall be construed to deprive any State south of said latitude 36° 30' of the right of abolishing involuntary servitude at its will SEC. 9. The regulation and control of the right to labor

or service in any of the States south of latitude 36° 30' is hereby recognized to be exclusively the right of each State within its own limits; and this Constitution shall not be altered or amended to impair this right of each State without its consent: *Provided*, This article shall not be construed to absolve the United States from rendering assistance to suppress insurrections or domestic violence, when called upon by any State, as provided for in section four, article four, of this Constitution.

SEC. 10. No State shall pass any law in any way interfer-ing with or obstructing the recovery of fugitives from justice, or from labor or service, or any law of Congress made under article four, section two, of this Constitution; and all laws in violation of this section may, on complaint made by any person or State, be declared void by the Supreme Court of the United States.

SEC. 11. As a right of comity between the several States south of latitude 36° 30' the right of transit with persons held to involuntary labor or service from one State to another shall not be obstructed, but such persons shall not

be brought into the States north of said latitude. Szo. 12. The traffic in slaves with Africa is hereby forever

SEO. 12. The traffic in slaves with Africa is hereby forever prohibited on pain of death and the forfeiture of all the rights and property of persons engaged therein; and the descendants of Africans shall not be citizens. SEC. 13. Alleged fugitives from labor or service, on re-quest, shall have a trial by jury before being returned. SEC 14. All alleged fugitives charged with crime com-mitted in violation of the laws of a State shall have the right of trial by jury, and if such person claims to be a citizen of another State, shall have a right to appeal or of 17 17

writ of error to the Supreme Court of the United States.

SEC. 15. All acts of any inhabitant of the United States tending to incite persons held to service or labor to insur-rection or acts of domestic violence, or to abscond, are hereby prohibited and declared to be a penal offence, and all the courts of the United States shall be open to suppress and punish such offences at the suit of any citizen of the United States or the suit of any State.

Sec. 16. All conspiracies in any State to interfere with lawful rights in any other State or against the United States shall be suppressed; and no State or the people thereof shall withdraw from this Union without the consent of three fourths of all the States, expressed by an amendment proposed and ratified in the manner provided in article five of the Constitution.

SEC. 17. Whenever any State wherein involuntary servitude is recognized or allowed shall propose to abolish such servitude, and shall apply for pecuniary assistance therein, the Congress may in its discretion grant such relief, not exthe Congress may in its discretion grant such reliet, not ex-ceeding one hundred dollars, for each person liberated; but Congress shall not propose such abolishment or relief to any State. Congress may assist free persons of African descent to emigrate and civilize Africa. SEC. 18. Duties on imports may be imposed for revenue, but shall not be excessive or prohibitory in amount.

SEC. 19. When all of the several States shall have abol-shed slavery, then and thereafter slavery or involuntary servitude, except as a punishment for crime, shall never be established or tolerated in any of the States or Territories of the United States, and they shall be forever free. SEC. 20. The provisions of this article relating to invol-

untary labor or servitude shall not be altered without the consent of all the States maintaining such servitude.

Which was rejected, without a division.

The joint resolution proposing the amendment, as reported from the Committee on the Judiciary, was then passed—yeas 38, nays 6, as follows :

YEAS-Messrs. Anthony, Brown, Chandler, Clark, Col-lamer, Conness, Cowao, Dixon, Doolittle, Fessenden, Foot, Foster, Grimes, Hale, Harding, Harlan, Harris, Fessenden, Howard, Howe, Johnson, Lane of Indiana, Lane of Kansas, Howard, Howe, Sofitson, Lane of Findada, Lane of Kansas, Morgan, Morrill, Nesmith, Pooneroy, Ramsey, Sherman, Sprague, Sumner, Ten Eyck, Trumbull, Van Winkle, Wade, Wilkinson, Willey, Wilson–38. NATS-Messrs. Davis, Hendricks, McDougall, Powell, Riddle, Saulsbury–6.

#### VOTE IN THE HOUSE OF REPRESENTATINES.

May 31-This joint resolution coming up, Mr. HOLMAN objecting to its second reading, the vote on rejecting the bill was yeas 55, nays 76, as follows :

YEAS—MESSTS. James C. Allen, Ancona, Bliss, Brooks, James S. Brown, Chanler, Coffroth, Cox, Cravens, Dawson, Denison, Eden, Edgerton, Eldridge, Finck, Ganson, Grider, Hall, Harrington, Herrick, Holman, Philip Johnson, Wil-liam Johnson, Kalbfleisch, Kernan, King, Knapp, Law, Long, Mallory, Marcy, McAllister, McDowell, James R. Morris, Morison, Noble, Odell, Pendleton, Frugn, Radford, William H. Randall, Ross, John B. Steele, Wil-liam G. Steele, Stiles, Sweat, Voorhees, Wallsworth, Ward, Wheeler, Chilton A. White, Joseph W. White, Wingled, Fer-nando Wood—55.

NAYS-Messrs. Alley, Allison, Ames, Anderson, John D. Baldwin, Baxter, Beaman, Blaine, Jacob B, Blair, Boyd, Broomall, Ambrose W. Clark, Cobb, Cole, Creswell, Henry Winter Bavis, Thomas T. Davis, Daves, Donnelly, Driggs, Eckley, Eliot, Fenton, Frank, Garfield, Gooch, Grinnell, Eckley, Eliot, Fenton, Frank, Garfield, Gooch, Grinnell, Griswold, Hale, Higby, Hooper, Hotchkiss, Asahel W. Hubbard, John H. Hubbard, Hulburd, Ingersoll, Jenckes, Kelley, Francis W. Kellogg, Orlando Kellogg, Littlejohn, Loan, Longyear, Marvin, McClurg, Morrill, Daniel Morris, Amos Myers, Leonard Myers, Charles O'Neill, Orth, Pat-terson, Perham, Pike, Pomeroy, Price, Alexander H. Rice, John H. Rice, Edward H. Rollins, Schenck, Scofield, Shan-non, Sloan, Spalding, Stevens, Thomas, Tracy, Ugson, Whilams, Wilder, Wilson, Windom-76.

June 14-Mr. WHEELER offered an amendment, to add this proviso:

That this article shall not apply to the States of Kentucky, Missouri, Delaware, and Maryland until after the expirations of ten years from the time the same shall be ratified.

June 15-The amendment offered by Mr. WHEELER was disagreed to. Also the amendamendments to the Constitution be submitted to conventions instead of the Legislatures of the States, so that the ratification, if at all, shall be by conventions of three fourths of the States.

The joint resolution of the Senate was then rejected-yeas 95, nays 66, (two-thirds being necessary,) as follows:

Becessairy, J as 1010 ws: YEAS-Messrs. Alley, Allison, Ames, Anderson, Arnold, Baily, J. D. Baldwin, Baxter, Beaman, Blaine, J. B. Blair, Blow, Boutwell, Boyd, Brandegee, Broomall, Ambrose W. Clark, Freeman Clatke, Cobb, Cole, Creswell, Dawes, Dem-ing, Dixon, Donnelly, Driggs, Eckley, Eliot, Farnsworth, Fenton, Frank, Garfield, Gooch, Griswold, Hale, Higby, Hooper, Hotchkiss, Asahel W. Hubbard, John H. Hubbard, Hulburd, Iogersoil, Jenckes, Julian, Kasson, Kelley, Fran-cis W. Kellogs, Orlando Kellogs, Littlejohn, Loan, Long-year, Marvin, McClurg, McIndoe, Samuel F. Miller, Moor-head, Morrill, Daniel Morris, Amos Myers, Leonard Myers, Norton, Oddl, Charles O'Niell, Orth, Patterson, Perham, Pike, Price, Alexander H. Rice, John H. Rice, Edward H. Rellins, Schenck, Scofield, Shannon, Sloan, Smith, Smithers,

 Mike, Price, Alexander II. Rice, John H. Rice, Edward H.
 Reic, Price, Alexander II. Rice, John H. Rice, Edward H.
 Rellins, Schenck, Schfeld, Shannon, Slean, Smith, Smithers,
 Spalding, Starr, Stevens, Thayer, Thomas, Tracy, Upson,
 Van Valkenburgh, Ellihu B. Washburne, Wm. B. Washburn, Webster.
 Wheley, Whileser, Williams, Wilder, Wilson, Windom, Woodbridge-95.
 NArs-Messrs. James C. Allen, Wm. J. Allen, Ancoma,
 Ashley, Augustus C. Baldwin, Eliss. Brooks, James S. Broun,
 Chanter, Caffroth, Cox, Cravens, Dawson, Denison, Eden,
 Edgetcan, Eldridge, English, Finek, Ganson, Grider, Harding, Harrington, Herrick, Holman, Hatchins, Philip Johnson, William Johnson, Kalbifeisch, Kernan, King, Law,
 Lasear, Le Blond, Long, Moltory, Marcy, McAllister, Mo Dowell, McKinney, Wm. H. Miller, James R. Morris, Morrison, Noble, J. O'Neill, Pendleton, Perry, Pruyn, Raulford, S.
 J. Rundall, Robinson, Rogers, James S. Kollins, Ross, Scott,
 Joinn B, Steele, Win. G. Siele, Siles, Strouse, Stuart, Sweat,
 Wadsworth, Ward, Chilton A. White, Joseph W. While, Fernando Iyood-C6. nando Wood—66.

Mando Wood-to. Nor Vortuso-Messrs. William G. Brown, Clay, Henry Winter Davis, Thomas T. Davis, Dumont, Grinnell, Hall, Benjamin G. Harris, Charles M. Harris, Knapp, McBride, Middleton, Nclson, Pomeroy, Wm. H. Randall, Stebbins, Voorhees, Winfield, Benjamin Wood, Feaman-20.

Same day, Mr. Ashley entered a motion to reconsider the above vote.

#### OTHER PROPOSITIONS.

#### IN HOUSE.

 1864, February 15-Mr. ARNOLD offered this resolution

Resolved, That the Constitution should be so amended as to abolish slavery in the United States wherever it now exists, and to prohibit its existence in every part thereof forever.

The House refused to table it—yeas 58, nays 79; and passed it-yeas 78, nays 62, as follows:

YEAS-Messrs. Allison, Anderson, Arnold, Ashley, Baily, John D. Baldwin, Baxter, Beaman, Jacob B. Blair, Blow, Boutwell, Boyd, Brandegee, Broomall, Cobb, Cole, Creswell, Henry Winter Davis, Thomas T. Davis, Dawes, Dixon, Don-nelly, Driggs, Dumont, Eckley, Eliot, Farnsworth, Frank, Garfield, Goech, Grinnell, Higby, Hooper, Asahel W. Hub-hard, John H. Hubbard, Hulburd, Jenckes, Julian, Kelley, Francis W. Kellogg, Loan, Longyear, Marvin, McClurg, Samuel F. Miller, Moorhead, Morrill, Daniel Morris, Amos Nurse, Locanard Murge, Nearton, Charles O'Neill, Orit, Bot

Branuels W. Reinogg, Loan, Löngyear, Marvin, McChirg, Samuel F. Miller, Moorhead, Morrill, Daniel Morris, Amos Myers, Leonard Myers, Norton, Charles O'Neill, Orth, Pat-terson, Perhan, Pike, Pomeroy, Price, Alexander II. Rice, John H. Rice, Edward H. Rollins, Schenck, Scofield, Shan-on, Smithers, Spalding, Stevens, Thayer, Thomas, Upson, Van Valkenburgh, Ellihn B. Washburne, William B. Wash-burn, Whaley, Williams, Wilder, Wilson, Windom-78.
 NAYS-Messrs, Janes C. Allen, William J. Allen, Ancona, Augustus C. Baldwin, Eliss, Brooks, James S. Brown, Clay, Coffroth, Cox, Cravens, Dawson, Denison, Eden, Edgerton, Eldridge, Fücke, Ganson, Grider, Hall, Harding, Harring-ton, Benjamin G. Harris, Herrick, Holman, William Johnson, Netlands Kellogg, Kernan, King, Knapp, Law, Lazer, Long, Mollory, McAluster, McBride, McDowell, McKinney, Wm. H. Miller, James R. Morris, Morrison, Neison, Noble, Odell, John O'Neill, Pendleton, Perry, Samuel J. Randall, William H. Randall, Robinson, James S. Rollins, Ross, Soxt, Join B. Steele, Siles, Strouse, Stuart, Sweat, Chilton A. White, Joseph W. White, Winfield, Fernando Wood-62.
 February 15-Mr, WINDOM offered this reso-

February 15-Mr. WINDOM offered this reso-

#### ment of Mr. PENDLETON, that the proposed | lution, which was referred to the Committee on the Judiciary:

Resolved by the Senate and House of Representatives, &c., That (two thirds of both Houses concurring) the following article be proposed to the Legislatures of the several States as an amendment to the Constitution of the United States, which, when ratified by three fourths of said Legislatures, shall be valid, to all intents and purposes, as part of said Constitution, namely:

#### ARTICLE 13.

SEC. 1. Neither slavery nor involuntary servitude, except as a punishment for crime whereof the party shall have been duly convicted, shall exist within the United States or any place subject to their jurisdiction. SEC. 2. Congress shall have power to enforce this article

by appropriate legislation.

1864, March 28-Mr. STEVENS offered the following joint resolution :

Resolved by the Senate and House of Representatives of the United States of America in Congress assembled, That the following articles be proposed to the several States as amendments to the Constitution of the United States:

ARTICLE I. Slavery and involution servitude, except for the punishment of crimes whereof the party shall have been duly convicted, is forever prohibited in the United States and all its Territories.

ART. 2. So much of article four, section two, as refers to the delivery up of persons held to service or labor escaping into another State is annulled.

On a motion to reject the resolution, the yeas were 45, nays 75, as follows :

yeas were 45, nays 75, as follows: YEAS-Messrs. James C. Allen, William J. Allen, Ancona, Augustus C. Baldwin, Eliss, Brooks, James S. Brown, Chan-ler, Cluy, Cravens, Dawson, Denison, Eldridge, English, Finele, Ganson, Grider, Hall, Harding, Harrington, Benjamin G. Harris, Herrick, Holman, Philip Johnson, Knupp, Law, Mallory, Marcy, James R. Morris, Morrison, Noble, John O'Neill, Pendlelon, Perry, Fruyn, Samuel J. Randall, Rogers, James S. Rollins, Ross, Scott, John B. Skeele, William G. Steele, Sweat, Winfield, Framan-45. NATS-Messrs. Alley, Allison, Ames, Anderson, Ashley, Boily, John D. Baldwin, Baxter, Beaman, Blaine, Francis P. Blair, ir., Blow, Boutwell, Boyd, Brandegee, Broomall, Am-brose W. Clark, Cobb, Cole, Henry Winter Davis, Thomas T. Pavis, Deming, Dison, Drizgs, Dumont, Eckley, Eliot, Frank, Garfield, Grinnell, Hale, Higby, Hooper, Hotchkiss, Asshel W. Hubbard, John H. Hubbard, Hulburd, Jenckes, Sulien, Kasson, Kelley, Francis W., Kellögg, Longvear, Me-

Assnel W. Hubbard, John H. Hubbard, Huburd, Jenckes, Julian, Kasson, Kelley, Francis W. Kellogg, Longyear, Mc-Bride, McClurg, Sanuel F. Miller, Moorhead, Morrill, Leonard Myers, Norton, Charles O'Neill, Orth, Perham, Pike, Pomeroy, Price, Alexander H. Rice, John H. Rice, Edward H. Rollins, Schenck, Scofield, Shannon, Smithers, Spalding, Stevens, Thomas, Tracy, Upson, Van Valkenburgh, Ellihu B. Washburne, William B. Washburn, Webster, Whador, Wilcon William Whaley, Wilson, Windom-75.

It was then laid over.

On motion of Mr. STEVENS the second article was stricken out.

#### IN SENATE.

1864, Feb. 8-Mr. ANTHONY offered a joint resolution to repeal the joint resolution to amend the Constitution of the United States, approved March 2, 1861, which was as follows:

Resolved, dc., That the following article be proposed to the Legislatures of the several States as an amendment to the Constitution of the United States, which, when ratified by three fourths of said Legislatures, shall be valid to all intents and purposes as part of the said Constitution, near other said Legislatures, shall be valid to all namely

ART, XIII. No amendment shall be made to the Consti-tution which will authorize or give to Congress the power to abolish or interfere within any State with the domestic institutions thereof, including that of persons held to labor or service by the laws of said State.

May 11-On motion of Mr. TRUMBULL, the Committee on the Judiciary were discharged from its further consideration.

#### OTHER PROPOSED AMENDMENTS.

A convention held in Alleghany City, Penn., January 27, 1864, adopted the following resolutions:

Resolved, First, That we deem it a matter of paramount

ν.

1

importance to the life and prosperity and permanency of our nation that the Constitution he so amended as fully to express the Christian national character. *Resolved*, Second, That we are encouraged by the success attending the labors of the friends of this movement to the success attending the labors of the friends of the success of the fit will

persevere in the hope that, with the blessing of God, it will

speedily result in the consumation of its great object. *Resolved*, Third, That the late proclamation of his Excellency the President of the United States, recommending cellency the President of the United States, recommending the observance of days of national fasting, humiliation, and prayer, as suggested by the Senate, for the purpose of confessing our national sins, which have provoked the divine displeasure, and of imploring forgiveness through Jesus Christ; and also days of national thanksgiving for the purpose of making grateful acknowledgments of God's mercies; we have pleasing evidence that God is graciously inclining the hearts of those who are in authority over us to recognize His hand in national affairs, and to cherish a sense of our dependence as a nation upon Him.

Which was subsequently presented to the President by a committee of the convention, with an address. The President made this reply:

GENTLEMEN: The general aspect of your movement I cordially approve. In regard to particulars I must ask time to deliberate, as the work of amending the Constitu-tion should not be done hastily. I will carefully examine your paper in order more fully to comprehend its contents than is possible from merely hearing it read, and will take such action upon it as my responsibility to our Maker and our country demands.

Numerously signed petitions were presented to Congress during the late session.

1864, April 11-Mr. CRAVENS offered these resolutions, which went over, under the rule :

Resolved, That in the present condition of the country, when the passions of the people are inflamed and their prejudices are excited, it is nowise and dangerous to attempt to alter or amend the Constitution of the United States; that ample power is contained within its limits as it now exists to protect and defend the national life, and the exor-cise of power not warranted by its provisions would be to enter the field of revolution, and dangerous to the liberties of the people, trading to the establishment of military despotism and the final overthrow of free government in Amer-ica.

Resolved, That any attempt by Congress to reduce States to the condition of Territories is as odious and as revolutionary in its character and tendency as secession itself. Resolved, Thut it is the duty of the Government to listen

to and consider any proposition for reconciliation that may be offered by the insurgents which does not involve the question of separation.

Resolved, That the thanks of the nation are due, and are hereby tondered, to the officers, soldiers, and seamen who have so gallantly horne our flag in this hour of peril to our country.

March 24-Mr. BLAINE offered this resolution, which was adopted:

Resolved, That the Judiciary Committee be directed to inquire into the expediency of proposing an amendment to the Constitution of the United States, by striking out the fifth clause of section nine, article one, which forbids the levying of a tax on articles exported from any State.

#### IN SENATE.

March 3-Mr. DAVIS presented, for printing, an amendment he proposed to offer to the amendment of the Constitution reported by the Judiciary Committee:

First. That no negro or person whose mother or grand-mother is or was a negro shall be a citizen of the United States, or he eligible to any civil or military office, or to any place of trust or profit under the United States. Second. That the States of Maine and Massachusetts shall

form and constitute ene State of the United States, to be called East New England, and the States of New Hampshire, Rhode Island, Connecticut, and Vermont shall form and constitute one State of the United States, to be called West New England.

#### Resolutions on Slavery. First Session Thirty-Eighth Congress.

1864, Jan. 18-Mr. HARDING offered this resolution :

Resolved, That the maintenance inviolate of the rights

of the States, and especially the right of each State to or-der and control its own domestic institutions according to its own judgment exclusively, is essential to that balance of nower on which the perfection and endurance of our political fabric depend.

Mr. STEVENS moved to lav it on the table: which was lost-yeas 73, nays 75, as follows:

YEAS-Messrs. Alley, Allison, Ames, Ashley, John D. Baldwin, Baxter, Beaman, Jacob B. Blair, Boutwell, Boyd, Brandegee, Broomall, Ambrose W. Clark, Freeman Clarke, Cole, Creswell, Thomas T. Davis, Deming, Dixon, Donnelly,

Brandegee, Broomall, Ambrose W. Clark, Freeman Clarke, Cole, Creswell, Thomas T. Davis, Deming, Dixon, Donnelly, Driggs, Eckley, Eliot, Farnsworth, Frank, Garfield, Gooch, Grinnell, Higby, Hooper, Hotchkiss, Asahel W. Hubbard, Hubburd, Jenckes, Julian, Kelley, Francis W. Kellogg, Orlando Kellogg, Loan, Longyear, Marvin, McBride, McClurg, McIndoe, Samnel F. Miller, Moorhead, Morrill, Daniel Vlorris, Leonard Myers, Charles O'Neill, Patterson, Perham, Pike, Pomeroy, Price, John H. Rice, Edward H. Rollins, Schenck, Smithers, Spalding, Stevens, Thayer, Thomas, Upson, Ellihu B. Washburne, William B. Washburn, Whaley, Williams & Wilder, Wilson, Windom, Woodbridge–73. Nars-Meesrs. Junes C. Allen, William J. Allen, Ancona, Augustus C. Baldwin, Francis P. Blair, ir., Bliss, Brooks, James S. Brown, W. G. Brown, Chanler, Clay, Coffroh, Coz, Cravens, Dawson, Denison, Eden, Edgerton, Eldridge, English, Finck, Ganson, Grider, Grisvold, Hale, Halt, Harding, Beingmin G. Harris, Merrick, Holman, Hüchrins, William Johnson, Kernan, King, Knapp, Lazeer, Le Blond, Long, Marcy McAllister, McDowl, McKinney, Middleton, Wm. H. Miller, James S. Rollins, Ross, Scofield, John G. Scott. Smith, John B. Steele, Stiles, Stroues, Stuart, Sweeat, Tracy, Voorhees, Wadsworth, Websler, Wheler, Chilton A. While, Joseph W. White, Winfield, Fernando Wood, Yeaman–75. Yeaman-75.

It was then referred to the Committee on the Rebellious States—yeas 83, nays 68, as follows :

YEAS—Wessrs. Alley, Allison, Ames, Arnold, Ashley, John D. Baldwin, Baxter, Beaman, Blaine, Jacob B. Blair, Boutwell, Boyd, Broomall, Ambrose W. Clark, Freeman Clarke, Clay, Cole, Creswell, Thomas T. Davis, Dawes, Deming, Dixon, Donnelly, Driggs, Eckley, Eliot, Farnsworth, Fenton, Frank, Garfield, Gooch, Grinnell, Higby, Hotchkiss, John H. Hubbard, Hulburd, Jenckes, Julian, Kasson, Kelley, Orlando Kellogg, Loan, Longyear, Lovejoy, Marvin, McBride, McClurg, McIndos, Samuel F. Miller, Moorhead, Morrill, Daniel Morris, Amos Myers, Leonard Myers, Charles O'Neill, Orth. Patterson, Perham. Pike, Pomerov, Price, Alexander Damei Morris, Amos Myers, Leonard Myers, Charles U'Neill, Orth, Patterson, Perham, Pike, Pomeroy, Price, Alexander II. Rice, John II. Rice, Edward H. Rollins, Schenck, Sco-field, Shannon, Smithers, Spalding, Stevens, Thayer, Thomas, Upson, Van Valkenburgh, Ellihu B. Washburne, William B. Washburn, Webster, Whaley, Williams, Wilder, Wilson, Windom, Woodbridge—S2.

Devel, university, Harrison, Kullar, Williams, Wilder, Willson, Wandom, Woodbridge—S3. NAYS—Messrs. James C. Allen, Williams, Wilder, Wilson, Mars-Messrs. James C. Baldwin, Francis P. Blair, jr., Brooks, James S. Brown, John W. Chenler, Cox, Cravens, Dawson, Denison, Eden, Edgerton, Eldridge, English, Finck, Ganson, Grider, Grisvold, Hale, Hall, Harding, Harrington, Benjanin G. Harris, Herrick, Holman, Hutchins, Wiltiam Johnson, Kernan, King, Knapp, Le Blond, Long, Marey, McAllister, McDoueld, McKinnei, Middleion, William H. Miller, James R. Morris, Morrison, Nelson, Noble, John O'Neill, Pendleton, Radford, Sanuel J. Randall, Robinson, James S. Rollins, Ross, Scott, Smith, John B. Steele, Stiles, Strouse, Stuart, Sweat, Tracy, Foorhees, Wadsworth, Wheeler, Chilton A. White, Joseph W. White, Winfield, Fernando Wood, Feaman–68.

1864, February 29-Mr. MORRISON offered the following resolution ; which was laid over under the rule :

Resolved, That slavery legally exists in some of the States of the Union hy virtue of the Constitution and laws of such States, and that neither the Government of the United States nor the people, as such are responsible therefor, nor have they any legal duty to perform in relation thereto ex-cept such as is enjoined by section two, article four, of the Federal Constitution, in these words: "No person held to service or labor in one State, under the laws thereof, escap-ion in the section of the section two sections are the section of the service of the section o ing into another, shall in consequence of any law or regu-lation therein, be discharged from such service or labor, but shall be delivered up on claim of the party to whom such service or labor may be due."

### Bureau of Freedmen's Affairs. IN HOUSE.

1864, March 1-The House passed a bill to

establish a Bureau of Freedmen's Affairs-yeas 69, nays 67, as follows :

YEAS—Messrs. Alley, Allison, Ames, Anderson, Arnold, John D. Baldwin, Baxter, Beaman, Blow, Boutwell, Boyd, Brandegee, Ambrose W. Clark, Cobb, Cole, Creswell, Dawes, Dixon, Donnelly, Driggs, Dumont, Eckley, Eliot, Farns-worth, Fenton, Frank, Garfield, Grinnell, Higby, Hooper, Hotchkiss, Asahel W. Hubbard, John II. Hubbard, Jonckes, Julian, Kasson, Kelley, Francis W. Kellogg, Orlando Kel-logg, Longyear, Lovejoy, Marvin, McClurg, McIndoe, Sam-uel F, Miller, Moorhead, Morrill, Daniel Morris, Amos Myers, Voorloa, Churles, O'Neill, Partham Pike, Pomerow, Fried.

Indgy, Milgy, and Gy, Markin, Janiel Morris, Amos Myers, Norton, Charles O'Neill, Perham, Pike, Pomeroy, Price, Alexander II. Rice, John H. Rice, Schenck, Shannon, Sloan, Smithers, Stevens, Thayer, Van Valkenburgh, William B. Washburn, Wilder, Wilson, Windom, Woodbridge–69. NATS-Messrs. Ancona, Baily, Augustus C. Baldwin, Fran-cis P. Blair, jr., Jacob B. Blair, Brocks, James S. Brown, William G. Brown, Chanler, Cluy, Coffroht, Goz, Dawson, Derö-ison, Eden, Eldridge, Finck, Ganson, Grider, Griswold, Hale, Mall, Harding, Harrington, Harris, Herrick, Holman, Hutch-ins, Kalbfeisch, Kernan, King, Knapp, Law, Long, Mallory, McAllister, McBride, McDowell, McKinney, Middleton, Wil-tiam H. Miller, Morrison, Nelson, Noble, Fendleton, Radford, Samuel J. Randall, Ropers, Ross, Scott, Stebbins, John E. Steele, William G. Steele, Stiles, Strouse, Staur, Sweat, Thomas, Tracy, Foorhees, Wadsworth, Webster, Whaley, Chilton A. White, Joseph W. White, Williams, Winfield–67. [The bill created in the War Department a

[The bill created in the War Department a Bureau of Freedmen's Affairs, with a Commissioner, (at a compensation of \$4,000 per annum,) to whom shall be referred the adjustment and determination of all questions concerning persons of African descent, and per- sons who are or shall become free by virtue of any proclamation, law, or military order issued during the present rebellion, or by virtue of any State act of Emancipation, or who shall be otherwise entitled to their freedom. The Commissioner is authorized to make all needful rules and regulations for the general superintendence, direction, and management of all such persons, and to appoint a chief and other clerks. All military and civil officers charged with the execution of any law or order liberating slaves, are required to make returns of their proceedings to the Commissioner, who is authorized to establish regulations for the treatment and disposition of all freedmen, that their rights and those of the Government may be duly determined and maintained. Assistant Commissioners of Freedmen are to be appointed in the rebellious States when brought under military authority, (each with an annual salary of \$2,500,) with power to permit freedmen to cultivate lands in those districts which have been, or may be, abandoned by their former owners. and all real estate within such districts to which the United States shall have acquired title and not previously appropriated to other uses, to adjust wages, receive returns, &c.]

#### IN SENATE.

1864, April 12-Mr. SUMNER reported from the Committee on Slavery and Freedmen a bill to establish a Bureau of Freedmen, which was read and passed to a second reading

May 25 — Mr. SUMNER reported back the House bill with an amendment in the nature <sup>®</sup> of a substitute.

June 8-It was considered in the Senate.

June 15-Mr. CARLILE moved to postpone the bill until the first Monday of December next; which was rejected-yeas 13, nays 23, as follows:

YEAS-Messrs. Buckalew, Carlile, Davis, Grimes, Hen-dricks, Hicks, Johnson, Powell, Richardson, Riddle, Sauls-bury, Van Winkle, Willey-13.

NAYS-Messrs. Anthony, Brown, Clark, Conness, Dixon, Doolittle, Fessenden, Foot, Hale, Harlan, Harris, Howe, Lane of Indiana, Lane of Kansas, Morgan, Morrill, Ramsey, Sherman, Sunner, Ten Eyck, Trumbull, Wade, Wilson-23.

The amendment of Mr. SAULSBURY (offered on the 13th) to add this new section:

That all white persons in the States not in revolt shall be protected, in their constitutional rights, and that no such person shall be deprived of life, liberty, or property, with-out due process of law; nor shall any such person be held to answer for a capital or otherwise infamous crime unless on a presentment or indictment of a grand jury except in cases arising in the land or naval forces, or in the militän be in calculation of the start in the start of the start in the start of the start is a start of the start in the start of the start is a be in cases arising in the land or naval forces, or in the militän when in actual service in time of war or public danger; non shall any such person, except as aforesaid, be tried for any crime or offence whatever by court-martial or military commission.

### Was rejected—yeas 8, nays 29, as follows :

YEAS-Messrs. Buckalew, Carlile, Davis, Hendricks, Pow-

KEAS—Messrs. Buckalew, Cartale, Davis, Henaricks, Fou-ell, Richardson, Riddle, Saulsbury—8. NATS—Messrs. Anthony, Brown, Clark, Conness, Dixon, Doolittle, Fessenden, Foot, Foster, Grimes, Hale, Harlan, Harris, Hicks, Howe, Johnson, Lane of Indiana, Morgan, Morriil, Pomeroy, Ramsey, Sherman, Sumner, Ten Eyck, Trumbull, Van Wiukle, Wade, Willey, Wilson—29.

June 28-The bill passed-yeas 21, nays 9, as follows :

as follows: YEAS-Messrs. Anthony, Chandler, Clark, Conness, Doo-little, Foot, Foster, Harlan, Howe, Lane of Indiana, Mor-gan, Morrill, Pomeroy, Ransey, Sprague, Sunner, Trum-bull, Van Winkle, Wade, Wilkinson, Wilson-21. Nars-Messrs. Buckalew, Carlile, Cowan, Davis, Hen-dricks, McDaugall, Powell, Riddle, Willey-9.

During the pendency of the bill in the Senate as in Committee of the Whole, Mr. WILLEY offered this as a new section:

That whenever the said Commissioner cannot find aban-That whenever the said commissioner cannot find aban-doned real estate on which to employ all of the freedmen who may come under his care and control by virtue of this act, it shall be his duty, so far as may be practical, to provide for them homes and employment with humane and suitable persons at fair and just compensation for their services; and that in order the more effectually to accomplish this purpose the said Commissioner shall open a correspondence with the Governors and the various municipal authorities of the different States requesting their co operation in this behalf.

Which was agreed to-yeas 19, nays 15, as follows:

YEAS-Messrs. Anthony, Brown, Clark, Davis, Doolittle, Foot, Grimes, Harlan, Harris. Henderson, Hicks, Howe, Lane of Indiana, Ramsey, *Riddle, Saulsbury*, Sprague, Van Winkle, Willey-19. NATS-Messrs. Buckalew, Conness, Cowan, Foster, Hen-dricks, Lane of Kansas, Morgan, Morrill, Fouell, Sumner, Ten Eyck, Trumbull, Wade, Wilkinson, Wilson-15.

In open Senate-

Mr. Wilson moved to strike out these words; which was rejected-yeas 14, nays 14, as follows:

YEAS-Messrs. Buckalew, Chandler, Clark, Conness, Hendricks, Morgan, Morrill, Pomeroy, Sumner, Ten Eyck, Trumbull, Wade, Wilkinson, Wilson-14. NAYS-Messrs. Brown, Davis, Foot, Grimes, Harlan, Harris, Hicks, Johnson, Lane of Indiana, Powell, Ramsey, Saulsbury, Van Winkle, Willey-14.

The last clause, "and that," &c., was afterwards stricken out as unnecessary, the Commissioner having full discretion over the subjectmatter, and being accessible to all persons interested.

June 30-The bill as amended by the Senate, was referred to the Select Committee in the House, who recommended a non-concurence in the Senate amendments; when the bill was postponed to December 20th next.

The postponement of the bill leaves unsettled the proposition to repeal the joint resolution amendatory of the confiscation act. Each House passed a repealing section, but neither passed the other's.

# LEGISLATION, ORDERS, PROCLAMATIONS,

AND

# **PROPOSITIONS RELATIVE TO THE WAR AND TO "PEACE."**

# The Enrollment Acts.

ACT OF MARCH 3, 1863. Third Session, Thirty-Seventh Congress.

The bill passed the Senate without a call of the yeas and nays.

#### IN HOUSE.

February 25---Mr. Cox moved to insert the word "white" so as to limit the enrolment to "white" able-bodied males.

Mr. LOVEJOY called the yeas and navs, but they were not ordered. And the amendment

was disagreed to—yeas 53, nays 85. Mr. WICKLIFFE, of Kentucky, moved to add the following to the thirteenth section:

Provided, That the men thus called into the service shall be by the Governor of the State organized into companies and regiments, and officers to command them shall be appointed and commissioned by authority of the State according to the provisions of the constitution and laws thereof. and in obedience to the Constitution of the United States.

Which was negatived—yeas 55, navs 104. The YEAS were :

THE TEAS WEIE: YEAS—Messrs. William Allen, William J. Allen, Ancona, Baily, Biddle, Clements, Conway, Corning, Cox, Cravens, Crittenden, Erglish, Fisher, Foule, Granger, Grider, Hale, Hall, Harding, Holman, Johnson, Kerrigan, Killinger, Knapp, Law, Lazear, Mallory, May, Menzies, Morris, Noble, Norton, Nugen, Pendleton, Perry, Price, Robinson, James S. Rollins, Shiel, Smith, John B. Steele, William G. Steele, Stiles, Fallandigham, Voorhees, Wadsworth, Ward, Webster, Whaley, Chilton A. White, Wickliffe, Wood, Woodruff, Wright, Feaman-55.

### THE \$300 COMMUTATION CLAUSE.\*

Mr. HOLMAN moved to strike out the thirteenth section, which provides for a commuta-tion, not exceeding \$300, where parties are drafted; which was disagreed to-yeas 67, nays 87, as follows:

YEAS-Messrs. William Allen, William J. Allen, Alley, An-cona, Beaman, Samuel S. Blair, Blake, Clements, Colfax, Roscoe Conkling, Conway, Corning, Cox, Cravens, Davis, Dawes, Delaplaine, Diven, Eliot, Fouke, Franchot, Granger, Hall, Holman, Hutchins, Johnson, Julian, Francis W. Kel-logg, William Kellogg, Kerrigan, Knapp, Law, McPherson,

\* Pending the engineer bill in Senate, March 2-Mr. TRUMBULL offered a section repealing the commutation clause; which was rejected-yeas 10, nays

Za, as follows: YEAS-Messis. Grimes, Harlan, Howe, Lane of Indiana, Powell, Saulsbury, Ten Eyck, Trumbull, Wilkinson, Wilson

Fowelt, Szutsbury, 101 Eyes, Humbury, Humbury, Colamer, Cowan, of Missouri-10. Navs-Messre. Arnold, Chandler, Clark, Collamer, Cowan, Davis, Dixon, Fessenden, Foot, Foster, Harris, Hender-ton, Hicks, Howard, Kennedy, King, Lane of Kansas, Mor-rill, Nesmith, Pomeroy, Sherman, Sumner, Turpie, Willey, Wilson of Massachusetts-25.

May, Morris, Noble, Norton, Nugen, Pendleton, Perry, Por-ter, Price, Riddle, Robinson, Edward H. Rollins, James S. Rollins, Shanks, Sherman, Shiel, John B. Steele, William G. Steele, Stiles, Vallandigham, Van Wyck, Foorhees, Wads-worth, Wall, Ward, Washburne, Alhert S. White, Chilton A. White, Wickliffe, Wilson, Windom, Wood, Worcester, Feaman-67.

Molett, Markinger, Wilson, Windom, Mour, Mouse, Forman-67.
 Yeas-Messrs. Aldrich, Arnold, Ashley, Babbitt, Baily, Baker, Baxter, Biddle, Bingham, Jacob B. Blair, William G. Brown, Buffnton, Calvert, Campbell, Casey, Chamberlain, Clark, Frederick A. Conkling. Crisfield, Cutter, Delano, Dunn, Edgerton, Edwards, Ely, English, Fenton, Samuel C. Fessenden, Fisher, Flanders, Frank, Gooch, Goodwin, Gurley, Halun, Harrison, Hickman, Hooper, Horton, Kelley, Killinger, Lansing, Lazear, Leary, Lchman, Loomis, Lovejor, Low, McIndoe, McKean, McKnight, Marston, Maynard, Moorhead, Anson P. Morrill, Nixon, Olin, Patton, Timothy G. Phelps, Pike, Pomeroy, Alexander H. Rice, John II. Rice, Surgent, Sedgwick, Segar, Shefield, Shellabarger, Sloan, Smith, Spaulding, Stratton, Benjamin F. Thomas, Francis Thomas, Train, Trimble, Trowbridge, Van Horn, Verree, Walker, Wallace, Walton, Webster, Wheeler, Woodruff, Wright-87.

features of which are embodied in these sections :

That such militia of the several States shall be enrolled under the authority of the respective States, and when any number thereof shall be called out by the President of the United States as authorized by law, the same shall be or-ganized into companies and regiments by the Governor of each State cubiert to the prelations extilision by the des such state, subject to the regulations established by the President of the United States in pursuance with existing law, and the company, field, and staff officers of the forces so organized shall be commissioned by such Governor as so organized shall be commissioned by such dovernor as now provided by law for commissioning officers for the vol-unteer forces: *Provided*, That such militia shall not be called ont for a period of more than one year: *And pro-vided further*, That the President of the United States in calling out the militia shall apportion the number from each State according to population, having regard to the number of volunteers or militia already furnished by the States reserveively. States respectively.

That the pay of the privates in the regular army and volun-teers and militia in the service of the United States shall be \$15 per month, from and after the 1st day of March, 1863, until otherwise provided by law.

Which was negatived-yeas 44, nays 108. The yeas wcre :

YEAS-Messre. William Allen, Ancona, Biddle, Conway, Corning, Cox, Cravens, Crittenden, Delaplaine, English, Fouke, Granger, Hall, Harding, Holman, Johnson, Kerrigan, Killinger, Knapp, Law, Malbry, May, Menzies, Morris, Noble, Nortom, Nugen, Pendleton, Price, James S. Rollins, Shiel, Smith, John B. Steele, William G. Steele, Skiles, Voor-hees, Wudsworth, Ward, Whaley, Chillon A. White, Wood, Woodruff, Wright, Yeaman-44.

The bill then passed-yeas 115, nays 49, as follows:

YEAS-Messrs. Aldrich, Alley, Arnold, Ashley, Babbitt, Baily, Baker, Baxter, Beaman, Bingham, Jacob B. Blair, Samuel S. Blair, Blake, William G. Brown, Buffinton, Calvert, Campbell, Casey, Chamberlain, Clark, Colfax, Frederick A. Conkling, Roscoe Conkling, Crisfield, Cutler, 261

Davis, Dawes, Delano, Diven, Dunn, Edgerton, Edwards, Eliot, Ely, Fenton, Samuel C. Fessenden, Thomas A. D. Fessenden, Fisher, Flanders, Franchot, Frank, Gooch, Goodwin, Granger, Gurley, Hahn, Haight, Hale, Harrison, Hickman, Hooper, Horton, Hutchins, Julian, Kelley, Fran-tis W. Kellogg, William Kellogg, Killinger, Lausing, Leary, Lehman, Loomis, Lovejoy, Low, McIndoe, McKean, Mc-Knight, McPherson, Marston, Maynard, Mitchell, Moor-head, Anson P. Morrill, Justin S. Morrill, Nixon, Olin, Pat-ton, Timothy G. Phelps, Pike, Pomeroy, Porter, Alexander H. Nice, John H. Rice, Riddle, Edward If. Rollins, Sar-gent, Sedgwick, Segar, Shanks, Shelfield, Shellabarger, Sherman, Sloan, Spaulding, Stevens, Stratton, Benjanin P. Thomas, Francis Thonas, Train, Trimble, Trowbridge, Walton, Washburne, Webster, Wheeler, Albert S. White, Wilson, Windom, Worcester–115.
 NATS-Messrs. William Atlen, Hültiam J. Allen, Ancona, Biddle, Clements, Conway, Corning, Coz, Cravens, Critten-den, Delapiane, Dunlang, English, Fauke, Grider, Hall, Har-dilory, Homan, Johnson, Kerrigan, Knapp, Law, Lazear, Mallory, May, Menzies, Morris, Noble, Norton, Nugen, Pen-fleton, Perry, Price, Robinson, James S. Rollins, Shiel, John B. Steele, William C. Steele, Stiles, Vallanadigham, Foorhees, Watsworth, Wird, Whaley, Chilton A. White, Wickligfe, Wood, Woodruff, Yeaman-49.

#### IN SENATE.

February 28-The Senate resumed the consideration of the amendments made by the House, as recommended by the Military Committee, when, after debate,

Mr. BAYARD moved to postpone the bill indefinitely, which was negatived-yeas 11, nays 35, as follows:

YEAS-Messrs. Bayard, Carlile, Davis, Kennedy, Latham, Powell, Rice, Richardson, Saulsbury, Wall, Wilson of Missouri-11.

Sourn-11. NArs-Messrs. Anthony, Arnold, Chandler, Clark, Colla-mer, Cowan, Dixon, Doolittle, Fessenden, Foot, Foster, Grimes, Harding, Harlan, Harris, Henderson, Hicks, How-ard, Howe, King, Lane of Indiana, Lane of Kansas, McDou-gall, Morrill, Nesmith, Pomeroy, Sherman, Sumner, Ten Eyck, Trumbull, Wade, Wilkinson, Willey, Wilmot, Wilson of Macsedbuette, 25. of Massachusetts-

The amendments of the House were then concurred in, and the bill was declared passed.

#### THE SUPPLEMENT TO THE ENROLLMENT ACT, FEBRUARY 24, 1864.

#### First Session, Thirty-Eighth Congress.

1864, January 18-The bill passed the Senate-yeas 30, nays 10, as follows :

YEAS-Messrs. Anthony, Clark, Collamer, Conness, Cowan, Davis, Dixon, Doolittle, Fessenden, Foot, Foster, Hale, Hard-Davis, Dixon, Doonttle, Fessenden, Foot, Foster, Hale, Hard-ing, Harlan, Harris, Howard, Johnson, Lane of Kansas, Morgan, Morrill, Nesmith, Pomeroy, Ramsey, Sprague, Sum-ner, Ten Eyck, Van Winkle, Wade, Willey, Wilson-30. NATS-Messrs. Buckalew, Carlile, Grimes, Hendricks, Howe, Lane of Indiana, Powell, Saulsbury, Wilkinson, Usatet 10.

Wright-10.

The bill contained a provision that colored troops enlisted and mustered into the service of the United States in any State shall not be credited upon the quota of any other State; which was adopted-yeas 27, nays 11, as follows:

YEAS-Messrs. Brown, Buckalew, Conness, Cowan, Davis, PEAS—Messrs. Brown, Buckalew, Conness, Cowan, Darts, Doolittle, Foot, Grimes, Hale, Harding, Harlan, Henderson, Hendricks, Howe, Johnson, Lane of Indiana, Lane of Kan-sas, Morgan, Morrill, Nesmith, Pomeroy, Ramsey, Sauls-bury, Sherman, Ten Eyck, Trumbull, Wright—27. NAYS—Messrs. Anthony, Clark, Dixon, Fessenden, Foster, Harris, Howard, Sprague, Sumner, Wilkinson, Wilson—11.

these was the following substitute for the 27th section, as reported :

#### ENROLLMENT OF COLORED PERSONS.

SEC. 27. That nothing contained in this act shall be so construed as to prevent or prohibit the enlistment of men in the States in rebellion under the orders of the War Department.

And in lieu thereof insert:

All able-bodied male persons of African descent, between the ages of twenty and forty-five years of age, whether citizens or not, resident in the United States, shall be enrolled according to the provisions of the act to which this is a supplement, and form part of the national forces; and when a slave of a loyal citizen shall be drafted and mustered when a slave of a loyal citizen shall be draited and mustered into the service of the United States his master shall have a certificate thereof. The bonnty of \$100, now payable by law for each drafted man, shall be paid to the person to whom such drafted person owes service or labor at the time of his muster into the service of the United States, on free-ing the person. The Secretary of War shall appoint a commission in each of the slave States represented in Congress charged to award a just compensation, not exceeding \$300, to each loyal person to whom the colored volunteer may owe service, who may volunteer into the service of the United States, payable out of the commutation money upon the master freeing the slave.

Which was agreed to-yeas 84, nays 71, as follows:

follows: YEAS-Messrs. Alley, Allison, Anderson, Arnold, Ashley, John D. Baldwin, Baxter, Beaman, Jacob B. Blair, Bont-well, Boyd, Brandegee, Broomall, William G. Brown, Cobb, Creswell, Henry Winter Davis, Thomas T. Davis, Dawes, Deming, Dixon, Driggs, Eckley, Eliot, Frank, Garfield, Gooch, Grinnell, Hale, Hooper, Hotchkiss, Asahel W. Hub-bard, John H. Hubbard, Hulburd, Julian, Kasson, Kelley, Francis W. Kellogg, Orlando Kellogg, Loan, Longyear, Marvin, McBride, McClurg, McIndee, Samuel F. Miller, Moorbead, Morrill, Daniel Morris, Amos Myers, Leonard Myers, Norton, Charles O'Neill, Orth, Patterson, Perham, Pike, Pomeroy, Alexander H. Rice, John H. Rice, Edward H. Rollins, Schenck, Scofield, Shannon, Sloan, Smith, Smithers, Spalding, Starr, Stevens, Thayer, Thomas, Tracy, Upson, Yan Valkenburgh, Elihu B. Washburne, William E. Washburn, Webster, Whaley, Williams, Wilder, Wilson, Windom, Woodbridge-S4.

D. Washburn, Webster, Whaley, Williams, Wilder, Wilson, Warss-Messrs, James C. Allen, William J. Allen, Ancona, Baily, Augustus C. Balduön, Eliss, Erooks, James S. Broum, Chanler, Coffroth, Cole, Coz, Cravens, Dawson, Denison, Eden, Edgerion, Eldridge, Finck, Ganson, Grider, Griswold, Hell, Harding, Harrington, Benjamin G. Harris, Herrick, Higby, Holman, Hutchins, William Jolmson, Kalbfeisch, Kernan, King, Knapp, Lavo, Lazeer, LeBlond, Long, Mallory, Marcy, McDowell, McKinney, James R. Morris, Morrison, Nelson, Noble, Odell, John O'Neill, Pendleton, Radford, Sameel J. Randall, William H. Randall, Robinson, Rogers, James S. Rollins, Ross, Scott, John B. Steele, Win G. Steele, Stiles, Strouse, Stuart, Sweat, Yoorhees, Wadsworth, Wheeler, Chilton A. While, Joseph W. While, Winfield, Fernando Wood-71. Wood-71.

The bill passed-yeas 94, nays 65, as follows :

The bill passed—yeas 94, nays 65, as follows: YEAS—Messrs. Alley, Allison, Anderson, Arnold, Ash-ley, Baily, John D. Baldwin, Baxter, Beaman, Jacob B. Blair, Boutwell, Boyd, Brandcgee, Broomall, William G. Brown, Cobb, Cole, Creswell, Henry Winter Davis, Thomas T. Davis, Dawes, Deming, Dixon, Driggs, Eckley, Eliot, Farnsworth, Frank, Garfield, Gooch, Grinnell, Griswold, John H. Hubbard, Hulburd, Jenckes, Julian, Kasson, Kel-ley, Francis W. Kellogg, Orlando Kellogg, Loan, Longyear, Marvin, McAltister, McBride, McClurg, McIndoe, Samuel F. Miller, Moorhead, Morrill, Daniel Morris, Amos Myers, Leonard Myers, Norton, OddL, Charles O'Neil, Orth, Pat-terson, Perham, Pike, Pomeroy, William H. Randall, Alex-ander H. Rice, John H. Rice, Edward H. Kollins, Schenck, Scofield, Shannon, Stoan, Smith, Smithers, Spaldiug, Starr, Scofield, Shanoon, Sloan, Smith, Smithers, Spalding, Starr, Sterens, Thayer, Thomas, Tracr, Upson, Van Valkenburgh, Filhin B. Washburne, William B. Washburn, Webster, Whaley, Wheeler, Williams, Wilder, Wilson, Windom, Whaley, Wheeler Woodbridge-94.

MATS-Messrs Anthony, Clark, Dixon, Foster,
 Harris, Howard, Sprague, Sumner, Wilkinson, Wilson-11.
 February 1—The House considered, in Committee of the Whole, the substitute for the Senate bill reported by the House Committee on Military Affairs.
 February 11—The House agreed to the amendments made by the Committee of the Whole, reserving a few for special vote; among
 Woodbridge-94.
 Woodbridge-94.
 Nars-Messrs. James C. Allen, William J. Allen, An-Charler, Ciffroth, Cca, Cravens, Dauson, Grüher, Hall, Hording, Harrington, Benjamin G. Harris, Herrick, Holman, Hulchins, William Johnson, Kallfeisch, Kernan, King, Knapp, Law, Lazear, Le Blond, Mallory, Marcy, MoDovell, McKinney, William H. Miller, James R. Morris, Morrison, Nelson, Weil, John O'Neill, Pendleon, Radford, Samuel J. Randall, Robinson, Rogers, J. S. Rollins, Ross, Scott, John B. Steele, W. G. Steele, Stiles, Strouse, Stuart, Sweat, Voorhees, Wads-

February 15-The Senate non-concurred in the amendments of the House.

February 16-The House insisted on its amendments, and asked a Committee of Conference.

February 19-The Scnate concurred in the report of the Committee of Conference-being the law as it now is-yeas 26, nays 16, as follows:

YEAS-MOSSTS. Anthony, Clark, Dixon, Doolittle, Fessen-den, Foot, Foster, Grimes, Hale, Harding, Harris, Hender-son, Johnson, Lane of Kausas, Morgan, Morrill, Nesmith, Ramszy, Sherman, Sprague, Sumner, Ten Eyck, Van Win-kle, Wado, Willey, Wilson-28. NAYS-Messrs. Buckalew, Carbile, Chandler, Conness, Davis, Harlan, Hendricks, Howard, Howe, Lane of Indiana, Powell, Riddle, Saulsbury, Trumbull, Wilkinson, Wright, -16.

-16.

February 19-The House concurred in the report of the Committee of Conference-yeas 71, navs 23, as follows:

nays 23, as follows: YEAS-MESSTES. Alley, Allison, Anderson, Arnold, John D. Baldwin, Baxter, Beaman, Francis P. Elair, Blow, Boutwell, Brandegee, Broomall, Ambrose W. Clark, Cobb, Cole, Henry Winter Davis, Dawes, Deming, Dixon, Dumont, Eckley, Eliot, Farnsworth, Fenton, Frank, Gooch, Grinnell, Hale, Higby, Hooper, Asahel W. Hubbard, Jenckes, Julian, Kas-son, Kelley, Orlando Kellogg, Loan, Longyear, Marvin, McBride, McClurg, Samuel F. Miller, Moorhead, Amos Myers, Loonard Myers, Odell, Charles O'Neil, Orth, Patter-son, Perham, Pike, Fomercy, Price, William H. Randall, Alexander H. Rice, John H. Rice, Edward H. Rolbins, Schenck, Scofeld Shannon, Smithers, Spalding, Starr, Skeb-bins, Upson, Van Valkenburgh, Ellin B. Washburne, Wi-liams, Wilder, Wilson, Windom-71. Nays-Messrs. Augustus C. Baldwin, Brooks, Chanler, Coz, Dawson, Edgerton, Eldridge, Benjamin G. Harris, Morrison, Noble, John O'Neill, Pendleton, Samuel J. Ran-dall, Janes S. Rolins, Ross, Ward, Chilton A. While, Jo-seph W. White-23.

seph W. White-23.

# SUPPLEMENT OF JULY 4, 1864.\* IN SENATE.

1864, June 8-The Senate took up this bill reported from the Military Committee :

That so much of the actentitled "An act for enrolling and the so much of the actentitled for other purposes," apcalling out the national forces, and for other purposes,

\* Additional legislation is asked by the President, Secre-tary of War, and Provost Marshal General, whose communi-cations are appended:

To the Senate and

House of Representatives: I have the honor to submit for the consideration of Congress a letter and inclosure from the Secretary of War, with my concurrence in the recommendation therein made WASHINGTON, D. C., June 8, 1864.

### WAR DEPARTMENT.

WASHINGTON CITY, June 7, 1864. SIR: I beg leave to submit to you a report made to me by the Provost Marshal General, showing the result of the draft now going on to fill the deficiency in the quotas of certain States, and recommending a repeal of the clause in the enrollment act commonly known as the \$300 clause. The recommendation of the Provost Marshal General is approved by this Department, and I trust that it will be recommended by you to Congress.

The recent successes that have attended our arms lead to The recent successes that have attended our arms lead to the hope that by maintaining our military strength and giving it such an increase as the extended field of opera-tions may require, an early termination of the war may be attained. But to accomplish this it is absolutely necessary that efficient means be taken, with vigor and promptness, to keep the Army up to its strength and supply deficiencies occasioned by the losses sustained by casualties in the field. To that end resort must be had to a draft, but ample expe-fence has new shown that the necuniary exemution from rience has now shown that the pecuniary exemption from service frustrates the object of the enrollment law by furnishing money instead of men. An additional reason for repealing the \$300 clause is that

worth, C. A. While, Joseph W. While, Winfield, Fernando Wood-65. Exhauser 15. The Senate non-concurred in is hereby, repealed: *Provided*, That nothing contained in this act shall be construed to alter the provisions of exist-

ing laws relative to persons actually furnishing substitutes. SEC. 2. That nothing in the act approved February 24, 1864, amending the act approved March 3, 1863, for enrolling and calling out the national forces, shall be construed to repeal that part of the said act approved March 3, 1863, which requires that the board of enrollment, in making drafts, shall "make a draft of the required number and fifty per centum in addition."

SEC. 3. That section twelve of the act for enrolling and calling out the national forces, and for other purposes, approved March 3, 1863, be, and is hereby, so amended that the notice to be served on drafted men may be served within ten days after such draft or at any time within six months therefrom.

June 9-Mr. COLLAMER moved the following additional sections :

That all calls for drafts hereafter made under the act entitled "An act for enrolling and calling out the national forces, and for other purposes," approved March 3, 1863, and of any act in addition to or amendment thereof, shall be

for a term not exceeding one year. That this act shall not extend to or include drafts to be made in any district or subdivision thereof, to fill its quota on calls already made, but the same shall be completed un-der the laws in force before the passage hereof. That no person drafted on future calls shall be liable to

be again drafted until the present enrollment shall be exhausted.

That the number of men furnished from any district for the service of the United States beyond and above its quota on calls heretobefore made, and the term of service of such men, shall be considered and allowed to said district in calls hereafter made.

The first section of the amendments was agreed to-yeas 22, nays 17, as follows :

YEAS -- Messrs. Anthony, Buckalew, Clark, Collamer,

it is contemplated to make the draft for a comparatively short term. The burden of military service will therefore be lightened, but its certainty of furnishing troops is an absolute essential to success.

I have the honor to be, your obedient servant, EDWIN M. STANTON Secretary of War.

To the PRESIDENT.

WAR DEPARTMENT PROVOST MARSHAL GENERAL'S OFFICE

WASHINGTON, D. C., June 6, 1864. Sir: In accordance with the amended enrollment act approved February 24, 1864, and your orders on the subject, I am now conducting a draft in various sub-districts for their am how conducting a triat in various sub-districts for here respective deficiencies on quotas of troops heretofore as-signed. The results of this draft, so far as shown by reports of this date, are worthy of attention. They are, briefly, as follows:

Number of drafted men examined	14,141
Number exempted for physical disability 4,374	-
Number exempted for all other causes 2,632	
Number exempted for all other causes	
Total exempted	7,016
•	
Number paid commutation money 5,050	
Number who have furnished substitutes 1,416	
Number held for personal service 1,259	

(This last includes some who may yet pay commutation money.)

Total not exempted..... 7,725

These reports come from sub-districts in eight different States. I invite your attention to the small proportion of soldiers being obtained under the existing law. I see no soldners being obtained under the existing law. I see no reason to believe that the army can be materially strength-ened by draft so long as the \$300 clause is in force, nor do I think it safe to assume that the commutation paid by a drafted man will enable the Government to procure a vol-unteer or substitute in his place. I do not think that large bounties by the United States should be again resorted to for raising troops. I recommend that the \$300 clause, as it is known, be renealed. is known, be repealed.

I am, sir, very respectfully, your obedient servant, JAMES B. FRY, Provost Marshal General,

Hon. E. M. STANTON, Secretary of War.

Davis, Doolittle, Fessenden, Foot, Foster, Harris, Howard, Howo, Lane of Kansas, Morrill, Powell, Richardson, Sum-ner, Van Winkle, Wade, Wilkinson, Willey, Wilson-22. NATS-Messrs. Brown, Curile, Chandler, Conness, Grimes, Harlan, Henderson, Johnson, McDougall, Morgan, Nesmith, Pomeroy, Ramsey, Sherman, Sprague, Ten Eyck, Trum-pull-12. bull-17

### The remaining sections were agreed to.

#### ENLISTMENT OF INDIANS>

#### June 20-Mr. Brown moved this new section :

That in any draft which may hereafter take place, all Indian tribes with whom treaties have been made by the United States and who are receiving annuities from the Government shall be required to furnish their respective quota of ernmentshall be required to turnish their respective quota of men; and that the duties of enrollment, or ascertaining the approximate numbers of said tribes, shall, whenever the same is necessary, be performed by the Indian agents as part of their appropriate duty, without further compensa-tion, under instructions from the Provost Marshal General. And is the covert that any trib reconvirgenmultiges as formaand in the event that any tribe receiving annuities as afore-said shall refuse or fail to furnish its required quota, then and in that event the whole or such part of their said an-nuities as the Secretary of the Interior shall deem adequate to provide substitutes shall be withheld from the annual payment, and shall be placed in the Treasury along with the commutation fund heretofore paid for a like purpose: And provided further, That the force thus raised may be employed by the Government for the purpose of maintaining peace and protecting from hostile incursion the Indian and other Territories, and of relieving such troops as are now engaged in that duty.

Mr. DOOLITTLE moved this substitute for the above:

That the Secretary of War is authorized to receive into the military service of the United States Indians of tribes in treaty with the United States, to be employed as a part of the military force of the United States for the purpose of maintaining peace and protecting from hostile incursion the Indian Territory and other Territories where the hostile or invading force is in whole or in part composed of hostile Indians.

Which was agreed to-yeas 24, nays 12, as follows:

YEAS-Messrs. Chandler, Clark, Collamer, Conness, Dixon, FEAS-JOSSIS. Chandler, Collamer, Conness, Dixon, Doolittle, Foot, Foster, Harlan, Hurris, Howard, Lane of Indiana, Lane of Kansas, McDougall, Morgan, Pomeroy, Sherman, Sprague, Sumner, Ten Eyck, Van Winkle, Wade, Willey, Wilson-24. NArs-Messrs. Brown, Buckalew, Carlile, Davis, Grimes, Hendricks, Johnson, Powell, Ramsey, Richardson, Sauls-bury, Wilkinson-12.

The amendment, as amended, was then rejected-yeas 10, nays 29, as follows:

YEAS-Messrs. Brown, Doolittle, Grimes, Harlan, Lane of Indiana, Lane of Kansas, McDougall, Pomeroy, Ramsey,

Indiana, Lane of Kalisas, *McDougati*, Pomercy, Ramsey, Sprague-10. NATS-Messrs. Anthony, *Buckalew, Carlile*, Chandler, Clark, Collamer, Cowan, *Davis*, Dixon, Foot, Foster, Harris, *Hendricks*, Howard, Johnsen, Morgan, *Powell*, *Richardson*, *Riddle*, *Saulsbury*, Sherman, Sumner, Ten Eyck, Trum-bull, Van Winkle, Wade, Wilkinson, Willey, Wilson-29.

#### NO SUBSTITUTION.

Mr. McDougall moved to insert this provision :

And from and after ten days from the passage of this act substitutes shall not be allowed in place of persons subject to draft and regularly drafted into the service of the United States.

Which was rejected-yeas 6, nays 35, as follows:

YEAS-Messrs. Chandler, Doolittle, Grimes, McDougall,

YEAS-Messrs. Chandler, Doolittle, Grimes, AcDougal, Ramsey, Wilkinson-6. NArs-Messrs. Brown, Buckalew, Carlile, Clark, Collamer, Conness, Cowan, Davis, Dixon, Foot, Foster, Harlan, Harris, Hendricks, Howard, Howe, Johnson, Lane of Indiana, Lane of Kansas, Morgan, Morrill, Nesmith, Pomeroy, Powell, Richardson, Riddle, Sherman, Sprague, Sunner, Ten Eyck, Trumbull, Van Winkle, Wade, Willey, Wilson-35.

Mr. HENDRICKS moved to strike out the first section ; which was agreed to-yeas 21, nays 18, as follows:

YEAS-Messrs. Buckalew, Carlile, Clark, Collamer, Cow-

an, Davis, Dixon, Foot, Foster Harris, Hendricks, Johnson, McDougall, Morrill, Powell, Richardson, Riddle, Saulsbury,

McDaugait, Morrin, Fourta, Richardson, Rudne, Saussory, Van Winkle, Willey, Wilson-21. NAYS-Messrs. Anthony, Brown, Chandler, Conness, Fes-senden, Grimes, Harlan, Howe, Lane of Indiana, Lane of Kansas, Morgan, Nesmith, Ramsey, Sherman, Sprague, Sumner, Ten Eyck, Trumbull-18.

Pending other propositions, the bill was recommitted to the Committee on Military Affairs, and all the amendments fell by the recommitment.

June 23-The Senate considered the bill, which was reported back in the shape in which it was originally reported.

Mr. MORGAN moved to amend by adding this section :

That in the calls for drafts hereafter made under the act "for enrolling and calling out the national forces," and the acts in addition to or amendatory thereof, the same may be made for such term of time as the President shall direct, not exceeding one year.

Mr. Wilson moved to amend the amendment by making it read "shall be made for one year ;" which was rejected-yeas 12, nays 18, as follows :

YEAS-Messrs. Clark, Collamer, Davis, Dixon, Foot, Har-ris, Hendricks, Howe, Johnson, Sumner, Willey, Wilson-12.

NATS-Messrs. Brown, Chandler, Foster, Grimes, Harlan, Howard, Lane of Kansas, Morgan, Pomeroy, Powell, Ram-sey, Riddle, Szulsbury, Sherman, Sprague, Ten Eyck, Trumbull, Wade-18.

Mr. CHANDLER moved to amend the amendment by striking out "not exceeding one year," and inserting "not less than one nor more than three years;" which was rejected—yeas 16, nays 23, as follows :

YEAS -- Messrs. Anthony, Brown, Chandler, Conness, Grimes, Harlan, Howard, Howe, Lane of Kansas, Pomeroy, Ramsey, Sherman, Sprague, Ten Eyck, Trumbull, Wilkinson-16.

NATS-Messrs. Buckalew, Carlide, Clark, Collamer, Davis, Dixon, Doolittle, Foot, Foster, Hale, Harris, Hendricks, Johnson, Morgan, Morrill, Fowell, Riddle, Saulsbury, Sum-ner, Van Winkle, Wade, Willey, Wilson-23.

The amendment offered by Mr. MORGAN was then agreed to-yeas 25, nays 14, as follows:

YEAS-Messrs. Buckalew, Clark, Collamer, Cowan, Davis, JEAS-Messes, Buckatew, Clark, Conlamer, Cowan, Dorsa, Dixon, Boolittle, Foct, Halo, Harris, Hendricks, Howe, Johnson, Lane of Kansas, Morgan, Morrill, Pomeroy, Pow-ell, Richardson, Sunner, Ten Eyck, Van Winkle, Wade, Willey, Willson-25. NATS-Messes, Brown, Carlile, Chandler, Conness, Foster, Cirrage Hwaved McDavaed Warmary, Evided, Shoreter,

Grimes, Howard, McDougall, Ramsey, Riddle, Sherman, Sprague, Trumbull, Wilkinson-14.

Mr. COLLAMER moved to strike out the first section and insert this:

That the thirteenth section of the act entitled "An act for enrolling and calling out the national forces, and for other purposes," approved March 3, 1863, is hereby so amended that the sum to be paid by a drafted man to the Government for the procuration of a substitute shall not exceed \$500, instead of \$300.

Which was rejected without a division.

Mr GRIMES moved a new section, which was agreed to:

That the number of men furnished from any district for the service of the United States beyond and above its quota on calls heretofore made, and the term of service of such men, shall be considered and allowed to said district in calls hereafter made.

Mr. WILSON offered this new section, which he afterwards withdrew, decided opposition to it being manifested:

That any persons resident in the States of Virginia, North Carolina, South Carolina, Georgia, Florida, Alabama, Mis-sissippi, Louisiana, Texas, or Arkansas, who may voluntarily enlist in the military service of the United States for the term of three years or during the war, shall be entitled to the benefits and privileges of existing laws, and such per-sons shall be mustered into the regiments or other organirations of whatsoever State they may elect, or, in the case of colored troops, shall be assigned as now provided by law. And the States or subdivisions of States proving such et-listments shall receive credit for such persons, in accordance with the laws in other cases; but such enlistments as are authorized in any State, under the provisions of this act, shall only continue until such State shall have been made subject to a call for troops.

Mr. GRIMES offered the following new section, which was agreed to :

That no person drafted on future calls or who shall vol-unteer to fill the same shall be liable to be again drafted, until the existing enrollment shall be exhausted.

The bill, as amended, passed-yeas 24, nays 7, as follows :

YEAS-Messrs. Anthony, Brown, Chandler, Clark, Doolittle, Foot, Foster, Grimes, Harlan, Howard, Howe, Lane of Kansas, Morgan, Pomeroy, Ramsey, Sherman, Sprague, Ten Eyck, Trumbull, Van Winkle, Wade, Wilkinson, Wil-ley, Wilson-24. NATS-Messrs. Buckalew, Carbile, McDougall, Powell, Picker Kuldhum, Carbile, McDougall, Powell,

Richardson, Riddle, Saulsbury-7.

June 29--The House having returned a new bill, it was considered in the Senate and amended by substituting the former Senate bill, with an additional section or two.

Mr. SHERMAN offered this new section :

That for the purpose of paying the bounties and of en-forcing the draft provided for in this act, there be levied and collected in addition to the dutics imposed by law, a special daty of five per cent. on all incomes exceeding \$600, accru-ing during the year 1564, which duty shall be assessed and collected in the mode and according to the provisions, penalties, and restrictions provided in the act approved —, en-titled "An act to provide internal revenue to support the Government, to pay interest on the public debt, and for other purposes." And this duty shall be payable on the 1st day of October next, and the Secretary of the Treasury is authorized to prescribe such rules and regulations as to the time and mode of assessment as will secure the collection of this special tax.

Which was agreed to-yeas 25, nays 7, as follows:

YEAS-Messrs. Anthony, Brown, Chandler, Clark, Con-ness, Doolittle, Foot, Foster, Grimes, Hale, Harlan, Howe, Lane of Indiana, Lane of Kansas, Morgan, Pomeroy, Ram-sey, Sherman, Sprague, Sunner, Ten Eyck, Wade, Wilkin-son, Willey, Wilson-25. NATS-Messrs. Euclalev, Davis, Harris, Hendricks, Mo-Dougall, Powell, Riddle-7.

Mr. POWELL offered this, which was agreed to:

That no officers or persons engaged in the military service of the United States shall enlist, recruit, or muster into the military service of the United States any person in any State of the United States to fill the quota of any State in which the person so enlisted, recruited, or mustered into the ser-vice of the United States does not reside. Any recruiting or military officer who shall violate this section shall be dismissed the service of the United States with forfeiture of all pay and allowances, and shall be subject to such further punishment as a court-martial may direct.

The bill then passed the Senate.

June 30 — The Senate, after receiving the message from the House stated in House proceedings of this date, reconsidered their action, and struck out the tax section.

The bill went to a Committee of Conference, whose report being the law as passed (page 116) the Senate, June 2, first rejected it-yeas 16, nays 18, as follows:

YEAS-Messrs. Anthony, Clark, Foot, Foster, Hale, John-Son, Lane of Indiana, Lane of Kansas, Morgan, Morrill, Pomeroy, Ramsey, Sumner, Wade, Wilkinson, Wilson-16, NATS-Messrs. Buckalew, Carlile, Conness, Cowan, Davis, Henderson, Hendricks, Hicks, McDougall, Powell, Riddle, Saulsbury, Sherman, Sprague, Ten Eyck, Trumbull, Van Winblo Willer, 18. Saulsbury, Sherman Winkle, Willey-18.

Afterward the vote was reconsidered, and the report adopted-yeas 18, nays 17, as follows:

YEAS-Messrs. Anthony, Chandler, Clark, Conness, Fes-

senden, Foot, Foster, Hale, Lane of Kansas, Morgan, Morrill, Pomeroy, Ramsey, Sumner, Van Winkle, Wade, Wilkinson, Wilson-18. NATS-Messrs. Buckalevo, Carlile, Davis, Doolittle, Harlan, Harris, Henderson, Hendricks, Howe, Lane of Indiana, Mo-Dougall, Powell, Riddle, Saulsbury, Sherman, Trumbull, Willey-17.

#### IN HOUSE.

1864, June 13-Mr. SCHENCK introduced this bill amendatory of the conscription act:

SEO. 1. That so much of the act entitled "An act for en-rolling and calling out the national forces, and for other purposes," approved March three, eighteen hundred and sixty-three, and of the several acts amendatory thereof, as provides for a commutation in money, to be paid by persons enrolled or drafted for military service, in lieu of actually rendering such military service, be, and the same is hereby, "meashed" and hereafter no puyment of money shall be ac-SEC. 1. That so much of the act entitled "An act for enrepealed; and hereafter no payment of money shall be ac-cepted or received by the Government to release any enrolled or drafted man from obligation to perform military duty.

SEC. 2. That hereafter no person shall be received or ac-Sec. 2. That necessary in person shall be received or ac-cepted to serve in the army of the United States, as sub-stitute for any other person liable to military duty and who may have been enrolled or drafted for that purpose, unless such substitute be the father, brother, or son of the person so enrolled or drafted, and for whom he proposes to become such substitute

SEC. 3. That the President of the United States may, at his discretion, at any time hereafter, order a draft under the provisions of the "Act for enrolling and calling out the national forces, and for other purposes," approved March third, eighteen hundred and sixty-three, and of the several acts amendatory thereto, for soldiers to scrve for a less pe-riod than three years: *Provided*, however, That no such draft shall be for a less term of service than one year.

SEC. 4. That the President shall accompany any order for a draft of men for military service with a notice that he will accept volunteers in lieu of such drafted men prior to the day appointed for the draft, to till the quota or any part day appointed for the drift, to full the quota or any part thereof of any town, township, precinct, or election district; and every person so volunteering, in lieu of a man to be drafted, shall be credited to such town, township, precinct, or election district; and if he volunteers and is accepted and mustered into the service for a term of one ycar, unless sooner discharged, shall receive and be paid by the United sooner discharged, shall receive and be paid by the United States a bounty of one hundred dollars, and if for a term of two years, unless sconer discharged, a bounty of two hun-dred dollars, and if for a term of three years, unless sconer discharged, a bounty of three hundred dollars, one half of which said bounty shall be paid to the soldier at the time of his being mustered into the service, one fourth at the expiration of one half his term of service, and one fourth at the end of his term of service; and the President in any call or order for a draft shall specify the exact time of service for which such draft is to be made, and the volunteers accepted in lieu of the whole or any part of the quotas to be provided under that draft shall be for not less than the term of service for which that draft is ordered. Sec. 5. That section three of an act entitled "An act to

Sec. 5. That section three of an act entitied "All act to amend an act entitled an act for enrolling and calling out the national forcee, and for other purposes," approved Feb-ruary 24th, 1864, be, and the same is hereby, aniended so as to authorize and direct district provost marshals, under the d rection of the Provost Marshal General, to make a draft for fifty per centum in addition to the number required to

fill the quota of any district, as provided by said section. SEC, 6. That, instead of travelling pay, all drafted persons reporting at the place of rendezvous shall be allowed transportation from their places of residence; and persons dis-charged at the place of rendezvous shall be allowed transportation to their places of residence.

Mr. SCHENCK demanded the previous question; but the House refused to second it-yeas The bill was laid over. 45, nays 60.

June 21-The subject was resumed. On a motion to reject the bill, the yeas were 75, nays 75--the SPEAKER voting nay.

VOTE ON SECTION REPEALING THE \$300 COM-MUTATION CLAUSE.

The first section was then stricken out-yeas 100, nays 50, as follows:

YEAS-Messrs, James C. Allen, William J. Allen, Alley, Ames, Ancona, Baily, Augustus C. Baldwin, John D. Bald-win, Blaine, Bliss, Boutwell, Brooks, Broomall, James S. Brown, William G. Brown, Charler, Freeman Clarke, Cof-froth, Cruvens, Thomas T. Davis, Dawes, Dawson, Deming,

Denison, Elen, Edgerton, Eldridge, Eliot, English, Fenton, Finck, Frank, Ganson, Gooch, Grider, Criswold, Ilale, Hard-ing, Harrington, Benjamin G. Harris, Herrick, Holman, Hooper, Hotchkiss, Hutchins, Philip Johnson, Wm. Johnson, Kalbfleisch, Knapp, Law. Lazear, LeBlond, Mallory, Marcy, McDowell, McKinney, Middleton, Samuel F. Miller, Won. H. McDowell, McKinney, Middleton, Samuel F. Miller, Wn. H. Miller, Daviel Morris, James R. Morris, Morrison, Amos Myers, Leonard Nyers, Nelson, Noble, Odell, John O Niell, Patterson, Pendleton, Perhan, Perry, Pruyn, Radford, Samuel J. Randall, Alexander H. Rice, Robinson, Rogers, Edward H. Rollins, Scofield, Scott, John B. Steele, Wm. G. Steele, Stevens, Stiles, Strouse, Stuart, Sweat, Thomas, Upson, Wadsworth, Ward, William B. Washburn, Webster, Whaley, Wheeler, Chilton A. White, Joseph W. White, Wil-liams, Winfeld-100.

Hams, Winfield—100. NArS—Messrs. Arnold, Ashley, Baxter, Beaman, Bleir, Blow, Boyd, Ambrese W. Clark, Cobb, Cole, Dixon, Jonnel-ly, Driggs, Eckley, Karnsworth, Garfield, Higby, Asahel W. Hubbard, John H. Hubbard, Hulburd, Ingersoll, Julian, Kelley, Francis W. Kellogg, Knox, Loan, Lougyear, Mar-vin, McClurg, Moorhead, Morrill, Norton, Charles O'Niell, Orth, Pike, Pomeroy, Price, John H. Rice, Koss, Schenek, Shannon, Sloan, Smithers, Thayer, Tracy, Van Valken-burgh, Ellihu B. Washburne, Wilder, Wilson, Windom—50.

The second section was stricken out without a division.

Several substitutes were offered, but no votes were taken on them.

June 25-Mr. SCHENCK offered a substitute for the bill, the first sections of which were:

That so much of the act entitled "An act for enrolling and calling out the national forces, and for other purposes," approved March 3, 1863, and of the several acts amendatory thereof, as provides for a commutation in money, to be paid by persons enrolled or drafted for military service, in lieu of actually rendering such military service, be, and the same is hereby, repealed is and hereafter no payment of money shall be accepted or received by the Government to release any enrolled or drafted man from obligation to perform military duty.

SEC. 2. And be it further enacted, That the President of the United States may at his discretion, at any time here-after, order a draft for soldiers to serve for a less period than three years: *Provided, however*, That no such draft shall be for a less term of service than one year.

The third section provided for the acceptance of volunteers before the draft, with \$100 bounty for a one year volunteer, \$200 for a two year, and \$300 for a three year. Drafted men, substitutes, and volunteers to be organized together according to States, and, as far as possible, to select their own regiments. Recruiting of persons under sixteen prohibited, and the Secretary of War authorized to discharge persons under eighteen.

After debate,

Mr. Cox moved that the bill be tabled, which was rejected-yeas 57, nays 78, as follows:

Was rejected—yeas 57, nays 78, as follows: YEAS—Messrs. William J. Allen, Anoona, Auguslus C. Baldwin, James S. Broon, Chanler, Cuffroth, Cax, Cravens, Dawson, Eden, Edgerton, Eldridge, English, Finck, Grider, Harding, Harrington, Benjamin G. Harris, Charles M. Harris, Herrick, Holman, Hutchins, Philip Johnson, Wil-biam Johnson, Kernan, Lav, Lazear, Le Blond, Mallory, Marcy, McDowelt, McKinney, William H. Miller, James R. Morris, Morrison, Nelson, Noble, John O Neill, Pendleton, Pruyn, Radford, Samuel J. Randall, Robinson, Rogers, James S. Rolluns, Ross, John B. Steele, William G. Steele, Stüles, Strouse, Stuurt, Sweat, Chillon A. White, Joseph W. White, Winfield, Benjamin Wood, Fernando Wood—57. NATS—Messrs. Alley, Allison, Ames, Anderson, Arnold,

NATS—Messrs. Alley, Allison, Ames, Anderson, Arnold, Ashley, John D. Baldwin, Baxter, Beaman, Blaine, Blair, Blow, Boutwell, Boyd, William G. Brown, Ambrose W. Clark, Cobb, Cole, Creswell, Thomas T. Davis, Dawes, Deming, Dixon, Donnelly, Driggs, Eliot, Farnsworth, Fen-tor Corfield Goodb Grain of Mele Highty Honcore Acaded Deming, Dixon, Donnelly, Driggs, Eliot, Farnsworth, Fen-ton, Garfield, Gooch, Griswo d, Hale, Higby, Hooper, Asahel W. Hubbard, John II. Hubbard, Hulburd, Jenckes, Julian, Kelley, Francis W. Kellogg, Orlando Kellogg, Knox, Little-john, Loan, Longyear, Marvin, McClurg, Moorhead, Mor-rill, Daniel Morris, Amos Myers, Leonard Myers, Norton, Oddz, Charles O'Neill, Orth, Patterson, Perham, Pike, Pomeroy, Price, John H. Rice, Edward H. Rollins, Schenck, Sloan, Smithers, Stevens, Thayer, Thomas, Tracy, Van Valkenburgh, Elihu B. Washburne, William B. Washburn, Webster, Whaley, Wilson, Windom-78.

June 27-The substitute of Mr. SCHENCK, as above, with the addition of a section authorizing persons resident in Virginia, North Carolina, South Carolina, Georgia, Florida, Mississippi, Louisiana, Texas, and Arkansas, enlisting for the war, or not less than one year, to have the benefit of existing laws, credit for them to go to the States procuring the enlistments; provided, that such enlistments as are authorized in any State, under the provisions of this act, shall only continue until such State shall have been made subject to a call for troops; and provided further, that no enlistments shall be made of any soldiers, either in or out of any State, except those enumerated herein, unless full credit is given to the State to which the enlisted soldier belongs, was rejected-yeas 62, nays 92, as follows:

YEAS-Messrs. Allison, Anderson, Arnold, Ashley, Bax-ter, Beaman, Blair, Blow, Boyd, William G. Brown, Am-brose W. Clark, Cobb, Cole, Creswell, Henry Winter Davis, brose W. Chark, Cobb, Cole, Cresweit, Henry Winter Davis, Dixon, Donnelly, Driggs, Garfield, Higby, Hotchkiss, Asahel W. Hubbard, Jno. H. Hubbard, Hulburd, Ingersoll, Jenckes, Julian, Kelley, Knox, Littlejohn, Loxan, Marvin, McBride, McClurg, McIndoe, Samuel F. Miller, Moorhead, Morrill, Daniel Morris, Amos Myers, Leonard Myers, Norton, Charles O'Neill, Orth, Pike, Pomeroy, William H. Randall, John H.

 Daniel Morris, Amos Myers, Leohard Myers, Korola Charles
 O'Neill, Orth, Pike, Pomeroy, William II. Randall, John H.
 Rice, Schenck, Shannon, Sloan, Smith, Smithers, Spalding,
 Thayer, Tracy, Upson, Yan Yulkenburgh, Ellihu B. Washburne, Wilder, Wilson, Windom—62.
 NAYS—Messrs. William J. Allen, Alley, Ames, Ancona,
 Bealty, Augustus C. Eadiwin, John D. Baldwin, Blaine, Bliss,
 Boutwell, Brooks, Broomall, James S. Brown, Chanler, Ceff Toth, Car, Cravens, Dawes, Dawson, Deming, Denison, Eden,
 Edgerton, Eldridge, Eliot, English, Finck, Frank, Ganson,
 Gooch, Grüder, Griswold, Ilale, Harding, Harrington, Belge,
 Miroy, McAllister, McDowell, McKinney, Wm. Hutchins, Philip Johnson, William Johnson, Kallyleisch, Orlando
 Keilorg, Kernan, Kraup, Law, Lacear, LeDlond, Long, Mallory, Matroy, McAllister, McDowell, McKinney, Vm. H. Miler, Janterson, Pendleon, Perham, Prayn, Radford,
 Samuel J. Randall, Alexander H. Rice, Rohinson, Rogers,
 Edward H. Rollins, James S. Stilas, Strouse, Suart, Sweat,
 Thomas, Wadsworth, William B. Vashburn, Webster, Whae,
 Win, G. Steele, Stevens, Stiles, Strouse, Suart, Sweat,
 Thomas, Wadsworth, William B. Washburn, Webster, Whae, Winfield, Fernando Wood-92.

Mr. BROOMALL's substitute was rejected without a division :

That hereafter no person shall be received or accepted to serve in the army of the United States as a substitute for

serve in the army of the United States as a substitute for any other person liable to military duty and who may have been enrolled or drafted for that purpose. SEC. 2. That the term of service of all soldiers hereafter volunteering or being drafted shall be one year unless sooner discharged, and that, in lieu of all bounties, their pay shall be thirty dollars per month. SEC. 3. That all persons hereafter volunteering shall be evolved to the site or county in which they are lighted to

credited to the city or county in which they are liable to drait, if so liable, and if not, then to the city or county which they shall elect.

SEC. 4. That the payment of commutation money under existing laws shall exempt the person paying it from draft for one year, unless within that period all persons liable to draft in his city or county shall be drafted and mustered into service, or shall pay commutation money, in which case he shall be again liable to draft.

SEC. 5. That section three of an act entitled "An act to amend an act entitled 'An act for enrolling and calling out the metional forces and for other nurroses" "approved the national forces, and for other purposes," approved February 24, 1864, be, and the same is hereby, amended so as to authorize and direct district provost marshals, under the direction of the Provost Marshal General, to make a draft for fifty per cent. in addition to the number required to fill the quota of any district, as provided by said section.

SEC. 6. That, instead of traveling pay, all drafted persons reporting at the place of rendezvous shall be allowed trans-portation from their places of residence; and persons dis-charged at the place of rendezvous shall be allowed transportation to their places of residence.

Mr. STEVENS proposed this substitute :

Be it enacted, &c., That the President of the United States is authorized to call into military service not exceeding 500,000 men, in addition to those already called for, to serve for two years, unless sooner discharged, and that if not otherwise obtained a draft may be ordered to take place within forty days, or at such time thereafter as the President may direct.

SEC. 2. That any person who is liable to draft and has been regularly enrolled, may purchase exemption from draft for the term of two years or until the roll is exhausted, by paying \$300 at any time not less than ten days before the time fixed for such draft. Any person who may be drafted may purchase the like exemption by paying \$500 at any time not less than ten days after he shall be duly notified that he is drafted. The commutation moncy thus paid shall go into the Treasury for the purpose of paying the bounties herein provided for. SEC. 3. That to enable the several States to raise the num-

SEC. 5. That to enable the several States to raise the number of troops allotted to them respectively, the Governors of said States may appoint such number of recruiting officers as they may deem proper, and when companies and regiments shall be enlisted, shall commission such company and regimental officers as shall be deemed competent.

Sec. 4. That when, in the judgment of the President, the regiments of the Army are so reduced as to require consolidation, they may be consolidated, and the supernumerary officers may be detailed for the purpose of raising new companies and regiments.

SEC. 5. That if the requisite number of soldiers shall not be raised at the first draft, a second draft may be ordered to take place within ten days, or any longer period which the President may deem proper, of the men remaining unexempted and undrawn, and may be repeated from time to time until the full number shall be obtained.

Size. 6. That every person who shall volunteer as a soldier and be regularly enlisted, shall receive a bounty of \$500, one half thereof to be paid to him when mastered into the service of the United States, and the other half at the end of his term of service, or if honorably discharged; or if he shall be killed or die in the service before the expiration of his term of enlistment, it shall go to his widow, if there be one, and in case there be no widow or children, then to his personal representatives.

SEC.7. That it shall be lawful for any of the States to send recruiting agents into any of the rebel States to enlist soldiers, who shall be credited to the State that may procure their enlistment; and it shall be the duty of the Secretary of War to approve the appointment of agents by the Governors of the respective States.

SEC. 8. That enrolled men may furnish substitutes either before or after they may be drafted according to the present law, who shall be accepted without regard to color, if otherwise competent.

SEC. 9. That the law with regard to persons conscientionsly opposed to bearing arms shall not be altered or affected by this act, except so far as it regards the amount of money to be paid for exemptions.

SEC 10. That it shall be lawful for the President, if he shall deem it expedient, to accept the services of any number not exceeding fifty thousand volunteers, between the ages of forty-five and fifty-five, to be nsed for post or garrison duty, who shall be called the "Old Guard." Such soldiers, when enlisted for two years, unless sooner discharged, shall be entitled to receive \$100 bounty, one half to be paid when mustered into service, the other half at the expiration of their term, if honorably discharged; and if they should be killed or die in the service of the United States, it shall be paid to their respective widows, children, or personal representatives, in the same manner as is provided in section six of this act; and they and their heirs shall be entited to such pensions as are now provided by law for other soldiers. When a drafted man shall claim exemption on account of physical disability. If found unfit for active field service, and fit for garrison or post service, he shall be certified accordingly and placed into the "Old Guard."

Section eleven is the same as the fifth section of the law.

Mr. BLAIR, of Virginia, offered this substitute:

That so much of the act entitled "An act for enrolling and calling out the national forces, and for other purposes," approved March 3, 1863, and of the several acts amendatory thereot, as provides for a commutation in money, to be paid by persons enrolled or drafted for military service, in lite of actually rendering such military service, be, and the same is hereby, repealed; and hereafter no payment of money shall be accepted or received by the Government to release an enrolled or drafted man from obligation to perform military duty: *Provided*, That hereafter no person shall be received or accepted to serve in the army of the United States as a substitute for any other person liable to military duty and who may have been enrolled or drafted for that purpose.

Which was rejected, on a count-yeas 25, nays 93.

#### Mr. SCHENCK then offered this substitute :

Be il enacted, dc., That so much of the act entitled "An act for enrolling and calling out the national forces, and for other purposes," approved March 3, 1863, and the acts amendatory thereot, as authorizes the discharge of any drafted person from liability to military service by reason of the payment of \$300 for the procuration of a substitute or otherwise, be, and the same is hereby, repealed: *Provided*, That nothing contained in this act shall be construed to alter the provisions of existing laws relative to persons actually furnishing substitutes.

SEC. 2. That in calls for drafts hereafter made under the act for enrolling and calling out the national forces, and the acts in addition to or amendatory thereof, the sume may be made for such term of time as the President shall direct, not exceeding one year.

SEC. 3. That the number of men furnished from any district for the service of the United States beyond and above its quota on calls heretofore made, and the term of service of such men, shall be considered and allowed to said district in calls hereafter made.

SEC. 4 That no person drafted on future calls, or who shall volunteer to fill the same, shall be liable to be again drafted until the existing enrollment shall be exhausted.

Which was rejected—yeas 58, nays 93, as follows:

YEAS-Messrs. Allison, Anderson, Arnold, Ashley, Baxter, Beaman, Blair, Blow, Boyd, Ambrose W. Clark, Cole, Creswell, Dixou, Donnelly, Driggs, Eckiey, Garfield, Higby, Hothkiss, Asabel W. Hobbard, John H. Hubbard, Hulburd, Ingersoll, Jenckes, Julian, Kelley, Knox, Littlejohn, Loan, Marvin, McBride, McClurg, McIndoe, Samael F. Miller, Moorhead, Morrill, Daniel Morris, Amos Myers, Leonard Myers, Norton, Charles O'Neill, Orth, Pike, Pomeroy, William H. Randall, John H. Rice, Schenck, Shannon, Sloan, Smuthers, Spalding, Thayer, Thomas, Tracy, Upson, Ellihu B. Washburne, Wilson, Windom-58.

B. Washburne, Wilson, Windom-58. NArs-Diessrs. Wullam J. Allen, Alley, Ames, Ancona, Baily, Augustus C. Baldwin, John D. Baltwin, Blaine, Bliss, Broomall, Boutwell, Brooks, James S. Brown, William G. Brown, Chanler, Cafroth, Cox, Cravens, Duwes, Dawson, Jemison, Eden, Edgerton, Eldridge, Eliot, English, Fenton, Frinck, Francis Annon, Gooch, Grider, Griswold, Harding, Harrington, Benjamin G Harris, Charles M. Harris, Herrick, Holman, Hutchins, Philip Johnson, William Johnson, Kaliptisch, Francis W. Kellogg, Orlaudo Kellogg, Kernan, Kaliptisch, Francis W. Kellogg, Orlaudo Kellogg, Kernan, Knapp, Law, Lazear, LeBlond, Long, Mallory, Marzy, McAllister, McDouch, McKinney, Middleon, William H. Miller, James R. Morris, Morrison, Nelson, Noble, Odell, John O Neill, Patterson, Pendleton, Perham, Pruyn, Radford, Samuel J. Randall, Alexander H. Rice, Robinson, Rogers, Edward H. Rollins, James S. Rollins, Schs, Scollidd, John B. Sleele, William G. Steele, Stiveus, Stivase, Staud, Sweat, Wadsworth, Ward, William D. Wushburn, Webster, Whaley, Wheeler, Chilton A. White, Joseph W. Winte, Winfield-98.

Mr. STEVENS'S substitute was then rejected veas 24, nays 120, as follows:

YEAS-Messrs. Alley, Ames, Baily, Boutwell, Broomall, William G. Brown, Eliot, English, Ganson, Griswold, Hale, Hotchkiss, Kernan, Lazear, Loan, Orth, Alexander H. Rice, Edward H. Rollins, Ross, Scofield, William G. Sleele, Stevens, Webster, Wilder-24.

Edward H. Rollins, Ross, Scofield, William G. Steele, Stevens, Webster, Wilder-24. Nars-Messrs. Wilder-24. Stevens, Webster, Wilder-24. Nars-Messrs. William J. Allen, Allison, Ancona, Anderson, Arnold, Ashley, Augustus C. Baldwin, John D. Baldwin, Baxter, Beaman, Blair, Büss, Blow, Boyd, Brooks, Chanler, Anubrose W. Clark, Coffroth, Cole, Cox, Cravens, Creswell, Dawes, Dawson, Denison, Dixon, Donnelly, Driggs, Eckley, Eden, Edgerion, Eldridge, Fenton, Fincle, Frank, Garfield, Gooch, Gridler, Harding, Harrington, Benjamin G. Harris, (Charles M. Harris, Herrick, Higby, Holman, Hulburd, Hutchins, Ingersoll, Jenckes, Philip Johnson, William Johnson, Julian, Kallyfleisch, Kelley, Francis W. Kellogg, Orlando Kellogg, Knox, Le Biond, Littlejohn, Long, Mallory, Marcy, Marvin, McBride, McClurg, McDowell, McIndoe, McKinney, Middleton, Samuel F. Miller, William H. Miller, Moorhead, Morril, Daniel Morris, James R. Morris, Morrison, Nelson, Noble, Norton, Odell, Charles O'Neill, John O'Neill, Pendleton, Perham, Pike, Chanles O'Neill, John O'Neill, Pendleton, Perham, Pike, Shannon, Sloan, Snithers, Spalding, John B. Steele, Silles, Strouze, Stant, Sweat, Thayer, Thomas, Tracy, Upson, Yan Valkenburgh, Wadkowrdt, Ward, Ellihu B. Washhurne, William B. Washburn, Whaley, Wheeler, Chiltor A. White, Joseph W. White, Williams, Wilson, Windom, Winfield-120. Mr. SMITHERS then offered a substitute:

The first section became the law, except that in the sub-stitute the bounty was fixed at \$200, \$300, and \$400, payable one half at muster, one fourth at expiration of half the term of service, and one fourth at expiration, and in case of death while in service the residue unpaid to be puid to the widow, or children, or legal representatives. If honorably dis-charged by reason of wounds or sickness incurred in the service, to receive full bounty.

Second section became law, except that "fifty" days was substituted for "sixty."

Third section became third section of the law.

Fourth section became fourth section of the law.

Fifth section became fifth section of the law.

Section six is the sixth section of the law, except that "fifty" per centum reads "one hundred."

Section seven was unchanged.

Section eight is substantially the eighth section of the law.

Section nine is the ninth section of the law.

Which was rejected - yeas 76, nays 77, as follows:

YEAS-Messrs. Alley, Allison, Ames, Anderson, Arnold, Ashley, John D. Baldwin, Baxter, Beaman, Blow, Boutwell, Boyd, Broomall, Ambrose W. Clark, Colb, Cole, Creswell, Boya, Broomail, Ambrose W. Clark, Cobb, Cole, Creswell, Dawes, Dixon, Donnelly, Driggs, Eckley, Eliot, Fenton, Gar-field, Gooch, Higby, Hooper, Hotchkiss, Asahel W. Hubbard, John H. Hubbard, Hulburd, Ingersoll, Jenckes, Julian, Kelley, Francis W. Kellogg, Orlando Kellogg, Knox, Little-john, Loan, Marvin, McBride, McClung, McIndoe, Samuel F. Miller, Moorhead, Morrill, Daniel Morris, Amos Myers, Leonard Myers, Norton, Charles O'Neill, Orth, Patterson, Berham, Borgeror, Willer, H. Raydell, Avander U. Briag, Perham, Pomeroy, William H. Randall, Alexander H. Rice, John II. Rice, Edward II. Rollins, Schenck, Scofield, Shan-non, Sloan, Smithers, Spaulding, Thayer, Tracy, Upson, Ellihu B. Washburne, William B. Washburn, Williams, Ellihu B. Washburne, Willis Wilder, Wilson, Windom-76.

Wilder, Wilson, Windom-76.
NARZ-Messrs. Willium J. Allen, Ancona, Baily, Augustus C. Baldwin, Blair, Bliss, Brooks, James S. Brown, William G. Brown, Chanler, Coffroth, Cox, Cravens, Daveson, Benison, Léden, Eldgerton, Eldridge, English, Finck, Ganson, Grider, Griswold, Hale, Harding, Harrington, Benjamin G. Harris, Charles M. Harris, Herrick, Holman, Hutchins, Philip Johnson, Wm. Johnson, Kallifeisch, Kernan, Kanpp, Law, Lazear, Le Blond, Long, Malloy, Marcy, McAllister, McDowell, McKinney, Midleton, Wm. H. Miller, James R. Morris, Morrison, Nelson, Noble, Odelt, John O'Neill, Pendleton, Fruyn, Radjord, Samuel J. Randall, Robinson Hu, Rogers, Jass. S. Rollins, Ross, John B. Steele, Wim G. Steele, Stevens, Stiles, Stronse, Stuart, Sweat, Thomas, Wadsworth, Ward, Webster, Whaley, Wheeler, Chilton A. White, Joseph W. Widte, Winded-71. W. White, Winfield-77.

June 28-Mr. BLAIR moved to reconsider this vote.

Mr. HOLMAN moved to table the motion, which was lost-yeas 73, nays 85. The motion to reconsider was then agreed to-yeas 83, nays 71, as follows :

71, as Iollows: YEAS-Messrs. Alley, Allison, Ames, Anderson, Arnold, Ashley, John D. Baldwin, Baxter, Beaman, Elair, Blow, Boutwell, Boyd, Broomall, Freeman Clarke, Cobb, Cole, Creswell, Henry Winter Davis, Thomas T. Davis, Dawes, Deming, Dixon, Donnelly, Driggs, Eckley, Eliot, Garfield, Gooch, Higby, Hooper, Hotchkiss, Asahel W. Hubbard, John H. Hubbard, Hulburd, Ingersoll, Jenckes, Julian, Kelley, Francis W. Kellogg, Orlando Kellogg, Knox, Itittlejohn, Loan, Longyear, Marvin, MeBride, McClurg, McIndoe, Samuel F. Miller, Moorhead, Daniel Morris, Amos Myers, Léonard Myers, Norton, Charles O'Neill, Orth, Pat-terson, Perham, Pike, Pomeroy, William H. Randall, Alexander H. Rice, John H. Rice, Edward H. Rollins, Schenck, Socfield, Shannon, Sloan, Smith, Smithers, Spald-Schenck, Scofield, Shannon, Sloan, Smith, Smithers, Spald-ing, Thayer, Thomas, Tracy, Upson, Van Valkenburgh, Ellihu E. Washburne, William B. Washburn, Williams, Wilder, Wilson, Windom-83. NAYS-Messrs. William J. Allen, Ancona, Baily, Augustus

NATS—Messrs. William J. Allen, Ancona, Baily, Augustus C. Baldwin, Blaine, Bliss, James S. Brown, Wm. G. Brown, Chanler, Coffroth, Crucrens, Davoson, Denison, Eiden, Edger-ton, Eldrüdge, English, Finck, Ganson, Grider, Hale, Hard-ing, Harrington, Benjamin G. Harris, Charles M. Harris, Herrick, Holman, Hutchins, Philip Johnson, Wm. Johnson, Kalbfeisch, Knapp, Law, Lazear, Le Blond, Long, Mallory, Marcy, McAllister, McDowell, McKinney, Middleton, Wm. H. Miller, Morrison, Nelson, Noble, Odell, John O'Neill, Pen-dleton, Perry, Pruyn, Radford, Samuel J. Randall, Robinson, James S. Rollins, Ross, John B. Steele, Wm. G. Steele, Stevens, Stiles, Strouse, Elsuart, Sweat, Wadsworth, Ward, Webster, Whaley, Wheeler, Chalton A. White, Joseph W. White, Win-field—71.

The substitute was modified by adding after " service" in the last line of the first section, the words: "in the line of his duty;" and was then agreed to-yeas 81, nays 75, as follows :

then agreed to—yeas 81, nays 75, as follows: YEAS—Messrs. Alley, Allison, Ames, Anderson, Arnold, Ashley, John D. Baldwin, Baxter, Besman, Blair, Blow, Boutwell, Boyd, Broomall, Cobb, Cole, Creswell, Henry Winter Davis, Thomas T. Davie, Dawes, Deming, Dixon, Donnelly, Driggs, Eckley, Eliot, Garfield, Gooch, Higby, Hooper, Hutchkiss, Asahel W. Hubbard, John H. Hubbard, Hubburd, Ingersoll, Jenckes, Julian, Kelley, Francis W. Kellogz, Orlando Kellogz, Knox, Littlejohn, Loan, Long-year, Marvin, McBride, McClurg, McIudoe, Sanuel F. Mil-er, Moorhead, Daulei Morris, Amos Myers, Leonard Myers, Norton, Charles O'Neill, Orth, Patterson, Perham, Pike, Pomeroy, William H. Randall, Alexander H. Rice, John I. Rice, Edward H. Rollins, Schenck, Scofield, Shanon, Sloan, Smith, Smithers, Spalding, Thayer, Tracy, Upson, Van Valkenburgh, Ellihu E. Washburne, William B. Wash-burn, Williams, Wilder, Wilson, Windom—81. burn, Williams, Wilder, Wilson, Windom-81. NATS-Messrs. William J. Allen, Ancona, Baily,

Augus tus C. Baldwin, Blaine, Bliss, James S. Brown, William G. Brown, Chanler, Coffolk. Crovens, Dawson, Denison, Eden, Edgerton, Elaridge, English, Finck, Frank, Ganson, Gri-der, Griswold, Hale, Harding, Harrington, Benjamin G. Harris, Charles M. Harris, Herrick, Holman, Hutchins, Philip Johnson, William Johnson, Kalbfeisch, Kernan, Knapp, Law, Lazear, Le Blond, Long, Mallery, Marcy, McAb lister, McDowell, McKinney, Middleton, William H. Miller, James R. Morris, Morrison, Nolson, Nolle, John O'Neill, Pendleton, Perry, Fruyn, Radford, Samuel J. Randall, Robinson, James S. Rellins, Ross, John B. Steele, William G. Steele, Stevens, Stiles, Strouse, Stuart, Sweat, Thomas, Wadsworth, Ward, Webster, Whaley, Wheeler, Chilton A. While, Joseph W. While, Winfield-75. tus C. Baldwin, Blaine, Bliss, James S. Brown, William G.

Mr. STEVENS moved to add this new section :

That nothing contained in this act shall be construed to alter or in any way affect the law relative to those conscientiously opposed to bearing arms.

Mr. PIKE moved to amend the amendment by adding:

That hereafter, persons between the ages of forty-five and fifty years shall be enrolled and subject to draft in the same manner as persons between the ages of twenty and forty-five.

Which was rejected-yeas 47, navs 102. The YEAS were :

Messrs. Alley, Allison, Ames, Arnold, Ashley, Augustus C. Baldwin, Baxter, Blow, Broomall, James S. Brown, Chan-ler. Freeman Clarke, Coffroth, Creswell, Henry Winter C. Baldwin, Baxter, Blow, Broomall, James S. Brown, Chan-ler, Freeman Clarke, Coffroit, Creswell, Henry Winter l'avis, Dawes, Deming, Lonnelly, Garfield, Gooch, Gris-wold, Hale, Asahel W. Hubbard, John H. Hubbard, Inger-soll, Julian, Littlejohn, Marrin, McBride, McClurg, Moor-head, Daniel Morris, Nelson, Orth, Patterson, Fendleton, Pike, Pomercy, Alexander H. Kice, John H. Rice, Smith, Stevens, Tracy, Van Valkenburgh, Wadsworth, Ellihu B. Washburne Chillon 4 White-47 Stevens, Tracy, Van Valkenburgh Washburne, Chilton A. White-47.

The amendment of Mr. STEVENS was then agreed to-yeas 77, nays 64, as follows :

YEAS-Messrs. Alley, Ames, Ashley, Baily, John D. Bald-win, Baxter, Beaman, Blaine, Blow, Boutwell, Broomall, Chanler, Creswell, Thomas T. Davis, Dawes, Dawson, Dem-Chanler, Creswell, Thomas T. Davis, Dawes, Daueson, Dem-ing, Dixon, Donnelly, Driggs, Eckley, Edgerton, Eliot, Frank, Ganson, Gooch, Grüder, Griswold, Hale, Charles M. Harris, Higby, Hooper, Hotchkiss, Asahel W. Hub-bard, John H. Hubbard, Hatchins, Ingersoll, Jenckes, Julian, Kelley, Francis W. Kellogg, Orlando Kellogg, Ker-nan, Knox, Loan, McAllister, McClurg, Middleton, Moor-head, Daniel Morris, Amos Myers, Leonard Myers, Norton, Charles O'Neill, Perham, Perry, Pike, Radford, Alexander H. Rice, John H. Ruce, Edward H. Rollins, Sloan, Smithers, Spalding, Stevens, Stuart, Sweat, Thayer, Thomas, Upson, Van Valkenburgh, Ellihu B. Washburne, Webster, Wheeler, Wilder, Wilson, Windom-77.

Van Valkenburgh, Ellihu B. Washburne, Webster, Wheeler,
 Wildor, Wilson, Windom-77.
 Nars-Messrs. William J. Allen, Allison, Ancona, Augustus C. Baldwin, Bliss, James S. Brown, Wm. G. Brown,
 Freeman Clarke, Cobb, Colfroth, Cole, Cracens, Denison,
 Eden, Eldridge, English, Finck, Garfield, Harding, Harrington, Benjamin G. Harris, Herrick, Holman, Hulburd, Philip Johnson, Kallgleisch, Knapp, Lav, Le Elmd, Littlejohn,
 Long, Longyear, Marcy, Marvin, McDowell, McIndoe, McKinney, Samuel F. Miller, William H. Miller, James R. Morris, Morrison, Noble, John O'Neill, Ortli, Pendleton, Pomeroy, Samuel J. Randall, Robinson, James S. Rollins, Ross, Schenck, Shannon, Smith, William G. Steeie, Stiles, Strouse, Tracy, Wadsworth, Ward, William B. Washburu, Whaley, Chillon A. While, Joseph W. While, Williams-64.

268

The bill then passed-yeas 82, nays 77, as follows:

The bill then passed—yeas 82, nays 77, as follows: Tras-Messrs. Alley, Allison, Ames, Anderson, Arnold, Ashley, John D. Baldwin, Baxter, Beaman, Blair, Blow, Boutwell, Boyd, Broomall, Freeman Clarke, Cobb, Cole, Cresswell, Thomas T. Davis, Dawes, Deming, Dixon, 'Don-nelly, Driggs, Eckley, Eliot, Farnsworth, Fenton, Garfield, Gooch, Higby, Hooper, Hotchkiss, Asahel W. Hubbard, John H. Hubbard, Hulburd, Ingersoll, Jenckes, Julian, Kiley, Francis W. Kellog, Orlando Kellogg, Knox, Little-john, Loan, Longyear, Marvin, McBride, McClurg, McIndoe, S. F. Miller, Moorhead, Morrill, Daniel Morris, Amos Myers, Iconard Nyers, Norton, Charles O'Neill, O'th, Patterson, Perham, Pike, Pomeroy, Alexander H. Rice, John H. Rice, Edward H. Rollins, Schenck, Shannon, Sloan, Smith, Smithers, Spalding, Thayer, Tracy, Upson, Van Valkenburgh, Elihu B. Washburne, William E. Washburn, Williams, Wilder, Wilson, Windom—82. Mars-Messrs. William J. Allen, Ancona, Baily, Augus tus C. Baldwin, Blaine, Bliss, Brooks, James S. Frown, Willam G. Brown, Chanler, Ogl?rdh, Coz, Cravens, Daw-son, Denison, Eden, Edgerton, Eldridge, English, Finck, Frank, Ganson, Grider, Grisvold, Hale, Harding, Eenjamin G. Harris, Charles M. Harris, Herrick, Hohman, Hutchins, Fhilip Johnson, William J. Millenson, Kublicisch, Kernan, Knapp, Law, Lazear, Le Blond, Long, Mallory, Marcy, Medlikirs, McDouell, McKiraney, Midleton, William H. Miller, James R. Morris, Morrison, Nelson, Noble, Odel, John O'Neill, Pendleton, Perry, Prayn, Radjord, Samuel J. Ramdall, James S. Rollins, Rross, Strouse, Stuart, Sueet, William G. Steele, Stevens, Stiles, Strouse, Stuart, Sueet, William G. Steele, Stevens, Stiles, Strouse, Stuart, Sueet, Thomas, Wadsworth, Ward, Webster, Whaley, Johne S. Seele, William G. Steele, Stevens, Stiles, Strouse, Stuart, Sueet, Thomas, Wadsworth, Ward, Webster, Whaley, Johnet, Steele, Steele, Stevens, Wilte, Joneth, Cane, Amedmeant, and

June 30-The bill having been returned from the Senate with sundry amendments, one of which imposed a five per cent. income duty, the House unanimously directed the return of the bill to the Senate as contravening the first clause of the seventh section of the first article of the Constitution, and as an infringement of the privileges of the House.

July 1-The House considered the Senate substitute for its bill, when several amendments were proposed.

Mr. INGERSOLL moved to add, after the word "soldiers," at the end of the fourth line, "except such States or Territories or parts thereof declared in rebellion."

Which was rejected on a count-yeas 28.

Mr. THOMAS offered this as a new section:

That it shall not be lawful for any of the States to send recruiting agents into other States and Territories to enlist soldiers to be credited to the States that may procure their enlistment; and no State shall be credited with soldiers re-cruited who are not citizens of the State claiming the credit, or foreigners who do not owe allegiance to the United

Which was rejected-yeas 63, nays 63, (the Speaker voting in the negative,) as follows:

YEAS — Messrs. Wm. J. Allen, Ancona, Baily, Blair, Brooks, James S. Brown, Chamler, Coffroth, Creswell, Henry Winter Davis, Dausson, Denison, Eden, Edgerton, Eldridge, English, Farnsworth, Finck, Ganson, Hale, Harding, Benja-English, Farnsworth, Firek, Ganson, Hale, Harding, Benja-min G. Harris, Charles M. Harris, Holman, Hutchins, Philip Johnson, Julian, Kalifleisch, Kernan, Knapp, Lavo, Lazear, Le Blond, Long, Mallory, McAllister, Middleion, William H. Miller, James R. Morris, Morrison, Neison, Noble, Odell, John O'Neill, Pendleton, Pruym, Radford, Robinson, James S. Rollins, Ross, Shannon, John E. Steele, Stiles, Strouse, Stuart, Thomas, Tracy, Wadsworth, Webster, Whaley, Wheeler, Chillon A. White, Winfield-63. NArs-Messrs, Alley, Allison, Ames, Ashley, John D. Baldwin, Baxter, Beaman, Boutwell, Boyd, Broomall, Cobh, Dawes, Deming, Dixon, Driggs, Eckley, Eliot, Fenton, Frank, Garfield, Gooch, Higby, Hooper, Hotchkiss, Asahel W. Hubbard, John H. Hubbard, Hulburd, Ingersoll, Jenckes, Kelley, Littlejohn, Loan, Marcy McBride, Samuel F. Miller,

W. Hubbard, John H. Hubbard, Hulburd, Ingersoll, Jenckes, Kelley, Litticjohn, Loan, Marcy, McBride, Samuel F. Miller, Mornhead, Morrill, Amos Myers, Leonard Myers, Norton, Charles O'Neill, Orth, Patterson, Perham, John H. Rice, Ed-ward H. Rollins, Schenck, Scofield, Sloan, Smithers, Spald-Ing, William G. Steele, Thayer, Upson, Van Valkenburgh, Ellihu B. Washburne, William B. Washburn, Williams, Wilder, Wilson, Windom, Woodbridge-62.

Mr. WILLIAM G. STEELE moved to reconsider this vote.

Mr. E. B. WASHBURNE moved to lay that motion on the table ; which was disagreed toyeas 61, nays 62.

The motion to reconsider was then agreed to-yeas 65, nays 62.

The proposition of Mr. THOMAS was then again rejected-yeas 63, nays 65, as follows :

ugain rejected—yeas 63, nays 65, as follows: YEAS—Messrs. William J. Allen, Ancona, Baily, Blair, Brooks, James S. Broun, Coffroth, Creswell, Henry Winter Davis, Dawson, Denison, Eden, Edgerton, Eldridge, English, Farnsworth, Finck, Ganson, Hale, Harding, Benjamin G. Harris, Charles M. Harris, Holman, Hutchins, Philip John-son, Kallyfeisch, Kernan, Knopp, Law, Lazear, Le Blond, Long, Mallory, McAllister, Middleton, William H. Miller, James R. Morris, Morrison, Nelson, Noble, Odell, John O'Neill, Fendleton, Pruym, Radford, William H. Randall, Robinson, James S. Rollins, Ross, Smith, John B. Steele, Wil-liam G. Steele, Stiles, Strouse, Stuart, Thomas, Tracy, Wads-worth, Webster, Whaley, Wheeler, Chillow A. White, Win-field-G3. Nars-Messrs, Alley, Allison, Ames, Arnold Ashley, John

feld-63. NAVS-Messrs. Alley, Allison, Ames, Arnold, Ashley, John D. Baldwin, Baxter, Beaman, Boutwell, Boyd, Broomall, Cobb, Cole, Dawes, Deming, Dixon, Donnelly, Driggs, Eliot, Fenton, Frank, Garfield, Gooch, Higby, Hooper, Hotchkiss, Asahel W. Hubbard, John H. Hubbard, Hulburd, Julian, Kelley, Littlejohn, Loan, McBride, McClurg, Samuel F. Willer, Moorhead, Morrill, Daniel Morris, Amos Myers, Leonard Myers, Norton, Charles O'Neill, Orth, Patterson, Perham, Alexander H. Rice, John H. Rice, Edward H. Rob, Ins. Schenck, Scoffeld, Shannon, Sloan, Smithers, Spalding, Fernan, Alexander H. Rice, John H. Rice, Loward H. Kol-lins, Schenck, Scofield, Shannon, Sloan, Smithers, Spalding, Thayer, Upson, Van Valkenburgh. Ellihu B. Washburne, William B. Washburn, Williams, Wilder, Wilson, Windom, Woodbridge-65.

Mr. ORTH offered the following amendment; which was agreed to:

That the number of men heretofore furnished by any of the States shall be credited to said State on her quota in any future draft in proportion to the length of time for which said men were furnished.

Mr. GARFIELD offered four new sections, the second, third, and fourth of which are the same as the fifth, sixth, and seventh sections of the act, and the first is as follows :

That any persons resident in Virginia, North Carolina. That any persons resident in Virginia, North Carolina, South Carolina, Georgia, Florida, Alabama, Mississippi, Louisiana, Texas, or Arkansas, who may voluntarily enlist in the military service of the United States for a term of not more than three years, or during the war, or not less than one year, shall be entitled to the benefits and privi-leges of existing laws; and such persons shall be mustered into the regiments or other organizations of whatsoever state there may close to rin the one of colored theorem, choll State they may elect, or, in the case of colored troops, shall be assigned as now provided by law. And the States or subdivisions of States procuring such enlistments shall receive credit for such persons in accordance with the laws in other cases: *Provided*, That such enlistments as are au-thorized in any State, under the provisions of this act, shall only continue until such State shall have been made subject to a call for troops: And provided further, That no or of any State, except those enumerated herein, unless full credit is given to the State to which the enlisted soldier belongs.

Which was agreed to-yeas 69, nays 53, as follows :

Follows.
YEAS—Messrs. Alley, Allison, Ames, Arnold, Ashley, John D. Baldwin, Baxter, Beaman, Boyd, Broomall, Cobb, Cole, Creswell, Henry Winter Davis, Dawes, Deming, Dixon, Donnelly, Driggs, Eckley, Eliot, English, Farnsworth, Fenton, Frank, Garfield, Gooch, Higby, Hotchkiss, Asahel W. Hubbard, John H. Hubbard, Hubburd, Ingersoll, Julian, Verlag, Utitiohn, Laan, McKide, McClurg, Samuel F. W. Huboard, John H. Hubbard, Hubbard, Hubbard, Hydron, Johnson K. Kelley, Littlejohn, Loan, McEride, McClurg, Samuel F. Miller, Moorhead, Morrill, Daniel Morris, Amos Myers, Leonard Myers, Norton, Charles O'Neill, Orth, Patterson, Leonard Myers, Norton, Charles O'Neill, Orth, Patterson, Januard H. Bola Miller, Jeonard Byers, Norton, Charles O'Nent, Orth, Fatterson, Perham, Alexander H. Rice, John H. Rice, Edward H. Rol-lins, Schenck, Scofield, Shannon, Sloan, Smithers, Spalding, Thayer, Upson, Yan Valkenburgh, Ellihu B. Washburne, William B. Washburn, Williams, Wilder, Wilson, Windom, Woodbridge-69.

Woodbridge-69. NAIS-Messrs. William J. Allen, Ancoma, Blair, Brooks, James S. Brown, Chanler, Coffroth, Denison, Eden, Edgertom, Eldridge, Finck, Ganson, Harding, Benjamin G. Harris, Charles M. Harris, Holman, Huichins, Philip Johnson, Kallfleisch, Kernan, Knapp, Law, LeBlond, Long, Malkory, Marcy, McAllister, Middleton, William H. Miller, James R. Morris, Morrison, Nelson, John O'Neill, Pendleton, Pruyn,

Radford, William H. Randall, Robinson, Jomes S. Rollins, Ross, John B. Sleele, Wm. G. Sleele, Stiles, Strouse, Sweat, Thomas, Wadsworth, Webster, Whaley, Wheeler, Chilton A. White, Winfield-53.

The other sections were agreed to without a division.

Mr. HENRY WINTER DAVIS moved that the House concur in the Senate amendments with this amendment:

That no person drafted shall be entitled to be exempted from personal service on payment of commutation money.

That all enrolled persons shall be divided into two classes, one comprising those between eighteen and twenty-five years of age, and the other those over twenty-five and under

forty years of age. That every year during the continuance of the rebellion to be drafted from the the President shall cause 250,000 men to be drafted from the first class, who shall be organized, drilled, and either sent into the field or held as a reserve to meet the exigencies of the service.

That in the event of the service requiring a levy of more than 250,000 men in any year, the residue beyond that number shall be drafted from the second class.

- That prior to any draft, and during the execution of the same, till the requisition is filled, the President shall call for volunteers to fill the same, and is authorized to offer and pay every volunteer \$300 who shall volunteer for three pay every volunteer solo who shall volunteer for distinct years, and proportionably for any shorter period designated by the President, one half at the time of mustering in and the other half on his discharge. That every drafted man not appearing by the assessment

for the internal revenue to have an income of \$300 on whom wife, parent, child, brother, or sister is dependent for support shall be allowed ten dollars a month for the support of every such dependent, payable directly to such dependent or the person charged with the guardianship of any of them : *Provided*, That not more than twenty dollars shall be paid for this purpose in any month on account of any drafted man.

That it shall be the duty of the President to order and execute a drait in each of the States heretofore declared in rebellion, so far as the territory of said States shall be under the military occupation of the United States; and any State may procure volunteers from any of the States declared in rebellion except Tennessee and have them credited to the quotas of the State procuring the same; but all persons residents of any loyal State volunteering in any other loyal State shall be credited to the State of his residence.

Which was rejected-yeas 26, nays 101. The YEAS were:

Messrs. Allison, Arnold, Ashley, Baxter, Broomall, Free-man Clarke, Creswell, Henry Winter Davis, Donnelly, Eck-ley, Garfield, Hooper, Asahel W. Hubbard, John H. Hub-bard, Hinburd, Julian, Longyear, McBride, McClurg, Moorhead, Morrill, Orth, Scofield, Sloan, Stevens, Woodbridge-26.

#### VOTE ON FINAL PASSAGE OF THE ACT OF JULY 4, 1864.

The House then non-concurred in the Senate amendment, and asked a Committee of Conference on the disagreeing votes of the two Houses, whose report, being the law as it stands, was adopted, July 2-yeas 66, nays 55, as follows:

adopted, July 2—yeas 66, nays 55, as follows: YEAS—Messrs. Allison, Ames, Arnold, Ashley, John D. Baldwin, Baxter, Beaman, Blair, Boutwell, Boyd, Cobb, Cole, Creswell, Henry Winter Davis, Dawes, Deming, Dixon, Driggs, Eckley, Eliot, Farnsworth, Fenton, Garfeid, Gooch, Higby, Hooper, Hotchkiss, A. W. Hubbard, John H. Hub-bard, Ingersoll, Jenckes, Julian, Kelley, Littlejohn, Loan, Longyear, McEride, McClurg, Samuel F. Miller, Moorhead, Morrill, Daniel Morris, Amos Myers, Leonard Myers, Nor-ton, Charles O'Neill, Orth, William H. Randall, John H. Rice, Schenck, Shannon, Sloan, Smith, Smithers, Spalding, Tracy, Upson, Van Valkenburgh, Ellihu E. Washburne, William B. Washburn, Whaley, Williams, Wilder, Wilson, Windom, Woodbrige—66. Windom, Woodbrige-66. NAVS-Messrs. William J. Allen, Alley, Ancona, Baily,

NAYS-MCSSTS. William J. Allen, Alley, Ancona, Baily, Blaine, Bliss, Chanler, Coffroth, Cox, Dawson, Denison, Eden, Edgerton, Eldridge, English, Frank, Ganson, Griswold, Ben-jamin G. Harris, Charles M. Harris, Hutchins, Kernan, Knapp, Law, Lazear, Le Blond, Long, Mallory, Marcy, Mid-dteton, William H. Miller, James R. Morris, Morrison, Noble, Odeil, Pattorson, Pendleton, Perham, Prwyn, Samuel J. Randall, Alexander H. Rice, Robinson, Edward H. Rollins, J. S. Rollins, Ross, Scofield, John B. Steele, W. G. Steele, Stevens, Stiles, Thomas, Wadsworth, Webster, Wheeler, Winfield-55.

A CALL FOR 500,000 MEN UNDER THE ACT.

1864, July 18 - The PRESIDENT issued a call for 500,000 men for one year, to be drafted for after September 5, if not furnished before.

RESOLUTIONS CONCERNING ENROLLMENT.

1863, December 22-Mr. PHILIP JOHNSON offered this preamble and resolution :

Whereas the supreme judicial tribunal of the State of Pennsylvania has solemnly decided that the act of Congress approved March 3, 1863, commonly called the conscription act, is in its provisions contrary to and in violation of the Constitution of the United States, and therefore null and void: Therefore.

Resolved, That it is the sworn duty of the executive de-Resolved, That it is the sworn duty of the executive de-partment of the Government to either acquiesce in that decision within that State, or to bring the questions in-volved before the Supreme Court of the United States for final adjudication, to the end that if Congress shall deem such legislation necessary, another bill may be prepared which shall not be subject to constitutional objections.

Which was tabled-yeas 81, nays 43. The NAYS were :

Messrs. Ancona, Augustus C. Baldwin, Bliss, Brooks, Coffroth, Cox, Dawson, Denison, Eden, Edgerton, Eldridge, Finck, Grider, Hall, Harding, Benjamin G. Harris, Charles M. Harris, Holman, Philip Johnson, William Johnson, Le Blond, Long, McDowell, McKinney, Middleton, Morrison, Netson, Noble, John O'Neill, Fenlleton, Perry, Rogers, Ross, Scott, John B. Steele, William G. Steele, Stiles, Strouse, Sweat, Wheeler, Chillon A. While, Joseph W. White, Fernando Wood-43.

1864, January 11-Mr. BROOMALL offered the following preamble and resolution :

Whereas the burden of government should be made to fail as nearly equally as possible upon all parts of the coun-try; and whereas the southern portion of the country has for several years contributed little, either in men or money, towards the support of the Government; and whereas almost the only way to get men from that portion is to take black men; and whereas for every black man enlisted in the South some man in the overburdened North may be exempted from draft:

It is therefore hereby declared to be the sense of this House that the Government should use its most stream ous efforts to procure the voluntary enlistment of persons claimed as slaves in the rebel territory, by giving them the full bounty and pay of other soldiers, and by guaranteeing their freedom, at once, upon enlistment.

Mr. Cox moved to lay them upon the table; which was disagreed to—yeas 61, nays 74; and they were then referred to the Military Committee.

1864, January 15-Mr. Ancona offered this resolution, which was laid over under the rule:

Whereas the act of Congress approved March 3, 1863, commonly called the conscription law, is oppressive, unjust, and unconstitutional; because, 1st, it takes from the States the control of their own militia; 2d, it subjects the rights of the States and the liberties of the people to the unlim-ited power of the Federal Government; 3d, it is calculated to create and build up a central military despotism which may be used for the worst and most dangerous purposes; may be used for the worst and most daligerous purposes; 4th, it falsely imputes the crime of deservion to every man whose name is drawn in the "lottery of death" and who fails to join the army, and subjects him to trial, condemna-tion, and capital punishment, without a jury of his peers, contrary to the fundamental law of the land: Therefore, *Be it resolved*, That the Committee on Military Affairs be instructed to bring in a bill for the unconditional repeal of soid act of Congress, and substitute in its place some con-stitutional and just mode of raising armies for the service of the United States.

of the United States.

1864, Feb. 1-Mr. ELDRIDGE offered the following preamble and resolution :

Whereas all conscription or other forced service of the Whereas all conscription of other forced service of the citizent to the State is contrary to the genius and principles of republican government and opposed to the principles of self-government, which is the true basis of the American republic; and whereas the laws for conscripting or drafting citizens into the military service of the United States have thus far proved, if not an entire failure, at least ineffectual for the averaging the Government the necessary number for the supplying to the Government the necessary number of men requisite for the military service in putting down the

rebellion ; and whereas the principles of equity and justice require in a government like ours, founded on the will of the majority, that the burdens of maintaining and preserving it should fail alike and equally upon all and every of the citizens, the rich as well as the poor, in proportion to their ability to bear the same; and whereas the military is a profession to which men are called as well from the in-ducements of personal gain and family advantage as from motives of patriotism and hopes of future fame: Therefore, *Resolved*. That the Committee on Military Affairs be, and they are hereby, instructed to examine and inquire imme-diately into the propriety and expediency of repealing or suspending, so far as any future or further draft is con-cerned, all acts and parts of acts authorizing or empower-ing the conscripting or drafting of, or in any way forcing. require in a government like ours, founded on the will of the

cerned, all acts and parts of acts authorizing or empower-ing the conscripting or drafting of, or in any way forcing, the citizen iuto the military service of the country, either in putting down rebellion or otherwise; and in lieu thereof providing by law for, and authorizing the President of the United States from time to time, and as he may deem it expedient and necessary, to offer the payment of such sum or sums of money for volunteers in bounties or monthly payments, or otherwise, as may be best to induce enlist-ments and secure such moneys to the soldier and his family, and a smill server birt ac more, and inst auth prones much be and as will secure just so many and just such men as may be requisite or necessary to put down the rebellion and restore the supremacy of the Constitution; and that said committee do report by bill.

Which were laid upon the table, on Mr. STEVENS'S motion - yeas 84, nays 42. The NAYS were:

Masses, James C. Allen, William J. Allen, Ancona, Au-gustus C. Baldwin, Bliss, Coffroth. Dawson, Denison, Eden, Edgerton, Eldridge, Finck, Hall, Harrington, Knapp, Law, Lazear, Long, Marcy, McDowell, McKinney, Middleton, Wm. H. Miller, Junes R. Morris, Morrison, Noble, John O'Neill, Pendleton, Pery, Pruyn, Samuel J. Randall, Robinson, Rogers, Ross, Scott, John B. Steele, Stiles, Strouse, Sturi, Sweat, Chilton A. White, Joseph W. White–42.

Of the Democrats, Messrs. Grider, Hutchins, and Yeaman voted aye.

1864, February 8-Mr. ELDRIDGE offered this resolution, which was laid over under the rule:

Resolved, That the Secretary of War be, and he is here-by, required to furnish to this House information as to the amount of moneys received up to this time for commuta-tion by drafted men; also what disposition has been made of said moneys. If substitutes have been purchased for drafted men, how many; where and who have been pro-cured as such substitutes; what sum has been paid for each, and whether for white or black, and how much for each.

April 4—The House considered the resolution, when Mr. STEVENS moved that it lie upon the table; which was agreed to-yeas 60, nays 46. The NAYS were:

Messrs. James C. Allen, Ancona, Augustus C. Baldwin, Messrs. James C. Allen, Ancona, Augustus C. Baldwin, Bliss, Brooks, James S. Brown, Chanler, Clay, Cox, Cravens, Dawson, Denison, Eden, Eldridge, English, Finck, Grider, Griswold, Harrington, Benjamin G. Harris, Herrick, Hol-man, Philip Johnson, Kallfleisch, King, Luw, Lazear, Long, Mallory, Marcy, McKinney, Middleton, James R. Morris, Morrison, Nelson, Odell, John O Neill, Price, Pruyn, Robin-son, John B. Steele, Strouse, Wheeler, Chilton A. White, Win-field, Yeaman-46.

1864, Feb. 1-Mr. GRINNELL offered the following preamble and resolution :

Whereas the war policy of the Government having brought into the military service as soldiers and laborers free colored men and persons claimed to be held by rebels, who have rendered invaluable service to the army; and whereas the more extended employment and enlistment of colored persons will be a relief to our northern soldiers, colored persons will be a reflet to our northern soldiers, unacclimated and unused to manual labor, and lessen the number to be taken from their homes and from the indus-trial pursuits in the United States, where there is now an unusual demand for labor: Therefore, *Resolved*, That a more vigorous policy to enlist, at an early day and in larger numbers, in our army persons of African descent would meet the approbation of this Humes.

African House.

Mr. STILES moved to lay them on the table, but the House refused-yeas 49, nays 76; and the resolution was then passed - yeas 80, nays 46.

BILL FOR THE PUNISHMENT OF GUERRILLAS.

June 6-The Committee on Military Affairs reported this bill to punish guerrillas:

Be il enacted, &c., That the provisions of the twenty-first section of an act entitled "An act for enrolling and calling out the national forces, and for other purposes," approved March 3, 1863, shall apply as well to the sentences of miliary commissions as to those of courts martial; and here-after the commanding general in the field, or the com-mander of the department, as the case may be, shall have power to carry into execution all sentences against guerrillas. for robbery, arson, burglary, rape, assault with intent to commit rape, and for violation of the laws and customs of war, as well as sentences against spies, mutineers, deserters, and murderers.

and murderers. SEC, 2. That every officer authorized to order a general court-martial shall have power to pardon or mitigate any punishment ordered by such court, including that of con-finement in the penitentiary, except the sentence of death or of cashiering or dismissing an officer, which sentences it shall be competent during the continuance of the present rebelliou for the general commanding the army in the field, or the department commander, as the case may be, to remit or mitigate; and the fifth section of the act approved July 17, 1862, chapter 201, be, and the same is hereby, repealed, so far as it relates to sentences of imprisonment in the ponitentiary.

Mr. Long moved that the bill be tabled; which was disagreed to.

After further proceedings,

Mr. ELDRIDGE moved that it be tabled; which was disagreed to-yeas 35, nays 67.

And the bill passed-yeas 72, nays 37. The NAYS were:

Messes. James C. Allen, Ancona, Augustus C. Baldwin, Bliss, Coffroth, Cravens, Henry Winter Davis, Dawson, Den-ison, Eden, Edgerton, Eldridge, Finck, Grider, Harding, Horrington, Charles M. Harris, Hutchins, King, Knapp, Le Blond, Long, Mallory, Marcy, McDowell, Morrison, Noble, Fendleton, Perry, Robinson, Royers, Ross, Strouse, Yoorhees, Wadsworth, Chilton A. White, Joseph W. Wiette-31.

Of the Democrats, Messrs. Baily and Griswold voted aye.

#### IN SENATE.

June 14-The Military Committee reported the House bill for the more speedy punishment of guerrillas; which was debated.

June 30-Mr. HENDRICKS offered this proviso to the first section :

Provided, That the term "guerrillas" herein contained shall not be held to include persons employed in the authorized military service of the enemy.

Which was agreed to, and the bill passed.

The House concurred, and the bill became a law

#### INCREASED PAY OF SOLDIERS.

First Session, Thirty-Seventh Congress.

1861, August 6-This section was passed by both Houses:

Be it enacted, &c., That the pay of the privates in the regular army and volunteers in the service of the United States be thirteen dollars per month for three years from and after the passage of this act, and until otherwise ordered.

#### First Session, Thirty-Eighth Congress.

#### IN HOUSE.

May 3-The Senate bill to increase the pay of soldiers was reported from the Committee on Military Affairs, making the pay, from May 1, of "all soldiers in the military service sixteen dollars per month ;" corporals, eighteen dollars ; sergeants, twenty dollars.

Mr. HOLMAN sought the floor to move an amendment that white soldiers should have twenty dollars per month, and colored eighteen dollars

Mr. Cox sought to move an amendment to

increase the pay of soldiers and non-commis- | respect to all persons arrested and detained, and in respect sioned officers two dollars per month.

Mr. DAWSON, to move to make it twenty dollars in the present currency.

But Mr. SCHENCK declined to yield the floor. After a brief explanation the bill passedyeas 135, nays none.

# IN SENATE.

May 11-The Senate Military Committee reported the bill with amendments, the principal one of which fixed the rates as now existing and stated in the summary of "our Military Legislation," which was agreed to, without While the bill was division in either House. pending in the Senate,

May 17-Mr. RICHARDSON offered the following:

That from and after the first day of May, 1864, the officers, non-commissioned officers, musicians, and privates in the regular army and volunteers and drafted forces in the service of the United States shall be paid in gold : *Provided*, That said officers, non-commissioned officers, musicians, and That shall once s, not commission of the statutes, and privates may be paid in Treasury notes or paper money when the Government cannot pay in gold. If not paid in gold, they shall be paid in paper an amount equal to the value of gold at the time of payment.

Which was rejected-yeas 6, nays 23, as follows:

YEAS-Messrs. Buckalew, Davis, Hendricks, Lane of Indiana, Powell, Richardson-

NATS-Messers. Anthony, Chandler, Clark, Collamer, Con-ness, Dixon, Doolittle, Foot, Foster, Grimes, Harlan, Harris, Henderson, Howe, Johnson, Lane of Kansas, Morgan, Morrill, Ramsey, Sumner, Ten Eyck, Van Winkle, Wilson-23.

Mr. Powell offered this amendment:

Provided, That the provisions of this act shall not apply to colored soldiers.

Which was rejected-yeas 5, nays 26, as follows:

YEAS--Messrs. Buckalew, Davis, Hendricks, Powell, Rich-

ardson-5. NAYS-Messrs. Anthony, Chandler, Clark, Collamer, Conness, Dixon, Doolittle, Foot, Foster, Grimes, Harlan, Harris, Henderson, Howard, Howe, Johnson, Lane of Indiana, Lane Morgan, Morrill, Pomeroy, Ramsey, Sumner, of Kansas, Morgan, Morrill, Pomer Ten Eyck, Van Winkle, Wilson-26.

#### ORDERS ISSUED BY THE SECRETARY OF WAR IN ENFORCING THE DRAFT.

The Secretary of War, in executing the draft of 1862, issued these orders:

WAR DEPARTMENT, WASHINGTON CITY, D. C.

August 8, 1862.

Orders to prevent the evasion of military duty and for the

Suppression of disloyal practices. First. By direction of the President of the United States, it is hereby ordered that, until further order, no citized liable to be drafted into the militia shall be allowed to go to a foreign country, and all marshals, deputy marshals, and militury discuss of the United States are discated and and military officers of the United States, are directed, and all police anthorities, especially at the ports of the United States on the seaboard and on the frontier, are requested to see that this order is faithfully carried into effect. And they are hereby authorized and directed to arrest and detain any person or persons about to depart from the United States in viclation of this order, and report to Major L. C. Turner, Judge Advocate, at Washington city, for further instruction respecting the person or persons so arrested and detained.

Second. Any person liable to draft who shall absent himself from his county or State before such draft is made, will be arrested by any provost marshal, or other United States or State officer, wherever he may be found within the jurisdiction of the United States, and conveyed to the nearest military post or depot, and placed on military duty for the term of the draft; and the expenses of his own arrest and conveyance to such post or depot, and also the sum of five dollars as a reward to the officer who shall make such arrest, shall be deducted from his pay.

Third. The writ of habeas corpus is hereby suspended in

to all persons arrested for disloyal practices. (Signed) EDWIN M. STANTON (Signed)

Secretary of War.

WAR DEPARTMENT, Washington City, Aug. 8. Ordered—First. That all United States Marshals, and superintendents, and chiefs of police of any town, city or district, be and they are hereby authorized and directed to arrest and imprison any person or persons who may be engaged, by any act of speech or writing, in discouraging volunteer enlistments, or in any way giving aid and comfort to the enemy, or in any other disloyal practice against the United States.

Second. That immediate report be made to Major L. C. Turner, Judge Advocate, in order that such persons may be tried before a military commission.

Third. The expenses of such arrest and imprisonment will be certified to the chief clerk of the War Department for scttlement and payment.

EDWIN M. STANTON, Secretary of War.

WASHINGTON, August 14. The following was issued to-day from the War Department:

ADDITIONAL REGULATIONS FOR THE ENROLLMENT AND DRAFT OF THE MILITIA.

Ordered-Eighth-That in filling all requisitions for militia the quotas of the several States will be apportioned by the Governors among the several counties, and where practicable among the subdivisions of counties, so that allowance shall be made to such counties and subdivisions for all volunteers heretofore furnished by them and mustered into the service of the United States, and whose stipulated terms of service shall not have expired. (Signed) E. M. STANTON,

Secretary of War.

WAR DEPARTMENT, WASHINGTON, September 7, 1862.

INSTRUCTIONS TO UNITED STATES MARSHALS, MILITARY COM-MANDANTS, PROVOST MARSHALS, POLICE OFFICERS, SHERIFFS, ETC.

The quota of volunteers and enrollment of militia having been completed in the several States, the necessity for a stringent enforcement of the orders of the War Department in respect to volunteering and drafting no longer exists, Arrests for violation of these orders and for disloyal practices will hereafter be made only upon express warrant or by direction of the military commanders or Governor of the State in which such arrests may be made. And restrictions upon travel imposed by these orders are rescinded. L. C. TURNER

Judge Advocate.

#### THE PRESIDENT'S PROCLAMATION RELATIVE TO ALIENS.

1863, May 8-The PRESIDENT issued a proclamation relative to an exemption from the draft on the plea of alienage, in which he declares:

Now, therefore, to avoid all misapprehensions concerning the liability of persons concerned to perform the service required by such enactment, and to give it full effect, I do hereby order and proclaim that no plea of alicnage will be received, or allowed to exempt from the obligations imposed by the aforesaid act of Congress any person of foreign birth who shall have declared on oath his intention to become a citizen of the United States, under the laws thereof, and who shall be found within the United States at any time during the continuance of the present insurrection and re-bellion, at or after the expiration of the period of sixty-five during the data of this present insurable works. days from the date of this proclamation; nor shall any such plea of alienage be allowed in favor of any such person who has so, as aforesaid, declared his intention to become a citiare so, as an other states, and shall have exercised at any time the right of suffrage, or any other political franchise within the United States, under the laws thereof, or under the laws of any of the several States.

JUDICIAL DECISIONS UPON THE CONSTITUTIONALITY OF THE CONSCRIPTION ACT.

In the Circuit Court of the United States for the eastern district of Pennsylvania, Judge Cadwalader delivered an opinion in September, 1863, a condensation of which is subjoined

The powers conferred by the Constitution upon Congress, o raise and support armies and make rules for their gov-mment, are distinct from the powers which are conferred rnment, are distinct from the powers which are conferred in it as to the militia of the respective States. Until the act in question, the national armies had been raised by vol-mary emistment. The system of enrollment and draft rad long been matured as to the militia of the States. But intil the summer of 1862 the utmost penalty for not serv-ng when drafted from such militia for the service of the United States had been pecuniary, with limited imprison-nent for non-payment. The act of Congress of 17th July, 1862, authorized impressment into the military service of the United States of these persons drafted from the military of the act, who, when erdered to attend at the place of inder that act, who, when ordered to attend at the place of

nuster, disobeyed. The specific power of impressment had not been previ-nusly conferred. But, under the former system, though the usly conterred. But, under the former system, though the fine for not serving had, when received, been considered an aquivalent for service, the payment had nevertheless been mforced, or the penalty of imprisonment inflicted by contrs-martial when the money was not otherwise collected. The constitutionality of this former jurisdiction of courts-martial may be considered as established, (5 Wheaton I.) It would not have been constitutional if discheimen attend tal may be considered as established, (5 Wheaton 1.) It would not have been constitutional if disobedience to attend at a place of muster had not been a military offence. Con-press, unless it had the power of absolutely subjecting a inafted person to military rule from the time of the draft, yould not have thus made his disobedience before he was mustered into service a military offence. The act of Congress of 1795, which fixed the time of mirval at the place of rendezvons as the period of the com-mencement of the military service, might constitutionally, a the onjuin of the Supreme Court have made the time

in the opinion of the Supreme Court, have made the time of draft the period. (5 Wheaton, 17, 18, 30, and see pp. 36, 37, 56, 64, 65.) The constitutionality of the act of 17th July, 1862, when the question was considered here in March last, 51 CoUM according the former to be actabilited by in McCall's case, appeared therefore to be established by authority. If the question had been thought an epen one, the same view of the effect of the Constitution would have been taken.

The act of 3d of March, 1863, has adopted a like system on an extended scale, for the purpose of raising national armies independently of the militia of the States. Under the former laws which have been mentioned, a question such as that new under consideration could net arise. The question under these laws could only have been that of a military court's exercise of jurisdiction over a person who having been lawfully drafted already owed military service. There could not have been any dispute that the primary question whether he had been lawfully drafted or was lable to serve, was open to decision by the ordinary tribu-nals under a writ of *habeas corpus*. Here, however, the question is whether a military commission can so decide the original question of liability to serve as absolutely as to deprive all other tribunals of cognizance of it. The enactments of the law in question are not so ar-ranged that its provisions for the preparatory emoliment,

ranged that its provisions for the preparatory enrollment, and those for the draft, are always separated. They must however, be kept distinct when they are considered with reference to the Constitution. The mest unlimited system of mere enrollment could not be constitutionally objectionable.

But a system of drafting might be arbitrary and latitu-dinarian to such an extent as to encroach upon constitu-tional rights. \* \* \* \* \* The constitutional authority to enact the law which is under consideration was derived exclusively from the power to raise armies. It cannot be confers to make all laws necessary and proper for carrying the powers delegated, this one included, into execution.

After citing the provision of the act, the judge says :

This review of the principal enactment of the law suffices to indicate its general purposes. The organization of armies under it is to cease on the termination of the civil war, for whose exigencies it provides; and the term of service of those drafted under it cannot exceed three years, though the war sheald continue longer. Such limitations of the time would have prevented the compulsory requirement of military service from being unconstitutional, though it had included every able-bodied male inhabitant.

[From the Illinois State Journal of June 17, 1864.]

THE CONSTITUTIONALITY OF THE ENROLLMENT ACT AFFIRMED.

In the United States Circuit Court, Judges David Davis and S. H. Treat upon the bench, an important decision was rendered on the 15th instant. It was on a motion to quash the indictment in the case of "The United States agt. John Graham and others," for resistance to the enrollment in 18

rom the New York *Tribune* of September 11, 1863: The powers conferred by the Constitution upon Congress, The contract and existent animing the construction of the constitution empowering Congress to raise and equip armies. This is the first decision under the enrollment act rendered in this State, and is therefore important. The motion to quash the indictment was granted, on the ground that the unit hence is of the restingue and the state. that the punishment of the particular class of offenders charged was not specifically provided for in the act of July, 1863. It was admitted, however, that the act as amended by the act of February, 1864, covers the whole point, and that, as the law now stands, resistance to the enrollment is beby the act of rest and the prover the provided of the prover the pr liable to the severest penalties. The decision was ren-dered by Judge Treat, Judge Davis concurring.

THE TWO DECISIONS OF THE SUPREME COURT OF PENNSYLVANIA.

A majority of the judges of the Supreme Court of Pennsylvania, as constituted in November, 1863, pronounced the enrollment law unconstitutional. Chief Justice Lowrie and Justices George W. Woodward and James Thompson concurred in this judgment, and Justices William Strong and John M. Read disscuted. These points are covered by the decision :

1. The Constitution of the United States recognizes only two sorts of military land forces, viz., "the militia," and the "regular or standing army.

2. The conscription act of March 3, 1863, is not founded on that clause of the Constitution which provides for call-ing forth the militia, because the persons drafted under the act are not to be armed, organized, and disciplined under the militia law, nor are they called forth under State offi-

 There is no power given to recruit the regular Army by forced levies. This cau only be done by voluntary en-betwarts. listments.

4. The mode of "raising armics" by forced recruiting for the suppression of rebellion or insurrection is not authorized by the Constitution, because such cases are expressly provided for by the power therein given for calling out the dormant forces, or militia.

5. The Constitution authorizes levies of the "militia of the States" in its organized form in cases of rebellion and invasion, but in no other case or mode than is therein previded.

6. The mode of coercion provided for this purpose by the act of March 3, 1862, is unconstitutional, because

(1.) It is incompatible with the provisions of the Constitution relative to the militia.

(2.) It exhausts the militia force of the several States, which existed as an institution before the formation of the Federal Government, and was not only not granted away but expressly reserved at the formation of the Constitution; annuls the remedy for insurrection expressly provided by the Constitution, and substitutes a new one not therein provided for; and converts into national forces as part of the regular army of the General Government the whole militia force of the States, not on the contingency therein unredded for particular the form the universe whole buy the deprovided for nor in the form therein prescribed, but entirely. irrespective thereof.

(3.) It incorporates into this new national force every civil officer of the State except the Governor, and every officer of its social institutions and military organization within the prescribed age, thus subjecting the civil, social, and military organizations of the States to the Federal power to "raise armies."

(4.) It provides for a thorough fusion of the army and the The second secon

(5.) It subjects the citizen to the rules and articles of war before he is in "actual service," and proposes to effect this purpose by merely drawing his name from a wheel and serving notice of that fact upon him.

The key-note of Judge Woodward's opinion is this paragraph :

The great vice of the conscript law is, that it is founded on an assumption that Congress may take away, not the State rights of the citizen, but the security and foundation of his State rights. And how long is civil liberty expected' to last, after the securities of civil liberty are destroyed? The Constitution of the United States committed the liber-tice of the citizen is the securities of civil states and the liberties of the citizen in part to the Federal Government, but, expressly reserved to the States, and the people of the. States, all it did not delegate. It gave the General Government a standing army, but left to the States their militia. Its purposes in all this balancing of powers were wise and good, but this legislation disregards these distinctions and upturns the whole system of government when it converts the State militia into "National forces," and claims to use and govern them as such.

Chief Justice Lowrie and Justice Thompson elaborate the same point.

1864, Jan. 16-The same court, then differently constituted by the defeat of Chief Justice Lowrie at the October election and the choice of Daniel Agnew, directed the orders granted in the cases heard in November to be vacated, and affirming the constitutionality of the enrollment act, overruled the motions for injunctions to restrain certain provost marshals from proceeding with the enrollment and draft. Messrs. Strong, Read and Agnew constituted the majority, and Woodward and Thompson the minority.

Justice Agnew closed his opinion with stating these conclusions :

The constitutional authority to use the national forces creates a corresponding duty to provide a number ade-quate to the necessity. The duty is vital and essential, failing back on the fundamental right of self-preservation, and the powers expressed to declare war, raise armies, maintain navies, and provide for the common defence. Power and duty now go hand in hand with the extremity until every available man in the nation is called into service, if the emergency requires it, and of this there can be no judge but Congress.

They may proceed, therefore, to the exhaustion of the whole element from which the State draws its militia, for the people, under the two powers, are the same ; while the supremacy of the national power, provided in section 2 of article 6, necessarily draws to itself the whole number, if required by the exigency, to the exclusion of the State power.

And in reason why should a major power be restricted by a minor? The power to raise armies comprehends for its purposes the whole scope of the purposes of armies, while the authority to call out the militia is confined to the enumerated three.

But it is a mistake in fact to say this case exhausts the militia. It enrolls probably all; for how can any be draft-ed without all be known? But the draft is confined to so ed without all be known? But the draft is confined to so many as are needed for the emergency, while the others remain in the militia. And if you deny the power to repeat the draft, what is that but to say your force shall not increase with the necessity?

Nor is it true that the enrollment under this law exhausts the militia. Neither the law of Congress, nor the laws of the finitual. Neither the taw of congress, hot the taws of the States, so far as we know them, have corrolled all able bodied men capable of militia duty. A wide margin yet exists in the law of the nation; but we do hear of this margin licing written all over in the second States.

As to the objection to the 13th section, providing the punishment of desertion for those who fail to appear, it is only necessary to say, we cannot presume the complainant will be guilty of failing to perform his legal duty subse-quent to the draft, when he finds the law valid which drafts him. He asks us to relieve him from the draft, not from a military trial for misconduct. Whenever he chooses to incur the proposed penalty for disobeying a valid law, it will be in time for the proper tribunal to arrest an illegal mode of punishment.

The question of jurisdiction is unnecessary to a decision. The point is too important, the cases too numerous, and the labor too great. It should therefore be left for a decision when it shall have to be met.

For all these reasons I concur in rescinding the order for a preliminary injunction.

#### GOVERNOR SEYMOUR AND THE DRAFT.

The result of the repeated conferences between the State authorities of New York and the War Department, was to ascertain that the State had been called to furnish 12,533 men more than her quota; and the House of Representatives of that State, during its late session, passed this resolution :

the attention of the General Government at Washington to the errors in the apportionment of the quota of this State under the enrollment act of 3d March, 1863, and for his prompt and efficient efforts in procuring a correction of the same.

GENERAL M'CLELLAN URGED A DRAFT IN 1861.

Soon after General McClellan assumed command of the army, succeeding General Scott, he wrote this letter to the President :

WASHINGTON, August 20, 1861. SIB: I have just received the inclosed dispatch in cipher Sile: I have just received the inclosed dispatch in cybes Colonel Marcy knows what he says, and is of the coolest judgment. I recommend that the Secretary of War ascer-tain at once by telegram how the enrollment proceeding New York and elsewhere, and that, if it is not proceeding with great rapidity, drafts to be made at once. We must With great rapidity, drafts to be made at once. We mu have men without delay. Respectfully your obedient servant, GEORGE B. MCCLELLAN, Maj. Gen. U. S. A.

DISPATCH FROM COL. R. B. MARCY TO GENERAL MCCLELLAN.

New York, August 20, 1861. I urge upon you to make a positive and unconditional de-mand for an immediate draft of the additional troops you require. Men will not voluuteer now, and drafting is the only successful plan. The people will applaud such a course, rely upon it. I will be in Washington to-morrow.

R. B. MARCY.

# Colored Soldiers.

### Second Session, Thirty-Seventh Congress.

1862, July 17-These provisions became law:

That the President be, and he is hereby, authorized to receive into the service of the United States, for the purpose of constructing intronchments, or performing camp ser-vice, or any other labor, or any military or naval service for which they may be found competent, persons of African descent; and such persons shall be enrolled and organized under such regulations, not inconsistent with the Constitution and laws, as the President may prescribe.

That when any man or boy of African descent, who by the laws of any State shall owe service or labor to any person who, during the present rebellion, has levied war or has borne arms against the United States, or adhered to their enemies by giving them aid and confort, shall render any such service as is provided for in this act, he, his mother, and his wife and children shall forever thereafter be free, and has whe state cliniter is not notice intercater be need, any law, usage, or custom whatsoever to the contrary not-withstanding: *Provided*, That the mother, wife and child-ren of such man or boy of African descent shall not be made free by the operation of this act except where such mother, wife or children owe service or labor to some person who, during the present rebellion, has borne arms against the United States or adhered to their enemies by giving them aid and comfort. That the expenses incurred to carry this act into effect

shall be paid out of the general appropriation for the army and volunteers.

That all persons who have been or shall be hereafter enrolled in the service of the United States under this act shall receive the pay and rations now allowed by law to soldiers, according to their respective grudes: *Provided*, That persons of African descent, who under this law shall be employed, shall receive ten dollars per month and one ration, three dollars of which monthly pay may be in clothing.

1863, May 3-Section 10 of the Engineer bill provides that the President of the United States be, and he is hereby, authorized to cause to be enlisted, for each cook, two under-cooks, of African descent, who shall receive for their full compensation ten dollars per month, and one ration per day-three dollars of said monthly pay may be in clothing.

1863, March 3-Includes all able-bodied males.

1864, Feb. 24-(Section 24) directs that all able-bodied male colored persons between twenty and forty-five, resident in the United States, shall be enrolled and form part of the national forces. When a slave of a loval mas-Resolved, That the thanks of this House he and are here-oy tendered to his Excellency Governor Seymonr for calling ter shall be drafted, his master shall have a

certificate and the bounty of \$100 and the slave | ing that when any man or boy of African shall be free. [For whole section, and for other votes on colored soldiers, see other pages.]

#### IN SENATE.

Pending the consideration of the bill of 1862, 1862, July 10-Mr. DAVIS, of Kentucky, moved to strike out the words "or any military or naval service for which they may be found competent;" which was rejected-yeas 11, nays 27, as follows :

YEAS-Messrs. Carlile, Cowan, Davis, Henderson, Ken-nedy, Powell, Saulsbury, Stark, Willey, Wilson of Missouri,

nedy, Fowel, Sumson y, Stars, Law, Chandler, Clark, Col-Wright-1. NAYS-Messrs. Anthony, Browning, Chandler, Clark, Col-lamer, Doolittle, Fessenden, Foot, Grimes, Hale, Harlan, Harris, Howard, King, Lane of Indiana, Lane of Kansas, Morrill, Pomeroy, *Rice*, Sherman, Simmons, Sumner, Ten Eyck, Trumbull, Wade, Wilkinson, Wilson of Massachu-Eyck, Trisetts-27.

Mr. HENDERSON moved to limit the section to "free" persons of African descent, and to "such persons of African descent as may owe service or labor to persons engaged in the rebellion ;" which was negatived-yeas 13, nays 22, as follows :

YEAS-MOSSTS. Anthony, Browning, Cowan, Davis, Hender-son, Lane of Indiana, McDougall, Powell, Rice, Stark, Wil-loy, Wilson of Missouri, Wright-13. NAYS-MOSSTS. Chandler, Clark, Collamer, Doolittle, Fes-senden, Foot, Grimes, Hale, Harlan, Harris, Howard, King, Lane of Kansas, Morrill, Pomeroy, Sherman, Simmons, Sumner, Trumbull, Wilkinson, Wilmot, Wilson of Massa-chusetts-22.

Mr. HENDERSON moved to add the following :

Provided, That all loyal persons entitled to the service or labor of such persons, according to the laws of the State in which the owner of such slave may reside, employed under the provisions of this act, shall be compensated for the loss of such service.

Which was agreed to-yeas 20, nays 17, as follows:

YEAS-Messrs. Anthony, Browning, Collamer, Cowan, Davis, Doolittle, Foot, Foster, Harlan, Henderson, Howe, Lanc of Indiana, McDougall, Powell, Simmons, Stark, Ten Eyck, Willer, Wilson of Missouri, Wright-20. Navs-Messrs. Chandler, Clark, Fessenden, Grimes, Hale, Harris, Howard, King, Lane of Kansas, Morrill, Pomeroy, Sherman, Sunner, Trumbull, Wilkinson, Wilmot, Wilson of Massachusetts-17.

Mr. LANE, of Kansas, moved to amend the section by "directing" as well as "authorizing" the President to receive into the service, &c., which was rejected.

Mr. SHERMAN moved to amend by making the section read-

That when any man or boy of African descent, who by the laws of any State shall owe service or labor to any person who, during the present rebellion, has levied war or borne urms against the United States, or adhered to their enemics by giving them aid and comfort, shall render any such service as is provided for in the first section of this act, he, his mother, and his wife and children shall forever thereafter be free, any law, usage, or custom whatsoever to the contrary notwithstanding.

Which was agreed to-yeas 22, nays 16, as follows:

YEAS — Messrs. Anthony, Browning, Collamer, Cowan, Davis, Doolittle, Foster, Harris, Henderson, Howard, Howe, Kennedy, Lane of Indiana, McDougall, Rice, Sherman Simmons, Stark, Ten Eyck, Willey, Wilson of Missouri, Simmons, S Wright-22.

NATS-Messrs. Chandler, Clark, Foot, Grimes, Hale, Har-lan, King, Lane of Kansas, Morrill, Pomeroy, Sumner, Trumbull, Wade, Wilkinson, Wilmot, Wilson of Mass-16.

The bill was then dropped, and, July 14,

Mr. WILSON reported a bill to amend the act calling forth the militia, containing the pro-visions which were finally passed, and provid-

descent shall render any such service, he, his mother, and his wife and children are forever thereafter to be free.

Mr. BROWNING moved to strike out the clause liberating the mother, wife, and children; which was negatived--yeas 17, nays 20:

TEAS-MESSIG. Browning, Collamer, Cowan, Davis, Doo-little, Foster, Harris, Henderson, Lane of Indiana, Powell, Rice, Saulsbury, Sherman, Skark, Ten Eyck, Wilson of Missouri, Wright-17.
 NATS-Messis. Chandler, Clark, Fessenden, Foot, Grimes, Hale, Harlan, Howard, Howe, King, Lane of Kansas, Mor-rill, Pomeroy, Simmons, Summer, Trambull, Wade, Wilkin-son, Wilmot, Wilson of Massachuetts-20.

Mr. BROWNING moved to limit the liberation of the mother, wife, and children to cases in which the owner has borne arms against the United States or adhered to their enemies, by giving them aid and comfort; which was agreed –yeas 21, nays 16, as follows:

YEAS-Messrs. Browning, Collamer, Cowan, Davis, Doo-HEAS-INCESTS, Browning, Contanter, Cowar, Davis, Boo-little, Fossenden, Foster, Harris, Henderson, Kennedy, Lane of Indiana, Powell, Rice, Saulsbury, Sherman, Simmons, Stark, Ten Eyck, Willey, Wilson of Missouri, Wright-21. NAYS-Messrs. Chandler, Clark, Foot, Grimes, Hale, Har-lan, Howard, Howe, King, Lane of Kansas, Summer, Trum-bull, Wade, Wilkinson, Wilmot, Wilson of Massachusetts--

16.

The bill was passed-yeas 28, nays 9, (Bayard, Carlile, Davis, Kennedy, Powell, Saulsbury, Stark, Willey, Wilson of Missouri.)

1862, May 16-Pending the consideration of a bill to suppress insurrection and to punish treason and rebellion,

Mr DAVIS moved to strike from the first section the words "and all his slaves, if any, shall be declared and made free," and make the imprisonment not less than five nor more than twenty years; which was rejected-yeas 7, nays 31, as follows :

YEAS—Messrs. Davis, McDougall, Pearce, Powell, Sauls-bury, Stark, Wilson of Missouri—7. NAVS—Messrs. Anthony, Browning, Chandler, Clark, Col-lamer, Cowan, Dixon, Doolittle, Fessenden, Foot, Foster, Grimes, Harris, Henderson, Howard, Howe, King, Lane of Indiana, Lane of Kansas, Morrill, Pomeroy, Sherman, Sim-mons, Summer, Ten Eyck, Trumbull, Wade, Willey, Wilmot, Wilson of Massachusetts, Wright—31.

May 19-Mr. Powell moved to strike out the eleventh section :

That the President of the United States is authorized to employ as many persons of African descent as he may deem necessary and proper for the suppression of this rebellion, and for this purpose he may organize and use them in such manner as he may judge best for the public welfare.

Which was rejected-yeas 11, nays 25, as follows:

YEAS-Messrs. Carlile, Davis, Henderson, Latham, Pearce, Powell, Saulsbury, Stark, Willey, Wilson of Missouri, Wright -11.

MAYS — Messrs. Anthony, Browning, Clark, Collamer, Cowan, Dixon, Doollittle, Fessenden, Foot, Foster, Grimes, Harlan, Harris, Howard, Howe, Lane of Kansas, Pomeroy, Sherman, Sumner, Ten Eyck, Trunbull, Wade, Wilkinson, Wilmot, Wilson of Massachusetts—25.

The bill did not get to a final vote.

#### IN HOUSE.

July 16-The Senate bill quoted above was taken up, when

Mr. HOLMAN moved that it do lie upon the table, which was negatived-yeas 30, nays 77, as follows :

Shiel, John B. Steele, William G. Steele, Stiles, Webster, Wickliffe-30.
Naxs-Messrs. Aldrich, Alley, Arnold, Ashley, Babbitt, Baxter, Beaman, Bingham, Samuel S. Blair, Blake, Buffinton, Campbell, Colfax, Roscoe Conkling, Cutler, Davis, Dawes, Duell, Dunn, Edwards, Ely, Fenton, Fessenden, Frank, Goodwin, Granger, Haight, Hale, Hanchett, Hooper, Julian, Kelley, Francis W. Kellogg, Wilhiam Kellogg, Leary, Loomis, Lovejoy, Low, McKnight, McPherson, Moorhead, Anson P. Morrill, Justin S. Morrill, Nixon, Noell, Olin, Patton, Timo-thy G. Phelps, Pike, Porter, Potter, Alexander H. Rice, John H. Rice, Riddle, Edward H. Rollins, Sargent, Sedg-wick, Shanks, Shellabarger, Sherman, Smith, Spaulding, Trowbridge, Van Horn, Verree, Wall, Wallace, Walton, Ward, Wilson, Windom, Worcester-77.
The bill then passed.

The bill then passed.

## PREVIOUS PROCEEDINGS ON THIS SUBJECT. Third Session, Thirty-Seventh Congress.

# IN HOUSE.

1863, January 27-Mr. STEVENS offered a bill for the enlistment of 150,000 soldiers, persons of color or of African descent, to serve for five years, and to receive a bounty of \$5, and pay at the rate of \$10 per month, recruiting stations to be established in free or slave States, regimental officers to be white, company officers white or colored as the President may direct. Slaves and freemen may be enlisted; such persons shall never again be slaves, but the United States shall pay for such of them as belong to persons not disloyal.

Mr. HICKMAN offered a substitute authorizing the organization of three hundred colored regiments to be uniformed in a special and marked manner, to serve for seven years, at a monthly pay of \$6 50, one half to be retained till discharged-their officers to be of collegiate education and receive twice the pay of other officers. The bill contained a colonization feature, and appropriated the proceeds of confiscation to be applied to educational purposes in the States in which the property confiscated, lies.

The bill was repeatedly before the House, was severely "filibustered," and was finally modified by Mr. STEVENS to read as follows:

Be it enacted by the Senate and House of Representatives the United States of America in Congress assembled, of the United States of America in Congress assembled, That the President be, and he is hereby, authorized to en-roll, arm, equip, and receive into the land and naval service of the United States, such number of volunteers of African descent as he may deem useful to suppress the present re-bellion, for such term of service as he may prescribe, not exceeding five years. The said volunteers to be organized encoding to the recruitations of the hereach of carrier in according to the regulations of the branch of service in which they may be enlisted, to receive the same rations, clothing, and equipments as other volunteers, and a monthly pay not to exceed that of other volunteers; to be officered by persons appointed and commissioned by the President, and to be governed by the rules and articles of war and such other rules and a reculations on more the presented hy such other rules and regulations as may be prescribed by the President: *Provided*, That nothing herein contained, or in the rules and articles of war, shall be so construed as to authorize or permit any officer of African descent to be apauthorize or permit any officer of African descent to be ap-pointed to rank, or to exercise military or naval authority over white officers, soldiers, or men in the military or naval service of the United States; norshall any greater pay than ten dollars per month, with the usual allowance of clothing and rations, be allowed or paid to privates or laborers of African descent which are, or may be, in the military or naval service of the United States: *Provided further*, That the slaves of loyal citizens in the States exempt by the President's proclamation of January 1, 1863, shall not be received into the armed service of the United States, nor shall there be recruiting offices opened in either of the States of Delaware, Maryland, West Virginia, Kentucky, Tennessee, or Missouri, without the consent of the Gover-nor of said State having been first obtained.

February 2—It passed the House—yeas 85, ays 57. The NAVS were nays 57.

NAXS-Messrs. William Allen, William J. Allen, Ancona,

Shiel, John E. Steele, William G. Steele, Stiles, Webster, Wickliffe-30.
 Waxse-Messra. Aldrich, Alley, Arnold, Ashley, Babbitt, Baxter, Beaman, Bingham, SanuellS. Blair, Blake, Buffhuson, Volkanov, Crawens, Crisheld, Crittenden, Delaplaine, Granger, Grider, Haight, Hale, Laul, Harding, Harrison, Holman, Horton, Campbell, Colfax, Roscoe Conkling, Cutler, Davis, Dawes, Duell, Dunn, Edwards, Ely, Fonton, Fessenden, Frank, Goodwin, Granger, Haight, Hale, Hanchett, Hooper, Julia, Kelley, Francis W. Kellogg, Wilham Kellogg, Leary, Loomis, Lovejoy, Low, McKnight, McPherson, Moorhead, Anson P.
 Morrill, Justin S. Morrill, Nixon, Noell, Olin, Patton, Ficher, The Songato Computities reported
 Methodia Marker, State State, State State, State State, State State, State State, State State State, State State, State State, State State, State State, State State, State State State, State State State, State State State State State, State S

February 13-The Senate Committee reported adversely to its passage.

#### IN SENATE.

#### Pending the engineer bill,

March 2-Mr. DAVIS offered this new section: That no negro, free or slave, shall be enrolled in the mili-tary, marine, or naval service of the United States.

Which was not agreed to-yeas 12, nays 23, as follows:

YEAS-Messrs. Davis, Henderson, Hicks, Kennedy, Nes-mith, Powell, Richardson, Saulsbury, Turpie, Wall, Willey, Wilson of Missouri-12.

Marson of Missiouri-12. Nars-Messrs. Chandler, Clark, Collamer, Cowan, Doo-little, Fessenden, Foot, Foster, Grimes, *Harding*, Harlan, Harris, Howard, Howe, Lane of Indiana, Lane of Kansas, Morrill, Pomeroy, Sumner, Ten Eyck, Trumbull, Wilkin-son, Wilson of Massachusetts-23.

Mr. Powell offered to amend the eighth section by adding this proviso:

Provided, That no person of African descent shall be commissioned or hold an office in the army of the United States.

Which was agreed to-yeas 18, nays 17, as follows:

YEAS-Cowan, Davis, Harding, Harris, Henderson, Hicks, Howe, Kennedy, Lane of Indiana, Nesmith, Powell, Rich-ardson, Saulsbury, Ten Eyck, Turpie, Wall, Willey, Wilson of Missouri-18.

NATS-Messrs. Chandler, Clark, Doolittle, Fessenden, Foot, Foster, Grimes, Harlan, Howard, King, Lane of Kan-sas, Morrill, Pomeroy, Sumner, Trumbull, Wilkinson, Wil-son of Massachusetts-17.

Mr. LANE, of Kansas, moved to add to the above proviso the following words :

Except company officers in companies composed exclu-sively of persons of African descent.

Which was agreed to-yeas 19, nays 17, as follows :

YEAS-Messrs. Chandler, Clark, Collamer, Doolittle, Fes-senden, Foot, Foster, Grimes, Harlan, Howe, King, Lane of Kansas, Morrill, Pomeroy, Sumner, Trumbull, Wilkinson, Wilmot, Wilkon of Massachusetts-19. NAYS-Messrs. Cowan, Davis, Harding, Harris, Hender-son, Hicks, Howard, Lane of Indiana, Nesmith, Powell, Richardson, Saulsbury, Sherman, Ten Eyck, Wall, Willey, Wilson of Missouri-17.

The eighth section was then stricken out without a division.

#### PAY OF COLORED SOLDIERS.

#### First Session, Thirty-Eighth Congress. IN SENATE.

1864, February 23-The Senate considered the joint resolution to equalize the pay of soldiers.

Mr. Cowan moved to strike out all after the enacting clause, and insert :

That from and after the passage of this joint resolution the soldiers of the United States of America, of the same grade and service, shall be entitled to the same pay, rations, and pensions.

Mr. DAVIS moved to insert the following as a substitute :

All negroes and mulattoes, by whatever term designated, in the military service of the United States, be, and the same are hereby declared to be, discharged from such ser-vice, and shall be disarmed as soon as practicable; but the President of the United States may retain such of said negroes and mulattoes as he shall deem proper in the military service as teamsters and laborers; and the commandants of the respective regiments to which said slaves may be at-tached shall issue to their owner or owners a certificate of their employment in the service of the Government.

That every loyal owner of any slave that has been here-tofore, or that may hereafter be taken into the service, or for the use of the United States, shall be entitled to a fair and easonable compensation for the service of such slave for the time such slave may have been, or may be, in such bit the time such state in the part of the part of the part of the state of the part of th reasonably satisfy them for all damages that he, she they may have sustained by reason of the death or disa-bility of any such slave or slaves. And where any such slave or slaves may become a fugitive, and be not returned to the owner, the United States shall pay to such owner or owners the reasonable value of the service of said slave or slaves

That the owner or owners of any slave, entitled to pay and compensation, as hereinbefore provided for, may make ont his, her, or their account therefor against the United States, and upon filing the same at the Treasury Department, sustained by vouchers and proofs as are required ordinarily to support accounts against the United States, the same shall be audited and paid by the proper officers out of any money in the Treasury not otherwise appropriated.

Which was disagreed to-yeas 7, nays 30, as follows:

Messrs. Buckalew, Carlile, Davis, Powell, Riddle, YEAS

FINS-MESSIS, Backalew, Cartae, Davis, Fobeli, Ruale, Sailsbury, Wright--. Navs-Messis, Chandler, Clark, Collamer, Conness, Dixon, Fessenden, Foot, Foster, Grimes, Hale, Harding, Harlan, Harris, Henderson, Howard, Howe, Johnson, Lane of In-diana, Lane of Kansas, Morgan, Morrill, Nesmith, Ramsey, Sherman, Sumner, Ten Eyck, Van Winkle, Wade, Willey, Wilson-30.

February 25-Mr. COLLAMER moved to amend by adding to the resolution the following :

All persons enlisted or mustered into service as volunteers under the call dated October 17, 1863, for three hundred thousand volunteers, who were at the time of enlistment actually, and for six months previous had been, resident in-habitauts of the State in which they volunteered, shall receive from the United States the same amount of bounty without regard to color: *Provided*, *however*, That the fore-going provision shall not extend to any State which the President by proclamation has declared in a state of insurrection.

Mr. SUMNER moved to amend the amendment by adding the words:

And provided also, That all persons whose papers of en-listment show that they were enlisted under the act of Congress of July, 1861, shall receive from the time of their enlistment the pay promised by that statute.

Which was agreed to-yeas 19, nays 18, as follows:

YEAS-Messrs. Clark, Collamer, Conness, Dixon, Doolit-tle, Foot, Foster, Hale, Harding, Howard, Lane of Kansas, Morgan, Morrill, Pomero Van Winkle, Wilson-19. Pomeroy, Sprague, Sumner, Ten Eyck,

NAVE-MCSSTS. Buckalew, Chandler, Davis, Grimes, Har-lan, Harris, Henderson, *Hendricks*, Howe, Johnson, Lane of Indiana, Nesmith, Powell, Ramsey, Saulsbury, Wilkin-son, Willey, Wright-18.

Feb. 29-The bill was recommitted to the Committee on Military Affairs, and again reported amended.

March 10-Mr. DAVIS offered a new section :

That for every slave or slaves that may have been here-United States, the loyal owner or owners of such slave or slaves shall be entitled to their fair and reasonable value, slaves shall be entitled to their fair and reasonable value, to be ascertained by three commissioners to be appointed by the United States district court for the district from which such slave or slaves may have been taken. And said commissioners shall take evidence and report on the points of the loyalty and ownership of the claimants, and the value of such slave or slaves; and when said report is confirmed by the district court aforesaid, it shall be con-clusing on the points compressed by it; and on the presents clusive on the points embraced by it; and on the presenta-tion of an authoritative copy of said report to the Treasury Department, any sum or sums therein appearing to be due any loyal owner or owners shall be paid out of any money in the Treasury not otherwise appropriated.

Which was disagreed to-yeas 6, nays 31, as follows:

YEAS-Messrs. Davis, Johnson, Nesmith, Powell, Riddle, Wright-6.

Wright-6. Naxs-Messrs. Anthony, Brown, Buckalew, Chandler, Clark, Collamer, Conness, Dixon, Doolittle, Fessenden, Foot, Foster, Grimes, Harding, Ilarlan, Ilarris, Howard, Lane of Indiana, Lane of Kansas, Morgan, Morrill, Pomeroy, Ram-sey, Sherman, Sumner, Ten Eyck, Van Winkle, Wade, Wil-kinson, Willey, Wilson-31.

Mr. DAVIS offered an additional section :

That in every case where heretofore any person has agreed to join, and has in fact joined, or hereafter may join, the military service of the United States, and has been received into the said service by any military officer thereof, the persou so joining and received into such service shall, for all purposes whatever, be deemed to have been regularly mustered into said service in the position of officer, non-commis-sioned officer, or private, in which he may have served, notwithstanding he may not have been mustered in according to law and the regulations of the War Department.

Which was lost-yeas 7, nays 29, as follows:

YEAS-Messrs. Davis, Hendricks, Johnson, Nesmith, Pow-

YEAS-Messrs. Davis, Henaricks, Johnson, Nesmut, Fou-ell, Wade, Wright-7. NArs-Messrs. Anthony, Brown, Buckalew, Chandler, Clark, Collamer, Conness, Dixon, Doolittle, Fessenden, Foot, Foster, Grimes, Harding, Harlan, Harris, Howard, Lane of Indiana, Lane of Kansas, Morgan, Morrill, Pomeroy, Ram-sey, Sherman, Sumner, Ten Eyck, Van Winkle, Willey, Wilson-29.

The bill then passed—yeas 31, nays 6, as follows :

YEAS-Messrs. Anthony, Brown, Chandler, Clark, Con-ness, Cowan, Dixon, Doolittle, Fessenden, Foot, Foster, Grimes, Harlan, Harris, Howard, Howe, Johnson, Lane of Indiana, Lane of Kansas, Morgan, Morrill, Nesmith, Pome-roy, Ramsey, Sherman, Sumner, Ten Eyck, Van Winkle, Wede Willow, Wilcon 21 Walter, J. K. Starberg, Sherman, Summer, Ten Eyck, Van Winkle, Wade, Willey, Wilson-31. NATS-Messrs. Buckalew, Davis, Hendricks, Powell, Rid-U. Walter, Starberg, Starberg,

dle, Wright-6.

### IN HOUSE.

It was referred to the Committee on Military Affairs, and not reported upon, when the Senate added propositions respecting the subject to the army appropriation bill, as follows :

#### IN SENATE.

Pending the army appropriation bill, 1864, April 22-Mr. WILSON offered these resolutions:

That all persons of color who have been or may be mustered into the service of the United States shall receive the same uniform, clothing, arms, equipments, camp equipage, rations, medical and hospital attendance, pay and emolu-ments, other than bounty, as other soldiers of the regular or volunteer forces of the United States of like arm of the service, from and after the 1st day of January, 1864, and that every person of color who shall hereafter be mustered into the service shall receive such sums in bounty as the President shall order in the different States and parts of

President shan order in the university states and parts of the United States, not exceeding one hundred dollars. That all persons enlisted and mustered into service as volunteers under the call dated October 17, 1864, for 300,000 volunteers, who were at the time of enlistment actually enrolled and subject to draft in the State in which they relationships and shall receive from the United States are encourvolunteered, shall receive from the United States the same

That all persons of color, who have been enlisted and mustered into the service of the United States shall be enittled to receive the pay and clothing allowed by law to other volunteers in the service from the date of their mus-ter into the service: *Provided*, That the same shall have been pledged or promised to them by any officer or person who, in making such pledge or promise, acted by authority of the War Department; and the Secretary of War is hereby authorized to determine any question of fact arising under this provision.

That the same premium shall be allowed for each colored recruit now imstered, or hereafter to be mustered, into the service of the United States, as is or shall be allowed for white recruits: *Provided*, That the Secretary of War shall proviously give his assent to the same.

Which were agreed to-yeas 31, nays 5, as follows :

YEAS-Messrs. Anthony, Brown, Clark, Collamer, Conness,.

Cowan, Dixon, Doolittle, Fessenden, Foster, Grimes, Hale, Harding, Henderson, Howard, Howe, Johnson, Lane of Indiana, McDougall, Morgan, Morrill, Pomeroy, Ramsey, Sherman, Sprague, Sumner, Ten Eyck, Trumbull, Van Winkle, Willey, Wilson-31.

NAYS-Messrs. Buckalew, Davis, Hendricks, Powell, Saulsbury-5.

Mr. DAVIS moved to add these words as a new section :

That when the existing insurrection and rebellion shall have been suppressed, and the authority and laws of the United States shall have been re-established in the States where the people are in rebellion, all negroes then in the military service of the United States shall thereupon be discharged from such service.

Which was disagreed to-yeas 10, (Messrs. Buckalew, Carlile, Cowan, Davis, Henderson, Hendricks, Johnson, McDougall, Powell, Saulsbury,) nays 27.

Mr. HENDRICKS offered a new section :

That the pay of the soldiers and non-commissioned officers of the army of the United States shall hereafter be fifty per cent. greater than is now allowed by law: *Provided*, That the pay of non-commissioned officers shall not exceed twenty-two dollars per month.

Which was disagreed to—yeas 6, nays 30, as follows:

YEAS-Messrs. Buckalew, Carlile, Davis, Hendricks, Lane of Indiana, Powell-6.

Of Indiana, Fowen-o. NATS-Messrs. Anthony, Brown, Chandler, Clark, Collamer, Conness, Cowan, Dixon, Doolittle, Fessenden, Foot, Foster, Grimes, Hale, Harding, Henderson, Howard, Howe, McDougall, Morgan, Morrill, Pomeroy, Ramsey, Sherman, Sumner, Ten Eyck, Trumbull, Wilkinson, Willey, Wilson --30.

The army appropriation bill was then passed —yeas 36, nay 1, (Mr. Powell.)

#### IN HOUSE.

April 30—The House Committee on Military Affairs proposed to amend by substituting a bill, making these changes. May 1st was fixed as the date of increased pay; the bounty to be given to those colored soldiers enlisted under the call of Oct. 17, 1863, is limited to \$100; and these two sections are substituted for the Senate's last two:

That in every case where it shall be made to appear to the satisfaction of the Secretary of War, that any regiment, or any battery, or any company of caralry, of colored troops, has been enlisted and mustered into the service of the United States, under any authorized assurance given by any officer or agent of the United States, or by any Governor of any State, authorized thereto by the President or the Secretary of War, that the non-commissioned officers and privates of such regiment, battery, or company, should be paid the same as other troops of the same arm of the service, then they shall be so paid for the period of time counting from the date of their being respectively mustered into the service to the 1st day of May, A.D. 1864. That there may be reserved at the discretion of the Sec-

That there may be reserved at the discretion of the Secretary of War, and under such regulation as he may pre-Scribe, a portion of the pay of any colored soldier, not exceeding in any case more than one-third thereof, to be applied to the support of the family of such soldier, or of other near relatives dependent on him for support.

• Mr. HOLMAN moved to strike from the section the word "pay," which was lost—yeas 53, nays 85.

• A division of the question was called; the first amendment of the House was disagreed to; and that of the Senate—being the first paragraph of bill passed, April 22, above—was agreed to—yeas 80, nays 51, as follows:

YEAS-Messre. Alley, Allison, Ames, Anderson, Arnold, John D. Baldwin, Baxier, Beaman, Blaine, John B. Blair, Blow, Boutwell, Boyd, Brandegee, Broomall, Cobb, Cole, Creswell, Henry Winter Davis, Dawes, Deming, Dixon, Donnelly, Driggs, Eckley, Eliot, Farnsworth, Fenton, Frank, Garfield, Grinnell, *Griswold*, Higby, Hooper, Hotchkiss, Asahel W. Hubbard, John H. Hubbard, Hulburd, Jenckes, Julian, Kel-

ley, Francis W. Kellogg, Orlando Kellogg, Loan, Longyear, Marvin, McBride, McClurg, McIndoe, Samuel F. Miller, Moorhead, Morrill, Daniel Morris, Amos Myers, Leonard Myers, Norton, Odell, Charles O'Neill, Orth, Patterson, Perham, Pomeroy, Price, Alexander II, Rice, John II, Rice, Edward II. Rollins, Schenck, Scofield, Shannon, Sloan, Spalding, Stevens, Thayer, Upson, Ellihu B. Washburne, William B. Washburn, Williams, Wilder, Wilson, Windom-80.

dom-50. NATS-Messrs. James C. Allen, Ancona, Brooks, William G. Brown, Chanler, Clay, Cox, Dawson, Denison, Eden, Eldridge, Finck, Grider, Harding, Harrington, Charles M. Harris, Holman, Kernan, Kina, Kaapp, Law, Lazear, Le Blond, Long, Marcy, McDouell, McKinney, William H. Miller, James R. Morris, Morrison, Noble, John O'Neill, Perry, Samuel J. Randall, Robinson, James S. Rollins, Ross, Scott, Smith, John B. Steele, Stiles, Strouse, Vorheeter, Chilton A. While, Joseph W. White, Fernando Wood, Yeaman-51.

The second amendment of the Committee (with regard to bounty) was agreed to, and the Senate amendment, as amended, was agreed to—yeas 79, nays 52.

The third and fourth House amendments were disagreed to. The following amendment was then adopted as a substitute for the third proposition of the Senate:

That all free persons of color who have been or may be mustered into the military service of the United States shall from the date of their enlistment receive the same uniform, clothing, arms, equipments, camp equipage, rations, medical and hospital attendance, pay and emoluments, and bounty, as other soldiers of the regular or volunteer forces of the United States of like arm of the service.

Yeas 73, nays 55. The NAYS were-

Messrs, James C. Allen, Ancona, Anderson, Augustus C. Baldwin, Jacob B. Blair, Brooks, William G. Brown, Chamler, Clay, Cox, Cravens, Dawson, Eden, Eldridge, Finck, Ganson, Grider, Griswold, Hall, Harding, Harrington, Benjamin G. Harris, Charles M. Harris, Herrick, Holman, Philip Johmson, Kernan, King, Knapp, Law, Le Blond, Long, Marcy, William H. Miller, James E. Morris, Morrison, Noble, John O'Neitl, Perry, Radford, Robinson, James S. Rollins, Rass, Scott, Smith, John B. Steele, Stiles, Strouse, Voorhees, Whaley, Wheeler, Chilton A. While, Joseph W. While, Fernando Wood, Feanan-55.

The fourth amendment of the Senate was non-concurred in, being in another bill.

May 3—The Senate insisted on its amendments, and appointed a Committee of Conference, which,

May 25—Reported, and recommended the adoption of this section as a substitute for those adopted by the two Houses:

That in every case where it shall be made to appear to the satisfaction of the Sceretary of War that any regiment of infantry, or any battery, or any company of cavalry, of colored troops has been enlisted and mustered into the service of the United States, under any authorized assurance given by any officer or agent of the United States, or by any Governor of any State authorized thereto by the President or the Sceretary of War, the non-commissioned officers and privates of such regiment, battery, or company shall be paid the same as other troops of the same arm of the service, then they shall be so paid for the period of time counting from the date of their being respectively mustered into the service to the 1st day of January, 1864: *Provided*, *however*, That this section shall not be construed to prevent like payment to other colored troops from the time of their being nustered into the service, if such shall be held by the proper authority to be their right under the law.

Which was rejected—yeas 25, nays 121. The YEAS were:

Messrs. Baily, Boyd, William G. Brown, Ambrose W. Clark, Thomas T. Davis, Farnsworth, Garfield, Griswold, Hooper, Kasson, Schenck, Shannon, Smith, Smithers, Thomas, Tracy, Whaley, Woodbridge-25.

#### ANOTHER CONFERENCE REPORT.

June 10—The second Committee of Conference on the army appropriation bill recommended the following section relating to the pay of colored troops, as a substitute:

That all persons of color who were free on the 19th day of

April, 1861, and who have been enlisted and mustered into the military service of the United States, shall from the time of their enlistment be entitled to receive the pay, bounty, and of their enlistment be entitled to receive the pay, bounty, and clothing allowed to such persons by the laws existing at the time of their enlistment. And the Attorney General of the United States is hereby authorized to determine any question of law arising under this provision. And if the Attorney General aforesaid shall determine that any such Autorney General Aloresald shall determine that any such enlisted persons are entitled to receive any pay, bounty, or clothing in addition to what they have already received, the Secretary of War shall make all necessary regulations to enable the pay department to make payment in accord-nce with such determination.

June 11-It was concurred in, without divisin in the Senate.

June 13-The report was adopted in the Huse--yeas 72, nays 58, as follows:

Hue Tay Page 72, nays 56, as follows:
 Exs-Messrs. Allison, Ames, Anderson, Arnold, Ashley, Big, Baxter, Beaman, Blaine, Bloir, Blow, Boyd, Brande, State, Colo, Thomas T. Davis, Dixon, Danelly, Driggs, Eckley, Fenton, Frank, Garfield, Higby, Hoper, Hotchkiss, Ashel W. Hubbard, John H. Hubbard, persoll, Jenckes, Julian, Kelley, Francis W. Kellogg, Lit-tion, Loan, Longyear, Marvin, McClurg, McIndoe, Sam-ui F. Miller, Moorhead, Morrill, Daniel Morris, Amos Frees, Leonard Myers, Norton, Charles O'Niell, Orth, Pat-teson, Perham, Pike, Price, John H. Rice, Scofield, Shan-m, Smith, Smithers, Spalding, Starr, Stevene, Thayer, Momas, Tracy, Upson, Van Valkenburgh, Webster, Wil-ame, Wilder, Wilsou, Windom, Woodbridge-72.
 MATS-Messrs. James C. Allen, William J. Allen, Alley, Jugustus O. Baldwin, John D. Baldwin, Biss, Boutwell, ames S. Brown, Chanler, Coz, Cravens, Henry Winter Da-its, Dawes, Dausson, Denison, Eden, Edgerton, Eldridge, Gliot, Finck, Ganson, Gooch, Grüfer, Harding, Harrington, Charles M. Harris, Herrick, Holman, Hutchins, Kallfeisch, Kernan, King, Law, Le Blond, Marcy, McDowell, McKin-ney, Wm. H. Miller, James R. Morris, Morrison, John O'Niell, Pendleuon, Radford, Alexander H. Rice, Robinson, James S. Roulins, Ross, John B. Steele, Wm. G. Steele, Sties, Strouse, Staart, Swead, Wadsworth, Vheeler, Chilton While, Strouse, Staart, Swead, Wadsworth, Vheeler, Chilton While, Joseph W. White, Fernando Wood-58.
 1863, Dec. 14--Mr. LovEJOY moved that the

1863, Dec. 14-Mr. LOVEJOY moved that the Committee on Military Affairs be instructed to inquire into the expediency of placing, in any bill or bills they may report on the subject, all regular enlisted soldiers on the same footing as to pay without distinction of color. Mr. Cox moved to table the motion; which

was disagreed to-yeas 68, nays 87.

The resolution was then adopted.

1864, May 16-Mr. RICE, of Maine, introduced a bill for the enrollment, organization, and service of certain persons of African descent in the militia of the several States; which was refer-red to the Committee on the Militia.

#### **OPINION OF ATTORNEY GENERAL BATES ON PAYING** A COLORED CHAPLAIN.

ATTORNEY GENERAL'S OFFICE,

April 23, 1864.

April 23, 1304. SiR: You have done me the honor to refer to me a communication to yourself from his excellency John A. Andrew, Governor of Massachusetts, with accompanying papers, relative to the claim of Rev. Samuel Harrison for pay as chaplain of the 54th regiment of Massachusetts volunteers.

It appears by Governor Andrew's letter and the other papers that Mr. Harrison, who is a colored man, was duly elected, and on the 8th day of September, 1863, commissioned by Governor Andrew as a chaplain of the 54th regiment of Massachusetts volunteers in theservice of the United States; accepted into the service of the United States at Morris islaccepted into the service of the United States at Morris isi-and, South Carolina, by the proper mustering officer, and actually performed the duties of chaplain of that regiment then and since serving in South Carolina. On demanding his pay as chaplain, he was met by the following refusal in writing, signed by the paymaster at Hilton Head: "Samnel Harrison, chaplain of the 54th regiment Massa-chuscts volunteers, (colored troops,) asks pay at the usual rate, \$100 per month and two rations, which, he being of African descent, I decline paying under act of Congress passed July 17, 1862, employing persons of African descent

in military service of the United States. The chaplain declines to receive anything less.

You have requested my opinion whether the paymaster should have paid as demanded, and, if he should, whether it is your duty to order him to do so.

The 54th regiment of Massachusetts volunteers was oranized in the same manner as were other regiments of State volunteers under the following order of the War De-partment, dated Jauuary 26, 1863, viz: "Ordered, That Governor Andrew, of Massachusetts, is au-

thorized, until further orders, to raise such number of vol-unteer companies of artillery for duty in the forts of Massachusetts and elsewhere, and such corps of infantry for the volunteer military service, as he may find conveni-ent, such volunteers to be enlisted for three years or until sooner discharged, and may include persons of African de-scent, organized into separate corps. He will make the usual needful requisitions on the appropriate staff bureaus and officers for the proper transportate staff bureaus supplies, subsistence, arms, and equipments of such vol-uniters."

"EDWIN M. STANTON, "Secretary of War."

I do not know that any rule of law, constitutional or statutory, even prohibited the acceptance, organization, and muster of "persons of African descent" into the military service of the United States as enlisted mon or volunteers. But whatever doubt might have existed on the subject had been fully resolved before this order was issued, by the eleventh section of the act of 17th July, 1862, chap-ter 195, which authorized the President to employ as many persons of African descent as he might deem necessary and proper for the suppression of the rebellion, and for that purpose to organize and use them in such manner as he may judge best for the public welfare, and the twelfth sec-tion of the act of same date, chapter 201, which authorized the President to receive into the service of the United States for the purpose of constructing entrenchments, or performing camp service, or any other labor, or any mili-tary or naval service for which they might be found competent, persons of African descent, such persons to be enrolled and organized under such regulations, not inconsistent with the Constitution and laws, as the President might prescribe

The 54th Massachusetts regiment was therefore organized and mustered into the service of the United States under clear authority of law.

But the fifteenth section of the act of 17th July, 1862, chapter 201, after directing that all persons who have been or shall be enrolled in the service of the United States under or shall be enrolled in the service of the United states under that act shall receive the pay and rations then allowed by law to soldiers, according to their respective grades, con-tains this *proviso*: "That persons of African descent who under this law shall be employed shall receive the dollars per month and one ration, three dollars of which monthly are more big advisor."

per month and one ration, three dollars of which monthly pay may be in clothing." Whether persons of African descent "enrolled in the ser-vice of the United States" as private soldiers are included within the words "persons of African descent who under this law shall be employed," thereby limiting their pay as soldiers to ten dollars a month, is not the question you have submitted to me; for Mr. Harrison was not a private sol-dier, but an officer, serving under the commission of the Governor of Massachusetts, the authenticity and validity of which were recognized and admitted by the United States when he was mustered into its service. But the question is, can a person of African descent lawfully hold the office and receive the pay of chaplain of a volunteer regiment in the service of the United States?

regiment in the service of the United States? I have already said that I knew of no provision of law, constitutional or statutory, which prohibited the accept-ance of persons of African descent into the military service of the United States; and if they could be lawfully ac-cepted as private soldiers, so also might they be accepted as commissioned officers, if otherwise qualified therefor. But the express power conferred on the President by the eleventh section of the act of 17th July, 1862, chapter 195, before cited, to employ this class of persons for the suppression of the rebellion as he may judge best for the public welfare, furnishes all needful sanction of law to the employment of a furnishes all needful sanction of law to the employment of a turnsness an necutu sanction of law to the employment of a colored chaplain for a volunteer regiment of his own race. Nor is any prohibition of the employment of such person found in the statutes which declare the qualifications of chaplains. The ninth section of the act to authorize the employment of volunteers, &c., of 22d July, 1861, chapter 9, provides that there shall be allowed to each regiment one chapted by the section of the such authorized by the section of 9, provides that there shall be allowed to each regiment one chaplain, who shall be appointed by the regimental commander on the vote of the field officers and company commanders on duty with the regiment at the time the appointment shall be made. The chaplain so appointed must be a regularly ordained minister of a Christian de-nomination, &c. The seventh section of the act of 3d August, 1861, chapter 42, for the better organization of the

military establishment, declares that one chaplain shall be allowed to each regiment of the army, to be selected and appointed as the President may direct, provided that none but regularly ordained ministers of some Christian denomination shall be cligible to selection or appointment. The eighth section of the act of 17th July, 1862, chapter 200, declares that the two sections last cited shall be construed

to read as follows: "That no person shall be appointed a chaplain in the United States army who is not a regular ordained minister of some religious denomination, and who does not present testimonials of his present good standing as such minister, with a recommendation for his appointment as an army chaplain from some authorized ecclesiastical body, or not ben down from some authorized belowing to coid unitions less than five accredited ministers belonging to said religious denomination."

denomination." The closest inspection of these provisions will discover nothing that precludes the appointment of a Christian minister to the office of chaplain because he is a person of African descent. I therefore conclude that Mr. Harrison was the lawfully appointed and qualified chaplain of the 54th Massachusetts regiment. The pitth section of the act of the 17th July 1862.

The ninth section of the act of the 17th July, 1862 chapter 200, provides that thereafter, the compensation of all chaplains in the regular or volunteer service or army the regulation in the regulat of volunteer events of a matter hospitals shall be one hundred dollars per month and two rations a day when on duty. Was Mr. Harrison entitled to this rate of compensation, or was he limited to the pay of ten dollars a month and one ration, fixed by the *proviso* to the fifteenth section of the act of 17th July, 1862, chapter 201?

It will be observed that this *proviso* declares that ten dollars a month and one ration shall be received by persons of African descent employed under the law of which it is a part, viz., the act of 17th July, 1862, chapter 201. Now, we have seen that it is not necessary to resort to that law to find authority for the appointment of Mr. Harrison as chap-in for every from the authority they have seen link autority in the autority which main field as dealer lain, for, apart from the authority which main be presumed to exist prior to the enactment of any of these statutes, the eleventh section of the act of 17th July, 1862, chapter 195, sufficiently warranted it. To bring him, then, within the sweep of this provise, and thus withdraw him from the reach of the act which specifically fixes the pay of the class of definitions. reach of the act which specifically fixes the pay of the crass of officers to which by clear law he belongs, would violate the plainest principles of construction. The act, of which the *proviso* is a part, was not intended, in my opinion, either to authorize the employment or fix the pay of any persons of African descent, except those who might be needed to perform the humble offices of labor and service for which they might be found competent. The twelfth section authorizes them to be received into service for the purpose of continuous production competent. authorizes them to be received into service for the purpose of constructing entrenchments, or performing camp service, or any other labor, or any military or naval service for which they might be found competent. The thirteenth section declares that when any man or boy of African descent, who, by the laws of any State, shall owe service or labor to any person aiding the rebellion, shall render such service as this act provides for, he, his mother, wife and children shall be free thereafter, with certain exceptions. And the fideenth section fives their nava subefore stated.

And the fifteenth section fixes their pay as before stated. While it is true that the words of the twelfth section are broad enough to embrace all persons of African descent who may be received into the military or naval service of the United States, it is yet quite evident from the terms of the orbit section evel or from the armying of foodom the whole section, as well as from the promise of freedom held out to such persons who were slaves, in the thirteenth section, that in limiting their pay to ten dollars a month and one ration, Congress had in view the class who were fitted only for the humbler kinds of service referred to, and not persons who, under the authority of other laws, might be appointed to positions requiring higher qualifications, and entitled to a higher rate of pay. To assume that be-cause Mr. Harrison is a person of African descent he shall draw only the pay which this law establishes for the class it obviously refers to, and be deprived of the pay which apother law smoothing the start of the pay which another law specifically affixes to the office he lawfully held, would be, in my opinion, a distortion of both laws, not only unjust to him, but in plain violation of the purpose of Congress.

I therefore think that the paymaster should have paid Mr. Harrison his full pay as chaplain of a volunteer regiment.

Your attention having been specially called to the wrong done in this case, I am also of opinion that your constitu-tional obligation to take care that the laws be faithfully executed makes it your duty to direct the Secretary of War to inform the officers of the pay department of the army that such is your view of the law, and I do not doubt that it will be accepted by them as furnishing the correct rule for their action.

I am, sir, very respectfully, your obedient servant, EDWARD BATES,

Attorney General.

The PRESIDENT.

# PROTECTION OF COLORED SOLDIERS.

THE PRESIDENT'S ORDER.

[General Orders No. 252.] WAR DEPARTMENT, ADJUTANT GENERAL'S OFFICE. WASHINGTON, July 31, 1863.

The following order of the President is published for the information and government of all concerned :

# EXECUTIVE MANSION,

WASHINGTON, July 30, 1863. It is the duty of every Government to givprotection to its citizens, of whatever class color, or condition, and especially to those wb are duly organized as soldiers in the pubk service. The law of nations, and the usags and customs of war, as carried on by civilizd powers, permit no distinction as to color in to treatment of prisoners of war as public end To sell or enslave any captured perso, mies. on account of his color, and for no offend against the laws of war, is a relapse into bar barism, and a crime against the civilization d the age.

The Government of the United States wil give the same protection to all its soldiers; and if the enemy shall sell or enslave any one because of his color, the offence shall be punished by retaliation upon the enemy's prisoners in our possession.

It is therefore ordered that for every soldier of the United States killed in violation of the laws of war, a rebel soldier shall be executed; and for every one enslaved by the enemy, or sold into slavery, a rebel soldier shall be placed at hard labor on public works, and continued at such labor until the other shall be released. and receive the treatment due to a prisoner of ABRAHAM LİNCOLN. war.

By order of Secretary of War.

E. D. TOWNSEND, Assistant Adjutant General.

### FROM THE RULES OF WAR ADOPTED BY THE GOV-ERNMENT.

57. So soon as a man is armed by a sovereign Govern-ment, and takes the soldier's oath of fidelity, he is a bel-ligerent; his killing, wounding, or other warlike acts are no individual crimes or offences. No beligerent has a right to declare that enemies of a certain class, color, or condi-tion, when properly organized as soldiers, will not be treated

by him as public enemies. 58. The law of nations knows of no distinction of color, and if an enemy of the United States should enslave and sell any captured persons of their army, it would be a case for the severest retaliation, if not redressed upon complaint. The United States cannot retaliate by enslavement; there-fore death must be the retaliation for this crime against the law of nations.

#### THE PRESIDENT'S SPEECH AT THE BALTIMORE FAIR, APRIL 18, 1864.

LADIES AND GENTLEMEN: Calling it to mind that we are in Baltimore, we cannot fail to note that the world moves. [Applause.] Looking upon the many people I see assem-bled here to serve as they best may the soldiers of the Union, it occurs to me that three years ago those soldiers could not pass through Baltimore. I would say, blessings upon the men who have wrought these changes, and the ladies who have assisted them. [Applause.] This change which has taken place in Baltimore is part only of a far wider change that is taking place all over the country.

which has taken place in Baltimore is part only of a far wider change that is taking place all over the country. When the war commenced three years ago, no one ex-pected that it would last this long, and no one supposed that the institution of slavery would be materially affected by it. But here we are. [Applause.] The war is not yet ended, and slavery has been very materially affected or interfered with. [Loud applause.] So true is it that man proposes and God disposes. The world is in wast of a good definition of the word liberty. We all declare ourselves to be for liberty, but we

do not all mean the same thing. Some mean that a man

can do as he pleases with himself and his property. [Applause.] With others it means that some men can do as they please with other men and other men's labor. Each of these things are called liberty, although they are entirely different. To give an illustration: A shepherd drives the wolf from the throat of his sheep when attacked by him, and the sheep of course thanks the shepherd for the preservation of his life; but the wolf denounces him as de spoiling the sheep of his liberty, especially if it be a black sheep. [Applause.]

This same difference of opinion prevails among some of the people of the North. But the people of Maryland have recently been doing something to properly define the meaning of the word, and I thank them from the bottom of my heart for what they have done and are doing. [Applause.]

It is not very becoming for a President to make a speech at great length, but there is a painful rumor afloat in the country in reference to which a few words shall be said. It country in restrict to winch a tew words shall be shall. It is reported that there has been a wanton massacre of some hundreds of colored soldiers at Fort Pillow, Tennessee, during a recent engagement there, and he thought it fit to explain some facts in relation to the affair. It is said by some persons that the Government is not in this matter do-ue its during the commensument of the area it more ing its duty. At the commencement of the war it was doubtful whether black men would be used as soldiers or not. The matter was examined into very carefully, and after mature deliberation, the whole matter resting as it were with himself, he in his judgment decided that they abould. [Applause.] He was responsible for the act to the American people, to a Christian nation, to the future his-torian, and, above all, to his God, to whom he would have torian, and, above all, to ins God, to whom he would have one day to render an account of his stewardship. [Ap-plause.] He would now say that in his opinion the black soldier should have the same protection as the white soldier, and he would have it. [Applause.] It was an error to say that the Government was not acting in the matter. The Government has no direct evidence to confirm the reports in existence relative to this massacre, but he himself be-lieved the facts in relation to it to be as stated. When the neven the facts in relation to it to be as stated. When the Government does know the facts from official sources, and they prove to substantiate the reports, retribution will be surely given. [Applause.] What is reported, he thought, would make a clear case. If it is not true, then all such stories are to be considered as false. If proved true, when the matter is thoroughly examined, what shape is to be given to the retribution? Can we take the man who was captured at Vicksburg and shoot him for the victim of this given to the retribution? Can be take the man who was captured at Vicksburg and shoot him for the victim of this massacre? If it should happen that it was the act of only one man, what course is to be pursued then? It was a mat-ter requiring careful examination and deliberation, and if it should be substantiated by sufficient evidence all might rest assured that retribution would be had. [Applause.]

# Confederate Use of Colored Persons. 1861.

A dispatch from Charleston, dated January 1861, from R. R. Riordan to Hon. Perry Walker, at Mobile, describes the preparations for war, and contains this closing paragraph:

Large gangs of negroes from plantations are at work on the redoubts, which are substantially made of sand-bags and coated with sheet-iron.

A Washington dispatch to the Evening Post says:

A gentleman from Charleston says that everything there betokens active preparations for fight. The thousand negroes basy in building batteries, so far from inclining to insurrection, were grinning from ear to ear at the prospect of shooting the Yankees.

The Charleston Mercury of January 3, 1861, announced :

We learn that one hundred and fifty able-bodied free colored men, of Charleston, yesterday offered their services gratuitously to the Governor to hasten forward the import-ant work of throwing up redoubts wherever needed along our coast.

In April, the Lynchburg (Va.) Republican announced :

We learn that about seventy of the most respectable free negroes in this city have enrolled themselves and design tendering their services to the Governor, to act in whatever capacity may be assigned them in defence of the State. Three cheers for the patriotic free negroes of Lynchburg.

A letter in the Petersburg (Va.) Express, dated at Norfolk on the 23d April, says:

free, are as loyal as could be desired. They freely proffer their services to the State, and zealously contend for the privilege of being allowed to work on the batteries. Yesterday Gen. Gwynn declined the services of three hundred from Hampton who solicited employment on the batteries, and twice and thrice that number could be obtained in this city and vicinity in a single day, if it was thought advisable to accept them. Indeed, the entire fortifications of this harbor might be constructed by the voluntary labor of nelege of being allowed to contribute their share toward the defence of the State, and the protection of their masters and mistresses, who had always extended a sheltering hand over them.

In June, the rebel Legislature of Tennessee passed this enlistment bill, which became a law:

SEC. 1. Be it enacted by the General Assembly of the State of Tennessee, That from and after the passage of this act the Governor shall be, and he is hereby, authorized, at his discretion, to receive into the military service of the State all male free persons of color between the ages of fifteen and fifty, or such numbers as may be necessary, who may be sound in mind and body, and capable of actual service.

2. That such free persons of color shall receive, each, eight dollars per month, as pay, and such persons shall be enti-tled to draw, each, one ration per day, and shall be entitled

to a yearly allowance each for clothing. 3. That, in order to carry out the provisions of this act, it shall be the duty of the sheriffs of the several counties in this State to collect accurate information as to the number and condition, with the names of free persons of color, sub-ject to the provisions of this act, and shall, as it is practica-ble, report the same in writing to the Governor.

4. That a failure or refusal of the sheriffs, or any one or more of them, to perform the duties required, shall be deemed an offence, and on conviction thereof shall be punished as a misdemeanor.

5. That in the event a sufficient number of free persons of color to meet the wants of the State shall not tender their services, the Governor is empowered, through the sheriffs of the different counties, to press such persons until the requisite number is obtained.

6. That when any mess of volunteers shall keep a servant to wait on the members of the mess, each servant shall be allowed one ration.

lowed one ration. This act to take effect from and after its passage. W. C. WHITTHORNE, Speaker of the House of Representatives. B. L. STOVALL, Passed June 28, 1861. Speaker of the Senate.

HOW SLAVES ARE EMPLOYED BY THE "CONFED-ERATES."

The following paragraph appears in the Memphis (Tennessee) Avalanche of the 3d of September:

A procession of several hundred stout negro men, mem-hers of the "domestic institution," marched through our streets yesterday in military order, under command of Con-federate officers. They were all armed and equipped with shovels, axes, blankets, &c. A merrier set were never seen. They were brimfull of patriotism, shouting for Jeff. Davis and singing war songs.

The Avalanche of the 7th September said:

Upwards of one thousand negroes, armed with spades and pickaxes, have passed through the city within the past few days. Their destination is unknown, but it is supposed that they are on their way to the "other side of Jordan."

NEW OBLEANS, November 23

Over twenty-eight thousand troops were reviewed to-day by Governor Moore, Major General Lovell, and Brigadier General Ruggles. The line was over seven miles long. One regiment comprised 1,400 free colored men.

#### 1862.

THE ENROLLMENT OF FREE NEGROES IN REBELDOM.

In the Virginia House of Delegates, on the 4th of February, on the subject of enrolling free negroes for the rebel army, the following debate We copy the official report of the took place. proceedings from the Richmond Examiner :

The bill amending the convention act for the enrollment of free negroes was, on motion of Mr. Prince, taken up. Among the amendments in this bill Mr. Prince called at-tention to the one allowing ten cents for each negro so en-The negroes in all this section of the country, slave and rolled to the sheriff or officer so enrolling them. He proposed to strike out this amendment and insert in lieu of the proposed compensation that if the said officers fail to comply with the requisitions of this law, they be subjected to a penalty of not less than fifty nor more than one hun-dred dollars. As these officers were exempt from military duty, he said, it was about as little as they could do to per-form the service of enrolling the free negroes of their terms of the service of the of the official duties. His respective counties as a part of their official dutics. His amendment was adopted.

Mr. Rives proposed that the amendment in the bill respecting the term of the enlistment of negroes be amended to make the term ninety days, instead of a hundred and eighty. His reason for this was the fact that the families of many of the free negroes so enlisted, having no other means of support, would, as had been the case in his own county, suffer very much from want.

Mr. PRINCE agreed to compromise with the gentleman on one hundred and twenty days.

Mr. ANDERSON, of Botetourt, hoped that the amendment would not pass. One hundred and cighty days were only six months; and if white men could be drafted for two years, he saw no reason why free negroes should be entitled to such charitable discrimination.

Mr. RIVES replied that he made the proposition from no particular friendship to free negroes; if it were in his power, he would convert them all into slaves to-morrow. But it was simply to call the attention of the House to the fact that in his power and the neuron concern for the house that, in his own county, many severe cases of suffering had occurred among the families of free negroes from this cause, and he thought that possibly some alleviation might be brought about by the amendment proposed.

The amendment was rejected, and the bill was then ordered to its engrossment.

In the debate in the Rebel Congress, March 10, Mr. FOOTE of Tennessee, alluding to the fall of Nashville, said :

Gen. Johnston had called for 1,000 or 1,500 slaves to work on the fortifications, and that the call had been fully answered, when Nashville was surrendered,

#### DRAFT OF NEGROES FOR MILITARY SERVICE.

[From the Richmond Examiner, October 18, 1862.]

We notice that in Texas, and in some portions of the Mis-sissippi valley, the proposition is urged to make a conscrip-tion or forced levy of slaves, where their labor is necessary for the army. Since the invasion of the South the Yankees have stolen tens of thousands of negroes, and made them useful as teamsters, laborers in camp, &c. It appears that slaveholders are averse, for some reasons, to hire their ne-groes in the Confederate army. The prejudice is certainly an ignorant and mean one. As the war originated and is carried on in great part for the defence of the slaveholder in his property, rights, and the perpetuation of the institu-tion, it is reasonable to suppose that he ought to be first and foremost in aiding and assisting, by every means in his power, the triumph and success of our arms. Good wages are offered, and proper care and attention will be given We notice that in Texas, and in some portions of the Misare offered, and proper cure and attention will be given every negro hired to the army, and the slaveholder ought to remember that for every negro he thus furnishes he puts a soldier in the ranks.

November 2-Governor Joseph E. Brown, of Georgia, issued a call announcing that if a sufficient supply of negroes be not tendered within ten days, General Mercer will, in pursuance of authority given him, proceed to impress, and asking of every planter of Georgia a tender of one fifth of his negroes to complete the fortifications around Savannah. This one fifth is estimated at 15,000.

#### 1863.

This paragraph was published in the New York Times, July 15, 1862:

The Adjutant General of the Confederate States publishes a general order from the Robel War Department, directing recruiting officers, duly accredited, to draft every white or mulatto male found throughout the South who is able to bear arms, and who is between the ages of twenty and fifty-five years, whether such persons may have obtained substitutes for themselves or not; and willful evasion of this order is to be severely punished.

The Governor of South Carolina in July, issued a proclamation for 3,000 negroes to work on the fortifications, "the need for them being pressing."

[From the Savannah, (Ga.) News, Sept 2.]

# EMPLOYMENT OF NEGROES IN THE REBEL ARMY.

A joint committee of the Alabama Legislature, just adjourned, reported a resolution in favor of the proposition to employ slaves in the military service of the Confederate States, which proposition, we see, is favored by many of the presses of Mississippi and Alabama. After discussion in the Alabama House, the resolution was adopted by a vote of sixty-eight yeas to twelve nays, after striking out the words "military" before service, and "soldiers" at the end of the resolution. The resolution was amended and read as follows

That it is the duty of Congress to provide by law for the employment in the service of the Confederate States of America, in such situations and in such numbers as may be found absolutely necessary, the able-bodied slaves of the country, whether as pioneers, sappers and miners, cooks, nurses and teamsters.

In this form we can see no objection to the resolution.

THE CHANGING SENTIMENT OF CONGRESS.

In the Rebel House of Representatives, December 29th, Mr. DARGAN, of Alabama, introduced a bill to receive into the military service all that portion of population in Alabama, Mississippi, Louisiana, and Florida, known as " Creoles."

Mr. Dargan supported the bill in some remarks. He said the Creoles were a mixed-blooded race. Under the treaty of Paris in 1803, and the treaty of Spain in 1810, they were or rais in rook, and the frequy of splan in 130, here we recognized as freemen. Many of them owned large estates, and were intelligent men. They were as much devoted to our cause as any class of men in the South, and were even anxious to go into service. They had applied to Mir. Randolph, they Sourchever of Way. We metable do divided against the then Secretary of War. Mr. Randolph decided against the application, on the ground that it might furnish to the enemy a pretext of arming our slaves against us. Some time after this he was again applied to by them, and he went to the present Secretary of War, Mr. Seddon, and laid the matter before him. Mr. Seddon refused to entertain the before on the mound the bit is did not some before proposition, on the ground that it did not come up before him through the military authorities. To obviate this ob-jection, Gen. Maury, at Mobile, soon afterwards represented their wishes to the War Department. Mr. Seddon refused the offer of their services, on the ground that it would be incompatible with the position we occupied before the world; that it could not be done.

Mr. Dargan said he differed with the Secretary of War. He cared not for "the world." He cared no more for their opinions than they did for ours. He was anxious to bring into service every free man, be he who he may, willing to strike for our cause. He saw no objection to employing strike for our cause. He saw no objection to employing Creoles; they would form a potent element in our army. In his district alone a brigade of them could be raised The crisis had been brought upon us by the enemy, and he believed the time would yet come when the question would not be the Union or no Union, but whether Southern men should be permitted to live at all. In resisting subjugation by such a barbarous foe hc was for employing all our available force. He would go further and say that he was for arming and putting the slaves into the military service. It was infavor even of employing them as a military service. He was infavor even of employing them as a military arm in the defence of the country.

#### 1864.

[From the Charleston Courier of January 27.]

NEGRO LABOR UPON FORTIFICATIONS.

The Mayor of Charleston, Charles Macbeth, summons all alweholders within the city to furnish to the military au-thorities forthwith, one-fourth of all their male slaves be-tween the ages of fifteen and fifty, to labor upon the fortifi-cations. The penalty aunounced, in case of failure to com-ply with this requisition is a fine of \$200 for every slave not fortherming. Compensation is allowed at the rate of \$400 a year.

All free male persons of color between the ages of fifteen and fifty are required to give themselves up for the same purpose. Those not complying will be imprisoned, and set to work upon the fortifications along the coast. To free negroes no other compensation than rations is allowed.

#### NEGROES IN THE ARMY.

The Richmond press publish the official copy of "An act to increase the efficiency of the army by the employment of free negroes and slaves in certain capacities," lately passed by the

The negroes are to perform Rebel Congress. "such duties as the Secretary of War or Com-nanding General may prescribe." The first secion is as follows:

And is as follows: The Congress of the Confederate States of America do en-tic, That all male free negroes, and other free persons of polor, not including those who are free under the treaty of paris, of 1803, or under the treaty of Spain, of 1819, resident in the Confederate States, between the ages of eighteen and fifty years, shall be held liable to perform such duties with the army, or in connection with the military defences of the country, in the way of work upon the fortifications, or in government works for the production or preparation of materials of war, or in military hospitals, as the Secretary of War or the Commanding General of the Trans-Mississiphi Department may, from time to time, prescribe; and while Department may, from time to time, prescribe; and while angaged in the performances of such duties shall receive ations and clothing and compensation at the rate of eleven A sign in the performances of scan dudes shall receive actions and clothing and compensation at the rate of eleven follars a month, under such rules and regulations as the aid Scretary may establish: *Provided*, That the Scretary of War of the Commanding General of the Trans-Mississippi Department, with the approval of the President, may ex-smpt from the operations of this act such free negroes as the interests of the country may require should be exempt-d, or such as he may think proper to exempt on the ground of justice, equity or necessity. The third section provides that when the Secretary of War shall be unable to procure the services of slaves in any military department, then he is authorized to impress the ervices of as many male slaves, not to exceed twenty thou-sand, as may be required, from time to time, to discharge the duties indicated in the first section of the act. The owner of the slave is to be paid for his services; or, if the bk killed or "escape to the enemy," the owner shall re-eive his full value.

Governor Smith, of Virginia, has made a call for five housand male slaves to work on the batteries, to be drawn from fifty counties. The call for this force has been made by the President under a resolution of Congress.

General Magruder's proclamation to Texans, March 4, thus closes :

March 4, thus closes: The slaveholding gentlemen of each county are respect-luly requested to meet together at their respective county ieats, or some convenient point, and appoint one or more of heir number to accompany their negroes to Houston, and see that they are made comfortable. One-fourth of all the male hands between the ages of seventeen and fifty years in sach county, without regard at this time to the numbers which have been previously furnished, will accomplish the purpose of fortifying Houston within a short time; and planters and all others are assured that the necessity of this sourse is manifest to the Major General commanding, and sould be made so to them, but that his plans would thus be made known to the enemy. By command of Maj. Gen. J. Bankhead Magruder : EDMUND P. TURNER, A. A. Gen. Housron, March 4, 1864.

HOUSTON, March 4, 1864.

"CONFEDERATE" LEGISLATION UPON NEGRO PRI-SONERS AND THEIR WHITE OFFICERS WHEN CAPTURED.\*

1863, May 1-An act was approved declaring that the commissioned officers of the enemy ought not to be delivered to the authorities of the respective States, (as suggested in Davis's message;) but all captives taken by the Con-federate forces ought to be dealt with and disposed of by the Confederate Government.

President Lincoln's emancipation proclama-tions of September 22, 1862, and January 1, 1863, were resolved to be inconsistent with the usages of war among civilized nations, and should be repressed by retaliation; and the

\* December 23, 1862—Jefferson Davis issued a proclama-tion of outlawry against Major General B. F. Butler, the last two clauses of which are: Third. That all negro slaves captured in arms be at once delivered over to the executive authorities of the respective States to which they belong, to be dealt with according to the laws of said States.

Fourth. That the like orders be executed in all cases with respect to all commissioned officers of the United States when found serving in company with said slaves in insur-rection against the authorities of the different States of this Confederacy.

President is authorized to cause full and complete retaliation for every such violation, in such manner and to such extent as he may think proper.

Every white commissioned officer commanding negroes or mulattoes in arms against the Confederate States shall be deemed as inciting servile insurrection, and shall, if captured, be put to death, or be otherwise punished, at the discretion of the court.

Every person charged with an offence made punishable under the act shall be tried by the military court of the army or corps of troops capturing him; and, after conviction, the President may commute the punishment in such manner and on such terms as he may deem proper.

All negroes and mulattoes who shall be engaged in war or taken in arms against the Confederate States, or shall give aid or comfort to the enemies of the Confederate States, shall, when captured in the Confederate States, be delivered to the authorities of the State or States in which they shall be captured, to be dealt with according to the present or future laws of such State or States.

PROPOSED PROHIBITION OF PAYMENT TO COLORED SOLDIERS.

First Session, Thirty-Seventh Congress.

### IN HOUSE.

Pending the army appropriation bill in the House,

1861, July 11-Mr. VALLANDIGHAM offered this proviso to come in at the end of the bill :

*Provided*, That no part of the money hereby appropri-ated shall be employed in subjugating, or holding as a con-quered province, any sovereign State now or lately one of the United States; nor in abolishing or interfering with African slavery in any of the States.

Which was rejected, without a call of the yeas and nays.

# Second Session, Thirty-Seventh Congress. IN SENATE.

1862, June 2-Mr. SAULSBURY offered this additional section:

That no part of the money arising from the taxes, im-posts, and excises in this bill provided for, shall be applied to or expended for the support or maintenance of fugitive slaves or free negroes.

Which was rejected-yeas 5, nays 34, as follows:

YEAS-Messrs. Davis, Nesmith, Powell, Saulsbury, Stark

-34.

June 6-Pending the internal revenue bill, Mr. DAVIS offered this as a new section :

That no money raised under this act shall ever be applied to pay or subsist any armed negroes in the service of the United States for any period of time after it goes into operation and effect.

Which was rejected without a division.

# Third Session, Thirty-Seventh Congress. IN SENATE.

Pending the army appropriation bill,

proviso to the bill :

**Provided**, That no part of the sums appropriated by this act shall be disbursed for the pay, subsistence, or any other supplies of any negro, free or slave, in the armed military service of the United States.

lows:

YEAS-Messrs. Carlile, Davis, Kennedy, Latham, Nesmith, Powell, Turpie, Wall-8. NATS-McSsrs. Anthony, Browning, Chandler, Clark, Col-lamer, Cowan, Dixon, Doolittle, Fesseuden, Foot, Foster, Grimes, Hale, Harlan, Harris, Howe, King, Lane of Indi-ana, Lane of Kansas, Morrill, Pomeroy, Sherman, Sumner, Ten Eyck, Trumbull, Wilkinson, Willey, Wilson of Mas-sachusetts-28.

# First Session, Thirty-Eighth Congress. IN HOUSE.

1863, December 21-Pending the deficiency bill, Mr. HARDING offered an amendment:

Provided, That no part of the moneys aforesaid shall be applied to the raising, arming, equipping, or paying of negro soldiers.

Which was disagreed to-yeas 41, nays 105. The YEAS were:

Mossrs. Ancona, Bliss, James S. Brown, Coffroth, Cox, Dawson, Denison, Eden, Edgerton, Eldridge, Finck, Grider, Hall, Harding, Harrington, Benjamin G. Harris, Charles M. Harris, Philip Johnson, Kring, Knapp, Law, Long, Marcy, McKinney, William H. Miller, James R. Morris, Morrison, Noble, John O'Neill, Pendleton, Samuel J. Randall, Rogers, Ross, Scott, Stiles, Strouse, Stuart, Chilton A. White, Joseph W. White, Yeaman-41.

1864, March 21-Pending the army appropriation bill—On concurring in the amendment made in Committee of the Whole, viz., to add to the end of the bill the words:

Provided, That no part of the money herein appropriated shall be applied or used for the purpose of raising, arming, equipping, or paying negro soldiers.

The yeas were •18, nays 81. The YEAS were :

Messrs. Ancona, Dawson, Denison, Eldridge, Harding, Harrington, Benjamin G. Harris, Long, Mallory, Marcy, Wm. H. Miller, Morrison, Samuel J. Randalt, James S. Rol-lins, Ross, Stiles, Stuart, Wadsworth–18.

Of the Democrats, Messrs. Baily, Augustus C. Baldwin, Kernan, Odell, Radford, Stebbins, Wheeler, and Winfield voted nay.

# Homesteads For Soldiers. First Session, Thirty-Eighth Congress. IN HOUSE.

1863, December 17-Mr. Wilson offered the following; which was agreed to:

Resolved, That the Committee on Public Lands be instructed to inquire what legislation is necessary to enable persons in the military and naval service of the United States to avail themselves, while engaged in said service, of the benefits of the homestead act, and to report by bill or otherwise.

1864, May 11-The House considered the bill to secure to persons in the military or naval service of the United States, homesteads from confiscated estates in insurrectionary districts; and

May 12-Passed it-yeas 76, nays 65, as follows:

YEAS-Messrs. Alley, Allison, Ames, Anderson, Ashley, John D. Baldwin, Baxter, Beaman, Boutwell, Boyd, Bran-degee, Broomail, Ambrose W. Clark, Freemau Clarke, Cole, Creswell, Henry Winter Davis, Dawes, Deming, Driggs, Farnsworth, Fenton, Garfield, Gooch, Grinnell, Hale, Higby, Hooper, Hotchkiss, Asahel W. Hubbard, John H. Hubbard, Hulburd, Jenckes, Julian, Kelley, Francis W.

1863, Jan. 28-Mr. DAVIS moved to add this roviso to the bill: Provided, That no part of the sums appropriated by this t shall be disbursed for the pay, subsistence, or any other pplies of any negro, free or slave, in the armed military rvise of the United States. Which was rejected—yeas 8, nays 28, as fol ws: YEAS-Messrs. Carbile, Davis, Kennedy, Latham, Nesmith, well Travis Wall.

C. Baldwin, Jacob B. Blair, Bliss, Brooks, James S. Broum, W. G. Brown, Chanler, Coffrolh, Coa, Cravens, Dawson, Eden, Edgerton, Eldridge, English, Finck, Grider, Griswold, Hall, Harding, Harrington, Benjamin G. Harris, Herrick, Hutch-ins, Philip Johnson, William Johnson, Kalifleisch, Kernan, King, Law, Lazear, Long, Mallory, Marcy, McAllister, Mo-Dowell, McKinney, Middleton, William H. Miller, James R. Morris, Morrison, Nelson, Noble, Odell, John O'Neill, Pen-dleton, Pruyn, Robinson, James S. Rollins, Ross, Scott, John B. Steele, Silles, Strouse, Stuart, Thomas, Voorhees, Wads-worth, Whaley, Wheeler, Fernando Wood, Yeanan-65.

The bill provides that all persons who have served, or may honorably serve in the army or navy of the United States for two years during the present rebellion, or have been or shall be discharged by reason of wounds or disease, shall be entitled to enter eighty acres, or less; and all who have rendered or shall render honorable service as soldiers or sailors in the army or navy, or as laborers therein, shall be entitled to forty acres, of all lands condemned under the confiscation act of July 17, 1862, and its amendments, and of all lands sold under the internal revenue law, and the direct tax law for insurrectionary districts. No distinction of color or race shall be made. In case of the death of the person entering lands, the required proof and payments may be made by his legal representatives.

This bill was not taken up in the Senate.

PREVIOUS ACTION ON THE SUBJECT.

1864, March 18-The House considered Senate bill amendatory of the homestead law and for other purposes, when

Mr. HOLMAN offered the following amendment:

Provided, however, That no non-commissioned officer or private soldier or seaman who shall be at the time of such entry in the military or naval service of the United States, or who shall have been in such service for a period not less than three months and honorably discharged therefrom, shall be required to pay the ten doltars fee of the commis-sions mentioned in this act and the act to which this is an amendment: And provided further, That any such non-commissioned officer, private soldier, or seaman, being a cit-izen of the United States, or having declared his intention to be come a dition thereof in our formity with how cod to become a citizen thereof in conformity with law, and having served in the military or naval service of the United States for a period of not less than two years and been honorably discharged, having made an entry of land in con-formity with this act and the act to which this is amendaformity with this declared the declared which this is all that tory, shall be entitled to the patent therefor at any time after having resided on and cultivated such land for a pe-riod of one year, and the widow, if unmarried, or if no widow, the children of such non-commissioned officer, private soldier, or scaman, shall be entitled to such patent on like condition.

Which was disagreed to-yeas 54, nays 58, as follows:

As follows: YEAS-Messre, James C. Allen, Ancona, Anderson, Augus-tus C. Baldwin, Jacob B. Blair, Bliss, James S. Brown, Chan-ler, Clay, Cafroth, Creswell, Dawson, Denison, Dumont, Edridge, Hall, Harding, Harrington, Charles M. Harris, Herrick, Hohman, Kalbjeisch, Kernan, Law, Long, Mallovy, Marcy, Middleton, Daniel Morris, Morrison, Annos Myers, Noble, Odell, John O'Neill, Orth, Pendleton, Perry, Pruyn, Radford, Samuel J. Randall, James S. Rollins, Ross, Smith, Stebbins, John B. Steele, Stiles, Stroues, Stuart, Wadworth, Ward, Webster, Whaley, Wheeler, Fcoman-54. NArS-Messrs. Alley, Allison, Ames, Ashley, John D. Baldwin, Baxter, Beaman, Boutwell, Brandegee, Broomall, Ambrose W. Clark, Cobb, Cole, Henry Winter Davis, Thomas T. Davis, Dawes, Deming, Donnelly, Driggs, Eckley, Eliot, Fenton, Frank, Gooch, Higby, Hotchkiss, John H. Hubbard,

Julian, Kasson, Kelley, Francis W. Kellogg, Loan, Long-year, Marvin, McBride, McClurg, Samuel F. Miller, Moor-head, Leonard Myers, Norton, Perham, Pike, Price, Alexan-der H. Rice, John H. Rice, Schenck, Scofield, Shannon, Smithers, Spalding, Starr, Stevens, Thayer, Upson, Ellihu B. Washburne, Wilder, Wilson, Windom-58.

The bill then passed.

# Unemployed Generals.

1863, December 13-Mr. FARNSWORTH offered this resolution, which was agreed to :

Resolved, That the Secretary of War be directed to inform this House of the names, number, pay, and emoluments of major generals and brigadier generals of volunteers, and of the regular army, and their staffs respectively, not on duty, and the length of time which has elapsed since each of them has been relieved from duty, and which of them, and how many, are not now on duty in consequence of wounds or lisability incurred in the service.

The Secretary of War replied, substantially, that twenty-five general officers-fourteen major generals and eleven brigadier generalswere entirely unemployed, and that certain officers of their personal staffs were also unemployed with them, numbering twenty-five. At the same time another list was furnished of general officers on duty who were not actually commanding or serving with troops, amounting in number to thirty-nine, making an aggregate of eighty-nine officers reported as being either unemployed altogether or not employed in the command of troops.

A similar resolution was adopted in the Senate, on motion of Mr. TRUMBULL:

Ate, on motion of Mr. TRUMBULL: Resolved, That the Secretary of War be directed to fur-rish the Senate with the names of all the major and brigad-dier generals who are without commands equal to a brigade, stating how long each has been without such command, and whether each has a staff; and if so, how numerous, and of what rank, and what amount of pay, including commu-tations and rations, each, including those of his staff, has been receiving while so without a command; and also that he inform the Senate how many major and brigadier gen-reals are in command of departments, districts, and posts in the loyal States; and whether any necessity exists that requires that these departments, districts, and posts should be commanded by officers of such high rank, with their numerous and expensive staffs.

1864, January 18-Mr. HOLMAN offered this preamble and resolution; which were agreed to:

Whereas this House has been officially informed that a

Whereas this House has been officially informed that a large number of officers of the army, including a number of major and brigadier generals, have been for a long period of time, relieved from active service, while still receiving the full pay pertaining to their rank; and whereas such policy, while embarrassing to the officers so relieved, is manifestly urjust to the coutry, and interfores with just and proper promotions in the army: Therefore, *Resolved*, That, in the judgment of this House, the policy of retaining in the pay of the Government officers who have been indefinitely relieved from active sorvice, not physically disabled by wounds, and who have not been placed on the retired list, ought to be discontinued; and that the Com-mittee on Military Affairs be instructed to inquire what legislation, if any, is necessary to effect a remedy in the premises, and reduce the number of general officers not em-ployed in active service, and report by bill or otherwise. May 12—The House resumed consideration of

May 12-The House resumed consideration of the joint resolution providing that all major and brigadicr generals, who, on the 1st of July, 1864, shall not be in the performance of duty or service corresponding to their respective grades and rank, and who shall not have been engaged in such duty or service for three months continuously next prior to that date, shall then be dropped from the rolls of the army-no officer to be included whose absence from duty shall have been occasioned by wounds received, or disease contracted in the line of his duty while

in the military service or while a prisoner. This to vacate volunteer commissions only.

Mr. Cox moved to add this proviso:

Provided, That when any officer comprehended in this act shall demand a board of inquiry according to the rules and regulations in such cases, and who shall be willing to serve, that such loard shall be forthwith convened; and if said board shall find him competent to command in the rank to which he is entitled, he shall be at once restored to active service with full pay: And provided further, That all officers who have received the thanks of Congress during the present war shall be exempted from the provisions of this act.

Which was disagreed to-yeas 46, nays 69.

Mr. KERNAN offered a substitute providing for a board to examine into the competency, fitness, and efficiency of major and brigadier generals, as described in the joint resolution; and each officer found by the board to be incapable of properly and efficiently following his duties shall be dropped, &c.; which was disagreed to-yeas 50, nays 69.

The joint resolution was then passed-yeas 72, nays 45, as follows :

72, nays 45, as follows:
YEAS-Messrs. Alley, Allison, Ames, Anderson, Arnold, Ashley, John D. Baldwin, Baxter, Beaman, Jacob E. Blair, Boutwell, Boyd, Ambrose W. Clark, Freeman Clarke, Cole, Creswell, Henry Winter Davis, Deming, Driggs, Eliot, Farnsworth, Garfield, Gooch, Grinnell, Higby, Hotchkiss, Jenckes, Julian, Kasson, Kelley, Francis W. Kellogg, Littlejohn, Loan, Longyear, McBride, McClurg, McIndoe, Moorhad, Morta, Marchael Morris, Amos Myers, Leonard Myers, Norton, Charles O'Neill, Orth, Patterson, Perham, Pike, Pomeroy, Price, William H. Rundall, Alexander H. Rice, John H. Rice, Edward H. Rollins, Schenck, Scoffeld, Shannon, Sloan, Smithers, Spalding, Thomas, Tracy, Upson, Van Valkenburgh, William B. Washburn, Webster, Whaley, Wilder, Wilson, Windom, Woodbridge, Framan-72 Marss. Adess. Brown, William G. Brown, Chanler, Cravens, Eden, Edgerton, Eldridge, English, Finck, Grider, Healt, Merrick, Huckins, Philb Johnson, Kalb-Heisch, Kernan, King, Long, Malloy, Marcy, McKinney, Middeld, William H. Miller, James R. Morris, Morrison, Noble, Odell, John O'Neill, Pendleton, Pruyn, Robinson, Ross, Scott, John B. Steele, Stevens, Sittes, Stuart, Wadsworth, Wheler, Fernando Wood-45.
The Senate did not consider the bill.

The Senate did not consider the bill.

# Resolutions relating to the War. ITS OBJECT.

First Session, Thirty-Seventh Congress.

## IN HOUSE.

1861, July 13-Mr. HOLMAN asked leave to offer these resolutions:

Resolved, That the sole object of the Government in its present and future military operations, resulting from the armed resistance to its authority, is, and ought to be, to maintain the integrity of the Union, as established by the Constitution, the enforcement of the laws, and the protec-tion of the constitutional rights of the loyal citizens of every State; and such operations ought not to be suspended until the authority of the Federal Government shall have been firmly established throughout its territorial limits. Resolved, That the Union must be preserved, and that no adjustment of pending difficulties can ever be sanctioned by the Government that is not based on the acknowledged in-

the Government that is not based on the acknowledged in-tegrity of the Union, and the supremacy of the Constitution of the United States.

Leave was not obtained.

1861, July 15-Mr. Allen, of Ohio, offered the following resolutions:

Resolved, That whenever the States now in rebellion against the General Government shall cease their rebellion and become loyal to the Union, it is the duty of the Gov-ernment to suspend the further prosecution of the present war.

Resolved, That it is no part of the object of the present war against the rebellious States to interfere with the insti-tution of slavery therein.

Which were ruled out of order by the Speaker.

August 5-Mr. Allen moved to suspend the rules, that he might offer a similar resolution, but it was disagreed to.

1861, July 22-Mr. Noble, of Ohio, offered this resolution :

Resolved, That the contest now existing between the Government of the United States and the disloyal organizations now existing in certain States which are now waging an unjustifiable war upon the constitutional authority of the Government, should be treated and regarded by all loyal citizens not as a sectional war, nor an anti-slavery war, nor a war of conquest or subjugation, but simply as a war for the maintenance of the Government, the suppres-sion of rebellion, and the preservation of all the rights of all the States full and undiminished, as they were purchased by the blood of the Pevolution of 1776, and secured by all the provisions and compromises of the Federal Constitution, and for no other purpose whatever.

The resolution went over, Mr. BURNETT indicating a desire to debate it. It was not again reached

1861, July 22-Mr. CRITTENDEN offered the following resolution :

Resolved, That the present deplorable civil war has been forced upon the country by the disunionists of the southern States now in revolt against the constitutional Government and in arms around the capital; that in this national emergency Congress, banishing all feeling of mere passion or resentment, will recollect only its duty to the whole country; that this war is not waged upon our part in any spirit of oppression, nor for any purpose of conquest or sub-jugation, nor purpose of overthrowing or interfering with the rights or established institutions of those States; but to defend and maintain the supremacy of the Constitution and to preserve the Union with all the dignity, equality, and rights of the several States unimpaired; that as soon as these objects are accomplished the war ought to cease.

The question being divided,

The House adopted the first clause of the resolution :

Resolved. That the present deplorable civil war has been forced upon the country by the disunionists of the southern States now in revolt against the constitutional Government, and in arms around the capital.

Yeas 122, nays 2, as follows :

YEAS-Messrs. Aldrich, Allen, Alley, Ashley, Babbitt, Goldsmith F. Bailey, Joseph Baily, Baker, Baxter, Beaman, Bingham, Francis P. Blair, ir., Samuel S. Blair, Blake, Buf-finton, Catvert, Campbell, Chamberlain, Clark, Cobb, Colfax, Bingnam, Francis F. Biar, Jr., Samuel S. Biar, Biake, But-finton, Caiver, Campbell, Chamberlain, Clark, Cobb, Colfax, Frederick A. Conkling, Roscoe Conkling, Conway, Cox, Crit-tenden, Curtis, Cutler, Davis, Dawes, Delano, Delaplaine, Diven, Duell, Dunlap, Dunn, Edgerton, Edwards, English, Fenton, Fessenden, Franchot, Frank, Gooch, Granger, Grider, Gurley, Laiglit, Hale, Harding, Harrison, Holman, Horton, Hutchins, Jackson, Julian, Kelley, Wm. Kellogg, Killinger, Lansing, Law, Leary, Lehman, Loomis, Lovejoy, McClernand, McKean, Mallory, Menzies, Moorhead, Anson P. Morrill, Justin S. Morrill, Morris, Nixon, Noble, Nugen, Odell, Olin, Patton, Penalton, Perry, Pike, Pomeroy, Porter, Potter, Alexander H. Rice, John H. Rice, Rid-dle, Edward H. Rollins, James S. Rollins, Shefileld, Shel-labarger, Sherman, Sloan, Smith, Spaulding, John B. Steele, William G. Steele, Stratton, Francis Thomas, Train, Trow-bridge, Upton, Vandever, Van Valkenburgh, Van Wyck, Verree, Vibbard, Wadsworth, Charles W. Walton, E. P. Walton, Ward, Webster, Wheeler, Whaley, Albert S. White, Chillon A. White, Wickliffe, Windom, Woodruff, Worcester, Wright-122.
 The second clause of the resolution :

The second clause of the resolution :

That in this national emergency Congress, banishing all Init in this national emergency Congress, banishing all feeling of mere passion or resentment, will recollect only its duty to the whole country; that this war is not waged upon our part in any spirit of oppression, or for any pur-pose of conquest or subjugation, or purpose of overthrow-ing or interfering with the rights or established institutions of the States, but to defend and maintain the supremacy of the Constitution and to preserve the Union with solution of the Constitution and to preserve the Union with all the dignity, equality, and rights of the several States unim-paired; and that as soon as these objects are accomplished the war ought to cease.

Was adopted-yeas 119, nays 2, as follows:

YEAS-Messrs. Aldrich, Allen, Alley, Babbitt, Goldsmith F. Bailey, Joseph Baily, Baxter, Beaman, Francis P. Blair, Sam-uel S. Blair, Blake, George H. Browne, Buffinton, Calvert,

Campbell, Chamberlain, Clark, Cobb, Colfax, F. A. Conkling, Roscoe Conkling, Cooper, Corning, Cox, Crillenden, Curtis, Cutler, Dawes, Delano, Diven, Duell, Dunlap, Dunn, Edwards, English, Fenton, Fessenden, Fouke, Franchot, Frank, Gooch, Granger, Grider, Gurley, Haight, Hale, Harding, Harrison, Holman, Horton, Jackson, Johnson, Kelley, William Kellogg, Killinger, Law, Lazer, Leary, Lehman, Logan, Loomis, McClernand, Mallory, Menzies, Mitchell, Moorhead, Anson P. Morrill, Justin S. Morrill, Morris, Nixon, Noble, Nugen, Odell, Olin, Patton, Pendleton, Perry, Pike, Pomeroy, Por-ter, Reid, Alexander H. Riles, John H. Rice, Richardson, Robinson, Edward H. Rollins, James S. Kaltins, Shefield, Shellabarger, Sherman, Smith, Spaulding, John B. Steele, William G. Steele, Stratton, Francis Thomas, Train, Trow-bridge, Upton, Vallandigham, Van Horn, Van Valkenburgh, Van Wyck, Verree, Fibbard, Wadsworth, Charles W. Walton, E. P. Walton, Ward, Webster, Wheeler, Whaley, Albert S. White, Chilton A. White, Wickliffe, Windom, Woodruff, Wor-cester, Wright-119. NAYS-Messrs. Potter, Riddle-2.

NAYS-Messrs. Potter, Riddle-2.

1861, July 24-Mr. JOHNSON, of Tennessee, offered substantially the same resolution in the Senate, which was ordered to be printed.

July 25-Mr. Polk, of Missouri, moved to strike out and insert so as to make it read in the first clause; "by the disunionists of the Southern and the Northern States ;" which was disagreed to-yeas 4, nays 33, as follows :

YEAS-Messrs. Johnson of Missouri, Kennedy, Polk, Saulsbury-4.

MATS-Messrs. Anthony, Bingham, Browning, Clark, Col-lamer, Cowan, Dixon, Doolittle, Fessenden, Foster, Grimes, Hale, Harlan, Harris, Howe, Johnson of Tennessee, King, Lane of Indiana, Lane of Kansas, Latham, Morrill, Nesmith Pomeroy, Sherman, Simmons, Sumner, Ten Eyck, Trumbull, Wade, Wilkinson, Willey, Wilmot, Wilson-33.

Mr. TRUMBULL moved to strike out the words, "and in arms around the capital," and the word "subjugation," which was rejected to.

The resolution was then passed-yeas 30, nays 5, as follows:

YEAS-Messrs. Anthony, Browning, Chandler, Clark, Cowan, Dixon, Doolittle, Fessenden, Foot, Foster, Grimes, Harlan, Harris, Howe, Johnson of Tennessee, Kennedy, King, Lane of Indiana, Lane of Kansas, Latham, Morrill, Nesmith, Pomeroy, Saulsbury, Sherman, Ten Eyck, Wade, Wilkinson, Willey, Wilson-30. NATS-Messrs. Breckinridge, Johnson of Missouri, Polk, Powell, Trumbull-5.

1861, July 31-Mr. PENDLETON, of Ohio, offered this joint resolution :

Resolved, That, under the Constitution, the rights, powers, and duties of all the States of the Union are equal; that the Union is founded in this equality; that in order to maintain the Constitution and the Union, this equality must be preserved; that every honest effort to perpetuate the Union must be made in accordance with the Constitution, and with a purpose to maintain this equality; that an at-tempt on the part of the Federal Government to subjugate any of the States, and hold them as territories or provinces, or in any position inferior to that of every other State, or to interfere with their State governments, or with their do-mestic institutions, or to abolish or interfere with slavery within their limits, would he an attempt to destroy this equality, and would, if successful, subvert the Constitution and the Union.

Resolved, therefore, &c., That Congress does hereby sol-emnly declare that hostilities against the so-called confed-erate States shall be so prosecuted as to enforce obedience to the obligations of the Constitution, and the laws passed in accordance therewith; and that they shall not be so prosecuted as to reduce to a position of inferiority any of the States, or to interfere with their State governments, or to abolish slavery within their limits.

Ruled out of order by the Speaker.

# Second Session, Thirty-Seventh Congress.

1861, December 2-Mr. ELIOT offered this joint resolution:

Resolved, &c, 1. That in behalf of the people of these States, we do again solemnly declare that the war in which we are engaged against the insurgent bodies now in arms against the Government has for its object the suppression of such rebellion and the re-establishment of the rightful

authority of the National Constitution and laws over the authority of the National Constitution and laws over the entire extent of our common country. 2. That while we disclaim all power under the Constitution to interfere, by ordinary legislation, with the institutions of the several States, yet the war now existing must be conducted ac-cording to the usages and rights of military service, and that during its continuance the recognized authority of the maxim that the safety of the State is the highest law, subordinates rights of property and dominates over civil relatious. 3. That therefore we do hereby declare that, in our judgment, the President of the United States, as the com-mander-in-chief of our army, and the officers in command judgment, the President of the United States, as the com-mander-in-chief of our army, and the officers in command under him, have the right to emancipate all persons held as slaves in any military district in a state of insurrection against the National Government, and that we respectfully advise that such order of emancipation be issued whenever the same will avail to weaken the power of the rebels in arms, or to strengthen the military power of the loyal forces.

Mr. DUNN moved that it be tabled, which was disagreed to-yeas 56, nays 70, as follows :

disagreed to-yeas 56, nays 70, as follows: YEAS-Messrs. Allen, Ancona, Joseph Baily, Biddle, Fran-cis P. Blair, Jacob B. Blair, Burnham, Caleert, Campbell, Oobb, Conway, Corning, Cox, Cravens, Delano, Dunlap, Punn, Fouke, Haight, Harding, Harrison, Holman, Korton, Yum, Kellogg, Law, Lazear, Leary, Lehman, McPherson, Maynard, May, Menzies, Morris, Noble, Noell, Norton, Odell, Pendleton, Perry, Alexander H. Rice, Richardson, Robin-on, Shejield, Shiel, Smith, John B. Steele, William G. Steele, Benjamin F. Thomas, Train, Trimble, Upton, Vallan-digham, Verree, Ward, Chilton A. White, Woodruff-56. Nave-Messrs. Aldrich, Alley, Arnold, Ashley, Babbitt, Goldsmith F. Bailey, Baker, Baxter, Beaman, Bingham, S. S. Blair, Blake, Buffnton, Clark, Colfax, F. A. Conkling, Roscoe Conkling, Davis, Dawes, Duell, Edgerton, Elwards, Ilot, Fessenden, Franchot, Frank, Goodwin, Granger, Gurley, Hickman, Hooper, Hutchins, Julian, Kelley, Lan-sing, Loomis, Lovejoy, McKean, Mitchell, Anson P. Morrill, Justin S. Morrill, Patton, T. G. Fhelps, Pike, Pomeroy, Por-ter, Potter, John H. Rice, Riddle, Edward H. Rollins, Sar-Spaulding, Stevens, Trowbridge, Van Horn, Van Valken-burgh, Van Wyck, Wall, Wallace, Chas. W. Walton, E. P. Walton, Washburn, Wheeler, Wilson, Worcester-70. Mr. Roscoe ConKLING moved to insert in the

Mr. ROSCOE CONKLING moved to insert in the third resolution, after the word "slaves," the

words, "held by rebels;" which was agreed to. It was then postponed; and December 17th, was referred to the Committee on the Judiciary-yeas 77, nays 57.

Same day, Mr. CAMPBELL offered this resolution :

Resolved, That in legislating to meet the exigencies of the present rebellion, Congress should confiscate the prop-erty, slaves included, of all rebels, and protect the property and rights, under the Constitution and laws, of all loyal citizens.

Which was postponed for the present.

Same day, Mr. STEVENS offered this preamble and resolution :

Whereas slavery has caused the present rebellion in the United States; and whereas there can be no solid and per-manent peace and union in this Republic so long as that institution exists within it; and whereas slaves are now used by the rebels as an essential means of supporting and protracting the war; and whereas by the law of nations it is right to liberate the slaves of an enemy to weaken his power : Therefore,

Be it resolved by the Senate and House of Representatives of the United States of America in Congress assembled. That the President be requested to declare free, and to direct all of our generals and officers in command to order freedom to all slaves who shall leave their masters, or who shall aid

in quelling this rebellion. Szc. 2. And be it further resolved, That the United States pledge the faith of the Union to make full and fair compensation to all loyal citizens who are and shall remain active in supporting the Union for all the loss they may sustain by virtne of this act.

Which, also, was postponed for the present. Dec. 4-Mr. HOLMAN offered the following resolution :

Whereas this House, on the 22d day of July last, by an almost unanimous vote, adopted the following resolution submitted to the House by Hon. J. J. CRITTENDEN, of Kentucky :

Resolved by the House of Representives of the Congress of

the United States, That the present deplorable civil war, has been forced upon the country by the disunionists of the southern States now in revolt against the constitutional Government and in arms around the capital; that in this national emergency Congress, hapishing all feelings of mere passion or resentment, will recollect only its duty to the whole country; that this war is not waged upon our part in any spirit of oppression nor for any purpose of conquest or subjugation, nor for the purpose of overthrowing or in-terfering with the rights or established institutions of the States, but to defend and maintain the supremacy of the Constitution and to preserve the Union with its dignities, equality, and the rights of the several States unimpaired, and that as soon as these objects are accomplished the war ought to cease.

And whereas since that time no event has occurred to change the policy of the Government: Therefore, *Resolved*, That the principles above expressed are sol-emply reafirmed by this House.

Which was laid upon the table-yeas 71, nays 65, as follows:

nays 65, as follows: YEAS-Mossrs. Aldrich, Alley, Arnold, Ashley, Goldsmith F. Bailey, Baker, Baxter, Beaman, Bingham, Francis P. Blair, Samuel S. Blair, Blake, Buffinton, Burnham, Cham-berlain, Clark, Coláx, Frederick A. Conkling, Conway, Davis, Delaplaine, Duell, Edgerton, Edwards, Eliot, Fenton, Fessenden, Franchot, Gooch, Gurley, Hooper, Hutchins, Julian, Kelley, Lansing, Loomis, Lovejoy, McPherson, Moorhead, Anson P. Morrill, Justin S. Morrill, Olin, Patton, T. G. Phelps, Pike, Pomeroy, Potter, John H. Rice, Riddle, Edward H. Rollins, Sargent, Sedgwick, Shanks, Sherman, Sloan, Spaulding, Stevens, Train, Trimble, Trowbridge, Van Hora, Verree, Wall, Wallace, Charles W. Walton, E. P. Walton, Washburne, Wheeler, Albert S. White, Wilson, Worcester-71.

Worcester-71. NAYS-Messrs. Allen, Ancona, Joseph Baily, Biddle, Jacob B. Blair, Cattert, Campbell, Cobb, Corning, Cox, Cravens, Crisfield, Dawes, Delano, Diren, Dunlap, Dunn, English, Fouke, Frank, Goodwin, Granger, Grider, Hanchett, Hard-ing, Harrison, Holman, Horton, William Kellogg, Law, Lazear, Leary, Lehman, McKnight, Maynard, Merzies, Mitchell, Morris, Nixon, Noble, Noell, Norton, Odell, Pen-dleton, Perry, Porter, Richardson, Sheffield, Shellabarger, Skiel, Smith, John B. Steele, William G. Steele, Stratton, Ben-jamin F. Homas, Francis Thomas, Upton, Van Valken-burgh, Wadsworth, Ward, Webster, Chilton A. White, Wicklife, Woodruff, Wright-65.

1862, January 20-Mr. Allen, of Ohio, offered the following resolution :

Resolved, That, in the judgment of this House, no part of the appropriation now made or hereafter made, nor of the taxes now or hereafter haid by Congress should be used in or applied to the prosecution of a war for the purpose of the emancipation of slaves in the slaveholding States of the Union.

Which was laid upon the table-yeas 90, nays 36, as follows :

36, as follows: YEAS-Messrs. Aldrich, Alley, Arnold, Babbitt, Goldsmith F. Bailey, Baker, Baxter, Beaman, Bingham, Francis P. Blair, Blake, Bufinton, Burnham, Campbell, Chamberlain, Clark, Clements, Colfax, Frederick A. Conkling, Roscoe Conkling, Conway, Cutler, Davis, Dawes, Delano, Duell, Dunn, Edgerton, Edwards, Eliot, Fenton, Fessenden, Frank, Gran-ger, Gurley, Hale, Hanchett, Hooper, Huorton, Hutchins, Kelley, Francis W. Kellogg, William Kellogg, Killinger, Lansing, Loomis, Lovejov, McKeau, McKnight, McPherson, Marston, Mitchell, Moorhead, Anson P. Morrill, Justin S. Morrill, Nixon, Olin, Patton, Tinnothy G. Phelps, Pomeroy, Porter, Alexander H. Rice, Riddle, Edward H. Rollins, Sar-gent, Shanks, Sheffield, Sherman, Shoan, Spaulding, Stevens, Stratton, Benjamin F. Thomas, Train, Trimble, Trowbridge, Van Horn, Van Valkenburgh, Van Wyck, Verree, Wall, Wal-hace, Charles W. Watton, E. P. Walton, Washburne, Wheeler, Albert S. White, Wilson, Windom, Worcester-90. NATS-Messrs. Allen, Ancoma, Joseph Baily, Jacoh B. Blair, Knapp, Law, Lazear, Leary, Morris, Noble, Norton, Nygen, Robinson, James S. Rollins, Shiel, John B. Steele, William G. Steele, Yallandigham, Yubbard, Yoonhees, Chilton A. White, Wickliffe, Woodruff, Wright-36. 1862, March 3-Mr. HOLMAN offered the fol-

1862, March 3-Mr. HOLMAN offered the following resolution :

Resolved, That in the judgment of this House, the unfor-tunate civil war into which the Government of the United States has been forced by the treasonable attempt of the southern secessionists to destroy the Union, should not be prosecuted for any other purpose than the restoration of the authority of the Constitution, and that the welfare of the whole people of the United States is permanently in-volved in maintaining the present form of Government under the Constitution, without modification or change.

Which was laid on the table-yeas 59, nays 59, as follows :

YEAS-Messrs. Aldrich, Alley, Arnold, Ashley, Babbitt, Baker, Daxter, Beaman, Bingham, Samuel S. Blair, Blake, Buffinton, Burnham, Camphell, Chamberlain, Clark, Colfax, Frederick A. Conkling, Roscoe Conkling, Conway, Outler, Davis, Delano, Duell, Ely, Fessenden, Franchot, Frank, Hoeper, Hutchins, Francis W. Kellegg, Lansing, Loomia, Lovejoy, McKnight, McPherson, Mitchell, Moorhead, Ansen P. Morrill, Justin S. Morrill, Patton, Pike, Pomeroy, John H. Rice, Riddle, Sargent, Sodgwick, Shanks, Støvens, Trowbridge, Van Wyck, Verree, Wallace, Charles W. Wal-ton, Wheeler, Albert S. White, Wilson, Windom, Worcester -59. - 59

-59. NAYS-Messrs. Joseph Baily, Biddle. Jacob B. Blair, George H. Browne, Wm. G. Brown, Calvert, Clements, Cobb, Corning, Coz, Cravens, Crisfield, Crittenden, Diven, Dunlap, Dunn, Geodwin, Granger, Hale, Hall, Harding, Harrison, Holman, Horton, Johnson, William Kellogg, Knapp, Law, Lazear, Leary, Mallory, Maynard, Menzies, Nixon, Noble, Noell, Norton, Nugen, Odel, Fendleton, Perry, Richardson, Robinson, James S. Rollins, Sheffield, Shellabarger, Smith, John B. Steele, Stratton, Benjamin F. Thomas, Francis Thomas, Trimble, Vibbard, Wadsworth, Webster, Whaley, Wickliffe, Woodruff, Wright-59. Thos Speaker (Mr. Graww) wated awa and the

The Speaker, (Mr. GRow,) voted aye, and the resolution was tabled.

### IN SENATE.

1861, December 20-Mr. WILLEY introduced the following, which was not acted upon :

Resolved, That the existing war, forced upon the country by the instigators of the rebellion without justifiable cause or provocation, was, and is, designed by them to destroy the Union and the Constitution; and their purpose, moreover, was at first, and is now, to disavow and repudiate the fundamental principles of republican government on which our fathers established the Union and the Constitution.

1861, December 16-Mr. TEN EYCK offered this resolution :

Resolved, That the present war is for the Union, according to the Constitution; that its object is to save the former and enforce the latter-was so in the beginning, is now as and enforce the latter—was so in the beginning, is now as carried on, and should be, to the last; that measures, ex-treme and radical, disruptive in themselves, involving in a common fate as well the loyal as disloyal, should not be resorted to; and that in crushing treason—wide-spread and hateful as it is—the Government itself cannot prove traitor to organic law.

# Third Session, Thirty-Seventh Congress. IN HOUSE.

1862, Dec. 4-Mr. STEVENS offered the following resolutions:

Resolved, That this Union must be, and remain, one and indivisible forever.

2. That if any person in the employment of the United States, in either the legislative or executive branch, should propose to make peace, or should accept, or advise the ac-ceptance, of any such proposition on any other basis than the integrity and entire unity of the United States and their Territories as they existed at the time of the rebellion, he will be guilty of a high crime. 3. That this Government can never accept the mediation

or permit the intervention of any foreign nation, during this rebellion, in our domestic affairs.

4. That no two Governments can ever be permitted to exist within the territory now belonging to the United States, and which acknowledged their jurisdiction at the time of the insurrection.

Which were read, and postponed for the present.

Same day, Mr. WICKLIFFE offered this amendment, which, being out of order, was not entertained :

That any officer of the United States, either executive, legislative, or judicial, who is opposed to close the present war upon preserving the Constitution as it is, with all its guarantees and privileges, and the union of the States as established by said Constitution, is unworthy to hold each office and should be discussed for propared from the area office, and should be dismissed or removed from the same.

1862, December 5 - Mr. VALLANDIGHAM offered the following resolutions :

1. Resolved, That the Union as it was must be restored and maintained one and indivisible forever under the Constitution as it is-the fifth article, providing for amendments, included.

2. That if any person in the civil or military service of the United States shall propose terms of peace, or accept or advise the acceptance of any such terms, on any other basis than the integrity and entirety of the Federal Union, and of the several States composing the same, and the Territories of the Union, as at the beginning of the civil war,

A be will be guilty of a high crime. 3. That this Government can never permit the interven-tion of any foreign nation in regard to the present civil war. 4. That the unhappy civil war in which we are engaged was waged in the beginning, professedly, not in any spirit of oppression or for unumers of consult country for the of oppression or for any purpose of conquest or subjugation, of oppression or for any purpose of conquest or sunjugation, or purpose of overthrowing or interfering with the rights or established institutions of those States, but to defend and maintain the supremacy of the Constitution and to preserve the Union with all the dignity, equality, and rights of the several States unimpaired, and was so understood and ac-cepted by the people, and especially by the army and navy of the United States; and that, therefore, wheever shall pervert, or attempt to pervert, the same to a war of con-usest and subjugation, or for the overthrowing or interferquest and subjugation, or for the overthrowing or interfer-ing with the rights or established institutions of any of the States, and to abolish slavery therein, or for the purpose of destroying or impairing the dignity, equality, or rights of any of the States, will be guilty of a flagrant breach of public faith and of a high crime against the Constitution and the Union.

5. That wheever shall propose by Federal authority to extinguish any of the States of this Union, or to declare any of them extinguished, and to establish territorial governments within the same, will be guilty of a high crime against the Constitution and the Union.

6. That whoever shall affirm that it is competent for this House or any other authority to establish a dictatorship in the United States, thereby supersoling or suspending the constitutional authorities of the Union, and shall proceed to make any move toward the declaring of a dictator, will be guilty of a high orime against the Constitution and the Union and public liberty.

Which were laid upon the table-yeas 79, nays 50, as follows:

YEAS-Messrs. Aldrich, Arnold, Ashley, Babbitt, Baker, Baxter, Beaman, Bingham, Samuel S. Blair, Blake, Buffin-ton, Chamberlain, Clark, Colfax, Frederick A. Conkling, Roscoe Conkling, Covode, Cutler, Davis, Dawes, Delano, Dueä, Edgerton, Eliot, Ely, Fenton, Samuel C. Fessenden, Thomas A. D. Fessenden, Fisher, Franchot, Frank, Good-win, Gurley, Hale, Harrison, Hickman, Ilooper, Horton, Hutchins, Julian, Kelley, Franctis W. Kellogg, Loomis, Love-joy, Low, McPherson, Mitchell, Moorhead, Justin S. Mor-rill, Nixon, Pike, Pomeroy, Porter, Potter, John H. Rice, Riddle, Edward H. Rollins, Sargent, Sedgwick, Shanka, Shellabarger, Sherman, Sloan, Spaulling, Stevens, Stratton, Venjan, Vandar, Stave, Torobridge, Van Horn, Van

Riddle, Edward 'H. Rollin's, Sargent, Sedgwick, Shanks,
 Shellabarger, Sherman, Sloan, Spaulding, Stevens, Stratton,
 Benjamin F. Thomas, Train, Trowbridge, Van Horn, Van
 Valkenburgh, Van Wyck, Walker, Wall, Wallace, Washburre, Wilson, Windom, Worcester-79.
 Nars-Messrs, William J. Allen, Ancona, Bailey, Biddle,
 William G. Brown, Clements, Cobb, Couway, Corning, Cox,
 Cravens, Crisfeld, Dunlap, English, Fouke, Granger, Grüder,
 Harding, Holman, Johnson, Knapp, Law, Lazear,
 Harding, Holman, Johnson, Knapp, Law, Lazear,
 Fendletom, Perry, Price, Richardson, Robinson, Shefjield,
 Shiel, Smith, John B. Steele, William G. Steele, Stiles, Vallandigham, Vibbard, Voorhes, Chilton A. White, Wickliffe,
 Woodruff, Wright, Yeanan-50.

1862, December 8-Mr. WRIGHT offered the following joint resolutions :

Resolved, &c., That the rebellion, on the part of the se-ceding States, against the Government and laws of this Union, was deliberately wicked, and without reasonable cause; the compact of the Union being perpetual, no State has the constitutional power to forcibly societ, and that there was no grievance, real or imaginary, upon the part of the second states, for the redress of which the Constitution does not furnish ample remedies. 2. That the rebellion being in contravention of the Con-

2. That the received is the duty of the Government to put stitution and laws, it is the duty of the Government to put it down, without regard to cost, or the consequences that may befail those engaged in it, and all necessary constitu-tional means for this purpose, and this alone, should be fur-nished by the people. That inasmuch as the great and wicked crime invoked the power of the sword, the war should be prosecuted with all the vigor and strength and

means of the Federal Government till rebellion be subdued, and no longer

3. That an honorable peace is desirable; but no peace while armed opposition menaces the Capital and threatens the overthrow of the Union; nor that peace which would be established upon the dismembered fragments of a mighty and prosperous nation; and that man who would entertain peace upon these conditions is a traitor to his country, and unworthy the protection of its laws. 4. That the war was inaugurated solely for the suppres-sion of the rebellion, and the restoration of the Union as it was: that any and all attempts to chonce a discust this

was; that any and all attempts to change or divert this line of policy is a fraud upon the nation, a fraud upon the memory of the gallant men who have sacrificed their lives, and a fraud upon the living soldiers who now stand up as a wall between their loved country and its wicked invaders. 5. That the value of dollars and cents does not enter into

the momentous question of the maintenance of popular liberty, or the preservation of a free Government, any more than the lives and comfort of the traitors who have conspired or leagued together for their destruction.

6. That the Union restored the war should cease, and the seceding States be received back into the Union with all the privileges and immunities to which they were originally entitled.

Mr. LOVEJOY moved to table them, which was rejected—yeas 43, nays 68, as follows :

rejected—yeas 43, nays 68, as follows: Yzas—Messrs. Aldrich, Alley, Eaxter, Beaman, Bingham, Samuel S. Blair, Buffinton, Chamberlain, Frederick A. Conkling, Cutler, Davis, Dawes, Edgerton, Eliot, Ely, Samuel C. Fessenden, Thomas A. D. Fessenden, Gooch, Goodwin, Gurley, Hooper, Horton, Hutchins, Julian, Fran-ris W. Kellogg, Lovejoy, Anson P. Morrill, Justin S. Mor-rill, Pike, Potter, Join H. Rice, Sargent, Sedgwick, Sloan, Spaulding, Stevens, Train, Van Horn, Walker, Wall, Albert White, Wilson, Windom—43 NATS—Messrs. Ancona, Bahlitt, Baily, Jacob B. Elair, Blake, William G. Brown, Burnham, Campbell, Clark, Blake, William G. Brown, Burnham, Campbell, Clark, Hale, Hall, Hording, Holman, Johnson, William Kellogg, Knapp, Lav, Lazear, Loomis, Low, McKnight, Mollovy, Menzies, Moorhead, Morris, Nixon, Noble, Olin, Patton, Pen-deton, Price, Alexander H. Rice, Edward H. Rollins, Shef-field, Shellabarger, Sherman, Smith, John B. Steele, Wm. G. Steele, Straton Benjamin F. Thomas, Francis, Shomas, Trimble, Trowbridge, Wallace, Washburne, Whaley, Chillow a. White, Wickliffe, Woodruff, Worcester, Wright, Yea-man-68. man-68.

They were then postponed.

## First Session, Thirty-Eighth Congress. IN HOUSE.

1863, December 14-Mr. FINCK offered this resolution, which was laid over under the rule:

Whereas, in the opinion of this House, the Federal Gov-ernment is invested by the Constitution of the United States with all necessary power and authority to suppress any resistance to the due execution of the laws thereof, and to employ the army and navy in aid of the civil authority to disperse all armed resistance to the rightful power and jurisdiction of the United States; and whereas, in the judg-ment of this House, the army and navy cannot be right-fully used to subjugate and hold as conquered territory any of the States of this Union : Therefore,

Be it resolved. That in this national emergency Congress will forego all feeling of mere passion or resentment, and will recollect only its duty to the country; that this war should not be waged on our partin any spirit of oppression. nor in any spirit of conquest or subjugation, or for the purborn any spiritor to found use of scoping atom, or for the par-pose of overthrowing or interfering with the rights or es-tablished institutions of the States, but to defend and main-tain the supremacy of the Coustitution and preserve the Union, with all the dignity, equality, and rights of the sev-eral States unimpared; and as soon as these objects are at-tained the unar such the space. tained the war ought to cease.

April 11-The resolution was laid on the table-yeas 81, nays 64, as follows:

Vias SI, Hays SF, as follows: Yias—Messis, Alley, Allison, Ames, Anderson, Arnold, Ashley, John D. Baldwin, Baxter, Beaman, Blaine, Bout-well, Boyd, Broomall, Wm. G. Brown, Ambrose W. Clark, Freeman Clarke, Colb, Cole, Henry Winter Davis, Thomas T. Davis, Deming, Driggs, Dumont, Eckley, Eliot, Farns-worth, Frank, Garfield, Gooch, Grinnell, Higby, Hooper, Hotchkiss, Asahel W. Hubhard, John H. Hubhard, Jenckes, Julian, Kasson, Kelley, Francis W. Kellogg, Orlando Kel-logg, Loan, Marvin, McBride, McClarg, Sanuel F. Miller, Morrill, Daniel Morris, Amos Myers, Leonard Myers, Nov-19 19

ton, Charles O'Neill, Orth, Patterson, Pike, Pomeroy, Price, William II. Randall, Alexander H. Rice, Edward II. Rol-lins, Schenck, Scofield, Shannon, Sloan, Smith, Smithers, Spalding, Starr, Stevens, Thayer, Thomas, Upson, Van Valkenlurgh, Ellihu B. Washburne, William B. Wash-burn, Whaley, Williams, Wilder, Wilson, Windom, Wood-burden-

burn, Whaley, Williams, Wilder, Wilson, Windom, 1992, bridge-Sl. NASE-Messrs. James C. Allen, William J. Allen, Ancona, Augustus C. Baldwin, Francis P. Blair, jr., Bliss, James S. Broon, Chanler, Cluy, Colfroit, Cox, Cravens, Dawson, Den-isson, Eden, Eldridge, English, Finck, Ganson, Grider, Hale, Harding, Harrington, Benjamin G. Harris, Herrick, Hol-man, Philip Johnson, Wm. Johnson, Kernan, King, Knapp, Law, Lazear, Long, Marcy, McAllister, McDowell, McKinn-ney, Müdleton, James R. Morris, Morrison, Nelson, Odell, Pendleton, Perry, Pruym, Samuel J. Randall, Robinson, Steele, Strouse, Stuart, Sweat, Foorhees, Webster, Wheeler, Chilton A. While, Joseph W. White, Winfield, Fernando Wood-64.

1863, Dec. 14-Mr. HOLMAN offered these resolutions :

Resolved, that the doctrine, recently announced, that the States in which an armed insurrection has existed against the Federal Government have ceased to be States of the Union, and shall he held, on the ultimate defeat of that in-Surrection, as Territories or subjugated provinces, and governed as such by the absolute will of Congress and the Federal Executive, or restored to the Union on conditions unknown to the Constitution of the United States, ought to he rebuked and condemned as manifestly unjust to the loyal citizens of those States, tending to prolong the war and to confirm the treasonable theory of secession, and if carried into effect must greatly endanger the public liberty and the constitutional powers and rights of all of the States, by centralizing and consolidating the powers of the government, State and national, in the Federal Executive. 2. That the only object of the war ought to be to subju-

gate the armed insurrection which for the time being suspends the proper relations of certain States with the Federal Covernment, and to re-establish the supremacy of the Constitution; and the loyal citizens of those States. and the masses of the people thereof, submitting to the authority of the Constitution, ought not to be hindered from restoring the proper relations of their respective States with the Federal Government, so far as the same is depend-ent on the voluntary act of the people, by any condition except unconditional submission to the Constitution and except unconditional submission to the Constitution and laws of the United States. In the language heretofore sol-emuly adopted by Congress, the war eught not to be waged on our part for any purpose of conquest or subjuga-tion, or purpose of overthrowing or interfering with the rights or established institutions of those States, but to de-fend and maintain the supremacy of the Constitution and to preserve the Union, with all the dignity, equality, and rights of the several States unimpaired; and as soon as those objects are accomplished the war ought to ccase. 3. That all necessary and proper appropriations of money

3. That all necessary and proper appropriations of money, ought to be promptly made by this Congress for the sup-port of the military and naval forces of the Government, and all measures of legislation necessary to increase and promote the efficiency of the army and navy and to mainpromote the enteries of the any and may and to many tain the public credit ought to be adopted; that through a vigorous prosecution of the war, peace, on the basis of the union of the States and the supremacy of the Constitution, may be the most speedily obtained.

Which were laid upon the table-yeas 88. nays 66, as follows:

nays 66, as follows: Trass-Messrs. Alley, Arnold, Ashley, John D. Baldwin, Baxter, Beaman, Blaine, Blow, Boutwell, Boyd, Brandegee, Broonall, Ambrose W. Clark, Freeman Clarke, Ctay, Cobb, Cole, Creswell, Henry Winter Davis, Dawes, Dixon, Don-nelly, Driggs, Dumont, Eckley, Eliot, Farnsworth, Fenton, Frank, Garfield, Gooch, Grinnell, Higby, Hooper, Hotchliss, Julian, Kasson, Francis W. Kellogg, Orlando Kellogg, Loan, Longvear, Lovejoy, Marvin, McBride, McClurg, McIndoe Samuel F. Miller, Moorhead, Morrill, Daniel Morris, Amos Myers, Leonard Myers, Norton, Charles O'Neill, Ortli, Pat-terson, Perham, Pike, Pomeroy, Price, William H. Randatl, Alexander H. Rice, John H. Rice, Edward H. Rollins, James S. Rollins, Schenck, Scofield, Shannon, Sloan, Snith-ers, Spalding, Stevens, Thayer, Thomas, Upson, Yan Valken-burgh, Ellibu B. Washburne, William B. Washburn, Waley, Williams, Wilder, Wilson, Windom, Woodbridge-Ss. NASS-Messrs, James C. Allen, William J. Allen, Ancona, Bady, Augustus C, Baldwin, Jacoh B. Blair, Eliss, Brooks, William G. Brown, Chanler, Coffroth, Cox, Crarens, Daw-Son, Denison, Eden, Edgerton, Eldridge, English, Finck, Ganson, Grider, Griswold, Hall, Harding, Harrington, Ben-jamin G. Harris, Charles M. Harris, Herrick, Holman,

William Johnson, Kernan, King, Knapp, Lavo, Lazear, Le Blond, Long, Mallory, Marcy, McDowell, McKinney, Wil-liam II. Miller, James R. Morris, Morrison, Nelson, Noble, Odell, John O'Nill, Pendlelom, Robinson, Royrrs, Ross, Scott, Smith, John B. Steele, Stuarl, Voorhees, Wadsworth, Ward, Wheeler, Chilton A. While, Joseph W. While, Win-field, Fernando Wood, Yeaman-66.

Feb. 3-Mr. COFFROTH asked consent to offer this preamble and resolution:

Whereas this once happy and prosperous nation has been The the second s its duty to the whole country; that this war is not waged on their part in any spirit of oppression, or for any purpose of conquest, or subjugation, or purpose of overthrowing or interfering with the rights or established institutions of these States, but to defend and maintain the supremacy of the Constitution, and to preserve the Union with all the the Constitution, and to preserve the Union with all the dignity, equality, and rights of the several States unim-paired; and that as soon as these objects are accomplished the war ought to ccase;" and whereas the President of the United States did, on the 22d of September, A. D. 1862, and on the 1st of January, A. D. 1803, and on the 8th of Decem-ber, A. D. 1863, issue proclamations in direct violation of this resolution; and whereas said proclamations have di-vided the Union people of the North, who at one time were united in their efforts to crush the rebellion: Therefore, in order to unite all the Union people and to carry out the spirit of said resolution and restore the "Union as it was" under the "Constitution as it is."

Be it resolved, That the President of the United States be respectfully requested to withdraw said proclamations, so that all the Union-loving people may again unite to man-tain the supremacy of the Constitution, and to preserve the Union with all the dignity, equality, and rights of the several States unimpaired.

To which Mr. GRINNELL objected.

1863, December 16-Mr. JAMES S. ROLLINS offered this resolution:

Resolved, That, prompted by a just patriotism, we are in favor of an earnest and successful prosecution of the war, and that we will give a warm and hearty support to all those measures which will be most effective in speedily overcom-ing the rebellion and in securing a restoration of peace, and which may not substantially infringe the Constitution and tend to subvert the true theory and character of the Government; and we hereby reiterate that the present deplorable civil war has been forced upon the country by the disunionists now in revolt against the constitutional govern-ment; that in the progress of this war, Congress, banisning all feeling of mere passion or resentment, will recollect only its duty to the whole country; that this war is not waged on our part in any spirit of oppression, nor for any purpose of conquest or subjugation, nor purpose of over-throwing or interfering with the rights or established insti-tutions of those States, but to defend and maintain the supremacy of the Constitution, and to preserve the Union with all the dignity, equality, and rights of the several States unimpaired; that as soon as these objects are accomplished, the war ought to cease.

A motion that the resolution be tabled was disagreed to-yeas 52, nays 115, as follows :

TEAS—Messrs, Alley, Allison, Ames, Anderson, Ashley, Baxter, Beaman, Blow, Boutwell, Boyd, Branderson, Ashley, Baxter, Beaman, Blow, Boutwell, Boyd, Brandergee, Broom-all, Cole, Thomas T. Davis, Dixon, Donnelly, Driggs, Du-mont, Eckley, Eliot, Garfield, Gooch, Grinnell, Hooper, Hotchkiss, Hulburd, Julian, Kelley, Francis W. Kellogg, Loaa, Longycar, Lovejoy, McClurg, McIndoe, Moorhead, Amos Myers, Leonard Myers, Norton, Charles O'Neill, Orth, Perhan, Pomeroy, Price, Schenck, Sloan, Spaulding, Ste-yens, Upson, Van Valkenburgh, William B. Washburn, Whaley Wilder—62 Whaley, Wilder-52. NATS-Messrs. James C. Allen, William J. Allen, Ancona,

Baily, Augustus C. Baldwin, John D. Baldwin, Blaine. Jacob
Blair, Eliss, Brooks, James S. Brown, Chanler, Ambrose
W. Clark, Clay, Cobb, Colyroth, Cox, Cravers, Creswell,
Henry Winter Davis, Dawes, Dawson, Deming, Denison,
Eden, Edgorton, Eldvidge, English, Farnsworth, Fenton,
Finek, Frank, Ganson, Grider, Griswold, Ilale, Hull, Harding, Harrington, Benjamin G. Hurris, Olarles M. Harris,
Pidlip Johnson, Ibn. Johnson, Kasson, Orlando Kellogg,
Kernan, King, Knapp, Law, Lazear, Le Blond, Long, Maller,
loy, Marey, Marvin, McAllister, McBride, McDowell, MoKinney, Middeton, Samuel F. Miller, Wortson, Nelson,
Pidle, Odell, John O'Neill, Patterson, Pendleton, Ferry, Baily, Augustus C. Baldwin, John D. Baldwin, Blaine. Jacob

Pike, Radford, Samuel J. Randall, William H. Randall, Price, Radjora, Samuel J. Randadi, William H. Randali, Alexander H. Rice, Robinson, Reogers, Edward H. Rollins, Jas. S. Rollins, Ross, Scofield, Scolt, Shannon, Smith, Smith-ers, Stebbins, John B. Steele, Win, G. Steele, Stiles, Strouse, Stuart, Sweat, Thayer, Tracy, Forchess, Wadsworth, Ward, Ellihu B. Washburne, Webster, Wheeler, Chillon A. White, Joseph W. White, Wilson, Winfield, Fernando Wood, Wood-bridge, Yeaman-115.

It then went over under the rule; and

May 30-Mr. FERNANDO WOOD moved that it be laid on the table, which was rejected-yeas 27, nays 114, as follows:

27, nays 114, as follows: YEAS-Messrs. Allison, Ames, Anderson, Arnold, John D. Baldwin, Brandegee, Creswell, Thomas T. Davis, Dawes, Donnelly, Eliot, Farnsworth, Garfield, Grinnell, Asahei W. Hubbard, John H. Hubbard, Hulburd, Littlejohn, Schenck, Sloan, Spanlding, Stevens, Elliun B. Washburne, Wilder, Wilson, Windom, Fernando Wood-27. NAVS-Messrs. James C. Allen, Alley, Ancona, Baily, Au-gustus C. Baldwin, Baxter, Beaman, Blaine, Jacob B. Blair, Eliss, Boyd, Brooks, James S. Broun, Chanler, Ambrose W. Clark, Freeman Clarke, Cobb, Coffroth, Cole, Cox, Cravens, Dawson, Eckley, Eden, Edgerton, Eldridge, English, Fenton, Finck, Frank, Ganson, Gooch, Griswold, Harding, Harring-ton, Charles M. Harris, Herrick, Higly, Holman, Hooper, Hotchkiss, Hutchins, Ingersoll, Jenckes, Philip Johnson, King, Knopp, Le Elond, Loan, Long, Longyear, Mallory, Marcy, Mavrin, McAllister, McBride, McClurg, McDouedl, Samuel F. Miller, Morrill, James R. Morris, Morrison, Amos Myers, Leonard Myers, Nelson, Noble, Oddl, Charles O'Hiller, Pomeroy, Price, Pruyn, Radford, Samuel J. Rundall, Wm, H. Randall, Alexander H. Rice, John H. Rice, Rogers, Ed-vard H. Rollins, James S. Rollins, Ross, Scofield, Scoti, Sonnadel, Alexander H. Rice, John Elice, Reid, Wing, Marker, Marker, Madord, Samuel J. Rundall, Wm, H. Randall, Alexander H. Rice, Rogers, Ed-vard H. Rollins, James S. Rollins, Ross, Scofield, Scoti, Sonnad H. Swithers & John P. Kiede, Will Word, Conduction, Scother, Scoler, Scone, Manger, Marker, Mcele, Will, Scother, Steve, Edward H. Rollins, James S. Rollins, Ross, Scofield, Scott, Sonnader H. Swithers & John P. Kiede, Will Word, Conduction, Scother, Scoler, Scoler, Scoler, Scoler, Scoler, Scother, Scoler, Sc H. Randall, Alexander H. Rice, John H. Rice, Rogers, Ed-ward H. Rollins, Jones S. Rollins, Ross, Scofield, Scott, Shannon, Smith, Smithers, John B. Steele, William G. Steele, Stiles, Swead, Thayer, Thomas, Upson, Van Valken-burgh, Voorhees, Wadsworth, William B. Washburne, Webster, Whaley, Wheeler, Joseph W. White, Williams, Woodbridge, Yeaman-114.

On motion of Mr. MORRILL, it was then referred to the Committee on the Rebellious States-yeas 81, nays 66.

# First Session, Thirty-Eighth Congress.

# IN SENATE.

1864, February 23-Mr. CARLILE introduced this joint resolution, which was ordered to be printed:

Resolved, de., 1. That the military power of the Govern-ment can only be rightfully exerted against individuals in arms opposing its authority. That the prosecution of hos-tilities against the citizens of the States in rebellion ought to be for the sole purpose of maintaining the constitutional Union, and for the restoration of the Union upon the basis of the Constitution, leaving to each State the regulation of its own domestic policy, and protecting each and all in the enjoyment of the right of self-government as recognized by the Constitution of the United States.

2. That the President be requested to declare by proclamation, whenever the people of any of the States now re-sisting the authority of the United States shall reorganize their State government by repudiating the ordinances of secession adopted in their name and shall recognize their obligations to the Union under the Constitution, full pardon and annesty to the people of such State, assuring the citizens thereof that all their rights of person and of prop-erty under the Constitution shall be restored to and en-joyed by them; excepting, however, from such pardon and amnesty such persons as shall be designated by the Legis latures of the several States as fit persons to be held for trial before the judicial tribunals of the United States under the laws thereof.

## ITS PROSECUTION. Special Session of Senate-1861.

1861, March 26-Mr. BRECKINRIDGE offered this resolution :

Resolved, That the Senate recommend and advise the removal of the United States troops from the limits of the Confederate States.

Mr. CLINGMAN offered this substitute :

Resolved, That, in the opinion of the Senate, it is expedi-ent that the President should withdraw all the troops of the United States from the States of South Carolina, Georgia,

Florida, Alabama, Mississippi, Louisiana, and Texas, and abstain from all attempts to collect revenues in the said States.

March 27-Mr. TRUMBULL offered this resolution :

Resolved, That, in the opinion of the Senate, the true way to preserve the Union is to enforce the laws of the Union; that resistance to their enforcement, whether under the name of anti-coercion or any other name, is encouragement to disunion; and that it is the duty of the President to use all the means in his power to hold and protect the public property of the United States, and enforce the laws thereof, as well in the States of South Carolina, Georgia, Florida, Mississippi, Alabama, Louisiana, and Texas, as within the other States of the Union.

No vote was taken upon either of these propositions.

# First Session, Thirty-Seventh Congress. IN SENATE.

1861, July 26-The Senate passed the following resolution, offered by Mr. CLARK:

Be it resolved, dc., That we, as representatives of the peo-ple and States, respectively, do hereby declare our fixed de-termination to maintain the supremacy of the Government and the integrity of the Union of all these United States; and to this end, as far as we may do so, we pledge the entire resources of the Government and people, until all rebels shall submit to the one and cease their efforts to destroy the other.

The vote was-yeas 34, nays 1, as follows:

YEAS-Messrs. Anthony, Bingham, Browning, Chandler, Clark, Collamer, Cowan, Dixon, Doolittle, Fessenden, Foot, Foster, Grimes, Hale, Harris, Howe, Johnson of Tennessee, King, Lane of Indiana, Lane of Kansas, Latham, McDougall, Morrill, Nesmith, Pomeroy, Saulsbury, Sherman, Sinnmons, Summer, Ten Eyek, Thomson, Trumbull, Wade, Willey-34. NATS-MR. Breckinridge-1.

## IN HOUSE.

July 15-Mr. McCLERNAND offered the following resolution :

Whereas a portion of the people of the United States, in violation of their constitutional obligations, have taken up arms against the National Government, and are now striving, by aggressive and iniquitous war, to overthrow it and break up the Union of these States: Therefore, Resolved, That this House hereby pledges itself to vote

for any amount of money and any number of men which may be necessary to insure a speedy and effectual suppres-sion of such rebellion and the permanent restoration of the Federal authority everywhere within the limits and jurisdiction of the United States.

The resolution was agreed to-yeas 121. nays 5, (Burnett, Grider, Norton, Reid, and Benjamin Wood.)

July 22-Mr. VANDEVER asked consent to offer this resolution :

Resolved, That the maintenance of the Constitution, the preservation of the Union, and the enforcement of the laws, are sarred trusts which must be executed; that no disaster shall discourage us from the most ample performance of this high duty; and we pledge to the country and the world the employment of every resource, national and individual, for the suppression, overthrow, and punishment of rebels in arms.

Consent was first refused, but it was afterwards offered, and by unanimous consent, considered and agreed to.

July 22-Mr. WRIGHT, of Peunsylvania, offered this resolution :

Resolution : Resolved, That the reverses of the army of the United States yesterday, the 21st instant, at Bull's Run, caused by the rebel army, have in no manner impaired the ultimate success of our arms, but that the cause of human liberty, the preservation of the Union, and the maintenance of the laws pervading the hearts and affections of more than twenty millions of people, are a sure and certain guarantee that the fag of our country shall be upheld, and the Union preserved to the people; that we call on all loyal people of the Union to respond manfully to the demand of the Government in furnishing men and money, and to stand together in its support with their lives and fortunes.

The resolution was laid on the table, witha call of the yeas and nays.

# Second Session, Thirty-Seventh Congress. IN SENATE.

#### 1862, May 26-Mr. SUMMER offered this resolution :

Resolved, That in the prosecution of the present war for the suppression of a wicked rebellion, the time has come for the Government of the United States to appeal to the loyalty of the whole people everywhere, but especially in the rebel districts, and to invite all, without distinction of ealer or place to myster their loyalty mainfeat hu resident the rebel districts, and to invite all, without disfinction of color or class, to make their loyalty manifest by ceasing to fight or labor for the rebels, and also by rendering every assistance in their power to the cause of the Constitution and the Union, according to their ability, whether by arms, or labor, or information, or in any other way; and, since protection and allegiance are reciprocal duties, dependent upon each other, it is the further duty of the Government of the United States to maintain all such loyal people, without distinction of color or class, in their rights as men, according to the principles of the Declaration of Independ-ence. ence.

#### June 18—Mr. GRIMES offered this resolution :

Resolved. (as the opinion of the Senate.) That it is the right and duty of the Government to call all loyal persons within the rebellious States to its armed defence against the traitors who are seeking its overthrow.

#### IN HOUSE.

1862, December 5-Mr. MORBILL offered the following resolution :

Resolved, That at no time since the commencement of the Resolved, That at no time since the commencement of the existing rebellion, have the forces and materials in the hands of the executive department of the Government been so ample and abundant for the speedy and triumphant termination of the war as at the present moment; and it is the duty of all loyal American citizens, regardless of minor differences of opinion, and especially the duty of every department of the Government—the legislative branch included—as a unit to corriculty and minord warding down the assess. as a unit, to cordially and unitedly strike down the assas-sins, at once and forever, who have conspired to destroy our Constitution, our nationality, and that prosperity and free-dom of which we are justly proud at home and abroad, and which we stand pledged to perpetuate forever.

Which was agreed to-yeas 105, nays 1, as follows:

follows: YEAS-Messrs. Aldrich, Arnold, Ashley, Babbitt, Baily, Baker, Baxter, Beaman, Bingbam, Blake, William G Brown, Buffinton, Chamberlain, Clark, Cloments, Cobb, Col-fax, Frederick A. Conkling, Roscoe Conkling, Corning, Coz, Cracens, Cutler, Davis, Dawes, Delano, Duell, Dunlap, Dunn, Edgerton, Eliot, Ely, Fenton, Samuel C, Fessenden, Thomas A. D. Fessenden, Fisher, Franchot, Frank, Gooch, Goodwin, Granger, Grider, Gurley, Hale, Hall, Harrison, Hickman, Holman, Hooper, Horton, Hutchins, Julian, Kel-ley, Francis W, Kellogg, Van, Kellogg, Leary, Loouis, Love-joy, Low, McPherson, Maynard, Mitchell, Moothead, Justin S. Morrill, Nixon, Noble, Nugen, Odell, Olin, Patton, Perry, Pike, Pomeroy, Porter, Price, John H. Rice, Riddle, E. H. Rollins, Sargent, Sedgwick, Segar, Shanks, Sheffield, Shella-arger, Smith, Spalding, Join B. Steele, Wm. G. Steele, Stevens, tratton, Benjamin F. Thomas, Francis Thomas, Train, Trowbridgo, Van Horn, Vau Valkenburgh, Van Wyck, Walkace, Wallace, Whaley, Wilson, Windom, Woodruff, Wor-cester, Wright-105. NAR-Mr. William J. Allen-1. Mr. Cox offered the following resolution :

Mr. Cox offered the following resolution :

Resolved, That the word "assassins," used in the resolu-tion this day offered by the member from Vermont, [Mr. MORELL,] is intended by this House to include all men, whether from the North or the South, who have been instrumental in producing the present war, and especially those in and out of Congress who have been guilty of fla-grant breaches of the Constitution, and who are not in favor of the establishment of the Union as it was and the Constitution as it is Constitution as it is.

Which was laid upon the table-yeas 85, nays 41, as follows:

YEAS-Messrs. Aldrich, Arnold, Ashley, Babbitt, Baily, Baker, Baxter, Beaman, Bingham, Jacob B. Blair, Samuel S. Blair, Blake, Buffinton, Chamberlain, Clark, Colfax, Frederick A. Conkling, Roscoe Conkling, Conway, Covode, Cutler, Davis, Dawes, Duell, Dunn, Edgerton, Eliot, Ely,

Fenton, Samuel C. Fessenden, Thomas A. D. Fessenden, Fisher, Franchot, Frank, Goodwin, Hale, Harrison, Hickman, Hooper, Horion, Hutchins, Julian, Kelley, Leavy, Loomis, Lovejoy, Low, McFherson, Mayuard, Mitchell, Moerhead, Justin S. Morrill, Olin, Patten, Pike, Pomeroy, Porter, Potter, John H. Rice, Riddle, Edward H. Rollins, Sargent, Sedgwick, Segar, Shanks, Sheffield, Shellabarger, Sherman, Sloan, Spaulding, Stevens, Stratten, Benjamin F. Thomas, Francis Thomas, Train, Trowbridge, Van Horn, Van Valkenburgh, Yan Wyck, Walker, Wallace, Washburne, Wilson, Windom, Worcester—85. Worcester-85.

Worcester-85. NAYS-Messrö. William J. Allen, Ancona, William G. Brown, Clements, Cobb. Corning, Cox, Dunlap, English, Grauger, Grider, Hall, Harding, Holman, Johnson, Lazear, Mallory, Menries, Noble, Norton, Nugen, Pendleton, Perry, Price, Richardson, Robinson, Shiel, Smith. John B. Steele, Wm. G. Steele, Stiles, Vallandigham, Vibbard, Voorhees, Ward, Whaley, Chilton A. While, Wickliffe, Woodruff, Wright, Yeaman-41.

1862, December 15-Mr. CONWAY offered the following resolutions:

Resolved, That freedom and slavery cannot co-exist in the same Government without producing endless strife and civil war; that "a house divided against itself cannot stand;" and that "this nation must be all free or all slave." 2. That the American Union consists of those States

which are now loyal to the Federal Constitution.

3. That a restoration of the Union, as it existed prior to the rehellion, would be a greater calamity than the rehel-lion itself, since it would give new life to the "irrepressible conflict," and entail upon the nation another cycle of bitter contention and civil war.

4. That the seceded States can only be put down, if at all, by being regarded as out of constitutional relations with the Union, and by being assailed upon principles of ordinary warfare as between separate nations.

5. That if any person in the employment of the United States, in either the legislative or executive branches, should propose to make peace, or should accept or advise the acceptance of any such proposition on any basis which would restore the slave power to its former supremacy in the Government, or by any new compromise or amendment

the Government, or by any new compromise or amendment of the Constitution recognize slavery as an element of power, such person will be guilty of a high crime. 6. That the superior resources and military provess of the North in the struggle are beyond dispute, and that the question of its success turns not upon its relative ability, but on the fitness of its chief Executive Magistrate to give effect to its power.

7. That it is unsafe to intrust the execution of any system of administration to persons who are not in cordial sym-pathy therewith, and that no change of policy in the con-duct of the war is more than nominal which is not accompanied by a complete change in the personnel of the executive department.

8. That it is a matter for serious reflection whether an-other election for President must not supervene before the rightful authority of the nation can be established, and whether, in the mean time, it is not a flagrant waste of our

whether, in the mean time, it is not a higrant waste of our energies to continue the war.

 That unless the army of the West shall have swept through the valley of the Missispipt to its mouth, and the army of the Potomac annihilated the legions of Lee and Jackson, thus subverting the military power of the rebel-lion, within a reasonable time, the best interests of the country and humanity will require a cessation of hostilities

10. That the States of the North, composing the American nation and wielding its power, must ever remain one and indivisible on the basis of freedom for all, without distinction of race, color, or condition; that their mission must ever be to extend their own civilization over the entire continent; and that whatever derangements, difficulties, checks, or defcats they may encounter, they must forever cherish and pursue the inspiring idea of nationality and continental dominion.

Which were laid upon the table-yeas 132, nay 1, as follows:

nay 1, as follows: YEAS-Messrs. Aldrich, William J. Allen, Alley, Ancona, Babbitt, Baily, Baker, Beaman, Biddle, Bingham, Samuel S. Biair, Blake, William G. Brown, Buffinton, Calvert, Cham-berlain, Clark, Clements, Cobb, Colfax, Frederick A. Conk-ling, Roscoe Conkling, Cox, Crisfield, Crittenden, Cutler, Dawes, Delano, Delaplaine, Duell, Dunlap, Dunn, Edgerton, Edwards, Eliot, English, Fenton, Samuel C. Fessenden, Thomas A. D. Fessenden, Fisher, Fouke, Franchot, Frank, Gooch, Goodwin, Granger, Grüder, Gurley, Haight, Hall, Harding, Harrison, Hickman, Holman, Hooper, Hudchins, Julian, Kelley, William Kellogg, Korrigan, Killinger, Lan-sing, Leary, Loounis, Lovejoy, Low, McKnight, McPherson, Mallory, Maynard, Menzies, Mitchell, Moorhead, Anson P.

Morrill, Justin S. Morrill, Morris, Nixon, Noble, Noell, Nor-ton, Odell, Patton, Pendleton, Pike, Pomeroy, Porter, Potter, Price, Alexander H. Rice, John H. Rice, Richardson, Rid-dle, Robinson, Edward H. Rollins, Sargent, Sedgwick, Se-gar, Shanka, Shefiled, Shellabarger, Shiel, Sloan, Smith, Spaulding, John B. Steele, Stiles, Stratton, Benjamin F. Thomas, Francis Thomas, Train, Trimble, Trowbridge, Val-landigham, Van Horn, Van Valkenburgh, Verree, Wads-worth, Walker, Wall, Wallace, Ward, Washburne, Whaley, Albert S. White, Chillon A. White, Wichkiffe, Wilson, Win-dom, Woodruff, Worcester, Wright, Yeaman-132. NAY-Mr. Conway-1.

1863, January 5-Mr. BLAKE offered the following resolution:

Resolved, That this House earnestly desires the most speedy and effectual measures taken to put down the rebellion; that any propositions for peace or cessation of hostilities at this time on any terms other than an unconditional submission of the rebels now in arms against the Government to the requirements of the Constitution and laws, would be pusillanimous and traitorous; that the members to the people of the United States that they will cheerfully co-operate with the President as Commander-in-Chief of the Army and Navy in any measures he may deem proper, sanctioned by the Constitution and the laws of civilized warfare, to strengthen the military power of our gallant soldiers in the field defending the Government, and to weaken that of the enemy laboring to destroy it.

2. That the only alternative Government can or ought at this time to offer to rebels, is, submit or be conquered.

Which was considered, but no action taken thereon.

January 8-Mr. HOLMAN offered the following resolution :

Resolved, That the duty of maintaining the integrity of the Union of the States under the present form of govern-ment, with the limitations of the Constitution unimpaired, is most sacred and obligatory, and no proposition tending to destroy the Union, or violate the obligations of the Consti-tution, can rightfully be entertained or considered by the representatives of the people in any of the departments of the Government.

2. That the free and unrestricted navigation of the Mis-sissippi river must be restored and maintained as the common and absolute right of the people of all of the States, and the duty to vindicate the same against every effort to impair it, is imperative, and cannot, under any circum-stances, be abandoned by the Government of the United States.

The consideration of which was postponed until the 14th instant, but not again called up.

# First Session, Thirty-Eighth Congress. IN HOUSE.

# 1864, January 7-Mr. Amos MyERS offered

the following, which were referred to the Select Committee on the Rebellious States :

Whereas, in the opinion of this House, the Federal Government is invested by the Constitution of the United States with all necessary power and authority to suppress any resistance, whether armed or unarmed, to the rightful power and jurisdiction of the United States : Therefore,

*Be it resolved*, That in this national emergency Congress will forego all feeling of mere passion, except that which loyalty dictates, all resentment except such as is due to treason; and that this war of national self-defence against armed rebels, insurrectionary traitors, and sympathizing abettors, should be waged on our part until such rebels and traitors are conquered into love for the Union, and made obedient to the Constitution and laws of the United States, and take the oath of allegiance to the country, and of sub-mission to the emancipation proclamation, and the procla-mation of December 8, 1863; and when those objects are accomplished, the leading rebels and chief traitors should be hung, and the war cease.

1864, January 18-Mr. GREEN CLAY SMITH offered this preamble and resolution :

Whereas a most desperate, wicked, and bloody rebellion exists within the jurisdiction of the United States, and the

Safety and security of personal and national liberty depend upon its utter and absolute extinction: Therefore, *Resolved*, That it is the political, civil, moral, and sacred duty of the people to meet it, fight it, crush it, and forever destroy it.

Mr. JAMES C. ALLEN moved to lay them on the table; which was disagreed to-yeas 26, nays 102.

' The YEAS were:

Messes, James C. Allen, Ancona, Brooks, Chanler, Dem-ing, Denison, Eden, Edgerton, Eldridge, Herrick, William Joinson, Knapp, Long, Marcy, McDowell, McKinney, Wm. H. Miller, Prendleton, Robinson, Ross, Stiles, Strouse, Voor-hees, Chilton A. White, Fernando Wood, Ycaman-26.

The resolution was then agreed to-yeas 112, nays 16, as follows:

nays 16, as follows:
YEAS — Messrs. Alley, Allison, Ames, Arnold, Ashley, Baily, Augustus C. Baldwin, John D. Baldwin, Baxter, Blaine, Francis P. Blair, jr., Jacob E. Blair, Boutwell, Boyd, Brandegee, Broomall, James S. Brown, William G. Brown, Ambrose W. Clark, Freeman Clarke, Cole, Cravens, Cres-well, Dawes, Deming, Dixon, Donnelly, Driggs, Eckley, El-dridge, Eliot, English, Farnsworth, Fenton, Frank, Ganson, Garfield, Gooch, Grinnell, Griswold, Hale, Harding, Higby, Holman, Hooper, Hotchkiss, Aschel W. Hubbard, Hutchins, Jenckes, Juliau, Kasson, Kelley, Francis W. Kellogg, Or-lando Kellogg, Kernan, Loan, Longyear, Lovejoy, Marvin, McBride, McClurg, McIndoce, Middleton, Samuel F. Miller, Moorhead, Morrill, Daniel Morris, Amos Myers, Leonard Myers, Nelson, Odell, Charles O'Neill, Orth, Patterson, Pike, Pomeroy, Price, Radford, William H. Randall, Alex-ander H. Rice, John H. Rice, Rogers, Edward H. Rollins, James S. Rollins, Schenck, Scofield, Shannon, Smith, Smith-sers, Spalding, Stebhorne, Strouse, Stuart, Sueat, Thayer, Thomas, Tracy, Upson, Van Valkenburgh, Wads-worth, Ellihu B. Washburne, William B. Washburn, Web-ster, Whaley, Wheeler, Williams, Wilder, Wilson, Windom, Winfeld, Woodbridge-112.
NAYS-Messrs. J. C. Allen, Ancona, Denison, Benjamin G. Harris, Long, Marcy, McDowell, William H. Miller, Morri-son, John O'Neill, Pendletom, Robinson, Stiles, Voorhees, Chultor A. White, Fernando Wood-26.
1864, Feb. 8-Mr. Wan. G. BROWN offered

1864, Feb. 8-Mr. WM. G. BROWN offered the following resolutions, which were laid over under the rule :

Whereas our beloved country, our highly cherished in-stitutions, Constitution, and Union of the States, are all imperilled by a causeless and wicked rebellion: Be it therefore.

Resolved, That it is the duty of every loyal citizen to give to the Government, and to the agents in its employ, both in the cabinet and in the field, all the legitimate aid and comfort in his power in their efforts to put down such rebellion.

2. That, as the rebels began the war, we will prosecute it until the last insurgent is disarmed and the authority of the United States acknowledged over every foot of ground belonging to the Republic.

3. That in the prosecution of the war we will use all the military power of the Government, but will combine with it all the means of conciliation calculated to give to the Government and country an honorable and lasting peace.

4. That it is the duty of the Government, so far as it is in is power to give equal protection to all loyal citizens with-out reference to their locality, whether residing within the second or loyal States; and one of the strong incentives to a vigorous prosecution of the war is to rescue our loyal brethren of the rebellious States from the domination of a military despotism.

1864, Feb. 8-Mr. JACOB B. BLAIR offered the following preamble and resolutions, which were laid over under the rule:

Whereas the present deplorable civil war was inauguwhereas the present deplorable etril war was inaugu-rated and is still carried on by a few desperate but daring men who, without any cause whatever, have not only filled the land with widows and orphans and caused almost un-told millions of treasure to be spent, but have put in peril the very life of that Government which never deprived them of one solitary right, but which was so mild and be-defore it means the pure by the blocking it conferred them of one solitary right, but which was so mild and be-neficent it was only known by the blessings it conferred. And whereas Jefferson Davis, the chief of rebels, is reported to have said in a speech delivered in Jackson, Mississippi, in December, 1862: "My only wonder is that we consented to live so long a time in association with such miscreants (referring to the people of the North) and have loved a Government rotton to the core. Were it ever to be pro-posed again to enter into a union with such a people I could no more consent to do it than to trust myself in a den of thieves." And whereas this same high official in the great synagogue of rebeldom has repeatedly since, in his messages to the Rebel Congress, utterly repudiated the

idea of ever ceasing his wicked designs and returning to his allegiance to the Government, whose Constitution and laws he has trampled under foot; and has also declared that no compromise would be entertained by him, or those that no compromise would be entertained by him, or those he represents, that did not secure to the States in rebellion their independence and final separation from the United States. And whoreas Alexander H. Stephens, the Vice President of the so-called Southern Confederacy, is reported to have said in a speech delivered in the month of July, 1863, at Charlotte, North Carolina, "As for reconstruction, such a thing was impossible; such an idea must not be tolerated for an instant. Reconstruction would not end the war, but would produce a more horrible war than that in which we are now engaged. The only terms on which we can obtain permanent peace is final and complete separa-tion from the North. Rather than submit to anything short of that, let us all resolve to die like men worthy of free-dom." And whereas John Letther in one of his message dom." And whereas John Lotcher, in one of his messages to the rebel legislature of the State of Virginia, declared : "The alliance between us is discoved, (meaning between the United States and the southern States,) never, I trust, to be renewed, at any time, under any conceivable state of circumstances." And whereas the *Richmond Enquire*, one of the organs and advocates of this imaginary South en Confederacy, in its issue of January 9, 1863, says : "Separation is inevitable. War has failed to prevent it, Peace cannot stop it. An armistice with propositions for reconstruction by constitutional amendments of conventions of States would very soon reveal the fact that separation so is states would very soon reveal the lact that separa-tion was final; and so far as one generation can speak for its successors, it is eternal." And whereas the *Richmond Dispatch* of January 10, 1863, another organ of the leaders of this wanton and upprovoked rebellion, said in response to a peace and reunion speech, delivered in New York by to a peace and reunion speech, delivered in New York by the editor of the *Express*, "That we assure him that the people of the Confederate States would infinitely prefer being the vassals of France or England; nay, they would prefer to be serfs of Russla, to becoming in any manner whatever associated politically or otherwise with the Yankoe States." And further, "that President Davis ex-pressed the sentiment of the entire Confederacy in his speech the other night, (in Richmond.) when he said 'the people would sconer unite with a nation of hyenas than with the detestable Yankee nation. Anything but that, English colonization, French vassalage, Russian serfdom— all, all are preferable to any association with the Yan-kees.'" And whereas the *Richmond Sentirel*, still another kees.''' And whereas the Richmond Sentizel, still another advocate of this new-fledged Confederacy, in its comments on the proceedings of what is known as the Frank Pierce meeting, held at Concord, New Hampshire, on the 4th of July, 1863, says, "Do the New Hampshire Democrats sup-pose for one moment that we could so much as think of a reunion with such a people? Rather tell one to be wedded to a corpse; rather join hands with the fiend from the pit, The blood of many thousands of martyrs is between us, A thousand feelings of horror repel the idéa of a renewal of affection.'' And whereas the Richmond Whig, another mouth-piece of treason and of crime, in its issue of the 10th of January, 1863, speaking of those who are opposed to of January, 1863, speaking of those who are opposed to breaking up the Union bequeathed to them by their fathers, says, "They are by nature menials, and fitted only for menial duties. They are in open and flagrant insurrection against their natural lords and masters, the gentlemen of the South. In the exercise of their assumed privileges they deport themselves with all the extravagant airs, the insolence, the cruelty, the cowardice and love of rapine, which have ever characterized the revolt of slaves. The former lenicacy of their masters only serves to aggravate the ferocity of their nature. When they are again reduced to subjection, and taught to know their place, we must take care to put such trammels about them that they will never have an opportunity to play their tricks again." It is,

therefore, 1. Resolved, That any attempt on the part of the Govern-ment of the United States to conciliate the leaders of the would be but an attempt on the one hand to rob the gallows of its own, and on the other to humiliate and bring into utter contempt this Government in the estimation of the civilized world.

2. That every State which has ever been is still a State in the Union, and that when this rebellion shall have been put the Union, and that when this rebellion shall have been put down each of the so-called secoding States will have the same rights, privileges, and immunities under the Constitu-tion as any of the loyal States, except so far as the holding of African slaves in bondage is affected by the President's proclamation of the 1st of January, 1863, the action of Con-gress on the subject, or the events of the war. 3. That this House utterly repudiate the dortrine ad-vanced by some, that the so-called seceding States have come Territories thereof, or stand in the relation of foreign powers at war therewith.

powers at war therewith.

resolutions:

Resolved, That the present war which this Government is carrying on against armed insurrectionists and others, banded together under the name of "Southern Confederacy," was brought on by a wicked and wholly unjustifiablo rebellion, and all those engaged in or aiding or encour-aging it are public enemies, and should be treated as such. 2. That this rebellion shall be effectually put down; and that, to prevent the recurrence of such rebellion, in future,

the causes which led to this one must be permanently removed.

a. That in this struggle which is going on for the saving of our country and free government, there is no middle ground on which any good eitizen or true patriot can stand; noutrality, or indifference, or anything short of a hearty support of the Government, being a crime where the ques-tion is between hereity and treason. tion is between loyalty and treason.

A division of the question having been called, The first resolution and the first branch of the second were agreed to. The second branch of the second resolution, "And that, to prevent the recurrence of such rebellions in future, the causes which led to this one must be permanently removed," was agreed to-yeas 124, nays none, as follows :

nays none, as ionows: YEAS-Messrs. James C. Allen, Alley, Allison, Ames, Ancona, Anderson, Arnold, Baily, Augustus C. Baldwin, John D. Baldwin, Baxter, Francis P. Blair, jr., Jaeob B. Bloir, Bliss, Blow, Boutwell, Boyd, Brandegee, Brooks, Ambrose W. Clark, Freeman Clarke, Clay, Cobb Colfroll, Cole, Coz, Creswell, Henry Winter Davis, Dawes, Dawson, Deming, Denison, Discon, Donnelly, Driggs, Dumont, Eck-ley, Eden, Eldridge, Eliot, Farnsworth, Fenton, Finck, Frank, Ganson, Grider, Griswold, Inle, Herrick, Highy, Versen Husper Huchkiss Asahel W. Hubbard, John H. Frank, Ganson, Grider, Griswold, Hale, Herrick, Highy, Holman, Hooper, Hotchkiss, Asabel W. Hubbard, John H. Holman, Hooper, Hotchkiss, Asahel W. Hubbard, John H. Hubbard, Hutchins, Jenckes, Julian, Kelley, Orlando Kel-logg, Kernan, King, Knapp, Law, Loan, Long, Lovejoy, Marcy, Marvin, McBride, McClurg, McIndoe, Samuel F. Miller, Wm. H. Müller, Moorhead, Morrill, Daniel Morris, Morrison, Amos Myers, Leonard Myers, Nelson, Noble, Norton, Charles O'Neill, John O'Neill, Patterson, Perham, Pomeroy, Price, Radford, Samuel J. Randall, William H. Randall, John H. Rice, Ross, Schenck, Scofield, Scott, Shan-on, Sloan, Smithers, Spalding, Starr, Stebbins, John B. Steele, Wm. G. Steele, Stevens, Niles, Stronse, Stuart, Sweat, Kora-Thayer, Thomas, Tracy, Upson, Van Valkenburgh, Foor-hees, William B. Washburn, Whaley, Williams, Wilder, Wilson, Windom, Winfield, Woodbridge-124.

NAYS-None.

The third resolution was agreed to-yeas 109, nays none, as follows:

YEAS-Messrs. James C. Allen, Alley, Allison, Ames, An-derson, Arnold, Baily, Augustus C. Baidwin, John D. Bald-win, Baxter, Jacob B. Blair, Boutwell, Boyd, Brandegee, Brooks, Ambrose W. Clark, Freeman Clarke, Cobb, Cole, Brows, Americes W. Clark, Freeman Charke, Cond. Cond. Coz, Creswell, Henry Winter Davis, Dawes, Demiog, Deni-son, Dixon, Donnelly, Driggs, Dumont, Eckley, Eldridge, Eliot, Fenton, Finck Frank, Ganson, Garfield, Grinnell, Griswold, Hale, Harrington, Iligby, Holman, Hotchkiss, Asahel W. Hubbard John H. Hubbard, Hutchins, Jenekes, Union Vellow Ferreig W. Kellow, Oxford W. Killow, For Asalet W. Hubbard John H. Hubbard, Hatchins, Jeneses, Julian, Kelley, Francis W. Kellogg, Orlando Kellogg, Kr-nan, King, Knapp. Loan, Lovejoy, Marvin, McBride, Mc Clurg, McIndoe, McKinney, William H. Miller, Moorhead, Morrill, Daniel Morris, Morrison, Annos Myers, Leonard Myers, Nelson, Noble, Norton, Charles O'Neill, John O'Neill, Pottorre Bachcone Dane wer Baise Division 2010 Myers, Nelson, Noble, Norton, Charles O'Neill, John O'Neill, Patterson, Perham, Pomeroy, Price, Radford, Samuel J. Randall, William H. Randall, John H. Rice, Rogers, Ross, Schenck, Scofield, Shannon, Sloan, Smithers, Spalding, Starr, Stebbins, John B. Steele, William G. Steele, Strouse, Stuart, Thayer, Thomas, Upson, Van Valkenburgh, Ellihu B. Washburne, William B. Washburn, Whaley, Williams, Wilder, Wilson, Windom, Winfield, Woodbridge-109. NATS-None.

1864, March 14-Mr. JOHN H. RICE offered the following resolutions, which were laid over under the rule :

Whereas the vital principle of our national life emanated from and survives in the grand and Heaven-inspired declara-tion "that all men are created equal; that they are endowed by their Creator with certatinialiering these are below we by their Creator with certatinialiering these are life, liberty, and the pursuit of happiness;" and whereas the Government of the United States was estab-lished and the Constitution adopted in the earnest desire and confident expectation that both would speedily and to devise finally operate in harmony with said "declaration," and

1864, Feb. 29-Mr. SCHENCK offered these | thereby secure to all native and naturalized citizens equal civil rights and privileges, regardless of all conditions of birth, race, descent, worldly possessions, or religious faith; and whereas the system of American slavery has been and is utterly subversive and destructive of the aforesaid principles, desires, and expectations, and has been the fruitful progenitor of all manner of evils—social, moral, and politi-cal—producing cruelty and oppression to the slave, de-moralization and degradation to the free laborer, and brutalization and arrogance in the slave-driver and the slave-master, and has finally culminated in robbery and nurder, rebellion and civil war, and has thus conclusively demonstrated that it cannot be longer tolerated with safety to the Government and peace to the Union, and that jus-tice, sound morality, and national unity, each and all, demand its entire extinction; and whereas our people of African descent have in the present war been more unani-mous in their loyalty to the Government and their devotedness to the Union than any other class, and have, at the call of Congress and the Executive, sprung to arms to pro-tect the one and maintain the other, and have bravely and nobly vindicated their conrage and their manhood upon the land and upon the water-on the battle-field and on the gun-deck; and whereas the freedmen in the District of Columbia and elsewhere in the United States have, by their obedience to the laws, their willingness to labor, their desire for improvement, and their ability to perform mili-tary service, evinced their capabilities as citizens and soldiers, and thus practically reversed and annihilated tho monstrous judicial dictum and heartless party dogma that "they have no rights which white men are bound to re-spect:" Therefore,

Resolved, That the Congress of the United States should, by positive and effective legislation, and in accordance with the true theory of our republican form of government, guaranty and secure equality of civil rights and privileges to all classes of persons residing within the District of Co-lumbia and the Territories, and whatever else the Government of the United States possesses sole and exclusive juris-diction, who are required and made liable, under the Constitution and the laws, to contribute to the support and maintenance of the Government by taxation and military service, and in like manner to protect, secure, and defend all persons in life, liberty, and lawful pursuits, throughout the length and breadth of the Republic.

2. That American slavery, having engendered the rebel-lion and sustained and prolonged the war, by which un counted thousands of the best citizens of the Republic have been made to suffer and bleed and die, and being subversive of natural right and justice, contrary to the spirit of our institutions, destructive of the best interests of society, disgraceful to our civilization, dangerous to the Republic, and accursed of God and all good men, should not be longer tolerated, but should, by force of law in the adhering States and the power of arms in the rebellious States, be forever abolished and extermioated.

3. That all statutes, legislative acts, and city ordinances, having the force of law, in the District of Columbia and the or a lized Territories of the United States, whereby persons of African descent residing therein are deprived of their civil rights and restrained of their just privileges, ought in justice to be repealed and declared void.

# May 23-Mr. KINNEY, of Utah Territory, offered this resolution; which was adopted :

Resolved, (as the sense of this House,) That the present crisis in the history of this causeless and unjustifiable rebellion calls loudly upon Congress for united patriotic legis-lation; that while our gallant and self-sacrificing soldiers are, with a courage unexampled either in ancient or modare, with a country the there is the internet of more than a country is a country and the hearty support of Congress; and, forgetting for the present all differences upon old party issues, it is the duty of Congress to sustain the constituted authorities of the country in their efforts to suppress the rebellion.

# **Propositions for "Peace."** First Session, Thirty-Seventh Congress. IN HOUSE.

1861, July 15-Mr. BENJAMIN Wood offered this resolution :

Resolved, That this Congress recommend the Governors of the several States, to convene their Legislatures for the purpose of calling an election to select two delegates from each Congressional District, to meet in general Convention at Louisville, in Kentucky, on the first Monday in September next, the purpose of the said Convention to be to devise measures for the restoration of peace to our

nays 51, as follows: YEAS-Messrs. Aldrich, Alley, Appleton, Aruold, Ashley, Babbitt, Baker, Baxter, Beaman, Bingham, Francis P. Birt, Sanuel S. Elair, Elake, Buffinton, Chamberlain, Clark, Coffax, Frederick A. Conkling, Roscoe Conkling, Conway, Curtis, Cutler, Davis, Dawes, Delano, Diven, Duell, Dunn, Edgerton, Edwards, Eliot, Ely, Fenton, Fessenden, Franchot, Gooch, Goodwin, Granger, Gurley, Hale, Hanchett, Harrison, Gooch, Goodwin, Granger, Gurley, Hale, Hanchett, Harrison, McKnight, McPherson, Mitchell, Moorhead, Justin S. Mor-rill, Nixon, Olin, Patton, Pike, Porter, Potter, John H. Rice, Riddle, Edward H. Rollins, Sedgwick, Shanks, Sheffield, Biollabarger, Sherman, Sloan, Spaulding, Stevens, Stratton, Benjamin F. Thomas, Trowbridge, Upton, Vandever, Van Horr, Van Valkenburgh, Van Wyck, Verree, Wall, Wallace, Charles W. Walton, E. P. Walton, Washburne, Wheeler, Albert S. White, Windom-92. NAVS-Messrs. Allen, Ancona, Joseph Baily, George H. Broune, Ewrnett, Cabert, Cobb, Cooper, Corning, Cox, Cra-cuss Crithenden, Delaylaine, Dunlap, English, Fisher, Fouke, Grider, Haight, Harding, Holman, Jackison, Johnson, Law, Lazear, Legan, McClernand, Mallory, Morris, Noble, Node, Vorton, Nugen, Oddt, Pendleton, Reid, Richardson, Robinson, James S. Rollins, Smith, John B. Sleele, William G. Steele, Vallandigham, Vibbard, Voorhees, Wadsworth, Ward, Chalton A. White, Wickliffe, Wood, Woodruff-51.

July 29-Mr. Cox asked leave to offer this resolution :

Whereas it is the part of rational beings to terminate their differences by rational methods, and inasmuch as the differences between the United States authorities and the by bitter hostility and extreme atrocity; and although the party in the seceeded States are guilty of breaking the national unity and resisting the national authority: Yet,

Be it resolved, First. That while we make undiminished and increased exertions by our navy and army to maintain the integrity and stability of this Government, the common the integrity and saturity of this Government, the common laws of war, consisting of those maxims of humanity, mod-eration, and honor, which are a part of the international code, ought to be observed by both parties, and for a stronger reason than exists between two alien nations, inasmuch as the two parties have a common ancestry, history, prosperity, glory, Government, and Union, and are now unhappily engaged in lacerating their common country. Second. That, resulting from these premises, while they ought to be left open, as between two alien natious, the same means for preventing the war being carried to outrageous extrem-ities, there ought also to be left open some means for the restoration of peace and union. Third, That to this endthe restoration of peace and union on the basis of the Constitution—there be appointed a committee of one member from each State, who shall report to this House, at its next session, such amendments to the Constitution of the United States, as shall assuage all grievances, and bring about a States, as shall assuage all grievances, and bring about a reconstruction of the national unity; and that for the prep-aration of such adjustment and the conference requisite for that purpose, there be appointed a commission of seven citizens of the United States, consisting of Edward Everett of Massachusetts, Franklin Pierce of New Hampshire, Mil-iard Fillmore of New York, Reverdy Johnson of Maryland, Martin Van Buren of New York, Thomas Ewing of Ohio, and James Guthrie of Kentucky, who shall request from the so-called Confederate States the appointment of a simi-par commission and who shall meat put confer on the sublar commission, and who shall meet and confer on the sub-ject in the city of Louisville on the first Monday of September next. And that the committee appointed from this tember next. And that the committee appointed from this House notify said commissioners of their appointment and function, and report their action to the next session as an amendment of the Constitution of the United States to be proposed by Congress to the States for their ratification, according to the fifth article of said Constitution.

The motion to suspend the rules and allow its introduction was lost-yeas 41, nays 85, as follows:

Iollows:
YEAS—Messrs. Allen, Ancona, Joseph Baily, Burnett, Gabert, Gaz, Cravens, Crisfield, Crittenden, Dunlap, Grider, Haight, Harding, Holman, Jackson, Johnson, Lau, Leary, Hogan, Mallory, May, Menzies, Morris, Noble, Norton, Nu-gen, Pendletom, Perry, Reid, Richardson, Robinson, Smith, William G. Steele, Vallandigham, Yoorhees, Wadsworth, Webster, Chilton A. White, Wickliffe, Wood, Woodruff—41. NAVS—Messrs. Aldrich, Alley, Arnold, Ashley, Babbitt, Baker, Baxter, Beaman, Bingham, Francis P. Blair, Sanuel Solfax, Frederick A. Conkling, Convasy, Covode, Cutler, Davis, Dawes, Delano, Diven, Duell, Edger-ton, Edwards, Eliot, Fessenden, Franchot, Frauk, Gooch,

The resolution was laid on the table—yeas 92, nays 51, as follows: YEAS-Messrs. Aldrich, Alley, Appleton, Aruold, Ashley, Babbitt, Baker, Baxter, Beaman, Bingham, Francis P. Blair, Sanuel S. Blair, Blake, Duffinton, Chamberlain, Clark, Colfax, Frederick A. Conkling, Roscoe Conkling, Conway, Curtis, Cutler, Davis, Dawes, Delano, Diven, Duell, Dunn, Edgerton, Edwards, Eliot, Ely, Fenton, Fessenden, Franch Karson, With, Wallone, Charles W. Walton, E. P. Walton, Wash-burne, Albert S. White, Windom, Worcester, Wright-85.

August 5 --- Mr. CALVERT offered the following resolution:

That whilst it is the duty of Congress, by appropriate legislation, to strengthen the hands of Government in its efforts to maintain the Union and enforce the supremacy of the laws, it is no less our duty to examine into the original causes of our dissensions, and to apply such remedies as are best calculated to restore peace and Union to the country: Therefore, it is

*Resolved*, (The Senate concurring herein,) That a joint committee, to consist of nine members of this House and four members of the Senate, be appointed to consider and report to Congress such amendments to the Constitution and laws as may be necessary to restore mutual confidence and insure a more perfect and durable Union amongst these States.

Which was laid on the table - yeas 72, nays 39, as follows:

YEAS—Messrs. Aldrich, Alley, Arnold, Ashley, Babbitt, Goldsmith F. Bailey, Baker, Baxter, Bingham, Francis P. Blair, Samuel S. Blair, Blake, Buffinton, Clark, Colfax, Frederick A. Conkling, Roscoe Conkling, Conway, Dunn, Edwards, Eliot, Fenton, Fessenden, Frank, Goodwin, Gran-ger, Gurley, Hale, Harrison, Hutchins, Julian, Kelley, Francis W. Kellogg, William Kellogg, Lansing, Loomis, Lovejoy, McKeen, McKnight, McPherson, Moorhead, Anson P. Morrill, Justin S. Morrill, Olin, Pike, Porter, Potter, Alexander H. Rice, John H. Rice, Riddle, Edward H. Rol-lins, Sedgwick, Shanks, Sucffield, Shellabarger, Sherman, Sloan, Spaulding, Stevens, Benjanin F. Thomas, Train, Sloan, Spaulding, Stevens, Benjawi, Liehnadiger, Dielinal, Sloan, Spaulding, Stevens, Benjamin F. Thomas, Train, Trimble, Trowbridge, Van Horn, Verree, Wall, Wallace, Charles W. Walton, E. P. Walton, Albert S. White, Win-dom, Worcester-72.

aoin, workster-12. NATE-Messrs. Alten, Ancona, Joseph Baily, George H. Broume, Calvert, Cox, Cravens, Crisfield, Dunlap, English, Fisher, Fouke, Grider, Haight, Harding, Inoton, Johnson, Law, Leary, Logan, May, McClernand, Morris, Noble, Odell, Pendleton, Phelps, Richdrason, Robinson, James S. Rollins, Stiel, Smith, William G. Steele, Francis Thomas, Vallandig-ham, Wadsworth, Ward, Webster, Wickliffe-29.

#### August 5-Mr. MAY offered the following :

Whereas the Government of the United States of America was created by its written Constitution, and derived its first powers alone from the consent of the people, as contained in that instrument, and it has no other powers, and force and arms can neither preserve nor rightfully be permitted to violate it under any authority whatsoever : and whereas Washington and other great sages and pa-triots, who founded our General Government, solemnly, warning their countrymen, predicted its destruction from, the establishment of a sectional political party; and they, also entreated a spirit of compromise whenever necessary to preserve the Union; and whereas a civil war now exists among the States which have been united, and which, having already prostrated the peace, prosperity, and happiness of the people, and destroyed many valuable citizens, now threatens their destruction in countless numbers, and by

Intreatens their destriction in countess furthers, and by its inevitable tendency, if not necessity, the final overthrow of free constitutional government: Therefore, *Be it resolved*, That the success of the Republican party, founded, as it is, on a sectional, social, and political ques-tion, is justly responsible for the origin of our present national misfortunes.

2. That the uncompromising spirit hitherto manifested by the representatives of that party has prevented a peace-ful compromise and adjustment of our unhappy difficulties when the same was practicable.

3. That if the present war continues, the only safety and 5. That if the present was continues, the only stater and refuge of constitutional government and civil liberty will be found in the constitutions and sovereignty of the several States, and afterwards, through them, the only hope of a future and more harmonious reconstruction of the Union. 4. That it is impossible by arms to subjugate the people of the graded String united as there are in such numbers.

4. That it is impossible by arms to subjugate the people of the seceed States, united as they are in auch numbers, so fully prepared and resolved, and actuated by motives which represent the just pride and dignity of equals, of trained freemen, of American citizens; and also believing, as they do to a man, that State, home, wife, children, prop-erty, all and every security and benefit of Government is at stake, and that the most cruel and merciless means, forced

by the necessities of an exhausting and desolating war, are to be employed against them.

5. That in view of all these public calumities, and to avoid them, recognizing the necessities which control human af-fairs, as our fathers of the revolution did, it becomes the duty of Congress, before it closes its present session, to pro-vide for the appointment of commissioners to procure an armistice between the contending armies, and restore peace at all events; and who shall be empowered to arrange a compromise to preserve the Union, if possible; but if not, then a peaceful separation of the respective States of the Union, as well such as now claim to have second, as others which may by the sovereign will of their citizens also hereafter ordain to secede; and that the said commission-ers be solemnly enjoined so to conduct their negotiations as to obtain, if possible, in the future, a happy, harmonious, and perpetual reconstruction of our Union of States.

Mr. MAY moved to suspend the rules so as to consider it at once; which was lost-yeas and nays not ordered.

## Second Session, Thirty-Seventh Congress. IN SENATE.

1861, December 4-Mr. SAULSBURY offered this joint resolution :

Whereas the people of the States of Virginia, North Carolina, South Carolina, Georgia, Florida, Alabama, Mississippi, Louisiana, Texas, Arkansas, and Tennesse are in revolt against the constitutional Government and authority of the United States, and have assumed to secede from the Federal Union, and to form an independent government under the name of the Confederate States of America: and whereas the Congress of the United States approving the sentiment expressed by the President, in his annual message, "that expressed by the President, in his annual message, "that the Union must be preserved, and hence all indispensable means must be employed," and believing that kind and fra-ternal feeling between the people of all the States is indis-pensable to the maintenance of a happy and prosperous Union, and being willing to manifest such feeling on their part, to the end that peace may be restored to a distracted eventure and the living and constitution be transmitted and country, and the Union and Constitution be preserved and maintained, and inviting the co-operation of the people of the aforesaid States in the accomplishment of objects so beneficial to each and all, do resolve as follows:

Beneficial to each and all, do resolve as follows: Resolved, dc., That Millard Fillmore, Franklin Pierce, Roger B. Taney, Edward Everett, George M. Dallas, Thomas Ewing, Horace Binney, Reverdy Johnson, John J. Critten-den, George E. Pugh, and Richard W. Thompson be, and they are hereby, appointed commissioners on the part of Congress, to confer with a like number of commissioners, to be appointed by the States aforesaid, for the preservation of the Union and the maintenance of the Constitution, and that they report the result of their said conference to Congress for approval or rejection.

2. That upon the appointment of commissioners, as hereby invited, by said States, and upon the meeting of the joint commission for the purpose of conference as atorcsaid, active hostilities shall cease, and be suspended, and shall not be renewed unless said commission shall be unable to agree, or in case of an agreement by them, said agreement shall be rejected either by Congress or by the aforesaid States.

It was laid on the table.

# Third Session, Thirty-Seventh Congress. IN SENATE.

1862, December 3-Mr. DAVIS offered the following joint resolution:

Resolved, &c., That it be, and is hereby, recommended to all the States to choose as many delegates, severally, as they are entitled to Senators and Representatives in Congress, to meet in convention in Louisville, Kentucky, on the first Monday in April next, to take into consideration the con-dition of the United States, and the proper means for the restoration of the Union; and that the Legislatures of the several States take such action on this proposition as they may deem fit at the earliest practicable day.

Which was ordered to lie on the table, and be printed.

speedy final settlement of the unhappy controversies which brought about this unnecessary and injurious civil war, by just and adequate security against the return of the like calamities in times to come; and this House desires to offer the most earnest assurances to the country that they will, in due time, cheerfully co-operate with the Executive and the States for the restoration of the Union, by such explicit and most solenn amendments and provisions of the Con-stitution as may be found necessary for securing the rights of the several States and sections within the Union under the Constitution.

The resolution giving rise to debate, was laid over.

#### CORRESPONDENCE BETWEEN PRESIDENT LINCOLN AND HON. FERNANDO WOOD.

MR. WOOD TO PRESIDENT LINCOLN.

### NEW YORK, December S, 1862.

#### HOD. ABRAHAM LINCOLN, President of the United States:

DEAR SIR: On the 25th of November last I was advised by an authority, which I deemed likely to be well informed as well as reliable and truthful, that the Southern States would send Representatives to the next Congress, provided that a full and general amnesty should permit them to do so. No guarantee or terms were asked for other than the annesty referred to. Decening this information of great value, if well founded, I communicated it in substance to the Hon. George Opdyke, the mayor of this city, whom I knew to hold confidential relations to members of your Administration, and proposing, through him, that if the Government would permit the correspondence, under its own inspection, I would undertake to procure something definite and positive from persons connected with the so-cilled Confederate authorities. Mr. Opdyke stated in reply that sev-eral Senators from New England States were then in this city on their way to Washington, to whom he would at once communicate the proposition, and advise nee of the an-swer. Knowing that these gentlemen were your friends, and supposing that they would immediately confer with you and the supposing that they would immediately confer with you on their arrival at the capital, and supposing that I should be speedily informed of the result, I have delayed until now

making a communication direct to you. I now learn, however, from Mr. Opdyke this day, that he failed to see these Senators when in New York, and that he had not made the proposition, and that therefore you are not in possession of it as coming from myself.

As an lumble but loyal citizen deeply impressed with the great necessity of restoring the Union of these States, I ask your immediate attention to this subject. The magni-tude of the interests at stake warrant some executive action predicated upon this information, if it be only to ascer-tain if it be grounded upon even probable foundation. If it shall prove groundless no harm shall have been done, provided the inquiry be made, as it can be, without compromising the Government or injury to the cause in which it is now engaged. If, however, it shall prove well founded, there is no estimate too high to place upon its national value.

Now, therefore, Mr President, I suggest that gentlemen whose former political and social relations with the leaders of the Southern revolt may be allowed to hold unofficial correspondence with them on this subject—the correspondence to be submitted to you. It may be thus ascertained what, if any, credence may be given to these statements, and also whether a peaceful solution of the present struggle may not be attainable. I am sure nothing that I can say can add to your own well known desire to produce this re-sult. Your exalted position, the embarrasments and re-sponsibilities which surround you upon all sides, the bleeding condition of the country, b'coming exhausted, not only in the impoverishment of its best life blood, of is dustrial production, but in the deterioration and consequent de-struction of our political institutions—all call upon you, as our chief ruler, to take one step upon the road or peaceful effort, by which to ascertain whether the time has not arrived when other methods than brute fighting may not ac-complish what military force has failed to do.

In the origin of this struggle, you foresaw, that such a time would come. Your inaugural address delivered near two years ago, pointed with prophetic vision the certain re-sults of the impending conflict of arms. Your language then IN HOUSE. Mr. VALLANDIGHAM offered the following resolution: Resolved, That this House does carnestly desire that the peace in America; and that no time may be lost in propos-ing an immediate cessation of hostilities, in order to the not the time arrived when, to quote your own language, we should "cease fighting," at least long enough to ascertain whether "the identical questions" about which we began the fight may not be amicably and honorably adjusted, and the "terms of intercourse" be once more peaceably estub-lished? It is to this end that I now address you-with confidence in your patriotism, and with no desire to inter-fere with your legitimate constitutional prerogatives. I am, with high respect, yours very truly. re with your legitimate constitutions, process I am, with high respect, yours very truly, FERNANDO WOOD.

# PRESIDENT LINCOLN TO MR. WOOD. EXECUTIVE MANSION, WASHINGTON, December 12, 1862.

HON. FERNANDO WOOD :

MY DEAR SIR : Your letter of the 8th, with the accompa-

My DEAR Site: Four letter of the stin, while the accompa-pying note of same date, was received yesterday. The most important paragraph in the letter, as I con-sider, is in these words: "On the 26th of November last I was advised by an authority which I deemed likely to be well informed as well as reliable and truthful, that the Southern States would send representatives to the next Congress, provided that a full and general amnesty should

congress, provided that a full and general amnesty should permit them to do so. No guarantee or terms were asked for other than the amnesty referred to." Istrongly suspect your information will prove to be groundless; nevertheless, I thank you for communicating it to me. Understanding the phrase in the paragraph above quoted—" the Southern States would send represen-tatives to the next Congress." To be subtrarticily the above quoted—" the Southern States would send represen-tatives to the next Congress"—to be substantially the same as that " the people of the Southern States would cease resistance, and would reinangurate, submit to, and maintain the national authority within the limits of such States, under the Constitution of the United States," I say that in such case the war would cease on the part of the United States; and that if within a reasonable time "a full and general amnesty" were necessary to such end, it would not be withheld. it would not be withheld.

I do not think it would be proper new to communicate this, formally or informally, to the people of the Southern States. My belief is that they already know it; and when they choose, if ever, they can communicate with me une-quivocally. Nor do I think it proper now to suspend mili-

Subjectives, in every liney can communicate with me une-quivocally. Nor do I think it proper now to suspend mili-tary operations to try any experiment of negotiation. I should nevertheless receive, with great pleasure, the exact information you now have, and also such other as you may in any way obtain. Such information might be more valuable before the last of January than afterward. While there is nothing in this latt a which I at a start of the

While there is nothing in this letter which is hall dread to see in history, it is, perhaps, better for the present that its existence should not become public. I therefore have to request that you will regard it as confidential. At you will regulate Your obedient servant, ABRAHAM LINCOLN.

# MR. WOOD TO PRESIDENT LINCOLN.

NEW YORK, December 17, 1862.

His Excellency ABBAHAM LINCOLN :

My DEAR SIR: Your letter of the 12th instant was handed to me on the afternoon of the 15th-instant by Mr. Wake-man, the postmaster of this city.

Pardon me, Mr. President, when I say that your reply has fi.led me with profound regret. It seclines what I had conceived to be an innocent effort to ascertain the foundation for information in my possession of a desire in the South to return to the Union. It thus appears to be an indication on your part to continue a policy which, in my judgment, is not only unwise, but, in the opinion of many, is in conflict with the constitutional authority vested in the Federal Government,

<sup>1</sup> I think, however, that my proposition is in keeping with your own expressed conditions upon which the war shall cease. You say that "when the people of the southern States would cease resistance, and would reinaugurate, submit to, and maintain the national authority within the limits of such States under the Constitution of the United States, that in such case the war would cease on the part of the United States."

Admitting this position as correct, you will see that as a condition precedent to such submission the opportunity to do so must be afforded. It cannot be expected that the Boothern people will cease resistance, so long as we pro-claim our intention to destroy their local institutions, their property, and their lives, and accompany the declaration with corresponding legislative, executive, social, and political action. They cannot ccase resistance, and reinaugu-rate, submit to, and maintain the Federal authority, if we will not let them alone long enough to do so. If they really desire acquisecence, and are willing to send delegates to the next Congress, as I am advised, how can they do so without the opportunity, and without some intimations or

guarantees as to the reception of their representatives at Washington? The act of sending representatives to Congress is within itself a full compliance with your own conditions. If thus represented by their own selected agents, chosen under the forms and in pursuance of their own local State laws governing such elections, they will compose an inte-gral portion of the Government, and thus give the assurance of an "acquiescence and submission" of the very highest and most satisfactory character.

My respectful suggestion was that you should put it in their power to take this course. It would require a sim-ple proclamation of general amnesty, to be qualified, if you please, by such conditions as to render it void in case of non-compliance within a limited period. You have established a precedent for this mode of speaking to those peo-ple. Your Emancipation Proclamation told of punish-ment. Let another be issued, speaking the language of mercy and breathing the spirit of conciliation.

The painful events which have occured since my com-munication of the 8th instant but embolden me to renew its suggestions. I hope you will now no longer refuse "to suspend military operations to try an experiment of ne-gotiation." I feel that military operations so bloody and exhausting as ours must sooner or later be suspended. The day of suspension must come. The only question is, whether it shall be before the whole American people, North and South, shall be involved in general ruin, or whether it shall be whilst there is remaining sufficient of the recuperative element of life by which to restore our once happy, prosperous, and peaceful American Union.

In compliance with your request that your letter shall not for the present become public, I shall withhold its pub-lication at this time.

n at this time. With high regard, yours, &c., FERNANDO WOOD.

# First Session, Thirty-Eighth Congress. IN SENATE.

1864, June 9-Mr. DAVIS, of Kentucky, sought to introduce this joint resolution, but objection was made :

A joint resolution to restore peace among the people of the United States.

Resolved, dc., That three years of civil war in which the enormous expenditure of blood and treasure has no parallel in the world's history, and whose wide-spread rapine and diabolical cruelties have shocked Christendom, and which, from alternating success, has produced no essential results, prove that war was not the proper remedy for our national troubles.

2. That if the people of America would save and restore their shattered Constitution and avert from themselves and their posterity the slavery of a military despotism and of a public debt, the interest upon which all the avails of their labor and economy will never meet, they must bring this war to a speedy close. 3. That the President of the United States be, and he is

hereby, authorized to propose a cessation of arms and an annosty to the authorized to a correction of the Confederate States of America, with a view to a convention of the people of all the States to reconstruct their Union; and if that cannot be effected, then that said convention agree upon the terms of a separation of the States without the further effusion of blood, and of a lasting peace among them.

#### IN HOUSE.

1863, Dec. 14-Mr. FERNANDO WOOD offered this resolution :

Whereas the President, in his message delivered to this House on the 9th instant, and in his recommendation to the people to assemble at their places of worship and give thanks to God for recent victories, claims that the Union cause has gained important and substantial advantages; and whereas, in view of these triumphs, it is no longer be-neath our dignity nor dangerous to our safety to evince a generous magnanimity becoming a great and powerful people by offering to the insurgents an opportunity to return to the Union without imposing upon them degrading or destructive conditions: Therefore, Resolved, That the President be requested to appoint

three commissioners, who shall be empowered to open ne-gotiations with the authorities at Richmond, to the end that this bloody, destructive, and inhuman war shall cease, and the Union be restored upon terms of equity, fraternity, and equality, under the Constitution.

Mr. ELLIHU B. WASHBURNE moved that the

resolution lie on the table, which was agreed | to-yeas 98, nays 59, as follows :

YEAS 50, https://www.science.com/ YEAS-Messrs Alley, Allison, Anderson, Arnold, Ashlay, Bally, J. D. Baldwin, Baxter, Beaman, Blaine, J. B. Blair, Blow, Boutwell, Boyd, Brandegee, Broomall, William G. Brown, Ambrose W. Clark, Freeman Clarke, Colb, Cole, Creswoll, Henry Winter Davis, Dawes, I eming, Dixon, Ion-nelly, Drigg, Dumont, Eckley, Eliot, Farnsworth, Fenton, Ganson, Garfield, Gooch, Grinnell, Griswold, Higby, Hooper, Horchkiss, Asahel W. Hubbard, J. H. Hubbard, Hubbard, Jenckes, Julian, Kasson, Francis W. Kellorg, Orlando Kel-lorg Loop Longveen Logiow, Martin McRide, MCChurg, Jences, Julian, Kasson, Francis W., Kenneg, Orlando Kel-logg, Loan, Longyear, Lovejoy, Marrin, WeBride, MrClurg, McIndoe, Samuel F. Miller, Moorhead, Morrill, Daniel Mor-ris, Amos Myers, Leonard Myers, Norton, Charles O'Neill, Orth, Patterson, Perham, Pike, Ponery, Price, William II. Randall, Alexander H. Rice, John H. Rice, Edward H. Rollins, Schenck, Schfield, Shannon, Sloan, Smith, Smithers, Spalding, Stevens, Thayer, Thomas, Tracy, Upson, Van Yalkenhargh, *Ward*, Elihuu B. Washburne, Wm. B. Wash-burn, Whaley, *Wheeler*, Williams, Wilder, Wilson, Windom, Woodbridge, *Icaman-98*, *Unv. En. J. Allen.* 

Woodbridge, Ycaman-98. NATS-Messrs, Janus C. Allen, Wm. J. Allen, Ancona. Augustus C. Baldovin, Eliss, Brooks, Chanler, Clay, Coffroh, Cox, Cruvens, Dawson, Denison, Eden, Edgerton, Eldridge, English, Finck, Gruder, Harding, Harrington, Denjemin G. Harris, Charles M. Harris, Herrick, Holman, William Johnson, Kernan, King, Knapp, Law, Lazear, Le Blond, Long, Mallory, Marcy, McDowell, McKinney, William H. Müller, Jimes R. Morris, Marrison, Nelson, Noble, Odell, John O. Neill, Fendlichon, Robinson, James S. Rollins, Ross, Scott, Stebbins, John B. Steele, Sluart, Sweat, Voorhees, Wudsworth, Chillon A. While, Joseph W. While, Winfield, Wood-59. Wood-59.

Dec. 17-Mr. GREEN CLAY SMITH offered these resolutions :

1. Resolved, That as our country, and the very existence of the best government ever instituted by man, are imperilled by the most causeless and wicked rebellion that the world has seen, and believing, as we do, that the only hope of saving this country and preserving this Government is by the power of the sword, we are for the most vigorous prosecution of the war until the Constitution and laws shall be enforced and obeyed in all parts of the United States; and to that end we oppose any armistice, or intervention, or mediation, or proposition for peace from any quarter, so long as there shall be found a rebel in arms against the Government ; and we ignore all party names, lines, and issues, and recognize but two parties in this war

the people to render every aid in their power to the constituted au horities of the Government in the crushing out of the rebellion, and in bringing the leaders thereof to con-

dign punishment. 3. That our thanks are tendered to our soldiers in the field for their galantry in defending and upholding the flag of the Union, and defending the great principles dear to every American patriot.

A division of the question having been called, and the question being on the first resolution :

Mr. ANCONA moved that it be laid on the table; which was disagreed to-yeas 60, nays 100. as follows:

100, us 10110WS: YEAS-MOSSTS. James C. Allen, William J. Allen, Ancona, Bliss, Brooks, Chanler, Clay, Caffroth, Cox, Cravens, Dav-son, Denison, Eden, Edgerton, Eldridge, English, Finck, Grider, Ital, Harding, Harrington, Benjamin G. Harris, Charles M. Harris, Herrick, William Johnson, Kernan, King, Knapp, Law, Le Blond, Long, Mallory, McDowell, McKinney, Middleton, William H. Miller, James R. Morris, Morrison, Nelson, Noble, John O'Neill, Pendleton, Radford, Samuel J. Randall, Robinson, Rogers, James S. Rollins, Ross, John B. Steele, Stiles, Strouse, Stuart, Sweat, Voorhees, Wadsworth, W heeler, Chilton A. White, Joseph W. White, Winfield, Wood-60.

Russ, out, B. Succe, Bates, Barouse, Surar, Succa, Volnies, Wadsworth, Wheeler, Chilton A. White, Joseph W. White, Wangield, Wood-e0.
NAVS-Messrs. Allison, Ames, Arnold, Ashley, Bady, Au-gustus C. Baldwin, John D. Baldwin, Beaman, Blaine, Blow, Boutweil, Boyd, Brandegee, Broomall, James S. Brown, William G. Brown, Ambrose W. Clark, Freeman Clarke, Cobb, Cole, Creswell, Henry Winter Davis, Thomas T. Davis, Dawes, Deming, Dixon, Donnelly, Driggs, Dumont, Eckley, Eliot, Farnsworth, Fenton, Frank, Ganson, Garfield, Gooch, Grinnell, Griswold, Hale, Higby, Hooper, Hotchkins, Jenckes, Julian, Kasson, Kelley, Francis W. Kellogg, Or-lando Kellogg, Loan, Lovejoy, Marvin, McAllister, McBride, McClurg, McIndoe, Samuel F. Miller, Moorhead, Morrill,

Daniel Morris, Amos Myers, Leonard Myers, Norton, Odell, Daniel Morris, Amos Myers, Leonard Myers, Norton, Odell, Charles O'Neill, Orth, Pattersson, Perham, Pike, Pomeroy, Price, William H. Randall, Alexander H. Rice, John H. Rice, Edward H. Rollins, Schenck, Schofield, Shannon, Sloan, Smith, Smithers, Spalding, Steblørs, Stovens, Thayer, Tracy, Van Valkenburgh, Ward, Ellihu B. Washburne, William B. Washburn, Whaley, Williams, Wilder, Wilson, William B. Washburn, Whaley, Williams, Wilder, Wilson,

The resolution was then agreed to-yeas 94, nays 65, as follows :

nays 65, as follows :
 YEAS-Messrs. Alley, Allison, Ames, Anderson, Arnold, Ashley, Euily, John D. Baldwin, Beaman, Elaine, Blow, Bontwell, Boyd, Brandegee, Broomall, Ambrose W. Clark, Cobb, Cole, Creswell, Henry Winter Davis, Thos. T. Davis, Dawes, Denuing, Dixon, Donnelly, Driggs, Dumont, Eckley, Eliot, Farnsworth, Fenton, Frank, Garfield, Gooch, Grinnell, Itabear, Hulburd, Jenckes, Kasson, Kelley, Francis W. Kellogg, Orlando Kellogg, Loan, Longyear, Lovejoy, Marvin, McAlüster, McBride, McChurg, McIndoe, Samuel F. Miller, Moorhead, Morrill, Daniel Morris, Amos Myers, Leonard Myers, Norton, Odell, Charles O'Neill, Orth, Patterson, Perham, Pike, Pomeroy, Price, William H. Randall, Alexander H. Rice, John H. Rice, Edward H. Rolling, Stevens, Thayer, Tracy, Van Valkenburgh, Ellihu B. Washburne, Wm. B. Washburn, Whaley, Williams, Wilder, Murss-Messrs, James C. Allen, William J. Alexander, E. Ganson, Grider, Gravens, Dawson, Denison, Eden, Edgerton, Eldridge, English, Finck, Gamson, Grider, Griswold, Harris, Herrick, Hutchins, William Johnson, Kernan, Knapp, Luw, Le Blond, Long, William G. Steve, Stanes R. Morris, Morrison, Neidel, Jong, Smiller, Jonse, James R. Morris, Merriek, Judelton, Wm. H. Miller, James R. Morris, Morrison, Neidel, Jong, Marly, Model, Jong, Milav, Johnson, Kernan, Knapp, Luw, Le Blond, Long, Wallion, Johnson, Kernan, Knapp, Luw, Le Blond, Long, Wallion, Meeler, Stiles, Strouse, Staud, Jonne, Neile, John B. Steele, William G. Steele, Stiles, Strouse, Staud, Jonnes, Steal, Strouse, Staud, Jonnes, Steal, Strouse, Staud, Jonne, Neiler, Garona, Kandal, Robinson, Rogers, James K. Rollins, Ross, Scott, John B. Steele, William G. Steele, Stiles, Strouse, Staud, Vorhees, Waster, Warled, Fernando Wood-65.
 The second resolution was agreed to—yeas

The second resolution was agreed to-yeas 153, nays 1, as follows:

YEAS-Messrs. Alley, Allison, Ames, Anderson, Arnold, Ashley, Baily, Augustus C. Baldwin, John D. Baldwin, Beaman, Blaine, Blizs, Blow, Boutwell, Boyd, Brandegee, Brooks, Broomall, James S. Brown, William G. Brown, Chanler, Ambrose W. Clark, Freeman Clarke, Clay, Cobb, Coffroith, Cole, Cox, Cravens, Creswell, Henry Winter Davis, Device Davis Device Device Device Device Device Thomas T. Davis, Dawes, Dawson, Deming, Denison, Dixon, Thomas T. Davis, Dawes, Dawson, Deming, Denison, Dixon, Donnelly, Driggs, Dumont, Eckley, Edgerton, Eldridge, Eliot, English, Farnsworth, Fenton, Finck, Frank, Ganson, Garfield, Gooch, Grider, Grinnell, Griswold, Itale, Halt, Harding, Charles M. Harris, Herrick, Higby, Holman, Hooper, Hotchkiss, Asahel W. Hubbard, John H. Hubbard, Hulburd, Hutchins, Jenckes, Wm. Johnson, Julian, Kasson, Kelley, Francis W. Kellogg, Orlando Kellogg, Kernan, King, Law, Lazear, Le Biond, Loan, Lovejoy, Mallory, Marrin, McAllister, McEride, McClurg, McIndoe, McKinney, Middleton, Samuel F. Miller, William H. Miller, Moorhead, Morrill, Daniel Morris, James, Morrison, Amos Middleton, Samuel F. Miller, William H. Mitter, Moorneau, Morrill, Daniel Morris, James R. Morris, Morrison, Amos Myers, Leconard Myers, Nelson, Noble, Norton, Odell, Charles O'Neill, John O'Neill, Orth, Patterson, Perham, Perry, Pike, Pomeroy, Price, Radford, Samuel J. Randall, William H. Randall, Alexander H. Rice, John H. Rice, Rogers, Edward H. Rollins, James S. Rollins, Schenck, Scofield, Scott, Shannon, Sloan, Smith, Smithers, Spalding, Manuel M. Kandall, Markan G. Strapes, Strapes Kogers, Edward H. Kollins, James S. Kollins, Schenck, Scofield, Scott, Shannon, Sloan, Smith, Smithers, Spalding, Stabins, John B. Stele, William G. Steele, Stevens, Strouse, Stuart, Sweat, Thayer, Thomas, Tracy, Van Valkenburgh, Ivadsworth, Ward, Ellihu B. Washburne, William B. Wash-burn, Whaley, Wheeler, Jaseph W. White, Williams, Wilder, Wilson, Windom, Winfeld, Woodbridge, Feaman-153. NAV-Mr. Benjamin G. Harris-1.

The third resolution was agreed to-yeas 168, nays 1, (Mr. B. G. Harris.)

On this vote, Messrs. James C. Allen, Wm. J. Allen, Ancona, Eden, Harrington, Knapp, Long, Longyear, McDowell, Pendleton, Robinson, Stiles, Voorhees, Webster, Chilton A. White and Fernando Wood-16, who are not recorded before, voted aye; and Mr. Hooper, who is recorded before, did not vote.

1864, Jan. 7-Mr. JOHN D. BALDWIN offered this preamble and resolution :

Whereas the organized treason having its headquarters at Richmond exists in defiant violation of the national Constitution and has no claim to be treated otherwise than

as an ontiaw; and whereas this Richmond combination of conspirators and traitors can have nor ghtful authority over the people of any portion of the national Union, and no warrant for assuming control of the political destiny of the pople of any state or section of this Union, and no apology but that of conspiracy and treason for any assumption of authority whatever: Therefore, *Resolved.* That any proposition to negotiate with the rebel leaders at Richmond (sometimes called "the author-ities at Richmond") for a restoration of loyalty and order

rever reasers as menmonal sometimes called "the author-ties at Richmond") for a restoration of loyalty and order in those portions of the Republic which have been disor-ganized by the robellion is, in effect, a proposition to rec-ognize the ringleaders of the rebellion as emittled to repre-sent and bind the loyal citizens of the United States whow they concess and the dive converse whom they oppress, and to give countenance and support to the pretensions of conspiracy and treason; and therefore every such proposition should be rejected without hesitation and delay.

Mr. Cox moved to lay the resolution on the table, which was disagreed to; and it was then passed-yeas 89, nays 24, as follows :

passed — yeas 89, nays 24, as follows: YEAS-Messrs. Alley, Allison, Ames, Anderson, Arnold, Baily, Augustus C. Baldwin, John D. Baldwin, Baxter, Beaman, Blaine, Jacob B. Blair, Blow, Boutwell, Boyd, Frandegee, Broomall, James S. Brown, Wiliam G. Frown, Cobb, Cole, Creswell, Henry Winter Davis, Dawes, Deming, Dixon, Donnelly, Eckley, Eliot, Farnsworth, Fenton, Garfield, Gooch, Grinnell, Griswold, Hale, Higby, Holman, Hooper, John H. Hubbard, Hulburd, Julian, Kasson, Kelley, Fran-cis W. Kellogz, Kernam, King, Loan, Longyear, Lovejoy, Marvin, McBride, McClurg, Middlelon, Morrill, Daniel Morris, Amos Myers, Leonard Myers, Norton, Odell, Charles O'Neill, Orth, Perham, Pike, Pomeroy, Price, William H. Ramidarl, John H. Rice, Scofield, Shanuon, Sloau, Smith, Smithers, Spalding, Stebbins, Stevens, Sweat, Thayer, Tracy, Upson, Van Valkenburgh, Ellihu B. Washburne, William B. Washburn, Webster, Williams, Wilson, Windom, Woo Dridge, Faman-89.

B. Washoffi, Webstel, Williams, Wilson, Wildon, Wood-bridge, Feaman-89. NAYS-Messis. Ancona, Bliss, Brooks, Cox, Denison, Edger-ton, Finck, Harrington, Charles M. Harris, Herrick, Knapp, Long, Marcy, William H. Miller, Morrison, Noble, Pendle-ton, Perry, Pruyn, Samuel J. Randall, Rogers, Ross, Strouse, Fernando Wood-24.

The preamble was then agreed to-yeas 102, nays none, as follows:

YEAS-Messrs. Alley, Allison, Ames. Anderson, Arnold, Baily, Augustus C. Baldwin, John D. Baldwin, Baxter, Blaine, Jacob B. Blair, Blow, Boutwell, Brandegee, Brooks, Broomall, James S. Brown, Wm. G. Brown, Cobb, Coffroth, Cole, Coz, Cravens, Creswell, Thomas T. Davis, Dawes, Dem-Cole, Coz, Gravens, Creswell, Thomas T. Davis, Dawes, Deming, Denison, Dixon, Eliot, Fenton, Finck, Frank, Garfield, Gooch, Grüder, Grinnell, Griswold, Hale, Harrington, Herrick, Higby, Holman. Hooper, John H. Hubbard, Hulburd, Julian, Kasson, Kelley, Francis W. Kellogg, Kernan, King, Lazear, Loan, Longyear, Lovejoy, Marvin, McBride, McClurg, Middleton, Wm. H. Miller, Morrill, Daniel Morris, Amos Myers, Leonard Myers, Noble, Norton, Odell, Charles O'Neill, Orth, Pike, Pomeroy, Price, Pruyn, Samuel J. Randall, William II. Randall, John II. Rice, Rogers, Scofield, Sloan, Smith, Smithers, Spalding, Stebbins, Join B. Steele, Wm. G. Steele, Stevens, Sirouse, Sweat, Thayer, Tracy, Upson, Van Valkenburgh, Elihu B. Washburne, William B. Washburne, Williams, Wilson, Windom, Woodbridge, Yeaman-102.

January 7 --- Mr. Rogens offered these resolutions :

Resolved, That as our country and the existence of the old Union are imperilled by a rebellion against the wisest and best government ever devised by man, we are for the most united, determined, and vigorous prosecution of the war for the purpose of enforcing the Constitution of the United States and the laws made in pursuance thereof in all parts of the United States; but at the same time we are for adding to force the power of conciliation and compro-mise so far as is consistent with an honorable and lasting peace, and founded solely upon a restoration of the Union pace, and founded solely upon a restoration of the Union under the Constitution, and in no event to agree to or coun-tenance a dissolution of the Union; and that we believe the appointment of commissioners upon the part of the The appointment of commissioners upon the part of the Federal Government, to meet commissioners similarly ap-pointed by the insurgent States, to convene in some suitable place for the purpose of considering whether any, and if any, what plan may be adopted consistent with the honor and dignity of the nation, and based upon a restoration of the whole Union, by which the present war may be brought to a close, and the lives, limbs, and health of the gallant officers and soldiers of the Union preserved, and the liber-ties of the neede meintained is not inconsistent with the ties of the people maintained, is not inconsistent with the

honor and dignity of the Federal Government, but, as an indication of the spirit which animates the adhering States, would, in any event, tend to strengthen us in the opinion of other nations and the loyal people of the insurgent States; and hoping, as we sincerely do, that the people of the southern States would reciprocate the peaceful indications thus evinced, and beheving, as we do, that under the blessings of God great henefits would arise from such con-ference, we most earnestly recommend such conference to the consideration of the President and Senate of the United States, and request their co-operation therein, and hope that the President will appoint commissioners for that purpose

2. That the people of the several States now in rebellion against the Government of the United States, whenever they shall desire to return to the Union and obey the Constitution of the United States, and laws made in pursuance thereof, have a right under and by virtue of the said Con-stitution to reorganize their respective State governments stitution to reorganize their respective state governments with all their domestic institutions as they were before the War, and to elect representatives to the Congres of the United States, and to be represented in the Union with all the rights of the people of the several States, and without any conditions precedent except that of being liable to be punished according to the Constitution and laws made in pursuance thereof, as their laws and acts of secession are unconstitutional and void.

On motion of Mr. STEVENS they were laid upon the table-yeas 79, nays 42, as follows:

YEAS--Messrs. Alley, Allison, Ames, Anderson, Arnold, Badly, John D. Baldwin, Baxter, Beaman, Blaine, Jacob B. Blair, Blow, Boutwell, Boyd, Brandegee, Broomall, William G. Brown, Cobb, Cole, Creswell, Thomas T. Davis, Dawes, Deming, Dixou, Donnelly, Eckley, Eliot, Farusworth, Fen-ton, Frank, Garfield, Gooch, Grinnell, Hale, Higby, Hooper, Hulburd, Julian, Kasson, Kelley, Francis W. Kellogg, Or-lando Kellogg, Loan, Lovejoy, Marvin, McBride, McClurg, Morrill, Daniel Morris, Amos Myers, Leonard Myers, Norton, Odell, Charles O'Neill, Orth, Perham, Pomeroy, Price, Wil-liam H. Randall, Schofield, Shannon, Sloan, Smith, Smithers, Spalding, *Stebbins*, Stevens, Thayer, Thomas, Tracy, Upson, Van Valkenburgh, Ellihu B. Washburne, William B. Washburn, Webster, Williams, Wilson, Windom, Wood-

Washburn, Webster, Wilhams, Wilson, Wildoli, Wood-bridge-79. NATS-Messrs. William J. Allen, Ancona, Augustus C. Baldwin, Brooks, James S. Brown, Chanler, Coz, Cravens, Dawson, Denison, Eden, Edgerton, Finck, Grider, Griswold, Harrington, Charles M. Harris, Herrick, Hoimon, Kernan, Knapp, Lazear, Long, Maroy, William H. Miller, Morrison, Noble, Pendleton, Perry, Pruyn, Samuel J. Raniall, Robin-son, Rogers, James S. Collins, Ross, John B. Steelc, William G. Steele, Strouse, Sweat, Joseph W. While, Fernando Wood, Fernan-U. Yeaman--42,

Jan. 18-Mr. Dawson, of Pennsylvania, offered the following preamble and resolution :

Whereas a great civil war like that which now afflicts the United states is the most grievous of all national calam-ities, producing as it does, spoliation, bloodshed, anarchy, public debt, official corruption, and private immorality, the American Government cannot rightfully wage such a war upon any portion of its people, except for the sole purpose upon any portion of its people, except for the sole purpose of vindicating the Constitution and laws and restoring both to their just supremacy; and whereas this House, on the 22d day of July, 186I, speaking in the name of the Ameri-can people and in the face of the world, solemnly and traly declared that it was waged for no purpose of conquest or oppression, but solely to restore the Union, with all the rights of the people and the State uninpaired; and whereas in every war, especially in every war of invasion, and most in every war, especially in every war of invasion, and most particularly if it be a civil war between portions of the same country, the object of it ought to be clearly defined and the terms distinctly stated upon which hostilities will cease; and the advancing armies of the Government should carry the Constitution in one hand, while they hold the sword in the other, so that the invading party may have its choice between the two: Therefore,

Resolved, That the President be requested to make known hereover, that the resident be requested to make known by public proclamation or otherwise, to all the country, that whenever any State now in insurrection shall submit hereaff to the authority of the Federal Government, as de-fined in the Constitution, all hostilities against her shall cease, and such State shall be protected from all external tester with the lead here we is within a content of the interference with the local laws and institutions, and her people shall be guarantied in the full enjoyment of all those rights which the Federal Constitution gave them.

Mr. STEVENS moved that they be laid upon the table, which was agreed to-yeas 79, nays 56, as follows:

YEAS-Messrs. Allison, Ames, Arnold, Ashley, John D.

Baldwin, Baxter, Beaman, Blaine, Francis P. Blair, jr., Jacob B. Blair, Boutwell, Brandegee, Broomall, Ambrose W. Clark, Freeman Clarke, Cole, Creswell, Henry Winter Davis, Thomas T. Davis, Dawes, Deming, Dixon, Driggs, Eckley, Eliot, Farnsworth, Fenton, Frank, Garfield, Gooch, Higby, Hoeper, Hotchkiss, Asahel W. Hubbard, Hul-burd, Jenckes, Julian, Kasson, Kelley, Francis W. Kel-logg, Longver, Lovejoy, Marvin, McBride, McClurg, Mc-Indoe, Samuel F. Miller, Moorhead, Morrill, Daniel Morris, Amos Myers, Leonard Myers, Charles O'Neill, Orth, Pat-terson, Pike, Price, William H. Randall, Alexander H. Rice, John H. Rice, Edward H. Rollins, Schenck, Scofield, Shan-non, Smith, Smithers, Spalding, Stevens, Thayer, Thomas, Upson, Van Valkenburgh, Ellihu B. Washburne, William B. Washburn, Whaley, Williams, Wilson, Windom, Wood-bridge-79.

Washburn, Whaley, Williams, Wilson, Windom, Woodbridge-79. NAYS-Messrs. James C. Allen, Ancona, Augustus C. Baldwin, Blits, Brooks, James S. Brown, William G. Brown, Chanler, Coffroth, Cox, Dateson, Denison, Eden, Edgerton, Eldridge, English, Finck, Ganson, Grider, Griswold, Hale, Holman, Hutchins, William Johnson, Kernan, Lazear, Le Blond, Long, Marcy, McAllster, McDwell, McKiney, Middleton, William H. Miller, James R. Morris, Morrison, Netson, Fendelton, Robinson, Ross, John B. Steele, Stilar, Stuart, Sweat, Torses, Wm. H. Wadsworth, Wheeler, Chilton A. While, Joseph W. White, Fernando Wood, Yeaman-56. man-56.

February 29-Mr. Long, of Ohio, offered this preamble and resolution :

Whereas history teaches that there never has been a civil war that was not settled in the end by compromise, and inasmuch as no possible harm can result either to the character or dignity of the United States from an honest effort to stop the effusion of fraternal blood, and restore the Union by the return of the States in rebellion to their allegiance under the Constitution; and whereas the Presi-dent, with a full knowledge of the lessons taught by history in relation to all civil wars, in his inaugural address said, "suppose you go to war, you cannot fight always; and when, after much loss on both sides, and no gain on either, when, after much loss on both sides, and no gain on either, you cease fighting, the identical old questions as to terms of intercourse are again upon you;" and whereas we now have an armistice, decreed by the Almighty, and executed for the past two months by the snows and ice of winter, thereby affording time and opportunity for reflection upon the past three years of horrible, relentless, and destructive civil war with all its calamities, and a prospective view of increased horrors in the approaching conflicts; and whereas Increased horrors in the approaching conflicts; and whereas a preamble and resolutions were, on the 7th of February, instant, introduced in the House of Representatives of the Confederate Congress at Richmond denying the statement of the President of the United States "that no propositions for peace had been made to the United States by the Con-federate States," and affirming that such propositions were federate States, and annumg that sour propositions actor prevented from being made by the President of the United States, in that he had refused to hear, or even to receive, two commissioners appointed to treat expressly for peace: Therefore, be it Resolved, That the President be, and he is hereby, most

earnestly. but respectfully, requested to appoint Franklin Pierce, of New Hampshire; Millard Fillmore, of New York; Thomas Ewing, of Ohio, and such other persons as the Pres. Thomas Ewing, of Ohio, and such other persons as the Pres-ident may see proper to select, as commissioners on behalf of the United States, who shall be empowered to meet a commission of like number when appointed for the same ob-ject on behalf of the Confederate States, at such time and place as may be agreed upon, for the purpose of ascertain-ing, before the renewal of hostilities shall have again com-menced, whether the war shall not now cease, and the Union be restored by the return of all the States to the al-herience and their inclus under the Constitution legiance and their rights under the Constitution.

Which was disagreed to-yeas 22, nays 96, as follows :

YEAS—Messers. James C. Allen, Ancona, Brooks, Coffroth, Denison, Ede., Eldridge, Finck, Knopp, Long, McDowell, Wm. H. Miller, Morrison, John O'Neill, Fendleon, Samuel J. Randall, Rogers, Ross, Stiles, Strouse, Voorhees, Chilton A. While-22. A. White-

A. White-22. NAYS-MOSSTS. Alley, Allison, Ames, Anderson, Arnold, Ashley, Augustus C. Bildwin, John D. Baldwin, Baxter, Jacob B. Blair, Blow, Boutwell, Boyd, Brandegee, William G. Brown, Anbrose W. Clark, Freeman Clarke, Cobb, Cole, Oreswell, Henry Winter Davis, Dawes, Deming, Dixon, Don-nelly, Driggs, Dumont, Eckley, Eliot, Farnsworth, Fenton, Frank, Ganson, Garfield, Grüneil, Griswold, Hale, Higby, Holman, Hooper, Hotchkiss, John II. Hubbard, Hutchins, Jenckes, Julian, Kelley, Orlando Kellogg, Kernan, King, Loan, Lovejoy, Marvin, McBride, McClurg, McIndoe, Samuel F. Miller, Moorhead, Morrill, Daniel Morris, Amos Myers,

Leonard Myers, Nelson, Norton, Charles O'Neill, Perham, Pike, Pomeroy, Price, Radford, William II. Randall, John II. Rice, Schenck, Scofield, Shannon, Sloan, Smithers, H. Rice, Schenck, Sconed, Shanhon, Floan, Smithers, Spalding, Starr, Stebbins, John B. Steele, Stevens, Thayer, Thomas, Tracy, Upson, Van Valkenburgh, Wadsworth, Ellihu B. Washburne, William B. Washburn, Whaley, Williams, Wilder, Wilson, Windom, Winfeld, Woodbridge -96.

May 30-Mr. LAZEAR asked consent to offer this resolution :

Whereas the fratricidal war which has for the last three years filled every neighborhood of our once united and happy country with mourning, and has drenched a hundred battle-fields with the blood of our fellow-citizens, and laid waste many of the fairest portions of the land, and yet has failed to restore the authority of the Federal Government in the seceded States; and whereas we believe a misapprehension exists in the minds of a large portion of the people of the South as to the feelings which actuate a large portion of the people of the free States, and which misapprehension we are called upon by every consideration of humanity and a sense of justice to correct and if possible remove, whether we regard in making this effort what we owe to ourselves. to our fellow-countrymen of the South, or to the world: Therefore.

Resolved, That no truly loyal citizen of the United States desires the application of any rule or law in determining the rights and privileges and the measure of responsibility of the people of any of the States but such as shall have been determined by the Supreme Court to be in accordance with and sanctioned by the Constitution and well-established usages of the country.

2. That the President, in his capacity of Commander-in-Chief of the Army and Navy of the United States, be, and he is hereby, required to adopt such measures as he may think best, with a view to a suspension of hostilities between the armies of the North and the South for a period not exceed-ing — days; and that he be also authorized to adopt or agree upon some plan by which the decision of the great body of the people North and South may be secured upon the question of calling a convention composed of delegates the question of chinng a convention composed or delegates from all the States, to which shall be referred the settlement of all questions now dividing the southern States from the rest of the Union, with a view to the restoration of the sev-eral States to the places they were intended to occupy in the Union, and the privileges intended to be granted to them by the framers of our national Constitution, who were is our criticity the most environment of the purest in our opinion the most enlightened statesmen and purest patriots that ever lived, and than whom we cannot hope to find wiser or better counsellors in the present exigency in our national affairs.

Objection was made.

June 30—Pending the consideration of the Enrollment Bill, the following amendment was offered by Mr. LE BLOND, of Ohio :

*Provided*, That no levy of troops shall be made under the provisions of this act, except by volunteering, till such time as the President of the United States shall have made a request for an armistice ; and shall have made such efforts as arc consistent with honor to restore harmony among the States, by the appointment of commissioners empowered to negotiate for peace upon the terms of a restoration of the Union under the Constitution, and until such offer shall have been rejected by the so-called Confederate government.

Which was rejected—yeas 13, nays 91, as follows:

IOHOWS:
 YEAS-Messrs. 'Ancona, Bliss, Edgerton, Eldridge, Finck, William Johnson, Long, James R. Morris, Noble, John O'Neill, Pendleton, Ross, Chillon A. While-13. Nars-Messrs. Alley, Allison, Ames, Ashley, Baily, John D. Baldwin, Baxter, Beeman, Jacob B, Blair, Boutwell, Broom-all, J. S. Brown, W. G. Bröwn, Cobb, Cole, Creswell, Dawes, Dixon, Donnelly, Driggs, Eckley, Eliot, English, Farnsworth, Fenton, Frank, Ganson, Garfield, Gooch. Griswold, Hale, Harding, Higby, Hooper, Hotchkiss, A. W. Hubbard, Hul-burd, Ingersoll, Jenckes, Julian, Kallfeick, Kel'ey, Kernan, Littlejohn, Loan, Longyear, Mallory, McAllister, McBride, McCiurg, Samuel F. Miller, Moorhead, Morrill, Amos Myers, Leonard Myers, Norton, Oddl, Charles O'Neill, Oth, Pat-terson, Radford, Randall, Alexander H. Rice, John H. Rice, Edward H. Rollins, Schenck, Scofield, Shannon, Slean, Smithers, Spalding, Stevens, Stuart, Sweed, Thayer, Thomas, Tracy, Upson, Van Valkenburgh, Wadsworth, Ellihn B, Wasbburne, William B. Washburn, Webster, Whaley, Wheeler, Williams, Wilder, Wilson, Windom, Winfield, Woodbridge-91.

## THE LATE PEACE NEGOTIATION AT NIAGARA FALLS.

The following correspondence explains itself:

[Private and confidential.]

CLIFTON HOUSE, NIAGRAF ALLS, CANDA WEST, July 12, 1864. DEAR STR: I am authorized to say that the Hon. Clement C. Clay, of Alabama, Prof. James P. Holcombe, of Virginia, and George N. Sanders, of Dixie, are ready and willing to go at once to Washington, upon complete and unqualified protection being given either by the President or Sccretary of War. Let the permission include the three names and one other. Very respectfully, GEORGE N. SANDERS.

GEORGE N. SANDERS To the Hon. HORACE GREELEY.

NIAGARA FALLS, N. Y., July 17, 1864. GENTLEMEN: I am informed that you are duly accredited from Richmond as the bearer of propositions looking to the establishment of peace; that you desire to visit Washington in the fulfillment of your mission, and that you further de-sire that Mr. George N. Sanders shall accompany you. If my information be thus far substantially correct, I am au-thorized by the President of the United States to tender you you at the earliest time that will be agreeable to you. I have the honor to be, gentlemen, yours, HORACE GREELEY.

Messis. CLEMENT C. CLAY, JACOB THOMPSON, JAMES P. HOL-COMBE, Clifton House, C. W.

CLIFTON HOUSE, NIAGARA FALLS, July 18, 1864. SIR: We have the honor to acknowledge your favor of the 17th instant, which would have been answered on yes-terday but for the absence of Mr. Clay. The safe conduct of the President of the United States

The same conduct of the President of the Onited States has been tendered us, we regret to state, under some mis-apprehension of facts. We have not been accredited to him from Richmond as the bearers of propositions looking

him from Richmond as the bearers of propositions looking to the establishment of peace. We are, however, in the confidential employment of our Government, and are entirely familiar with its wishes and opinions on that subject; and we feel authorized to declare that, if the circumstances disclosed in this correspondence that if the circumstances disclosed in this correspondence were communicated to Richmond we would be at once in-vested with the authority to which your letter refers, or other gentlemen clothed with full powers would be im-mediately sent to Washington with the view of hastening

mediately sent to Washington with the view of hastening a consummation so much to be desired, and terminating at the earliest possible moment the calamities of the war. We respectfully solicit through your intervention a safe conduct to Washington, and thence by any route which may be designated through your lines to Richmond. We would be gratified if Mr. George N. Sanders was embraced in this privilege. Permit us in conclusion to acknowledge our obligations

Permit us in conclusion to acknowledge our obligations to yon for the interest you have manifested in the further-ance of our wishes, and to express the hope that, in any event, you will afford us the opportunity of tendering them in person before you leave the Falls. We remain, very respectfully, &c., C. C. CIAY, JR., J. P. HOLCOMBE. B. S. It is proper to add that We Thermore is not have

# P. S. It is proper to add that Mr. Thompson is not here,

and has not been staying with us since our sojourn in Canada.

INTERNATIONAL HOTEL, NIGARA, N. Y., July 18, 1864. GENTLEMEN: I have the bonot to acknowledge the receipt of yours of this date by the hand of Mr. W. C. Jewett. The state of facts therein presented being materially different from that which was understood to exist by the President when he entrusted me with the safe conduct required, it seems to me on every account advisable that I should com-municate with him by telegraph and solicit fresh instruc-tions which label its once proceed to do. I hope to he abble tions, which I shall at once proceed to do. I hope to be able to transmit the result this afternoon; and, at all events, I

shall do so at the earliest moment. Yours, truly, HORACE GREE Messrs. CLEMENT C. CLAY and JAMES P. HOLCOMBE, HORACE GREELEY. Clifton House, C. W.

CLIFTON HOUSE. NIAGARA FAILS, July 18, 1864. To the Hon. H. GREELEY, Niagara Falls, N. Y. Sis: We have the honor to acknowledge the receipt of your note of this date by the hands of Col. Jewett, and will await the further answer which you propose to send us. We are, very respectfully, &c. C. C. CLAY, jr. JAMES P. HOLCOMBE.

INTERNATIONAL HOTEL, NIAGARA FALLS, N. Y., July 19, 1864. GENTLEMEN: At a late hour last evening (too late for communication with you) I received a dispatch informing me that further instructions left Washington last evening, which must reach me, if there be no interruptiou, at noen to-morrow. Should you decide to await their arrival, I feel confident that they will enable me to answer definitely your note of yesterday morring. Regretting a delay which I am sure you will regard as unavoidable on my part, I remain, yours, truly,

HORACE GREELEY. The Hon. Messrs. C. C. CLAY, jr., and J. P. HOLCOMBE, *Clifton House, Niagara, C. W.* 

CLIFTON HOUSE, NIAGARA FALLS, C. W., July 19, 1864.

Size: Colonel Jewett has just handed us your note of this date, in which you state that further instructions from Washington will reach you by noon to-morrow, if there be no interruption. One, or possibly both of us, will be obliged to leave the Falls to-day, but will return in time to receive the communication which you promise to-morrow. We remain, truly yours, &c., JAMES P. HOLCOMBE, C. C. CLAY, Jr.

The Hon. HORACE GREELEY, Now at the International Hotel.

EXECUTIVE MANSION WASHINGTON, July 18, 1864.

To whom it may concern: Any proposition which embraces the restoration of peace, the integrity of the whole Union, and the abandonment of slavery, and which comes by and with an authority that can control the armies now at war against the United States, will be received and considered by the Executive Govern-ment of the United States, and will be met by liberal terms on other substantial and collateral points, and the bearer or bearers thereof shall have safe conduct both ways. ABRAHAM LINCOLN.

INTERNATIONAL HOTEL, Wednesday. Major Hay would respectfully inquire whether Professor Holcombe and the gentlemen associated with him desire to send to Washington by Major Hay any messages in refer-ence to the communication delivered to him on yesterday. and in that case when he may expect to be favored with such messages.

Mr. Holcombe presents his compliments to Major Hay, and greatly regrets if his return to Washington has been delayed by any expectation of an answer to the communi-cation which Mr. Holcombe received from him on yester-day, to be delivered to the President of the United States. That communication was accepted as the response to a let-ter of Messrs. Clay and Holcombe to the Hon. II. Greeley, and to that gentleman an answer has been transmitted.

CLIFTON HOUSE, NIAGARA FALLS, July 21. [Copy of original letter held by me to deliver to the Hon. forace Greeley, and which duplicate I now furnish the As-sociated Press. WM. CORNELL JEWETT.]

NIAGARA FALLS, CLIFTON HOUSE, July 21, 1864.

TIGGARA FALLS, CLIFTON HOUSE, July 21, 1864. To the Hon. HORACE GREELEY. SIR: The paper handed to Mr. Holcombe, on yesterday, in your presence, by Major Hay, A. A. G., as an answer to the application in our note of the 18th instant, is couched in the full wing terms. in the following terms:

"Executive Mansion, "Washington, July 18, 1864.

· To whom it may concern:

"Any proposition which embraces the restoration of peace, "Any proposition which embraces us rescontion of peace, the integrity of the whole Union, and the abandonment of slavery, which comes by and with an authority that can control the armies now at war against the United States, will be received and considered by the Executive Govern-ment of the United States, and will be met by liberal terms, an achor substantial and collateral points and the heaves on other substantial and collateral points, and the bearer or bearers thereof shall have safe conduct both ways. "ABRAHAM LINCOLN."

.

"ABRAHAM LINCOLN." The application to which we refer was elicited by your letter of the 17th instant, in which you inform Mr. Jacob Thompson and ourselves that you were authorized by the President of the United States to tender us his safe conduct, on the hypothesis that we were "duly accredited from Richmond as bearers of propositions looking to the estab-lishment of peace," and desire a visit to Washington in the fulfilment of this mission. This assertion, to which we then gave, and still do, entire credence, was accepted by us as the evidence of an nnexpected but most gratifying change in the policy of the President—a change which we felt au-

peace mutually just, honorable, and advantageous to the North and to the So (th, exacting no condition but that we should be "duly accredited from Richmond as bearers of propositions looking to the estab ishment of peace.

Thus proffering a basis for conference as comprehensive as we could desire, it seemed to us that the President opened a door which had previously been closed against the Confederate States for a full interchange of sentiment, free dis-cussion of conflicting opinions, and untrammeled effort to remove all causes of controversy by liberal negotiations. We indeed could not claim the benefit of a sub-conduct which had been extended to us in a character we had no right to assume and had never affected to possess; but the uniform declarations of our Executive and Congress, and their thrice-repeated and as often repulsed attempts to open negotiations, furnish a sufficient pledge to us that this conciliatory manifestation on the part of the President of the United States would be met by them in a temper of equal magnanimity. We had, therefore, no hesitation in declaring that if this correspondence was communicated to the President of the Confederate States, he would promptly embrace the opportunity presented for seeking a peaceful solution of this unhappy strife. We feel confident that you must share our profound regret

that the spirit which dictated the first step toward peace had not continued to animate the councils of your President. Had the representatives of the two Governments met to consider this question, the most momentous ever met to consider this question, the most momentous ever submitted to human statesmanship, in a temper of becoming moderation and equity, followed, as their deliberations would have been, by the prayers and benedictions of every patriot and Christian on the habitable globe, who is there so hold as to pronounce that the frightful waste of indivi-bul heaviers and ublic arcmanic urbin is field to adden dual happiness and public prosperity which is daily saddening the universal heart might not have been terminated, or if the desolation and carnage of war must still be endured through weary years of blood and suffering, that there might not at least have been infused into its conduct something more of the spirit which softens and partially redeems its brutalities?

Instead of the safe conduct which we solicited, and which your first letter gave us every reason to suppose would be extended for the purpose of initiating a negotiation in which neither Government would compromise its rights or its dignity, a document has been presented which provokes as much indignation as surprise. It bears no feature of resemblance to that which was originally offered, and is unlike any paper which ever before emanated from the constitutional executive of a free people. Addressed "to whom it may concern," it precludes negotiation, and pre-scribes in advance the terms and conditions of peace. It returns to the original policy of "no bargaining, no negotiations, no truce with rebels except to bury their dead, until every man shall have laid down his arms, submitted to the Government, and sued for mercy.

Whatever may be the explanation of this sudden and entire change in the views of the President, of this rude with-drawal of a courteous overture for negotiation at the moof words of peace just uttered, and fresh blasts of war to the bitter end, we leave for the speculation of those who have the means or inclination to penetrate the mysteries of his Cabinet or fathom the caprice of his imperial will. Tt is enough for us to say that we have no use whatever for the paper which has been placed in our hands. We could not transmit it to the President of the Con-

federate States without offering him an indignity, dishonoring ourselves, and incurring the well-merited scorn of our countrymen. While an ardent desire for peace pervades the people of the Confederate States, we rejoice to believe that there are few, if any, among them who would purchase it at the expense of liberty, honor, and self-respect. If it can be secured only by their submission to terms of con-quest, the generation is yet unborn which will witness its restitution

If there be any military autocrat in the North who is en-titled to proffer the conditions of this manifesto, there is none in the South authorized to entertain them. Those who control our armies are the servants of the people-not their masters: and they have no more inclination than they have the right to subvert the social institutions of the sovereign States, to overthrow their established constitutions, and to barter away their priceless heritage of self-government.

This correspondence will not, however, we trust, prove wholly barren of good result. If there is any citizen of the Confederate States who has clung to a hope that peace was possible with this administration of the Federal Government it will strip from his eyes the last film of such delusion; or if there be any whose hearts have grown faint under the suffering and agony of this bloody strnggle, it will inspire them with fresh energy to eidure and brave what-ever may yet be requisite to preserve to themselves and their children all that gives dignity and value to life or

thorized to hope might terminate in the conclusion of a hope and consolation to death. And if there be any patriots peace mutually just, honorable, and advantageous to the or Christians in your land who shrink appelled from the which stratches before them, we pray that in their bosoms a resolution may be quickened to recall the abused authority and vindicate the outraged civilization of the country. For the solicitude you have manifested to inangurate a movement which contemplates results the most noble and humane, we return our sincere thanks, and are, most respectfully and truly, your obedient servants,

#### C. C. CLAY JAMES P. HOLCOMBE.

# CLIFTON HOUSE, NIAGARA FALLS

WEDNESDAY, July 20, 1864. Col. W. C. JEWETT, Cataract House, Niagara Falls : SR: We are in receipt of your note, admonishing us of the departure of the Hon. Horace Greeley from the Falls; that he regrets the sad termination of the initiatory steps taken for peace, in consequence of the change made by the Presidentin bis instructions to convey commissioners to Washington for negotiations unconditionally, and that Mr. Greeley will be pleased to receive any answer we may

We avail ouselves of this offer to enclose a letter to Mr. Greeley, which you will oblige us by delivering. We can-not take leave of you without expressing our thanks for your conress and kind offices as the intermediary through whom our correspondence with Mr. Greeley has been con-ducted, and assuring you that we are, very respectfully, your obedient servants,

C. C. CLAY, Jr. JAMES P. HOLCOMBE.

#### MR. GREELEY TO MR. JEWETT.

NIAGARA FALLS, (N. Y.,) July 20, 1864. DEAR SIR : In leaving the Falls, I feel bound to state that I have had no intercourse with the Confederate gentlemen at the clitton House but such as I was fully authorized to hold by the President of the United States, and that I have done nothing in the premises but in fulfillment of his injunc-tions. The notes, therefore, which you have kindly interchanged between those gentlemen and myself can in no case subject you to the imputation of unauthorized dealing with public enemies.

HORACE GREELEY. Yours. W. C. JEWETT, Esq.

MR. JEWETT TO MR. CLAY AND OTHERS.

NIAGARA FALLS, July 20, 1864.

HOD. C. C. CLAT, HOD. JACOB THOMPSON, HOD. 50, 1005. DERS, HOD. BEVERLY TUCKER, and the other HOD. Repre-sentatives of the Southern Confederacy. GENTLEMEN: I am directed by Mr. Greeley to acknowledge the receipt of the following telegram from Mr. Clay:

"ST. CATHERINE'S, July 20, 1864.

"To GEO. N. SANDERS: "Will be with you at five o'clock. Detain Greeley until I C. C. CLAY. see him.

And to state that, in view of his mission being ended, through the rejection of the terms of negotiation in the letter of the President of the United States, delivered to you by Major Hay, he does not feel himself authorized to take any further steps in the matter. He regrets the sad termination of the steps taken for peace, from the change made by the President in his instructions given him to convey commissioners to Washington for negotiations un-conditionally. He will be pleased to receive any answer you may have to make in writing through me or any mode you may desire.

I enclose you a copy of a note from Mr. Greeley addressed to me justifying the intercourse I have had with you during

In conclusion, I conder to you my heartfelt thanks for the kind and generous manner in which you have received me personally, and for the noble and magnanimous sentiments you have advanced in a desire to end the bloody conflict between the two sections. I can only regret that our Government should not have seen the policy, duty, and justice of meeting your generous offer to meet in council unconditionally—terms of a peace to depend upon circumstances transpiring during negotiations. My efforts shall be as ever unceasing for peace that shall secure to the section you represent that justice that shall neet with the approval of the civilized world, of the coming International Congress proposed by the wise and noble Napoleon. Very truly, WM. CORNELL JEWETT.

Very truly,

MAJOR HAY TO MR. HOLCOMBE.

INTERNATIONAL HOTEL, July 21, 1864. Major Hay has just received Mr. Holcombe's note of this. date, and thanking him for his prompt response, will start at once for Washington. Both Mr. Greeley and Major Hay understood Mr. Holeombe to say, yesterday, that he would bransmit to-day, and on that supposition Mr. Greeley set out for Now York yesterday, and Major Hay romained. It is a matter of no special importance. Major Hay only wishes to explain his note of to-day.

## [From the New York Tribune of July 22.]

[From the New York Tribune of July 22.] The telegraphic stories concerning peace conferences at Niagara Falls have a slender foundation in fact, but most of the details are very wide of the truth. The Editor of this paper has taken part in and been privy to no further or ther negotiations than were fully authorized, and more than authorized; but these related solely to bringing the untagonists face to face in amicable rather than beligerent titude, with a view to the initiation of an earnest effort for peace, to be prosecuted at Washington. The movement as had no immediate success. Of course all reports that the writer has been engaged in proposing, or receiving, or discussing hypothetical terms or asis of peace, whether with accredited agents of the Rich-mond authorities or others, are utterly mistaken. He has hever had the slightest authorization to do anything of the sort; and he is quite aware of those provisions of law which

aver man the slightest authorization to do anything of the sort; and he is quite aware of those provisions of law which relate to volunteer negotiators with public enemies. Those provisions he heartily approves, and is nowise inclined to riolate.

More than this he does not at yet feel at liberty to state, hough he soon may be. All that he can now add is his general inference that the pacification of our country is neither so difficult nor so distant as seems to be generally supposed.

# Rebel Views of "Peace." 1862.

In September, 1862, these proceedings are published as having taken place in the Rebel Congress :

In the House of Representatives Mr. FOOTE, of Tennessee, offered the following joint resolution proposing to send a commissioner to Washington, empowered to propose terms of just and honorable peace :

Be it enacted by the Congress of the Confederate States of America, That the signal success with which Divine Provi-dence has so continually blessed our arms for several months past would fully justify the Confederate Govern-ment in dispatching a commissioner or commissioners to the Government at Washington city, empowered to pro-pose the terms of a just an honorable peace.

Mr. Holt, of Georgia, asked the consent of the House to offer the following substitute for the resolution :

The people of the Confederate States are, and have been from the beginning, anxious that the war with the United States should be conducted with the sense established by the rules of civilized and Christian nations, and have, on their part, so conducted it, and the said people ardently de-sire that said war should cease and peace be restored, and

sire that said war should cease and peace be restored, and have so announced from the beginning: Therefore, *Resolved*, That, whenever the United States Government shall manifest a like anxiety and a like desire, it shall be the duty of the President of the Confederate States to ap-point — commissioners to treat and negotiate with the said United States Government upon said subjects, or either of them of them.

On motion of Mr. KENAN, of Georgia, the resolution and substitute were laid upon the table-yeas 59, nays 26.

#### 1863.

In January, Mr. CROCKETT, of Kentucky, introduced into the House a resolution with reference to the conditions on which peace should be negotiated; which was debated.

Mr. FOOTE, of Tennessee,\* also introduced

\* The Atlanta (Georgia) Intelligencer of the 20th of Janu-

"The resolution introduced by Mr. Foote in Congress "The resolution introduced by Mr. Foote in Congress bearing upon a reconstruction of our Government with the Sour Government Northwestern States, we desire now solemnly to protest against, and we trust that they will be tabled by the Con-

The people of the Confederate States of America having, in the progress of the pending war, most clearly demon-strated their ability to maintain by arms the claim to sepa-rate independence, which they have herefobre asserted be-fore the world, and being inflexibly resolved never to re-linquish the struggle in which they are engaged until the great object for which they are contending shall have been finally accomplished; in view of the fact that a great politi-cal reaction in any strike blocky and upnot rul war finally accomplished; in view of the fact that a great point-cal reaction in opposition to the bloody and unnatural war now in course of prosecution, has displayed itself in several of the most populous and influential States of what was once honorably known as "The United States of America;" and, in view of the additional fact that, even among the avoved opponents of despotism, and the recognized friends of peace opponents of despotsm, and the recognized friends of peace in the North, a grave and deplorable missipprehension has of late arisen in regard to the true condition of public senti-ment in the South touching the question of reconstructing that political Union once existing under the protection of what is known as the Federal Constitution. Now, in order that as when a superferred the order being a set the line as formed to that as a set of the set that no further misunderstanding of the kind referred to may hereafter prevail, and in order that the unchangeable determination of our Government and people, in reference to the terms upon which alone they would bring the sanguinary struggle to a close, may be made known, the Congress of the Confederate States of America do resolve as follows:

1. There is no plan of reconstructing what was formerly known as the Federal Union to which the people of the Con-federate States will ever consent. Wrongs too grievous and rederate states win ever consist. Wrongs too grevous and multiplied have been committed upon us and upon our most cherished rights by a united North, since this unprovoked and most wicked war commenced; a majority of the people of the northern States have too evidently shown themselves to be utterly incapable of self-government and unmindful of all the fundamental principles upon which alone repub-lican institutions can be maintained. They have too long submitted patiently to the iron rule of the basest and most degraded desposite that the world has yet known; for too long a period of time they have openly and unblushingly sympathized with the lawless and forceious miscreants who have hear south into the bases of the unefinding fourth for have been sent into the bosom of the unoffending South to spill the precious blood of our most valued citizens; to pol-lute and desecrate all that we hold in especial veneration: to rob us of our property; to expel us from our homes and

Interind descripte all that we hold in especial veneration; to rob us of our property; to expel us from our homes and wantonly to devastate our country, to allow even of the possibility of our ever again consenting to hold the least political connection with those who have so cruelly out-raged our sensibilities and so profoundly dishonored them-selves, and in association with whom we feel that we could not expect that freedom which we love, that self-respect which we are determined ever to cultivate, and the esteem and sympathy of civilized and Christian nations. 2. While the Confederate States of America are not at all responsible for the existing war, and have been at all times ready to participate in such arrangements as would be best suited to bring it to a close, in a manner consistent-with their own safety and honor, they could not yield their consent to an armistice of a single day or hour, so long as the incendiary proclamation of the atrocious monster, now bearing rule in Washington city, shall remain unrevoked; nor could the government of said Confederate States agree to negotiate at all in regard to a suspension of hostilities, except upon the basis of a formal and unconditional recog-nition of their independence. nition of their independence

a. Whenever the iriends of peace in the North shall grow strong enough to constrain Abraham Lincoln and his flagi-tions (Ashinet to withdraw said proclamation, and propose an armistice upon the basis aforesaid, the Government of the Confederate States will be ready to accede to said prop-osition of annihilice with a view to the settlement of all ex-tinguishing the settlement of all ex-tension of annihilice with a site of the settlement of all ex-tension. isting difficulties.

4. Should peace be at any time brought about, the Con-federate States of America would freely consent to the formation of a just and mutually advantageous commercial

federate Congress, whenever they are taken up to be con-sidered. We are fighting this war for Southern independ-ence and for a Government of Southern States, recognizing African slavery as an institution ordained of God, beneficial Afficiant states as an institution organized of you, pereficient to marking, a necessity in our social and political relations as States, and in our intercourse with all other nations or States. Hence the admission of any free State into our Union is not only repugnant to us, but it will be only a con-tinuance of that evil which has brought on the war, and which to get rid of we are now fighting. If the Northwest-ern States should shake off the North and East, and set up on the theoretical or a construction of the state as the peace for themselves a new Government, and desire to be at peace with the South, no barrier will be placed in their way by our Government, and we shall be willing to treat with them as an independent Government—in peace, as friends; in

treaty with all the States now constituting the United States, except New England, with whose people, and in whose ig-noble love of gold and brutifying fanaticism this disgraceful war has mainly originated; in consideration of which fact the people of the Confederate States of America are firmly and deliberately resolved to have no intercourse whatever hereafter, either direct or indirect, political, commercial, or social, under any circumstances which could be possibly imagined to exist, with said States of New England, or the people therein resident.

The Government of the Confederate States, in consideration of the change in public sentiment which has oc-curred in several of the Northern States, wherein political elections have been recently held—sympathizing most kindly with those by whose manly exertions that change Andly with those by whose many exertions that change has been brought about—would be willing to conclude a just and honorable peace with any one or more of said States who, renouncing all political connection with New England, may be found willing to stipulate for desisting at once from the further prosecution of the war against the South, and iu such case the Government of the Confederate States would be willing to enter into a league, offensive and defensive, with the States thus desisting, of a permanent and enduring character. 6. The Government of the Confederate States is now

willing, as it has heretofore repeatedly avowed itself to be, whenever the States bordering upon the Mississippi river, or any of them, shall have declared their inclination to withdraw from the further prosecution of the war upon the South, (which, could it be successful, would only have the effect of destroying their own best market,) to guaranty to them in the most effectual and satisfactory mauner the peaceful and uninterrupted navigation of the said Mississippi river and its tributaries, and to open to them at once the markets of the South, greatly enhanced in value to them as they would be by the permanent exclusion of all articles of New England growth or manufacture.

7. The course of practical neutrality in regard to the pending war heretofore pursued by the States and Territotories west of the Rocky Mountains has afforded the highest gratification to the people of the Confederate States of America; and it is hoped that the day is not far distant when said States and Territories, consulting their own ob-vious safety and future welfare, will withdraw from all Violas safety and future wenare, will without an inom an political connection with a Government which has hereto-fore been a source of continual oppression to them; and when said States and Territories, asserting their separate independence, shall appropriate to themselves the manifold advantages sure to result from such a movement, among which are be necessarily from such a movement, among which may be reckoned: 1st. Relief from grievous and ex-hausting tariff regulations, now being rigidly enforced. 2d. Relief from all the discredit resulting inevitably from the prosecution of the present unjust and unauthorized war. 3d. Relief from the pressure of a despotism the most heartless and atrocious ever yet established. 4th. Relief from the crushing weight of taxation unavoidably growing out of the war. 5th. The exclusive use and enjoyment of all of the war, but the exclusive use and enjoyment of an the rich mineral lands stretching along the slope of the Pa-cific. 6th. Free trade with all the nations of the earth, and a future maritime growth and power that has no parallel; and lastly, a monepoly of the trade of the Pacific occean. 8. Resolved, That the President be respectfully requested, if he then been maching the parameters to be

if he shall approve these resolutions, to cause them to be promulgated and transmitted to the States of the North by such means as he shall deem most judicious; and that he accompany them, if he think it advisable, with such an address or proclamation, expository of the matters embodied therein, as he shall judge most suitable and proper.

Jan. 21-Mr. FOOTE remarked in relation to proposed retaliation by Mr. Clopton:

He (Mr. Foote) did not certainly intend to call into ques-tion the motives or acts of the President in reference to this matter; but the gentleman from Virginia must be aware that the judiciarics of most of the States were most familiar with their own laws on the subject under consideration, and know what would be most satisfactory to their own people better than the Confederate Government could by boostilly know. But one point he desired especially to no-tice. It could not but be apparent to every one how these measures of retaliation would result—the amount of blood-shed and terrible atrocities to which it would lead. Therefore he desired that a messenger or messengers should be sent to the Northern Government, to propose terms of hon-orable peace-to let them know what was to be expected by a continuance of the war under present auspices. This would give heart to that great Peace party which is now springing up at the North and daily increasing in strength, springing up at the rotation and daily increasing in scrength, especially in the Northwestern States, where the people are already clamorous for peace. And if the Lincoln Gov-ernment still persisted in their atrocious course, our action would show the world that upon that Government rested all the future responsibility. It would show that if they

persisted in this demonlacal and hellish warfare, we are not to blame. He was sure, however, that these politics? uprisings in the States of Illinois, Indiana, and Ohio, were of no little moment, and the results would presently appear.

In September, certain "peace resolutions" were introduced into the Senate of Virginia, which proposed in substance that three commissioners should be appointed by each of the Confederate States, to repair to each of the States remaining in the old Union, and make known to the Governors of each of them that the Confederate States demand that they will. by the ballot-box, consent that the Confederate States be allowed thenceforth to be separated from them in peace.

These resolutions were discussed at some length, and then indefinitely postponed by a unanimous vote. When they were first offered and read-

Mr. ARMSTRONG moved that the rules be suspended for the purpose of acting on the resolutions forthwith.

Mr. Collier appealed to the Senate to allow the resolntions to lay on the table, in order that Senators might have full time to reflect on their importance.

Mr. SEDDON was unwilling to allow the silent sanction of the Senate to endorse for an hour these resolutions. He Was in favor of disposing of them without delay. The rules were then suspended. Mr. NASH moved the indefinite postponement of the reso-

lutions. He regretted painfully to see such resolutions offered. He objected to them  $toto \ c \alpha l o$ . He was as much opposed to sending commissioners to Gov. Morgan, or any other northern Governor, as to Mr. Lincoln. In their views

other northern Governor, as to Mr. Lincoln. In their views and public policy they were all alike. Mr. Collies thought the action of the Senate was hasty and inconsiderate. The scheme he proposed never had been attempted. There was no clause in the Constitution that provided for a peaceful dissolution of the Union. A sort of dread filled the public mind in regard to propositions of peace, because they might do harm. If there was any ex-mension in the present or which is undertained pression in the preamble or resolutions which inculcates the idea that we are to desist from the struggle till the inthe idea that we are to desix from the struggle till the in-dependence of this Confederacy was achieved he would like to know it. He was willing to fight on, to fight on till we should obtain our independence; but while fighting with one hand he would hold out the olive-branch of peace with the other. He hoped, if the resolutions were to be buried, they would be allowed the decent respect shown to all

corpses, of remaining twenty-four hours before interment. The roll was called, and the vote stood—ayes 38, noes 1, (Mr. Collier.) So the preamble and resolutions were indefinitely postponed.

# In the Legislature of Georgia, the same subject was considered, on which the Savannah Republican remarked:

We hope the Georgia Legislature will let this question alone, and turn their attention to war. The peace talk is designed to help the Northern Democrats, but it is a great mistake. It helps Lincoln, as we shall see to our sorrow.

In the Legislature of North Carolina.\* some

\* March 4, 1864, Wm. W. Holden issued this card:

To the People of North Carolina: In compliance with the wishes of many friends, I announce myself a candidate for the office of Governor of North Carolina, at the election to be held on the first Tuesday in August next

My principles and views, as a conservative, "after the strictest sect," are well known to the people of the State. These principles and views are what they have been. They will not be changed.

will not be changed. I am not disposed, at a time like this, to invite the people from their employments, and add to the excitement which prevails in the public mind by haranguing them for their votes. We need all our energies to meet the common enemy, and to provide means of subsistence for our troops in the field and the people at home. Let the people go calmly and firmly to the polls and vote for the man of their choice. I will cheerfully abide their decisions whatever it

If elected I will do everything in my power to promote the interests, the honor, and the glory of North Carolina, and to secure an honorable pace.

# resolutions were submitted, which are thus described by a Richmond paper :

They assert the right of the people to meet and consult for the good of the country; denounce mob violence and military aggression upon the freedom of the press; pledge the State to a firm maintenance of the decisions of its legal tribunals, and appland Gov. Yance for his many defence of the State judiciary; They turther compliment the army the State judiciary. They further compliment the army for its gallantry and heroism, and urge a faithful discharge of duty in vigorously prosecuting the war for national inde-pendence. They turther declare that formal negotiations for peace, on the basis of separation from the United States, for peace, on the basis of separation from the United States, should be instituted by the treaty-making power, and urge our Representatives in Congress to exert themselves to bring about such negotiations. They further recommend proposals from the Confederate authorities to the Federal Congress looking to the holding of a peace convention for the adjustment of difficulties where action should be authorities the adjustment of difficulties, whose action shall be subject to the ratification of the people.

### TERMS OF PEACE INDICATED BY THE RICHMOND ENQUIRER.

#### [From the Richmond Enquirer, Oct. 16.]

Notwithstanding the cheering rumors of an early recognition by France, notwithstanding the fact that the Cabinet at Washington has been considering "proposals for peace" -proposals of its owu, for it has none of ours to consider-notwithstanding the White Flag that was seen in the notwithstanding the White Flag that was seen in the heavens by a respectable woman, and the march of ærial troops northwards, witnessed by a man of good character, in the clouds—in spite of all this, there is not the slightest prospect of speedy peace visible to us. There will be no peace until the military power of the Yankce nation is en-tirely broken, and its people so thoroughly sickened of the war that we can exact our own terms.

Save on our own terms, we can accept no peace what-ever, and must fight till doomsday rather than yield an iota of them; and our terms are: Recognition by the enemy of the independence of the

Confederate States

Withdrawal of Yankee forces from every foot of Confederate ground, including Kentucky and Missouri.

Withdrawal of Yankee soldiers from Maryland until that State shall decide, by a free vote, whether she shall remain in the old Union or ask admission into the Confederacy.

Consent, on the part of the Federal Government to give up to the Confederacy its proportion of the navy as it stood

at the time of secession, or to pay for the same. Yielding up of all pretension, on the part of the Federal Government, to that portion of the old Territories which lies west of the Confederate States.

An equitable settlement—on the basis of our absolute independence and equal rights—of all accounts of the pub-lic debts and public lands, and of the advantages accruing from foreign treaties.

These provisions, we apprehend, comprise the minimum of what we must require before we lay down our arms. That is to say, the North must yield all; we nothing. The whole pretension of that country to prevent by force the separation of the States must be abandoned, which will be an equivalent to an avowal that our enemies were wrong from the first; and, of course, as they waged a causeless and wicked war upon us, they ought, in strict justice, to be required, according to usages in such cases, to reimburse to us the whole of our expenses and losses in the course of that war.

Whether this last proviso is to be insisted upon or not certain we are that we cannot have any peace at all until we shall be in a position, not only to demand and exact, but also to enforce and collect, the treasure for our own reimbursement out of the wealthy cities in the enemy's In other words, unless we can destroy or scatter country. their armies and break up their Government, we can have we peace; and if we can do that, then we can, and ought not only to extort from them our own full terms, and ample acknowledgment of their wrong, but also a handsome indemnity for the trouble and expense caused to us by their crime.

Now, we are not yet in a position to dictate those terms to our enemies, with Rosecrans's army still in the heart of our country, and Meade still on Virginia soil; but though it is too soon to propose such conditions to them, yet it is important that we should keep them plainly before our own Even as that we should keep black planty bend but own gives as the only admissible basis of any conceivable peace. This well fixed in the Confederate mind, there will be no more fearful looking for of news from Europe, as if that blessed peace were to come to us over the sea, and not to be compared on over the sea. conquered on our own ground; no more gaping for hints of recognition, "filling the belly with the east wind;" no more distraction or diversion from the single momentous ousiness of bracing np every nerve and sinew of the country for battle.

20

It is especially now, at a moment when great and perhaps decisive battles are impending at two or three points, that we think it most essential to insist upon the grand entire magnificence of the stake and of the cause. Once more we say, it is all or nothing. This Confederacy or the Yan-kee nation, one or the other, goes down, down to perdition; that is to say, one or the other must forfeit its national ex-istence, and lie at the mercy of its mortal enemy. We all know by this time the fate in store for us if we succumb. The other party has no smaller stake. As surely as we completely ruin their armies—and without that is no peace debt, though we wring it out of their hearts. And they know it well; and therefore they cannot make peace except through utter exhaustion and absolute inability to strike another blow.

The stake they have to forfeit, then, if they lose this dreadful game, is as vital as ours. So is the stake to be won if they win anything. It is no less than entire possession of our whole country, with us in it, and everything that is ours, from the Ohio to the Rio Grande, to have and to hold to them and their heirs forever. But, on the other hand, what we mean to win is utter separation from them for all We do not want to govern their country; but after time. levying upon it what seemeth good to us, by way of indem-nity, we leave it to commence its political life again from the beginning, hoping that the lesson may have made them sadder and wiser Yankees. We shut them out forever, with all their unclean and scoundrelly ways, intending to lead our lives here in our own Confederate way within our own well-guarded bounds; and without, as St. John says, "without are dogs."

And let no Confederate of feeble knees and tremulous backbone say to us, this complete triumph is impossible; say that we must be content with some kind of compromise, and give and take. On the contrary, we must gain all or lose all; and that the Confederates will indeed win the giant game, we take to be as certain as any future event in this nucertain world. Meade's army and Reservan's once this uncertain world. Meade's army and Rosecrans's once scattered Lincoln can get no more armics. The draft turns out manifestly fruitless. Both the German and the Irish elements are now for peace! The Yankees lave to bear the brunt of the war themselves; but, in the mean time, their inevitable bankruptcy is advancing like an armed man; 'hungry ruin has them in the wind.'' It cannot be long before the Cabinet at Washington will have indeed to con-sider seriously. "promosals of neaco?" muder ausnices and Sider seriously "proposals of peace," under auspices and circumstances very different from the present. For the present the war rolls and thunders on—and may God defend the right!

#### 1864.

[From the Richmond Examiner, February 8.]

The following extraordinary resolutions were yesterday introduced in the House of Representatives by Mr. Wright, of Georgia. The House went into secret session before taking any action upon them : Whereas the President of the United States, in a late pub-

lic communication, did declare that no propositions for peace had yet been made to that Government by the Confederate States, when, in truth, such propositions were pre-vented from being made by the President of the United States, in that he refused to hear, or to receive, two commissioners appointed to treat expressly of the preservation of amicable relations between the two Governments. Nevertheless that the Confederate States may stand justi-

Nevertheless that the Contractate States may stand justi-fied in the sight of the conservative men of the North of all parties, and that the world may know which of the two Governments it is that urges on a war unparalleled for the fierceness of the conflict, and intensifying into a sectional hared unsurpassed in the annals of mankind: Therefore, *Resolved*, That the Confederate States invite the United

States, through their Government at Washington, to meet them by Representatives equal to their Representatives and Senators in their respective Congress, at --, on the day of — next, to consider, First. Whether they cannot agree upon the recognition

of the Confederate States of America.

Second. In the event of such recognition, whether they can-not agree upon the formation of a new government, founded upon the equality and sovereignty of the States; but if this cannot be done, to consider—

Third. Whether they cannot agree upon treaties, offensive, defeusive, and commercial.

Resolved, In the event of the passage of these resolutions, the President be requested to communicate the same to the Government at Washington in such manner as he shall deem most in accordance with the usages of nations; and in the event of their acceptance by that Government, he do issue his proclamation of election of delegates, under such regulations as he may deem expedient.

In the House of Representatives, May 23, Mr. J. T. LEACH, of North Carolina, submitted the following preamble and resolutions, which were read by the clerk, creating quite a sensation on the floor, among the privileged seats, and in the galleries :

Whereas, The unconstitutional enactment of laws by the Congress of the United States, upon subjects of vital im-portance to the harmony and independence of the States, the happiness and prosperity of the people, the preserva-tion and perpetuation of the Union, against the demands of instinct the unreale and advections of the best rat where justice, the appeals and admonitions of her hest and wisest statesmen, made it our painful duty to fall hack upon the rights for which the colonies maintained the war of the Revolution, and which our forefathers asserted and maintained to be clear and inalienable:

Resolved, by the Congress of the Confederate States, That the delegates from each State, acting in sovereign and independent character, for the purpose of adding moral to our physical force, and placing ourselves properly before the civilized world, do niost earnestly appeal to the President. by and with the advice and consent of the Senate, to appoint commissioners whose duty it shall be to propose an armistice of ninety days to the proper authorities of the Federal Gov-ernment, preliminary to negotiation upon State sovereignty and independence; and the said commissioners shall report in writing to the President the answer received from the Federal Government upon the subject.

Resolved, That should the peace-making power of the Federal Government accede to the proposition for an armis-tice of ninety days, the President be requested to convene the Congress of the Confederate States for the purpose of appointing commissioners, by and with the advice and consent of the Senate; and that he be also requested to notify the executive of the several States of the fact, and ask their co-operation by appointing commissioners, either by the Legislature or by convention, to co-operate with the cominsistences appointed by the President and to negotiate with the commissioners appointed by the Federal Government upon such terms of peace as will be consistent with the honor, dignity, and independence of the States, and compat-

ible with the safety of our social and political rights. Resolved, That in maintaining the rights guaranteed to us by the blood and treasure of our revolutionary fathers, and dear at all times to freemen, we desire to be let alone. We take no man's property; we fight not for conquest, but for our rights, the independence of the States, our equality, our civil and religious liberties.

Resolved, That such terms of peace as agreed to by the commissioners ought to be endorsed by the President and Senate, and submitted to the people for their ratification or rejection.

Mr. FOOTE, of Tennessee, moved the reference of the pre-amble and resolutions to the Committee on Foreign Affairs.

Mr. McMullen inquired whether the subject of the resolutions was not a proper one for the sceret session. Mr. LEACH said he desired it to be discussed in open ses-

sion.

Mr. CONRAD rose to a point of order. A rule was adopted at the last session, which is still force, requiring that all such resolutions be received and discussed in secret session. The SPEAKER said that such a rule had been adopted, but

it had special reference to the last session. Mr. LEACH insisted that the importance of the question involved in the resolutions should commend them to the gravest considerations of the body. They werenot his views only, but the views of his constituents, and a good portion of the people of North Carolina. Mr. A. H. GARLAND, of Arkansas, hoped the gentleman

would not proceed until the Chair had settled the point of order that had been raised.

Mr. FOCTE renewed his motion to refer the resolutions to the Committee on Foreign Affairs.

Mr. MCMULLEN moved to lay the resolutions upon the table.

Mr. MARSHALL, of Kentucky, rose to a point of order. The gentleman from North Carolina [Mr. Leach] had the floor, and he had not yielded it. The gentleman from Virginia [Mr. McMullon] was usurping the floor. Mr. Foorz didn't want to cut off any remarks the gentle-man from North Carolina might have to make in support of his resolutions

of his resolutions

Mr. LEACH said the presentment of the resolutions was and the would be satisfied.

Mr. McMullen withdrew his motion, and Mr. HEISKELL,

lutions submitted by the gentleman from North Carolina, [Mr. Leach.] and he found that they look to separate State action in the prosecution of peace. Taking that view of them, he could not give them his countenance or support. He asked leave to change his vote on the motion to lay them upon the table. He had voted "aye;" he would now vote no.

Mr. J. T. LEACH said he held in his hand the address of the last Congress to the people of the Coufederate States. The principles there enunciated were embodied almost wholly in these resolutions. Mr. SWAN, of Tennessee, asked and obtained leave to

change his vote on the motion to lay the resolutions on the table.

Mr. READ, of Kentucky, moved to reconsider the vote by which the resolutions were disposed of, so that all the mem-bers might exercise their discretion in voting.

• Mr. MARSHALL, of Kentucky, was for treating the resolu-tions of the gentleman from North Carolina with becoming espect. They did not represent his own views more than the views of his congressional district. He thinks they are proper and right, and so think his constituents.

The debate soon began to be somewhat stormy and personal, and, after much wrangling, the resolutions were tabled by 62 ayes to 22 nays.

### CORRESPONDENCE BETWEEN JEFF. DAVIS AND GOV. VANCE, OF NORTH CAROLINA.

[From the Petersburg Express, May 24.]

STATE OF NORTH CAROLINA, EXECUTIVE DEPARTMENT,

RALEIGH, December 30, 1863.

His Excellency President DAVIS:

My DEAR SIR: After a careful consideration of all the sources of discontent in North Carolina, I have concluded that it will be impossible to remove it, except by making some effort at negotiation with the enemy

The recent action of the Federal House of Representatives, though meaning very little, has greatly excited the public hope that the northern mind is looking toward peace. I am promised by all men who advocate this course that if and incomised by all their into activity that the activity to strengthen and intensify the war feeling, and will rally all classes to a more cordial support of the Government.

And although our position is well known as demanding only to be let alone, yet it seems to me that for the sake of humanity, without having any weak or improper motives attributed to us, we might, with propriety, constantly lender negotiations. In doing so we would keep conspicuously before the world a disclaimer of our responsibility for the great slaughter of our race, and convince the humblest of our citizens—who sometimes forget the actual situation— that the Government is tender of their lives and happiness, and would not prolong their sufferings unnecessarily one moment.

Though statesmen might regard this as useless, the peo-ple will not, and I think our cause will be strengthened thereby. I have not suggested the method of these nego-tiations or their terms. The effort to obtain peace is the principal matter.

Allow me to beg your earnest consideration of this suggestion.

Very respectfully yours, Z. B. VANCE.

#### DAVIS'S REPLY.

EXECUTIVE OFFICE, RICHMOND, January 8, 1864. DEAR SIR: I have received your letter of the 30th ult.,

containing suggestions of the measures to be adopted for the purpose of removing "the sources of discontent" in North Carolina. The contents of the letter are substantially the Same as those of the letter addressed by you to Scattor Dortch, extracts of which were by him read to me. I remarked to Mr. Dortch that you were probably not aware of the obstacles to the course you indicated, and,

without expressing an opinion on the merits of the pro-posed policy, desired him in answering your letter to write for suggestions as to the method of opening negotiations, and as to the terms which you thought should be offered to the enemy.

I felt persuaded you would appreciate the difficulties as soon as your attention was called to the necessity of con-sidering the subject in detail. As you have made no sug-gestions touching the manner of overcoming the obstacles, I infer that you were not apprised by Mr. Dortch of my remarks to him.

Apart from insuperable objections to the line of policy you propose (and to which I will presently advert) I cannot see how the more material obstacles are to be surmounted. The ayes and noes were taken and the resolutions tabled. After a chort interval, Mr. CONRAD, of Louisiana, rose to a personal explanation. He had read over the peace resoA second time I sent a military officer, with a communi-cation addressed by myself to President Lincoln. The letter was received by General Scott, who did not permit the officer to see Mr. Lincoln; but promised that an answer would be sent. No answer has ever been received. The bird time a few boothe area of goutherman was sent whose wom he sent. No answer has ever over received. The third time, a few honths ago, a genithman was sent, whose position, character, and reputation were such as to insure his reception, if the enemy were not determined to receive no proposals whatever from the Government. Vice Presi-no proposals whatever from the Government. dent Stephens made a patricite tender of his services in the hope of being able to promote the cause of humanity, and although little belief was entertained of his success, I cheerfully yielded to his suggestion, that the experiment should be tried.

The enemy refused to let him pass through their lines, or to hold any conference with them. He was stopped be fore he ever reached Fortress Monroe on his way to Washington.\* To attempt again in the face of these repeated rejections of all conference with us, to send commissioners or agents to propose peace, is to invite insult and contumely, and to subject ourselves to indignity without the slightest chance of being listened to.

No true citizen, no man who has our cause at heart, can desire this; and the good people of North Carolina would be the last to approve of such an attempt, if aware of all the facts. So far from removing sources of discontent, such a course would receive, as it would merit, the condemnation a course would receive, as it would ment, the condemnation of those true patriots who have given their blood and their treasure to maintain the freedom, equality, and independ-ence which descended to them from the immortal herces of King's Mountain and other battle-fields of the Revolution. If, then, these proposals cannot be made through envoys, because the enemy would not receive them, how is it possible to communicate our desire for peace otherwise than by the public announcements contained in almost every mes-Bage I ever sent to Congress? I cannot recall at this time one instance in which I have

failed to announce that our only desire was peace, and the only terms which formed a *sine qua non*, were precisely those that you suggested, namely "a demand only to be let **a**lone." But suppose it were practicable to obtain a conference through commissioners with the Government of President Lincoln, is it at this moment that we are to consider

it desirable or even at all admissible? Have we not just been apprised by that despot that we can only expect his gracious pardon by emancipating all our slaves, swearing allegiance and obedience to him and bit smalls, swatch, and becoming, in point of fact, the slaves of our own negroes? Can there be in North Carolina one citizen so fallen beneath the dignity of his ancestors as to accept, or to enter conference on the basis of these terms?

That there are a few traitors in the State who would be That there are a few traitors in the state who would be willing to betray their follow-citizens to such a degraded condition, in hope of being rewarded for treachery by an escape from the common doom, may be true But I do not believe that the vilest wretch would accept such terms for himself. I cannot conceive how the people of your State, than which none has sent nobler or more guilant soldiers to the field of battle, (one of whom it is your honor to be,) can have been deceived by anything to which you refer in "the recent action of the Federal House of Representa-tives." I have seen no action of that House that does not indicate by a yerv decided maiority, the purpose of the indicate, by a very decided majority, the purpose of the enemy to refuse all terms of the South, except absolute, un-conditional subjugation or extermination. But if it were

\*This is the correspondence :

CONFEDERATE STATES STEAMER TORPEDO, IN JAMES RIVER, July 4, 1863.

SIR: As military commissioner, I am the bearer of a com-munication in writing from Jefferson Davis, Commander-in-Chief of the land and naval forces of the Confederate States, to Abraham Lincoln, Commander-in-Chief of the States, to Abraham Lincoln, Commander-in-Chief of the land and naval forces of the United States. The Hon. Robert Onld, Confederate States agent of exchauge, ac-companies me as secretary, for the purpose of delivering the communication in person and conferring upon the sub-ject to which it relates. I desire to proceed to Washington in the steamer Torpedo, commanded by Lieut. Hunter Davidson, of the Confederate States Navy, no person being on board but the Hon. Mr. Ould, myself, and the boat's offi-cern and crew. on oozru 32. ... cers and crew. Yours, most respectfully, ALEX. H. STEPHENS.

Acting Rear Admiral S. H. LEE, Hampton Roads: The request of Alex. H. Stephens is inadmissible. The customary agents and channels are adequate for all needful military communications and conference between the United States forces and the insurgents. GDEON WELLES

GIDEON WELLES.

Sec. of the Navy.

otherwise, how are we to treat with the House of Representatives?

It is with Lincoin alone that we ever could confer, and his own partisans at the North arow unequivocally that his purpose in his message and proclamation was to shut out all hope that he could ever treat with us on any terms. If we will break up our Government, dissolve the Confederadisband our armies, emancipate our slaves, take an oath of, instants of allegiance binding ourselves to obedience to him and of disloyalty to our own States, he proposes to pardon us and not to plunder us of anything more than the property al-ready stolen from us, and such slaves as still remain. In order to render his proposals so insulting as to secure their rejection, ho joins to them a promise to support with his army one-tenth of the people of any State who will attempt to set, up a government over the other nine-tenths, thus seeking to sow discord and suspicion among the people of the several States, and excite them to civil war in furtherance of his ends.

I know well it would be impossible to get your people, if they possessed full knowledge of these facts, to consent that they possessed this knowledge of bisse facts, to content that proposals should now he made by us to those who control the Government at Washington. Your own well-known devotion to the great cause of liberty and independence, to which we have all committed whatever we have of earthly possessions, would induce you to take the lead in repelling the bare thought of abject submission to the enemy.

the bare thought of abject submission to the enemy. Yet peace on other terms is now impossible. To obtain the sole terms to which you or I could listen, this struggle must continue until the enemy is beaten out of his vain confidence of our subjugation. Then, and not till then, will it be possible to treat for peace. Till then, all tender of terms to the enemy will be received as proof that we are ready for submission, and will encourage him in the atrocious warfare which he is waging. I fear much from the tenor of the news I receive from

North Carolina, that an attempt will be made by some bad men to inaugurate movements which must be considered as equivalent to aid and comfort to the enemy, and which all patriots should combine to put down at any cost.

All patriots should combine to put down at any cost. You may count on my aid in every effort to spare your State the scenes of civil warfare, which will devastate its homes if the designs of these traitors be suffered to make headway. I know that you will place yourself in your le-gitimate position in the lead of those who will not suffer the name of the Old North State to be blackened by such a stain.

Will you pardon me for suggesting that my only source of disquietude on the subject arises from the fear that you will delay too long the action which now appears inevitable, and that by an overearnest desire to reclaim by conciliation men whom you believe to be sound at heart, but whose loyalty is more than suspected elsewhere, you will permit them to gather such strength as to require more violent measures than are now needed?

than are now needed? With your influence and position, the promoters of the unfounded discontent now prevalent in your State would be put down without the use of physical force if you would abandon a policy of conciliation and set them at defiance. In this course, frankly and firmly pursued, you could rally around you all that is best and noblest in your State, and

around you are the is been taken to be bloodless. If the contrary policy be adopted, I much fear you will be driven to the use of force to repress treason. In either event, however, be assured that you will have my cordial event, however, be assured that you will have my contain concurrence and assistance in maintaining with you the honor, dignity, and fair fame of your State, and in your efforts to crush treason, whether incipient, as I believe it now to be, or more matured, as I believe, if not firmly met, it will in our future inevitably become.

I have the honor to be, very respectfully, yours, JEFFERSON DAVIS.

His Excellency Z. B. VANCE, Governor of North Carolina, Raleigh, N. C.

### THE LATEST FROM JEFF. DAVIS.

James R. Gilmore-otherwise known as "Edmund Kirke"-who recently visited Richmond with Col. Jacques, writes an explanatory note respecting his visit to the Boston Transcript of July 22, 1864, in which he says of their "mission'':

It will result in nothing. Jefferson Davis said to me last Sunday, (and with all his faults I believe him to be a man of truth): "This war must go on till the last of this gen-eration fails in his tracks, and his children scize his musket and fight our hattle, unless yon acknowledge our right to self-government. We are not fighting for slavery; we are fighting for independence, and that or extermination we will have."

# MILITARY ORDERS RESPECTING ELECTIONS.

### Maryland.

EXTRACT FROM AN ACT OF ASSEMBLY OF 1860.

No commissioned or non-commissioned officer, having command of any solider or soliders quartered or posted in any district of any county in this State, shall muster or embody any of the said troops, or march any recruiting party within the view of any place of election, during the time of holding said election, under the penalty of one hundred collars. This section not to apply to the city of Baltimore.

#### ELECTION OF 1861.

1861, Oct. 29-General Marcy, Chief of Mc-Clellan's Staff, issued this order :

## HEADQUARTERS ARMY OF THE POTOMAC

WASHINGTON, OCTOBER 29, 1861. GENERAL: There is an apprehension among Union citi-zens in many parts of Maryland of an attempt at interference with their rights of suffrage by disunion citizens on the oc casion of the election to take place on the 6th of November next

In order to prevent this, the major general commanding directs that you send detachments of a sufficient number of men to the different points in your vicinity where the elections are to be held to protect the Union voters, and to see that no disunionists are allowed to intimidate them, or in any way to interfere with their rights

He also desires you to arrest and hold in confinement till after the election all disunionists who are known to have returned from Virginia recently and who are known to have sat the polls, and to guard effectually against any invasion of the peace and order of the election. For the purpose of carrying out these instructions you are authorized to suspend the *habeas corpus.* General Stone has received similar instructions to these. You will please confer with him as to the particular points that each shall take the control of.

I am, sir, very respectfully, your obedient servant,

R. B. MARCY, Chief of Staff.

Major General N. P. BANKS,

Commanding Division, Muddy Branch, Md.

Nov. 1-Gen. John A. Dix issued this order : HEADQUARTERS, BALTIMORE, November 1, 1861.

To the United States Marshal of Maryland, and the Provost Marshal of the city of Baltimore.

Information has come to my knowledge that certain indi-viduals who formerly resided in this State, and who are known to have been recently in Virginia bearing arms against the authority and the forces of the United States, have returned to their former homes with the intention of taking part in the election of the 6th of November instant, thus carrying out at the poils the treason they have com-mitted in the field. There is reason also to believe that other individuals lately residents in Maryland who have been individuals lately residents in Maryland, who have been engaged in similar acts of hostility to the United States, or in actively aiding and abetting those in arms against the United States, are about to participate in the election for United States, are about to participate in the election for the same treacherous purpose, with the hope of carrying over the State by disloyal votes to the cause of rebellion and treason. I, therefore, by virtue of the authority vested in me to arrest all persons in rebellion against the United States, require you to take into custody all such persons in any of the election districts or precincts in which they may appear at the polls to effect their criminal attempt to convert the elective franchise into an engine for the subver-sion of the Government, and for the encouragement and support of its enemies

In furtherance of this object, I request the judges of elec-tion of the several precincts of the State, in case any such person shall present himself and offer his vote, to commit him until he can be taken into custody by the authority of the United States; and I call on all good and loyal citizens to support the judges of election, the United States marshal, and his dantities and the process marshel of Boltimore and and his deputies, and the provest marshal of Baltimore and police, in their efforts to secure a free and fair expression of the voice of the people of Maryland, and at the same time to prevent the ballot-box from heing polluted by treasonable votes.

JOHN A. DIX, Major General Commanding.

Same day, he addressed this letter to the in-spectors of the election at New Windsor, Carroll county:

HEADQUARTERS DEPARTMENT OF PENNSYLVANIA

BALTIMORE, MD., November 1, 1861. GENTLEMEN: I have received your letter of the 29th ultimo, asking me to is see a proclamation authorizing you to administer to all persons of doubtful loyalty, who offer their votes at the approaching election, an oath to support the Constitution of the United States. If I had the power I would most cheerfully do so, for no one who is false to the Government ought to be allowed to vote. But the conthe Government ought to be allowed to vote. But the con-stitution and laws of Maryland provide for the exercise of the elective franchise by regulations with which I have no right to interfere. I have this day issued an order, of which I enclose a copy, to the United States marshal and the pro-vost marshal of Baltimore to arrest any persons who have been in arms in Virginia if they appear at the polls and attempt to vote, as we are told some such persons intend, and to take into custody all who aid and abet them in their treasonable designs; and I have requested the judges of treasonable designs; and I have requested the judges of election, in case any such person presents himself at the polls and attempts to vote, to commit him until he can be taken into custody by the anthority of the United States. I consider it of the utmost importance that the election should be a fair one, and that there should be no obstruc-tion to the free and full expression of the voice of the peo-

tion to the free and full expression of the voice of the peo-ple of the State, believing, as I do, that it will be decidedly in favor of the Union. But it is in the power of the judges of election, under the authority given them, to satisfy them-selves as to the qualifications of the voters, to put to those who offer to poll such searching questions in regard to res-idence and citizenship as to detect traitors, and, without any violation of the constitution or laws of Maryland, to reverse the pollution of the hold to be the their votes. prevent the pollution of the ballot-hoxes by their votes. I am, very respectfully, yours,

JOHN A. DIX

Major General Commanding.

DANIEL ENGEL and WILLIAM ECKER, Inspectors of Election, New Windsor.

MAJOR GENERAL DIX'S ORDER TO PROVOST MAR-SHAL DODGE BEFORE THE MARYLAND ELECTION OF 1861.\*

BALTIMORE, November 5, 1861.

To Provost Marshal DODGE: Use all your power to-morrow to have the polls unob-structed. We have shown that we can control Maryland

\* The above, General Dix says, is in all essential particu-lars accurate, and is given by him from memory. The original of the order has been mislaid.

by force. We now wish to show that we can control it by the power of opinion, and we shall lose the whole moral in-fluence of our victory if the right of suffrage is not free, and maintained.

JOHN A. DIX Major General Commanding.

FURLOUGHING OF SOLDIERS TO VOTE IN MARY-

LAND.

SECRETARY OF WAR TO GEN. M'CLELLAN.

WAR DEPARTMENT WASHINGTON, October 28, 1841.

Major Gen. McCLELLAN, Commanding: SIR: In order to have a full vote in Maryland at the com-Sin: In order to have a lim, vote in mary and as the votes ing election, Wednesday, November 6, so that the legal voters may decide by their ballots all public questions, you are hereby directed to grant three days' furlough to the voters may decide by their balots all public dustions, you are hereby directed to grant three days furlough to the soldiers of the 1st, 2d, and 3d regiments of Maryland volunteers, all to return to duty on Thursday, Nov. 7. Very respectfully,

SIMON CAMERON, Secretary of War.

GEN. M'CLELLAN'S ORDER TO GEN. BANKS.

HEADQUARTERS ARMY OF THE POTOMAC,

WASHINGTON, October 29, 1861.

WASHINGTON, October 29, 1861. Major Gen. N. P. BANKS, Commanding, dc.: GENERAL: Pursuant to directions from the Secretary of War, of the 28th inst., of which I inclose you a copy, the Major General commanding directs that such soldiers of the 1st, 2d, and 3d regiments of Maryland volunteers as may be within the limits of your command receive fur-loughs for such a length of time as will enable them to reach the place wherein they may be entitled to vote by the fth of November. Wherever it may be necessary, in order reach the place wherein they may be entitled to vote by the 6th of November. Wherever it may be necessary, in order to facilitate the presence of these men at their places of voting, to furnish them transportation, it may be furnished. It is desired that the most liberal and prompt circulation may be given to these instructions, in order to secure with certainty the carrying into effect the design proposed. Sufficient time is to be allowed the soldiers thus fur-loughed to enable them to return after voting, without ex-ceeding the term fixed for their furlough; but a prompt return is desired.

return is desired.

Whenever it may be necessary, the absent soldier should be replaced for the time by other troops. The General Commanding desires that the receipt of this

communication be acknowledged at once.

I have the honor to be, General, very respectfully, your obedient servant,

S. WILLIAMS, Asst. Adj. Gen.

#### ELECTION OF 1863.

1863, October 27-General Schenck issued this order :

[General Orders No. 53.]

Ideneral Orders No. 53.] HEADQUARTERS, MIDDLE DEPARTMENT, EIGHTH ARMY CORPS, BALTIMORE, MARTIAND, October 27, 1863. It is known that there are many evil disposed persons, now at large in the State of Maryland, who have been en-gaged in rebellion against the lawful Government, or have given aid and comfort or encouragement to others so engaged, or who do not recognize their allegiance to the United States, and who may avail themselves of the indulg-ence of the authority which tolerates their presence to embarras the approaching election, or, through it, to foist enemies of the United States into power. It is therefore ordered.

1. That all provost marshals and other military officers

1. That all provost marshals and other military officers do arrest all such persons found at, or hanging about, or approaching any poll or place of election on the 4th of No-vember, 1863, and report such arrest to these headquarters. 2. That all provost marshals and other military officers commanding in Maryland shall support the judges of elec-tion on the 4th of November, 1863, in requiring an oath of allegiance to the United States, as the test of citizenship of any one whose vote may be challenged on the ground that he is not loyal, or does not admit his allegiance to the United States, which oath shall be in the following form and terms: and terms :

I do solemnly swear that I will support, protect, and de-fend the Constitution and Government of the United States Each and constitution and Government of the Onliced States Esclint all enemies, whether domestic or foreign; that I hereby pledge my allegiance, faith, and loyalty to the same, any ordinance, resolution, or law of any State convention, or State Legislature, to the contrary notwithstanding; that I will at all times yield a hearty and willing obelience to the said Constitution and Government, and will not, either directly or indirectly, do any act in hostility to the same,

either by taking up arms against them, or aiding, abetting, or countenancing those in arms against them; that, with-out permission from the lawful authority, I will have no communication, direct or indirect, with the States in insurrection against the United States, or with either of them, or with any person or persons within said insurrectionary States; and that I will u all things deport myself as a good and loyal citizen of the United States. This I do in good faith, with full determination, pledge, and purpose to keep this, my sworn obligation, and without any mental reservation or evasion whatsoever.

3. Provost marshals and other military officers are di-3. Provost marshals and other military officers are di-rected to report to these headquarters any indge of an elec-tion who shall refuse his aid in carrying out this order, or who, on challenge of a vote being made on the ground of disloyalty or hostility to the Government, shall refuse to require the oath of allegiance from such voter. By order of Major General Schenck :

W. H. CHESEBROUGH,

Lieutenant Colonel and Assistant Adjutant General

PROVOST MARSHAL GENERAL'S ORDER.

WAR DEPARTMENT

PROVOST MARSHAL GENERAL'S OFFICE,

WASHINGTON, D. C., October 31, 1863. Direct your provost marshals to give their aid in carry-ing out General Schenck's orders for preserving the purity of elections at the polls in Maryland.

JAMES B. FRY, Provost Marshal General.

Major JEFFRIES, Acting Assistant Provost Marshal General, Baltimore, Maryland.

November 2-Governor BRADFORD issued this proclamation :

To the citizens of the State, and more especially the judges of election:

A military order, issued from the headquarters of the "Middle Department," bearing date the 27th ult., printed and circulated, as it is said, throngh the State, though never yet published here, and designed to operate on the approaching election, has just been brought to my attention, and is of such a character and issued under such circumstances as to demand notice at my hands.

This order reciting, "that there are many evil disposed persons now at large in the State of Maryland, who have been engaged in rebellion against the lawful Government, or have given aid and comfort, or encouragement to others so engaged, or who do not recognize their allegiance to the United States, and who may avail themselves of the indul-gence of the authority which tolerates their presence, to embarrass the approaching election, or through it to foist enemies of the United States into power," proceeds, among other things, to direct "all provost marshals and other mil-itary officers, to arrest all such persons found at or hanging about, or approaching any poll or place of election, on the 4th of November, 1863, and report such arrest to these headquarters."

This extraordinary notice has not only been issued without any notice to, or consultation with the constituted authorities of the State, but at a time and under circumstances when the condition of the State, and the character of the candidates are such as to preclude the idea that the result of that election can in any way endanger either the safety of the Government, or the peace of the community.

It is a well known fact that, with perhaps one single ex-ception, there is not a congressional candidate in the State whose loyalty is even of a questionable character, and in not a county of the State outside of the same congressional district is there, I believe, a candidate for the Legislature or any State office, whose loyalty is not equally undoubted. In the face of this well known condition of things, the several classes of persons above enumerated are not only to be arrested at but "approaching any poll or place of election." And who is to judge whether voters thus on their several classes of persons above enumerated are not only to be arrested at but "approaching any poll or place of election." And who is to judge whether voters thus on their way to the place of voting have given "aid, comfort, or en-couragement." to persons engaged in the rebellion, or that they "do not recognize their allegiance to the United States." and may avail themselves of their presence at the polls "to foist enemies of the United States into power?" As I have already said, in a very large majority of the candidates any such "enemies of the United States." but the provest marshale-created for a very different purposethe provost marshals-created for a very different purpose the provost marshals—created for a very different purpose-and the other military officials who are thus ordered to arrest approaching voters are necessarily made by the order the sole and exclusive judges of who fall within the pro-scribed category; an extent of arbitrary discretion, under any circumstances the most odions and more especially offensive and dangerous in view of the known fact that two at least of the five provest marshals of the State are them-sulvage and date for inversated fices and sundary of their selves candidates for important offices, and sundry of their deputies for others.

This military order, therefore, is not only without justification when looking to the character of the candidates before the people, and rendered still more obnoxious by the to the people, and rendered still note oblighted by the means appointed for its execution, but is equally offensive to the sensibilities of the poople themselves and the au-thorities of the State, looking to the repeated proofs they have furnished of an unalterable devotion to the Govornment. For more than two years past there has never been a time when, if every traitor and every treasonable sympathizer in the State had voted, they could have controlled, whoever might have been their candidates, a single depart-ment of the State or jeopardized the success of the General Government. No State in the Union has been or is now actuated by more heartfelt or unwavering loyalty than Maryland—a loyalty intensified and purified by the ordeal Maryland—a loyary intensined and purified by the order through which it has passed; and yet looking to what has lately transpired elsewhere and to the terms and character of this military order, one would think that in Maryland and nowhere else is the Government endangered by the "many evil disposed persons that are now at large." Within less than a month the most important elections

have taken place in two of the largest States in the Union; in each of them candidates were before the poople, charged by the particular friends of the Government, with being hostile to its interests, and whose election was deprecated as fraught with the most dangerous consequences to its suc-One of the most prominent of these candidates was cess. cess. One of the most prominent of these candidates was considered so dangerously inimical to the triumph of the national cause, that he has been for months past banished from the country, and yet hundreds of thousands of voters were allowed to approach the polls, and to attempt "to foist" such men into power, and no provost marshals or other military officers were order to arrest them on the way, or, so far as we have ever heard, even test their allegi-

With these facts before us, it is difficult to believe that the suggestion that the enemies of the United States may be foisted into power at our coming election, was the considbeen that prompted this order; but whatever may have been that motive, I feel it to be my duty to solennly pro-test against such an intervention with the privileges of the ballot-box, and so offensive a discrimination against the rights of a loyal State.

I avail myself of the occasion to call to the particular attention of the judges of election the fact that they are on the day of election clothed with all the authority of conserthe curve of the peace, and may summon to their aid any of the executive officers of the county, and the whole power of the county itself to preserve order at the polls and secure the constitutional rights to voters.

It is also made their "special duty" to give information to the State's attorney for the county of all infractions of the State's attorney for the conney of an infactions of the State laws on the subject of elections, and by these laws it is forbidden to any "commissioned or non-commis-sioned officers, having command of any soldier or soldiers quartered or posted in any district of any county of the State, to muster or embody any of said troops, or march any law in the state of the State, to muster or ennough any or said toops, or march any recruiting party within the view of any place of election during the time of holding suid election." I need not, I am sure, remind them of the terms of the oath they are required to take before entering upon their

out they are required to take before entering upon their duties, and according to which they swear' to permit all per-sons to vote who shall offer to poll at the election, &c., who in their judgment, shall, according to the directions contained in the Constitution and laws, be entitled to poll at the same election, and not permit any person to poll at the same election who is not in their judgment qualified to vote as aforesaid."

It is the judgment of the judges of election alone, founded upon the provisions of the constitution and the laws of the State, that must determine the right to vote of any person offering himself for that purpose. I trust and believe that they will form that judgment, and discharge their duty, as their conscientious convictions of its requirements, under the solemn obligatione they assume shall dictate, undeterred by an order to provost marshals to report them to "headquarters."

Whatever power the State possesses shall be exerted to protect them for anything done in the proper execution of its laws.

Since writing the above, I have seen a copy of the Presi-dent's letter to the chairman of the Union State Central Committee, bearing the same date with the order, and evi-dently showing that the order was unknown to him, that it would not have been approved by him if he had known it, and that it is therefore all the more reprehensible.

A. W. BRADFORD.

By the Governor: WM. B. HILL, Secretary of State.

After the above was in print, at three o'clock this after-noon, I received from the President the following dispatch:

"I revoke the first of the three propositions in General Schenck's General Order No. 53, not that it is wrong in

principle, but because the military, being of necessity ex-clusive judges as to who shall be arrested, the provision is liable to abuse; for the revoked part I substitute the following:

"That all provost marshals and other military officers do prevent all disturbance and violence at or about the polls, whether offered by such persons as above described, or by any other person or persons whomsoever; the other two propositions I allow to stand; my letter at length will reach you to-night. "A. LINCOLN." you to-night.

Whilst this modification revokes the authority of the provost marshals and military officers to arrest the classes of persons enumerated in the preamble to the order, "found at or hanging about, or approaching any poll or place of election," it directs them to prevent all violence or disturbance about the polls, &c.

ance about the poils, &c. To meet such disturbances, the judges of election, as I have already stated, are clothed with ample powers, and I had received no previous intimation that there was any reason to apprehend a disturbance of any kind at the polls on the day of election. In the absence of any military dis-play, there would certainly seem to be as little cause for such apprehensions as ever before existed. A preparation by the decomponent by military present to pravide for such by the Government, by military means, to provide for such a contingency, will be quite as likely to provoke as to sub-due such a disposition. Not only so, but the military thus required to prevent violence or disturbance about the minury time must necessarily be empowered to arrest the parties they may charge with such disorder, and they are still left in effect "exclusive judges as to who shall be arrested"—a power they may as readily abuse as any other.

I regret, therefore, that I can perceive no such change in the general principles of the order as to induce me to change

the aforegoing proclamation. A. W. BRADFORD. BALTINORE, Monday evening, Nov. 2, 1363. Nov. 2.—The PRESIDENT wrote this letter to Governor Bradford :

### WAR DEPARTMENT

WASHINGTON, November 2, 1862. SIR: Yours of the 31st ultimo was received yesterday about noon, and since then I have been giving most earn-est attention to the subject-matter of it. At my call General Schenck has attended, and he assures mo it is al-most certain that violence will be used at some of the most certain that violence will be used at some of the voting places on election day, unless prevented by his pro-vost guards. He says that in some of those places the Union voters will not attend at all, or run a ticket, unless they have assurance of protection. This makes the Missouri case of my action, in regard to which you express your

The remaining point of your letter is a protest against any person offering to vote being put to any test not found in the laws of Maryland. This brings us to a difference between Missouri and Maryland. With the same reason in-both States, Missouri has, by law, provided a test for the voter with reference to the present rebellion, while Mary-land has not. For example, General Trimble, captured fighting us at Gettysburg, is, without recanting his treason, a legal voter by the laws of Maryland. Even General Schenck's order admits him to vote, if he recants upon oath. I think that is cheap enough. My order in Missouri, which you approve, and General Schenck's order here, reach pre-cisely the same end. Each assures the right of voting is all loyal men, and whether that man is loyal, each allows that man to fix by his own oath. Your suggestion that meerly all the candidates are loyal I do not think quite meets the caso. In this struggle for the nation's life, I cannot so confidently rely on those whose election may have depended upon disloyal votes. Such men, when elected, may prove true, but such votes are given them in expectation that they will prove false. Nor do I think that to keep the peace at the polls, and to prevent the persist-ently disloyal from voting, constitutes just cause of offence to Maryland. I think she has her own example for it. If I mistake not, it is precisely what General Dix did when your excellency was elected Governor. I revoke the first of the three propositions in General Schenck's General Order No. 53, not that it is wrong in principle, but because the mili-tary being, of necessity, exclusive judges as to who shall be arrested, the provision is liable to abuse. For the revoked part I substitute the following: That all provest marshals and other military officers do prevent all disturbance and violence at or about the polls, whether offered by such persons as above described, or by any other person or persons whatevere. The remaining point of your letter is a protest against

whether offered by such persons as above described, or by any other person or persons whatsoever.

The other two propositions of the order I allow to stand. General Schenck is fully determined, and has my strict order besides, that all loyal men may vote, and vote for whom they please. Your obedient servant,

A. LINCOLN. President of the United States. His Excellency A. W. BRADFORD, Governor of Maryland.

#### GENERAL SCHENCK'S MODIFICATION OF THE ORDER.

[General Orders No. 55.]

HEADQUARTERS, MIDDLE DEPARTMENT,

BALTIMOR, MIDDLE DEPARTMENT, EIGITH ARMY CORPS, BALTIMORR, MARILAND, November 2, 1863. Paragraph I, of Goneral Orders No. 53, from these head-quarters, is modified so as to read as follows: I. That all provest mersionly as to I. That all provost marshals and other military officers do

prevent all disturbance and violence at or about the polls, whether offered by such persons as above described, or by any other person, or persons, whomsoever. By command of Major General Schenck : WILLIAM H. CHESEBROUGH,

Lieutenant Colonel and Assistant Adjutant General.

November 3-General Schenck issued this address, in reply to the proclamation of Governor Bradford :

HEADQUARTERS MIDDLE DEPARTMENT, STH ARMY CORPS, BALTIMORF, MD., November 3, 1863. A very extraordinary proclamation was issued last even-ing by his Excellency A. W. Bradford, Governor of Maryland, in relation to General Order No. 53, from these headquarters. I will not presume, with my knowledge of Gov-ernor Bradford, that that proclamation was designed to produce collision between the military power and the citizens who may be assembled at the polls to vote at the election to-morrow; but I cannot doubt that its obvious When tendency is to invite and suggest such disturbance. that proclamation came to my knowledge, late last night, I felt it my duty to take measures for restricting, as far as possible, its circulation in those parts of the State to be most affected by it, until there could go out with it the letter of the President of the United States on the subject, written yesterday to Governor Bradford, a copy of which I have now obtained.

I will make, for myself, but one or two comments on the proclamation.

The intimation of the Governor that my order might have been prompted by some other consideration than patriotic purpose or official duty, is unworthy of reply, and unworthy of him. He knows, and the people of Marylaud and of this military department know, how single and earnest and constant has been my aim to avoid all side influences, and to keep in view, and act steadily upon, the idea of maintain-ing the just authority of the national Government against

is logal to all its forms, and for the general good only. It was in this spirit that I issued the general good only. It was in this spirit that I issued the general order in question. Its principal purpose is to prevent traitorous persons from controling, in any degree, by their votes, or taking part in the coming election. The order is not aimed at candidates, either individually or as a class, as the Gov-ernor would presume. Neither is it aimed at, nor can it by any proper interpretation in any way interfere with the rights of loyal voters. It is only framed and intended to exclude from a voice. In the lection of those who are to administer the affairs either of the national Government or of this loyal State such individuals as are hostile to that Government of which Maryland is a part. Will any good Government of which Maryland is a part. Will any good citizen pretend that the exclusion of such persons is not a wise and wholesome protection, due to those who adhere to and sustain the Constitution and lawful authority? And it is clearly not a hardship, to be complained of by the indivi-dual challenged for such disqualification, when he is per-mitted to purge himself by his own oath of allegiance to the Government, in the management of which he claims a share.

Blare. Governor Bradford himself cannot appreciate more highly than I do, the stering loyalty of the great majority of the people of Maryland, but he must know, as I do, that there still remains at large, from forbearance of the Government authorities, a very considerable number who are more or less actively engaged in aiding and encouraging rebels in arms. Even in his proclamation he admits the existence of such prevailing disbricts. But my separal order was only put forth after the re-

But my general order was only put forth after the re-ceipt, through all the last month, of a great number of letceipt, through all the last month, of a great humber of left-ters, petitions, and appeals in person, from respectable and loyal citizens, throughout the southern part of the State particularly, on both sides of the bay, imploring the issue of such an order. I have only failed in complying with this request by making its provisions less stringent than justice and fairness to loyal citizens seemed to them to de-mand mand.

I will add only, to show with what anxiety I have sought, on this occasion, to secure peace and good order at the polls, that officers intrusted with this duty have, in every in-stance, been furnished with written or printed instructions, of which the following is one clause: "The officers and men are to be cautioned not to commit

or permit any unlawful violence. They must not enter into political discussions, and are to remember, that while protecting the polls from rebel sympathizers, they are conservators of the pace, and are there to support the judges of election." Even Governor Bradford could scarcely ob-ject to this. I now repeat to the provost guards that in-struction, and enjoin upon them, that while they enforce the observance of the general order firmly and faithfully, as directed, they do it in every respect discretely and tem-perately perately.

I append copies of the President's letter, and of the general order, as modified.

ROBERT C. SCHENCK. Major General Commanding.

LIEUTENANT COLONEL TEVIS'S ORDER AT CHES-TERTOWN.

HEADQUARTERS THIRD MARYLAND CAVALRY, CHESTERTOWN, November 3, 1863. Whereas the President of the United States, in reply to whereas the residence of the other states, in repry or a letter addressed to him by Hon. Thomas Swann, of Balti-more city, has stated that all loyal qualified votors should have a right to vote, it therefore becomes every truly loyal citizen to avail himself of the present opportunity offered to place himself honorably upon the record or poll-book at the approaching election, by giving a full and ardent sup-port to the whole Government ticket, upon the platform adopted by the Union League convention. None other is recognized by the Federal authorities as loyal or worthy of the support of any one who desires the peace and restoration of this Union.

#### CHARLES CARROLL TEVIS, Lieutenant Colonel Commanding.

This order was at once suppressed by Gen-eral Schenck, and Colonel Tevis ordered under arrest. November 6, he made a statement of his action, and requested to be released from arrest and restored to his command, which was granted November 9, as stated in the subjoined order:

HEADQUARTERS MIDDLE DEPARTMENT, 87H ARMY CORPS, BALTIMORE, MARYLAND, November 9, 1863.

The within explanation and report having been submitted with permission by Colonel Tevis, under arrest and restored to duty-

to duty— The general commanding remarks that Colonel Tevis ap-pears to have acted himself in good faith and from a sense of duty, but to have been misled by the bad course and in-structions of Captain Frazier, provost marshal of the first congressional district of Maryland, who framed the order issued by him, and perhaps also by the mistaken advice of other over-zealous persons. But Colonel Tevis should have understored that the whole scone and nurnose of the General understood that the whole scope and purpose of the General Orders 53 and 55, and of accompanying instructions, was to prevent disloyal persons from voting and to suppress dis-turbances at the polls. The only test of loyalty prescribed was the oath of allegiance embodied in General Order No. 53. No directions were given to interfere with candidates, as such, nor with the voters because of their supporting any particular ticket, nor was any such interference conany particular tacket, for was any such interference con-templated or intended to be sanctioned by the general com-manding. The idea of a "Government ticket" was only with Colonel Tevis or the provost marshal, Frazier, and other indiscreet or had advisers, and was not put forth from these headquarters. Colonel Tevis's printed order was therefore rule law more during a

these headquarters. Colonel leviss printed order was therefore wholly unauthorized and wrong. The general commanding directs that a copy of this in-dorsement be furnished to Colonel Tevis, and also that a copy of the same be sent, duly certified, to Major H. L. Jeffries, assistant provost marshal general for the States of Maryland and Delaware.

By command of Major General Schenck: W. II. CHEESEBROUGH,

Lieut. Col. and Assistant Adjutant General.

ELECTION OF 1864.

Pending the constitutional convention election in Maryland, April 6, 1864, Major General Lewis Wallace and Gov. Bradford had a correspondence as to the oath prescribed by law to to be administered to the delegates chosen, and the power possessed by the judges of election to investigate and decide upon a charge of disloyalty as affecting the right to vote. The Governor recited the law, and the duties of the judges, and the penalties for non-performance, and expressed the opinion that the laws,

if faithfully executed, as he hoped they would | be, would be found "entirely sufficient to exclude disloyal voters from the polls."

#### Delaware.

An act to secure the freedom of elections in this State. Whereas the Constitution of this State declares that "all elections shall be free and equal," and whereas the freedom of elections and the free enjoyment of the right of suffrage according to the Constitution and laws of the State are essential to the enjoyment of public liberty; and whereas, evil disposed persons did cause armed soldiers to be brought into this State and to be present at different voting places in the State on the day of the last general election, and a free election was thereby prevented; and whereas, it is proper that a repetition of so grave an offence against the peace and dignity of the State shall be prevented: Therefore-

SECTION 1. Be it enacted, d.c., That if any person, being a citizen or inhabitant of this State, shall send or cause to be sent, bring or canse to be brought, into this State, or shall aid, abet, procure, advise, counsel, or in any manner assist in sending or bringing into this State any armed soldier or soldiers, to be present at any voting place in this State, or within five miles thereof, on the day of any general, special, or other election hereafter to be holden in this State, or shall aid, abet, procure, advise, counsel, or in any manner sharh and, anet, product, navise, counsel, or in any manner assist the presence or attendance of any armed soldier or soldiers at any such voting place, or within five miles thereof, on any such election day, every person so offending shall be guilty of felony, and upon conviction thereof by indictment, shall forfeit and pay to the State a fine not less than one thousand dollars and not more than ten thousand dollars, and shall be imprisoned for a term not less than one nor more than five years, and shall forever thereafter be incapable of exercising the right of suffrage in this State.

State. Sec. 2. That if any person, being a citizen or inhabitant SEC. 2. That if any person, being a citizen or inhabitant of this State, shall aid, abet, procure, advise, counsel, or in any manner assist or be guilty of military interference in any manner with the freedom of any election in this State, every person so offending shall be guilty of felony, and upon conviction thereof by indictment shall forfielt and pay to the State a fine of not less than one thousand dol-lars nor more than ten thousand dollars, and shall be im-prisoned for a term not less than one nor more than five years, and shall forever thereafter be incupable of exercis-ing the right of suffrare in this State.

ing the right of suffrage in this State. SEC. 3. That if any inspector, presiding officer, or judge of any election in this State, or any other person, shall administer or cause to be administered to any legal voter in minister or cause to be administered to any legal voter in this State, any oath or affirmation not authorized by the Constitution and laws of this State for that purpose, as a pre-requisite or condition of voting at any election in this State, except when such oath or affirmation shall be ad-ministered in order to satisfy such inspector, presiding offiministered in order to satisfy such inspector, presiding offi-eer, or judge, that such vote is a logal vote according to the Constitution and laws of this State, such inspector, presid-ing officer, judge, or other person so offending, shall be guilty of a misdemeanor, and upon conviction thereof by indictment shall, for every such offence forfeit and pay to the State a fine of not less than five hundred dollars and not more than one thousand dollars, and shall be imprisoned for a term of ten days. Passed at Dover, March 25, 1863.

#### GENERAL SCHENCK'S ORDER.

HEADQUARTERS MIDDLE DEPARTMENT, STH ARMY CORPS, BALTIMORE, MD., November 13, 1863.

[General Orders No. 59.]

It is known that there are many evil disposed persons now at large in the State of Delaware, who have been en-gaged in rebellion against the lawful Government, or have given aid or comfort or encouragement to others so engaged, or who do not recognize their allegiance to the United States, and who may avail themselves of the indulgence of the authority which tolerates their presence to attempt to take part in or embarrass the approaching special election in that State. It is therefore ordered:

I. That all provost marshals and other military officers do prevent all disturbance and violence at or about the polls, whether offered by such persons as above described, or by any other person or persons whomsover.

II. That all provost marshals and other military officers commanding in Delaware, shall support the judges of elec-tion on the 19th of November, 1863, in requiring an oath of allegiance to the United States, as a test of citizenship of any one whose vote may be challenged on the ground that he is not loyal or does not admit his allegiance to the Uni-ted States, which oath shall be in the following form and terms:

I do solemnly swear that I will support, protect, and de-

fend the Constitution and Government of the United States and the Constitution in the contract of the constitution of the contract providence, resolution, or law of any State convention or State legislature to the contrary notwithstanding; that I will at all times yield a hearty and willing obodience to the said Constitution and Government, and will not, directby chalters and constitution and sover in hostility to the same, either by taking up arms against them, or aiding, abetting, or countenancing those in arms against them; that, without permission from the lawful authority, I will have no com-nunication, direct or indirect, with the States in insurrecmunication, aread or indirect, with the States in insurrec-tion against the United States, or with either of them, or with any person or persons within said insurrectionary States; and that I will in all things report myself as a good and loyal citizen of the United States. This I do in good faith, with full determinstion, pledge, and purpose to keep this, my sworn obligation, and without any mental reservation or evasion whatsoever.

III. Provost marshals and other military officers are di-111. Provost marshals and other minitary onleters are di-rected to report to these headquarters any judge of election who shall refuse his aid in carrying out this order, or who, on challenge of a vote being made on the ground of die-loyalty or hostility to the Government, shall refuse to re-quire the oath of allegiance from such voter. By command of Major General Schenck:

W. H. CHEESEBROUGH, Lieut. Col. and Asst. Adj. Gen.

GOVERNOR CANNON'S PROCLAMATION ON THE FOREGOING ORDER. STATE OF DELAWARE, EXECUTIVE DEPARTMENT,

Dover, November 13, 1863. All civil officers and good citizens of this State are enjoined to obey the above military order, issued by the com-manding general of the Middle department, and to give all needful aid for the proper enforcement of the same. WILLIAM CANNON.

Governor of Delaware.

### GENERAL TYLER'S ORDER.

HEADQUARTERS DISTRICT OF DELAWARE, MIDDLE DEPARTMENT.

WILMINGTON, DELAWARE, Nov. 15, 1863. The following instructions have been received from the General Commanding, and will be strictly and carefully observed by all detachments of officers and soldiers within the "District of Delaware," while carrying out the provis-ions of Department General Orders No. 59, during the election to be held on the 19th instant:

Every officer or non-commissioned officer in command of a detachment will be held strictly accountable for the good conduct and obedience of the men in his charge.

Officers and soldiers must be strictly sober, and, while preventing disturbance by others, must avoid all distur-bance themselves, and are required not to hang around the polls, or engage in political discussions. It will be borne in mind that the whole object of the

order of the Commanding General is to preserve peace at order of the Commanding General is to preserve peace at the places of voting, to sustain the judges or inspectors of election, to protect loyal voters, and to prevent from voting disloyal and traitorously disposed persons, who refuse to take the oath of allegiance. In enforcing Paragraph I, of the General Order, to pre-vent violence and disturbance at the polls, care will be taken that disloyal citizens of other States do not, as it is suprohended they will attemut to do intrude themesizes of

apprehended they will attempt to do, intrude themselves at the places or voting, and eudeavor to intimidate the lawful and loyal voters of Delaware.

DANIEL TYLER. Brigadier General Commanding.

# Kentucky.

1863, March 11-The Legislature passed this bill by a two-thirds vote over the veto of Governor Magoffin:

An act to amend chapter 15 of the Revised Statutes, entitled "Citizens, expatriation, and aliens."

SEC. 1. Be it enacted, dc., That any citizen of this State who shall enter into the service of the so-called Confederate States, in either a civil or military capacity, or enter into the service of the so-called provisional government of Ken-tucky in either a civil or military capacity, or, having here-tofore entered such service of either the Confederate States totore entered such service of either the Confederate States or provisional government, shall continue in such service after this act takes effect, or shall take up and continue in arms against the military forces of the United States or the State of Kentucky, or shall give voluntary aid and assist-ance to those in arms against said forces, shall be deemed to have expatriated himself, and shall no longer be a cittern of Vectories and the bar citien and the states are the state. of Kentucky, nor shall he again be a citizen, except by per-mission of the Legislature by a general or special statute.

2

SEC. 2. That whenever a person attempts or is called on to exercise any of the constitutional or logal rights and privi-leges belonging only to citizens of Kontucky, he may be re-quired to negative on oath the expatriation provided in the first section of this act, and upon his failure or refusal to do so, shall not be permitted to exercise any such right or priv-

ilege. SEC. 3. This act to be of force in thirty days from and after its passage.

1863, July 10-Governor J. F. Robinson issued this proclamation :

#### COMMONWEALTH OF KENTUCKY,

COMMONWEALTH OF RENTCEX, Executive Department. For the information and guidance of all officers at the approaching election, I have caused to be herewith pub-lished an act of the Legislature of Kentucky entitled "An act to amend chapter 15 of the Revised Statutes, entitled "Citizens, expatriation, and alions." The strict observance and enforcement of this and all other laws of this State reg-ultion clucicar can canned the unique dat zonined are and enforcement of this and all other laws of this State reg-ulating elections are earnestly enjoined and required as being alike due to a faithful discharge of duty, to the purity of the elective franchise, and to the sovereign will of the people of Kentucky expressed through their Legislature. Given under my hand, as Governor of Kentucky, at Frankfort, this 10th day of July, 1863, and in the seventy-second year of the Commonwealth. LE PORTSON

J. F. ROBINSON.

By the Governor:

D. C. WICKLIFFE, Secretary of State.

July 31-General Burnside issued this order:

HEADQUARTERS DEPARTMENT OF THE OHIO. CINCINNATI, OHIO, July 31, 1863.

CINCINNATI, OHIO, July 31, 1863. [General Orders No. 120.] Whereas the State of Kentucky is invaded by a rebel force with the avowed intention of overawing the judges of elections, of intimidating the loyal voters, keeping them from the polls, and forcing the election of disloyal candi-dates at the election on the 3d of August; and whereas the military power of the Government is the only force that can defeat this attempt, the State of Kentucky is hereby declared under martial law, and all military officers are commanded to aid the constituted authorities of the State in support of the laws and of the purity of suffrage as de-fined in the late proclamation of his Excellency Governor Robinson. Robinson.

Robinson. As it is not the intention of the commanding general to interfere with the proper expression of public opinion, all discretion in the conduct of the election will be, "as usual, in the hands of the legally appointed judges at the polls, who will be held strictly responsible that no disloyat person be allowed to vote, and to this end the military power is or-dered to give them its utmost support. The civil authority, civil courts, and business, will not be suspended by this order. It is for the purpose only of pro-tecting, if necessary, the rights of loyal citizens and the freedom of election. By command of Major General Burnside:

By command of Major General Burnside LEWIS RICHMOND, Assistant Adjutant General.

LIEUT. COLONEL THOS. JOHNSON'S ORDER.

#### HEADQUARTERS UNITED STATES FORCES.

MEADQUARTERS UNITED STATES FORCES, SMITHLAND, KY., July 16, 1863. The county court judges of the counties of Trigg, Cald-well, Lyon, Crittenden, and Livingston, are hereby direct-ed, in appointing judges and clerks for conducting the State elections in August next, to observe stircity the laws of Kentucky, which require that such judges and clerks shall be unconditional Union men.

the unconditional Union men. Judges and clerks so appointed are hereby directed not to place the name of any person on the poll-books to be voted for at the said election who is not a Union man, or who may be opposed to *furnishing men and money for a tigorous prosecution of the war* against the rebellion against the Unitod States Government. The judges and clerks are further directed to permit no person to vote at said elec-tion without taking the oath required by the laws eff. Kensel and the said person so presenting himself to vote is personally known to the judges to be a Union man. Any person violating this order will be regarded as an enemy to the Government of the United States, and will be arrested and punished accordingly. By order of THOMAS JOHNSON, *Lieutenant Colonel Commanding*.

The oath prescribed by Lieutenant Colonel Johnson, to be taken by the voters, is in substance similar to the oath attached to the proclamation of General Shackleford.

## MAJOR GENERAL HURLBUT'S ORDER.

HEADQUARTERS SIXTEENTII ARMY CORPS.

MEADQUARERS SIXTEENTIL ARAY CORPS, MEMPHIS, TENN., July, 1863. I. In so much of the State of Kentucky as is within the district of Columbus, it is ordered— 1. That no person be permitted to be a candidate for office who is not avowedly and unconditionally for the Union and the suppression of the rebellion.

2. That no person shall exercise the privilege of an elec-tor and vote at the said elections who is not avowedly and unconditionally for the Union and the suppression of the rebellion.

3. The military authorities in said district of Columbus will see to it that this order be carried out. Judges of elec-tion will be governed by the principles herein set forth, and will demand evidence upon oaths in such cases as may be in doubt, and allow no person to exercise the franchise of In doubt, and allow ho person to exercise the frammse voting who does not take the oath required. By order of Major General S. A. Hurlbut: HENRY DINMORE, Assistant Adjutant General.

#### BRIGADIER GENERAL ASBOTH'S ORDER.

HEADQUARTERS DISTRICT OF COLUMEUS,

SIXTH DIVISION, SIXTEENTH ARMY CORPS, COLUMBUS, KENTUCKY, July 15, 1863.

The above orders of the general commanding corps are communicated to the civil and military authorities for their information. Military officers making arrests for trief of these orders will be governed by the circular from office of Commissary General of Prisoners, dated Washington, May 11, 1863.

By order of Brigadier General Asboth :

T. H. HARRIS, Assistant Adjutant General.

# BRIGADIER GENERAL ASBOTH'S ORDER.

# DISTRICT OF COLUMBUS.

DISTRICT OF COLUMNUS, HEADQUARTERS SIXTH DIVISION, SIXTEENTH ARMY CORPS, COLUMNUS, KY., July 29, 1863. That no further doubt may exist as to the intent and meaning of Special Orders No. 159, dated Headquarters Sixteenth Army Corps, July 14, 1863, it is ordered that no person shall be permitted to be voted for, or be a candidate for office, who has been or is now under arrest or bonds, but person subtrait, for utforing disloval language as as by proper authority, for uttering disloyal language or sentiments

tuments. County judges within this district are hereby ordered to appoint, as judges and clerks of the ensuing August elec-tion, only such persons as are avowedly and unconditionally for the Union and the suppression of the rebellion, and are further ordered to revoke and recall any appointment of judges and clerks already made who are not such loyal persons.

Judges and clerks of elections are hereby ordered not to o uggs and clerks of elections are hereby ordered hot to place the name of any person upon the poll-books, to be voted for at said election, who is not avowedly and uncon-ditionally for the Union and the suppression of the rebellion, or who may be opposed to furnishing men and money for the suppression of the rebellion.

The following oath is prescribed and will be administered by judges of elections to voters and to such candidates as reside within the district:

"I do solemnly swear that I have never entered the ser-vice of the so-called Confederate States; that I have not been engaged in the service of the so-called 'provisional government of Kentucky' either in a civil or military ca-pacity; that I have never, either directly or indirectly, aided the rebellion against the Government of the United aided the robellion against the Government of the United States or the State of Kentacky; that I am unconditionally for the Union and the suppression of the rebellion, and am willing to furnish men and money for the vigorous prose-cution of the war against the rebellious league known as the 'Confederate States;' so help me God.' Any voter, judge, or clerk of elections, or other person, who may evade, neglect, or refuse compliance with the pro-visions of this order will be arrested and sent before a mili-tary commission as soon as the facts are substantiated.

By order of Brigadier General Asboth:

T. H. HARRIS, Assistant Adjutant General.

BRIGADIER GENERAL SHACKLEFORD'S ORDER.

HEADQUARTERS FIRST BRIGADE,

Second Division, Twenty-Third Army Corrs, Russeltville, KY., July 30, 1863. In order that the proclamation of the Governor and the In order that the proclamation of the Governor and the laws of the State of Kentucky may be observed and en-forced, post commandants ard officers of this command will see that the following regulations are strictly complied with at the approaching State election : No one but loyal citizens will act as officers of the election.

No one will be allowed to offer himself as a candidate for

office, or be voted for at said election, who is not in all things loyal to the State and Federal Government, and in favor of a vigorous prosecution of the war for the suppression of the rebellion.

The judges of election will allow no one to vote at said election unless he is known to them to be an undoubtedly loyal citizen, or unless he shall first take the oath required by the laws of the State of Keutucky. No disloyat man will offer himself as a candidate, or at-

tempt to vote, except for treasonable purposes; and all such efforts will be summarily suppressed by the military authorities.

All necessary protection will be supplied and guaranteed at the polls to Uniou men by all the military force within this command.

By order of Brigadier General J. M. Shackleford, commanding: J. E. HUFFMAN,

# Assistant Adjutant General.

Colonel John W. Foster, of the sixty-fifth Indiania regiment, commanding post at Henderson, Kentucky, issued an order similar to the above order of General Shackleford.

#### COLONEL MUNDY'S ORDER.

All loyal citizens, who have not forfeited their citizen-ship, can safely and quietly cast their votes for the candi-dates of their choice; but all who have forfeited their right of citizenship, under the provisions of the act of Assembly, who shall present themselves at the polls and fraudulently attempt to vote, will be immediately arrested by the guard detailed for that purpose at such precinct, and confined in the military prison.

## Missouri.

1862, June 12-The convention of Missouri adopted this ordinance:

SEC. 1. No person shall vote at any election to be here-emnly swear (or affirm, as the case may be) that I will supbort, protect, and defend the Constitution of the United States and the constitution of the State of Missouri against all enemies and opposers whether domestic or forcing; that I will bear true faith, loyalty, and allegiance to the United States, and will not, directly or indirectly, give aid and comfort, or countenance, to the enemies or opposers thereof, or of the provisional government of the State of Missouri, any ordinance, law, or resolution of any State convention or Legislature, or of any order or organization, secret or otherwise, to the contrary notwith standing; and that I do this with a full and honest determination, pledge, and purpose, faithfully to keep and perform the same, without any mental reservation or evision whatever. And I do further solemnly swear (or affirm) that I have not, since the 17th day of December, A. D. 1861, wilfully taken up arms or levied war against the United States, or against the provisional government of the State of Missouri: So help me God."

#### CONGRESSIONAL ELECTION OF 1862.

[General Orders No. 45.]

HEADQUARTERS STATE OF MISSOURI,

ADJUTANT GENERAL'S OFFICE St. Louis, October 23, 1862.

I. A general election is to take place throughout the State the first Tuesday in November next.

This is the first attempt of the people to choose their officers since the war of the rebellion commenced. It will be an occasion when angry passions, excited by this war, might produce strife, and prevent the full expression of the popular will in the selection of officers.

Lar will in the selection of officers. The convention has provided by ordinance that every voter shall, before voting, take a prescribed oath, and that no vote shall be counted in favor of any candidate for a State or county office, unless he shall have taken an oath prescribed for candidates. The ordinance of the convention force heavy negative upon these who she has the convention fixes heavy penalties upon those who take the oath falsely. These are the safeguards which the convention has judged necessary to keep unfaithful and disloyal persons from ex-ercising power in the State. They are sufficient. No per-son must be allowed to interfere with the freedom of those

Government that all qualified voters shall be allowed, without molestation of any kind, to cast their votes as they please.

II. It is required of all officers and men of the enrolled militia that they keep perfect order at the polls on the day of election, and that they see that no person is either kept from the polls by intimidation or in any way interfered with in voting at the polls for whatever candidate he may choose

III. If any officer or private shall either interfere with the rights of voters, or countenance such interference by others, it will be treated as a high military offence, and punished with the utmost rigor.

1V. Whenever there is any reason to apprehend any in-terference with the election on the part of bands of guerrillas, the commanding officer of the nearest regiment will detail a sufficient force to prevent any such interference, and station it where there is apprehended danger.

V. In case of disturbance arising which cannot be arrested by the civil authorities, any commissioned officer present is hereby ordered, at the request of any judge, sheriff, or justice of the peace, to use the necessary military force to suppress

it. VI. Commanding officers of the enrolled Missouri militia are hereby directed to see that the foregoing orders are strictly obeyed. By order of the commander-in-chief:

WILLIAM D. WOOD, Acting Adjutant General Missouri.

#### BRIGADIER GENERAL HALL'S ORDER.

HEADQUARTERS SEVENTH MILITARY DISTRICT, ST. JOSEPH, MISSOURI, November 1, 1862. [General Orders, No. 33.]

The attention of all officers and soldiers of the militia of this district is called to General Order No. 45, dated "Head-quarters State of Missouri, Adjutant General's office, St. Louis, October 23, 1862" with reference to the election on Tuesday next. The military should bear in mind that they are not the judges of the qualifications of voters. That duty is devolved by law on the judge of the election. If those officers either admit improper persons to vote, or exclude proper persons from voting, the statutes of this State pro-vide an ample remedy. The militia will carefully abstain from all acts calculated to interfere with the freedom of election. All officers who interfere with the rights of voters will be reported to the commander-in-chief, to be dealt with as he may decide. All soldiers guilty of the same offence will be punished as a court-martial shall determine.

By order of Brigadier General Willard P. Hall: ELWOOD KIRBY,

Assistant Adjutant General.

# MAJOR GENERAL SCHOFIELD'S ORDERS OF 1863.

HEADQUARTERS DEPARTMENT OF THE MISSOURI. ST. LOUIS, MISSOURI, September 28, 1863.

[General Orders No. 101.] The right of the people to peaceably assemble for all law-ful purposes, and the right to freely express their will at the polls according to law, are essential to evil liberty. No interference with these rights, either by violence, therefore an etherwise with the statements.

threats, intimidation, or otherwise, will be tolerated. Any commissioned officer who shall incite or encourage any interference with any lawful assemblage of the people, or who shall fail to do his utmost to prevent such inter-ference, shall be dismissed the service; and any officer, soldier, or civilian, who shall, by violence, threats, or otherwise, actually interfere with any such lawful assemblage of the people, shall be punished by imprisonment or otherwise, at the discretion of a court-martial or military commission.

Any officer, soldier, or civilian, who shall attempt to in-timidate any qualified voter in the exercise of his right to vote, or who shall attempt to prevent any qualified voter from going to the polls or voting, shall be punished by im-prisonment or otherwise, at the discretion of a court-martial or military commission. Special attention is called to the fifth article of war,

which will be applied to commissioned officers of Missouri militia not in active service, as well as to officers and soldiers in active service.

# ldiets in active service. By command of Major General Schofield: C. W. MARSH, Assistant Adjutant General.

HEADQUARTERS DEPARTMENT OF THE MISSOURI, ST. LOUIS, October 20, 1863.

and the set of the set

of the State convention, adopted June 10, 1862, and pub-lished herewith, be enforced in every case. It is the duty of judges of election at each precinct in the

It is the day of Judge of election at each product in the State to see that every person qualified by the constitution and laws of the State shall be permitted to exercise the elective franchise without let or hindrance; and it is equally their duty to see that those who are not qualified under the constitution and laws, or who refuse to qualify according to the terms of the annexed ordinance, shall not be allowed to vote; and any action on their part excluding qualified

to vote; and any action on their part excluding quanted voters from the polls, or admitting those who are not quali-fied as stated, will be punished as a military offence. Any person who has borne arms against the Government of the United States, or voluntarily given aid and comfort to its enemies during the present rebellion, and who shall to its enemies during the present rebellion, and who shall presume to act as judge or clerk at said election, and any county judge who shall knowingly appoint any such per-son as above described to act as judge at said election, will be deemed guilty of violation of military orders, and upon conviction thereof, will be punished accordingly. In those parts of the State where there is danger of in-terference by guerrilla bands, or by combinations of persons intended to overawe or intimidate legal voters, district commanders will so dispose their troops as will most cer-ticibly present such interference.

tainly prevent such interference.

Where no such protection may be deemed necessary, all troops will absent themselves from the polls entirely; and in all cases when ordered by the distinct commanders to be present, their action will be strictly confined to the suppression of violence and removing the interference above named, to the end that the laws may be enforced and the purity of the ballot-box maintained. Under the pretext of punty of the halfot-box maintained. Under the pretext of guarding against violence at the polls, no officer or soldier will be permitted to interfere with the peaceful and legal expression of public sentiment, and no officer will be ex-cused for a willful failure to remove any interference in-tended to prevent such expression. District commanders will designate, on the day previous to the alcosing those counties in their securities districts

to the election, those counties in their respective districts in which such protection may be deemed necessary, and cause their troops to be disposed accordingly. Any willful violation of this order will be promptly pun-

ished as a military offence.

Missouri troops will vote at the company polls, opened for that purpose at the headquarters of their posts, camps, or detachments, in accordance with the ordinance of the State convention, passed June 12, 1862, and at no other place.

Special attention is directed to General Orders No. 101 from these headquarters, dated September 28, 1863, and its observance strictly enjoined upon all in Missouri. By command of Major General Schofield:

O. D. GREENE, Assistant Adjutant General.

# **Orders Respecting Impressments of** Property.

MAJOR GENERAL HARTSUFF'S ORDER.

HEADQUARTERS TWENTY-THIRD ARMY CORPS.

LEXINGTON, KENTUCKY, July 24, 1863. For the information and guidance of officers in impressing property, it is hereby directed that, whenever its impress-ment may become necessary for the troops of the twenty-third army corps, it will be taken exclusively from rebels and rebel sympathizers; and so long as the property needed is to be found belonging or pertaining to either of the above-named classes, no man of undoubted loyalty will be molested.

Among rebel sympathizers will be classed those persons in Kentucky, nominally Union men, but opposed to the Government and to the prosecution of the war, whose acts and words alike hinder the speedy and proper termination of the rebellion.

Property will only be taken by the proper staff officers, who will in every case give receipts for it. Appropriate blank receipts will be furnished by the chief commissary and chief quartermaster at these headquarters. By command of Major General Hartsuff:

GEORGE B. DRAKE, A. A. G.

BRIGADIER GENERAL BOYLE'S ORDER.

HEADQUARTERS DISTRICT OF KENTUCKY, LOUISVILLE, July 25, 1863.

By authority of the general commanding the department, the following general order is made: 1. It is ordered that no forage or other property belong-ing to loyal eitizens in the State of Kentucky be seized or impressed except in cases of absolute necessity, and then only on the written authority from the headquarters of the transit they are form the productments of the

wenty-third army corps or from these headquarters.
Whenever it becomes necessary to seize or impress private property for military purposes, the property of sym-

pathizers with the rebellion and of those opposed to furnishing any more men or any more money to maintain the Federal Government and suppress the rebellion will be first seized and impressed.

3. The negroes of loyal citizens will not be impressed on the public works and military roads unless absolutely necessary. The negroes of citizens who are for no more men and no more money to suppress the rebellion, and the supporters, aiders, and abettors of such, will be first im-pressed, and officers detailed for this purpose are required

strictly to observe this order in the execution of their duties. 4. All horses of the enemy captured or subject to capture will be taken possession of by quartermasters and reported to Captain Jenkins, chief quartermaster, Louisville, who is ordered to allow loyal citizens to retain horses to supply the places of those stolen by the orient horses to supply sons mentioned in paragraphs two and three, who encourage raids by the enemy, will not in any case be allowed to re-tain captured horses or horses justly subject to capture.

5. For all property seized or impressed proper and regular vouchers will be given, with indorsement as to the loyalty or disloyalty of the owners of the property. By order of Brigadier General Boyle: A. C. SEMPLE, A. A. G.

# Action of Congress.

First Session, Thirty-Eighth Congress. TO PREVENT OFFICERS OF THE ARMY AND NAVY

FROM INTERFERING IN ELECTIONS.

#### IN SENATE.

1864, January 5-Mr. Powell brought in a bill to prevent officers of the army and navy, and other persons engaged in the military service, from interfering in elections in the States; which was ordered to be printed.

January 6-Mr. Powell moved it be referred to the Committee on the Judiciary; which was lost-yeas 16, (Messrs. Buckalew, Carlile, Cowan, Davis, Harding, Hendricks, Hicks, Johnson, Lane of Indiana, Nesmith, Powell, Saulsbury, Sherman, Sprague, Van Winkle, Willey,) nays 21. It was then referred to the Military Committee.

January 16-The Senate adopted this resolution :

Resolved, That the Secretary of War be directed to trans-mit to the Senate all orders or proclamations, in his department, concerning elections issued hy military authority in the States of Kentucky, Missouri, Maryland, and Delaware.

February 12-Mr. HOWARD, of Michigan, made an adverse report upon the bill.

June 22-The bill was amended so as to read as follows:

Be it enacted, &c., That it shall not be lawful for any mil-itary officer of the United States, or any person engaged in the civil, military, or naval service of the United States, to order, bring, keep, or have under his authority or control, any troops or armed men within one mile of the place where any general or special election is held in any State of the United States of America, unless it shall be necessary to repel the armed enemies of the United States or to keep the reper the armed enemies of the Onice States of the Point States of the Point And that its shall not be lawful for any officer of the army or navy of the United States to prescribe or fix, or attempt to prescribe or fix, by proclamation, order, or otherwise, the qualifications of voters in any State of the United States of America, or in any manner to interfere with the freedom of any election in any State, or with the exercise of the free right of suffrage in any State of the United States. Any officer of the army or navy of the United States, or other person engaged in the civil, military, or naval ser-vice of the Umited States, who violates this section of this act, shall, for every such offence, be liable to indictment as for a misdemeanor, in any court of the United States having for a misdemeanor, in any court of the United States having jurisdiction to hear, try, and determine cases of misde-meanor, and on conviction thereof shall pay a fine not ex-ceeding \$5,000, and suffer imprisonment in the penitentiary not less than three months nor more than five years, at the discretion of the court trying the same; and any person convicted as aforesaid shall, moreover, he disqualified from holding any office of honor, profit, or trust under the Gov-ernment of the United States: *Provided*. That nothing herein contained shall be so construed as to prevent any officer solider earlier or marine from exercising the right officer, soldier, sailor, or marine from exercising the right

of suffrage in any election district to which he may belong, if otherwise qualified, according to the laws of the State in which he shall offer to vote.

Which he shall offer to vote. SEC. 2. That any officer or person in the military or naval service of the United States who shall order or advise, or who shall directly or indirectly, by force, threat, menace, intimidation, or otherwise, prevent or attempt to prevent any qualified voter of any State of the United States of America from freely exercising the right of suffrage at any unreach or aversion electric in care of the of the United general or special election in any State of the United States, or who shall in like manner compel, or attempt to compel, any officer of an election in any such State to compet, any other of an election in any such state to receive a vote from a person not legally qualified to vote, or who shall impose or attempt to impose any rules or regulations for conducting such election different from those prescribed by law, or interfere in any manner with any officer of said election in the discharge of his duties, shall for every such offense be liable to indictment as for a mis-demeanor, in any court of the United Slates having juris-diction to hear try and deterpine cases of misdemeanor diction to hear, try, and determine cases of misdemeanor, and on conviction thereof shall pay a fine not exceeding \$5,000, and suffer imprisonment in the penitentiary not ex-ceeding five years, at the discretion of the court trying the same, and any person convicted as aforesaid shall, moreover, be disqualified from holding any office of honor, profit, or trust under the Government of the United States.

### And passed—yeas 19, nays 13, as follows:

YEAS-Messrs. Buckalew, Carlile, Davis, Grimes, Hale, Harlan, Hendricks, Hicks, Johnson, Lane of Kansas, Mc-Dougail, Pomeroy, Fowell, Richardson, Riddle, Saulsbury, Trumbull, Wade, Willey-19. NATS-Messrs. Anthony, Chandler, Clark, Collamer, Dixon, Foot, Foster, Harris, Howard, Morgan, Sumner, Ten Eyck, Wilson-13.

Mr. HARLAN entered a motion to reconsider. June 28-The Senate refused to reconsider it--yeas 19, nays 23, as follows:

YEAS-Messrs. Anthony, Chandler, Clark, Collamer, Con-ness, Doolittle, Foot, Foster, Howard, Lane of Indiana, Lane of Kansas, Morgan, Morrill, Ramsey, Sprague, Sumner, Ten Eyck, Wilkinson, Wilson-19. NATS-Messrs. Brown, Buckalew, Carkile, Cowan, Davis, Grimes, Hale, Harlan, Harris, Henderson, Hendricks, Hicks McDougall, Nesmith, Pomeroy, Powell, Riddle, Saulsbury, Sherman, Trumbull, Van Winkle, Wade, Willey-23.

The bill was not acted upon in the House.

# RECONSTRUCTION OF STATES.

# Bill for Reconstruction. First Session, Thirty-Eighth Congress. IN HOUSE.

1863, December 15-Mr. HENRY WINTER DAvis moved that so much of the President's message as relates to the duty of the United States to guarantee a republican form of government to the States in which the governments recognized by the United States have been abrogated or overthrown, be referred to a select committee of nine to report the bills necessary and proper for carrying into execution the foregoing guarantee; which was agreed to-yeas 89, nays 80.

May 4-The House passed the bill reported from the committee-yeas 74, nays 66, as follows:

lows:
 YEAS-Messrs. Alley, Allison, Ames, Anderson, Arnold, Ashley, John D. Baldwin, Baxter, Beaman, Blow, Boutwell, Brandegee, Broomall, Cole, Creswell, Henry Winter Davis, Dawes, Deming, Dixon, Donnelly, Driggs, Eliot, Farnsworth, Penton, Frank, Garfield, Higby, Hooper, Hotchkiss, A. W. Hubbard, J. H. Hubbard, Hulburd, Julian, Kelley, Francis W. Kellogg, Orlando Kellogg, Littlejohn, Loan, Longver, Marvin, McBride, McClurg, McIndoe, Sanuel F. Miller, Moorhead, Morrill, Daniel Morris, Amos Myers, Leonard Myers, Norton, Charles O'Neill, Orth, Patterson, Perham, Pike, Fomeroy, Price, Alexander H. Rice, John H. Rice, Edward H. Rollins, Schenck, Scofield, Shannon, Sloan, Smithers, Spalding, Thayer, Upson, William B. Washburn, Williams, Wilder, Wilson, Windom, Woodbridge–74.
 MAYS-Messrs. William J. Allen, Ancona, Augustus C. Baldwin, J. B. Blair, Brooks, J.S. Brouon, W. G. Brown, Charler, Clay, Cox, Cravens, Dauson, Denison, Eden, Eldridge, English, Finck, Ganson, Grider, Hale, Hail, Harding, Benjamin G. Harris, Charles M. Harris, Merrick, Holman, Philip Johnson, William Johnson, Kernan, King, Knapp, Law, Lozar, LeBlond, Long, Marcy, McAllister, McDouell, McKinney, Midleton, Morris, Morrison, Noble, Odell, John O'Neill, Schne, Steele, Stilliam H. Randall, Robinson, James S. Rollins, Ross, Sott, Smith, Worbster, Whaley, Wheeler, Ohilton A. White, Teaman–66.
 The preamble to the bill, which was in these

The preamble to the bill, which was in these words-

Whereas the so-called Confederate States are a public enemy, waging an unjust war, whose injustice is so glaring that they have no right to claim the mitigation of the ex-treme rights of war which are accorded by modern usage to an enemy who has a right to consider the war a just one; and whereas none of the States which, by a regularly re-corded majority of its citizens, have joined the so-called southern Confederance and persidered and treated as an Sonthern Confederacy, can be considered and treated as en-titled to be represented in Congress, or to take any part in the political government of the Union: Therefore—

was rejected-yeas 57, nays 75, as follows:

YEAS-Messrs. Alley, Allison, Ames, Anderson, Ashley, John D. Baldwin, Baxter, Beaman, Boatwell, Boyd, Broom-all, Cole, Henry Winter Davis, Donnelly, Driggs, Eckley, Eliot, Frank, Garfield, Grinnell, Higby, Hooper, Hotchkiss, Asahel W. Hubbard, John H. Hubbard, Julian, Kasson,

Kelley, Francis W. Kellogg, Littlejohn, Loan, Longyear, McBride, McClurg, Moorhead, Daniel Morris, Amos Myers, Leonard Mycrs, Norton, Charles O'Neill, Orth, Patterson, Perham, Pike, Price, John H. Rice, Edward H. Rollins, Schenck, Shannon, Sioan, Spalding, Stevens, Upson, Wil-iam B. Washburn, Villiams, Wilder, Woodbridge-57. NAYS-Messrs. Wm. J. Allen, Ancona, Arnold, Augustus C. Bidwin, Jacob B. Blair, Blow, Brooks, James S. Broun, William G. Brown, Chanler, Clay, Cox. Cresvell, Dauson, Denison, Eden, Eldridge, Farnsworth, Fenton, Finck, Gan-son, Grider, Hale, Halt, Harding, Berjamin G. Harris, Chas. M. Harris, Herrick, Holman, Hulburd, Philip John-son, Orlando Kellogg, Kernan, Knopp, Law, Lazeer, Le Blond, Long, Murcy, Marvin, McAllister, McIndoe, Midole-ton, Morrill, Morrison, Noble, Oddl, John O'Neill, Fendledon, Perry, Pomeroy, Radford, Samuel J. Randall, William H. 1007, Morrison, Novie, Okar, John O Neuli, Fendelow, Perry, Pomeroy, Radford, Szmuel J. Randall, William H. Randall, Alexander H. Rice, James S. Rollins, Ross, Sco-field, Scott, Smith, Smithers, John B. Steele, Wm. G. Steele, Stilles, Strouse, Stuart, Thayer, Ward, Webster, Whaley, Wheeler, Chillon A. While, Joseph W. White, Wilson, Win-dom, Varguere 755 dom, Yeaman-75

The bill authorizes the President to appoint in each of the States declared in rebellion, a Provisional Governor, with the pay and emoluments of a brigadier; to be charged with the. civil administration until a State government therein shall be recognized. As soon as the military resistance to the United States shall have been suppressed, and the people sufficiently returned to their obedience to the Constitution and laws, the Governor shall direct the marshal of the United States to enroll all the white male citizens of the United States, resident in the State in their respective counties, and whenever a majority of them take the oath of allegiance, the loyal people of the State shall be entitled to elect delegates to a convention to act upon the re-establishment of a State government—the proclamation to contain details prescribed. Qualified voters in the army may vote in their camps. No person who has held or exercised any civil, military, State, or Confederate office, under the rebel occupation, and who has voluntarily borne arms against the United States, shall vote or be eligible as a delegate. The convention is required to insert in the constitution provisions-

1st. No person who has held or exercised any civil or military office, (except offices merely ministerial and mili-tary offices below a colonel.) State or Confederate, under the usurping power, shall vote for, or Le a member of the

legislature or governor. 2d. Involuntary servitude is forever prohibited, and the

freedom of all persons is guaranteed in said State. 3d. No debt, State or Confederate, created by or under the sanction of the nsurping power, shall be recognized or paid by the State.

Upon the adoption of the constitution by the convention, and its ratification by the electors

317

of the State, the Provisional Governor shall so certify to the President, who, after obtaining the assent of Congress, shall, by proclamation, recognize the government as established, and none other, as the constitutional government of the State; and from the date of such recognition, and not before, senators and representatives and electors for President and Vice President may be elected in such State. Until reorganization the Provisional Governor shall enforce the laws of the Union and of the State before the rebellion

The remaining sections are as follows:

SEC. 12. That all persons held to involuntary servitude or labor in the States aforesaid are hereby emancipated and discharged therefrom, and they and their postcrifty shall be forever free. And if any such persons or their posterity shall be restrained of liberty, under pretence of any claim t such service or labor, the courts of the United States

to such service or labor, the courts of the United States shall, on *habcas corpus*, discharge them. SEC. 13. That if any person declared free by this act, or any law of the United States, or any proclamation of the President, be restrained of liberty, with intent to be held in or reduced to involuntary servitude or labor, the person convicted before a court of competent jurisdiction of such act shall be punished by fine of not less than \$1,500, and be imprisoned not less than five, nor more than twenty years.

SEC. 14. That every person who shall hereafter hold or exercise any office, civil, or military, except offices merely ministerial and military offices below the grade of colonel, in the rebel service, State or confederate, is hereby declared not to be a citizen of the United States.

#### IN SENATE.

May 27-Mr. WADE, from the Committee on Territories, reported the bill with two amendments; one fixing the compensation of the provisional Governor at \$3,000 a year, and the other striking out the word "white" wherever it occurs.

The Senate as in Committee of the Whole,

July 1-Mr. BROWN offered this as a substitute for the bill:

That when the inhabitants of any State have been deand when the financiality of any state have been de-elared in a state of insurrection against the United States by proclamation of the President, by force and virtue of the act entitled "An act to provide for the collection of duties on imports, and for other purposes," approved July 13, 1561, they shall be, and are hereby declared to be, in-capable of casting any vote for electors of President or Vice President of the United States, or of electing Senators or Representatives in Concress until said insurrection in or Representatives in Congress, until said insurrection in said State is suppressed or abandoned and said inhabitants have returned to their obedience to the Government of the United States, nor until such return to obedience shall be declared by proclamation of the President issued by virtue of an act of Congress, hcreafter to be passed authorizing the same.

Which was agreed to-yeas 17, nays 16, as follows:

YEAS-Messrs. Brown, Carlile, Cowan, Davis, Doolittle, Grimes, Henderson, Hendricks, Johnson, Lane of Indiana, McDougall, I'oucell, Richardson, Riddle, Saulsbury, Trum-bull, Van Winkle-17. NAYS-Messrs. Chandler. Clark, Conness, Hale, Harlan, Lane of Kansas, Morgan, Morrill, Pomeroy, Ramsey, Sher-man, Sprague, Sumner, Wade, Wilkinson, Wilson-16.

Mr. SUMMER proposed the following new section :

TO MAKE THE EMANCIPATION PROCLAMATION A STATUTE.

That the proclamation of emancipation issued by the President of the United States on the first day of January, 1863, so far as the same declares that the slaves in certain designated States and parts of States thenceforward should be free, is hereby adopted and enacted as a statute of the United States, and as a rule and article for the government of the military and naval forces thereof.

Which was rejected-yeas 11, nays 21, as follows :

gan, Morrill, Pomeroy, Ramsey, Sumner, Wade, Wilkinson, Wilson-11. NAYS-Messrs. Brown, Carlile, Davis, Doolittle, Grimes, Hale, Harris, Henderson, Hendricks, Johnson, Lane of In-diana, McDougall, Powell, Richardson, Riddle, Saulsbury, Sherman, Sprague, Ten Eyck, Trumbull, Van Winkle-21.

The Senate adopted the amendment made in Committee of the Whole-yeas 20, nays 13, as follows :

YEAS-Messrs. Brown, Carlile, Davis, Doolittle, Grimes, Harlan, Harris, Henderson, Hendricks, Johnson, Lane of Indiana, McDougall, Pomeroy, Powell, Richardson, Riddle, Saulsbury, Sprague, Trumbull, Van Winkle-20. NAYS-Messrs. Clark, Connese, Hale, Lane of Kansas, Morgan, Morrill, Ramsey, Sherman, Sumner, Ten Eyck, Wade, Wilkinson, Wilson-13.

And the bill passed-yeas 26, nays 3, as follows:

YEAS—Messrs. Brown, Chandler, Conness, Doolittle, Grimes, Harlan, Harris, Henderson, Johnson, Lane of In-diana, Lane of Kansas, *McDougall*, Morgan, Pomeroy, Ram-sey, *Riddle*, Sherman, Sprague, Sunner, Ten Eyck, Trum-bull, Van Winkle, Wade, Wilkinson, Wilson—26. NAYS—Messrs. Davis, Powell, Saulsbury—3.

July 2-The House non concurred in the amendment of the Senate, and asked a Committee of Conference; but the Senate, on motion, receded from its amendments-(thus passing the House bill)-yeas 18, nays 14, as follows:

YEAS-Messrs. Anthony, Chandler, Clark, Conness, Foot, Harlan, Ilarris, Howe, Lane of Kansas, Morgan, Pomeroy, Ramsey, Sherman, Sprague, Sumner, Wade, Wilkinson, Wilson-18.

NATS-MCSSTS. Buckalew, Carlile, Davis, Doolittle, Hen-derson, Hendricks, Lane of Indiana, McDougall, Powell, Riddle, Saulsbury, Ten Eyck, Trumbull, Van Winkle-14.

THE PRESIDENT'S PROCLAMATION RESPECTING IT.

The President failed to approve this bill; and, July 8, 1864, issued this proclamation respecting it:

Whereas, at the late session, Congress passed a bill "to guarantee to certain States, whose governments have been usurped or overthrown, a republican form of government," a copy of which is hereunto annexed ;

And whereas the said bill was presented to the President of the United States for his approval less than one hour before the sine die adjournment of said session, and was not signed by him;

And whereas the said bill contains, among other things, a plan for restoring the States in rebellion to their proper practical relation in the Union, which plan expresses the sense of Congress upon that subject, and which plan it is now thought fit to lay before the people for their consideration :

Now, therefore, I, ABRAHAM LINCOLN, President of the United States, do proclaim, declare, and make known, that, while I am (as I was in December last, when by proclamation I propounded a plan for restoration) unprepared, by a formal approval of this bill, to be inflexibly committed to any single plan of restoration; and, while I am also unprepared to declare that the free State constitutions and governments already adopted and installed in Arkansas and Louisiana shall be set aside and held for nought, thereby repelling and discouraging the loyal citizens who have set up the same as to further effort, or to declare a constitutional competency in Congress to abolish slavery in YEAS-Messrs. Chandler, Conness, Lane of Kansas, Mor- | States, but am at the same time sincerely hop-

ing and expecting that a constitutional amendment abolishing slavery throughout the nation may be adopted, nevertheless I am fully satisfied with the system for restoration contained in the bill as one very proper plan for the loyal people of any State choosing to adopt it, and that I am, and at all times shall be, prepared to give the Executive aid and assistance to any such people, so soon as the military resistance to the United States shall have been suppressed in any such State, and the people thereof shall have sufficiently returned to their obedience to the Constitution and laws of the United States, in which cases Military Governors will be appointed, with directions to proceed according to the bill.

ELECTORAL VOTE OF REBEL STATES.

June 13-Mr. GARFIELD introduced a joint resolution resolving that no State declared to be in rebellion by proclamation of the President is entitled to appoint electors of President and Vice President, and that no electoral vote from any such tate shall be received or counted until both Houses of Congress, by concurrent action, shall have recognized a State government in such State.

It was read a first and second time; was ordered to be engrossed, and read the third time, under the operation of the previous question, when the following proceedings took place:

Mr. McKINNEY. Is it in order to inquire whether Mr. Johnson could be a candidate for the Vice Presidency under that rule?

The SPEAKER. It is not in order to make the inquiry. Mr. BLAINE. Is it too late now to raise a point of order? It is entirely too late. The joint resolution has received its third reading, and the main question has been ordered on its passage.

Mr. BLAINE. Has the morning hour expired? The SPEAKER. It has; but the House has ordered the main question to be now put. Mr. BLAINE. I move to lay the joint resolution on the

table.

Which was agreed to-yeas 104, nays 33, as follows:

follows: YEAS-MCSSTS. James C. Allen, William J. Allen, Allison, Ames, Anderson, Baily, Augustus C. Baldwin, John D. Bald-win, Blaine, Blair, Biss, Boutwell, Brooks, James S. Erown, Charler, Ambrose W. Clark, Freeman Clarke, Cobb, Chi-roth, Coz, Cravens, Dawes, Denison, Dixon, Driggs, Eden, Edgerton, Eldridge, Ellot, English, Farnsworth, Fenton, Frank, Ganson, Gooch, Grider, Grisvold, Harding, Harrington, Charles M. Harris, Herrick, Holman, Notch-kiss, Asahel W. Hubbard, Hutchins, Ingersoll, Jenekes, William Johnson, Kallpleisch, Francis W. K-llogg, Orlando Rellogg, Kernan, King, Kneyp, Law, Le Blond, Littlejohn, Marcy, Marvin, McDowell, McIndoe, McKinney, Samuel F. Miller, Willer, Mochead, James R. Morris, Annos Myers, Leonard Myers, Odell, Charles O'Neill, John O'Neill, orth, P.endleton, Perham, Pike, Price, Pruyn, Radford, Alex-ander H. Rice, John H. Rice, Robinson, James S. Koltins, Ross, Scofield, Sloan, Smith, John B. Steele, William G. Steele, Sites, Stronse, Stuart, Sweat, Thayer, Thomas, Wadsworth, Wuhte, Wilson, Windom, Errando Wood-Lot. Navé-Messrs, Alley, Ashley, Baxter, Beaman, Blow, Favis, Donnelly, Eckley, Gartield, Higby, Hooper, John H. Hubbard, Julian, Kelley, Lazear, Longyear, McClurg, Mor-Hubbard, Julian, Kelley, Lazear, Longyear, McClurg, Mor-Hubbard, Julian, Kelley, Chazear, Longyear, McLurg, Mor-Hubbard, Julian, Kalley, Ashlen, Baxter, Wiliams, Wood-Hubbard, Julian, Kelley, Ashlenburgh, Williams, Wood-Hubbard, Julian, Kelley, Chazear, Longyear, McClurg, Mor-Hubbard, Julian, Kelley, Ashlenburgh, Williams, Wood-Hubbard, Julian, Kelley, Ashlenburgh, Williams, Kood-Hubbard, Julian, Kelley, Ashlenburgh, Williams, Woo

June 20-Mr. AshLEY asked leave to offer this joint resolution, but it was objected to :

Resolved, dc., That when the inhabitants of any State have been declared in a state of insurrection against the United States by proclamation of the President, by force and virtue of the act entitled "An act further to provide for the collection of duties on imports, and for other pur-poses," approved July 13, 1861, they shall be and are hereby

declared to be incapable of casting any vote for elec-tors of President or Vice President of the United States, or of clecting Senators or Representatives in Congress, nntil said insurrection in said State is suppressed or abandoned, and said inhabitants have returned to their obschence to the Constitution and Government of the United States, nor until such return to obedience shall be declared by procla-mation of the President, issued by virtue of an act of Congress authorizing the same.

#### PROPOSED APPOINTMENT OF COMMISSION OF INQUIRY.

June 22-Mr. DAWES, from the Committee on Elections, in the application of certain persons to be received as Representatives from Arkansas, made a report, which closes with this joint resolution :

Resolved, &c., That there be appointed by the President, by and with the consent of the Senate, a commission conby all with the consent of the Printer, a commission con-sisting of three persons, residents of States not involved in the present rebellion, whose duty it shall be during the re-cess of the present Congress to visit those States declared by the proclamation of the President to have been in rebellion, and which have already taken or may before the next session of the present Congress take measures to es tablish or reorganize State governments, and after careful examination and hearing testimony report to the President for the information of Congress at as early day in the next session as possible all such evidence as they may be able to obtain upon the question, whether the loyal people in any such States have succeeded in re-establishing a State gov-ernment, to what extent sad: State government represents and has the support of the loyal people in such State, and what is the ability of such people therein to maintain the same against domestic violence.

Resolved further, That until Congress shall be satisfied upon evidence submitted to them that the rebellion has so far been suppressed in any such State that there has been established therein a State government, republican in form, and prolibiting the existence of slavery in the same, and so firmly established as to be able tomaintain itself against domestic violence, representation from any such Sta ought not to be admitted into either branch of Congress.

Mr. JAMES S. BROWN, from the Minority Committee, presented these resolutions :

Whereas by article six of the Constitution of the United States it and the laws made in pursuance thereof are de-clared to be the supreme law of the land, and every act of secession by any State is in direct violation of such supreme laws: Therefore, Resolved, That the acts of secession by the Legislatures

Resoured, That the acts of secession by the negasiatures of the several States whose people are now in rebellion are mere nullities, having no force or effect to change the rela-tion either of States themselves or of the people thereof toward the General Government; and that by such acts the people neither freed themselves from the penalties attach-near by the to trace now now host to a difference of the ing by law to treason nor lost any rights as citizens of the States and United States, except such as may follow upon conviction of crime; that the duty of the people of such States to send true and loyal men to Congress, and the right so to do as consequent upon the duty, still remain by force of the Constitution, requiring no act of the President or Congress to confirm them; that no State can under the Constitution assent to the presence of armed rebels from other States within its borders, and that any act of the authorities of a State giving such assent is a nullity; that the entrance of such armed rebels of one State upon Territories of another is an invasion from which by article four of the Constitution the United States are bound to protect the Consumption of the control of protection of protection of the part of the United States is due to each citizen individually as a consequence of his duty of allegiance, and continues so long as there is a single loyal citizen in a State oppressed by such invasion; that so long as the Constitution and laws of the United States cannot be enforced in any con-gressional district on account of the presence of armed rebels there can be no free election, and a person claiming a seat through an election under such circumstances should

be rejected. Be it further resolved, That the Constitution, in article two, determines the qualifications of electors for Representatives, and that any order of the President or act of Congress changing such qualifications would be a usurpation and a

Be it further resolved, That whenever by pestilence, foreign invasion, or domestic conspiracy, the officers of a State required by its laws to conduct an election have been state required by the State does not thereby cease to destroyed or carried off, the State does not thereby cease to exist, nor do its people forfeit their rights as citizens of the States or of the United States, but, from the very necessity

of the case, and by virtue of the power impliedly reserved to the people, they may, in a practicable and reasonable manner, supply the deficiency, and hold an election, con-ducting it, however, as far as possible, in conformity with the existing laws and Constitution of the State; and the duty of Congress in passing upon such an election claimed to be held under such circumstances is limited to ascertaining whether it was a fair expression of a majority of the people, and in the mode of conducting it departed from the general laws of the State only so far as was necessary to supply the deficiency of officers required to conduct the election.

Be it further resolved, That the right of the claimants from Arkansas should be determined by the principles here enunciated; and if they shall satisfy the House that the Constitution and laws of the United States and of the State Constitution and laws of the United States and of the States held peaceful sway over their respective districts, that in their elections they departed in nothing from the Constitu-tion and existing laws of that State, save in supplying requisite officers, and that they received a vote of a ma-jority in their respective districts, then they are entitled to cost but not otherwise seats, but not otherwise.

A motion to table the whole subject was lost -yeas 43, nays 63.

A motion to postpone until next session, was lost-yeas 50, nays 78.

June 29-The report was further debated, when, on motion of Mr. DAVIS, the whole subject was leid on the table-yeas 80, nays 47, as follows:

as follows: YEAS-Messrs. William J. Allen, Allison, Ancona, Augus-tus C. Baldwin, Beaman, Blair, Bliss, Blow, Boyd, Brooks, Broomall, James S. Brown, Cole, Cox, Henry Winter Davis, Thomas T. Davis, Dawson, Deming, Denison, Dirogs, Eden, Edgerton, English, Finck, Griswold, Hale, Herrick, Higby, Holman, Asahel W. Hubbard, John H. Hubbard, Hulburd, Jenckes, William Johnson, Kalbfleisch, Kelley, Francis W. Kellogg, Kernan, Knapp, Law, Lazzar, Loan, Longyear, McAllister, McDowell, McIndoe, McKinney, Sam-uel F. Miller, Moorhead, James R. Morrison, Amos Myers, Leonard Myers, Norton, Charles O'Neill, Orth, Pen-dleton, Perham, Perry, Radford, Samuel J. Randall, Robinson, Kos, Schenck, Shannon, Sloan, Smith, Smithers, Wil-liam G. Sleele, Stevens, Stiles, Strouse, Stuart, Thayer, Web-ster, Chilton A. White, Joseph W. White, Williams, Windom -So. -86

NAYS-Messrs. Alley, Ames, Anderson, Ashley, Baily, John D. Baldwin, Blaine, Boutwell, William G. Brown, Chanler, Cobb, Dawes, Donnelly, Eckley, Eliot, Farnsworth, Ganson, Gooch, Harding, Benjamin G. Harris, Cluarles M. Ganson, Gooch, Harang, Berjaman G. Harris, Guartes M. Harris, Hooper, Hutchins, Julian, Knox, Littlejohn, Long, Mallory, Marcy, McClurg, Morrill, Odell, John O'Neill, Pike, Pomeroy, John H. Rice, Edward H. Rollins, James S. Roh-lits, Scofield, Thomas, Upson, Van Valkenburgh, Ellihn B. Washburne, William B. Washburn, Whaley, Winfield—47.

FREE STATE GOVERNMENT IN ARKANSAS.

June 10-Mr. LANE of Kansas, introduced this joint resolution:

For the recognition of the free State government of the State of Arkansas. Whereas the President of the United States, by procla-mation of the 1st of January, A. D. 1863, did, among other things, proclaim and declare that the "people" of Arkansas "are this day in rebellion" against the United States; and Program hore and the State of Arkansas "are this day in rebellion" against the United States; and whereas the loyal people of the State of Arkansas have, since that time, by a free and untrammelled vote, organized and have in operation a State government upon a free busis and republican in form; and whereas, pending the organ-ization of said government, the President of the United States did human business in the State of Decomposition. States did, by proclamation of the 8th day of December, A. D. 1863, hvite, among others, the people of Arkansas to organize a loyal State government upon a free basis; and whereas the President of the United States approved said organization in the State of Arkansas and officially recog-

biganization in the brack of minimum and children is the problem in the brack of minimum and the same is therefore, Be it resolved, dc., That so much of the proclamations of the President of the United States, and so much of all laws of Congress, as declares the people or State of Arkansas in rebellion, be, and is hereby, declared inop-

State of Arkansab we had its because and is hereby, declared hop-state of Arkansab be, and it is hereby, recognized, upon the condition that slavery and involuntary servitude shall never exist in said State, except as a punishment for crime.

additional resolution :

and after the acceptance of its provisions by the people c\* the said State and proclamation of the same by the Presi-dent of the United States.

He also offered this resolution, which was adopted :

That the President of the United States be requested to furnish to the Senate copies of all correspondence, orders, and documents on file in the Departments in relation to the organization by the loyal people of Arkansas of the free State government of that State.

June 13-The Senate considered the joint resolution ; refused--yeas 5, (Messrs. Chandler, Howard, *Richardson*, Sumner, Wade,) nays 32, to lay the subject on the table, and then referred it to the Committee on the Judiciary, and with it the resolution of Mr. SUMNER, offered May 27, as follows:

That a State pretending to secede from the Union and battling against the national Government to maintain this pretension must be regarded as a rebel State, subject to military occupation, and without title to representation on this floor until it has been readmitted by a vote of both Houses of Congress; and the Sonate will decline to enter-tain any application from any such rebel State until after such vote of beth Houses of Congress.

## ADMISSION OF SENATORS FROM ARKANSAS.

## IN SENATE.

### First Session, Thirty-Eighth Congress.

1864, June 27--Mr. TRUMBULL made this report:

The Committee on the Judiciary, to whom were referred the credentials of William M. Fishback and Elisha Baxter, claiming seats from the State of Arkansas, report :

That the credentials are presented in due form, purport-ing to be under the seal of the State of Arkansas, and to be signed by Isaac Murphy, Governor thereof; and if the right to seats were to be determined by an inspection of the credentials, Messrs. Fishback and Baxter would be en-titled to be sworn as members of this body. It is, howtuted to be sworn as memoers of this body. It is, how-ever, admitted by the persons claiming seats, and known to the country, that in the spring of 1861, the State of Ar-kansas, through its constituted authorities, undertook to secede from the Union, set up a government in hostility to the United States, and maintain the same by force of arms, Congress, in view of the condition of affairs in Arkansas and competers future similar situated accorded accord some other States similarly situated passed an aci, July 13, 1861, authorizing the President, in case of an in-Survection in any State against the laws of the United States, and when the insurgents claimed to act under au-thority of the State, and such claim was not repudiated, nor the insurrection suppressed by the persons exercising the functions of government in such State, to declare the inhabitants of such State or part thereof where such insurrection existed, to be in a state of insurrection against the United States; and that, thereupon, all commercial inter-course by and between the same and the citizens of the United States, except under license and upon certain conditions, should cease and be unlawful so long as such condition of hostility should continue.

In pursuance of this act, the President, August 16, 1861, issued his proclamation declaring the inhabitants of the State of Arkansas, except the inhabitants of such parts thereof as should maintain a loyal adhesion to the Union and the Constitution, or might be from time to time occupied and controlled by forces of the United States engaged in the dispersion of said insurgents, to be in a state sengence surrection against the United States, and that all commer-cial intercourse between them and citizens of other States was and would be unlawful, except when carried on under special license, until such insurrection should ccase. At the date of this proclamation no part of the State of Arkan-sas was occupied and controlled by the forces of the United Be it resolved, dc., That so much of the proclamation or The analysis of the President of the United States, and so anch of all laws of Congress, as declares the people or State f Arkansas in rebellion, be, and is hereby, declared inop-rative and void. SEC. 2. That the present organized government in the tate of Arkansas had it is hereby, recognized, upon the issuing of said proc-lamation, a state of hostility or civil war existed between the inhabitants of the State of Arkansas and the United States, and these was not at that time ony organized author-the inhabitants of the State of Arkansas and the United States, and there was not at that time ony organized author-the inhabitants of the State of Arkansas ever exist in said State, except as a punishment for crime. June 11--Mr. LANE of Kansas, offered this dditional resolution : SEC. 3. That this joint resolution shall be in force from SEC. 3. That this joint resolution shall be in force from State government, and have the right, through the Legislature they have instituted, to choose two Senators for said State.

The Constitution declares that "the Senate of the United States shall be composed of two Senators from each State, chosen by the Legislature thereof, for six years," and makes cations of its own members." In the investigation of the claimants' right to scats, the first question to be determined is, was the body by whom they were elected clothed with without to alext scatters in the scatter of the scat authority to elect Senators; in other words, was it, in a constitutional sense, "the Legislature of Arkansas?"

A question similar to this aross some years since between Robbins and Potter, each claiming to have been elected Senator by the Legislature of Rhode Island, though by dif-ferent bodies. In that case the Senate was called upon to ferent bodies. In that case the Senate was called upon to decide, and did decide, which of the two bodies, each claim-ing to be legitimate, was the Legislature contemplated by the Constitution. The Supreme Court of the United States, in the case of Luther vs. Borden, growing out of the politi-cal difficulties in Rhode Island in 1841 and 1842, held that "when the Senators and Representatives of a State are ad-mitted into the councils of the Union, the authority of the covernment under which they are annointed as well as its government under which they are appointed, as well as its republican character, is recognized by the proper constitudepartment of the Government." The claimants laid before the committee a statement of

the circumstances attending the assembling of the body by which they were elected, in which, after detailing the con-dition of the State while under rebel control, and prior to september, 1863, they say: "Upon the advent of the Union army the rebels in the State, guerrillas and all, for the most art left with their armies, leaving about two-thirds of the

part left with their armies, reaving about violations of a State comparatively free from guerrilla depredation. "The Union men came flocking from the mountains, where they had lain for two years, to the Federal standard, and nearly every man whom the medical examiners would receive joined the Federal army.

"Those who were rejected, (and their number was enormous, their constitutions having been broken by exposure and their hardships,) and those whom circumstances prevented from joining the army, found themselves, so far as law was concerned, in a state of chaos. Many of them, living remote from military posts, had not even the protection of military law.

"Immediately they began to agitate the question of a re-organization of their State government. They first moved in primary meetings, and on the 80th of October, 1863, they held a mass meeting in the city of Fort Smith, in which some twenty counties are said to have been represented, and at which they realled upon all counties in the State to and at which they called upon all counties in the State to elect delegates (after having elected commissioners of elec-tion) to a State convention, to be held in the city of Little Rock on the 8th day of January, 1864, for the purpose of so amending the constitution as to abolish slavery. Simul-taneously with this meeting, meetings were held in a number of other counties. In every single one (in ignorance of the action of others in many instances) they declared for a convention and the abolition of slavery.

"Commissioners of election were first elected, and they held the election for the delegates.

"All this was prior to the President's amnesty proclamation.

"When the convention met, forty-five delegates were present, representing about one-half of the State. (Several of the delegates failed to attend.) They repudiated the rebel debt, State and Confederate, abolished slavery, and sub-mitted the constitution to the people for their ratification. They also provided for taking the vote for State and county officers and members of the lowing the vote the serie time officers, and members of the legislature at the same time with the vote for the ratification of the constitution.

"The result of those elections was 12,177 for the constithis net result of those elections was 12,111 of the construction and 226 against it, an election of state and county officers, an election of delegates to the lower house of Congress, and a representation in the State legislature from forty-six of the fifty-four counties of the State."

forty-six of the fifty-four counties of the State." The number of persons in Arkansas who voted for Presi-dent in 1860 was 54,053, less than one-fourth of whom as appears from the statement of the claimants, took part in the reorganization of the State government. This, however, would not be fatal to the reorganization, if all who were loyal to the Union had an opportunity to participate, and the State was free from military control. Such, however, is understood not to have been the case. The President had not then, nor has he un to this time, recalled his proc-But distribution in the interview of the case. The restorms had not then, nor has he up to this time, recalled his proc-lamation, which declared the inhabitants of Arkansas in a state of insurrection, against the United States, nor was there any evidence before the committee that said insurrection had ceased or been suppressed. At the time when the body which chose the claimants was clected, when it as-sembled, and at this time, the State of Arkansas is occupied by basile armien which compare are the view of the state of th by hostile armies, which exercise supreme authority within

the districts subject to their control. While a portion of Arkansas is at this very time, as the committee are in-formed, in the actual possession and subject to the con-trol of the enemies of the United States, other parts of the State are only held in subordination to the laws of the Union by the strong arm of military power. While this state of things continues, and the right to exercise armed authority over a large part of the State is claimed and exerted by the military power, it cannot be said that a civil government, set up and continued only by the sufferance of the military, is that republican form of government which the Constitution requires the United States to guarantee to every State in the Union.

When the rebellion in Arkansas shall have been so far suppressed that the loyal inhabitants thereof shall be free to re-establish their State government upon a republican foundation, or to recognize the one already set up, and by the aid and not in subordination to the military to maintain the same, they will then, and not before, in the epin-ion of your committee, be entitled to a representation in Congress, and to participate in the administration of Federal Government. Believing that such a state of things did not at the time the claimants were elected, and does not now, exist in the State of Arkansas, the committee recommend for adoption the following resolution : *Resolved*, That William M. Fishback and Elisha Baxter

are not entitled to seats as Senators from the State of Arkansas.

1864, June 29-The resolution of the Committee on the Judiciary was adopted-yeas 27, nays 6, as follows:

YEAS-Messrs. Anthony, Brown, Buckalew, Carlile, Chandler, Clark, Cowan, Davis, Fessenden, Foot, Foster, Hale, Harlan, Harris, McDougall, Morgan, Morrill, Powell, Ramsey, Riddle, Saulsbury, Sherman, Sumner, Ten Eyck, Trumbull, Wade, Wikinson-27. NAYS-Messrs. Doolittle, Hicks, Howe, Lane of Kansas, Nesmith, Pomeroy-6.

#### Process of Reconstruction.

In ARKANSAS, a new State government is organized with Isaac Murphy, Governor, who was reported to have received nearly 16,000 votes at a called election. The other State officers are:

Lieutenant Governor, C. C. Bliss; Secretary of State, R. J. T. White; Auditor, J. B. Berry; Treasurer, E. D. Ayers; Attorney General, C. T. Jordan; Judges of the Supreme Court, T. D. W. Yowley, C. A. Harper, E. Baker.

The legislature also elected Senators, but neither Senators nor Representatives obtained their seats.

In LOUISIANA, a free State government has also been organized-Michael Hahn, Governor. The clause in the constitution abolishing slavery was adopted in convention-yeas 72, nays 15. The legislature is prohibited from passing any law recognizing the right of property in slaves. At the election for Governor, 10,725 votes were polled; one parish, (Terrebonne,) 630 votes, not in the official count. J. Madison Wells is Lieutenant Governor; Stanislaus Wrotnowski, Secre tary of Sate; J. G. Belden, Treasurer; B. Lynch, Attorney General; A. P. Dostie, .:

tor; John McNair, School Superintendent.

In VIRGINIA, the constitutional convention, recently in session in Alexandria, abolished slavery-yeas 13, nays 1, as follows:

YEAS-Messrs. Beach, Boush, Downey, Dix, Edwards, Go-ver, Henshaw, Hawxhurst, Penn, Thomas, Tennis, Wood, Watkins-13. NAYS-Mr. Moore.

## IN LOUISIANA.

1863, June 19-The President wrote this letter:

### EXECUTIVE MANSION,

WASHINGTON, June 19, 1863. GENTLEMEN: Since receiving your letter, reliable information has reached me that a re-

21

spectable portion of the Louisiana people desire to amend their State constitution, and contemplate holding a convention for that object. This fact alone, it seems to me, is a sufficient reason why the General Government should not give the committee the authority you seek, to act under the existing State constitution. I may add, that while I do not perceive how such a committee could facilitate our military operations in Louisiana, I really apprehend it might be so used as to embarrass them.

As to an election to be held in November, there is abundant time without any order or proclamation from me just now. The people of Louisiana shall not lack an opportunity for a fair election for both Federal and State officers by want of anything within my power to give them.

Your obedient servant, A. LINCOLN.

# IN ARKANSAS.

## EXECUTIVE MANSION, WASHINGTON, January 20, 1864.

Major General STEELE: Sundry citizens of the State of Arkansas petition me that an election may be held in that State, at which to elect a Governor; that it be assumed at that election, and thenceforward, that the constitution and laws of the State, as before the rebellion, are in full force, except that the constitution is so modified as to declare that there shall be neither slavery nor involuntary servitude, except in the punishment of crimes, whereof the party shall have been duly convicted; that the General Assembly may make such provisions for the freed people as shall recognize and declare their permanent freedom, and provide for their education, and which may yet be construed as a temporary arrangement, suitable to their present condition as a laboring, landless, and homeless class; that said election shall be held on the 28th March, 1864, at all the usual places of the State, for all such voters as may attend for that purpose; that the voters attending at each place at eight o'clock in the morning of said day may choose judges and clerks of election for that purpose; that all persons qualified by said constitution and laws, and taking the oath presented in the President's proclamation of December 8, 1863, either before or at the election, and none others, may be voters; that each set of judges and clerks may make returns directly to you on or before the -—— day of – ---- next; that in all other respects said election may be conducted according to said modified constitution and laws; that, on receipt of said returns, when 5,406 votes shall have been cast, you can receive said votes and ascertain all who shall thereby appear to have been elected; that, on the ----- day of -— next, all persons so appearing to have been elected, who shall appear before you at Little Rock and take the oath, to be by you severally administered, to support the Constitution of the United States and the modified constitution of the State of Arkansas, shall be declared by you qualified and empowered to immediately enter upon the duties of the offices to which they shall have been respectively elected.

You will please order an election to take place on the 28th of March, 1864, and returns to be made in fifteen days thereafter,

# A. LINCOLN.

#### WASHINGTON, February 18, 1864. To WILLIAM M. FISHBACK:

When I formed a plan for an election in Arkansas, I did it in ignorance that your convention was at the same work. Since I learned the latter fact, I have been constantly trying to yield my plan to them. I have sent two letters to General Steele, and three or four dispatches to you and others, saying that he (General Steele) must be master, but that it will probably be best for him to keep the convention on its own plan. Some single mind must be master, else there will be no agreement on anything, and General Steele, commanding the military and being on the ground, is the best man to be that master. Even now citizens are telegraphing me to postpone the election to a later day than either affixed by the convention or me. This discord must be silenced.

# A. LINCOLN.

RELATIONS OF THE REBELLIOUS STATES TO THE GOVERNMENT.

# IN SENATE.

# Second Session, Thirty-Seventh Congress.

1862, Feb. 11-Mr. SUMNER offered the following resolutions:

Resolutions declaratory of the relations between the United States and the territory once occupied by certain States, and now usurped by pretended governments, without constitutional or legal right.

Whereas certain States, rightfully belonging to the Union of the United States, have through their respective governments wickedly undertaken to abjure all those duics by which their connection with the Union was maiotained; to renounce all allegiance to the Constitution; to levy war upon the national Government; and, for the consummation of this treason, have unconstitutionally and unlawfully confederated together, with the declared purpose of putting au end by force to the supremacy of the Constitution within their respective limits; and whereas this condition of insurrection, organized by pretended governments, openly exists in South Carolina, Georgia, Florida, Alabama, Mississippi, Louisiana, Texas, Arkausas, Tennessee, and Virginia, except in Eastern Tennessee and Western Virginia, and has been declared by the President of the United States, in a proelamation duly made in conformity with an act of Congress, to exist throughout this territory, with the exceptions already named; and whereas the extensive territory thus usurped by these pretended governments and organized into a hostile confederation, belongs to the United States, as an inseparable part thereof, under the sanctions of the Constitution, to be held in trust for the inhabitants in the present and future generations, and is so completely interlinked with the Union that it is forever dependent thereupon ; and whereas the Constitution, which is the supreme law of the land, cannot be displaced in its rightful operation within this territory, but must ever continue the supreme law of the land, exanot be displaced in its rightful operation within this territory, but must ever continue the supreme law of the land, supremacy : Therefore—

In order to put an end to its supremacy : Interelore— 1. Resolved, That any vote of secession or other act by which any State may undertake to put an end to the supremacy of the Constitution within its territory is inoperative and void against the Constitution, and when sustained by force it becomes a practical *abdication* by the State of all rights under the Constitution, while the treason which it involves still further works an instant forfeiture of all those functions and powers essential to the continued existence of the State as a body politic, so that from that time forward the territory falls under the exclusive jurisdiction of Congress as other territory, and the State boing, according to the language of the law, felo-de-se, ceases to exist.

exist. 2. That any combination of men assuming to act in the place of such State, attempting to insnare or coorce the inhabitants thereof into a confederation hostile to the Union is rebellions, treasonable, and destitute of all moral authority; and that such combination is a usurpation incapable of any constitutional existence and utterly lawless, so that everything dependent upon it is without constitutional or legal sapport.

support. 3. That the termination of a State under the Constitution necessarily causes the termination of those peculiar local institutions which, having no origin in the Constitution or in those natural rights which exist independent of the Constitution, are upheld by the sole and exclusive authority of the State.

4. That slavery, being a peculiar local institution, derived from local laws, without any origin in the Constitution or in natural rights, is upheld by the sole and exclusive authority of the State, and must therefore cease to exist legally or constitutionally when the State on which it depends no longer exists; for the incident cannot survive the principal.

5. That in the exercise of its exclusive jurisdiction over the territory once occupied by the States, it is the duty of Congress to see that the supremacy of the Constitution is maintained in its essential principles, so that everywhere in this extensive territory slavery shall cease to exist practically, as it has already ceased to exist constitutionally or legally.

6. That any recognition of slavery in such territory, or any surronder of slaves under the pretended laws of the oxtinct States by any officer of the United States, civil or military, is a recognition of the pretended governments, to the exclusion of the jurisdiction of Congress under the Constitution, and is in the nature of aid and comfort to the rebellion that has been organized.

7. That any such recognition of slavery or surrender of pretended slaves, besides being a recognition of the pretended governments, giving them aid and comfort, is a denial of the rights of persons who, by the extinction of the States have become free, so that under the Constitution, they cannot again be enslaved.

8. That allegiance from the inhabitant and protection from the Government are corresponding obligations, dependent upon each other, so that while the allegiance of every inhabitant of this territory, without distinction of color or class, is due to the United States, and cannot in any way be defeated by the action of any pretended Government, or by any pretence of property or claim to service, the corresponding obligation of protection is at the same time due by the United States to every such inhabitant, without distinction of color or class; and it follows that inhabitants held as slaves, whose paramount allegiance is due to the United States, may justly look to the national Government for protection.

9. That the duty directly cast upon Congress by the extinction of the States is reinforced by the positive prohibition of the Constitution that "no State shall enterinto any confederation," or "without the consent of Congress keep troops or ships-of-war in time of peace, or enter into any agreement or compact with another State," or "grant letres of marque and reprisal," or "coin money," or "emit bills of credit," or "without the consent of Congress lay any duties on imports or exports," all of which have been done by these protended governments, and also by the positive injunction of the Constitution, addressed to the nation, that "the United States shall guaranty to every State in this Union a republican form of government;" and that in pursuance of this duty cast upon Congress, will assume complete jurisdiction of such vacated territory where such unconstitutional and illegal thrings have been attempted, and will provide carefully for the protection of all the inhabitants thereof, for the security of families, the organization of labor, the encouragement of industry, and the welfare of society, and will in every way discharge the duties of a just, merciful, and paternal Government.

Feb. 13-Mr. GARRETT DAVIS offered these resolutions:

1. Resolved, That the Constitution of the United States is the fundamental law of the Government, and the powers established and granted, and as parted out and vested by it, the limitations and restrictions which it imposes upon the legislative, executive, and judicial departments, and the States, and the rights, privileges, and liberties which it assures to the people of the United States, and the States respectively, are fixed, permanent, and immutable through all the phases of peace and war, fintil changed by the power and in the mode prescribed by the Constitution itself; and they cannot be abrogated, restricted, enlarged, or differently apportioned, or vested, by any other power, or in any other mode.

2. That between the Government and the citizen the obligation of protection and obedience form mutual rights and

obligations; and to enable every citizen to perform his obligations of obedience and loyalty to the Government it should give him reasonable protection and security in such performance; and when the Government fails in that respect, for it to hold the citizen to be criminal in not performing his duties of loyalty and obedience would be unjust, inhuman, and an outrage upon this age of Christian civilization.

3. That if any powers of the Constitution or Government of the United States, or of the States, or any rights, privileges, immunities, and liberties of the people of the United States, or the States, are, or may hereafter be, suspended by the existence of this war, or by any promulgation of martial law, or by the suspension of the writ of habeas corpus, rights, privileges, immunities, and liberties would be resumed, and would have force and effect as though they had not been suspended.

4. That the duty of Congress to guarantee to every State a republican form of government, to protect each of them against invasion, and, on the application of the legislature or executive thereof, against domestic violence, and to enforce the authority, Constitution, and laws of the United States in all the States, are constitutional obligations which abide all times and circumstances.

5. That no State can, by any vote of secession, or by rebellion against the authority, Constitution, and laws of the United States, or by any other act, abdicate her rights or obligations under that Constitution or those laws, or absolve her people from their obelience to them, or the United States from their obligation to guarantee to such State a republican form of government, and to protect her people by causing the due enforcement within her territories of the authority, Constitution, and laws of the United States. 6. That there cannot be any forfeiture or confiscation of

6. That there cannot be any forfeiture or confiscation of the rights of person or property of any citizen of the United States who is loyal and obedient to the authority, Constitution and the laws thereof, or of any person whatseever, unless for acts which the law has previously declared to be criminal, and for the punishment of which it has provided such forfeiture or confiscation.

criminal, and for the pumisiment of which it has provided such forfeiture or confiscation. 7. That it is the duty of the United States to subdue and punish the existing rebellion by force of arms and civil trials in the shortest practicable time, and with the least cost to the people, but so decisively and thoroughly as to impress upon the present and future generations as a great truth that rebellion, except for grierous oppression of government, will bring upon the rebels incomparably more of evil than obedience to the Constitution and the laws.

8. That the United States Government should march their armise into all the insurgent States, and promptly put down the military power which they have arrayed against it, and give protection and security to the loyal men thereof, to enable them to reconstruct their legitimate State governments, and bring them and the people back to the Union and to obedience and duty under the Constitution and the laws of the United States, bearing the sword in one hand and the olive branch in the other, and whilst inflicting on the guilty leaders condign and exemplary punishment, granting amnesty and oblivion to the comparatively innocent masses; and if the people of any State cannot, or will not, reconstruct their State government and return to loyalty and duty. Congress should provide a government for such State as a territory of the United States, securing to the people thereof their appropriate constitutional rights.

1862, June 11-Mr. DIXON offered this resolution:

Resolved, That all acts or ordinances of secession, alleged to have been adopted by any legislature or convention of the people of any State, arc, as is the Federal Union, absolutely null and void; and that while such acts may and do subject the individual actors therein to forfeitures and penalties, they do not, in any degree, affect the relations of the State wherein they purport to have been adopted to the Government of the United States, but are as to such Government acts of rebellion, insurrection, and hostility, on the part of the individuals engaged therein, or giving such acts or ordinances, members of the Federal Union, and as such are subject to all the obligations and duties imposed upon them by the Constitution of the United States; and the loyal citizens of such States are entitled to all the rights and privileges thereby guarantied or conferred.

#### IN HOUSE.

1862, May 26—Mr. WICKLIFFE presented these resolutions:

Resolved, That the Constitution, and the laws of the United States made in pursuance thereof, and all treaties made under the authority of the United States, are the supreme law of the land in time of war or rebellion, as well as in

.

any part thereof, by either of the departments of the Government, does not exist any more in time of war or rebellion than in time of peace.

3. That the powers vested in Congress by the Constitution of the United States are not enlarged by a state of war, but are ample and full to enable the Government to suppress a rebellion, or to prosecute a war, without the exer-cise of powers not granted therein. 4. A disregard of the provisions of the Constitution, un-

der the plea of necessity, by Congress, is dangerous to free government, and ultimately will make it a despotism if controlled by an uuprincipled majority, dangerous to pri-

vate rights and destructive of public liberty. 5. That no State has the right, by any act, ordinance, or law, to absolve itself or any citizens from their obligations and duties imposed by the Constitution of the United States.

6. A rebellion and open resistance to the Constitution and the laws of the United States made in pursuance thereof, cannot abrogate the same; but during such rebellion and resistance the rights and powers of the Constitution do exand the he citizen in every portion of the United States, and the right and power of the Government, though ob-structed, are not destroyed. And when that obstruction shall be overcome, the State governments and the citizens thereof are restored to all the rights and privileges secured by the Constitution.

7. That open and forcible resistance, by arms, by the citizens of any portion of the United States is treason, to be punished as such by the power of the Government under the provisions and limitations of the Constitution.

8. That a citizen not engaged in actual war, but who resides within the territory of a *de facto* Government, which is in rebellion and at war with the United States, is not guilty of treason by obeying the orders of such *de facto* Government, or by contributing to its support.

9. That neither the Executive nor military has the right to seize the property of a non-combatant in any of the second States, and confiscate the same, even under the authority of an act of Congress. "No person shall be de-prived of life, liberty, or property, without due process of law," is the language of the Constitution. Therefore, all helpenet bills configurations active a numerative form kind wholesale bills confiscating estates, or property of any kind, whether lands, slaves, or chattels, are in violation of the laws of Christian warfare, and forbidden by the Constitution of the United States.

#### Third Session, Thirty-Seventh Congress.

1863, January 8-Mr. CLEMENTS offered this resolution, which went over :

Resolved, That the acts of secession claiming to have been passed by a part of the people of a part of the States of this Government were null and void, and of no effect; and that the State Constitutions and laws of the States so claim-ing to have seceded are the same as they were previous to said acts; they have not committed State suicide, for their Constitutions and laws still exist in the loyal people.

## IN SENATE.

#### First Session, Thirty-Eighth Congress.

1864, Jan. 5-Mr. GARRETT DAVIS offered these resolutions:

1. Resolved, That the Government of the United States was established by the people of States which before had been separate, sovereign, and independent; and they formed their common national Government by a written Constitution, and delegated to it so much of their sover-oign political power as they adjudged to be necessary and proper to enable it to manage all their affairs with foreign nations and among the several States; and, both by its leading principle and express provision, they reserved "to the States, respectively, or to the people," "all powers not delegated to the United States, nor prohibited by it to the States."

2. That our system consists of a limited national Gov-ernment for the whole United States, of supreme authority as to all the powers with which the Constitution has in-vested it, and State governments for each State, formed by the people thereof, and holding the entire residuum of political sovereignty within their respective States-each government, within its sphere, being alike supreme; and as the governors, and all other civil and military officers of the States, as other individuals, may commit treason against the United States, "by lavying war against them, or in ad-hering to their enemies, giving them aid and comfort," so the President of the United States, and the civil and milithe President of the United States, and the civil and mill-tary officers thereof, may commit treason against any State whose government is in the performance of its duties under shall have been adjudged by the civil courts in the mode

time of peace, anything in the constitution or laws of any State to the contrary notwithstanding. 2. The right to disregard and violate the Constitution, or resisting with an armed force the execution of its laws, or

adhering to such armed force, giving it aid and comfort. 3. That in all the States, and parts of States, where the laws of the United States and the States can be executed, the military authorities should not be brought into conflict with the civil power, but should be strictly held to be, as they rightfully are, in subordination to it. 4. That all elections to civil offices, Federal or State, should be in strict accordance with the Constitution and laws of

the United States and of the States, respectively, and be understood by officers appointed by the proper authorities for that purpose; and where from the presence or apprehension of force, violence, or other cause, any election can-not be so conducted, it ought not to be held at all; and every election at which any military force may interfere by imposing additional oaths or qualifications for the electors, or regulations for conducting thesaid election, or by changing or modifying the oaths and qualifications of the electors, or regulations to govern it, provided as by law, or to con-strain, control, or direct the officers of such election in con-ducting it, should be held to be void and of no effect.

ducting it, should be held to be void and of no effect. 5. That the experience of the world proves that there can be neither security nor liberty in any country without wise and just laws, firmly sustained and uniformly executed; that this is the life, the spirit, the soul of the nation; and all neglect and departures from law, and particularly from constitutional law, by agents appointed to administer it, although sometimes attended with seeming advantage, are used to reduce alway a concorr on lot or which more and sure to produce always, sooner or later, much greater and more enduring mischief: wherefore, disregard of law by

6. That the powers of the Government of the Will States are derived wholly from, and limited by the Consti-tution, and by it are divided into legislative, executive, and judicial; and each class of those powers is vested in a sep-arate department; that the President is the chief of the ex-ecutive department, and has no legislative or judicial power whatever, and only such executive powers as are enumerated in the second and third sections of the second article atea in the second and third sections of the second article of the Constitution, and such other powers as may be, from time to time, conferred upon him by Congress in virtue of this provision: "Congress shall have power to make all laws which shall be necessary and proper for carrying into execution the foregoing powers, and all other powers vested by this Constitution in the Government of the United States, or any department or officer thereof."

7. That the President cannot be divested of any powers with which he is directly invested by the Constitution, nor controlled or interfered with in their execution, but all powers conferred on him by law of Congress he holds in subordination to that department, which may supervise, modify, and correct his execution of them, or resume them,

biology, and correct insected on them, or result them, by repealing the laws intrusting their execution to him. 8. That the power of the President to recognize the exist-ence of a state of case amounting to "an invasion, or immi-nent danger of an invasion, of the United States," or "in-surrection in any State against the government thereof," or "obstruction to the execution of the laws of the United States by combinations too powerful to be suppressed by the ordinary course of judicial proceedings," and to call forth the military power to meet such conditions, is con-ferred on him by the laws of Congress; and the repeal of these laws would withdraw from the President all that power.

9. That Congress is invested with the power "to lay and 9. That congress is invested with the power to tay and collect taxon, duties, imposts, and excises, to pay the debts and provide for the common defence and general welfare of the United States," to declare war, grant letters of marque and reprisal, and make rules concerning captures on land and water," to raise and support armies," to provide and maintain a navy," "to provide for calling forth the militia maintain a navy," "to provide for calling forth the militia to execute the laws of the Union, suppress insurrections, and repel invasions," "to provide for arming, organizing, and disciplining the militia, and governing such part of them as may be employed in the service of the United States," "to guarantee to every State in the Union a republican form of government, and to protect each of them against invasion," and thus the entire war power and quasi war power, exter-nal and internal, of the Government, is vested by the Con-stitution in Congress, and no part of it whatever in the President President.

10. That whenever there is an insurrection in the United States, Congress is vested with the power to suppress it, and States, congress is vested with the power to suppress it, and with no other power whatever over the insurrection; and when it is suppressed, either by the arms of the United States or by the submission of the insurgonts to the Consti-tution, laws, and authorities thereof, thereupon the power of Congress over the whole subject is exhausted, and the insurgents are immediately remitted to all the rights, privprescribed by the Constitution; and Congress, much less the President, has no power to impose upon them any other terms or conditions.

11. That the whole power and duty of the President, in the existing insurrection, is to grant pardons to those engaged in it, and, as Commander-in-Chief of the Army and Navy, to direct their operations for its suppression; and as such, his powers are strictly military, and are not different or greater than would be those of the senior general in the Service, if the Constitution had designated him to be the Commander-in-Chief: the power to devise a policy or measures for its suppression is legislative, to which the President is incompetent, whether as the first executive officer of the Government, or Commander-in-Chief of the Army and Navy.

Navy. 12. That the law of military necessity is not established, but only tolerated, in the United States. It does not, nor cannot, in peace or war abrogate or suspend the Constitution in whole or in part. It cannot authorize arbitrary arrests and imprisonment, or in any way interfere with the person of the citizen, but only with his property. It does not appertain to the *President*, or to the *Commandor-in-Chief*, unless he be in the actual command of a military force, and then only under particular circumstances. It results from a present and urgont need of an army or military corps, which is so pressing that it cannot await other modes, but must be supplied anywhere within its reach by its own power and action. It is not an *expediency*, but a *vecessity* of a military hody, and creates a law and confors a power, for the occasion only, on its commander, of whatever grado he may be, to supply that necessity by taking property with summary military force, without depriving the owner of his right to be compensated for it by the United States. Each case of military necessity makes its own law, dapted to its own peculiar circumstances, and expirning with that particular necessity. There is not, and cannot be, any uniform, permanent, or even continuing law of military necessity. The ida that a law always accidental, evanescent, and, in truth, so inconsiderable, should have the magic force to onable Abraham Lincoln to bound over the Constitution and all its limitations and restrictions, and clutch the vast powers which he claims under it, is a gimatical suredity.

Clutter the two points and the t

14. That the present Executive Government of the United States has subverted, for the time, in large portions of the loyal States, the freedom of speech, the freedom of the press, and free suffrage, the Constitutions and laws of the United States, the civil courts and trial by jury; it has ordered, ad *liditum*, arbitrary arrests by military officers, not only without warrant, but without any charge or imputation of crime or offence; and has hurried the persons so arrested from home and vicinage to distant prisons, and kept them incarcerated there for an indefinite time; some of whom it discharged without trial, and in utter ignorance of the cause of their arrest and imprisonment; and others it caused to be brought before courts, created by itself, and to be tried and punished without law, in violation of the constitutional guarantee to the citizen of his right to keep and bear arms, and of his rights of property; it has forcibly deprived as well the loyal as the disloyal of both; it has usurped the power to suspend the writ of *habeas corpus*, and to proclaim martial law, and establish military tribunals in States and hes States by the civil courts and authorities; and ordered many citizens, who were not connected with the army or may, to be dragged before its dram-lead courts, and to be tried by them for new and strange offences, declared by itself, and by undefined and indefinable law, being but the arbitrary will of the court; it has ordined at pleasure a military despotism in the loyal States, by means of courts-martial, provost marshals, and military forces, governed neither by law, principles, nor rules, from whose tyranny and oppressions no man can claim immunity; all of which must be repudiated and swept away by the sovereign people.

Its avore ign people. 15. That a free press, free speech, and free elections are the great and peaceful forces by which the mal-administration of our Government, whether in the legislative or executive departments, is prevented, reformed, and reversed, and its authors brought to public condemnation and punishment; and these bulwarks of constitutional Government and popular liberty are formidable to malversators, usurpers, and tyrants only, and they must be held by the people at all hazards.

16. That as the Constitution and laws afford no means to exclude from the office of President a man appointed to it by military power, or who is declared to be chosen to it by reason of the suppression of the freedom of election, or by the exclusion of legal voters from the polls, or by any other means, the people of the United States would be incompetent to defend, and unworthy to have received the rich heritage of freedom bequeathed to them by their fathers, if they permit that great offices so to be filled, or in any other mode than by their own free suffrages.

insurgent States is open to many and insuperable objections. The pardon and amnesty offered by him is upon the condition that those who accept it shall renounce their right to their slave property, and swear to support his unconstitu-tional proclamation, and unconstitutional acts of Congress, which attempted to take it from them. He must have intended to put this condition in a form so obnoxious as to secure its rejection by most of those to whom it was offered. He affects the position that ten of the insurgent States have free aneces the position that ten of the insurgent states have forfeited or dissolved their State governments, and requires that they be reconstructed on a condition prescribed by himself, and this against the true principle which he and the legislative department of the Government had previously recognized: that all the acts of the insurgent States and people, tending to their secession, separation, and inde-pendence, were void, and when the inundation with which their insurrection covered over the authority of the Con-stitution and laws of the United States in them passed away, it would leave the constitutions, laws, property, and institutions of those States in every respect the same that they were previously, excepting only the changes that were they were previously, excepting only the changes that were produced by the mere shock of arms, the principle status ante bellum being applicable. He ignores the constitutions of Tennossee and Arkansas, and others that have not been altered in any particular, but are the same that they were be-fore their revolt; and he requires those States to repudiate their constitutions that governed them many years peace-fully in the Union, and to form new ones. He has no right to take cognizance, in any way, of the governments and constitutions of those States or of any other States; to the extent that such a power is vested in the Government of the United States, it is congressional, not presidential. He has no authority whatever to impose any conditions on the insurgents, and they are subject to none but what are pre-scribed by the Constitution and laws of the United States, to be determined by their courts. What right has the President to proclaim that one-tenth of as many of the voters of those States as voted at the last presidential election may pull down and revolutionize their State governments, and erect new ones for the other nine-tenths, which he will recognize and uphold with the armies and navies of the United States? This project is to continue the war upon slavery by his fur-ther usurpations of power, and to get together and buy up a desperate faction of mendicants and adventurers in the rehel States, give them possession of the polls by interpo-sing the bayonet, as in Maryland, Delaware, and portions of Mineset and Kontacher and the two of the poll of advent of Missouri, and Kentucky, and to keep off loyal pro-slavery voters, and thus to form bastard constitutions to abolitionize those States.

18. That the impending destiny of our country can no longer he blinked. The people of the loyal States are resolved into two great parties, the destructives and the comservatives. The first consists of Abraham Lincoln, his officeholders, contractors, and other followers; the second of all men who are for ejecting Lincoln and bis party from officeand power. The professed objects of the first are, to preserve the Union and to aboitsh slavery in all the States; they have about ceased to make a pretence of supporting the Constitution and the laws. Their real objects are to perpetuate their party power, and to hold possession of the Governmentto continue the aggrandizement of their leaders, great and small, by almost countless offices and employments, by myriads of plundering contracts, and by putting up to sale their largest amount of spoils that were ever offered to market: by any Government on earth. Their object is not to eradicate slavery, but only to abolish its form and the mastery ; to subjugate wholly the rebel States, and utterly to revolutionize their political and social organization; to destropr

people, secessionists and anti-secessionists, loyal and dis-loyal, combatants and non-combatants, old men, women, and children, the decrepit, and the non compos mentis, all and children, the decrepit, and the non composementis, all whom they cannot abolitionize, and to distribute the lands of the subjugated people among their followers, as was done by the Roman conquerors of their own countrymen; to proclaim a mock freedom to the slaves, but by military power to take possession of the freedmen, and work them for their own profit; to do all this, and also to enslave the white man by the unsofted on the focus titution and layer of the their own profit; to do all this, and also to enslave the white man, by tranpling under foot the Constitution and laws of the United States and the States, by the power of a subsidized army, and, lest it should failter, by hundreds of thousands of negro janizaries, organized for that, purpose by the Sec-retary of War and the Adjutant General. The first and paramount object of the conservatives is to preserve their own liberties by saving the Union, the Constitution, and the laws from utter and final overthrow by the destructives, not the preserve to be combard under the reveated of giving a themselves to be enslaved under the pretext of giving a fictitious freedom to the negro, and to restore and perpetuate the Union, and to bring back the people in revolt by renewed and sufficient guarantees of all their constitutional rights. There is no choice left to any man but to be a destructive or a conservative.

January 7-Mr. CARLILE offered these resolutions:

Resolved. That the Government of the United States, in the language of Mr. Webster, is "the result of compact be-tween the States," each State for itself adopting for its government the Constitution of the United States.

2. That the people of each State adopted for their government the Constitution of the United States, in the same way as they adopted their State constitution. 3. That the Constitution of the United States was not

binding, nor had it force or effect in any State until it was binding, for had to force or effect in any state main it was ratified and adopted by the people thereof, as is evidenced by the fact that the people of North Carolina were not bound by it, nor was North Carolina a member of the Union created by it, for some time after the Government had been organized under it, and after it had been ratified by eleven other States.

4. That the Government created by the Constitution of the United States is "neither wholly national nor wholly federal. Were it wholly national, the supreme and ultimate rederat. Were it wholly hadonal, incorporate and distinct authority would reside in the majority of the people of the Union, and this authority would be competent at all times, like that of a majority of every national society, to alter or abolish its established government. Were it wholly federal, units and the characteristic government. We follow he would be Union would be essential to every alteration that would be binding on all." The mode provided by the Constitution for its amendment "is not founded on either of these principles." "In requiring more than a majority, and particu-larly in computing the proportion by *States*, not by *citizens*, it departs from the national and advances towards the federal character. In rendering the concurrence of less than the whole number of States sufficient, it loses again the federal and partakes of the national character." The Government created by the Constitution of the United States, therefore, is neither a national nor a federal government, "but a composition of both." "In its foundation it is fedean, not national; in the sources from which the ordinary powers of the Government are drawn, it is partly federal and partly national; in the operation of these powers it is national, not federal; in the extent of them it is federal, not national.

5. That it is essential to the preservation of the Government created by the Constitution that the military should be subordinate to the civil power, and that such was the in-tention of the founders of the Government is evidenced by the fact that the President was made Commander-in-Chief of the Army and Navy.

6. That Congress alone can declare what are crimes against the United States, and can alone legislate for the punish-The United States, and can alone legislate for the purish-ment thereof. That it is not competent for the President, in any character, or for any military commander, to re-strict the right of suffrage in any State, or to impose con-ditions precedent to the exercise thereof; nor can the will or judgment of the President enlarge his powers beyond what are conferred by the Correlitation

or judgment of the President enlarge his powers beyond what are conferred by the Constitution. 7. That all the powers of the General Government are derived from the Constitution; that the Constitutions of the several States confers power, and the Constitutions of the several States limit power; that the General Government can only exercise such powers as are conferred by the Con-stitution of the United States; that but for the State Con-stitutions each State, within its own jurisdiction, could every ender the construction of the States of the States and the Constitutions each State, within its own jurisdiction, could every ender the construction of the States of t excrcise all governmental power.

8. That Governments are instituted for the protection of minorities, the chief objects being to secure to the citizen his right to personal liberty, and to protect him in the possession and enjoyment of his private property.
9. That there is no such power as "the war power"

known to the Government of the United States, outside of the powers conferred by and enumerated in the Constitution of the United States.

10. That the Government of the United States enforces obcdience to the Constitution and laws, as do the State Governments, by acting directly upon the citizons; that it cannot create a State Government or a State Constitution, or alter one; that it has no right to interfere, directly or indirectly, to alter, abolish, or procure a change in the Constitution of any State, but is bound to protect existing Constitutions in each and every State from any attempt to substitute for them a Government not republican in form, and cannot act against States or political communities as such but is confined in its action to the punishment of persons, there being no right of eminent domain in the Government of the United States, the soil of each State belonging to it and the people thereof. 11. That any attempt on the part of the Government of

the United States, or any department thereof, to destroy the State Governments, and substitute for them Territorial the State Governments, and substitute for them Territorial or any other form of Government, would be an exercise of arbitrary and usurped power, destructive to the liberties of the people, violative of the Constitution, and an overthrow of the Government created by it. 12. That it is the duty of the servants of the people to whom the administration of the Government is intrusted, and

to whom its powers have been confided, to put down rebelto whom its powers have been conduct, to put down rebe-lion against its authority whonever organized, to the end that the law-abiding citizen may be protected in the en-joyment of the blessings the Government was designed to confer; that, if necessary to this end, the whole military and naval power of the Government can and should be used, not against the States as such, nor in a war against populations and homes, but against the citizens and persons so resisting the authority of the Government and defying

its power. 13. That the whole millitary power of the Government should be directed against the armies of the Confederates, and that, until they are broken and dispersed, there can be no peace upon the basis of a re-union of the States.

#### Feb. 8-Mr. SUMNER offered these resolutions:

Resolved, That, in order to determine the duties of the national Government at the present moment, it is of the first importance that we should see and understand the real character of the contest which has been forced upon real character of the contest which has been forced upon the United States, for a failure truly to appreciate this con-test must end disastrously in a failure of those proper ef-forts which are essential to the re-establishment of unity and concord; that, recognizing the contest in its real char-acter, as it must be recorded by history, it will be apparent beyond controversy, that this is not an ordinary rebellion, or an ordinary war, but, that it is absolutely without pre-cedent, differing clearly from every other rebellion and overy other war, inasmuch as it is an audacious attempt, for the first time in history, to found a wicked power on the corner-stone of slavory; and that such an attempt having the first time in history, to found a wicked power on the corner-stone of slavery; and that such an attempt having this single object—whether regarded as rebellion or as war —is so completely penetrated and absorbed, so entirely filled and possessed by slavery, that it can be justly re-garded as nothing else than the huge impersonation of this crime, at once rebel and belligerent, or in other words, as slavery in arms.

slavery in arms.
2. That, recognizing the unquestionable identity of the rebellion and of slavery, so that each is to the other as another self, it becomes plain that the rebellion cannot be crushed without crushing slavery, as slavery cannot be crushed without crushing the rebellion; that every forbearance to the one is a forbearance to the other, and every blow at the one is a blow at the other; that all who tolerate slavery tolerate the rebellion, and all who strike at slavery strike at the rebellion; and that, therefore, it is our superposed duty in which all other, means duty is a slavery duties are consupremest duty, in which all other present duties are con-tained, to take care that the barbarism of slavery, in which alone the rebellion has its origin and life, is so utterly trampled out that it can never spring up again anywhere in the rebel and belligerent region; for leaving this duty undone nothing is done, and all our blood and treasure have been lavished in vain.

3. That in dealing with the rebel war the National Gov-3. That in dealing with the rebel war the National Government is invested with two classes of rights—one the rights of sovereignty, inherent and indefeasible everywhere within the limits of the United States, and the other the rights of war, or beligerent rights, which have been superinduced by the nature and extent of the contest; that, by virtue of the rights of sovereignty, the rebel and belligerent region is now subject to the National Government as its only rightful Government, bound under the Constitution to the during of sovereignty and us special mendate is only right the duties of sovercignty, and by special mandate bound also "to guarantee to every State a republican form of government, and to protect it against invasion;" that, by virtue of the rights of war, this same region is subject to all

.

the conditions and incidents of war, according to the cstablished usages of Christian nations, out of which is derived the familiar maxim of public duty, "Indemnity for the past and security for the future."

4. That, in seeking the restoration of the States to their That, in seeing the resolution the bars to then proper places as members of the Republic, so that every State shall enjoy again its constitutional functions, and every star on our national flag shall represent a State, in reality as well as in name, care must be taken that the rebellion is not allowed, through any negligence or mistaken or the least germ of future life; that, whether proceeding by the exercise of sovereign rights or of belligerent rights, by one exercise or sovereign rights of of beiligerent rights, the same precautions must be exacted against future peril; that, therefore, any system of "reconstruction" must be rejected, which does not provide by irreversible guarantees against the continued existence or possible revival of sla-very, and that such guarantees can be primarily obtained only through the agency of the petionel Government only through the agency of the national Government, which to this end must assert a temporary supremacy, military or civil, throughout the rebel and belligerent region, of sufficient duration, to stamp upon this region the character of freedom.

That, in the excrcise of this essential supremacy of the 5 national Government, a solemn duty is cast upon Congress to see that no rebel State is prematurely restored to its constitutional functions until, within its borders, all proper rafeguards are established, so that loyal citizens, including the new-made freedmen, cannot at any time be molested by evil-disposed persons, and especially that no man there may be made a slave; that this solemn duty belongs to Congress under the Constitution, whether in the exercise of rights of sovereignty or rights of war, and that in its performance that system of "reconstruction" will be found the best, that system of "reconstruction" will be found the best, howsoever it may be named, which promises most surely to accomplish the desired end, so that slavery, which is the synonym of the rebellion, shall absolutely cease through-out the whole rebel and belligerent region, and the land which it has maddened, impoverished, and degraded, shall become safe, fertile, and glorious, from assured emancipation.

6. That, in the process of "reconstruction," it is not enough to secure the death of slavery throughout the rebel enough to secure the death of slavery throughout the rebel and belligerent region only; that experience testifies against elavery wherever it exists, not only as a crime against hu-manity, but as a disturber of the public peace and the spoiler of the public libertis, including the liberty of the press, the liberty of speech, and the liberty of travel and transit; that obviously, in the progress of civilization, it has become incompatible with good government, and espe-cially with that "republican form of government" which the United States are bound to guarantee to avery State. the United States are bound to guarantee to every State; that from the outbreak of this rebel war, even in States professing loyalty, it has been an open check upon patriotic duty and an open accessory to the rebellion, so as to be a source of unquestionable weakness to the national cause; that the defiant pretensions of the master, claiming the con trol of his slave, are in direct conflict with the paramount Tights of the national Government; and that, therefore, it is the further duty of Congress, in the exercise of its dou-ble powers, under the Constitution, as guardian of the na-tional safety, to take all needful steps to secure the extinction tional satety to take an needuli steps to secure the extinction of slavery, even in States professing loyality, so that this crime against humanity, this disturber of the public peace, and this spoiler of the public liberties shall no longer exist anywhere to menace the general harmony; that civiliza-tion may be no longer shocked; that the constitutional guaranty of a republican form of government to every State may be fulfilled; that the rebellion may be deprived of the traitorous aid and comfort which slavery has in-stinctively volunteered; and that the master claiming an unnatural property in human flesh, may no longer defy the national Government.

7. That in addition to the guaranties stipulated by Con-That in author to be guarantes submated by Con-gress, and as the cap-stone to its work of restoration and reconciliation, the Constitution itself must be so amended as to prohibit slavery everywhere within the limits of the Republic; that such a prohibition, leaving all personal claims, whether of slave or master, to the legislation of Congress and of the States, will be in itself a sacred and in-table memory memory the collection will of the violable guaranty, representing the collective will of the people of the United States, and placing universal emanci-pation under the sanction of the Constitution, so that freepation under the sanction of the Constitution, so that irre-dom shall be engraved on every foot of the national soil, and be worven into every star of the national flag, while it elevates and inspires our whole national existence, and the Constitution, so often invoked for slavery, but at last in interpret the Declaration of Independence, will beharmony with the Declaration of Independence, will become, according to the holy aspirations of its founders, the sublime guardian of the inalienable right of every human being to life, liberty, and the pursuit of hap-piness; all of which must be done in the name of the Union, in duty to humanity, and for the sake of permanent peace.

0

### IN HOUSE.

1863, Dec. 14-Mr. HARDING offered this resolution which was laid over under the rule :

Resolved, That the Union has not been dissolved, and that whenever the rebellion in any of the seceded States shall be put down and subdued, either by force of the Federal arms or by the voluntary submission of the people of such State to the authority of the Constitution, then such State will be thereby restored to all its rights and privileges as a State of the Union, under the constitution of such State and the Constitution of the United States, including the right to regulate, order and control its own domestic institutions according to the constitutution and laws of such State, free from all congressional or executive control or dictation.

May 2-The resolution was laid on the table yeas 67, nays 56, as follows :

YEAS-Messrs. Alley, Allison, Ames, Anderson, Arnold, Ashley, John D. Baldwin, Baxter, Beaman, Blow, Bout-well, Boyd, Brandegee, Broomall, Cole, Henry Winter Davis, Deming, Donnelly, Driggs, Eckley, Eliot, Farnsworth, Fen-ton, Frank, Garfield, Grinnell, Hooper, Hotchkiss, A. W. Hubbard, J. H. Hubbard, John H. Hubbard, Hulburd, Julian,

Hubbard, J. H. Hubbard, John H. Hubbard, Hubburd, J. Ukar, Hubbard, J. H. Hubbard, John H. Hubbard, Hubburd, J. Ukar, Kelley, Francis W. Kellogg, Loan, Longyear, McClurg, Mo-Indoe, S. F. Miller, Moorhead, Morrill, Daniel Morris, Amoa Myers, Leonard Myers, Norton, Charles O'Neill, Orth, Per-ham, Pike, Pomeroy, Price, Alexander H. Rice, John H. Rice, Edward H. Rollins, Schenck, Scofield, Slaannon, Sloan, Spalding, Stevens, Thayer, Upson, William B. Wash-burn, Williams, Wilder, Wilson, Windom-eff. NAYS-Messrs. Ancona, Baily, Augustus C. Baldwin, Jac. b B. Blair, Brooks, William G. Brown, Chanler, Clay, Coz, Daw-son, Denison, Eden, Eldridge, Finck, Ganson, Grider, Gris-wold, Hall, Harding, Harrington, Charles M. Harris, Her-rick, Holman, Hutchins, Philip Johnson, William Johnson, Kernan, King, Knapp, Law, Luzear, Le Blond, Long, Mal-lory, Marcy, McDowell, McKinney, Morrison, Noble, Perry, Radford, Robinson, James S. Rollins, Ross, Scott, Smith, John B. Steele, Stiles, Strouse, Stuart, Whaley, Wheeler, Chilton A. White, Joseph W. White, Fernando Wood, Yea-man-56. man - 56.

December 14-Mr. WADSWORTH offered this resolution, which was laid over under the rule :

Resolved. That the powers not delegated to the United States by the Constitution nor prohibited by it to the States are reserved to the States respectively, or to the people, and the Federal Executive can, neither directly nor indirectly, exercise any of the powers thus reserved or law-fully restrict or obstruct the exercise thereof by the people.

May 2-The resolution was referred to the Committee on the Rebellious States-yeas 69, nays 50, as follows :

nays 50, 28 10110WS: YEAS-Messrs. Alley, Allison, Ames. Anderson, Arnold, Ashley, John D. Baldwin, Baxter, Beaman, Blaine, Blow, Boutwell, Boyd, Brandegee, Broomall, Cobb, Cole, Dawes, Deming, Dixon, Donnelly, Driggs, Eliot, Farnsworth, Fen-ton, Frank, Garfield, Grinnell, Higby, Hotchkiss, Asahel W. Hubbard, Hulburd, Julian, Kelley, Orlando Kellogg, Loan, Longyear, McBride, McClurg, McIndoe, Samuel F. Miller, Moorhead, Amos Myers, Leonard Myers, Norton, Charles O'Neill, Orth, Perham, Pike, Price, Alexander II. Rice, John H. Rice, Edward II. Rollins, Schenck, Scofield, Shannon, Smith, Spalding, Stevens, Thayer, Thomas, Upson, William B. Washburn, Whaley, Williams, Wilder, Wilson,

William B. Washburn, Whaley, Williams, Wilder, Wilson, Windom-69.
NATS-MOSSTS. William J. Allen, Ancona, Baily, Augustus C. Baldwin, Brooks, William G. Brown, Chanler, Cox, Deni-son, Eden, Eldridge, Finck, Ganson, Grider, Griswold, Hall, Harding, Benjamin G. Harris, Charles M. Harris, Herrick, Holman, Hutchins, Philip Johnson, Wm. Johnson, Kernan, King, Knapp, Law, Le Blond, Long, Marcy, McDowell, McKinney, Morrison, Noble, Perry, Radford, Robinson, James S. Rollins, Ross, Scott, John B. Steele, Stiles, Strouse, Stuart, Wheler, Chillon A. While, Joseph W. While, Fer-nando Wood, Yeaman-50.

1863, December-Mr. YEAMAN offered these resolutions; which were refered to the Judieiary Committee:

Joint resolutions concerning the restoration of the civil authority of the United States and of certain States within regions once or now under the control of the existing rebellion.

Be it resolved, &c., 1. That a combination of persons, in the name of a State, or an assumed confederation of States, for levying war against the United States, or for withdrawing such States from the Union, does not alter the legal character of the act done, nor excuse those engaged in it, nor does any such combination, levying of war, or attempted withdrawal amount to any destruction, forfeiture, or abdication of the right of those who at any time acknowledge allegiance and render obedience to the United States to administer, amend. or establish a State government.

antegrates and render outerine of the onited States to administer, amond, or establish a State government. 2. That a formal return or readmission of any State to the Union is not necessary. It is sufficient that the people, or those who are loval in any State, and qualified by the election laws thereof in force before the rebellion, shall at any time resume the functions of a State government compatible with the Union and with the Constitution and laws of the United States, and doing this is sufficient evidence of loyalty for the purpose of doing it.

3. That all questions touching property-rights and inter ests arising out of confiscation and emancipation, and the effect of any law, proclamation, military order, or emergency of war, or act of rebellion, upon the title to any property, or upon the status of any persons heretofore held to service or labor in any State, are left for the judicial determination of the courts of the United States.

4. That the whole power of the nation is pledged for the suppression of the rebellion, the execution of the laws, the defense of loyal citizens in any State, the territorial integrity of the Republic, and the nationality of the Constitution.

5. That nothing herein contained shall be construed to abridge or lessen any valid defense, or as waiving the right of the Government to inflict punishment; the purpose being to declare the nullity of sacession as a State ordinance, to define the objects of the war, and to express the sense of Congress as to the proper mode of restoring harmonicous relations between the Government and certain of the States and the disaffected people thereof.

Subsequently, December 21, he offered these resolutions, being, in part, a modification of the others, upon which he addressed the House, January 12, 1864:

Resolved, That a conspiracy of persons combined together, and assuming the name of a State, or a confederation of States, for levying war upon the United States, or for withdrawing such States from the Union, does not extinguish the political franchises of the loyal citizens of such States: and such loyal citizens have the right, at any time, to administer, amend, or establish a State government without other condition than that it shall be republican in form.

2. That a formal return or readmission of any State into the Union is not necessary. It is sufficient that the people, or those who arc loyal in any State, and qualified by the election laws thereof in force before the rebellion, shall, at any time, resume the functions of a State government compatible with the Union, and with the Constitution and laws of the United States; and doing this is sufficient evidence of loyalty for the purpose of doing it.

3. That all questions touching property-rights and interests, arising out of confiscation and emancipation, and the effect and validity of any law, proclamation, military order, emergency of war, or act of rebellion, upon the title to any property, or upon the status of any persons heretofore held to service or labor in any State under the laws thereof, are left for the judicial determination of the courts of the United States.

The House refusing to second the demand for the previous question on their passage, they were, on motion of Mr. LOVEJOY, referred to the Select Committee on the Rebellious States.

Feb. 16-Mr. BOUTWELL proposed these resolutions:

Resolved, That the Committee on the rebellious States be instructed to consider and report upon the expediency of recommending to this House the adoption of the following

# Declaration of Opinions:

In view of the present condition of the country, and especially in view of the recent signal successes of the national arms, promising a speedy overthrow of the rebellion, this House makes the following declaration of opinions concerning the institution of slavery in the States and parts of States engaged in the rebellion, and embraced in the proclamation of emancipation issued by the President on the 1st day of January, A. D. 1863; and also concerning the relations now subsisting between the people of such States and parts of States on the one side and the American Union on the other.

It is therefore declared, (as the opinion of the House of Representatives,) That the institution of slavery was the cause of the present rebellion, and that the destruction of slavery in the rebellious States is an efficient means of weakening the power of the rebels; that the President's proclamation, whereby all persons heretofore held as slaves in such States and parts of States have been declared free, has had the effect to increase the power of the Union and to diminish the power of its enemies; that the freedom of such persons was desirable and just in itself, and an efficient means by which the Government was to be maintained and its authority re-established in all the territory and over all the people within the legal jurisdiction of the United States; that it is the duty of the Government and of loyal men everywhere to do what may be practicable for the enforcement of the proclamation, in order to secure in fact as well as by the forms of law the extinction of slavery in such States and parts of States; and finally, that it is the paramount duty of the Government and of all loyal men to labor for the restoration of the American Union on the basis of freedom.

And this House does further declare, That a State can exist or cease to exist only by the will of the people within its limits, and that it cannot be created or destroyed by the external force or opinion of other States, or even by the indegment or action of the nation itself; that a State when created by the will of its people can become a member of the American Union only by its own organized action and the concurrent action of the existing National Government; that when a State has been admitted to the Union, no vote, resolution, ordinance, or proceeding, on its part, however formal in character or vigorously sustained, can deprive the National Government of the legal jurisdiction and sovereignty over the territory and people of such State which existed previous to the act of admission, or which were acquired thereby; that the effect of the so-called acts, resolutions, and ordinances of sccession adopted by the eleven States engaged in the present rebellion is, and can only be, to destroy those political organizations as States, while the legal and constitutional jurisdiction and authoriy of the National Government, over the people and territory remain unimpaired; that these several communities can be organized into States only by the will of the loyal people expressed freely and in the absence of all coercion; that States so organized can become States of the American their admission shall have been authorized by the existing national Government; that when a people have organized a State upon the basis of allegiance to the Union, and applied for admission, the character of the institutions of such astate upon the basis of allegiance to the union, and applied for admission that the existence of human slavery is incompatible with a republican form of Government in the several States or in the United States, and inconsistent with the people and of all men in authority to resist the duty of the people and of all men in authority to resist

1864, March 14-Mr. WILLIAMS offered these resolutions, which were laid over under the rule:

Resolved, That the existing relation between the Union and the rebel States constitutes a condition of public war, with all the consequences attaching thereto under the law of nature and of nations.

2. That the appeal of the rebel States from the jurisdiction of the ordinary tribunals established by the Constitution to the arbitrament of the sword has not, however, withdrawn the case beyend the purview of the Constitution, which, in conferring the war power on the General Government, has made the law of war the rule of conduct in the prosecution and adjustment of the pending controversy.

The solution and adjustment of the pending controversy. 3. That while the rebel States are, by that law and by the solemn recognition thereof in the proclamation made by the President of the United States on the 10th of August, A. D. 1861, under and in pursuance of the authority conferred on him by the act of Congress of the 13th of July of the same year, interdicting all commercial intercourse between their citizens and those of the loyal States, in the attitude of belligerents, and outside of the Union as States, by construction of law as well as in point of fact, and have thereby either abdicated or forfeited their rights to membership therein, the jurisdiction and powers of the Government over their territory and citizens continue unimpaired, and the latter are still amenable to the law and the judicial tribunals of the United States for their treason and other crimes against the same

4. That so long as those States continue under the armed occupation of the forces of the United States employed in suppressing the rebellion against its authority, the local laws are necessarily subordinated, and the functions of the civil anthorities so far suspended therein as to prevent the exercise of all the rights arising out of their relations to this Government, and to disable them from electing members of either branch of Congress, or electors for the choice of a President of the United States.

e

5. That the occupation of these States by the armed forces That the occupation of these States by the armed forces of the United States, either under military commanders or provisional Governors appointed by the Executive, resting for its lawfulness upon the condition of insurrection existing therein, is a purely military one, and ought to determine with the necessity which produced it.
 That as soon as the rebellion is suppressed in any of the rest of the rest of the rest of the rest of the rest.

b. That as soon as the redenion is suppressed in any of the revolted States, by the reconquest and occupation thereof by the armies of the United States, and the same are so tranquilized as to furnish adequate assurance against the recurrence of disturbance therein, it will become the duty of the President to communicate the fact to Congress in order that it may take the proper measures for the reorganization of the civil governments and the re-establish-ment of the civil functionaries therein, and prescribe such terms as it may deam wise and proper and consistent with the public safety for the readmission of those districts as s of this Union. State

7. That it is the exclusive right of the legislative power of the Government to say upon what terms those Territorics shall be allowed to return to the Union; and that in the adjustment of the existing controversy in the Government ad interim of the reconquered territory, and in the arrange-ment of the terms of reorganization and readmission, it will be entirely within their competency to putish the treason of individuals and provide indemnity for the expenses of the war and security against any future outbreak of the like kind by removing its causes and confiscating absolutely the property and estate of the guilty authors and abettors thereof

# Rebel Views of "Reconstruction." 1861.

By JEFFERSON DAVIS, February 16, 1861, at Montgomery:

The time for compromise has now passed, and the South is determined to maintain her position, and make all who oppose her smell southern powder and feel southern steel if coercion is persisted in. He had no doubts as to the result. He said we will maintain our rights and Government at all he said would be ask nothing, we want nothing; we will have no complications. If the other States join our confederation they can freely come in on our terms. Our separation from the old Union is now complete. No compromise, no recon-struction is now to be entertained.

By WALTER BROOKE, of Mississippi, in "Confederate" Provisional Congress, March 5, 1861, quoting Davis and Stephens:

These men have long since given up all hope of receiving satisfiction from the General Government, and the entire people of Mississippi stand to-day upon the same platform, I am authorized, I think, to speak their sentiments on this floor, from the information I am daily receiving. I do not believe that there is a man in Mississippi who desires a reconstruction of this Government, or who will not fully in-dorse the sentiments uttered by you, Mr. President, that "the separation is perfect, complete, and perpetual," and likewise the sentiments of our distinguished President of the Confederate States, when he declared that "a reconstruction is neither practicable nor desirable."

T. R. R. COBB, of Georgia, a member of the Provisional Congress, spoke at Atlanta on reconstruction:

Iam against it now and forever. What have we worked for? Simply a new Constitution? No! we sought to be relieved of the North because they were fleecing us; giving fishing bounties and otherwise squandering 'the public treasure, and filling their pockets from our labors. I would not unite with there if they rune to hind thereadynce any any other which then if they were to bind themselves in amounts more than they were worth, and give me a distress warrant to sell them out. I wish the people of Georgia to say-this shall be a slaveholding confederacy, and nothing else.

JEFF. DAVIS, in his message of November, 1861, says:

If, instead of being a dissolution of a league, it were indeed a rebellion in which we are engaged, we might find ample vindication in the course we have adopted in the scenes which are now being enacted in the United States. Our people now look with contemptuous astonishment on those with whom they have been so recently associated. They shrink with aversion from the bare idea of renewing such a connection.

JOINT RESOLUTIONS OF GEORGIA,

#### DECEMBER 11, 1861.

Resolved, by the Senate and House of Representatives of Georgia, in General Assembly met, That it is the sense of

this General Assembly that the separation of those States now forming the Confederate States of America from the United States is, and ought to be, final and irrevocable; and that Georgia will, under no circumstances, entertain any proposition from any quarter which may have for its object a restoration or reconstruction of the late Union, on any terms or conditions whatever.

Resolved, That the war which the United States is waging upon the Confederate States should be met on our part with the utmost vigor and energy, until our independence and nationality are unconditionally acknowledged by the United States.

Resolved, That Georgia pledges herself to her sister States the struggle; she will contribute all the means which her resources will supply, so far as the same may be necessary to the support of the common cause, and will not consent to lay down arms until peace is established on the basis of the foregoing resolutions. WARREN AIKEN,

foregoing resolutions. Speaker of the House of Representatives. L. CARRINGTON, Clerk of the House of Representatives. JOHN BILLUPS,

President of the Senate.

JAMES M. MORLEY, Secretary of the Senate. Approved December 11, 1861. JOSEPH E. BROWN, Governor.

See Roger A. Pryor's declaration at Charleston, page 112.

#### 1862.

Feb. 3-The Congress adopted this preamble and resolution unanimously :

Whereas, the United States are waging war against the Confederate States, with the avoved purpose of compelling the latter to re-unite with them under the same constitu-tion and government; and whereas, the waging of war with such an object is in direct opposition to the sound re-publican maxim, that "all government rests upon the con-sent of the governed," and can only tend to consolidation in the General Government, and the consequent destruction of the rights of the States; and whereas, this result being at-bained the two sections can only aviit theorem. the rights of the states; and whereas, this result being at-tained, the two sections can only exist together in the re-lation of the oppressor and the oppressed, because of the great prependerance of power in the Northern section, coupled with dissimilarity of interests; and whereas, we, the representatives of the people of the Confederate States, in Congress assembled, may be presumed to know the sen-timents of said people, having just been elected by them: Therefore be it. Therefore be it

Resolved, That this Congress do solemnly declare and publish to the world that it is the unalterable determina-tion of the people of the Confederate States (in humble retion to the people of the confiderate States (in humble re-liance upon Almighty God) to suffer all the calamities of the most protracted war, but that they will *never*, on any terms, politically affiliate with a people who are guilty of an invasion of their soil and the butchery of their citizens.

JOHN C. BRECKINRIDGE, announcing himself a candidate in the Eleventh District of Kentucky for the permanent "Confederate" Congress, at the election held Jan. 22, said :

I am utterly opposed to a reconstruction of the old Gov-I am utterly opposed to a reconstruction of the old Gov-ernment, or any measure which in the remotest degree tends in that direction. For one, I shall never consent that peace shall be made until the very last of all the ene-mies of our liberty shall be driven, not only from our hal-lowed soil, but from every foot of territory which, by its geographical position, naturally belongs to the South. God grant that the day may not be far distant when Kentucky will arise, free and disenthralled, and assume her true po-sition as one, the fairest, among the sisters of the South.

ALEXANDER R. BOTELER, of Virginia, about the same time said the same thing :

In regard to the canvass for Congress, I have been studi-ously silent, as I have a special repugnance to whatever may seem like thrusting myself on the public; but you can say for me that I have consented to become a candidate, say for me that I have consented to become a candidate, which I suppose will be sufficient. In doing so, however, it is but proper that I should say that, having done all that I could, consistent with self-respect, to preserve the Union upon its original basis of constitutional equality, I am equal-ly resolute in my determination to resist all attempts, should any be made, for its restoration; being unalterably opposed to reconstruction, at any time or on any terms. This much is due to the people that I should make known before the election, so that they may be aware of the course I shall pursue. if elected. I shall pursue, if elected.

In the winter of 1862, GEORGE N. SANDERS, in his letter to "Governor Seymour, Dean Rich-

mond, John Van Buren, Charles O'Connor, Washington Hunt, Fernando Wood, and James Brooks, representative men of the triumphant revolutionary party of New York," under date of December 24, said :

Not only do you owo it to yourselves to repudiate every dollar of this unconstitutional debt, by you owe it equally to your posterity to pay the half, if not all the debt the peo-ple of the South have had to incur to maintain the rights of citizens and of States, in the establishment of free trade.

Let heart and brain into the revolution; accelerate and direct the movement; get rid of the Baboon, (or What is it!) Abraham Lincoln, pacifically if you can, but by the blood of his followers if necessary. Withdraw your support, material and moral, from the invading armies, and the South will make quick work with the Abolitionists that remain on her soil. Suffer no degenerate son of the South, tupon however plausible pretext, to idly embarruss your action by throwing into your way rotten plauks of recon-struction. Unity is no longer possible. The very word Union, once so dear, has been made the cover for so many atrocious acts that the mere mention of it is odious in the cars of Southern people. The State Legislature will be called upon to obliterate the hated name from counties and towns.

In the fall of 1862, after the Democratic successes in the election, the Richmond papers took occasion to assure the North that there could be no peace except by recognizing the independence of the Southern Confederacy.

The Examiner said of the North :

They do not yet understand that we are resolute to be rid of them forever, and determined rather to die than to live with them in the same political community again.

The Dispatch anticipating Democratic ascendency in the Union Congress, said :

It is probable, therefore, that they might propose a re-construction of the Union, with certain pledges, guaranties, &c. To this the South will never consent. They will never set to this the solution with never consent. They will never exist in the same political association, be its name what it may, be its terms what they will, and be the guaranties what-ever the good will of the Democrats may make them. They want nothing more to do with the Yankees, and they are determined to have nothing more to do with them. Let them pay off their own debt with their own resources, we have as much as we can do to pay off ours. We cannot consent to return to the state of vassalage from which we have emerged. We cannot consent to sit in a Congress in have emerged. We cannot consent to sit in a Congress in which nothing is debated but the *nigger* from the begin-ning of the longest session to the end of it. We can never again affiliate with people who made a martyr of the cold-blooded assassin, John Brown, and thought he was doing a glorious deed when he was dying his hands in the blood of our people. We of the Confederate States have made up our our people. We so the content at states have made up our minds to endure the worst extremity to which war can re-duce a people. We are prepared for it. The Government that should propose to reunite us with the Yankees could not exist a day. It would sink so deep beneath the deluge of popular indignation that even history would not be able to fish up the wreck.

# Again, October 18, the Dispatch said :

Nor, after the sacrifices which the South has suffered at

Northern hands, could she ever consent, of her own free will, to live under the same Government with that people. The blood of our murdered children would cry from the ground against their tathers if they could ever be guilty of such unnatural and monstrous ingratitude. If the South has given her blood without a murmur to this contest, it is not because she does not value that blood but because she values freedom more than life or any earthly possession. Precious, more than aught else save her honor, are the jewels she has laid upon the altar of liberty, and never can she consent to shake hands again under one Government with men who have made so many vacant places in southern households, and whose steel is dripping with the blood of our brethren and children.

Henceforth we are two people.

December 2-The Legislature of North Carolina adopted these, among other, resolutions :

Resolved, That the Confederate States have the means and the will to sustain and perpetuate the Government they have established, and to that end North Carolina is determined to contribute all of her power and resources. Resolved, That the separation between the Confederate States and the United States is final, and that the people of

North Carolina will never consent to reunion at any time or upon any terms.

December 8-Governor Letcher, of Virginia, wrote .

It cannot be that the people of the Confederate States can again entertain a feeling of affection and respect for the Government of the United States. We have, therefore, separated from them; and now let it be understood that the separation is and ought to be final and irrevocable; that Virginia "will under no circumstances entertain any properties from them questre which mer have for its obproposition from any quarter which may have for its object a restoration or reconstruction of the late Union, on any terms or conditions whatever.'

Jeff. Davis addressed the Mississippi Legislature, December 26, 1862, and is reported in the Jackson Mississipian to have said :

He alluded to it, however, as a matter of regret that the best affections of his heart should have been bestowed upon an object so unworthy—that he should have loved so long a Government which was rotten to its very core. He had predicted from the beginning a fierce war, though it had assumed more gigantic proportions than he had calculated upon. He had predicted war not because our right to seupon. He had predicted war not because our right to se-cede was not an undoubted one, and defined in the spirit of that declaration which rests the right to govern upon the consent of the governed; but the wickedness of the North would entail war upon the country. The present war, waged against the rights of a free people, was unjust, and the fruit of the evil passions have heen brought out and develoned; and so fra from remuting

been brought out and developed; and so far from reuniting with such a people—a people whose descendants Cromwell had gathered from the bogs and fens of Ireland and Scotland—a people whose intolerance produced discord and trouble wherever they went—who persecuted Catholics, Episcopalians, and every other sect that did not subscribe to their bigoted and contracted notions-who hung witches, and did a thousand other things calculated to make them forever infamous.

The President was emphatic in his declaration that under no circumstances would he consent to reunion. He drew a glowing picture of the horrors of war, and the rav-ages of the enemy, and while his tears flowed for those who suffered, yet all these would be endured cheerfully before our manhood and our liberties would be surrendered."

#### 1863.

From the speech of Jeff. Davis, delivered in Richmond, as reported in the Richmond Enquirer of January 7:

You have shown yourselves in no respect to be degenerate sons of your fathers. You have fought mighty battles, and sons of your lathers. You have longht might parties, and your decds of valor will live among the richest spoils of Time's ample page. It is true, you have a cause which binds you together more firmly than your fathers were. They fought to be free from the usurpation of the British crown, but they fought against a manly foe. *Fou fight the offseourings of the carth.* [Applause.] \* \* \* offscourings of the carth. [Applause.] \* \* \* They have come to disturb your social organizations on

the plea that it is a military necessity. For what are they waging war? They say to preserve the Union. Can they preserve the Union by destroying the social existence of a portion of the South? Do they hope to reconstruct the Union by striking at everything which is dear to men? By showing themselves so utterly disgraced that if the question was proposed to you whether you would combine with Hy-enas or Yankees, I trust every Virginian would say, give me the Hyenas. [Cries of "Good! good!" and applause.]

[From the Richmond Dispatch of January 11.]

Reconstruction! Can they reconstruct the family circles which they have broken—can they reconstruct the family effects which they have broken—can they reconstruct the fortunes which they have scattered—can they reconstruct the bodies of our dead kindred, which by tens of thousands they have destroyed? When they can do this they can reconstruct the old Union. When they can do this—when they can breathe the breath of life into the pallid faces of our sons and broknes and next for them one onco this when they can and brothers, and restore them once more, living and happy, to our desolate firesides, they may dream of bringing back that Union whose only principle of cohesion was the mutual love and confidence of its people. \*

We warn the Democrats and conservatives of the North the old Union. If the North will allow us to write the Constitution ourselves, and give us every guarantee we would ask, we would sooner be under the Government of England or France than under a Union with men who have shown that

they cannot keep good faith, and are the most barbarous and inhuman, as well as treacherous of mankind. If the reconstructionists want peace, they can easily have it upon the terms on which they could have always had it—letting us alone. We ask neither more nor less. We are making no war on them. We are not invading their territory, nor giving their homes to the flames, their populations to prison and the sword, their women to a fate worse than death. Let us alone! That is all we ask. Let us alone, and peace will return once more to bless a dis-tracted land! But do not expect us to degrade ourselves and sat dishonor upon the graves of our kindred by ever return-ing to the embrace of those whose hands are dripping with the tears and blood of our people.

The Richmond Sentinel refers to the address of the Democrats of New Hampshire declaring that if the South will "come back into the Union, the Democracy of the North will do all in their power to gain for them (the Southern States) such guarantees as will secure their safety," and remarks that the proposition is safety," and remarks that the proposition frank and courteous, but inadmissible.  $\mathbf{It}$ adds :

"They (the Democrats) are powerless to secure for us those guarantees of which they admit the necessity. Less than three years ago the States which now form the Con-federacy, sought, in the spirit of conservatism and forbear-ance, to avoid disruption, with an importunity that now appears to us amazing. When we look back at it now it in the two think they two offends to take the ance, to avoid disciplicity, with an importantly that now it makes us tremble to think that we offered to take the Crittenden compromise. But conciliation on our part was met only by contumely and define by the Republican majority. From that time the men who willfully destroyed the University of the second second second second second second to be the second the Union have been assailing us with all the engines of destruction. They have evinced towards us a malignity which has seldom been paralleled in human history.

"Do the New Hampshire Democrats suppose for one "Do the New Hampshire Democrats suppose for one moment that we could so much as think of reunion with such apeople? Rather tell one to be wedded to a corpsel Rather join hands with a fiend from the pit. Since that time the only greeting of kind words which has come to us from the North, the New Hampshire men have sent. All, or nearly all beside, has been configration, sword, demonia denunciation, and brutal menace of destruction. When those in the United States who are disposed to deal fairly with us shall gain the rule, we may in time begin to bury the many bitter memories which now add energy to our resentment, and may make with them treaties that shall be mutually advantageous. Perhaps, hereafter, good will may be revived again. But Union-never let it be mentioned! It is impossible."

#### FROM ALEXANDER H. STEPHENS.

The Richmond Dispatch of July 23, gives a sketch of ALEX. H. STEPHENS'S speech at Charlotte, N. C., on his way to Richmond, in which after alluding to Lee's invasion of Maryland and Pennsylvania, and the loss of Vicksburg, and exhorting the people to give the government a cordial support, he said :

As for reconstruction, such a thing was impossible-such an idea must not be tolerated for an instant. Reconstructhe near flux hot be to be not at in the tail. Reconstruc-tion would not end the war, but would produce a more hor-rible war than that in which we are now engaged. The only terms on which we can obtain permanent peace is final and complete separation from the North. Rather than submit to anything short of that, let us all resolve to die like men worthy of freedom.

#### FROM ROBERT TOOMBS.

# WASHINGTON, GA., August 17, 1863.

Washington, GA., August 11, 1863. My DEAR SIR: Your letter of the 15th instant, asking my authority to contradict the report that "I am in favor of reconstruction," was received this evening. I can conceive of no extremity to which my country could be reduced in which I would for a single moment entertain any proposi-tion for any union with the North on any terms whatever. When all else is lost, I prefer to unite with the thousands of aur own comptyment who have found hourable destha

tember, published in the Richmond Examiner. in which he "denounced, in bitter terms, as an utter impossibility, any thought of reconstruction on any conditions."

#### THE QUESTION ABOUT NORTHERN " TONE AND TEMPER.'

# [From the Richmond Sentinel, September 29.]

The House of Delegates yesterday, in the most summary manner, disposed of a resolution for inquiring into the tone and temper of the people of the United States on the sub-ject of peace, with a view to responding, if favorable. The House knew what everybody knows—that such resolutions are both idle and mischierous, for they will only be taken by our enemies as evincing more or less readiness on our part for reconstruction. The House, by a unanimous vote, put its foot on the resolution, without a word ot discussion or a moment of delay. In this they but fairly represented the manliness and the unanimity of our people.

#### 1864

Gov. ZEBULON B. VANCE of North Carolina, in his elaborate speech at Wilkesboro', used this language:

#### RECONSTRUCTION.

It is a favorite idea with a great many, that possibly the old order of things could be restored; that our rights un-der that Constitution could be guaranteed to us, and every-thing move on peacefully as before the war. My friends, there are a great many desirable things; but the question, not what may be wished, but what may be obtained, is the one reasonable men may consider. It is desirable to have a lovely wife and plenty of pretty children : but every man ear't have them. I tell you now, candidly, there is no more possibility of reconstructing the old Union and rein-stating things as they were four years ago, than exists for stating things as they were four years ago, than exists for you to gather up the scattered bones of your sons who have fallen in this struggle, from one end of the country to the other, re-clothe them with flesh, fill their veins with the blood they have so generously shed, and their lungs with the same breath with which they breathed out their last prayer for their country's triumph and independence. [Immense applause.]

The new rebel Governor of Louisiana, HENRY W. Allen, in his inaugural, says :

Peace is not so sweet as to be purchased at the cost of reconstruction. Reconstruction means subjugation, ruin, death. Lose negroes, lose lands, lose everything, lose life itself, but never think of reconstruction. He says "I speak to-day by authority, i speak as the Governor of Louisiana, and I wish it known at Washington and else-where that rather than reconstruct this Government and go back to the Union, on any terms whatever, the people go back to the Union, on any terms whatever, the people of Louisiana will, in convention assembled, without a dis-senting voice, cede the State to any European Power. \* \* I speak to-day not only for the loyal citizens of Louisiana, who have stood by her in all trials, but in behalf of the misguided individuals who have been com-pelled to take the cath of allegiance to the Federal Gov-ernment. In their hearts they are true to us, and are daily praying for the triumph of our arms. They have flet the iron in their souls, and know full well the curse of reconstruction. I speak by authority, for they write to me daily that they would rather, by 'ten thousand times, be the subjects of the Emperor of France than the slaves of Lincoln.''

The Richmond Dispatch, in March, discussed President Lincoln's amnesty proclamation, and adds:

No one, however, knows better than Abraham Lincoln that any terms he might offer the southern people which contemplate their restoration to his bloody and brutal Gor-ernment, would be rejected with scorn and exectation. If, instead of devoting to death our President and military and of no extremity to which my country could be reduced in which I would for a single moment entertain any proposi-tion for any union with the North on any torms whatever. When all else is lost, I prefer to unite with the thousands of our own countrymen who have found honorable desk if not graves, on the battle-field. Use this letter as you please. Very truly, your friend, &c. R. TOOMBS. Dr. A. BEES, Americus, Ga. WM. SMITH, then recently elected Governor of Virginia, made a speech in Richmond in Sep-

#### JOINT RESOLUTION OF THE REBEL LEGISLATURE OF LOUISIANA.

Be it resolved, &c., That the barbarous manner in which our enemics have waged war against us deserves the exccration of all men, and has confirmed and strengthened us in the determination to oppose to the last extremity a reunion with them, and that the spirit of our people is unabated in the resolution to resist every attempt at their subjugation.

1864, August 5—Messrs. BENJAMIN F. WADE of Ohio, and HENRY WINTER DAVIS of Maryland, published in the New York *Tribune* a paper arraigning the President for his course in reference to the reconstruction bill. A few extracts are subjoined:

The President, by preventing this bill from becoming a law, holds the electoral votes of the rebel States at the dictation of his personal ambition.

If those votes turn the balance in his favor, is it to be supposed that his competitor, defeated by such means, will acquiesce?

If the rebel majority assert their supremacy in those States, and send votes which clect an enemy of the Government, will we not repel his claims?

And is not that civil war for the Presidency inaugurated by the votes of rebel States?

By the votes of refer States? Seriously impressed with these dangers, Congress, "the proper constitutional authority," formally declared that there are no State governments in the rebel States, and provided for their erection at a proper time; and both the Senate and the House of Representatives rejected the Senators and Representatives closen under the authority of what the President calls the free constitution and government of Arkansa.

The President's proclamation "holds for naught" this judgment, and discards the authority of the Supreme Court, and strides headlong toward the anarchy his proclamation of the 8th of December inaugurated.

If electors for President be allowed to be chosen in either of those States, a sinister light will be cast on the motives

which induced the President to "hold for naught" the will of Congress rather than his government in Louisiana and Arkansas.

That judgment of Congress which the President defies was the exercise of an authority exclusively vested in Congress by the Constitution to determine what is the established government in a State, and in its own nature and by the highest judicial authority binding on all other departments of the Government.

A more studied outrage on the legislative authority of the people has never been perpetrated. Congress passed a bill; the President refused to approve

Congress passed a bill; the President refused to approve it, and then by proclamation puts as much of it in force as he sees fit, and proposes to execute those parts by officers unknown to the laws of the United States and not subject to the confirmation of the Senate!

to the confirmation of the Senate! The bill directed the appointment of Provisional Governors by and with the advice and consent of the Senate.

The President, after defeating the law, proposes to appoint without law, and without the advice and consent of the Senate, *Military* Governors for the rebel States!

Senate, Milliary Governors for the rebel States ! He has already exercised this dictatorial usurpation in Louisiana, and he defeated the bill to prevent its limitation. \* \* \* \* \* \* \* \* \* \* \*

The President has greatly presumed on the forbearance which the supporters of his Administration have so long practiced, in view of the arduous conflict in which we are engaged, and the reckless ferocity of our political opponents.

nents. But he must understand that our support is of a cause and not of a man; that the authority of Congress is paramount and must be respected; that the whole body of the Union men of Congress will not submit to be impeached by him of rash and unconstitutional legislation; and if he wishes our support, he must confine himself to his executive duties—to obey and execute, not make the laws—to suppress by arms armed rebellion, and leave political reorganization to Congress.

If the supporters of the Government fail to insist on this, they become responsible for the usurpations which they fail to rebuke, and are justly liable to the indignation of the people whose rights and security, committed to their keeping, they sacrifice. Let them consider the remody for these usurpations, and,

Let them consider the remedy for these usurpations, and, having found it, fearlessly execute it.

# REMAINING PAPERS OF PRESIDENT LINCOLN.

# EXPLANATORY OF GOVERNMENT PURCHASES IN | I also directed the commandant of the navy-MAY, 1861. yard at Boston to purchase or charter, and arm

1862, May 29—The PRESIDENT sent this message\* to Congress:

To the Senate and

# , House of Representatives:

The insurrection which is yet existing in the United States, and aims at the overthrow of the Federal Constitution and the Union, was clandestinely prepared during the winter of 1860 and 1861, and assumed an open organization in the form of a treasonable provisional government at Montgomery, in Alabama, on the 18th day of February, 1861. On the 12th day of April, 1861, the insurgents committed the flagrant act of civil war by the bombardment and capture of Fort Sumter, which cut off the hope of immediate conciliation. Immediately afterwards all the roads and avenues to this city were obstructed, and the capital was put into the condition of a siege. The mails in every direction were stopped, and the lines of telegraph cut off by the insurgents, and military and naval forces, which had been called out by the Government for the defence of Washington, were prevented from reaching the city by organized and combined treasonable resistance in the State of Maryland. There was no adequate and effective organization for the public de-Congress had indefinitely adjourned. fence. There was no time to convene them. It became necessary for me to choose whether, using only the existing means, agencies, and processes which Congress had provided, I should let the Government fall at once into ruin, or whether, availing myself of the broader powers conferred by the Constitution in cases of insurrection, I would make an effort to save it with all its blessings for the present age and for posterity. I thereupon summoned my constitutional advisers, the heads of all the Departments, to meet on Sunday, the 20th day of April, 1861, at the office of the Navy Department, and then and there, with their unanimous concurrence, I directed that an armed revenue cutter should proceed to sea, to afford protection to the com-

mercial marine, and especially the California

treasure ships then on their way to this coast.

yard at Boston to purchase or charter, and arm as quickly as possible, five steamships, for purposes of public defence. I directed the commandant of the navy-yard at Philadelphia to purchase, or charter and arm, an equal number for the same purpose. I directed the commandant at New York to purchase, or charter and arm, an equal number. I directed Commander Gillis to purchase, or charter and arm, and put to sea two other vessels. Similar directions were given to Commodore Du Pont, with a view to the opening of passages by water to and from I directed the several officers to the capital. take the advice and obtain the aid and efficient services in the matter of his Excellency Edwin D. Morgan, the Governor of New York, or, in his absence, George D. Morgan, William M. Evarts, R. M. Blatchford, and Moses H. Grinnell, who were, by my directions, especially empowered by the Secretary of the Navy to act for his Department in that crisis, in matters pertaining to the forwarding of troops and supplies for the public defence.

On the same occasion I directed that Governor Morgan and Alexander Cummings, of the city of New York, should be authorized by the Secretary of War, Simon Cameron, to make all necessary arrangements for the transportation of troops and munitions of war, in aid and assistance of the officers of the Army of the United States, until communication by mails and telegraph should be completely re-established between the cities of Washington and New York. No security was required to be given by them, and either of them was authorized to act in case of inability to consult with the other.

On the same occasion I authorized and directed the Secretary of the Treasury to advance, without requiring security, two millions of dollars of public money to John A. Dix, George Opdyke, and Richard M. Blatchford, of New York, to be used by them in meeting such requisitions as should be directly consequent upon the military and naval measures necessary for the defence and support of the Government, requiring them only to act without compensation, and to report their transactions when duly called upon.

The several departments of the Government at that time contained so large a number of disloyal persons that it would have been impos-

<sup>\*</sup> Called forth by the passage of a resolution, April 30, in the House—yeas 79, nays 45—censuring Secretary Cameron for a supposed responsibility for, and connection with, the circumstances detailed.

sible to provide safely, through official agents only, for the performance of the duties thus confided to citizens favorably known for their ability, loyalty, and patriotism.

The several orders issued upon these occurrences were transmitted by private messengers, who pursued a circuitous way to the seaboard cities, inland, across the States of Pennsylvania and Ohio and the Northern Lakes. I believe that by these and other similar measures taken in that crisis, some of which were without any authority of law, the Government was saved from overthrow. I am not aware that a dollar of the public funds thus confided without authority of law to unofficial persons was either lost or wasted, although apprehensions of such misdirection occurred to me as objections to those extraordinary proceedings, and were necessarily overruled.

I recall these transactions now because my attention has been directed to a resolution which was passed by the House of Representatives on the 30th day of last month, which is in these words:

Resolved, That Simon Cameron, late Secretary of War, by investing Alexander Cummings with the control of large sums of the public money, and authority to purchase military supplies without restriction, without requiring from him any guarantee for the faithful performance of his dutics, when the services of competent public officers were available, and by involving the Government in a vast number of contracts with persons not legitimately engaged in the business pertaining to the subject-matter of such contracts, es-pecially in the purchase of arms for future delivery, has adopted a policy highly injurious to the public service, and deserves the censure of the House.

Congress will see that I should be wanting equally in candor and in justice if I should leave the censure expressed in this resolution to rest exclusively or chiefly upon Mr. Cameron. The same sentiment is unanimously entertained by the heads of Departments, who participated in the proceedings which the House of Representatives has censured. It is due to Mr. Cameron to say that, although he fully approved the proceedings, they were not moved nor suggested by himself, and that not only the President but all the other heads of Departments were at least equally responsible with him for whatever error, wrong, or fault was committed in the premises.

# ABRAHAM LINCOLN. WASHINGTON, May 26, 1862.

#### THE PRESIDENT'S REMARKS AT A UNION MEET-ING IN WASHINGTON, AUGUST 6, 1862.

FELLOW-CITIZENS: I believe there is no precedent for my appearing before you on this occasion, but it is also true that there is no precedent for your being here yourselves, and I offer, in justification of myself and of you, that, upon examination. I have found nothing in the Constitution against it. I, however, have an impression that there are

against if. I, however, have an impression that there are younger gentlemen who will entertain you better, and better address your understanding than I will or could, and therefore I propose but to detain you a moment longer. I am very little inclined on any occasion to say anything unless I hope to produce scome good by it. The only thing I think of just now not likely to be better said by some one else, is a matter in which we have heard some ofter persons blamed for what I did myself. There has been avery wide-scread atternat to have a quarrel between Gen, McClellan blamed for what I did myself. There has been avery wide-spread attempt to have a quarrel between Gen. McClellan and the Secretary of War. Now, I occupy a position that enables me to observe, that these two gentiemen are not nearly so deep in the quarrel as some pretending to be their friends. General McClellan's attitude is such that, in the very selfishness of his nature, he caunot but wish to be suc-cessful, and I hope he will—and the Secretary of War is in precisely the same situation. If the military commanders

in the field cannot be successful, not only the Secretary of War, but myself, for the time being the master of them both, cannot but be failures. I know General McClellan wishes to be successful, and I know he does not wish it any more than the Secretary of War for him, and both of them together no more than I wish it. Sometimes we have a dispute about how many men General McClellan has had. dispute about how many men General McClellan has had, and those who would disparage him say that he has had a very large number, and those who would disparage the Secretary of War insist that General McClellan has had a very small number. The basis for this is, there is always a wide difference, and on this occasion, perhaps a wider one than usual, between the grand total on McClellan's rolls and the men actually fit for duty; and those who would disparage him talk of the grand total on paper, and those a who would disparage the Secretary of War talk of those at present fit for duty. General McClellan has sometimes asked for things that the Secretary of War did not give him. asked for things that the Secretary of War did not give him. General McClellan is not to blame for asking for what he wanted and needed, and the Secretary of War is not to blame for not giving when he had none to give. And I say here, as far as I know, the Secretary of War has withheld no one thing at any time in my power to give him. I have no accusation against him. I believe he is a brave and able man, and I stand here, as justice requires me to do, to be here any set of what has hear chorged on the Sacretary. take upon myself what has been charged on the Secretary

of War, as withholding from him. I have talked longer than I expected to do, and now I avail myself of my privilege of saying no more.

# The President's Letters on Politics.

# TO HORACE GREELEY.

#### EXECUTIVE MANSION WASHINGTON, Friday, Aug. 22, 1862.

Hon HORACE GREELEY: DEAR SIR: I have just read yours of the 19th instant, ad-dressed to myself through the New York *Tribune*. If there be in it any statements or assumptions of fact which I may know to be erroneous, I do not now and here controvert them.

If there be any inferences which I may believe to be falsely drawn, I do not now and here argue against them.

If there be perceptible in it an impatient and dictatorial tone, I waive it in deference to an old friend whose heart

I have always supposed to be right. As to the policy I "seem to be pursuing," as you say, I have not meant to leave any one in doubt. I would save the Union. I would save it in the shortest way under the Constitution.

The sooner the national authority can be restored, the nearer the Union will be-the Union as it was. If there be those who would not save the Union unless

they could at the same time save slavery, I do not agree with them.

If there he those who would not save the Union unless they could at the same time destroy slavery, I do not agree with them.

My paramount object is to save the Union and not either to

Buy paramount object is to sate the brash and the case of sate or destroy slavery. If I could save the Union without freeing any slave, I would do it—and if I could save it by freeing some and I would do it—and if I could save it by freeing some and

I would do it—and if I could save it by freeing some and leaving others alone, I would also do that. What I do about slavery and the colored race, I do be-cause I believe it helps to save the Union, and what I for-bear, I forbear because I do not believe it would help to save the Union.

I shall do less whenever I shall believe what I am doing hurts the cause, and shall do more whenever I believe doing more will help the cause. I shall try to correct errors when shown to be crrors, and

I shall adopt new views so fast as they shall appear to be true views.

I have here stated my purpose according to my view of official duty, and I intend no modification of my off-expressed personal wish that all men everywhere could be free. Yours, A. LINCOLN.

# THE PRESIDENT'S RESPONSE TO A SERENADE. JULY, 1863.

FELLOW-CITIZENS: I am very glad indeed to see you to night, and yet I will not say I thank you, for this call; but I do most sincerely thank Almighty God for the occasion on which you have called. How long ago is it,—eighty odd years since on the Fourth of July, for the first time, in the history of the world, a nation, by its representatives, as-sembled and declared as a self-evident truth, "that all men on concidence." are created equal.<sup>2</sup> That was the birthday of the United States of America. Since then the Fourth of July has had several very peculiar recognitions. The two men most dis-tinguished in the framing and support of the Declaration

were Thomas Jefferson and John Adams—the ono having penned it, and the other sustained it the most forcibly in debate—the only two of the fifty-live who signed it, and were alected Presidents of the United States. Precisely fifty years after they put their hands to the paper, it pleased Almighty God to take both from this stage of action. This was indeed an extraordinary and remarkable event in our history. Another President, five years after, was called from this stage of existence on the same day and month of the year; and now on this last Fourth of July, just passed, when we have a gigantic rebellion, at the bottom of which is an effort to overthrow the purinciple that all men were created equal, we have the surrender of a most powerful position and army on that very day. And not only so, but in a succession of battles in Pennsylvania, near to us, through three days, so rapidly fought that they might be called one great battle, on the first, second, and third of the month of July; and on the fourth the cohorts of those who opposed the Declaration that all men are created equal, "turned tail" and run. [Long continued cheers.] Gentlebut I am not prepared to make one worthy of the occasion. I would like to speak in terms of praise due to the many fave officers and soldiers who have fought in the cause of the Union and liberties of their country from the beginning of the war. These are trying occasions, not only in snccess, but for the want of success. I dislike to mention the mame of one single officer, lest I might do wrong to those I might forget. Recent events bring up glorious names, and particularly prominent ones; but these I will not mention.

#### THE PRESIDENT'S LETTER TO THE ILLINOIS CON-VENTION.

EXECUTIVE MANSION, WASHINGTON, August 26, 1863.

#### HOD. JAMES C. CONKLING :

MY DEAR SIR: Your letter inviting me to attend a mass meeting of unconditional Union men, to be held at the capital of Illinois, on the 3d day of September, has been received. It would be very agreeable for me thus to meet my old friends at my own home: but I cannot just now

my old mends at my own home: but I cannot just how be absent from here so long as a visit there would require. The meeting is to be of all those who maintain unconditional devotion to the Union; and I am sure that my old political friends will thank me for tendering, as I do, the nation's gratitude to those other noble men whom no partisan malice or partisan hope can make false to the nation's life.

There are those who are dissatisfied with me. To such I would say: you desire peace, and you blame me that we do not have it. But how can we attain it? There are but three conceivable ways: First—to suppress the Rebellion by force of arms. This I am trying to do. Are you for it? If you are, so far we are agreed. If you are not for it, a second way is to give up the Union. I am against this. Are you for it? If you are, you should say so plainly. If you are not for force, nor yet for dissolution, there only remains some imaginable compromise. I do not believe that any compromise ambrecing the

I do not believe that any compromise embracing the maintenance of the Union is now possible. All that I learn leads to a directly opposite belief. The strength of the Rebellion is its military, its army. That army dominates all the country, and all the people within its range. Any offer of terms made by any man or men within that range, in opposition to that army, is simply nothing for the presant; because such man or men have no power whatever to enforce their side of a compromise, if one were made with them.

To illustrate: Suppose refugees from the South and peace men of the North get together in convention, and frame and proclaim a compromise embracing a restoration of the Union. In what way can that compromise be used to keep Lee's army out of Pennsylvania? Meade's army can keep Lee's army out of Pennsylvania, and, I think, can ultimately drive it out of existence. But no paper compromise to which the controllers of Lee's army are not agreed Gan at all affect that army. In an effort at such compromise we would waste time, which the enemy would improve to our disadvantage: and that would be all.

We would waste time, which the energy would many over a our disadvantage; and that would be all. A compromise, to be effective, must be made either with those who control the robol army, or with the people, first liberated from the domination of that army by the success of our own army. Now, allow me to assure you that no word or infimation from that rebel army, or from any of the men controlling it, in relation to any peace compromise, has ever come to my knowledge or belief. All charges and insincations to the contrary are deceptive and groundless. Aud promise you that if any such proposition shall bereafter come, it shall not be rejected and kept a secret from you. I freely acknowledge myself to be the servant of the

people, according to the bond of service, the United States Constitution; and that, as such, I am responsible to them. But, to be plain. You are dissatisfied with me about the

But, to be plain. You are dissatisfied with me about the negro. Quite likely there is a difference of opiniou between you and myself upon that subject. I certainly wish that all men could be free, while you, I suppose, do not. Yet, I have neither adopted nor proposed any measure which is not consistent with even your view, provided that you are for the Union. I suggested compensated emancipation; to which you replied you wished not to be taxed to buy negroes. But I had not asked you to be taxed to buy negroes. But I had not asked you to be taxed to buy negroese. Buch a way as to save you from greater taxation to save the Union exclusively by other means. You dislike the Emancipation Proclamation, and perhaps would have it retracted. You sayit is unconstitutional. I

You dislike the Emancipation Proclamation, and perhaps would have it retracted. You say it is unconstitutional. I think differently. I think the Constitution invests its Commander-in-Chief with the law of war in time of war. The most that can be said, if so much, is, that slaves are property. Is there, has there ever been, any question that by the law of war, property, both of enemies and friends, may be taken when needed ? And is it not needed whenever it helps us and hurts the enemy? Armies, the world over, destroy enemies' property when they cannot use it; and even destroy their own to keep it from the enemy. Civilized belligerents do all in their power to help themselves or hurt the enemy, except a few things regarded as barbarous or cruel. Among the exceptions are the massacre of vanquished foes and non-combatants, male and female.

But the Proclamation, as law, either is valid or is not valid. If it is not valid it needs no retraction. If it is valid it cannot be retracted, any more than the dead can be brought to life. Some of you profess to think its retraction would operate fivorably for the Union. Why better after the retraction than before the issue? There was more than a year and a balf of trial to suppress the rebellion before the Broclamation was issued, the last one hundred days of which passed under an explicit notice that it was coming, unless averted by those in revolt returning to their allegiance. The war has certainly progressed as favorably for us since the issue of the Proclamation as before.

I know as fully as one can know the opinion of others that some of the commanders of our armies in the field, who have given us our most important victories, believe the emancipation policy and the use of colored troops constitute the heaviest blows yet dealt to the rebellion, and that at<sup>#</sup> least one of those important successes could not have been achieved when it was but for the aid of black soldiers.

Among the commanders who hold these views are some who have never had an affinity with what is called "abolitionism," or with "Republican party politics," but who hold them purely as military opinions. I submit their opinions as entitled to some weight against the objections often urged that emancipation and arming the blacks are unwise as military measures, and were not adopted as such in good faith.

You say that you will not fight to free negroes. Some of them seem willing to fight for you; but no matter. Fight you, then, cxclusively, to save the Union. I issued the proclamation on purpose to aid you in saving the Union. Whenever you shall have conquered all resistance to the Union, if I shall urge you to continue fighting, it will be an apt time then for you to declare you will not fight to free negroes. I thought that in your struggle for the Union, to whatever extent the negroes should cease helping the enemy, to that extent it weakened the enemy in hisrosistance to you. Do you think differently? I thought whatever negroes can be got to do as soldiers, leaves just so much less for white soldiers to do in saving the Union. Does it appear otherwise to you? But negroes, like other people, act upon motives. Why should they do anything for us if we will do nothing for them? If they stake their lives for the promise of freedom. And the promise, being made, must be kept.

must be kept. The signs look better. The Father of Waters again goes unvexed to the sea. Thanks to the great Northwest for it; nor yet wholly to them. Three hundred miles up they met New England, Empire, Keystone, and Jersey, hewing their way right and left. The sunny South, too, in more colors than one, also lent a helping hand. On the spot, their part of the history was jotted down in black and white. The job, was a great national one, and let none be slighted who bore an honorable part in it. And while those who have cleared the great river may well be proud, even that is not all. It is hard to say that anything has been more bravely and well done than at Antietam, Murfreesboro, Gettysburg, and on may fields of less note. Nor must Uncle Sam's web feet be forgotten. At all the watery margins they and where resent, not only on the deep sa, the broad bay, and the rapid river, but also up the narrow, muddy bayou, and made their tracks. Thanks to all. For the great Republic-for the principle it lives by and keeps alive-for | man's vast future-thanks to all.

Peace does not appear so distant as it did. I hope it will come scot and come to stay; and so come as to be worth the keeping in all future time. It will then have been proved that among freemen there can be no successful ap-peal from the ballot to the bullet, and that they who take such appeal are sure to lose their case and pay the cost. And there will be some black men who can remember that with silent tongue, and clinched teeth, and steady eye, and well-poised bayonet, they have helped mankind on to this great consummation, while I fear there will be some white ones unable to forget that with malignant heart and deceitful speech they have striven to hinder it.

Still, let us not be over-sanguine of a speedy, final triumph. Let us be quite sober. Let us diligently apply the means, never doubling that a just God, in his own good time, will give us the rightful result. A. LINCOLN.

Yours, very truly,

THE PRESIDENT'S LETTER TO THE NORTH AMERICAN BEVIEW.

#### EXECUTIVE MANSION WASHINGTON, January 16, 1864.

MASHINGTON, January 10, 1304. MASHINGTON, January 10, 1304. GENTLEMEN: The number for this month and year of the North American Review was duly received, and for which please accept my thanks. Of course, I am not the most impartial judge, yet, with due allowance for this, I venture to hope that the article entitled "The President's Policy," will be of value to the country. I fear I am not quite worthy of all which is therein kindly said of me personally. The sentence of twelve lines, commencing at the ton of

The sentence of twelve lines, commencing at the top of page 252, I could wish to be not exactly as it is. In what is there expressed the writer has not correctly understood me. I have never had a theory that secession could absolve States or people from their obligations. Precisely the con-trary is asserted in the inaugural address; and it was be-cause of my belief in the continuation of these obligations that I was puzzled, for a time, as to denying the legal rights of those citizens who remained individually innocent of treason or rebellion. But I mean no more now than to merely call attention to this point. Yours respectfully, A. LINCOLN.

The sentence in the January number, refered to by Mr. Lincoln, is as follows :

Even so long ago as when Mr. Lincoln, not yet convinced of the danger and magnitude of the crisis, was endeavoring to persuade himself of Union majorities at the South, and to persuade nimseif of union majorities at the south, and to carry on a war that was half peace in hope of a peace that would have been all war, while he was still enforcing the fugitive slave law, under some theory that secession, how-ever it might absolve States from their obligations, could not escheat them of their claims under the Constitution, and that it probades in scheduling had a long one man. and that slaveholders in rebellion had alone, among mortals, the privilege of having their cake and eating it at the same time—the enemies of free government were striving to persuade the people that the war was an abolition cru-sade. To rebel without reason was proclaimed as one of the rights of man, while it was carefully kept out of sight that to suppress rebellion is the first duty of the Government.

To this the editors of the Review append a note, as follows:

Nothing could have been further from the intention of the editors than to misrepresent the opinions of the Presi-dent. They merely meaut that, in their judgment, the policy of the administration was at first such as practically to concele to any rehel who might choose to profess loyalty, rights under the Constitution whose corresponding obligations he repudiated.

#### THE PRESIDENT'S LETTER TO COLONEL HODGES OF KENTUCKY.

EXECUTIVE MANSION, WASHINGTON, April 4, 1864.

WASHINGTON, April 4, 1804. A. G. HODEES, Eeg., Frankfort, Ky: MY DEAR SIR: You ask me to put in writing the sub-stance of what I verbally said the other day, in your pres-ence, to Governor Bramlette and Senator Dixon. It was about as follows: "I am naturally anti-slavery. If slavery is not wrong,

"I am naturally anti-slavery. If slavery is not wrong, nothing is wrong. I cannot remember when I did not so think and feel, and yet I have never understood that the Presidency conferred upon me an unrestricted right to act officially upon this judgment and feeling. It was in the oath I took that I would to the best of my ability preserve, protect, and defend the Constitution of the United States.

I could not take the office without taking the oath. Nor was it my view that I might take an oath to get power, and was it my view that I might take an earn to get power, and break the oath in using the power. I understood, too, that in ordinary and civil administration this eath even forbade me to practically induge my primary abstract judgment on the moral question of slavory. I had publicly declared this many times, and in many ways. And I aver that, to this day, I have done no official act in mere deforence to my abday, I have done no official act in mere deforence to my ab-stract judgment and feeling on slavery. I did understand, however, that my oath to preserve the Constitution to the best of my ability, imposed upon me the duty of preserv-ing, by every indispensable means, that Government—that nation, of which that Constitution was the organic law. Was it possible to lose the nation and yet preserve the Con-tution? By general law, life and limb must be protected; yet often a limb must be amputated to save a life; but a life is never wisely given to save a limb. I felt that meas-ures, otherwise unconstitutional, might become lawful, by becoming indispensable to the measervation of the Constitubecoming indispensable to the preservation of the Constitu-tion, through the preservation of the nation. Right or wrong, I assumed this ground, and now avow it. I could not feel that, to the best of my ability I had even tried to preserve the Constitution, if, to save slavery, or any minor matter, I should permit the wreck of Government, country, and Constitution, altogether. When early in the war, Gen-eral Fremont attempted military emancipation, I forbade it, because I did not then think it an indispensable necess-sity. When a little later, General Cameron, then Secretary of War, suggested the arming of the blacks, I objected, be-cause I did not yet think it an indispensable necessity. When, still later, General Hunter attempted military eman-cipation, I again forbade it, because I did not yet think the indispensable necessity had come. When in March and May, and July, 1862, I made earnest and successive appeals to the border States to favor compensated emancipation, I believed the indispensable necessity for military emancipabecoming indispensable to the preservation of the Constitubelieved the indispensable necessity for military emancipabelieved the indispensable necessity for military emancipa-tion and arming the blacks would come, unless averted by that measure. They declined the proposition, and I was, in my best judgment, driven to the alternative of either sur-rendering the Union, and with it, the Constitution, or of laying strong hand upon the colored element. I chose the latter. In choosing it, I hoped for greater gain than loss, but of this I was not entirely confident. More than a year of trial now shows no loss by it in our foreign relations, none in our home popular sentiment, none in our white military force, no loss by it any how, or anywhere. On the contrary, it shows a gain of quite a hundred and thirty thousand sol-diers. seamen, and laborers. These are palable facts, abont diers, seamen, and laborers. These are palpable facts, about which, as facts, there can be no cavilling. We have the men; and we could not have had them without the measure.

"And now let any Union man who complains of this measure, test himself by writing down in one line, that he is for subduing the rebellion by force of arms; and in the next, that he is for taking three hundred and thirty thou-sand men from the Union side, and placing them where they would be best for the measure he condemns. If he cannot face his case so stated, it is only because he cannot face the truth."

I add a word which was not in the verbal conversation. In telling this tale, I attempt no compliment to my own In teiling this tale, I attempt no compliment to my own sagacity. I claim not to have controlled events, but confess plainly that events have controlled me. Now at the end of three years' struggle, the nation's condition is not what either party, or any man devised, or expected. God alone can claim it. Whither it is tending seems plain. If God now wills the removal of a great wrong, and wills diso that we of the North, as well as you of the South, shall pay fairly for our complicity in that wrong, impartial history will find therein new causes to attest and revere the justice and therein new causes to attest and revere the justice and goodness of God. Yours, trnly. Yours, truly,

A. LINCOLN.

# TO A NEW YORK MEETING.

EXECUTIVE MANSION, WASHINGTON, June 3, 1864.

Hon. F. A. CONKLING, and others:

GENTLEMEN: Your letter, inviting me to be present at a mass meeting of loyal citizens to be held at New York, on the 4th inst. for the purpose of expressing gratitude to Licutenant General Grant for his signal services, was received yesterday. It is impossible for me to attend. I approve, nevertheless, whatever may tend to strengthen and sustain General Grant and the noble armies now under his direction.

My previous high estimate of General Grant has been My previous high estimate of General Grant has been maintained and heightened by what has occurred in the remarkable campaign he is now conducting. While the magnitude and difficulty of the task before him do not prove less than I expected, he and his brave soldiers are now in the midst of their great trial, and I trust at your

# SPEECH OF MR. LINCOLN AT THE PHILADELPHIA FAIR, JUNE 16, 1864.

rain, some ic, foot. I suppose that this toast was intended to open the way for me to say something. War, at the best, is terrible, and this war of ours, in its magnitude and in its duration, is one of the most terrible. It has deranged business, totally in many localities, and partially in all localities. It has destroyed property and ruined homes; it has produced a national debt and taxation unprecedented, at least in this country; it has carried mourning to almost every home, until it can almost be said that the "heavens aro hung in black."

that the "heavens are hung in black." Yet the war continues, and several relieving coincidents have accompanied it from the very beginning which have not been known, as I understand, or have any knowledge of, in any former wars in the history of the world. The Sanitary Commission, with all its benevolent labors; the Obristian Commission, with all its Unstain and benevolent labors; and the various places, arrangements, so to speak, and institutions, have contributed to the comfort and relief of the soldiers. You have two of these places in this city— the Cooper Shop and Union Volunteer Refreshment Saloons. of the soldiers. You have two of these places in this diry-the Cooper Shop and Union Volunteer Refreshment Saloons. And, lastly, these Fairs, which, I believe, began only in last August, if I mistake not, in Chicago, then at Boston, at I is this: That Grant is this evening, with General I and almost tempted to make one. If I were to haz it is this: That Grant is this evening, with General I and almost tempted to make one. If I were to haz it is this: That Grant is this evening, with General I and almost tempted to make one. If I were to haz it is this: That Grant is this evening, with General I and almost tempted to make one. If I were to haz it is this: That Grant is this evening, with General I and Senter Hat General Hancock, and the brave officers and so with him, in a position form whence he will never loded until Richmond is taken, and I have but one proposition to put now, and, perhaps, I can best put it form of an interogative. If I shall discover that G greatly facilitated in their work by a sudden pouri ward of men and assistance, will you give them to the soldier hand of woman, or from any other source, it is much, very much. But I think that there is still that which is of as much value to him in the continual re-ho is yet remembered by the loved ones at home. Another view of these various institutions, if I may so call them, is worthy of consideration, I think. They are voluntary con-tributions, given zealously and earnestly, on top of all the disturbances of business, of all the disorders, of all the disturbances of business, of all the disorders, of all the

meeting you will so shape your good words that they may turn to men and guns, moving to his and their support. Yours, truly, A. LINCOLN. I taxation, and of all the burdens that the war has imposed upon us, giving proof that the national resources are not at all exhausted, and that the national spirit of patriotism is even firmer and stronger than at the commencement of the

It is a pertinent question, often asked in the mind pri-vately, and from one to the other, when is the war to end? Surely I feel as deep an interest in this question as any stream but I do not wish to name a day, a month, or a Surely I feel as deep an interest in this question as any other can, but I do not wish to name a day, a month, or a year when it is to end. I do not wish to run any risk of seeing the time come, without our being ready for the end, for fear of disappointment, because the time had come and not the end. We accepted this war for an object, a worthy object, and the war will end when that object is attained.

object, and the war will end when that object is attained. Under God, I hopeit never will end until that time. Speak-ing of the present campaign, General Grant is reported to have said, I am going through on this line if it takes all summer. This war has taken three years; it was begun or accepted upon the line of restoring the national authority over the whole national domain, and for the American peo-ple, as far as my knowledge enables me to speak, I say we are going through on this line if it takes three years more. My friends, I did not know but that I might be called upon to say a few words before I got away from here, but I did not know it was coming just here. I have never been in the habit of making predictions in regard to the war, but I am almost tempted to make one. If I were to hazard it, it is this: That Grant is this evening, with General Meade and General Hancock, and the brave 'officers and soldiers with him, in a position from whence he will never be dis-lodged until Richmond is taken, and I have but one single lodged until Richmond is taken, and I have but one single lodged until Kichmond is taken, and I have but one single proposition to put now, and, perhaps, I can best put it in the form of an interrogative. If I shall discover that General Grant and the noble officers and men under him can be greatly facilitated in their work by a sudden pouring for-ward of men and assistance, will you give them to me? Are you ready to march? [Gries of "yes."] Then, I say, stand ready, for I am watching for the chance. I thank you, gentlemen. [Langhter and cheers.]

This speech was repeatedly interrupted by applause, and at its close three cheers were given for the army of the Potomac, and successive cheers for Grant and Meade and Hancock, and their brave soldiers. In the mean

# OUR FOREIGN RELATIONS.

# The Trent Affair.

# SECRETARY SEWARD TO MR. ADAMS.

# November 30, 1861.

\* \* Since that conversation was held Captain Wilkes, in the steamer San Jacinto, has boarded a British who were proceeding to Europe on an errand of treason against their own country. This is a new incident, unknown to and unforescen, at least in its circumstances, by Lord Palmerston. It is to be met and disposed of by the two Governments, if possible, in the spirit to which I have adverted. Lord Lyons has prudently refrained from opening the subject to me, as, I presume, waiting instructions from home. We have done nothing on the subject to anticipate the discussion, and we have not furnished you with any explanations. We adhere to that course now, because we think it more prudent that the ground taken by the British Government should be first made known to us here, and that the discussion, if there must be one, shall be had here. It is proper, however, that you should know one fact in the case, without indicating that we attach much importance to it, namely, that, in the capture of Messrs. Mason and Slideli on board a British Possel, Captain Wilkes having acted without any instructions from the Government, the subject is therefore free from the embarrassment which might have resulted if the act had been specially directed by us. I trust that the British Government will consider the

I trust that the British Government will consider the subject in a friendly temper, and it may expect the best disposition on the part of this Government.

# EARL RUSSELL TO LORD LYONS.

#### NOVEMBER 30, 1861.

My LORD: Intelligence of a very grave nature has reached her Majesty's Governmeut.

This intelligence was conveyed officially to the knowledge of the Admiralty by Commander Williams, agent for mails on board the contract steamer Trent. It appears from the letter of Commander Williams, dated

It appears from the letter of Commander Williams, dated "Royal Mail Contract Packet Trent, at sea, November 9." that the Trent left Havana on the 7th instant, with her Majesty's mails for England, having on board numerous passengers. Commander Williams states that shortly after noon, on the 8th, a steamer having the appearance of a man-of-war, but not showing colors, was observed ahead. On nearing her, at 1.15 p. m., she fired a round shot from her pivotgun across the bors of the Trent and showed American colors. While the Trent was approaching her slowly, the American vessel discharged a shell across the bows of the Trent, exploding half a cable's length ahead of her. The Trent then stopped, and an officer with a large armed guard of marines boarded her. The officer demanded a list of the passengers; and, compliance with this demand being refused, the officer said he had orders to arrest Messrs. Mason, Slidell, McFarland, and Eustis, and that he hads ure information of their being passengers in the Trent. While stepped forward and told the American officer that the four persons he had ander were then standing bofore him. The commander of the Trent and Commander Williams protested fag. But the San Jacinto was at that time only two humdred yards from the Trent, her ship's company at quarters, her ports open, and tompions out. Resistance was therefore out of the question, and the four gentlemen before named were forcibly taken out of the ship. A further de-

mand was made that the commander of the Trent should proceed on board the San Jacinto, but he said he would not go unless forcibly compelled likewise, and this demand was not insisted upon.

It thus appears that certain individuals have been forcibly taken from on board a British vessel, the ship of a neutral Power, while such vessel was pursuing a lawful and innoent voyage—an act of violence which was an afront to the British flag and a violation of international law.

Her Majesty's Government, bearing in mind the friendly relations which have long subsisted between Great Britain and the United States, are willing to believe that the United States naval officer who committed the aggression was not acting in compliance with any authority from his Government, or that if he conceived himself to be so authorized he greatly misunderstood the instructions which he had received. For the Government of the United States must be fully aware that the British Government could not allow such an affront to the national honor to pass without full reparation, and her Majesty's Government are unwilling to believe that it could be the deliberate intention of the Government of the United States unnecessarily to force into discussion between the two Governments a question of so grave a character, and with regard to which the whold British nation would be sure to entertain such unanimity of feeling.

feeling. Her Majesty's Government, therefore, trust that when this matter shall have been brought under the consideration of the Government of the United States that Government will, of its own accord, offer to the British Government such redress as alone could satisfy the British nation, namely, the liberation of the four gentlemen and their delivery to your lordship, in order that they may again be placed under British protection, and a suitable apology for the aggression which has been committed.

Should these terms not be offered by Mr. Seward, you will propose them to him.

You are at liberty to read this dispatch to the Secretary of State, and, if he shall desire it, you will give him a copy of it.

#### SECRETARY SEWARD TO LORD LYONS. DECEMBER 26, 1861.

MY LORD: Earl Russell's despatch of November the 30th, a copy of which you have left with me at my request, is of the following effect, namely:

actopy of wing effect, namely: That a letter of Commander Williams, dated Royal Mail Contract Packetboat Trent, at sea, November 9th, states that that vessel left Havana on the 7th of November, with her Majesty's mails for England, having on board numerous passengers. Shortly after noon, on the 8th of November, the United States war steamer San Jacinto, Capt. Wilkes, not showing colors, was observed ahead. That steamer, on being neared by the Trent, at one o'clock fifteen minutes in the afternoon, fired a round shot from a pivot-gun across her bows, and showed American colors. While the Trent was approaching slowly towards the San Jacinto, she discharged a shell across the Trent's bows, which exploded at half a cable's length before her. The Trent then stopped, and an officer with a large armed guard of mairnes boarded her. The officer said he had orders to arrest Messrs. Mason, Slidell, McFarland, and Eustis, and had sure information that they were passengers in the Trent. While some parley was going on upon this matter, Mr. Slidell stepped forward and said to the American officer that the four persons be had name were stading before him. The commander of the Trent and Commander Williams protested against the act of taking those four passengers out of the Trent, they then being under the protection of the British flag. But

338

the San Jacinto was at this time only two hundred yards distant, her ship's company at quarters, her ports open and tompions out, and so resistance was out of the question. The four persons before named wore then forcibly question. The four persons before named were then forcibly taken out of the ship. A further demand was made that the commander of the Trent should proceed on board the San Jacinto, but he said he would not go unless forcibly compelled likewise, and this demand was not insisted upon. Upon this statement Earl Russell remarks that it thus appears that certain individuals have heen forcibly taken

from on board a British vessel, the ship of a neutral power, while that vessel was pursuing a lawful and inocent voyage-an act of violence which was an affront to the British flag and a violation of international law. Earl Russell next says that her Majesty's Government,

bearing in mind the friendly relations which have long subsisted between Great Britian and the United States, are willing to believe that the naval officer who committed this aggression was not acting in compliance with any authority from his Government, or that, if he conceived himself to be so authorized, he greatly misunderstood the instructions which he had received.

Earl Russell argues that the United States must be fully aware that the British Government could not allow such an affront to the national honor to pass without full reparaan already of the national holds to pass which the repart tion, and they are willing to believe that it could not be the deliberate intention of the Government of the United States unnecessarily to force into discussion between the two Governments a question of so grave a character, and with regard to which the whole British nation would be were to the track of the unconsisting of follows:

sure to entertain such unanimity of feeling. Earl Russell, resting upon the statement and the argu-ment which I have thus recited, closes with saying that her Majesty's government trust that when this matter shall have been brought under the consideration of the Government of the United States, it will, of its own accord, offer to the British Government such redress as alone could satto the British dovernment such redress as alone could sat-isfy the British nation, namely, the liberation of the four prisoners taken from the Trent, and their delivery to your lordship, in order that they may again be placed under British protection, and a suitable apology for the aggres-sion which has been committed. Earl Russell finally instructs you to propose those terms to me, if I should not first offer them on the part of the Government.

This despatch has been submitted to the President.

The British government has rightly conjectured, what it is now my duty to state, that Captain Wilkes, in conceivin and executing the proceeding in question, acted upon his own suggestions of duty, without any direction or in-struction, or even foreknowledge of it, on the part of this Government. No directions had been given to him, or any other naval officer, to arrest the four persons named, or any of them, on the Trent or any other British vessel, or on any other neutral vessel, at the place where it occurred or elsewhere. The British government will justly infer from these facts that the United States not only have had no purpose, but even no thought, of forcing into discussion the question which has arisen, or any other which could affect in any way the sensibilities of the British nation.

It is true that a round shot was fired by the San Jacinto from her pivot-gun when the Trent was distantly approaching. But, as the facts have been reported to this govern-ment, the shot was nevertheless intentionally fired in a direction so obviously divergent from the course of the Trent as to be quite as harmless as a blank shot, while it should be regarded as a signal.

So also we learn that the Trent was not approaching the San Jacinto slowly when the shell was fired across her bows, but, on the contrary, the Trent was, or seemed to be, moving under a full head of steam, as if with a purpose to

The second state of the state of state and a state of the the Trent. He stated us had used used is a tour capter which to search for the four persons named, in a respectful and courteous, though decided manner, and he asked the cap-tain of the Trent to show his passenger list, which was re-fused. The lieutenant, as we are informed, did not employ absolute force in transferring the passengers, but he used just so much as was necessary to satisfy the parties con-cerned that refusal or resistance would be unavailing. So, also, we are informed that the captain of the Trent was not at any time or in any way required to go on board the force the set of t

the San Jacinto.

These modifications of the case, as presented by Com-mander Williams, are based upon our official reports.

mander Williams, are based upon our official reports. I have now to remind your lordship of some tacts which the very proper and becoming motive of allowing them to be brought into the case, on the part of the United States, iu the way most satisfactory to this Government. These facts are, that at the time the transaction occurred an insurrection was existing in the United States which this Government

was engaged in suppressing by the employment of land and naval forces; that in regard to this domestic strife the United States considered Great Britain as a friendly power, while she had assumed for herself the attitude of a neuhad assumed the same attitude as Great Britain.

It had been settled by correspondence that the United States and Great Britain mutually recognized as applicable to this local strife these two articles of the declaration to this local strife these two articles of the declaration made by the Congress of Paris in 1856, namely, that the neutral or friendly flag should cover enemy's goods not contraband of war, and that usural goods not contraband of war are not liable to capture under an enemy's flag. These exceptions of contraband from favor were a negative acceptance by the parties of the rule hitherto everywhere recognized as a part of the law of nations, that whatever is contraband is liable to capture and confiscation in all cases.

James M. Mason and E. J. McFarland are citizens of the United States and residents of Virginia. John Slidell and George Eustis are citizens of the United States and residents of Louisiana. It was well known at Havana when these parties embarked in the Trent that James M. Mason was proceeding to England in the affected character of a minister plenipotentiary to the Court of St. James, under a pretended commission from Jefferson Davis, who had assumed to be president of the insurrectionary party in the Uni-ted States, and E.J. McFarland was going with him in a like unreal character of secretary of legation to the pretended mission. John Slidell, in similar circumstances, was going to Paris as a pretended minister to the Emperor of the French, and George Eustis was the chosen secretary of legation for that simulated mission. The fact that these persons had assumed such characters has been since avowed by the same Jefferson Davis in a pretended message to an unlawful and insurrectionary Congress. It was, we think, rightly pre-sumed that those ministers bore pretended credentials and instructions, and such papers are in the law known as despatches. We are informed by our consul at Paris that these despatches, having escaped the search of the Trent, were actually conveyed and delivered to emissaries of the insurrection in England. Although it is not essential, yet it is proper to state, as I do also upon information and belief, hope to state as the set upon information and benet, including Commander Williams, had knowledge of the as-sumed characters and purposes of the persons before named when they embarked on that vessel.

Your lordship will now perceive that the case before us, instead of presenting a merely flagrant act of violence on the part of Captain Wilkes, as might well be inferred from the incomplete statement of it that went up to the British the incomplete statement of it that went up to the British government, was undertaken as a simple legal and cus-tomary belligerent proceeding by Captain Wilkes to arrest and capture a neutral vessel engaged in carrying contra-band of war for the use and benefit of the insurgents. The question before us is, whether this proceeding was authorized by and conducted according to the law of na-tions. It involves the following inquiries: Let Wore the pursue named at the insurgent descathes

tions. It involves the following inquiries: Ist. Were the persons named and their supposed despatches

2d. Might Capitain Wilkes lawfully stop and search the Tront for these contraband persons and despatches? 3d. Did he exercise that right in a lawful and proper

manner?

4th. Having found the contraband persons on board and in presumed possession of the contraband dispatches, had he a right to capture the persons?

5th. Did he exercise that right of capture in the manner allowed and recognized by the laws of nations? If all these inquires shall be resolved in the affirmative

the British government will have no claim for reparation.

I address myself to the first inquiry, namely, were the four persons mentioned, and their supposed despatches, contraband?

Maritime law so generally deals, as its professors say, rem, that is with property, and so seldom with persons, that it seems a straining of the term contraband to apply it to them. But persons, as well as property, may become con-traband, since the word means broadly "contrary to proc-lamation, prohibited, illegal, unlawful."

All writers and judges pronounce naval or military per-sons in the service of the enemy contraband. Vattel says war allows us to cut off from an enemy all his resources, and to hinder him from sending ministers to solicit assist-And Sir William Scott says you may stop the ambasance. sador of your enemy on his passage. Despatches are not less clearly contraband, and the bearers or couriers who un-dertake to carry them fall under the same condem-

William Scott, speaking of civil magistrates who are arrested and detained as contraband, says: "It appears to me on principle to be but reasonable that

when it is of sufficient importance to the enemy that such persons shall be sent out on the public service at the pub-lic expense, it should afford equal ground of forfeiture against the vessel that may be let out for a purpose so inti-mately connected with the hostile operations." I trust that I have shown that the four persons who were then from the Grant by Contin Willie and the index

taken from the Trent by Captain Wilkes, and their des-

taken from the frent by Captain Wilkes, and their dee-patches, were contraband of war. The second inquiry is, whether Captain Wilkes had a right by the law of nations to detain and search the Trent, The Trent, though she carried mails, was a contract or merchant vessel—a common carrier for hire. Maritime law knows only three classes of vessels-vessels of war, revenue vessels, and merchant vessels. The Trent falls within the latter class. Whatever disputes have existed concerning a right of visitation or search in time of peace, none, it is supposed, has existed in modern times about the right of a belligerent in time of war to capture contraband in neutral and even friendly merchant vessels, and of the right of visitation and search, in order to determine whether they are nentral, and are documented as such according to the law of nations.

I assume in the present case what, as I read British au-thorities, is regarded by Great Britain herself as true maritime law: That the circumstances that the Trent was protime law: That the circumstances that the Trent was pro-ceeding from a neutral port to another neutral port does not modify the right of the belligerent captor. The third question is whether Captain Wilkes exercised the right of search in a lawful and proper manner. If any doubt hung over this point, as the case was pre-sented in the statement of it adopted by the British Gor-erment think is much once already used annuk here

ernment, I think it must have already passed away before the modifications of that statement which I have already submitted.

I proceed to the fourth inquiry, namely: Having found the suspected contraband of war on board the Trent, had Captain Wilkes a right to capture the same? Such a capture is the chief, if not the only recognized, object of the permitted visitation and search. The principle

or the law is, that the belligerent exposed to danger may prevent the contraband persons or things from applying themselves or being applied to the hostile uses or purposes designed. The law is so very liberal in this respect that when contraband is found on board a neutral vessel not only is the contraband forfeited, but the vessel which is the vehicle of its passage or transportation, being tainted, also becomes contraband, and is subjected to capture and confiscation.

Only the fifth question remains, namely: Did Captain Wilkes exercise the right of capturing the contraband in conformity with the law of nations?

It is just here that the difficulties of the case begin. It is just here that the difficulties of the case begin. What is the manner which the law of nations prescribes for disposing of the contraband when you have found and seized it on board of the neutral vessel? The answer would be easily found if the question were what you shall do with the contraband vessel. You must take or send her into a convenient port, and subject her to a judicial prosecution there in admirality, which will try and decide the questions of bellicemeng neutrality contraband and canture. So of belligerency, neutrality, contraband, and capture. So, again, you would promptly find the same answer if the questhen were, What is the manner of proceeding prescribed by the law of nations in regard to the contraband, if it be prop-erty or things of material or pecuniary value?

But the question here concerns the mode of procedure in regard, not to the vessel that was carrying the contraband, nor yet to contraband things which worked the forfeiture of the vessel, but to contraband persona. The books of law are dumb. Yet the question is as im-portant as it is difficult. First, the belligerent captor has a

right to prevent the contraband officer, soldier, sailor, minister, messenger, or courier from proceeding in his unlawful voyage and reaching the destined scene of his injurious service. But, on the other hand, the person captured may be innocent—that is, he may not be contraband. He, therefore, has a right to a fair trial of the accusation against him. The neutral State that has taken him under its flag is bound to react this if he is not contraband and it is the bound to protect him if he is not contraband, and is therefore entitled to be satisfied upon that important question. The faith of that State is pledged to his safety, if innocent, as its justice is pledged for his surrender if he is really conas its justice is pleaged for his surrender if he is really con-traband. Here are conflicting claims, involving personal liberty, life, honor, and duty. Here are conflicting national claims, involving welfare, safety, honor, and empire. They require a tribunal and a trial. The captors and the eap-tured are equals; the neutral and the belligerent State are equals. • While the law authorities were found silent, it was sug

only courts of admiralty have jurisdiction in maritime cases, and these courts have formulas to try only claims to contraband chattels, but none to try claims concerning contraband persons. The courts can entertain no proceed inge and render no judgment in favor of or against the al-

leged contraband me. It was replied all this was true; but you can reach in those courts a decision which will have the moral weight of a judicial one by a circuitous proceeding. Convey the of a judicial one by a circuitous proceeding. Convey the suspected men, together with the suspected vessel, into port, and try there the question whether the vessel is con-traband. You can prove it to be so by proving the suspect-ed men to be contraband, and the court must then determ-ine the vessel to be contraband. If the men are not con-traband the vessel to be contraband. The the vessel to be contraband. If the men are not con-traband the vessel will secape condemnation. Still, there is no judgment for or against the captured persons. But it was assumed that there would result from the determina-tion of the coart concerning the vessel a legal certainty concerning the character of the men. This course of proceeding seemed open to many objec-tions. It elevates the incidental inforior private interest tota, the proper place of the mein normeunt sublicant

tions. It elevates the incidental inferior private interest into the proper place of the main paramount public one, and possibly it may make the fortunes, the safety, or the existence of a nation depend on the accidents of a merely personal and pecuniary litigation. Moreover, when the judgment of the prize court upon the lawfulness of the cap-ture of the vessel is rendered, it really concludes nothing, and binds neither the beligerent State nor the neutral upon the great question of the disposition to be mede of upon the great question of the disposition to be made of the captured contraband persons. That question is still to be really determined, if at all, by diplomatic arrangement

Or by war. One may well express his surprise when told that the law of nations has furnished no more reasonable, practical, and perfect mode than this of determining questions of such grave import between sovereign powers. The regret we may feel on the occasion is nevertheless modified by the reflection that the difficulty is not altogether anomalous. Similar and equal deficiencies are found in every system of municipal law, especially in the system which exists in the greater portions of Great Britain and the United States. The title to personal property can hardly ever be resolved by a the to personal property can hardly ever be resolved by a court without resorting to the foction that the claimant has lost and the possessor has found it, and the title to real es-tate is disputed by real litigants under the names of imagi-nary persons. It must be confessed, however, that while all aggrieved nations demand, and all impartial once con-cede, the need of some form of judicial process in determinting the characters of contraband persons, no other form than the illogical and circuitous one thus described exists, nor has any other yet been suggested. Practically, there-fore, the choice is between that judicial remedy or no judi-

cial remedy whatever. If there be no judicial remedy, the result is that the ques-tion must be determined by the captor himself, on the deck tion must be determined by the captor himself, on the deck of the prize vessel. Very grave objections arise against such a course. The captor is armed, the neutral is unarmed. The captor is interested, prejudiced, and perhaps violent; the neutral, if truly neutral, is disinterested, subdued, and helpless. The tribunal is irresponsible, while its judgment is carried into instant execution. The captured party is compelled to submit, though bound by no legal, moral, or treaty obligation to acquiesce. Reparation is distant and problematical, and depends at last on the justice, magnan-imity, or weakness of the State in whose behalf and by whose authority the capture was made. Out of these dis-putes reprisals and wars necessarily arise, and these are so putes reprisals and wars necessarily arise, and these are so frequent and destructive that it may well be doubted whether this form of remedy is not a greater social evil than all that could follow if the belligerent right of search were universally renounced and abolished forever. But carry the case one step further. What if the State that has made the capture unreasonably refuse to hear the complaint of the

capture unreasonably refuse to hear the complaint of the neutral or to redress it? In that case, the very act of cap-ture would be an act of war-of war begun without notice, and possibly entirely without provocation. I think all unprejudiced minds will agree that, imperfect as the existing judical remedy may be supposed to be, it would be, as a general practice, better to follow it than to adopt the summary one of leaving the decision with the captor, and relying upon diplomatic debates to review his decision. Practically, it is a question of choice between law, with its imperfections and delays, and war, with its evils and desolations. Nor is it ever to be forgotten that neutral-ity, honestly and justly preserved, is always the harbinger of peace, and therefore is the common interest of nations, which is only saying that it is the interest of humanity itself. itself.

At the same time it is not to be denied that it may sometimes happen that the judicial remedy will become impos-sible, as by the shipwreck of the prize vessel, or other cirtake the captured persons into a convenient port, and insti-take the captured persons into a convenient port, and insti-take into port for confiscation. In such a case the right of tute judicial proceedings there to try the controversy. But the captor to the custody of the captured persons, and to dispose of them, if they are really contraband, so as to de-feat their unlawful purposes, cannot reasonably be denied. What rule shall be applied in such a case? Clearly, the captor ought to be required to show that the failure of the indicial remedy results from circumstances beyond his con-trol, and without his fault. Otherwise, he would be allowed to derive advantage from a wrongful act of his own.

In the present case, Captain Wilkes, after capturing the In the present case, Captain whites, after capturing the contraband persons and making prize of the Trent in what seems to be a perfectly lawful manner, instead of sending her into port, released her from the capture, and permitted her to proceed with her whole cargo upon her voyage. He thus effectually prevented the judicial examination which might otherwise here occurred

might otherwise have occurred. If, now, the capture of the contraband persons and the capture of the contraband vessel are to be regarded, not as wo separate or distinct transactions under the law of na tions, but as one transaction, one capture only, then it follows that the capture in this case was left unfinished, or was abandoned. Whether the United States have a right to retain the chief public benefits of it, namely, the custody of to retain the chief public benefits of it, hamely, the custody of the captured persons on proving them to be contraband, will depend upon the preliminary question whether the leaving of the transaction unfinished was necessary, or whether it was nunccessary, and therefore voluntary. If it was necessary, **Great** Britain, as we suppose, must, of course, waive the de-fect, and the consequent failure of the judicial remedy. On the other hand, it is not seen how the United States can in-sist unon ber waiver of that indicial, remedy if the defect sist upon her waiver of that judicial remedy, if the defect of the capture resulted from an act of Captain Wilkes,

which would be a fault on their own side. Captain Wilkes has presented to this Government his reasons for releasing the Trent. "I forehore to seize her;" he says, "in consequence of my being so reduced in officers and crew, and the derangement it would cause innocent and crew, and the derangement it would cause innocent persons, there being a large number of passengers who would have been put to great loss and inconvenience, as well as disappointment, from the interruption it would have caused them in not being able to join the steamer from St. Thomas to Europe. I therefore concluded to sacrifice the Interest of my officers and crew in the prize, and suffered her to proceed after the detention necessary to effect the transfer of those commissioners, considering I had obtained the important end I had in view, and which affected the interest of our country and interrupted the action of that of the confederates."

I shall consider, first, how these reasons ought to affect the action of this Government; and secondly, how they ought to be expected to affect the action of Great Britain.

The reasons are satisfactory to this Government, so far as Captain Wilkes is concerned. It could not desire that the San Jacinto, her officers and crew, should be exposed to danger and loss by weakening their number to detach a prize crew to go on board the Trent. Still less could it disavow the humane motive of preventing inconveniences, losses, and perhaps disasters, to the several hundred innolosses, and perhaps disasters, to the several hundred inno-cent passengers found on board the prize vessel. Nor could this Government perceive any ground for questioning the fact that these reasons, though apparently incongruous, did operate in the mind of Captain Wilkes and determine him to release the Trent. Human actions generally proceed upon mingled, and sometimes conflicting motives. He measured the sacrifices which this decision would cost. It munifiestly, however, did not come to his the the trend at the manifestly, however, did not occur to him that beyond the sacrifice of the private interests (as he calls them) of his officers and crew, there might also possibly be a sacrifice even of the chief and public object of his capture, namely, the right of his Government to the custody and disposition of the captured persons. This Government cannot consure of the captured persons. This Government cannot censure him for this oversight. It confessess that the whole subject came unforeseen upon the Government, as doubtless it did upon him. Its present convictions on the point in question are the result of deliberate examination and deduction now

wade, and not of any impressions previously formed. Nevertheless, the question now is, not whether Captain Wilkes is justified to his Government in what he did, but what is the present view of the Government as to the effect of what he has done. Assuming now, for argument's sake only, that the release of the Trent, if voluntary, involved a waiver of the claim of the Government to hold the capa waiver of the claim of the Government to hold the cap-tured persons, the United States could in that case have no hesitation in saying that the act which has thus already been approved by the Government must be allowed to draw its legal consequence after it. It is of the very na-ture of a gift or a charity that the giver cannot, after the **exercise** of his benevolence is past, recall or modify its heavier. benefits.

We are thus brought directly to the question whether we are entitled to regard the release of the Trent as involuntary, or whether we are obliged to onsider that it was vol-untary. Clearly the release would have been involuntary had it been made solely upon the first ground assigned for it by Captain Wilkes, namely, a want of a sufficient force to send the prize vessel into port for adjudication. It is not the duty of a captor to hazard his own vessel in order to secure a judicial examination to the captured party. No large prize crew, however is legally necessary, for it is the duty of the captured party to acquiesce, and go will-ingly before the tribunal to whose jurisdiction it appeals. If the captured party indicate purposes to employ means of resistance which the captor cannot with probable safety to himself overcome, he may properly leave the vessel to go forward; and neither she nor the State she represents can ever afterwards justly object that the captor deprived her

ever afterwards justly object that the captor deprived her of the judicial remedy to which she was entitled. But the second reason assigned by Captain Wilkes for re-leasing the Trent differs from the first. At best, therefore, it must be held that Captain Wilkes, as he explains him-self, acted from combined sentiments of prudence and gen-erosity, and so that the release of the prize vessel was not strictly necessary or involuntary. Secondly. How ought we to expect these explanations by Contain Wilkes of this reasons for leaying the canture in-

Secondly. How ought we to expect mess explanations by Captain Wilkes of his reasons for leaving the capture in-complete to affect the action of the British government? The observation upon this point which first occurs is, that Captain Wilkes's explanations were not made to the author-ities of the captured vessel. If made known to them, they might have approved and taken the release upon the condition of waivings indicidi ursatingtion of the whole condition of waiving a judicial investigation of the whole transaction, or they might have refused to accept the re-lease upon that condition.

But the case is one not with them, but with the British government. If we claim that Great Britain ought not to insist that a judicial trial has been lost because we volun-Insist that a junch that has been lost because we volumi-tarily released the offending vessel out of consideration for her innocent passengers. I do not see how she is to be bound to acquiesce in the decision which was thus made by ns without necessity on our part, and without knowledge of conditions or consent on her own. The question between Great Britain and ourselves thus stated would be a ques-tion not of right and of law, but of favor to be conceded by then not of right and of raw, but of rayer to be concered by her to us in return for favors shown by us to her, of the value of which favors on both sides we ourselves shall be the judge. Of course the United States could have no thought of raising such a question in any case. I trust that I have shown to the satisfaction of the British

Government, by a very simple and natural statement of the facts, and analysis of the law applicable to them, that this Government has neither meditated, nor practiced, nor approved any deliberate wrong in the transaction to which what has happened has been simply an inadvertency, that with a has happened has been simply an inadvertency, con-sisting in a departure, by the naval officer, free from any wrongful motive, from a rule uncertainly established, and probably by the several parties concerned either imperfectly understood or entirely unknown. For this error the British Government has a right to expect the same reparation that we, as an independent State, should expect from Great Britain or from any other friendly nation in a similar case.

I have not been unaware that, in examining this question, have fallen into an argument for what seems to be the British side of it against my own country. But I am re-lieved from all embarrassment on that subject. I had hardly fallen into that line of argument when I discovered that I was really defending and maintaining, not an exclusively British interest, but an old, honored, and cherished American cause, not upon British authorities, but upon principles that constitute a large portion of the distinctive policy by which the United States have developed the resources of a which the United States have developed the resources of a continent, and thus becoming a considerable maritime power, have won the respect and confidence of many ma-tions. These principles were laid down for us in 1804, by James Madison, when Secretary of State, in the administra-tion of Thomas Jefferson, in instructions given to James Monroe, our Minister to England. Although the case before him concerned a description of persons different from those who are incidentally the subjects of the present discussion, the ground he assumed then was the same I now occupy, and the arguments by which he sustained himself upon it

and the arguments by which he sustained himself upon it have been an inspiration to me in preparing this reply. "Whenever," he says, "property found in a neutral vessel is supposed to be liable on any ground to capture and con-demnation, the rule in all cases is, that the question shall not be decided by the captor, but be carried before a legal tribunal, where a regular trial may be had, and where the captor himself is liable to damages for an abuse of his power. Can it be reasonable, then, or just, that a beligerent commander who is thus restricted, and thus responsible in a case of mere proparity of thrisi a prount should be negative. a case of mere property of trivial amount, should be permitted, without recurring to any tribunal whatever, to examine the crew of a neutral vessel, to decide the important question of their respective allegiances, and to carry that decision into execution by forcing every individual he may choose into a service abborrent to his feelings, cutting him off from his most tender connexions, exposing his mind and his person to the most humilating discipline, and his life itself to the greatest danger. Reason, justice, and humanity units in matteria protesting against so extravagant a proceeding.

If I decide this case in favor of my own Government, I must disavow its most cherished principles, and reverse and forever abandon its essential policy. The country cannot afford the sacrifice. If I maintain those principles, and adhere to that policy, I must surrender the case itself. It will be seen, therefore, that this Government could not deny the justice of the claim presented to us in this respect upon its merits. We are asked to do the British nation just what we have always insisted all nations ought to do to us.

The claim of the British government is not made in a discourteeus manner. This Government, since its first organization, has never used more guarded language in a similar case.

In coming to my conclusion I have not forgotten that, if the safety of this Union required the detention of the captured persons, it would be the right and duty of this Government to detain them. But the effectual check and waning proportions of the existing insurrection, as well as the comparative unimportance of the captured persons themselves, when dispassionately weighed, happily forbid me from resorting to that delence.

Nor am I unaware that American citizens are not in any case to be unnecessarily surrendered for any purpose into the keeping of a foreign State. Only the captured persons, however, or others who are interested in them, could justly raise a question on that ground.

Nor have I been tempted at all by the suggestions that cases might be found in history where Great Brittan refused to yield to other nations, and even to curselves, claims like that which is now before us. Those cases occurred when Great Britain, as well as the United States, was the home of generations, which, with all their peculiar interests and passions, have passed away. She could in no other way so offectually disavow any such injury as we think she does by assuming now as her own the ground upon which we then stood. It would tell little for our own claims to the character of a just and magnanimous people if we should so far consent to be guided by the law of retaliation as to lift up buried injuries from their graves to oppose against what national consistency and the national conscience compel us to regard as a claim intrinsically right.

Pufting behind me all suggestions of this kind, I prefer to express my satisfaction that, by the adjustment of the present case upon principles confessedly American, aud yct, as I trust, mutually satisfactory to both of the nations concerned, a question is finally and rightly settled between them, which, heretofore exhausting not only all forms of peaceful discussion, but also the arbitrament of war itself, for more than a half a century alienated the two countries from each other, and perplexed with fears and apprehensions all other nations.

The four persons in question are now held in military custody at Fort Warren, in the State of Massachusetts. They will be cheerfully liberated. Your lordship will please indicate a time and place for receiving them.

#### EARL RUSSELL TO LORD LYONS. JANUARY 10, 1862.

Mr LORD: In my despatch to you of the 30th of November, after informing you of the circumstarces which had occurred in relation to the capture of the four persons taken from on board the Trent, I stated to you that it thus appeared that certain individuals had been forcibly taken from on board a British vessel, the ship of a neutral power, while such vessel was pursuing a lawful and innocent voyage; an act of violence which was an affront to the British flag; and a violation of international law. I concluded by directing you, in case the reparation which her Majesty's government expected to receive should not be offered by Mr. Seward, to propose to that minister to make such redress as alone would satisfy the British nation, namely : first, the liberation of the four gentlemen taken from on board the Trent, and their delivery to your lordship, in order that they might again be placed under British protection; and, secondly, a suitable apology for the aggression which had been committed.

I received yesterday your lordship's despatch of the 27th ultimo, enclosing a note to you from Mr. Seward, which is in substance the answer to my despatch of the 30th of November.

Proceeding at once to the main points in discussion between us, her Majesty's government have carefully examined how far Mr. Seward's note, and the conductitannounces, comply substantially with the two proposals I have recited.

With regard to the first, viz: the liberation of the prisoners with a view to their being again placed under British protection, I find that the note concludes by stating that the prisoners will be cheerfully liberated, and by calling upon your lordship to indicate a time and place for receiving them. No condition of any kind is coupled with the liberation of the prisoners.

With regard to the suitable apology which the British government had a right to expect, I find that the Government of the United States distinctly and unequivocally declares that no directions had been given to Captain Wilkes, or to any other naval officer, to arrest the four persons named or any of them, on the Trent, or on any other British vessel, or on any other neutral vessel, at the place where it occurred or elsewhere.

Where it occurred or elsewhere. I find further that the Secretary of State expressly forbears to justify the particular act of which her Majesty's government complained. If the United States Government had alleged that, although Captain Wilkes had no previous instructions for that purpose, he was right in capturing the persons of the four prisoners, and in removing them from the Trent on board his own vessel, to be afterwards carried into a port of the United States, the Government which had thus sanctioned the proceeding of Captain Wilkes would have become responsible for the original violence and insults of the act. But Mr. Seward contents himself with stating that what has happened has been simply an inadvertoncy, consisting in a departure by a naval officer, free from any wrongful motive, from a rule uncertainly established, and probably by the several partice concerned either imperfectly understood or entirely unknown. The Secretary of State goes on to affirm that for this error the British which the United States as an independent State should expect from Great Britain or from any other friendly nation in a similar case.

Her Majesty's government having carefully taken into their consideration the liberation of the prisoners, the delivery of them into your hands, and the explanations to which I have just referred, have arrived at the conclusion that they constitute the reparation which her Majesty and the British nation has a right to expect.

It gives her Majesty's government great satisfaction to be enabled to arrive at a conclusion favorable to the maintenance of the most friendly relations between the two nations. I need not discuss the modifications in my statement of facts which Mr. Seward says he has derived from the reports of officers of his Government. I cannot conclude, however, without adverting shortly to the discussions which Mr. Seward has raised upon points not prominently brought into question in my despatch of the 30th of November. I there objected, on the part of her Majesty's government, to that which Captain Wilkes had done. Mr. Seward, in his answer, points out what he conceives Captain Wilkes might have done without violating the law of nations.

It is not necessary that I should here discuss in detail the five questions ably argued by the Secretary of State. But it is necessary that I should say that her Majesty's government differ from Mr. Seward in some of the conclusions at which he has arrived; and it may lead to a better understanding between the two nations on several points of international law which may, during the present contest, or at some future time be brought into question, that I should state to you, for communication to the Secretary of State, wherein those differences consist. I hope to do so in a few days.

days. In the meantime, it will be desirable that the commanders of the United States cruisers should be instructed not to repeat acts for which the British government will have to ask for redress, and which the United States Government cannot undertake to justify.

You will read and give a copy of this despatch to the Secretary of State.

#### COMMENT OF THE LONDON TIMES.

The London *Times* of January 11, 1863, commenting on the proper reception to be given these visitors, remarked :

So we do sincerely hope that our conntrymen will not give these fellows anything in the shape of an ovation-The civility that is due to a foe in distress is all that they can claim. We have returned them good for evil, and, sooth to say, we should be exceedingly sorry that they should ever be in a situation to choose what return they will make for the good we have now done them. They are here for their own interest, in order, if possible, to drag us into their own quarrel, and, but for the unpleasant contingencies of a prison, rather disappointed perhaps that their detention has not provoked a new war. When they stepped on board the Trent they did not trouble themselves with the thought of the mischief they might be doing an unoffending neutral; and if now, by any less perilous device, they could entangle us in the war, no doubt they would be only too happy. We trust there is no chance of their doing; this, for, impartial as the British public is in the matter, it certainly has no prejudice in favor of slavery, which, if anything, these gentlemen represent. What they and their secretaries are to do here passes our conjecture. They are personally nothing to us. They must not suppose, because

we have gone to the very verge of a great war to rescue them, that therefore they are precious in our eyes. We should have done just as much to rescue two of their own should have done just as much to rescue two of their own negroos; and, had that been the object of the rescue, the swarthy Pompcy and Czesar would have had just the same right to triumphal arches and municipal addresses as Messrs. Mason and Slidell. So, please, British public, let's have none of these things. Let the commissioners come up quietly to town, and have their say with anybody who may have time to listen to them. For our part, we cannot see how anything they have to tell can turn the scale of British duty and deliberation. There have been so many cases of peoples and nations establishing an actual independence, and compelling the recognition of the world, that all we have to do is what we have done before, up to the very last year. This is now a simple matter of precedent. Our statesmen and lawyers know quite as much on the subject as Messrs. Mason and Slidell, and are in no need of their information or advice. information or advice.

# ACTION OF CONGRESS. Second Session, Thirty-Seventh Congress. IN HOUSE.

1861, December 2-Mr. Lovejov, by unanimous consent, offered this joint resolution :

That the thanks of Congess are due, and are hereby ten-dered, to Captain Wilkes of the United States Navy, for his brave, adroit, and patriotic conduct in the arrest and detention of the traitors James M. Mason and John Slidell.

Mr. EDGERTON moved this as a substitute:

That the President of the United States be requested to present to Captain Charles Wilkes a gold medal, with suit-able emblems and devices, in testimony of the high sense entertained by Congress of his good conduct in promptly arresting the rebel ambassadors James M. Mason and John Slidell.

Which was agreed to; and the resolution passed.

1862, February 19-The Senate indefinitely postponed it.

1861, December 16-Mr. VALLANDIGHAM offered this resolution :

Whereas the Secretary of the Navy has reported to this House that Captain Charles Wilkes, in command of the Sau Jacinto, an armed public vessel of the United States, did, on the 8th of November, 1861, on the high seas, intercept the Trent, a British mail steamer, and forcibly remove therefrom James M. Mason and John Slidell—"disloyal citizens, leading conspirators, rebel enemies, and dangerous men"—who, with their suite, were on their way to Europe "to promote the cause of the insurrection," claiming to be ambassadors from the so-called Confederate States; and whereas the Secretary of the Navy has further reported to this House, that " the prompt and decisive action of Captain Wilkes on this occasion merited and received the emphatic approval of the Department," and moreover, in a public letter has thanked Captain Wilkes for the act; and whereas this House, on the first day of the session, did propose to tender the thanks of Congress to Captain Wilkes, for his "brave, adroit, and patriotic conduct in the arrest and de-tention of the traitors James M. Mason and John Sliddl;" and whereas further, on the same day, this House did re-quest the President to confine the said James M. Mason and John Slidell in the cells of convicted felons until certain military officers of the United States, captured and held by the so-called Confederate States, should be treated as pris-pages of ward Theorefore oners of war: Therefore,

Be it resolved, (as the sense of this House.) That it is the duty of the President to now firmly maintain the stand thus taken, approving and adopting the act of Captain Wilkes, in spite of any menace or demand of the British government; and that this House pledges its full support to him in upholding now the honor and vindicating the cour-age of the Government and people of the United States against a foreign power.

Which, on motion of Mr. FENTON, was referred to the Committee on Foreign Affairsyeas 109, nays 16, as follows :

YEAS TOO, Bays TO, as FOROWS.
YEAS—MESSTS. Aldrich, Alley, Arnold, Babbitt, Joseph Baily, Baker, Baxter, Beaman, Biddle, Bingham, Francis P. Blair, Jacob B. Blair, Samuel S. Blair, Blake, William G. Brown, Buffinton, Burnham, Calvert, Chamberlain, Clark, Cobb, Colfax, Roscoe Conkling, Cooper, Covode, Davis, Delano, Diven, Duell, Dunn, Edwards, Eliot, Eng-ush, Fenton, Fossenden, Franchot, Frank, Gooch, Granger,

Grider, Gnrley, Hale, Harding, Harrison, Hickman, Hutch-ins, Julian, Francis W. Kellogg, William Kellogg, Knapp, Law, Lazear, Leary, Lehman, Loomis, Lovejoy, McKnight, McPherson, Mallory, Maynard, Menzies, Mitchell, Moorhead, Justin S. Morrill, Noell, Oddil. Olin, Patton, Perry, T. G. Phelps, Pike, Pomeroy, Porter, Alexander II. Rice, John H. Rice, Richardson, Edward H. Rollins, Sargent, Sedg-wick, Shanks, Shefield, Shellabarger, Sherman, Smith, Spaulding, William G. Skele, Stevens, Benjamin F. Thom-as, Francis Thomas, Train, Trimble, Trowbridge, Van Horn, Verree, Wadsworth, Wall, E. P. Walton, Ward, Wash-burn, Wheeler, Whaley, Albert S. White, Wickliffe, Wil-son, Windom, Woodruff, Worcester, Wright-109. NATS-Messrs, Allen, George H. Browne, Frederick A. Conkling, Cox, Cravens, Haight, Holman, Morris, Noble, Nugen, Pendleton, Skiel, John B. Steele, Vallandigham, Vandever, Chilton A. White-16.

# Monarchical Intrigues in Central and South America.

1864, March 15-Mr. McDougall offered this resolution, which was adopted :

Resolved, That the President be requested to communicate to the Senate, if not incompatible with the public in-terest, any correspondence or other information in posses-sion of the Government; relating to any plan or plans now projected or being projected with a view to the establish-ment of monarchical governments in Central and South America

March 24-The PRESIDENT transmitted this paper from the Secretary of State, in reply :

The Secretary of State, to whom has been referred the resolution of the Senate of the 15th instant, requesting the President "to communicate to the Senate, if not incompatible with the public interest, any correspondence or other information in possession of the Government relating to ary plan or plans now projected, or being projected, with a view to the establishment of monarchical governments in Central and South America," has the honor to report, that central and south America, has the noted to report, that surmises and jealousies are constantly arising on the sub-ject to which the resolution refers, which are brought to the notice of the Department by our representatives abroad. But there is no correspondence or other form of informa-tion which furnishes any reliable facts showing the exist-ence of "plans" for the accomplishment of the object mentioned.

Any correspondence which might be regarded as embraced in the resolution, besides being very vague, is, in its nature, confidential, and its publication at the present time would be incompatible with the public interest.

# Alleged Foreign Enlistments.

1864, June 28-The PRESIDENT transmitted, in reply to a Senate resolution of 24th, these reports:

#### FROM THE SECRETARY OF STATE.

WASHINGTON, June 25, 1864. The Secretary of State, to whom has been referred the resolution of the Senate of the 24th instant, requesting the President to inform that body "if any authority has been given any one, either in this country or elsewhere, to obtain recruits in Ireland or Canada for our army and navy; and whether any such recruits have been obtained, or whether, to the knowledge of the Government, Irishmen or Canadians have been induced to emigrate to this country in order to be recruited; and if so, what measures, if any, h been adopted in order to arrest such conduct," has has the honor, in reply to the inquiries thus submitted, to report, that no authority has been given by the Executive of this Government, or by any executive department, to any one, either in this country or elsewhere, to obtain recruits either in Ireland, or in Canada, or in any foreign country, for either the army or the navy of the United States; and on the contrary, that whenever application for such authority has been made, it has been refused and absolutely withheld. If any such recruits have been obtained, either in the provided spaced in the vace being obtained, either in the

If any such recruits have been obtained, either in the provinces named in the resolution, or in any foreign country, they have been obtained by persons who are not even citi-zens of the United States, but subjects or citizens of the country where the recruits were obtained. The persons who obtained such recruits, if any were so obtained, were answerable to the laws of the foreign province or country where their offences were committed, and at the same time they were not within the reach of our own laws and tribunals; and such persons acted without any authority or con-sent, and even without the knowledge of this Government. This Government has no knowledge that any such recruits

have been obtained in the provinces named, or in any foreign country. In two or three instances it has been reported to this department that recruiting agents crossed the Canadian This department that retributing agents crossed the cantantian frontier, without authority, with a view to engage recruits or reclaim deserters. The complaints thus made were im-mediately investigated, the proceedings of such recruiting agents were promptly disavowed and condemned, the re-cruits or deserters, if any had been brought into the United States, were at once returned, and the offending agents

States, were at once returned, and the offending agents were dismissed from the public service. In the land and naval forces of the United States there are found not only some Canadians, some Englishmen, and some Irishmen, but also many subjects of continental European powers. All of these persons were voluntary immigrants into the United States. They enlisted after their arrival on our shores, of their own free accord, within our own limits and jurisdiction, and not in any foreign country. The executive government has no knowledge of the nature of the special inducements which led these vol-unteers to emigrate from their native countries, or of the purposes for which they emigrated. It has, however, neither directly nor indirectly invited their immigration by any when such persons were found within the United States, When such persons were found within the United States, exactly the same inducements to military service were open to them which by authority of law were offered at the same time to citizens of the United States.

He same time to citizens of the United States. Having thus answered the inquiries contained in the res-olution of the Senate, the Secretary of State might here without impropriety close this report. Nevertheless, the occasion is a proper one for noticing complaints on the sub-ject of recruitment in our army and navy which have re-The cently found utterance in the British House of Lords. Secretary of State has, therefore, further to report that the Government of the United States has practiced the most scrupulous care in preventing and avoiding in Great Britain, and in all other foreign countries, any violation of in-ternational or municipal laws in regard to the enlistment of soldiers and seamen. Moreover, when the British government, or any other

foreign government, has complained of any alleged viola-tion of the rights of its subjects within the United States, this Government has listened to the complaints patiently, investigated them promptly, and where redress was found due, and was practicable, has cheerfully accorded it. This Government, on the other hand, has been obliged to subwar from such ports against the United States.

It is a notorious fact, manifest to all the world, that a vigorous and continual tide of emigration is flowing from Eu-rope, and especially from portions of the British empire, and from Germany and Sweden, into the United States. This immigration, like the immigration which preceded it, results from the reciprocal conditions of industrial and results from the reciprocal conditions of industrial and social life in Europe and America. Of the mass of immi-grants who arrive on our shores, far the largest number go immediately into the occupations of peaceful indus-try. Those, on the contrary, who are susceptible to the attractions of military life, voluntarily enter the national service with a similar class of our own native citizens, upon service with a similar class of our own hative citizens, upon the same equal inducements, and with the same patriotic motives. There is no law of nations, and no principles of international comity, which requires us to refuse their aid in the cause of the country and of humanity. This Government does not repudiate or discourage immi-gration. The Government frankly avows that it encourages imminentia formed locustic but only here you caved

gration. The Government frankly avows that it encourages immigration from all countries, but only by open, lawful, and honorable agencies and means. However statesmen in other countries may have at the beginning misunderstood the nature and direction of the present civil war, that na-ture and that direction were not misunderstood by the Gov-ernment of the United States. It was foreseen here that the seditious attempt to divide the American Union, if not discouraged by other commercial and maritime powers, would not merely produce great commercial and social em-barrassment in the United States, but that, if it should be persisted in and produce fract must seriously disturb the persisted in and protected, it must seriously disturb the commerce and industry of other nations. Upon this ground, among others, the Government of the United States earn-estly remonstrated with foreign States against their award of upmend commercial and ball increase ariticle or to the in estly remonstrated with foreign States against their award of unusual commercial and belligerent privileges to the in-surgents, in derogation of the sovereignty of the United States. When, however, it was fully disclosed that the in-surrection aimed at nothing less than to separate fifteen of these States from the rest, and to re-establish them within our own lawful territory as one single independent nation upon the foundation of African slavery, this Government did not hesitate, so far as authorized by law, to draw upon all the resources of the country, and to call into activity all the energies of the American people to prevent so great a crime. If further resolved to devote its best efforts within the limits of international law and the Constitution of the

United States, first to bring African slavery to an end throughout the world, and secondly to strengthen the in-terest of free labor upon the American continent. It recog-nized and entered into commercial relations with free States founded on African colonization. It refused a market for slaves, and it pursues the slave trader on the high seas, and denies to him an asylum on our own shores. On the con-trory, it inits honest and inductions from form trary, it invites honest and industrious freemen hither from all parts of the world, and gives them free homes and ample fields, while it opens to them virgin mines and husy work-shops, with all the privileges of perfect civil and religious liberty. So far as increase of immigration has resulted from the action of the Government during the present civil war, it is due exclusively to what has thus lawfully been done It is one exclusively to what has thus lawfully been done with those two ends of extinguishing slavery and fortifying freedom always in view. Nor has this Government any reason to be disappointed with the results. The country has sustained a very destructive war for the period of three years. Yet it is not here that national resources or credit fails. It is not here that patriots are wanting to defend fails. It is not here that patriots are wanting to defend the country of their birth or their choice, nor is it here that the country of their birth of their birth of their birth, birth of the their times, merchants, artisans, and laborers lack either subsistence or employment, with abundant rewarda. The number of slaves is rapidly diminishing, and the number of sheemen continues to augment, even during the convulsions of domestic war, more rapidly than over a free population advanced in any other country, or even in our own.

Respectfully submitted,

The PRESIDENT.

#### WILLIAM H. SEWARD.

#### FROM THE SECRETARY OF WAR.

WAR DEPARTMENT, WASHINGTON CITY, June 27, 1864. SIR: In answer to the Senate resolution of inquiry, passed June 24, 1864, and referred by you to this department, I have the honor to reply: First. That no subsetue to

First. That no authority has been given by this department to any one, either in this country or elsewhere, to obtain recruits in Ireland or Canada for the army of the United States.

Second. That no recruits have been obtained in Ireland or in Canada for the army of the United States with my knowledge or consent, and, to the best of my information and belief none have been obtained, nor any effort mado

Third. That neither Irishmen nor Canadians have, with my knowledge, approbation, or consent, or with the knowl-edge, approbation, or consent of any one in this department, been induced to emigrate to this country in order to enlist into the army. Fourth. That no measures have been adopted by this de-

partment to arrest any such conduct, because no informa-tion of any such conduct has reached the department, and I do not believe that it has been practiced in any instance.

I will add that no encouragement or inducement whatever has been extended by this department to any person or persons to obtain recruits for the army anywhere beyond the limits of the United States.

I have the honor to be, very respectfully, your obedient ervant, EDWIN M. STANTON. servant. Secretary of War.

# FROM THE SECRETARY OF THE NAVY.

#### NAVY DEPARTMENT, June 27, 1864.

SIR: I have the honor to acknowledge the reference to this Department of a resolution passed in the Senate of the United States on the 24th instant, requesting the President of the United States "to inform the Senate if any authority of the Oniced States to not in the Scherce it all studies in all states in the Scherce it is country or else-where, to obtain recruits in Ireland or in Canada for our army or navy; and whether any such recruits have been obtained, or whether, to the Knowledge of the Government, Irishmen or Canadians have been induced to emigrate to be accounted and the second the and if or whether Irishmen of Canadians have been induced to emigrate to this country in order to be so recruited; and if so, what measures, if any, have been adopted to arrest such con-duct;<sup>1</sup> and to state, in reply, that no such order as that indicated in the resolution has been given by the Navy De-partment to any one, either in this country or elsewhere, nor is the Navy Department aware that any such recruits here here able to the divergence here here one offered have been obtained, or that inducements have been offered to Irishmen or Canadians to emigrate to this country in order to be so recruited.

On the occasion of a visit of the United States steamor Kearsarge to Queenstown, Ireland, in November last, several Irishmen secreted themselves on board the vessel, were upon the foundation of Aincan slavery, this Government did not hesitate, so far as authorized by law, to draw upon all the resources of the country, and to call into activity all the energies of the American people to prevent so great a the energies of the American people to prevent so great a the limits of international law and the Constitution of the or any intention of permitting others under his command to do so. Explanations have been made to the British gov-ernment, and it is presumed the matter has been satisfac-torily settled.

I am, sir, with very great respect, your obedient servant, GIDEON WELLES.

Secretary of the Navy.

# Foreign Mediation.

SECRETARY SEWARD'S LETTER TO GOV. HICKS.

DEPARTMENT OF STATE, April 22, 1861. His Excellency THOS. H. HICKS, Governor of Maryland: SR: I have had the honor to receive your communica-tion this morning, in which you inform me that you have falt it to be your duty to advise the President of the United States to order elsewhere the troops then off Annapolis, and also that no more may be sent through Maryland; and that you have further suggested that Lord Lyons be re-quested to act as mediator between the contending parties in our country to prevent the effusion of blood. \* \* \*

If eighty years could have obliterated all the other noble entiments of that age in Maryland, the President would be semiments of that age in Maryland, the President would be hopeful, nevertheless, that there is one that would forever remain there and everywhere. That semiment is that no domestic contention whatever, that may arise among the parties of this Republic, ought in any case to be referred to any foreign arbitrament, least of all to the arbitrament of Worment measure.

an European monarchy. I have the honor to be, with distinguished consideration, J have the honor to be, with distinguished events. your Excellency's most obedient servant. WILLIAM H. SEWARD.

# FRENCH MEDIATION.

During 1862, the French Government proposed to the Russian and English governments to join it in trying to bring about an armistice for six months between "the Federal Government and the Confederates of the South;" which they declined.

Jan. 9, 1863-M. Drouyn de l'Huys, Min-ister of Foreign Affairs, addressed M. Mercier, the French Minister at Washington, on this subject.

# We add a few paragraphs:

SR: In forming the purpose of assisting, by the proffer of our good offices, to shorten the period of those hestilities which are desolating the American continent, had we not been guided, beyond all, by the friendship which actuates the government of the Emperor in regard to the United States, the little success of our overtures might chill the interests with which we follow the fluctuations of this contest : but the sentiment to which we have vielded is too sincere for indifference to find a place in our thoughts, and that we should cease to be painfully affected, whilst the

war continues to rage. We cannot regard without profound regret this war, worse than civil, comparable to the most terrible distractions of the ancient republics, and whose disasters multi-The government of his Majesty have therefore seriously examined the objections which have been made to us.

Where we have suggested the idea of a friendly mediation, and we have asked ourselves whether they are truly of a nature to set aside as premature every tentative to a recon-ciliation, on one part has been opposed to us the repug-nance of the United States to admit the intervention of foreign influence in the dispute, on the other, the hope, which the Federal Government has not abandoned, of attaining its solution by force of arms.

Assuredly, sir, recourse to the good offices of one or sev-eral neutral powers contains nothing incompatible with the pride so legitimate amidst a great nation, and wars purely international are not those alone which furnish examples of the useful character of mediation.

SECRETARY SEWARD'S LETTER TO MR. DAYTON.

1

DEPARTMENT OF STATE, WASHINGTON, 6th Feb., 1863. The intimation given in your dispatch of January SIB: Site: The intimation given in your dispatch of Jahuary 15th, No. 255, that I might expect a special visit from M Mercier, has been realized. He called on the 3d instant, and gave me a copy of a dispatch which he had just then received from M. Drouyn de l'Huys under the date of the 21st of January. I have taken the President's instructions, and how proceed to give you his views upon the subject in function question.

It has been considered with seriousness resulting from It has been considered with seriousness resulting from the reflection that the people of France are known to be faultless sharers with the American nation in the misfor-tunes and calamities of our unhappy civil war; nor do we on this, any more than on other occasions, forget the tradi-tional friendship of the two countries, which we unhesita-tingly believe has inspired the counsels that M. Drouyn de l'Huys has imparted. He says, " the Federal Government does not despair, we

He says, "the rederat Government does not acceptar, no know, of giving more active impulse to hostilities;" and again he remarks, "the protraction of the struggle, in a word, has not shaken the confidence (of the Federal Gov-ernment) in the definite success of its efforts." These passages seem to me to do unitentional injustice to the language, whether confidential or public, in which this Government has constantly spoken on the subject of the war. It certainly has had and avowed only one purposea determination to preserve the integrity of the country. So far from admitting any laxity of effort, or betraying any despondency, the Government, has on the contrary, horne itself cheerfully in all vicissitudes with unwavering confldence in an early and complete triumph of the national cause. Now, when we are, in a manner, invited by a friendly power to review the twenty-one months' history of the conflict, we find no occasion to abate that confidence. Through such an alternation of victories and defeats as is the appointed incident of every war, the land and naval forces of the United States have steadily advanced, re-claiming from the insurgents the ports, forts, and posts, which they had treacherously seized before the strife actually began, and even before it was seriously apprehended. So many of the States and districts which the insurgents included in the field of their projected exclusive slaveincluded in the field of their projected exclusive slave-holding dominions, have already been re-established under the flag of the Union, that they now retain only the States of Georgia, Alabama, and Texas, with half of Virginia, half of North Carolina, two thirds of South Carolina, half of Mississippi, and one third respectively of Arkansas and Louisiana. The national forces hold even this small territory in close blockade and siege.

This Government, if required, does not hesitate to submit its achievements to the test of comparison; and it maintains that in no part of the world, and in no times, ancient or modern, has a nation, when rendered all unready for combat by the enjoyment of eighty years of almost unbroken peace, so quickly awakened at the alarm of sedition, put forth en-ergies so vigorous, and achieved successes so signal and effective, as those which have marked the progress of this

contest on the part of the Union. M. Drouyn de l'Huys, I fear, has taken other light than the correspondence of this Government for his guidance, in ascertaining its temper and firmness. He has probably read of divisions of sentiment among those who hold themselves forth as organs of public opinion here, and has given to them an undue importance. It is to be remembered to them an undue importance. It is to be remembered that this is a nation of thirty millions, civilly divided into forty-one States and Territories, which cover an expanse hardly less than Europe; that the people are a pure democharding less than hardpe, that the people at each part canon-racy, exercising everywhere the ntimost freedom of speech and suffrage; that a great crisis necessarily produces vehe-ment as well as profound debate, with sharp collisions of individual, local, and sectional interests, sentiments and different sections interests. ambitions, and that this heat of controversy is increased by the intervention of speculations, interests, prejudices, and passions from every other part of the civilized world. It is, passions from every other part of the civilized world. It is, however, through such debates that the agreement of the nation upon any subject is habitually attained, its resolu-tion formed, and its policy established. While there has been much difference of popular opinion and favor con-cerning the agents who shall carry on the war, the princi-ples on which it shall be waged, and the means with which it shall be prosecuted, M. Drouyn de l'Huys has only to refer to the statute book of Congress and the Executiv e ordi-The to the statute book of congress and the Execute of the nances, to learn that the national activity has hitherto been, and yet is, as efficient as that of any other nation— whatever its form of government—ever was, under circum-stances of equally grave import to its peace, safety and welfare. Not one voice has been raised anywhere, out of the importion of the important on the provestion of the import the immediate field of the insurrection, in favor of foreign intervention, of mediation, of arbitration, or of compro-mise, with the relinquishment of one acre of the national domain, or the surrender of even one constitutional fran-At the same time, it is manifest to the world that chise. our resources are yct abundant, and our credit adequate to the existing emergency. What M. Drouyn de l'Huys suggests is that this Govern

ment shall appoint commissioners to meet, on neutral ground, commissioners of the insurgents. He supposes that in the conferences to be thus held reciprocal complaints could be discussed, and in place of the accusations which the North and the South now mutually cast upon each other, the conferees would be engaged with discussions of the interests which divide them. He assumes, further, that the commissioners would seek, by means of well-ordered and profound deliberation, whether these interests are definitively irreconcilable, whether separation is an extreme that can no longer be avoided, or whether the memories of a common existence, the ties of every kind which have made of the North and the South one whole Federative State, and have borne them on to so high a degree of prosperity, are not more powerful than the quese which have placed arms in the hands of the two populations. The suggestion is not an extraordinary one, and it may well have been thought by the Emperor of the French, in

The suggestion is not an extraordinary one, and it may well have been thought by the Emperor of the French, in the carnestness of his benevolent desire for the restoration of peace, a feasible one. But when M. Drouyn de l'Huys shall come to review it in the light in which it must necessarily be examined in this country. I think he can hardly fail to perceive that it amounts to nothing less than a proposition that, while this Government is engaged in suppressing an armed insurrection, with the purpose of maintaining the constitutional national authority, and preserving the integrity of the country, it shall enter into diplomatic discussion with the insurgents upon the questions whether that authority shall not be renounced, and whether the country shall not be delivered over to disunion, to be quickly followed by ever increasing anarchy.

If it were possible for the Government of the United States to compromise the National authority so far as to enter into such debates, is it not easy to perceive what good results could be obtained by them.

The commissioners must agree in recommending either that the Union shall stand, or that it shall be voluntarily dissolved; or clse they must leave the vital question unsettled, to abide at last the fortunes of the war. The Government has not shut out knowledge of the present temper, any more than of the past pnrposes of the insurgents. There is not the least ground to suppose that the controling actors would be persuaded at this moment, by any arguments which national commissioners could offer, to forego the ambition that has impelled them to the disloyal position they are occupying. Any commissioners who should be appointed by these actors, or through their dictation or influence, must enter the conference imbuded with the spirit, and pledged to the personal fortunes of the insurgent chiefs. The loyal people in the insurrectionary States would be unheard, and any offer for peace by this Government, on the condition of the maintenance of the Union

On the other hand, as I have already intimated, this Government has not the least thought of relinquishing the trust which has been confided to it by the nation, under the most selemn of all political sanctions; and if it had any such thought, it would still have abundant reason to know that peace proposed at the cost of dissolution would be immediately, unreservedly, and indignantly rejected by the American people. It is a great mistake that European statesmen make, if they suppose this people are demoralized. Whatever, in the case of an insurrection, the people of France, or of Great Britain, or of Switzerland, or the Netherlands would do to save their national existences, no matter how the strife might be regarded by or affect foreign nations, just so much, and certainly no less, the people of Atlantic coasts, and by the shores of the Guifs of 5t. Lawrence and Mexico, together with the free and common navigation of the Rio Grande, Missouri, Arkaneas, Mississippi, Ohio, St. Lawrence, Hudson, Delaware, Potomac, and other natural highways by which this land, which to them is at once a land of inheritance and a land of promise, is opened and watered. Even if the agents of the American people now exercising their power, should, through fear or faction, fall below this height of the national virtue, they would be speedily, yet constitutionally, replaced by others of thermer character and patriotism.

fail below the second state of the speed of the speed state of the spe

I have on many occasions heretofore submitted to the French Government the President's views of the interests, and the ideas more effective for the time than even interests, which lie at the bottom of the American Government and people to sustain the Federal Union. The President has done the same thing in his messages and other public declarations. I refrain, therefore, from reviewing that argument in connection with the existing question. M. Drouyn de l'Huys draws to his aid the conferences

M. Drouyn de l'Huys draws to his aid the conferences which took place between the colonies and Great Britain, in our Revolutionary war. He will allow us to assume, that action in the crisis of a nation must accord with its neces-

sities, and therefore can seldom be conformed to precedents. Great Britain, when entering on the negotiations had manifestly come to entertain doubts of her ultimate success; and it is certain that the councils of the colonics could not fail to take new courage, if not to gain other advantage, when the parent State compromised so far as to treat of peace on the terms of conceding their independence.

It is true, indeed, that peace inust come at some time, and that conferences must attend, if they are not allowed to precede the pacification. There is, however, a better form for such conferences than the one which M. Drouyn de 'Huys suggests. The latter would be palpably in derogation of the Constitution of the United States, and would carry no weight because destitute of the sanction necessary to bind either the disloyal or the loyal portions of the people. On the other hand, the Congress of the United States furnishes a constitutional forum for debates between the alienated parties. Senators and Representatives from the loyal portion of the people are there already, freely empowered to confer; and seats also are vacant, and inviting Senators and Representatives of the discontented party who may be constitutionally sent there from the States involved in the insurrection. Moreover, the conferences which can thus be held in Congress have this great advantage over any that could be organized upon the plan of M. Drouyn de l'Huys, namely: that the Congress, if it were thought wise, could call a national convention to adopt its recommendations, and give them all the soleminity and binding force of organic law. Such conferences between the alienated parties may be said to have already begun. States which are claimed by the insurgents—are already leading in a proper spirit, their advice upon the course best calculated to bring about, in the shortest time, a firm, lasting, and honorable peace. Representatives have been sont, also, from Louisiana, and others are understood to be coming from Arkansas.

There is a proponderating argument in favor of the congressional form of conference over that which is suggested by M. Drouyn de l'Huys, namely: that while an accession to the latter would bring this Government into a concurrence with the insurgents in disregarding and setting aside an important part of the Constitution of the United States, and so would be of pernicious example, the congressional conference, on the contrary, preserves and gives new strength to that sacred writing which must continue through future ages the sheet anchor of the Republic.

You will be at liberty to read this dispatch to M. Drouyn de l'Huys, and to give him a copy, if he shall desire it.

To the end that you may be informed of the whole case, I transmit a copy of M. Drouyn de l'Huys's dispatch. I am, sir, your obedient servant,

WILLIAM H. SEWARD.

#### RESOLUTIONS OF CONGRESS UPON FOREIGN MEDIATION.

#### Third Session, Thirty-Seventh Congress.

Whereas it appears from the diplomatic correspondence, submitted to Congress that a proposition; friendly in form, looking to pacification through foreign mediation, has been made to the United States by the Emperor of the French and promptly declined by the President; and whereas the idea of mediation or intervention in some shape may be regarded by foreign governments as practicable, and such governments, through this misunderstanding, may be led to proceedings tending to embarrass the friendly relations which now exist between them and the United States; and whereas, in order to remove for the future all chance of misunderstanding on this subject, and to secure, for the United States the full enjoyment of that freedom from foreign interference which is one of the highest rights of independent States; it eems fit that Congress should declare its convictions thereon: Therefore, *Resolved*, (the House of Representatives concurring.) That

Resolved, (the House of Representatives concurring.) That while, in times past, the United States have sought and accepted the friendly mediation or arbitration of foreign powers for the pacific adjustment of international questions, where the United States were the party of the one part and some other sovereign power the party of the other part ; and while they are not disposed to misconstrue the natural and humane desire of foreign powers to aid in arresting domestic troubles, which, widening in their influence, have afflicted other countries, especially in view of the circumstance, deeply regretted by the American people, that the blow aimed by the rebellion at the national life has fallen heavily upon the laboring population of Europe; yet, notwithstanding these things. Congress cannot hesitate to regard every proposition of foreign interference in the present contest as so far unreasonable and inadmissible that its only

true state of the question, and of the real character of the war in which the Republic sengaged. 2. That the United States are now grappling with an un-provoked and wicked rebellion, which is seeking the de-truction of the Republic that it may build a new power, whose corner-stone, according to the confession of its chiefs, shall be slavery; that for the suppression of this rebellion, and thus to save the Republic and to prevent the establish-ment of such a power, then national Government is now moly depends, any proposition from a foreign power, what ere form it may take, having for its object the arrest of these offorts, is, just in proportion to its influence, an er-couragement to the rebellion and to its declared principles, that, with these convictions, and not doubting that every auch openone interests, Congress will be obliged treasure, and to postpone the much desired day of peace; that, with these convictions, and not doubting that every auch orposition, although made with good intent, is inju-treasure, and to postpone the much desired day of peace; that, with these convictions, and not doubting that every auch orposition, although made with good intent, is inju-tious to the national interests, Congress will be obliged to known with the second the rebellion and to doubting that every auch orposition, although made with good intent, is inju-tions to the national interests, Congress will be obliged to known work with the second the rebellion and to doubting that every auch orposition, although made with good intent, is inju-tions to the national interests, Congress will be obliged to known work the second the readed area principles and on this scloutated the good intent, is inju-tions to the national interests, Congress will be obliged to known work the second the second interest is to the same direction as an aution with a state with good intent is inju-tions to the attional the second is the same direction as an aution with ext with it eaversets to the national interests. Congress will be obl and of this account, is calculated to prolong and imbitter the conflict, to cause increased expenditure of blood and freasure, and to postpone the much desired day of peace; that, with these convictions, and not doubting that every unch proposition, although made with good intent, is inju-rious to the national interests, Congress will be obliged to look upon any further attempt in the same direction as an unfriendly act, which it earnestly deprecates, to the end that nothing may occur abroad to strengthen the robellion or to weaken those relations of good will with foreign pow-ers which the United States are happy to cultivate.

3. That the rebellion from its beginning, and far back even in the conspiracy which preceded its outbreak, was encouraged by the hope of support from foreign powers; that its chiefs frequently boasted that the people of Europe were so far dependent upon regular supplies of the great were so far dependent upon regular supplies of the great southern staple that sooner or later their governments would be constrained to take side with the rebellion in some effective form, even to the extent of forcible interven-tion, if the milder form did not prevail; that the rebellion is now sustained by this hope, which every proposition of foreign interference quickens anew, and that without this life-giving support it must soon yield to the just and pater-nal authority of the national Government; that, consider-ing these things, which are aggravated by the motive of the resistance thus encouraged, the United States regret that foreign powers have not frankly told the chiefs of the rebel-lion that the work in which they are engaged is hateful, and that a new government, such as they seek to found, with slavery as its acknowledged corner-stone, and with no other declared object of separate existence, is so far shockthe factor of section of separate existence, is so far shock-ing to civilization and the moral sense of mankind that it must not expect welcome or recognition in the common-wealth of nations.

4. That the United States, confident in the justice of their cause, which is the cause also of good government and of human rights everywhere among men; anxious for the ppedy restoration of peace, which shall secure tranquillity at home and remove all occasion of complaint abroad; and awaiting with well-assured trust the final suppression of the awarting with well-assured trust the final suppression of the rebellion, through which all these tluings, rescued from present danger, will be secured forever, and the Republic, one and indivisible, triumphant over its enemies, will con-tinue to stand an example to mankind, hereby announce, as their unalterable purpose, that the war will be vigorously prosecuted, according to the humane principles of Christian States, until the rebellion shall be suppressed; and they reverently invoke upon their cause the belssings of Almighty God.

God. 5. That the President be requested to transmit a copy of these resolutions, through the Secretary of State, to the ministers of the United States in foreign countries, that the declaration and protest herein set forth may be com-municated by them to the government to which they are accredited.

1863, March 3-The resolutions were passed by the Senate-yeas 31, nays 5, as follows:

The Schatt -- yeas 31, nays 5, as follows: Yras-Messrs. Anthony, Arnold, Chandler, Clark, Colla-mer, Davis, Dixon, Doolittle, Fessenden, Foot, Foster, Grimes, Harding, Harlan, Harris, Henderson, Hicks; How-ard, Howe, King, Lane of Indiana, Morrill, Nesmill, Pom-eroy, Sunner, Ten Eyck, Trumbull, Wade, Willey, Wilmot, Wilson of Massachusetts-31. NAYS-Messrs. Carbile, Latham, Powell, Saulsbury, Wall -5.

1863, March 3-The above resolutions were considered in the House.

Mr. VALLANDIGHAM moved to lay them on the table; which was negatived-yeas 29, nays 92.

They were then passed-yeas 103, nays 28, as follows:

YEAS-Messers. Aldrich, Alley, Arnold, Ashley, Babbitt, Baily, Baker, Baxter, Bingham, Jacob B. Blair, Samuel S. Blair, Blake, Bridges, William G. Brown, Buffinton, Camp-

#### LETTER OF LORD LYONS TO EARL RUSSELL RE-SPECTING MEDIATION.

WASHINGTON, November 17, 1862. In his dispatches of the 17th and of the 24th ultimo, and of the 7th instant, Mr. Stuart reported to your lordship the result of the elections for members of Congress and State result of the elections for memoers of Congress and state officers, which have recently taken place in several of the most important States of the Union. Without repeating the details, it will be sufficient for me to observe that the success of the Democratic, or (as it now styles itself) the Conservative Party, has been so great as to manifest a change in public feeling, among the most rapid and the most complete that has ever been witnessed, even in this country.

country, On my arrival at New York on the 5th instant I found the Conservative leaders exulting in the crowning success achieved by the party in that State. They appeared to re-joice, above all, in the conviction that personal liberty and freedom of speech had been secured for the principal State of the Union. They believed that the Government must at once desist from exercising in the State of New York tho extraordinary (and as they regarded them) illegal and un-constitutional powers which it had assumed. They were confident that at all events after the 1st of January next, on which day the newly elected Governor would come into confident that at all events atter the 1st of January next, on which day the newly elected Governor would come into office, the suspension of the writ of habeas corpus could not be practically maintained. They seemed to be persuaded that the result of the elections would be accepted by the President as a declaration of the will of the people; that he would increase the moderate and conservative element in the Cabinet; that he would seek to terminate the war, not the cannet; that he would seek to terminate the wir, hot to push it to extremity; that he would endeavor to effect a reconciliation with the people of the South, and renounce the idea of subjugating or exterminating them. On the following morning, however, intelligence arrived from Washington which dashed the rising hopes of the Conservatives. It was announced that General McClellan hed hear dismissed from the commend of the arrow of the

had been dismissed from the command of the army of the Potomac, and ordered to repair to his home; that he had, in fact, been removed altogether from active service. The general had been regarded as the representative of Con-servative principles in the army. Support of him had been made one of the articles of the Conservative electoral pro-gramme. His dismissal was taken as a sign that the Presgramme. His dismissi was taken as sign that the res-ident had thrown himself entirely into the arms of the extreme radical party, and that the attempt to carry out the policy of that party would be persisted in. The irrita-tion of the Conservatives at New York was certainly very great; it seemed, however, to be not unmixed with con-tinuation and dependence.

sternation and despondency. Several of the leaders of the Democratic party sought in-terviews me, both before and after the arrival of the intel-ligence of General McClellan's dismissal. The subject upngence of General acclenate distinsual. The subject up-permost in their minds while they were speaking to me, was naturally that of foreign mediation between the North and South. Many of them seemed to think that this medi-ation must come at last, but they appeared to be very much afraid of its coming too soon. It was evident that they apafraid of its coming too soon. It was evident that they ap-prehended that a premature proposal of foreign interven-tion would afford the Radical party a means of reviving the violent war spirit, and of thus defeating the peaceful plans of the Conservatives. They appeared to regard the present moment as peopliarly unfavorable for such an offer, and indeed, to hold that it would be essential to the success of any proposal from abroad that it should be deferred until the control of the Executive Government should be in the bards of the Conservatives.

I gave no opinion on the subject. I did not say whether or no I myself thought foreign intervention probable or ad-

visable, but I listened with attention to the accounts given | me of the plans and hopes of the Conservative party. At the bottom I thought I perceived a desire to put an end to the war, even at the risk of losing the southern States altogether; but it was plain that It was not thought prndent to avow this desire. Indeed some bints of it, dropped be-fore the elections, were so ill received that a strong decla-ration in the contrary sense was deemed necessary by the Democratic leaders.

At the present moment, therefore, the chiefs of the Conservative party call loudly for a more vigorous prosecution of the war, and reproach the Government with slackness as well as with want of success in its military measures. But they repudiate all idea of Interfering with the institutions of the southern people, or of waging a war of subjugation or extermination. They maintain that the object of the military operations should be to place the North in a posiin which such as the such as t of its slave property, and would enable the North and the South to reunite and to live together in peace and harmony. The Conservatives profess to think that the South might be induced to take part in such a convention, and that a res-toration of the Union would be the result.

The more sagacious members of the party must, however, look upon the proposal of a convention merely as a last experiment to test the possibility of reminin. They are no doubt well aware that the more probable consequence of an armistice would be the establishment of Southern independence, but they perceive that if the South is so utterly alienated that no pessible concessions will induce it to return voluntarily to the Union, it is wiser to agree to separa-tion than to prosecute a cruel and hopeless war.

It is with reference to such an armistice as they desire to attain, that the leaders of the Conservative party regard the question of foreign mediation. They think that the offer of mediation, if made to a radical administration, would be rejected; that, if made at an unpropitious moment, it might increase the virulence with which the war is prosecuted. If their own party were in power, or virtually conobtain an armistice without the aid of foreign governments, but they would be disposed to accept an offer of mediation If it appeared to be the only means of putting a stop to hostilities. They would desire that the offer should come from the great powers of Europe conjeintly, and in particu-lar that as little prominence as possible should be given to Great Britain.

At Washington I have had fewer opportunities than I had at New York of ascertaining the present views of the chiefs of the political parties. At the interview which I had with Mr. Seward the day after my arrival he showed had with Mr. Seward the day after my arrival he showed no disposition to enter upon political matters. He did not appear to expect or to desire to receive from me any special communication from her Majesty's government. The President, when I waited upon him, talked to me only on ordinary topics. I, for my part, gladly shunned all allusion to foreign intervention, my principal object being to avoid saying anything which might embarrass me in carrying with arritetration of the chieft which I may receive out any instructions on the subject which I may receive from your lordship.

All things considered, my own opinion certainly is that the present moment is not a favorable one for making an offer of mediation. It might embarrass the peace party, and even oblige them, in order to maintain their popularity, to make some declaration against it, and this might make it difficult for them to accept a similar offer at a more propitions time. It would in all probability be rejected by the President, who appears to have thrown himself into the arms of the extreme radical party. The views of that party are clear and definite. They contend the throma is a view of a supervision with the state of the sta

declare that there is no hope of reconciliation with the Southern people; that the war must be pursued, per fas et nefas, until the disloyal men of the South are ruined and subjugated, if not exterminated; that not an inch of the subjugated, if not exterminated; that not an inch of the old territory of the Republic must be given up; that foreign intervention, in any shape, must be rejected and resented. This party would desire to turn an offer of mediation to account, for the purpose of inflaming the war mirit and producing reaction carries the Concernities spirit and producing a reaction against the Conservatives. Is is probable, too, that the Government would urge, in

answer to an offer of mediation, that it has by no means abandened the hope of putting down the rebellion within a abandoned the hope of putting down the rebellion within a reasonable time; that at all events, this is not a moment at which it can reasonably be called upon to put a stop to bostilities. It would observe that the armies of the United States are everywhere advancing, and that expeditions are prepared against Texas, as well as against Charleston, Mo-bile, and other points on the coast. It would point out nostinities. It would coverve that the armies of the United of France to subject the Republic of MeXico to her author-States are everywhere advancing, and that expeditions are ity by armed force is a violation of the established and prepared against Texas, as well as against Charleston, Mo-bile, and other points on the coast. It would point out that it had equipped a considerable number of warvessels, iron-clad as well as others, at a vast expense; that the sea-son had just arrived when the autumn rains would render

the rivers navigable by armed vessels, and when the Sonth ern coast would be free from epidemic disease. It might even represent an advaace of the Army of the

Potomac to Richmond as a probable event. The experience of the past is certainly not calculated to inspire any great of the past is certainly not calculated to inspire any great confidence in the results of these warlike preparations, but the political interests of the party now in power render a continuance of the war a necessity to it. Its only chance of regaining its lost popularity lies in successful military oper-ations. Unless it can obtain a much higher place in public estimation than it now occupies, not only will its tenure of power become extremely precarious, but some of its lead-ing members may be called to a severe account for their extra-legal proceedings. During the session of Congress which begins next month, the present Administration has which begins next month, the present Administration has indeed reason to expect an uncompromising support from a majority of both Houses of Congress. But on the 4th of March next, the existing House of Representatives is dis-solved by the terms of the Constitution, and at the same time several of the present Senators go out of office. The majority of the members chosen at the recent elections for the new House of Representatives are of the Democratic or Conservative party, and in some States, Senators of that Conservative party, and in some states, Senators of that party will be returned in the room of those whose term of office expires next March. The new Congress is likely to be hostile to the Administration and to the Radical party; and although it will not, in the ordinary course of things, assemble until the last month of next year, the President will hardly be able to persist in his present policy and in his assumption of extraordinary powers, unless he can, in virtue of military successes, obtain a reputation with the people which will sustain him in a contest with the Legislature.

It would seem, then, to be vain to make an offer of me-diation to the present Government, in their present mood with any notion that it would be accepted. A change of mood may, however, take place after the 4th of March, if no great military successes occur in the interval. Such a The great initiary successes occur in the interval. Such as change may possibly be produced sconer by military re-verses. A propesal, however, to mediate, made even under the present circumstances, by three or more of the great powers of Europe conjointly might not produce any great înconvenience.

It is, indeed, urged by some people that mediation should be offered, not so much with a view to its being accepted as to its clearing the way for a recognition of the Southern Confederacy. And, indeed, if it were determined that the time had come for recognizing that Confederacy, no doubt an offer of mediation would be a suitable preliminary. But I do not clearly understand what advantage is expected to result from a simple recegnition of the southern government; and I presume that the European powers do not contemplate breaking up the blockade by force of arms, or en-gaging in hostilities with the United States in support of the independence of the South.

I have, indeed, heard it maintained that Great Britain should recognize the independence of the South as soon as possible, with a view to impede the success of the efforts of the conservative party to reconstruct the Union. The advocates of this opinion consider a re-union as a probable event, and apprehend that the first result of it would be that the combined forces of the North and the South would that the communed forces of the North and the South would be let loose upon Canada. I certainly do not at present share these apprehensions. All hope of the reconstruction of the Union appears to be fading away, even from the minds of those who mest ardently desire it. But if the ro-construction be still possible, I do not think we need con-clude that it would lead to an invasion of Canada, or to any consequences injurious to Great Britain. At any rate dangers of this kind are remote. The immediate and obvious interest of Great Britain, as well as of the rest of Europe, is that peace and prosperity should be restored to this country as soon as possible. The point chiefly worthy of consider-ation appears to be whether separation or reunion be the more likely to effect this object.

## The French in Mexico.

# Third Session, Thirty-Seventh Congress. IN SENATE.

1863, January 19-Mr. McDougall offered the following concurrent resolutions:

Resolved by the Senate, (the House of Representatives concurring,) That the present attempt by the government of France to subject the Republic of Mexico to her anthor-

presentatives of the allies by letter of the 30th day of No-vember, 1861, and particularly and repeatedly assured to this Government through its ministers resident at the Court of France

Recoved further, That the attempt to subject the Repub-lic of Mexico to the Franch authority is an act not merely unfriendly to this Republic, but to free institutions every-where, and that it is regarded by this Republic as not only

where, and that it is regarded by this Republic as not only unfriendly, but as hostile. Resolved further, That it is the duty of this Republic to require of the government of France that her armed forces be withdrawn from the territories of Mexico. Resolved further, That it is the duty and proper office of this Republic now, and at all times, to lead such aid to the Beaublic of Maxico sai or may be required to prevent the

Republic of Mexico as is or may be required to prevent the forcible interposition of any of the States of Europe in the

political affairs of that Republic. Resolved further, That the President of the United States be requested to cause to be communicated to the goverument of Mexico the views now expressed by the two Houses of Congress, and be further requested to cause to be nego-tiated such treaty or treaties between the two Republics as will best tend to make these views effective.

1863, February 4-The resolutions were, on motion of Mr. SUMNER, laid on the table-yeas 34, nays 10, as follows:

YEAS-Messrs. Anthony, Arnold, Carlile, Chandler, Clark, Collamer, Covan, Davis, Dixon, Doolittle, Fessenden, Foot, Foster, Grimes, Hale, Harding, Harlan, Harris, Hen-derson, Howard, Howe, King, Lane of Iudiana, Lane of Kansess, Morrill, Pomeroy, Shermau, Sunner, Ten Eyck, Wade, Wilkinson, Wilmot, Wilson of Massachusetts-34. NATS - Mesrs. Kennery, Latham, McDongall, Powell, Rice, Richardson, Saulsbury, Trumbull, Turpie, Wilson of

Missouri-10.

## First Session, Thirty-Eighth Congress. IN SENATE.

1864, Jan. 11-Mr. McDougall offered this joint resolution, which was referred to the **Committee on Foreign Relations:** 

IN RELATION TO THE OCCUPATION OF MEXICO.

Resolved, &c., That the occupation of a portion of the territory of the Republic of Maxico by the armed forces of the government of France, with the purposes arowed by the government of France, is an act unfriendly to the Re-

public of the United States of America. SEC. 2. And be it further resolved, That it is the duty of the proper department of this Government to demand of the government of France the withdrawal of her armed forces from the Mexican territory within a reasonable time.

SEC. 3. And be it further resolved, That in the event the government of France shall decline or refuse to so with-draw her armed forces, or shall fail to take measures to that effect, on or before the fifteenth day of March next, then it will become the duty of the Congress of the United States of America to declare war against the government of France.

June 14-Mr. McDougall sought to introduce this resolution, but objection was made :

Resolved, That the people of the United States can never regard with indifference the attempt of any foreign power to overthrow by force or to supplant by fraud the institutions of any republican government on the western conti-nent, and that they will view with extreme jealousy, as menacing to the peace and independence of their own country, the efforts of any such power to obtain any footholds for monarchical Governments sustained by foreign military force in near proximity to the United States.

## IN HOUSE.

1864, April 4-Mr. H. WINTER DAVIS, from the Committee on Foreign Affairs, reported to the House of Representatives the following joint resolution, which was passed-yeas 109, nays none, as follows :

YEAS-Messrs. James C. Allen, William J. Allen, Alley, YEAB-Messrs, James C. Allen, William J. Allen, Alley, Allison, Ames, Ancorson, Arnold, Ashley, Baily, Augustus C. Baldwin, John D. Baldwin, Baxter, Beaman, Blaine, Francis P. Blair, ir., Birs, Elow, Boutwell, Boyd, Brooks, Broomall, James S. Brown, William G. Brown, Chanter, Ambrose W. Clark, Clay, Cobb, Cole, Cox, Cravens, Cresswell, Henry Winter Davis, Thomas T. Davis, Dawson, Denison, Dixon, Driggs, Dumont, Eckley, Eden, Eldridge, Eliot, English, Fenton, Finck, Frank, Ganson, Garfield, Gooch, Grider, Grinnell, Griswold, Hale, Harding, Harring-

ton, Herrick, Higby, Holman, Hooper, Hotchkiss, Asahel W. Hubbard, Jenckes, Philip Johnson, Julian, Kalbfelsch, Kasson, Kelley, Francis W. Kellogg, Orlando Kellogg, Ker-nan, King, Law, Lazear, Long, Longyear, Mallory, Marey, Marvin, McBride, McClurg, McKinney, Middleon, Samuel F. Miller, Moorhead, Morrill, Daniel Morris, James R. Mor-ris, Morrison, Amos Myers, Leonard Myers, Nelson, Norton, Odell, Charles O'Neill, John O'Neill, Orth, Patterson, Pen-dleton, Perham, Pike, Pomeroy, Price, Pruyn, Samuel J. Randall, William H. Randall, Alexander H. Rice, John H. Rice, Regers, Edward H. Rollins, James S. Rollins, Schenck, Scofield, Stoft, Shannon, Sloan, Smithers, Spalding, Starr, John B. Stele, Stevens, Strouse, Stuart, Sweat, Thayer, Tracy, Upson, Yan Valkenburgh, Foorhees, Ward, Ellihu B. Washburne, William B. Washburn, Webster, Whaley, Wheler, Chilton A. White, Joseph W. White, Williams, Wider, Wilson, Windom, Winfield, Benjamin Wood, Wood-bridge, Faaman-109. NATS-None.

NAYS-None.

# The resolution is as follows:

Relative to the substitution of monarchial for republican Government in Mexico, under European auspices.

Resolved, &c., That the Congress of the United States are unwilling, by silence, to leave the nations of the world under the impression that they are indifferent spectators of the deplorable events now transpiring in the Republic of Mexico; and they therefore think fit to declare that it does not accord with the policy of the United States to acknowl-edge a monarchical government, erected on the ruins of any republican government in America, under the auspices of any European power.

## IN SENATE.

1864, April 5-The resolution was referred to the Committee on Foreign Relations, and it remained unreported at the close of the session.

April 27-Mr. McDougall offered a resolution that the Committee on Foreign Relations be instructed to report to the Senate the joint resolution printed above.

May 28-Mr. McDougall offered a resolution that the committee be discharged from the subject, both of which went over.

## IN HOUSE.

May 23-Mr. H. W. DAVIS offered this resolution, which was agreed to without a division :

Whereas the following announcement appeared in the Moniteur, the official journal of the French Government:

"The Emperor's Govern-

ment has received from that of the United States satisfactory explanations as to the

sense and bearing of the res-olution come to by the House

of Representatives at Wash-

"It is known, besides, that the Senate had indefinitely postponed the examination

of that question, to which in any case the executive power would not have given its sanction."

ington relative to Mexico.

"Le gouvernement de l'Empereur a reçn du gon-vernement des Etats-Unis des explications satisfaisantes sur le sens et la portée de la résolution prise par l'assemblée des représentans à Washington, au sujet des affairs du Mexico.

"On sait, d'ailleurs, que le Sénat avait déjà ajourné indéfinement l'examen de cetto résolution, à laquelle, dans tous les cas, le pouvoir exécutif n'eût point accordé sa sanction."—Moniteur.

Therefore, Resolved, That the President be requested to communi-cate to this House, if not inconsistent with the public interest, any explanations given by the Government of the United States to the Government of France respecting the sense and bearing of the joint resolution relative to Mexico. which passed the House of Representatives unanimously on the 4th of April, 1864.

May 24-The PRESIDENT transmitted the following correspondence, communicated by the Secretary of State, in response to the resolution of the House :

MR. SEWARD'S LETTER TO MR. DAYTON, DATED DEPARTMENT OF STATE, WASHINGTON, APRIL 7, 1864.

I send you a copy of a resolution which passed the House. of Representatives on the 4th instant, by a manimous vote, and which declares the opposition of that body to a recog-nition of a monarchy in Mexico. M. Geofrey has lost no time in asking for an explanation of this proceeding.

It is hardly necessary, after what I have heretofore writ-ten with perfect candor for the information of France, to,

say that this resolution truly interprets the unanimous sen-timent of the people of the United States in regard to Mex-It is, however, another and distinct question whether ico. 1co. It is, however, abother and distinct question whether the United States would think it necessary or proper to express themselves in the form adopted by the House of Representatives at this time. This is a practical and purely Executive question, and the decision of its constitutionality belongs not to the House of Representatives, nor even to Congress, but to the President of the United States. You will, of course, take notice that the declaration made by the House of Representatives is in the form of a joint resolution, which, before it can acquire the character of a legislative act, must receive, first, the concurrence of the Senate, and secondly, the approval of the President of the United States; or, in case of his dissent, the renewed assent of both houses of Congress, to be expressed by a majority of two-thirds of each body. While the President receives the declaration of each body. While the President receives the declaration of the House of Representatives with the profound respect to which it is entitled, as an exposition of its sentiments upon a grave and important subject, he directs that you inform the Government of France that he does not at present con-template any departure from the policy which this Government has hither to pursued in regard to the war which ex-ists between France and Mexico. It is hardly necessary to say that the proceeding of the House of Representatives was adopted upon suggestions arising within itself, and not upon any communication of the Executive department; and that the French Government would be seasonably apprised of any change of policy upon this subject which the President might at any future time think it proper to adopt.

## MR. DAVION'S LETTER, DATED PARIS, APRIL 22, 1864.

STR: I visited M. Drouyn de l'Huys yesterday at the de-partment of foreign affairs. The first words he addressed to me, on entering the room, were: "Do you bring us peace, or bring us war?" I asked him to what he referred, and he or oring us war? I asked nim to what he referred, and he said he referred more immediately to those resolutions recently passed by Congress in reference to the invasion of Mexico by the French, and the establishment of Maximilian upon the throne of that country. I said to him, in reply, that I did not think France had a right to infer that we were about to make war against her on account of anything contained in those resolutions; that they embodied nothing more than had been constantly held out to the Freuch government from the beginning; that I had always repre-sented to the government here that any action upon their part interfering with the form of government in Mexico would be looked upon with dissatisfaction in our country, and they could not expect us to be in haste to acknowledge a monarchical government, built upon the foundations of a republic which was our next neighbor; that I had reason to believe you held the same language to the French Min-ister in the United States. This allegation he did not seem to deny, but obviously viewed the resolutions in question as a serious step upon our part; and I am told that the leading secessionists here build largely upon these resolutions leading secessionists here build largely upon these resolutions as a means of fomenting ill-feeling between this country and some others and ourselves. Mr. Mason and his secretary have gone to Brussels to confer with Mr. Dudley Mann, who is their commissioner at that place. Mr. Slidell, it is said, was to have gone to Austria, although he has not yet got off.

# SECOND LETTER FROM MR. DAYTON, PARIS, MAY 2, 1864, BEING IN REPLY TO MR. SEWARD'S OF APRIL 7.

SIR: Immediately upon the receipt of your dispatch No. 525, I applied to M. Drouyn de l'Huys fort special inter-view, which was granted for Saturday last. I then said that I knew that the French government had felt some anxiety in respect to the resolution which had recently been passed In respect to the resolution which had recently been passed by the House of Representatives, in reference to Mexico; and inasmuch as I had just received a copy of that resolu-tion, together with the views of the President of the United States, I begged, if agreeable, to read to him your dispatch in reference to the latter. To this he assented, and as the shortest and most satisfactory mode, following out my in-turnations I read to bim that there of our dispatch structions, I read to him that entire portion of your dispatch which applies to this subject, stating, at the same time, that I thought it was a remarkable illustration of the frankness and straightforwardness of the President. When the read-ing was closed, M. Drouyn de l'Huys expressed his gratif-cation, and after asking some questions in regard to the effect of laying a resolution upon the table in the Senate; the conversation terminated.

The extreme sensitiveness which was manifested by this government when the resolution of the House of Representatives was first brought to its knowledge has, to a consider-able extent, at least, subsided.

Mr. SEWARD's responses of May 9 and May 21 :

SIR: Your dispatch of April 22 (No. 454) has been received. What you have said to M. Drouyn de l'Huys on the subject of the resolution of the House of Representatives con-

cerning Mexico, as you have reported it, is entirely approved.

The resolution yet remains unacted upon in the Senate. Mr. Corwin was to leave Vera Cruz on the 3d instant under the leave of absence granted to him by this department on the 8th of August last. SIR: I have the honor to acknowledge the receipt of your

dispatch of May 2, (No. 401,) and to approve of your pro-ceedings therein mentioned. We learn that Mr. Corwin, our minister plenipotentiary to Mexico, is at Havana, on his return to the United States, under leave of absence.

June 4-Mr. DAVIS, of Maryland, asked unanimous consent to make a written report from the Committee on Foreign Affairs, in response to the above message of the President. Mr. BROOMALL objected.

June 6—He asked consent, and Mr. ARNOLD objected. He then moved to suspend the rules, which was rejected—yeas 43, nays 55, (two thirds being required,) as follows:

thirds being required,) as follows: YEAS-Mesers. Allison, Ancone, Augustus C. Baldwin, Cox, Henry Winter Davis, Dawson, Eden, Edgerton, Eldridge, Finck, Ganson, Garfield, Grider, Griswold, Harding, Har-rington, Charles M. Harris, Herrick, Holman, Jenckes, King, Knapp, Lazear, Le Blond, Long, Mallory, Marcy, James R. Morris, Morrison, Noble, Odell, Orth, Fendleton, Perry, Robinson, Ross, Scott, Spalding, Strouse, Sweat, Wadsworth, Chilton A. White, Joseph W. White-43. NAYS-Messrs. Alley, Ames, Arnold, Baily, John D. Bald-win, Beaman, Blaine, Jacob B. Blair, Broomall, Ambrose W. Clark, Freeman Clarke, Cobb, Cofroth, Cole, Thomas T. Javis, Dawes, Dixon, Donnelly, Driggs, Eliot, Farnsworth, Fenton, Frank, Gooch, Grinnell, Hale, Hotchkiss, John H. Hubbard, Ingersoll, Kelley, Orlando Kellogg, Littlejohn, Longyear, Marvin, Samuel F. Miller, Moorhead, Daniel Mor-ris, Amos Myers, Charles O'Neill, Patterson, Perham, Price, Alexander H. Rice, John H. Rice, Edward H. Rollins, Shan-non, Smithers, Thayer, Tracy, Upson, Elihu B. Washburne, William B. Washburn, Whaley, Wilson, Windom-55. June 27--Mr. DAVIS made this report, which

June 27-Mr. DAVIS made this report, which was ordered to be printed :

The Committee on Foreign Affairs have examined the correspondence submitted by the President relative to the joint resolution on Mexican affairs with the profound re-spect to which it is entitled, because of the gravity of its subject and the distinguished source from which it emanated.

They regret that the President should have so widely departed from the usage of constitutional governments as to make a pending resolution of so grave and delicate a character the subject of diplomatic explanations. They regret still more that the President should have thought proper to inform a foreign government of a radical and se-rious conflict of opinion and jurisdiction between the de-positories of the legislative and executive power of the United States.

No expression of deference can make the denial of the right of Congress constitutionally to do what the House did with absolute unanimity, other than derogatory to their dignity.

They learn with surprise that, in the opinion of the President, the form and term of expressing the judgment of the United States on recognizing a monurchical governmentim-posed on a neighboring republic is a "purely executive question, and the decision of it constitutionally belongs not to the House of Representatives, nor even to Congress, but to the President of the United States."

This assumption is equally novel and inadmissible. No President has ever claimed such an exclusive authority. No Congress can ever permit its expression to pass without dissent.

It is certain that the Constitution nowhere confers such authority on the President.

The precedents of recognition, sufficiently numerous in this revolutionary era, do not countenance this view; and if there be one not inconsistent with it, the committee have not found it.

All questions of recognition have heretofore been debated An questions of recognition have herefore become detacted and considered as grave questions of national policy, on which the will of the people should be expressed in Congress assembled; and the President, as the proper medium of for-eign intercourse, has executed that will. If he has ever anticipated its expression, we have not found the case.

The declaration and establishment of the independence of the Spanish American colonies first brought the question of recognition of new governments or nations before the Government of the United States; and the precedents then set have been followed ever since, even by the present Administration.

The correspondence now before us is the first attempt to

depart from that usage, and deny the nation a controlling deliberative voice in regulating its foreign policy. The following are the chief precedents on recognition of new governments, and the policy of the United States Gov-

ernment on that topic : On the 9th of February, 1821, Henry Clay moved in the House of Representatives to amend the appropriation bill

by the following clause: "For an outfit, and one year's salary, to such minister as the President, by and with the consent of the Senate, may send to any government of South America, which has established and is maintaining its independency on Spain, a sum not exceeding \$18,000.<sup>37</sup> It failed. On the 10th of February, he moved that the House of Representatives participates with the people of the United

Representatives participates with the people of the United States in the deep interest which they feel for the success of the Spanish provinces of South America, which are strug-gling for their liberty and independence, and that it will give its constitutional support to the President of the United States whenever he may deem it expedient to recognize the sovereignty and independence of any of the said provinces. provinces.

provinces. A motion to amend by the proviso "that this resolution hall not be construed to interfere with the independent exercise of the treaty-making power," and another, "that the House approves of the course heretofore pursued by the President of the United States with regard to the said provinces," were negatived. The resolution was adopted, and a committee appointed to lay it before President Monroe. The committee, on the 17th February, reported— "That the President assured the committee that, in com-mon with the people of the United States and the House of

mon with the people of the United States and the House of Representatives, he felt great interest in the success of the Mepresentatives, he left great interest in the success of the provinces of Spanish America, which are struggling to es-tablish their freedom and independence, and that he would take the resolution into deliberate consideration, with the most perfect respect for the distinguished body from which it had emanated."

So the House of Representatives took the initiative to-wardsrecognizing the new republics. The amendment to the appropriation bill would have been a legislative recognition. The resolution was a formal statement of the opinion of the House to the President, which he did not think beyond their constitutional authority.

At the first session of the next Congress the House of Representatives, on motion of Mr. Nelson, of Virginia, resolved.

"That the President of the United States be requested to lay before this House such communications as may be in the possession of the Executive, from the agents of the United States with the governments south of the United States, which have declared their independence, and the ocommunications from the agents of such governments in the United States with the Secretary of State, as tend to show the political condition of those governments, and the state of the war between them and Spain, as it may be consistent with the public interest to communicate."

estent with the public interest to communicate." President Monroe answered the application on the 8th of March in an elaborate message, of which the following extracts sufficiently show that he did not think recogni-tion "a purely excentive question," and that the decision of it constitutionally belongs, "not to the House of Rep-resentatives, nor even to Congress, but to the President of the United States:" "In transmitting to the House of Representatives the documents called for by the resolution of that House of January, I consider it my duty to invite the attention of *Congress* to a very important subject, and to communicate the sentiments of the Excentive on it; that should *Con-gress* entertain similar sentiments, there may be such co-operation between the two departments of the Government

operation between the two departments of the Government as their respective rights and duties may require." "This contest has now reached such a stage, and been attended with such decisive success on the part of the provinces, that it merits the most profound consideration, whether their right to the rank of independent nations. with all the advantage incident to it in their intercourse with the United States, is not complete." After narrating the events, he proceeds:

"When the result of such a contest is manifestly settled, the new governments have a claim to recognition by other powers which ought not to be resisted."

"When we regard, then, the great length of time which the war has been prosecuted, the complete success which has attended it in favor of the provinces, the present con-dition of the parties, and the utter inability of Spain to produce any change in it, we are compelled to conclude that its fate is sattled, and that the provinces which have declared their indexeduces and are in the misurement of it declared their independence, and are in the enjoyment of it,

ought to be recognized." "In proposing this measure, it is not contemplated to change thereby in the slightest manner our friendly rela-tions with either of the parties."

"The measure is proposed under a thorough conviction "The measure is proposed under a thorough conviction that it is in strict accord with the law of nations, and that the United States owe it to their essential interests, to adopt it." "Should Congress concur in the view herein presented, they will doubless see the propriety of making the neces-sary appropriations for carrying it into effect." It is quite apparent that President Monroe was far from countenancing the opinion that the form and time in which the United States would think it necessary to express

the United States would think it necessary to express themselves on the policy of the recognition is a purely executive question, and that he did not think the decision of it constitutionally belongs not to the House of Representatives, nor even to Congress, but to the President of the United States, to the exclusion of Congress. Had he so thonght, he would have refused the production of the papers, as president Washington did the diplomatic instructions

relative to the English treaty. He would have announced his purpose to recognize the republics, and left it to Congress to provide for diplomatic

For the second s effect.

He consulted the will of the nation at the mouth of Con-gress, and proposed to concur in its execution. So Congress understood him, for the papers and message were referred to the Committee on Foreign Affairs. That

committee considered, in an elaborate report, the question of independence of the republics, the policy and principles involved in their recognition, and on the 19th of March, 1822, they submitted it to the House. It concluded as folfollows:

"Your committee having thus considered the subject referred to them in all its aspects, are unanimously of opin-ion that it is just and expedient to acknowledge the independence of the several nations of Spanish America without any reference to the diversity in the form of their governments; and in accordance with this opinion they respect-

fully submit the following resolutions: "Resolved, That the House of Representatives concur in the opinion expressed by the President in his message of the 8th of March, 1822, that the American provinces of Spain which have declared their independence, and are in the en-joyment of it, ought to be recognized by the United States as independent nations.

"Resolved, That the Committee of Ways and Means be instructed to report a bill appropriating a sum not exceeding \$100,000 to enable the President to give due effect to such recognition."

It is, therefore, equally apparent that the House of Rep-resentatives of the 17th Congress was clearly of opinion with President Monroe that the question of recognition was not purely executive, but that it constitutionally belongs to Congress as well as to the President; that the Legislature declares the will of the United States, which the Executive gives effect to-each concurring in the act of recognition according to their respective constitutional functions.

In obedience to that resolution the following bill, recog-nizing the new nation, was reported and passed and ap-proved on the 4th of May, 1862: "That for such missions to the independent nations on the

American continent as the President may deem proper, there be, and hereby is, appropriated a sum not exceeding \$100,000, to be paid out of any money in the Treasury not otherwise appropriated." The approval of that law completed the recognition of the comprised.

the new nations. The sending ministers to some or all of them was a matter of executive discretion, not at all essenthem was a matter of executive inscretion, not at all essen-tial to or connected with the fact of recognition. Ministers were appointed to Mexico, Colombia, Buenos Ayres, and Chili, on the 27th of January, 1823. None was appointed to Peru till May, 1826; yet it is certain Peru was as much recognized by the United States as the other governments from the 4th of May 1829.

recognized by the United States as the other governments from the 4th of May, 1822. This great precedent has governed all that follow it. The acknowledgment of the independence of Texas stands next in our history. It is a most instructive prece-dent, strictly following the forms observed respecting the overnments of Spanish America. On the 18th of June, 1836, on the motion of Henry Clay,

the Senate adopted the resolution-"That the independence of Texas ought to be acknowl-edged by the United States whenever satisfactory information shall be received that it has in successful operation a

civil government capable of performing the duties and ful-filling the obligations of an independent power." The House of Representatives, on the 4th of July, 1836, adopted a resolution in the same words; and added a

"That the House of Representatives perceive with satis-faction that the President of the United States has adopted measures to ascertain the political, military, and civil condition of Texas."--(Congressional Globe, 1st session 24th | Congress, pp. 453, 486.) Those resolutions were not formal acknowledgments of a

government of Texas; the report of the Sonate Committee showed the circumstances were not sufficiently known; and both Senate and House awaited further information at the hands of the President.

On the 2d December, while communicating the information. President Jackson accepted the oceasion to express to Congress his opinion on the subject. The following passa-ges are very instructive, touching the authority to recognize new States :

"Nor has any deliberative inquiry ever been instituted in Congress, or in any of our legislative bodies, as to whom belonged the power of recognizing a new State; a power, the exercise of which is equivalent, under some circumstances, to a declaration of war; a power nowhere ex-pressly delegated, and only granted in the Constitution, as it is necessarily involved in some of the great powers given to Congress, in that given to the President and Senate to form treaties and to appoint ambassadors and other public ministers, and in that conferred on the President to receive ministers from foreign nations."

"In the preamble to the resolution of the House of Rep-resentatives, it is distinctly intimated that the expediency of recognizing the independence of Texas should be left to the decision of Congress. In this view, on the ground of exthe decision of Congress. In this view, on the ground of ex-pediency, I am disposed to concur; and do not, therefore, think it necessary to express any opinion as to the strict constitutional right of the Executive, either apart from, or in conjunction with, the Senate over the subject. It is to be presumed that on no future occasion will a dispute arise, are a non-based for converse between the Frequeries and as none has heretofore occurred, between the Executive and Legislature in the exercise of the power of recognition. It will always be considered consistent with the spirit of the Constitution and most safe that it should be exercised, when probably leading to war, with a previous understanding with that hody by whom war can alone be declared, and by whom all the provisions for sustaining its perils must be furnished. Its submission to Congress, which represents in one of its branches the States of this Union, and in the other the people of the United States, where there may be reasonable ground to apprehend so grave a consequence, would certainly afford the fullest satisfaction to our own country, and a perfect guarantee to all other countries, of the justice and prudence of the measures which ought to be adopted." After forcibly stating why he thought "we should still stand aloof," he closed with the following declaration:

"Having thus discharged my duty, by presenting with simplicity and directness the views which, after much re-flection, I have been led to take of this important subject, I have only to add the expression of my confidence that ′ if Congress should differ with me upon it, their judgment will be the result of dispassionate, prudent, and wise delibera-tion; with the assurance that, during the short time which I shall continue connected with the Government, I shall promptly and cordially units with you in such measures as may be deemed best fitted to increase the prosperity and perpetuate the peace of our favored country."

The concurrent resolutions of the Senate and House of Representatives, and that message of President Jackson, leave no doubt that the views which presided over the recognition of the South American Governments still prevailed, and that the President was as far from asserting as Congress from admitting that the recognition of new na-tions and the foreign policy of the United States is a purely Executive question.

The independence of Texas was finally recognized in pursuance of the following enactment in the appropriation bill of the second session of the Twenty-Fourth Congress which appropriates money-

"For the outfit and salary of a diplomatic agent to be sent to the republic of Texas, whenever the President of the United States may receive satisfactory evidence that Texas is an independent power, and shall deem it expedient to appoint such minister."

That law was approved by President Jackson. Not only is this exclusive assumption without counte-nance in the early history of the Republic, but it is irrecon-cilable with the most solemn acts of the present Administration. The independence of Hayti is nearly as old as that tion. The independence of Hayti is nearly as out as that of the United States; it antedated that of the South Amer-ican republics, and the republic of Liberia has long been recognized by European nations. Both were first recognized by act of Congress, approved by President Lincoln on the 5th of July, 1862, which enacted—

"That the President of the United States be, and he is hereby, authorized, by and with the advice and consent of the Sente, to apoint diplomatic representatives of the United States to the republics of Hayti and Liberia respectirely. Each of the representatives so appointed shall be accredited as commissioner and consul general, and shall receive the compensation of commissioners," &c., &c. That was a formal recognition of those republics by *law*,

whether the President sent diplomatic representatives or not.

Quite in the spirit of these precedents is the well-consid-ered language of the Supreme Court: "Those questions which respect the rights of a part of a

foreign empire which asserts and is contending for its independence, belong more properly to those who can declare what the law shall be, who can place the nation in such a position with respect to foreign powers as to their own judgment shall appear wise, to whom are entrusted its for-eign relations, than to that tribunal whose power as well as duty is confined to the application of the rule which the But the joint resolution of the 4th of April does more

than declare the refusal of the United States to recognize a monarchical usurpation in Mexico. It declares a general rule of policy, which can be authentically and authorita-tively expressed only by the body charged with the legislative power of the United States. "Resolved, &c., That the Congress of the United States

"*Resolved*, dc., That the Congress of the United States are unwilling, by silence, to leave the nations of the world under the impression that they are indifferent spectators of the deplorable events now transpiring in the republic of Mexico; and they, therefore, think fit to declare that it does not accord with the policy of the United States to acknowledge a monarchical government erected on the ruins of any republican government in America, under the auspices of any European power."

The committee are of opinion that this authority, to speak in the name of the United States, has never, before the correspondence in question, been considered a purely executive function.

The most remarkable declaration of this kind in our history, which events seem now likely to make of as grave practical interest as when it was uttered, is President Mouroe's declaration in his message of the 2d December, 1823: "With the governments which have declared their inde-

pendence and maintained it, and whose independence we have, after great consideration and on just principles acknowledged, we could not view any interposition, for the purpose of oppressing them or controlling in any other manner their destiny by any European power, in any other light than as the manifestation of an unfriendly disposition toward the United States? toward the United States."

But though always the accurate expression of the feelings of the American people, it was not regarded as the settled policy of the nation, because not formally declared by Con-gress. By the administration of President John Quincy Adams, which followed, it was treated as merely an executive expression on behalf of the people, which Congress alone could elevate to the dignity of a national policy by its formal adoption.

In 1826, Mr. Poinsett, the minister to Mexico, having used language supposed to commit the United States to that policy in behalf of Mexico, a resolution was promptly in-troduced into the House of Representatives and adopted on the 27th of March, 1826-

That the Committee on Foreign Affairs inquire and re port to this House upon what authority, if any, the minla-ter of the United States to the Mexican republic, in his offiter of the United States to the Mexican republic, in his offi-cial character, declared to the plenipotentiary of that gov-ernment that the United States have pledged themselves not to permit any other power than Spain to interfere either with their (the South American republics) inde-pendence, or form of government," &c., &c.-2. Cong. Deb., 19th Con., let eess., p. 1820.) Mr. Poinsett hastened to axplain by his letter of the 6th of May, 1826, to Henry Clay, then Secretary of State: "I cannot rest satisfied without stating explicitly that, in the observations I made during my conference with the Mexican plenipotentiaries, I alluded only to the message of the President of the United States to Congress in 1823. "That message, dictated, in my opinion, by the soundes

"That message, dictated, in my opinion, by the soundes policy, has been regarded both in Europe and America as a solemn declaration of the views and intentions of the Executive of the United States, and I have always considered that declaration as a pledge, so far forth as the language of the *President* can pledge the *nation*, to defend the new American republics from the attacks of any of the powers Amorican republics from the attacks of any or the powers of Europe other than Spain. That the people of the United States are not bound by any declarations of the Executive is known and understord as well in Mexico, where the govern-ment is modelled on our own political institutions, as in the United States themselves. But in order to correct any er-roneous impressions these words might have made on the roneous impressions these words might have made on the minds of the Mexican plenpotentiaries. I explained to them in the course of our conference this morning their precise meaning: that the declaration of Mr. Monroe in his message of 1823, to which I had alluded, indicated only the course of policy the Executive of the United States was disposed to pursue towards these countries, but was not binding on the nation unless sanctioned by the Congress of the United States; and when I spoke of the United States having pledged themselves not to permit any other power than Spain to interfere with the independence or form of government of the American republics, I meant only to all de to the above-cited declaration of the President of the

United States in his message of 1823, and to nothing more." This explanation is the more significant from the fact that Mr. Clay's instructions to Mr. Poinsett directed him to that Mr. Clay's instructions to Mr. Poinsett directed him to bring to the notice of the Mexican government the mes-sage of the late President of the United States to their Congress on the 2d of December, 1823, asserting certain important principles of intercontinental law in the rela-tions of Europe and America; and, after stating and en-larging on them, Mr. Clay proceeds: "Both principles were laid down after much and anxious deliberation on the part of the late administration. The President, who then formed a gravt of it, continues entirely to coincide in both, and you will urge upon the government of Mexico the propriety and expediency of asserting the same principles on all proper occasions." occasions."

occasions." And in reply to the resolution of inquiry of the 27th of March, Mr. Clay accompanied his instructions with the de-claration—entirely in the spirit of Mr. Poinsett's letter— "that the United States have contracted no engagement, nor made any pledge to the governments of Mexico and South America, or either of them, that the United States would not permit the interference of any foreign power with the independence or form of government of those nations, \* \* \* \* \* \* \* \* \* \* \* \* \* \*

"If, indeed, an attempt by force had been made by allied Europe to subvert the liberties of the southern nations on this continent, and to erect upon the ruins of their free in-stitutions momarchical systems, the people of the United States would have stood pledged, in the opinion of the Ex-cettice, not to any foreign State, but to themselves and their posterity, by their dearest interests and highest duties, to resist to the utmost such attempt; and it is to a pledge of that character that Mr. Poinsett above refers."—(2 Cong. Debates, 19th Congress, 1st session, App. 83, 84.) Such were the views of the administration of President

John Quincy Adams, whose Secretary of State was Henry Clay, and whose minister to Mexico was Mr. Poinsett, upon the supremacy of the legislature in declaring the foreign policy of the United States, the diplomatic execution and conduct of which is confided to the President.

It is impossible to condense the elaborate message of President Adams of the 15th of March, 1826, dedicated to persuading Congress to concur in and sanction the Panama mission; but that message and the great debate which consumed the session in both houses are unmeaning on the assumptions of this correspondence with the French gov-ernment; and the consideration and approval of its recom-mendations elevate President Monroe's declaration to the dignity and authority of the policy of the nation solemnly and legally proclaimed by Congress.

That message was in reply to a resolution requesting the President to inform the House of Representatives "in re-gard to what objects the agents of the United States are expected to take part in the deliberations of that congress" of Panama.

Among the subjects of deliberation, the President enu-merated the declaration of President Monroe above quoted and on that topic said:

"Most of the new American republics have declared their entire assent to them; and they now propose, among the subjects of consideration at Panama, to take into consider-ation the means of *making effectual* the assertion of that principle, as well as the means of resisting interference from abroad with the domestic concerns of the American governments.

"In alluding to these means, it would obviously be pre-mature at this time to anticipate that which is offered merely as matter for consultation, or to pronounce upon those measures which have been or may be suggested. The purpose of this government is to concur in none which would import hostility to Europe, or justly excite resent-ment in any of her States. Should it be deemed advisable to contract any conventional engagement on this topic, our views would extend no further than to a mutual pledge of the parties to the compact, to maintain the principle in application to its own territory, and to permit no colonial lodgments or establishments of European jurisdiction upon its own soil; and with respect to the obtrustive interference from abroad, if the future character may be inferred from that which has been, and perhaps still is, exercised in more than one of the new States, a joint declaration of its char-acter and exposure of it to the world would be probably of the decreasion would result all that the occasion would require.

"Whether the United States should or should not be parties to such a declaration may justly form a part of the deliberation. That there is an evil to be remedied needs little insight into the secret history of late years to know, and that this remedy may best be considered at the Panama meeting deserves at least the experiment of consideration." Upon this message, after elaborate debates, Congress

23

passed in May an appropriation "for carrying into effect the appointment of a mission to the congress of Panama;" and the President, by and with the advice and consent of the Senate appointed ministers to that congress, and fur-nished them with instructions in conformity with the message, and in execution of the policy approved by Congress.

Accident and delays prevented the arrival of our mission before the dissolution of the congress; but President Adams thought the gravity of the precedent justified him in com-numicating to Congress, in 1829, Mr. Clay's instructions to

municating to congress, in 1829, Mr. Clay's instructions to the ministers for our information; and the precedent re-mains, forever to vindicate the authority of Congress to de-clare and present the forcign policy of the United States. The great name of Daniel Webster is justly considered authoritative on any question of constitutional power; and in that debate, when the enemies of the Administration strove to insert particular instructions to the diplomatic agents sent to that congress, he clearly defined the limits of executive and congressional authority, in declaring tha policy and conducting the negotiations to effectuate it. On the 4th of April, 3226, he is reported to have said in

the House of Representatives :

"He would ask two questions: First, Does not the Constitution vest the power of the Executive in the President? Second, Is not the giving of instructions to ministers abroad an exercise of Executive power? Why should we take this responsibility on ourselves? He denied that the President had devolved, or could devolve, his own constitutional rehad devolved, or could devolve, his own constitutional re-sponsibility, or any part of it, on this house. The President had sent this subject to the House for its concurrence, by voting the necessary appropriation. Beyond this the House was not called on to act. We might refuse the appropria-tion if we saw fit, but we had not the power to make our conditional, and to attach instructions to it.

"There was a way, indeed, in which this House might express its opinion in regard to foreign politics. That is by resolution. He agreed entirely with the gentleman that, if the House were of opinion that a wrong course was given to our foreign relations, it ought to say so, and say so by some measure that should affect the whole, and not a part, of our diplomatic intercourse. It ought to control all missions, and not one only.

"There was no reason why the ministers to Panama should act under these restrictions that did not equally apply to other diplomatic agents-for example, to our minister at Colombia, Mexico, or other new States. A resolution expressive of the sense of the House would, on the contrary, lead to instructions to be given to them all. A resotrary, lead to instructions to be given to them all. A reso-lation was, therefore, the regular mode of proceeding. We saw, for instance, looking at these documents, that our government had declared to some of the governments of Europe, perhaps it has declared to all the principal powers, that we could not consent to the transfer of Cuba to any European power. No doubt the executive government can maintain that ground only so long as it receives the appro-hation and support of Congress. If Congress be of opinion that this course of policy is wrong, then he agreed it was bation and support of Congress. If Congress be of opinion that this course of policy is wrong, then he agreed it was in the power, and, he thought, indeed, the duty of *Congress* to interfere and to express dissent. If the amendment new offered prevailed, the declarations so distinctly unde on this point could not be reported, under any circumstances, at Panama; but they might, nevertheless, be reported any-where and everywhere else. Therefore, if we dissent from this oping, but dissent should he declared by resolution. this opinion, that dissent should be declared by resolution, and that would change the whole course of our diplematic correspondence on that subject in all places. If any gen-tleman thinks, therefore, that we ought to take no measure, under any circumstances, to prevent the transfer of Cuba into the hands of any government, European or American, let him bring forward his resolution to that effect. If it should pass, it will effectually prevent the repetition of such declarations as have been made."—(2 Cong. Debates, 10th Congress, 1st session, pp. 2021, 2022.) This view is, in the opinion of the committee, at once the just view and the traditional practice of the government; the will of the neonle expressed in the lexislative form by

the will of the people expressed in the legislative form by the legislative power can declare authoritatively the foreign policy of the nation; to the President is committed the diplomatic measures for effecting it. The constitutional authority of Congress over the foreign

relations of the United States can hardly be considered an open question, after the concurrent resolutions of Mr. Senator Summer, adopted in the last Congress, it is believed, at the suggestion, certainly with the approval of the Presi-dent, and by him officially notified to foreign governments, as the most authentic and authoritative expression of the national will respecting intervention, mediation, and every other form of foreign intrusion into the domestic struggle in the United States.

The committee are not inclined to discuss theoretically questions of relative power. The Constitution is a practical, and not a theoretical instrument. It has been adminis-tered and construed by men of practical sagacity, and in their hands the voice of the people has been heard authori-

tatively in the executive chamber, on the conduct of foreign affairs

But this correspondence requires us to say, that in view of the historic precedents, it is not a purely executive quea-tion whether the United States would think it necessary to express themselves in the form adopted by the House of Representatives at this time; it does belong to Congress to declare and decide on the foreign policy of the United States, and it is the duty of the President to give effect to that policy by means of the diplomatic negotiations, or military power if it be authorized.

The President is not less bound to execute the national will expressed by law in its foreign than in its domestic concerns.

The President appoints all officers of the United States, but their duties are regulated, not by his will, but by law. He is the commander of the army and navy, but he has no power to use it except when the law points out the das no sion and the object. He appoints foreign ministers, but neither in this case are they, by reason of their appointment, anything but the ministers of the law. If it be true that the appointment of an ambassador to a nation implies the recognition of the nation, it is just as sound logic to argue that none can be appointed to a nation that does not exist by the recognition of Congress, as that the President can recognize alone, because he can appoint. But we prefer to waive the question. We are anxious not to depart from the approved precedents of our history. Our

desire is to preserve, not to change. We will not inquire what would be the effect of a recognition of a new nation by the President against the will of Congress. We prefer to indulge the hope so wisely expressed by President Jack-son, that "it is to be presumed that on no future occasion will a dispute arise, as none has heretofore occurred, be-tween the Executive and Legislature in the exercise of the power of recognition."

Hitherto new nations, new powers, have always been recognized upon consultation and concurrence of the executive and legislative departments, and on the most impor-tant occasions by and in pursuance of law in the particular cas

Changes of the person or dynasty of rulers of recognized powers, which created no new power, have not been treated always with the same formality; but usually the general law providing for diplomatic intercourse with the power whose internal administration had changed remained on the statute book and conferred a plenary discretion on the President, under the sanction of which he has accredited ministers to the new possessors of power. It is not known that hitherto the President has ever undertaken to recognize a new nation or a new power not before known to the history of the world, and not before acknowledged by the United States, without the previous authority of Congress. It is peculiarly unfortunate that the new view of the ex-

ecutive authority should have been announced to a foreign government, the tendency of which was to diminish the force and effect of the legislative expression of what is admitted to be the unanimous sentiment of the people of the United States, by denying the authority of Congress to pronounce it.

Of the prudence of that expression at this time Congress is the best and only judge under the forms of the Constitu-tion, and the President has no right to influence it otherwise than in the constitutional expression of his assent or his dissent when presented to him for his consideration.

It is vain to suppose that such a declaration increases the danger of war with France. The Emperor of the French will make war on the United States when it suits his convenience, and it can be done without danger to his dynas-tic interests. Till then, in the absence of wrong or insult tic interests. on our part, there will be no war. When that time arrives we shall have war, no matter how meek, inoffensive, or pu-sillanimous our conduct may be, for our sin is our freedom and our power, and the only safety of monarchical, impe-rial, aristocratic, or despotic *rule*, lies in our failure or our overthrow.

It postpones the inevitable day to be ready and powerful at home, and to express our resolution not to recognize acts of violence to republican neighbors on our borders perpe-trated to our injury. That declaration will encourage the republicans of America, to resist and endure, and not to submit. It is not perceived how an attack on the United States can promote the establishment of a monarch in Mexico. It might seriously injure us, but it would be an addi-tional obstacle to the accomplishment of that enterprise. It is fortunate that events in Europe, in great measure, embarrass any further warlike enterprise on this continent, emolarrass any article wains out of the or one containing and the ruler who has not thought fit to mingle in the strife of Poland or Schleswig-Holstein will hardly venture to provoke a war with the United States. The committee are content to bide their time, confident

in the fortune and fortitude of the Americau people, but resolved not to encourage by a weak silence complications with foreign powers inimical to our greatness and safety, which, in the words of Mr. Webster, "a firm and timely assertion of what we hold to be our own rights and our own interests would strongly tend to avert."

The committee recommend the adoption of the following

Resoluted, That Congress has a constitutional right to an authoritative voice in declaring and prescribing the foreign policy of the United States, as well in the recognition of new powers as in other matters; and it is the constitutional duty of the Prosident to respect that policy, not less in di-plomatic negotiations than in the use of the national force when authorized by law; and the propriety of any declara-tion of foreign policy by Congress is sufficiently proved by the vote which pronounces it; and such proposition while pending and undetermined is not a fit topic of diplomati explanation with any foreign power.

FRANCE, MEXICO, AND THE UNITED STATES.

The Courier du Dimanche, of Paris, publishes a circular letter addressed by M. Drouyn de l'Huys, the French Minister of Foreign Affairs, to the agents of the Empire abroad, respecting the relation of France to the American Government. This letter is a sequel to the correspondence between Mr. Seward and Mr. Dayton with regard to the Mexican question, and is as follows:

PARIS, May 7, 1864 .- Mr. Dayton has called on me to read to me a despatch addressed to him by the Secretary of State of the Union, in order to define the responsibility of the Government of Washington, and to show that a or the government of washington, and to show that a vote of the House of Representatives, or of the Senate, or even of the two Houses, while it naturally recommends itself to the attention of the Government, did not oblige it to modify its policy and take from it its liberty of action.

Mr. Seward sees no reason to follow in the Mexican ques tion a line of conduct other than that which he had adopted heretofore; and if his disposition should happen to hemod-ified, we should be directly and in good time informed of this resolution and its motives.

Ins resolution and its motives. I have replied to Mr. Dayton that in the opinion of the Government of the Emperor, nothing could justify this change; that our confidence in the wisdom and enlighten-ment of the American Cabinet was too great to permit us to suppose it to have any idea of compromising, by thought less action, the true interests of the United States. While American Colored and the American the Mr.

While expressing to Mr. Dayton the entire satisfaction which the assurances he was charged with giving to us caused to the government of the Emperor, I added that I thought, in effect, that, even from the point of view of the United States, the choice would not be doubtful between the establishment in Mexico of a stable and regular gov-erument, and the perpetuation of an anarchy of which they had been the first to suffer and to point out the great inconvenience.

The reorganization of a vast country which, after the The reorganization of a vast country which, after the restoration of order and security, is expected to play an important economical part in the world, would be for the United States especially a real source of advantage, since it would open a new market to them from which they, be-cause of their proximity, would profit more than others. The prosperity of Mexico would therefore agree with his widdher a lower data the state of the state of the state.

their rightly understood interests, and I certainly do not believe that the Government of Washington could misunderstand this truth.

DROUYN DE L'HUYS.

# The Arguelles Case.

1864, May 28-In the SENATE, Mr. JOHNSON offered this resolution, which was agreed to :

Resolved, That the President be requested to inform the Senate, if he shall not deem it incompatible with the public interest, whether he has, and when, authorized a person alleged to have committed a crime against Spain or any of its dependencies, to be delivered up to officers of that gov-ernment; and whether such delivery was had; and if so, under what authority of law or treaty it was done.

May 31-The PRESIDENT transmitted a reply covering a report from the Secretary of State and other documents, by which it appears that Don Josè Augustin Arguelles, an officer in the Spanish army in Cuba, had captured a slave expedition, while he was acting as Lieutenant Governor of the district of Colon, in Cuba. It

4.0

was subsequently discovered that he had, with | the connivance of the curate of Colon, made representations to the Spanish Government that one hundred and forty-one of the recap-tured negroes had died of small-pox, though, in fact, he had sold them into slavery, and succeeded in escaping to the United States, where he was arrested by the officers of the United States and surrendered to the Cuban authorities.

In explanation of this act on the part of the Government of the United States, the Secretary of State reports as follows :

DEPARTMENT OF STATE, WASHINGTON, May 30, 1864. The Secretary of State, to whom was referred the resolu-tion of the Senate of the 28th instant, requesting the Presi-dent to inform that body, "if he shall not deem it incom-patible with the public interest, whether he has, and when,

attible with the public, where whether he has, and when, anthorized a person, alleged to have committed a crime gainst Spain, or any of its dependencies, to be delivered up to officers of that government; and whether such deliv-ery was had; and if so, ander what authortly of law or of treaty it was done," has the honor to submit to the Presi-dent a copy of the papers which are on file or on record in this department relative to the subject of the resolution. By the act of Congress of the 15th of May, 1820, the Af-fican slave trade is declared to be piracy. By the ninth article of the treaty of 1842 with Great Fortian, it is stipulated that, "Whereas, notwithstanding all efforts which may be made on the coast of Africa for sup-pressing the slave trade, the facilities for carrying on that traffic, and avoiding the vigilance of cruisers, by the fraud-ulent use of flags and other means, are so great, and the four slaves, so strong, as that the desired result may be ong delayed, unless all market be hord four shorts the pur-chase of African negroes, the parties to this treaty agree chase of African negroes, the parties to this treaty agree that they will unite in all becoming representations and remonstrances with any and all powers within whose domininconstrainces with any and an powers within whose domin-ions such markets are allowed to exist, and that they will urge upon all such powers the propriety and duty of closing such markets effectually at once and forever.<sup>37</sup> There being no treaty of extradition between the United States and Spain, nor any act of Congress directing how fugitives from justice in Spanish dominions shall be de-

livered up, the extradition in the case referred to in the resolution of the Senate is understood by this department to have been made in virtue of the law of nations and the Constitution of the United States.

Although there is a conflict of authorities concerning the expediency of exercising comity towards a foreign government by surrendering, at its request, one of its own sub-jects charged with the commission of crime within its territory, and although it may be conceded that there is no national obligation to make such a surrender upon a denand therefor, niles its acknowledged by treaty or by statute law, yet a nation is never bound to furnish asylum to dangerous criminals who are offenders against the hu-man race; and it is believed that if in any case the comity could with propriety be practiced, the one which is under-stood to have called forth the resolution furnished a just occasion for its exercise.

Respectfully submitted. To the PRESIDENT.

WILLIAM H. SEWARD.

Upon the arrival at Havana of the fugitive Arguelles, General Dulce, the Captain-General

of Cuba, addressed to the Spanish minister at Washington the following note :

MOST EXCELLENT SIR : My aide-de-camp, with the person

expected, arrived in the steamer Eagle. I request your excellency to render thanks in my name to Mr. Seward for the service which he has rendered to to hir. Seward for the service which he has rendered to humanity by furnishing the medium through which a great number of human beings will obtain their freedom, whom the desertion of the person referred to would have reduced to slavery. His presence alone in this island a very few hours has given liberty to eighty-six.

I also render thanks to your excellence for the effciency of your action.

God preserve your excellency many years. DOMINGO DULCE.

HAVANA, May 19, 1864.

Murray, United States Marshal for the Southern District of New York, was indicted by the grand jury for the arrest of Arguelles, on a charge of kidnapping, and the matter came up before the General Sessions in New York, upon a petition of the counsel for defendant to transfer the case to the United States District Court, on the ground that under the act of Congress passed March 3, 1863, relating to habeas corpus, &c., any officer of the Federal Government, exposed to criminal prosecution, might remove the case in the Federal courts by filing a petition setting forth that the offence charged was done by the authority of the President or the Congress of the United States.

July 6-Judge Russell and Recorder Hoffmann concurring, decided that the petition to remove the case to the United States courts must be denied, the State courts having jurisdiction thereof. The counsel for the defence then made a motion to quash the indictment.

# IN HOUSE.

June 6-Mr. Cox offered this resolution:

Resolved, That the recent extradition of a Spanish sub-ject, by the action of the Chief Executive of the United States, in the absence of a law or treaty on the subject, was a violation of the Constitution of the United States and of the law of nations, and in derogation of the right of asylum, which has ever been a distinguishing feature of our political system.

The House refused to second the demand for the previous question-yeas 38, nays 57.

It was then referred to the Committee on the Judiciary-yeas 72, nays 43. The NAYS were-

While-43.

# THE FINANCES.

# Our Financial Legislation.

'he financial legislation has been as follows : 60, December 17-Authorized an issue of \$10,000,000 in ASURY NOTES, to be redeemed after the expiration of one from the date of issue, and bearing such a rate of in-st as may be offered by the lowest bidders. Authority given to issue these notes in payment of warrants in r of public creditors at their par value, bearing six per interest per annum.

61, February 8-Authorized a LOAN of \$25,000,000, bearinterestat a rate not exceeding six per cent. Per annum, reimbursable within a period not beyond twenty years less than ten years. This loan was made for the pay-t of the current expenses, and was to be awarded to the favorable bidders.

arch 2-Authorized a LOAN of \$10,000,000, bearing intert a rate not exceeding six per cent. per annum, and reursable after the expiration of ten years from July 1,

. In case proposals for the loan were not acceptable, ority was given to issue the whole amount in TREASURY is, bearing interest at a rate not exceeding six per cent. Anthonity was also given to substitute TREAS-Authority was also given to substitute TERAS-NOTES for the whole or any part of the loans for which Secretary was by law authorized to contract and issue is, at the time of the passage of this act, and such treas-notes were to be made receivable in payment of all pub-ues, and redeemable at any time within two years from the 2, 1861. arech 2—Authorized an issue, should the Secretary of Treasury deem it expedient, of \$2,800,000 in coupon is, bearing interest at the rate of six per cent. per an-, and redeemable in twenty years, for the payment of mess incurred by the Territories of Washington and yon in the suppression of Indian hostilities during the s 1855-'56. Up 17—Authorized a loan of \$250,000,000, for which al be issued BOXB bearing interest at a rate not exceed-ly per cent. per annum, irredeemable for twenty years, after that redeemable at the pleasure of the United est.

LEASURY NOTES bearing interest at the rate of 7.30 per . per annum, payable three years after date; and nited States NoTES without interest, payable on demand, ac extent of \$50,000,000. (Increased by act of February 862, to \$60,000,000.)

ie bonds and treasury notes to be issued in such pro

ions of acid treasury works to be issued in such pro-ions of acid as the Secretary may deem advisable. ugust 5—Authorized an issue of nonns bearing 6 per . interest per annum, and payable at the pleasure of the red States after twenty years from date, which may be ad in exchange for 7.30 treasury notes; but no such is to be issued for a less sum than \$500, and the whole which works bench part to arread the whole empower of unt of such bonds not to exceed the whole amount of

unt of such bonds not to exceed the whole amount of treasury notes issued. 62, February 25—Authorized the issue of \$150,000,000 gal tender United States Nones, \$50,000,000 of which to a lieu of demand notes issued under act of July 17, 1861, 000,000 in 6 per cent. bonds, redeemable after five years, payable twenty years from date, which may be ex-ged for United States notes, and a temporary loan of 000,000 in United States notes for not less than thirty , payable after ten days' notice at 5 per cent. interest annum.

anuum. arch 17-Authorized an increase of TEMPORARY LOANS of 000,000, bearing interest at a rate not exceeding 5 per . per annum.

ily 11-Authorized a further increase of TEMPORARY

LOANS of \$50,000,000, making the whole amount authorized \$100,000,000.

\$100,000,000. March 1—Authorized an issue of CERTIFICATES OF INDEET-EDNESS, payable one year from date, in settlement of andited claims against the Government. Interest 6 per cent, per annum, payable in gold on those issued prior to March 4, 1863, and in lawful currency on those issued on and after that date. Amount of issue not specified. 1862, July 11—Authorized an additional issue of \$150,000,-000 legal tender NOTES, \$35,000,000 of which might be in de-nominations less than five dollars. Fifty million dollars of this issue to be reserved to pay temporary loans promptly

this issue to be reserved to pay temporary loans promptly

July 17—Authorized an issue of NOTES of the fractional July 17—Authorized an issue of NOTES of all dues. except but of one dollar, receivable in payment of all dues, except customs, less than five dollars, and exchangeable for United States notes in sums not less than five dollars. Amount of issue not specified.

1863, January 17—Authorized the issue of \$100,000,000 in United States NOTES for the immediate payment of the army and navy; such notes to be a part of the amount provided for in any bill that may hereafter be passed by this Con-gress. The amount in this resolution is included in act of Numb 2, 1860 March 3, 1863.

gress. The amount in this resolution is included in act of March 3, B463. March 3, B463. Ston,000,000 for the next fiscal year, for which could be is-sued bonds running not less than ten nor more than forty years, principal and interest payable in coin, bearing inter-est at a rate not exceeding 5 per cent. per annum, payable on bonds not exceeding \$100, annually, and on all others semi-annually. And TEASERY NOTES (to the amount of \$400,000,000) not exceeding three years to run, with interest at not over 6 per cent. per annum, principal and interest payable in lawful money, which may be made a legal tender for their face value, excluding interest, or convertible into United States notes. And a further issue of \$150,000,000 in United States notes. And a further issue, if necessary, for the payment of the army and navy, and other creditors of the Government, of \$150,000,000 in United States Nortes, which amount includes the \$100,000,000 authorized by the joint resolution of Congress, January 17, 1863. The whole amount of bonds, treasury notes, and United States notes issued under this act not to exceed the sum of \$900,000,000 in FLACTIONAL CURRENCY, (in lieu of postage or other stamps).

in FRACTONAL CURRENCY, (in lieu of postage or other stamps,) exchangeable for United States notes in sums not less than three dollars, and receivable for any dues to the United States less than five dollars, except duties on imports. The whole amount issued, including postage and other stamps issued as currency, not to exceed \$50,000,000. Authority was given to prepare it in the Treasury Department, under the sumervision of the Scenterv

per annum, payable annually on bonds not over \$100, and on all others semi-annually. These bonds to be exempt from taxation by or under State or municipal authority. 1864, June 30—Authorized a LOAN of \$400,000,000, for which may be issued bonds, redeemable after five nor more than thirty years, or if deemed expedient, made payable at any period not more than forty years from date—interest not exceeding six per cent. semi-annually, in coin. Secre-tary of the Treasury is authorized to dispose of these bonds; 356

356

or any part, and of the remainder of the five-twenties, in the United States or in Europe, on such terms as he may deem most advisable, for lawful money of the United States, or, at his discretion, for Treasury notes, certificates of indebtedness, or certificates of deposit issued under any act of Con-gress. And all bonds, Treasury notes, and other obligations of the United States shall be exempt from taxation by or of the United States shall be exempt from taxation by of under State or municipal authority. In lieu of an equal amount of bonds, not exceeding \$200,000,000, the Sccretary is authorized to issue Treasury notes of not less than \$10, payable at any time not exceeding three years, or, if thought expedient, redeemable at any time after three years, at an expendent, redemands at any time after time yeaks, at an interest not exceeding seven and three tenths per-cent., payable in lawful money at maturity, or, at the discretion of the Secretary, semi-annually. And the said Treasury notes may be disposed of by the Secretary of the Treasury, on the best terms that can be obtained, for lawful money; and such of them as shall be made payable, principal and and such of them as shall be made payable, principal and interest, at maturity, shall be a legal tender to the same extent as United States notes for their face value, oxcluding interest, and may be paid to any creditor of the United States at their face value, excluding interest, or to any creditor willing to receive them at par, including interest, and any Treasury notes issued under the authority of this act may be made convertible, at the discretion of the Secreact may be made convertible, at the discretion of the Secre-tary of the Treasury, into any bonds issued under the au-thority of this act. And the Secretary of the Treasury may redeem and cause to be cancelled and destroyed any Treas-ury notes or United States notes heretofore issued under authority of previous acts of Congress, and substitute, in lieu thereof, an equal amount of Treasury notes such as are authorized by this act, or of other United States notes: *Pro-ided* That the total amount of Treasury notes under a super-scient the states and the supersonance of the supersonance of the super-sonance of the supersonance of the supersonance of the super-sonance of the supersonance of the *athorized* by the first and second sections of this act shall not exceed \$400,000,000 in addition to the amounts heretofore issued; nor shall the total amount of United States notes, issued or to be issued, ever exceed \$400,000,000, and such additional sum, not exceeding \$50,000,000, as may be temporarily required for the redemption of temporary loan; nor shall any Treasury note bearing interest, issued under his act, be a legal tender in payment or redemption of any notes issued by any bank, banking association, or banker, calculated or intended to circulate as money.

SEC. 3. That the interest on all bonds heretofore issued, payable annually, may be paid semi-annually; and in lieu of such bonds authorized to be issued, the Secretary of the of such bonds authorized to be issued, the Secretary of the Treasury may issue bonds bearing interest payable semi-annually. And he may also issue in exchange for Treasury notes heretofore issued bearing seven and threa-tenths per centum interest, besides the six per cent honds heretofore authorized, like bonds of all the denominations in which said Treasury notes after maturity shall be paid in lawful money, and they may be exchanged for such bonds at any time within threemonths from the date of notice of redemution within three months from the date of notice of redemption by the Secretary of the Treasury, after which the interest on such Treasury notes shall case. And so much of the law approved March 3, 1864, as limits the loan authorized therein to the current fiscal year, is hereby repealed; and the authority of the Sceretary of the Treasury to borrow The authority of the Secretary of the Treasury to borrow money and issue therefor bonds or notes conferred by the first section of the act of March 3, 1863, entitled "An act to provide ways and means for the support of the Govern-ment," shall cease on and after the passage of this act, except so far as it may effect [affect] \$75,000,000 of bonds blood not dusting already advertised.

## SPECIAL WAR INCOME TAX.

Resource, dc., That, in addition to the income duty alrea-dy imposed by law, there shall be levied, aasessed, and col-lected on the first day of October eighteen hundred and sixty-four, a special income duty upon the gains, profits, or income for the year ending the thirty-first day of Decem-ber next preceding the time herein named, by levying, as-sessing, and collecting said duty of all persons residing within the United States, or of citizens of the United States residing abroad, at the rate of five per centum on all sums exceeding six hundred dollars, and the same shall be levied, according of minuted to collected as and such that the level, assessed, estimated, and collected as to the rates, according to the provisions of existing laws for the collec-tion of an income duty, annually, where not inapplicable hereto; and the Secretary of the Treasury is hereby authorized to make such rules and regulations as to time and thorized to make such rules and regulations as to time and mode, or other matters, to enforce the collection of the special income duty herein provided for, as may be neces-sary: *Provided*. That in estimating the annual gains, pro-fits, or income, as aforesaid, for the foregoing special income duty, no deductions shall be made for dividends or interest received from any association, corporation, or company, nor shall any deduction be made for any salary or pay received.

This resolution passed the House, July 4 yeas 53, nays 47, as follows:

YEAS-Messis. Alley, Allison, Ames, Arnold, Ashley,

Baxter, Boutwell, Boyd, Cobb, Cole, Creswell, Dawes, Dem-ing, Dixon, Driggs, Eckley, Ekot, Garfield, Gooch, Higby, Hooper, Hotchkiss, J. H. Hubbard, Ingersoll, Jonckes, Julian, Kelley, Longyear, McClurg, Moorhead, Morrill, D. Morris, Leonard Myers, Norton, Charles O'Neill, Perham, Alexander H. Rice, John H. Rice, Edward H. Rollins, Schenck, Shannon, Sloan, Smith, Smithers, Francis Thomas, Tracy, Upson, Ellihu B. Washburno, William B. Washburn, Webster, Wilder, Willson, Windom, Woodbridg\_-53. Nars-Messrs. Ancona, Baily, Bliss, Brooks, Chanler, Coffroth, Cox, Dawson, Denison, Eden, Edgerton, Eldridge, Harris, Hutchins, William Johnson, Kernan, Knapp, Law, Harris, Hutchins, William Johnson, Kernan, Knapp, Law, H. Killer, James R. Morris, Morrison, Noble, Odetl, Johnson, James S. Rollins, Ross, Scofield, John B. Steele, William G. Steele, Stovens, Stiles, Sweat, Ward, Williams, Winfield -41.

Seventy-eight absentces.

It passed the Senate same day-yeas 28, nay 7, as follows:

YEAS-Messrs. Anthony, Clark, Conness, Cowan, Doo-little, Foot, Foster, Hale, Harlan, Harris, Hicks, Howe, Johnson, Lane of Iodiana, Lane of Kansas, *McDougall*, Mor-gan, Morrill, Pomeroy, Ramsey, Sherman, Sunner, Ten Eyck, Trumbull, Van Winkle, Wilkinson, Willey, Wilson -98 -28

NATS-Messrs. Buckalew, Carlile, Davis, Powell, Richardson, Riddle, Saulsbury-7.

# "Legal Tenders."

# Second Session, Thirty-Seventh Congress.

1862, Feb. 6--The House came to a vote on two propositions for the issue of \$150,000,000 in Treasury notes.

The one, for notes without interest, and denomination not below \$5-of which \$50,000,000 should be in lieu of so many of the "Demand notes"-to be receivable for all duties, imposts, excises, debts, and demands of every kind due to the United States, and all salaries, &c., from the United States, "and shall also be lawful money, and a legal tender in payment of all debts, public and private, within the United States," to be exchangeable for twenty year six per cent. bonds, interest payable semi-annually, or five year seven per cent. bonds with interest payable semi-annually in coin. Such United States notes to be received the same as coin in pay-ment for loans. Five hundred millions of bonds authorized, payable after twenty years, at six per cent. interest payable semi-anuually.

The other, offered as an amendment, authorized \$100,000 of Treasury notes at 3.65 per cent. per annum, payable in two years, to be receivable for all public dues except duties on imports, and for all salaries, debts, and de-mands owing by the United States to individuals, corporations, and associations within the United States, at the option of such individuals, corporations, and associations, exchangeable for 7.30 bonds with interest payable semiannually in coin, and receivable the same as coin in payment of loans. \$500,000,000 of bonds authorized—\$200,000,000 at 7:30 interest payable semi-annually in coin, and redeemable after ten years, and \$300,000,000, redeemable after twenty-four years, at 6 per cent., payable semi-annually in coin.

The latter was rejected—yeas 55, nays 95, as follows :

YEAS-Messrs. Ancona, Baxter, Biddle, George H. Browne, William G. Brown, Cobb, Frederick A. Conkling, Roscoe Conkling, Conway, Corning, Cox, Cravens. Cristeld, Crit-tenden, Diven, Eliot, English, Goodwin, Grider, Harding, Holman, Horton, Johnson, Law, Lazear, Lovejoy, May,

ies, Justin S. Morrill, Morris, Nixon, Noble, Norton, m. Oddil, Pendleton, Perry, Pomeroy, Porter, Edward ollins, Scdgwick, Skeffeld, Shick, William G. Skele, ton, Benjamin F. Thomas, Francis Thomas, Train, mdigham, Wadsworth, E. P. Walton, Ward, Wobster, on A. While, Wright-55.
Ts-Messrs. Aldrich, Alley, Arnold, Ashley, Babbitt, unith F. Baliey, Joseph Baily, Baker, Beaman, Bing-Francis P. Blair, Jacob B. Blair, Sanuel S. Blair, S. Minton, Burnham, Campbell, Chamberlain, Clark, X, Cutter, Davis, Delano, Delaplaine, Duell, Lundap, Heigerton, Edwards, Ely, Fenton, Fessenden, Fisher, Thot, Frank, Gooch, Granger, Gurley, *Haight*, Hale, bett, Harrison, Hickman, Hooper, Hutchins, Julian, y, Francis W. Kellogg, William Kellogg, Killinger, uder H. Rice, John H. Rice, Richardson, James S. Na, Sargent, Shanks, Shellabarger, Sherman, Sloan, ding, John E. Skele, Steveus, Timble, Trowbridge, Q, van Horn, Van Valkenburgh, Yan Wyck, Verree, Wallace, Charles W. Walton, Whaley, Albert S. e, Wickliff, Wilson, Windom, Worcester-95.
10. affirmative vote on the passage of the

ne affirmative vote on the passage of the er was the same as the negative on the ndment except that Messrs. W. G. Brown Crittenden, who voted aye on the amendt, did not vote on the passage of the bill; rs. Dunlap, Knapp, Richardson, and Wick-who voted nay on the amendment, voted on the passage; Messrs. Mallory, Robinson, Voorhees, who did not vote on the amendt, voted nay on the passage; Mr. Nugen, voted aye on the amendment, voted aye on passage; and Messrs. Dunn and Riddle, did not vote on the amendment, voted aye he passage.

## IN SENATE.

ebruary 12-The Committee of Finance mmended instead of making these notes reable for all demands due to, and all deds owing by, the United States, this substi-

d such notes herein authorized shall be receivable in ent of all public dues and demands of every descrip-and of all claims and demands against the United s of every kind whatsoever, except for interest upon s and notes, which shall be paid in coin.

r. SHERMAN moved to include with these s, "the notes authorized by the act of July 1861;" which was agreed to.

he amendment as amended was agreed to. eb. 13-Mr. Collamer moved to strike out e words:

d such notes herein authorized and the notes author-by the act of July 17, 1861, shall be receivable in pay-of all public dues and demands of every description, of all claims and demands against the United States ery kind whatsoever, except for interest upon bonds notes, which shall be paid in coin, and shall also be il money and a legal tender in payment of all debts, c and private, within the United States, except inter-s aforce aid s aforesaid.

'hich was rejected-yeas 17, nays 22, as ws:

As-Messrs. Anthony, Bayard, Collamer, Cowan, Fes-en, Foot, Foster, Kennedy, King, Latham, Nesmith, ce, Powell, Sauksbury, Simmons, Thomsone, Willey-17. Ns-Messrs. Chandler, Clark, Davis, Dixon, Doolittle, an, Harris, Henderson, Howard, Howe, Lane of Indi-McDougall, Morrill, Pomeroy, Rice, Sherman, Sumner, Eyck, Wade, Wilkinson, Wilson of Massachusetts, on of Missouri-22.

r. DOOLITTLE moved to amend the bill so as take the notes a legal tender "in payment Il public debts, and all private debts herer contracted within the United States ;" ch was rejected without a division. ch was rejected without a division. r. KING offered an amendment, proposing, Congress. The condition of the Treasury certainly renders immedi-ate action on the subject of affording provision for the e<sub>4</sub>;

among other things, to strike out the legal tender clause; but it was rejected without a division.

The bill was then passed-yeas 30, nays 7, as follows :

YEAS-Messrs. Anthony, Chandler, Clark, Davis, Dixon, Doolittle, Fessenden, Foot, Foster, Grimes, Hale, Harlan, Harris, Henderson, Howard, Howe, Lane of Indiana, La-tham, McDougall, Morrill, Pomeroy, Rice, Sherman, Sum-ner, Ten Eyck, Trumbull, Wade, Wilkinson, Wilson of Massachusetts, Wilkon of Missouri-30. NAYS-Messrs. Collamer, Cowan, Kennedy, King, Pearce, Ponell Scruishuru-5.

Powell, Saulsbury-7.

Feb. 20-In the House, the question being on concurring in the amendment of the Senate making the interest upon bonds and notes payable in coin,

Mr. STEVENS moved to include also "payments to officers, soldiers, and sailors, in the Army and Navy of the United States, and for all supplies purchased for the said Government;' which was rejected-yeas 67, nays 72.

The amendment of the Senate, making interest payable in coin was then concurred in-yeas 88, nays 55, as follows:

Ses, nays 55, as follows:
 YEAS-Messrs. Ancona, Arnold, Ashley, Baxter, Beaman, Biddle, Jacob B. Blair, George H. Browne, William G. Brown, Burnham, Catteert, Clements, Cobb, Frederick A. Conkling, Roscoe Conkling, Conway, Covode, Cox, Cravens, Crittenden, Diven, Dunden, Dunn, Eliot, English, Goodwin, Grider, Garley, Haight, Hall, Harding, Hoiman, Horton, Johnson, Kelley, Knapp, Law, Leary, Lehman, Loomi, Grider, Garley, Kalip, Law, Leary, Lehman, Horton, Johnson, Kelley, Knapp, Law, Leary, Lehman, Korne, Crittenden, Diven, Dunden, Dunn, Eliot, English, Goodwin, Grider, Garley, Haight, Hall, Harding, Hoiman, Horton, Johnson, Kelley, Knapp, Law, Leary, Lehman, Korander, H. Rice, Riddle, Robinson, Edward H. Rollins, James S. Rollins, Sargent, Sedgwick, Sheffield, Sherman, Shiel, Smith, John E. Steele, William G. Steele, Stratton, Bealjamin F. Thomas, Francis Thomas, Train, Trimble, Vallandigham, Yibbard, Voorlees, Charles W. Walton, E. P. Walton, Yard, Washburne, Webster, Whaley, Wheeler, Wickliffe, Woodruff, Wright-8.
 Navs-Messrs, Aldrich, Alley, Babbit, Joseph Baily, Baker, Bingham, Francis P. Blair, Samuel S. Blair, Blake, Buffinton, Campbell, Chamberlain, Clark, Davis, Dawes, Duell, Edwards, Ely, Fenton, Fessenden, Fisher, Franchot, Frank, Granger, Hale, Hanchet, Harrison, Hickmeth, Hooper, Julian, William Kellogg, Killinger, Lansing, McPherson, Marston, Maynard, Moorhead, Anson P. Morrill, Novell, Olin, John H. Rice, Shanks, Sloan, Spaulding, Stevens, Trowbridge, Van Horn, Van Valkenburgh, Verree, Wal, Wallace, Albert S. White, Wilson, Windom, Worcester-55.

Other amendments were non-concurred in, and a Committee of Conference agreed upon the bill as it became a law.

One feature of this report was to provide that the Treasury notes issued under the bill should not be a legal tender in payment of duties, and the duties on imports, made payable in coin, should be pledged for the payment of interest on the bonds.

The report was agreed to in the House-yeas 97, nays 22. The NAYS were-

Messrs. Baker, Biddle, Buffinton, Coz, Edwards, English, Haight, Hooper, Johnson, Justin S. Morrill, Odell, Peraleton, Perry, Pike, Robinson, Sheffield, William G. Steele, Van Wyck, Voorhees, Wickliffe, Wood, Woodruff-22.

The Senate concurred without a division.

While this question was pending before the Committee of Ways and Means, the Secretary of the Treasury, Mr. CHASE, addressed them a letter, from which this is an extract :

TREASURY DEPARTMENT, January 29, 1862. SIR: I have the honor to acknowledge the receipt of a resolution of the Committee of Ways and Means, referring me to House bill No. 240, and requesting my opinion as to the propriety and necessity of its immediate passage by

penditures of the Government, both expedient and neces-sary. The general provisions of the bill submitted to me, seem to me well adapted to the end proposed. There are, however, some points which may, perhaps, be usefully amended.

The provision making United States notes a legal tender has doubtless been well considered by the committee, and their conclusion needs no support from any observation of mine. I think it my duty, however, to say, that in respect to this provision my reflections have conducted me to the same conclusions they have reached. It is not uuknown same conclusions they have reacned. It is not unknown to them that I have feit, nor do I wish to conceal that I now feel, a great aversion to making anything but coin a legal tender in payment of debts. It has been my anxious wish to avoid the necessity of such legislation. It is, however, to a provide the payment of debts. It has been my anxious wish to avoid the necessity of such legislation. It is, however, at present impossible, in consequence of the large expendi-tures entailed by the war, and the suspension of the banks, therefore, become indispensably necessary that we should resort to the issue of United States notes. The making them a legal tender might, however, still be avoided if the willingness manifested by the people generally, by railroad companies, and by many of the banking institutions, to receive and pay them as money in all transactions were absolutely or practically universal; but, unfortunately, there are some persons and some institutions which refuse to re-ceive and pay them, and whose action tends not merely to cerve and pay them, and whose action tends not merely to the unnecessary depreciation of the notes, but to establish discrimination in business against those who, in this mat-ter, give a cordial support to the Government, and in favor of those who do not. Such discriminations should, if pos-sible, be prevented; and the provision making the notes a legal tender, in a great measure at least, prevents it, by put-ting all citizens, in this respect, on the same level both of rights and dutios.

rights and duties. The committee, doubtless, feel the necessity of accom-panying this measure by legislation necessary to secure the highest credit as well as the largest currency of these notes. This security can be found, in my judgment, by proper pro-tisions for funding them in interest bearing bonds, by well-guarded legislation authorizing banking associations with circulation based on the bonds in which the notes are funded, and by a judicious system of adequate taxation, which will not only create a demand for the notes, but-by securing the prompt payment of interest-raise and sustain the credit of the bonds. Such legislation, it may be hoped, will divest the legal lender clause of the bill of injurious tendencies, and secure the earliest possible return to a sound currency of coin and promptly convertible notes. I beg leave to add that vigorous military operations and

the unsparing retrenchment of all necessary expenses will also contribute essentially to this desirable end.

I have the honor to be, with very great respect, yours sulv. S. P. CHASE. truly.

Hon. THADDEUS STEVENS, Chairman.

During the pendency of this question,

1862, January 15-Mr. CORNING offered this joint resolution :

Resolved, &c., That in order to pay the ordinary expenses of the Government, the interest on the national loans, and have an ample sinking fund for the ultimate liquidation of the public debt, a tax shall be imposed, which shall, with the tariff on imports, secure an annual revenue of not less than \$150,000,000.

Which was adopted—yeas 133, nays 6. The NAYS were :

Messrs. Allen, Norton, Robinson, Shiel, Voorhees, Wood-6.

The following Democrats voted aye :

The following Definition States worked aye: Messrs. Ancona, Joseph Baily, Charles J. Biddle, G. H. Browne, C. B. Cutvert, G. T. Cobb, E. Corning, S. S. Coa, J. W. Cristield, J. J. Crittenden, J. E. English, H. Grider, H. Haight, A. Harding, W. S. Holman, P. Johnson, J. Lazear, W. E. Lehman, R. Mallory, J. W. Menzies, J. R. Morris, W. P. Noble, J. W. Noell, R. H. Nugen, Odell, G. H. Pen-deton, N. Perry, J.S. Rollins, W. P. Sheffield, J. B. Steele, W. G. Steele, C. L. Vallandigham, C. Vibbard, Wadsworth, E. Word, C. A. Wichkliffe, G. C. Woodruff, H. B. Wright—38. Lorence, M. M. S. Charles, M. S. Steele, J. Steele, J. B. Steele, C. Moodruff, H. B. Wright—38.

January 17-The Senate passed the resolution-yeas 30, nays 1, (Mr. Powell.)

SMALL BANK NOTES-VETO MESSAGE. To the Senate of the United States:

The bill which has passed the House of Representatives and the Senate, entitled "An act to repeal that part of an act of Congress which | which was agreed to-yeas 61, nays 44.

prohibits the circulation of bank notes of a less denomination than five dollars in the District of Columbia," has received my attentive consideration, and I now return it to the Senate, in which it originated, with the following objections :

1. The bill proposes to repeal the existing legislation, prohibiting the circulation of bank notes of a less denomination than five dollars within the District of Columbia, without permitting the issuing of such bills by banks not now legally authorized to issue them. In my judgment, it will be found impracticable, in the present condition of the currency, to make The banks have genersuch a discrimination. ally suspended specie payments; and a legal sanction given to the circulation of the irredeemable notes of one class of them, will almost certainly be so extended, in practical operation, as to include those of all classes, whether authorized or unauthorized. If this view be correct, the currency of the District, should this act become a law, will certainly and greatly deteriorate to the serious injury of honest trade and honest labor.

2. This bill seems to contemplate no end which cannot be otherwise more certainly and beneficially attained. During the existing war, it is peculiarly the duty of the National Government to secure to the people a sound circulating medium. This duty has been, under existing circumstances, satisfactorily performed, in part at least, by authorizing the issue of United States notes, receivable for all Government dues except customs, and made a legal tender for all debts public and private, except interest on the public debt. The object of the bill submitted to me, namely, that of providing a small note currency during the present suspension, can be fully accomplished by authorizing the issue-as part of any new emission of United States notes made necessary by the circumstances of the country-of notes of a similar character, but of less denomination than five dollars. Such an issue would answer all the beneficial purposes of the bill; would save a considerable amount to the treasury, in interest; would greatly facilitate payments to soldiers and other creditors of small sums; and would furnish to the people a currency as safe as their own Government.

Entertaining these objections to the bill, I feel myself constrained to withhold from it my approval, and return it for the further consideration and action of Congress.

ABRAHAM LINCOLN.

June 23, 1862.

# State Taxation.

## First Session, Thirty-Eighth Congress.

June 22-The loan bill before the House in Committee of the Whole, and the question being on the first section, authorizing a loan of \$400,000,000, closing with this clause:

And all bonds, Treasury notes, and other obligations of the United States shall be exempt from taxation by or under State or municipal authority.

Mr. HOLMAN moved to strike out the clause,

dr. HOLMAN moved to insert at the end of first section these words :

nd that the bonds and other obligations issued under act shall be subject to State and municipal taxation.

Ir. Noble moved to amend the amendment substituting the following :

nd all bonds, Treasury notes, and other obligations of United States, shall be subject to State and municipal ution, on equal terms, the same as other property.

Vhich was rejected.

Ir. KERNAN moved this substitute for Mr. LMAN'S amendment :

nd that the owners of the bonds and obligations issued ar and by virtue of the provisions of this act shall be le to State and nuncipal taxation upon the value coft to the same extent as they are liable to such taxa-upon any other securities or similar personal estate cd by them.

Ir. HOLMAN accepted the amendment, which s rejected—yeas 56, nays 59.

une 23-Pending the consideration of the n bill in the House, being in Committee of Whole,

Ir. STEVENS offered this substitute for the

: hat the Secretary of the Treasury be, and he is hereby, torized to borrow, from time to time, on the credit of United States, \$400,000,000, for the service of the fiscal ending June 30, 1865, and to issue therefor, coupon or stered bonds of the United States, redeemable, at the sure of the Government, after any period not less than nor more than thirty years, and, if deemed expedient, e payable at any period not more than forty years from , payable in coin. And said bonds shall be of such minations as the Secretary of the Treasury shall direct, ess than fifty dollars, and bear an annual interest not eding eight per cent., payable semi-annually, and the rest on all bonds heretofore issued, payable annually, be paid semi-annually; and in lieu of such bond, an-ized to be issued, the Secretary of the Treasury may bonds, bearing interest, payable semi-annually. And hay also issue in exchange for Treasury notes hereto-issued bearing seven and three tenths per cent. inter-besides the six per cent bonds heretofore authorized, bonds of the denomination of S100 and of 550. And bonds, Treasury notes, and other obligations of the ed States shall be exempt from taxation by or under or municipal authority. Ir. HOLMAN moved to strike out the last

Ir. HOLMAN moved to strike out the last tence, which was rejected, on division—yeas nays 73.

he amendment was agreed to in Committee, s 72, nays 51; but immediately after was cted in the House—yeas 59, nays 81, as ows:

Stan-Messre. William J. Allen, Anderson, Baily, Augus-J. Baldwin, Baxter, Blair, Blow, Boyd, Brooks, Broom-William' G. Brown, Cole, Dawson, Denison, Donnelly, , Eldridge, Farnsworth, Grider, Harding, Benjamin G. ris, Higby, Holman, Hotchkiss, Asahel W. Hubbard, resoll, Philip Johnson, William Johnson, Kallyfeisch, pp, Law, Loan, Long, Marcy, McAllister, McClurg, Jouell, William H. Miller, Moorhead, James R. Morris, Shannon, John B. Stele, Stevens, Sweat, Thayer, Yan tenburgh, Whaley, Wheeler, Chilton A. White, Joseph White, Wilson, Wingleid-59.
 Yas-Messrs. Alley, Allison, Ames, Ancona, Arnold, ley, John D. Baldwin, Beaman, Blaine, Boutwell, Am-e W. Clark, Freeman Clarke, Cobb, Cresswell, Thomas avis, Dawes, Dixon, Driggs, Eckley, Edgerton, Eliot, *Isah*, Ferenan Clarke, Cobb, Cresswell, Thomas avis, Dawes, Dixon, Driggs, Eckley, Edgerton, Eliot, *isah*, Ferenan Clarke, Cobh, Hubbard, Hul-J, Jenckes, Julian, Kelley, Francis W. Kellogg, Orlando ogg, Kornar, Knox, Longyear, Marvin, McBride, Sam-e, Miller, Morrill, Daniel Morris, Leouard Myers, Noble, ion, Odell, Charles O'Neill, Patterson, Pendleton, Per-Jaul, John H. Rice, Edward H. Rollins, James S. Rol-Schenck, Scofield, Sloon, Smithers, Spalding, William Isele, Stiles, Straus, Staur, Upson, Ward, Ellihu B. hburne, William B. Washburn, Webster, Williams, *jamin Wood, Fernando Wood*-S1.

6

Mr. POMEBOY moved this substitute for the second section of the bill :

Second section of the birl. SEC. 2. And be il further enacted, That the Secretary of the Treescury may issue, upon the credit of the United States, bonds of any denomination not less than \$100, pay-able in lawfol money, three years from the date thereof, and bearing interest not exceeding eight per cent. per an-num, payable semi-annually in lawful money, and may receive at par therefor the lawful money of the United States, Treasury notes, certificates of indebtedness, or cer-tificates of deposit issued under any act of Congress. And the Secretary of the Treasury, in addition to the total amounts of bonds authorized by the first and second sec-tions of this act, shall issue at par, in redemption of any The secretary of the Treasury, in addition to the body sec-tions of this act, shall issue at par, in redemption of any outstanding notes, certificates of deposit, certificates of in-debtedness of the United States, bonds similar to those hereinbefore in this second section authorized, in denomi-nations of not less than \$100, or of like denominations similar to those anthorized by the first section, and pay-able five years from date, with interest at six per cent, psyable semi-annually. And the Secretary of the Treas-ury is further authorized to issue, in lieu of any bonds hereto for authorized by law, and not now issued in pur-suance thereof, bonds similar to and in the denominations United States notes, shall cease to be a legal tender in pay-ment of public or private indebtedness on and after the 1st day of October, 1564. And on notes, other than United States notes, shall hereafter be issued or reissued. Nor shall the total amount of United States notes issued or to be issued ever exceed \$400,000,000, and such additional sum, not exceeding \$50,000,000, as may be temporarily required for the redemption of temporary loan. Which was rejected—yeas 44, nays 81, as

Which was rejected—yeas 44, nays 81, as follows :

follows: YEAS-Messrs. Ancona, Augustus C. Baldwin, Brooks, William C. Brown, Freeman Clarke, Coffroth, Cole, Cravena, Creswell, Dawes, Dawson, Edgerton, English, Farnsworth, Ganson, Griswold, Harrington, Herrick, Holman, Hotch-kiss, Jenckes, Kallifleisch, Kernan, Law, Marcy, Samuel F. Miller, James R. Morris, Morrison, Nelson, Odell, Pike, Pomeroy, Price, Prum, James S. Rollins, Ross, Scofield John E. Steele, Thayer, Van Valkenburgh, William B. Wash-burn, Whaley, Wheeler, Wilson-44. NAXS-Messrs. William J. Allen, Alley, Allison, Ames, Arnold, Ashley, Baily, John D. Baldwin, Baxter, Beaman, Blair, Blow, Boutwell, Boyd, Broomall, James S. Brown, Ambrose W. Clark, Cobb, Thomas T. Davis, Denison, Dixon, Donnelly, Driggs, Eden, Eldridge, Eliot, Frank, Gooch, Hale, Harding, Benjamin G. Harris, Hooper, Asahel W. Hubbard, John H. Hubbard, Hulburd, Plaibp Johnson, Ju-Han, Kelley, Francis W. Kellogg, Orlando Kellogg, Knapp, Knox, Loan, Long, Longyear, McAllister, McBride, McClurg, Ym. H. Miller, Moorhead, Morrill, Daniel Morris, Amos Myers, Leonard Myers, Noble, Norton, Charles O'Neill, John o'Neill, Orth, Petham, John H. Rice, E. H. Rollins, Schenck, Shannon, Sloan, Smithers, Spalding, Wm. G. Steele, Stevens, Stiles, Strouse, Stuart, Tracey, Upson, Ellihu B. Washburne, Webster, Williams, Windom, Winfield, Benjamin Wood, Fer-nando Wood-S1. nando Wood-81.

The bill then passed without the yeas and nays being ordered.

June 27-The bill was slightly amended in the Senate, and passed without a division.

## STATE TAXATION-AGAIN.

June 28-On concurring in Senate amendments, Mr. HOLMAN moved to add this proviso to one of them:

Provided, That nothing in this act shall impair the right of the States to tax the bonds, notes, and other obligations issued under this act.

Which was rejected-yeas 71, nays 77, as follows:

10110WS: YEAS-MESSTE. William J. Allen, Ancona, Bliss, Brooks, James S. Brown, Chanler, Coffroth, Coz, Cravens, Dawes, Dawson, Denison, Eden, Edgerton, Eldridge, English, Finck, Ganson, Grider, Griswold, Harding, Harrington, Charles M. Harris, Herrick, Holman, Hotchkinse, Hulchins, Philip Johnson, William Johnson, Kalbfeisch, Kernan, Knapp, Law, Lazear, Le Blond, Long, Mallory, Marcy, McDowell, McKinney, Middleton, Samuel F. Miller, William H. Mil-ler, James R. Morris, Morrison, Noble, John O Neull, Pen-delton, Perry, Pomeroy, Pruyn, Radford, Samuel J. Ran-dall, Robinson, Ross, John B. Steele, William G. Steele, Stites, Strouse, Stuart, Sweat, Thomas, Tracy, Van Valken

burgh, Wadsworth, Ward, Whaley, Wheeler, Chillon A. White, Joseph W. White, Winfeld—11.
NATS—Messrs. Alley, Allison, Ames, Anderson, Arnold, John D. Baldwin, Baxter, Beaman, Blaine, Blair, Blow, Bontwell, Boyd, Broomall, William G. Brown, Cobb, Cole, Creswoll, Henry Winter Davis, Thomas T. Davis, Deming, Jono, Donnelly, Driggs, Eckley, Eliot, Fenton, Garfield, Gooch, Hale, Higby, Hooper, Asahel W. Hubbard, John H. Hubbard, Hilburd, Ingersoll, Jennekes, Julian, Kelley, Jongerson, Kellogg, Littlejohn, Loan, Longyear, Marvin, McBride, McClurg, McIndoe, Moorhead, Daniel Morris, Amos Myers, Leonard Myers, Norton, Chas. O'Neill, Orth, Perham, Pike, Alexander H. Rice, John H. Riee, Edward H. Rollins, Schenck, Scofield, Shannon, Sloan, Smith, Smithers, Spalding, Stevens, Thayer, Upson, Elihu
B. Washburne, Wm. B. Washburn, Webster, Williams, Wilder, Wilson, Windem-77.

## Taxation.

THE INTERNAL REVENUE AND TARIFF ACT OF 1861. First Session, Thirty-Seventh Congress.

The bill to provide increased revenue from imports, &c., passed the House August 2, 1861 yeas 89, nays 39. The NAVS were :

Messrs. Allen, Ancona, Beaman, Burnett, Coz, Cravens, Crit-tenden, Dunlap, English, Grider, Haight, Harding, Holman, Jackson, Johnson, Law, Logan, Mallory, May, Menzies, Mor-ris, Noble, Norton, Odell, Pendleton, Reid, James S. Rollins, Shiel, Smith, Trowbridge, Vallandigham, Vibbard, Voorhees, Wadsworth, Ward, Webster, Chillon A. While, Woodruff, Wright-39.

Same day, it passed the Senate-yeas 34, nays 8, (Messrs. Breckinridge, Bright, Johnson of Missouri, Kennedy, Latham, Polk, Powell, Saulsbury.)

THE INTERNAL REVENUE ACT OF 1862.

Second Session, Thirty-Seventh Congress.

1862, April 8-The House passed the bill to provide internal revenue, support the Government, and pay interest on the public debtyeas 126, nays 15. The NAYS were-

Messrs. William Allen, George H. Browne, Buffinton, Cox, Kerrigan, Knapp, Law, Norton, Pendleton, Richardson, Shiel, Vallandigham, Voorhees, Chilton A. White, Wickliffe -15.

June 6-The bill passed in the Senate-yeas 37, nay 1, (Mr. Powell.)

TAX ON SLAVES.

# While this bill was pending in the House,

April 8-Mr. BLAIR, of Missouri, offered this as a new section :

SEC. -.. That any person who shall claim to own the service or labor for life of any person under the laws of any State, shall pay on account of the service of each person so held the sum of two dollars.

Which was rejected—yeas 51, nays 76, as follows:

follows: Tass-Messrs. Aldrich, Arnold, Babbitt, Baxter, Beaman, Francis P. Blair, Samuel S. Blair, Blake, Campbell, Clark, Glements, Colfax, Frederick A. Conkling, Roscoe Conkling, Ovde, Davis, Dawes, Duell, Edgerton, Eliot, Fessenden, Gurley, Hanchett, Hickman, Kelley, Francis W. Kellegg, Hillinger, Lansing, Loomis, McPherson, Mitchell, Moor-ead, Anson P. Morrill, Justin S. Morrill, Olin, Patton, Pike, John H. Rice, Edward H. Rollins, Shanks, Sherman, Sloan, 2000 Status, Stevens, Van Valkenburgh, Verree, Vallace, E. Walton, Wheeler, Wilson, Windom-Si. Mars-Messrs. Allen, Alley, Baker, Biddle, Bingham, Facob B. Bain, William G. Brown, Buffinton, Calvert, Cham-Berlan, Cobb, Corning, Cox, Cravens, Crittenden, Delaplaine, Juraleo, Dunn, Edwards, English, Fisher, Frank, Franker, Grider, Haight, Hale, Hall, Harding, Harrison, Hoger, Horton, Hutchins, Julian, Kerrigan, Knapp, Law, Jacez, Leeary, Lekman, Lovejoy, Malloy, Menzies, Noble, Noten, Norton, Nugen, Pendicton, Perry, Timothy G. Phelps, Sond, Norton, Nugen, Pendicton, Perry, Jimothy G. Phelps, Sond, Norton, Nugen, Steffield, Sheilabarger, Shiel, Smith, Sohr B. Stelet, William G. Steele, Stratton, Benjamin F. Morn, Wadsworth, Webster, Albert S. White, Challton A. White, Wickliffe, Wright.-76.

# IN SENATE.

May 29-Mr. SUMNER offered this new section:

That an annual tax of five dollars shall be paid by every person or persons, corporation, or society, for and on ac-count of the service or labor of every other person between Count of the service or labor of every other person between the ages of ten and sixty-five years, whose service or labor, for a term of years or for life, is claimed to be owned by such first mentioned person or persons, corporation, or so-ciety, whether in a fiduciary capacity, or otherwise, under and by virtue of the laws or customs of any State; and said annual tax shall be levied and collected of the person or persons, corporation, or society, making such claim, and of their goods, chattels, or lands, as is hereinbefore pro-vided; but in no case shall the person or persons whose service or labor is so claimed, or their service or labor, be sold for the purpose of collecting said tax: *Provided*, That this tax shall not apply to service due to parents.

Mr. HENDERSON moved to add this proviso:

That the tax herein prescribed shall not be levicd or collected in any State where a system of gradual emanci-pation may have been adopted at the time of the collection.

Which was rejected-yeas 15, nays 20, as follows:

YEAS-Messrs. Browning, Dixon, Doolittle, Grimes, Hale, Harlan, Harris, Howe, Lane of Indiana, Pomeroy, Sher-man, Ten Eyck, Willey, Wilson of Massachusetts, Wright, -15.

NAYS-Messrs. Carlile, Chandler, Clark, Collamer, Cow-an, Davis, Fessenden, Foster, Howard, Kennedy, King, Lane of Kanses, Morrill, Foweil, Saulsbury, Sumner, Trum-bull, Wilkinson, Wilmot-20.

Mr. FESSENDEN moved to amend Mr. SUM-NER's, by reducing the tax from five to two dollars, which was agreed to:

YEAS—Messrs. Anthony, Browning, Chandler, Clark, Col-lamer, Cowan, Davis, Dixon, Doolittle, Fessenden, Foster, Grimes, Hale, Harlan, Harris, Howard, Howe, Kennedy, Lane of Indiana, Lane of Kansas, Morrill, Powell, Sher-man, Simmons, Ten Eyck, Wade, Willey, Wright—28. NATS—Messrs. Carlile, King, Latham, McDougall, Pome-roy, Saulsbury, Sunner, Trumbull, Wilmot, Wilson of Mas-sachusetts—10.

Mr. SUMNER's amendment was then rejected yeas 14, nays 22, as follows:

YEAS-MUSSIS. Anthony, Clark, Fessenden, Grimes, Har-lan, Howard, Howe, King, Lane of Kansas, Morrill, Sim-mons, Sumner, Trumbull, Wade-14. NAYS-MESSTS. Bayard, Browning, Carlile, Cowan, Davis, Dixon, Doolittle, Foster, Hale, Harris, Kennedy, Lane of Indiana, Latham, McDougal, Pomeroy, Powell, Saulsbury, Sherman, Ten Eyck, Willey, Wilson of Massachusetts, Wright-22.

June 5-Mr. SUMNER offered this as a new section :

That every person claiming the service or labor of any other person as a slave, shall pay a tax of two dollars on account of every person so claimed. But in no case shall any person so claimed be sold for the purpose of collecting the tax.

Which was agreed to-yeas 19, nays 16, as follows:

YEAS-Messrs. Anthony, Chandler, Clark, Collamer, Fes-senden, Foot, Grimes, Harlan, Howard, Howe, King, Mor-rill, Pomeroy, *Rice*, Simmons, Sumner, Wilkinson, Wilmot-19.

NATS-Messrs. Browning, Carlile, Cowan, Davis, Dixon, Doolittle, Foster, Hale, Lane of Indiana, Latham, Nesmith, Powell, Willey, Wilson of Massachusetts, Wright-16.

June 6-Mr. ANTHONY moved to reconsider this vote; which was agreed to-yeas 22, nays 18, as follows :

YEAS-Messrs. Bayard, Browning, Carlile, Cowan, Davis, Dixon, Doolittle, Foster, Hale, Harris, Kennedy, Lane of Indiana, Latham, McDougall, Nesmith, Powell, Rice, Sauls-bury, Stark, Ten Byck, Willey, Wright-22. NATS-Messrs. Anthony, Chandler, Clark, Fessenden, Foot, Grimes, Harlan, Howard, Howe, King, Morrill, Pomeroy, Simmons, Sumner, Trumbull, Wade, Wilkinson, Wilmot-18.

Mr. Hows moved to amend the amendment by inserting after the word "claimed" the words, "except those under ten and over sixty years of age," which was agreed to, and the amendment, as amended, was then rejectedyeas 17, nays 23, as follows:

YEAS — Messrs. Anthony, Chandler, Clark, Fessenden, Foot, Grimes, Harlan, Howard, Howe, King, Morrill, Sim-mons, Sumner, Trumbull, Wade, Wilkinson, Wilmot-17. NAYS-Messrs. Bayard, Browning, Carlile, Cowau, Davis, Dixon, Deolittle, Foster, Hale, Harris, Kennedy, Lane of Indiana, Latham, McDougall, Nesmith, Powell, Rice, Sauls-bury, Stark, Ten Eyck, Willey, Wilson of Mass., Wright -23.

# First Session, Thirty-Eighth Congress.

INTERNAL REVENUE ACT OF 1864.

April 28-The House passed the act of 1864 -yeas 110, nays 39. The NAYS were:

Messrs. James C. Allen, William J. Allen, Ancona, Brooks, Messrs. James C. Allen, William J. Allen, Ancoma, Brooks, Chanler, Cox, Dawson, Denison, Eden, Eldridge, Finch, Harrington, Benjamin G. Harris, Herrick, Philip Johnson, William Johnson, Knapp, Law, Le Blond, Long, Marcy, McDowell, McKinney, James R. Morris, Morrison, Noble, John O'Nell, Pendleton, Perry, Robinson, Ross, Stiles, Strouse, Stuart, Forchees, Ward, Chillon A. White, Joseph W. White, Fernando Wood—39.

June 6-The Senate amended and passed the bill--yeas 22, nays 3, (Messrs. Davis, Hendricks, Powell.)

The bill, as finally agreed upon by a Committee of Conference, passed without a division.

## TARIFF ACT OF 1862.

# Second Session, Thirty-Seventh Congress. IN HOUSE.

1862, July 1-The House passed, without a division, a bill increasing temporarily the duties on imports, and for other purposes.

July 8-The Senate passed it without a division.

THE TARIFF ACT OF 1864.

June 4-The House passed the bill-yeas 81, The NAYS were: nays 28.

Messrs. James C. Allen, Bliss, James S. Brown, Cox, Edger-ton, Eldridge, Finck, Grider, Harding, Harrington, Chas. M. Harris, Herrick, Holman, Hutchins, Le Blond, Long, Mallory, Marcy, McDowell, Morrison, Noble, Pendleton, Perry, Pruyn, Ross, Wadsworth, Chillon A. White, Joseph W. White-28.

June 17-The Senate passed the bill-yeas 22, nays 5, (Messrs. Buckalew, Hendricks, Mc-Dougall, Powell, Richardson.)

## TAXES IN INSURRECTIONARY DISTRICTS, 1862.

# Second Session, Thirty-Seventh Congress.

1862, May 12-The bill for the collection of taxes in the insurrectionary districts passed the Senate—yeas 32, nays 3, as follows :

YEAS-Messrs. Anthony, Browning, Chandler, Clark, Davis, Dixon, Doolittle, Fessenden, Foot, Foster, Harlan, Harris, Henderson, Howe, King, Lane of Indiana, Lane of Kansas, Latham, McDougall, Morrill, Nesmith, Pomeroy, Rice, Sherman, Summer, Ten Eyck, Trumbull, Wade, Wil-kinson, Willey, Wilson of Massachusetts, Wright-32. Nusce Messrs Howard Bound Scattering NAVS-Messrs. Howard, Powell, Saulsbury-3.

May 28-The bill passed House-yeas 98, nays 17. The Navs were :

Messrs. Biddle, Calvert, Cravens, Johnson, Kerrigan, Law, Mallory, Menzies, Noble, Norton, Pendleton, Perry, Francis Thomas, Vallandigham, Ward, Wickliffe, Wood-17.

The Democrats who voted Ave were:

Messrs. Ancona, Baily, Cobb, English, Haight, Holman, Leh-man, Odell, Phelps, Richardson, James S. Rollins, Sheffield, Smith, John B. Steele, Wm. G. Steele-14.

TAXES IN INSURRECTIONARY DISTRICTS, 1864. IN SENATE.

June 27-The bill passed the Senate without a division.

July 2-It passed the House without a division.

# The National Currency Act\* of 1863. Third Session, Thirty-Seventh Congress. IN SENATE:

1863, February 12-The bill passed-yeas 23, nays 21, as follows:

YEAS-Messrs. Anthony, Arnold, Chandler, Clark, Doo-little, Fessenden, Foster, *Harding*, Harlan, Harris, Howard, Howe, Lanc of Kansas, Morrill, *Nesmith*, Pomeroy, Sherman, Sumner, Ten Eyck, Wade, Wilkinson, Wilmot, Wilson of Massachusetts-23.

\* January 17-The President sent this message to Conress

To the Senate and House of Representatives: I have signed the joint resolution to provide for the imme-diate payment of the army and navy of the United States, assed by the House of Representatives on the 14th and by the Senate on the 15th instant.

the Senate on the 15th instant. The joint resolution is a simple authority, amounting, however, under existing circumstances, to a direction to the Secretary of the Treasury to make an additional issue of \$100,000,000 in United States notes, if so much money is needed, for the payment of the army and navy. My approval is given in order that every possible facility, may be afforded for the prompt discharge of all arrears of pay due to our soldiers and our sallors. While giving this approval, however, I think it my duty to express my sincere regret that it has been found necessary to authorize so large an additional issue of United

sary to authorize so large an additional issue of United States notes when this circulation, and that of the sus-pended banks together, have become already so redundant pendeu bans togenet, have become already so redundant as to increase prices beyond real values, thereby augment-ing the cost of living to the injury of labor, and the cost of supplies to the injury of the whole country. It seems very plain that the continued issues of United States notes, without any check to the issues of suspended banks and without adocuate movision for the reliance

banks and without adequate provision for the raising of money by loans, and for funding the issues so as to keep them within due limits, must soon produce disastrous con-sequences. And this matter appears to me so important that I feel bound to avail myself of this occasion to ask the president trains of Community of this occasion to ask the

that I feel bound to avail myself of this occasion to ask the especial attention of Congress to it. That Congress has power to regulate the currency of the country can hardly admit of a doubt; and that a ju-dicious measure to prevent the deterioration of this cur-rency, by a reasonable taxation of bank circulation or otherwise, is needed, seems caulty clear. Independently of this general consideration, it would be unjust to the people at large to exempt banks enjoying the special priv-ilege of circulation from their just proportion of the public burdens. burdens.

burdens. In order to raise money by way of loans most easily and cheaply, it is clearly necessary to give every possible sup-port to the public credit. To that end, a uniform currency, in which taxes, subscriptions to loans, and all other ordi-nary public dues, as well as all private dues, may be paid, is almost if not quite indispensable. Such a currency can be furnished by banking associations, organized under a general act of Congress, as suggested in my message at the beginning of the present session. The securing of this cir-culation by the pledge of United States bonds, as therein suggested, would still further facilitate loans, by increasing the present and causing a future demand for such bonds. the present and causing a future demand for such bonds. In view of the actual financial embarrassments of the

Government, and of the greater embarrassments sure to covernment, and of the greater embarrassments sure to come if the necessary means of relief be not afforded, I feel that I should not perform my duty by a simple an-nouncement of my approval of the joint resolution which proposes relief only by increasing circulation, without ex-pressing my earnest desire that measures, such in sub-stance as those I have just referred to, may receive the early sanction of Congress. By such measures, in my opinion, will payment he most

By such measures, in my opinion, will payment be most certainly secured, not only to the army and navy, but to all honest creditors of the Government, and satisfactory provision made for future demands on the Treasury. ABRAHAM LINCOLN.

January 17, 1863.

NATS-Messrs. Carlile, Collamer, Cowan, Davis, Dixon, Foot, Grimes, Henderson, Hicks, Kennedy, King, Latham, McDougall, Powell, Rice, Richardson, Saulsbury, Trumbull, Turpie, Wall, Wilson of Missouri-21.

## IN HOUSE.

February 20-The bill passed-yeas 83, nays 66, as follows:

66, as follows: YEAS-Messrs. Aldrich, Ashley, Babbitt, Beaman, Jing-ham, Jacob B. Blair, Blake, William G. Brown, Buffinton, Calvert, Campbell, Casey, Chamberlain, Clements, Colfax, Conway, Covode, Cutler, Davis, Delano, Diven, Dunn, Ed-gerton, Eliot, 'Ely, Fenton, Samuel C. Fessenden, Thomas A. D. Fessenden, Fisher, Frank, Goodwin, Granger, Hahn, Kaiyht, Hale, Hickman, Hooper, Hutchins, Julian, Kelley, Francis W. Kellogg, William Kellogg, Lansing, Leary, Iovejoy, Low, McIndee, McKean, McCherson, Marston, Maynard, Moorhead, Anson P. Morrill, Noell, Olin, Patton, Timothy G. Phelps, Potter, Alexander H. Rice, John H. Rice, Riddle, Sargent, Sedgwick, Segar, Shanks, Shellabar-ger, Sherman, Sloan, Spaulding, Stevens, Francis Thomas, Trimble, Trowbridge, Yan Horn, Yan Wyck, Verree, Wall, Wallace, Washburne, Albert S. White, Wilson, Windom, Worcester-33. MATS-Messrs. *Hulliam Allen, Ancona, Baily*, Baker,

Worcester-S3.
 NATS-Messrs. William Allen, Ancona, Baily, Baker,
 Baxter, Biddle, Cobb, Frederick A. Conkling, Roscoe Conk-ing, Coz, Cravens, Crisfield, Crittenden, Dawas, Edwards, Fouke, Gooch, Grider, Gurley, Hall, Harding, Harrison, Holman, Horton, Johnson, Kerrigan, Knapp, Lave, Lazear, Loomis, Mallory, May, Menzies, Justin S. Morrill, Morris, Nixon, Noble, Norton, Nugen, Odell, Pendleton, Perry, Pike, Pomeroy, Porter, Price, Robinson, James S. Rollins, Sheffield, Sheil, John E. Steele, William G. Steele, Silles, Stratton, Benjamin F. Thomas, Vallandigham, Voorhees, Wadsworth, Walton, Wheeler, Whaley, Chilton A. White, Wickliffe, Woodruff, Wright, Yeaman-66.

## THE ACT OF 1864.

April 18-The bill passed the House-yeas 80, nays 66, as follows:

YEAS—Messrs. Alley, Allison, Ames, Anderson, Arnold, Ashley, John D. Baldwin, Baxter, Beaman, Blaine, Jacob B. Blair, Boutwell, Boyd, Broomall, W. G. Brown, A. W. Clark, Freeman Clarke, Cobb, Cole, Creswell, Dawes, Deming, T. T. Davis, Donnelly, Driggs, Dumont, Eckley, Farnsworth, Fenton, Frank, Garfield, Gooch, Grinnell, Higby, Hooper, Hotchkiss, Asahel W. Hubbard, John II. Hubbard, Jenckes, Julian, Kasson, Kelley, Francis W. Kellogg, Orlando Kel, Gogg, Loan, Longyear, Marvin, McBride, McClurz, McIndoe, Samuel F. Miller, Moorhead, Morrill, Daniel Morris, Anos Myers, Norton, Charles O'Neill, Orth, Patterson, Perham, Price, William H. Randall, John H. Rice, Edward H. Rol-ins, Shannon, Sloan, Smith, Stevens, Thayer, Tracy, Upson, Yan Valkenburgh, Ellihu B. Washburne, William B. Wash-burn, Webster, Williams, Wilder, Wilson, Windom, Wood-hridge=-80.

burn, Webster, Williams, Wilder, Wilson, Windom, Wood-bridge-80. NAVS-Messrs. J. C. Allen, W. J. Allen, Ancona, Baily, A. C. Baldwin, Francis P. Blair, Bliss, Brooks, Chanler, Clay, Coffroth, Cravens, Dawson, Deinson, Eden, Eldridge, Finck, Ganson, Grider, Griswold, Hall, Harding, Harrington, Ben-jamin G. Harris, Charles M. Harris, Herrick, Holman, Hutchins, William Johnson, Kernan, King, Knapp, Law, Lazear, Long, Mallory, Marcy, McDowell, McKinney, Mid-eldeon, William H. Miller, James R. Morris, Morrison, Nel-son, Noble, Odell, Pendleton, Pruyn, Radford, Samuel J. Randall, Robinson, Rogers, Ross, Soott, John B. Steele, W. G. Steele, Stiles, Strouse, Stuart, Thomas, Wheeler, Chillon A. While, Joseph W. White, Winfield, Fernando Wood, Fea-man-65. man-66.

May 10-The Senate passed it-yeas 30, nays 9, as follows:

Yass-Mowsr. Anthony, Chandler, Clark, Collamer, Con-ness, Dixon, Doolittle, Fessenden, Foot, Foster, Hale, Har-lan, Howard, Howe, Johnson, Lane of Indiana, Lane of Kansas, Morgan, Morrill, Pomeroy, Ramsey, Sherman, Sprague, Sumner, Ten Eyck, Trumbull, Van Winkle, Wil-kinson, Willey, Wilson-20, Kansey, Grimes, Hender-enn, Pawell, Richardson, Riddle; Szulsbury-9.

## IN HOUSE.

## SMALL NOTES.

Pending the consideration of this bill,

April 6-A section was adopted, authorizing the issue to those banks of notes of the denominations of one, two, three, five, ten, twenty, Government.

fifty, one hundred, five hundred, and one thou-sand dollars—yeas 76, nays 54, as follows:

Sand dollars—yeas 76, nays 54, as follows:
 YEAS—Messrs. Alley, Allisou, Ames, Anderson, Ashley, John D. Baldwin, Baxter, Beaman, Blaine, Elow, Boutwell, Boyd, Broomall, Ambrose W. Clark, Cobb, Cole, Dixon, Donnelly, Driggs, Eckley, Eliot, Frank, Ganson, Gooch, Grinnell, Griswold, Hale, Hotchkiss, Asahel W. Hubbard, John H. Hhbbard, Jenckes, Julian, Kasson, Kelley, Francis W. Kellogg, Orlando Kellogg, Loan, Longyear, Marvin, McBride, McClurg, Samuel F. Miller, Moorlhead, Morrill, Daniel Morris, Amos Myers, Leonard Myers, Charles O'Neill, Orth, Patterson, Perham, Pomeroy, Price, Pruyn, Radford, William H. Randall, Alexander H. Rice, John H. Riee, Edward H. Rollins, Schenck, Scofield, Shannon, Spalding, Starr, Stevens, Thayer, Upson, Van Valkenburgh, Elihu B. Washburne, William B. Washburn, Whaley, Williams, Wilder, Wilson, Windom, Woodbridge—76.
 NAYS—Messrs. James C. Allen, Wiltiam J. Allen, Ancona, Baily, Jugustus C. Baldwin, Büss, Brooks, James S. Broun, William G. Brown, Chanler, Cox, Dawson, Denison, Eden, Eldridge, English, Finek, Grider, Harrington, Herrick, Holman, Philip Johnson, William Johnson, Kelbfleisch, Kernan, Law, Long, Mallory, Mary, McKinney, Middleton, William H. Midler, James R. Morris, Morison, Nelson, John O'Neill, Fendelton, Samuel J. Randall, Robinson, Rogers, James S. Rollins, Scott, John B. Steele, William G. Steele, Strouse, Sweat, Thomas, Foorhees, Wheeler, Chillon A. White, Joseph W. White, Winfeld, Benjamin Wood, Yeaman—54.

While, Jo man-54.

## INTEREST.

The section enacting that seven per cent, interest shall be deemed the lawful interest in all States where no rate is established, but each bank shall be bound by the State law regulating interest in the State where it is located, was agreed to-yeas 89, nays 45.

### STATE TAXATION.

This section :

That nothing in this act shall be construed to prevent the taxation by States of the capital stock of banks organized under this act, the same as the property of other moneyed corporations, for State or municipal purposes; but no State shall impose any tax upon such associations or their capi-tal, circulation, dividends, or business, at a higher rate of taxation than shall be imposed by such State upon the same amount of moneyed capital in the hands of individual citizens of such State.

Was adopted—yeas 78, nays 56, as follows:

Was adopted—yeas 78, nays 56, as follows: YEAS—Messrs. James C. Allen, William J. Allen, Ancona, Baily, Augustus C. Baldwin, Bitiss, Broots, Broomall, James S. Brown, William G. Brown, Chanler, Clay, Coz, Cravens, Dawson, Denison, Eden, Eldridge, English, Finck, Ganson, Grider, Griswold, Hall, Herrington, Benjamin G. Harris, Herrick, Holman, Hotchkiss, Philip Johnson, William John-son, Kalbfeisch, Orlando Kellogg, Kernan, Law, Lazear, Long, Mallory, Marcy, McKinney, Middlelon, Samuel P. Miller, William H. Miller, James P. Morris, Morrison, Amos Myers, Nelson, Odell, John O'Neill, Orth, Pecalleton, Pike, Pomeroy, Pruyn, Radford, Samuel J. Randall, William H. Randall, John H. Rice, Robinson, Rogers, Scofield, Soci, Traev, Van Valkenburgh, Ward, Whaley, Wheeler, Chilton A. While, Joseph W. White, Windom, Winfield, Benjamin Wood, Yeaman-78.

Wood, Yeaman-75. Nars-Messrs. Alley, Allison, Ames, Anderson, Ashley, John D. Baldwin, Baxter, Beaman, Blaine, Blow, Boutwell, Boyd, Ambrose W. Clark, Cobb, Cole, Thomas T. Davis, Dixon, Donnelly, Driggs, Eckley, Eliot, Frank, Gooch, Grin-nell, Hale, Hooper, Asahel W. Hubbard, John H. Hubbard, Jenckes, Julian, Kasson, Kelley, Francis W. Kellogg, Loan, Longyear, Marvin, McBride, McClurg, Morrill, Daniel Mor-ris, Leonard Myers, Charles O'Neill, Perham, Price, Alexan-der H. Rice, Edward H. Rollins, Schenck, Shannon, Smithers, Spalding, Stevens, Thayer, Upson, Ellihu B. Washburne, William B. Washburn, Wilder-56.

## MR. STEVENS'S SUBSTITUTE.

Upon these and other amendments being adopted, Mr. STEVENS offered a substitute for the whole bill, which he explained as differing from the amended bill in these respects only:

The substitute provides for a uniform rate of interest at seven per cent., and withdraws these national banks from State taxation and leaves them to be taxed by the national

Which was rejected-yeas 59, nays 78, as follows

YEAS-Messrs. Alley, Allison, Ames, Anderson, Ashley, John D. Baldwin, Baxter, Beaman, Blow, Boutwell, Boyd, Broomall, Ambrose W. Clark, Cobb, Cole, Thomas T. Davis, Dixon, Donnelly, Driggs, Eckley, Eliot, Frank, Garfield, Gooch, Grinnell, Hale, Hooper, John H. Hubbard, Jencks,

Dixon, Donnelly, Driggs, Eckley, Eliot, Frank, Garfield, Gooch, Grinnell, Hale, Hooper, John H. Hubbard, Jonekcs, Julian, Kassou, Kelley, Francis W. Kellogg, Loan, Longyar, Marvin, McBride, McChurg, Morrill, Daniel Morris, Leonard Myers, Charles O'Neill, Patterson, Perham, Alexander H. Nice, Edward H. Rollins, Schenck, Scofield, Shanon, Spalding, Starr, Stevens, Thayer, Thomas, Upsou, William B. Washburn, Wilder, Windom, Woodbridge-59.
 NAYS-McSsts. James C. Allen, William J. Allen, Anona, Baity, Augustus C. Baldwin, Blaine, Bliss, Brooks, James S. Brown, Ochanler, Cluy, Ox, Cravens, Dawson, Denison, Eden, Eldridge, English, Finol, Garson, Grüder, Griswold, Hall, Harrington, Benjamin G. Harris, Joneson, William Johnson, Kalbfleisch, Orlando Kellogg, Kernan, Law, Long, Mallory, Marcy, McKinney, Middleon, Price, Pruyn, Redford, Smuel, J. Randell, William I. Randall, John H. Nice, Robinson, Rogers, James S. Stott, Smithers, John B. Steele, W. G. Steele, Strouse, Sweat, Tracy, Van Valkenburgh, Ward, Elihn B. Washburne, Wieler, Childon A. While, Joseph W. White, Wilson, Win-field, Benjamin Von, Karther, Sond, Stevel, Washburne, Wieler, Chillon A. While, Joseph W. White, Wilson, Win-field, Benjamin Wood, Jeaman-78.

On Mr. STEVENS'S motion, the bill was then tabled-yeas 91, nays 44.

## STATE TAXATION.

April 16-A new bill, previously introduced, was considered, containing (among others) this provision :

"Every organization under this act shall pay to the Treasurer of the United States a duty of one per cent. each half year, from and after the 1st day of April, in the year 1864, upon the maximum amount of their circulating notes during the six months; and in default of such payment the Treasurer of the United States is hereby authorized to retain one per cent. of the amount of bonds required to be retain one per cent. of the amount of bonds required to be deposited as security for such circulation at each semi-an-nual payment of the interest thereon; and such duty and the taxes or duties imposed by Congress from time to time shall be in lieu of all other taxes on such associations: *Provided*, That nothing in this act shall be construed to prevent the market value of the shares in any of the said backing back back of the shares in any of the said prevent the market value of the shares in any of the part banking associations, held by any person or body-corporate created by State law, being included in the valuation of the aggregate personal property of such person or State corpo-ration in assessing any tax imposed by any State or munjcipal authority on the aggregate personal estate of all persons subject to the authority of such State or munici-pality."

## Mr. FENTON moved to substitute this :

And that nothing in this act shall be construed to prevent the taxation by States of the capital stock of banks organized under this act, the same as the property of other mon-eyed corporations for State or municipal purposes; but no State shall impose any tax upon such associations or their capital, circulation, dividends, or business, at a higher rate of taxation than shall be imposed by such State upon the Same amount of moneyed capital in the hands of individual citizens of such State: *Provided*, That no State tax shall be imposed on any part of the capital stock of such associa-tion invested in the bonds of the United States, deposited as security for its circulation.

Which was agreed to-yeas 70, nays 60, as follows :

YEAS-Messrs. Alley, Allison, Ames, Arnold, Ashley, Baily, John D.Baldwin, Baxter, Beaman, Blaine, Boutwell, Broomall, William G. Brown, Ambrose W. Clark, Freeman Broomall, William G. Brown, Ambrose W. Clark, Freeman Clarke, Clay, Cobb, Cole, Dawes, Driggs, Dumont, Eckley, Farnsworth, Fenton, Frank, Gooch, Grinnell, Higby, Hooper, Hotchkiss, John H. Hubbard, Jenckes, Julian, Kasson, Francis W. Kellogg, Orlando Kellogg, Loan, Longyear, Mar-vin, McClurg, McIndoe, Samuel F. Miller, Moorhead, Mor-rill, Daniel Morris, Amos Myers, Charles O'Neill, Orth, Patterson, Perham, Pike, Pomeroy, Price, William H. Ran-dall, Alexander H. Rice, John H. Rice, Edward H. Rollins, Shannon, Sloan, Smith, Tracy, Upson, Van Valkenburgh, Ellihu B. Washburne, William B. Vashburn, Webster, Wilder, Wilson, Windom, Woodbridge-70.

NATS--Messrs. James C. Allen, William J. Allen, Au-gustus C. Baldwin, Brooks, James S. Brown, Chanler, Cra-rens, Creswell, Houry Winter Davis, Dawson, Eden, Eldridge, Finck, Ganson, Halt, Harding, Harrington, Benjamin G. Hurris, Horrick, Holman, Asahel W. Hubbard, Hutchins, William Johnson, Kallyleisch, Kelley, Kernan, King, Knapp, Law, Lazear, Long, Marcy, McBride, McDoweld, McKinney, William H. Miller, James R. Morris, Morrison, Nelson, Noble, Odell, Penlleton, Pruyn, Radford, Samuel J. Randall, Robinson, James S. Rollins, Ross, Scott, John B. Stole, Strouse, Stuart, Thayer, Thomas, Wheeler, Chilton A. While, Joseph W. White, Williams, Winfield, Fernando Wood -60.

The bill was then passed.

## IN SENATE.

## ON TAXATION.

April 29-The Senate committee reported a substitute for Mr. FENTON's amendment adopted by the House, which proposed that-

In lieu of all other taxes every association shall pay to the Treasurer of the United States, in the months of January and July, a duty of one half of one per cent. each half year and only, a duty of one half of one per cent, each half year from and after the 1st day of January, 1864, upon the aver-age amount of its notes in circulation, and a duty of one quarter of one per cent, each half year upon the average amount of its deposits, and a duty of one quarter of one per cent, each half year, as aforesaid, on the average amount of its actual atout barrand the amount invested in View of its capital stock beyond the amount invested in United States bonds. \* \* \* *Provided*, That nothing in this act shall be construed to prevent the market value of the shares in any of the said associations, held by any per-son or body-corporate, from being included in the valuation of the personal property of such person or corporation in the assessment of all taxes imposed by or under State authe assessment of all taxes imposed by or under State au-thority for State, county, or municipal purposes; but not at a greater rate than is assessed upon all other moneyed cap-ital in the hands of individual citizens of such State. And all the remedies provided by State laws for the collection of such taxes shall be applicable thereto: *Provided*, also, That nothing in this act shall exempt the real estate of associa-tions from either State county or municipal taxes to the tions from either State, county, or municipal taxes to the same extent, according to its value, as other real estate is taxed.

Mr. POMEROY moved to strike out the first proviso, and insert:

Provided, That nothing in this act shall be construed as exempting the capital stock of an association, beyond the amount invested in United States bonds and deposited with the Treasurer of the United States as part of its capital or as security for its circulating notes, from being subject to the same rate of State and municipal taxation as is imposed upon other personal property in the State, city, or town in which the association is located.

Which was rejected-yeas 11, nays 28, as follows:

YEAS-Messrs. Chandler, Conness, Harding, Howard, Lane of Indiana, Pomeroy, Ramsey, Sherman, Sprague,

Sumer, Wilkinson-11. Nars-Messrs. Buckalew, Curlie, Clark, Collamer, Cowan, Davis, Dison, Fessenden, Foot, Foster, Grimes, Hale, Har-lan, Henderson, Hendricks, Howe, Johnson, Lane of Kansas, McDougall, Morgan, Morrill, Nesmith, Powell, Riddle, Ten Eyck, Van Winkle, Willey, Wilson-28.

Mr. Howard moved to amend the proviso so as to make it read:

Provided, That nothing in this act shall be construed to prevent the market value of the shares in any of the said associations, held by any person or body-corporate, from being included in the valuation of the personal property of such person or corporation in the assessment of all taxes imposed by or under State authority for State, county, or municipal purposes in the State where the bank is situated; &c.

Which was rejected-yeas 11, nays 27, as follows:

YEAS-Messrs. Chandler, Conness, Harlan, Hendricks, Howard, Morrill, Pomeroy, Ramsey, Sherman, Sumner, Wilkinson-11.

MARS-Messres. Anthony, Buckalew, Clark, Collamer, Cowan, Davis, Dixon, Doolittle, Fessenden, Foot, Foster, Grimes, Hale, Henderson, Howe, Johnson, Lane of Indiana, Lane of Kansas, Morgan, Nesmith, Fowell, Riddle, Spragne, Ten Eyck, Van Winkle, Willey, Wilson-27.

May 6 - Mr. SUMNER's substitute :\*

And in licu of all other taxes on the capital, circulation. And in heu of all other taxes on the capital, circulation, deposits, shares, and other property, every association shall pay to the Treasurer of the United States, in the months of January and July, a duty of one per cont. each half year from and after the 1st day of January, 1864, upon the aver-age amount of its notes in circulation, and the duty of one ball of one now cont. each balf. For upon the average amount of its notes in circulation, and the duty of one half of one per cent. each half year upon the average amount of its deposits, and a duty of one half of one per cent. each half year, as aforesaid, on the average amount of its capital stock beyond the amount invested in United States bonds: \* *Provided*, That nothing in this act shall exempt the real estate of associations from either State, county, or municipal taxes to the same extent, accord-ing to its value, as other real estate is taxed: Provided also, That all taxes imposed by this or any. future act on banking associations organized under national legislation shall be

## \* Mr. SUMNER read this letter from Socretary Chase: TREASURY DEPARTMENT

May 5, 1864.

SIR: Nothing but my deep sense of the importance of sustaining by every possible means the public credit, upon which the sole dependence of the Government to suppress the insurrection must rest, would induce me to address you this letter upon a subject which has already received so much consideration.

The bill in relation to the national banking system now under debate is in the nature of an amendment to the act of the last session. Though a complete bill in itself, it con-tains few provisions not substantially embraced in that act, among which that in relation to the measure and distribu-tion of twestion men by accorded perform act the meet in tion of taxation may be regarded perhaps as the most im-portant. Under ordinary circumstances there might be no insuperable objection to leaving the property organized under the national banking law, subject as are almost all descriptions of property to general taxation; State, national, and municipal. But in the present condition of the coun-try, I respectfully submit that this particular description of property should be placed in the same category with imand be subjected to exclusive national taxation.

and be subjected to exclusive national taxation. At the present moment the duties on imports form the sole reliance of the Government for means to pay the inter-est on the public debt. If to these means the taxes to be paid by the national banks shall be added a most important addition will be made to these measures. The mere fact that these taxes are made payable to the national Govern-ment and so rendered available for the payment of interest on the public debt, and for the reduction of its principal, will strengthen the public credit and facilitate the negotia-tion of the public debt, and for the reduction of its principal, tion of the necessary loans at moderate rates of interest. have no doubt that such a disposition of these taxes would be worth more to the Government during the present strug-gle in practical results than three times the actual value of the taxes themselves.

I do not at all suggest that this description of property should not be taxed as heavily as any other description. On the contrary, I think it just that it should bear its full proportion of the public burdens. I am only anxious that the taxation upon it shall be made to contribute as largely the taxation upon it shall be made to contribute as largery as possible to the general welfare, and it is the conviction deeply impressed on my mind that it will contribute more when aggregated in one mass, and made to tell upon the general public credit, than when distributed between the nation and the States and numerous municipal corporations,

that prompts me to address these views to you. Under any plan of partition that may be adopted the amount of taxation distributable to the several States and municipalities will be comparatively small and unimportnuncipalities will be comparatively small and unimport-ant, and it is quite possible that the total taxation of bank-ing property for all purposes, will be less than it will if taxed exclusively for national purposes. The advantages of partition to States and municipalities will therefore be small, and the banks may not lose by it. The nation alone will be injured. It will not be understood, of course, that the foregoing suggestions are intended to apply to real estate held by any banking institutions; that description of property must necessarily be held by titles under State laws, and should properly be subjected exclusively to State taxation excent in the event of a direct tax by Con-State taxation, except in the event of a direct tax by Con-

gress. The case is otherwise with the personal property and credits of the banking associations. These receive their organization from national law and for great national pur-poses, and may therefore be with great propriety, and—as I have endeavored to show at the present time—with great public advantage be subjected to exclusive national taxa-

S. P. CHASE.

tion. Respectfully yours, The Hon. WILLIAM PITT FESSENDEN, Chairman of the Committee on Finance, Senate

applied exclusively to the payment of the interest and prin-cipal of the national debt of the United States.

Which was rejected-yeas 11, nays 24, as follows:

YEAS-Messra. Chandler, Conness, Howard, Lane of In-diana, Pomeroy, Ramsey, Sherman, Sprague, Sumner, Wil-kinson, Wilson-11.

MINSON, WIISON-11. NAYS-MOSSTES. Anthony, Buckalew, Carlile, Clark, Colla-mer, Cowan, Davis, Dixon, Doolittle, Fossenden, Foot, Fos-ter, Grimes, Hale, Henderson, Howe, Johnson, Morgan, Powell, Richardson, Riddle, Ten Eyck, Trumbull, Van Winkle-24.

## ON SMALL NOTES.

May 10-Mr. BUCKALEW moved to strike out of the twenty-second section the words allowing one, two, and three dollar bills ; which was rejected-yeas 8, nays 27, as follows :

Was Fejected—Jeas 6, Rays 11, as follows. YEAS—Messrs. Buckalew, Cowan, Doolittle, Henderson, Fowell, Richardson, Riddle, Saulsbury—S. NAYS—Messrs. Anthony, Olark, Collamer, Conness, Dixon, Foot, Foster, Grimes, Hale, Harlan, Howe, Johnson, Lane of Indiana, Lane of Kansas, Morgan, Morrill, Ramsey, Sherman, Sprague, Sumner, Ten Eyck, Trumbull, Van Winkle, Wade, Wilkinson, Willey, Wilson—27.

ON REPEALING THE BANKING SYSTEM.

Mr. POWELL moved to substitute for the whole bill, a section repealing the banking act of February 25, 1863; which was rejected—yeas 6, (Messrs. Buckalew, Henderson, Powell, Richardson, Riddle, Saulsbury,) nays 31.

The bill then passed.

The House non-concurred in the Senate's amendments, when a Committee of Conference was appointed, who reported, June 1. The tax question was settled by adding these words to the thirty-second section:

And nothing in this act shall be construed to prevent all the shares in any of the said associations, held by any person or body corporate, from being included in the valuation son or body corporate, from being included in the valuation of personal property of such person or corporation, in the assessment of taxes imposed by or under State authority, at the place where such bank is located, and not elsewhere jo but not at a greater rate than is assessed on other moneyed capital in the hands of individual citizens of said States : *Provided further*. That the tax is imposed under the law of any State upon the shares of any of the associations author-ized by this act shall not exceed the rate imposed on shares in any of the bank organizations under the authority of the State where such association is located.

The bill provides for a tax of one per cent. on the circulation of national banks, one half of one per cent. on their deposits, and one per cent. on their capital above the amount invested in United States bonds.

The report was concurred in, without a division in either house.

## The Public Debt.

## THE PUBLIC CREDIT UNDER BUCHANAN.

December 28, 1860-Bids for \$5,000,000 in six per cent. Treasury notes, authorized by act. of December 17, 1860, were opened by Philip F. Thomas, Secretary of the Treasury, when it. was found that only \$2,500,000 were bid for, and this at a rate of discount from seven to thirty-six per cent. Eight thousand five hundred were bid for at seven per cent.; \$151,600 at from seven to ten per cent; \$1,087,000 at twelve per cent.; \$140,000 at from twelve to, twenty per cent.; \$325,000 at from twenty to, thirty-six per cent.

January 19, 1861-The balance of this loam was taken, the bids being for nearly triple, the amount and ranging from eight and as half to fifteen per cent. discount. One bidl

of \$10,000 was made at twenty per cent. dis-count. February 23, 1861—John A. Dix, Secretary of the Treasury, opened the bids for \$8,000,000 of six per cent. twenty years' stock of the United Over \$4,000,000 were bid for at \$90 15.

# Debt of the United States, from June 30, 1860, to May 14, 1864.

June 30, 1860, the public debt was	\$61,769,703	02
Of which \$45.079,203 08 were in bonds, and \$19,690,500 in Treasury notes.		
June 30, 1861, the public debt was	90,867,828	68
May 29, 1862, the public debt was	491,445,984	11
as follows:		

Under what act.	Rate of interest.	Amount.	Total.
Loan of 1842 Do 1847 Do 1848 Do 1858 Do 1858 Do 1860 Do 1861 Texan indemnity	do do 5 per cent 6 per cent	$\begin{array}{r} 9,415,250 & 00 \\ 8,908,341 & 80 \\ 20,000,000 & 00 \\ 7,022,000 & 00 \\ 18,415,000 & 00 \end{array}$	\$70,104,955 9 <b>1</b>
Treasury notes issued prior to 1857 Treasury notes under act of December 23, 1857 Treasury notes under act of December 17, 1860 Treasury notes under acts of June 22, 1860, and February and March, 1861 Treasury notes under acts of March 2, July 17, and August 5, 1861	do do 6 per cent	175,900 00 221,650 00	
Three years' 7.30 bonds Twenty years' bonds	7.30 pe <del>r</del> cent 6 per cent	120,523,450 00 50,000,000 00	3,382,161 64
Oregon war debt United States notes Certificates of indebtedness	No interest 6 per cent	47,199,000 00	170,523,450 00 878,450 00 145,880,000 00
Four per cent. temporary loan Five per cent. temporary loan	4 per cent 5 per cent	5,913,042 21 44,865,524 35	49,898,400 00 50,778,566 58
			\$491,445,984 11

Of this, these items belonged to the old debt:		
Funded	\$70,104,955	5 9
Treasury notes	3,270,561	6
Treasury notes Oregon war debt	878,450	) ()
-		
	74,253,967	5
Leaving due to the war	417,195,016	5 5
Inne 30, 1862, the debt was	\$517.372.809	2 q
June 30, 1862, the debt was	1.098.793.181	2
As follows:		Ŭ
Loan of 1842	\$302,620	7
	9,415,250	
" 1848,	8,908,341	
" 1858,	20,000,000	
1800,	7,022,000	
1:01, (Feb. 6,) b per cent	18,415,000	
" 1861 (March 2) 6 per cent	776,750	
Id funded and unfunded debt (1800 and 1812,) 3 and 6 per cent	114,115	
Did funded and unfunded debt (1800 and 1812,) 3 and 6 per cent Ecxan indemnity, 5 per cent	3,461,000	0
Ireasury notes issued prior to 1857, (int. stopped,)		
Treasury notes issued under act of Dec. 23, 1857		
Treasury notes issued under act of Dec, 17, 1860		
Demand notes July 17, and Aug. 5, 1861		
J. S. notes, Feb. 25, 1862		
J. S. notes, July 11, 1862		
Postal currency, July 17, 1862		
5, 5, 20(05) (ICW ISSUES,) Ind(CII 5, 1805	411,308,626	1 3
Three years' 7.30 bonds, 7.30 per cent (two issues,)	139,970,500	
wenty years' bonds, 6 per cent. (two issues,)	50,028,500	0
Jregou war uebl ber cent	1.1124.394	ì
Jertificates of indebtedness, 6 per cent	156,784,241	. 6
Dertificates of indebiedness, 6 per cent	168,880,250	0 0
femporary loan, 4 and 5 per cent.	102,384,085	3
		-
	1,098,793,181	
Feb. 2, 1864, the debt was	1,473,225,714	: (
May 10, 1864, the public debt was	1,726,248,411	6
May 14, 1864, the debt was	1,730,870,926	, 0
		-

# THE FINANCES.

# Statement of the Public Debt of the United States, June 30, 1864.

## DEBT BEARING INTEREST IN COIN.

LATE OF INTEREST.	CHARACTER OF ISSUE.	AMOUNT OUTSTANDING.	INTEREST.
6 per cent	Bonds	\$9,415,250 00	\$564,915 00
6 per cent	Bonds	8,908,341 80	534,500 50
5 per cent	Bonds	20,000,000 00	1,000,000 00
	Bonds	7,022,000 00	351,100 00
6 per cent			1,104,900 00
	Bonds		3,000,000 00
6 per cent		30,923,600 00	1,855,416 00
per cent	Bonds, 5-20's*	510,756,900 00	30,645,414 00
per cent		72,963,850 00	3,648,192 50
5 ner cent	Bonds, Texan Indemnity	2,159,000 00	107.950 00
6 per cent	Bonds, Oregon War	1,016,000 00	60,960 00
3.10 per cent	Notes, Three years*	109,075,750 00	7,962,529 75
	Bonds		1,188,965 79
Aggregate	of debt bearing Coin Interest	\$860,471,788 45	\$52,024,843 54

### DEET BEARING INTEREST IN LAWFUL MONEY.

4 per cent		\$662,474 49	\$26,495 97
6 per cent			370,731 12 4.024.343 09
6 per cent	Certificates of Indebtedness	159,570,000 00	9,574,200 00
5 per cent	One Year Notes	44,520,000 00	2,226,000 00
5 per cent	Two Years Notes	16,480,000 00	824,000 00
5 per cent	Two Years Notes, with Coupons, \$150,000,000 00 Less withdrawn and destroyed		
1	or ready to be destroyed 57,279,150 00	92,720,850 00	4 696 049 50
6 per cent	3 years Compound Interest Notes	3,880,000 00	4,636,042 50
Aggregate	of debt bearing Lawful Money interest	\$392,328,665 20	\$21,682,315 68

h

DEET ON WHICH INTEREST HAS CEASED.		RECAPITULATION.		
CHARACTER OF ISSUE.	AMOUNT OUTSTANDING.	DEBT	AMOUNT OUTSTANDING.	INTEREST.
Bonds Treasury Notes Treasury Notes Treasury Notes	\$203,808 45 104,511 64 9,900 00	Bearing interest in coin Bearing interest in lawful	\$860,471,788 45	\$52,024,843 54
Treasury Notes Treasury Notes Temporary Loan Coin	$\begin{array}{c} 600 & 00 \\ 600 & 00 \\ 47,150 & 00 \\ 4.200 & 00 \end{array}$	On which interest has	392,328,665 20	21,682,315 68
		ceased	370,170 09	
Aggregate of debt on which Inter- est has ceased Aggregate of debt not bearing In-		Bearing no interest	486,866,065 79	
terest†	486,866,065 70		\$1,740,036,659 53	\$73,707,159 22

\* TREASURY DEPARTMENT, May 18, 1864.

\*TREASURY DEPARTMENT, May 15, 1864. Size: Your letter of the 13th instant, making inquiries in regard to the kind of currency with which the five-twenty years six per cent. and the three years seven-thirty per cent. notes are to be redeemed, has been received. It has been the constant usage of the Department to redeem all coupon and registered bonds forming part of the funded or permanent debt of the United States in coin, and this usage has not been deviated five during my adminis-tration of its allairs.

tration of its allairs. All the treasury notes and other obligations forming part of the temporary loan are payable and will be redeemed in lawful money: that is to say, in United States notes until after the resumption of specie pay-aents, when they also will doubtless be redeemed in coin, or equivalent notes. The five-twenty sixes being payable twenty years from date, though redeemable after five years, are considered as belonging to the funded or permanent debt, and so also are the twenty years sixes into which the three years seven-thirty notes are convertible. These bonds, therefore, according to the usage of the Government, are physical in coin. The three years seven-thirty treasury notes are part of the temporary loan, and will be paid an United States notes, unless holders prefer conversion to payment. Very respectfully,

S. P. CHASE, Secretary of the Treasury.

This item is thus composed:	
United States Notes, July 17, August 5, 1861, and February 12, 1862	\$781.073.00
" " February 25. July 11, 1862, and July 17, 1863	399 218 927 00
1 This item is thus composed: United States Notes, July 17, August 5, 1861, and February 12, 1862 " " February 25, July 11, 1862, and July 17, 1863 " " in redemption of temporary loan	26 160 569 00
" in redemption of temporary loan Fractional Currency, July 17, 1862, and March 3, 1863	22 210 483 10
	22,210,100 10
and the second se	448.371.052 10
Unpaid requisitions\$50,262,000 00	110,011,008 10
Amount in Treasury11,766,986 40	
	38,495,013 60
	\$186 866 065 70

367

\$486,866,065 70

as follows:

Days.	Increase of Debt per day.	Amount out.
June 30, 1863 92 Sept. 30, 1863 92 Feb. 2, 1864123	\$1,340,000 2,041,000	\$1,098,793.181 1,222,113,559 1,473,225,714
March 2, 1864 29 " 15, 1864 13 May 10, 1864 55 " 14, 1864 4	1,395,000 5,550,000 2,330,000 1,130,629	$\begin{array}{c} 1,513,702,837\\ 1,596,999,429\\ 1,726,248,411\\ 1,730,870,926 \end{array}$

# The Rebel Debt.

December 31, 1862, the receipts of the Treasury from the commencement of the "Permanent Government," (February 18, 1862,) were as follows:

RECEIPTS.	- E	
Patent fund	00	
Customs		
Miscellaneous 2,291,812 (	00	
Repayments of disbursing officers	00	
Interest on loans 26,583	00	
Call loan certificates 59.742.796	00	1
One hundred million loan 41,398,286 (	DO	
Treasury notes	DO	
Interest bearing notes 113,740,000 (		
	00	
War tax         16,664,513           Loan 28th of February, 1861         1,375,476		1
Coin received from Bank of Louisiana 2,539,799		
Com received from bank of Louisiana	_	
Wetel \$157 \$55 704 (	nn	1
Total	ا م	
Total debt up to December 51, 1502 550,105,100	<sup>00</sup>	1
Estimated amount at that date necessary to	0.0	
support the Government to July, 1863, was 357,929,229		
Up to December 31, 1862, the issues of the	ıe l	
	- 1	
Treasury were:	- (	
Notes	00	
Redeemed	50 l	
	- 1	
	- 1	
Outstanding	50	
Outstanding	50	
	=	
From January 1, 1863, to September 30, 186	=	
From January 1, 1863, to September 30, 186	=	
From January 1, 1863, to September 30, 186 the receipts of the Treasury were :	= 3,	
From January 1, 1863, to September 30, 186 the receipts of the Treasury were : For 8 per cent. stock	= 3, 70	
From January 1, 1863, to September 30, 186 the receipts of the Treasury were: For 8 per cent. stock	= 3, 70 70	
From January 1, 1863, to September 30, 186           the receipts of the Treasury were :           For 8 per cent. stock           For 7 per cent. stock           85,757,650           For 6 per cent. stock           6,810,050           6,810,050	= 3, 70 70 00	
From January 1, 1863, to September 30, 186           the receipts of the Treasury were :           For 8 per cent. stock		
From January 1, 1863, to September 30, 186           the receipts of the Treasury were :           For 8 per cent. stock	3, 70 70 00 00	
From January 1, 1863, to September 30, 186           the receipts of the Treasury were :           For 8 per cent. stock           For 7 per cent. stock           For 6 per cent. stock           6510,050           For 5 per cent. stock           22,992,000           For 4 per cent. stock           22,992,000           For 4 per cent. stock           22,992,000           For 4 per cent. stock           2000,000	= 3, 70 70 00 00 00	
From January 1, 1863, to September 30, 186           the receipts of the Treasury were :           For 8 per cent. stock           For 7 per cent. stock           Stor 6 per cent. stock           For 5 per cent. stock           For 6 per cent. stock           For 7 per cent. stock           Stor 7 per cent. stock           Por 4 per cent. stock           Cotton certificates           1ntcrest on loans           140,210		
From January 1, 1863, to September 30, 186           the receipts of the Treasury were :           For 8 per cent. stock.         \$107,292,000           For 6 per cent. stock.         \$8,757,650           For 6 per cent. stock.         6,810,050           For 5 per cent. stock.         22,992,900           For 5 per cent. stock.         22,902,900           Cotton certificates.         2,000,000           Interest on loans.         140,210           War tax         4,122,988	= 3, 70 70 00 00 00 97	
From January 1, 1863, to September 30, 186           the receipts of the Treasury were :           For 8 per cent. stock.         \$107,292,900           For 7 per cent. stock.         \$3,757,650           For 6 per cent. stock.         \$22,992,900           For 4 per cent. stock.         \$22,992,900           For 5 per cent. stock.         \$22,992,900           For 4 per cent. stock.         \$22,992,900           Cotton certificates.         \$2,000,000           Interest on loans.         \$140,210           War tax.         \$4,123,988           Treasury notes.         \$391,633,530	= 3, 70 70 00 00 00 00 97 00 00	
From January 1, 1863, to September 30, 186           the receipts of the Treasury were:           For 8 per cent. stock         \$107,292,900           For 7 per cent. stock         \$3,757,650           For 5 per cent. stock         6,810,050           For 5 per cent. stock         22,902,900           For 5 per cent. stock         22,902,900           For 4 per cent. stock         22,902,900           Ottom certificates         20,000,000           Interest on loans         140,210           War tax         4,128,988           Treasury notes         391,623,550           Sequestration         1,862,550	= 3, 70 70 000 000 000 000 000 000 000 000	
From January 1, 1863, to September 30, 186           the receipts of the Treasury were :           For 8 per cent. stock.         \$107,292,900           For 7 per cent. stock.         \$6,510,550           For 6 per cent. stock.         \$22,992,000           For 4 per cent. stock.         \$22,992,000           For 4 per cent. stock.         \$22,992,000           For 5 per cent. stock.         \$22,992,000           For 4 per cent. stock.         \$2,900,000           Interest on loans.         \$140,210           War tax.         \$4,128,988           Treasury notes.         \$301,623,520           Sequestration.         \$934,798		
From January 1, 1863, to September 30, 186           the receipts of the Treasury were :           For 8 per cent. stock         \$107,292,900           For 7 per cent. stock         \$3,757,650           For 6 per cent. stock         \$2,992,900           For 5 per cent. stock         \$2,992,900           For 4 per cent. stock         \$2,992,900           Interest on loans         \$2,902,900           Interest on loans         \$2,902,900           Yreasury notes         \$31,623,550           Sequestration         \$32,550           Chotoms         \$31,623,550           Sequestration         \$32,750           Sequestration         \$34,769           Storm         \$31,01	= 3, 70 70 00 000 000 000 000 000 000 000 0	
From January 1, 1863, to September 30, 186           the receipts of the Treasury were :           For 8 per cent. stock         \$107,292,900           For 7 per cent. stock         \$8,77,650           For 6 per cent. stock         6,810,050           For 7 per cent. stock         22,992,900           For 4 per cent. stock         22,992,900           For 4 per cent. stock         22,992,900           For 4 per cent. stock         22,992,900           For 5 per cent. stock         4,82,200           Cotton certificates.         2,000,000           Interest on loans         140,210           War tax.         4,128,988           Treasury notes         301,623,550           Sequestration         1862,550           Customs         934,798           Sport duty on cotton         8,101           Patent fund         10,794	= 3, 70 70 00 000 000 000 000 000 000 000 0	
From January 1, 1863, to September 30, 186           the receipts of the Treasury were :           For 8 per cent. stock.           For 7 per cent. stock.           83,757,650           For 6 per cent. stock.           65,810,950           For 4 per cent. stock.           22,992,000           For 4 per cent. stock.           22,992,000           For 4 per cent. stock.           2000,000           Interest on loans.           2000,000           Treasury notes.           Sequestration.           1882,820           91,623,530           931,623,530           934,798           Export duty on cotton.           8,101           Patent fund.           Miscellancous, including repayments by dis-	= 3, 70 70 70 70 000 000 000 000 000 000 00	
From January 1, 1863, to September 30, 186           the receipts of the Treasury were :           For 8 per cent. stock.           For 7 per cent. stock.           83,757,650           For 6 per cent. stock.           65,810,950           For 4 per cent. stock.           22,992,000           For 4 per cent. stock.           22,992,000           For 4 per cent. stock.           2000,000           Interest on loans.           2000,000           Treasury notes.           Sequestration.           1882,820           91,623,530           931,623,530           934,798           Export duty on cotton.           8,101           Patent fund.           Miscellancous, including repayments by dis-	= 3, 70 70 70 70 000 000 000 000 000 000 00	
From January 1, 1863, to September 30, 186           the receipts of the Treasury were :           For 8 per cent. stock         \$107,292,900           For 7 per cent. stock         \$3,757,650           For 6 per cent. stock         \$2,992,900           For 4 per cent. stock         \$2,992,900           Interest on loans         \$149,210           War tax         \$123,988           Streasury notes         \$31,632,550           Sequestration         \$1,862,550           Constoms         \$107,74           Miscellancous, including repayments by disburg officers         \$24,498,217	<b>3</b> , <b>70</b> , <b>70</b> , <b>70</b> , <b>000</b> ,	
From January 1, 1863, to September 30, 186           the receipts of the Treasury were :           For 8 per cent. stock         \$107,292,900           For 7 per cent. stock         \$3,757,650           For 6 per cent. stock         \$2,992,900           For 4 per cent. stock         \$2,992,900           Interest on loans         \$149,210           War tax         \$123,988           Streasury notes         \$31,632,550           Sequestration         \$1,862,550           Constoms         \$107,74           Miscellancous, including repayments by disburg officers         \$24,498,217	<b>3</b> , <b>70</b> , <b>70</b> , <b>70</b> , <b>000</b> ,	
From January 1, 1863, to September 30, 186           the receipts of the Treasury were :           For 8 per cent. stock.           For 7 per cent. stock.           83,757,650           For 6 per cent. stock.           65,810,950           For 4 per cent. stock.           22,992,000           For 4 per cent. stock.           22,992,000           For 4 per cent. stock.           2000,000           Interest on loans.           2000,000           Treasury notes.           Sequestration.           1882,820           91,623,530           931,623,530           934,798           Export duty on cotton.           8,101           Patent fund.           Miscellancous, including repayments by dis-	<b>3</b> , <b>70</b> , <b>70</b> , <b>70</b> , <b>000</b> ,	

EXPENDITURES DURING THAT TIM	
	- S

War Department	\$377,988,244	00
Navy Department	38,437,661	00
Civil, miscellancous, &c	11,629,278	00
Customs		00
Public debt		00
Notes canceled and redeemed	59,011,119	00
Total expenditures	\$519,368,559	00
Total receipts	601,522,893	00
Balance in treasury	\$82,154,334	00

But from this amount is to be deducted the amount of all Treasury notes that have been funded, but which have not ret received a truc estimation, \$65,000,000; total remaining, \$17,154,334.

CONDITION OF THE TREASURY, JANUARY 1, 1864.

Jan. 25-The Secretary of the Treasury (C. G. Memminger) laid before the Senate a statement in reply to a resolution of the 20th, ask-

The increase of debt between periods has been | ing information relative to the funded debt, to call certificates, to non-interest and interestbearing Treasury notes, and other financial matters. From this it appears that, January 1. 1864, the funded debt was as follows:

1, 1864, the funded dept was as for	10.42.
Act Feb. 28, 1861, 8 % cent, \$15,000,000 00 Act May 16, 1861, 8 % cent, \$774,900 00 Act Aug. 19, 1861, 8 % cent, 100,000,000 00 Act April 12, 1862, 8 % cent, 3,612,300 00 Act Feb. 20, 1863, 7 % cent, 63,615,750 00 Act March 23, 1863, 6 % cent, 2,831,700 00	_
Act April 30, 1863 (cotton in- tcrest coupons	\$297,871,650 00 89,206,770 00
Non-interest-bearing Treasury notes out- standing: Act May 16, 1861—payable two years after date	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,
Act Aug. 19, 1861—General currency	-
Act March 23—All denomi- nations	720,898,095 00
ing Amount of Treasury notes under \$5, outstanding Jan. 1, 1864, viz :	102,465,450 00
Act April 17, 1862, denomina- tion of \$1 and \$2         4,860,277 50           Act Oct. 13, 1862, \$1 and \$2         2,344,800 00           Act March 23, 1363, 50 cents, Total under \$5         3,419,000 00	10,424,077 50
Lotal ubuci gommenterine	10,202,011 30

Total debt, Jan. 1, 1864.....\$1,220,866,042 50

## ITS CONDITION, MARCH 31, 1864.

The Register of the Treasury, Robert Tyler, gave a statement, which appeared in the Richmond Sentinel after the passage of the funding law, which gives the amount of outstanding non-interest-bearing Treasury notes, March 31, 1864, as \$796,264,403, as follows :

Act Aug. 19, 1861– Act April 19, 1862– Act Oct. 18, 1862–	Ten-year notes General currency -Ones and twos General currency -General currency	$\substack{\$7,201,375\\154,356,631\\4,516,509\\118,997,321\\511,182,566}$	00 00 50
Total		\$708 961 409	00

Total..... \$796,264,403 00

He also publishes this statement of the issue of non-interest-bearing Treasury notes since the organization of the "Confederate" Government:

Fifty cents         \$911,258         50           Ones         4,882,000         00           Twos         6,086,320         00
Emos 6.056.290.00
Fives
Tens
Twenties
Fifties
Total

# Rebel Financial Legislation.

The following is the funding act:

[From the Richmond Sentinel, Feb. 17, 1864.]

From the richmond sentine, Feb. 11, 100-1 SEc. 1. The Congress of the Confederate States of America do enact, That the holders of all Treasury notes above the denomination of five dollars not bearing interest, shall be allowed until the first day of April, 1804, east of the Mis-sissippi River, and until the first day of July, 1864, west of the Mississippi River, to fund the same, and until the peri-ods and at the places stated, the holders of all such Treas-ury notes shall be allowed to fund the same in registered bonds perable typenty years after they bearing interest at bonds payable twenty years after, they bearing interest at

the rate of four per cent. per annum, payable on the first

day of January and July of each year. Szc. 2. The Secretary of the Treasury is hereby author-ized to issue the bonds required for the funding provided for in the preceding section; and until the bonds can be provided, he may issue certificates to answer the purpose. Such bonds and certificates shall be receivable without interest, in payment of all Government dues payable in the year 1864, except export and import duties. SEC. 3. That all Treasury notes of all denominations of

Sec. 3. That all Treasury notes of all denominations of ome hundred dollars, not bearing interest, which shall not be presented for funding under the provisions of the first section of this act, shall, from and after the first day of April, 1864, east of the Mississippi River, and the first day of July, 1864, west of the Mississippi cease to be receivaof July, 1864, west of the Mississippi, case to be receiva-ble in payment of public dues, and said notes, if not so presented at the time, shall, in addition to the tax of thirty-three and one-third cents imposed in the fourth section of this act, be subject to a tax of ten per cent, per month until so presented; which taxes shall attach to said notes wherever circulated, and shall be deducted from the face of said notes wherever presented for payment or for fund-ing, and such notes shall not be exchangeable for the new issue of Treasury notes provided for in this act. SEC. 4. That on all the said Treasury notes not funded or sed in navment of taxes at the dates and places presented

used in payment of taxes at the dates and places prescribed in the first section of this act, there shall be lovied at said dates and places a tax of 33% cents for every dollar funded on the face of said notes. Said tax shall attach to said notes wherever circulated, and shall be collected by deducting the same at the Treasury, its depositories and by the collec-tors, and by all Government officers receiving the same, tors, and by all Government officers receiving the same, wherever presented for payment, or for funding, or in pay-ment of Government dues, or for postage or in exchange for new notes as hereinafter provided, and said Treasury notes shall be fundable in bonds as provided in the first section of this act, until the first day of January, 1865, at the rate of sixty-six cents and two thirds on the dollar, and it shall be the duty of the Secretary of the Treasury at any time between the first of April east and the first of January, 1865, west of the Mississippi river, and the first of January, 1865, to substitute and exchange new Treasury notes for same, at the rate of sixty-six and two thirds cents on the dollar: *Promided*. That notes of the denomination of 5100 shall not **Provided**, That notes of the denomination of \$100 shall not be entitled to the privilege of said exchange: Provided, further, That on the right to fund all such Treasury notes which may remain outstanding on the first day of January, 1863, and which may not be exchanged for new Treasury notes, as herein provided, a tax of one hundred per cent is hereby

SEC. 5. That after the 1st day of April next, all authority beretofore given to the Secretary of the Treasury to issue Treasury notes shall be, and is hereby, revoked, provided the Secretary of the Treasury may after that time issue new Treasury notes in such forms as he may prescribe, payable two years after the ratification of a treaty of peace with the United States, said new issue to be receivable in payment of all public dues except export and import duties, and to be issued in exchange for old notes at the rate of \$2 of the new issue for \$3 of the old issues, whether said old notes be surrendered for exchange by the holders thereof or be received into the Treasury under the provisions of this act; and the holders of the new notes or of the old notes, except those of the denomination of \$100, after they are reduced to 66% cents on the dollar by the tax aforesaid, may convort them into call certificates bearing interest at the rate of four per cent per annum, and payable two years after a ratification of a treaty of peace with the United States, un-less sooner converted into new notes.

SEC. 6. That to pay the expenses of the Government, not otherwise provided for, the Secretary of the Treasury is hereby authorized to issue six per cent bonds to an amount hereby authorized to issue six per cent bonds to an amount not exceeding \$500,000,000, the principal and interest where-of shall be free from taxation, and for the payment of the interest thereon, the entire net receipts of any export duty hereinafter laid on the value of all cotton, tobacco, and naval stores, which shall be exported from the Confederate States, and the net proceeds of the import duties now laid, or so much thereof as may be necessary to pay the interest are herchy specially pledged: *Provided*, That the duties now laid upon imports, and hereby pledged, shall hereafter be paid in specie or in sterling exchange, or in the coupons of said bonds. said bonds.

Said bonds. SEC, 12. That any State holding Treasury notes received before the times herein fixed for taxing said notes shall be allowed until the 1st day of January, 1865, to fund the same in six per cent, bonds of the Confederate States, payable twenty years after date, and the interest payable semi-an-nually. But all Treasury notes received by any State after the time definition between a comparing table to held the time fixed for taxing the same, as aforesaid, shall be held to have been received, diminished by the amount of said tax. The discrimination between the notes subject to the tax and those not so subject shall be left to the good faith of 24

.

each State, and the certificate of the Governor thereof shall in each case be conclusive. SEC. 13. That Treasury notes heretofore issued, bearing

interest at the rate of seven dollars and thirty cents on the Interest at the rate of seven dollars and thirty cents on the hundred dollars per annun, shall no longer bo received in payment of public dues, but shall be deemed and considered bonds of the Confederate States, payable two years after the ratification of a treaty of peace with the United States, bearing the rate of interest specified on their face, payable the 1st of January in each and every year.

## SPECULATIONS ON THE FUNDING.

The Richmond *Examiner* of the 11th of April, 1864, gives the following statistics of the rebel currency, from which it will be seen that a desperate attempt is making to retrieve the financial disasters of the South :

The depletion of the Confederate currency under the recent legislation is much greater than is generally supposed: and in this connection it will be interesting to refer to well established figures. The entire issue of the old circulation we may take at \$\$00,000,000. The number of one hundred dollar bills in circulation has been about \$250,000,000. Of lesser denominations that will be funded, there are, at lesser states at \$50,000,000. Deduct now the \$300,000,000 funded, and we have \$500,000,000. This, reduced by the discount of thirty-three and one-third per cent., will, in round numbers, leave us \$330,000,000.

The tax levied for 1864 is estimated considerably above \$40,000,000. There being only \$320,000,000 funded in four per cents., it follows that \$100,000,000 of currency must be used in addition to the above for the payment of taxes for 1864, which will still further reduce the circulating medium to \$230,000,000. From the last named sum there must be to \$230,000,000. From the last memod sum there must be substracted the amount required to pay the additional taxes imposed by the late Congress on the income tax of 1863, as well as some portion of the old taxes that will not be paid on the first of April, 1864. The circulation would thus be reduced to \$200,000,000, without reference to the

But here comes up the important question of the mew issue, which involves the vitality of the whole scheme. The first interpretation of the currency act was that it de-nied power to the Sceretary of the Treasury to issue one nied power to the Secretary of the Treasury to issue one dollar notes except in exchange for the present currency at the rate of two dollars of the new for three dollars of the old, which may remain unfunded on the first of April. Others construct he act to empower the Secretary of the Treasury to issue two dollars of the new issue for three dollars of the old, whether funded or unfunded—whether exchanged or model to for targe exchanged or paid in for taxes.

The latter construction is said to be favored by Mr. Memminger, namely, that he is authorized to issue new notes minger, namely, that he is authorized to issue new notes to the amount of two-thirds of the whole of the old issue. In other words, supposing the old notes in circulation amounted to \$\$00,000 the first of April, the Secretary of the Treasury is empowered to issue two-thirds of this amount, that is, \$533,333,333, affording a supply to the Treasury for about eight months, irrespective of the sum that may be raised by the sale of six per cent, bonds.

## [From the Richmond Sentinel.]

There is but little doubt that the funding, east and west f the Mississippi, will amount to \$300,000,000. The of total issue outstanding March 31 is thereby reduced say to \$485,000,000. Of this, a considerable portion, probably \$100,000,000, is in \$100 notes. Excluding these, we have \$100,000,000, is in \$100 notes. Excluding these, we have \$385,000,000 left, of which the issues of \$5 and under amount to \$90,969,\$98 50. Suppose \$\$5,000,000 of these are now in circulation, and we have for all others \$200,000,000, which the tax of one third has reduced to \$200,000,000. The total circulation at the present time, irrespective of the new issue, is therefore, largely less than \$300,000,000, and of this a con-iderable ouncer to elument to be used is the bandle of the siderable amount is always to be found in the hands of the

disbursing officers and depositaries. We have not included in the above any estimates of the amount of circulation lost or destroyed, and thereby gained to the Government. It is doubtless considerable.

The amount of currency which has been canceled and destroyed (irrespective of the operations of the present cur-rency law) is nearly twenty per cent. of the whole issue. If this reduction be applied to the five dollar notes, the amount of these in circulation would appear to be \$63,-272,252,

# [From The Examiner of the 21st March, 1864.]

Only ten days now intervene hefore the currency remaining in circulation is taxed *one-third*, and, consequently, during the week commencing Monday, the holders of Treasury notes *must* decide whether they will keep them until the lst of April and submit to the Government shave, fund the amount in four per cent. bonds, or exchange it for bonds or personal property. The necessity of coming to a conclusion is "sharpening the wits" of the people, though it is not improbable that some who esteem themselves

"wondrons wise " in financial matters will commit a blunder in disposing of their surplus cash. The ability to penetrate the future is a power which few, if any, possess, penetrate the future is a power winen few, if any, possess, and hence the views expressed in regard to the effect of the financial legislation of Congress, after the currency is re-duced, are diverse and vaguely theoretical. Everybody knows that a \$10 note, after 1st April, will only represent a net value of \$0.66%, and accordingly the universal desire is to dispose of the currency in hand at

this time, so as to avoid this apparent loss, very few being willing to hold it with the expectation that \$66 of the new currency will, in a few weeks, buy more of any article than \$100 will purchase now. They must see the fact before they will believe it, but then it will be too late to profit by the development.

In the meantime, all kinds of " cornering " processes are in vogue, and it must be admitted that some of them are plausible enough. For instance, it is argued that certain bonds and stocks may be bought now, and sold in the new currency at a decline not exceeding fifteen to twenty-five per cent .- thereby saving ten to fifteen in the transmutory fax

On the last day of the session, June 15, President Davis vetoed the bill allowing further time to persons within the enemy's lines to fund their Treasury notes. The consideration of the subject was postponed until the next session.

## RICHMOND, April 22, 1864.

The ontstanding amount of Treasury notes, of the denomination of five dollars and under, is about seventy-five millions.

The funding returns sum up an aggregate of two hundred and thirty-seven million dollars. Twenty small depositories yet to hear from. The only State from which complete re-turns have been received is Georgia, where the amount funded is seventy-two millions one hundred and eighty-four thousand dollars.

## TAXATION.

## THE TAX ACT OF JULY, 1861.

The Richmond Enquirer gives the following summary of the act authorizing the issue of Treasury notes and bonds, and providing a war tax for their redemption :

Section one authorizes the issue of Treasury notes, paya-ble to bearer at the expiration of six months after the ratification of a treaty of peace between the Confederate States and the United States. The notes are not to be of a less and the United States. The notes are not to be of a res-denomination than five dollars, to be releasure, to be received in payment of all public dues, except the export duty on cotton, and the whole issue outstanding at one time, including the amount issued under former acts, are not to exceed one hundred millions of dollars.

Section two provides that, for the purpose of funding the said notes, or for the purpose of purchasing specie or mili-tary stores, &c., bonds may be issued, payable not more than twenty years after date, to the amount of one hundred millions of dollars, and bearing an interest of eight per cent. per annum. This amount includes the thirty millions already authorized to be issued. The bonds are not to be issued in less amounts than \$100, except when the subscription is for a less amount, when they may be issued as low as \$50.

Section three provides that holders of Treasury notes

Section three provides that holders of Treasury notes may at any time exchange them for bonds. Section four provides that, for the special purpose of paying the principal and interest of the public debt, and of supporting the Government, a war tax shall be assessed and levied of fifty cents upon each one hundred dollars in value of the following property in the Confederate States, namely: Real estate of all kinds; slaves; merchandise; mans stocks; railwead and other conversion stocks; man bank stocks; railroad and other corporation stocks; money at interest or invested by individuals in the purchase of bills, notes, and other securities for money, except the bonds of the Confederate States of America, and cash on bonds of the Concuerate States of America, and cash of hand or on deposit in bank or elsewhere; cattle, horses, and nules; gold watches, gold and silver plate; pianos and pleasure carriages: *Provided*, *however*. That when the taxable property, herein above conuncrated, of any head of a family is of value less than five hundred dollars, such taxable property shall be exempt from taxation under this act. It provides further that the property of colleges, schools, and religious associations shall be exempt.

The remaining sections provide for the collection of the tax.

## THE TAX ACT OF DECEMBER 19, 1861.

An act supplementary to an act to authorize the issue of Treasury notes, and to provide a war tax for their redemption.

SEC. 1. The Congress of the Confederate States of Amer-ica do enact, That the Secretary of the Treasury is hereby authorized to pay over to the several banks, which have made advances to the Government, in anticipation of the issue of Treasury notes, a sufficient amount, not exceeding \$10,000,000, for the principal and interest due upon the said advance, according to the engagements made with them. SEC. 2. The time fixed by the said act for making assign-

ments is hereby extended to the 1st day of January next, and the time for the completion and delivery of the lists is extended to the 1st day of March next, and the time for the report of the said lists to the chief collector is extended to the lst day of May next; and in cases where the time thus fixed shall be found insufficient, the Secretary of the Treasury shall have power to make further extension, as circumstances may require.

SEC. 3. The cash on hand, or on deposit in the bank, or elsewhere, mentioned in the fourth section of said act, is hereby declared to be subject to assessment and taxation, and the money at interest, or invested by individuals in the purchase of bills, notes, and other securities for money, shall be deemed to include securities for money belonging to non-residents, and such securities shall be returned, and the tax thereon paid by any agent or trustee having the same in possession or under his control. The term merchandise shall be construed to include merchandise belonging to any non-resident, and the property shall be returned, and the tax paid by any person having the same in posses-sion as agent, attorney, or consignee: *Provided*, That the words "money at interest," as used in the act to which this act is an amendment, shall be so construed as to include all notes, or other evidences of debt, bearing interest, without reference to the consideration of the same. The exception allowed by the twentieth section for agricultural products shall be construed to embrace such products only when in the hands of the producers, or held for his account. But no tax shall be assessed or levied on any money at interest when the notes, bond, bill, or other security taken for its payment, shall be worthless from the insolvency and total inability to pay of the payor or obligor, or person liable to make such payment; and all securities for money payable under this act shall be assessed according to their value, and the assessor shall have the same power to ascertain the value of such securities as the law confers upon him with respect to other property.

SEC. 4. That an amount of money, not exceeding \$25,000, shall be and the same is hereby appropriated, out of any shall be and the same is hereby appropriated, out of any money in the treasury not otherwise appropriated, to be disbursed under the authority of the Secretary of the Treasury, to the chief State tax collectors, for such expenses as shall be actually incurred for salaries of clorks, office hire, stationary, and incidental charges; but the books and printing required shall be at the expense of the department, and unbine to the composed. and subject to its approval.

SEC. 5. The lien for the tax shall attach from the date of the assessment, and shall follow the same into every State in the Confederacy; and in case any person shall attempt to remove any property which may be liable to tax, beyond the jurisdiction of the State in which the tax is payable, without payment of the tax, the collector of the district may distrain upon and sell the same, in the same manner as is provided in cases where delault is made in the payment of the tax.

SEC. 6. On the report of any chief collector, that any county, town or district, or any part thereof, is occupied by the public enemy, or has been so occupied as to occasion destruction of crops or property, the Secretary of the Treasury may suspend the collection of tax in such region until the same can be reported to Congress, and its action had thereon.

SEC. 7. In case any of the Confederate States shall undertake to pay the tax to be collected within its limits before the time at which the district collectors shall enter upon the discharge of their duties, the Secretary of the Treasury may suspend the appointment of such collectors, and may direct the chief collector to appoint assessors, and to take proper measures for the making and perfecting the returns, assessments and lists required by law; and the returns, assessments and and lists so made, shall have the same legal validity, to all intents and purposes, as if made according to the provisions of the act to which this act is sup-

plementary. SEC. 8, That tax lists already given, varying from the provisions of this act, shall be corrected so as to conform thereto.

THE TAX ACT OF APRIL 24, 1863.

[From the Richmond Whig, April 21.]

We present below a synopsis of the bill to lay taxes for

the common defence and to carry on the government of the Confederate States, which has passed both branches of Con-gress. It is substantially the bill proposed by the commit-tee on conference:

1. The first section imposes a tax of eight per cent. upon the value of all naval stores, salt, wines and spirituous liquors, tobacco, manufactured or unmanufactured, cotton, liquors, tobacco, manufactured or unmanufactured, cotton, wool, flour, sugar, molasses, syrup, rice, and other agricul-tural products, held or owned on the 1st day of July next, and not necessary for family consumption for the unex-pired portion of the year 1863; and ot the growth or pro-duction of any year preceding the year 1863; and a tax of one per cent. upon all moneys, bank notes or other curren-cy on hand or on deposit on the 1st day of July next, and on the value of all credits on which the interest has not been paid, and not employed in a business, the income derived from which is taxed under the provisions of this act: Pro-vided, That all moneys owned, held or deposited beyond the limits of the Confederate States shall be valued at the the limits of the Confederate States shall be valued at the corrent rate of exchange in Confederate treasury notes. The tax to be assessed on the 1st day of July and collected on the 1st day of October next, or as soon thereafter as may be practicable.

2. Every person engaged, or intending to engage, in any business named in the fifth section, shall, within sixty days after the passage of the act, or at the time of beginning Dusiness, and on the 1st of January in each year thereafter, register with the district collector a true account of the name and residence of each person, firm, or corporation engaged or interested in the business, with a statement of the time for which, and the place and manner in which the same is to be conducted, &c. At the time of the registry there shall be paid the specific tax for the year ending on the next 31st of December, and such other tax as may be due upon sales or receipts in such business.

3. Any person failing to make such registry and pay such tax, shall, in addition to all other taxes upon his business imposed by the act, pay double the amount of the specific tax on such business, and a like sum for every thirty days of such failure.

4. Requires a separate registry and tax for each business mentioned in the fifth section, and for each place of conducting the same; but no tax for mere storage of goods at a place other than the registered place of business. A new registry required upon every charge in the place of con-ducting a registered business, upon the death of any person conducting the same, or upon the transfer of the business to another, but no additional tax.

5. Imposing the following taxes for the year ending 31st of December, 1863, and for each year thereafter :

Bankers shall pay \$500. Auctioneers, retail dealers, tobacconists, pedlers, cattle brokers, apothecaries, photographers, and confectioners, \$50, and two and a half per centum on the gross amount of sales made.

Wholesale dealers in liquors, \$200, and five per centum on gross amount of sales. Retail dealers in liquors, \$100, and ten per centum on gross amount of sales.

Wholesale dealers in groceries, goods, wares, merchan-dise, &c., \$200, and two and a half per centum. Pawnbrokers, money and exchange brokers, \$200. Distillers, \$200, and twenty per centum. Brewers, \$100,

and two and a half per centum.

Hotels, inns, taverns, and eating-houses, first class, \$500; second class, \$300; third class, \$200; fourth class, \$100; fifth scalas, S20. Every house where food or refreshments are sold, and every boarding house where there shall be six boarders or more, shall be deemed an eating house under this act.

Commercial brokers or commission merchants, \$200, and

Commercial products or commission increments, yave, and two and a half per centum. Theatres, \$500, and five per centum on all receipts. Each circus, \$100, and \$10 for each exhibition. Jugglers and other persons exhibiting shows, \$50. Bowling alleys and billiard rooms, \$40 for each alley or

table registered

Livery stable keepers, lawyers, physicians, surgeons, and dentists, \$50.

Butchers and bakers, \$50, and one per centum. 6. Every person registered and taxed is required to make returns of the gross amount of sales from the passage of the act to the 30th June, and every three months thereafter.

7. A tax upon all salaries, except of persons in the military or naval service, of one per cent. when not exceeding \$1,500, and two per cent. upon an excess over that amount: Provided, That no taxes shall be imposed by virtue of this act on the salary of any person receiving a salary not exceeding \$1,000 per aunum, or at a like rate for another period of time, longer or shorter.

8. Provides that the tax on annual incomes, between \$500 and \$1,500, shall be five per cent.; between \$1,500 and \$3,000, five per cent. on the first \$1,500 and ten per cent. on the excess; between \$3,000 and \$5,000, ten per cent;

between \$5,000 and \$10,000, twelve and a half per cent.; over \$10,000, fifteen per cent., subject to the following de-ductions: On incomes derived from rents of real estate, manufacturing, and uning establishments, e., a sum suffi-cient for necessary annual repairs; on incomes from any mining or manufacturing business, the rent, (if rented,) cost of labor actually hired, and raw material; on incomes from of halor actuary mires, and raw materiar, or moments from navigating enterprizes, the hire of the vessel, or allowance for wear and tear of the same, not exceeding ten per cent; on incomes derived from the sale of merchandise or any other property, the prime cost, cost of transportation, sala-ries of clerks, and rent of buildings; on incomes from any other occupation, the salaries of clerks, rent, cost of labor, material, &c; and in case of mutual insurance companies, the amount of losses paid by them during the year. Incomes derived from other sources are subject to no deductions whatever.

All joint stock companies and corporations shall pay one tenth of the dividend and reserved fund annually. If the annual earnings shall give a profit of more than ten and less than twenty per cent. on capital stock, one eighth to be paid; if more than twenty per cent, one sixth. The tax to be collected on the 1st of January next, and of each year thereafter.

9. Relates to estimates and deductions, investigations,

referees, &c. 10. A tax of ten per cent. on all profits in 1862 by tho purchase and sale of flour, corn, hacon, pork, oats, hay, rice, salt, irou or the manufactures of iron, sugar, molasses made of cane, butter, woolen cloths, shoes, boots, blankets, and cotton cloths. Does not apply to regular retail business.

11. Each farmer, after reserving for his own use fifty bushels sweet and fifty bushels Irish potatoes, one hundred bushels corn or fifty bushels wheat produced this year, shall pay and deliver to the Confederate Government one tenth of the grain, potatoes, forage, sugar, molasses, cotton, wool, and tobacco produced. After reserving twenty bushels peas or beans he shall deliver one tenth thereof.

12. Every farmer, planter, or grazier, one tenth of the hogs slaughtered by him, in cured bacon, at the rate of sixty pounds of bacon to one hundred pounds of pork; one per cent. upon the value of all neat cattle, horses, mules, not used in cultivation, and asses, to be paid by the owners of the same; beeves sold to be taxed as income.

13. Gives in detail the duties of post quartermasters under the act.

14. Relates to the duties of assessors and collectors.

 Ideates trustees, guardians, &c., responsible for faxes due from estates, &c., under their control.
 Exempts the income and moneys of hospitals, asy-lums, churches, schools, and colleges from taxation under the act.

17. Authorizes the Secretary of the Treasury to make all rules and regulations necessary to the operation of the act.

18. Provides that the act shall be in force for two years To Provides that the act shart be in force for two years from the expiration of the present year, unless sooner re-pealed; that the tax on naval stores, flour, wool, cotton, tobacco, and other agricultural products of the growth of any year preceding 1863, imposed in the first section, shall be levied and collected only for the present year.

The tax act of February 17, 1864, levies, in addition to the above rates, the following, as stated in the Richmond Sentinel of February, 1864:

SEC. 1. Upon the value of real, personal, and mixed prop-erty, of every kind and description, except the exemptions hereafter to be named, five per cent.; the tax leviced on property employed in agriculture to be credited by the value of property in kind.\*

\* This is the section in full:

\* This is the section in full: That on the 1st day of Jannary, 1863, there shall be levied and assessed on each person residing in the Confederate States, for the support of the Government and the defence of the country, the following tax, to wit: One fifth the value of all the wheat, corn, rice, rye, oats, potatoes, hemp, flax, peas, beans, barley, hay, wool, rosin, tar, pitch, turpentine, cotton, sugar, mokases, and tobacco produced by him in these States during the previous calendar year; also, ono fifth of the value of the increase for the preceding calendar year of the horses, asses, cattle, sheep, and also, nith of the value of the increase for the preceding calculative year of the horses, assess, cattle, sheep, and swine; and also, one fifth of the profits made in the preceding calcular year of the feeding of swine, sheep, cattle, or mules; also, oue fifth of each person's yearly income for the preceding cal-endar year, from all sources whatsoever, except from the sources hereinafter described, and except from the interest or One found horizont is a stream protocol of the sources of the sources are stream protocol. on Confederate bonds, certificates, or treasury notes; Pro-vided, That said tax so levied and assessed shall be due and payable on the 1st day of April, 1863. Provided further,

On gold and silver ware, plate, jewels, and watches, ten ] per cent.

The tax to be levied on the value of property in 1860, except in the case of land, slaves, cotton, and tobacco, pur-chased since January 1st, 1862, upon which the tax shall be levied on the price paid. SEC. 2. A tax of five per cent on the value of all shares in

joint stock companies of any kind, whether incorporated or not. The shares to be valued at their market value at the time of assessment.

SEC. 3. Upon the market value of gold and silver coin or bullion, five per cent.; also the same upon moneys held abroad, or all bills of exchange drawn therefor. A tax of five per cent. on all solvent credits, and on all

bank bills and papers used as currency, except non-interest bearing Confederate Treasury notes, and not employed in a registered business taxed twenty-five per cent.

SEC. 4. Profits in trade and business taxed as follows:

On the purchase and sale of agricultural products and mercantile wares generally, from January 1, 1863, to Jan-uary 1, 1865, ten per cent in addition to the tax under the act of April 24, 1863.

The same on the purchase and sale of coin, exchange, stocks, notes, and credits of any kind, and any property not

Bucks, note, and require a standard sta on such excess.

on such excess. SEC. 5. The following arc exempted from taxation: Five hundred dollars' worth of property for each head of a family, and a hundred dollars additional for each minor child; and for each son in the army or navy, or who has fallen in the service, and a member of the family when he enlisted, the further sum of \$500.

One thousand dollars of the property of the widow or mi-nor children of any officer, soldier, sailor, or marine, who has died in the service.

A like amount of property of any officer, soldier, sailor, or marine, engaged in the service, or who has been disabled therein, provided said property, exclusive of furniture, does not exceed in value \$1,000. When property has been injured or destroyed by the

enemy, or the owner unable temporarily to use or occupy it by reason of the presence or proximity of the enemy, the assessment may be reduced in proportion to the damage sustained by the owner, and the tax in the same ratio by the district collector.

the district collector. SEC. 6. The taxes on property for 1864 to be assessed as on the day of the passage of this act, and collected the 1st of June next, with ninety days extension west of the Missis-sippi. The additional tax on incomes, &c., for 1863, to be paid forthwith; the tax on incomes, &c., for 1864, to be collected according to the acts of 1863.

SEC.7. Exempts from tax on income for 1864, all property herein taxed *ad valorem*. The tax on Confederate bonds in no case to exceed the interest payable on the same; and said bunds exempt from tax when held by minors or lunatics, if the interest do not exceed one thousand dollars.

### THE TAX LAW.

4

We learn that, according to the construction of the re-cent tax law in the Treasury Department, tax payers will be required to state the articles and objects subjected to a specific or ad valorem tax, held, owned, or possessed by them on the 17th day of February, 1864, the date of the act.

The daily wages of detailed soldiers and other employés of the Government are not liable to taxation as income, although they may amount, in the aggregate, to the sum of \$1,000 per annum.

A tax additional to both the above was imposed as follows, June 1, 1864:

 $\Lambda$  bill to provide supplies for the army, and to prescribe the mode of making impressments.\*

SEC. 1. The Congress of the Confederate States of America do enact, Every person required to pay a tax in kind, under

That foreigners resident within the Confederate States shall not be required to pay, except from the aforesaid articles produced by or for them, or from income or profits derived from business conducted by them within those States; nor How business contained up then wright note batters, not shall any tax be leviced upon the produce of residents where the total value of such products during said years is less than §500; nor shall any tax be leviced on the income of residents where the total value of such income is less than

the provisions of the "Act to lay taxes for the common defense and carry on the Government of the Confederate States," approved April 24, 1863, and the act amendatory thereof, approved February 17, 1864, shall, in addition to the one touth required by said acts to be paid as a tax in kind, deliver to the Confederate Government, of the pro-ducts of the present year and of the year 1865, one other tenth of the several products taxed in kind by the acts aforesaid, which additional one truth shall be assortained, assessed and collected, in all respects, as is provided by law assessed and collected, in all respects, as is provided by law for the said tax in kind, and shall be paid for, on delivery, by the Post Quartermasters in the several districts at the assessed value thereof, except that payment for cotton and tobacco shall be made by the agents of the Treasury Do-partment appointed to receive the same.

SEC. 2. The supplies necessary to the support of the pro-ducer and his family, and to carry on his ordinary business, shall be exempted from the contribution required by the preceding section, and from the additional impressments authorized by the act: *Provided*, however, That nothing herein contained shall be construed to repeal or affect the provisions of an act entitled "An act to authorize the im-pressment of meat for the use of the army, under certain of any article or product so necessary cannot be agreed upon between the assessor and the producer, it shall be as-certained and determined by disinterested freeholders of the vicinage, as is provided in cases of disagreement as to be estimates and assessments of tax in kind. If required by the assessor, such freeholder shall ascertain whether a producer, who is found unable to furnish the additional one tenth of any one product, cannot supply the deficiency by SEC. 2. The supplies necessary to the support of the prothe delivery of an equivalent in other products, and upon what terms such commutation shall be made. Any com-mutation thus awarded shall be enforced and collected, in all respects, as is provided for any other contribution re-quired by this act. SEC. 3. The Secretary of War may, at his discretion, de-cline to assess, or, after assessment, may decline to collect the whole or any part of the additional one tenth herein provided for, in any district or headity - add is be be

provided for, in any district or locality, and it shall be his duty promptly to give notice of any such determination, specifying, with reasonable certainty, the district or locality and the product, or the proportion thereof, as to which he so declines.

SEC. 4. The products received for the contribution herein required, shall be disposed of and accounted for in the same manner as those received for the tax in kind; and the Secremanner as those received for the tax in kind; and the Secre-tary of War may, whenever the exigencies of the public service will allow, anthorize the sale of products received from either source, to public officers or agents charged in any Stato with the duty of providing for the families of sol-diers. Such sale shall be at the prices paid or assessed for the products sold, including the actual cost of collec-tions. tions.

SEC. 5. If, in addition to the tax in kind and the contri-SEC. 5. If, in addition to the tax in kind and the contri-bution herein required, the necessities of the army or the good of the service shall require other supplies of food or forage, or any other private property, and the same can-not be procured by contract, then impressments may be made of such supplies or other property, either for absolute ownership or for temporary use, as the public nocessities may require. Such impressments shall be made in accord-ance with the provisions, and subject to the restrictions of the existing impressment laws, except so far as is herein otherwise provided. SEC. 6. The right and the duty of making impressments is hereby confided exclusively to the officers and agents

SEC. 6. The right and the duity of making impressments is hereby confided exclusively to the officers and agents charged in the several districts with the assessment and col-lection of the tax in kind and of the contribution herein required; and all officers and soldiers in any department of the army are hereby expressly prohibited from undertak-ing in any manner to interfere with these officers and agents in any part of their duits in respect to the tax in kind, the contribution, or the impressment herein provided for: *Provided*. That this rechibition shall not be applicable and, the control of the impression in the impression provided for  $\cdot Provided$ , That this prohibition shall not be applicable to any district, county, or parish in which there shall be no officer or agent charged with the appointment and collec-tion of the tax in kind.

SEC. 7. Supplies or other property taken by impressment shall be paid for by the post quartermasters in the several districts, and shall be disposed of and accounted for by them as is required in respect to the tax in kind and the contri-bution herein required; and it shall be the duty of the post quartermasters to equalize and apportion the impressments within their districts, as far as practicable, so as to avoid oppressing any portion of the community.

residents where the total value of such income is less than \$500. \* The Georgia Supreme Court has made an important de-cisicu in the case of impressment of sugar in the hands of a merchant. One of the points was that "the Congress of made or tendered to the owner."

SEG. 8. If any one not anthorized by law to collect the tax in kind or the contribution herein required, or to make impressments, shall undertake, on any pretence of such authority, to seize or impress, or to collect or receive any such property, or shall, on any such pretence, actually ob-tain such property, he shall, upon conviction thereof, he promished by fine not exceeding five times the value of such property, and be imprisoned not exceeding five years, at the discrition of the court having jurisdiction. And it the assessment and collection of the tax in kind and of the contribution herein required, promptly to report, through the post quartermasters in the several districts, any viola

tion or disregard of the provisions of this act by any officer or soldier in the service of the Confederate States. SEO. 9. That it shall not be lawful to imprese any sheep, milch cows, brood marce, stud horses, jacks, bulls, or other stock kept or necessary for raising horses, mules, or cattle.

The following is the vote by which the bill passed the Senate:

YEAS - Messre. Caperton, Graham, Haynes, Jemison, John-son (Ark.), Johnson (Mo.), Mitchell, Orr, Walker, Watson-

10. NAVS-Messrs. Baker, Burnett, Henry, Hunter, Maxwell,

ţ

# MISCELLANEOUS.

1862, August 14-The President received a deputation of colored persons relative to emigration. The interview is thus reported :

WASHINGTON, THURSDAY, August 14, 1862. This afternoon the President of the United States gave an andience to a committee of colored men at the White House. They were introduced by Rev. J. Mitchell, Commissioner of Emigration. E. M. Thomas, the chairman, remarked that they were there by invitation to hear what the Executive had to say to them.

had to say to them. Having all been scated, the President, after a few pre-liminary observations, informed them that a sum of money had been appropriated by Congress, and placed at his dispo-sition, for the purpose of aiding the colonization in some country of the people, or a portion of them, of African des-cent, thereby making it his duty, as it had for a long time been his inclination, to favor that cause; and why, he asked, should the people of your race be colonized, and where? Why should they leave this country? This is, perhaps, the first question for proper consideration. You and we are first question for proper consideration. You and we are different races. We have between us a broader difference than exists between almost any other two races. Whether it is right or wrong I need not discuss; but this physical difference is a great disadvantage to us both, as I think. Your race suffer very greatly, many of them by living among us, while ours suffer from your presence. In a word we suffer on each side. If this is admitted, it affords a reason, at least, why we should be separated. You here are freemen, I suppose. A VOICE-Yes, sir.

The PRESIDENT-Perhaps you have long been free, or all The PRESERVA-Perhaps you have long been free, of all your lives. Your race are suffering, in my judgment, the greatest wrong inflicted on any people. But even when you cease to be slaves, you are yet far removed from being placed on an equality with the white race. You are cut off from many of the advantages which the other race enjoys. The aspiration of men is to enjoy equality with the best The aspiration of men is to enjoy equality with the best when free, but on this broad continent not a single man of your race is made the equal of a single man of ours. Go where you are treated the best, and the ban is still upon you. I do not propose to discuss this, but to present it as a fact, with which we have to deal. I cannot alter it if would. It is a fact about which we all think and feel alike, I and you. We look to our condition. Owing to the exist-ence of the two races on this continent, I need not recount to you the effects upon white men growing ont of the instience of the two races on this continent, **I** need not recount to you the effects upon white men, growing out of the insti-tution of slavery. I believe in its general evil effects on the white race. See our present condition—the country en-gaged in war! our white men cutting one another's throats —none knowing how far it will extend—and then consider what we know to be the truth. But for your race among us there could not be war, although many men engaged on either side do not care for you one way or the other. Never-theless, I repeat, without the institution of slavery, and the colored race as a basis, the war could not have an existence. It is better for us both, therefore, to be senarated. I know It is better for us both, therefore, to be separated. I know It is better for us both, therefore, to be separated. I know that there are free men among you who, even if they could better their condition, are not as much inclined to go out of the country as those who, being slaves, could obtain their freedom on this condition. I suppose one of the principal difficulties in the way of colonization is that the free col-ored man cannot see that his comfort would be advanced by it. You may believe that you can live in Washington, or elsewhere in the United States, the remainder of your life: nerheas more so than you can in any foreign country. life; perhaps more so than you can in any foreign country,

The President on Colonization, in Au-gust, 1862. and hence you may come to the conclusion that you have nothing to do with the idea of going to a foreign country. This is (I speak in no unkind sense) an extremely selfsh view of the case. But you ought to do something to help those who are not so fortunate as yourselves. There is an unwillingness on the part of our people, harsh as it may be, for you free colored people to remain with us. Now if you could give a start to the white people you would open a wide door for many to be made free. If we deal with those who are not free at the beginning, and whose intellects are clouded by slavery, we have very poor material to start with. If intelligent colored men, such as are before me, would move in this matter, much might be accomplished. Would move in this matter, much might be accomplished. It is exceedingly important that we have men at the begin-ning capable of thinking as white men, and not those who have been systematically oppressed. There is much to en-courage you. For the sake of your race you should sacrifice something of your present comfort for the purpose of being as grand in that respect as the white people. It is a cheeras grade in throughout life, that something can be done to ameliorate the condition of those who have been subject to the hard usages of the world. It is difficult to make a man miscrable while he feels he is worthy of himself and claims miserable while he feels he is worthy of himself and claims kindred to the great God who made him. In the American Revolutionary war sacrifices were made by men engaged in it, but they were cheered by the future. General Washing-ton himself endured greater physical hardships than if he had remained a British subject, yet he was a happy man, because he was engaged in benefiting his race, in doing something for the children of his neighbors, having none of bis arm of his own

The colory of Liberia has been in existence a long time. In a certain sense it is a success. The old President of Li-beria, Roberts, has just been with me the first time I ever saw him. He says they have within the bounds of that colony between three and four hundred thousand people, or more than in some of our old States, such as Rhode Island or Delaware, or in some of our newer States, and less then in each of the states o Island or Delaware, or in some of our newer States, and less than in some of our larger ones. They are not all Amer-ican colonists or their descendants. Something less than 12,000 have been sent thither from this country. Many of the original settlers have died, yet, like people elsewhere, their offspring outnumber those deceased. The question is, if the colored people are persuaded to go anywhere, why not there? One reason for unwillingness to do so is, that some of you would rather remain within reach of the coun-try of your nativity. I do not know how much attachment not there? One reason for unwillingness to do so is, that some of your would rather remain within reach of the coun-try of your nativity. I do not know how much attachment you have the greatest reason to love them. But still you are attached to them at all events. The place I am think-ing about having for a colony is in Central America. It is nearer to us than Liberia-not much more than one fourth as far as Liberia, and within seven days' run by steamers. Unlike Liberia, it is a great line of travel-it is a highway. The country is a very excellent one for any people, and with great natural resources and advantages, and especially be cause of the similarity of climate with your native soil, thus being suited to your physical condition. The particu-lar place I have in view is to be a great highway from the Atlantic or Caribbean Sea to the Pacific Ocean, and this particular place has all the advantages for a colony. On both sides there are harbors among the fuest in the world. Again, there is evidence of very rich coal mines. A certain amount of coal is valuable in any country. Why I attach so the inhabitants for immediate employment till they get ready to settle permanently in their homes. If you take colonists where there is nothing to cultivate, and of anow; and so where there is nothing to cultivate, and of

which to make a farm. But if something is started so that for a great advantage. Coal land is the best thing I know of with which to commence an enterprise. To roturn—you have been talked to upon this subject, and told that a spec-nlation is intended by gentlemen who have an interest in the country, including the coal mines. We have been mis-taken all our lives if we do not know whites, as well as blacks, look to their self-interest. Unless among those de-ficient of intellect, everybody you trade with makes some thing. You meet with these things here and everywhere. If such persons have what will be an advantage to them the question is, whether it cannot be made of advantago to It such persons have what will be an advantage to their the question is, whether it cannot be made of advantage to you? You are intelligent and know that success does not as much depend on external help as on self-reliance. Much, therefore, depends upon yourselves. As to the coal mines, I think I see the means available for your self-reliance. I hell if the set a configurate of who are more do have a more than the set of the I think I see the fields available for your sententance. If you will be aball, if I get a sufficient number of you engaged, have pro-vision made that you shall not be wronged. If you will engage in the enterprise, I will spend some of the money intrusted to me. I am not sure you will succeed. The Gov-ernment may lose the money, but we cannot succeed unless to avail the ability of a point. we try; but we think with care we can succeed. The polit-ical affairs in Central America are not in quite as satisfac-tory condition as I wish. There are contending factions in tory condition as I wish. There are contending factions in that quarter; but its true, all the factions are agreed alike on the subject of colonization, and want it, and are more generous than we are here. To your colored race they have no objection. Besides, I would endeavor to have you made equals, and have the best assurance that you should be the equals of the best. The practical thing I want to ascertain is, whether I can get a number of able-bodied men, with their wives and children, who are willing to go, when I pre-sent evidence of encouragement and protection. Could I get a hundred tolerably intelligent men, with their wives and children, and able to "cut their own fodder." so to speak? Can I have fitly? If I could find twenty-five able-bodied men, with a mixture of women and children—good things in the family relation, I think—I could make a suc-cessful commencement. I want you to let me know whether this can be done or not. This is the practical part of my wish to see you. These are subjects of very great import-ance—worthy of a month's study, of a speech delivered in an hour. I ask you, then, to consider seriously, not per-taining to yourselves merely, nor for your race and ours for the present time, but as one of the things, if successfully managed, for the good of mankind—not confined to the present time, but as that quarter; but it is true, all the factions are agreed alike present generation, but as

"From age to age descends the lay To millions yet to be, Till far its echoes roll away Into eternity."

The above is merely given as the substance of the President's remarks.

Gent's remarks. The chairman of the delegation briefly replied, that "they would hold a consultation, and in a short time give an answer." The President said, "Take your full time—no hurry at all." The delegation then withdrew.

It was proposed to settle these persons on a tract of country in New Grenada, but that Government objected, and no further attempt has been made in that direction. (For further particulars see page 212.)

# Incompatibility of Civil and Military Office.

ROBERT C. SCHENCK and FRANCIS P. BLAIR, Jr., were elected in the fall of 1862 members of the Thirty-Eighth Congress, the first named then being a Major General. The latter was commissioned a Brigadier, and then a Major General, subsequently to the election. The former resigned his commission November 13, 1863, to take his seat in Congress, the resignation to have effect December 5, 1863; the Pres-The latter reident accepted it November 21. signed January 1, 1864, which the President accepted January 12, giving it effect that day. He took the oath of office as Representative in Congress, January 12. April 23, he requested to withdraw his resignation as Major General of Volunteers; and, same day, the President bill being under consideration in the Senate,

qualifies and enters on the discharge of the duties of a member, is thereby, in either case, out of the office previ-ously held, and cannot be restored to it without a new ap-pointment, in the manner prescribed by the Constitution.

June 30-This resolution passed without a division.

## IN HOUSE.

June 13-The Committee on Elections made a report, and submitted these resolutions:

Resoluted, That ROBERT C. SCHENCK, having resigned the office of Major General of Volunteers which he then held on the 13th day of November, 1863, which resignation was accepted November 21, 1863, to take effect December 5, 1863, was not, by reason of having held such office, disquali-fied from holding a seat as a Representative in the Thirty-Eighth Congress, whose first ession commenced on the 7th day of December, 1863. Resolved, That FRANCIS P. BLAIR, Jr., by continuing to hold the office of Major General of Volunteers to which he

was appointed November 29, 1867, volunteers to which he was appointed November 29, 1862, and to discharge the dn-ties thereof till January 1, 1864, the date of his resignation, did thereby decline and disqualify himself to hold the office of Representative in the Thirty-Eight Congress, the first session of which commenced on the first Monday in December, 1863.

June 29-They were adopted without a division.

# Repeal of the Fishing Bounties.

First Session, Thirty-Sixth Congress.

Pending the consideration of a tax bill in the Senate,

July 29, 1861, Mr. SAULSBURY moved to add this section :

That from and after the 6th day of October, 1861, all acts and parts of acts granting allowances or bounties on the tonnage of vessels employed in the bank or other cod fisheries, be, and the same are hereby, repealed.

Which was rejected - yeas 15, nays 19, as follows:

YEAS-Messrs. Browning, Carlile, Chandler, Doolittle, Grimes, Harlan, Johnson of Missouri, Polk, Powell, Rice, Saulsbury, Sherman, Trumbull, Wilkinson, Willey-15. NATE-Messrs. Anthony, Clark, Collamer, Dixon, Fessen-den, Foot, Foster, Harris, King, Lane of Indiana, Lane of Kansas, McDougall, Morrill, Pomeroy, Simmons, Summer, Ten Eyck, Wade, Wilson-19.

# Third Session, Thirty-Seventh Congress.

# IN SENATE.

1863, Feb. 2-Pending the legislative bill, Mr. POWELL offered this new section :

That all laws or parts of laws allowing or giving bounties on the tonnage of vessels engaged in the cod or other bank fisheries, be, and the same are hereby, repealed.

Which was rejected-yeas 8, nays 35, as follows:

YEAS -- Messrs. Carlile, Kennedy, Powell, Richardson, Sherman, Turpie, Wall, Wilson of Missouri-S. NAYS--Messrs. Anthony, Arnold, Chandler, Clark, Colla-mer, Cowan, Davis, Dixon, Dooltitle, Fessenden, Foot, Foster, Grimes, Hale, Harding, Harlan, Harris, Hicks, Howard, Howe, King, Lane of Indiana, Lane of Kansas, Latham, McDougall, Morrill, Pomeroy, Rice, Sumner, Trumbull, Wade, Wilkinson, Willey, Wilmot, Wilson of Mass-35.

# First Session, Thirty-Eighth Congress.

1864, April 12-The Naval appropriation

YEAS-Messrs. Buckalew, Cowan, Davis, Harding, Har-lan, Henderson, Hendricks, Lane of Indiana, Lane of Kan-sas, McDougall, Nesmith, Pomeroy, Powell, Saulsbury, Trumbull, Wilkinson, Willey, Wright-18. NAYS-Messrs. Anthony, Chandler, Conness, Doolittle, Fessenden, Foot, Foster, Grimes, Hale, Harris, Howe, John-son, Morgan, Ramsey, Sherman, Sprague, Sumner, Ten Wark, Wedo, Wilson-20.

son, Morgan, Ramsey, S Eyck, Wade, Wilson-20.

May 31-Pending the Internal Revenue bill, Mr. Powell offered the same section, which was rejected-yeas 11, nays 24, as follows:

YEAS-Messrs. Buckalew, Conness, Davis, Grimes, Hen-dricks, Nesmith, Powell, Richardson, Saulsbury, Sherman, Trumbull--11.

NAYS-Messrs. Anthony, Chandler, Clark, Dixon, Doo-little, Fessenden, Foot, Foster, Hale, Howard, Howe, John-son, Lane of Kansas, Morgan, Morrill, Pomeroy, Ramsey, Summer, Ten Eyck, Van Winkle, Wade, Wilkinson, Willey, Wilson-24.

# To prohibit Polygamy in Utah. Second Session, Thirty-Seventh Congress.

## IN HOUSE.

1862, April 28--The House passed, without a division, a bill to punish and prevent the practice of polygamy in the Territories of the United States and other places, and disapproving and annulling certain acts of the territorial legislature of Utah. (It is the identical bill passed at the first session of the Thirty-Sixth Congress, with the difference that this bill strikes out the exception of the District of Columbia from its

provisions, which was contained in the other.) June 3—The bill, amended, passed the Sen-ate—yeas 37, nays 2, (Messrs. Latham and Mc-Dougall.)

# **Declaring Certain Persons Ineligible** to Office.

# Second Session, Thirty-Seventh Congress. IN HOUSE.

1861, December 23-Mr. MOORHEAD offered this resolution, which was adopted :

Resolved, That the Judiciary Committee be instructed to Insured, that the subcary committee be instructed to imquire into the expediency of reporting a bill providing that any person or persons engaged or implicated in the present rebellion against the Constitution of the United States be forever hereafter rendered ineligible to hold any office under the Constitution and laws of the United States.

1862, March 13-Mr. Wilson gave notice of a bill declaring certain persons ineligible to office.

June 4-The House passed a bill declaring certain persons ineligible to office.

certain persons ineligible to office. It provides that any person elected or appointed to any office of honor or profit under the Government of the United States, either in the civil, military, or naval department, shall, before entering on the duties of such office, and be-fore being entitled to any of the salary or other emoluments thereof, take and subscribe an oath that he had never vol-untarily borne arms against the Government of the United States since he had been a citizen thereof; had voluntarily given no aid, countenance, counsel, or encouragement to persons engaged in armed hostility thereto; had never sought or accepted or attempted to exercise the functions of any office whatever under any authority or pretended au-thority in hostility to the Government of the United States;

Mr. POWELL offered the following as a new section: That from and after the first day of July, 1864, all acts and parts of acts granting allowances or bounties on the tonnage of vessels engaged in the Bank or other cod fish-aries be, and the same are hereby, repealed. Which was rejected—yeas 18, nays 20, as follows: Users Rueleden Comp. Davis Harding Hardi which he is about to enter. This oath is to be preserved among the files of the court, House of Congress, or depart-ment to which such office may appertain; and any per-son falsely taking such eath shall be guilty of perjury, and on conviction thereof, shall, in addition to the penalties now prescribed, be deprived of his office, and rendered incapa-ble forever thereafter of holding any office under the Gov-errment of the United States. ernment of the United States.

## Yeas 78, nays 47. The NAYS were:

Messrs. William J. Allen, Ancona, Jacob B. Blair, George H. Browne, Calverl, Casey, Cobb, Corning, Cox, Dunlap, Eng-lish, Grider, Haight, Harding, Holman, Johnson, Kerrigan, Knapp, Law, May, Menzies, Noble, Noell, Norton, Nugen, Pendleton, John S. Phelps, Richardson, Robinson, James S. Rollins, Segar, Shiel, Smith, John B. Steele, William G. Steele, Stiles, Benjamin F. Thomas, Francis Thomas, Fallan-digham, Vibbard, Voorhees, Wadsworth, Ward, Chilton A. White, Wickliffe, Woodruff, Wright-41.

June 23-The bill was amended and passed in the Senate-yeas 23, nays 5, (Messrs. Bayard, Carlile, Kennedy, Powell, Saulsbury.)

June 24-The House non-concurred in the amendments of the Senate. A Committee of Conference arranged the differences, and their report, being the existing law, was adopted in the House without a division, and in the Sen-ate-yeas 27, nays 8, (Messrs. Bayard, Carlile, Davis, Henderson, Nesmith, Powell, Saulsbury, Stark.)

# Bill to Punish Conspiracies.

The bill to define and punish conspiracies provides that if two or more persons within any State or Territory of the United States shall conspire together to overthrow, or to put down, or to destroy by force, the Government of the United States, or to levy war against the United States, or binded blacks, to the authority of the Government of the United States; or by force to prevent, hinder, or delay the execution of any law of the United States; or by force to seize, take, or possess any property of the United States against the will, or contrary to the authority of the United States; or by force, or intimidation, or threat, to prevent any person from accepting or holding any office, or trust, or place of confidence, under the United States; each and every person so offending shall be guilty of a high crime, and upon conviction in any district or circuit court of the United States having jurisdiction, or district or supremo court of any Territory of the United States having jurisdic-tion, shall be punished by a fine not less than \$500 and not more than \$5,000, or by imprisonment, solitary or social, and with or without hard labor, as the court shall determto oppose by force the authority of the Government of the and with or without hard labor, as the court shall determ-ine, for a period not less than six months nor greater than six years; or by both such fine and imprisonment.

1861, July 15-It passed the House-yeas 123, nays 7, (Messrs. Ashley, Burnett, Diven, Edgerton, Goodwin, Pomeroy, Wood.)

July 26-The bill passed the Senate without a division.

Same day-Mr. Powell presented this protest against its passage:

Protest of the minority of the Senate of the United States against the passage of the House bill No. 45, entitled "An act to define and punish certain conspirators." The undersigned, members of the Senate, dissent from the passage of the bill on the following grounds: The government of the United States is a Government of

The government of the United States is a Government of specially delegated powers; and though treason is one of the highest crimes known to the law, it is a political offence. To guard against the abuses which in times of high ex-citement had, in the history of England previous to the rev-olution of 1688, too often sacrificed able, virtuous, and innocent men on charges of treason and kindred offences processing by acts the Constitution of the United unaccompanied by acts, the Constitution of the United

States expressly defines the crime of treason in the follow-

States expressive denies the state of the United States shall are terms: Arr. 3, Stc. 3, "Treason against the United States shall consist only in levying war against them, or in adhering to their enemies, giving them aid and comfort." It further provides that "no person shall be convicted of the termine of the terminery of two witnesses to the

It further provides that "no person shall be convicted of treason unless on the testimony of two witnesses to the same overt act, or on confession in open court." The intent to restrict Congress in the creation of crimes of the nature created by this bill seems obvious; for in trea-son all are principals, and in any conspiracy of the kind stated in the bill, an overt act in pursuance of it, proved by two witnesses, would be treason against the United States. Thus the creation of an offence, resting in intention alone, without overt act round pender nucefory the nervoision last

without overt act, would render nugatory the provision last quoted, and the door would be opened for those similar opquoted, and the door would be opened for those similar op-pressions and crueltics which, under the excitement of po-litical struggles, have so often disgraced the past history of the world. The undersigned can conceive no possible object in defining the crime of treason by our ancestors, and re-quiring proof by two witnesses to the same overt act to instity the conviction of the accused, unless it he to restrict the power of Congress in the creation of a political crime kindred to treason, and charged as resting in intent alone, which would, if accompanied by an overt act, be treason. It matters not that the punishment prescribed in the law is not death, but imprisonment; for the passage of the bill.

is not death, but imprisonment; for the passage of the bill, though it might not affect the life of an innocent man, would give, from the uncertainty of the offence charged, and the proof requisite to sustain it, the utmost latitude to prosecutions founded on personal enmity and political ani-mosity and the suspicions as to intention which they inevitably engender.

JAMES A. BAYARD, L. W. POWELL, J. D. BRIGHT, W. SAULSBURY, TRUSTEN POLK. J. A. PEARCE, A. KENNEDY JOHN C. BRECKINRIDGE, WALDO P. JOHNSON.

## Letters of Marque.

## Second Session, Thirty-Seventh Congress.

Be il enacted, dc., That in all domestic and foreign wars the President of the United States is authorized to issue to private armed vessels of the United States commissions, or letters of marque and general reprisal, in such form as he shall think proper, and under the seal of the United States, and make all needful rules and regulations for the govern-tand the government and conduct thereof, and for the adjudication and disposal of the prizes and salvages made by such vessels: *Provided*, That the authority conferred by this act shall cease and terminate at the end of three years from the passage of this act.

## IN SENATE.

Fébruary 17, 1863-Mr. SUMNER offered the following amendment :

That, to aid in putting down the present rebellion, the President of the United States is authorized to issue to private armed vessels of the United States, &c.

Which was rejected-yeas 13, nays 22.

The bill passed the Senate-yeas 27, nays 9, as follows:

YEAS—Messrs. Anthony, Arnold, Chandler, Clark, Colla-mer, Cowan, Doolittle, Fessenden, Foet, Foster, Grimes, *Harlang*, Harlan, Harris, Hicks, Howe, King, Lane of Kan-sas, Latham, McDougall, Morrill, Nesmith, Rice, Sherman, Turpie, Wade, Wilson of Massachusetts—27. NATS—Messrs. Davis, Dixon, Henderson, Heward, Lane of Indiana, Pomercy, Sumner, Trumbull, Wilson of Mis-souri-09.

souri-9.

March 2-The bill passed the House without division.

# Enabling Act for Nebraska.

1864, March 17-The House considered the bill to enable the people of Nebraska to form a constitution and State government, when Mr. Cox moved to add a proviso :

That the said Territery shall not be admitted as a State until Congress shall be satisfied, by a census taken under anthority of law, that its population shall be equal to that required as a ratio for one member of Congress under the apportionment.

Which was disagreed to-yeas 43, nays 72. The YEAS were:

Messre. James C. Allen, Ancona, Augustus C. Baldwin, James S. Brown. Chanler, Cox. Dawson, Denison, Eldridge, James S. Brown, Chanter, Coz. Dawson, Denison, Eldradge, Hall, Harding, Harrington, Herrick, Holman, Kallyfeisch, Kernan, Law, Long, Mallory, Marcy, McAtlister, McDowell, Middleton, James R. Morris, Noble, Odell, John O'Neill, Perddleton, Ferry, Pruyn, Rogers, James S. Rollins, Ross, John B. Steele, William G. Steele, Stiles, Strouse, Sweat, Voorhees, Wadsworth, Webster, Wheeler, Yeaman-43.

April 14-The Senate passed the bill without a division.

# State of West Virginia.

# Second Session, Thirty-Seventh Congress. IN SENATE.

1862, July 14-The bill providing for the admission of the State of West Virginia into the Union, passed-yeas 23, nays 17, as follows:

YEAS-Messrs. Anthony, Clark, Collamer, Fessenden, Foot, Foster, Grimes, Hale, Harlan, Harris, Howe, Lane of Indiana, Lane of Kansas, Morrill, Pomeroy, *Rice*, Sherman, Simmons, Ten Eyck, Wade, Wilkinson, Willey, Wilson of Messnehr 2014 Massachusetts-23.

NATS -- Messrs. Bayard, Browning, Carlile, Chandler, Cowan, Davis, Howard, Kennedy, Kiog, McDougall, Pow-ell, Saulsbury, Stark, Sumner, Trumbull, Wilson of Missouri, Wright-17.

During the pendency of this bill, July 14, 1862, Mr. SUMNER moved to strike from the first section of the second article the words: "the children of all slaves born within the limits of said State shall be free," and insert:

Within the limits of the said State there shall be neither slavery nor involuntary servitude, otherwise than in pun-ishment of crimes whereof the party shall be duly convicted.

Which was rejected-yeas 11, nays 24, as follows:

YEAS-Messrs. Chandler, Clark, Grimes, King, Lane of Kansas, Pomeroy, Sumner, Trumbull, Wilkinson, Wilmot, Wilson of Massachusetts-11.

WISON OF MASSACHUSETIS-11. NAYS-Messrs. Anthony, *Bayard*, Browning, *Carlile*, Col-lamer, Doolittle, Foot, Foster, Harris, Henderson, Howe, *Kennedy*, Lane of Indiana, *Powell*, *Rice*, *Saulsbury*, Sher-man, Simmons, *Stark*, Ten Eyck, Wade, Willey, *Wilson* of Missouri, *Wright*-24.

Mr. WILLEY proposed to strike out all after the word "That" in the first section, and insert:

That the State of West Virginia be, and is hereby, de-clared to be one of the United States of America, and adthat to be one of the ormed states of America, and and States in all respects whatever, and until the next general census shall be entitled to three members in the House of Representatives of the United States: *Provided always*, That this act shall not take effect until after the preclama tion of the President of the United States hereinafter provided for.

SEC. 2. It being represented to Congress that since the convention of the 26th of November, 1861, that framed and proposed the constitution for the said State of West Vir-ginia, the people thereof have expressed a wish to change the seventh section of the eleventh article of said constituthe seventh section of the eleventh article of said constitu-tion by striking out the same, and inserting the following in its place, namely, "The children of slaves born within the limits of this State after the 4th day of July, 1863, shall be free and no slave shall be permitted to come into the State for permanent residence therein:" Therefore, *Be it further enacted*, That whenever the people of West Virginia shall, through their said convention, and by a vote to be taken at an election to be held within the limits of the State at such time as the convention may provide, makes and ratify the change agence in an encourted set if the

and ratify the change aforesaid and properly certify the same under the hand of the president of the convention, it, shall be lawful for the President of the United States to issue his proclamation stating the fact, and thereupon this act shall take effect and be in force from and after sixty deer form the detect of acid prediverging days from the date of said proclamation.

Mr. LANE, of Kansas, moved to amend the

amendment by inserting after the word "there-| in" and before the word "Therefore" the words

And that all slaves within the said State who shall at the time aforestid be under the age of ten years shall be free when they arrive at the age of twenty-one years; and all slaves over ten and under twenty-one years shall be free when they arrive at the age of twenty-five years.

Which was agreed to-yeas 25, nays 12, as follows:

YEAS—Messrs. Anthony, Clark, Collamer, Doolittle, Foot, Foster, Grimes, Harlan, Harris, Howard, Howe, King, Lane of Indiana, Laue of Kansais, Morrill, Pomeroy, Sherman, Simmons, Summer, Ten Eyck, Trumbull, Wade, Wilkinson, Wilmot, Wilson of Massachusetts—25. Nars—Messrs. Browning, Carlie, Davis, Henderson, Ken-nedy, McDougall, Powell, Saulsbury, Stark, Willey, Wilson of Missoni Ukright.—12.

of Missouri, Wright-12.

The amendment as amended was then agreed to

A motion to postpone the bill to the first Monday of the next December was lost-yeas 17, nays 23.

## IN HOUSE.

July 16-The bill was postponed until the second Tuesday of the next December-yeas 63, nays 33.

## Third Session, Thirty-Seventh Congress.

1863, Dec. 10-The House passed the bill\*yeas 96, nays 57, as follows:

YEAS-Messrs. Aldrich, Arnold, Babbitt, Baker, Baxter, Beaman, Bingham, Jacob B. Blair, Samuel S. Blair, Blake, William G. Brown, Buffinton, Burnham, Campbell, Casey, Chamberlain, Clark, Clements, Colfax, Frederick A. Conk-ling, Covode, Cutler, Davis, Duell, Dunn, Edgerton, Ed-wards, Eliot, Ely, Fenton, Samuel C. Fessenden, Thomas A. Ing, Covode, Curller, Davis, Duch, Junn, Edgerton, Edwards, Eliot, Ely, Fenton, Samuel C. Fessenden, Thomas A. \*D. Fessenden, Thranchot, Frank, Goodwin, Gurley, Haight, Hale, Harrison, Hickman, Hooper, Horton, Hutchins, Julian, Kelley, Francis W. Kellogg, William Kellogg, Killinger, Lansing, Lehman, Loomis, Lovejoy, Low, McKnight, McPherson, Maynard, Mitchell, Moorhead, Anson P. Morrill, Justin S. Morrill, Nixon, Neell, Olin, Patton, Timothy G. Pholps, Pikg, Pomeroy, Porter, Potter, John H. Rice, Riddle, Edward II. Rollins, Sargent, Sedgwick, Shanks, Sheffield, Shellabarger, Sherman, Stoan, Spaulding, Stevens, Stratton, Trimble, Trowbridge, Van Horn, Van Valkenburgh, Van Wyck, Verree, Walker, Wall, Washburne, Whaly, Maxs—Messrs. William J. Allen, Alley, Ancona, Ashley, Baily, Biddle, Cobk, Roscoe Conkling, Convas, Coar, Cravens, Crisfield, Crittenden, Delano, Delaplaine, Diven, Dunlap, English, Gooch, Granger, Grider, Hall, Hurding, Holman, Johnson, Kerrigan, Knapp, Law, Mallory, Marzis, Morie, Nobe, Norton, Odell, Pendleton, Price, Alexander H. Rice, Richardson, Kerrin, Tam, Kalley, Stolins, Songer, Sheil, Sonth, John E. Steele, William G. Steele, Stiles, Benjamin F. Thomas, Francis Thomas, Trin, Yallandigham, Forhes, Wadworth, Ward, Chillon A. Whule, Wickliffe, Wright, Yeaman—57.

Ward, Chillon A. White, Wickliffe, Wright, Yeaman-57

1863, April 20-The President issued a proclamation announcing the compliance, by West Virginia, of the conditions of admission.

# Colored Men as Citizens.

# OPINION OF ATTORNEY GENERAL BATES.

ATTORNEY GENERAL'S OFFICE November 29, 1862.

Hon. S. P. CHASE, Secretary of the Treasury :

SIR: Some time ago I had the honor to receive your let-ler submitting, for my opinion, the question whether or not colored men can be citizens of the United States. The urgency of other unavoidable engagements, and the great im-

\* It includes these counties: Hancock, Brooke, Ohio, Marshall, Wetzel, Marion, Monongalia, Preston, Taylor, Ty-Marshall, Wetzei, Marton, Monongaha, Freston, Taylor, Ty-ler, Pleasants, Ritchie, Doddridge, Harrison, Wood, Jack-son, Wirt, Roane, Calhoun, Gilmer, Barbour, Tucker, Lewis, Braxton, Upshur, Randolph, Mason, Putnam, Kanawba, Clay, Nicholas, Cabell, Wayne, Boone, Logan, Wyoming, Mercer, McDowell, Webster, Pocahontas, Fayette, Raleigh, Greenbrier, Monroe, Pendleton, Hardy, Hampshire, and Morgan.

portance of the question itself, have caused me to delay the answer until now.

x our letter states that "the schoner Elizabeth and Mar-garet, of New Brunswick, is detained by the revenue cutter Tiger, at South Amboy, New Jersey, because commanded by a 'colored man,' and so by a person not a citizen of the United States. As colored masters are numerous in our coasting trade, I submit, for your opinion, the question sug-gested by Captain Martin, of the Tiger: Are colored men cilizens of the United States, and therefore competent to command American vessels?" The question would have been more clearly stated is : Your letter states that " the schooner Elizabeth and Mar-

The question would have been more clearly stated if, instad of saying are colored men citizens, it had been said, can colored men be citizens of the United States; for within our borders and upon our ships, both of war and commerce. there may be colored men, and while men, also, who are not citizens of the United States. In treating the subject, I shall endeavor to answer your question as if i imported only this: Is a man legally incapacitated to be a citizen of the United States by the sole fact that he is a colored, and not a white man?

Who is a citizen? What constitutes a citizen of the Untted States? I have been often pained by the fruitless scarch in our law books and the records of our courts, for a clear and satisfactory definition of the phrase cilizen of the Uni-ted States. I find no such definition, no anthoritative es-tablishment of the meaning of the phrase, neither by a course of judicial decisions in our courts, nor by the con-tinued and consentaneous action of the different branches of any neither by a set of the set of the set of the set of the set. of our political government. For anght I see to the con-trary, the subject is now as little understood in its details and elements, and the question as open to argument and to speculative criticism, as it was at the beginning of the government. Eighty years of practical enjoyment of citizen-ship, under the Constitution, have not sufficed to teach us either the exact meaning of the word, or the constituent

elements of the thing we prize so highly. In most instances, within my knowledge, in which the matter of citizenship has been discussed, the argument has matter of citizenship has been discussed, the argument has not turned upon the existence and the intrinsic qualities of citizenship itself, but upon the claim of some right or pri-ilege as belonging to and inhering in the character of citi-zen. In this way we are easily led into errors both of fact and principle. We see individuals, who are known to be citizens, in the actual enjoyment of certain rights and privileges, and in the actual exercise of certain powers, so-cial and political, and we, inconsiderately, and without any cial and political, and we, inconsiderately, and without any regard to legal and logical consequences, attribute to those individuals, and to all of their class, the enjoyment of those rights and privileges and the exercise of those powers as incidents to their citizenship, and belonging to them only in their quality of citizens.

In such cases it often happens that the rights enjoyed and the powers exercised have no relation whatever to the quality of citizen, and might be as perfectly enjoyed and exercised by known aliens. For instance, General Bernard, a distinguished soldier and devoted citizen of France, for a a distinguished solider and devoted childen of France, for a long time filled the office of general of engineers in the service of the United States, all the time arowing his French allegiance, and, in fact, closing his relations with the United States by resigning his commission and returning to the service of his own native country. This, and all such in-stances, (and they are many,) go to prove that in this coun-try the legal capacity to hold office is not confined to citi-cars, and therefore that the fact, of holding aux office for zens, and therefore that the fact of holding any office for which citizenship is not specially prescribed by law as a qualification, is no proof that the incumbent is an American citizen.

can citizen. Again, with regard to the right of suffrage, that is, the right to choose officers of government, there is a very com-mon error to the effect that the right to vote for public officers is one of the constituent elements of American citizenship, the leading faculty indeed of the citizen, the test at once of his legal right, and the sufficient proof of his membership of the body politic. No error can be greated then this and the more injurious to the right understand. than this, and few more injurious to the right understanding of our constitutions and the actual working of our po-Ing of our constitutions and the actual working of our po-litical governments. It is not only not true in law or in fact, in principle or in practice, but the reverse is conspicu-ously true; for I make bold to affirm that, viewing the na-tion as a whole, or viewing the States separately, there is no district in the nation in which a majority of the known and recognized citizens are not excluded by law from the right of suffrage. Besides those who are excluded specially on account of some nersonal defect such as nauners idiots on account of some personal defect, such as paupers, idiots, on account of some personal defect, such as paupers, idiots, lunatics, and men convicted of infamous crimes, and, in some States, soldiers, all females and all minor males are also excluded. And these, in every community, make the majority; and yet, I think, no one will venture to deny that women and children, and lunatics, and even convict felons, may be citizens of the United States. Our code (unlike the codes of France, and perhaps some other nations) makes no provision for loss or legal depriva-tion of dividence and the states.

tion of citizenship. Once a citizen (whether natus or datus,

as Sir Edward Coke expresses it.) always a citizen, unless changed by the volition and act of the individual. Neither infancy nor madness nor crime can take away from the subject the quality of citizen. And our laws do, in express terms, declare women and children to he citizens. See, for ope instance, the act of Congress of February 10, 1855, 10

The Constitution of the United States does not declare who are and who are not citizens, nor does it attempt to describe the constituent elements of citizenship. It leaves that quality where it found it, resting upon the fact of home birth, and upon the laws of the several states. Even nome-pirit, and upon the lawsof the several states. Even in the important matter of electing members of Congress, it does no more than provide that "the House of Repre-sentatives shall be composed of members chosen every sec-ond year by the people of the several States; and the electors in the several States shall have the qualifications requisite for the decorage of the most numerup branch of the State for the electors of the most numerous branch of the State legislature." Here the word *citizen* is not mentioned, and is a legal fact, known of course to all awyers and pub-licists, that the constitutions of several of the States, in specifying the qualifications of electors, do altogether omit and exclude the word citizen and citizenship. I will refer, in proof, to but three instances:

1. The constitution of Massachusetts, adopted in 1779-'80. in article 4 of section 3, chapter 1, provides as follows: "Every male person (being twenty-one years of sge, and resident of a particular town in this Common wealth for the

resident of a particular town in this Commonwealth for the space of one year next preceding) having a freehold estate within the same town of the annual income of three pounds, or any estate of the value of sixty pounds, shall have the right to vote in the choice of representatives or representatives for said town." 2. The constitution of North Carolina, adopted in 1776, after a bill of rights, and after reciting that "whereas alle-giance and protection are, in their nature, reciprocal, and the one should of right be refused when the other is with-drawn," declares, in section eight, "that all freemen at the ge of twenty-one years, who have been whadbitatiof any one county within the State twelve months immediately preceding the day of any election, and shall have paid pubpreceding the day of any election, and shall have paid pub-lic taxes, shall be entitled to vote for members of the Honse

of Commons for the county in which he resides." 3. The constitution of Illinois, adopted in 1818, in article two, section twenty-seven, declares that "in all elections all white male inhabitants above the age of twenty-one years, having resided in the State six months next preceding the election, shall enjoy the right of an elector; but no person shall be entitled to vote except in the county or district in which he shall actually reside at the time of the election."

These three constitutions belong to States widely separ-eted in geographical position, varying greatly from each other in habits, manners, and pursuits, having different climates, soils, productions, and domestic institutions; and yet not one of the three has made *citizensity* a necessary qualification for a voter; all three of them exclude all fo-males, but only one of them (Illinois) has excluded the black man from the right of suffrage. And it is historically true that the practice has conformed to the theory of those constitutions, respectively; for, without regard to citizen-ship, the colored man has not voted in Illinois, and freemen of all colors have voted in North Carolina and Massachu-These three constitutions belong to States widely scparof all colors have voted in North Carolina and Massachusetts.

From all this it is manifest that American citizenship does not necessarily depend upon nor coexist with the legal capacity to hold office and the right of suffrage, either or both of them. The Constitution of the United States, as I have said, does not define citizenship; neither does it de clare who may vote, nor who may hold office, except in regard to a few of the highest national functionaries. And regard to a ter of the nighest national functionaries. And the several States, as far as I know, in exercising that power, act independently and without any controlling an-thority over them, and hence it follows that there is no limit to their power in that particular but their own pru-dame and dispertion and the set of the se dence and discretion; and therefore we are not surprised to find that these factions, and therefore we are not surprised to find that these factilies of voting and holding office are not uniform in the different States, but are made to depend upon a variety of facts, purely discretionary, such as age, eard length of residence there.

On this point, then, I conclude that no person in the United States did ever exercise the right of suffrage in virtue of the naked, unassisted fact of citizenship. In every instance the right depends upon some additional fact and cumulative qualification, which may as perfectly exist without equip differentiation.

my predecessors,) without telling what particular rights my predecessors,) without terming what particular lights they are nor what relation they have, if any, with citizen-ship. And they suggest, without affirming, that there may be different grades of citizenship of higher and lower degree in point of legal virtue and efficacy; one grade "in the sense of the Constitution," and another inferior grade made but of state and not recommind but the Constitution by a State and not recognized by the Constitution.

In my opinion the Constitution uses the word citizen only to express the political quality of the individual in his re-lations to the nation; to declare that he is a member of the body politic, and bound to it by the reciprocal obligation of allegiance on the one side and protection on the other And I have no knowledge of any other kind of political citizenship, higher or lower, statal or national; or of any other sense in which the word has been used in the Constiother sense in which the word has been used in the Consti-tution, or can be used properly in the laws of the United States. The phrase "a citizen of the United States," with-out addition or qualification, means neither more nor less than a member of the nation. And all such are, politically and legally, equal—the child in the cradle and its father in the Senate are equally citizens of the United States. And it needs no argument to prove that every citizen of a State is, necessarily, a citizen of the United States; and to me it is equally clear that every citizen of the United States is a citizen of the particular State in which he is domiciled. And as to voting and holding office, as that privilege is

not essential to citizenship, so the deprivation of it by law is not a deprivation of citizenship. No more so in the case of a negro than in case of a white woman or child.

In common speech the word citizen, with more or less of truth and pertinency, has a variety of meanings. Some-times it is used in contrast with *soldier*; sometimes with *farmer or countryman*; sometimes with *alien* or *foreigner*. Speaking of a particular man we ask, is he a citizen or a sol-dier? meaning, is he engaged in civil or military pursuits? Is he a citizen or a countryman? meaning, does he live in Is he a children of a country limit i meaning, does he life in the city or in the country? Its he a citizen or an alien? meaning, is he a member of our body-politic or of some other nation. The first two predicates relate only to the pursuits and to the place of abode of the person. The last is always and wholly political, and concerns only the political and governmental relations of the individual. And it

is only in this last sense, the political, that the word is ever used in the Constitution and statutes of the United States. We have *natural-born* citizens, (Constitution, article 2, sec. 5,) not made by law or otherwise, but *born*. And this lass is the uncompositive in fact the more factorial than the sec. class is the large majority; in fact, the mass of our citizens; for all others are exceptions specially provided for by law. As they became citizens in the natural way, by *birth*, so they remain citizens during their natural lives, nuess, by their own voluntary act, they expatriate themselves and their own voluntary act, they explained themselves and become citizens or subjects of another nation. For we have no law (as the French have) to *decilizenize* a citizen, who has become such either by the natural process of birth, or by the legal process of adoption. And in this connection the Constitution says not one word, and furnishes not one hint, in relation to the color of the ancestral race of the "natural-born citizen." Whatever may have been said, in the opinion of judges and lawyers, and in State statutes, about negroes, mulattees, and persons of color, the Consti-tution is wholly silent upon that subject. The Constitution itself does not make the citizens, (it is, in fact, made by them.) It only intends and recognizes such of them as are natural-home-born-and provides for the naturalization of

them.) It only intends and recognizes such of them as the natural—home-born-and provides for the naturalization of such of them as were alien—foreign-born-making the latter, as far as nature will allow, like the former. And I am not aware of any provision in our laws to-war-rant us in presuming the existence in this country of a class of persons intermediate between citizens and aliens. In England there is such a class, clearly defined by law, and called *denizens*. "A denizen (says Sir William Blackstone) is an alien born, but who has obtained, ex donatione regis, letters patent to make him an English subject; a high and incommunicable branch of the royal prerogative. A denizen is in a kind of middle state between an alien and A denizen is in a kind of middle state between an alien and a natural-born subject, and partakes of both of them." (I Sharswood's BL Com., 374.) In this country I know of but one legal anthority tending to show the existence of such a class among us. One of my learned predecessors, Mr. Legaré, (4 Opin., 147.) supposes that there may be such a class, and that free colored persons may be ranked in it. Yet, in that same opinion, he declares that a "free man of color, a *native* of this country, may be admitted to the priv-ileges of a pre-emptioner under the 10th section of the act of the 4th September, 1841." And that act declares that a pre-emptioner must be either a citizen of the United States I an aware that some of our most learned lawyers and ablo writers have allowed themselves to speak upon the context that some of our most learned lawyers and ablo writers have allowed themselves to speak upon the subject in loose and indeterminate language. They speak of "all the rights, privileges, and immunities guarantied by the Constitution to the citizen" without telling us what the constitution to the citizen "without telling us what and imperfect, because he is supposed not to have "ail the given in the jura civitatis, as expressed by one of

Decause natural-born in the country, and not a free white

person, because, though free, confessedly "a man of color." It occurs to me that the discussion of this great subject of national citizenship has been much embarrassed and obscured by the fact that it is beset with artificial difficulties, extrinsic to its nature, and having little or no relation to its great political and national characteristics. And these difficulties, it seems to me, flow mainly from two sources: First, the existence among us of a large class of people whose physical qualities visibly distinguish them from the mass of our people, and mark a different race, and who, for the most part, are held in bondage. This visible difference and servile connection present difficulties hard to be conand service connection present dimensions and to be con-quered; for they unavoidally lead to a more complicated system of government, both legislative and administrative, than would be required if all our people were of one race, and undistinguishable by outward signs. And this, with-out counting the effect upon the opinions, passions, and prejudices of men. Second, the common habit of many of our bost and most hearred upon (the wise antitude of which our best and most learned men (the wise aptitude of which I have not been able to perceive) of testing the political status and governmental relation of our people by standards drawn from the laws and history of ancient Greece and Rome, without, as I think, taking sufficient account of the

The second secon Greece consistent in the circumstance that the ree states which were formed were nothing but citizes with their dis-tricts; and their constitutions were, consequently, only forms of city governments. This point of view (the learned author warns us) must never be lost sight of." And the wise observation of the author applies to Italy as well, for the conjust free iting of Italy must be foreign.

well; for the earliest free cities of Italy were but Grecian colonies, which (bringing along with them the higher civcolonics, which (bringing along with them the higher civ-lization of their parent country, and its better notions of civil polity) by degrees diffused the light of knowledge, and consequently the love of liberty among the then bar-barous people of the Italian peninsula. The Italians, profit-ing by the good example, founded cities of their own upon the Grecian models, and each new Italian city became an independent State. How long this condition of thing con-tinued I know not; but it continued until Rome outgrew all the neithboring communities and subdued them all (the all the neighboring communities, and subdued them all (the Grecian colonies included) under its power. Still the city ruled, and irom time to time granted to such as it would (and withheld from such as it would) the title of *Roman*, and the rights of Roman citizens.

In the process of time, when the dominant power of Rome had expanded over Greece and western Asia, the same civil polity was still continued. As it had been in Italy, so it was in Greece and Asia. In the countries and kingdoms subdued by the Roman arms and transformed into Roman

Subdued by the Roman arms and transformed into Roman provinces, the same system of government still prevailed. Rome, by her pro-consuls and other governors, ruled the conquered nations with absolute sway. And the ruling power at Rome, whether republican or imperial, granted, from time to time, to communities and to individuals in the conquered east the title of *Roman* and therights of Roman citizens.

A striking example of this Roman naturalization, of its controlling authority as a political law, and of its benefi cent power to protect a persecuted citizen may be found in the case of St. Paul, as it is graphically reported in the Acts of the Apostles. Paul, being at Jerusalem, was in great period in this life from his own countrymen, the Jews, who accused him of crimes against their own law and faith, and were about to put him to death by mob violence, when he was rescued by the commander of the Roman troops and taken into a fort for security. He first explained, both and taken into a tor for security. In site explanates, both to the Roman officer and to his own countrymen, who were clamoring against him, his local status and municipal rela-tions, that he was a Jew of Tarsus, a *natural-born citizen* of no mean city, and that he had been brought up in Jerusa-lem in the strictest manner according to the law and faith of the fittee. But this did not marge of the new more drawn of the fathers. But this did not appease the angry crowd of the fattlets. But this that not appears are spin-who were proceeding with great violence to kill him. And then "the chief captain commanded that he be brought into the castle, and bade that he should be examined by scourging," (that is, tortured to enforce confession.) "And as they bound him with thongs, Paul said unto the centu-rion that stood by, Is it lawful for you to scourge a man that is a Roman and uncondenned? When the centurion heard that he went out and told the chief captain saying, Take heed what thou doest, for this mean is a Roman. Then the chief captain came and said, Tell me, art thou a Ro-man? He said yea, and the chief captain said, With a great sum obtained I this freedom. And Paul said, But I was free born. Then straightway they departed from him which should have examined him. And the chief captain also was a freid, after he knew he was a Roman, and be cause he had bound him." who were proceeding with great violence to kill him. And cause he had bound him."

Thus Paul, under circumstances of great danger and ob-

loquy, asserted his immunity, as " a Roman uncondemned," from ignominious constraint and cruel punishment, a con-straint and punishment against which, as a mere provincial subject of Rome, he had no legal protection. And thus the Roman officers instantly, and with fear, obeyed the law of their country and respected the sacred franchise of the Roman citizen.

Paul, as we know by this record, was a natural-born citizen of Tarsus, and as such, no doubt, had the municipal freedom of that city; but that would not have protected him against the thougs and the lash. How he became a Roman we learn from other historical sources. Cæsar granted to the people of Tarsus (for some good service done, probably for taking his side in the war which resulted in the establishment of the empire) the title of *Roman*, and In the escape of the empire the three of the chief of normar, and the freedom of Roman citizens. And, considering the chro-nology of events, this grant must have been older than Paul; and therefore he truly said *I was free born*—a free citizen of Rome, and as such exempt by law from degrading punishment.

And this immunity did not fill the measure of his rights And this inhumity that hot init the measure of his right as a citizen. As a Roman, it was his right to be tried by the supreme authority, at the capital of the empire. And when he claimed that right, and appealed from the jurisdi-tion of the provincial governor to the Emperor at Rome, his appeal was instantly allowed, and he was remitted to Cæsar's judgment."

I have dwelt the longer upon this case of Paul, because is is a *leading case* in Roman jurisprudence in the matter of the "*jus Romanum.*" And in so far as there is any analogy between Roman and American citizenship, it is strictly applicable to us. Its authenticity is unquestionable, and by its lucid statement of facts in minute detail leaves no room to doubt the legal merits of the case. It establishes the great protective rights of the citizen, but, like our own national constitution, it is silent about his powers. It pro-tected Paul against oppression and outrage, but said noth-ing about his right of suffrage or his eligibility to office.

As far as I know, Mr. Secretary, you and I have no better title to the citizenship which we enjoy than the "accident of birth"-the fact that we happened to be born in the United States. And our Constitution, in speaking of natural-born citizens, uses no affirmative language to make them such, but conly recognizes and realized the universal prin-ciple, common to all nations, and as old as political society, that the people born in a country do constitute the nation, and, as individuals, are *natural* members of the body politic.

If this be a true principle, and I do not doubt it, it fol-lows that every person born in the country is, at the mo-ment of birth, prima facie a citizen; and he who would deny it must take upon himself the burden of proving some great disfranchisement strong enough to override the "natural-born" right as recognized by the Constitution in terms the most simple and comprehensive, and without any reference to race or color, or any other accidental circumstance.

That nativity furnishes the rule, both of duty and of right as between the individual and the government, is a right as between the individual and the government, is a historical and political truth so old and so universally ac-cepted that it is needless to prove it by authority. Never-theless, for the satisfaction of these who may have doubts upon the subject, I note a few books, which, I think, cannot fail to remove all such doubts—Kent's Com., vol. 2, part 4, sec. 25; Bl. Com., book 1, ch. 10, p. 365; 7 Co. Rep., Calvin's case; 4 Term Rep., p. 300, Doe vs. Jones; 3 Pet. Rep., p. 246, Shanks vs. Dupont; and see a very learned treatise, at-tributed to Mr. Binney, in 2 Am. Law Reporter, 193. In every civilized country the individual is *born* to duties and rights—the duty of allegiance and the right to proteo-tion; and these are correlative obligations, the one the price

tion; and these are correlative obligations, the one the price of the other, and they constitute the all-sufficient bond of union between the individual and his country; and the coun-try he is born in *is*, *prima facie*, *his* country. In most country he is born in *is, prima facie, his* country. In most coun-tries the old law was broadly laid down that this natural con-nection between the individual and his native country was perpetual; at least, that the tie was indissoluble by the act of the subject alone.—(See Bl. Com. supra; 3 Pet. Rep. supra.)

But that law of the perpetuity of allegiance is now changed, both in Europe and America. In some countries by silent acquiescence, in others by affirmative logislation. In England, while asserting the perpetuity of natural alle giance, the King, for centuries past, has exercised the power to grant letters of denization to foreigners, making them English subjects, and the Parliament has exercised at pleasure the power of naturalization.

ure the power of naturalization. In France the whole subject is regulated by written law, which plainly declares who are citizons (*citoyens Français*) and who are only *the French*, (*Français*,) meaning the whole body of the French people. (See Les Codes Français, *thre premier.*) And the same law distinctly sets forth by what means citizenship and the quality of *French* may be lost and regained; and maintains fully the right of expatriation

in the subject, and the power of naturalization in the nation to which he goes.

In the United States it is too late now to deny the political rights and obligations conferred and imposed by nativity; for our laws do not pretend to create or enact them, but do assume and recognize them as things known to all men, because pre-existent and natural; and therefore things of which the laws must take cognizance. Acting out this guiding thought, our Constitution does no more than grant to Congress (rather than to any other department) the power "to establish auniform rule of naturalization." And our laws made in pursance thereof indue the made citizen with all the rights and obligations of the natural citizen. And so strongly was Congress impressed with the great legal fact that the child takes its political status in the nation where it is born, that it was found necessary to pass a law to prevent the *alienage* of children of our known fellow-citizens who happen to be born in foreign countries. The act of February 10, 1852, 10 Statutes, 604, provides that "persons," (not while persons) "persons heretofore born, or hereafter to be born, out of the limits and jurisdiction of the and considered, and are hereby declared to be, citizens of the United States: *Provided, however*, That the rights of citizenship shall not descend to persons whose fathers mere resided in the United States.

"SEC. 2. And be it further enacted, That any woman who might lawfully be naturalized under the existing laws, married, or who shall be married to a citizen of the United States, shall be deemed and taken to be a citizen."

But for that act, children of our citizens who happen to be born at London, Paris, or Rome, while their parents are there on a private visit of pleasure or business, might be brought to the native home of their parents, only to find that they themselves were aliens in their father's country, incapable of inheriting their father's land, and with no right to demand the protection of their father's Government.

That is the law of birth at the common law of England, clear and unqualified; and now, both in England and America, modified only by statutes, made from time to time, to meet emergencies as they arise.

In most a mergencies as they arise. I have said that, prima facie, every person in this country is born a citzen; and that he who denies it in individnal cases assumes the burden of stating the exception to the general rule, and proving the fact which works the disfranchisement. There are but a few exceptions commonly made and urged as disqualifying facts. I lay no stress upon the small and admitted class of the natural-born composed of the children of foreign ministers and the like; and

1. Slavery, and whether or no it is legally possible for a slave to be a citizen. On that point I make no question, because it is not within the scope of your inquiry, and does not concern the person to whom your inquiry relates.

2. Color.—It is strenuously in south Hquiry Fentles. 2. Color.—It is strenuously in south Hquiry Pentles. Sons of color," though born in the country, are not capable of being citizens of the United States. As far as the Constitution is concerned, this is a naked assumption; for the Constitution contains not one word upon the subject. The exclusion, if it exist, must then rest upon some fundamental fact which, in the reason and nature of things, is so inconsistent with citizenship that the two cannot coexist in the same person. Is mere color such a fact? Let those who assert it prove that it is so. It has never been so understood nor put into practice in the nation from which we derive our language, laws, and institutions, and our very morals and modes of thought; and, as far as I know, there is not a single nation in Christendom which does not regard the new-found idea with increduity, if not disgust. What can there be in the mere color of a man (we are speaking now not of race, but of color only) to disqualify him for bearing true and faithful allegiance to his native country, and for demanding the protection of that country? And these two, allegiance and protection, constitute the sum of the duties and rights of a " natural-born citizen of the United States."

United States." 3. Race.—There are some who, abandoning the untenable objection of color, still contend that no person descended from negroes of the African race can be a citizen of the United States. Here the objection is not color but race only. The individual objected to may be of very long descent from African negroes, and may be as white as leprosy, or as the intermixture for many generations with the Caucasian race can make him; still, if he can be traced back to negroes of the African race, ho caunot, they say, be a citizen of the United States! And why not? The Constitution certainly does not forbid it, but is silent about the race as it is about color.

Onr nationality was created and our political Government exists by written law, and inasmuch as that law does not exclude persons of that descent, and as its terms are manifastly broad enough to include them, it follows inevitably that such persons, born in the country, must be citizens,

unless the fact of African descent be so incompatible with the fact of citizenship that the two cannot exist together. If they can coexist, in nature and reason, then they do coexist in persons of the indicated class, for there is no law to the contrary. I am not able to perceive any antagonism, legal or natural, between the two facts.

legal or natural, between the two facts. But it is said that African negroes are a degraded race, and that all who are tainted with that degradation are forever disqualified for the functions of citizenship. I can hardly comprehend the thought of the absolute incompatibility of degradation and citizenship. I thought that they often went together. But if it be true with regard to races, it seems to me more cogently true with regard to individuals. And if I be right in this, there are many sorrowful examples in the legislation and practice of various States in the Union to show how low the citizen may be degraded by the combined wisdom and justice of his fellow-citizens. In the early legislation of a number of States the most lummiiating punishments were denounced against persons guilty of certain crimes and misdemeanors—the lash, the pillory, the cropping of the ears, and the branding of the face with an indelibe mark of infamy. And yet a lower depth: in several of the States the common punishment of the crime of *vagrancy* was sale into bondage at public auction 1 And yet I have not read that such unfortunates thereby lost their natural-born citizenship, nor that their descendants are doomed to perpendual exclusion and degradation. I am inclined to think that these objections, as to color

I am inclined to think that these objections, as to color and ancestral race, arise entirely from a wrong conception of the nature and qualities of citizenship, and from the loose and unguarded phraseology too often used in the discussion of the subject. I have already given, at some length, my own views of the word and the thing-citizenship. And now I will add only a few observations before drawing your attention to certain authorities upon the subject mostly relied upon by those who support the objections.

In my opinion it is a great error, and the fruitful parent of errors, to suppose that *cilizens* belong exclusively to ropublican forms of government. English subjects are as truly cilizens as we are, and we are as truly subjects as they are. Imperial France (following imperial Rome) in the text of her laws calls her people cilizens.—(*Les Codes Françats*, book 1, tit. 1, ch. 1, and notes.) And we have a treaty with the present Emperor of the French, stipulating for rociprocal rights in favor of the *clizens* of the two countries respectively. (10 Stat., p. 906, art. 7.)

It is an error to suppose that citizenship is ever hereditary. It never "passes by descent." It is as original in the child as it was in the parents. It is always either born with him or given to him directly by law.

In discussing this subject it is a misleading error to fail to mark the natural and characteristic distinction between political *rights* and political *powers*. The former belong to all citizens allike, and cohere in the very name and nature of citizenship. The latter (participation in the powers of government by voting and exercising office) does not belong to all citizens alike, nor to any citizen, merely in virtue of citizenship. This *power* always depends upon extraneous facts and superadded qualifications; which facts and **qual**ifications are common to both citizens and aliens.

In referring to the authorities commonly adduced by those who deny the citizenship of colored people, I do not pretend to cite them all, but a few only of such as I believe to be most usually relied upon. And I will not trouble you with a detailed examination of the reasoning employed in each case, for I have already stated my own views of the principles and laws involved in the question; and where they conflict with the arguments upon which the contrary opinion is founded, I still adhere to my own.

each case, for I have already stated my own views of the principles and laws involved in the question; and where they conflict with the arguments upon which the contrary opinion is founded, I still adhere to my own. The first of these authorities of which I will treat is the opinion of my predecessor, Mr. Wirt, upon a case precisely like the present, except that in that case the "free person of color" was a Virginian, and the objections to his competency were founded mainly, if not entirely, upon Virginia law.—(See [Opinions of Attorneys General, vol. 1, p. 506, date November 7, 1821.) I have examined this opinion for learning and his known and varied excellencies as a man.

In that case the precise question was, "whether free persons of color are, in Virginia, citizens of the United States, within the intent and meaning of the acts regulating foreign and coasting trade, so as to be qualified to command vessels." And thus Mr. Wirt was in a manner invited to consider the question rather in a statal than a national point of view; and hence we ought not to be surprised to find the whole argument for the exclusion based upon local institutions and statal laws.

As a general answer to all such arguments, I have this to say: Every eitizen of the United States is a component member of the nation, with rights and dutics, under the Constitution and laws of the United States, which cannot be destroyed or abridged by the laws of any particular State. The laws of the State, if they conflict with the laws of the suance thereof, and all treaties, &c., shall be the supreme law of the land, and the judges in every State shall be bound thereby, anything in the constitution or laws of any State to the contrary notwithstanding.<sup>3</sup> And from this I assume that every person who is a citizen of the United States, whether by birth or naturalization, holds his great franchise by the laws of the United States, and above the control of any particular State. Citizenship of the United States is an integral thing, incapable of legal existence in fractional parts. Whoever, then, has that franchise is a whole citzen and a citizen of the whole nation, and cannot be (as the argument of my learned predecessor seems to suppose) such citizen in one State and not in another.

I fully concur in the statement that "the description, citizer of the United States, used in the Constitution, has the same meaning that it has in the several acts of Congress passed under the authority of the Constitution." And I freely declare my inability to conceive of any second or subordinate meaning of the phrase as used in all those instruments. It means in them all the simple expression of the political status of the person in connection with the nation—that he is a member of the body politic. And that hathon-that he is a member of the body point. And that is all it means, for it does not specify his rights and duties as a citizen, nor in any way refer to such "rights, privileges, and immunities" as he may happen to have, by State laws or otherwise, over and beyond what legally and natulaws or otherwise, over and beyond what regary and natural rally belong to him in his quality of citizen of the United States. State laws may and do, nay must, vest in individ-nals great privileges, powers, and duties which do not be long to the mass of their fellow-citizens, and in doing so they consult discretion and convenience only. One citizen, whe however to hose index may under prome incomestioned who happens to be a judge, may, under proper circumstances, sentence another to be hanged, and a third, who happens to be a governor, may grant a pardon to the condemned man, who, as a citizen, is the undoubted peer of both the judge and the governor.

As to the objection (not in law, but sentiment only) that If a negro can be a citizen of the United States, he might, possibly, become President, the legal inference is true. There would be such a legal possibility. But those who make that objection are not arguing upon the Constitution as it is, but upon what, in their own minds and feelings, they think it ought to be. Moreover, they seem to forget that all limitations upon eligibility to office are less restrictions upon the rights of aspirants than upon the powers of electors. Even the legislature of the State, however unanimous, have no power to send to the Senate of the United States their wisest and best man unless he be thirty years old. And all the people of the nation, speaking with one united voice, cannot, constitutionally, make any man President who happens to be under thirty-five. This is, obviously, a restriction upon the appointing power—that is, for our popular government, a restriction upon the people themselves. As individuals, we may like it or dislike it, and flatter ourselves into the belief that we could make a wiser and better frame of government than our fathers made. Still it is our Constitution, binding upon us and upon every citizen from the moment of birth or naturalization.

The Constitution, I suppose, says what it means, and does not mean what it does not say. It says nothing about "the high characteristic privileges of a citizen of the State" (of Virginia or any other.) I do not know what they were, but certainly in Virginia, for the first half of the existence of the community the what he what a sufficiency and are of the commonwealth, the right of suffrage was not one of them. For during that period no man ever voted there because he was a free white adult male citizen. He voted on his freehold, in land; and no candidate, in soliciting his election, appealed to the people or the citizens, but to the freeholders only, for they alone could vote. I shall not trouble you with any argument touching the

list of disabilities declared by the laws of Virginia against free negroes and mulattoes, as stated in the opinion, because they are such only as the Legislature, if so minded, might have denounced as well against a portion of its own acknowledged citizens, whose weakness might necessitate submission.

mission. It is said in the opinion that "the allegiance which the free man of color owes to the State of Virginia is no evi-dence of citizenship, for he owes it not in consequence of an oath of allegiance." This proposition surprises me; per-haps I do not understand it. I did verily believe that the oath of allegiance was not the cause but the sequence of citizenship citizen cut as a clean gravement of the pericitizenship, given only as a solemn guarantee for the per-formance of duties already incurred. But if it be true that the oath of allegiance must either create or precede citizenship, then it follows, of necessity, that there can be no nat-ural-born citizen, as the Constitution affirms, because the child must needs be born before it can take the oath.

The opinion, supported by the arguments upon which I have commented, is in these words: "Upon the whole, I am of the opinion that free persons

nation, are of no force. The Constitution is plain beyond | of color in *Firginia* are not citizens of the United States, cavil upon this point. Article 6: "This Constitution, and the laws of the United States which shall be made in pur-tion in the laws of the United States which shall be made in purvessels."

As an authority this opinion is rebutted by the opinion of Attorney General Legaré, above cited.—(4 Op. A. G., 147, date March 15, 1843.) Under an act of Congress which limited the pre-emption of public land to citizens of the United States and aliens who had declared intention to be come citizens, according to the naturalization laws, Mr Le-

garé was of opinion that a free colored man was competent to pre-empt the land. In that same opinion Mr. Legaré makes a just distinction between political and civil rights, which, I believe, is com-mon to most nations. The French code expresses it very plainly, thus: "L'exercice des droits civils est ind pendant de la qualité de *citoyen*, laquelle ne s'acquiert et ne se con-serve que conformément à la loi constitutionelle."

serve que conformément à la loi constitutionelle." The next authority I shall consider is a decision of the Department of State made in Mr. Marcy's time, November 4, 1856, and evidenced by a letter of that date from Mr. Thomas, Assistant Secretary, to Mr. Rice, of New York.\* That decision is entitled to great consideration, because upon such political questions the Secretary of State is of high authority. The case was an application for passports to travel in foreign parts, in favor of certain free blacks of some of the northern State and the time was for work to some of the northern States, and the time was a few months after the passage of the act of August 18, 1856, (the first act directing the issuing of passports to individuals and re-stricting the issue to *citizens* of the United States, though

The practice is much older.) The letter, after stating the case, declares emphatically that "if this be so (*i. e.*, if they be negroes) there can be no doubt that they are not citizens of the United States." If this stood alone there could be no doubt of the opinion of the department at that time. But it does not stand alone, the department at that time. But it does not stand alone, The letter, after eiting several authorities, and among them one from Tennessee, to which I will have occasion to refer by name, concludes with this qualifying paragraph, which have a several authorities of the set of the several seve by hand, concludes what has qualitying paragraph, which leaves some doubt as to what was the real practical opinion of Mr. Secretary Marcy at that time. The letter, assuming that a passport is a certificate of citizenship, proceeds to

say: "Such being the construction of the Constitution in reand to be persons of color, it is conceived that they can-not be regarded, when beyond the jurisdiction of this gov-ernment, as entitled to the full rights of citizens, but the Secretary directs me to say that though the department could not certify that such persons are citizens of the Uni-ted States, yet, if satisfied of the truth of the facts, it would \* This is the letter alluded to : DEPARTMENT OF STATE, Nonember 4, 18

WASHINGTON, November 4, 1856.

SIR: Your letters of the 29th ultimo and 3d instant, requesting passports for eleven colored persons, have been received, and I am directed by the Secretary to inform you that the papers transmitted by you do not warrant the de-partment in complying with your request. The question whether free negroes are citizens is not now presented for the first time, but has repeatedly arisen in the administration of both the national and State governments. In 1821 a controversy arose as to whether free persons of color were citizens of the United States, within the intent and meaning of the acts of Congress regulating foreign and coasting trade, so as to be qualified to command vessels, and Wirt, Attorney General, decided that they were not, and he moreover held that the words "citizens of the United States" were used in the acts of Congress in the same sense as in the Constitu-This view is also fully sustained in a recent opinion tion. of the present Attorney General.

The judicial decisions of the country are to the same effect

In Kent's Commentaries, volume 2, page 277, it is stated that in 1833, Chief Justice Daggett of Connecticut, held that free blacks are not "citizens" within the meaning of the free blacks are not "citizens, within the meaning of the term, as used in the Constitution of the United States; and the Supreme Court of Tennessee, in the case of the State against Clairbone, held the same doctrine.

Such being the construction of the Constitution in regard to free persons of color, it is conceived that they cannot be regarded, when beyond the jurisdiction of this Government, as entitled to the full rights of citizens; but the Secretary directs me to say that, though the department could not certify that such persons are citizens of the United States, yet, if satisfied of the truth of the facts, it would give a certificate that they were born in the United States, are free, and that the Government thereof would regard it to be its duty to protect them if wronged by a foreign govern-ment while within its jurisdiction for a legal and proper purpose.

purpose. I am, sir, respectfully, your obedient servant, J. A. THOMAS, Assistant Secretary. H\*H. RICE, Esq., New York City.

give a certificate that they were born in the United States, are free, and that the government thereof would regard it to be its duty to protect them, if wronged by a foreign govern-ment, while within its jurisdiction for a legal and proper purpose."

It seems to me that the certificate proposed to be given It seems to me that the certificate proposed to be given would be, in substance and fact, a good passport, for the act of Congress prescribes no form for the passport, and re-quires no particular fact to appear upon its face. And I Confidently believe that there is not a government in Europe which, in view of our laws of citizenship, would question the validity of a passport which declares upon its face that the bearer is a free natural-born inhabitant of the United States.

I turn now to the consideration of the Tennessee case, referred to and relied upon in the letter from the State De-partment—the State of Tennessee vs. Ambrose, (1 Meigs' R., 831,) adjudged in 1838. Ambrose, being a free negro emancipated in Kentucky, moved to and settled in Tennessee. He was indicted for that crime against the Tennessee statute, made to prevent the ingress of that sort of people. He demurred to the indictment upon the ground that he was a citizen of Kentucky, and as such had a right under the Con-stitution of the United States (art. 4, soc. 2,) to go to and abide in Tennessee in spite of the State statute. The court in public the indictment was found surfaced the demurse The public prosecutor took the case up to the summer. The public prosecutor took the case up to the supreme court, where the judgment below was reversed, and it was held by the court that Ambrose, under the circumstances, could not be a citizen of Kentucky, and therefore could not claim the protection of the national constitution as against the Tennessee statute. the Tennessee statute.

I must rouble you with a few remarks upon certain pas-sages in the opinion of the court, which constitute the foundation of the judgment, and without which the judg-ment itself, having no legal basis to rest upon, ought not to have any authority as a precedent.

The court, after stating the case and citing from the Con-stitution, (art. 4, sec. 2,) "the citizens of each State shall be stitution, (art. 4, sec. 2.) "the citizens of each State shall be entitled to all the privileges and immunities of citizens in the several States," proceeds: "the citizens here spoken of (says the supreme court of Tennessee) are those who are entitled to 'all the privileges and immunities of citizens," But free negroes, by whatever appellation we call them, were never in any of the States entitled to all the privileges and immunities of citizens, and consequently were not in-tended to be included when this word was used in the Con-stitution stitution.

"In this country," (continues the court,) "under the free government created by the Constitution, whose language we are expounding, the humblest *white* citizen is entitled to all 'the privileges and immunities' which the *most exalted* one enjoys. Hence, in speaking of the rights which a citizen of one State shall enjoy in every other State, as applicable of one State shall enjoy in every other State, as applicable to white men, it is very properly said that he should be en-titled to all the privileges and immunities of citizens in each other State. The meaning of the language is that no privi-lege enjoyed by, or immunity allowed to, the most favored class of citizens in said State shall be withheld from a citi-zen of any other State. How can it be said that he enjoys all the privileges, when he is scarcely allowed a single right in common with the mass of the citizens of the State? "It cannot be; and therefore either the free negro is not a citizen in the sense of the Constitution or, if a citizen he

a citizen, in the sense of the Constitution, or, if a citizen, ht is entitled to 'all the privileges and immunities' of the mose favored class of citizens. But this latter consequence will be contended by no one. It must then follow that they are not citizens."

These are the foundations of the judgment in the case of Ambrose, and not only in that but in almost every similar case which I have had occasion to examine. A good deal case which I have had occasion to examine. A good deal of what I have already said is strictly applicable here, and in trying to show the fallacy of the reasoning of the court in Tennessee, I must take the risk of some needless repetition.

The leading thought, that indeed which seems to have compelled the judgment against Ambrose, is, in my opinion, a naked assumption, not supported by any word of written law, nor maintainable by logical argument. It is assumed that a person to be a citizen at all must have all the rights, privileges, and immunities which the *favored* one enjoys; all of the most *favored* class of citizens. Now, if there be all of the most favored class of citizens. Now, if there be grades and classes of citizens, (which I am not exactly wil-ling to admit,) it would seem that there must be something to distinguish the grades; some difference in the rights, privileges, and immunities of the different classes. And yet the court, while asserting the existence of different classes of citizens, asserts also their equality, by declaring that "the humblest white citizen is entitled to all the 'privileges and immunities' which the most exalled one enjoys." Then what marks the difference of classes? By what line con yea

and unmunities' which the most exaited one enjoys." Incn what marks the difference of classes? By what line can we separate humility from exaltation, as applied to a citizen? In fact, it seems to me that the difficulties which sur-round the subject are artificial, created by the habitual confounding of things different in their nature and origin,

and by the persistent abuse of language. No distinction is drawn between the rights and duties as a member of society, without regard to his citizenship. The first are political merely—the last civil and social only. And the society, without regard to ins citizenship. The first are political mercly—the last civil and social only. And the words rights, privileges, immunities are abusively used, as if they were synonymous. The word rights is generic, common, embracing whatever may be lawfully claimed. *Privileges* are special rights belonging to the individual or class, and not to the mass. *Immunities* are rights of ex-emption only—freedom from what otherwise would be a duty, obligation, or burden. For instance, the constitution of Tennessee (art. 4, sec. 1) declares that "all free men of color shall be exempt from military duty, in time of peace, and also from paying a free poll-tax." This is immunity. But whether there be or be not grades and classes of citizens, higher or lower, more or less favored, is wholly immaterial to this question. For the Constitution speaks of citizens only, without any reference to their rank, grado, or class, or to the number or magnitude of their rights, privileges, and immunities—citizens simply, without an ad-capacities. Therefore, if there be grades and classes of citizens, still the lowest individual of the lowest possible class is a citizen, and as such fills the requirement of the

class is a citizen, and as such fills the requirement of the Constitution.

If we must have grades and classes of citizens, higher and lower, more and less favored, it seems to me im-possible to sustain the proposition of the court that the humblest and most exalted are entitled to equal privileges and immunities. A free, white, natural-born female infant is certainly a citizen, and I suppose it would be but reason-able to place her in the lowest class. And I assume that it would not be deemed unreasonable to call that class the highest, out of which the President must be chosen. high body of the theorem interaction make the constant in the distribution of the second state of the sec

Those who most indulge in the assumption that to constitute a citizen at all the person must have all the privi-leges and immunities which any citizen can enjoy, rarely venture to specify precisely what they mean. Generally, I think, the inference is plain that they mean suffrage timins, the inference is plain that they mean sumfage and eligibility; and, in that connection, I think I have al-ready shown that suffrage and eligibility have no necessary connection with citizenship, and that the one may, and often does, exist without the other. Again, "immunities" are enjoyed to a very large extent in the processing of the start of the start.

by free negroes in all the slaveholding States. They are generally exempted by law from the oncrous duties of jn-rors in the courts, and militia men in the field; and these are immunities eagerly desired by many white men in all the States.

In another part of that opinion, the court declares that the word "freemen," as used in the constitution of Tennes-see, is equivalent to *citizen*; and yet the court denies that the plrase "freemen of color," used in the same constitu-tion, is a proper designation of citizens! I close my remarks upon that case with an extract from the constitution of Tennessee, (which was originally made in 1765, and amended in 1835.) reminding you only that, until 1760. Tennessee was a part of North Carolina, and subject to its constitution and laws, and hence the peculiar phraseology of the automatic

of the extract: "Article 4, section 1. Every free white man, of the age of twenty-one years, being a citizen of the *United States* and a citizen of the county wherein he may offer his yote six months next preceding the day of election, shall be entitled to vote for members of the general assembly and other civil officers for the county or district in which he resides: *Provided*, That no person shall be disqualified for voting, in Provided, like the person shart be disjudning to rooms, in any election, on account of color, who is now, by the laws of this State, a competent witness in a court of justice against a white man. All freemen of color shall be exempt from military duty in time of peace, and also from paying a free poll-tax."

Finally, the celebrated case of Scott vs. Sandford, 19 Howard's Reports, 393, is sometimes cited as a direct authority against the capacity of free persons of color to be citizens of the United States. That is an entire mistake. The case, determine, that question. It was an ordinary suit for freedom, very common in our jurisprudence, and especially provided for in the legislation of most of the slaveholding States, as it is in Missouri, For convenience the form of the action usually is (and is in this case) trespass, alleging The action is used is in this case presponse, altegring an assault and battery and false imprisonment, so as to en-uble the defendant, (the master,) if he choose, to make a direct issue upon the freedom or slavery of the plaintiff, which is the real point and object of the action, by plead-ing, in justification of the alleged trespass, that the plain-tiff is a slave—his own or another man's. Such an action Dred Scott, if entitled to freedom, might have browcht in the State court without any allegation of

have brought in the State court, without any allegation of

citizenship, and without being, in fact, a citizen. But it seems he desired to bring his action in the circuit court of the United States in Missouri; and, to enable him to do that, he had to allege citizenship, because Mr. Sandford, the defendant, was a citizen of New York, and unless the plaintiff were a citizen of Missouri (or some other State) the na-tional court had no jurisdiction of the case.

The plaint four that no jursaic ton of the case. The plaintiff having made his election to suo in the Uni-ted States court, the defendant might, if he would, have pleaded in bar to the merits of the action, but he exercised his election to plead in abatement to the jurisdiction of the court; thus, that the action, if any, "accrued to the said Dred Scott out of the jurisdiction of this court, and exclu-Sively within the jurisdiction of the courts of the State of Missouri, for that, to wit, the said plaintiff, Dred Scott, is not a citizen of the State of Missouri, as alleged in his dec-laration, [not because he was not born there, and born free, but] because he is a negro of African descent; his ancestors were of pure African blood, and were brought into this country and sold as negro slaves, and this the said Sandford country and sold as negro shaves, and this the said saladford is ready to verify. Wherefore he prays judgment whether this court can or will take further cognizance of the action aforesaid." To this plea the plaintiff demurred, and the circuit court sustained the demurrer, thereby declaring that the facts stated in the plea, and confessed by the do-murrer, did not disqualify Sect for being a citizen of Mis-souri, and so that the United States circuit court had jurisdiction of the cause

The circuit court having taken jurisdiction, the defendant had, of course, to *plead over* to the merits of the action. He

had, of course, to plead orer to the merits of the action. If e did so, and issues were joined, and there was an elaborate trial of the facts, which resulted in a verdict and judgment in favor of the defendant. And thereupon the plaintiff brought the case up to the Supreme Court by writ of error. The power of the Supreme Court over the proceedings and judgments of the circuit court is appellate only, and this for the sole purpose of enabling the court above to affirm what has been rightly done, and reverse what has been wrongly done in the court below. It the error of the court below consist in the illegal assumption of power to hear below consist in the illegal assumption of power to hear and determine the merits of a case not within its jurisdic-tion, of course the court above will correct that error, by tion, of course the coart above will correct that error, by setting aside whatever may have been done by that usurped authority. And in doing this the court above has no more power than the court below had to hear and determine the merits of the case. And to assume the power to determine a case not within the jurisdiction is as great an error in the court above as in the court below; for it is equally true, in all courts, that the jurisdiction must first be ascertained before proceeding to judgment.

before proceeding to judgment. In this particular case the Supreme Court did first exam-ine and consider the plea in abatement, and did adjudge that it was a good plea, sufficient to oust the jurisdiction of the circuit court. And hence it follows, as a necessary legal consequence, that whatever was done in the circuit court after the plea in abatement, and touching the merits of the case, was simply void, because done corram rom judice. Pleas in abatement were never favorites with the courts in Fundance A merica. Lord Cole to the wet that they muct

In England or America. Lord Coke tells us that they must be "certain to a certain intent, in overy particular," and in practice they are always dealt with very strictly. When, therefore, the Supreme Court affirmed the plea in abate-ment in this case, I assume that it is affirmed, in manner and form, as written, and not otherwise. And this not merely because pleas in abatement are always considered Mercy because pleas in anatoment are always considered strict legis, but also, and chiefly, because the decision tends to abridge the valuable rights of persons natural-born in the country, which rights ought not to be impaired, except upon the clearest evidence of fact and law.

Taking the plea, then, strictly as it is written, the persons who are excluded by this judgment from being citizens of Who are excluded by this potential to be of the blacks, nor multiple of the blacks of even though they come of the blackest Malays in soum-eastern Asia. They must have had ancestors, (yet that may be doubtful, if born in slavery, of putative parents, who were slaves, and being slaves, incapable of contracting matrimony, and therefore every child must needs be a bas-tard, and so, by the common law, nullius filius, and inca-pable of ancestors.) His ancestors, if he had any, must have been of pure African blood, not mixed with the tawny Moor of Moroece or the dusky Arab of the desert, both of whom had their origin in Asia. They must have been brought to this country, not come voluntarily; and sold, not kept by the importer for his own use, nor given to his friends.

In this argument I raise no question upon the legal valid-ity of the judgment in Scott vs. Sandford. I only insist that the judgment in that case is limited in law, as it is, in fact, limited on the fence of the record to the place the judgment in that case is innited in law, as it is, in fact, limited on the face of the record, to the plea in abatement; and, consequently, that whatever was said in the long course of the case, as reported, (240 pages,) respecting the legal merits of the case, and respecting any supposed legal disability resulting from the mere fact of color, though en-

titled to all the respect which is due to the learned and up-right sources from which the opinions come, was " dehors

right sources from which the opinions come, was "dehors the record," and of no authority as a judicial decision. To show that, notwithstanding all that was said upon other subjects, the action of the court was strictly confined to the plea in mathematic, I copy the judgment: "Upon the whole, therefore, it is the judgment of this court that it appears by the record before us that the plain-

tiff in error is not a citizen of Missouri, in the sense in which that word is used in the Constitution, and that the circuit that word is used in the Constitution, and that the circuit court of the United States, for that reason, had no jurisdic-tion in the case, and could give no judgment in it. Its judg-ment for the defendant must, consequently, be reversed, and a mandate issued, directing the suit to be dismissed for want of jurisdiction." And now, upon the whole matter, I give it as my opinion that the free man of color, mentioned in your letter, if born in the United States, is a citizen of the United States, and, if otherwise qualified is computed according to the acts

if otherwise qualified, is competent, according to the acts of Congress, to be master of a vessel engaged in the coasting trade.

All of which is respectfully submitted by your obedient servant.

EDWARD BATES, Attorney General.

# Pay of Colored Soldiers.

Attorney General Bates has decided that persons of color who were free on the 19th of April, 1861, and who were enlisted and mustered into the military service of the United States between December, 1862, and 16th of June. 1864, are entitled by law to receive the same amounts of pay, bounty and clothing, as are, by the law existing at the times of their enlistment, allowed to other soldiers in the volunteer forces of the United States of like arms of the service.

> THE OPINION IN FULL. ATTORNEY GENERAL'S OFFICE July 14, 1864.

## TO THE PRESIDENT :

SR: By your communication of the 24th ultimo, you re-quire my opision in writing, as to what amounts of pay, bounty and clothing are allowed by law to persons of color who were free on the 19th day of April, 1851, and who have been enlisted and mustered into the military service of the United Stotac hatmos the month of December 1862 and United States between the month of December, 1862, and the 16th of June, 1864. I suppose that whatever doubt or difficulty may exist

I suppose that whatever doubt or dimiculty may exist with regard to the amount of pay and allowances to which the soldiers to whom you refer are entitled, has mainly its origin in the several provisions of the act of July 17th, 1862, chap. 201, (12 Stat. 599.) relative to the employment and en-rollment of persons of *African descent* in the service of the United States. The 12th section of that statute provides, "That the President be and he is hereby authorized to receive into the service of the United States, for the purpose of constructing intreachments or performing camp service, or any other labor, or any military or naval service for which they may be found competent, persons of African descent, and such persons shall be enrolled and organized under such regulations, not inconsistent with the Constitution and laws, as the President may prescribe." The 15th section of the same statute enacts, that "persons of African de-scent who under this law shall be employed shall receive ten dollars per month and one ration, three dollars of which

dollars per month and one ration, three dollars of which monthly pay may be in clothnig.<sup>29</sup> The first and main question, therefore, is, whether the persons of color referred to in your letter, who were mustered into the military service of the United States du-ring the period of time which you indicate, are "persons of African descent," employed unxer the statute of July 17th, 1862, chap. 201. If they are not thus employed, their com-perention should not be governed and is not regulated by pensation should not be governed and is not regulated by the words of the 15th section of the statute, which I have just quoted.

Now I think that it is clear-too clear indeed to admit of doubt or discussion-that those persons of color who have volnntarily enlisted and have been mustered into our military service—who have been organized with appropriate officers into companies, regiments, and brigades of soldiers— and who have done and are doing in the field and in gar-rison the duty and service of soldiers of the United States are not persons of African descent employed under the statute to which I have referred. I do not find, indeed, in the act any anthority to enlist

persons of African descent into the service, as soldiers. It will be observed that the said twelfth section enumerates two kinds of employment for which those persons are authorized to be enrolled, namely, constructing intrenci-ments and performing camp service. The section then con-tains a more general authority—authority to receive such persons into the service for the purpose of performing "any other labor or any military or naval service for which they may be found competent." I am bound, however, by every rule of law, respecting the construction of statutes, to con-strue these words of more general authority with reference strue these words of more general authority with reference strue these words of more general authority with reference to the character, nature and quality of the particular kinds of labor and service which are, in the first instance, specific-ally enumerated in the statute, as those for the performance of which persons of *African descent* are authorized to be received into the service, and, therefore, I must suppose that Congress, when it conferred authority upon the Presi-dent to receive into the service of the United States persons of African descent for the numers of not forming any affer of African descent for the purpose of performing any other labor or any military service for which they may be found competent, meant and intended that that other labor and competent, meant and intended that that other idoor and military service should be of the same general character, nature and quality as that which it had previously in the statute, specially named and designated. "Always in statutes," says Coke, "relation shall be made according to the matter precedent." Dwarris says: "sometimes works

the matter precedent." Dwarris says: "sometimes works and sections are governed and explained by conjoined words and clauses: noscitur a socis." (Dwarris on stat. 604.) Applying these rules of construction then to the act be-fore me, I am constrained to hold, that if the authority to enlist and muster into the military service soldiers of African descent depended upon this statute, (as it does not.) the wild be independent of a more authority. It is African descent depended upon this statute, (as it does not.) it would furnish no foundation for such authority. It is manifest that the labor and service that United States soldiers are enlisted to perform, are of an essentially differ-ent character from, and are essentially of a higher nature, order and quality than those kinds of labor and service specifically named in the statute, and for the performance of which the President is specially authorized to employ "persons of African descent." In my late opinion in the case of the claim of Rev. Samuel Harrison for full pay as Cheplain of the 54th Regiment of Massachusetts Volun-teers, I expressed the same view when I said that the act of July 17, 1862, chap. 201, "was not intended either to authorize the employment or to fix the pay of any persons of African descent, except those who might be needed to perform the humbler offices of labor and service for which they might be found competent."

they might be found competent." This view finds confirmation in a statute that received the approval of the President on the same day as the act before me—the statute of July 17, 1862, chap. 195 (12 Stat., 592)— which conferred upon the President the authority to employ as many persons of African descent ashe might deem neces-sary and proper for the suppression of the rebellion, and gave him power to organize and use them in such manner as he might judge best for the public welfare. In these words we may find clear and ample authority for the enlist-its *under* this act, if under either of the acts of July 17, 1862, that colored volunteer soldiers may be said to have been employed. There is no need to resort, therefore, to 1862, that colored volunteer soldiers may be said to have been employed. There is no need to resort, therefore, to the satute of July 17, 1862, chap. 201, for any authority with respect to their employment, or for any rule in regard to their compensation. Persons of African descent em-ployed as soldiers are not embraced at all, as I have shown, by the act of July 17, 1862, chap. 201, as objects or subjects of legislation; and we must therefore look to some other law for the measure of their compensation. If find the law for the compensation of the persons of color referred to in work letter to me in the acts of Concress in

referred to in your letter to me in the acts of Congress in force at the dates of the enlistments of these persons, re-specting the amount of pay and bounty to be given and the amount and kind of clothing to be allowed to soldiers in the voluntee service of the United States. For, after a careful and critical examination, I believe, of every statute enacted since the foundation of the present Government relative to the enlistment of soldiers in the regular and volunteer forces of the United States, I have found no law which at forces of the United States, I have found no haw which at any time prohibited the emlistment of free colored men into either braach of the national military service. The words of Congress descriptive of the recruits competent to enter the services were, in the act of April 30, 1790, "able-bodied men not under tive feet six inches in height without shoes, not under the age of eighteen nor above the age of forty-five;" in tho act of March 3, 1795, "able-bodied, of at least five feet six inches in height, and not under the age of oighteen nor above the age of forty-six years:" in the act of March 3, 1799, "able-bodied and of a size and age suitable for the unblie service according to the directions which the of March 3, 1799, "able-bodied and of a size and age suitable for the public service according to the directions which the President of the United States shall and may establish;" in the act of March 16, 1802, "effective able-bodied citizens of the United States, of at least five feet six inches high and between the ages of eighteen and forty-five years;" in the acts of December 24, 1811, January 11, 1812, January 20,

1813, and January 27, 1814, "effective able-bodied men;" in the act of December 10, 1814, "free, effective able-bodied men between the ages of eighteen and fifty years;" and in the act of January 12, 1847, "able-bodied men." Some of the foregoing statutes are obsolete; others of them are still a foregoing statutes are obsolete; others of them are still the foregoing statutes are obsolete; others of them are still in force, and furnished, before the suspension of the writ of habeas corpus, the rule by which the validity of the enlist-ments of persons alleged to have been minors was every day tried in the State and Federal courts. They organized the military establishments of the United States in time of peace and in time of war. They embrace the periods of all the wars, previously to the present, in which the United States has been engaged. By no one of them was or is the enlistment of free colored men into the military service of the United States whether as volunteers or services of the United States, whether as volunteers or as regulars, prohibited. After the war of 1812 claims for bounty land preferred by persons of color who had enlisted and served pretered by persons of color who had emissed and served in the army under the statutes of 24th December, 1811, Jan-nary 11, 1812, and December 10, 1814, were sustained as valid by the then Attorney General, William Wirt. (1 Opin, 603.) And when I turn to more recent statutes—those which authorized the raising and regulate the organization of the whole body of the volunteer forces now in the field, and provided for the maintenance and increase of the regular forces in the sorvice-I discover throughout them no other statutory qualifications for recruits than those established by the earliest legislation to which I have referred. It is not needed that I should specially recite the words

of those acts of Congress that provide for the pay, bounty, of mose acts of congress that provide for the pay, bounds, and clothing to be allowed to soldiers in the volunteer military service of the United States. It is enough to say that under the statutes relative to those subjects, and in force during the period of time mentioned in your commu-nication, all volunteers competent and qualified to be mem-bers of the national forces, are entitled respectively to re-ceive like amounts of pay, bounty, and clothing from the Government. Government.

In view, therefore, of the foregoing considerations, I give In view, therefore, of the foregoing considerations, 1 give it to you unhesitatingly, as my opinion, that the same pay, bounty, and clothing are allowed by law to the persons of color referred to in your communication, and who were en-listed and mustered into the military service of the United States between the months of December, 1862, and the 16th of June, 1864, as are by the laws existing at the times of the enlistments of said persons, authorized and provide dor, and allowed to, other soldiers in the volunteer forces of the United States of like arms of the service.

I have the honor to be, very respectfully, your obedient rvant. EDWARD BATES. servant.

## Gen. McClellan's Letters.

## ON POLITICAL ADMINISTRATION, JULY 7, 1862.

IIEADQUARTERS ARMY OF THE POTOMAC, CAMP NEAR HARRISON'S LANDING, VA., July 7, 1862.

MR. PRESIDENT: You have been fully informed that the rebel army is in the front, with the purpose of overwhelm-ing us by attacking our positions or reducing us by blocking us by attacking our positions or reducing us by block-ing our river communications. I cannot but regard our condition as critical, and I earnestly desire, in view of pos-sible contingencies, to lay before your excellency, for your private consideration, my general views concerning the existing state of the rebellion, although they do not strictly existing state of the rebending, actuding they do battery relate to the situation of this army, or strictly come within the scope of my official duties. These views amount to convictions, and are deeply impressed upon my mind and heart. Our cause must never be abandoned; it is the cause of free institutions and self-government. The Constitution and the Union must be preserved, whatever may be the cost in time, treasure, and blood. If secession is successful, other dissolutions are clearly to be seen in the future. Let other military disaster, political faction, nor foreign war shake your settled purpose to enforce the equal operation of the laws of the United States upon the people of every State.

The time has come when the government must determine upon a civil and military policy, covering the whole ground

upon a civil and military policy, covering the whole ground of our national trouble. The responsibility of determining, declaring, and support-ing such civil and military policy, and of directing the whole course of national affairs in regard to the rebellion, must now be assumed and exercised by you, or our cause will be lost. The Constitution gives you power, even for the present terrible exigency. This rebellion has assumed the character of a war; as such it should be recarded and it should be conducted upon

such it should be regarded, and it should be conducted upon the highest principles known to Christian civilization. It should not be a war looking to the subjugation of the people of any State, in any event. It should not be at all a war upon population, but against armed forces and political organizations. Neither confiscation of property, political executions of persons, territorial organization of States, or

forcible abolition of slavery, should be contemplated for a moment.

In prosecuting the war, all private property and unarmed persons should be strictly protected, subject only to the necessity of military operations; all private property taken for military use should be paid or receipted for; pillage and wasto should be treated as high crimes; all unnecessary trespass sternly prohibited, and offensive demeanor by the military towards citizens promptly rebuked. Military arrests should not be tolerated, except in places where active hostilities exist; and oaths, not required by enactments, constitutionally made, should be neither demanded nor received.

Military government should be confined to the preservation of public order and the protection of political right. Military power should not be allowed to interfere with the relations of servitude, either by supporting or impairing the authority of the master, except for repressing disorder, as in other cases. Slaves, contraband under the act of Congress, seeking military protection, should receive it. The right of the government to appropriate permanently to its own service claims to shave labor should be asserted, and the right of the owner to compensation therefor should be recognized. This principle might be extended, upon grounds of military necessity and security, to all the slaves of a particular State, thus working manunission in such State; and in Missonri, perhaps in Western Virginia also, and possibly even in Maryland, the expediency of such a measure is only a question of time. A system of policy thus constitutional, and pervaded by the influences of Christianity and freedom, would deeply impress the rebel masses and all foreign nations, and it might be humbly hoped that it would commend itself to the favor of the Almighty.

Unless the principles governing the future conduct of our struggle shall be made known and approved, the effort to obtain requisite forces will be almost hopeless. A declaration of radical views, especially upon slavery, will rapidly disintegrate our present armies. The policy of the government must be supported by concentrations of military power. The national forces should not be dispersed in expeditions, posts of occupation, and numerous armies, but should be mainly collected into masses, and brought to bear upon the armies of the Confederate States. Those armies thoroughly defeated, the political structure which they support would soon cease to exist.

In carrying out any system of policy which you may form, you will require a commander-in-chief of the army, one who possesses your confidence, understands your views, and who is competent to execute your orders, by directing the military forces of the nation to the accomplishment of the objects by you proposed. I do not ask that place for myself. I am willing to serve you in such position as you may assign me, and I will do so as faithfully as ever subordinate served superior.

I may be on the brink of eternity; and as I hope forgiveness from my Maker, I have written this letter with sincerity towards you and from love for my country. Very respectfully, your obedient servant,

GEORGE B. McCLELLAN,

Major General Commanding. His Excellency A. LINCOLN, President.

IN FAVOR OF THE ELECTION OF GEORGE W. WOOD-WARD AS GOVERNOR OF PENNSYLVANIA.

#### ORANGE, NEW JERSEY, October 12, 1863.

DEAR STR: My attention has been called to an article in the Philadelphia Press, asserting that I had written to the managers of a Democratic meeting at Allentown, disapproving the objects of the meeting, and that if I voted or spoke it would be in favor of Goternor Curtin, and I am informed that similar assertions have been made throughout the State.

It has been my earnest endeavor heretofore to avoid participation in party politics. I had determined to adhere to this course, but it is obvious that I cannot longer maintain silence under such misrepresentations. I therefore request you to deny that I have written any such letter, or entertained any such views as those attributed to me in the Philadelphia *Press*, and I desire to state clearly and distinctly, that having some days ago had a full conversation with Judge Woodward, I find that our views agree, and I regard his election as Governor of Pennsylvania called for by the interests of the nation.

Inderests of the nation. I understand Judge Woodward to be in favor of the prosecution of the war with all the means at the command of the loyal States, until the military power of the rebellion is destroyed. I understand him to be of the opinion that while the war is urged with all possible decision and energy, the policy directing it should be in consonance with the

principles of humanity and civilization, working no injury to private rights and property not demanded by military necessity and recognized by military law among civilized nations.

And, finally, I understand him to agree with me in the opinion that the sole great objects of this war are the restoration of the unity of the nation, the preservation of the Constitution, and the supremacy of the haws of the country. Believing our opinions entirely agree upon these **points**, **I** would, were it in my power, give to Judge Woodward my voice and vote.

I am, very respectfully, yonrs, GEORGE B. McCLELLAN.

Hon. CHARLES J. BIDDLE.

## Proposed Censures of Officials.

#### OF PRESIDENT LINCOLN.

First Session, Thirty-Seventh Congress. IN HOUSE.

### July 15-Mr. VALLANDIGHAM offered the following :

Resolved, That the Constitution of the United States confers upon Congress alone the power to "raise and support armies," and to "provide and maintain a navy," and therefore the President, in the proclamation of May 3, 1861, and the orders and action, by his authority, of the War and Navy Departments, increasing the Army and Navy, and calling for and accepting the services of volunteers for three years without warrant of law, usurped powers belonging solely to Congress, and so violated the Constitution. 2. That the right to declare a blockade as against an incompadent power is a belligerent right deconding upon

2. That the right to declare a blockade as against an independent power, is a belligerent right, depending upon the existence of a state of war; and that as Congress, and Congress alone, have the power to declare or recognize the existence of war, the President has no right to order a blockade until after Congress shall have declared or recognized war with the power whose ports are to be blockaded; and further, that Congress alone can abolish or shut up the perts of entry of any State within the Union; and that, the authority of Congress, violated the Union, without the authority of Congress, violated the Constitution.

3. That Congress alone have the constitutional power to suspend the writ of habcas corpus; and that until the writ has been suspended by act of Congress, it is the duty of the President, and all other officers, civil and mailtary, to obey it; and that, therefore, the President, in suspending said writ himself, or attempting to authorize certain military officers to suspend it, or to disobey it, or in sustaining them in disobedlence to it, violated the Constitution.

writ himself, or attempting to anthorize certain military officers to suspend it, or to disobey it, or in sustaining them in disobedience to it, violated the Constitution. 4. That by the Constitution "no money shall be drawn from the Treasury but in consequence of appropriations made by haw;" and that in ordering the drawing from the Treasury of money unappropriated or appropriated for one purpose, and applying the same to purposes for which no appropriations.

5. That the search of certain telegraph offices in the month of May last, by several officers and agents of the Executive, without search warrant upon probable cause, supported by oath or affirmation, and particularly describing the place to be searched, and the things to be seized; and the seizure of papers and despatches in said offices was a violation of the constitutional "right of the people to be secure in their persons, houses, papers, and effects again unreasonable searches and seizures;" and that the Predent, in ordering such search and seizures, violated the Constitution.

6. That neither Congress, nor the President, nor the judiciary, have any constitutional power to abridge the freedors of speech or of the press; and that the suspension of new *A* paper presses by military authority and force, and the arrest of citizens by military or civil authority, for the expression by speech, or through the press, of opinions upon political subjects, or subjects of any kind, is a violation of the Constitution.

7. That the arrest without civil process of persons net subject to the rules and articles of war, nor in cases arising in the land or inval forces or in the millitia, whea in actual service, by soldiers in the service of the United States, is a breach of the Constitution, and a violation of the constitutional liberty of the person.

On motion of Mr. LOVEJOY, these resolutions were at once laid upon the table—the House refusing to order the yeas and nays.

#### OF EX-PRESIDENT BUCHANAN.

## Third Session, Thirty-Seventh Congress. IN SENATE.

1862, Dcc. 15-Mr. DAVIS, of Kentucky, offered this resolution :

Resolved. That after it had become manifest that an insurrection against the Upited States was about to break out in several of the southern States, James Buchanan, then Pres-ident, from sympathy with the conspirators and their treasonable project, failed to take necessary and proper measures to prevent it; wherefore he should receive the censure and condemnation of the Senate and the American people.

Dec 16-Mr. SAULSBURY offered this amendment:

Resolved further, That a copy of the foregoing resolution be served upon the said James Buchanan, and that he be notified that he has liberty to defend himself before the Senate against the charges in said resolution contained, if he shall choose so to do.

Same day - The resolution and proposed amendment were laid upon the table-yeas 38, nays 3, as follows:

YEAS-Messrs. Anthony, Arnold, Browning, Carlile, Clark, Collamer, Cowan, Dixon. Doolittle, Fessenden, Field, Foot, Foster, Grimes, Hale, Harding, Harlan, Harris, Henderson, Kennedy, King, Laue of Indiana, Lane of Kanasa, Latham, Morrill, Nesmith, Pomeroy, Pwell, Rice, Saulsbury, Sher-man, Ten Eyck, Trumbull, Wade, Willey, Wilson of Mass-achusetts, Wilson of Missouri, Wright-38. NATS-Messrs. Davis, Howe, Wilkinson-3.

## OF MESSRS. LONG AND HARRIS. First Session, Thirty-Eighth Congress. IN HOUSE.

1864, April 9-Mr. COLFAX, the Speaker, (Mr. Rollins, of New Hampshire, in the Chair, ) offered this preamble and resolution :

Whereas on the 8th of April, 1864, when the House of Representatives was in Committee of the Whole on the state of the Union, ALEXANDER LONG, a Representative from the second district of Ohio, declared himself in favor of recognizing the independence and nationality of the so-called confederacy now in arms against the Union; and whereas, the said so-called confederacy, thus sought to be whereas, the said so-called confederacy, thus sought to be recognized and established on the ruins of a dissolved or destroyed Union, has as its chief officers, civil and military, those who have added perjury to their treason, and who seek to obtain success for their particidal efforts by the killing of the loyal soldiers of the nation who are seeking to save it from destruction; and whereas the oath required of all members, and taken by the said ALEXANDE LONG on the first day of the present Congress, declares "that I have voluntarily given no aid countenance coursel or encours voluntarily given no aid, countenance, counsel, or encour agement to persons engaged in armed hostility to the United States," thereby declaring that such conduct is re-garded as inconsistent with membership in the Congress of

garded as inconsistent with membership in the Congress of the United States: Therefore, Resolved, That ALEXANDER LONG, a Representative from the second district of Ohio, having, on the 8th of April, 1864, declared himself in favor of recognizing the indepen-dence and nationality of the so-called confederacy now in arms against the Union, and thereby "given aid, counte-mance, and encouragement to persons engaged in armed hostility to the United States," is hereby expelled.

Pending which, April 9, Mr. WASHBURNE, of Illinois, offered this resolution :

Whereas Hon. BENJAMIN G. HARRIS, a member of the Honse of Representatives of the United States from the State of Maryland has on this day used the following lan-guage, to wit: "The South asked you to let them go in peace. But, no; you said you would bring them into sub-jection. That is not done yet, and God Almighty grant that it never may be. I hope that you will never subjugate the South." And whereas such language is treasonable, and is South." And whereas such language is treasonable, and is a gross disrespect of this Honse: Therefore, *Be it resolved*, That the said BEXJAMIN G. HARRIS be ex-pelled from this House.

Which was rejected—yeas 84, nays 58, (two thirds being required :)

YEAS-Messrs. Alley, Allison, Ames, Anderson, Arnold,

Ashley, Baily, John D. Baldwin, Baxter, Beaman, Blaine, Francis P. Blair, Boutwell, Boyd, Broomall, William G. Brown, Ambrose W. Clark, Freeman Clarke, Cobb, Cole, Creswell, Henry Winter Davis, Thomas T. Davis, Dixon, Donnelly, Driggs, Dumont, Eckley, Eliot, Frank, Garlield, Gooch, Grin-nell, Hale, Nigby, Hooper, Hotchkiss, Asahel W. Hubbard, John H. Hubbard, Jenckes, Julian, Kasson, Kelley, Francis

heir, Fride, Jrigoy, Hooger, Hofenkiss, Asande W. Hubbard, John H. Hubbard, Jenckes, Julian, Kasson, Kelley, Francis W. Kellogg, Orlando Kellorg, Loan, Marvin, McBride, Mc-Clurg, McIndoe, Sanuel F. Miller, Morrill, Daniel Morris, Amos Myers, Leonard Myers, Norton, Orth, Patterson, Pike, Pomeroy, Price, William H. Randall, Edward H. Rollins, Schenck, Scofield, Shannon, Smith, Smithers, Spalding, Starr. Thayer, Thomas, Tracy, Upson, Van Valkenburgh, Ellihu B. Schenck, Scofield, Shannon, Smith, Smithers, Spalding, Starr. Mayer, Thomas, Tracy, Upson, Van Valkenburgh, Ellihu B. Washburne, William B. Washburn, Webster, Whaley, Wil-Mans, Wilder, Wilson, Windom, Woodbridge—Sta. Nars-Messrs, James C. Allen, Ancona, Augustus C. Bald-win, Bliss, James S. Broum, Clanler, Clay, Cox, Cravens, Dawson, Denison, Elen, Eldridge, English, Finck, Ganson, Grider, Harding, Marrington, Herrick, Holman, Hulchins, Philip Johnson, William Johnson, Kernan, Law, Lazear, Morrison, Neleon, O.J.tl, Pendleton, Pruyn, Samuel J. Renris, Morrison, Neleon, O.J.tl, Pendleton, Pruyn, Samuel J. Ren-dall, Robinson, Roger, James S. Rollins, Koss, Scott, John B. Stecle, William G. Steele, Strouse, Swocat, Toorhees, Ward, Chillon A., White, Joseph W. White, Winfield, Fernando Wood—58. Wood-58.

#### Mr. SCHENCK then offered this resolution :

Resolved, That BENJAMIN G. HARRIS, a Representative from the fifth district of the State of Maryland, having spoken words this day in debate, manifestly tending and designed to encourage the existing rebellion and the ene-mies of this Union, is declared to be an unworthy member of this House, and is hereby severely consured.

A motion to table the resolution was lostyeas 23, nays 80; two motions to adjourn were made and voted down; and the resolution was then adopted—yeas 98, nays 20, as follows :

YEAS—Messrs. Alley, Allison, Ames, Anderson, Arnold, Ashley, Baily, Augustus C. Baldwin, John D. Baldwin, Bax-ter, Beaman, Blaine, F. P. Blair, Boutvell, Boyd, Broomall, James S. Brown, Ambrose W. Clark, Freeman Clarke, Cobb, Cole, Creswell, Cox, Henry Winter Davis, Thomas T. Davis, Dixon, Donnelly, Driggs, Dumont, Eckley, Eliot, Eug-lish, Frank, Gancon, Garfield, Gooch, Grinnell, Hale, Har-rington, Higby, Holman, Hotchkiss, Asahel W. Hubhard, John H. Hubbard, Jenckes, Julian, Kasson, Kelley, Francis W, Kelloge, Orlando Kellogez, Kernan, Loan, Marvin, Me-Sonn R. Hubbard, Jenckes, Junar, Nasson, Kenley, Francis W. Kellogg, Orlando Kellogg, Kernan, Loan, Marvin, Mo-Allister, McBride, McClurg, McIndoe, Middleton, Samuel F. Miller, Morrill, Daniel Morris, Amos Myers, Leonard Myers, Nelson, Norton, Odell, Orth, Patterson, Pike, Fomeroy, Price, William II, Randall, Edward H. Rollins, Schenck, Social Shannon, Storn, Start, Smithers, Snelding, Storp. Price, William H. Randall, Edward H. Rollins, Schenck, Scofield, Shannon, Stoan, Smith, Smithers, Spalding, Starr, John B. Steele, William G. Steele, Thayer, Thomas, Tracy, Upson, Van Valkenburgh, Ellibu B. Washburne, William B. Washburn, Webster, Whaley, Williams, Wilder, Wilson, Windom, Winfield, Feanuan-98. NATS-Messrs. James C. Allen, Ancona, Bliss, Chanler, Denison, Eden, Eldridge, Law, Le Blond, Long, Wm. H. Mü-ler, Morrison, Fendleton, Praym, Samuel J. Randall, Ross, Strouse, Voorhees, Chilton A. White, Fernando Wood-20.

The question recurring upon the resolution offered by Mr. Colfax-

April 14-Mr. BROOMALL offered this amendment, as a substitute, which Mr. COLFAX accepted :

Whereas ALEXANDER LONG, a Representative from the whereas ALEXANDER LOSG, a Representative from the second district of Ohio, by his open declarations in the na-tional Capitol and publications in the city of New York, has shown himself to be in favor of a recognition of the so-called confederacy now trying to establish itself upon the ruins of our country, thereby giving aid and comfort to the enemy in that destructive purpose—aid to avoved traitors in cre-ating an illegal government within our borders—comfort to the human conference of their within our borders—comfort to them by assurances of their success, and affirmations of the justice of their cause; and whereas such conduct is at the same time evidence of disloyalty and inconsistent with his oath of office and his duty as a member of this body: Therefore

Resolved, That the said ALEXANDER LONG, a Representa-tive from the second district of Ohio, be, and he is hereby, declared to be an unworthy member of the House of Representatives.

Resolved, That the Speaker shall read these resolutions to the said ALEXANDER LONG during the session of the House.

Mr. Cox moved to lay the preamble and reso-

lution on the table; which was disagreed to-A division of the question yeas 70, nays 80. was called, when

The first resolution was agreed to-yeas 80, nays 70, as follows :

nays 70, as follows : YEAS-Messrs. Alley, Allison, Ames, Anderson, Arnold, Ashley, Bady, John D. Baldwin, Baxter, Beaman, Blaine, Boutwell, Boyd, Broomall, Ambrose W. Clark, Cobb, Cole, Creswell, Dawes, Deming, Driggs, Dumont, Eckley, Farns-worth, Frank, Garfield, Gooch, Grinnell, Higby, Hooper, Hotchkiss, Jehn H. Hubbard, Jenckes, Julian, Kasson, Kelley, Francis W. Kellogg, Orlando Kellogg, Loan, Long-year, Marvin, McBride, McClurg, MoIndoe, Sanuel F. Miller, Morrill, Daniel Morris, Amos Myers, Leonard Myers, Nor-ton, Charles O'Neill, Orth, Patterson, Perham, Pike, Pome-roy, Price, William H. Randall, Alexander H. Rice, John H. Rice, Edward H. Rollins, Schenck, Shannon, Sloan, Smith, Smithers, Starr, Stovens, Thayer, Thomas, Upson, Yan Valkenburgh, Ellihu B. Washburne, William B. Wash-burn, Webster, Whaley, Wilder, Wilson, Windom, Wood-bridge=80.

Jurn, Webster, Whaley, Wilder, Wilson, Windom, Woodbridge-80.
 NArs-Messre, James C. Allen, William J. Allen, Ancona, Augustus C. Baldwin, Francis P. Blair, Bliss, James S. Brown, William G. Brown, Chanler, Clay, Coffroth, Cox, Graeens, Dewson, Denison, Eden, Eldridge, Finel, Ganson, Grider, Hall, Harding, Harrington, B. G. Harris, Herriele, Holman, Hutchins, P. Johnson, Wm. Johnson, Kalbfleisch, Kernan, King, Knapp, Law, Lazear, Mallory, Marcy, McDowell, McKinney, William H. Miller, James R. Morris, Morrison, Nelson, Noble, Odell, John O'neill, Pendleton, Perry, Pruyn, Radford, Samuel J. Randall, Robinson, Rogers, James S. Rollins, Ross, Scott, Stebhns, John B. Steele, William G. Steele, Strouse, Stuart, Sweat, Vonchees, Ward, Wheeler, Chilton A. While, Joseph W. White, Winfield, Fernando Wood, Yeanan-70.

The second resolution was laid on the tableyeas 71, nays 70, as follows:

Yeas 71, nays 70, as follows: YEAS-Messrs. James C. Allen, William J. Allen, Ancona, Baily, Augustus C. Baldwin, Eliss, James S. Brown, Wm. G. Brown, Chanler, Clay, Coffrolh, Cox, Dawson, Denison, Eden, Eldridge, Finele, Ganson, Grider, Hall, Harding, Har-ringlon, Benjamin G. Harris, Herrick, Holman, Hutchins, William Johnson, Kalbfeisch, Kernan, Kirg, Knapp, Lau, Lazear, Mallory, Marcy, McDowell, McKinney, William H. Muller, James E. Morris, Morrison, Nolson, Noble, Odell, John O'Neill, Pendleton, Perry, Pruyn, Radford, Samuel J. Randall, William H. Randall, Robinson, Rogers, James S. Rollins, Ross, Scott, Smith, Stebbins, John B. Stele, William G. Steele, Strouse, Stuart, Sweat, Voorhees, Ward, Webster, Wheeler, Chillon A. While, Joseph W. While, Winfield, Pernando Wood, Feaman-Tl. NATS-Messrs. Alley, Allison, Ames, Anderson, Arnold, Ashley, John D. Baldwin, Baxter, Beaman, Blaine, Bout-well, Boyd, A. W. Clerk, Cobb, Cole, Creswell, Dawes, Dem-ing, Driggs, Dumont, Eckley, Franswerth, Frank, Garfield, Jenckes, Julian, Kasson, Kelley, Francis W. Kcllogg, Or-lando Kellogg, Loan, Longyear, Marvin, McBride, McClurg, McIndoe, Morrill, Daniel Morris, Amos Myers, Norton, Charles O'Neill, Orth, Patterson, Perham, Pike, Pomeroy, Charles O'Neill, Orth, Patterson, Perham, Disano, Ward, Wilder, Wilson, Windom, Woodbridge-70. The preamble was then agreed to-yeas 78, Nare Men March March March March March March March March March Nare Colument Milliam B. Washburne, William B. Washburn,

The preamble was then agreed to-yeas 78, nays 63, as follows:

häys 65, as 10110WS: YEAS-Messrs. Alley, Allison, Ames, Anderson, Arnold, Ashley, Eaïly, John D. Baldwin, Baxter, Beaman, Blaine, Boutwell, Boyd, Broomall, Ambrose W. Clark, Cobb, Cole, Creswell, Dawes, Driggs, Dumont, Eckley, Francis, Garfield, Gooch, Grinnell, Higby, Hooper, Hotchkiss, John H. Hub-bard, Jenckes, Julian, Kasson, Kelley, Francis W. Kellogg, Orlando Kellogg, Lean, Longyear, Marvin, McBride, Mc-Clurg, McIndøe, Samuel F. Miller, Morrill, Daniel Morris, Amot Myers, Leonard Myers, Norton, Charles O'Neill, Orth, Patterson, Perham, Pike, Pomeroy, Price, William H. Ran-dall, Alexander H. Rice, John H. Rice, Edward H. Rollins, Schenck, Shannon, Sloan, Smith, Smithers, Starr, Stevens,

Thayer, Thomas, Upson, Van Valkenburgh, Ellihu B. Washburn, William B. Washburn, Webster, Whaley, Wilder, Wilson, Windom, Woodbridge-78. NATS-Mossrs. James C. Allen, William J. Allen, Ancona, Augustus C. Baldwin, Bliss, James S. Brown, William G. Brown, Chanter, Clay, Coffroth, Cox, Davson, Denison, Geider, Eldvilge, Finck, Ganson, Grider, Hall, Harding, Benjamin G. Harris, Herrick, Holman, Hutchins, William Johnson, Kallfeitsek, Kernan, Knapp. Law, Lazer, Marey, McDowell, McKinney, William H. Miller, James F. Morris, Morrison, Nelson, Nole, Odell, John O'Neill, Pendleton, Ferry, Pruyn, Radford, Samuel J. Randall, Foldinson, Rogers, James S. Rollins, Ross, Scott, Stebbins, John E. Steele, Chilton A. While, Joseph W. While, Winfield, Fernande Wood, Feaman-63.

## Completed Vote on the Passage in the House, of the Bill to Repeal the Fugitive Slave Laws.

After the vote given on page 237, was taken, six other members received consent to record their votes, viz: Henry C. Deming, John R. McBride, Edward H. Rollins, and William B. Washburn in the affirmative, and Philip Johnson and Samuel J. Randall in the negativemaking the full vote, as recorded on the Journal, yeas 90, nays 62, as follows:

YEAS-Messrs. Alley, Allison, Ames, Arnold, Ashley, John D. Baldwin, Baxter, Beaman, Blaine, Blair, Blow, Boutwell, Boyd, Brandegee, Broomall, Ambrose W. Clark, Freeman Clarke, Cobb, Cole,Creswell, Henry Winter Davis, Boutwell, Boyd, Brandegee, Broomall, Ambrose W. Clark, Freeman Clarkc, Cobb, Cole, Creswell, Henry Winter Davis, Thomas T. Davis, Dawes, Deming, Dixon, Donnelly, Drigga, Eckley, Eliot, Farnsworth, Fenton, Frank, Garfield, Gooch, Griswold, Higby, Hooper, Hotchkiss, Asahel W. Hubbard, John H. Huhbard, Hulburd, Ingersoll, Jenckes, Julian, Kelley, Francis W. Kellogg, Orlando Kellogg, Littlejahn, Loan, Longyear, Marvin, McBride, McClurg, McIndee, Samuel F. Miller, Moorhead, Morrill, Daniel Morris. Amos Myers, Leonand Myers, Norton, Chas. O'Noill, Orth, Patter-son, Perham, Pike, Price, Alexander H. Rice, John H. Rice, Edward H. Rollins, Schenck, Scofield, Shannen, Sloan, Spalding, Starr, Stevons, Thayer, Thomas, Tracy, Upson, Van Valkenburgh, Wm. B. Washburn, Webster, Whaley, Williams, Wilder, Wilson, Windom, Woodbridge-90. NARS--Messes, James C. Allen, William J. Alten, Ancoma, Augustus C. Baldwin, Bliss, Brocks, James S. Brown, Chan-ler, Cofroth, Coz, Cravens, Dawson, Denison, Eden. Edger-ton, Eldridge, English, Finck, Ganson, Grider, Harding, Philip Johnson, Kalbfeisch, Kernan, King, Knapp, Law, Iazear, Le Blond, Mallory, Marcy, McDowell, McKinney, William H. Miller, James R. Morris, Morrison, Odell, Pen-dleton, Pruym, Radford, Samuel J. Randall, Robinson, James S. Rolins, Rose, Smines, Jonen S. Melawand, Muse, Stele, Stikes, Strouse, Stuart, Sweat, Wadsworth, Ward, Wheeler, Chilton A. White, Joseph W. White, Fernande Wood-62.

## California Representatives

IN THE THIRTY-SEVENTH CONGRESS.

By an accident, the names of the Representatives from California in the Thirty-Seventh Congress, were not inserted on page 122, as they should have been. They were not elected in time to be present at the first session of that Congress, in July, 1861. Messrs. Timothy G. Phelps and Aaron A. Sargent qualified December 2, 1861, the first day of the regular session. Frederick F. Low, now Governor of the State, was admitted to a seat June 3, 1862.

THE CONSPIRACY OF DISUNION.

In the slaveholding States, a considerable body of men have always been disaffected to They resisted the adoption of the the Union. National Constitution, then sought to refine away the rights and powers of the General Government, and by artful expedients, in a series of years, using the excitements growing out of passing questions, finally perverted the sentiments of large masses of men, and prepared them for revolution.

I had prepared an extensive collection of statements and facts bearing upon this point, but am obliged to omit them for want of space. The well-read in our politics can readily recur to a multitude of proofs. I append a few con-spicuous points, the first of which is less well known, being from an unpublished journal by Hon. William Maclay, United States Senator from Pennsylvania, from March 4, 1789, to March 3, 1791, being the First Congress under the Constitution. This journal is the property of his relative, George W. Harris, Esq., of Harrisburg, Pa.

## An early Threat of Dissolution.

#### FROM SENATOR MACLAY'S JOURNAL.

1789, June 9--In relation to the Tariff bill, the affair of confining the East India Trade to the citizens of America to mining the dates that a track to the formation of the second s the Senate) since Friday.

*Pierce Buller*, from South Carolina, had taken his seat, and flamed like a meteor. He arraigned the whole Im-post law, and then charged (indirectly) the whole Congress with a design of oppressing South Carolina. He cried out for encouraging the Danes and Swedes, and foreigners of mere kind to go one out take a provider In Sect. every kind, to come and take away our produce. he was for a navigation act reversed. June 11—Attended at the hall as usual. In fact,

Mr. Izard\* and Mr. Butler opposed the whole of the draw-

backs in every shape whatever. Mr. Graysont of Virginia, warm on this subject, said we were not ripe for such a thing. We were anew nation, and had no business for any such regulations-a nation sui generi:

Mr. Lect said drawbacks were right, but would be so much abused, he could not think of admitting them. Mr. Elsworth said New England rum would be ex-ported, instead of West India, to obtain the drawback.

I thought it best to say a few words in teply to each. We were a new nation, it was true, but we were not a new people. We were composed of individuals of like manners, habits, and customs with the European nations. What, therefore, had been found useful among them, came well accounted by consider the accounted by the second second accounted by the second second accounted by the second secon recommended by experience to us. Drawbacks stand as an

*Ralph,	†William.	tRichard Henry, from Virginia.
-	Oliver.	of Connecticut.

example, in this point of view, to us. If the thing was right in itself, there could be no just argument drawn against the use of a thing from the abuse of it. It would be the duty of Government to guard against abuses, by prudent appoint-ments and watchful attention to officers. That as to change institution of the advection of the second se ing the kind of run. I though the collection bill would provide for this, by limiting the exportation to the *original* casks and packages. I said a great deal more, but really did not feel much interest either way. But the debates were very lengthy.

were very lengthy. Butler flamed away, and THREATENED A DISSOLUTION OF THE UNION, with regard to his State, as sure as God was in the firmament. He scattered his remarks over the whole impost bill, calling it partial, oppressive, &c., and solely calculated to oppress South Carolina, and yet ever and anon declaring how clear of local views and how candid and dis-passionate he was. He degenerates into mere declamation. His State would live free; or die glorious.

Richard Henry Lee, grandfather of General Robert E. Lee, writing in 1790 on the national Constitution, is reported as saying:

When we (the South) attain our natural degree of popu-lation, I flatter myself that we shall have the power to do ourselves justice, with dissolving the bond which binds us together.

Referring to the modus operandi of southern disunionists, General JACKSON'S recently-discovered letter to Rev. A. J. Crawford is curious for the keenness of its perceptions, and the accuracy of its prediction :

#### ["Private."]

#### WASHINGTON, May 1, 1833.

"My DEAR SIR: \* \* \* I have had a laborious task here. but nullification is dead: and its actors and courtiers will but nullification is dead; and its actors and counters will only be remembered by the people to be executed for their wicked designs to sever and destroy the only good Govern-ment on the globe, and that prosperity and happiness we enjoy over every other portion of the world. Haman's gallows ought to be the fate of all such ambitious men-ples would swelve division country in a juil may and all the gallows ought to be the fate of all such ambitious men who would involve their country in civil war, and all the evils in its train, that they might reign and ride on its whirly inds and direct the storm. The free people of these United States have spoken, and consigned these wicked demagogues to their proper doom. Take care of your null-ifiers; you have them among you; let them meet with the isolarout formers of arows may who have be accurated a constru-tion of the states of the second results of the second results. indignant forwns of every man who loves his country. The tariff, it is now known, was a mere pretext—its burden was on your coarse woolens. By the law of July, 1832, coarse woolen was reduced to five per cent. for the benefit of the South. Mr. Clay's bill takes it up and classes it with woolens at fifty wor cont woolens. of the South. AIT. Clays bill takes it up and classes it with woolens at fifty per cert, reduces it gradually down to twenty per cent, and there it is to remain, and Mr. Cal-houn and all the nullifiers agree to the principle. The cash duties and home valuation will be equal to fitteen per cent. more, and after the year 1842, you pay on coarse woolens thirty-five per cent. If this is not protection, I cannot understand; therefore the tariff was only the pre-text and disjunion and a souther perform the real obtext, and disunion and a southern confederacy the real ob-

"My health is not good, but is improving a little. Pre-sent me kindly to your lady and family and believe me to be your friend. I will always be happy to hear from you. "ANDEEW JACKSON."

389

BENTON in his Thirty Years' View, says:

The regular inauguration of this slavery agitation dates from the year 1835; but it had commenced two years before, and in this way: nullification aud disunion had commenced in 1830, upon complaint against protective tariff. That, being put down in 1833 under President Jackson's proclamation and energetic measures, was immediately substituted by the slavery agitation. Mr. Calhoun, when he went home from Congress in the spring of that year, told his friends that "the South could never be united against the North on the tariff question—that the sugar interest of Louisiana would keep her out—and that the basis of southern union must be shifted to the slave question." Then all the papers in his interest, and especially the one at Washington, published by Mr. Duff Green, dropped tariff agitation, and commenced upon slavery, and in two years had the agitation ripe for inauguration on the slavery question. And in tracing this agitation to its present stage, and to comprehend its rationale, it is not to be forgotten that it is a mere continuation of old tariff disunion, and preferred because more available.—Thirty Tears in the Senate, vol. 2.

Mr. CLAY, in a letter to an Alabamian in 1864, (see his private correspondence, p. 490,) said:

From the developments now being made in South Carolina, it is perfectly manifest that a party exists in that State seeking a dissolution of the Union, and for that purpose employ the pretext of the rejection of Mr. Tyler's abominable treaty. South Carolina being surrounded by slave States, would, in the event of a dissolution of the Union, suffer only comparative evils; but it is otherwise with Kentucky. She has the boundary of the Ohio extending four hundred miles on three free States. What would our condition be in the event of the greatest calamity that could befall this nation?

Hon. Nathan Appleton, of Boston, member of Congress in 1832-3, in a letter dated December 15, 1860, said that when in Congress he "made up his mind that Messrs. Calhoun, Hayne, McDuffie, &c., were desirous of a separation of the slave States into a separate confederacy, as more favorable to the security of slave property."

About 1835, some South Carolinians attempted a disunion demonstration. It is thus described by Ex-Governor Francis Thomas of Maryland, in his speech in Baltimore, October 29, 1861:

Full twenty years ago, when occupying my seat in the House of Representatives, I was surprised one morning, after the assembling of the House, to observe that all the members from the slaveholding States were absent. Whilst reflecting on this strange occurrence, I was asked why I was not in attendance on the Southern cancus assembled in the room of the Committee on Claims. I replied that I had received no invitation.

I then proposed to go to the committee room, to see what was being done. When I entered I found that little cocksparrow, Governor Pickens of South Carolina, addressing the meeting, and strutting about like a rooster around a barn-yard coop, discussing the following resolution, which he was urging on the favorable consideration of the meeting:

"Resolved, That no member of Congress representing a Southern constituency shall again take his seat until a resolution is passed satisfactory to the South on the subject of glavery."

I listened to his language, and when he had finished, I of slavery." I listened to his language, and when he had finished, I obtained the floor, asking to be permitted to take part in the discussion. I determined at once to kill the treasonable plot hatched by John C. Calhoun, the Cataline of America, by asking questions. I said to Mr. Pickens, "What next do you propose we shall do? Are we to tell the people that Republicanism is a failure? If you are for that I am not. I came here to sustain and uphold American institutions; to defend the rights of the North as well as the South; to secure harmony and good fellowship between all sections of our common country." They dared not answer these questions. The southern temper had not then been gotten up. As my questions were not answered, I moved an adjournment of the caucus *sine die*. Mr. Craig, of Virginia, seconded the motion, and the company was broken up. We returned to the House, and Mr. Ingersoll, of Pennsylvania, a glorious patriot then as now, introduced a resolution which temporarily calmed the excitement.

Respecting this event, the National Intelli-

gencer of November 4, 1861, makes these remarks:

However busy Mr. Pickens may have been in the cauens after it met, the most active man in getting it up and pressing the southern members to go into it was Mr. R. B. Rhett, also a member from South Carolina. The occasion, or alleged cause of this withdrawal from the House into secret deliberation was an anti-slavery speech of Mr. Slade, of Vermont, which Mr. Rhett violently denounced, and proposed to the southern members to leave the House and go into conclave in one of the committee rooms, which they generally did, if not all of them. We are able to state, however, what may not have been known to Governor Thomas, that at least three besides himself of those who did attend it went there with a purpose very different from an intention to consent to any treasonable measure. These three men were Henry A. Wise, Balic Peyton, and Wm. Cost Johnson. Neither of them opened his lips in the caucus; they went to observe; and we can assure Governor Thomas that if Mr. Pickens or Mr. Calhoun (whom he names) or any ono else had presented a distinct proposition looking to disunion, or revolt, or secession, he would have witnessed a scene not soon to be forgotten. The three whom we have a scene not soon to be forgotten. The three whom we have have a brave as they were determined. Fortunately, perhaps, the man whom they went particularly to watch remained siltent and passive.

#### EARLY HOPES OF THE REBELS.\*

Mr. LAWRENCE M. KEITT, when declaiming in Charleston in November, 1860, in favor of the separate secession of that State, used this language, as reported in the Charleston *Mercury*:

But we have been threatened. Mr. Amos Kendall wrote a letter, in which he said to Col. Orr, that if the State went out, threa hundred thousand volunteers were ready to march against her. I know little about Kendall—and the less the better. He was under General Jackson; but for him the Federal treasury seemed to have a magnetic attraction. Juckson was a pure man, but he had too many around him who made fortunes far transcending their salaries. [Applause.] And this Amos Kendall had the same good fortune under Van Buren. He (Kendall) threatened us on the one side, and John Hickman on the other. John Hickman said, defiantly, that if we went out of the Union, eighteen millions of Northern men would bring us back. Let me tell you, there are a million of Democrats in the North, who, when the Black Republicans attempt to march upon the South, will be found a wall of fire in the front. [Cries of " that's sol?" and applause.]

Recently-found letters in Fredericksburg, Virginia, noticed editorially in *Harpers' Weekly* of May 28, 1864, show that the South calculated confidently upon the defection of large masses of men at the North. The *Weekly*, commenting on M. F. Maury's letters, says:

How far Maury and his fellow-conspirators were justified in their hopes of seducing New Jersey into the rebellion, may be gathered from the correspondence that took place in the spring of 1861 between Ex-Governor Price, of New Jersey, who was one of the representatives from that Stato in the Peace Congress, and L. W. Burnet, Esq., of Newark. Mr. Price, in answering the question what ought New Jersey to do, says: "I believe the Southern Confederation permanent. The proceeding has been taken with forethought and deliberation—it is no hurried impulse, but an irrevocable act, based upon the sacred, as was supposed, 'equality of the States;' and in my opinion every slave State will in a short period of time be found united in one confederacy. \* Before that event happens we cannot act, however much we may suffer in our material interests. It is in that contingency, then, that I answer the second part of your question—i what position for New Jersey will best accord with her interests, honor, and the patriotic inreason." Ex-Governor Price proceeds to say that he is confident the States of Pennsylvaniat and New York will "choose also to cast their lot with the South," and after them the western and northwestern States.

<sup>+</sup> January 16, 1861—A meeting of Democrats was held in National Hall, Philadelphia, Charles Macalester presiding, at which Robert P. Kane offered this, among other resolutions which were put to the meeting, and declared adopted, and which, read in the light of this revelation, appear to

390

<sup>\*</sup> See page 20.

#### LETTER OF EX-PRESIDENT FRANKLIN PIERCE TO JEFFERSON DAVIS, OF JANUARY 6, 1860, FOUND IN DAVIS'S MISSISSIPPI HOME, WHEN TAKEN BY OUR TROOPS:

CLARENDON HOTEL, January 6, 1860.

MY DEAR FRIEND: I wrote you an unsatisfactory note a day or two since. I have just had a pleasant interview with Mr. Shepley, whose courage and fidelity are equal to his learning and talents. He says he would rather fight the fearming and talents. He says he would rather light the hattle with you as the standard-bearer in 1800 than under the anspices of any other leader. The feeling and judgment of Mr. S. in this relation is, I am confident, rapidly gaining ground in Now England. Our people are looking for "the coming man," one who is raised by all the elements of his coming man," one who is raised by an ine treathed by pol-iticians, a man really fitted for this exigency by his ability, courage, broad statesmanship, and pariotism. Colonel Seymonr (Thos II.) arrived here this morning, and expressed his views in this relation in almost the identical language used by Mr. Shepley. It is true that, in the present state of things at Washington and throughout the country, no man can predict what changes two or three months may bring can predict what changes two or three months may bring forth. Let me suggest that, in the running debates in Con-gress, full justice seems to me not to have been done to the Democracy of the North. I do not believe that our friends at the South have any just idea of the state of feeling, hur-rying at this moment to the pitch of intense exasperation, hetween those who respect their political obligations and those who have apparently no impelling power but that which fanatical passion on the subject of domestic slavery immarts. Without discussing the onestion of right, of abimparts. Without discussing the question of right, of ab-stract power to secede, I have never believed that actual disruption of the Union can occur without blood; and if, through the madness of northern abolitionism, that dire through the madness of northern abolitionism, that dire calamity must come, the fighting will not be along Mason's and Dixon's line merely. It [will] he within our own hor-ders, in our own streets, between the two classes of citizens to whon I have referred. Those who defy law and scout constitutional obligations will, if we ever reach the arbitra-ment of arms, find occupation enough at home. Nothing but the state of Mrs. Pierce's health would induce me to leave the coantry now, although it is quite likely that my presence at home would be of little service. Thave tried to presence at home would be of little service. presence at home would be of little service. I have tried to impress upon our people, especially in New Hampshire and Connecticut, where the only elections are to take place during the coming spring, that while our Union meetings are all in the right direction, and well enough for the pres-ent, they will not be worth the paper upon which their resolutions are written unless we can overthrow political abolitionism at the polls and repeat the unconstitutional and obnoxious laws which, in the cause of "personal lib-erty," have been placed upon our statute-books. I shall look with deep interest, and not without home for a decided I have tried to look with deep interest, and not without hope, for a decided change in this relation. Ever and truly, your friend,

FRANKLIN PIERCE.

Hon. JEFF. DAVIS, Washington, D. C.

## The Disunion Programme.

From the National Intelligencer of Friday, January 11, 1861:

The subjoined communication, disclosing the designs of those who have undertaken to lead the movement now threatening a permanent dissolution of the Union, comes to us from a distinguished citizen of the South,\* who for-

disclose a plan of which ex-Governor Price was likely aware:

Tweifth, That in the deliherate judgment of the Democracy of Philadelphia, and, so far as we know it, of Penasylrania, the dissolution of the Union by the separation of the whole Sonth, a result we shall most sincerely lanent, may release this Commonwealth to a large extent from the bonds which now connect her with the Confederacy, except so far as for temporary convenience she chooses to subanit to them, and would authorize and require her citizens, through a convention, to be assembled for that purpose, to determine with whom her lot should be cast, whether with the North and the East, whose fanaticism has precipitated this misery upon us, or with our brethren of the South, whose wrongs we feel as our own; or whether Pennsylvania should stand by herself, as a distinct community, ready when occasion offers to bind together the broken Union, and resume her place of loyalty and devotion.

\* Understood to be Hon, Lemuel D. Evans, Representative from Texas in the 34th Congress, from March 4, 1855, 20 March 3, 1857.

merly represented his State with great distinction in the popular lwanch of Congress. Temporarily sojourning in this city he has become anthentically informed of the facts recited in the subjoined letter, which he communicates to us under a sense of duty, and for the accuracy of which he makes himself responsible. Nothing but assurances coming from such an intelligent, reliable source could induce us a accept the authenticity of these startling statements, which so deeply concern not only the welfare but the honor of the Southern people. To them we submit, without present comment, the programme to which they are expected to yield their implicit adhesion, without any semples of conscience as without any regard to their own safety.

#### WASHINGTON, January 9, 1861.

I charge that on last Saturday night a caucus was held in this city by the Southern Secession Senators from Florida, Georgia, Alahama, Mississippi, Louisiana, Arkansas, and Texas. It was then and there resolved in effect to assumd to themselves the political power of the South, and, to control all political and military operations for the present, they telegraphed to complete the plan of seizing forts, arsenals, and custom-houses, and advised the conventions now in session, and soon to assemble, to pass ordinances for immediate secession; but, in order to thwart any operations of the Government here, the Conventions of the Seceding States are to retain their representations in the Senate and the House.

the flouse. They ilso advised, ordered, or directed the assembling of a Convention of delegates from the secoding States at Montgomery on the 13th of February. This can of course only be done by the revolutionary Conventions usurping the powers of the people and sending delegates over whom they will lose all control in the establishment of a Provional Government, which is the plan of the dictators.

This cancus also resolved to take the most effectual means to dragoon the Legislatures of Tennessee, Kentucky, Missouri, Arkansas, Texas, and Virginia into following the seceding States. Maryland is also to be influenced by such appeals to popular passion as have led to the revolutionary steps which promise a conflict with the State and Federal Governments in Texas.

They have possessed themselves of all the avenues of information in the South—the telegraph, the press, and the general control of the postmasters. They also confidently rely upon defections in the army and navy. The spectacle here presented is startling to contemplate. Senators entrusted with the representative sovereignty of the spectrum arm in the unrest the Constitution of the

The spectacle here presented is startling to contemplate. Senators entrusted with the representative sovereignty of the States, and sworn to support the Constitution of the United States, while yet acting as the privy councillors of the President, and anxiously looked to by their constituents to effect some practical plan of adjustment, deliberately conceive a conspiracy for the overthrow of the Government through the military organizations, the dangerous secret order, the Knights of the Golden Circle, "Committees of Safety," Southern Leagnes, and other agencies at their command; they have instituted as thorough a military and civil despotism as ever cursed a maddened country.

It is not difficult to foresee the form of government which a convention thus hurriedly thrown together at Montgomery will irrevocably fasten upon a deluded and unsuspecting people. It must essentially be "a monarchy founded upon military principles," or it cannot endure. Those who usurp power never fail to forge strong chains.

upon military principles," or it cannot endure. Those who usurp power never fail to forge strong chains. It may be too late to sound the alarm. Nothing may be able to arrest the action of revolutionary tribunals whose decrees are principally in "secret sessions." But I call upon the people to pause and reflect before they are forced to surrender every principle of liberty, or to fight those who are becoming their masters rather than their servants. EATON.

As confirming the intelligence furnished by our informant we may cite the following extract from the Washington correspondence of yesterday's Baltimore San: "The leaders of the Southern movement are consulting

"The leaders of the Southern movement are consulting as to the best mode of consolidating their interests into a Confederacy under a *Provisional Covernment*. The plan is to make Senator Hunter, of Virginia, Provisional President, and Jefferson Davis Commander-in-Chief of the army of defence. Mr. Hunter possesses in a more eminent degree the philosophical characteristics of Jefferson than any other statesman now living. Colonel Davis is a graduate of West Point, was distinguished for gallentry at Buena Vista, and served as Secretary of War mider President Pierce, and is not second to General Scott in military science or courage."

The Charleston *Mercury* of January 7, 1860, published the following telegraphic dispatch :

#### [From our own Correspondent.]

WASHINGTON, Jan. 6.-The Senators from those of the Southern States which have called Conventions of their resolutions:

" Resolved, That we recommend to our respective States immediate secession.

"Resolved, That we recommend the holding of a General Convention of the said States, to be holden in the city of Montgomery, Alabama, at some period not later than the 5th day of February, 1861."

These resolutions were telegraphed this evening to the Conventions of Alabama, Mississippi, and Florida. A third resolution is also known to have been adopted, but it is of a confidential character, not to be divulged at present. There was a good deal of discussion in the cancus on the question of whether the secoding States ought to continue their del-egations in Congress till the 4th of March, to prevent unreactions in congress in the 4th of March, to prevent un-friendly legislation, or whether the Representatives of the secoling States should all resign together, and leave a clear field to the Opposition to pass such bills, looking to coer-cion, as they may see fit. It is believed that the opinion that they should remain prevailed.

#### CERTAIN "SECRET" DOCUMENTS.

## LETTER FROM U. S. SENATOR YULEE OF FLORIDA.

#### WASHINGTON, Jan. 7, 1861.

MY DEAR SIR: On the other side is a copy of resolutions adopted at a consultation of the Senators from the seceding States-in which Georgia, Alabama, Louisiana, Arkansas,

States—in which deorgia, Alabama, Joursina, Alamsao, Texas, Mississipi, and Florida were present. The idea of the meeting was that the States should go out at once, and provide for the early organization of Confederate Government, not later than 15th February. This time is allowed to enable Louisiana and Texas to participate. It seemed to be the opinion that if we left here, force, loan, and volunteer bills might be passed, which would put Mr. LINCOLN in immediate condition for hostilities; whereas, by remaining in our places until the 4th of March, it is thought we can keep the hands of Mr. Buchanan tied, and disable the Republicans from effecting any legislation which will dewelte the head of the incomposed the interview.

assoute the Republicans from Epicating any repristance which we will strengthen the hands of the incoming Administration. The resolutions will be sent by the delegation to the President of the Convention. I have not been able to find Mr. Mallory this morning. Hawkins\* is in Connecticut. I have therefore thought it best to send you this copy of the resolutions. resolutions.

In haste, yours truly, D. L. YULEE.

JOSEPH FINEGAN, Esq., "Sovereignty Convention," Tallahasee, Fla.

The following were the resolutions referred to:

Resolved 1. That in our opinion each of the Southern States should, as soon as may be, secode from the Union. Resolved 2. That provision should be made for a conven-tion to organize a Confederacy of the secoding States, the convention to meet not later than the 15th of February, at the clin of Monteconvery in the State of Alduma. the city of Montgomery, in the State of Alabama.

Resolved, That in view of the hostile legislation that is threatened against the seceding States, and which may be consummated before the 4th of March, we ask instructions whether the delegations are to remain in Congress until that date for the purpose of defeating such legislation. *Resolved*, That a committee be and are hereby appointed.

consisting of Messrs. Davis, Slidell, and Mallory, to carry out the objects of this meeting.

The preceding letter was found in Fernandina, Florida, upon the capture of that city in the winter of 1862, and the orginal letter was forwarded to the editor of the N.Y. Times by its correspondent who accompanied the expe-The Times of Saturday, March 15, dition. 1862, comments on this development:

The telegraphic columns of the Times of January 7, 1861, contained the following Washington despatch : "The South-ern Senators last night (Jan. 5) held a conference, and tel-egraphed to the conventions of their respective States to advise immediate secession." Now, the present letter is a second by W. Yulee who was avecand at this "conputte advise initiation secession." Now, the present actions in a report by Mr. Yulee, who was present at this "consulta-tion," as he calls it, of the resolutions adopted on this oc-casion, transmitted to the said Finegan, who, by the way, was a member of the "Sovereign Convention" of Fiorida,

then sitting in the town of Tallahassee. It will thus be seen that this remarkable letter, which breathes throughout the spirit of the conspirator, in reality lets us into one of the most important of the numerous secret conclaves which the plotters of treason then held in the capital. It was then, as it appears, that they deter-mined to strike the blow and precipitate their States into

\* The Representative from Florida.

people met in caucus last night, and adopted the following | secession. But at the same time they resolved that it would secession. But at the same time they resolved that it would be imprudent for them openly to withdraw, as in that case Congress might pass "force, loan, and volunteer bills, which would put Mr. Lincoln in immediate condition for hostilitics," No, no 1 that would not do. (So much patri-otic virtue they half suspected, half feared was left in the country.) On the contrary, "by remaining in our places until the 4th of March it is thought we can keep the hands of Mr. Buchanan tied, and disable the Republicans from effecting any legislation which will strengthee the hands of the incoming Administration." All what a tragic back-ground, full of things unatterable, is there there ! It appears, however, that events were faster than they,

It appears, however, that events were faster than they, and instead of being able to retain their seats up to the 4th of March, they were able to remain but a very few weeks. Mr. Davis withdrew on the 21st of January—just a fort-night after this "consultation." But for the rest, mark how faithfully the programme here drawn up by this knot of traitors in secret session was realized. Each of the named States represented by this cabal did, "as soon as may be, secede from the Union"—the Mississippi Convention passing its ordinance on the heels of the receipt of these resolutions, on the 9th of January; Florida and Alabama on the 11th; Louisiana on the 26th, and Textus on the 1st of February; while the "organization of the Confede-rate Government" took place at the very time apointed, Davis being inaugurated on the 15th of February.

And here is another plot of the traitors brought to light. These very men, on withdrawing from the Senate, urged These very men, on within awing from the schutc, firged that they were doing so in obscilence to the command of their respective States. As Mr. Davis put it, in his parting speech, "the ordinance of secession having parsed the Convention of his State, he felt obliged to obey the sum-mons, and retire from all official connection with the Fed-eral Government." This letter of Mr. Yulee's clearly re-yeals that they had themselves pushed their State Conven-tions to the adoption of the arey measure which they had tions to the adoption of the very measure which they had which they could not disobey. It is thus that treason did its work.

Mr. James L. Pugh, member of Congress from Alabama, in a letter, Nov. 24, 1860-"made public his reasons for going to Washington," and taking his seat in Congress. He says: "The sole object of my visit is to promote the cause of secession."

## Douglas's Farewell Words.

#### IN CHICAGO.

The election of Mr. Lincoln is a mere pretext. The pres-ent secession movement is the result of an enormous con-spiracy formed more than a year since—formed by leaders in the Scuthern Confederacy more than twelve months ago. in the Scuthern Confederacy more than twelve months ago. They use the slavery question as a means to aid the accom-plishment of their ends. They desired the election of a northern candidate by a sectional vote, in order to show that the two sections cannot live together. When the history of the two years from the Lecompton question down to the Presidential election shall be written, it will be shown that the scheme was deliberately made to break up this Union. They desired a northern Republican to be elected by a purely northern vote, and theu assign this fact as a reason why the sections cannot live together. If the dismion can-

why the sections cannot live together. If the distinion can-didate in the late Presidential contest had carried the united South their scheme was the northern candidate suc-cessful, to seize the Capital last spring, and by a united South and divided North, hold it. Their scheme was de-feated, in the defeat of the disunion candidates in several of the southern States.

But this is no time for a detail of causes. The conspiracy is now known ; armies have been raised, war is levied to accomplish it. There are only two sides to the question. Every man must be for the United States or against it. There can be no neutrals in this war; only patriots or traitors! (Cheer after cheer.)

## The Conspiracy in Maryland.

On page 152, the arrest of Marshal Kane\*

\* Marshal Kane, on his way to burn the bridges on the railroads to Philadelphia and to Harrisburg, sent this telegram for troops:

MARSHAL KANE'S DESPATCH TO BRADLEY T. JOHNSON, OF FREDERICK.

BALTIMORE, April 19, 1861.

Thank you for your offer ; bring your men in by the first train, and we will arrange with the railroad afterwards.

and the Board of Police of Baltimore in June, 1861, by General Banks, is noticed. After their arrest, the minutes of their proceedings during the "reign of terror" in Baltimore were found, from which these interesting extracts are taken:

EXTRACTS FROM THE "MINUTES" OF THE POLICE COMMISSION-ERS OF BALTIMORE, AND FROM THEIR LETTER BOOK; FOUND IN THEIR OFFICE AFTER THEIR ARREST BY GENERAL BANKS.

1861, April 19th, 20th, and 21st-No entry.

April 22d—After debate, Resolved, That notice be imme-diately given of election for Legislature on Wednesday 24th April, and the sheriff be requested to unite in giving the

April, and the sherin be requested to ante in giving the necessary call therefor. The Board declared itself to be in permanence. D. J. Foley & Bros' powder purchased, and the disposition of the same and all other powder to be purchased committed to Col. Isaac Trimble.<sup>#</sup> Henry Thompson, Esq., appointed Quartermaster General, to act under the Mayor,<sup>+</sup> Order passed to prevent the transportation of provisions

without special permission. Colonel Trimble appointed to enforce the order.

Charles Pitts ‡ appeared and offered 200 to 300 negroes to perform any service which the authorities may call on them to do. The Mayor returns thanks, and assures them they shall be called on when any occasion arises when their services can be availed of.

Coleman Yellott & authorized to charter a steamer to sum-mon the members of the State Legislature.

Unnecessary parades forbidden in the streets.

Extract from a note from Col. Trimble to Howard, Presi-dent of the Board, found among the papers :

"The display of military will be a sorry one as to the strength of the military of the city, and calculated to dishearten our own citizens, and if represented abroad will rather invite and encourage attempts from the north to defy us and pass through the city, whereas without this display many will think that the military force of the city is much stronger than it really is."

Order passed authorizing Col. Trimble to permit steam-boats to leave for the eastern shore, to and below the Sassafras river, upon condition that in going and returning they shall not stop at Annapolis.

April 23. U. S. SHIP ALLEGHANY, April 23. U. S. SHIP ALLEGHANY, BAITINGRE, April 23, 1861. CHARLES HOWARD, Esq., President of the Board of Police. SIR: Having occasion to employ a steamtug in the service of the United States, I have to request that you will author-ize me to use one this day in the harbor of Baltimore and the adjacent waters.

I am, respectfully, WM. H. HUNTER, Com. U. S. N.

Respectfully declined.

Communication from Col. Trimble in regard to the re-moval of the Alleghany. Answer given through Charles Wethered, that the matter was attended to, and *the removal* forbidden.

Mr. Zenas Barnum called in regard to repairing the telegraph wires on the road to Philadelphia; no action determined on. Mr. Barnum was informed that no communica-tions can pass over the wires for Washington, whether for the War Department or citizens, without being subjected to the inspection of the police board.

Commissary Lee applied for permission to convey pro-visions into Fort McHenry. He was informed of the impos-sibility of ensuring him protection in carrying provisions to the fort during the present excitement of public feeling. Application received from Mr. Clarke, Superintendent of the continuous for the public of the present excitement of public the continue Constant Pailword for public in the scheduler.

the Northern Central Railroad, for permission to rebuild the bridges [which had been destroyed by Kane on the night of the 19th] at Melville and the Relay House. This was refused.

Streets red with Maryland blood! Send expresses over the mountains and valleys of Maryland and Virginia for the riflemen to come without delay. Fresh hordes will he down on us to-morrow, (the 20th.) We will fight them or die. GEO. P. KANE.

This was posted in Frederick, with a placard as follows, signed by Bradley T. Johnson: All men who will go with me will report themselves as soon

as possible; providing themselves with such arms and ac-coutrements as they can. Double-barrelled shot guns and buck-shot are efficient. They will assemble after reporting themselves at 101/2 o'clock, so as to go down in the 111/2 train.

\*Now Major General in rebel service, and a prisoner in our hands, captured at Gettysburg.

† The Mayor was one of the Board.

‡ Elected to the Legislature at the election of 24th April. ¿ Member of the State Senate.

Application by Messrs. Tucker [British subjects] to complete the loading of the Queen Victoria, [British.] In rcply were informed that instructions would be given to Col. Trimble to permit the Queen Victoria to depart with her present loading, but that no other articles prolibited to be removed from the city can be permitted to be shipped on board the vessel. *American* vessels party laden shall not depart without further and distinct action being taken by the board.

The following were loose papers accompanying the "minutes" but not recorded:

1. Despatch by Morse's line, dated

"HARPER'S FERRY, 1861. Received Baltimore 22, 1861." To POLICE COMMISSIONERS :

I want to communicate.

What have you to say? KENTON HARPER, [Rebel] Commanding.

2. Baltimore and Ohio Railroad Telegram, dated HARPER'S FERRY, April 22, 1861. Received 11 o'clock, 8 nin., A. M. To W. II. GATCHELL, *Police Commissioner* : Very satisfactory interview here with General Harper

C. J. M. GWINN,

3. Memorandum in pencil-omited in the "minutes;" April 23, 1861.

"Telegraphed General Harper, Harper's Ferry, to the effect, that the town is quiet, expressing thanks for his communication, and promising to let him know when any exigency requiring it shall arise; mentioning that Gwinn had been seen by us."

Gwinn reported "six thousand men ready to come down." 4. Another memorandum:

"Gwinn asserts that six thousand troops are at Harper's Ferry.

A PRT. 25 \* " All police officers and others in the employ of the Board, and all other parties whatsoever, are requested to offer no obstruction to the running of the trains." (*i. e.*, to Washington.)

In a letter to John Garrett, Esq., Col. Trimble directed to allow and grant facility for the transportation of 40 kegs of powder to be used by the Baltimore and Ohio Railroad Company in Virginia.

Restrictions on the export of provisions, breadstuffs, and bituminous coal, removed.

Propositions to repair the telegraphic lines to Cockeys-

Appointed is to the rest and telegraphic must be occasioned. Appointment made with Col. Trimble and Gen. Thompson by the Mayor and Board, for 6 p. m., to consider and determine certain matters in reference to the disbursement of moneys under the ordinance, appropriating \$500,000.

Col. Huger furnished plans and specifications for a bombproof.

Letter from General Stewart + found among the papers : April 22, 1861.

"MY DEAR HOWARD : I will endeavor to put on duty the same company of cavalry that was ordered out last night. ‡ I know not what to think of the rumors from Annapolis ; but if the Massachusetts troops are on the march from that place to Washington, I shall be in motion very early to-morrow morning to pay my respects to them, of course making arrangements for an adequate force being here in my absence.

LETTER FROM THE SECRETARY OF THE MAYOR TO THE PRESIDENT OF POLICE BOARD.

MAYOR'S OFFICE, April 23, 1861.

CHARLES HOWARD, Esq. : CHARLES HOWARD, ESQ.: DEAR SIR: A messenger from Virginia called to inform the Mayor that Senator Mason will be in the city either on to-morrow evening or the next (azy, and wishes an inter-yiew with Governor Pratt, Hen. Robert McLean, and J. Pratt, Hon. ... Respectfully, D. H. BLANCHARD, Secretar Mason Campbell.

Secretary.

April 26-Colonel Kane reports that the powder purchased by Colonel Trimble's orders, and stored in a church in the west end, has been ordered to be transferred to Vicker's house.

Negroes said to be offering northern notes (1) General

\* The day after the election for members of House of Delegates, when but 9,000 votes were cast, of the 30,000 in Baltimore.

† Stewart was Major General of the 5th Division Maryland militia; is now in the rebel service and our prisoner, hav-ing been captured in one of the battles of the Wilderness, ‡ To take Fort McHenry.

Ing reports a revolving cannon\* in William Wilkens & Co's. warehouse. But Colonel Iluger and Ross Winans do not approve of such butteries.

Application made for permission to repair telegraph lines to Havre de Grace. Granted; it being understood that the

to Havre de Grace. Granicet, it being understood that the board shall have cognizance of all communications made by the American Telegraph Company. (Letter Book, *folio* 97.) Certain directors of the Philadelphia, Wilmington, and Baltimore railroad [O'Donnel, Pratt, Cohen, and T. Donel-son] ask for the privilege of reconstructing the bridges and marking the order way their way folgered by Marshal repairing the rails upon their road, [destroyed by Marshal Kane.] They assumed that a promise could be had by them from the Ger eral Government not to ask for the passage of troops over the road. Answer given that they should first learn what time would be required for repairing the road, and what as arance they could get from the Government that troops shall not seek that means of transportation, and then that the application for permission to rebuild the road shall be renewed to the board.

Prohibit on to remove flour and breadstuffs re-enacted. April 27 1861—Full Board and Mayor. Resignation of David Daneker and William T. Butler, of

the police force, received.

#### DANEKER'S LETTER.

TO THE BUARD OF POLICE:

GENTS I hereby tender my resignation as a member of the police force of Baltimore. As an American citizen I cannot condescend to pull down the American flag.

DAVID DANEKER, 20th April.

Colonel Trimble instructed to allow shipments of breadstuffs, d.c., in limited quantities, within the state; must use his discretion: "Keep a list of parties shipping the ar-ticles, and quantities, and make daily reports."

Gene. al Stewart appeared and stated that he had infor-mation of 2,000 stand of arms having arrived at Camden Station, (Baltimore and Ohio railroad,) which he claimed as officer of the State.

Reply, (Letter Book, page 111:)

"A gentleman representing the house of W. T. Walters & Co., has just called. Their house has bill of lading for the arms, and desires to hand them over to the police depot solely for safe-keeping.

April 28th-General Stewart notifies the Board that 2.000 guns were yesterday morning at Harper's Ferry, awaiting his orders, and that he had given directions for them to be forwarded to his orders

## LETTER FROM GENERAL STEWART TO THE BOARD.

HEADQUARTERS, 1st LIGHT DIVISION, M. V. BALTIMORE, 2d May, 1861.

To the BOARD OF POLICE OF THE CITY OF BALTIMORE :

GENTS: I have the pleasure to lay before you the official dispatches of Governor Letcher, of 24th April, addressed to me, and several other communications between myself and Major General Kenton Harper, commanding at Harper's Ferry, as well as Brightier General Cocke, commanding at Alexandria, all showing that the 5,000 fint-lock guns fur-nicled by the State of Virginia are for the use of Maryland, and are destined to be delivered to my order.

Understanding that 2,030 of these arms, which arrived two or three days ago from Harper's **T**erry, consigned to Mr. W. T. Walters, (but contrary to the instructions of Major General Harper, as he assures me in a dispatch.) are now in a warehouse at the corner of Second and Gay streets, occupled by your Board or by the city authorities, I, as the officer of the State of Maryland, commanding here, and agent of the State, to whom the arms were destined, require that of the State, to whom the arms were destined, require time the whole be d-livered to my order, it being understood that I assume the responsibility of receiving those arms on account of the State of Maryland. You are aware that upon special application to me on behalf of the companies of Calvert county, I agreed to delivery of 120 of those arms."

#### Extract from the reply of the Commissioners (C. Howard) to General Stewart-same date.

"And you having also understood that a gentleman now in Frederick has full power to control or alter the destination of them, we were anwilling to exercise any authority over them. I showed you a copy of my letter to that gen-tleman from whom I have yet no reply."

\* This was afterwards captured en route for Harper's Ferry.

+ The arguarat of the rebels then was that otherwise Government would suddenly buy up and remove all the supplies.

1 It had been determined to suppress the American flag and the minitary had been put in array to put this through vide hers from Stewart to Howard, 26th April, and th replice of the Beard in Letter Book.

Letter referred to, Letter Book, page 116, to Senator Mason, dated April 30, 1861.

MY DEAR SIR: Since I had the pleasure of seeing you I have had a good deal of annoyance in consequence of the irregular manner in which some of our townshen obtained in your State some arms. I do not question their motives, but it was unfortunate that we had no information of what had been done by them until we learned it from you. Can you inform me whether these arms mentioned by General Harper are a part of the original quantity mentioned by you, and of which I understand you have full power to control the destination?

If so, do you desire General Stewart, as a military officer of the State, to receive them for its use? CHARLES HOWARD.

(Signed) GOVERNOR LETCHER'S LETTER TO GENERAL STEWART.

DEAR SIR : I called this morning to see you, having received a dispatch from J. S. Barbour, Esq., Alexandria, giving information respecting matters in Baltimore. I have issued an order to General Harper to send 1,000 stand of arms to General Stewart. Stirring times in your State. Truly, JOHN LETCHER.

#### THE ORDINANCE APPROPRIATING \$500,000.

This ordinance (No. 22) was approved April 20, 1861, and a portion of the money expended. The following report, made April 4th, 1862, shows how :

The Joint Select Committee, to whom was referred the resolution for the appointment of a committee " to examine and dispose of all papers and vouchers relating to the expenditure of money under the provisions of the ordinance for the preservation of the peace of the city, approved April 20th, 1861, to inquire and report what has become of the articles then purchased and in whose possession they now are, and also what disposition shall be made of the bal-ance of the funds now in the city treasury," have discharged the duty imposed upon them, and submit to the Councils the following report, with accompanying resolutions: The duties imposed upon the committee were three-fold;

first, to examine and dispose of all papers and vouchers re-lating to the expenditure of money, under the provisions of the ordinance specified; secondly, to inquire and report what has become of the articles thus purchased, and in whose possession they now are; and thirdly, to decide what disposition shall be made of the balance of the funds now in the city treasury.

On the first head, your committee have to report that no "papers" have come under their cognizance, except a note addressed by Col. Henry A. Thompson to the Mayor, which accompanied a statement of his agency, in the capacity of Quartermaster General, in the disbursement of \$38,000 of Guarterinater othera, in the unstanding of 00,000 to the money in question, and in which he claims to have saved a considerable sum for the city treasury, by efforts to prevent lavish expenditure, and by annulling numerous contracts and returning many articles already delivered.

Your committee regard it as a cause of profound regret, that such an agency had not been earlier in action; but it is due to the gentleman referred to to say, that he appears to have performed the difficult duties confided to him, with fidelity and discretion. Acknowledgments are also due to Col. Thompson, for a courteous note, tendering any assistance in his power, in the examination of the "vouch ers," so far as his agency was concerned; but no occasion has arisen for making use of the proffered aid. The "vouchers," as placed in the hands of your commit-

tee by the city Register, have been carefully examined, and found to be in due form, authenticated by the parties having control of affairs at that period, and accounting for the expenditure of the gross amount reported, allowing for file expenditure of the gross amount reported, allowing for interest and the balance on hand. Each account has been singly inspected, the objects of expenditure have been classified, and present the results stated below.

classified, and present the results stated below. Having served the purposes of your committee in afford-ing the points of information it was necessary to obtain, no other disposition of them seems necessary than to re-store them to the custody of the city officer, in whose charge they properly belong, as vouchers for disbursements under the avilances under the ordinance.

The labors of your committee under the second head of instructions, "to inquire and report what has become of the articles then purchased, and in whose possession they now are," have resulted in the following classified summary of expenditures for different objects:

#### SUMMARY.

	Arms, ammunition, &c	\$24,174	74	
or.	Blankets, mattresses, &c	2,825	57	
h :	Surgical instruments and medicines	99	37	
he	Marine and navy (embracing hire and altera-			
	tions of steamers, and wages of men)	5,461	31	

Carpenters, bricklayers, materials, &c	\$2,568 05
Rations	9,914 39
Pay of officers and men	7,736 30
Horse and hack hire, and hauling arms	3,472 86
Advertising, stationery, &c	234 78
Rent of armories and repairs	1,748 02
Workmen on parks, by resolution, approved	,
June 11, 1861	8,508 19
Poor Association, for use of indigent widews,	
sewing women, &c	2.000 00
Interest	5,775 56
Cash in Bank	24,576 86

Total...... \$99,096 00

From the above summary it will be seen that the expend-itures for "Arms and ammunition" reached the large amount of nearly \$25,000. The articles purchased include fixtures for cannon, carbines, rifles, muskets, pistols, swords, spears, drums, canister and grape shot, bullets, lead, pow-der, cartridges, caps and other kindred materials, the pre-cise number and quantities of which, as nearly as could be constrained and spears of the start of the star ascertained, are given in the two following schedules:

1,217 carbines, from F. W. Bennett.

407 Hall's patent rifles, from Denson & Buck. 80 carbines, from Denson & Buck.

- 6 cutlasses, from Denson & Duck.
  12 gun carriages, from A. & W. Denmead & Son.
  2,285 iron spears, from A. & W. Denmead & Son.
  2 muskets, from Levi Cromwell. 16 pistols, from Wm. Harris.
- 20 pikes from Hayward, Bartlett & Co. 23 guns, from Hayward, Bartlett & Co. 23 guns, from Merrill, Thomas & Co. 12 flasks, from Merrill, Thomas & Co. 4 ship pistols, from Levi Cromwell.

- 1 sword, from Canfield, Bro. & Co. 414 pikes, from George Page & Co. 7 revolvers, from C. Schumaker. 3 drums, from Eisenbrandt.

AMMUNITION.

- 1 keg rife powler, from F. Devlin.
   2 bags balls, from F. Devlin.
   2,000 caps, from F. Devlin.
   4,958 lbs. lcad, from Baltimore Water Board.
- 200 blank cartridges, from Green & Yee.
  - 2 boxes caps, from Green & Yoe.
- 1 bag shot, from Green & Yoe. 113 yards flannel, for cartridges, from -10,000 ball cartridges, from, Wm. Harris.
- 10,000 ball cartridges, from, Wm. Harris. 10,860 do do do 800 do do Hoffman. do. do Poultney & Trimble. 429 cylinders for 6-pound guns, from 0. II. Cromwell. 86 lbs. antimony, for balls, from Regester & Wabb. 102 lbs. tin, for balls, do do 102,000 caps, from Merrill, Thomas & Co. 13 bags and 14 lbs. balls, from Merchants' Shot Tower. 3031 cartridges, from Merrill, Thomas & Co.

- 8 boxes cartridges, from Merrill, Thomas & Co.
- 30,000 caps,
- 0,000 caps, do additional definition and a second de additional de addit
- de 473 cylinders, do
- 54 lbs. match rope do do

The expenditures for "Blankets, Mattresses, &c.," were \$2,825 57; and the articles purchased were as follows:

BLANKETS, MATTRESSES, ETC.

- 445 blankets, from Whitely, Stone & Co., and others.

- 445 bilankets, from (mitre), store a do, do 40s frans, do do do do 425 fiannel shirts, from John H. Rea, and others. 225 mattresses, from C. S. Frey & Co., A. Pollack, and others. 240 billows, do do do do 240 pillows, do do do 400 caps, from White & Rosenberg. 1 desk, 10 chairs, 1 washstand, by S. S. Lee.

For "Surgical Instruments, &c.," the expenditures, con-sidering the fermidable nature of the preparations, offensive and defensive, were moderate, reaching only the sum of \$99 37; and of this amount a portion is for damage to in-struments returned. The list of these articles is here given:

SURGICAL INSTRUMENTS AND MEDICINES.

3 tourniquets, ordered by Dr. Robinson. do

6 ternaculums, do 2 bullet forceps,

- Sponges, Wadding, Muslin, Spirits, &c.

do Having thus ascertained what articles had been pur-chased, your committee proceeded "to inquire what has

become of them, and in whose possession they now are." This investigation was attended with considerable difficulty, This investigation was attended with considerable difficulty, owing to the various changes which have taken place since the period of the purchases; the abdication of the former Police Commissioners, the absence of the Mayor, the substi-tution of the United States Provest Marshal and his police force, and the seizure, by way of precaution, by the United States, of arms and military materials stored in different parts of the city. Many articles and equipments have un-doubtedly disappeared; but your committee are enabled to present the following list of arms and other articles, fur-nished by Mr. James L. McPhail, Deputy Provest Marshal, which will account for a portion of the articles. Identifi-cation of those purchased was, of course, impossible: 28 rifles, found at the old City Hall.

old City Hall. 38 ns.

99	rines, i	iouna a	u une ora
3	single	barrel	shot gun
1	double	do	do
27	single	do	do

- 2 double do dø
- 2 rifles.

49 Hall's carbines.

7 U. S. do

2 Hall's rifles.

1 percussion musket.

1 ship gun.

2 six pound field pieces, mounted. 2 field pieces, complete. 20 rounds of canister shot.

Of the above articles, (your committee are informed by Mr. McPhail.) the beds and bedding were distributed to the several station-houses, they having been found, on taking possession, to be generally destitute of such articles, and those found being in bad repair and very dirty. Desides the bedding thus appropriated, a considerable number of mattresses, some tin ware, &c., were transferred, with the approbation of the mayor, to the warden of the jail, for use at that institution, where they were much needed. A por-tion of the arms purchased by the parties who had control of affairs during the period of the "crisis" are still at Fort McHenry, where they were placed after seizure by the Uni-ted States authorities. According to a return made to the mayor on the 12th of August, 1801, by Samuel W. Bowen, Captain of the Middle District Station, the following arms, taken by order of General Dutler from the corner of Gay and Second streets, were then at Fort McHenry: Of the above articles, (your committee are informed by and Second streets, were then at Fort McHenry:

	boxes,	marke	d "Va. m	uskets,"	20 in each	bex	1,160
15	66	<b>66</b>	" A	rmory,	do.		300
40	Hall's	carbin	$es, 20 in \epsilon$	each box	and 141 loc	se	941
	Hall's			44			
					s		46
38	old mu	iskets,	flint locks	· · · · · · · · · · · · · · · · · · ·			38
23	66	44 · ·	percussion	<b>n</b>			23
27	new	**	flint				27
50	Yager	rifles.		· • • • • • • • • • • • • • • • • • • •			50
	0					-	

2,919

1 box accoutrements.

115 boxes pikes.

Of the above, issues to a considerable extent have been made by the United States authorities, and some have been returned to Denson & Buck. Your committee are obliged to Deputy Marshal McPail for the following statement of arms at present to be found at Fort McHenry:

60 boxes of pikes, containing 60 each.

11 carbines. 300 carbines.

5 guns, 7 carriages, 13 rammers.

It is understood that these articles, and such others be-longing to the city as may be found in possession, will be restored by the Government of the United States as soon as their restoration is deemed compatible with a proper regard for its own protection.

A few words more are necessary, in explanation of the remaining items of the summary. Of the amount charged to the "Marine and Navy," the

Of the amount charged to the "Marine and Navy," the principal portion is made up of sums expended on the hire and alterations of steam-tugs Horcules and Tiger, charges for use of the Potomac, George's Creek, and other steamers, amounting to \$1,500 53, and \$55 pail for the purchase of a boat and cars. The balance was principally expended in the payment of officers and "harbor police." The only ar-ticles known to be remaining from this department are a

- 9 dez. bexes Hicks's hat caps.
- 26 do grape do do 11/2 tons de ao 11/2 tons de ao 11 lot of 24 and 32 lb. shot. 33 pillows. 87 blankets. 6 sheets. 25 pillow-cases. 5 beds.

boat and oars, said to have been left in the care of John Honderson, Esq., and two spy-glasses in the possession of the city register.

In the charges for "carpenters, bricklayers, materials," &c., is included a considerable sum, say about \$1,250, for reconstruction of the bridge at Canton. Considerable quantities of bricks and lumber are also included, nossly used in the erection or repair of a "powder-house," of the locality of which your Committee are not informed. Wherever situated, if not retained in use, the second-hand materials may be of some value to the city.

be of some value to the city. The payments for "rations," "pay-rolls," "horse and hack hire," "hauling," "armories," "advertising," &c., amounting in the aggregate to \$23,106 35, represent so much ontlay, from which the city reaps no tangible return.

from which the city reaps no tangible return. The amounts of \$3,518 10 for "workmen on parks," and \$2,000 for "indigent widows, sewing-women," &c., were withdrawn from the fund by special resolutions of the council and expended for the benevolent purposes designated.

cit and expended for the benevolent purposes designated. The charge for "interest," is a necessary result of the loan, authorized by the ordinance of April 20th, 1861. The remaining item is the "cash in bank"—and this

The remaining item is the "cash in bank"—and this brings your committee to the third and last clause of the duties assigned them, viz: "to decide what disposition shall be made of the remaining funds now in the City Treasury."

On this point your committee are unanimonsly of the opinion that the halance should be returned, pro rata, to the several banks from which the whole amount was borrowed, thus canceling so much of the city's indebtedness, and stopping the payment or accumulation of interest.

In accordance with which statements and views, your committee respectfully recommend the adoption of the following resolutions :

S. F. STREETER, WM. SULLIVAN, JOHN DUKEHART, *First Branch.* JOSEPH ROBB, WM. SWINDELL, ASA HIGGINS, *Second Branch.* 

Resolved, By both Branches of the City Council of Baltitimore, that the papers and vouchers relating to expenditures under Ordinance No. 22, approved April 20th, 1861, be restored to the custody of the City Register. *Resolved furthermore*, That the Mayor be and is hereby authorized and requested to have collected, at the earliest possible moment the arms and military materials below

Resolved furthermore, That the Mayor be and is hereby authorized and requested to have collected, at the carliest possible moment, the arms and military materials belonging to the city, in the custody of the U.S. Marshal, or elsewhere, and, reserving such as, in his judgment, may be necessary for public use, to dispose of the balance, and place the proceeds in the City Treasury. *Resolved*, By the Mayor and City Council of Baltimore, that the Register be and he is hereby authorized and di-

*Resolved*, Ey the Mayor and City Council of Baltimore, that the Register be and he is hereby authorized and directed to repay, *pro rata*, to the several banks, the balance remaining on hand from the sum borrowed of them, under authority of ordinance No. 22, approved April 20th, 1861, "to appropriate \$500,000, or so much thereof as may be necessary for the defence of the city against any danger that may arise out of the present crisis."

These resolutions passed both branches.

THE LEGISLATURE CONFIRMED THE LOAN.

The Legislature of Maryland passed the bill, reported by Mr. Wallis from the Special Committee consisting of the Baltimore eity delegation, "to ratify an ordinance of the Mayor and City Council of Baltimore appropriating \$500,000 for the use of the city."

April 27, 1861, (the second day of the session,) this bill, under a suspension of the rules, passed the House of Delegates — yeas 58, nay 1, as follows:

YEAS-Messrs. E. G. Kilbourn, Speaker, Clarke J. Durant, George H. Morgan, Philip F. Rasin, Albert Medders, James T. Briscoc, Benjamin Parran, Bernard Compton, F. B. F. Burgess, Thos. C. Worthington, Robt. M. Denison, Leonard J. Quinlan, T. W. Renshav, J. Laurence Jones, Alex. Chaplain, James U. Dennis, Edward Long, John R. Keene, Wm. Holland, John R. Brown, James W. Maxwell, E. P. Bryan, Richard Wooten, Ethan A. Jones, Wm. H. Legg, William L. Starkey, Curtis W. Jacobs, George W. Landing, Stephen P. Dennis, Andrew Kessler, Jr, Thomas J. Clargett, John A. Johnson, William E. Salmon, Henry Straughn, G. W. Goldsborough, John C. Brune, Henry M. Warfield, Charles H. Pitts, William G. Harrison, J. Hanson Thomas, S. Teakle Wallis, T. Parkin Scott, Ross Winans, H. M. Morfit, Lau

renco Sangston, Joshua Wilson, William F. Baylees, Richard M'Coy, Janes Condy, Martin Eakle, John C. Brining, Howard Griffith, Josiah H. Gordon, W. R. Barnard, David Roop of A., John W. Gorsuch, Bernard Mills, William Turner-58.

NAYS-Mr. William T. Lawson-1.

Same day, the bill passed the Senate-yeas 16, nays none, as follows :

YEAS-Messrs. John B. Brooke, Thomas J. M'Kaig, John E. Smith, Coleman Yellott, John S. Watkins, Tengle Townsend, Stephen J. Bradley, Janues F. Dashiell, H. H. Goldsborough, Charles F. Goldsborough, Daniel C. Blackistone, Franklin Whitaker, Tilghman Nuttle, Anthony Kimmel, Oscar Miles, Washington Duvall-16.

#### \$500,000 MORE.

### IN HOUSE OF DELEGATES.

May 9—The House passed — yeas 56, nays none, a bill giving authority to the Mayor and City Council of Baltimore, to issue certificates of debt to any amount not exceeding \$500,000, in addition to the amount authorized to be issued by section 939 of the fourth article of the code of public laws.

May 10—It passed the Senate—yeas 16, nays none.

INDEFINITE AUTHORITY TO RAISE MONEY.

April 27—Mr. Wallis, from the Special Committee of the Baltimore city delegation, reported a bill "to add to the fourth article of the code of public local laws the following sections, empowering the Mayor and City Council of Baltimore to raise and appropriate money for the defence of the said city;" which was passed yeas 58, nays none.

Same day, it passed the Senate—yeas 16, nays none, as on the previous bill.

ACTS OF INDEMNITY.

#### IN THE HOUSE OF DELEGATES.

May 4—Mr. Pitts, from Special Committee, reported a bill, entitled "an act to relieve the Mayor and Board of Police of the city of Baltimore, and all persons who acted under the orders of the Mayor or of the Board of Police of Baltimore city, in their efforts to maintain peace and good order and prevent further strife, on and after the occurrences of the 19th April, 1861, in said city from prosecution in consequence of their acts of obedience to said order."

Pending the bill-

May 5-Mr. Pitts offered these amendments; which were adopted :

That no prosecuting attorney nor any officer in any court in this State, shall be entitled to charge or receive any compensation or cost in any prosecution which has been or may be instituted against any person for whose relief the act is intended.

That this act may be given as evidence under the general issue.

And the bill passed—yeas 43, nays none, as follows:

YEAS-Messrs. Kilbourn, Sp'r, Durant, Morgan, Rasin, Medders, Briscoe, Parran, Burgess, Ford, Denison, Renshaw, Chaplain, Long, Lawson, Maxwell, Wotten, Landing, Dennis of Wor, Routzahn, Johnson, Naill, Straughn, Goldsborough, Warfield, Brune, Winaus, Pitts, Harrison, Thomas, Wallis, Sangston, Morfit, Scott, Wilson of Har., Bayless, Coudy, Eakle, Brining, Stake, Barnard, Roop, Mills, Brown, -43.

May 8-Under the suspension of the rules, the bill passed—yeas 17, nays 2, (Kimmel and Stone.)

June 20-Mr. MILLS, from the Military Committee, reported a bill entitled an act to repeal sections 41, 42, 43, 46, 47, 48, and 112, of article | LXIII, of the Code of Public General Laws, and sections 741, 742, 745, of article IV, of the Code of Local Laws of this State, in relation to the uniformed and ununiformed militia and volunteer force in the several counties of this State, and in the city of Baltimore and the reclamation, custody, control, and redelivery of arms heretofore delivered to the officers and men under them; and to add to the said article LXV of the Code of Public General Laws a section providing for the prevention and suspension of proceedings upon bonds heretofore given under the laws of this State for the return of such arms; which was agreed to-yeas 40; nays 12.

The nays were Messrs. Medders, Keene, Mc-Intire, Routzahn, Naill, Wilson, M'Coy, Fiery, Stake, McCleary, Roop, Gorsuch.

#### IN SENATE.

June 22-The bill passed-yeas 12, nays none.

#### PROTEST AGAINST THE WAR, AND RECOGNITION OF REBEL INDEPENDENCE DEMANDED.

#### IN HOUSE OF DELEGATES.

May 9-Mr. WALLIS, from the Committee on Federal Relations, reported these resolutions :

Whereas, in the judgment of the General Assembly of Maryland, the war now waged by the Government of the United States upon the people of the Confederate States is unconstitutional in its origin, purposes and conduct; repugunconstitutional in its origin, purposes and conduct; repng-nant to civilization and sound policy; subversive of the free principles upon which the Federal Union was founded, and certain to result in the hopeless and bloody overthrow of our existing institutions; and whereas, the people of Maryland, while recognizing the obligations of their State, as a member of the Union, to submit in good faith to the exercise of all the legal and constitutional powers of the General Government, and join as one man in fighting its authorized battles, do reverence nevertheless, the great American principle of self-government and sympathize deeply with their Sonthern brethren in the noble and manly determination to uphold and defend the same; and whereas not merely on their own account, turn away from their own soil the calamities of civil war, but for the blessed sake of humanity, and to arrest the wanton shedding of fraternal blood, in a miserable contest which can bring nothing with to boost in a mark and constant which can bring not may not the set of Mary-land are enlisted with their whole hearts, upon the side of reconciliation and peace; Now therefore, it is hereby Resolved by the General Assembly of Maryland, That the

State of Maryland owes it to her own self-respect and her respect for the Constitution, not less than to her deepest and most honorable sympathics, to register this, her solemn protest, against the var which the Federal Government has declared upon the Confederate States of the South, and our sister and neighbor Virginia, and to announce her resolute determination to have no part or lot, directly or indirectly, in its prosecution.

2. That the State of Maryland earnestly and anxiously desires the restoration of peace between the belligerent sections of the country; and the President, authorities and people of the Confederate States having over and over officially and unofficially, declared, that they seek only peace and self-defence and to be let alone, and that they are willing to throw down the sword, the instant the sword now drawn against them shall be sheathed, the Senators and Delegates of Maryland do fervently besech and implore the President of the United States, to accept the olive branch which is thus held out to him, and in the name of God and of humanity, to cease this unholy and most wretched and unprofitable strife, at least until the assembling of the Congress at Washington shall have given time for the prevalence of cooler and better counsels. 2. That the State of Maryland earnestly and anxiously

a) Choice of cooler and better counsels.
3. That the State of Maryland desires the peaceful and immediate recognition of the independence of the Confederate States, and hereby gives her cordial consent thereunto as a member of the Union—entertaining the profound conviction that the willing return of the Southern people to their former federal relations is a thing beyond hope, and that the attempt to coerce them will only add slaughter and hat the nessent military occumation of Maryland

4. That the present military occupation of Maryland, nays 3, as the first.

being for purposes which, in the opinion of this Legislature, are in flagrant violation of the Constitution, the General Assembly of the State, in the name of her people, does hereby protest against the same, against the arbitrary restrictions and illegalities with which it is attended, calling upon all good citizens at the same time, in the most carnest and authoritative maner, to abstain from all violent and unlawful interference of every sort, with the troops in transit through our territory or quartered among us, and patiently and peacefully to leave to time and reason the ultimato and certain re-establishment and vindication of the right.

5. That under existing circumstances it is inexpedient to call a Sovereign Convention of the State at this time, or to take any measures for the immediate organization for arm -

ing of the militia. 6. That when the Legislature adjourn, it adjourn to meet , on the ---- day of at - next.

The report was accepted, and 10,000 copies ordered to be printed-yeas 50, nays 11

May 10-They were considered, and Mr. McCleary moved these as a substitute :

Whereas, it is right and proper, that the Gen wal Assemby of Maryland should give such expression of opinion as will call forth the united voice of the whole people of Maryland in the present emergency ; and the border States, with the exception of Maryland, have already, through their legislatures or conventions, defined their positions and course of action: Therefore be it Resolved by the General Assembly of Maryland, The pres-sent position of Maryland in the Union, is that of strict un-realized and will avaid unbeau and an an a Workington

such position of mary main unchanged so long as Washington continues the seat of Government: *Provided*, the United States affords ample protection to slaves and other property. *Resolved*, That when the seat of the United States Gov-ernment ceases to be at Washington, and a division of the united states along the prophetical here the free wight and

country takes place, the people shall have the free right and choice of deciding which section they will be attached to, by a free expression and decision of the popular will at the ballot-box.

Which were rejected-yeas 13, nays 43, as follows :

follows: YEAS-MESSIS. Medders, Lawson, Keene, Miller, Jonathan Routzahn, Salmon, Naill, Joshua Wilson of Har'd, Bayless, McCoy, Stake, David W. McCleary, Gorsuch—13. Navs-Messis. Kilbourn, Spr., Morgan, Rasin, Briscoe, Compton, Ford, Jacobs, Landing, Dennis of Wor'r, Kessler, Claggett, Johnson, Sangston, Morfit, Scott, Coudy, Eakle, Brining, Denison, Quinlan, Renshaw, Chaplain, Holland, Bryun, Jones of P. G's, Legg, Starkey, Goldsborough, Wati-Field, Brune, Winans, Pitts, Harrison, Thomas, Wallis, Flery, Griffith, Harding, Gordon, Barnard, Mills, Turner, Brown-43. Brown-43.

The blank in the resolutions reported, was then filled by "Frederick," and "Tuesday, the 4th day of June, at one o'clock, p. m."

And the resolutions were then adopted—yeas 45, nays 12, as follows:

45, nays 12, as follows: YEAS—Messrs. Kilbourn, Speaker, Morgan, Rasin, Briscoe, Compton, Ford, Worthington, Denison, Quinian, Renshaw, Jones of Talbot, Chaplain, Holland, Bryan, Wootten, Legg, Jones of Prince Georges, Starkey, Jacobs, Landing, Dennis of Worcester, Kessler, Johnson, Salmon, Goldsborough, Warfield, Brune, Winans, Pitts, Harrison, Thomas, Wallis, Sangston, Morfit, Scott, Coudy, Eakle, Brining, Griffith, Harding, Gordon, Barnard, Mills, Turner, Brown—45. Nars—Messrs. Medders, Lawson, Keene, Routzahn, Nail, Wilson of Harford, Bayless, McCoy, Fiery, Stake, McCleary, Gorsuch—12.

Gorsuch-12.

#### IN SENATE.

May 14-The committee recommended certain amendments; when these votes were taken :

The first resolution was adopted-yeas 11, nays 3, as follows :

YEAS-Messrs. John B. Brooke, President, Wash. Duvall, Thos. Franklin, J. F. Gardiner, John J. Heckart, Andrew A. Lynch, Thomas J. McKaig, Teagle Townsend, John S. Wat-kins, Franklin Whitaker, Coleman Yellott. NATS-Messrs. H. H. Goldsborough, Anthony Kimmel, John G. Stone

John G. Stone.

The second resolution was adopted—yeas 14, being all the above-named.)

The fifth resolution received all the above votes, and the third and fourth passed-yeas 11,

ing the following for the sixth:

That a committee be appointed to consist of four membors of the Senate and four members of the House of Delebors of the Senate and four members of the House of Delo-gates, four of which committee, (to be selected of them-selves.) shall as early as possible, wait on the President of the United States at Washington, and the other four of said committee shall wait on the President of the Southern Confederacy, for the purpose of laying the foregoing reso-lutions before them, and that said committee be and is hereby especially instructed to obtain, if possible, a general cesation of hostilities, now impending, until the meeting of Congress in July next, in order that said body may, if possible, arrange for an adjustment of existing troubles by means of negotiation, rather than the sword. Resolved, That said committee consist of Messrs. Brooke,

Yellott, McKaig, and Lynch, of the Senate, and Messrs. , , , , and , , of the Ilouse of Delegates. *Resolved*, That said committee bo requested to report, if

practicable, to the General Assembly, on the 5th day of June next.

Passed-yeas 11, nays 3, as before; and the resolutions, as amended, were then adopted by the same vote.

May 14-The House filled the blanks with the names of Harding, Morgan, Compton, and Goldsborough, and passed the resolution-yeas 24, nays 18, as follows:

YEAN-Messrs, Kilbourn, (Speaker,) Morgan, Ford, Quin-lan, Renshaw, Jones of Talbot, Chaplain, Jones of P. G., Starkey, Kessler, Straughn, Goldsborough, Warfield, Brune, Windns, Fitts, Harrison, Thomas, Wallis, Sangston, Scott, Griffith, Harding, Turner-24. NAYS-Messrs. Welch, Worthington, Denison, Keene, Holland, Maxwell, Landing, Claggett, Routzahn, Johnson, Salmon, Morfit, Fiery, Barnard, McCleary, Roop, Milts, Brown-18.

Brown-18.

#### RECOGNITION AGAIN DEMANDED.

June 10-The House passed resolutions previously offered by Mr. Chaplain, after having amended them to read as follows :

Resolved by the General Assembly of Maryland, That the Representatives of the State of Maryland in the Congress of the United States at the approaching extra session of that legislative body be, and they are hereby, earnestly desired and requested to urge and vote for an immediate recognition of the independence of the government of the Confederate

of the independence of the government of the Connectation States of America. *Resolved*, That the Speaker of the House of Delegates and the President of the Senate, together, forward to Hon. James Alfred Pearce and Hon. Anthony Kennedy, the Representatives of the sovereign State of Maryland in the Senate of the United States, a copy of the report of the Committee on Federal Relations and the accompanying resolutions together with these resolutions. resolutions, together with these resolutions.

## Yeas 31, nays 22, as follows:

YEAS-Messrs. Kilbourn, Speaker, Durant, Morgan, Rasin, Welch, Briscoe, Parran, Compton, Jones of Tallot, Chap-lain, Holland, Maxwell, Bryan, Wootten, Legg, Starkey, Landing, Kessler, Goldsborough, Warfield, Brune, Winans, Harrison, Thomas, Wallis, Sangston, Morfit, Scott, Griffith,

Martison, Anomes, and Sangeron, Monre, Every Grinner, Gordon, Mills-31. NAVS-Messrs. Medders, Ford, Denison, Quinlan, Ren-shaw, Dennis of Somerset, Stanford, Lawson, McIntire, Mil-ler, Routzahn, Johnson, Salmon, Naill, Stranghn, Wilson, Bayless, McCoy, Fiery, Brining, McCleary, Gorsuch-22.

#### IN SENATE.

June 22—The Senate passed these resolutions -yeas 9, nays 3, as follows :

YEAS-Messrs. Brooke, President, Blakistone, Dashiell, Franklin, Gardiner, McKaig, Townsend, Whatkins, Whitaker-9.

NAVS-Messrs. Duvall, Heckart, Lynch-3.

THE "PUBLIC SAFETY" BILL.

#### IN SENATE.

May 2-It was taken up-yeas 14, nays 8, as follows:

YEAS-Messrs. Brooke, President: Blackistone, Duvall, Dashiell, Franklin, Grahame, Gardiner, Heckart, Lynch, McKaig, Townsend, Watkins, Whitaker, Yellott-14. NATS-Messrs. Bradley, Goldsborough of Talbot, Golds-nessee league, note, page 5.

The amendment of the committee, substitut- | borough of Dorchester, Kimmel, Miles, Nuttle, Smith, Stone

Mr. Goldsborough of Talbot moved to amend so as to give the power of appointment of the commissioners provided for to the Governor, one from each of the gubernatorial districts; rejected-yeas 6, nays 12.

A motion to recommit was lost-yeas 10, navs 12.

Motions to adjourn, recommit, change the names of the commissioners, were made and lost.

May 3-Mr. Goldsborough of Talbot moved to give the people, on the 13th of June, the election of three commissioners, one from each gubernatorial district, which was lost-yeas 8, nays 13, as follows:

YEAS-Messrs. Bradley, Grahame, Goldsborough of Talbot, Goldsborough of Dorchester, Kimmel, Nuttle, Smith, Stone

NAYS-Messrs. Brooke, President; Blackistone, Duvall, Dashiell, Franklin, Gardiner, Heckart, Lynch, Miles, Town-send, Watkins, Whitaker, Yellott-13.

A like motion, for six commissioners, was lost--yeas 8, nays 12.

On the section respecting disbursements of money, Mr. Goldsborough of Talbot moved this proviso:

Provided, That the same shall not extend to the exercise of any powers for the disbursement of any moneys that may be hereafter appropriated for the arming of the military forces of this State and for the formation of any alliance offensive or defensive, with any other State in this Confederacy.

Which was rejected-yeas 8, nays 12, as follows:

YEAS-Mossrs. Bradley, Grahame, Goldsborough of Talbot, Kimmel, Miles, Nuttle, Smith, Stone-8. NATS-Messrs. Brooke, President; Blackistone, Duvall,

Watkins, Whitaker, Yellott-12.

Mr. STONE offered an additional section :

That no member of the board shall act as such, until he has taken the same oath as is now administered to the members of the Legislature.

Which was rejected-yeas 7, nays 12, as follows:

YEAS-Messrs. Bradley, Grahame, Goldsborough of Tal-bot, Kimmel, Yuttle, Smith, Stone-7. NAVS-Messrs. Brooke, President; Blackistone, Duvall,

Dashiell, Franklin, Gardiner, Heckart, Lynch, Townsend, Watkins, Whitaker, Yellott-12.

At this stage, and after a most obstinate contest, the bill was abandoned by its friends, and Maryland was spared a bloody baptism.

## Votes on Secession Ordinances in Scuth Carolina\* and Florida.

In South Carolina it passed unanimouslythe following persons voting for it :

the following persons voting for it: James H Adams, R. T. Allison, D. C. Appleby, S. T. Ab-kinson, L. M. Ayer, R. W. Baruwell, A. L. Barron, D. R. Barton, T. W. Beaty, A. W. Bethea, C. St. P. Bellinger, S. Bobo, P. P. Bonneau, J. J. Brabham, A. H. Brown, C. P. Brown, J. Buchanan, A. W. Burnet, W. Cain, John A. Cah-houn, Joseph Caldwell, W. H. Campbell, M. E. Carn, Jas, H. Carlisle, J. P. Carroll, H. J. Canghman, W. C. Cauthen, E. W. Charles, Jas. Chesnut, jr., Langdon Cheves, E. M. Clarke, H. W. Conner, R. L. Crawford, William Curtis, J. A. Dargan, R. J. Davant, H. C. Davis, W. F. DeSaussure, R. DeTreville, A. W. Dozier, Perry E. Duncan, B. F. Dunkin, A. Q. Dunovant, R. G. M. Dunovant, D. Upre, W. K. Easley, W. J. Ellis, T. R. English, C. D. Evans, Simeon Fair, W. P. Finley, Daniel Flud, A. M. Forster, B. B. Foster,

\* Vote on Virginia ordinance, note, page 7; and on Ten-

J. E. Frampton, J. C. Furman, James M. Gadberry, H. W. Gardington, J. C. Geiger, William H. Gist, T. W. Glover, E. W. Goodwin, R. M. Gourdin, T. L. Gourdin, H. D. Green, Maxog Greeg, William Gregg, W. S. Grisham, A. J. Hummond, T. M. Hanckol, W. W. Harlee, James Ilarison, J. M. Koitt, J. B. Kershaw, B. F. Kilgore, J. P. Kiluson, John A. Inglis, J. J. Ingruham, S. Jackson, James J. Grifts, J. Ohn Jenklins, J. E. Kershaw, B. F. Kilgore, J. P. Kiluson, J. M. Koitt, J. B. Kershaw, B. F. Kilgore, J. P. Kuster, J. G. Landrum, B. W. Lawton, A. F. Lewis, R. C. Logan, W. S. Lyles, Edward McCrady, Henry McIver, John McKee, A. McLoed, A. G. Magrath, G. Maniguit, John L. Manning, B. F. Mauldin, John Maxwell, M. P. Mayes, Alexander Mazyek, John H. Means, C. G. Memminger, J. Isard Middleton, W. Middleton, W. Middle, T. K. Simons, J. R. Simpson, J. R. Shingler, Y. Simons, Jr. R. F. Simgson, J. P. Robinson, W. B. Rowell, B. H. Rutledge, E. Scott, E. M. Scabrook, G. W. Seabrook, S., B. E. Sessions, J. M. Shingler, W. Shingler, T. V. Simons, J. T. F. Simpson, J. M. Timmons, J. Tompkins, J. Townsend, T. D. Wagner, John Wanamaker, D. L. Wardlaw, F. H. Wardtaw, W. D. Watts, Thomas W. More, R. N. Mornen, T. J. Mangar, F. B. Simps, P. Stokes, J. S. Sims, R. A. Thomson, J. M. Timmons, J. Tompkins, J. Townsend, T. D. Wagner, John Wanamaker, D. L. Wardlaw, F. H. Wardtaw, W. Wattit calls "their having been the back door," through which the creep and get "an interview with a 'full interchange of sentiments."

#### In Florida, this was the vote :

YEAS-Mr. President, Judge John C. McGehee, Messrs. Alderman, Allison, Anderson, Baker of Calhoun, Barring-ton, Beard, Bethel, Chanler, Collier, Coon, Coopor, Daniel, Davis, Dawkins, Devall, Dilworth, Finegan, Folsom, Gary, Gettis, Glazier, Golden, Helvenston, Henry, Hunter, Irwin, Kirksey, Ladd, Lamar, Lamb, Lea of Madison, Leigh of Sumter, Lewis, Love, McGahagin, McLean, McIntosh, Mc-Nealer, Mays, Morton, Newmans, Nicholson, Owens, Pal-mer, Parkhill, Pelot, Pinckney, Sanderson, Saxon, Sever, Spencer, Simpson, Solana, Stephens, Thomas, Tift, Turman, Ward, Wright of Escambia, Wright of Columbia, Yates-62.

NAYS-Messrs. Baker of Jackson, Gregory, Hendricks,

The Legislatures of Georgia and Mississippi have passed resolutions disapproving, as unconstitutional, the bill suspending the privilege of the writ of habeas corpus and demanding its

Alabama is reported to have furnished 40,000 men to the army; Florida, 4,000; Georgia, 54,000; Louisiana, 36,000; Mississippi, 40,000, North Carolina, 25,000; South Carolina, 25,-000; Texas, 29,000; Tennessce, 34,000; Virginia, 103,000; Arkansas, 28,000; Kentucky, 20,000; Maryland, 20,000; Missouri, 35,000; total, 501,000. The estimate of Maryland must

The Richmond Examiner of July 26 denounces Clay and Holcombe for their late attempt at "negotiation," at Niagara Falls, and rejoices at what it calls "their having been kicked from the back door," through which they sought to creep and get "an interview with Lincoln for a 'full interchange of sentiments.'"

The same paper, of June 13, alludes to the nomination at Baltimore " of the Illinois railsplitter " and the "Tennessee tailor, one of the meanest of that craft," and predicts that " whether they shall ever be elected or not depends upon the Confederate army altogether."

The Richmond Sentinel publishes the speech of Mr. Long of Ohio on peace, and compliments it as "able, truthful, deliberate, and brave."

# THE REBEL ADMINISTRATION.

## The "Provisional."

FEBRUARY 18, 1861, TO FEBRUARY 18, 1862. Those formerly members of the Congress of

the United States are italicized.

President-Jefferson Daris, of Mississippi. Vice President-Alexander H. Stephens, of Georgia.

#### THE CABINET.

Secretary of State-Robert Toombs, of Georgia, from February 21 to July 30; resigned, and succeeded by Robert M. T. Hunter, of Virginia. Secretary of the Treasury—Charles G. Memminger, of South

Carolina, from February 21. Scoretary of War-Leroy Pope Walker, of Alabama, from

- February 21 to September 10; resigned, and succeeded by Judah P. Benjamin, of Louisiana. Secretary of the Navy-Skephen R. Mallory, of Florida, from
- March 4.
- Atlorney General-Judah P. Benjamin, of Louisiana, from February 21 to September; resigned, and succeeded by
- Costary at to september; resigned, and succeeded by Thomas II. Watts, of Alabama. Postmaster General—John II. Reagan, of Texas, from March 6, Henry J. Ellet, of Mississippi, first appointed, having declined.

#### MEMORANDUM.

- Secretary of War-It was reported that the position was offered to General Braxton Bragg, of Louisiana, but de-clined, after Mr. Walker's resignation. Szeretary of the Navy-Also, that John Perkins, Jr., of

- Screttary of the Navy—Also, that John Ferkins, Jr., of Louisiana, was first appointed, but declined.
   Mr. Toombs was appointed a brigadier general in July, 1861, and Mr. Walker August 20, 1861.
   The President and Vice President were elected by the Con-gress, February 9, 1861, receiving the unanimous vote of the six States then composing the Confederacy.

#### THE CONGRESS.

The Deputies who organized the Confederacy at Montgomery remained in office one year, the "permanent" Government having been instituted in February, 1862.

There were four sessions of that body :

The first at Montgomery, from February 4 to March 16, 1861

The second at Montgomery, from May 6 to May 21, 1861. The third at Richmond, from July 20 to September 2, 1861.

The fourth at Richmond, from November 18, 1861, to February 17, 1862.

The following are the names of members :

The following are too hamos or momosts.
 President—Howell Colb, of Georgia.
 Alabama\*—Robert H. Smith, Richard Wilde Walker, Colin J. McRae, John Gill Shorter, William P. Chilton, Stephen F. Hale, David P. Lewis, Thomas Fearn, Jabez L. M. Curry.
 Arkansas\*—Robert W. Johnson, Albert Rust, Augustus H. Garland, William W. Watkins, Hugh F. Thomasson.
 Florida\*—Jackson Morton, J. Patton Anderson, James B. Guens

Owens.

Georgia\*—Robert Toombs, Howell Cobb, Francis S. Bartow, Martin J. Crawford, Eugenius A. Nisbei, Benjamin H. Hill, Augustus R. Wright, Thomas R. R. Cobb, Augus-tus II, Konan, Alexander H. Stephens.

Kentucky\*—Henry C. Burnett, -Machen, Thomas B. Munroe. - Thomas, Willis B.

- Louisiana-John Perkins, Jr., A. de Clouet, Charles M. Con-rad. Duncan F. Kenner, Edward Sparrow, Henry Marshall.
- sissippi-Wiley P. Harris, Walter Brooke, W. S. Wilson, A. M. Clayton, William S. Barry, James T. Harrison, Mississippi-
- J. A. P. Campbell. Missouri<sup>\*</sup>—William H. Cook, Thomas A. Harris, Caspar W. Bell, A. H. Conrow, George G. Vest, Thomas W. Freeman, Samuel Hyer.
- North Carolina\*-George Davis, William W. Avery, William N. H. Smith, Thomas Ruffin, Thomas D. McDowell, Abram W. Venable, John M. Morehead, Robert C. Puryear, Burton Craige, Andrew J. Davidson.
- South Carolina<sup>\*</sup>—R. Bornwell Rhett, Sr., Robert W. Barn-well, Lawrence M. Keitt, James Chesaut, Jr., Charles G.
- Well, Lawrence M. Kett, James Chessul, Jr., Charles G.
   Memninger, William Porcher Miles, Thomas J. Withers, William W. Boyce.
   Tennessec<sup>8</sup>-Joseph B. Heiskell, Thomas Menees, William G. Swan, <u>Mecker</u>, <u>House</u>, George W.
   Jones, John D. C. Atkins, <u>De Witt</u>, <u>Monthematical Science</u> Monthematical Mechanisms. Ihery.
- ast-Louis T. Winfall, John H Reagan. John Hemphill, T. N. Waul, William B. Ochiltree, W. S. Oldham, John Texas\*-
- 1. A. Watt, H. S. Seddon, William Ballard Preston, Robert M. T. Hunter, John Tyler, Sr., William H. Mao-farland, Roger A. Pryor, Thomas S. Boeock, William C. Rives, Robert E. Scott, James M. Mason, J. W. Brockon-Line Constant W. Russell. Robert Johnson, Waller brough, Charles W. Russell, Robert Johnson, Waller R. Staples, Walter Preston.

#### MEMORANDUM.

J. Johnson Heeper, of Alabama, was Secretary of the Con-

J. Johnson Hooper, of Alabama, was severing --gress. Alabama-David P. Lewis and Thomas Fearn, elected De-putics, were succeeded during the year by II. C. Jones and Nicholas Davis, Jr. Mr. Shorter was elected Governor in November, 1861. Mr. McRae was after-wards appointed a brigadier general, and is now the cotton loan agent in Paris. Mr. Yancey was nominated for, but declined election to, the Provisional Congress, and was sent to Europe to negotiate for "Confederate" recognition.

recognition. Arkansas-Admitted, at second session, in May, 1861. Florida—J. Patton Anderson resigned early in the term, and was sncceeded by G. T. Ward. Georgia—Mr. Toombs resigned, February 21, to become Provisional Scoretary of State; in July was appointed a brigadier general. Mr. Howell Cobb subsequently became brigadier general, and now is a major general. Mr. Coverbod subsequentity became a grafty coloned in became brigader general, and how is a halo general. Mr. Crawford subsequently became a cavalry colonel in the army, and was once a prisoner. Mr. Thomas R. R. Cobb became colonel of the Georgia Legion late in 1861 or early in 1862, and was killed at the first Fredericks-burg battle, December 13, 1862. Mr. Stephens accepted burg battle, December 13, 1862. Mr. Stephensaccepted the Vice Presidency, February 11, 1861. *Kentucky*—Admitted, at fourth session, in December, 1861; *Missouri*—Admitted, at fourth session, in December, 1861;

\* See memorandum at the close of the list.

400

the Delegates were self-elected, and held over the first Congress. Mr. Hyer is reported to have taken the oath of allegiance to the Government of the United States.

- or aneguance to the Government of the United States.
   North Carolina—These Delegates were elected by the Con-vention, June 18, 1861. Mr. Rufin afterwards became e cavalry colonel, and died in the spring of 1864, a prisoner in Alexandria, Va., of wounds received in ba.tle.
- South Cirolina—Mr. Memminger became Secretary of the Treas.ry, February 21, 1861. Mr. Keitt died in Rich-mond, June 2, 1664, of wounds received July 31 in bat-the colonel of the 20th South Carolina regiment. Mr. Chesnut ærved as aid to Beauregard at the bombard-ment of Sumter; and Mr. Miles as an aid at the battle of Bull Ran.
- Tennessee-Aganitted, at second session, in May, 1861; members took their seats at the third session
- Texas-Admitt.d, at first session, March 2, 1861. Mr. Rea-
- Texas—Admitt.d, at first session, March 2, 1861. Mr. Reagan resigned to become Postmaster General, March 0, 1861. Mr. Vigfall was appointed a brigadier general, October 29, 1261, but did not yield his seat in the "Provisional" or the "Permanent" Congress. Mr. Hemphill died January 4, 1862.
  Virginia—Admit.ed, at second session, May 7, 1861, when Messus. Brockenbrough and Staples took their seats; the others w.se sworn at the third session, at Richmond, July 20, 1861. Mr. Hunter became Secretary of State, July 50, and resigned. Mr. Mason resigned in the fall of 1861 to go to England, and November 19 the State Convention elected Alexander R. Boteler to succeed him. ceed him.

### The "Permanent" Administration. FROM FEBRUARY 19, 1862.

President—Jefferson. Davis, of Mississippi. Vice President—Alexander H. Stephens, of Georgia.

THE CABINET.

## CONFIRMED, MARCH 23, 1862.

Secretary of State-JuJah P. Benjamin, of Louisiana.

- Secretary of the Treasury-Charles G. Memminger, of South
- Carolina; rosigned, in June, 1864, and succeeded by George A. Trenhola, of South Carolina. Secretary of War-George W. Randolph, of Virginia; re-signed, and succeeded by James A. Seddon, of Virginia.
- Scretary of the Navy-Stephen R. Mallory, of Florida. Attorney General-Thoms II. Watts, of Alahama; resigned on election as Governon of Alahama, in November, 1863, and succeeded by George Davis, of North Carolina. Postmaster General-John H. Reagan, of Texas.

#### MEMORANDUM.

- Mr. Randolph was appointed a colonel of Virginia troops by Governor Letcher, in the tall of 1861; tendered his resignation but withdrew it, and in November of that year appointed a brigadier general, and assigned to the command of the district between Suffolk, in Nanse-mond, and Weldon, on the Roanoke; he was a candi-date for Congress in November, 1861, but withdrew on the morning of the election.
- Mr. Memminger was born in Wirtemberg, Germany, Janu-Memininger was born in wreenberg, Germany, Jahu-ary 7, 1503; was brought to this country when nine years old; was early left an orphan; adopted by Gover-nor Thomas Bennett, and educated in South Carolina college, graduating in 1820; began the practice of law in 1820; in 1832–33 he was against nullification; for nearly twenty years he was at the head of the Finance Committee of the lower house of the Legislature of South Carolina, retiring in 1852; he filled other State offices.

## THE FIRST CONGRESS.

## FEBRUARY, 1862, TO FEBRUARY, 1864.

It held four sessions :

The first from February 18 to April 21, 1862. The second from August 12 to October 13, 1862. The third from January 12, 1863, to May -, 1863. The fourth from December 7, 1863, to February 18, 1864. SENATORS.

Alabama<sup>\*</sup>— William L. Yancey, Clement C. Clay, Jr. Arkansas<sup>\*</sup>—Robert W. Johnson, Charles B. Mitchel. Florida—James M. Baker, Augustus E. Maxweil. Georgia<sup>\*</sup>—Benjamin H. Hill, Robert Toombs. Kentucky—Henry C. Burnett, William E. Simms. Louisiana—Edward Sparrow, Thomas J. Semmes. Mississippi<sup>\*</sup>—Albert G. Brown, James Phelan.

\* See memorandum at the end of the list. 26

Missouri-John B. Clark, Robert L. Y. Peyton. North Curolina\*-George Davis, William T. Dortch. Noric Cardina-George Davis, William T. Dortch. Sould Cardina-Robert W. Barnwell, James L. Orr. Tranessee\*-Landon C. Haynes, Gustavus A. Henry. Virginia\*-Robert M. T. Hunter, William Ballard Presion. Texas-Louis T. Wigfall, William S. Oldham.

#### MEMORANDUM.

- Jefferson Davis was inaugurated as "Permanent" President of the "Confederate" States, February 22, 1862, in Richmond.
- on the first day of the session, Vice President Stephens pre-siding, Robert M. T. Hunter, of Virginia, was clected President pro tempore; James II. Nash, of South Caro-lina, Secretary; and James Page, of North Carolina,
- Mild, Secretary, and sames rage, or North Carolina, Doorkceper.
   Alabama—Mr. Yancey diel, and Robert Jemison was elected, Angust 22, 1863, to the vacancy.
   Arkansas—Mr. Mitchel had been elected, shortly before secession, to the United States Senate for six years, from March 4, 1861.
- Gorgia-Mr. Toombs having accepted a brigadier's com-mission did not take his seat, and he was succeeded, March, 1862, by Dr. John W. Lewis, appointed by Gov-ernor Brown, and, December, 1882, by Herschel V. John-
- son, elected by the Legislature. Mississippi-Mr. Brown, when elected, was captain of a company in the 17th Mississippi volunteers. Mr. Wal-ter Brooke was at first announced elected over Mr. Phelan, but the latter appeared and was qualified at
- the first session. North Carolina-Mr. Davis, when he resigned to become Attorney General, was succeeded by William A. Graham. .
- Tennessee-Mr. Henry, early in 1862, was A. A. G. on Gen-
- eral Pillow's staff. Wirginia-Mr. Preston was succeeded, January 2S, 1863, by Allen T. Caperton.

#### REPRESENTATIVES.

- Speaker—Thomas S. Bocock, of Virginia. Alabama—Thomas J. Foster, William R. Smith, John P. Ralls, Jabez L. M. Curry, Francis S. Lyon, William P. Chilton, David Clopton, James L. Pugh, Edward S.

- Chilton, David Clopton, James L. Pugh, Edward S. Dargan-9.
  Arkansas<sup>8</sup>-Felix I. Batson, Grandison D. Royston, Augustus H. Garland, Thomas B. Hanley.—4.
  Florida—James B. Duwkins, Robert B. Hilton—2.
  Georgia—Julien Hartridge, C. J. Munnerlyn, Hines Holt, Angustus H. Kenan, Daniel W. Lewis, William W. Clark, Robert P. Trippe, Lucius J. Gartrell, Hardy Strickland, Augustus R. Wright—10.
  Kentucky<sup>3+</sup>—Willis B. Machen, John W. Crockett, Henry E. Read, George W. Ewing, James S. Chrisman. Theodore L. Burnett, H. W. Bruce, G. B. Hodges, Ely M. Bruce, James W. Moore, Robert J. Breckinidge, Jr., John M. Elliott-12.
- Bruce, James W. Moore, Robert J. Breckinridge, Jr., John M. Eliott-12. Louisiana—Charles J. Villere, Charles M. Conrad, Duncan F. Kenner, Lucius J. Dupre, Henry Marshall, John Perkins, Jr.-6. Mississippi<sup>\*</sup>-J. W. Clapp, Reuben Davis, Israel Welsh,

\* See memorandum at the end of the list.

- + Members sworn August 18, 1862. The Provisional Le-
- Thembers sworn August 15, 1802. The Provisional Le-gislature of Kentucky thus districted the State: First District—Fulton, Hickman, McCracken, Graves, Cal-loway, Marshall, Livingston, Lyon, Caldwell, Trigg, Ballard.
- Ballařd.
   Second District—Union, Webster, Hopkins, Christian, Todd, Henderson, Daviess, Muldenburch, Crittenden.
   Thérd District—Hancock, Olio, Gruyson, Breekinridge, Meade, Hardin, Larne, Eutler, Hart.
   Fourth District—Logan, Simpson, Allen, Monroe, Barren, Edmondson, Warren, Metcalfe.
   Fifth District—Cumberland, Clinton, Wayne, Pulaski, Casey, Lincoln, Taylor, Green, Adair, Russell.
   Sixth District—Spencer, Bullitt, Nelson, Washington, Ma-rion, Mercer, Boyle, Garrard, Anderson.
   Seenth District—Hefrson, Shelby, Oldham.
   Eighth District—Henry, Trimble, Carroll, Boone, Gallatin, Grant, Kenton, Campbell.

- Grant, Kenton, Campbell.
- Winth District—Pendleton, Bracken, Nicholas, Harrison, Bourbon, Fleming, Mason.
  Tenth District—Bath, Lewis, Greenup, Boyd, Carter, Law-rence, Montgomery, Powell, Morgan, Rowan, Wolfe, Estill, Magodin.
- Estin, Magonal. Eleventh District-Franklin, Woodford, Jessamine, Fayette, Madison, Clarke, Owen, Scott. Twel/th District-Rockcastle, Knox, Harlan, Laurel, Whit-ley, Clay, Perry, Owsley, Letcher, Breathitt, Floyd, Pike, Johnson, Jackson.

٩

Henry C. Chambers, Olho R. Singleton, Ethelbert Barks- |

- dale, John J. Mc Ree-7.
  Missouri-Thomas A. Harris, Casper W. Bell, A. H. Conrow, George G. Vest, Thomas W. Freeman, William II. Cook -6,
- -G.
   North Carolina--W. N. H. Smith, Robert R. Bridgers, Owen R. Kenau, Thomas D. McDowell, A. H. Arrington, J. B. McLean, Thomas S. Ashe, William Lander, Bur-gess S. Gaither, A. J. Davidson--10.
   South Curolina\*-John McQuern, William Porcher Miles, Milledge L. Bonham, William D. Simpson, James Far-row, William W. Rouse-6
- Milledge L. Bonham, William D. Entropols, Conserved for row, William W. Boyce-6.
   Tennessee\*—Joseph B. Heiskell, William G. Swan, William B. Tibbs, E. L. Gardenhire, Henry K. Foole, Micredith P. Gentry, George W. Jones, Thomas Menoces, John D. C. Atkins, John V. Wilght, David M. Currin-11.
   Tezas\*—John A. Wilcor, Clark C. Herbert, Peter W. Gray, William B. Wilght
- B. F. Sexton, Malcolm D. Graham, William B. Wright -6.
- —0. Virginia\*—Muscoe R. H. Garnett, John R. Chambliss, James Lyons, Roger A. Pryor, Thomas S. Bocock, John Goole, Jr., James P. Holcombe, D. C. De Jarnette, William Smith, Alexander R. Boteler, John B. Baldwin, Waller R. Staples, Walter Preston, Albert G. Jenkins, Robert R. Staples, Walter Preston, Albe. Johnson, Charles W. Russell-16.
- October 9, 1862, at the second session, Elias C. Boudinot was admitted a Delegate from the Cherokee nation.

#### MEMORANDUM.

- Emmet Dixon, of Georgia, was elected Clerk of the House, and R. H. Wynn, of Alabama, Doorkeeper. Arkansas-Mr. Garland's seat was contested by Jilson P.
- Johnson.
- Keniucky-Mr. Hodges was not sworn until August 16, 1862
- Mississippi-Mr. Davis resigned, and was succeeded by
- William D. Holder, M. Boham was elected Governor in Jannary, 1863, and was succeeded by Lewis M. Ayer, Tennessee—Mr. Currin died during the Congress, after his
- election to the second Congress.
- Texas-Mr. Wilcox died during the Congress, after his clec-
- Jezas—Mr. Whicox died diffing the Congress, after his elec-tion to the second Congress. *Virginia*—Mr. Garnett died, January 12, 1864. Mr. Pryor was appointed a brigadier general in the fall of 1862, and was succeeded by Charles F. Collier. Mr. Smith accepted a colonel's commission, was succeeded by David Funsten, and was elected Governor in 1863.
  Mr. Belwin was exercisted a colonal of Urarija traver Mr. Baldwin was appointed a colorel of Virginia troops in the fall of 1861, by Governor Letcher. Mr. Jenkins was appointed brigadier general, and resigned in June or July, 1862; was succeeded by Samuel A. Miller; and died in the summer of 1864, in Southwestern Virginia, of succeeding bette of wounds received in battle.

## THE SECOND CONGRESS.

FEBRUARY 19, 1864, TO FEBRUARY 18, 1866.

The first session closed June 15; next session in November SENATORS.

The following are the changes from the first Congress:

Alabama-Richard Wilde Walker, in place of Clement C. Clay, Jr. Mississippi-J. W. C. Watson, in place of James Phelan.

\* Sce memorandum at the end of the list.

• 1

#### Missouri-Waldo P. Johnson, in place of Mr. Peyton. REPRESENTATIVES.

Speaker-Thomas S. Bocock, of Virginia.

- Alabama-Thomas J. Foster.\* William R. Smith.\* William-Alabama -- Hormas J. Poster, "Waltam R. Smith," Waltamson R. W. Colb, H. Marcus H. Cruickshanl, Francis S. Lyon,\* William P. Chilton,\* Darid Clopton,\* James L. Picgh,\* J. S. Dickinson. Arkansas-Felix I. Batson,\* Rufus K. Garland, Augustus H. Garland,\* Thomas B. Hanley.\*
- H. Garland, \* Holmas B. Hanley,\* Florida-St. George Robert B. Hilton.\* Georgia-Julien Hartridge,\* William E. Smith, Mark H. Blanford, Clifford Anderson, J. T. Shewmake, J. H. Echola, James M. Smith, George N. Lester, H. P. Bell.
- Warren Aikon. Kentucky-Willis B. Machen,\* George W. Triplott, Henry E. Road, \* George W. Ewing, \* James S. Chrisman, \* Theo dore L. Burnott, \* H. W. Bruce, \* *Humphrey Marshall*, Ely M. Bruce, \* James W. Moore, \* Benjamin F. Bradley,
- John M. Elliott.\* Louisiana-Charles J. Villero,\* Charles M. Conrad,\* Duncan F. Kenner,\* Lucius J. Dupre,\* B. L. Hodge, John Perkins, Jr.\*

- F. Kenner,\* Lucius J. Dupre,\* B. L. Hodge, John Perkins, Jr.\*
  Mississippi John A. Orr, William D. Holdor,\* Israel Welsh,\* Henry C. Chambers,\* Otho R. Singleton,\* Ethelbert Barksdale,\* J. T. Lumpkin.
  Missouri, (supposed) Thomas L. Snead, N. L. Norton, Thomas H. Price, A. H. Conrow,\* G. G. Vest,\* Thomas W. Freeman,\* R. A. Hatcher.
  North Carolina-William N. H. Smith,\* Robert R. Bridgers,\* J. T. Leach, Thomas C. Fuller, Josiah Tarner, John A. Gilmer, James M. Leuch, J. G. Ramsay, Burgess S. Gaither,\* Georg W. Logan.
  South Carolina-William N. H. Smith,\* Robert R. Bridgers,\* J. T. Leach, Thomas C. Fuller, Josiah Tarner, John A. Gilmer, James M. Leuch, J. G. Ramsay, Burgess S. Gaither,\* Georg W. Logan.
  South Carolina-James M. Witherspoon, William Porcher Miles,\* Lewis M. Ayer,\* William G. Swan,\* A. S. Colver, John P. Murray, Henry S. Foole,\* E. A. Keeble, James McCollum, Thomas Nences,\* John D. C. Atkins,\* John V. Wright,\* one vacancy.
  Tezas-C. C. Herbert,\* A. M. Branch, B. F. Sexton,\* A. R. Baylor, S. H. Morzan, one vacancy.
  Virginia-Robert L. Montague, R. H. Whitfield, William C. Wickham, T. S. Gholson, Thomas C. Decock,\* John Boode, Jr.,\* William C. Kues, Daniel C. De Jarnette,\* David Funsten,\* F. W. M. Holhalay, John B. Baldwin,\* Waller, R. Staples,\* Fugate McMallea, Samuel A. Miler,\* Robert Johnson,\* Charles W. Russell.\* ler,\* Robert Johnson,\* Charles W. Russell.\*

Those marked thus \* were members of the last Congress.

+1864, May 3-Mr. Chilton offered this resolution, which

representative of a loyal constituency : Therefore,

Resolved, That a committee of five members be appointed by the Speaker to inquire into such reports, and to collect and report upon the testimony bearing upon the loyalty or disloyalty of said member, and report the same to this House, with such recommendation as to its further action in the premises as to said committee shall seem proper; and that Mr. Cobb be notified by the committee, if practicable, of the sitting of the committee, and that said com-mittee have power to send for persons and papers. Mr. Cobb has since come within our lines.

# NATIONAL POLITICAL CONVENTIONS.

## Union National Convention.

This body met at 12 o'clock, noon, on Tuesday, June 7, at Baitimore, in accordance with the call of the National Executive Committee:

The undersigned, who by original appointment, or subse-quent designation to fill vacancies, constitute the Executive Committee created by the National Convention held at Chi-cago on the 16th day of May, 1860, do hereby call upon all qualified voters who desire the unconditional maintenanco of the Union, the supremacy of the Constitution, and the complete suppression of the existing rebellion, with the cause thereof, by vigorous war, and all apt and efficient means, to sent delegates to a convention to assemble at Baltimore on Tuesday, the 7th day of June, 1864, at 12 o'clock, noon, for the purpose of presenting candidates for the offices of President and Vice President of the United States. Each State having a representation in Congress will be entitled to as many delegates as shall be equal to whice the number of electors to which such State is settiled The undersigned, who by original appointment, or subse-

will be entitled to as many delegates as shall be equal to twice the number of electors to which such State is entitled in the Electoral College of the United States.
EDWIN D. MORGAN, New York, Chairman. CHARLES J. GILMAN, Maine.
E. H. ROLLINS, New Hampshire.
J. Z. GOODRICH, Massachusetts.
THOMAS G. TURNER, Rhode Island.
GIDEON WELLES, Connecticut.
DENNING DUER, New Jersey.
EDWARD MCPHERSON, Pennsylvania.
N. B. SMITHERS, Delaware.
J. F. WAGNER, Maryland.
THOMAS SPOONER, Ohio.
H. S. LANE, Indiana. THOMAS SPOONER, Ohio.
H. S. LANE, Indiana.
SAMUEL L. CASEY, Kentucky.
E. PECK, Illinois.
HERBERT M. HOXIE, Iowa.
AUSTIN BLAIR, Michigan.
CARL SCHURZ, Wisconsin.
W. D. WASHBURN, Minnesota.
CORNELLUS COLE, California.
YM. A. PHILLIPS, Kansas.
O. H. IRISH, Nebraska.
JOS. GERHIARDT, District of Columbia.
Grox, February 22, 1864. WASHINGTON, February 22, 1864.

The Convention was called to order by the Chairman of the Executive Committee, Senator Morgan, of New York, who said :

MEMBERS OF THE CONVENTION: It is a little more than MEMPERS OF THE CONVENTION: It is a little more than eight years since it was resolved to form a national party, to be conducted upon the principles and policy which had been established and maintained by those illustrious states-men—George Washington and Thomas Jefferson. A con-vention was held in Philadelphia under the shade of the trees that surround the Hall of Independence, and our condidates Meaner Demonstrate the state and trees that surround the Harl of Interpendence, and our candidates, Messrs. Fremont and Dayton, were there nomi-nated, who were to espouse our cause and maintain it; but the State of Pennsylvania gave its electoral vote to James Buchanan, and the election in 1856 was lost. Nothing daunted by defeat, it was immediately determined to fight ment this further survey and ender our outputs without upon this line—[applausc]—not only all summer, but four summers and four winters: and, in 1860, the party banner was again unfurled with the names of Abraham Lincoln [enthusiastic applause] and Hannibal Hamlin [renewed applause] inscribed thereon. This time it was successful. [Analows] [Applause.] But with success came rebellion; with rebel-

lion of course came war, a terrible civil war, which has continued up to the period when it becomes necessary uncontinued up to the period when it becomes necessary un-der our Constitution to prepare for another Presidential election; and it is for this highly responsible purpose that you are to-day assembled. It is not my duty nor my pur-pose to indicate any general plan of action for this conven-tion, and yet I trust I may be permitted to say, that, in view of the dread realities of the past, and of what is pass-ing at this moment, the fact that the bones of our soldiers are bleaching in every State of this Union, and with a knowledge of the further fact that this has all been caused by slavery, the party of which you, gentlemen are the hon-Knowledge of the further lack that takes that all been called by slavery, the party of which you, gentlemen, are the hon-ored representatives, will fall short of accomplishing its great mission unless, among its other resolves, it shall de-clare for such amendment of the Constitution as will posi-tively prohibit African slavery in the United States. [Most with mission and the second are here the whole semanth rising to its feet and cheering lustily, the ladies waving their handkerchiefs, and the men their hats and canes.]

Order being restored, Senator Morgan proceeded.

On behalf of the National Committee, I now propose for temporary president of the convention, Robert J. Breckin-ridge, of Kentucky. [Loud applause.]

On being introduced, Dr. Breckinridge, who was most enthusiastically received, said :

GENTLEMEN OF THE CONVENTION : You cannot be more sen-sible than I am, that the part which I have to perform here to-day is merely a matter of form ; and, acting upon the principle of my whole life, when the soggestion was made to me from various quarters that it was in the mind of to me from various quarters that it was in the mind of many members of the convention to confer this distinction upon me, I honestly declined to accept of it, because I have never sought honors. I have never sought distinction; I have been a working man and nothing else. But two con-siderations led me to change my mind. One was personal sucrations for me to change my mind. One was personal to a class of men in the country, far too small for the good of the country—those men who, merely by their example, by their pens, and by their voice, try to do good, and all the more in perilous times, without regard to the rewards that may come. It was good to make such men understand but the divinction conformed upon upon of the humblest of the more in perilous times, without regard to the rewards that may come. It was good to make such men understand by the distinction conferred upon one of the humblest of their class, that they were men that the country would cherish and who would not be forgotten. [Applause.] The other motive related to yourselves and to the country at large; and it is good for you-- it is good for every nation, every State, and every party--to cherish all generous im-pulses, to follow all noble instincts; and there are none more generous and none more noble than to purge your-selves of all self-seekers and betrayers, and to confer your favors, if it be only in the mere form, upon those who are worthy to be trusted, and ask nothing more. [Applause.] Now according to my convictions of propriety, having said this, I should say nothing more. [Ories of "go on."] But it has been inimated to me from may quarters, and in a way which I cannot disregard, that I should disap-point the wishes of my friends, and perhaps the just expec-tations of the Convention, if I did not as briefly, and yet as precisely as I could, say somewhat upon the great matters which have brought us here. Therefore, in a very few words, and as plainly as I can, I will endeavor to draw your attention to one and another of these great matters in which we are all engaged. In the first place, mathing can be more plain than the

which we are all engaged.

In the first place, nothing can be more plain than the fact that you are here as the representatives of a great na-tion—voluntary representatives chosen without forms of law, but as really representing the feelings, the principles, and if you choose, the prejudices of the American people, 402

403

as if it were written in laws and already passed by votesfor the man that you will nominate here for the Presidency of the United States, and ruler of a great people in a great cricis, is just as certain I suppose to become that ruler as anything under heaven is certain before it is done. [Prolonged cheering.] And, moreover, you will allow me to say, though perhaps it is hardly strictly proper that I should-but as far as I know your opinions I suppose it is just as certain now before you utter it whose name you will utter, and which will be responded to from one cud to the other of this nation, as it will be after it has been uttered and recorded by your sceretary. Does any man doubt that this Convention intends to say that A braham bincoln shall be the nominee? [Great applause.] What I wish, however, to call your attention to is the grandeur of the mission upon which you are met, and therefore the dignity and solemnity, earnestness and conscientiousness with which, representing one of the greatest and certainly one of the first people of the world, you ought to discharge these duties. [Applause.] Now, besides the nomination of President and Vice Pres-

Now, besides the nomination of President and Vice President, in regard to which second office I will say nothing, because I know there is more or less difference of opinion among you; but besides these nominations, you have other most solemu duties to perform. You have to organize this party thoronghly throughout the United States. You have to put it in whatever form your wisdom will suggest that will unite all your wisdom, energy, and determination to gain the victory which I have a already said was in our power. More than that, you have to lay down with clearness and precision the principles on which you intend to carry out this great political contest and prosecnte the war which his before us if we succed. Plainly—not in a double sense—briefly—not in a treatise—with the dignity and precision of a great people to utter, by its representatives, the political principles by which they intend to live, and for the sake of which they are willing to die. So that all nen everywhere may understand precisely what we mean, and Ly that furrow so deeply and clearly that while every man who is worthy to associate with freemen may see it and pass over it, every man who's unworthy may be cither unable to pass it or may be driven far from us. We want

Now, among these principles, if you will allow me to say it, the first and most distinct is, that we do not intend to permit this nation to be destroyed. [Applause.] We are a nation—no doubt a peculiar one—a nation formed of States, and no nation except as these States form it. And these States are no States except as they are States in that nation. They had no more right to repudiate the nation than the nation has to repudiate them. None of them had even the nation has to reput ate them. Note of them had even the shadow of a right to do this, and God helping us, we will vindicate that truth so that it shall never be disputed any nore in this world. [Applause.] It is a fearful alternative that is set before us, but there are great compensations for Those of you who have alluded to this subject know, or it. ought to know, that from the foundation of the present Government, before and since our present Constitution was formed, there have always been parties that had no fuith in our Government. The men that formed it were doubtful of its success, and the men that opposed its formation did not desire its success. And I am bold to say, without detaining you on this subject, that with all the outcry about our violations of the Constitution, this present living genera-tion and this present Union party are more thoroughly devoted to that Constitution than any generation that has ever lived under it. [Applause.] While I say that, and solemnly believe it, and believe it is capable of the strongest proof, I may also add that it is a great error which is being propagated in our land, to say that our national life depends increiv upon the sustaining of that Constitution. Our fathers made it, and we love it. We intend to maintain it. But if it suits us to change it we can do so. [Applause.] And when it suits us to change it we will change it. [Ap-plause.] If it were torn into ten thousand pieces the nation would be as much a nation as it was before the Constitution was made-a nation always that declared its independence as a united people, and lived as a united people independence as a united people, and lived as a united people until now—a nation independent of all particular institu-tions under which they lived, and capable of modelling them precisely as their interests require. We ought to have it distinctly understood by friends and enemies that while we love that instrument we will maintain it, and will, with undoubted certainty, put to death friend or foe who undertakes to trample it under foot; yet, beyond a doubt, we will reserve the right to alter it to suit ourselves from time to time and from generation. [An. from time to time and from generation to generation. [Applause.] One more idea on that subject. We have incor-porated in that instrument the right of revolution, which gives us, without a doubt, the right to change it. It never existed before the American States, and by the right to change there is no need of rebellion, insurrection, or civil

war, except upon a denial of the fundamental principles of all free governments-that the major part must rule; and there is no other method of carrying on society, except that the will of the majority shall be the will of the whole—or that the will of the minority shall be the will of the whole. So that, in one word, to deny the principles I have tried to state is to make a dogmatic assertion that the only form of government that is possible with perfect liberty and acknowledged by God is a pure and absolute despotism. The principles therefore which 1 am trying to state before you are principles which, if they be not true, freedom is impossible, and no government but one of pure force can impossible, and no government but one of pure force can exist or ought to endure among men. But the idea which I wished to carry out, as the remedy for these troubles and sorrows, is this: Dreadful as they are, this fearful truth runs through the whole history of mankind, that whatever else may be done to give stability to authority, whatever else may be done to give perpetuity to institutions-however wise, however glorious, practicable, and just may be the philosophy of it—it has been found that the only enduring, the only imperishable cement of all free institutions, has been the blood of traitors. No Government has ever been built upon imperishable foundations which foundations were not laid in the blood of traitors. It is a fearful truth, but we had as well avow it at once, and every lick you strike, and every rebel you kill, every battle you win, dreadful as it is to do it, you are adding, it may be, a year-it may be ten years-it may be a contury-it may be ten centuries to the life of the Government and the freedom of

Continues to an other applause.] Now, passing over that idea—passing over many other things which it would be right for me to say, did the time serve and were this the occasion, let use add—you are a Union party. [Applause.] Your origin has been referred to as having occurred eight years ago. In one sense it is true. But you are far older than that. I see before me not only primitive Republicans and primitive Abolitonists, but I see also primitive Democrats and primitive Whigs—primitive Americans, and, if yon will allow me to say so, I myself arn here, who all my life have been in a party to myself. [Laughter and applause.] As a Union party I will follow you to the ends of the earth and to the gates of death. [Applause.] But as an Abolition party—as a Republican American party, I will not follow you one foot. [Applause.] But it is true of the mass of the American people, however you may divide and scatter while this war lasts, while the country is in peril, while you call yourselves as you do in the erals of the Carth and in my judgment, one of the greatest errors that has been committed by our administration of the Federal Government, the Chief of which we are about to nominate for another term of office—one of the arrers has been to believe that we have succeeded where ye have not succeeded, and to act in a manner which is precisely as if we had succeeded. You will not, you cannot, succeed until you have mitterly broken up the military power of these people. [Applause.]

Jewin of these people. Lappanes, I will not detain you upon these incidental points, one of which has been made prominent in the remarks of the excellent Chairman of the National Committee. I do not know that I would be willing to go so far as probably he would. But I cordially agree with him in this—I think, considering what has been done about slavery, taking the thing as it now stands, overlooking altogether, either in the way of condemnation or in the way of approval, any act that has brought us to the point where we are, but believing in my conscience and with all my heart, that what has brought us where we are in the matter of slavery, is the original sin and folly of treason and secession, because you remember that the Chicago Convention itself was understood to say, and I believe it virtually did explicitly say, that they would not touch slavery in the States. Leav go further than the clicago Convention in the way we came where we are, on that particular point, we are prepared to go further than the original Republicans themsolves were prepared to go. We are prepared to demand not only that the whole territory of the United States shall not be made slave, but that the General Government of the American people shall do one of two things—and it appears to me that there is nothing else that can be dome—either to use the whole power of the Government, both the war power and the peace power, to put slavery as nearly as possible back where it was—for, although that would be a fearful state of society, it is better than anirely; or else to use the whole power of the Government, both do war and peace, and all the practical power that the people of the United States will give them to exterminate and extinguish slavery. [Prolonged applause.]

ery. [Prolonged applause.] I have no hesitation in saying for myself that if I were a pro-slavery man, if I believed this institution was an ordinance of God, and was given to man, I would unhesitatingly join those who demand that the Government should be put back where it was. But I am not a pro-slavery man-I never was: I unite myself with those who believe it is contrary to the brightest interests of all men and of all governments, contrary to the spirit of the Christian religion, and incompatible with the natural rights of man. I join myself with those who say away withit forever; [appRusse:] and I fervently pray God that the day may come when throughout the whole land every man may be as free as you are, and as capable of onjoying regulated liberty. [Prolonged applause.]

Integed applause.] I will not detain you any longor. One single word you will allow me to say in behalf of the State from which I come, one of the smallest of the thousands of Israel. We know very well that our eleven votes are of no consequence in the Presidential election. We know very well that in our present unh-ppy condition, it is by no means certain that we are here to-day representing the party that will east the majority of the votes in that unhappy State. I know very well that the sentiments which I am uttering will cause me great odium in the State in which I was born, which I love, where the bones of two generations of my ancestors and some of my children are, and where very soon I shall lay my own. I know very well that my colleagues will incur odium if they indorse what I say, and they, too, know it. But we have put our faces toward the way in which We intend to go, and we will go in it to the end. If w<sub>2</sub> are to perish, we will perish in that way. All I have to say to you is, help us if you can; if you cannot, believe in your hearts that we have fid like men.

Rev. J McKendree Reilly, of the Methodist Episcopal church, offered a prayer, when those States which are represented in Congress were called for lists of delegates.

At the evening session of Tuesday the permanent organization was made, with Hon. WIL-LIAM DENNISON, of Ohio, as President. On taking the chair, he said:

I thank you for the honor you have conferred upon me, and while I will bring to the discharge of the duties of the chair little experience in parliamentary rules, it will be my pleasure, as my duty, to spare no effort in contributing, to the extent of my ability, to the facilitating of the business of the Convention, and securing such results from your deliberations as will meet the loyal expectations of the contry.

We meet here as representatives of the true friends of the Government and of impartial liberty—of that large portion of the people who gratefully appreciate the unmatched blessings which flow from our institutions well administered, and reject any form of human enslavement, not in punishment of crime, as no less incompatible with the rights of humanity than with the genius and the peaceful workings of republican government. [Prolonged applause.] In no sense do we meet as menbers or representatives of

In no sense do we meet as members or representatives of either of the old political parties which bound the people, or as the champions of any principle or doctrine peculiar to either. The extraordinary condition of the country since the outbreak of the rebellion has, from necessity, taken from the issues of these parties their practical significance, and compelled the formation of substantially new political organizations; hence the origin of the Union party—if party it can be called—of which this Convention is for the purpose of its assembling the accredited representatives, and the only test of membership in which is an unreserved, unconditional loyalty to the Government and the Union.

commona royary to the covernment and the Onion. Let me congratulate you upon the favorable auspices of your meeting. While the deepest anxiety is felt by all patrotic men as to the result of the war unjustifiably forced upon the Government by the bad, ambitious men and their deceived followers in the rebellions States, and the country is filled with distress and mourning over the loss of so many of our brave men who have fallen in battle, or died in hospitals from wounds received in defence of the constitutional authorities of the Government, we yet have, in what has been accomplished towards the suppression of the rebellion and the extinguishment of its cause—in the heroic deeds of our noble armies and gallant navy—in the renewal of the partiotism of the country that almost seemed to be paralyzed under the influence of our national prosperity—in the unprecedented generosity of the people, awakened by the wants of the Government and the necessities of its defenders—much, very much of the highest felicitation, and for which the country is grateful to Almighty God. [Applause.]

Almighty God. [Applause.] And may I not add to these causes of congratulation the formation of the political organization of which this Convention is a representative, which has so nobly sustained the Government in its efforts to put down the rebellion, and to the complete accomplishment of which its energies are

consecrated; the patriotic harmony that has marked our assembling and will characterize all our proceedings, and presenting that harmony which will display itself in the unminous nomination for the Presidency of the United States of the wise and good man whose unselfish devotion to the country, in the administration of the Government, has secured to him not only the admiration, but the warmest affection of every friend of constitutional liberty? [Applause.]

I need not remind you of the very grave responsibilities that devolve upon you as members of this convention. The loyal people of the country have authorized and expect you to renew on their part the pledge of their faith to support the Government, in the most vigorons prosecution of the war, to the complete suppression of the rebellion, regardless of the time or the resources required to that end, and they equally expect and call upon you to declare the cause and the support of the rebellion to be slavery, which, as well for its treasonable offences against the Government as for its incompatibility with the rights of humanity and the permanent peace of the country, must, with the termination of the war, and as much speedic as possible, be made to cease forever in every State and Territory of the Union. But I must not refer to other subjects of interest that will challenge your attention.

Let me repeat my thanks for your expressions of confidence in me in having selected me to preside over your deliberations. [Applause.]

#### REPORT OF COMMITTEE ON CREDENTIALS.

Mr. PRESTON KING, of New York, submitted the report of the majority committee; which was substantially as follows:

1st. That the delegations from the States of Maine, New Hampshire, Massachusetts, Connecticut, Vermont, Rhodo Island, New York, New Jersey, Pennsylvania, Delaware, Maryland, Ohio, Kentucky, Indiana, Illinois, Iowa, Minnesota, Oregon, California, Kansas, and West Virginia were all regular, and are admitted to seats with all the rights and privileges of members, except one district of Pennsylvania, which had elected four instead of two members. The committee admit the two who received the largest number of votes as delecates, and the other two as alternates.

2d. That there being two delegations from the State of Missouri, claiming seats, the committee recommend that those styling themselves the Union Radical Delegation be awarded the seats. [Applause and cheering.]

awarded the seats. [Applause and cheering.] 3d. That the delegates from Virginia, Tennessee, Louisiana, and Arkansas be admitted to all the privileges of the floor, except that of voting. 4th. That the delegations from the Territories and tho

4th. That the delegations from the Territories and the District of Columbia be admitted to seats and all the privileges except that of voting.

5th. That the persons presenting themselves as delegates from the State of South Carolina are not entitled to the rights of delegates on the floor.

Mr. W. E. STEVENSON, of Virginia, and Mr. HIRAM SMITH, of Oregon, made a minority report, and recommended that the delegates from the States of Virginia, Louisiana, Arkansas, Kansas, Tennessee, and Florida, and from all the Territories, be admitted, with the right to vote.

Mr. A. H. INSLEY, of Kansas, made a report arguing that, especially in the cases of the Territories of Nebraska, Colorado, and Nevada, the delegates be admitted with the right to vote.

That part of the report of the majority relating to the uncontested seats was then adopted.

Mr. KING, of New York, offered a substitute covering three points in report of the majority:

1st. He proposed to admit both of the Missouri delegations, and that where they agree they cast the vote to which the State is entitled; where they disagree, the vote of tho State shall not be cast.

24. He proposed to give all the delegates admitted all the rights and privileges of delegates, without exception; but that the District of Columbia and the Territories should have but two votes each, and that no State, District, or Territory should cast more votes than it has delegates present in the Convention.

A division of the question was called,

When Mr. KING's amendment relative to Missouri was lost; and the report of the committee, on that point, was adopted, by States, and as follows:

· Aves.	Nays.
Maine	õ
New Hampshire 10	Ō
Vermont 10	Ō
Massachusetts 24	ŏ
Rhode Island 8	Ő
Connecticut	Ó
New York 66	Ő
New Jersey 14	ŏ
Pennsylvania 49	ŝ
Delaware 6	õ
Maryland 14	ŏ
Kentucky 21	i
Ohio 42	õ
Indiana	õ
Illinois	ŏ
Michigan 16	ŏ
Wisconsin	ŏ
Iowa	ŏ
Minnesota 8	ŏ
California 10	ŏ
Oregon 6	Ó
West Virginia 10	Ó
Kansas 6	ŏ
	-
440	4

The question was still further divided, and a vote then taken upon admitting the fifteen delegates from Tennessee, with the right to vote; which was agreed to, by States, as follows:

For.	Against.
Maine 3	11
New Hampshire 0	10
Vermont	8
Massachusetts 0	24
Rhode Island 2	6
Connecticut 10	$\tilde{2}$
New York	õ
New Jersey 14	ŏ
Pennsylvania	21
Delaware 1	4
Maryland 1	13
Missouri	3
Kentucky 4	18
Ohio	Ĩõ
Indiana	2
Illinois	õ
Michigan	14
Wisconsin 15	1
Iowa	7
Minnesota 1	7
California	ò
Oregon	ŏ
West Virginia 10	ŏ
Kansas	ŏ
	0
Total	151
100000000000000000000000000000000000000	191

The delegates from Louisiana and Arkansas were then admitted, by States, as follows:

, <b>,</b> , , ,	For.	Agains
Maine	3	11
New Hampshire	0	10
Vermont.		5
Massachusetts	0	24
Rhode Island	. 1	7
Connecticut	10	2
New York	61	$^{2}_{3}$
New Jersey	11	ŏ
Pennsylvania	. 5	46
Delaware	" õ	5
Maryland	ĩĩ	13
Missouri	17	5
Tennessee	15	ő
Kentucky	12	10
Ohio	42	10
Indiana		4
Illinois	20	õ
Michigan	10.	6
Wisconsin	15	1
Iowa	14	2
Minnesota	0	8
California		0 4
Oregon		Ô
West Virginia	10	0
Kansas	10:	0
		0
Total	307	167
		101

The portions of the majority report, as thus amended, were then agreed to.

The Delegates from Nebraska, Colorado, and Nevada, were then admitted with the right to vote.

And the balance of the report-admitting the Delegates from Virginia and Florida, without the right to vote, rejecting the Delegates from South Carolina, and admitting the Delegates from the remaining Territories without the right to vote, was adopted.

This PLATFORM was then adopted unanimously, as reported by Mr. Raymond, of New York, Chairman of the Committee:

Resolved, That it is the highest duty of every American citizen to maintain against all their enemies the integrity of the Union and the paramount authority of the Constitu-tion and the paramount authority of the Constitu-tion and the laws of the United States; and that, laying aside all differences and political opinions, we pledge ourselves as Union men, animated by a common sentiment, and aim-ing at a common object, to do everything in our power to aid the Government in quelling by force of arms the rebel-lion paw radius acciment the uthority and in binder the In now raging against its authority, and in bringing to the punishment, due to their crimes, the rebels and traitors arrayed against it. [Prolonged applause.] *Resolved*. That we approve the determination of the Gov-ernment of the United States not to compromise with rebels.

or to offer them any terms of peace except such as may be based upon an "unconditional surrender" of their hostilibased upon an "unconditional surrender" of their nostil-ty and a return to their just allegiance to the Constitution and the laws of the United States; and that we call upon the Government to maintain this position and to prosecute the war with the utmost possible vigor to the complete suppression of the rebellion, in full reliance upon the self-constitues the articitient the luminizations and the sede sacrifices, the patriotism, the heroic valor, and the undy-ing devotion of the American people to their country and

Resolved, That as slavery was the cause and now consti-tutes the strength of this rebellion, and as it must be always and everywhere hostile to the principles of republican government, justice and the national safety demand its utter and complete extirpation from the soil of the Republic, (applause.) and that we uphold and maintain the acts and proclamations by which the Government, in its own de-fense, has aimed a death-blow at this gigantic evil. We are in favor, furthermore, of such an amendment to the Constitution, to be made by the people in conformity with its provisions, as shall terminate and forever prohibit the existence of slavery within the limits or the jurisdiction of the United States. [Applause.] Resolved, That the thanks of the American people are due

to the soldiers and sailors of the army and the navy [ap-plause] who have periled their lives in defence of their country, and in vindication of the henor of the flag; that the nation owes to them some permanent recognition of their patriotism and their valor, and ample and permanent provision for those of their survivors who have received disabling and honorable wounds in the service of the country ; and that the memories of those who have fallen in its defense shall be held in grateful and everlasting remem-

brance. [Loud applanse.] Resolved, That we approve and applaud the practical wis-dom, the unselfish patriotism and unswerving fidelity to 'the Constitution and the principles of American Liberty with which Abraham Lincoln has discharged, under circum-stances of unparalleled difficulty, the great duties and re-sponsibilities of the Presidential office; that we approve and inderse as demanded by the emergence and essential to indorse, as demanded by the emergency and essential to the preservation of the nation, and as within the Constitution, the measures and acts which he has adopted to defend the nation against its open and secret foes ; that we approve especially the Proclamation of Emancipation, and the em-ployment as Union soldiers of men heretofore held in slavery (applause ;) and that we have full confidence in his determination to carry these and all other constitutional measures essential to the salvation of the country into full and complete effect.

Resolved, That we doom it essential to the general welfare Resolved, That we deem it essential to the general welfare that harmony should prevail in the national conneils, and we regard as worthy of public confidence and official trust those only who cordially indorse the principles proclaimed in these resolutions, and which should characterizz the ad-ministration of the Government. [Applause,] *Resolved*, That the Government owes to all men employed in its armies, without regard to distinction of color, the full protection of the laws of war, [applause,] and that any violations of these laws or of the usages of civilized nations in the times of war by the rebels now in arms should be

made the subject of full and prompt redress. [Prolonged

applause.] Resolved, That the foreign immigration which in the past Resolved, That the foreign immigration which in the past Accorded, that the of eight mining attort which the pase has added so much to the wealth and development of re-sources and increase of power to this nation, the asylum of the oppressed of all nations, should be fostered and encour-aged by a liberal and just policy. *Resolved*, That we are in favor of the speedy construction of the willowed to the Decific

of the railroad to the Pacific. *Resolved*, That the national faith pledged for the redemp-tion of the public debt must be kept inviolate, and that for this purpose we recommend economy and rigid responsibility in the public expenditures, and a vigorous and just system of taxation; that it is the duty of any loyal State to sustain the credit and promote the use of the na-

State to sustain the credit and promote the use of the na-tional currency. [Applause.] *Resolved*, That we approve the position taken by the Gov-ernment that the people of the United States can never re-gard with indifference the attempt of any European power to overthrow by force or to supplant by fraud the institu-tions of any republican government on the Western Conti-ent forward republican det thet they will not writted the nent, [prolonged applause,] and that they will view with extreme jealousy, as menacing to the peace and independ-ence of this our country, the efforts of any such power to obtain new footholds for monarchical governments, sustained by a foreign military force in near proximity to the United States. [Long-continued applause.]

RE-NOMINATION OF PRESIDENT LINCOLN.

The vote was taken by States, when all were found to have voted for Mr. LINCOLN, except Missouri-for General Grant.

Before the announcement of the result, Mr. HUME, of Missouri, moved that the nomination of ABRAHAM LINCOLN be declared unanimous. His delegation had been instructed to vote for General Grant, but he was now in favor of declaring the nomination already made to be unanimous.

The motion was agreed to amidst a furore of applause.

The president then announced the result, stating that ABRAHAM LINCOLN, of Illinois, was the unanimous choice of the Union National party of the country for the next Presidency.

States and Territòries.	Johnson.	Dickinson.	Hamlin.	Butler.	Rosscau.	Burnside.	Colfax.	Holt.	Tod.	King.
Maine New Hampshire Vermont Massaclusetts Rhode Island Connecticut New York New York	1 5  12 32 2	$     \begin{array}{c}             3 \\             1 \\           $	14 4 2 3 3 6	2 2 2 2 2 2 2		2	••••	2		
Pennsylvania Delaware Maryland Louisiana Arkansas Missouri Tennessee Kentucky Ohio Indiana Ilinois	2 7 10 2 15  26	6 11 7	52 1   32 16	20	21		•		1	
Michigan Wisconsin. Iowa Oregon. West Virginia. Kansas. Nebraska. Colorada.	2 16 5 10 2 3 6	10 3 5  2 1 6	10 4 5  2 1			•••	6		•••	1
Nevada	200	113	145	28	21	2	6	2	1	1

VOTE FOR VICE-PRESIDENT.

Several of the States changed their votes to Presidency.

JOHNSON, and the final result reached was as follows:

L	0110 1 5 .	-		
	Johnson.	Dickinson.	Hamlin.	
	Maine 14	0	0	
	New Hampshire 10	0	0	
	Vermont 10	0.	0	
	Massachusetts 21	3	0	
	Rhode Island 7	1	0	
	Connecticut 12	0	0	
	New York 66	0	0	
	New Jersey 14	0	0	
	Pennsylvania 52	Ó	Ó	
	Delaware 6	Ó	Ó	
	Maryland 14	Ó	Ó	
	Louisiana 14	Ó	Ō	
	Arkansas 10	Ó	Ó	
	Missouri 22	0	Ō	
	Tennessee 15	Ō	Ó	
	Kentucky 21	Ó	Ō	
	Ohio 42	Ó	0	
	Indiana	0	0	
	Illinois 32	0	0	
	Michigan 16	Ó	0	Ì
	Wisconsin 2	10	4	
	Iowa 16	0	0	
	Minnesota 0	3	5	
	California 10	Ó	0	
	Oregon 6	0	0	
	West Virginia 10	Ó	0	
	Kansas 6	0	Ō	
	Nebraska	0	0	
	Colorado	0	0	
	Nevada	Ō	õ	
	Total	17	9	

The PRESIDENT. Andrew Johnson, having received a majority of all the votes, is declared duly nominated as the candidate of the National Union Party for Vice President of the United States.

On motion of Mr. TREMAIN, of New York, the nomination of Andrew Johnson was declared unanimous.

After the transaction of some routine business, the convention adjourned.

THE PRESIDENT'S RECEPTION OF THE NOMINATION.

Thursday, June 9-The committee to ratify the nominees called upon the President, when the following proceedings took place:

Governor Dennison, president of the convention and chairman of said committee, addressed the President as follows:

MR. PRESIDENT: The National Union Convention, which MR. PRESIDENT: The National Union Convention, which closed its sittings at Baltimore yesterday, appointed a com-mittee consisting of one from each State, with myself as its chairman, to inform you of your unanimous nomination by that convention for election to the office of President of the United States. That committee, I have the hoffer of now informing you, is present. On its behalf, I have also the honor of presenting you with a copy of the resolutions or platform which were adopted by that convention, as expres-sive of its sense, and of the sense of the loval neople of the sive of its sense, and of the sense of the loyal people of the country which it represents; of the principles and the policy that should characterize the administration of the Government in the present condition of the country. I need not say to you, sir, that the convention, in thus unanimously not say to you, sit, that the convention, in this unanimously nominating you for re-election, but gave utterance to the almost universal voice of the loyal people of the country. To doubt of your triumphant election would be little short of abandoning the hope of the final suppression of the re-bellion, and the restoration of the authority of the Govern-root corr the incorrent States ment over the insurgent States.

ment over the insurgent States. Neither the convention nor those represented by that body entertained any doubt as to the final result. Under your administration, sustained by that loyal people and by our noble army and gallant navy, neither did the conven-tion nor do this committee doubt the speedy suppression of this most wicked and unprovoked rebellion. [A copy of the resolutions were here handed to the President.] I should add, Mr. President, it would be the pleasure of the committee to communicate to you, withiu a few days, through one of its most accomplished members, Mr. Curtis, of New York, by letter, more at length the circumstances under which you have been placed in nomination for the Presidency.

## THE PRESIDENT'S RESPONSE.

The President, taking the resolutions from his pocket where he had placed them, and unfolding the same, said :

MR. CHAIRMAN AND GENTLEMEN OF THE COMMITTEE : I will neither conceal my gratileation nor restrain the expression of my gratitude that the Union people through their convention, in the continued effort to save and advance the na-tion, have deemed me not unworthy to remain in my present position.

I know no reason to doubt that I shall accept the nomination tendered; and yet, perhaps, I should not declare definitely before reading and considering what is called the platform.

I will say now, however, I approve the declaration in favor of so amending the Constitution as to prohibit slavery throughout the nation. When the people in revolt, with a hundred days of explicit notice that they could within those days resume their allegiance without the overthrow of their institutions, and that they could not resume it afterwards, elected to stand out, such amendment to the Constitution as is now proposed became a fitting and necessary conclusion is now proposed because a mining into accessive production in the final success of the Union cause. Such alone can meet and cover all cavils. Now, the unconditional Union men, North and South, preceive its importance, and embrace it. In the joint names of Liberty and Union, let us labor to give it legal form and practical effect.

Same day, a delegation of the National Union League called upon the President, to congratulate him upon his re-nomination, to whom he made this reply:

GENTLEMEN: I can only say in response to the kind remarks of your chairman, as I suppose, that I am very grate-ful for the renewed confidence which has been accorded to me both by the Convention and by the National League. I am not insensible at all to the personal compliment there is in this, and yet I do not allow myself to believe that any Is in this, and yet 1 do not allow hyself to believe that any but a small portion of it is to be appropriated as a personal compliment. That really the Convention and the Union League assembled with a higher view—that of taking care of the interests of the country for the present and the great future—and that the part I am entitled to appropriate as a compliment is only that part which I may lay hold of as being the opinion of the Convention and of the League, that I am not entirely unworthy to be entrusted with the place which I have occupied for the last three years. But I do not allow myself to suppose that either the Convention or the League have concluded to decide that I am either the greatest or best man in America, but rather they have concluded that it is not best to swap horses while crossing the river, and have further concluded that I am not so poor a horse that they might not make a botch of it in trying to swap. [Laughter and applause.]

## To a delegation from Ohio he said :

#### SPEECH OF PRESIDENT LINCOLN.

GENTLEMEN: I am very much obliged to you for this compliment. I have just been saying, and as I have just said it. I will repeat it: The hardest of all speeches which I have to answer is a screnade. I never know what to say on such occasions. I suppose that you have done me this kindness in connection with the action of the Baltimore Convention, which has recently taken place, and with which, of course, I am very well satisfied. [Laughter and applause.] What we want still more than Baltimore Conventions or Presidential elections is success under General Grant. [Crics of "Good," and applause.] I propose that you constantly bear should that the support you owe to the brave officers and soldiers in the field is of the very first importance, and we should therefore bend all our energies to that point. Now, without detaining you any longer, I propose that you help me to close up what I am now saying with three rousing cheers for General Grant and the officers and soldiers under his command.

#### PRESIDENT LINCOLN'S ACCEPTANCE.

1

#### NEW YORK, June 14, 1864.

Hon. ABRAHAM LINCOLN: SR: The National Union Convention, which assembled in Baltimore on June 7, 1864, has instructed us to inform you that you were nominated with enthusiastic unanimity for the Presidency of the United States for four years from the 4th of March next.

The resolutions of the Convention, which we have already 1 had the bonor of placing in your hands, are a full and clear statement of the principles which inspired its action, and

which, as we believe, the grate body of Union men in the country heartily approve. Whether those resolutions ex-press the national gratitude to our soldiers and sailors; or the national scorn of compromise with Rebels, and conse-

quent dishonor; or the patriotic duty of union and success, whether they approve the Proclamation of Emancipation, whicher they approve the Proclamation of Edualicity attom, the constitutional amendment, the engloyment of former slaves as Union soldiers, or the solemn obligation of the Government promptly to redress the wrongs of every sol-dier of the Union, of whatever color or race; whether they declare the inviolability of the pledged faith of the nation, or offer the national hospitality to the oppressed of every bedress the declared by the barries of the Admitioned Bu land, or urge the union by railroad of the Atlantic and Pacific oceans; whether they recommend public economy and vigorous taxation, or assert the fixed popular opposition to the establishment by armed force of foreign monarchies in the innucliate neighborhood of the United States, or de-clare that those only are worthy of official trust who ap-

clare that those only are worthy of official trust who ap-prove unreservedly the views and policy indicated in the resolutions—they were equally hailed with the heartiness of profound conviction. Believing with you, sir, that this is the people's war for the maintenance of a Government which you have justly described as "of the people, by the people, for the people," we are very sure that you will be glad to know, not only from the resolutions themselves, but from the singular barrony and enthusiasm with which they were adouted. harmony and enthusiasm with which they were adopted, how warm is the popular welcome of every measure in the how warm is the popular welcome of every measure in the prosecution of the war, which is as vigorous, numistakable, and unfultering as the national purpose itself. No right, for instance, is so precious and sacred to the American heart as that of personal liberty. Its violation is regarded with just, instant, and universal jealousy. Yet in this hour of peril every faithful citizen concedes that, for the sake of national existence and the common welfare, individual liherty may, as the Constitution provides in the case of rebellion, be sometimes summarily constrained, asking only with painful anxiety that in every instance, and to the least detail, that absolutely necessary power shall not be hastily or unwisely exercised.

We believe, sir, that the honest will of the Union men of the country was never more truly represented than in this Convention. Their purpose we believe to be the overthrow of armed rebels in the field, and the security of permanent peace and union by liberty and justice under the Constitu-That these results are to be achieved amid cruel per tien. plexities, they are fully aware. That they are to be reached only by cordial unanimity of counsel, is undeniable. That good men may sometimes differ as to the means and the time, they know. That in the conduct of all human affairs the highest duty is to determine, in the angry conflict of is their sincere persuasion. They have watched your official course, therefore, with unflagging attention; and amid the bitter taunts of eager friends and the fierce denunciation of energies, now moving too fast for some, now too slowly for others, they have seen you throughout this tremendous contest patient, sagacious, faithful, just; leaning upon the heart of the great mass of the people, and satisfied to be moved by its mighty pulsations.

It is for this reason that long before the Convention met, the popular instinct had plainly indicated you as its candi-date; and the Convention, therefore, merely recorded the popular will. Your character and career prove your unwerving fidelity to the cardinal principles of American Liberty and of the American Constitution. In the name of that Liberty and Constitution, sir, we earnestly request your acceptance of this nomination. Reverently commend-ing our beloved country, and you, its Chief Magistrate, with all its brave sons who, on sea and land, are faithfully de-fording the good old the waring mours of course highly to the fending the good old American cause of equal rights, to the blessing of Almighty God,

We are, sir, very respectfully, your friends and fellowcitizens,

WILLIAM DENNISON, Ohio, Chairman. JOSIAII DRUMMOND, Maine.
THOMAS E. SAWYER, New Hampshire.
BRADLEY BARLOW, Vermont.
A. H. BULLOCK, Massachusetts.
A. M. GAVMELL, Rhode Island.
C. S. BUSHINELL, Connecticut.
G. W. CURTIS, New York.
W. A. NEWELL, New Jersey.
HENRY JOHNSON, Pennsylvania.
N. B. SMITHERS, Delaware.
W. L. W. SEABROOK, Maryland.
JOHN F. HUME, Missouri.
G. W. HITE, Kentucky.
E. P. TYFFE, Ohio.
CYRUS M. ALLEN, Indiana.
W. BUSHINELL, Illinois.
L. P. ALEXANDER, Michigan.
A. OLIVER, Iowa.
THOMAS SIMPSON, Minnesota. WILLIAM DENNISON, Ohio, Chairman. JOIN AS SIMPSON, Minnesota, JOIN BIDWELL, California. THOMAS II. PEARNE, Oregon,

LEROY KRAMER, West Virginia.

A. C. WILDER, Kansas, A. C. WILDER, KARSAS, M. M. BRIEN, Tennessee,
J. P. GREVES, Nevada.
A. A. ATOCHA, Louisiana.
A. S. PADDOCK, Nebraska, vALENTINE DELL, Arkansas.
JOHN A. NYE, Colorado. A. B. SLOANAKER, Utah.

EXECUTIVE MANSION, WASHINGTON, June 27, 1864. BOD. WILLIAM DENNISON and others, a Committee of the Na-tional Union Convention: GENTLEMEN: Your letter of the 14th instant formally po-

tigning me that I have been nominated by the convention you represent for the Presidency of the United States for four years from the 4th of March next, has been received. The nomination is gratefully accepted, as the resolutions of the

e convention, called the platform, are heartily approved. While the resolution in regard to the supplanting of re-While the resolution in regard to the supplanting of re-publican government upon the western continent is fully concurred in, there might be misunderstanding were I not to say that the position of the Government in relation to the action of France in Mexico as assumed through the State department and indorsed by the Convention, among the measures and acts of the Executive, will be faithfully upintoined to long as the the state of feater chellion to rest. maintained so long as the state of facts shall leave that position pertinent and applicable.

I an especially gratified that the soldier and the seaman were not forgotten by the Convention, as they forever must and will be remembered by the grateful country for whose salvation they devote their lives.

Thanking you for the kind and complimentary terms in which you have communicated the nomination and other proceedings of the convention, I subscribe myself,

Your obedient servant.

6 mm

ABRAHAM LINCOLN.

HON. ANDREW JOHNSON'S LETTER OF ACCEPTANCE.

NASHVILLE, TENN., July 2, 1864. H n. WM. DENNISON, Chairman, and others, Committee of the National Union Convention:

**GENTLEMEN:** Your communication of the 9th ult., inform-ing me of my nomination for the Vice Presidency of the United States, by the National Union Convention, held at Baltimore, and enclosing a copy of the resolutions adopted by that body, was not received until the 25th ult.

A reply on my part had been previously made to the ac-A reply of my part has been previously made to the the tion of the Convention in presenting my name, in a speech delivered in this city, on the evening succeeding the day of the adjournment of the Convention, in which I indicated my acceptance of the distinguished honor conferred by that by acceptance of the dissinguished noise contented by that body, and defined the grounds upon which that acceptance was based, substantially saying what I now have to say, From the comments made upon that speech by the various presses of the country to which my attention has been di-

presses of the country to which my attention has been an rected, I considered it to be regarded as a full acceptance. In view, however, of the desire expressed in your com-munication, I will more fully allude to a few points that have be en heretofore presented. My opinions on the lead-ing questions at present agitating and distracting the public wind evaluate the exceedible in a few points. mind, and especially in reference to the rebelliou now being waged against the Government aud authority of the United States, I presume, are generally understood. Before the southern people assumed a belligerent attitude, (and fre-quently since.) I took occasion most frankly to declare the views I then entertained in relation to the wicked purposes of the southern politicians. They have since undergone but little, if any change. Time and subsequent events have rather confirmed than diminished my confidence in their correctness.

At the beginning of this great struggle I entertained the same opinion of it I do now, and in my place in the Senate I denotinced it as treason, worthy the purisher in the Senate I denotinced it as treason, worthy the purishment of death, and warned the Government and people of the impending danger. But my voice was not heard or counsel heeded until it was too late to avert the storm. It still continued until it was too late to avert the storm. It still continued to gather over us without molestation from the authorities at Washington, until at length it broke with all its fury upon the country. And now, if we would save the govern-mentfrom being overwhelmed by it, we must meet it in the true spirit of patrioism, and bring traitors to the punish-ment due their crime, and by force of arms crush out and subdue the last vestige of rebel authority in every State. I felt then as now that the destruction of the government west deliverator heteration of the punctual out desire was deliberately determined upon by wicked and designing conspirators, whose lives and fortunes were pledged to carry conspirators, whose lives and fortunes were pledged to carry it out, and that no compromise, short of an unconditional recognition of the independence of the southern States, could have been or could now be proposed which they would accept. The clamor for "southern rights," as the rebel journals were pleased to designate their rallying cry, was not to secure their assumed rights in the Union and ander the Constitution, but to disrupt the government and truly,

establish an independent organization based upon slavery, which they could at all times control.

which they could at all times control. The separation of the Government has for years been the cherished purpose of the southern leaders. Baffled, in 1832, by the stern, patriotic heroism of Andrew Jackson, they sullenly acquiesced, only to mature their diabolical schemes, and await the recurrence of a more favorable opportunity to execute them. Then the pretext was the tariff, and Jackson, after folling their schemes of nullification and dis-puted with averback presenting. union, with prophetic perspicacity, warned the country against the renewal of their efforts to dismember the Government.

In a letter dated May 1, 1823, to the Rev. A. J. Crawford, after demonstrating the heartless insincerity of the south-

after demonstrating the heartless insincerity of the south-ern nullifiers, he said: "Therefore the tariff was only a pretext, and disunion and a southern confederacy the real object. The next pre-text will be the negro, or slavery question." Time has fully verified this prediction, and we have now not only "the negro, or slavery question." as the pretext, but the real cause of the rebellion, and both must go down together. It is vain to attempt to reconstruct the Union with the distracting element of slavery in it. Experience has demonstrated its incompatibility with free and republi-can governments, and it would be unwise and unjust longer to continue it as one of the institutions of the country to continue it as one of the institutions of the country. While it remained subordinate to the Constitution and laws of the United States I yielded to it my support, but when it became rebellious and attempted to rise above the Gov-ernment, and control its action, I threw my humble influence against it.

The authority of the Government is supreme, and will admit of no rivalry. No institution can rise above it, whether it be slavery or any other organized power. In our happy form of government all must be subordinate to the will of the people, when reflected through the Constitution and laws made pursuant thereto-State or Federal. This great principle lies at the foundation of every government, and cannot be disregarded without the destruction of the gov-ernment itself. In the support and practice of correct principles we can never reach wrong results, and by rigor principles we can never react wrong results, and by figor-ously adhering to this great fundamental truth the end will be the preservation of the Union and the overthrow of an institution which has made war upon and attempted the destruction of the government itself. The mode by which this great change—the emanipation

The mode by which this great change—the emancipation of the slave—can be effected, is properly found in the power to amend the Constitution of the United States. This plan is effectual, and of no doubtful authority: and while it does not contravene the timely exercise of the war power by the President in his emancipation proclamation, it comes stamped with the anthority of the people themselves, acting in accordance with the written rule of the supreme law of the land, and must, therefore, give more general suffaction the land, and must, therefore, give more general satisfaction and quictude to the distracted public mind.

By recurring to the distracted public mind. By recurring to the principles contained in the resolu-tions so unanimously adopted by the Convention, I find that they substantially accord with my public acts and opinions heretofore made known and expressed, and arc, therefore, most cordially indersed and approved; and the nomination, having been conferred without any solicitation on my part,

it is with the greater pleasure accepted. In accepting the nomination I might here close, but I cannot forego the opportunity of saying to my old friends of the Democratic party *proper*, with whom I have so long and pleasantly been associated, that the hour has now come when their tract party are not justly vinderate its description to and pleasantly been associated, that the hour has now come when that great party can justly vindicate its devotion to true democratic policy and measures of expediency. The war is a war of great principles. It involves the supremacy and life of the Government itself. If the rebellion triumphs free government North and South fails. If, on the other hand, the Government is successful, as 1 do not doubt, its destiny is fixed, its basis permanent and enduring, and its career of honor and glory just begun. In a great contest like this for the existence of free government the path of duty is patriotism and principle. Minor considerations and ouestions of administrative policy should give way to the duty is patriotism and principle. Minor considerations and questions of administrative policy should give way to the higher duty of first preserving the Government, and then there will be time enough to wrangle over the men and measures pertaining to its administration. This is not the hour for strife and division among our-selves. Such differences of opinion only encourage the enemy, prolong the war, and waste the country. Unity of exting and consentration of news rebord be our withword

action and concentration of power should be our watchword and rallying cry. This accomplished, the time will rapidly approach when their armies in the field—the great power approach when their armies in the field—the great power of the rebellion—will be broken and crushed by our gallant officers and brave soldiers, and ere long they will return to their homes and fresides to resume again the avocations of peace, with the proud consciousness that they have aided in the noble work of re-establishing upon a surer and more-permanent basis the great temple of American freedom. I am, gentlemen, with sentiments of high regard, yours truly, ANDREW JOHNSON.

#### The National Union League-The Platform Adopted.

BALTIMORE, June 8, 1864.

The following are the resolutions passed by the Grand National Council of the Union League of America, assembled at the New Assembly Rooms in this city. The injunction of secrecy has been removed :

1. Resolved, That we will support the Administration in the vigorous prosecution of the war to the complete and final suppression of the rebellion, and to this we pledge all our energies and efforts.

2. Resolved, That slavery, being the cause of the rebellion and the bound of union among traitors, ought to be abolished without delay, and it is the sense of this organization that slavery, in all its forms, should be prohibited by an amend-to the Federal Constitution.

3. Resolved, That we hereby approve of the principles involved in the policy known as the Monroe doctrine.

4. Resolved, That the confiscation acts of Congress should be promptly and vigorously enforced, and that homesteads on the lands confiscated under it should be granted to our soldiers and others who have been made indigent by the acts of traitors and rebels.

5. Resolved, That every person who hears arms in defence of the national flag, is entitled, without distinction of color or nationality, to the protection of the government he de-

fends, to the full extent of that government's power. 6. Resolved, That we hereby tender our thanks to the soldiers of the army and the sailors of the navy.

## The Cleveland Convention.

May 31-A convention of about three hundred and fifty persons, as reported, met in Cleveland, pursuant to sundry calls.

A CALL TO THE RADICAL MEN OF THE NATION.

Whereas a Convention has been called by certain parties favorable to changing the present Administration, and for the purpose of "counseling concerning the approaching Presidential election," to meet in the city of Cleveland, Ohio, on Thesday, the 31st of the present month; and whereas we are glad to learn that such a convention is to assemble, and having confidence that the objects of those issning the call are in unison with those of the radical men of the country;

Therefore, the notest signed, having been appointed by the "Central Fremont Club" of the city of New York, for that purpose, do hereby invite their radical fellow-citizens in purpose, do nerey invite their function entowentizens in every State, county, and town thronghout the country to meet them in the above-named Convention, on the said Tuesday, the 31st of this month, in order, then and there, to recommend the nomination of John C. Fremont for the Presidency of the United States, and to assist in organizing for his election.

The imbecileand vacillating policy of the present Admin-istration in the conduct of the war, being just weak enough to waste its men and means to provoke the enemy, but not strong enough to conquer the rebellion-and its treachery to justice freedom, and genuine democratic principles in its plan of reconstruction, whereby the honor and dignity of the nation have been sacrificed to conciliate the still existing and arrogant slave power, and to further the ends of an unscrupulous partisan ambition—call in thunder tones upon the lovers of justice and their country to come to the rescue of the imperiled nationality and the cause of impar-tial and universal freedom, threatened with betrayal and overthrow.

overthrow. The way to victory and salvation is plain. Justice must be throned in the seats of national legislation, and guide the executive will. The things demanded, and which we ask you to join us to render sure, are, the immediate extinc-tion of slatery throughout the voloel United States by Con-gressional action, the absolute equality of all men before the two, without regard to race or color, and such a plan of re-construction as shull conform entirely to the policy of free-dom for all, placing the political power alone in the hands of the lowal, and executing with vior the line for confiscatof the loyal, and executing with vigor the law for confiscal-ing the property of the rebels. Come, then, in formidable numbers, and let us take coun-

sel together, in this crisis of the nation's calamity, and, with one united effort, endeavor to redeem the country from slavery and war, that it may be consecrated to FREEDOM and PEACE FOREVER MORE. Men of God! Men of lumanity! Lovers of justice! Patriots and freemen! One and all, rally !!

Most respectfully, your fellow-citizens DAVID PLUMB,

EDWARD GILBERT,

FREDERICK KAPP, ERNEST KRACKOWIZER, WILLIAM J. DEMAREST Committee

NEW YORK, May 6, 1864.

The undersigned join in the foregoing call: Geo. B. Cheever, N. Y. Henry T. Cheever, Mass. Henry T. Cheever, Mass J. W. Alden, N. J. F. O. Irish, N. Y. William Goodell, N. Y. S. S. Jocelyn, N. Y. E. Cady Stanton, N. Y. Wm. F. Knowles, N. Y. W. H. Woodruff, N. J. C. Feromout, N. Y. W. H. Woourin, N. J.
 C. Fromont, N. Y.
 Ira H. Cobb, N. Y.
 Doct. H. Joslyn, N. Y.
 H. L. Green, N. Y.
 H. L. Green, N. Y.
 J. Warner, N. Y. David C. Harrington, N. A. S. Betts, N. Y. David Downs, N. Y. W. H. Bathaway, N. Y. T. C. Harrison, N. J. J. R. Johnson, Ya. James W. Vail, Wis, Elisha Galpin, Mich, B. A. Fay, Mich, A. J. Fay, Mich, Thomas C. Post, N. Y. Thomas C. Post, N. Y.

Pantaleon Candidus, N. Y. R. F. Hibbard, N. Y. Edmund Tuttle, Conn. Peter G. Tuttle, Conn. F. N. Bixby, Conn. James R. Surtliff, Conn. James Tuttle, Conn. James Tuttle, Conn. E. B. Hall, Conn. Edward 11. Tuttle, Conn. S. B. Hall, Cogn. George II. Scars, N. Y. Nathaniel R. Harris, N. Y. Doct. H. Joslyn, N. Y. H. L. Green, N. Y. T. O. Warner, N. Y. J. Henry Warner, N. Y. C. B. Smith, N. Y. J. Henry Warner, N. Y. E. M. Mason, Mich. Chas. A. Lane, N. Y. David C. Harrington, N. Y. Henry B. Harrington, N. Y. Bail C. Harrington, N. Y. Stoulen Betts, N. Y. Stoulen Betts, N. Y. Stephen Betts, N. Y. Wm. H. H. Downs, N. Y. C. S. Middlebrook, Conn. A. B. Pratt, Mich. Wm. Cunnning, Mich. Ira Chase, Mich. C. C. Foote, Mich. Elisha Hill, Mich. Robert Garner, Mich.

TO THE PEOPLE OF THE UNITED STATES.

After having labored ineffectually to defer as far as was in our power the critical moment when the attention of the people must inevitably be fixed upon the selection of a can-didate for the Chief Magistracy of the country; after hav-ing interrogated our conscience and consulted our duty as citizens, obeying at once the sentiment of a mature conviction and a profound affection for the common country, we feel ourselves impelled, on our own responsibility, to declare to the people that the time has come for all independent men, fedlons of their liberties and of the national great-ness, to confer together and unite to resist the swelling inness, to concer together and unite to result the sweining m-vasion of an open, shameless, and unrestrained patronage which threatens to engulf under its destructive wave the rights of the people, the liberty and dignity of the nation. Deeply impressed with the conviction that, in a time of

revolution, when the public attention is turned exclusively to the success of armies, and is consequently less vigilant of the public liberties, the patronage derived from the organization of an army of a million of men, and an administration of affairs which seeks to control the remotest parts of the country in favor of its supreme chief, constitute a danger seriously thratening to the stability of republican institutions, we declare that the principle of ono term, which has now acquired nearly the force of law by the con-sceration of time, ought to be inflexibly adhered to in the approaching election. We further declare that we do not recognize in the Balti-

more Convention the essential conditions of a truly National Convention. Its proximity to the centre of all the interested influences of administration, its distance from the centre of the country, its mode of convocation, the corrupting prac-tices to which it has been and inevitably will be subjected, do not permit the people to assemble there with any expectation of being able to deliberate at full liberty. Convinced, as we are, that in presence of the critical circumstances in which the nation is placed, it is only in the energy and good sense of the people that the general safety can be found, satisfied that the only way to consult it is to indicate a satisfied that the only way to consult it is to indicate a central position to which every one may go without too nuch expenditure of means and time, and where the assembled people, far from all administrative influence, may consult freely and deliberate peaceably with the presence of the greatest possible number of men whose known principles guarantee their sincere and enlightened devotion to the rights of the people and to the preservation of the true basis of republican government—we carnestly invite our fellow-citizens to unite at Cleveland, Ohio, on Tuesday, the 31st of May next, for consultation and concert of action in respect to the approaching Presidential election.

B. Gratz Brown, Mo. B. Gratz Brown, Mo. Stephen S. Foster, Mass. A. Van Antwerp, N. Y. Bird B. Chapman, Ohio. Ezra C. Andrews, Me. Henry A. Clover, Miss. Peter Engleman, Wis. Frederick Kapp, N. Y. Charles E. Moss, Mo. E. G. Parker, Me. Ernest Pruessing, Ill. Wm. D. Robinsin, Me. John S. Savery, N. Y. E. Cluseret.

410

Caspar Butz, Ill. George Field, N. Y. Edward Gilbert, N. Y. Peter Gillen, N. Y. Isaac W. Haff, N. Y. Wm. Herries, N. Y. James Ilill, Me. K. Heinzen, Mass. S. P. Dinsmore, D. C. And. Humbert, Pa. J. W. Alden, N. J. L. Sieboldt, Iowa. Wm. Morris Davis, Pa. E. M. Davis, Pa. Wm. F. Johnston, Pa.

Emil Pretorius, Mo. Nath. P. Sawyer, Pa. Ernest Schmidt, Ill. Ernest Schmidt, James Redpath, Mass. Walter H. Shupe, Ohio. Wm. 11, Smith, Me. P. W. Kenyon, N. Y. James Taussig, Mo. Ph. Stoppelrein, N. Y. Wm. II. Dwinelle, N. Y. Samuel Taylor. Jas. S. Thomas, Mo. F. Munch, Mo. a. J. Q. Westbrook, Mc. ton, Pa. J. F. Whipple, N. Y. THEO. OLSHAUSEN, Missouri, of the People's Committee.

#### TO THE PEOPLE.

Citizens of the United States who mean to uphold the Union, who believe that the rebellion can be suppressed with-Union, why believe that the recention can be suppressed what out infringing the rights of individuals or of States, who regard the extinction of slavery as among the practical effects of the war for the Union, and favor an amendment of the Federal Constitution for the exclusion of slavery, and of the rederat constitution for the exclusion of shavery, and who demand integrity and economy in the administration of government, are respectfully invited to meet in Mass Convention, at Cleveland, on Tuesday, the 31st day of May inst, for consultation and concert of action in respect to the approaching Presidential election.

bbrogening ricelandent en	
Lucius Robinson.	Charles F. French, Ky.
John Cochrane.	Rob. Kraus, Mo.
Andrew J. Colvin.	Hanson Breut, Mo.
Thomas B. Carroll.	J. B. Clairbour, Mo.
Edward Wade.	Wm Freel, Mo.
George W. Demers,	Charles H. Frost, Mo.
Ira Porter.	Thomas J. Riddle, Mo.
Brace Millerd.	Wm. L. Bookstaver, Mo.
Howard Holdridge.	Fred. L. Braden, Ill.
Francis G. Fiue.	Caspar O. Fitch, Ill.
Lemon Thomson.	Wallace Furman.
Charles Requa.	Frederick Smith.
Smith Requa.	Peter B. Lent, Ind.
Thomas P. White, Ky.	Andrew F. Butler.
Edward Cole, Ky.	Thomas Willks.
Francis F. Williams, Ky.	O. Whaley, Mass.
Smith Thompson, Ky.	Johnson Stemmer.
Leroy McArdle, Ky.	Alfred Moses,
William Bentley, Ky.	Leonard J. Timon.
John F. Smithers, Ky.	John F. Pendleton, N. J.
David S. Whiteley, Ky.	Patrick Clare.
Peter McCall, Ky.	Simon Munson.

#### LETTER FROM MRS. STANTON.

#### TO THE CENTRAL FREMONT CLUB:

MAY 14, 1864.

GENTLEMEN: To your call "to the radical men of the na-tion," taking it for granted you use "men" in its largest sense, I desire to append my name, and for the following reasons:

1. This is the only call ever issued for a political conven-tion, demanding the right of suffrage for the black man— that safeguard of civil liberty, without which emancipation

that sateguare of error needsy, ...... is a mockery. 2. When a body of men thus consecrate themselves to "freedom and peace," and declare their high resolve to found a republic on the eternal principles of justice, they have lifted politics into the sphere of morals and religion, and made it the duty of all true men and women to unite with them in building up the *New Nation*. Yours respectfully, E. CADY STANTON.

BOSTON, April 21.

## LETTER FROM WENDELL PHILLIPS.

JUDGE STALLO:

JUDGE STALLO: DEAR SIR: Since you asked my judgment as to the course to be taken in nominating a candidate for the Presidency, I have been requested to sign a call for a convention for that purpose, to meet at Cleveland, in May next. Let me tell you the national policy I advocate: Subdue the South as rapidly as possible. The moment territory comes under our flag reconstruct States thus: Confiscate and divide the lands of rebels; extend the right

of suffrage broadly as possible to whites and blacks; let the Federal Constitution prohibitslavery throughout the Union, and forbid the States to make any distinction among their citizens on account of color or race.

I shall make every effort to have this policy pursued. Believing that the present Administration repudiates it and is carrying us to a point where we shall be obliged either to acknowledge the Southern Confederacy or to reconstruct the Union on terms grossly unjust, intolerable to the masses, and sure soon to result in another war, I earnestly masses, and sure soon to result in another war, I earnestly advise an unpledged and independent convention, like that proposed, to consider public affairs and nominate for the Presidency a statesman and a patriot. Yours, faithfully, WENDELL PHILLIPS.

The Convention was called to order by Edward Gilbert, of New York, on whose motion Ex-Governor William F. Johnston, of Pennsylvania, was chosen temporary Chairman. Mr. B. H. Brooks, of California, and Mr. Walfe, of the District of Columbia, were chosen Secre-taries. A Committee on Credentials was proposed but not created. It was subsequently reported that the following States were repsented:

Ohio, Illinois, Massachusetts, New York, Iowa, Missonri, Michigan, Pennsylvania, Maryland, Wisconsin, Tennessee, Maine, Iudiana, New Hampshire, New Jersey, and the Dis-trict of Columbia.

The Committee on Permanent Organization reported officers, with General John Cochrane, of New York, as President.

Gen. Cochrane, on returning his thanks to the Convention, said:

tion, said: GENTLENEN OF THE CONVENTION: The formal routine of duty for the presiding officer, prescribes that he return set thanks for the honor conferred upon him. I am not dis-posed to follow in this path. The formality is too heartless for the solemnity and importance of the occasion. I as-sume my duties with cheerfulness and I trust that in this grand army of freedom I may well perform my humble part, and that that duty may be so performed that we may command the universal applause of all men. [Cheers.] I see before me representatives of the West, on which de-pends sommch of the interests and destinies of the country— of the great central region of the country—its support and sustenance. I look further to the East and see before me the commanions of my curry life, assembled now by a common sustenance. I look further to the East and see before me the companions of my early life, assembled now by a common motive from the devious paths in which the exigencies of politics had lead them—the War Democrats of the State of New York. [Applause.] For them party possesses no chains when it is not identified with principle. There are also those here, who while they wait auxiously the tidings from the shattered cohorts under the banner of freedom, still turn affectionately to a Fatherland on the other side of the raging sea. [Applause.] We need with such emotions suggested by the nelancholy vista of the past—such reflections upon the scenes of the present. We have come together regard-ing party as nothing—country as everything. Our national existence is at issue. Three years ago the question of national life fell like a timuderbolt at the feet

question of national life fell like a thunderbolt at the feet of the people and they sprang to arms, with a wild shout in of the people and they spring to arms, with a wild shout in which all faction, all party, went down. Everywhere was heard the steady tramp of armed men and the patriotism and power of the North hasswept on until the hour isalmost striking when time shall proclaim the rededs defeated and the Union triumphant. [Applause.] In this hour men should review the past and speculate as to the dangers and viciositudes of the future. For this purpose you assemble here—intending to support the army in the field, and at the same time to organize a great civil army, to fight for prinhere—intending to support the army in the field, and at the same time to organize a great civil army, to fight for prin-ples, and to save for all generations the precions legacy ob-tained for us by the sacrifices of the soldiers of the Union, [Applause.] The rebellion, it must be suppressed—The Union, it must be preserved. [Great applause.] But we shall allow no criticism of the Government which repre-sents ns all, shall cust no impediment in the way of our Union soldiers, shall entertain no thought unworthy of American citizens American citizens.

American citizens. The speaker then referred to the various discussions of the past. Since 1757 slavery has been the root of every politi-cal party of the country. The convention had assembled from all positions on this question, and now occupied one common ground. All are now united in demanding that slavery be destroyed and its last vestige wiped out. [Cheers.] In this connection the speaker referred to the War Dem-ocrats of New York—virtual and virtuous—denying that the convention at Streages tenses that have no sum demante-

the convention at Syracuse represented them, and denouncing that convention as a medley party of trading, sourcy politicians. If e solemnly declared before high heaven that, since the war broke out, he had never belonged to party, that his feet had not been soiled by its touch, that he never that ins feet had not been solid by its tolicit, that he never crossed its lines, that he would never till the war was over consent to be "cabined, cribbed, confined" by party influ-ences. As he had gone up and down the Hudson, he had indeed heard echoes from these "convocations of most po-litical worns," and they had all been tuned to the key of "roast beef and cabbage on a trencher." But he had been betrayed into prolisity by the interest of the occasion. Before closing he wished to speak of one other thing. All men on this continent are free and equal, and our Government must regard the private rights of civilians. If private rights are not respected, public liberty dies, We contend for individual rights, and wheever attacks them wounds the vital parts of the Republic. Not even the pleat of necessity allows any one to trample upon them. To be sure these rights depend on circumstances, and may be superceded by martial law. Till that is proclaimed they must not be infringed. Law is the reflex of order, which is the principle of the universe and God hinself. When it is the principle of the universe and God hinself. When it must need the grand, noble old liberty of the press.

Most served is the grand, noble old liberty of the press. Over that, in far Enrope the struggle for freedom has been most sternly and pertinncionsly waged. Let a free people gnard with jealens care the liberty of the press, and declare the administration who would strike at it as guilty of incivism and little less than traitors. [Applause.] Gen. Cecharae referred to his early belief that America

Gen. Cochrace referred to his early belief that America was the light-house of the world—the asylum of the oppressed. The had heard that the light had been extinguished. He demanded that America should remain the sanctuary of freedom, the asylum of the oppressed throughout the world. The refugee from other lands must be held innocent until, in accordance with law, you pronounce him guilty. (Applause.)

The speaker closed with a brief reference to the Monroe doctrine, and an enthusiastic assertion of his belief in the speedy triumph of our arms, which was rapturously applauded.

After the adoption of the platform, nominations were made.

JOHN C. FREMONT was nominated for President by acclamation.

Gen. JOHN COCHRANE was nominated for Vice President, with but few voices in the negative.

The following letters were addressed to the Convention: that of Wendell Phillips, read in Convention, is reported to have been warmly applauded:

#### FROM WENDELL PHILLIPS.

#### BOSTON, May 27, 1864.

DEAR SIR: I deeply regret that it is out of my power to attend the Cleveland Convention. Allow me to suggest one or two things which I hoped to urge on its attention.

Without denying what the friends of the Administration claim—that it has done something toward crushing the rebellion—my charge against it is that it has not done half that it should and could have done toward that end, had it used the means in its hands with an earnest and single purpose to close the strife thoroughly and forever. It has thought more of conciliating rebels than of subduing them. It has avowedly forborne the use of lawful and efficient means (to wit, the abolition of slavery) until it was thought indispensable, and even then has used it in a half-hearted, halting way, wishing to save the feelings of rebels.

We have three tools with which to crush the rebellionmen, money, and the emancipation of the negro. We were warned to be quick and sharp in the use of these, because every year the war lasted hardened the South from a rebellion into a nation, and doubled the danger of foreign interference. Slavery has been our great trouble in the past, and, as every man saw, was our great danger in the future. Statesmanship, said, therefore, seize at once the God-given opportunity to end it, at the same time that you, in the quickest, shortest, and cheapest manner, annihilate the rebellion.

For three years the Administration has lavished money without stint, and drenched the land in blood, and it has not yet throroughly and heartily struck at the slave system. Confessing that the use of this means is indispensable, the Administration has used it just enough to irritate the rebels and not enough to save the State. In sixty days after the rebellion broke out the Administration suspended *habaes corpus* on the plea of military necessity—justy. For three years it has poured out the treasure and blood of the country like water. Meanwhile slavery was too sacred to be used; that was saved lest the feelings of the rebels should be hurt. The Administration weighed treasure, blood, and civil ilierty against slavery, and, up to the present moment, has decided to exhaust them all before it uses freedem, heartily, as a means of battle. Mr. Lincoln's friends tell us that it ho is re-elected he is

Mr. Lincoln's triends tell us that it he is re-elected he is re-elected to pursue the same policy and obey the same Cabi-

net. What will be the result of another four years of such policy? Unless the South is recognized the war will continue; the taxation needed to sustain our immense debt, doubled by that time, will grind the laboring man of the North down to the level of the pauper labor of Europe; and we shall have a Government accustomed to despotic power for eight years—a fearful peril to democratic institutions.

If, Linedu's model of reconstruction is the experiment in Louisiana, which puts all power into the hands of the unchanged white race, soured by defeat, hating the laboring classes, plotting constantly for aristocratic institutions. To reconstruct the rebel States on that model is only continuing the war in the Senate Chamber after we have closed it in the field. Such reconstruction, leaving the South with its labor and capital at war, puts the whole payment of the debt on the industrious North, and in that way it will hang on us for a century. Such reconstruction makes the freedom of the negro a sham, and perpetuates slavery under a softer name. Such reconstruction, leaving the seeds of discontent and division in the South in the places of power, tempts and facilitates another rebellion, at the instigation or with the aid of French-Mexico. Such reconstruction dooms us to a second or third-rate place among nations, and provokes foreign insult and agression.

There is no plan of reconstruction possible within twenty years, unless we admit the black to citizenship and the ballot, and use him, with the white, as the basis of States. There is not in the rebel States sufficient white basis to build on. If we refine this method we must subdue the South and hold it as territory until this generation of white men have passed away, and their sons, with other feelings, have taken their places, and northern capital, energy, and immigration have forced their way into the South. Should we adopt that plan, and wait for those changes, twenty years must elapse before we can venture to rebuild States. Meanwhile a large and expensive army, and the use of despotic power by a Government holding half its territory and citizens as subjects, make every thoughtful man tremble for the fate of new government.

A quick and throogh reorganization of States on a democratic basis—overy man and race equal before the law—is the only sure way to save the Union. I urge it not for the black man's sake alone, but for ours—for the nation's sake. Against such recognition of the blacks Mr. Lincoln stands pledged by prejudice and avowal. Men say, if we elect him he may change his views. Possibly. But three years have been a long time for a man's education in such hours as these. The nation cannot afford more. At any rate, the Constitution gives this summer an opportunity to make President a man fully educated. I prefer that course. The Administration, therefore, I regard as a civil and mil-

The Administration, therefore, I regard as a civil and military fullure, and its avowed policy ruinous to the North in every point of view. Mr. Lincoln may wish the end—neace and freedom—but he is wholly unwilling to use the means which can seeme that end. If Mr. Lincoln is re-elected I do not expect to see the Union reconstructed in my day, unless on terms more disastrous to liberty than even disunion would be. If I turn to General Fremont, I see a mau whose first act was to use the freedom of the negro as his weapon, I see one whose theorogh loyalty to democratic institutions, without regard to race, whose earnest and decisive character, whose clear-sighted statesmaship and rare military ability justify my confidence that in his hands all will be done to save the State that foresight, skill, decision, and statesmaship can do.

I think the Convention should incorporate in its platform the demand for an amendment to the Constitution prohibiting slavery everywhere within the Republic, and forbidding the States to make any distinction among their citizens on account of color or race. I think it should demand a reconstruction of States as specifily as possible on the basis of every loyal man, white or black, sharing the land and the ballot. But if some of these points are not covered I shall still support its action with all my heart if it puts the name of Fremont or Butler on its flag. Fremont is my first choice, but I can support either of them; and this is an hour of such peril to the Republic that I think men should surrender all party and personal partiality, and support any man able and willing to save the State.

If the Baltimore Convention shall nominate Mr. Lincoln, then I hope we shall fling our candidate's name, the longhonored one of J. C. Fremont, to the breeze, and appeal to the patriotism and common sense of the people to save us from another such three years as we have seen. If, on the contrary, the Baltimore convention shall give us the name of any man whom the Radicals of the Loyal States can trust, I hope we shall be able to arrange some plan which, will unite all on a common basis and carry our principles into the Government.

Wishing you all success, and prepared to second your effort to remove the Administration, I am, yours &c., WENDELL PHILLIPS.

EDWARD GILBERT, Esq., New York.

#### FROM LUCIUS ROBINSON.

### STATE OF NEW YORK, COMPTROLLER'S OFFICE, ALBANY, May 28.

To Hon. A. J. COLVIN: My official duties and the illness of my deputy will pre-vent me from attending the meeting called for consultation at Cleveland, the 31st instant. I trust that you will be there with your judicious advice.

There was never a time when the safety and welfare of the country more imperatively demanded careful delibera-tion, with wise and resolute action. We have lived through three years of war, and have survived many bad mistakes, simply because the popular mind has been intensely fixed maps the single purpose of suppressing the releasing the norm the single purpose of suppressing the releasing the trolling power, and the masses have followed so steadily, that it has served as a substitute for proper governmental leadership. But it is evident that the time is near at hand when the re-establishment of order, the removal of the cause of the rebellion and the repairing of the terrible desolation it has produced, will require at the head of the Government the very highest qualities of leadership. How can we hope to live as a nation through the crisis before us with a weak Executive and Cabinet in a state of discord and anarchy? Will not the country be in imminent danger of falling into the same condition when it ceases to be held taming into the same constant when it ceases to be near together by the pressure of war? It appears to me that a firm assertion of sound principles and the election of the greatest men, regardless of former party organizations, are essential to the safety of the nation.

Whether a nomination should be made at Cleveland or not can best be determined after meeting and consulting with those who will assemble there ; but it it shall be decided to nominate I have no hesitation in saying that I helieve the hopes of the people throughout the country are resting upon General Grant as the candidate. He has displayed the qualities which give all men confidence. H- Das shown himself possessed of great ability and skill, the most indomitable courage and most unselfish devotion to the cause of his country. Victory has attended him wherever he has gone. One year ago all confidence in the Adminis-tration was lost. The brilliant victories which have since been won by Gen. Grant have so far restored it as to en-courage the Administration to attempt to rc-elect itself on the strength of his achievements. But in my judgment we should let him who has won the honors wear them, and should entrust power to one who has shown that he knows how to wield it; we shall then have a leader at the head of affairs in whom all loyal men will have confidence, against whom there will be no prejudices, and whom all will aid Yours, LUCIUS ROBINSON. with alacrity.

#### FROM FREDERICK DOUGLASS.

ROCHESTER, May 23, 1864.

SIR: I mean the complete abolition of every vestige, form, modification of slavery in every part of the United States, perfect equality for the black man in every State before the perfect equality for the black man in every state before me law, in the jury-box, at the ballot-box, and on the battle-field; ample and salutary retaliation for every instance of enslavement or slaughter of prisoners of any color. I mean that in the distribution of offices and honors under this Government no discrimination shall be made in favor of or private one close of divinges whether black or white, or against any class of citizens, whether black or white, of native or foreign birth. And supposing that the Conven-tion which is to meet at Cleveland means the same thing, I cheerfully give ny name as one of the signers of the call. Yours, respectfully, FREDERICK DOUGLASS.

E. GILBERT, ESq.

#### THE PLATFORM.

Mr. Carroll, Chairman of the Committee on Resolutions.\* reported the following resolutions :

First. That the Federal Union shall be preserved.

\* During the Convention, Mr. Langer, of Iowa, offered these resolutions, which, on the suggestion of the Chair, were referred to the committee:

"Resolved, That the members of this Convention, or of any Convention arising from this, to nominate or participate in the nomination of a candidate for the next Presidential term, and the Presidential electors of this party. pledge themselves upon their honor not to accept offices of trust, honor, or profit from the Administration in power during the next Presidential term, and not to be connected directly the fact resentation in the for confected uncerty of indirectly with any contract or business transaction in the power of the Administration. *"Resolved,* To make it obligatory on the Presidential can-didate of this party, if successful, to act accordingly. *"Resolved,* That this is not to be construed to prevent any member from become an active combetration the the second

member from becoming an active combatant in the Navy and Army of the United States, in such capacity as his respective State may elect to employ him."

Second. That the Constitution and laws of the United States must be observed and obeyed. Third. That the rebellion must be suppressed by force of

arms, and without compromise. Fourth. That the rights of free speech, free press, and the

habeas corpus be held inviolate, save in districts where mar-

tial law has been proclaimed. *Fifth.* That the rebellion has destroyed slavery, and the Federal Constitution should be amended to prohibit its reestablishment, and to secure to all men-absolute equality before the law.

Sixth. That integrity and economy are demanded at all times in the administration of the Government; and that in time of war the want of them is criminal.

Seventh. That the right of asylum, except for crime and subject to law, is a recognized principle of American lib-erty; that any violation of it caunot be overlooked, and

erry that any rotation is called as orteriories, and must not go unreluked. Eighth That the national policy known as the "Monroe doctrine" has become a recognized principle, and that the establishment of an anti-republican Government on this continent by any foreign power cannot be tolerated. Nonth. That the gratifude and support of the nation are

due to the faithful soldiers and the earnest leaders of the Union army and navy for their heroic achievements and deathless valor in defence of our imperiled country and of civil liberty.

Tenth. That the one-term policy for the Presidency, adopted by the people, is strengthened by the force of the existing crisis, and should be maintained by constitutional amendments.

Eleventh. That the Constitution should be so amended that the President and Vice President shall be elected by a

The direct vote of the people. The direct vote of the reconstruction of the red f(d). That the question of the reconstruction of the rebellions States belongs to the people, through their representatives in Congress, and not to the Executive.

That the confiscation of the lands of the Thirtcenth. rebels, and their distribution among the soldiers and actual settlers, is a measure of justice.

Mr. Carroll stated that the committee were unanimous on all the resolutions, save one-the last. As a matter of expediency it was thought not advisable to recommend it, but the majority of the committee had instructed him to report it for the consideration of the Convention.

At this juncture, Mr. Gilbert announced the receipt of a letter from Wendell Phillips, which was warmly applauded while being read by the secretary.

Mr. Ransom of New Jersey, moved that the question on the resolutions be taken separately.

It was then decided to take up the resolutions separately. The first, second, third, and fourth were adopted without dissent.

Mr. Goodell moved to amend the fifth by declaring that slavery shall die, instead of saying that it is dead; for, said the mover, although it is legally dead, the fact is that there are over three millions of people now enslaved in the southern States. He moved to substitute these words:

"That the rebellion must be suppressed by the destrnotion of its motive cause, slavery."

This was lost after a debate, and the resolution was adopted.

The remainder were then adopted.

LETTERS OF ACCEPTANCE.

### FROM GENERAL FREMONT.

#### NEW YORK, June 4, 1864.

CENTLEMEN: In answer to the letter which I have had the honor to receive from you, on the part of the representa-tives of the people assembled at Cleveland, the 21st of May, I desire to express my thanks for the confidence which led them to offer me the honorable and difficult po-sition of their confidence in the supersching. Duraidential sition of their candidate in the approaching Presidential election.

Very honorable, because in offering it to me you act in the name of a great number of citizens who seek above all things the good of their contry, and who have no sort of selfash interest in view. Very difficult, because in accept-ing the candidacy 'you propose to me, I am exposed to the

reproach of creating a schism in the party with which I | inderse a policy and renew a power which has cost us the have been identified.

Had Mr. Lincoln remained faithful to the principles he was elected to defend, no schism could have been created, and no contest could have been possible. This is not an ordinary election. It is a contest for the right even to have candidates, and not merely, as usual for the choice among them. Now, for the first time since '76, the question of constitutional liberty has been brought directly before the people for their serious consideration and vote. The ordinary rights secured under the Constitution and the laws of nary rights secured under the Constitution and the laws of the country have been violated and extraordinary powers have been usurped by the Executive. It is directly before the people now to say whether or not the principles estab-lished by the Revolution are worth maintaining. If, as we have been taught to believe, those guarantees for liberty which made the distinctive name and glory of us country are in truth invited by second then here worth

our country, are in truth inviolably sacred, then here must be a protest against the arbitrary violation which had not oven the excuse of a necessity. The schism is made by those who force the choice between a shameful silence or a how who here the choice between a share in sectors a reprotest against wrong. In such considerations originated the Cleveland Convention. It was among its objects to arouse the attention of the people to such facts, and to bring them to realize that, while we are saturating Southern soil with the best blood of the country in the name of liberty, we have really parted with it at home.

To-day we have in the country the abuses of a military dictation without its unity of action and vigor of execution-an Administration marked at home by disregard of constitutional rights, by its violation of personal liberty and the liberty of the press, and as a crowning shame, by its abandonment of the right of asylum, a right especially dear to all free nations abroad. Its conrse has been characterized by a feeleness and want of principle which has misled Ea-ropean powers and driven them to a belief that only com-mercial interests and personal aims are concerned, and that no great principles are involved in the issue. The admira-ble conduct of the people, their readiness to make every coarding demonded driven the features to make every sacrifice demanded of them, their forbearance and silence under the suspension of everything that could be suspended, their many acts of heroism and sacrifices, were all rendered personal ends for which the war was managed. This incapersonal ends for which the war was managed. This inca-pacity and schishness naturally produced such results as led the European powers, and logically enough, to the convic-tion that the North, with its greatly superior population, its immense resources, and its credit, will never be able to recover the South. Sympathies which would have been with us from the outset of this war were turned against us, and in this way the Administration has done the country a double wrong shroad. It created hostility, or at host in and in this way the Administration has done the country a double wrong abroad. It created hostility, or at best in-difference, among those who would have been its friends if the real intentions of the people could have been better known, while, at the same time, it neglected no occasion for making the most humiliating concessions.

Against this disastrous condition of affairs the Cleveland Convention was a protest.

The principles which form the basis of its platform have The principles which form the basis of its platform have my unqualified and cordial approbation, but I cannot so heartily concur in all the measures which you propose. I do not believe that confiscation extended to the property of all rebels, is practicable, and if it were so, I do not think it a measure of sound policy. It is, in fact, a question belong-ing to the people themselves to decide, and is a proper oc-cusion for the exercise of their original and sovereign au-thority. As a way measure in the hearinging of a revolt thority. As a war measure, in the beginning of a revolt which might be quelled by prompt severity, I understand the policy of confiscation, but not as a final measure of re-

construction after the suppression of an insurrection. In the adjustments which are to follow peace no considerations of vengeance can consistently be admitted.

The object of the war is to make permanently secure the peace and happiness of the whole country, and there was but a single element in the way of its attainment. This element of slavery may be considered practically destroyed in the country, and it needs only your proposed amendment of the Constitution, to make its extinction complete.

With this extinction of slavery the party divisions cre-ated by it have also disappeared. And if in the history of the country there has ever been a time when the American people, without regard to one or another of the political divisions, were called upon to give solemnly their voice in a matter which involved the safety of the United States, it

a matter which involved the safety of the United States, it is assuredly the present time. If the Convention at Baltimore will nominate any man whose past life justifies a well-grounded confidence in his fidelity to aur cardinal principles, there is no reason why there should be any division among the really patriotic men of the country. To any such I shall be most happy to give a cordial and active support.

My own decided preference is to aid in this way, and not to be myself a candidate. But if Mr. Lincoln should be nominated—as I believe it would be fatal to the country to

on the road to bankruptcy-there will remain no other alternative but to organize against him every element of conscientious opposition with the view to prevent the misfortune of his re-election.

In this contingency, I accept the nomination at Cleve-land, and, as a preliminary step, I have resigned my commission in the arms. This was a sarrifice it gave me point to make. But I had for a long time fruitlessly endeavored to obtain service. I make this sacrifice now only to regain liberty of speech, and to have nothing in the way of discharging to my utmost ability the task you have set for me.

With my earnest and sincere thanks for your expressions of confidence and regard, and for the many honorable terms in which you acquaint me with the actions of the Convention, I am, gentlemen,

tion, I am, gentlemen, Very respectfully and truly yours, J. C. FREMONT.
To Worthington G. Snethen of Maryland, Edward Gilbert of New York, Casper Butz of Illinois, Charles E. Moss of Missouri, N. P. Sawyer of Pennsylvania, a Committee, &c.

## FROM GENERAL COCHRANE.

### NEW YORK, June 4, 1864.

GENTLEMEN: I have received your note informing me officially of my nomination by the Radical Democracy at Cleveland, on the 31st ultimo, as their candidate for Vicc. President of the United States, on the ticket with John C Fremont as their candidate for President.

I have been accustomed to regard simply as a duty per formed what you are pleased to represent as personally meritorions, and to regret the physical disability which alone withdrew mc from the immediate scene of war.

I concur in the action and agree with the principles of the Convention. Where by its twelfth resolution the question of reconstruction is referred to the constitutional ac-tion of the people, it wisely committed to them an issue peculiarly within the province of the luture, and not yet sufficiently emerged from war, to warrant positive opinion. While I have ever supposed confiscation and use of the

property of an enemy in arms to be a laudable exercise of established and essential rule of civilized war, I am ān pleased to observe that the Convention, when asserting the justice of the principle, intended to remit its exercise to the discretion of the people, bereafter manifested through their representatives in Congress, when considering the parafor, indeed, so blended must be the various methods-see This was judicions; questration, confiscation, military absorption and occupadescription that shall hereafter co-operate to evolve order from confusion, and to restore the Government, that it is difficult, if not impossible, now, when affirming the principle, to provide for its application.

I have the honor, gentlemen, to accept the nomination for the Vice President of the United States, which you have tendered to me under the direction of the Convention.

I am, very respectfully, yours, JOHN COCHRANE. To the COMMITTEE.

Speech of Colonel Cochrane.

DELIVERED TO HIS REGIMENT, FIRST UNITED STATES CHAS-SEURS, NOVEMBER 13, 1861.

It having been announced that Colonel John Cochrane would speak to his regiment, at their camp, on the occasion of their first appearace in new uniforms, on the afternoon of Friday, the 13th of November, instant, a large assemblage of ladies and gentlemen was congregated to hear him. A staging had been improvised beneath a spreading oak, where, conspicuous among the audience, sat the Secretary of War, dignified and composed. In front, in enclosing linez, stood the imposing regiment—the first United States Chasseurs, steady, exact, and attentive. Within the square a regimen-tal band uttered harmonious music, while the reddening rays of the descending sun enveloped the audience, soldiers, Secretary, and orator, in rosy rays, that imparted a soft beauty to the scene, and conveyed the pleasing illusion at-tendant upon dramatic effect. The Colonel then advanced, and notified his hearers that one of the companies of the regiment had selected this as a fitting opportunity to present to one of their lieutenants (Morton) a small token of their admiration and esteem. The preliminary ceremony was agreeably and satisfactory finished. It was then that Colonel Cochrane arose, and, justly inspired by the scene and the circumstances which produced it, spoke as follows:

and the circumstances which produced it, space as follows: Soldiers of the First United States Chasseurs: [Brave Colonel.] I have a word to say to yon to-day. You have engaged in an arduous struggle. You have prosecuted it; you intend to prosecute it; you have stood unflinchingly before the enemy; you have proved yourselves patriotic, able, and tried soldiers, and you are entitled to the meed of

**praise.** I, your commander, this day feel that it is a proud duty to extend to you the hand of approbation, and to declare that you are worthy of your country.

Soldiers, you have undergone labor; you have faced the enemy; you have stood without retreating before their fire; you have borne the inclemencies of the season, and you are ready to advance with that grand army of which you are a part. Your country opens its arms, and receives you to its bosom. It will always praise and applaud you. Its commanders stand at the head of the column, and, with you behind them, they are not to be deterred. But the command is forthconning—forward, march! toward the enemy. Take his possessions, for they are yours; they are yours to occupy; they are yours to enjoy; you are no marauders, you are no plunderers of property not your own, but you are the avengers of the law; yon are the right arm of the Constitution; under your flag march patriotism and order, and republican institutions; in your train follow peace, prosperity and liberty; you are the servants of these high out his occasion, the Secretary of War.

arm of the worthy public servant who stands behind me on this occasion, the Secretary of War. Soldiers, you have been called to the field, not as ma-randers and mercenaries, but as the defenders of our high faith, defenders of our glorious reputation, defenders of our honor and renown, around which cluster the memories of the past, and whose feats and performances will yet distin-guish the future. You are led forward by a commander guish the future. You are led forward by a commander under whom to serve it is a pride for the highest among us. He enjoys the confidence of the people, and his reputation already renders powerless the arms of your enemies. By him we have assurances of triamph yet to come. Beau-fort is ours—Charleston may be ours—the whole country now disintegrated may shortly be united by the force of those arms of which you are a part, and the Union once more signify to the world the intent of that glorious motto, *E Pluribus Unum*. Then no longer shall be heard that fell doctrine of seconsion which would the runs sanuder and disdoctrine of secession, which would tear us as under, and dis-tract, part from part, this glorions Union; but we shall all be as we have been, one and inseparable, under the flag of our glorions nationality, won by our fathers, and preserved by you. [Applause.] Here is assembled upon the banks of the Potomac an army the like of which the world has never The motive which has gathered that army together seen. never before was presented to the eye of history. It was congregated by no despotic order; it was the voluntary wish, the motive power, of every man composing it—the power of men rushing, as with one purpose, to reinstate the flag of our Union and save the Republic. That, soldiers, is your mission; and you have a commander who with lightning speed will lead you to conquest, and with equal speed will transmit the glory of your labors to the remotest corners of transmit the glory of your labors to the remotest corners of our country. And now permit me, though the shades of night are falling upon us, to indulge in a few words as to the cause of the war, and the means by which it is to be brought to a successful termination. The material aid I have already averted to; the motive power remains to be commented upon. On the one side you have the Confeder-Federal army: Now, the difference between these two words, in their common acceptation, is the cause for which these two armies are fighting. It is Secession against Fed-eration, Federation against Secession. Nationality against disunion; confusion against order; anarchy against a good, free, and liberal Government—a Government made equally by the Fathers of the South and the Fathers of the North. We are in a revolutionary period. The South contends for the right of revolution. We admit the right; but, while we the right of revolution. We admit the right; but, while we admit it, we invoke the sole umpire which may be invoked on such occasions—the umpire of the sword, the umpire of force, the *ultima ratio*, that last effort to which men appeal

bites the attack that the tend to which may appear when they have differences otherwise irreconcileable. They—the South—have resorted to arms, and they have compelled us to the same resort; and if they claim that it is a war of self-preservation on their part, it is equally a war of self-preservation on ours; and if we are in controversy for very existence, then I contend that all the resources, all the means within ourselves, individually, collectively, and hationally, nust be resorted to and adopted. [Applause.] But some friend—a doubter—exclaims; "Would you disrupt and tear asunder the Constitution?" Where is the Constitution? Would you tread and trample upon that sacred instrument, and no longer acknowldgs its binding force, no longer be bound by its compromises and decrees? I answer, no. The Constitution, by the necessity of the controversy, is cast behind the arena of the strife. May it rest there safe, until the present strife being over it shall be restored to its original purity and force. Like the sibyl leaves when lost, the remainder become more valuable in our eyes, and in the midst of the carnage we will clasp to our bosons that instrument whose worth has never been transcended by human efforts. Soldiers, to what means shall we resort for our existence? This war is devoted not merely to victory and its

mighty honors, not merely to the triumph which moves in mightly honors, not merely to the triumph which moves in glorious procession along our streets. But it is a war which moves towards the protection of our homes, the safety of our families, the continuation of our domestic altars, and the protection of our firesides. In such a war we are jus-tified, are bound to resort to every force within our power. The protection of our mestals. In such a we are jus-tified, are bound to resort to every force within our power. Having opened the port of Beaufort, we shall be able to export millions of cotton bales, and from these we may supply the sinews of war. Do you say that we should not seize the cotton? No; you are clear upon that point. Sup-pose the munitions of war are within our reach, would we not be guilty of shameful neglect, if we availed not our-selves of the opportunity to use them? Suppose the ene-my's slaves were arrayed against you, would you, from any squeamishness, refrain from pointing against them the hos-open their ports, and even destroy their lives, I ask you whether you would not use their slaves? Muether you would not arm their slaves [great applause.] and carry them in battalions against their masters? [Reneved and turnul-tuous applause.] If necessary to save this Government, I would playe, their whole country black and white, into one indiscriminate sea of blood, so that we should in the end indiscriminate sea of blood, so that we should in the end have a Government which would be the vicegerent of God. Let us have no more of this dilletante system, but let us work with a will and a purpose that cannot be mistaken. Let us not put aside from too great a delicacy of motives. Let us not put aside from too great a dencacy of motives. Soldiers, you know no such reasoning as this. You have arms in your hands, and those arms are placed there for the purpose of exterminating an enemy unless he submits to law, order, and the Constitution. If he will not submit, co-place every thing that comes in your way. Set fire to the cotton. Explode the cotton. Take property wherever you may but Explate the course and bestow him upon the non-slave holder if you please. [Great applause.] Do to them as they would do to us. Raise up a party of interest against the absent slaveholder, distract their counsels, and if this should not be sufficient, take the slave by the hand, place a musket in not be sufficient, take the slave by the hand, place a musket in it, and in God's name bid kin strike for the liberty of the human race. [Immense applause.] Now, is this emancipa-tion? Is this abolitionism? I do not regard it as either. It no more parta<sup>2</sup>/es of Abolitonism than a spaniel partakes of the nature of the lion. Abolitionism is to free the slaves. It is to make war upon the South for that purpose. It is to place them above their masters in the social scale. It is to assert the great abstract principles of equality among men. But to take the slave and make him an implement of war in overcoming your en-eux that is a military scheme. It is a military necessity emy, that is a military scheme. It is a military necessity, and the commander who does not this, or something equivand the commander who does not this, or something equiv-alent to it, is unworthy of the position he holds, and equally unworthy of your confidence. Emancipation! Are we en-gaged in a war of emancipation? If so, who commenced the war? Not we. And if we did not commence the war, we cannot be charged with its consequences. Where had it is origin? I thad its origin in the South. It was and has been a war of the South against the free institutions of the North. Let me illustrate. A re we to free their slaves? the North. Let me illustrate. Are we to free their slaves? We do not intend it. Do you recollect the resolution which was passed the last session of Congress, which distinctly declares that it never was intended by anybody in this wide land to free the slave. "Compromise," too, has been talked of this matter. "Why did they not compromise? Because it was not their object. I say this fearlessly, for I infer it

It was not their object. I say this learnessly, for 1 miner it from scenes in which I was an actor. At Charleston, I remember, that when satisfied that Mr. Donglas could not, while they remained in the convention, be nominated for the Presidency, they nevertheless withdrew. It was, if my remembrance is not at fault, near the midnight hour, at the prominent headquarters of the sonthern array, that Messrs, Yancey and Percy Walker entered the room. Those present had previously concluded, upon careful calculations, that the South abiding by the convention, Judge Douglas could not receive two thirds of the vote of the convention. This conclusion was communicated to these gentlemen, who, as I understand, having reviewed the calculations, and expressed their reliance upon them, declared, when leaving, their determination to rouention on the very next morning, that Mr. Walker sent to the chair the act of secession of Alkhama therefrom. The morning deed declared marvelonsly, when contrasted with the inidnight profession, that the act of secession was but a foregone conclusion, necessary to precipitate the only issue to which they desired to be a party. Nothing could be satisfactory to these, except that arms should be resorted to, and the fate of revolution abided by.

Nothing could be satisfactory to these, except that arms should be resorted to, and the fate of revolution abided by. I declare, therefore, that the war is not of our originating, but it has been forced upon us by a crafty enemy-an enemy resolved to do or die; to destroy our free Government, or perish in the attempt. And what is their object? Why, their object is to thar down this proud, noble, and beneficent Government, to establish a reign of terror, anarchy, and

confiscation, in the land; to implant upon this our soil the hideous doctrine of the right of secession, so that when one State secodes another may secode, and still another, and still another, so that within forty-eight hours, by the light of their reason and them holy eight hours, you may of their reason and the exactness of their judgment, you may establish on this continent thirty-four independent governments. Thirty-four, did I say? Why, no, not thirty-four merely, but every county and every city, and every village and hamlet; nay, every person who suffers from indigestion at the dinner table may claim the same right; and thus, soldiers, we shall have the confusion and disorder which will plunge into dismay and ruin the best and most benevolent government in the world. Now, what is our object? It is simply to arrest the sway of this fell spirit of secession. It is to maintain our Government, to establish and vindicate law and order, without which neither happiness nor prosperity can exist. You are engaged, too, by the strength of your arms, to protect our commerce with other nations, and when victory crowns your devotion to your country's cause—as it assuredly will—you will be proudly pointed at as the champions of American rights, as men who have maintained their dearest principles, and as those who, from this time forward, shall live in the most grateful with marks of imperishable honor to the remotest pos-But, soldiers, to accomplish all this, not merely terity. terity. But, soldiers, to accomptish all this, not merely arms are necessary, not merely mon to carry them, but that powerful and overwhelming spirit which constitutes and nakes us men, that spirit which lifts us above the creeping things of the earth, and brings near the Deity, in accomplishing his work on earth. Oh, then, let us not think that the "battle is to the strong "—let us not merely decoud a disclution can dealer, but with that formidence of depend on discipline and order, but with that fervidness of soul which inspired our fathers at Bunker Hill, and Saratoga, and Yorktown, come forward and give effect to all that is valuable in the name of patriotism, and honor, and religion. rane. [Loud and prolonged cheering.]

Never, no never, will you succeed until that spirit's once more manifested and developed which actuated the soldiers of Cromwell, who, on the field, invoked the Lord their God to arise. So let it be with us. We must be at least one to arise. So let it be with ns. We must be at least one with Him in spirit. Let us, like Cronwell, invoke the Al-mighty blessing, and, clothed with the panoply of patriot-ism and religion, stike for our homes and our country. [Im-mense cheering.] Let us—oh, let us—without reference to any differences of the past, keep our eyes steadfastly on the great object to be achieved, the nationality and independ-nce of this country. the solved in a faith of the country. ence of this country, the salvation of civilization from the insults and assaults of barbarism; and then, but not till then, will you be worthy to be recognized as a distinguished portion of onr great American army. [Long continued cheering from the whole regiment.)

Upon the conclusion of Colonel Cochrane's speech, loud and repeated calls being made for the Secretary of War, Mr. Cameron came before the regiment, and said :

Soldiers : It is too date for me to make you a speech tonight, but I will say that I heartily approve every sentinight, but I will say that I heartily approve every senti-ment uttered by your noble commander. The doctrines which he has laid down I approve of as if they were my own words. They are my sentiments—sentiments which will not only lead you to victory, but which will in the end reconstruct this our glorious Federal Constitution. It is idle to talk about treating with these rebels upon their own We must meet them as our enemies, treat them as terms. enemies, and punish them as enemies, until they shall learn enemies, and purish them as chemics, with they shall learn to behave themselves. Every means which God has placed in our hands it is our duty to use for the purpose of protecting ourselves. I am glad of the opportunity to say here, what I have already said elsewhere, in these few words, that I approve the destrines this evening enunciated by Colonel Coch-

## INDEX.

- ADAMS, Charles Francis, Representative in Thirty-Sixth Con-gress, 43; member of Committee of Thirty-three, 53; report of, 57; Secretary Seward's letter to, on Trent affair, 333.
- allair, 338. ADAMS, Green, Representative in Thirty-Sixth Congress, 49. ADAMS, James H., Commissioner to Washington, 2; cor-respondence with President Buchanan, 29-32. ALMAS, Wirt, Commissioner from Mississippi to Louisiana, 11; received by Convention, 3.
- ADDRESS of South Carolina Convention to Slaveholding States,
- ADDRESS of South Carolina Convention to Slaveholding States, 12; of loyal Governors in favor of Emancipation, 233. ADUSTREY, propositions of, in Thirty-Sixth Congress—In House, by Messrs. Thayer, John Cochrane, Haskin, Mallory, Stevenson, English, Kilgore, Holman, Niblack, McClernand, Noell, Hindman, Larrabee, Anderson of Missouri, Smith of Virginia, Sickles, Dunn, Adrain, Morris of Pennsylvania, Stewart of Maryland, Leake, Jenkins, Cox, Hutchins, Sherman, Etheridge, Davis of Indiana, Florence, Fenton, Kellogg of Illinois, and Val-landigham, 53-57; report of Committee of Thirty-three and votes upon, 57-62; on Constitutional Convention, 62-64; on Mr. Kellogg's, 63; on Mr. Clemens's, 63. In Senate, Mr. Pugh's and vote, 64; Mr. Grimes's and vote, 64; Mr. Bingham's and vote, 64; Mr. Grimes's and vote, 64; Mr. Rowell's amendment to Mr. Critten-den's proposition, 55; Mr. Clark's, 66; re-consideration and vote, 64; Mr. Powell's amendment to Mr. Critten-den's proposition, 65; Mr. Olark's, 66; re-consideration and rejection of, 66; final defeat of the Crittenden proposition, 66; final defeat of the Crittenden tham thereon, 66, 67; the Peace Conference, members and proposition of, votes on, 68, 69; Senate Committee of Thirteen and votes therein, 70–72; propositions of Senators Joseph Lane, and Andrew Johnson, Represen-tatives Barrett, Pryor, Colfax, Vallandigham, and Mor-ris of Illinois, and Border-States meeting, 73; of New York Representatives, Alex. H. H. Stuart, National Intelligencer, Albany Evening Journal, and Joseph W. Taylor, 74; New York Sun, Baltimore Exchange, "A Distinguished Carolinian," and a correspondent of the National Intelligencer, 75; of Mr. Seward, in commit-tee, 71; Senator Hunter, 86–88; and President Buchanan, 50. 50.

ADJUSTMENT, Mr. Lincoln's reported opinion on, 67.

- ADMINISTRATION of Abraham Lincoln, 105-194. ADMINISTRATION of Abraham Lincoln, 105-194. ADMAIN, Garnett B., Representative in Thirty-Sixth Con-gress, 48; proposition of adjustment by, 54; resolu-tion of, on unconstitutional State laws, and yote upon,
- tion of, on unconstitutional State laws, and vote upon, 75; on approving Major Anderson's course, 76.
  Advertiser, Mobile, comments of, on Floyd's transfer of United States arms South, in 1839-760, 33.
  ARBCAN Slave Trade, propositions respecting and votes, 52-72; treaty for the suppression of, 150; modification of, 151; bill to carry out the, 151; bill of rebel Congress concerning, 151; veto of President Davis, 151; intercepted "Confederate" dispatch upon, 151, 152.
  AGNEW, Daniel, Justice Supreme Court of Pennsylvania, decision on enrollment act, 274.

AIKEN, Warren, Speaker of Georgia rebel House of Repre-sontatives, 329; Representative in Second Congress, 402.

ALABAMA, vote of, for President, 1; members of Thirty-Sixth Congress, 45, 49; deputies in rebel Provisional Con-gress, 11, 400; members of First Congress, 401; of Second, 402; secession movement in Convention and Legislature of, 4; ratified Confederate Constitution, 4; Inter-State Commissioners of, 11; makes a loan to the Provisional Government, 12; scizures and surrenders, in 22, commissioners and from the tract with United in, 27; commissioner sent from, to treat with United 27

States Government, 34; purchase of Government arms by, in 1860, 35; resolutions of Legislature of, on mili-tary employment of slaves, 282; soldiers in the army, 399

- ALBANT Evening Journal, adjustment proposition of, 74. ALBANT Evening Journal, adjustment proposition of, 74. ALBANCH, Cyrus, Representative in Thurty-Sixth Congress, 49; in Thirty-Seventh, 122. ALEXANDER, William C., member of Peace Conference, 67.

- ALEXANDER, William C., member of Peace Conference, 67. ALEXANDERIA, Virginia, opening of port of, to trade, 149. ALEXANDERIS, President's proclamation relative to, 272. ALLEOR foreign enlistment, 343-345. ALLEN, Charles, member of Peace Conference, 67. ALLEN, Henry, W., rehel Governor of Louisiana, on reconstruction, 331.
- ALLEN, James C., Representative in Thirty-Eighth Con-
- ALLEN, James C., Kepresentative in Thirty-Lignth Congress, 140.
  ALLEN, William, Representative in Thirty-Sixth Congress, 49; in Thirty-Seventh, 122; resolutions of, rospecting the object of the way, 285; resolution of, not to use money appropriated in emancipating slaves, 287.
  ALLEN, William J., Representative in Thirty-Seventh Congress, 123; in Thirty-Eighth, 140.
  ALLENS, William B., Representative in Thirty-Eighth Congress, 140.

- gress, 140. ALLEY, John B., Representative in Thirty-Sixth Congress, 48; in Thirty-Seronth, 122; in Thirty-Eighth, 140. ALVORD, Lieutenant Henry E., captured Moseby's military
- papers, 119. AMERICA, Central and South, monarchical intrigues in, 343.

- AMERICA, Central and South, monarchical intrigues in, 343.
   American, Baltimore, on arrests, 153.
   AMES, James T., purchase of Government arms by, in 1800, 35.
   AMES, Oakes, Representative in Thirty-Eighth Congress, 140.
   AMES, Samuel, member of Peace Conference, 67.
   AMESTY, President's argument upon, 146: proclamations of, 147, 148; accompanying circular of Attorney General, 148.
   ANCONA, Svdenham E., Representative in Thirty-Seventh
- ANCONA, Sydenham E., Representative in Thirty-Seventh Congress, 122; in Thirty-Eighth, 140; resolution of, concerning enrollment, 270.
- ANDERSON, Clifford, Representative in Second Rebel Con-gress, 402.

- gress, 402.
   ANDERSON, Fulton, commissioner from Mississippi to Virginia, 11; received by Virginia, 6.
   ANDERSON, Josiah, M., member of Peace Conference, 68.
   ANDERSON, J. Patton, Deputy in rebel Provisional Congress, 11,400; resigned, 400; vote on secession ordinance, 399, 1990; Construction of Congression Contraction Congress. ANDERSON, Lucien, Representative in Thirty-Eighth Con-gress, 140.
- gress, 140, ANDERSON, Major Robert, Floyd's instructions to, 31; cor-respondence before firing upon Sumter, 113, 114; report of, 114; resolution approving of course of, 76. ANDERSON, Thomas L., Representative in Thirty-Sixth Con-
- gress, 49.
- gress, 40. ANDERSON, William C., Representative in Thirty-Sixth Con-gress, 49. ANDREW, John A., Governor of Massachusetts, signer of Al-toona Address, 233.
- ANTHONY, Henry B., Senator in Thirty-Sixth Congress, 49; Thirty-Seventh, 122; Thirty-Eighth, 140; proposition of, to repeal proposed constitutional amendment of 1861, 258.
- ANTHONY, Lieutenant Colonel D. R., order on fugitive slaves, 251.
- APPEAL for recognition to Earl Russell by rebel commis-sioners, extract from, 27; Russell's letter to Lyons re-specting, 27; President Lincoln's appeal to Border State Representatives and their replies, 213-220.

417

APPLETON, William, Representative in Thirty-Seventh Con-

- members of First Congress, 401; of Second, 402; secession movement in Convention and Legislature, 4; sciances and surrenders, 28; free State government, 320, 321; report of Lyman Trumbull on admission of Senators from, and vote, 320, 321; officers of new State government, 321; Representatives and Senators from, fail to obtain seats, 321; President Lincoln's letters to Major General Steele and William M. Fishback respect-
- ing election in, 322; soldiers in the army, 399. Arxs, United States, transfer of, South in 1859-60, 34; pur-chase by Alabama and Virginia in 1860, 35; purchase of, by Governor of Virginia prior to secession, 36; sale of, to States, 36; Jefferson Davis's bill to authorize such sale, 36; discussion and vote on in the Senate, 36, 37; Mobile Advertiser's comment on Floyd's transfers, 35;
- Mobile Advertiser's comment on Floyd's transfers, 50; Burch's resolution relative to California's quota of, 41. ARMY, rebel, strength of, 117, 121, 399. ARNOLD, Isaac N., Representative in Thirty-Seventh Con-gress, 122; in Thirty-Eighth, 140; bill of, respecting the President's emancipation proclamation, 229, 230; proposition to amend the Constitution and vote, 258. ARNOLD, Samuel G., member of Peace Conference, 67; Sen-ator in "Unity-Seventh Concress, 123.

ator in Thirty-Seventh Congress, 123.

- ARRESTS of citizens, 152-154; President Lincoln's and Secretary Stanton's orders respecting, 154; of Vallandigham, [12] M. Pendleton's resolution in Iouse denoncing, 182; proceedings respecting sundry, 179–181; resolu-tions, votes, and legislation respecting persons in ar-rest, 183-185; General Augur's order respecting military, 187; W. Saulsbury's amendment respecting, and vote, 236, 237
- ARRINGTON, A. H., Representative in First Rebel Congress, 402.

ARTICLE of War, new, 237, 238. ASBOTH, Brigadier General, orders of, respecting the election in Kentucky, 313.

- ASHE, Thomas S., deputy in rebel Provisional Congress, 402.
- 49: withdrew, 1. ATENNS, John D. C., deputy in rebel Provisional Congress,
- 400; Representative in First Congress, 402; in Second, 402.
- ATLANTA (Georgia) Intelligencer on reconstruction with the Northwestern States, note, 303. ATTORNEY General, circular of respecting amnesty, 148, 149;
- opinion of respecting the President's powers, 51, 52; converting habeas corpus, 158-161; concerning execution of fugitive slave law, 235; concerning pay of colored chaptains: 279, 280; concerning citizenship, 378-281; on pay of volored soldiers, 384, 385.
  AUGRE, C. C., Major General, order of respecting military

arrests, 187.

AVERELL, A. A. General, order of, to prohibit slaves leaving the city of Washington, 245.

AVERY, William T., Representative in Thirty-Sixth Congress, 49.

AVERY, William W., deputy in rebel Provisional Congress, 400.

ATER, L. M., member of First Rebel Congress, 400; bf Sec-ond, 402; vote on secession ordinance, 398.

- BADDITT, Elijah, Representative in Thirty-Sixth Congress, 45; in Thirty-Seventh, 122.
  BACKUS, Franklin T., member of Peace Conference, 68.
  BALL, bill for, of persons arrested, 153.
  BALLEY, Goldsmith F., Representative in Thirty-Seventh Congress, 122; Death of, 123.
  BALLY, Joseph, Representative in Thirty-Seventh Congress, 122; in Thirty-Eighth, 140.
  BAKER, Edward D., Senator in Thirty-Sixth Congress, 48; in Thirty-Seventh, 122; death of, 123; colloquy with Senator Hunter, January 11, 1861, 88.
  BAKER, James M., Senator in First Rebel Congress, 401; Second, 402; vote on secession ordinance, 399.
  BAKER, Stephen, Representative in Thirty-Seventh Congress, 122.
- gress, 122.
- BALDWIN, Augustus C., Representative in Thirty-Eighth Congress, 140.
- BALDWIN, John B., vote on secession ordinance in Virginia Convention, note, 7: Representative in First Rebel Congress, 402; in Second, 402.
- BALDWIN, John D., Representative in Thirty-Eighth Con-

gress, 140; resolution of, denouncing negotiation with the rehel leaders, and votes, 295, 299. BALDWIN, Roger S., member of Peace Conference, 67; prop-

- osition for Constitutional Convention, 69. BALTIMORE, special election in, in 1861, 0; arrest of Police Marshal and Commissioners by General Banks, and Marshal and Commissioners by General Banks, and proceedings concerning, 152; further proceedings re-specting, 179; Emancipation Resolutions of the City Union Convention of, 226; Fair, President Lincoln's remarks at, 250, 251: Minutes of Police Commissioners in April 1861, 303, 304; Report in Conneils on the expenditure of \$500,000 appropriation, same period, 394-396: Union National Convention in, 403.

BALTIMORE American on arrests, 153. BALTIMORE Evening Transcript, suppression of, 192.

- BALTIMORE Exchange, adjustment proposition of, 75. BANISUMENT, proclamation of, by Jefferson Davis, 121.
- BANKING System, proceedings on repealing the, 365. BANK Notes, small, President Lincoln's veto of bill for, 359; proposed issue of, 363.
- Proposed issue 0, 505. BANES, A. P., connected with rebel commissioners, 108. BANES, Nathaniel P., Major General, instructions to Lient. Col. Ruger, respecting arrests of Maryland Legislature, 153; arrest of Police Commissioners, Marshal Kane, and others, in Baltimoro, and reasons therefor, 152
- BARKJALE, Ethelbert, Representative in First Rebel Con-gress, 402: in Second, 402.
   BARKJALE, William, Representative in the Thirty-Sixth Congress, 40; sent telegram South, 37; withdrew, 3.
   BARWELL, Robert W., commissioner to Washington, 2; and convergencement with Bresidert Breakington, 2; 2;
- and correspondence with President Buchanan, 29-32; deputy to Provisional Congress, 2, 11, 400; Senator in First Congress, 401; in Second, 402; vote on secession, 398.
- BARR, Martin W., arrest of, 153. BARR, Thomas J., Representative in Thirty-Sixth Congress, 48; adjustment proposition of, 74.
- BARRETT, J. R., Representative in Thirty-Sixth Congress, 49; adjustment proposition of, 73; member of Border
- States Committee, 73. BARRINGER, David M., member of Peaco Conference, 68. BARROW, Washington, Commissioner of Tennessee in leagu-ing her with the "Confederate" States, 5.
- BARNY, Capt. G., purchase of Government arms by, in 1860, 35.
- BARRY, Wm. S., deputy in rebel Provisional Congress, 11, 400.
- BARTOW, Francis S., deputy in Rebel Provisional Congress, 11, 400.
- BATES, Daniel M., member of Peace Conference, 67.
- BATES, Danier M., memorr of reace conference, of: BATES, Edward, Attorney General, 108; circular of, respecting ammesty, 148, 149; opinion of, concerning habeas corpus, 153-161; letter of, respecting excention of fu-gitive slave law, note, 235; opinion of, concerning pay of colored chaplains, 270, 280; citizenship, 373-384; pay of colored soldiers, 384, 355.
- BATES, James P., member of bogus legislative conneil of Kentucky, S.
   BATSON, Felix I., Representative in First Rebel Congress, 401; Second, 402.
   BATTLE, Judge, North Carolina, view on habeas corpus, 1200
- 120.
- BATTELLE, Robbins, member of Peace Conference, 67.

- BATTER, Lisha, claimed seat as Senator from Arkansas in Thirty-Eighth Congress, 140.
   BATTER, II. Henry, member of Peace Conference, 67.
   BATTER, Portus, Representative in Thirty-Seventh Con-gress, 122; in Thirty-Eighth, 140.
   BATARD, James A., Senator in Thirty-Sixth Congress, 43; Thirty-Sarahth 122; Thirty-Eighth and resignations;
- Thirty-Seventh, 122; Thirty-Eighth, and resignation of, 140; resolution respecting the arrest of the Baltimore Police Commissioners, 179; proposition on enrollment, 262.
- BAYLOR, A. R., Representative in Second Rebel Congress, 402. BEALE, Charles L., Representative in Thirty-Sixth Congress, 48.
- BEAMAN, Fernando C., Representative in Thirty-Seventh Congress, 122; in Thirty-Eighth, 140.BEAUFORT (S. C.) opened to trade, 149.BEAUERGARD, G. T., correspondence of, with Major Ander-

- son, before benbarding fort Sunter, 113, 114. Bee, San Francisco, on a Pacific Republic, 42. BELL Casper W., Deputy in rebel Provisional Congress,

400.

- BELL, H. P., Representative in First Rebel Congress, 401; in Second, 402.
- BELL, John, vote for President, 1.
- BELL, Joshua F., member of Peace Conference, 68. BELJ, Joshua F., member of Peace Conference, 68. BENJAMN, Judah P., Senator in Thirty-Sixth Congress, 48; telegram South.27: withdrew, 4: Attorney General and Secretary of War in Rebel Provisional Government, 12, 40); Secretary of War, and Secretary of State in Per-manent Government, 401; intercepted dispatch to Lamar, 151, 152; order of arrest of Judge Patterse

and Colonel Pickens, of East Tennessee, 187; order on bridge burning, 187.

- BENNIK, Henry L., Commissioner from Georgia to Virginia, 11; received by Convention, 6. BENNOK, Thomas H., on the disunion conspiracy, 390. BEENARD, General of Engineers in the service of the United
- States, a citizen of France, allusion of Attorney General
- Bates, 378. BERRET, James G., arrest and release of, 153. BERRY, N. S., Governor of New Hampshire, and signer of the Altoona Address, 233.
- BIDDLE, Charles J., Representative in Thirty-Seventh Con-
- Biolic, Charles J., Representative in Thirty-Seventh Congress, 122.
   Bielen, William, Senator in Thirty-Sixth Congress, 48; member of Committee of Thirteen, 70; proposition of adjustment and votes, 71, 72.
   Binlurgs, John, Speaker of Georgia Rebel House of Repre-sentatives, 329.
- BIGHAIVes, 525. BINGHAM, John A., Representativo in Thirty-Sixth Con-gress, 49; in Thirty-Seventh Congress, 122; proposition of adjustment, 55; bill of, to collect duties on imports, 78: substitute of, for emancipation bill, and vote upon, 201.
- BINOHAM, Kingsley S., Senator in Thirty-Sixth Congress, 48; in Thirty-Seventh, 122; death of, 123; proposition of, respecting amendment of the Constitution, 64.
- BINNEY, Horace, views of, on suspension of writ of habeas
- BINNEY, HOFACE, Views OI, on suspension of wirt of Adveas corpus, 161.
   BLACK, J. S., Attorney General and Secretary of State, 28; opinion of, upon the powers of the President, 50.
   BLAINE, James G., Representative in Thirty-Eighth Con-gress, 140; resolution respecting amendment of the Constitution, 259.
- BLAIR, Austin, Governor of Michigan, and signer of Altoona Address, 233.
- BLAIR, Francis P., jr., Representative in Thirty-Seventh Congress, 122; in Thirty-Eighth, 140: unseated, 141; Congress, 122; in Thirty-Binn, 140; unseated, 141; amendment of, to the emancipation bill, and vote upon, 199, 200; on select committee on compensated emanei-pation, 213; letter of General Halleck to, note, 248; Senate resolution supposed to refer to his case, 375; report of Honso Committee of Elections in case of, and proceedings upon, 375.
- BLAIR, Jacob E., Representative in Thirty-Seventh Congress, 122; in Thirty-Eighth, 141; substitute of, for the en-rollment bill, and vote, 267; resolutions of, relating to the war, 293.
- BLAIR, Montgomery, Postmaster General, 108; report of, on
- BLAIR, Montgomery, Postmaster General, 108; report of, on exclusion of newspapers, 188-191.
   BLAIR, Samnel S., Representative in Thirty-Sixth Congress, 48; Thirty-Seventh, 122.
   BLARE, Harrison G., Representative in Thirty-Sixth Con-gress, 49; in Thirty-Seventh, 122; resolution of, relating to the prosecution of the war, 292.
   BLANFERD, Mark H., Representative in Second Rebel Con-gress, 402.
- BLISS, George, Representative in Thirty-Eighth Congress, 140.
- BLOCKADE, President's proclamations of, 149; proclamations
- raising, of certain ports, 149. BLOW, Henry T., Representative in Thirty-Eighth Con-gress, 140.
- BOCOCK, Thomas S., Representative in Thirty-Sixth Con-BOODS, HOMBA'S, Representative in Hinty-Staff Con-gress, 49; address to people of Virginia, 40; member of rebel "Provisional" Congress, 400; member and Speaker of First Rebel Congress, 401; of second, 402. BONDS of United States, proposed State taxation of, 359, 360. BOMBAN, Milledge L., Representative in Thirty-Sixth Con-gress 40, withdraw L. Computied and Minimized 11.
- gress, 49; withdrew, 1; Commissioner to Mississippi, 11; statement of, respecting forts in Charleston harbor, 30; member of First Robel Congress, 400; elected Governor of South Carolina, 402.
- BORDER Slave State Representatives, interviews of President Lincoln with, 210, 211, 213-220. BORDER State embassies, Charleston Mercury on, 3.
- BORDER States meeting, and adjustment proposition of, 73. BOTELER, Alexander R., Representative in Thirty-Sixth Congress, 49; motion for adjustment committee of thirty-three, 52; hopeful despatch of, 39; remarks on re-con-struction, 329; member of rebel Provisional Congress, 401; of First Congress, 402.
- 401; of First Congress, 402.
  BODDINCT, E. C., Delegate from the Cherokee Nation at second session of First Rebel Congress, 402.
  BOLDIN, Wood, notion in Virginia Convention, 7; vete on secession ordinance, note, 7.
  BOULIONY, John E., Representative in Thirty-Sixth Congress, 49; refusal to withdraw, 4.
  BOUTWELL, efforgate Congress, 140; remarks on Issue of generative in Thirty-Eighth Congress, 140; remarks on Issue of generative in Thirty-Eighth Congress, 140; remarks

- on issue of emancipation proclamation, 232; resolutions on re-construction, 328.
- BOWDEN, Lemuel J., Senator in Thirty-Eighth Congress, and death of, 140.
- Borce, William W., Representative in Thirty-Sixth Con-

- gress, 49; withdrew, 1; appointed on Committee of Thirty-three and declined, 53; deputy to Rebel Pro-visional Congress, 2, 11, 400; member of First Congress, 402; second, 402; statement respecting forts in Charles ton harbor, 30; vote on secession ordinance, 398. Boyn, Sempronius H., Representative in Thirty-Eighth Con-
- gress, 140.
- BorLE, Brigadier General, order on impressments, 315. BorLE, Llewellyn, request of, for special session of Legisla-
- ture of Maryland, 8. BRABSON, Robert B., Representative in Thirty-Sixth Con-
- gress, 49.
- BRADFORD, Alexander W., member of Peace Conference, 68; Governor of Maryland, election affinirs of, 309–311. BRADLEY, Benjamin F., Representativo in Second Rebel Con-
- gress, 402. Brass, Braston, reported offered, and declined, rebel Secre-taryship of War, 402.Brass, Homas, Senator in Thirty-Sixth Congress, 43.BRANCH, A. M., Representative in Second Rebel Congress,
- 402
- BRANCH, Lawrence O'B., Representative in Thirty-Sixth Congress, 49; on select committee of investigation, 80; resolution on keeping troops in the District of Columbia, 81; reports on disposition of the navy and resignations therein, on milita bill, Buchanan's correspondence with rebel commissioners, and secession, 84, 85. BRANDEGEE, Augustus, Representative in Thirty-Eighth Congress, 140.
- BRANTON, William D., Representative in Thirty-Sixth Congress, 48.
- BRECKINRINGE, Robert J., Jr., Representative in First Rebel Congress, 401.
- BRECKINRIDGE, Robert J., Sr., temporary chairman of Balti-more Union Convention, and remarks, 403–405.
- BECKINEGE, John C., vote of, for President, 1; Vice Pres-ident, Thirty-Sixth Congress, 48; Senator in Thirty-Seventh, 122; expulsion, 123; escape, 153; proposition to withdraw troops from rebellious States, 290; candidate for Rebel Congress and views on reconstruction, 329.
- BRESHWOOD, Captain, surrender of his vessel to rebels, 28. BREWER, T. N., arrest of, 153.

- BRIDGE-BURNERS, in East Tennessee, Benjamin's order for
- hanging, 187. BRIDGERS, John L., commissioner from North Carolina to Rebel "Provisional" Congress, 12.
- BRIDGERS, Robert R., Representative in First Rebel Con-gress, 402; Second, 402.
- BRIDGES, George W., Representative in Thirty-Seventh Con-
- gross, 122. BRIGGS, George, Representative in Thirty-Sixth Congress, 48; adjustment proposition of 74. BRIGHT, Jesse D., Senator in Thirty-Sixth Congress, 48;
- Thirty-Seventh, and expulsion, 123.
- Inrry-Sevenin, and expulsion, 123. BRISTOW, Francis S., Representative in Thirty-Sixth Con-gress, 49; member of Committee of Thirty-three, 53. BROCKENERGUEN John W., member of Peace Conference, 63; deputy in Rebel Provisional Congress, 400. BRONSON, Greene C., member of Peace Conference, 67. BRONKE, John B., in favor of meeting of Legislature of Mary-land 9.

- land, 9.
- BROKE, Walter, deputy in Rebel Provisional Congress, 11, 400; views on re-construction, 329; see MEM. 401.
- Brooklyn Eagle, presented by grand jury and excluded from the mails, 188-192.
- the mails, 188-192.
  BROOKS, James, Representative in Thirty-Eighth Congress, 140; letter of George N. Sanders to, 330.
  BROOMALL, John M., Representative in Thirty-Eighth Con-Congress, 140; substitute for enrollment bill, 266; resolution concerning, 270; substitute of, for Mr. Colfax's resolution to expel Alexander Long, 387.
  BROWN, Albert Gallatin, Senator in Thirty-Sixth Congress, 48; telogram south, 37; withdrew, 5; on protecting slavery in the Territories, 90; captain in rebel service, 401: Sonator 402;
- 401; Senator in First Rebel Congress, 401; Second, 402; on repealing substitute law, 121.
- BROWN, B. Gratz, Senator in Thirty-Sixth Congress, 140; Indian enlistments, 264; substitute for reconstruction bill, 318.
- BROWN, George William, police commissioner and Mayor of Baltimore, arrest of, and reasons for, 152; minutes of the board, 393, 394.
- BROWN, James S., Representative in Thirty-Eighth Con-gress, 140; resolutions respecting reconstruction, 319, 320.
- BROWN, John Young, Representative in Thirty-Sixth Congress, 49.
- BROWN, Joseph E., Governor of Georgia, 2; proclamation of, concerning northern indebtedness, 2; call for negroes, 282.
- BROWN, William G., vote on secession ordinance, note, 7; Representative in Thirty-Seventh Congress, 122; Thirty-Eighth, 140; resolutions relating to the War, 293.

BROWNE, George H., member of Peace Conference, 67; Representative in Thirty-Seventh Congress, 122.
 BROWNE, William M., rebel Assistant Secretary of State, 12.
 BROWNEG, O. H., Senator in Thirty-Seventh Congress, 122; superseded, 123; amendments to milita bill, 275.

- BROWNSTILLE, port opened, 149.
   BRUCE, E. M., member of bogus legislative council of Kentucky, 8; member of First Rebel Congress, 401; Second, 402.
- BRUCE, H. W., member of bogus legislative council of Ken-tucky, 8; member of First Rebel Congress, 401; Second, 402.
- <sup>402,1</sup>
   BRUCE, James C., Commissioner of Virginia in making military league, 8; vote on Secession ordinance, nole, 7.
   BRUNE, John C., arrest of, 152.
   BRUNE, T.S., Assistant Secretary of Russellville (Ky.,) Congress, 8.
- gress, 5. BUCHANAN, James, changes in Cabinet of, 28; letter to the South Carolina Commissioners, 29, 32-34; repert of special committee on, 85; declines to receive Commis-sioner from Alabama, 34; jast annual message of, 49; the power to coerce a State discussed in, 49; explana-tors emergement to the Coercitivitien receiver and the tory amendment to the Constitution recommended by, 50; message of January 8, alluded to, 80; public credit during his administration, 365, 366; proposed censure of,
- utring ins administration, 303, 300; proposed censure of, and proceedings in the Senatc, 287. BUCKALEW, Charles R., Senator in Thirty-Eighth Congress, 140; amendment to civil appropriation bill, respecting testimony of colored persons, parties to suits, and vote, 243; amendment of, to national currency bill, and vote, 265 365
- BUCKNER, Aylett H., member of Peace Conference, 68.

- BUCKANS, Aylett H., Menuber of reace Conference, 6S.
   BUELL, Don Carlos, Brig, Gen., letter of, to J. R. Underwood, affering to return fugitive slaves, 250.
   BUFFINTON, James, Representative in Thirty-Sixth Congress, 48; in Thirty-Seventh, 122.
   BURCH, John C., Representative in Thirty-Sixth Congress, 49; resolution relative to California's quota of arms, 41;
- 49; resolution relative to California's quota of arms, 41; member of Committee of Thirty-three, 53; report of, 58; proposition for Constitutional Convention, 62; for a Pacific republic, 41.
  BUREAU of Freedmer's Affairs, bill to establish, and vote upon in House, 259, 260; bill reported in Senate, amendment of W. Saulsbury, and vote, 260; W. T. Wil-ley's amendment, and vote, 260; Henry Wilson's amendment, and vote, 260; postponement of bill to De-cember 20th part 260 cember 20th next, 260.
- BURKE, Martin, Col., refusal of, to obey writ of Judge Garrison, 152.
- BURLINGAME, Anson, Representative in Thirty-Sixth Congress, 48.
- BURNETT, Theodore L., Representative in First Rebel Cen-gress, 401; Second, 402.
- BUBNHAN, Alfred A., Representative in Thirty-Sixth Con-gress, 49; in Thirty-Seventh, 123.
- BURNS rescue case, resolution concerning, and Col. T. W.
- Dums resource case, resculation concerning, and Col. 7. W. Higginson, 244.
  BURNSUPE, A. E., Brig, Gen., proclamation of, to the people of North Carolina, 249, 250; order No. 38, 162; desig-nation by, of Fort Warren for confinement of Mr. Val-

- nation by, of Fort Warren for confinement of Mr, Vallandigham, 162.
  BUTENSTER, I. P., member of bogus legislative council of Kentucky, 8.
  BUTLER, Benjamin F., Maj. Gen., letter of Governor Hicks to, reply of, and Governor Hicks's protest to, 9; to General Scott, 244; affair with Erig. Gen. Phelps, 252; extract from Davis's proclamation of outlawry, note, 283.
  BUTLER, Pierce, arrest of, 153.
  BUTLER, Pierce, arrest of, 153.
  BUTLER, William O., membor of Peace Conference, 68.
  BUTLER, William O., membor of Peace Conference, 68.
  BUTLER, Flerce, arrest of, 123; Ohairman of Russell-ville (Ky.) Congress, 8; deputy in Rebel Provisional Congress, 400; Senator in First Congress, 401; in Second, 402.
- ond, 402. BURTON, GOVERDOT, of Delaware, reply of, to President's call for troops, 114.

C

- CABINET of Mr. Buchanan, 28; Mr. Lincoln, 108; Jeff. Davis, 400, 401.
- 200, 201. CADWALADER, George, General, refusal to obey a writ in the Merryman case, 154; remarks of Chief Justice Ta-
- The interfyman case, 102, remarks of other outside Larney, 155.
   CADWALADER, John, Judge United States, decision on en-rollment act, 273.
   CALHOUN, A. P., commissioner from South Carelina'to Ala-bana, 11; received by Convention, 4.
- CALHOUN, John C., bill of, relative to security of the mails. 191.
- FORNIA, vete of, for President, 1; not represented in Peace Conference, 69; members of Thirty-Sixth Congress, CALIFORNIA

69; of the Thirty-Seventh, 388; of the Thirty-Eighth, 141; secession-attempt in, 41; Union resolutions of the Assembly, 41.

- CALL for negrees, of General Joseph E. Brown of Georgia, 282; of General Magruder, 283.
- CALLS for troops, the President's, 114, 115, 270. CALVERT Charles B., Representative in Thirty-Seventh Congress, 122; views on compensated emancipation, 213-217;
- peace resolution, 295. CAMERON, Simon, Senator in Thirty-Sixth Congress, 48; Secretary of War, and resignation of, 108; order of ar-Secretary of War, and resignation of, 105; order of ar-rest of the members of the Maryland Legislature, 155; letters of, to General Butler, 245; instructions of, to Gen-eral Sherman, 247–248; extra throm Report of, and med-ification of, by the President, 249; letter of, to General McClellan respecting the Maryland election, 309; Pres-identify Mergenerative the the second ident's Message respecting the House resolution of censure of, 333, 334; remarks, Nevember 13, 1861, on col-ored troops, 416.
- ored treops, 416.
   CAMPBELL, James, opinion, as Postmaster General, on excluding newspapers from the mails, 189.
   CAMPBELL, James II., Representative in Thirty-Sixth Congress, 48; in Thirty-Seventh, 122; member of Committee of Thirty-three, 53; resolution of, recommending confiscation of slaves, &c., of orbels, 287.
   CAMPBELL, John A., letters of, to Scoretary Seward, 110, 111; to Jefferson Davis, 111.
- CAMPBELL, J. A. P., deputy in Rebel Provisional Congress, 400.
- CAMPBELL, J. Mason, Senator Mason's desire to see, 393.
- CANBY, E. R. S., General, President's order through, for C. L. Vallandigham's deportation, 162.

- Vallandigham's deportation, 162.
  CANNON, William, member of Peace Conference, 68; Governor of Delaware, proclamation of, 312.
  CAPERTON, Allen T., vote en secession ordinance in Virginia. Convention, 7, note; Senator in First Rebel Congress, 401; Second, 402.
  CARLIE, John S., Representative in Thirty-Seventh Congress, 122; Senator in Thirty-Seventh Congress, 123; in Thirty-Eighth, 140; offer of Peace Conference proposition in Virginia Convention, and vote, 7; vote of, on secession ordinance, note, 7; substitute of, for indemnity bill and vote upon, 185; on compensated emancipation, 213-217; resolution of, relating to the prosecution of the war, 290; resolutions cf, on the relations of the rebellicus States to the Government, 326.
  CARROLL, James, of Maryland, Disunien letter of, 9.

CARROLL, James, of Maryland, Disunion letter of, 9. CARTER, Luther C., Representative in Thirty-Sixth Congress,

48.

- CAREY, John, Representative in Thirty-Sixth Congress, 49. CARUTHERS, Robert L., member of Peace Conference, 68. CASE, Charles, Representative in Thirty-Sixth Congress, 49.
- 49.
  CASEY, Samuel L., Representative in Thirty-Seventh Congress, 123; on Select Committee on compensated emancipation, and views on, 213, 217, 218.
  CASS, Lewis, Scerctary of State, and resigned, 28.
  CHAMBERLAIN, Jacob P. Representative in Thirty-Seventh Congress, 123.
  CHAMBERLAIN, Levi, member of Peace Conference, 67.
  CHAMBERS, Ezekiel F., proposed member of Board of Public Safety in Maryland, 9; vote on emancipation in Maryland, 277.

- land, 227.
- CHAMBERS, Henry C., Representative in First Rebel Congress, 402.

- gress, 402.
  CHAMBLESS, John R., voto on Secession Ordinance in Virginia Convention, note, 7; Representative in First Rebel Congress, 401.
  CHANDLER, L. H., claimant for seat as Representative In Thirty-Eighth Congress, 141.
  CHANDLER, Theophilus P., member of Peace Conference, 67.
  CHANDLER, Zachariah, Senator in Thirty-Sixth Congress, 48; Thirty-Seventh, 122; Thirty-Eighth, 140; amendment to enrollment bill, 264.
  CHANDLER, Lehn W. Representative in Thirty-Eighth Congress, 40.
- CHANLER, John W., Representative in Thirty-Eighth Congress, 140.

- gress, 140. GRAPLATS, Alexander, resolutions in the Legislature of Maryland, 308. Charlotte Bulletin, Clingman's telegram to, 41. Clusse, Salmon P., member of Peace Conference, 68; Secre-tary of the Treasury and resignation, 108; letter on "Legal Tenders," 358, 359; National Currency bill, note, 305; redemption of Treasury leans and netes, note, 307. CHESNET, James, Jr., Senator in Thirty-Sixth Congress, 48;
- CHESNET, James, Jr., Senator in Thirty-Sixth Congress, 48; resigned, 2; deputy in Rebet Provisional Congress, 2, 11,400; Aid to Beauregard, 113; vote on Secession Ordinance, 398.
- CHILTON, William P., deputy in Rebel Provisional Con-gress, 11, 400; Representative in First Rebel Congress, 401; Second, 402.
- CHITTENDEN, Lucius E., member of Peace Conference, 67. CHRISMAN, James S., member of bogus legislative council of Kentucky, 8; deputy in Rebel Provisional Congress,

400; Representative in First Robel Congress, 401; Second, 402

CINCINNATI Gazette, upon a Northwestern Confederacy in-trigue in 1860, 42.

CITIZENS, arrest of, 152-154. See "Arrests."

CITIZENSHIP, Attorney General Bates's opinion on, 378-384; Secretary Marcy's, through J. A. Thomas, Assistant Sec-

- Secretary Marcy's, through J. A. Thomas, Assistant Secretary of State, note, 382.
   CLAPF, J. W., Representative in First Rebel Congress, 401.
   CLARE, Ambrose W., Representative in Thirty-Seventh Congress, 122; Thirty Lighth, 140.
   CLARE, Daniel, Senator in Thirty-Sixth Congress, 48; Thirty-Seventh, 121; Thirty-Eighth, 140; amendment to the Crittenden proposition of adjustment, 65; vote upon, reconsideration, and final vote, 66; bill to ratify the emancipation proclamation, 229; resolution on prosecuting the war, 201.
   CLARE, Morace F., Representative in Thirty-Sixth Congress,
- CLARK, Horace F., Representative in Thirty-Sixth Congress, 48; adjustment proposition, 74.
- CLARK, JOHN B., Representative in Thirty-Sixth Congress, 49, Thirty-Seventh, 122; expulsion, 123; Senatorin First Rebel Congress, 401; Second, 402; remarks in favor of repealing the substitute bill, 121.
- CLARE, William W., Representative in First Rebel Congress, 401.
- CLARKE, Freeman, Representative in Thirty-Eighth Congress, 140.
- CLAY, Brutus J., Representative in Thirty-Eighth Congress, 140.
- CLAY, Clement C., Jr., Senator in Thirty-Sixth Congress, 48;
- CLAY, Ja 153.
- CLAYTON, A. M., deputy in Rebel Provisional Congress, 11, 400.
- CLANTON, Philip, rebel Assistant Secretary of Treasury, 12. CLEGGETT, Thomas, arrest of, 153. CLEMENS, Jeremiah, statement respecting the bombardment
- of Sumter, 112
- CLEMENS, Sherrard, Representative in Thirty-Sixth Congress, 48; adjustment proposition of, 63; vote on secession ordinanco in Virginia convention, note, 7.
- CLEMENTS, Andrew J., Representative in Thirty-Seventh Congress, 122; on Select Committee on compensated emancipation, 213; views on, 217, 218; resolution re-specting the insurrectionary States, 324. CLEERE, Justice, adverse opinion on indemnity act, 186, 187. CLEERE, Justice, adverse opinion on indemnity act, 186, 187.

CLEVELAND, Chauncey F., member of Peace Conference, 67. CLINGMAN, Thomas L., Senator in Thirty-Sixth Congress, 48; re-elected, 5; telegram to the Charlotte Bulletin, 41; on Peace Conference proposition, 70; resolution re-specting slave protection in the Territories, 90; proposition to withdraw troops from the rebellious States, 290.

- 290. Clipper, Baltimore, on loyalty of Baltimore, S. CLOPTON, David, Representative in First Rebel Congress, 401; Sccond, 402. COLIER, John D., member of Peace Conference, 68. COASTWISE slave trade, repeal of laws regulating, 243, 244. COBB, Amasa, Representative in Thirty-Bighth Congress, 141. Comp. Generator T. Benressentative in Thirty-Seventh Con-

- Comp. Annasa, Representative in Thirty-Eighth Congress, 141.
   Comp. George T., Representative in Thirty-Seventh Congress, 122.
   Comp. Howell, resignation of, as Secretary of the Treasury, 38; deputy in Rebel Provisional Congress, and president thereof, 12, 400; pronounces secession a "fixed and irrevocable fact," & c., 12; major general in rebel service, 400 400.
- COBB, Thomas R. R., deputy in Rebel Provisional Congress,
- COBE, THOMAS K. R., deputy in Kepel Provisional Congress, 11, 400; on reconstruction, 229; death of, 400.
   COBE, Williamson R. W., Representative in Thirty-sixth Con-gress, 40; withdrew, 3; elected to Second rebel Con-gress, 402; inquiry on his loyalty, 400; came within our lines, note, 402.
- note, 45; member of Peace Conference, 67. COGHRANE, Clark B., Representative in Thirty-Sixth Con-gress, 43.
- COCHRANE, John, Representative in Thirty-Sixth Congress, 48; adjustment propositions of, 53, 74; views on im-posts bill, 79; on select investigating committee, 80; committee, 80; investigating committee, 80; poses unit, 19; on select investigating committee, 80; reports on President Buchanan's correspondence with rebel commissioners, and on seccession, 55; president of Cleveland Convention, 411; candidate for Vice-Presi-dent, 412; acceptance of, 414; address to his regiment, 414-416;
- COERCION, right of, Buchanan's opinion, 49; vote and reso-
- lution upon, 77. Correy, Titian J., ac nesty, 148-149. acting Attorney General, circular on am-

COFFROTH, Alexander H., Representative in Thirty-Eighth

Congress, 140; resolution on emancipation proclamation, 290.

- Colf, Cornelius, Representative in Thirty-Eighth Congress, 140.
- COLFAX, Schuyler, Representative in Thirty-Sixth Congress,
- COLFAX, Schuyler, Representative in Thirty-Sixth Congress, 49; in Thirty-Seventh, 122; in Thirty-Eighth, and Speaker of the House, 140; adjustment proposition of, 73; resolution for amending the fugitive slave law, 236; resolution of, to expel Alexander Long, 357.
  COLLANER, Jacob, Senator in Thirty-Sixth Congress, 48; amendment of, to resolution for prohibition of slavery in the Territories, 90; Sonator in Thirty-Seventh Con-gress, 122; in Thirty-Eighth, 140; resolution of, to reduce the period of service of drafted men to one, year, &c., and vote, 263; amendment of, to enrollment, bill, and vote, 264; amendment to joint resolution to equalize the pay of soldiers, 277; amendment of, to legal tender bill, 358.
  COLLER, Charles F. Representative in First Rebel Congress,
- COLLIER, Charles F. Representative in First Rebel Congress, 402
- COLONIZATION, the President's views on, 213; interview with colored persons in August, 1862, 374, 375; the Isle & Veche experiment, note, 212.
- COLOR, to remove disqualification of, in carrying United States mails, 239, 240. COLORADO Territory, bill for temporary government, 89, 90; enabling act for, 255.
- COLORED men as citizens, opinion of Attorney General Bates, respecting, 378-384; letter of Assistant Secretary of State
- respecting, 318-354; letter of Assistant secretary of state J. A. Thomas on, note, 382.
  CoLORED persons, Confederate use of, 281-283.
  COLORED persons, disqualification of, in carrying the mails, proceedings to remove, and votes, 239, 240; letter of Gideon Granger, nucle, 239; disqualification of, to testify, proceedings to remove, 242, 243; all able-bodied males af suitable age to be enrolled, 274; to be liable to, the draft 274; combine of Attornay General Bates to the draft, 274; opinion of Attorney General Bates on citizenship of, 378-384.
- con citizenship of, 375-334.
  COLORED persons, exclusion of from the cars, proceedings respecting, 241, 242; Charles Summer's resolution on, and vote, 242; Waitman T. Willey's report upon, and vote, 242; Charles Summer's proviso forbidding and vote, 242; Charles Summer's proviso respecting, to the bill to amend the Washington and Georgetown Railroad Company, and votes, 242.
  Cotonens schols, legislation respecting, 244.
  Cotonens choliers, to have same pay, &C., except bounty, as white men-bounty of, not to exceed \$100, to be fixed by the President, 116, 272; the several laws concerning, 274; mother, wife, and children of, to be free, 274; proceedings in Congress respecting, 274-276; respecting pay of, 276-279; opinions of Attorney General Bacts re-
- pay of, 276-279; opinions of Attorney General Bates re-specting pay of, 270, 280, 384, 385; Secretary Stanton's order to Governor Andrew respecting enlistment of, 279; the President's order for the protection of, 230; rules of war (extract) concerning, 230; allusion to, in speech of President Lincoln at the Baltimore fair, 230, 231; proposed prohibition of pay to, 283, 284; rebel facts concerning, 281–283. Colvin, A. J., letter of Lucius Robinson to, 413.
- COMMERCIAL intercourse, proclamations forbidding, with in-
- COMMERCIAL intercourse, proclamations forbidding, with in-surrectionary States, 149, 150.
  Commsstorers, inter-State, names of, 11; rebel to England, appeal of, to Earl Russell, 27; of "Confederate" States, correspondence of, with President Buchanan and Secre-tary Holt, 29-34; with Secretary Seward, 108-110.
  Committee of Thirteen in Senate, votes in, 70, 72.
  Committee of Thirty-three, in House, 52; members of, 53; propositions submitted to, 52-55; report of, 57, 58; mi-nority reports of, 57, 58; votes on propositions of, 58-62.

- norty reports of, 94, 55; votes on propositions of, 58-52. COMMUTATIONS, or S200 clause, votes on, 201; repealed, 263-270; report on, node, 263. COMPEONISK, Mr. Lincoln's opinion upon, 67. "COMPEDERATE," Legislation—see "Rebell" Legislation. COMPEDERATE, "Legislation—see "Rebell" Legislation. COMPEDERATE, "Legislation proceedings upon its passage, 195, 196; act of 1862, and for other purposes and pro-ceedings upon its passage, 196, 197; joint resolution giving construction to last, 197; President's messago thereon 197, 198; other proceedings respective, 198-203; giving construction to last, 197; President's message thereon, 197, 198; other proceedings respecting, 198-203; votes on repealing the joint resolution, 202, 206, 260; judicial action under, 206, 207; military confiscation suspended by the President, note, 207; President's proc-lamation under, 208; proceedings respecting, 275, 276.
  Cóxensss, of United States, meeting, and members of Thirty-Sixth, second session, 48, 49; Thirty-Seventh, 121, 122; Thirty-Eighth, 140, 141; of "Confederate" States, "pro-visional," and members of, 11, 400; of First Congress, 401, 402; of Second, 402.
  Coxkling, James C., President Lincoln's letter to, 235, 236.
  Coxkling, Frederick A., Representative in Thirty-Seventh Congress, 122.
- Congress, 122.
- CONKLING, ROSCOC, Representative in Thirty-Sixth Congress, 48; Thirty-Seventh 122; amendment to confiscation

bill, 198; to Mr. Eliot's joint resolution, 287; joint resolution on compensated emancipation, 209, 210.
CONNECTICT, vote for President, 1; members of peace conference, 67; of Thirty-Sixth Congress, 48; Thirty-Seventh, 122; Thirty-Eighth, 140.
CONNESS, John, Senator in Thirty-Eighth Congress, 140.
CONNESS, Charles M., deputy in rebel Provisional Congress, 11, 400; Representative in First Rebel Congress, 401; of Second 402.

- of Second, 402. Connow, A. H., deputy in Rebel Provisional Congress, 400;

- CONROW, A. H., deputy in Rebel Provisional Congress, 400; Representative in First Congress, 401; Second, 402.
  CONSCRIPTION-See "Enrollment."
  CONSCRIPTION-SEE "Enrollment."
  CONSCRIPTION, Rebel, acts of, 118, 119; judicial decisions upon, 120; proclamations and order upon, 118, 119.
  CONSTRUTION, Rebel, acts of, 118, 119; judicial decisions 376, 377; Douglas's proposition respecting, 71.
  CONSTRUTION of the United States, 91-97; index to, 100-3; explanatory amendment to, proposed by President Buchanan, 50; adoption of proposition to amend, 59, 600; William H. Seward's proposition, 71; votes on proposi-tion to amend, 59, 60, 62-65; proceedings to amend, 255 -259; proposition of John B. Henderson, 255; Charles Summer, 255; Lazarus W. Powell, 255, 256; Lyman Trumbull, 256; Garrett Davis, 256; Willard Saukbury's proposition, 257; final vote on amendment in Senate, proposition, 257; final vote on amendment in Senate, 257; Ezra Wheeler's proposition, 257; final vote in the House, 258; resolutions of Isaac N. Arnold, 258; Wil-liam Windom, 258; Thaddeus Stevens, 258; Henry B. Anthony's joint resolution to repeat the joint resolu-tion of Measure 1, 1881; 1884; resolution of convention of tion of March 2, 1861, 258; resolutions of convention at Allegheny City, Pennsylvania, proposing amendment to, 258, 259; James A. Cravens's resolutions respecting amendment of, 259; James G. Blaine's proposition to amend, 259.
- Constitutional amendment for the extinction of slavery, 257, 258; other propositions, 255-259-see "Adjustment" and "Constitution."

CONSTITUTIONAL Convention, votes on, 62, 64; in Peace Con-ference, 69, 70.

- CONSTITUTION of the "Confederate" States, 98-100; index to, 100-103; A. H. Stephens's comment upon, 103, 104; ratification of, 4, 5.
- CONTRADANDS, order of Major General Wool respecting, 248; his order respecting employment of, and their wages, 248; general instructions concerning, 244, 245, 247, 248; order of President Lincoln respecting, in the District of Columbia, 248; extract from report of Secretary Camer-or 200, instructions of Secretary Camer-Contact, 253, contractions of Secretary Stanton to General Saxton, 251, 252; orders of General Tuttle at Natchez, 253; Josiah B. Grinnell's resolutions of inquiry respect-ing General Tuttle's orders, 253.
- Convention of South Carolina, conflicting views of fugitive slave law in, 18; vote on secession ordinance, 398, 399; address of, and papers in, 12-20.
- CONVENTIONS, Political, at Baltimore, 403-407; at Cleveland, 410-413.

- CONTENTIONS, Political, at Baltimore, 403-407; at Cleveland, 410-413.
  CONWAY, Martin F., Representative in Thirty-Sixth Congress, 49; in Thirty-Seventh Congress, 122; resolutions on prosecution of the way. 292.
  CONK, Burton C., member of Peace Conference, 68.
  CONT, Burton C., member of Peace Conference, 68.
  CONTER, S., rebel Adjutant and Inspector General, 119.
  COOPER, N., R., Representative in Thirty-Seventh Congress, 122; death, 123.
  CONNNO, Erastus, member of Peace Conference, 67; Representative in Thirty-Seventh Congress, 122; resolution relative to taxation, 350; correspondence with the President on Vallandigham's cares, 162-167; same with Ohio Democrats, 167-175; of President Lincoln and Governor Bradford, 209, 201; President Lincoln and Governor Bradford, 209, 301; President Lincoln-and Fernando Wood, 296, 207; the Niagara Falls, 301-302; on Foreign Affairs, 339-355.
  or Jeff, Davis and Governor Vance, 306, 307; of Alexander H. Stephens and Gideon Welles, note, 307.
  CONWIX, Thomas, Representative in Thirty-Sixth Congress, 49; chairman of Adjustment Committee of Thirty-three, 53; report of, 57; votes upon, 58-62; Minister to Mexico, granted leave of absence, 350.

Costr, Capitalit, surrender of his vessel to rebels, 27.
 Costruin, surrender of his vessel to rebels, 27.
 Cottuin, Thomas, claimed seat as Representative from Louisiana, in Thirty-Eighth Congress, 141.
 Cortox, rebel regulation of destruction of, 117.
 Cortoner, John, Representative in Thirty-Sixth Congress, 48; in Thirty Sourcet 122.

- in Thirty-Seventh, 122. Cowax, Edgar, Senator in Thirty-Seventh Congress, 122: in
- Thirty-Eighth, 140; amendment on pay of soldiers, 222, 11 Thirty-Eighth, 140; amendment on pay of soldiers, 276. Cox, Sanuel S., Representative in Thirty-Sixth Congress, 49; in Thirty-Seventh, 122; in Thirty-Eighth, 140; adjustment proposition, 55; resolution respecting arrests,

181; amendment to bill for recognition of Haytl and Liberia, 239; amendment of, to Montana bill, 254, 255; motion to insert the word "white" in the enrollment bill, 261; amendment of, to bill relating to unemployed bill, 261; amendment of, to bill relating to unemployed generals, 285; resolution of, defining the word "assass-ins," 291, 292; peace resolution of, 295; resolution on the Arguelles ease, 355; relative to a colony of blacks, *mote*, 212; proviso to Nebraska enabling act, 377; pro-posed increase of the pay of soldiers, 271. Cox, Walter S., letter respecting secession, 9. Craster, James, Representative in Thirty-Sixth Congress, 49. Outgo, Thurker Development of Thirty Sixth Congress, 49.

- Cn.xia, James, Representative in Thirty-Sixth Congress, 49;
  Cn.xia, Burton, Representative in Thirty-Sixth Congress, 49;
  telegram sont South, 37; remarks upon Peace Conference proposition, 70; resolution on recognition, 89;
  deputy in Rebel Provisional Congress, 400.
  Cn.vENS, James A., Representative in Thirty-Seventh Congress, 122; in Thirty-Eighth, 140; resolutions of, respecting amending the Constitution, 259.
  CRAWFORD, Martin J., Representative in Thirty-Sixth Congress, 49; telegram South, 37; withdrew, 3; deputy in rebel Provisional Congress, 11, 400; resolution respecting avery. 76; rebel commissioner to Washington.
- ing slavery, 76; rebel commissioner to Washington, 108; colonel of cavalry, 400.
- CREOUES, rebel proposed military employment of, 282; reason given for declining, 282. CRESWELL, John A. J., Representative in Thirty-Eighth Con-gress, 140.
- CEISFIELD, John W., member of Peace Conference, 68; Rcp-resentativo in Thirty-Seventh Congress, 122; memoranda by, of an interview between the President and some border State Representatives, March 10, 1862, 210, 211; views on compensated emancipation, 213–217. CRITTENDEN, John J., Senator in Thirty-Sixth Congress, 483 member of the Committee of Thirteen, 70; of border
- member of the Committee of Infreen, 10; of border State Committee, 73; telegram of, stating that his ad-justment was lost in the Senate by refusal of southern Senators to vote, 39; hopeful dispatch of, 39; various votes on his proposition, 64-67, (also see Adjustment;) remarks of Senators Johnson and Latham respecting defeat of the adjustment of, 66, 67; telegram respecting same, 38; alleged authorship of the measure, 75; vote of the obst of eact Viewing Convortion 6; renorabiliton of. of thanks to, of Virginia Convention, 6; proposition of, in Senate Committee of Thirteen, 70; amendment respecting fugitive slave law, 70.
- CRITENDER, Join J., Representative in Thirty-Seventh Con-gress, 122; resolution of, respecting the cause and the objects of the war, 286; at border State interview with the President, 211; views on compensated emancipation; 213-217.
- CROCKETT, John W., member of bogus legislative council of Kentucky, 8: of First Rebel Congress, 401; peace prop-osition of, 303. CROSET & Nichols, President Lincoln's letter to, 336.

CROWNINSHIELD, Francis B., member of Peace Conference, 67. CRUICESHANN, Marcus H., Representative in Second Rebel Congress, 402.

- CULLON, Alvin, member of Pcace Conference, 68. CURLON, Alvin, member of Pcace Conference, 68. CURRY, Jabez L. M., Representative in Thirty-Sixth Com-gress, 49; telegram South, 37; withdrew, 4; deputy in rebel Provisional Congress, 11, 400; Representative in First Congress, 401. CURTIN, Andrew G., Governor of Pennsylvania, signer of
- Altoona Address, 233.
- CURTIS, Samuel R., Representative in Thirty-Sixth Congress, 49; in Thirty-Seventh, 122; resigned, 123; member of Committee of Thirty-three, 53; member of Peace Conference, 68.

CUSHING, Caleb, opinion as Attorney General on newspaper exclusion from the mails, 189, 190. CUTLER, William P., Representative in Thirty-Seventh Con-

gress, 122.

## D

- Daily News, New York, presented by grand jury and excluded from the mails, 188; report thereon, 188-192
   DAROTAH, bill to organize, 89.
   DARGAN, Edward S., Representative in First Rebel Congress, 401; remarks on creoles, 282.
   DARGAN, J. A., in South Carolina Convention, 17, 20; voteom second constrained and second constrained and second constrained and second constraints.

secession ordinance, 398.

seccession orunance, 093. DAVIDSON, Andrew J., deputy in rebel Provisional Congress, '400; Representative in First Congress, 402. DAVIDSON, Thomas G., Representative in Thirty-Sixth Con-gress, 40; withdrew, 4. DAVIS, David, United States Judge, decision on enrollment act, 273.

DAVIS, George, member of Peace Conference, 68: Senatorin

DATIS, George, member of Peace Conference, 05: Senatorin First Rebel Congress, 401; rebel Attorney General, 401.
 DATIS, H. Winter, Representative in Thirty-Sixth Congress, 49; in Thirty-Eighth, 140; member of Committee of Thirty-three, 53; Address of for the Union, 9; amend-ment to the enrollment bill, 270; bill for reconstruc-tion, 317; resolutions respecting the French in Mexico,

349, and report thereon, 350-354: paper of, and Benjamin F. Wade, respecting failure of the reconstruction bill, 332.

- DAVIS, Garrett, Senator in Thirty-Seventh Congress, 123; in Thirty-Eighth, 140; views on compensated emanci-In Thirty-Eighth, 140; Yiews on compensated enhanc-pation, 213-217; motion to amend the new Article of War, 233; resolution respecting Col. T. W. Higginson and the Anthony Burns affair, 244; proposition to amend the Constitution, 256; proposition to divide New England, 259; amendment to bill to employ col-ored persons in the public service, 275; amendment to bill to suppress insurrection, 275; amendment to the engineer bill, 276; substitute for and amendments to joint resolution to equalize pay of soldiers, 277; amend-ment to the army apmromriation bill. 278: amendment ment to the army appropriation bill, 278; amendment to internal revenue bill, that no money raised by it shall be used to pay negroes, 253; like amendment of, to army appropriation bill, 254; resolution for a peace convention at Louisville, Kentucky, 296; joint resolution for negotiation for peace, 297; resolutions respect-ing the relations of the rebellious States to the Gov-ernment, 222-526; resolution censuring Ex-President Buchanan, 387.
- DAVIS, Jefferson, Senator in Thirty-Sixth Congress, 48; member of Committee of Thirteen, and proposition in, T1; withdrew, 3; intermediary between Col. Hayne and President Buchanan, 32; opinion concerning the militia of the States, 37; on newspaper exclusion from the mails, note, 100; bill to authorize sale of United States arms to the States, 36; discussion thereof in the United States Control 20; States and Stat United States Senate, 36, 37; resolutions of, on with-drawal of United States troops, 85, 86; Franklin Pierce's letter to, 391.
- DAVIS, Jefferson, elected Rebel Provisional President, 12; inaugurated, 12; Provisional Cabinet, 12; elected permanent President, 401; martial law declared in Rich-mond, 121; proclamation of banishment by, 121; corres-pondence with Governor Vance respecting peace, 300; 307; views on reconstruction, 329, 330, 331; declaration to Gilmore, 307; veto of African slave trade bill,
- 151; proclamation on conscription, 118.
   DATE, John G., Representative in Thirty-Sixth Congress, 49; proposition of adjustment, 56.
   DAVIS, Nicholas, Jr., deputy in Rebel Provisional Con-
- gress, 400.
- DAVIS, REUBEN, Representative in Thirty-Sixth Congress, 49; member of Committee of Thirty-three, 53; mani-festo sent South, 37; statement to slaveholding caucus, 37; withdrew, 3; Representative in First Rebel Congress, 401; and resigned, 402.
- DAVIS, Thomas T., Representative in Thirty-Eighth Congress, 140.
- BAVIS, William Morris, Representative in Thirty-Seventh Congress, 122.
- DAWES, Henry L., Representative in Thirty-Sixth Congress, 48; in Thirty-Seventh, 122; in Thirty-Eighth, 140; member of Select Committee on military and naval affairs, 80; report of, on disposition of United States vessels, 82; also respecting resignations in the navy, 82-84; vote upon same, 85; report and resolutions by, respecting commission of inquiry, 319. DAWENNS, James B., Representative in First Rebel Congress
- 401
- DAWSON, John L., Representative in Thirty-Eighth Congress, 140; resolution of, for amnesty, and vote, 299, 800; pro-posed increase of pay of soldiers, 272.
- Darrow, William L., letters of Secretary Seward to, on me-diation, 345, 349, 350; letters of, to Secretary Seward, on the French policy in Mexico, 350; Secretary Sew-ard's letter to, 350; report of the Committee on Foreign Affairs upon, 351-354.
- DEBATE on adoption of the papers of the South Carolina Convention, 16-18; in Virginia rebel Legislature on enroll-ing free negroes, 281, 282; and on peace resolutions, Sol; in rebel Congress, on pace proposition, 306; on sequestration, 205; on repealing substitution, 121. rs, payable in "legal tenders," 357, 358; Secretary Chase's letter respecting payment of United States,
- DEETS, note, \$67.
- DELANO, Charles, Representative in Thirty-Sixth Congress, 48; in Thirty-Seventh, 122.
- DECLARATION of Independence of South Carolina, 15, 16. DECLARATORY resolutions of Committee of Thirty-three, and
- vote on, 58, 59. DE CLCUET, A., Deputy in rebel Provisional Congress, 11,
- 400.
- DE JARFETTE, Daniel C., Representative in Thirty-Sixth Congress, 49; address to people of Virginia, 40; Rep-resentative in First Rebel Congress, 402; Second, 402.
- DELAPLAINE, Isaac C., Representative in Thirty-Seventh Congress, 122.
- DELAWARE, vote of for President, 1 ; members of Peace Con-ference, 67 ; of Thirty-Sixth Covgress, 48 ; of Thirty-Seventh, 122 ; of Thirty-Eighth, 140 ; act to secure the

freedom of elections in the State of, 312; General Schenck's order concerning election in, 312; Governor William Cannon's proclamation upon, 312. General

- DEMING, Henry C., Representative in Thirty-Eighth Congress, 140
- DEMOGRATS Of New York, on foreign mediation, estated by Lord Lyons, 347; on C. L. Vallaudigham's arrest, 163. DEMOGRATS of Obio, on C. L. Vallaudigham's arrest, 167-
- 175. DENISON, Charles, Representative in Thirty-Eighth Con-
- gress, 140. DENNIS, John U., arrest of, 152; votes of, in Legislature of
- Maryland, 379, 398, DENNISON, Mr., arrest of, 152. DENNISON, William, President of Baltimore Convention, and
- remarks, 405; announcement of nomination to the Pres-
- ident, 407. DENT, John F., member of Pcace Conference, 68. DE WITT, Mr., deputy in Rebel Provisional Congress, 400. DICKINSON, Henry, Commissioner from Mississippi to Delaware, 11.
- DICKINSON, J. S., Representative in Second Rebel Congress, 402.
- DIMMICK, William H., Representative in Thirty-Sixth Congress, 48.
- Dispatch, Richmond, on re-construction, 330; on amnesty. 331.
- DISTRICT of Columbia, bill to reorganize militia of, 77; com-
- pensated emancipation in, 212, 213. DISUNION, the conspiracy of, 389-399; an early threat of dis-union, extract from Senator Maclay's journal, 389; sentiment of Richard Henry Lee, 389; General Jackson's letter to Rev. A. J. Crawford on, 389; Thomas H. Ben-ton's views, 300; Henry Clay's views, 380; Nathan Ap-pleton's views, 390; extract from speech of Ex-Gover-nor Francis Thomas, respecting, and extract from the nor Francis Thomas, respecting, and extract from the National Intelligencer, concerning, 300; early hopes of the rebels, 390; extract from the speech of Lawrence M. Keitt, 300; recently found letters in Fredericksburg, 300; proceedings of a Democratic meeting in Philadel-phia in January, 1861, note, 300; letter of Ex-President Franklin Pierce to Jeff. Davis, of January 6, 1860, 391; extract from the National Intelligencer of January 11, 1861, and letter of "Eaton," disclosing the programme of, 391; distacth to the Charleston Mercarv of January 7, 1860, disclosing plans of, 391, 392; letter of David L. 7. 1860, disclosing plans of, 391, 392; jetter of David L. Yulee, and resolutions referred to, and comments of New York *Times* on. 392; James L. Pugh's confession, 392; Stephen A. Douglas's farewell words, 392; the conspiracy in Maryland, 392-400; Marshal Kane's dis-patch to Bradley T. Johnson, *note*, 302; arrest of Mar-shal Kane and the Board of Police of Baltimore, by Cameral Parkar methands. General Banks, mentioned, 393; extracts from the minutes of the Police Commissioners, and from their Letter Book, found in their office, 393; proceedings of the Maryland Legislature of 1861, 396-398-(see "So-cession.")
- Dix, John A., appointment of Secretary of the Treasury, 28; bids for public leans, under, 366; major general, seizure of newspaper offices in New York, arrest of, and subsequent proceedings, 192-194; proclamation of, in Virginia, 248; orders and letters concerning elections, 308, 309.
- DIXON, James, Senator in the Thirty-Sixth Congress, 48; in Thirty-Seventh, 122; in Thirty-Eighth, 140. on, Nathan, Representative in Thirty-Eighth Congress,
- DIXON. 140.
- DOCK-vards of United States, employment of slaves in, 238; DODGE, George P., provost marshal, arrests by, 152; General
- DOUDER, GENERGY, PROVOST MARSHAL, ATTESTS by, 152; General Dix's order to; 308, 309.
   DODER, William E., member of Peace Conference, 67.
   DONTELT, Ignatius, Representative in Thirty-Eighth Con-gress, 141.
- DOOLITTLE, James R., Senator in Thirty Sixth Congress, 48; Thirty-Seventh, 122; Thirty-Eighth, 140; member of Peace Conference, 68; of Committee of Thirteen, 70; pro-position respecting secession, 63; amendment to fug-tive slave law, 70; explanation of votes, note, 243; amendment to enrollment bill, respecting Indians, 264; amendment to "legal tender" bill, 358. Doвтся, George T., Senator in First Rebel Congress, 401; Seo-
- ond, 402.
- DOUBLEDAY, Abner, General, order forbidding the rendition of fugitive slaves, 250.
- DOUGLAS, Stephen A., Senator in Thirty-Sixth Congress, 48; vote of, for President 1, 1 member of Committee of Thirteen, 70: hopeful despatch of, to James Barbour, 39: propositions of, in Senate Committee of Thirteen, 72: amendment respecting conspiracies, 71; his farowell words, 392.
- DOUGLASS, Frederick, letter to Cleveland Convention, 413. DRAFTS, the several, ordered, 115, 270: orders issued by the Secretary of War to enforce the, 272; President's proc-

lamation concerning, 272; judicial decisions respecting | EMANCIPATION, compensated, James B, McKcan's resolution

constitutionality, 272–274; instructions to United States Marshals respecting, 272; Governor Seymour and, 274; Gen. McIlellan on, 274; R. B. Marcy's despatch on, 274, Drages, Jon F., Representative in Thirty-Eighth Con-

gress, 140. DROUYN de l'Huys, M., letter of, to M. Mercier, on media tion, (extract.) 345; Secretary Seward's reply, 345, 346; circular letter of, respecting the relation of France to the American Government, 354.

DUELL, R. Holland, Representative in Thirty-Sixth Congress, 48; in Thirty-Seventh, 122

Duce Domingo, Captain General of Cuba, letter respecting Arguelles Case, 355.

DUMONT, Ebenezer, Representative in Thirty-Eighth Con-

gress, 140. DUNAR, Jordensen, Alexander, member of Peace Conference, 67. DUNAR, George W., Representative in Thirty-Seventh Con-gress, 122; on compensated emanchation, 213-217.

DUNN, William McKee, Representative in Thirty-Sixth Con-gress, 49; in Thirty-Seventh, 122; member of Com-mittee of Thirty-Three, 53; adjustment proposition, 54.

DUPRE, Lucius J., Representative in First Rebel Congress, 401; second, 402.

DURANT, C. J., arrest of, 153.

DURKEE, Charles, Senator in Thirty-Sixth Congress, 48; member of Peace Conference, 68.

 $\mathbf{E}$ 

- EARL RUSSELL, to Lord Lyons on Rebel appeal for recogni-tion, 27; on Trent affair, 338-342; Lord Lyons, to, res-pecting New York democrats or conservatives on foreign mediation in American affairs, 347, 348.
- EAST TENNESSEE, Col. W. M. Churchwell's order to people of, 121; bridge-burning order of Judah P. Benjamin,
- and arrests in, 187. \* EATON," communication of, revealing proceedings of dis-uuion caucus, 301; understood to be, note, 391.

ECHOLS, J. H., Representative in Second Rebel Congress, 402

ECKLEY, Ephraim R., Representative in Thirty-Eighth Congress, 140.

EDEN, John R., Representative in Thirty-Eighth Congress, 140.

EDGERTON, Joseph K., Representative in Thirty-Eighth Congress, 140; resolution of denouncing coercion, &c., and vote upon, 220; resolution on emancipation proclama-tion, &c., 231.

EDGERTON, Sidney, Representative in Thirty-Sixth Congress, 49; in Thirty-Seventh, 122; substitute of, for resolution of thanks to Captain Wilkes, and vote, 343.

- ButUNSOY, Henry A. Representative in Thirty-Sixth Congress, 40; address to the people of Virginia, 40.
   EDWINSOY, Thomas M., Representative in Thirty-Sixth Congress, 48; in Thirty-Seventh, 122.
   ELDEDEL, Charles A., Representative in Thirty-Eighth Congress, 40; and 40.
- gress, 141 resolution on political arrests, 153; on en-rollment, 271; on commutation, 271. ELECTION: of President, in 1860, 1; propositions concerning the, 54, 55, 57, 73, 87, 255, 256.
- ELECTIONS, military orders respecting, 308-316; bill to pre-vent officers of army and navy from interfering in, 315, 316.

ELECTORAL vote of Rebel States, action respecting, 319-see "Reconstruction."

- ELIOT, Thomas D., Representative in Thirty-Sixth Congress, 48; Thirty-Seventh, 122; Thirty-Eighth, 140; bill to repeal fugitive slave law in Thirty-Third Congress, 233; joint resolution on the objects of the war, 287; bill to establish bureau of Freedmen's affairs, 260; confiscation and emancipation bills, 198-202.
- ELEINS, Joseph, arrest of, 153.
- ELLET, Henry J., declined as Rebel Post Master General, 400.

ELLIOTT, John M., Representative in First Rebel Congress, LLEOTI, John M., Representative in First Rebel Congress, 401; Second, 402.
 ELLIOTT, Robert, Secretary Seward's letter on arrest of, 154.
 ELLIS, E. W. H., member of Peace Conference, 68.
 ELLIS, John W., Governor of North Carolina, reply to Presi-dent's call, 114.

ELMORE, J. A., Commissioner from Alabama to South Caro-

lina, 11. , Alfred, Representative in Thirty-Sixth Congress, 48;

- ELV, Alfred, Represent Thirty-Seventh, 122.
- EMANCIPATION, to follow secession, prophecy of, by Alexan-
- EXIANCIPATION, to follow secession, prophecy of, by Alexander H. Stephens, 25.
   EMANCIPATION bill, reported, 199; amendment of F. P. Blair, Jr. 199, 200: substitute of C. B. Sedgwick, 200; substitute of P. P. Walton, 200; substitute of J. S. Morrill, 200, 201; vote on original, 201: vote reconsidered and bill recommitted with instructions offered by Albert G. Porter, 201; another bill reported, 201; substitute of Z. A. Bingham, 201; substitute of E. P. Walton, 202 ton, 202.

- in Thirty-Sixth Congress respecting, 209; President's Message, Match 6, 1862, 209; votes in House and Senate on resolution proposing, to border slave States, 209, 210; interview between the President and some berder slave States Congressmen, 210, 211; in District of Co-lumbia, 211-213; colonization clause of said bill repealed, with report on, note, 212; President's Message, and amendatory bill, 213; Select Committee on compensated emancipation, 213; President's appeal to the border slave States Congressmen, with their replies, 213–220; extracts from the President's second annual Message upon, 220-224; the Missouri bill, 224-226; the Maryland bill, and subsequent action by the people of Maryland, 226, 227.
- Maryland, 226, 227.
  EMAXCHATION, proclamations of the President, September 22, 1862, 227; January 1, 1863, 228: votes thereon in House, 229; in Senate, 318; Daniel Clark's, James F. Wilson's, and Isaac N. Arnold's bills respecting, 229; resolutions respecting, 230, 231; visit of the Chicago deputation to the President respecting its issue, 231, 232; Mr. Boutwell's statement in House respecting, 232; address of loyal Governors, 232; letters from Charles Sunner and Owen Lovejoy, 233; the President's modification of General Fremont's proclamation, and correspondence respecting, 245-247; rescinding of General Hunter's, 250: Alexander H. Stephens's prediction of, 25. of, 25.
- ot, 20, ENGLISH, James E., Representative in Thirty-Seventh Con-gress, 122; Thirty-Eighth, 140. ENGLISH, T. R., remarks in South Carolina Convention, 17; vote on secession, 398. ENGLISH, William H., Representative in Thirty-Sixth Con-

- ENGLISH, William H., Representative in Thirty-Sixth Congress, 49: adjustment proposition of, 54.
   ENLISTMENTS, alleged foreign, 343.
   Enquirer, Cincinnati, excluded from Kentucky, 192.
   Enquirer, Richmond, denunciation of Convention bill, 5 on full of Sumter, 114: on terms of peace, 305.
   ENROLLMENT act of March 3, 1863, 115; of February 24, and July 4, 1864, 116, 117; colored persons liable to, 116; certain aliens liable to, 116; all votes upon passage of, 261-270; proposition to enroll persons between forty. 261-270; proposition to enroll persons between forty-five and fifty, 268; reports of Secretary of War and Provost Marshal General respecting operation of, note, 263; judicial decisions upon, 272-274.
- ENROLLMENT of free negroes in rebeldom, 281, 282.

ENEOLLMENT, rebel, 118, 119. (See "Conscription.") ESCAPE of fugitive slaves, 1850 to 1860, 237.

- Excerpt of lighte states, 1550 to 100, 251. ETHERIDGE, Emerson, Representative in Thirty-Sixth Con-gress, 49; adjustment proposition of, 55. Evans, Lenuel D., understood to be "Eaton," of the Na-tional Intelligencer in January, 1861, revealing the proceedings of the disunion caucus, 391.
- Ewine, George W., Representative in First Rebel Congress, 401; Second, 402.

Ewing, Thomas, member of Peace Conference, 68.

Examiner, Richmond, on fall of Sunter, 114; on Wright's peace proposition, 305; on the North, 330; on Niagara Falls "peace conference," 399; on Union Baltimoro nominations, 399.

F

FALL, George R., commissioner from Mississipi to Arkansas, 11.

- FARNSWORTH, John F., Representative in Thirty-Sixth Con-gress, 49; in Thirty-Eighth, 140; resolution concerning
- unemployed generals, 255. FARNOW, James, Representative in First Rebel Congress, 402; Second, 402.
- FEARN, Thomas, deputy in rebel Provisional Congress, 11, 400; retired, 400.
- 400; retired, 400.
   FEATHERSTON, William S., commissioner from Mississippi to Kentucky, 11.
   FENTON, Reuben E., Representative in Thirty-Sixth Con-gress, 48; in Thirty-Seventh, 122; in Thirty-Eighth, 140; proposition of adjustment, 56; substitute of, for tax section of national currency bill, 364; Senate sub-ritigate for lost 2001.
- stitute for last, 364. FERRY, Orris S., Representative in Thirty-Sixth Congress. 48; member of Committee of Thirty-three, 53; report of. 58.
- FESSENDEN, Samuel C., Representative in Thirty-Sevenila Congress, 122; resolution of, approving the emancipa-tion proclamation and vote upon, 229. FESSENDEN, Thomas A. D., Representative in Thirty-Seventh
- FESSENDEX, Infolmas A. D., Representative in Thirty-Sectorial Congress, 123.
   FESSENDEX, William Pitt, Senator in Thirty-Eighth, 140; member of Peace Conference, 67; amendment to bill to anthorize the sale of arms to States, 36; Secretary of the sale of arms to States, 36; Secretary of the Treasury, 108.
- FIELD, David Dudley, member of Peace Conference, 67; proposition in. ou secession, 69. FIELD, Richard S., Senator in Thirty-Seventh Congress, 123; superseded, 123.

FIELDS, A. P., claimant of a seat in Thirty-Eighth Congress, | 141.

FINANCES, our, 356-368.

FINANCES, rebel, 368-373. FINANCES, William E., Representative in Thirty-Eighth Con-

- gress, 140; resolution on the war 259. FINEGAN, Joseph, letter of D. L. Yulee to, 392; vote on secession ordinance, 399.
- Pismack, William M., claimant of a seat in Senate, Thirty-Eighth Congress, 141; President's letter to, 322. Fismak, George P., Representative in Thirty-Seventh Con-gress, 122; on Select Committee on compensated eman-
- cipation, 213; views on, 217, 218. FISHING bounties, discussed, 21; votes on abolishing, 375, 376; prohibited in "Confederate" Constitution, 98, 100.
- Frich, Graham N., Scnator in Thirty-Sixth Congress, 48.
   Frizzarrance, Benjamin, Senator in Thirty-Sixth Congress, 48; withdrew, 4; intermediary between Col. Hayne and President Buchanan, 32.
- FLAG of the United States, ordered to be suppressed in Bal-
- timore, 304. FLANDERS, Benjamin F., Representative in Thirty-Seventh Congress, 123.
- Congress, 123. FLORENCE, Thomas B., Representative in Thirty-Sixth Con-gress, 45; adjustment proposition of, 56. FLORENA, vote for President in 1860, 1; members of Thirty-Sixth Congress, 48, 49; deputies in Rebel Provisional Congress, 11, 400; in First Congress, 401; Second, 402; seccession movement in, 3; seizures and surrenders in, 77. with a convention of the control of the 27; vote on secession ordinance, 399; contribution of men to the war, 399.
- FLOYD, John B., Secretary of War and resignation, 38; in-structions to Major Anderson, 31; removal of public arms, 34, 35; authorized to raise rebel troops, 120.
- arms, 04, 05; authorized to raise repet troops, 120. Foor, Solomon, Senator in Thirty-Sixth Congress, 48; in Thirty-Seventh, 122; in Thirty-Eighth, 140. Foorn, Henry S., Representative in First Robel Congress, 402; Second, 402; peace propositions, 303, 304; views on retaliation, 304.
- FORBES, John M., member of Peace Conference, 67. FOREIGN cnlistments, alleged, 343-345.
- Youndst timisturis, integrated, offored and declined, 345– 345; resolutions of Congress upon, 347; letter of Lord Lyons to Earl Russell stating views of New York dem-ocrats or conservatives upon, 347, 348.
- Octats of conservatives apon, or., e.e., Foreign relations, our, 235-255. Forsynt, John, robel commissioner to Washington, 108.
- FORSYRI, John, recei commissioner to Washington, 108.
  FORT Sumten, surrender dered, 27; surrender demanded by Governor Pickens, and correspondence, 32; Mobile Mercury on bombardment of, 112; Pryor's declaration, 112; Jeremiah Clemens's, 112; correspondence preceding bombardment between Anderson and Beauregard, 113, 114; Anderson's report, 114; rebel press on fall, 114.
- FORTRESS Monroe, Governor Letcher's designs upon, 28.

- FORTRESS Monroe, Governor Letchor's designs upon, 28.
  FORTRESS Monroe, Governor Letchor's designs upon, 28.
  FORTER, Lafayette S., Senator in Thirty-Sight Congress, 48; Thirty-Seventh, 122; Thirty-Eighth, 140.
  FOSTER, Stephen C., Representative in Thirty-Sixth Congress, 43; member of Peace Conference, 67.
  FORTER, Thomas J., Representative in Thirty-Sixth Congress, 49; Thirty-Seventh, 122.
  FOWLER, Asa, member of Peace Conference, 67.
  FRAKCHOT, Richard, Representative in Thirty-Sixth Congress, 49; Thirty-Seventh, 122.
  FOWLER, Asa, member of Peace Conference, 67.
  FRAKCHOT, Richard, Representative in Thirty-Sixth Congress, 43; Thirty-Seventh, 122.
  FEANCHS, John T., receiver under rebel sequestration, 204.
  FRANKLA, Augustus, Representative in Thirty-Sixth Congress, 48; Thirty-Seventh, 122.
  FEANCHS, John T., receiver under rebel sequestration, 204.
  FRANKLA, Augustus, Representative in Shirty-Sixth Congress, 48; Thirty-Seventh, 122.
  FEANCH, J. S., request for special session of Maryland Legislature in 1861, 8.
  FEANKLN, Thomas E., member of Peace Conference, 68;

- gislature in 1507, 8.
   FRANKLIN, Thomas E., member of Peace Conference, 68; proposition in, on secession, 69.
   FREEDMEN'S AFFAIRS, bureau of, 259, 260.
   FREEMAN, Thomas W., deputy in Rebel Provisional Congress, 400; member of First Congress, 402; Second, 402.
- gress, so; member of First Congress, 402; Second, 402.
   Freeman's Journal, presented by grand jury and excluded from the mails, 188; report upon, 188-191.
   FREE NEGROES, Mr. Rives's desire to enslave, 282; legisla-tion upon, in rebeldom, 282.
   FRELINGHUYSEN, Frederick T., member of Peace Confer-ence of the second - ence, 67. FREMONT, John C., Major General, emancipation proclama-tion and special military order of, and correspondence
- concerning, 245-247; nominated for President, 412; ac-ceptance, 413. FERNCH, Exra B., Representative in Thirty-Sixth Congress, 48; member of Peace Conference, 67. the Administra
- FRENCH in Mexico, action of Congress and the Administration, 348-354. FRENCH mediation, proffered and declined, 345, 346.
- Firs, James B., Provost Marshal General, order of, respecting elections, 309; report on commutation clause, note, 263.

- FUGITIVE slave laws, conflicting views on, in South Carolina Convention, 18; amendment reported by House Con-mittee of Thirty-Three, 61; propositions respecting in Senate Committee of Thirteen, 70; votes on acts of 1793 and 1850, 234; the several bills to amend or repeal, and resolutions relating thereto, 23–230 proceedings and roots, in Senate and House, on the repealing bill, 236, 237; completed vote of the House thereon, 388; Attorney General Bates's opinion respecting enforcement of, note, 235.
- FUGITIVE slaves, orders concerning, in District of Columbia, 245; General Halleck's order No. 3, explanation of, and vote in Congress concerning, 248, and note, 253, 254; Halleck's proclamation excluding them from camps, 250; General Buell's letter, offering to return, 250; Solo General Duch's letter, onering to return, 200; General Hocker's order directing return and report of facts respecting, 250; General Williams's order to turn out all, and arrest of Col. E. P. Paine for refusing to obey, 251; Lieut. Colonel D. R. Anthony's order concerning, and arrest, 251; the President's order to Gen-eral Schofield, respecting, 252, 253. FUGITVES from justice, bill respecting, 61. FULLER, Thomas C., Representative in Second Robel Con-
- gress, 402.

FUNDING act, Rebel, 368-370.

FUNSTEN, David, Representative in First Rebel Congress, 402; Second, 402.

G

- GAITHER, Burgess S., Representative in First Rebel Con-gress, 402; Second, 402.
  GANSON, John, Representative in Thirty-Eighth Congress,
- 140.
- GARDENHIRE, E. L., Representative in First Rebel Congress, 402.
- GABLAND, Augustus H., Representative in First Rebel Con-gress, 401; Second, 402. GABLAND, Rufus K., Representative in Second Rebel Con-
- gress, 402. GAEFIELD, James A., Representative in Thirty-Eighth Con-gress, 140; amendment to the enrollment bill, and vote, 269.
- GARLINGTON, Albert G., Secretary of Interior of the Repub-
- Gambardon, Ander G., cooledy of a second of the second lie of South Carolina, 2. GanxErr, Muscoe R. H., Representative in Thirty-Sixth Congress, 40; address to the people of Virginia, 40; re-marks upon Peace Conference proposition, 70; mem-

- marks upon Peace Conference proposition, 70; mem-ber of Rebel Congress, 402; death of, 402. GARRET, John W., commissioner from Alabama to North Carolina, 11; received by Convention, 5. GARRISO, Judge, of Brooklyn, proceedings of, 152. GARTBLL, Lucius J., Representative in Thirty-Sixth Con-gress, 40; sent telegram South, 37; withdrew, 3; Re-presentative in First Rebel Congress, 401; Second, 402. GATOREL, William H., Police Commissioner, arrest of, and reason for, 152; minutes of Board, 393, 394. GENERALS unemployed, proceedings in Congress, concern-ing, 285; joint resolution proposed, to drop from the rolls of the army certain, 255; reply of Secretary of War to resolution of enquiry concerning, 285. GENERA, 402.
- gress, 402.
- Genera, vote of, for President, 1; members of Thirty-Sixth Congress, 48, 49; deputies in R-bet Provisional Con-gress, 11, 400; members of First Rebel Congress, 401; of Secoud, 402; secession movement in Convention and Legislature of, 3; ratified Contederate Constitution, 3; Inter-State Commissioners of, 11; Supreme Court of, confirm the constitutionality of the anti-substitute law, 120; peace measure and discussion of in Legislature of, 304; action of Legislature on reconstruction, 323; on habeas corpus bill, 399; men in the army, 399. GHOLON, T. S., Representative in Second Rebel Congress,
- 402.
- 402.
  402.
  GLEERT, Colonel, resolution respecting conduct of, in dispersing a convention in Kentucky, 181.
  GLES, Thomas T., receiver under rebel sequestration, 204.
  GLMER, Frank, commissioner from Alabama to Virginia, 11.
  GLMER, John A., Representative in Thirty-Sixth Congress, 49; member of Border States Committee, 73; Representative in Scool Rebel Congress, 402.
  GLMORE, James R., Jeff. Davis's declaration to, 307.
  GIST, William IL, Governor of South Carolina, message of, 2; vote on secession ordinance, 399.
  GLASS, R. H., postmaster at Lynchburg, Va., and New York *Tribune*, 192.
  GLENN, Luther J., commissioner from Georgia to Missouri,

- GLENN, Luther J., commissioner from Georgia to Missouri,
- 11; received by Missouri Convention, 9. Googan, William L., speech of, in Virginia Convention, 6;
- vote on secession ordinance, note, 7. GOLD, motion to pay soldiers in, or equivalent, 272; propo-sition to pay officers, soldiers, contractors for supplies,

public loans, in, note, 367.

GOIDSBOROUGH, William T., member of Peace Conference, 68.

- 68.
  Goorn, Daniel W., Ropresentative in Thirty-Sixth Congress, 48; in Thirty-Seventh, 122; in Thirty-Eighth, 140.
  Goorg, John, Jr., resolution of, in Virginia Convention, 6; vote on ordinance of secession, note, 7; Representative in First Rebel Congress, 402; in Second, 402.
  Goorner, John Z., member of Peace Conference, 67.
  Goorner, John Z., member of Peace Conference, 67.
- GOODWIN, John N., Representative in Thirty-Seventh Con-
- Gorbanova, Sohn R., Representative in Hinty Seventia Coargress, 122.
   Gorbon, J. H., arrest of, 152, 153.
   Gorbanova, of the northern States, responses of, to the President's call for troops, 114; of southern States, defant responses of, to same, 114, 115; address of loyal Governors, 232, 233.
- GEAHAM, James H., Representative in Thirty-Sixth Congress, 48.
- GRAHAM, Malcolm D., Representative in First Rebel Congress, 402.
- GRAHAM, William A., Senator in First Rebel Congress, 401; Second, 402
- GRANGER, Bradley F., Representative in Thirty Seventh Congress, 122
- "GRANGER, Gidcon, letter of, respecting colored mail carriers, note, 239. note, 239. GRANT, Ulysses S., appointed Lieutenant General, 116; Lu-
- cius Robinson suggested his nomination at Cleveland, 413.
- Peter W., Representative in First Rebel Congress, GRAY. 402.
- GREELET, Horace, correspondence of, and a Virginia post-master, respecting suppression of the Tribune, 192; President's letter to, 334; the Niagara Conference, 301-400 303.
- GREEN, James S., Senator in Thirty-Sixth Congress, 48; telegram to Missouri on first vote on Mr. Crittenden's adjustment proposition, 39.
  - GREEN, N. H., request of, for special session of the Legislature of Maryland, 8. GEEGG, John, deputy in rebel Provisional Congress, 11, 400.
  - GREGG, Maxcy, remarks of, in South Carolina Convention, 17; vote on secession ordinance, 389.

  - 17; vote on secession ordinance, 389.
    GRIDER, Henry, Representative in Thirty-Seventh Congress, 122; in Thirty-Eighth, 140; on compensated emancipation, 213-217.
    GRIDER, James W., Senator in Thirty-Sighth, 140; member of Peace Conference, 68; of Committee of Thirteen, 70; proposition of, for constitutional convention, and vote, 64; bill of, for relief of Robert Small, 229; amendments of, to concliment bill, and votes 204, 265; resolution of, relating to the war, 291.
    GRINNELL, Josiah B., Representative in Thirty-Eighth Congress, 140; resolutions approving the enlistment of colored persons, and vote, 271, touching General J. M. Tuti-lo's Natchez order, 253, on newspaper suppression, 194.
  - sion, 194
  - GRISWOLD, John A., Representative in Thirty-Sixth Congress, 140.

  - GROESBECK, William S., member of Peace Conference, 68. GROW, Galusha A., Representative in Thirty-Sixth Congress, 48; Representative in, and Speaker of House, Thirty-Seventh Congress, 122.
  - GURLEY, John A., Representative in Thirty-Sixth Congress, 49; in Thirty-Seventh, 122. GUERRILAS, our legislation upon, 271; bill reported in the
  - House to punish, 271; proceedings upon in Senate, 271.
  - GUERRILLAS, "rebel" legislation upon, 119; commission under, 119; Virginia resolutions respecting, 119. GUTHRIE, James, member of Peace Conference, 68.

GWIN, William M., Senator in Thirty-Sixth Congress, 48. GWIN, C. J. M., telegram respecting interview with Gen-eral Kenton Harper, 393.

Ħ

- HABEAS corpus, writ of, opinion of Chief Justice Taney, 155– 158; of Attorncy General Bates, 158–161; Horace Bin-ney, 161, 162; Theophilus Parsons, 162; suspension of, in certain cases, 177; general suspension, 177, 178; in Kentucky, 178; Gen. McClellan's order to Gen. Banks to suspend, if necessary, 179; power given to Military Governors, 179; resolutions concerning, 182, 183; act for suspension, and proceedings and votes upon, 183–185; reply of Reverdy Johnson to Chief Justice Taney, note, 155. note, 155.
- HABEAS corpus, rebel suspensions of, 121; legislation upon, 187, 188; State Legislatures upon, 399; North Carolina Supreme Court on, 120. HACKLEMAN, Pleasant A., member of Peace Conference, 68.

- in, 358; Mr. Chase's letter, respecting redemption of | HAHN, Michael, Representative in Thirty-Seventh Congress, 123; Free State Governor of Louisiana, 321. HAIGHT, Edward, Representative in Thirty-Seventh Con
  - gress, 122.
  - HALE, James T., Representative in Thirty-Sixth Congress, 48; Thirty-Seventh, 122; Thirty-Eighth, 140; member
  - Finity-Seventh, 123, Thirty-Eighth, 140, Johnov of Border States Committee, 73.
     HALE, John P., Senator in Thirty-Sixth Congress, 48; Thirty-Seventh, 121; Thirty-Eighth, 140; motion to amend bill to repeal fagitive slave laws, 237.
     HALE, Stephen F., Commissioner from Alabama to Ken-therker, 11, download, Brankieval, Congress, 11
  - tucky, 11; deputy in rebel Provisional Congress, 11, 400.

HALL, Ohapin, Representative in Thirty-Sixth Congress, 48, HALL, Hiland, member of Peace Conference, 67, HALL, Thomas W., arrest of, 153, HALL, Willard P., Brigadier General, order respecting elec-

- tions, 314.
- Lions, 544.
   Indat, William A., Representative in Thirty-Seventh Congress, 133; Thirty-Eighth, 140; at Border State interview with the President, 211; views on compensated emancipation, 213–217.
   HALLEEK, Henry Wager, Major General, order and proclaming functions claves from his lines. and
- mation excluding fugitive slaves from his lines, and explanation of the former, 248, 250.
- HAMILTON, Andrew J., Representative in Thirty-Sixth Con-gress, 49.
- HAMLIN, Hannibal, Senator in Thirty-Sixth Congress and resignation, 48; elected Vice President in November, 1860, and President of the Senate, Thirty-Seventh and Thirty-Eighth Congresses, 122, 140.
- Thirty-Eighth Congresses, 122, 140.
   IAMMOND, James H., Senatorin Thirty-Sixth Congress, 48; resignation, 2; extract from speech in 1858, 26.
   HANCHERT, Luther, Representative in Thirty-Seventh Congress, 122; death of, 123.
   HANDY, A. H., commissioner from Mississippi to Maryland, and the second - and answer of Governor Hicks, 8; speech at Baltimore, 8; appointment of, 11.
- 8; appointment of, 11.
  HANKINS, Isaac R., member of Peace Conference, 68.
  HIANLEY, Thomas B., Representative in First Rebel Congress, 401; Second, 402.
  HARDEMAN, Thomas, Jr., Representative in Thirty-Sixth Congress, 49; withdrew, 2.
  HARDING, Aaron, Representative in Thirty-Seventh Congress, 49; type Field Physics and Seventative Physics and Se
- COURTESS, 49, WILLING, J. HARDING, Aaron, Representative in Thirty-Seventh Con-gress, 122; Thirty-Eighth, 140; views on compensated emancipation, 213-217; resolution on State rights, 259; propositions to prohibit the payment of colored soldiers, 284; resolutions on the insurrectionary States, 327.
- 284; resolutions on the insurrectionary States, 327.
  HARDING, Charles B., commissioner from Legislature of Maryland to visit Jefferson Davis, and report of, 9.
  HARLAN, James, Senator in Thirty-Sixth Congress, 49; Thirty-Seventh, 122; Thirty-Eighth, 140; member of Peace Conference, 68.
  HARLEE, W. W., Postmaster General of the republic of South Carolina, 2: vote on secession ordinance, 389.
  HARRINGTON, Henry W., Representative in Thirty-Eighth Congress, 140; resolutions on habeas corpus, &c., 182.
  HARRINGT, Jalien, Representative in First Rebel Congress, 401; Second, 402.
  HARRIS, B. D., member of Peace Conference, 67.
  HARRIS, B. D., member of Peace Conference, 67.
  HARRIS, Benjamin G., Representative in Thirty-Eighth Con-ress, 140; votes on his expulsion, and his censure.

- gress, 140; votes on his expulsion, and his censure, 387.
- HARRIS, Charles M., Representative in Thirty-Eighth Comgross, 140. HARRIS, Ira, Senator in Thirty-Seventh Congress, 122; in
- Thirty-Eighth, 140. HARRIS, Isham G., Governor of Tennessee, proclamation of.

- HARRIS, ISIAM G., GOVERNOTO I Chressee, FOCLARIATOR O, 5; reply to President Lincoln's call for troops, 115.
  HARRIS, J. Morrison, Representative in Thirty-Sixth Con-gress, 49; member of Border States Committee, 73.
  HARRIS, John T., Representative in Thirty-Sixth Congress, 49; hopeful dispatch of, 39; member of Border States committee 73. committee, 73.
- HARRIS, Thomas A., deputy in rebel Provisional Congress, 400; member of First Congress, 402. HARRIS, William L., commissioner from Mississippi to Geor-
- gia, 11. HARRIS, Wiley P., deputy in rebel Provisional Congress, 11, 400.
- HARRISON, James T., deputy in rebel Provisional Congress, 11, 400.
- HARRISON, Richard A., Representative in Thirty-Seventh Congress, 122. HARRISON, William G., arrest of, 152; minutes of the Board
- of Police Commissioners, 393, 394. HARTSUFF, George L., Major General, order respecting im-
- pressments, 315.
- HARVE, Lewis E., resolution and report in Virginia Conven-tion. 6; vote on secession, note, 7; commissioner in making military league, 8.
- Maryland in 1861, 8.

- HASKIN, John B., Representative in Thirty-Sixth Congress, 48; adjustment proposition of, 53.
- HATCHER, R. A., Representative in Second Rebel Congress, 402.
- HATTON, Robert, Representative in Thirty-Sixth Congress, 49: member of Border States Committee, 73.
- HAWKINS, George S., Representative in Thirty-Sixth Con gress, 49; telegram South, 37; on House Committee of Thirty-three and declined, 53; declaration on secession, 52; withdrew, 3; Yule's allusion to, 392. Hay, Major John, at Niagara Falls conference, 301-303.
- HATNE, Isaac W., commissioner from South Carolina to Washington, 32; correspondence with President Buch-Mashington, 32; correspondence with President Juch-anan, 32-34; vote on secession ordinance, 389.
   **HAYXES**, Landon C., receiver under rebel sequestration, 204; Senator in First Rebel Congress, 401; Second, 402.
- HATTI, bill for recognition of, 239; President Lincoln's recommendation of, 130.
- HECKART, Franklin, arrest of, 152.
  HEISKELL, Joseph B., deputy in rebel Provisional Con-gress, 400; Representative in First Congress, 402; Secoud, 402.
- HELMICK, William, Representative in Thirty-Sixth Congress, 49.
- 49.
   HERPHIL, John, Senator in Thirty-Sixth Congress, 48; intermediary between Colonel Hayne and President Buchanan, 32; deputy in rebel Provisional Congress, 11, 400; death of, 401.
   HERDERSON, John B., Senator in Thirty-Seventh Congress, 122; in Thirty-Eighth, 140; reply to the President's appeal to the border States, 218-220; proposed amendments to the Congritition of Sterman Provided Amendments of the Congress, and the States, 218-220; proposed amendments to the Congrestion of the States, 218-220; proposed amendments of the Congress, and the States, and the States, and the
- 123; in Thirty-Eighth, 140; reply to the President's appeal to the border States, 218-220; proposed amend-ments to the Constitution, 255; amendments to bill to employ colored persons, 275
   HENDRICKS, Thomas A., Senator in Thirty-Eighth Congress, 140; notion on confiscation, 203; motion on enroll-ment, 264; amendment to bill to purish guerrillas, 271; muscher and then are fer allowed for the senator of the senator
- amendment on the pay of soldiers, 278. HERBERT, Clark C., Representative in First Rebel Congress,
- 402; Second, 402. HENEY, Gustavus A., Commissioner for Tenessee in leagning
- her with the Confederate States, 5; aid to General Pil-low, 401; Senator in First Rebel Congress, 401; Second, 402
- Herald, Shasta, upon a Pacific republic, 42. HERRICK, Auson, Representative in Thirty-Eighth Congress. 140
- HIGERBON, William, member of Peace Conference, 68. HIGERBAN, John, Representative in Thirty-Sixth Congress, 48; in Thirty-Seventh, 122; proposition on colored sol-diers, 370; Keitt's comment on Hickman's declaration against disunion, 390.
- against disunion, 390. HIGES, Thomas H., Governor of Maryland, declined to call special session of the Legislature, 8; reply of, to A. H. Handy, Commissioner from Mississippi, 8; address of, against secession, 9; letter of John B. Brooke and E. G. Kilbourn to, asking that the Legislature be con-vened, 9; letter of, to General Butler, reply of latter, and protest of former, 9; proposed member of Board of Public Safety, 9; reply of, to President's first call, 114; Secretary Seward's letter to, on foreign mediation, 345; Senator in Thirty-Seventh Congress, 123; Thirty-Bichth, 140. Eighth, 140. HIGBY, William, Representative in Thirty-Eighth Congress,
- 141.
- HIGGINSON, Colonel T. W., resolution concerning, 244. HILL, Joshua, Representative in Thirty-Sixth Congress, 49; resigned, 3.
- Hengkned, S.
   Hengker, S.
   In making the league with Tennessee, 5.
   Hut, Beujamin H., deputy in rebel Provisional Congress, 11,400; Senator in First Rebel Congress, 401; Second,
  - 402; views on sequestration 205. HILTON, Robert B., Representative in First Rebel Congress,
  - 401; Second, 402.
  - HINDMAN, Thomas C., Representative in Thirty-Sixth Con-gress, 49; sent telegram South, 37; adjustment propo-sition of, 54; remark upon Peace Conference proposition, 70. HINKS, Charles D., Police Commissioner, arrest of, and rea-
  - sons for, 152; minutes of board, 393, 394. HITCHCOCK, Reuben, member of Peace Conference, 68. HOARD, Charles B., Representative in Thirty-Sixth Con-

  - gress, 48.
- HODGE, B. L., Representative in Second Rebel Congress, 402.

  - HODGES, Col. A. G., President Lincoln's letter to 336.
    HODGES, George B., member of bogus legislative council of Kentucky, 8; Representative in First Rebel Congress, 401.

  - HOGAN, John, arrest of, 153. HOGEN, Mr., elected adjutant general of North Carolina, 5. HOLCOMBE, James P., vote on secession ordinance in Vir-ginia Convention, node, 7; commissioner of Virginia in making a league with the Confederate States, 8; Rep-

- resentative in First Rebel Congress, 402; Niagara Falls correspondence, 301-303. DEN, William W., card of, to the people of North Car-
- HOLDEN, William

- HotDex, William W., card of, to the people of North Carolina, note, 304.
   HotDers, William D., Representative in First Rebel Congress, 402; Second, 402.
   HOLLADAY, F. W. M., Representative in Second Rebel Congress, 402.
   HOLLADAY, F. W. M., Representative in Thirty-Sixth Congress, 402.
   HOLMAN, William S., Representative in Thirty-Fighth, 140; propositions of adjustment, 54, 56; propositions of adjustment, 54, 56; propositions of acquisetment for the curolinear bill, 261; amendment for the arm superprincipal unit 275: amendment mutation, 50', substruct for the enrolment mit, 20'; amendment to the army appropriation bill, 25'; amend-ment to homestead bill, 25'; resolution respecting unemployed army officers, 25'; resolution on maintain-ing the integrity of the Union, 25'; resolution affirming Ing the integrity of the china, 5.5; resolution animing the resolution of J. J. Crittenden, adopted by the House July 22, 1861, 287; resolutions respecting the prosecution of the war, 257, 288, 259, 292; amendments to the loan bill, respecting State taxation, 559, 500; pro-
- to the total bill, respecting state taxation, 559, 560; pro-posed increase of pay of soldiers, 271. HOLT, Joseph, appointed Secretary of War, and resigned as Postmaster General, 28; letter to Fitzpatrick, Mallory, and Sildell, in regard to Fort Sumter, 32; report on the subject of quartering troops in the District of Columbia. 81-82; opinion on newspaper exclusion from the mails, 190.
- HOLT, Hines, member of First Rebel Congress, 401; peace proposition, 303.
   HOMESTEADS for soldiers, proceedings and votes respecting, and votes respecting, proceedings.
- 284, 285.
- HOCKER, C. E., commissioner from Mississippi to Sonth Carolina, 11.
- HOOKER, Joseph, General, order of, directing fugitive slaves to be returned to their masters, 250
- Hooper, Samuel, Representatives in Thirty-Seventh Con-gress, 123; Thirty-Eighth, 140.
  Hopkins, A. F., commissioner from Alabama to Maryland,
- 11.
- HOPKINS, Judge, Richmond, John S. Preston's telegram to. 40,
- HOPPIN, William W., member of Peace Conference, 67. HORTON, Valentine B., member of Peace Conference, 68; Representative in Thirty-Seventh Congress. 122.
- Horcu Kiss, Giles W., Representative in Thirty-Eighth Congress, 140.
- gress, 140. HOUGH, Harrison, member of Peace Conference, 68. HOUSE, Mr., deputy in Rebel Provisional Congress, 400. HOUSER, Edward, arrest of, 153. HOUSEN, George S., Representative in Thirty-Sixth Con-
- gress, 49; withdrew, 4; member of Committee of Thir-ty-three, 53. HOUSTON, John W., member of Peace Conference, 68.
- HOUSTON, Sam, Governor of Texas, deposed as, 4; protest of,
- HOWARD, Benjamin C., member of Peace Conference, 68.
- Howard, Charles, Police Commercial Feace Conference, 68.
  Howard, Charles, Police Commissioner, arrest of, and reason for, 152; application to Congress and result, 179; minutes of the Board, 393,394.
  Howard, Confederacy, 9, in 1861, resolution of, for Southern Confederacy, 9, 160ward, Jacob M., Senator in Thirty-Seventh Congress, 123;

- in Thirty-Eighth, 140; amendment to national curren-cy act. 364.
- HOWARD, William, Representative in Thirty-Sixth Congress, 49
- HOWARD, William A., Representative in Thirty-Sixth Congress, 49; resolution of, for Select Committee on Mill-tary and Naval Affairs, chairman of, 80; report of, 85;
- reports of committee, 80-85. Howe, Timothy O., Senator in Thirty-Seventh Congress, 122; in Thirty-Eighth, 140; bill to repeal the fugitive slave law, 235.
- HUBBARD, Asahel W., Representative in Thirty-Seventh
- Congress, 140. HUBBARD, John H., Representative in Thirty-Eighth Con-gress, 140; resolution of, to instruct the Judiciary Committee to report bill to repeal fugitive slave law
- HUGHES, George W., Representative in Thirty-Sixth Congress, 49. HULBURD, Calvin T., Representative in Thirty-Sixth Con-
- gress, 140.
- HUMPHREY, James, Representative in Thirty-Sixth Con-gress, 48; member of Committee of Thirty-three, 56; recommendations of New York City members to, 74. HUXT, Washington, letter of George N. Sanders to, 330.
- HUNT, Washington, letter of George A. Sanders of Sol. HUNTER, Lavid, Major General, emancipation proclamation of, 250; President's revocation, 251.
  HUNTER, Robert M. T., Senator in Thirty-Sixth Congress, 48; address to people of Virginia, 40; member of Com-mittee of Thirteen, 70; extract from speech at Char-lottesville, 26; plan of adjustment of, 56-88; commis-missioner of "Confederate" government in making

military league with Missouri, 11; colloquy with Sen-ator Baker, 88; deputy in rebel Provisional Congress, 400; Secretary of State in Provisional administration,

- 40); Seator in First Congress, 401; in Second, 402. HUNTER, William W., United States Navy, denied the use of a steamtug in Baltimore waters, 393.
- HURLBUT, S. A., Major General, order of, respecting elections
- in Kentucky, 313. HUTCHINS, John, Representative in Thirty-Sixth Congress, 49; in Thirty-Seventh, 122; adjustment proposition of, 55.
- HUTCHINS, Wells A., Representative in Thirty-Eighth Congress, 140.

HYEE, Samuel, deputy in rebel Provisional Congress, 400. Т

- ILLINOIS, vote of, for President, 1; members of Peace Con-ference, 68; of Thirty-Sixth Congress, 48, 49; of Thirty-Seventh, 122; of Phirty-Eighth, 140; no "personal liberty" law, 47; proposed negro suffrage, and residence, in, note, 241. Imports, Mr. Bingham's bill to provide for collection of du-
- ties on, 78; Mr. John Cochrane's, with his views thereon, 79, 80,
- INAUGURAL Address, Abraham Lincoln's, 105-108. Income tax, special war, 357.
- INCOMPABILITY of civil and military office, 375.
- INDEMNITY act, passage of, 183-185; protest against by thirty-seven Representatives, 185; sustained by New York Supreme Court, 185, 186 ; like acts by Legislature of Maryland, 396.
- INDEPENDENCE, rebel demand of, and terms, 303, 307; 329-332.
- INDEX to the Constitutions of the United States, and the
- "Confederate" States, 100-103. INDEXA, vote of, for President in 1860, 1; members of Peace Conference, 63; Thirty-Sixth Congress, 43, 49; Thirty-Seventh, 122; Thirty-Eighth, 140; no" personal
- liberty'' law, 47. INDIANS, proposed enlistment of, votes in Senate on, 264.
- IMMANS, proposed entrainent of, votes in Senate on, 204.
   INGERSOL, Ebon C., Representative in Thirty-Eighth Congress, 141; amendment to enrollment act, 209.
   INGLIS, John A., remarks of, in South Carolina Convention, 16, 17; vote on secession ordinance, 300.
   INSELF, A. H., report of, in Baltimere Convention, 405.
   INSERF. 2016.

- INSTRUCTIONS, general, of the War Office, 247, 243. INSURRECTION, boundaries of, declared, 150; bill to suppress, 77, 78; bill to suppress, and for other purposes, 196, 197.
- INSURPECTIONARY States, enlistments in, 265, 266, 269; elec-toral vete of, 318-320; resolutions concerning, 322-329; vote of representatives in Baltimore Convention, 405.
- INTERCOURSE, commercial, with insurrectionary proclamations respecting, 149, 150. States,
- INTEREST, provision respecting, in national currency act, 363.
- INTERNAL revenue, act of 1862, proceedings on passage, 361; act of 1864, 362.

- INTER-STATE commissioners, names of, 11. Iowa, vote of, for President, 1; members of Peace Confer-ence. 68; of Thirty-Sixth Congress, 48, 49; of Thirty-Secuth, 122; of Thirty-Eighth, 140; personal liberty law, has no, 46.
- IRVINE, William, Representative in Thirty-Sixth Congress, 48
- IVERSON, Alfred, Senator in Thirty-Sixth Congress, 48; withdrew, 3; intermediary between Col. Hayne and President Buchanan, 32; sent telegram South, 37.

.3

- JACKS, T. M., claimed seat as Representative from Arkansas, 141.
- JACKSON, Andrew, letter on nullification, secession, and disunion, 589.
- JACKSON, Claiborne F., Governor of Missouri, reply of, to President's first call for troops, 115. JACKSON, James, Representative in Thirty-Sixth Congress,
- withdrew, S.
   yuthdrew, S.
   JAOKSON, James S., Representative in Thirty-Seventh Congress, 122: death of, 123; on compensated emancipation, 213-217.
- JAMES, Anaziah B., member of Peace Conference, 67. JAMESA, David B., Sceretary of War of the republic of South Carolina, 2: despatches of throughout the South, 39; vote on secession, 599.
- JEMISON, Robert, Senator in First Rebel Congress, 401; Second, 402. JEXCKES, Thomas A., Representative in Thirty-Eighth Con-
- gress, 140. JENKINS, Albert G., Representative in Thirty-Sixth Con-
- gress, 497; address to the people of Virginia, 40; ad-justment proposition of, 55; member of First Rebel Congress, and resignation, 402; death of, 402.

- JOHNSON, Andrew, Senator in Thirty-Sixth Congress, 48; in Thirty-Seventh, 122; remarks on defeat of the Crittenden compromise, 66; adjustment proposition of, 73; resolution respecting the cause and objects of the war, 286; nominated for Vice President, 407; acceptance, 409

- 409. JOHNSON, Bradley T., Marshal Kane's telegram to, and re-ply, note, 302, 393. JOHNSON, George W., Governor of Kentucky, 8. JOHNSON, Herschel V., Senator in First Rebel Congress, 401; Second, 402; views on sequestration, 205. JOHNSON, James M., claimed seat as Representative from Arkanse 111
- Arkansas, 141.
- JOHNSON, Jilson R., contestant for seat in First Rebel Con-
- gross, Jison K., contestant for sect in First field Con-gross, 402.
   Jomssox, Lieut. Col., order respecting election in Kentucky, 313; onth preseribed by, 313.
   Jomssox, Philip, Representative in Thirty-Seventh Con-gross, 122; in Thirty-Eighth, 140; resolution on en-rollment, 270.
- Jonnson, Reverdy, letter of John C. Le Grand to, 9; mem-mer of Peace Conference, 68; Senator in Thirty-Eighth Congress, 140; report against the passage of joint res-olution amending confiscation act, 203; amendment to fugitive slave law, 237; resolution on the Arguelles case, and the President's reply, 354, 355; reply to Chief
- case, and the President's reply, 504, 555; reply to Chief Justice Taney on *habcas corpus*, *note*, 155.
   JoHNSON, Robert, deputy in Rebel Provisional Congress, 400; member of First Congress, 402; Second, 402.
   JOHNSON, Robert W., Senator in Thirty-Sixth Congress, 43; deputy in rebel Provisional Congress, 400; Senator in First Congress 401; Second 402; viewa on sequestra. First Congress, 401; Second, 402; views on sequestra-
- First Congress, 401; Second, 402; Views on sequestra-tion, 205.
  JoHNSON, Waldo P., member of Peace Conference, 68; Sen-ator in Thirty-Seventh Congress, 122; expulsion of, 123; Senator in Second Rebel Congress, 402.
- Johnson, William Cost, purpose in attending Southern caucus in 1835, 390. Johnson, William, Representative in Thirty-Eighth Con-
- 30 Sols, William F., temporary chairman of the Cleve-land Convention, 411.
- Jones, George W., member of Peace Conference, 68; deputy in rebel Provisional Congress, 400; Representative in
- First Congress, 402. JONES, Henry C., deputy in Rebel Provisional Congress, 400.
- JONES, John J., Representative in Thirty-Sixth Congress, 49; sont telegram South, 37; withdrew, 3. JONES, T. Laurence, arrest of, 153; votes in Legislaturo of Maryland, 396-398.
- Journal of Commerce, New York, presented by Grand Jury, 188; excluded from the mails, 188; report of Post Master General Blair respecting, 188-192; seizure of office of by the military authorities, and subsequent proceedings, 192-194.

- proceedings, 192-194.
   JUDICIAL rulings on indemnity act, 185, 186; enrollment, 272-274; confiscation, 206, 207.
   JUDICIAL rulings (rebel) upon conscription, 120; on antisubstitute law, 120; habeas corpus, 130; on impressment act, note, 372.
   JULLIAN, George W., Representative in Thirty-Seventh Congress, 122; in Thirty-Eighth, 140; resolutions for amendment or repeal of fugitive slave laws, 235, 236; homesteads for soldiers, 284.
   JUNER, Scholmer, B. Representative in Thirty-Sixth Congress, 122; in Thirty-Eighth, 140; resolutions for amendment or repeal of fugitive slave laws, 235, 236; homesteads for soldiers, 284.
- JUNKIN, Benjamin F., Representative in Thirty-Sixth Congress, 48
- JUSTICE, fugitives from, bill respecting, 61.

## к

- KALBFLEISCH, Martin, Representative in Thirty-Eighth Congress, 140.
- gress, 140.
   KANE, George P., marshal of police of Baltimore, letter of, respecting safety of transit through or sojourn in Baltimore of President Lincoln, 9; arrest of, by General Banks, 152; telegram to Bradley T. Johnson, note, 393.
   destroyed railroad bridges, 393.
   KANE, Robert P., secession resolution of, at Philadelphia meeting, 390.
- KANSAS, Representative in Thirty-Sixth Congress, 49; members of Thirty-Seventh Congress, 122; of Thirty-Eighth, 141.
- KASSON, John A., Representative in Thirty-Eighth Congress, 140.
- KEEBLE, E. A., Representative in Second Robel Congress, 402.
- KEIN, William H., General, agency in the arrest of John
- KEIM, Wilham H., General, agoncy in the artest read-Merryman, 155.KEITT, Lawrence M., Representative in Thirty-Sixth Con-gress, 40; withdrew, 1; remarks of, in South Carolina Convention, 17; statement of, respecting forts in Charles-ton harbor. 30; Toombs's telegram to, 37; vote on se-cession ordinance, 390; deputy in rebel Provisional.

Congress, 2, 11, 400; death of, 401; expectations of northern help in rebellion, 390.

- northern help in rebellion, 300. KELLEY, William D., Representative in Thirty-Seventh Con-gress, 122; in Thirty-Eighth, 140. KELLOGG, Francis W., Representative in Thirty-Sixth Con-gress, 49; in Thirty-Seventh, 122; in Thirty-Eighth Con-gress, 140. KELLOGG, William, Representative in Thirty-Eighth Con-gress, 140. KELLOGG, William, Representative in Thirty-Sixth Congress, 49; in Thirty-Seventh, 122; adjustment propositions of, 56, 62, 63; member of Committee of Thirty-Three, 53. 53
- KENAN, Augustus H., dcputy in rebel Provisional Congress, 11,400; Representative in First Congress, 401; motion to table peace proposition, 306. KENAN, Owen R., Representative in First Rebel Congress,
- 402

- 402.
  KENDALL, Amos, opinion of, respecting suppression of mail matter, 189-191; Keitt's opinion of, 390.
  KENNER, Duncan F., deputy in Provisional Congress, 11, 400; Representative in First Congress, 401; Second, 402.
  KENNEDT, Anthony, Senator in Thirty-Sixth Congress, 45; in Thirty-Seventh, 122; letter of, respecting the secession of Maryland, 9.
  KENTUCKY, vote of, for President, 1; members of Peace Conference, 67; of Thirty-Eighth, 140; proceedings in Legislature and borgs Convention of, in respect to second. Legislature and bogus Convention of, in respect to se-cession, 8; proceedings in Congress respecting arrest of citizens of, 180; respecting dispersion of Convention in, 181; act of Legislature of, and veto of Governor in, 181; act of Legislature of, and veto of Governor Magoffin, 312, 313; military orders respecting elections in, 313-314; orders respecting impressment of proper-ty, 315; adjustment proposition of a "distinguished citizen" of, 74; Judge Nicholas's proposed amend-ment to the Constitution, 255, 256; men in the rebel army, 399; rebel congressional districts, 401.
   KENYON, William S., Representative in Thirty-Sixth Con-gress, 48.
   KERNAN, Francis, Representative in Thirty-Eighth Con-cress 140; substitute reletive to uncomployed concrels
- gress, 140; substitute relative to unemployed generals, 285; amendment to Ioan bill, 360.
- KERRIGAN, James E., Representative in Thirty-Seventli Con-gress, 122. KERSHAW, J. B., commissioner from South Carolina to Tex-
- as, 11; vote on secession ordinance, 399.

- KESSLER, Andrew, arrest of, 153. KIDD, Milton, arrest of, 153. KIDD, Milton, arrest of, 153. KIBOURN, E. G., letter of, to Governor Hicks, asking that the Legislature of Maryland be convened, 9; at Howard conty meeting, 9. RELINGER, John W., Representative in Thirty-Sixth Con-gress, 48; in Thirty-Sventh, 122. KILGORE, David, Representative in Thirty-Sixth Congress,
- 49; adjustment proposition of, 54. KILGORE, N., arrest of, 153.
- KING, Austin A., Representative in Thirty-Eighth Congress, 140.
- 140. KING, John A., member of Pence Conference, 67. KING, Preston, Senator in Thirty-Sixth Congress, 43; in Thirty-Seventh, 122; resolution respecting trial of per-sons arrested, 179; a mendment to confiscation bill, 202; amendment to the "legal tender," 358; in Baltimore Convention, 405, 406. KINNEY, John F., Delegate from Utah, Thirty-Eighth Con-gress, resolution on the war, 294. KIENKON Sanuel J. Governor of Lows, signer of Altoons
- KIRKWOOD, Samuel J., Governor of Iowa, signer of Altoona
- Minkwood, Camber S., Governor of Iowa, sight of Alcona address, 233.
   Krotesn, B. M., claimant for seat as Representative in Thirty-Eighth Congress, 141.
   KNAPP, Anthony L., Representative in Thirty-Seventh Con-gress, 123; in Thirty-Eighth, 140.
   KNox, Samuel, Representative in Thirty-Eighth Congress, 141.
- KUNKEL, Jacob M., Representative in Thirty-Sixth Congress, 49.

L

- LAMAR, G. B., purchase of Government arms by, in 1860,
- Lamar, Lucius Q. C., Representative in Thirty-Sixth Con-gress, 49; withdrew, 3; Benjamin's intercepted dis-patch to, relative to African slave trade, 151. LaNDER, William, Representative in First Rebel Congress, 402.

- 402.
  LANDEUM, John M., Representative in Thirty-Sixth Congress, 49; telegram sent South, 37; withdrew, 4.
  LANDEW, Mr., arrest of, 152.
  LANE, Henry S., Senator in Thirty-Seventh Congress, 122; in Thirty-Eighth, 140.
  LANE, Henry S., Senator in Thirty-Seventh Congress, 122; in Thirty-Eighth, 140; amendment to the bill to employ colored persons, 275; amendment to the engineer bill, 276; joint resolutions respecting free State of Ar-

kansas, 320; amendment to bill to admit West Virginia, 378.

- Joseph, Senator in Thirty-Sixth Congress, 48; a/Just-ment proposition of, 72.
   LANSING, William E., Representative in Thirty-Screnth Congress, 122; substitute of, for Mr. Lovejoy's resolution respecting General Halleck's order No. 3, and vote, 954
- LARRABEE, Charles H., Representative in Thirty-Sixth Con-
- gress, 49; adjustment proposition of, 54. LATHAM, Milton S., Senator in Thirty-Sixth Congress, 49; in Thirty-Seventh, 122; declaration for the Union, 41; remarks upon the defcat of the Crittenden proposition, of adjustment, 67; upon Peace Conference proposition, 70.
- LAW, John, Representative in Thirty-Seventh Congress, 122;
- in Thirty-Eighth, 140. LAWS, James M. Mason's resolution to suspend certain, in the secoded States, 86.
- LAZERA, Jesses, Representative in Thirty-Seventh Congret, 122; in Thirty-Eighth, 140; resolution of, for sue-pending hostilities, 300. LEACH, De Witt C., Representative in Thirty-Sixth Congress,
- 49.
- LEACH, James M., Representative in Thirty-Sixth Congress, 49; Representative in Second Rebel Congress, 402. LEACH, J. T., Representative in Second Rebel Congress, 402;
- peace proposition, 306.
- LEAGUE between Tennessee and insurrectionary States, 5; Virginia and same, 8; Missouri and same, 11. LEAKE, Shelton F., Representative in Thirty-Sixth Congress,

- LEAKE, Shelton F., Representative in Thirty-Sixth Congress, 49; address to people of Virginia, 40; adjustment prop-osition, 55; remark on that of Peace Conference, 70.
  LEAKT, Cornelius L. L., Representative in Thirty-Seventh Congress, 122; on select committee on compensatul emancipation, 213; views on, 213-217.
  LEAYTH, Humphrey H., refusal to issue writ of habeas cor-pus in Vallandigham case, 162; latter's opinion of, 176.
  LE BLOND, Francis C., Representative in Thirty-Eighth Con-gress, 140; amendment to enrollment, proposing an armistice, 300.
  LE GRAND, John C., letter on secession, 9.
- LE GRAND, John C., letter on secession, 9.
- LEE, M. Lindley, Representative in Thirty-Sixth Congress, 43. LEE, Richard Henry, reported disunion sentiment of, 389. LEE, Stephen D., aid to Beauregard, 113.
- LEGAL tenders, proceedings creating, 357-359.
- LEHMAN, William E., Representative in Thirty-Seventh Con-
- gress, 122; on select committee on compensated eman-cipation, 213.
- LEONARD, Justice, opinion on indemnity act, 185, 186. LESTER, George N., Representative in Second Rebel Con-gress, 402.
- LETCHER, John, Governor of Virginia, on resolutions of New York, 5; of Ohio, 6; purchase of arms prior to seces-sion, 36; designs on Fortress Monroe, 28; reply to President's call for troops, 114; on reconstruction, 330; sent muskets to Baltimore Police Commissioners, and controversy concerning, 303, 394. LETTERS of marque authorized, 377; by rebels, 117. LEWIS, Daniel W., Representative in First Rebel Congress,
- 401
- LEWIS, David P., deputy in rebel Provisional Congress, 11. 400; retired, 400. LEWIS, John W., Senator in First Rebel Congress, 401; su-
- perseded, 401. LIBERIA, bill to recognize, 239; President Lincoln's recom-
- mendation of, 130.
- LIEUTENANT General, grade revived, 116; Ulysses S. Grant
- LIEUTENANT General, grade revived, 116; Ulysses S. Grant appointed, 116. LINCOLN, Abraham, election of, as President in 1860, 1; dis-cussion of, in South Carolina Convention, 17; opinion on a compromise, 67; ADMINSTRATON of-his luagn-ral Address, 105-108; Cabinot, 108; reply to Virginia delegates, 112; acts of, legalized and ratified, 115, 150, and proposed censure of, 586; PROCLAMATIONS and CALLS for 75,000 troops, and answers of Governors, 114, 115; all other calls, 115, 270; the several drafts ordered, 115, 270; proclamations of blockade, 149; commercial inter-course, 149; declaring the boundaries of the insurce-270; proclamations of blockade, 149; commercial inter-course, 149; declaring the boundaries of the insurrec-tion, 150; amnesty, 147, 148; habeas corpus, 177, 178; under confiscation act, 208; rescinding General Hun-ter's, 251; respecting reconstruction bill, 318, 319; on emancipation, 227, 228; admission of West Virginia, 378; respecting aliens, 272.
  ORDERS of, on habeas corpus, 177; State prisoners, 154; in Vallandigham's case, 162; on seizing supplies, 252; rescinding Fremont's proclamation, 247; to General Schofield, 253; for the protection of colored soldiers, 280; on contrabands in the District of Columbia, 248; annulling General Wallace's confiscation orders at Bat-timore, 207.

  - timore, 207. MESSAGES of —first, 123-129; first Annual, 129-136; second Annual, 136-140, 220-229; third Annual, 140-147; in Annual, 136-140, 220-229; third Annual, 140-147; in General Stone's case, 180; on emancipation bill in Dis-

trict of Columbia, 212, 213; on compensated emancipation, 209; on repeal of the commutation clause, note, 263; explanatory of Government purchases in May, 1861, 333, 334; respecting monarchical intrigues in Cen tral and South America, 343; on the French-Mexican question, 349; on the Arguelles case, 354, 355; vetoing

- question, 349; on the Arguelles case, 354, 355; vetoing bill concerning small notes, 358; on financial legisla-tion, 362; confiscation, 197, 198. CORRESPONDENCE of—with General Fremont, 246, 247; with Fernando Wood, 296, 207; with Governor Brad-ford, 309, 310; the Nigara Falls, 301-303; Ohio and New York Democrats, 167-175. LETTERS of—to Horace treeley, 334; the Illinois Mass Convention, 335, 336; North American Review, and re-ply, 336; Colonel A. G. Hodges, of Kentucky, 336; the Grant meeting in New York, 336, 337; of acceptance as candidate for re-election, 408; on Constitutional Con-vention in Louisiana, 321; to General Steele and W. M. Fishback, on Arkansas, 321, 322. INTERVIEWS of—with committee of Allegheny City Con-
- Mismorek, of Arkinski, 031, 022. Mirekviews of—with committee of Allegheny City Con-vention, 259; some Borler State Representatives, 210, 211, and subsequent appeal to, and replies from, 213–220; deputation of colored persons on colonization, 374, 375.
- REMARKS of -At Union meeting, August, 1862, 334; to a serenade, July, 1863, 334, 335; at Philadelphia Fair, 337; at Baltimore, 280, 281; on re-nomination for President, 407, 408.
- IITLEJOHN, DeWitt C., Representative in Thirty-Eighth Congress, 140.
- LOAN, Benjamin F., Representative in Thirty-Eighth Congress, 140.
- LOANS, acts anthorizing the various, condensed, 356, 357; Secretary Chase's letter respecting redemption of in gold, note, 367.
- LOCHEANE, Judge, of Georgia, opinion on the anti-substitute law, 120.
- LOGAN, George W., Representative in Second Rebel Congress, 402. gress, 402. 106aN, John A., Representative in Thirty-Sixth Congress, 49: in Thirty-Seventh, 122: resigned, 123. 106aN, Stephen T., member of Peace Conference, 68.

- LONDON Times, comment of, respecting a reception to Messrs. Mason and Slidell, 342, 343.
- LONG, Alexander, Representative in Thirty-Eighth Congress, 140; resolution of, for appointment of commissioners to negotiate a peace, 300; Richmond Sentinel on his speech, 399; resolutions to expel and censure, 387, 388.
- LONGNECKER, Henry C., Representative in Thirty-Sixth Congress, 48.
- LONGYEAR, John W., Representative in Thirty-Eighth Congress, 140.
- LOOMIS, Andrew W., member of Peace Conference, 68. LOOMIS, Dwight, Representative in Thirty-Sixth Congress,
- 48; in Thirty-Seventh, 122.
- 48; in Thirty-Seventh, 122.
  LORD, Daniel, Scretary Seward's letter to, respecting arrest of Algernon S. Sulliran, 133.
  LOUISIANA, voie of, for President, 1; members of Thirty-Sixth Congress, 43, 49; of Thirty-Seventh, 123; claimants in Thirty-Eighth, 141; deputies in rebel Provisional Concress, 14, 400: members of First Concress, 101; of Congress, 11, 400; members of First Congress, 401; of Congress, 11, 100, memorie of rate Congress, 14, 55 Second, 402; secession movement in Convention and Legislature of, 3; ratified Confederate Constitution, 4; seizures and surrenders in, 23; free State government organized in, 321; letter of the President respecting Convention to amend constitution of, 321, 322; joint resolution of the rebel Legislature of, 332; men in the army, 399.
- Love, Peter E., Representative in Thirty-Sixth Congress, 49; withdrew, 3; member of Committee of Thirty-
- 49; withdrew, 3; member of Committee of Thirty-three, 53; report of, 58.
  LOVELOY, Owen, Representative in Thirty-Sixth Congress, 49; in Thirty-Seventh, 122; in Thirty-Eighth, 140; death of, 141; letter of, respecting President Lincoln and his Administration, 233; resolution of, respecting the capture and return of fugitive slaves 238; reso-lution of, calling for a revocation of General Halleek's order, No. 3, and vote, 253, 254; bill of, to prohibit sla-very in the Territories and vote, 254; his substitute and vote, 254; motion of, to instruct the Committee on Mil-ityry Affairs to place all soldiers on same footing, with-out regard to color, and vote, 279; resolution of thanks out regard to color, and vote, 279; resolution of thanks to Captain Wilkes for his action in arresting Messrs. Mason and Slidell, 343.
- Low, Frederick F., Governor of California, Representative in Thirty-Seventh Congress, 388.
- In Thirdy-Soventh Congress, 358.
   LOWE, Look, proposed member of board of public safety of Maryland, 9.
   LOWELL, Colonel, raid of, 119.
   LOWELT, Walter H., Chief Justice, decision on enrollment act, 273, 274.
   LUMPETIN, J. T., Representative in Second Rebel Congress, 402.

- 402. LYNCH, Dr., arrest of, 152.

- LYON, Francis S., Representative in First Rebel Congress, 401: Second, 402
- LYONS, James, Representative in First Rebel Congress, 402. LYONS, James, Representative in First Rehei Congress, 402. LYONS, LORD, Earl Russell's letter to, respecting robel ap-peal for recognition, 27; to Earl Russell respecting foreign mediation, giving views of New York Demo-crats, 347, 345; Secretary Seward's letter to, on the Trent affair, 338-342; Earl Russell's to, on the Trent affair, 342.

M

- MACFARLAND, William H., vote on secession ordinance in Virginia Convention, note, 7; deputy in rebel Provis-visional Congress, 400.
- MACHEN, Willis B., member of bogus legislative council of Kentucky, 8; deputy in rebel Provisional Congress, 400; Representative in First Congress, 289; extract MACLAY, William, Senator in First Congress, 289; extract
- from his journal containing first record of disunion threat, 389.
- threat, 389.
  MACLAY, William B., Representative in Thirty-Sixth Congress, 48; adjustment proposition, 74.
  MAGOFFIN, B., GOVERNOF OF KENUCKY, reply of, to President's first call, 114; bill passed over veto, 312, 313; declines tender of Indiana troops, 8; neutrality proclamation of, 8; veto of resolution ordering rebel troops from the soil of Kentucky, 8.
  MAGRATH, A. G., resigned as United States judge, 2; appointed Secretary of State for the republic of South Carolina, 2; decision on sequestration, 206; vote on seccession ordinance, 399.
- MAILS of the United States, to remove disqualification of color in carrying, 239, 240; opinions of Post Masters General Kendall, Campbell, Holt, and Blair, and Attoney General Cushing on freedom of the, 188-192; Cal-
- Maine Joshing Orania, respecting, 191.
   Maine, vote for President in 1860, 1; members of Peace Conference, 67; Thirty-Sixth Congress, 48; Thirty-Seventh, 121; Thirty-Eighth, 140; "personal liberty" law, 44-47.
- MALDER, Robert, Representative in Thirty-Sixth Congress, 49; Thirty-Seventh, 122; Thirty-Eighth, 140; at Border State interview with the President, 211; views on com-pensated emancipation, 213-217; amendment relating to colored persons as witnesses, 243; to Colorado bill, or colored persons as witnesses, 243; to Colorado bill, 255.
- MALLORY, Stephen R., Senator in Thirty-Sixth Congress, 48; withdrew, 3; intermediary between Col. Hayne and President Buchanan, 32; Secretary of Navy in rebel Provisional administration, 12, 400; in Permanent administration, 401.
- MANIFESTO of southern members of Congress, 37.

MANLEY, Judge, of North Carolina, opinion on habeas corpus. 130.

- MANN, A. Dudley, rebel Commisioner to England, letter of to Earl Russell, 27; reported conversation with Earl Russell, 27.
- MANNING, J. L., Commissioner from South Carolina to Lou-isiana, 11; received by convention, 4; vote on secession ordinance, 399.
- MANSFIELD, Joseph K. F., Brigadier General, order of, con-cerning fugitive slares, 245. MARCF, Daniel, Representative in Thirty-Eighth Congress,
- 140. MARCY, William L., opinion on citizenship of colored per-
- sons, note, 382
- MARQUE, letters of, bill to authorize, 377; rebel legislation on, 117. MARSHALL, Henry, deputy in rebel Provisional Congress, 11, 400; member of First Congress, 401.
- MARSHALL, Humphrey, Representative in Second Rebel Con-gress, 402.
   MARSHALL, Humphrey, Representative in Second Rebel Con-gress, 402.
   MARSHAL, 843.
   Marshall, 1843.
   Marshall, 1843.
   Marshall, 1844.
   Marshall, 1844.</
- MARTIN, Charles D., Representative in Thirty-Sixth Con-
- gress, 49. MARTIN, Eglert S., Representative in Thirty-Sixth Con-gress, 49; address to the people of Virginia, 40. MARTIN, Thomas, member of Peace Conference, 68. MARVIN, James M., Representative in Thirty-Eighth Con-

- gress, 140.
- gress, 140. MARTAND, vote for President in 1360, 1; members of Peace-Conference, 67; of Thirty-Sixth Congress, 48; Thirty-Seventh, 122; Thirty-Eighth, 140; secession movement in, 9; State Conference Convention, 9; Howard County meeting and resolutions, 9; Legislature on secession and recognition, 9, 397; proceedings of Legislature, 396-398; arrest of members, 152, 153; orders for, and report on, 153; emancipation in, 209-211, 226, 227; law aud military orders respecting elections, 308-311; Governor Bradford's proclamation, 309, 310; soldiers in the rebel army, 399.
  - in the rebel army, 399.

MASON, James M., Senator in Thirty-Sixth Congress; 48;

address of, to people of Virginia, 40; on voting ou se-cession ordinance in Virginia, 7; resolution on suspend-ing certain laws in the insurrectionary States, 80; con-respondence respecting surrender of, to Great Britain, 338 -342; comment of the London Times on, 342, 343; deputy in rebel Provisional Congress, 400; retired, 401; expected in Baltimore during "reign of terror," 393.

MASON, Mr., arrest of, 153.

- Massachuserrs, voic of, for President, 1; members of Peace Conference, 67; of Thirty-Sixth Congress, 48; of Thirty-Seventh, 122; of Thirty-Eighth, 140; "personal lib-erty" laws of, 44, 45-47.
- MATTHEWS, Joseph W., Commissioner from Mississippi to
- Alabama, II. MAUKA, M. F., recently discovered letters of, 330. MAXWELL, Augustus E., Senator in First Rebel Congress, 401; Second, 402.
- MAY, Henry, Representative in Thirty-Seventh Congress 122; arrest of 152; resolution of, respecting arrest of Baltimore Police Commissioners, 179; respecting refu-sing counsel to prisoners in Fort Warren, and vote, 181; respecting interference of Major General Schenck with
- respecting interference of Magor General Schenck with freedom of refiguous worship, and vote, 182; peace res-olutions of, 295, 296.
   MAYNARD, Horace, Representative in Thirty-Sixth Congress, 49; in Thirty-Seventh, 122; substitute of for confisca-tion bill, and vote, 198, 199; reply of, to the President's appeal to the border States, 218.
   MALUERE, Available Representative in Thirty-Fighth
- MCALLISTER, Archibald, Representative in Thirty-Eighth Congress, 140.
- Coursess, 140. MCBRIDS, John R., Representative in Thirty-Eighth Con-gress, 141. MCCLEARY, David W., resolutions in Legislature of Mary-land, 397.
- MCCLELAN, George B., Major General, President Lincoln's allusion in Message to appointment of, as general-in-ohief, 135; order to General Banks for the arrest of members of the Legislature of Maryland, 153; order respecting President's emancipation proclamation, note, 227; proclamation of, in Western Virginia, 244; order promulgating and enforcing the President's order respect-ing seizure of supplies, 252; order of chief of staff of, respecting election of 1861 in Maryland, 308, 309; urged a draft in 1861, 274; letter on political administration, 385, 386; letter in favor of election of George W. Woodward as Governor of Pennsylvania, 386.
- MCCLERNAND, John A., Representative in Thirty-Sixth Con-gress, 49; Thirty-Seventh, 122; resignation, 123; ad-justment proposition, 54; member of Border State Com-

- justment proposition, 54; member of Border State Com-mittee, 73; resolution on the war, 201. McCurre, Joseph W., Representative in Thirty-Eighth Con-gress, 140. McCurry, James, Representative in Second Rebel Con-gress, 402. McCurry, Charles J., member of Peace Conference, 67. McCurry, Charles J., member of Peace Conference, 67. McCurry, Charles J., member of Peace Conference, 67. McDurdarL, James A., Senator in Thirty-Seventh Congress, 122; in Thirty-Eighth, 140; resolutions on General Stone's arrest, 179, 180; on monarchical governments in Central and South America, 353, and French occupation in McSuco, 343, 349; amendment to enrollment bill to in Mexico, 348, 349; amendment to enrollment bill to abolish substitution, 264.
- McDowell, Irvin, Major General, order to protect property in Virginia, 251.
- McDowell, James F., Representative in Thirty-Eighth Con-gress, 140; resolution on arrests and habeas corpus, 183.

- gress, 140; resolution on arrests and habeas corpus, 183;
   McDoweLL, J. L., letter of Attorney General Bates to, re-respecting fugitive slave law, note, 235.
   McDoweLL, Thomas D., deputy in rebel Provisional Con-gress, 400; Representative in First Congress, 402.
   McLILERX, Mr., deputy in rebel Provisional Congress, 400.
   McLANDE, Walter D., Representative in Thirty-Seventh Con-gress, 123; Thirty-Eighth, 140.
   McKARG, Thomas J., commissioner from Legislature of Maryland to Jefferson Davis, and report, 9, 10; wetes in Senate of Maryland 396-395.
- Senate of Maryland, 396-398. McKEAN, James B., Representative in Thirty-Sixth Con-gress, 48; Thirty-Seventh, 122; resolution on compensated emancipation, 209.
- McKEE, R., Secretary of Russellville (Kentucky) Congress, S.
- McKENNAN, William, member of Peace Conference, 68. McKENTY, Jacob K., Representative in Thirty-Sixth Con-gress, 48.
- MCKINNEY, John F., Representative in Thirty-Eighth Con-

.

- gress, 140. MCKINNET, Robert J., member of Peace Conference, 6S. MCKNET, Robert, Representative in Thirty-Sixth Con-gress, 48; Thirty-Seventh, 122; agency in preventing the removal of heavy guns from Allegheny Arsenal, note, 35.
- McLANE, Robert M., commissioner from Legislature of Mary-land to Jefferson Davis, and report, 9; Senator Mason's desire to see, 393.

- MCLEAN, J. B., Representativo in First Rebel Congress, 402. MCLEAN, John, Justice, President Lincoln's allusion to-death of, 131.
- McMAHON, John V. L., proposed member of board of public safety in Maryland, 9. McMullen, Fayette, Representative in Second Rebel Con-
- gress, 402
- MCPHERSON, Edward, Representative in Thirty-Sixth Congress, 48; Thirty-Seventh, 122. McQuEEN, John, Representative in Thirty-Sixth Congress,
- 49; withdrew, 1; statement of, respecting forts in Charleston harbor, 30; Representative in First Rebel Congress, 402. McRAE, Colin J., deputy in rebel Provisional Congress, 11,
- 400; appointed brigadier and cotton loan agent at Paris, 400. McRar, John J., Representative in Thirty-Sixth Congress,
- 49; withdrew, 3; Representative in First Rebel Congress, 401.
- MEDIATION, 345-348-see "Foreign Mediation."
- MEDIATION, 345-345-360 "Foreign Mediation." MEEKER, Mr., deputy in robel Provisional Congress, 400. MEMMINGER, Charles G., vote on secession ordinance, 309; Secretary of the Treasury of the republic of South Carolina, 2; report in South Carolina Convention, 15, 16; remarks of, 17, 18; deputy in rebel Provisional Congress, 11, 400; Secretary of the Treasury under the Provisional Administration, 400, under the Permanent and resignation of 401; robitical history of 401
- and resignation of 401; political history of 401. MENEES, Thomas, deputy in rebel Provisional Congress, 400; Representative in First Congress, 402; Second, 402. MENZIES, John W., Representative in Thirty-Seventh Con-gress, 122; at interview of Border State Representatives and President, 211; views on compensated emancipa-
- Mercury, Charleston, on Border State Embassies, 3; on Prace Conference, 41; telegram in, respecting the disunion conspiracy, 301, 392.
   MEREDITI, S. A., Colonel, Gen. McDowell's order to, 251.
   MEREDITI, William M., member of Peace Conference, 67.
   MERRYMN, John, arrest of, 154; proceedings in case, in-cluding Chief Justice Taney's opinion, 154–158.
   Metropolity, Record, archide from Decortment of Mis.

- Metropolitan Record, excluded from Department of Mis-
- Metropolitän Record, excluded from Department of Missouri, 192.
  MESSAGES, last annual, of President Buchanan, 49, 50; President Lincoln's first, 123-129; first annual, 129-136; second, 136-140, 220-229; third, 141-147; in General Stone's case, 180; on confiscation, 197, 198; compensated emancipation, 209; on District of Columbia bill, 212, 213; on enrollment act, *node*, 263; on early proceedings to save the Capital, 333, 334; vetoing bill, authorizing small notes, 259; relative to national banking system, *node*, 362; on monarchical intrigues in Central and tem, note, 362; on monarchical intrigues in Central and South America, 343; the French Mexican question, 349; the Arguelles case, 354, 355.
  MEXICO, the French occupation of, proceedings of Govern-ment, 348-354; Mr. McDougall's resolution respecting, 348, 349; resolution of the House, and subsequent pro-ceedings, 349-354.
  MIQUIGAN, vote for President in 1860, 1; members of Thirty-Sixth Congress, 45, 49; Thirty-Sventh, 122; Thirty-Eighth, 140; personal liberty law, 46, 47; not repre-sented in Peace Conference, 69.
  MIDDLEFON, George, Representative in Thirty-Eighth Con-gress, 140.

- MIDDLETON, Mr., remarks in South Carolina Convention, 17;
- MIDLETON, MF, FEIRARS IN South Contention, At , vote on secession ordinance, 399.
   MILES, District Attorney, Charleston, on sequestration, 206.
   MILES, William Porcher, Representative in Thirty-Sixth-Congress, 49, withdrew, 1; deputy in rebel Provisional Congress, 11, 400; Representative in First Congress, 402; Second 402; vote on secession ordinance, 399; statement respecting forts in Charleston harbor, 22; declaration respecting secession, 53; on Beauregard's staff, 401.
- MILITARY arrests, General Augur's order respecting, 187.
- MILITARY Governors, letter of appointment, 170; power to-suspend the writ of habeas corpus, 179. MILITARY interference with elections, bill to prevent, 315,
- 316.
- MILITARY legislation, summary of our, 115-117.
- MILITARY legislation, summary of rebel, 117-121. MILITARY orders respecting elections, 308-316. MILITIA, bill to reorganize, in District of Columbia, 77; Da-
- vis's opimion of militia of the States, 37.
- MILLER, H. H., commissioner from Mississippi to Texas, 11., MILLER, Samuel A., Representative in First Rebel Congress, 402; Second, 402.
- MILLER, Samuel F., Representative in Thirty-Eighth Con-
- gress, 140. MILLER, William H., Representativo in Thirty-Eighth Congress, 140.

1

ţ

MILLIGAN, Samuel, member of Peace Conference, 68.

432

- MILLS, Bernard, arrest of, 153; motions and votes in Mary-land Legislature, 396-308.
- MILLSON, John S., Representative in Thirty-Sixth Congress, 49; member of Committee of Thirty-three, 53; extract of letter to James Barbour, 39.

MILLWARD, William, Representative in Thirty-Sixth Congress, 48.

- MINNESOTA, vote for President in 1860, 1; members of Peace
- MINESOTA, Vote for Freshent in 1500, 1 memory of Fence Conference, 60; of Thirty-Fixth Congress, 48, 49; of Thirty-Seventh, 122; of Thirty-Eighth, 140.
  Mississivel, vote for President in 1860, 1; members of Thir-ty-Sixth Congress, 48, 49; deputies in Rebel Provisional Congress, 11, 400; members of First Congress, 401, 402; Second, 402; secession movement in, 3; inter-State
- Second, 402; seecssion movement in, 5; inter-State commissioners, 11; legislature of, on *habcas corpus* bill, 399; men in the army, 300.
   Mussouti, vote for President in 1800, 1; members of Peace Conference, 67; Thirty-Sixth Congress, 48, 49; Thirty-Seventh, 122; Thirty-Eighth, 140; deputies in Rebel Dreminged Computers, 11, 000; numbers of Nizt Cong. Provisional Congress, 11, 400; nembers of First Con-gress, 401, 402; Second, 402; secession movement in,10, 11: seizures and surrenders of United States property, 28; ordinance of convention respecting elections, 314; military orders respecting elections in, 314, 315; pro-posed compensated emancipation, 224-226; men in the rebel army, 399.
- MITCHEL, Charles B., Senator in First Rebel Congress, 401; Second, 402. CHELL, Walter,
- proposed member of board of public MITCHELL, Walter, proportion safety in Maryland, 9.
- MITCHELL, William, Representative in Thirty-Seventh Congress, 122.
- MONARCHICAL intrigues in Central and South America, ac-
- Second Rebel Congress, 402.

- MONTAXA territory, bill to organize, 240, 241; proposed ne-gro suffrage in, 240; prohibition of slavery, 254, 255. MONTGOMERY, William, Representative in Thirty-Sixth Con-
- gress, 48.
- Gotes, James W., member of bogus legislative council of Kontucky, 8; Representative in First Rebel Congress, 401; of Second, 402.
- MOORE, Laban T., Representative in Thirty-Sixth Congress, 49

- 49.
  Moore, S. McD., vote on secession ordinance, note, 7; commissioner of Virginia in making military league, 8.
  Moore, Sydenham, Representative in Thirty-Sixth Congress, 40; telegram south, 37; withdrew, 4.
  MOORHEAD, James K., Representative in Thirty-Sixth Congress, 40; Thirty-Seventh, 122; Thirty-Eighth, 140; agency in preventing the removal of heavy guns from Allogium Arsonal and as a resolution on insclination. Allegheny Arsenal, note, 35; resolution on ineligibility to office, 376. MOREHEAN, Charles S., member of Peace Conference, 68;

- MOREMEAD, Charles S., member of Peace Conference, 68; arrest of, 153.
   MOREHEAD, John M., member of Peace Conference, 68; dep-uty in Rebel Provisional Congress, 400.
   MORFIT, Henry M., arrest of, 153; votes in Maryland Legis-lature, 396-398.
   MOREAN, Edwin D., Senator in Thirty-Eighth Congress, 140; amendment to enrollment bill, 265; remarks in open-ing the Baltimore Union Convention, 403.
   MOREAN, S. H. Representative in Second Rebel Congress
- MORGAN, S. H., Representative in Second Rebel Congress, 402.
- MORRILL, Anson P., Representative in Thirty-Seventh Congress, 122.
- MORRILL, Justin S., Representative in Thirty-Sixth Congress, 48; Thirty-Seventh, 122; Thirty-Eighth, 140; member of the Committee of Thirty-three, 53; substitute for confiscation bill, 199; for emancipation bill, 200, 201; resolution on the war. 291.
- MORRILL, Lot M., Sonator in Thirty-Sixth Congress, note, 43; Thirty-Seventh, 121; Thirty-Eighth, 140; member of Peace Conference, 67; amendment to Washington city residue bill Odday. registry bill, 241; resolution on bailing arrested per sons, 183.
- MORRIS, Daniel, Representative in Thirty-Eighth Congress,
- MORRIS, Daniel, Representative in Thirty-Eighth (Congress, 140; bill repealing fugitive slave laws, 237.
   MORRIS, Edward Joy, Representative in Thirty-Sixth Con-gress, 45; adjustment proposition, 55.
   MORRIS, Isaac N., Representative in Thirty-Sixth Congress, 49; adjustment proposition, 73; resolution on the Union, 75, 76.

- Union, 75, 76.
   MORRIS, James R., Representative in Thirty-Seventh Congress, 122; Thirty-Eighth, 140.
   MORRISON, Captain, surrender of his vessel to rebels, 23.
   MORRISON, William R., Representative in Thirty-Eighth Congress, 140; resolution on slavery, 259.
   MORSE, Freeman II., Representative in Thirty-Sixth Congress, 45; member of Committee of Thirty-three, 53; of Peace Conference, 67..

- Morron, Jackson, deputy in rebel Provisional Congress, 11, 400; vote on secession ordinance, 309. MORTON, Oliver P., Governor of Indiana, signer of Altoona
- address, 233.
- Moseny, John, commission of, as major of partisan rangers, 119.
- MULATTOES, drafting of, in rebel States, 282.
- MUNNERLYN, C. J., Representative in First Rebel Congress, 401. MUNROE, Thomas B., deputy in rebel Provisional Congress,
- 400. MURRAY, John P., Representative in Second Robel Congress,
- 402. MURRAY, United States Marshal of southern district of New

MURAY, Onlied States Marsha of southern district of New York, indicted by the Grand Jury for the arrest of Arguelles, 355; proceedings in court upon, 355.
 MURRETT, Robert T., arrest of, 153, MYERS, Amos, Representative in Thirty-Eighth Congress, 140; resolution of, relating to the war, 292.
 MYERS, Leonard, Representative in Thirty-Eighth Congress, 140

140.

## N

NATIONAL banks, proposed State taxation on, 363, 364;

- National Intelligencer, adjustment proposition of, 74; pro-position of writer in, 75; on disclosure, by "Eaton," of the distantion conspiracy, 391; on distantion caucus of 1835, 390.
- NATIONAL currency, act of 1863, 362, 363; act of 1864, 363-365.
- NAVY, disposition of vessels in, during 1860, report on, 8.
- NATY, disposition of vessors in, darking loss, report on, o.
   NEBRASEA, enabling act for, 377.
   NEGOTIATION, John D. Baldwin's resolution against, with rebels, 298, 299; Niagara Falls, 301-303.
   NEGRO suffrage, bills to authorize, in Montana Territory,
- NEERO suffrage, bills to authorize in Montana Territory, and Washington city, amendments to , and votes upon, 240, 241; vote on proposed, in New York and Illinois, note, 241; former, in Tennessee, 383.
  NEGROES, enrollment of in "rebeldom," 252, 283; draft of, for military service, 282; to work on fortifications, 282; impressment of, in Georgia, 282; clanging sentiment in the Rebel Congress respecting the employment of, in military service, 282; in the army 282, 283; act of Rebel Congress respecting, 283; General Magruder calls for, 253; legislation concerning negro prisoners, 283.
  NELSON, Homer A., Representative in Thirty-Eighth Con-gress, 140.
- gress, 140.
- NELSON, Thomas A. R., Representativo in Thirty-Sixth Con-gress, 49; member of Committee of Thirty-Three, 53; report of, 58.
- report of, 93. NESMTER, James W., Senator in Thirty-Seventh Congress, 122; in Thirty-Eighth, 140. NEYADA Territory, bill to organize, 89, 90. New England, proposition of Garrett Davis to divide, 259. New Hampshire, vote of, for President, 1; members of Peace Conference 77. Somators and Benrasontalives in Thirty-

- Conference, 67; Senators and Representatives in Thirty-Conterence, 61; Scintors and Representatives in Intry-Sixth Congress, 48; in Thirty-Seventh, 121; in Thirty-Eighth, 140; "personal liberty" law of, 44; Democrats of, Richmond Sentinel's response to resolution of, 331.
   New Jersey, vote of, for President, 1; members of Peace Conference, 67; of Thirty-Sixth Congress, 48; of Thirty-Seventh, 122; of Thirty-Eighth, 140; "personal liberty" law has no 46
- erty" law, has no, 46.

NEW Mexico, proposed admission of, 59.

- NEW Orleans, opening of port of, to trade, 149.
   NEWSPAPERS, exclusion and suppression of, 187; proceed-ceedings, orders, presentments, reports, letters, arrests, charges, and resolutions respecting, 188-194; rebel lim-
- charges, and resolutions respectively. Its-194; receir im-itations upon privileges of, 177.
   New York, vote of, for President, 1; members of Peace Conference, 67; of Thirty-Sitch Congress, 48; of Thirty-Seventh, 140; of "personal liberty" law, 46; negro suffrage in, note, 241.
   New York city, Mayor Wood's recommendation that it be-come a free city, 42-44.
   New Congressmen, adjustment proposition of 74.

- New York Congressmen, adjustment proposition of, 74. New York Democrats, Lord Lyons's statement of views of, respecting foreign mediation, 347; correspondence with President Lincoln, respecting Vallandigham's release, 167-175.
- NEW YORK (Grant) Meeting, President Lincoln's letter to. 336.

- 336.
  New York Sun, adjustment proposition of, 75.
  New York Times, on the disunion programme, 392.
  New York Tribute, on the prospect of peace, 303.
  Nusdara Falls, peace negotiations, 301-303.
  Nuslack, William E., Representativo in Thirty-Sixth Congress, 49; adjustment proposition of, 54.
  NICHOLSS, Judge, of Kentucky, proposition of, te amend the Constitution, 255, 256.
  NicHOLSS, A. O. P., Senator in Thirty-Sixth Congress, 43.
  Nirens, Eugenius A., deputy in rebel Provisional Congress, 400.

400 NIXON, John T., Representative in Thirty-Sixth Congress,

48; in Thirty-Seventh, 122; member of Border States |

4S; in Thirty-Sevenue, i.z., according to Committee, 73.
 NORLE, WARREN P., Representative in Thirty-Seventh Congress, 1:2; in Thirty-Eighth, 140; resolution of, respecting the purposes of the war, 286; amendment of, to the loan bill, 360.
 The transform Review President Lincoln's letter to public to the public sector.

lishers of, 236.

- Name 2010, 2000 Norm Carolina, vote of, for President, 1; members of Peace Conference, 9; of 1hirty-Sixth Congress, 40, 40; depu-ties in rebel Provisional Congress, 400; members of First Congress, 401, 402; of Second, 402; Secession movement in Convention and Legislature of, 4; seizures and surrenders in, 28; Supreme Court of, refuse writs of haleas corpus to persons who had furnished substitutes, habeas corpus to persons who had furnished substitutes, 120; peace proposition in Legislature of, 304, 305; res-olutions on peace, 330; adjustment proposition of "dis-tinguished citizen" of, 75; men in the army, 399.
   NOELL, John W., Representative in Thirty-Sixth Congress, 49; in Thirty-Seventh, 122; adjustment proposition of, 54; on compensated emancipation in the border States, 910, 912, 915.
- 211, 217, 218; bill reported by, to promote emancipa-tion in Missouri, and votes, 224, 225.
- NORTHWESTERN Confederacy, Cincinnati Gazette upon the intrigue for, 42. Norron, Elijah H., Representative in Thirty-Seventh Con-
- gress, 122.
- NORTON, Jesse O., Representative in Thirty-Eighth Congress, 140
- NORTON, N. L., Representative in Second Rebel Congress, 402.
- Nores, William Curtis, member of Peace Conference, 67. NUCEN, Robert H., Representative in Thirty-Seventh Cou-
- gress, 122. n

- OCHILTREE, William B., deputy in rebel Provisional Con-
- gress, 12; in Thirty-Eighth, 140.

- gress, 122; in Thirty-Eighth, 140. OFFICE, certain persons ineligible to, act declaring, 376. OFFICE, civil and military, incompatibility of, 375. OHIO Democrats, correspondence with President Lincoln re-lative to Mr. Vallandigham's return, 167-175. OHIO, vote of, for President, 1; members of Peace Confer-ence, 68; of Thirty-Sixth Congress, 48: of Thirty-Sev-enth, 122; of Thirty-Eighth, 140; no "personal liberty" law, 47. OLDHAM, William S., deputy in rebel Provisional Congress, 11, 400; Senator in First Rebel Congress, 401; Second, 402; riews on seconsetration. 205.
- 402; views on sequestration, 205. OLDEN, Charles S., member of Peace Conference, 67. OLDN, Abraham B., Representative in Thirty-Sixth Congress,
- 43; in Thirty-Seventh, 122. O'NEIL, Charles, Representative in Thirty-Eighth Con-
- gress, 140. O'NELL, John, Representative in Thirty-Eighth Congress,
- 140.
- OPINIONS of the Attorney General-J. S. Black, upon the powers of the President, 50-52; Edward Bates, on the power of the President to make arrests and suspend bow of the writ of habeas corpus, 153-161; Calch Cushing, on inviolability of U. S. mails, 189, 190; Edward Bates, on pay of colored chaplains, 279, 280; on pay of colored goldiers, 384, 385; on citizenship of colored persons, 378-384.

- ORDERS, military, respecting elections, 308-316. ORDERS of Abraham Lincoln—see "Lincoln, Abraham." ORDERS, of Secretary Cameron for the arrest of members of Dates of Abraham Lincolm-see "Lincoln, Abraham." Dates of Secretary Cameron for the arrost of members of the Legislature of Maryland, 153; of Secretary Stan-ton on State prisoners, 154; of General Mugur respect-ing military arrests, 157; of General Mansfield on fugi-tive slaves, 245; of Washington city Provost Marshal on same, 245; of General Fremont, 246; of General Wool, 245; General Halleck, 248, and proposed censure of, 253, 254; of General Fremont, 246; of General Wool, 245; General Halleck, 248, and proposed censure of, 253; General McDowell, 251; General T. Williams, 251; Liout, Col. D. R. Anthony, 251; Secretary Stanton to General Saxton, 252; General J. M. Tuttle, 253; Gene-ral Roberts, 253; of Secretary Stanton, in 1861, en-forcing the draft, 272; to Governor Andrew for raising colored troops, 279; General McClellan at Maryland election of 1801, 308, 509; General Dix, General Schenck, General Fry, 309; General Schenck, 311, 312; Lieut. Col. Tevis, 311; General Tyler, 312; General Burn-side, Lieut, Col. Johnson, Generals Furthbut, Asboth, and Shackleford, 313, 314; Colonels Foster and Mun-dy, 314; Generals Hall and Schöfeld, 314, 315; Gene-rals Hartsuff and Boyle, 315.
- rals flutistif and Boyle, 315. Omners, robel, of Colonel W. M. Churchwell, 121; of Judah P. Benjamin on bridge burning, and arrest of Judge Patterson and Colonel Pickens, 187; respecting emancipation, 118. 28

- OREGON, vote of, for President, 1: not represented in Peace
- OREGON, vote of, for President, 1; not represented in Pence Conference, 49; members of Thirty-Sith Congress, 48, 49; of Thirty-Seventh, 122, 123; of Thirty-Eighth, 141. ORR, James L., conmissioner to Washington, 2; and cor-respondence with President Buchanan, 29-32; com-missioner from South Carolina to Georgia, 11; received by Convention, 3; Senator in First Rebel Congress, 401; in Second, 402; objections to bill repealing substitute law, 121; vote on secession ordinance, 290. ORR, John A., Representative in Second Rebel Congress, 402.

102

- ORTH, Godlove S., member of Peace Conference, 68; Representative in Thirty-Eighth Congress, 140; amondment of, to the enrollment bill, 200.
   OWENS, James B., deputy in rebel Provisional Congress, 11,
- 400; vote on secession ordinance, 399.

P

- PACIFIC republic, intrigues for a, 41; Shasta Herald upon, 41; San Francisco Bee, 42.
   PAINE, Colonel E. T., arrested for declining to return fugi-
- LAINE, COIONEL E. T., arrested for declining to return fugi-tive slaves, 251.
  PALMER, John M., member of Peace Conference, 6S.
  PALMER, George W., Representative in Thirty-Sixth Con-gress, 48.
- PATTERSON, James W., Representative in Thirty-Eighth Con-
- gress, 140. PATTERSON, Judge, East Tennessee, Benjamin's order of ar-
- rest, note, 157. PATERSON, Robert, Major General, proclamation, 244. PATTON, John, Representative in Thirty-Seventh Congress, 122.
- PAY of colored soldiers, opinions of Attorney General upon, 279, 280, 384, 385; proposed prohibition of pay to, 283. Pay of soldiers, increase, 116, 272.
- PEACE Conference, propositions of, rejected in Virginia Con-vention, 7; Washington telegram of Charleston Mer-ventue, 7; washington telegram of charleston her-eary respecting, 11; delegated to officers of, proceed-ings, and propositions, and votes in, 67-69; Congres-sional votes on proposition of 60, 70.
   PEACE propositions in Congress, 294-200; correspondence between President Lincoln and Fernando Wood, 206, 201; his core Fully as and an and Vent Work Weak.
- 297; Niagara Falls conference, and New York Tribune on prospects of peace, 301-303; Richmond Examiner on, 399.
- CC propositions, rebel, in their Congress, and debate, 203-307; in Senate of Virginia, 304; Legislatures of Georgia and North Carolina, 304, 330; Richmond press PEACE
- upon, 305. PEARCE, James Alfred, Senator in Thirty-Sixth Congress, 48; Thirty-Seventh, 122; death of, 123.
- PEARSON, Judge, of North Carolina, opinion on habeas cor-pus, 120.
- PENDLEDO, George H., Representative in Thirty-Sixth Con-gress, 49; Thirty-Seventh, 122; Thirty-Eighth, 140; resolutions relative to Baltimore Police Commissioners, 180, alleged proposition of release to prisoners, 182, ar-rest of Mr. Vallandigham, 182, and the objects of the war, 265; remarks on newspaper exclusion from the mails, 191, 192. NINGTON, William, Representative in, and Speaker of
- mails, 191, 192.
   PENNISCON, William, Representative in, and Speaker of House, Thirty-Sixth Congress, 48.
   PENNSUTANIA, vote for President in 1860, 1; members of Peace Conference, 67; of Thirty-Sixth Congress, 48; Thirty-Seventh, 122; Thirty-Eighth, 140; personal lib-erty law, 46, 47; decisions of Supreme Court on enroll-ment act, 273, 274.
   PENROSE, Samuel, arrest of, 153.
   PEORLE'S Committee, address of, 410.
   PENROME, Schuer, Excent for the District Fighth Congress

- PERHAM, Sidney, Representative in Thirty-Eighth Congress, 140.
- PERKINS. KINS, John, jr., deputy in rebel Provisional Congress, 11, 400; Representative in First Congress, 401; Second, 11, 400; Representative in First Congress, 401; Second, 402; reported offered and declined Secretaryship of the Navy, 400.
   PERRY, Joseph J., Representative in Thirty-Sixth Congress, 48; member of Pcace Conference, 67.
   PERRY, M. S., Governor of Florida, message of, 2.
   PERRY, Nehemiah, Representative in Thirty-Seventh Con-gress 122; Thirty Eichelt, 10 in Thirty-Seventh Con-gress 123; Thirty Eichelt, 10 in Thirty-Seventh Con-gress 123; Thirty Eichelt, 10 in Thirty-Seventh Con-

- PERET, Nchomiah, Kepresentative in Thrty-Seventh con-gress, 122; Thirty-Eighth, 140.PERSONAL liberty laws of Maine, New Hampshire, Vermont, Massachusetts, Connecticut, Rhode Island, New Jersey, Pennsylvania, Michigan, Iowa, and Wisconsin, with movements to modify or repeal, 44-47; complained of in South Carolina Convention, 16; Alexander H. Sto-phens upon, 24, 25; propositions in Senate Committee of Thirteen, 70; resolution in House, 58, 75; remarks of Senator Simmons upon, 47. Senator Simmons upon, 47.

Schator Simmons upon, 41.
 PETORU, James L., argument on sequestration, 205, 206.
 PETTIT, John U., Representative in Thirty-Sixth Congress, 40; member of Border State Committee, 73.
 PETUS, E. W., commissioner from Alabama to Missis-sippl, 117; received by Convention, 3.

PETTON, Balie, National Intelligencer's statement respecting in 1835, 590.

PEYTON, Robert L. Y., Senator in First Robel Congress, 402. PEYTON, Samuel O., Representative in Thirty-Sixth Congress, 49.

PHELAN, James, Senator in First Rebel Congress, 401.

- PHELPS, John S., Representative in Thirty-Sixth Congress, 49; Thirty-Seventh 122; member of Committee of Thirty-Three, 53; report 58; views on compensated emancipation, 213-217.
- PHELPS, Timothy G., Representative in Thirty-Seventh Congress, 388.

PHILADELPHIA fair, President Lincoln's remarks at, 337.

- PHILLIPS county, (Arkansas,) purchase of United States arms by, in 1860, 35. Phillips, Wendell, letter to Judge Stallo, 411; to Cleveland
- Convention, 412. Picayune, New Orleans, on fall of Sumter, 114.
- PICKENS, Colonel, of Last Tennessee, Benjamin's order of arrest, 187.
- PICKENS, Francis W., Governor of South Carolina, 2; cabi-net, 2; proclamation announcing independence, 2; dep-utation of, and instructions to Isaac W. Hayne, commissioner to Washington, 32.
- PICKETT, J. T., Secretary to rebel commissioners, 109, 110. PIERCE, Franklin, letter to Jeff. Davis, January 6, 1860, 391.
- PIERPOINT, Francis II., Governor of Virginia, and signer of
- Altona address, 233. PIRE, Frederick A., Representative in Thirty-Seventh Con-gress, 122; in Thirty-Eighth, 140; amendment to ex-tend enroliment to include persons of fifty years of age, 268.
- PINCENEY, J., request for special session of Maryland Legis-lature of 1861, 8.
- PITTS, Charles H., arrest of, 152; offer of negroes to Balti-more Police Commissioners, 393; motions and votes in

- More role commissioners, 305, motions and votes in Maryland Legislature, 390–398.
   Prinsburgi, excitement in 9 on attempted removal of guiss from Allegheny Arsenal, 35.
   PLATFORMS, of Baltimore Convention, 406; of Cleveland Convention, 413; of Union League, 410.
   Pouter Commissioners, Baltimore, arrest of, and reasons, 152; application to Congress, 179,180; minutes of, 393, 304. 394.
- POLICEMAN David Daneker, of Baltimore, refusal to "con-descend" to pull down the American flag, and resignation, 394; William T. Butler resigned, 394.

POLLOCK, James, member of Peace Conference, 67.

- r, Trusten, Senator in Thirty-Sixth Congress, 48; in Thirty-Seventh, 122; telegram to Missouri, 39; amend-ment to resolution on the cause of the war, 286; ex-POLK, pulsion of, 123.
- POLYGAMY, to prohibit in Utah, 376. FOMEROY, Samuel C., Senator in Thirty-Seventh Congress,
- Pointo', Sullior C, Schuto In Finty-Sevent Congress, 122: in Thirty-Eighth, 140.
   Pointstor, Theodore M, Representative in Thirty-Seventh Congress, 122; in Thirty-Eighth, 140; substitute for loan bill, 360; amendment to national currency bill, 264.
- PORTER, Albert G., Representative in Thirty-Sixth Congress, 40; in Thirty-Seventh, 122; motion and proposition ou cmancipation bill, 200, 201. Porr Royal opened to trade, 149.

- POSTMANTER General, opinions of, 188-191. POTTER, John F., Representative in Thirty-Sixth Congress, 49; in Thirty-Seventh, 122; member of Peace Conference, 68.
- POTTER, R. B., General, President of the Vallandigham mili-tary court, 162.
  POTTE, Eurory B., Representative in Thirty-Sixth Congress,
- 48.
- PowELL, Lazarus W., Senator in Thirty-Sixth Congress, 48; in Thirty-Seventh, 122; in Thirty-Eighth, 140; ohair-man of Committee of Thirteen, 70; amendment to Grittenden proposition, 65, 66; resolution on arrest of Publication and Commission and Seventh in Keel Baltimore Police Commissioners, 179; arrests in Ken-tucky, 180; dispersion of convention, 181; amendment to indemnity bill, 185; propositions to abolish fishing bounties, 375, 376; to amend the Constitution, 255, 256; 316; protest against the bill to punish conspiracies, 377, 378.
- Powers of the President, opinion of Attorney General Black, 50-52; Chief Justice Taney, 155-158; Attorney Gen-eral Bates, 158-161.
- PRAT, James T., member of Peace Conference, 67.
  PRAT, Thomas G., request for special session of Maryland Legislature of 1861, 8; proposed member of board of

public safety, 9; Senator Mason's desired interview with, 293.

- PRESIDENT, election of, propositions to change the mode and term, 54, 55, 57, 73, 87, 255, 256.
   PRESIDENT, powers of the, Attorney General Black, 50-52; Chief Justice Taney, 155-158; Attorney General Pates, 158-161.
- PRESIDENTIAL Election of 1860, 3.
- PRESS, freedom of the, Justice Story upon, 188; other proceedings respecting, 117, 188-194.
- Prestors, John S., Commissioner from South Carolina to Virginia, 11; received by Legislature, 6; h.s denunca-ation of the Union, 6; telegram to Judge Hopkins, 40,
- Representative in First Congress, 400; Representative in First Congress, 400; Representative in First Congress, 402; Second, 402, PRESTON, William Ballard, vote on secession ordinance, 7 : commissioner of Virginia in making infiltary league, 8; delegate from Virginia convention to President Lincoln, 100 Acoustria in wheel Representations (2000) 112; deputy in rebel Provisional Congress, 400; Sena-tor in First Congress, 401; retired, 401.
- PRICE, Hiram, Representative in Thirty-Eighth Congress, 140.
- PRICE, Rodman M., member of Peace Conference, 67; letters of, in 1860, found in Fredericksburg, Va., 390.
   PRICE, Thomas H., Representative in Second Robel Con-
- gress, 402.
- PRICE, Thomas L., Representative in Thirty-Seventh Con-gress, 123; views on compensated emancipation, 213-217.
- PROCEEDINGS of the Government in relation to the secession movement, 48-90. PROCLAMATIONS of President Lincoln, for troops, 141; am-
- nesty, 147, 148; blockade, 149; commercial intercourse, 149, 150; declaring the boundaries of the insurrection,
- 149, 150; declaring the boundaries of the insurrection, 150; suspending the privilege of the writ of habcas cop-pus, 177, 173; under confiscation act, 208; reseinding General Hunter's proclamation, 251; respecting recon-struction bill, 318; of emancipation, 227, 228.
  PROCLAMATIONS, of General McClellan in Western Virginia, 244; General Patterson, 244; General Fremont on emancipation, 245, 246; of General Thomas W. Sher-man, 248; of General John A. Dix, 248; General Burn-sido in North Carolina, 249; General Halleck, 250; General Hunter on emancipation 250; of Governor General Hinter, on emacipation, 230; of Governor Bradford, 309, 310; of Governor Cannon of Delaware, 312; Governor J. F. Robinson of Kentucky, 313. PROCLAMATIONS, Rebel, Davis's of banishment, 121; on con-
- FROCKMARTIONS, InCOLF, DATES & OF CONSIMILATION, 12.1: OUTCOMPT scription, 113; martial law, 121; outlawary of General Butter, note, 233; Governor J. E. Drown on northern dotts, 2; General M. Jeff, Thompson, note, 245, 246. PROFERTY, rebel regulations for destroying, 117.
- PROPOSED censure of officials, proceedings of Congress on, 386-388.

PROPOSITIONS of adjustment—see "Adjustment." PROTESTS, of 36 members of House of Representatives against the annesty bill, 184, 185; of nine Senators against the bill to purish conspiracies, 317. **PROVISIONAL** Cabinet, of Jeff. Davis, 12, 400. **PROVISIONAL** Constitution, rebel, adopted, 12; particulars

of. 12.

- PROVISIONAL Congress, rebel, names of deputies and officers, 11, 400; proceedings of, 12.
   PRUXN, John V. L., Representative in Thirty-Eighth Con-gress, 140; resolution relative to the seized newspaper offices in New York, 194.
- PRYOR, Roger A., Representative in Thirty-Sixth Congress, 49; address to the people of Virginia, 40; adjustment proposition of, 73; resolution on coercion, 76; on bom-bardment of Sunter, 112; deputy in rebel Provisional Congress, 400; Representative in First Congress, 402; retired, 402. "PUBLIC Safety " bill in Legislature of Maryland, 398

- "PUBLIC Safety " bill in Legislature of Maryland, 398.
   PUGI, George L., Schattor in Thirty-Sixth Congress, 45; ad-justment proposition of, and vote upon, 63, 64.
   PUGI, James L., Representative in Thirty-Sixth Congress, 49; his purpose in taking his seat, 392; declaration re-specting secession, 53; sent telegram South, 37; with-drow the Perspective Line King Perdol Congress, 401. drew, 4; Representative in First Rebel Congress, 401; Second, 402.
- PUBLIC credit, under Euchanan, 365, 366. PUBLIC debt-statements of amount at different periods, 365-368; Secretary Chase's letter respecting the redemption of in gold, note, 367; of rebels, 386.
- PURYEAR, Robert C., deputy in rebel Provisional Congress, 400.

QUARLES, James M., Representative in Thirty-Sixth Congress, 49. QUINLAN, Mr., arrest of, 152.

R

RADFORD, William, Representative in Thirty-Eighth Congress, 140.

- Raleigh Standard, sensation article of, 38.
- RALLS, John P., Representative in First Robel Congress, 401; Second, 402. RAMAAY, J. G., Representative in Second Robel Congress,
- 40 .
- RAMSEY, Alexander, Senator in Thirty-Eighth Congress, 140
- RANDALL, Samuel J., Representative in Thirty-Eighth Congress, 140. RANDALL, William H., Representative in Thirty-Eighth Con-
- gress, 140.
- RANDOLPH, George W., vote on secession ordinance, 7, note; delegate from Virginia to President Lincoln, 112; declined to receive Creoles into the military service, and reason for, 282; Provisional Secretary of War, 401; re-

- RAYMOND, Henry J., report of, in Baltimore Convention, 406.
- READ, Henry E., Representative in First Rebel Congress, 401; Second, 402.
  READ, John M., Justice, opinion on enrollment, 273.
- REAGAN, John II., Representative in Thirty-Sixth Congress, 49; deputy in Rebel Provisional Congress, 11, 400; Postmaster General in Provisional and Permanent administration, 400, 401.
- REBEL Administration, names of persons composing, 400-402.
- AUL. REBEL army, strength of, 121, 399. REBEL commissioners, appeal of certain, to Earl Russell, and reply of latter, 27; correspondence of certain, with President Buchanan, 29-34; intercourse of certain, with Secretary Seward, 108-110. Purp Concession process of neuronal effect. 400, 102.
- REBEL Congresses, names of members and officers, 400, 402. REBEL debt, 368.
- REBEL financial legislation, 368-373.
- REPEL indical legislation, 305-313.
  REPEL independence, demands and terms, 303-307, 329-332.
  REPEL legislation, military, 117-120; letters of marque, 117; conscription laws, 118, 119; on habeas corpus, 157; sequestration, 203; military employment of colored persons, 251-253; on peace, 300-307; reconstruction, 329-332; taxation, 365-373.
  PERF Sete cloptory 104:06 218, 210, recentions on their
- REBEL States, electoral vote of, 318, 319; resolutions on their relations to the Government, 322-329; reconstruction of, 317-322
- RECOGNITION of Hayti and Liberia, recommendation of Pres-
- Reconstruction of real recommendation of Pres-ident Lincoln, 130; bill for, 239; of rebel independence demanded by Legislature of Maryland, 397, 398; of in-dependence, the sine qua non of rebels, 303-307, 329-332.
   RECONSTRUCTION, proceedings respecting, 317-331; rebel press and politicians on, 330, 331; George N. Sanders on terms of, 330.
- RECONSTRUCTION, rebel view of, views of various rebel leaders, Legislatures, Congress, and press upon, 329-332. RECTOR, Governor, of Arkansas, reply of, to President's first
- call for troops, 115.
- call for (Foops, Lto. RED, David S, member of Peace Conference, 68. RED, John W., Representative in Thirty-Seventh Congress, 122; expelled, 123. REPORT to the South Carolina Convention of Mr. Rhett, 12-
- 15; of Mr. Memminger, 15, 16; of Mr. Withers, 19; of Major Anderson to the Secretary of War, 114; of committee and Secretary Holt on secret organiza-tion to attack the capital, 80-82; of Mr. Dawes on the disposition of United States ressels, also respecting re-directions; in the new 69 Site and minority the Mr. as possible of the navy, 82-84; and minority, by Mr. Branch, 84; majority by Mr. Reynolds and minority by Mr. Branch upon bill to authorize the President to call between the President and the "special envoy" of South Carolina, 85; of Committee operations of solution suppression of newspapers, 188-191; on transfer of arms South in 1850-60, 34-36; of House Committee of Thirty-three, 57-62; of Peace Conference, 03; Senate Committee of Thirteen, 69-73; of John Cochrane on imports bill, 79, 80. RESIGNATIONS in the Navy-see "Report."
- RESOLUTIONS for protection of slave property in the Territories, 90.

tories, 90. Resolutions offered at second session of Thirty-Sixth Con-gress, 58-59; in Thirty-Seventh and Thirty-Eighth Congresses, on arrosts, 170-183; on newspaper suppres-sion, 194; compensated emancipation, 209, 210, 226; emancipation proclamation, 229-231; Gen. Halleck's order, 253; slavery, 259; enrollment act, 270, 271; the objects of the war, 285-290; the prosecution of the war, 290-294; "peace," 294-296, 297-300; military or-ders concerning elections, 315; the relations of the re-bellious States to the Government, 322-329; arrest of Mason and Slidell, 343; monarchical intrigues in Cen-

- tral and South America, 343; foreign mediation, 346, 347; the French in Mexico, 348, 319; the Arguelles case, 354, 355; taxation, 359; incompatibility of civil and military office, 575; declaring certain persons in-eligible to office, 376; proposed censure of President Lincoln, 386, of ex-President Buchman, 387, 388, Rep-
- resentatives Long and Harris, 357, 358, RESOLUTIONS, rebel, of Congress on independence, 329; Legislature of Virginia, on guerrillas, 119: Legislature of Georgia, 320, 509; of North Carolina, 530; of Louisiana, 332; of Mississippi, 300.
- REVNOLDS, Edwin R., Representative in Thirty-Sixth Con-gress, 48.
- REYNOLDS, John H., Representative in Thirty-Sixth Con-
- RELYODDS, JOHN H., Representative in Inity-Sixin Con-gress, 45; member of Scleet Committee on Military and Naval Affairs, 80; report of, 85.
  RHETT, Robert Barnwell, Sr., deputy in rebel Provisional Congress, 2, 11, 400; report in South Carolina Conven-tion, 12-15; remarks on, 18; view on fugitive slave law, 18; with on scoresion exclusions, 200
- tion, 12-15; remarks on, 18; view on fugitive slave law, 18; vote on secession ordinance, 200.
  RHODE Island, vote for President in 1860, 1; members of Peace Conference, 67; Thirty-Sixth Congress 48; Thirty-Seventh, 122; Thirty-Eighth, 140; "personal liberty" law, 46, 47.
  RICE, Alexander II., Representative in Thirty-Sixth Con-gress, 48; in Thirty-Seventh, 122; in Thirty-Eighth, 140.
- RICE, Henry M., Senator in Thirty-Sixth Congress, 48; in Thirty-Seventh, 122; member of Committee of Thirteen, 70; proposition of adjustment, 72.
- (10) proposition of adjustment, 42.
  Rice, John H., Representative in Thirty-Seventh Congress, 122; Thirty-Eighth, 140; resolution on the war, 204; proposed militia bill, 279.
  Ricekneson, William A., Representative in Thirty-Seventh Congress, 122; in Thirty-Eighth, 123; Senator in Thir-ty-Eighth, 123; resolution of, respecting imprisonment of citizens of Illiveis and varia unon 151; anondment. V-Eighth, 125; resolution of, respecting imprisonment of citizens of Illinois, and vote upon, 181; amendment of, to bill to increase pay of goldiers, and vote, 272. Richmond Enquirer, 5, 114, 305, 350, 590—see "Enquirer." Richmond Extinel, 331, 390—see "Sentincl." Richmond Sentinel, 331, 390—see "Sentincl." Richmond Multia votilianstitute full 121.

- Richmond Whig, on anti-substitute bill, 121. Richmed, Albert G., Representative in Thirty-Seventh Con-
- gress, 122. RIDDLE, George R., Senator in Thirty-Eighth Congress, 140.
- RIDGELT, Henry, member of Peace Conference, 68. RIGGS, Jetur R., Representative in Thirty-Sixth Congress, 48.

RILEY, E. S., arrest of, 153.

- RIVES, William C., member of Peace Conference, 68; deputy in rebel Provisional Congress, 400; Representative in Second Congress, 402. RoBERTS, Benjamin S., Brigadier General, order of, con-
- cerning contrabands, 253.
- ROBERTSON, Judge, Commissioner from Virginia to South Carolina, 2.
- ROBINSON, Christopher, Representative in Thirty-Sixth Con-gress, 48; member of Committee of Thirty-three, 53.
- ROBINSON, James C., Representative in Thirty-Sixth Congress, 49; in Thirty-Seventh, 122; in Thirty-Eighth, 140.
- ROBINSON, J. F., Governor of Kentucky, proclamation of, 313.
- ROBINSON, Lucius, letter on the Cleveland Convention, 413.
- RODER, George B., member of Peace Conference, 63. RODERS, A. A. C., claimed seat as Representative from Ar-kansas, 141.
- ROGERS, Andrew J., Representative in Thirty-Eighth Congress, 140; resolutions of, asking for the appointment of peace commissioners, and vote, 200.
- ROGERS, St. George, Representative in Second Rebel Con-
- Roulas, B. Coupe, Representative in Level Con-gress, 402.
   Roulas, Edward H., Representative in Thirty-Seventh Con-gress, 122; in Thirty-Eighth, 140.
   Roulas, Janes S., Representative in Thirty-Seventh Con-
- ROLLINS, James S., Representative in Thirty-Seventh Con-gress, 122; in Thirty-Eighth, 140; on compensated emancipation, 213-217; resolution of, relating to the prosecution of the war, and votes, 250. ROMAN, A. B., rebel commissioner, 110. ROMAN, J. Dixon, member of Peace Conference, 68. Ross, D. G., proxy of Governor Morton at Altoona meeting, application of the second second second second second second provide the second second second second second second provide second secon

- 623.
- ROSECRANS, William S., Major General, exclusion of Metro-politan Record from his department, 192.
- Ross, Lewis W., Representative in Thirty-Eighth Congress, Ross, Lewis W., Representative in Thirty-Eignth Congress, 140; resolution of, to give pay and mileage to civilians arrested and released without trial, 153.
   Ross, Robert J., commissioner to the President from the Legislature of Maryland, 9; report of, 9.
   Rost, P. A., commissioner to England, letter of, to Earl Rus-sell, 27; reported conversation with Earl Russell, 27.
   Rorce, Homer E., Representative in Thirty-Sixth Congress, 48.

gress, 401.

- RUFFIN, Thomas, Representative in Thirty-Sixth Congress, 49; telegram sent South, 37; member of Peace Conference, 63; deputy in rebel Provisional Congress, 400; death of, 401.
- RUGER, Lieutenant Colonel, instructions of General Banks to, 153, 154.
- to, 153, 154.
  RUSSELL, Charles W., deputy in rebel Provisional Congress, 400; Representative in First Congress, 402; Second, 402.
  RUSSELL (EARL), rebel Commissioners' letter to, 27; reply of, 27; letter of Lord Lyons to, on Foreign Mediation, 347-348; letter of, to Lord Lyons, on the Trent affair 338; same to same, on same, 342.
  RUSS, Albert, Representative in Thirty-Sixth Congress, 49; member of Committee of Thirty-three, 33; report of, 55; deputy in school Provisional Congress, 409.
- 5S; deputy in rehel Provisional Congress, 400.

S

- SALMON, B. S., arrest of, 153.
- SALOMON, Edward, Governor of Wisconsin, signer of Altoona address, 233.
- San Francisco Bee, upon a Pacific republic, 42. Sanbers, George N., letter of, to Governor Seymour and others, on reconstruction, &c., 330; Niagara Falls Conference, 301-303.
- SANGSTON, Lawrence, arrest of, 152.
- SAEDENT, Aaron A., Representative in Thirty-Seventh Con-gress, 389.
- SAUERWEIN, Peter G., emancipation resolutions offered by, and adopted in Baltimore City Union Convention in 1862, 226.
- 1862, 226. SAUSJURY, Willard, Senator in Thirty-Sixth Congress, 43; Thirty-Seventh, 122; Thirty-Eighth, 140; member of Border States Committee, 73; resolutions on arrests in Delaware, 181, and release of prisoners, 182; amend-ments to bill repealing the fugitive slave laws, 226, 227; to the bill creating a new article of war, 238; to freedmen's bill, 260; to army appropriation bill, 238; to tax bill, 250; and to resolution of censure of Exto tax bill, 375; and to resolution of censure of Ex-President Buchanan, 387; joint resolution for peace commissioners, 296; proposed constitutional amend-
- commissioners, 200; proposed constitutional amendment, 157.
   Sawanah News, December 24, 1800, publication in, of Senator Toombs's address to Georgia, 37.
   SCHENCK, Robert C., Major General, orders, &e., concerning elections, 300, 311; reply to Governor Bradford, 311.
   SCHENCK, Fobert C., Hepresentative in Thirty-Eighth Con-
- gress, 140; resolution upon eligibility of, 375; resolu lution on the war, 294; resolution to censure Benjamin G. Harris, 375; amendments to the enrollment bill, 265, 166, 268.
- SCHNABLE, Ellis B., arrest of, 153.
- SCHOFIELD, John M., Major General, orders concerning elections, 314, 315.
- SCHOOLS, colored, legislation respecting, 244. SCOFIELD, Glenni W., Representative in Thirty-Eighth Con-gress, 140.
- Scorr, Charles L., Representative in Thirty-Sixth Congress, 49; for a Pacific republic, 41.
   Scorr, John G., Representative in Thirty-Eighth Congress,
- 140. Scorr, Otho, commissioner to President Lincoln from the
- Legislature of Maryland, and report, 9. rr, Robert E., deputy in Rehel Provisional Congress, SCOTT,
- 400.
- 400.
   Scorr, T. Parkin, arrest of, 152; votes and motions in Maryland Legislature, 393-393.
   Scorr, Winfield, Lieutenant General, President Lincoln's allusion to retirement of, 135; letter of, relative to quotas of arms, 35, 36; order of, to Col. Martin Burke, and the tambination of the second relative to military prisoners, 152; President's orders to, respecting suspension of habeas corpus, 177. SCRANTON, George W., Representative in Thirty-Sixth Con-
- gress, 43. SEDASTIAN, William K., Senator in Thirty-Sixth Congress, 45; member of Border States Committee, 73.
- SECESSION movement developed, 2-47.
- SECESSION Ordinances, vote on, in South Carolina, 339, 239; Florida, 339; Virginia, note, 7; Tennessee military hours. league, 5.
- league, 5.
  SECESSION, right of, Mr. Doolittle's amendment respecting, 63; Thomas E. Franklin's, in Peace Conference, and David Dudley Field's, 69; President Buchanan's opinion, 49.
  SEDDON, James A., member of Peace Conference, 63; report on its proposition, 6; adjustment proposition, 69; deputy in rebel Provisional Congress, 400; rebel Secretary of War, 401; declined to receive creoles into the military service, and the reason given, 252.
  SEDEWIC, Charles B., Representative in Thirty-Sixth Congress, 48; Thirty-Seventh, 122; substitute for emancition Uill, 200.
  SEGAR, Joseph, Representative in Thirty-Seventh Congress.
- EZGAR, Joseph, Representative in Thirty-Seventh Congress, 123; claimant in Thirty-Eighth, 141.

- ROYSTON, Grandison D., Representative in First Rebel Con- SEIZURES of Government property in insurrectionary States, 28; Governor Letcher's proposed attempt on Fortress Monroe, 28.
  - SEMMES, Thomas J., Senator in First Rebel Congress, 401: Second, 402.

  - Second, 402.
     SENATIC Committee of Thirteen, 70; proceedings of, 70-72.
     SENATIC (*Micl. Mond.*) on reconstruction and on northern temper, 331; on Mr. Long's speech, 209.
     SEWARD, William H., Senator in Thirty-Sixth Congress, 49; on Committee of Thirteen, 70; proposition therein, 71; also respecting armed invasion, 71.
  - on Committee of Thirteen, 70; proposition therein, 71; also respecting armed invasion, 71. SZWARD, William H., Secretary of State 108; the commis-sioners of the Southern Confederacy 108-111; to Daniel Lord and Gov. Washburn respect 1g arrests, 153; to Mr. Adams on the Trent affair, 328; to Lord Lyons, 328; on foreign enlistments, 343, 344; to G.y. Hicks on foreign mediation, 345; to Mr. Dayton in .seponse to letter of M. Drouyn de l'Huys on mediation, 345, 346, letters of, and Mr. Dayton respecting the French in Mexico, 349, 350; to the President on the Arrenelles case, 355. 350; to the President on the Arguelles case, 355.
  - SEXTON, B. F., Representative in First Rebel Congress, 402; Second, 402.
  - SETMOUR, Horatio, Governor of New York, letters to District Attorney Itali respecting the suspension of the Journal of Commerce and World newspapers, 192, 193; and the draft in New York city, 274.
  - SHACKELFORD, James M., Brigadicr General, orders respect-ing elections, 313, 314. SHANKS, John P. C., Representative in Thirty-Seventh Con-
  - gress, 122; resolution respecting return of fugitive slaves, 233. SHANNON, Thomas B., Representative in Thirty-Eighth Con-

  - gress, 140. Stasta (Cal.) Herald, on Pacific republic, 41. SHEFFIELD, William P., Representativo in Thirty-Seventh Congress, 122.
  - SHELLABARGER, Samuel, Representative in Thirty-Seventh
  - Congress, 122; resolution of, on the war, 230. SHILMAN, John, Representative in Thirty-Sixth Congress, 48; member of Border States Committee, 73; proposi-tion of adjustment, 55; resolution respecting slavery in the States, 77.
  - SHERMAN, John, Senator in Thirty-Seventh Congress, 122; in Thirty-Eighth, 140; amendment to confiscation bill In run vy-figure, 189; antendment to consist of bill, and votes, 202, 265; amendments to carolinear bill, and votes, 202, 265; amendment to bill repealing fugitive slave law, and vote, 256; explanatory remark of vote, note, 245; amendment to bill to employ colored per-sons in the United States service, 275; amendment to the law transfer bill  $\frac{65}{2}$ the legal tender bill, 358.
  - SHERMAN, Socrates N., Representative in Thirty-Seventh Congress, 122.
  - SHERMAN, Thomas W., Brigadier General, proclamation to the people of South Carolina, 248.
  - SHEWMAKE, J. gress, 402. J. T., Representative in Second Rebel Con-
  - grees, 122. SHIEL, George K., Representative in Thirty-Seventh Con-gress, 123. SHORTER, John Gill, deputy in rebel Provisional Congress,
  - 11, 400; elected Governor of Alabama, 400.

  - 11, 400; elected Governor of Alabama, 400.
    SICKLES, Daniel E., Representative in Thirty-Sixth Congress, 48; adjustment proposition of 54; report to, in fugitive slave case, 250.
    SIMNOS, James F., Senator in Thirty-Sixth Congress, 49; Thirty-Seventh, 122; resigned, 123; remarks on "personal liberty" laws, 47.
    SIMMIS, William E., Representative in Thirty-Sixth Congress, 49; Senator in First Rebel Congress, 401; Second, 402.
    SINGLERON, Otho R., Representative in Thirty-Sixth Congress, 402; Second, 402.
    SINGLERON, Otho R., Representative in Thirty-Sixth Congress, 49; telegrum south, 37; withdrew, 3; declaration respecting secession, 52; Representative in First Rebel Congress, 401; Second, 402.
    SLAVERY, the avowed cause of secession, 17; Crawford's respective and an end second and cause of secession, 17; Crawford's respective and second and cause of secession, 17; Crawford's respective and second cause of secession, 17; Crawford's respective and second cause of secession, 17; Crawford's respective second cause of se

- SLAVERT, the avowed cause of secession, 17; Crawford's res-olution respecting the Constitution and, 76; Sherman's resolution on non-interference with, in the States, 77; resolution on non-interference with, in the States, 77; proposed constitutional amendment to protect, in States, 59, 60; other propositions, 63-75; Davis's proposition respecting, in Senate Committee of Thirteen, 71; pro-ceedings to prohibit in the Territories, 254, 255; Owen Lovejoy's bill and substitute, 254; proceedings to so amend the Constitution as to p withit, 255-250; Aaron Harding's resolution on, 259; William R. Morrison's resolution on, 250. resolution on, 259.
- SLATES, bill to prohibit employmen : of, in dock-yards, 238; resolution respecting, and vota 238. SLAVES to be enrolled and draftel 274; the master to re-ceive bounty of \$100, 275; if drafted to be free, 275;
- proposed tax on, 361.
- ELAVES, rebel employment of, 281, 282; draft of, for military service, 282; views of Legislature of Alabama, :..; tax imposed in 1861, 370.

SLAVE trade, African, treaties and legislation concerning, 150-152.

- SLAVE trade, constwise, repeal of laws regulating, 243, 244. SLIDELL, John, Senator in Thirty-Sixth Congress, 48; with
  - oran, conin, centator in inity-sixth Congress, 45; with-drew, 4; intermediary between Col. Hayne and Presi-dent Buchanan, 327; correspondence respecting sur-render of to Great Britain, 538, 342; comments of the London Times, 542, 343; votes in Congress on arrest of, 343; allusion to by Mr. Dayton, 350.
- SLOAN. A. Scott, Representativo in Thirty-Seventh Congress, 122. SLOAN, Ithamar C., Representative in Thirty-Eighth Con-
- gress, 140.
- SLOOPS-OF-WAR, construction of, anthorized, 89.
  SMALL notes, proposed legislation, 363-365; President Lin-coln's veto of bill for, 359.
- Cons vewon on nor. 339. SMALR, Robert, bill for relief of, 239. SMTH, Caleb R., member of Peace Conference, 63; Secre-tary of the Interior, 108, and resigned, 108. SmTH, Edward H., Representative in Thirty-Seventh Con-
- gress, 122.
- SMTH, Green Clay, Representative in Thirty-Eighth Con-gress, 140; resolutions relating to the war, 292, 293, 298, SMTH, James C., member of Pace Conference, 67.
  SMTH, James M., Representative in Second Rebel Congress,
- 402
- SMITH, Robert H., deputy in rebel Provisional Congress, 11, 400.
- 400.
   Smith, William, Representative in Thirty-Sixth Congress, 49; adjustment proposition of, 54; Representative in First Polyd Congress, 402: resigned to accent military First Rebel Congress, 402; resigned to accept military commission, 402; elected Governor of Virginia, 402; views on reconstruction, 331.
- SMITH, William E., Representative in Second Rebel Congress, 402.
- SMITH, William N. H., Representative in Thirty-Sixth Con-Representative in First Congress, 400; Representative in First Congress, 402; in Second, 402; statement of strength of rebel army, 121, 399.
- ·SMITH, William R., Representative in First Rebel Congress, 401; Second, 402
- SMITHERS, Nathaniel B., Representative in Thirty-Eighth Congress, 140; amendment to enrollment act, 268. SNEAD, Thomas L., Representative in Second Rebel Con-
- gress, 402.
- SOLDIERS, bi for, 284. bills to increase pay, 116, 271, 272; homesteads
- SOLDIERS, colored, legislation concerning, 274-279; proposed prohibition of payment to, 283, 284; rebel facts con-cerning, 281-283-see "Colored Soldiers."

- Cerrmng, 251-255-see "Colored Soliders." Sours, Daniel L., Representative in Thirty-Sixth Congress, 48; member of Peace Conference, 67. Sourn, overtures of, to the Northwest, 42. Sourn Carolina, vote of, for President, 1; members of Thirty-Sixth Congress, 45, 49; deputies in rebel Provisional Congress, 14, 400, members of Fint Congress 10, 400. Sixth Congress, 45, 45, 46, 49, deputtes in rever Provisional Congress, 11, 400; members of First Congress, 401, 402; Second, 402; secession movement in, 2; ratified Con-federate Constitution, 3; Convention rejects co-opera-tive action, profiered by Virginia, 2; resolutions rela-tive to, 2; inter-State commissioners of, 11; declaration of independence, 12; seizures and surrenders in, 27; letter of commissioners of, to Mr. Buchanan, 29; Gov. letter of commissioners of, to Mr. Buchanan, 29; Gov. Pickens of, demands the surrender of Fort Sunter, 32;
  Fort Sumter surrendered, 27; secession of, not the work of a day, but maturing for thirty years, 18; Convention of, seads despatch to Mayor of Macon, 38; also to Mayor of New Orleans, 39; personal liberty laws, subject of complaint in convention of, 16; 17; report of Judge Withers to Convention of, 19; vote on secession ordinance of, 308, 599; men in the army, 399.
  SOUTHEIN Congress, rebel Provisional, officers and proceedings of, 12, 400; tariff act passed, 12; commissioners admitted from North Carolina, 12; Constitution of Provisional Government adopta, 12; assumes charge of all questions and differences between the "sovereign States of the Confederacy" and the United States Government, 12; Texas deputies received, 12; Alexander H. Stephens, of Georgia, elected Yresident, 12; Alexander H. Stephens, of States of the Confection Prosident, 12; Alexander H. Stephens, of States of the Confection of and the sum of the sovereign for the sum of the sovereign for the sovereign for the source of the sovereign of the confections of the confection of the sovereign of the confections of the sovereign of th
- H. Stephens, of Georgia, elected Vice President, 12.
- Southern manifesto, 37. Sparn, A. C., commissioner from South Carolina to Arkansas, 11; vote on secession ordinance, 399.

- 11; vote on secession ordinance, 339. SPARROW, Edward, deputy in robel Provisional Congress, 11, 400; Senator in First Congress, 401; Second, 402. SPALDING, Rufus P., Representative in Thirty-Eighth Con-gress, 140; bill to repeal fugitive slave law, 237. SPAUDING, Ellnidge G., Representative in Thirty-Sixth Congress, 48; Thirty-Seventh, 122. SPENCER, William A., letter for Couvention in Maryland, 9. SPINNEE, Francis E., Representative in Thirty-Sixth Con-gress, 48. gress, 48. SPBATT, L. W., commissioner from South Carolina to Florida,
- 11; received by Convention, 3; vote for secession ordinance, 399.

- SPRINOFIELD Mass Convention, President Lincoln's letter to. 335. 336
- SPRAGUE, William, Governor of Rhode Island and signer of Altoona address, 233; Senator in Thirty-Eighth Congress, 140.

- gross, 140. STALLO, Judge, letter of Wendell Phillips to, 411. STALLWORTH, James A., Representative in Thirty-Sixth Con-gross, 49; sent telegram South, 37; withdrew, 4. STALLEY, Edward, appointed Military Governor of North Carolina, 179.
- STANTON, Benjamin, Representative in Thirty-Sixth Con-
- STANTON, Benjamin, Representative in Thirty-Sixth Congress, 49; bill to re-organize the militia of the District of Columbia, 71; to suppress insurrection, 77; report on transfer of United States arms south, in 1850-60, 34.
   Stanton, Edwin M., Attorney General, 25; Scortary of War, 108; orders on military arcesis, 154; form of appointment of military governor, 179; instructions to General Saxton, 251, 252; letter to General Butler on the latter's controversy with General Phelps, 252; orders, &c., enforcing the draft, 272; letter on Commutation clause of enrollment act, 263; to Governor Andrew cespecting colored enlistments, 279; reply to Senate on alleged foreign enlistments, 343.
   STANTON, Mrs. E. Gady, letter to Fremont Club, 411.
   Startes, Waller R., deputy in rebel Provisional Congress, 400; Representative in First Congress, 402; in Second, 402.
- 402
- STAR of the West, fired upon by the rebels, 27. Star, Washington Evening, publication in, of vote on Vir-ginia secession ordinance, note, 7. STARB, John F., Representative in Thirty-Eighth Congress,
- 140
- 140. STATE of West Virginia, bill to erect, 377, 378. STATE taxation, on United States bonds and national banks, 358, 859, 360-364.
- STATES, militia of, part of the army of the United States, Jefferson Davis's opinion, 37.
- STEBBINS, Henry G., Representative in Thirty-Eighth Congress, 140. STEELE, Frederick, Major General, President Lincoln's let

- STEELE, Frederick, Major General, President Lincours not ter to, 322.
  STEELE, John B., Representative in Thirty-Seventh Congress, 122; in Thirty-Eighth, 140.
  STEELE, William G., Representative in Thirty-Seventh Congress, 122; in Thirty-Eighth, 140.
  STEPHENS, Alex. H., deputy in rebel Provisional Congress, 11, 400; commissioner for Confederate States in making the league with Virginia, 8; elected rebel Provisional Vice Fresident 12; permanent, 401; speech before the Georgia Legislature in 1860, 20-25; fishing bounties discussed by, 21; prediction of universal bounties discussed by, 21; prediction of universal conversion, 25; extract from speech in Georgia State Convention, 25; extract from address in 1559, 25, 26; "Confederate" Constitution expounded by, 103, 104; "correspondence with Secretary Welles, respecting pass to
- correspondence with Secretary Welles, respecting pass to Washington, note, 307; views of, on reconstruction, 331.
  STEVENS, Thaddeus, Representative in Thirty-Sixth Con-gress, 43; Thirty-Seventh, 122; Thirty-Eighth, 140; in-demity bill of, 184; propositions to amend the Consti-tution, 253; substitute for the enrollment bill, 266, 267; amendment to, 268; bill for enlistment of 250,000 colored soldiers and modification, 276; joint resolution offering to free all slaves who leave their masters or aid in supressing the rehelion. 277; resolution relative to othering to free all slaves who leave their masters or all in suppressing the rebellion, 257; resolution relative to the war, 283; amendment to "legal tender" bill, 358; letter of Sceretary Chase to, on "legal tenders," 358, 359; substitute for loan bill, 360; substitute for na-tional currency bill, 363, 364. STEVENS, William H., member of Peace Conference, 68.
- STEVENSON, John W., Representative in Thirty-Sixth Con-gress, 49; adjustment proposition of, 54. STEVENSON, William E., motion in Baltimore Convention,
- 405. STEWART, George H., of Maryland, proceedings of, 393, and
- note, 394.

- note, 394.
  STEWART, James A., Representative in Thirty-Sixth Congress, 49; adjustment proposition of, 55.
  STEWART, William, Representative in Thirty-Sixth Congress, 43; Thirty-Seventh, 122.
  STILES, John D., Representative in Thirty-Seventh Congress, 123; Thirty-Eighth, 140.
  STOCKTON, Robert F., member of Peace Conference, 67.
  STOCES, William B., Representative in Thirty-Sixth Congress, 49.

- gress, 49. Sronz, Charles P., brigadier general, resolution respecting arrest of, 179; the President's reply, 180. Sronz, Joseph, Justice, upon freedom of the press, 188. Sronz, Lansing, Representative in Thirty-Sixth Congress, 49; member of Committee of Thirty-three, 53; report of, 58. Sroyer, R. L. Saccharter
- STOVELL, B. L., Speaker of Tennessee rebel Senate, 231. STRATTON, John L. N., Representative in Thirty-Sixth Con-gross, 43; Thirty-Seventh, 122; member of Committee of Thirty-three, 53.

STRICKLAND, Hardy, Representative in First Rebel Congress. 1 401.

- STRONG, William, justice, opinion on enrollment act, 273.
- STROUSE, Myer, Representative in Thirty-Eighth Congress, 140.
- STRYKER, Thomas J., member of Peace Conference, 67.
- STUART, Alexander II. II., vote on secession ordinance, note, 7; adjustment proposition, 74; delegate from Virginia to President Lincoln. 112.
- STUART, John T., Representative in Thirty-Eighth Congress, 140.
- SUBSTITUTION, votes on abolishing, 264, 266; rebel abolition of. 119.
- SULLIVAN, Algernon S., letter of Secretary Seward on arrest of, 154.
- SUMNAN, Agenon S., Feter of Sectearly Securitor artesis of 154.
  SUMMERS, George W., vote on secession ordinance, note, 7; member of Peace Conference, 68.
  SUMMER, Charles, Senator in Thirty-Fixth Congress, 48; Thirty-Sevendi, 122; Thirty-Eighth, 140; resolution respecting arrests, 180; proposition in Thirty-Second Congress to repeal the fugitive slave law, 234, 235; propositions on suffrage to colored persons, 241, on excluding them from the cars, 241, 242, on making them witnesses, 242, 243, and on repealing the laws regulating the toxastwise slave trade, 243; proposed constitutional amendment, 255; freedmen's bill, 260; on equalizing the pay of soldiers, 277; amendments to the reconstruction bill, 318, the Misssouri bill, 225, 226, to the national currency bill, 355, to letters of marque, 377, and West Virginia bill, 377; resolutions on Foreign Mediation, 346, 347, on the prosecution of the war, 291, and the relations of the insurrectionary States, 322, 3:3, 326, 327; letter respecting the President and emancipation, 327; letter respecting the President and emancipation, 203
- SUMTER, Fort, surrendered, 27; correspondence pending boundardment, 113, 114; spirit of the rebel press, 114.
   SUPREME Court of the United States, President Lincoln's recommendation concerning, 181; Vallandigham's case in, 175.
- SUBRENDERS of Government property in the insurrectionary States, 28.
- SWAIN, David L., commissioner from North Carolina to Provisional Congress, 12.
- SWAN, William G., deputy in rebel Provisional Congress, 400; member of First Rebel Congress, 402; Second, 402. SWLAT, Lorenzo D. M., Representative in Thirty-Eighth Con-
- gress, 140. SWIFT, W. C. N., purchase of Government arms by, in 1860,
- 35.

711

- TANEY, Roger B., Chief Justice, opinion of, in Merryman case, 155-158; rémarks on General Cadwalader's refusal
- CRSC, 130-135; FEMALAS OF GENERAL CAUMANNEES SERIESAT to obey his writ, 155.
  TAPPAN, Mason W., Representative in Thirty-Sixth Con-gress, 48; in Thirty-seventh, 122; member of Commit-tee of Thirty-three, 53; report of, 57.
  TARTF acts of 1801, 1802, and 1864, enactment of, 301, 362.
- TARIFF laws of the United States alluded to as a cause of secession, 17
- TARIFF, rebel, 12.

TAX, income, special war, 357.

- TAX on slaves, proposed, and votes, 361, 362; rebel, laid in 1861, 370.

- ISOL 370.
   TAXLIOS, our, 357, 361-368.
   TAXAITON, rebel, 368-373.
   TAXAITON, State, proceedings in Congress respecting, 359, 360, 364.
   TAXLOR, Dayard, chargé d'affaires, intercepts Confederate dispatch, 151.

- With 191.
   TATLOR, Joseph W., adjustment proposition of, 74, 75.
   TATLOR, Jaseph V., adjustment in Thirty-Sixth Congress, 49; withdrew, 4; member of Committee of Thirty-three, 53; report of, 58.
- TELEGRAPH, how it aided secession, 37. TELEGRAPH, how it aided secession, 37. Tex Ercz, John C., Senator in Thirty-Sixth Congress, 48; in Thirty-Seventh, 122; in Thirty-Eighth, 140; resolu-
- TENNESSEL, vote of, for President, 1; members of Peace Con-ference, 63; of Thirty-Sixth Congress, 43; of Thirty-Seventh, 122; deputies in rebel Provisional Congress, Seventil, 12.5, acputtes in rebet Provisional Congress, 201, 400; members of First Congress, 401, 402; of Sec-oud, 402; secession movement in Legislature of, 5; vote on adopting the league with the Confederate States, 5. TENNESSE, act of rebel Legislature of, for enrollment of free persons of color into the military service of the State '91. man in the army 200
- Territorius, bill passed setting disputed questions con-cerning, S9, 90; to prohibit slavery in, 254, 255; to pro-hibit polygamy in Utah, 376; enabling act for Nebraska, 377.
- TEVIS, Lieut. Col., order of, respecting Maryland election, his arrest and release, 311.

TEXAS, vote of, for President, 1; members of Thirty-Sixth

Congress, 43, 49; deputies in rebel Provisional Con-gress, 11, 400; members of First Congress, 402; Second, 402; secession movement in Convention and Legislature of, 4; ratified confederate constitution, 4; scizures and surrenders in, 28; men in the army, 300. THAYER, Andrew J., Representative in Thirty-Seventh Con-

- gress, 122; unseated, 123.
- THAYER, Eli, Representative in Thirty-Sixth Congress, 48; proposition of adjustment, 53.
- THAYER, M. Russell, Representative in Thirty-Eighth Congress, 140. Тпомля, Benjamin F., Representative in Thirty-Seventh
- Congress, 122. THOMAS, Francis, Representative in Thirty-Seventh Con-
- THOMAS, Francis, Representative in Thirty-Seventh Congress, 122; in Thirty-Eighth, 140; on compensated emancipation, 213-217; amendment of, to the enrollment bill, and vote, 209; on disunion caucus of 1535, 390.
   THOMAS, J. A., Assistant Sceretary of State, letter on citizenship of colored persons, note, 382.
   THOMAS, James II., Representative in Thirty-Sixth Congression.

- THOMAS, James II., Representative in LINEY-BLACK Congress, 40.
  THOMAS, J. Hanson, arrest of, 152.
  THOMAS, Mr., deputy in robel Provisional Congress, 400.
  THOMAS, Philip F., Sceretary of the Treasury, and resignation, 25; bids for public loan under, 363.
  THOMPSON, Jacob, Secretary of the Interior, and resignation, 25; commissioner from Mississippi to North Carolina, 11; received by letter, 5.
  THOMPSON, James Lucido, design on enrollment act, 273,
- THOMPSON, James, justice, decision on enrollment act, 273, 274.
- THOMPSON, Philip B., member of bogus legislative council
- Thomson, Finite X, memory of logist registative council of Kentucky, 8.
   Thomson, John R., Senator in Thirty-Sixth Congress, 43; Thirty-Seventh, 122; death of, 123.
   Timas, William G., Representative in First Rebel Congress, 402.
- TOBACCO, rebel regulations for destroying, 117.
- Ton, David, Governor of Ohio, and signer of Altoona Ad-dress, 233.
  ToLEX, John, Major, report of, in fugitive slave case, 250.
  ToMPKINS, Cydnor B., Representative in Thirty-Sixth Con-gress, 49.

- gress, 43; Tooms, Robert, Senator in Thirty-Sixth Congress, 48; on Senate Committee of Thirteen, 70; propositions in, 71; dispatch to L. M. Keitt, 37; telegraphic manifesto of, and sensational dispatch of, 37, 38; deputy in robel Provi-sional Congress, II, 400; provisional Secretary of State and anophytical dispatch of grant and anophytical provisional Secretary Secretarian and anophytical dispatch of grant grant dispatch of grant and anophytical dispatch of grant grant dispatch of and resignation, and appointed brigadier general, 12, 400; elected Senator in First Rebel Congress, 401; letter on reconstruction, 331.
- TOTTEN, A. O. W., member of Peace Conference, 68; com-missioner for Tennessee in making a league with Con-
- federate States, 5. TRACK, Henry W., Representative in Thirty-Eighth Congress, 140.
- TRAIN, Charles R., Representative in Thirty-Sixth Congress, 48; in Thirty-Seventh, 122.

- TRAIL, Amos, member of Peace Conference, 67. TREAT, Mos, member of Peace Conference, 67. TREAT, S. IL, Judge, decision on enrollment act, 273. TRENT affair, the, 635-946; Secretary Seward to Mr. Adams, 335; Earl Russell to Lord Lyons, 535; Secretary Seward to Lord Lyons, 338-342; Earl Russell to Lord Lyons, 342; comment of the London Times, 342, 343; action of Congress on, 343.
- TRINDLE, Carcy A., Representative in Thirty-Sixth Congress, 49; in Thirty-Seventh, 122.
- TRIMBLE, Isaac G., in command at Baltimore, and subsequent career, 393, and note.
- TRIPLETT, George W., Representative in Second Rebel Congress, 402.
- TRIPPE, Robert P., Representative in First Rebel Congress, 401
- TROWBEIDGE, Rowland E., Representative in Thirty-Seventh Congress, 122. TRUMBULL, Lyman, Senator in Thirty-Sixth Congress, 43;
- in Thirty-Seventh, 122; in Tlurty-Eighth, 140; amendment to Jeff. Davis's bill for sale of United States arms, 36; report on Baltimore police commissioners, 179; res-olution respecting arrests, 179; proposition to repeal joint resolution of July 17, 1862, on confiscation, 202-see 200; joint resolution proposing amendments to the Constitution, 256; proposition to repeal the commuta-tion clause of enrollment bill, note, 261; resolution on unemployed generals, 255; amendment to Andrew John-son's resolution respecting the cause of the war, 286; proposition to enforce the laws in all the States, 201; substitute for Mr. Noell's bill to encourage emancipa-tion in Missouri, 225, 226.
  TUCK, Amos, member of Peace Conference, 68.
  TUTXER, Josiah, Representative in Second Rebel Congress, ment to Jeff. Davis's bill for sale of United States arms,
- TURNER, Josiah, Representative in Second Rebel Congress, 402
- TURNER, Lewis C., Judge Advocate, instructions of, to United States Marshals, &c., 272.

TURNER, Thomas J., member of Peace Conference, 68. TURPIE, David, Senator in Thirty-Seventh Congress, 123. TUTTLE, James M., Brigadier General, orders at Natchez, respecting contrabands, 253.

TWENTY-negro exemption law. rebel, and repeal, 120. TWIGGS, David E., Major General, surrender by, of his com-

- TwieGs, David E., Major General, surrender by, of his command in Texas, 28.
  Trier, Frastus B., Brigadier General, order of, respecting the election in Delaware, 312.
  Trier, John, Sr., member of Peace Conference, 68; elected President of, 68; report of Peace Conference, 6; commissioner for Virginia in making a league with the "Confederato States," 8; deputy in rebel Provisional Concress 400 Congress, 400.

U

- UNDERWOOD, John C., Judge, opinion of, on confiscation act, 206. 207.
- UNDERWOOD, Joseph R., letter of Gen. Bnell, to, 250. UNDERWOOD, J. W. H., Representative in Thirty-Sixth Con-gress, 49; sent telegram South, 37; withdrew, 3. UNDERWOOD, Levi, member of Peace Conference, 67.

- UNDERWOOD, Levi, nicmoer of Feace Conference, 67. UNENFLOTE Generals, proceedings in Congress concerning, 285-see "Generals." Enox, preservation of national, resolutions concerning, 75, 285. UNION League, platform of 1864, 410. UNION National Convention, proceedings of, 403-410.

- UNITED States flag to be suppressed in Baltimore, note 394 ; Policeman Daneker's refusal to "condescend" to pull it down, and resignation of place, 394; William T. But ler's resignation, 394.
- UPSON, Charles, Representative in Thirty-Eighth Congress, 140.
- UPTON, Charles H., Representative Thirty-Seventh Con-
- Gross, Carlos H., Representative Thirty-Seventh Con-gress, 122; unseated, 23. Usues, John P., Secretary of the Interior, 108; report of, on colonization experiment, *note*, 212. Uтан, polygamy in, act to prohibit, 376.

- VALLANDIGHAM, Clement L., Representative in Thirty-Sixth Congress, 49; in Thirty-Seventh, 122; propositions of adjustment, 55, 73; arrest and trial of, 162; refusal of Judge Leavit to grant writ of habeas corpus, and his opinion, 162; order, directing that he be sent beyond builds, 102; order, infecting that he besch beyond the lines, 162; letter of, on retaliation, 175; case of, in United States Supreme Court, 175; correspondence concerning, 167–175; return of and address at Hamil-ton, 176; at Daytou, 176; notices of, by the press, 177; amendment of, to army appropriation bill, 283; reso-lutions relating to the war, 285; peace resolution of, 296; resolution respecting the Trent affair, 343; resolu-tion of censure of President Lincen, 386 tion of censure of President Lincoln, 386.
- VAN BUREN, John, letter of George N. Sanders, to, 330.
- VANCE Zebulon B., Representative in Thirty-Sixth Congress, 497; correspondence with Jefferson Davis on peace, 306, 307; views on reconstruction, 331.
   VANDEVER, William, Representative in Thirty-Sixth Con-
- VANDEVER, William, Representative in Thirty-Sixth Con-gress, 49; in Thirty-Seventh, 122; member of Peace Conference, 68; of Border States Committee, 73; reso-lution on the war, 291.
- VAN HORN, Burt, Representative in Thirty-Seventh Congress, 122
- VAN VALKENBURGH, Robert B., Representative in Thirty-Seventh Congress, 122; in Thirty-Eighth, 140.
  VAN WINKLE, Peter G., Senator in Thirty-Eighth Congress,
- 140.
- VAN WYCK, Charles H., Representative in Thirty-Sixth Con-gress, 48; in Thirty-Seventh, 122.VENALLE, Abram W., deputy in rebel Provisional Congress, 400
- 400.
- VERMON, vote of, for President, 1; members of Peace Con-ference, 6; of Thirty-Sixth Congress, 43; of Thirty-Seventh, 122; of Thirty-Eighth, 140; "personal liberty" laws of, 44, 47.
- VERREE, John P., Representative in Thirty-Sixth Congress, 48; in Thirty-Seventh, 122.
- VESSELS, United States, disposition of-"see Report."
- VEST, George G., deputy in rebel Provisional Congress, 400; Representative in First Rebel Congress, 402; Second, 402.
- VIBBARD, Chauncey, Representative in Thirty-Seventh Congress, 122. VICKSEURG, President Lincoln's remarks on fall of, 335, 335.
- VILLERE, Charles J., Representative in First Rebel Con-
- gress, 401; Second, 402. Gloss, 301, Second, 40.2. Vincinia, vote of for President, 1; members of Peace Con-ference, 67; of Thirty-Sixth Congress, 48, 49; of Thir-ty-Seventh, 122; Thirty-Eighth, 141; proposal to co-operate with South Carolina, 2; secession movement in, 5-8; vote on ordinance of secession, note, 7; league with Confederate States, 8; Governor of, purchase of arms prior to secession, 35, 36; address of members of

- Congress from, to the people of, 39; answer of Presi-dent Lincoln, to the delegates from, 112; resolutions of Convention of, 112; Legislature of, on guerrillas, 119; peace resolutions and discussion of in Legislature of, 304; men in the army, 399; reconstruction movement and abulition of slavery in, 321; West Virginia bill, 377.378
- Vooruers, Daniel W., Representative in Thirty-Seventh Congress, 122; in Thirty-Eighth, 140.
- VOLUNTEERING, rebel restrictions upon, 120. VOLUNTEERS, the several calls for, 114, 115, 270,
- VECOM, Peter D., member of Peace Conference, 67.

W

- WADE, Benjamin F., Senator in Thirty-Sixth Congress, 43; Thirty-Serenth, 122; Thirty-Eighth, 140; on Commit-tee of Thirteen, 70; amendments to reconstruction bill, 718; paper of on failure of said bill, 322. WADE, Edward, Representative in Thirty-Sixth Congress, 49.
- WADSWORTH, James S., member of Peace Congress, 67. WADSWORTH, William II., Representative in Thirty-Seventh Congress, 122; Thirty-Eighth, 140; resolution on the powers of the President, 327.
- WATTE, Carlos A., colonel, appointed to supersede General Twiggs in Texas, 28.
- WALDRON, Henry, Representative in Thirty-Sixth Congress, 49.
- WALKER, Amasa, Representative in Thirty-Seventh Con-
- Walker, Hand, Toplesonation in the potential of gress, 173.
  WALKER, Leroy Pope, commissioner from Alabama to Ten-nessee, 11; Provisional Secretary of War, and resigna-tion, 12, 400; correspondence with Beauregard before bombardment of Suntar, 113, 114; speech after, 114;
- WALL ASIAN STATES OF A STATES
- WALL, William, Representative in Thirty-Seventh Congress, 122.
- WALLACE, John W., Representative in Thirty-Seventh Con-
- WALLACE, John W., Representative in Link 9 contents of gress, 122.
   WALLACE, Lewis, Major General, suppression by, of Eve-ning Transcript, Baltimore, 192; confiscation orders revoked, 207, 205; correspondence with Governor Brad-ford, 311, 312.
   WALLIG, S. Teackle, arrest of, 152; motions and votes in Maryland Legislature, 396-398.
   WALTON, Charles W., Representative in Thirty-Seventh Con-gress, 122.
- gress, 122. WALTON, E. P., Representative in Thirty-Sixth Congress, 48;
- Thirty-Seventh, 122; substitutes for emancipation bills, 200, 202.
- WAR, new article of, 237, 238; resolutions relating to its objects, 285-290; to its prosecution, 290-294; extract from rules of, for the protection of colored soldiers, 280; special income, tax, 357.
- WAR, the, how and why made upon the United States, 112-114.
- WARD, Elijah, Representative in Thirty-Seventh Congress, 122; Thirty-Eighth, 140.
- WARD, G. T., deputy in robel Provisional Congress, 400.
- WARDLAW, D. Louis, remarks in South Carolina Convention, 17; vote on ordinance of secession, 399. WARFIELD, Henry M., arrest of, 152; votes of, in Maryland Legislature, 390-398.
- WASHBURN, Cadwallader C., Representative in Thirty-Sixth
- Congress, 49; on Comnuittee of Thirty-three, 53; report
- Congress, 49; on Commuttee of Larry-three, 53; report of, 57; member of Peace Conference, 68.
   WASHBURN, Israel, jr., Representative in Thirty-Sixth Con-gress, and resignation, 45; Sceretary Seward's letter to, respecting Robert Elliott's arrest, 154; Governor of Maine, and signer of Altoona address, 233.
   WASHBURN, William B, Representative in Thirty-Eighth Congress 10.
- WASIBURN, Wilham B., Representative in furry-Light Congress, 140.
   Wasunuxne, Ellihu B., Representative in Thirty-Sixth Con-gress, 49; Thirty-Seventh, 122; Thirty-Lighth, 140; resolution to expel Alexander Long, 387.
   Wartsex, J. W. C., Senator in Second Rebel Congress, 402.
   WATL, T. N., deputy in rebel Provisional Congress, 11, 400.
   WAYNE, James M., Justice Supreme Court of the United States, oplinion in the Vallandigham case, 175; the latter segmenta et Hamilton 176.

- States, opinion in the valuation and case, 173; the latter's comment at Hamilton, 176.
   WEBSTER, Edwin H., Representative in Thirty-Sixth Con-gress, 43; Thirty-Seventh, 122; Thirty-Eighth, 140; views on compensated emancipation, 213-217; instruc-tions to Conference Committee on Montana bill, 240, 000 241.
- WEED, Thurlow, letter replying to Judge Campbell's reflections on Secretary Seward, 111, 112.
   WELLES, Gideon, Secretary of the Navy, 108; correspondence with Alexander II. Stephens, 307; letter on alleged foreign enlistments, 344, 345.

WELLS, Alfred, Representative in Thirty-Sixth Congress, 48. | WELSH, Israel, Representative in First Rebel Congress, 401; Second, 402.

- WEST Virginia, bill to crect State of, 377, 378, and note.
  WIALEY, Kellian V., Representative in Thirty-Seventh Congress, 122; Thirty-Eighth, 141: on select commitee on compensated emancipation, 213; amendment to West Virginia bill, 377.
- WHARTON, T. J., commissioner from Mississlppi to Tennessec. 11.
- WHEELER, Ezra, Representative in Thirty-Eighth Congress, 141; amendment to proposed constitutional amend-
- ment, 257. WHEELER, William A., Representative in Thirty-Soventh Congress, 122.

Whig, Richmond, on repealing substitution, 121.

- WHITE, Albert S., Representative in Thirty-Seventh Congress, 122; resolution for select committee on compen-sated emancipation, 213; Missouri and Maryland bills, 224, 226.
- WHITE, Chilton A., Representative in Thirty-Seventh Con-
- gress, 122; Thirty-Eighth, 140. WHITE, Joseph W., Representative in Thirty-Eighth Con-gress, 140.
- WHITE, Thomas, member of Peace Conference, 67.
- gress, 49; on Committee of Thirty-three, 53; report of, 58. WHITELEY, William G., Representative in Thirty-Sixth Con-
- WHITFIELD. R. H., Representative in Second Rebel Congress, 402.
- WHITTHORNE, W. C., Speaker of Tennessee Rebel House of
- Representatives, 231. WICKHAM, William C., Representative in Second Rebel Con-gress, 402.
- WICKLIFF, Charles A., member of Peace Conference, 68; Representative in Thirty-Seventh Congress, 122; at
- Representative in Thirty-Seventh Congress, 122; at Border States interview with the Prosident, 211; views on compensated emancipation, 213-217; amendment to enrollment bill, 261; propositions relating to the war, 253; to the rebellious States, 323, 324.
   WIGFALL, Louis T., Senator in Thirty-Sixth Congress, 48; intermediary between Colonel Hayne and President Buchanan, 32; deputy in rebel Provisional Congress, 14, 400; appointed Brigadier General, 401; Senator in First Congress, 401; Second, 402.
   WIGCX, John A., Representative in First Rebel Congress, 402; death of, 402.
- 402; death of, 402.
- WILDER, A. Carter, Representative in Thirty-Eighth Congress, 141.
- WILKINSON, Morton S., Senator in Thirty-Seventh Congress,
- WILKINSON, Morton S., Schator in Thirty-Seventh Congress, 122; Thirty-Eighth, 140; amendment to the Montana bill, 240.
   WILLER, Waitman T., vote on secession ordinance of Vir-ginia, note, 7; Senator in Thirty-Seventh Congress, 123; Thirty-Eighth, 140; report on excluding colored per-sons from street cars, 242; amendment to the freed-men's bill, 260; resolutions on the war, 288.
   WILLIAMS T., General, order to turn all fugitives from his camps, 251.

- WILLIAMS T., General, order to turn all fugitives from his camps, 251.
  WILLIAMS, Thomas, Representative in Thirty-Eighth Congress, 141; resolutions on the relations of the rehellious States, 323, 329.
  WILLIAMSON, Benjamin, member of Peace Conference, 67.
  WILMON, David, member of Peace Conference, 67; Senator in Thirty-Seventh Congress, 122.
  WILSON, Henry, Senator in Thirty-Fixth Congress, 48; Thirty-Eighth, 140; substitute for resolution relative to General Stone's arrest, 180; bill to amend the functive glave law, 255; proposition respectoutton renative to General Stone's arrest, 189; bill to amend the fugitive slave law, 255; proposition respect-ing slaves in dock-yards, 238; amendments to freed-men's bill, 260; enrollment bill, 261; army appropria-tion bill, 277; the militia bill, 275. WLSON, James, Representative in Thirty-Sixth Congress, 40
- 49.
- WILSON, James F., Representative in Thirty-Seventh Congress, 123; Thirty-Eighth, 140; a mendatory joint reso-lution on confiscation, 202; resolution to ratify the emancipation proclamation, 230; on Select Committee on compensated emancipation, 213; resolutions respecting an additional article of war, 238, and homesteads for soldiers, 284.
- WILSON, Robert, Senator in Thirty-Seventh Congress, 123; views on compensated emancipation, 213-217; amend-ments to the Missouri bill, 225.
- WILSON, W. S., deputy in rebel Provisional Congress, 11, 400
- WINANS, Ross, arrest of, 152; votes in Maryland Legislature, 396-398; also, 394.
- WINANS, Thomas, proposed member of Board of Public Safety
- of Maryland, 9. Wispost, William, Representative in Thirty-Sixth Congress, 49; Thirty-Seventh, 122; Thirty-Eighth, 141; on Com-mittee of Thirty-three, 53; proposed constitutional amendment, 258.

- WINFIELD, Charles H., Representative in Thirty-Eighth Congress, 140. WINSLOW, Warren, Representative in Thirty-Sixth Congress,
- 49; on Committee of Thirty-three, 53; report of 56. WINSTON, John Anthony, Commissioner from Alabama to
- Arkansas, 11. WISCONSIN, vote for President in 1860, 1; members of Thir-
- ty-Sixth Congress, 49; Thirty-Seventh, 122, 123; Thirty-
- Eighth, 140. WISE, Henry A., proposition in Virginia Convention for Confederate recognition, 6, 7; vote on secession ordi-
- WITHERS, Thomas J., deputy in rebel Provisional Congress, 2, 11, 400; report in South Carolina Convention, 18, 19, 20 ortho on Second on evidence 200 20; vote on secession ordinance, 399.
- WITHERSPOON, James M., Representative in Second Rebel Congress, 402.
- WITNESSES, colored persons as, in United States courts, 242, 243.

WOLCOTT, C. P., member of Peace Conference, 68, note.

- Woon, Benjamin, Representative in Thirty-Seventh Con-gress, 122; Thirty-Eighth, 140; peace proposition, 294, gres 295.
- Woon, Fernando, Representative in Thirty-Eighth Congress, 140; New York as a free city, 42; correspondence with President Lincoln on peace, 296, 297; proposition for commissioners to settle terms of peace, 297, 298; letter of George N. Sanders to, 330.
- Woon, John, member of Peace Conference, 68.
   Woon, John, Representative in Thirty-Sixth Congress, 48.
   Woon, William B., Colonel, Sceretary Benjamin's order to, respecting East Tennessee bridge-burners, 187.
   WoonBRIDGS, Frederick E., Representative in Thirty-Eighth Congress, 140.
- Congress, 140.
- WOODRUFF, John, Representative in Thirty-Sixth Congress, 49.
- WOODRUFF, George C., Representative in Thirty-Seventh Congress, 122.
- Congress, 122.
   WOONSON, Samuel H., Representative in Thirty-Sixth Congress, 49; on Peace Conference proposition, 70.
   WOONWARD, George W., Justiee, decision on enrollment act, 273, 274; General McClellan's letter respecting the election of, as Governor of Pennsylvania, 386.
   WOOL, John E., Major General, member of Peace Conference, 67; orders respecting contrubands, 243.
   WORCESTER, Samuel T., Representative in Thirty-Seventh Concress 123.

- Congress, 122. World, New York, seizure of office, and subsequent proceed-
- Words, New York, Seizure of once, and subsequent proceed-ings, 192-194.
   WRIGHT, Angustus R., deputy in rebel Provisional Congress, 11, 400; Representative in First Congress, 401; peace proposition, 305. WRIGHT, Crafts J., secretary of Peace Conference, 63. WRIGHT, Hendrick B., Representative in Thirty-Seventh Congress, 122; resolutions on the war, 258-291.

- WRIGHT, John C., member of Peace Conference, 68; death of, 68.
- WRIGHT, John V., Representative in Thirty-Sixth Cougress, 49; Representative in First Rehel Congress, 402; Second, 402.

WRIGHT, Joseph A., Senator in Thirty-Seventh Congress, 123; superseded, 123. WRIGHT, William, Senator in Thirty-Eighth Congress, 140. WRIGHT, William B., Representative in First Rebel Congress, 402.

Y

- YANCEY, William L., rebel commissioner to England, letter of, to Earl Russell, 27; reported conversation of, with Earl Russell, 27; Senator in First Rebel Congress, 401; death of, 401.
- YATES, Richard, Governor of Illinois, signer of Altoona address, 233.
- YEAMAN, George H., Representative in Thirty-Seventh Con-gress, 123; in Thirty-Eighth, 140; resolutions on the President's emancipation proclamation, 229; resolution on the status of the rebellious States, 327, 328.
- YELLOT, Coleman, letter for a convention in Maryland, 93 introduced public safety bill into the Senate of Mary-land, 9; proceedings on, 398; commissioner from Legis-lature of Maryland to visit Jefferson Davis, 9; report of, 9; also, 393.
- YERGER, E. M., commissioner from Mississippi to Florida,
- II.
   YULEE, David Levy, Senator in Thirty-Sixth Congress, 48; withdrew, 3; intermediary between Col. Hayne and President Buchanan, 32; "secret" disunion letter of,

7.

ZACHARRE, J. W., & Co., purchase of Government arms by, in 1860, 35.

ZOLLICOFFER, Felix K., member of Peace Conference, 68.









No. 872	Sect	Shelf_2
CONTENTS		
	·	
	·	
Lincoln National Life Foundation		
Collateral Lincoln Library		

2009.084.062.96

